COMPREHENSIVE ORGANIC SYNTHESIS

Selectivity, Strategy & Efficiency in Modern Organic Chemistry

Editor-in-Chief BARRY M. TROST Stanford University, CA, USA

Deputy Editor-in-Chief IAN FLEMING University of Cambridge, UK

Volume 3 CARBON–CARBON σ-BOND FORMATION

> Volume Editor GERALD PATTENDEN, FRS University of Nottingham, UK



PERGAMON PRESS OXFORD • NEW YORK • SEOUL • TOKYO

© 1991 Elsevier Ltd. All rights reserved.

All rights reserved. No part of this publication may be reproduced, stored in a retrieval system or transmitted in any form or by any means: electronic, electrostatic, magnetic tape, mechanical, photocopying, recording or otherwise, without permission in writing from the publishers.

First edition 1991 Second impression 1993 Third impression 1999 Fourth impression 2002 Fifth impression 2005

Library of Congress Cataloging in Publication Data

Comprehensive organic synthesis: selectivity, strategy and efficiency in modern organic chemistry/editor[s] Barry M. Trost, Ian Fleming. p. cm. Includes indexes. Contents: Vol. 1.-2. Additions to C-X[pi]-Bonds — v. 3. Carbon-carbon sigma-Bond formation — v. 4. Additions to and substitutions at C-C[pi]-Bonds — v. 5. Combining C-C[pi]-Bonds — v. 6. Heteroatom manipulation — v. 7. Oxidation — v. 8. Reduction — v. 9. Cumulative indexes. 3. Organic Compounds — Synthesis I. Trost, Barry M. 1941-II. Fleming, Ian, 1935-QD262.C535 1991 547.2—dc20 90-26621

British Library Cataloguing in Publication Data

Comprehensive organic synthesis 3. Organic compounds. Synthesis I. Trost, Barry M. (Barry Martin) 1941– 547.2

ISBN 0-08-040594-0 (Vol. 3) ISBN 0-08-035929-9 (set)

Contents

Preface		vii
Contributors to Volume 3		ix
Abbrevi	ations	xi
Content	s of All Volumes	xv
Alkylati	on of Carbon	
1.1	Alkylations of Enols and Enolates D. CAINE, University of Alabama, Tuscaloosa, AL, USA	1
1.2	Alkylations of Nitrogen-stabilized Carbanions R. E. GAWLEY & K. REIN, University of Miami, Coral Gables, FL, USA	65
1.3	Alkylations of Sulfur- and Selenium-containing Carbanions A. KRIEF, Facultés Universitaires Notre-Dame de la Paix, Namur, Belgium	85
1.4	Alkylations of Other Heteroatom-stabilized Carbanions D. R. CHESHIRE, Fisons Pharmaceutical Division, Loughborough, UK	193
1.5	Alkylations of Nonstabilized Carbanions J. M. KLUNDER & G. H. POSNER, Johns Hopkins University, Baltimore, MD, USA	207
1.6	Alkylations of Vinyl Carbanions D. W. KNIGHT, University of Nottingham, UK	241
1.7	Alkylations of Alkynyl Carbanions P. J. GARRATT, University College London, UK	271
1.8	Friedel–Crafts Alkylations G. A. OLAH, R. KRISHNAMURTI & G. K. S. PRAKASH, University of Southern California, Los Angeles, CA, USA	293
1.9	Polyene Cyclizations J. K. SUTHERLAND, University of Manchester, UK	341
1.10	Transannular Electrophilic Cyclizations D. C. HARROWVEN & G. PATTENDEN, University of Nottingham, UK	379
Couplin	g Reactions	
2.1	Coupling Reactions Between sp ³ Carbon Centers D. C. BILLINGTON, Institut de Recherches Servier, Paris, France	413
2.2	Coupling Reactions Between sp ³ and sp ² Carbon Centers K. TAMAO, Kyoto University, Japan	435
2.3	Coupling Reactions Between sp ² Carbon Centers D. W. KNIGHT, University of Nottingham, UK	481
2.4	Coupling Reactions Between sp ² and sp Carbon Centers K. SONOGASHIRA, Osaka City University, Japan	521
2.5	Coupling Reactions Between sp Carbon Centers K. SONOGASHIRA, Osaka City University, Japan	551
2.6	Pinacol Coupling Reactions G. M. ROBERTSON, Glaxo Group Research, Ware, UK	563
2.7	Acyloin Coupling Reactions R. BRETTLE, University of Sheffield, UK	613
2.8	Kolbe Reactions H. J. SCHÄFER, Westfälische Wilhelms Universität, Münster, Germany	633

vi	Contents	
2.9	Oxidative Coupling of Phenols and Phenol Ethers D. A. WHITING, University of Nottingham, UK	659
Rearr	angement Reactions	
3.1	Wagner-Meerwein Rearrangements J. R. HANSON, University of Sussex, Brighton, UK	705
3.2	The Pinacol Rearrangement B. RICKBORN, University of California, Santa Barbara, CA, USA	721
3.3	Acid-catalyzed Rearrangements of Epoxides B. RICKBORN, University of California, Santa Barbara, CA, USA	733
3.4	The Semipinacol and Other Rearrangements D. J. COVENEY, Avondale Chemical Company, Rathdrum, Republic of Ireland	777
3.5	Dienone–Phenol Rearrangements and Related Reactions D. A. WHITING, <i>University of Nottingham, UK</i>	803
3.6	Benzil–Benzilic Acid Rearrangements G. B. GILL, University of Nottingham, UK	821
3.7	The Favorskii Rearrangement J. MANN, <i>University of Reading, UK</i>	839
3.8	The Ramberg–Bäcklund Rearrangement J. M. CLOUGH, ICI Agrochemicals, Bracknell, UK	861
3.9	The Wolff Rearrangement G. B. GILL, University of Nottingham, UK	887
3.10	The Stevens and Related Rearrangements I. E. MARKÓ, University of Sheffield, UK	913
3.11	The Wittig Rearrangement J. A. MARSHALL, University of South Carolina, Columbia, SC, USA	975
Other	Carbon–Carbon Bond Forming Reactions	
4.1	Carbonylation and Decarbonylation Reactions D. J. THOMPSON, ICI Specialty Chemicals, Wilmington, DE, USA	1015
4.2	Carbon–Carbon Bond Formation by C–H Insertion D. F. TABER, University of Delaware, Newark, DE, USA	1045
Autho	r Index	1063
Subject Index		1131

Preface

The emergence of organic chemistry as a scientific discipline heralded a new era in human development. Applications of organic chemistry contributed significantly to satisfying the basic needs for food, clothing and shelter. While expanding our ability to cope with our basic needs remained an important goal, we could, for the first time, worry about the quality of life. Indeed, there appears to be an excellent correlation between investment in research and applications of organic chemistry and the standard of living. Such advances arise from the creation of compounds and materials. Continuation of these contributions requires a vigorous effort in research and development, for which information such as that provided by the *Comprehensive* series of Pergamon Press is a valuable resource.

Since the publication in 1979 of *Comprehensive Organic Chemistry*, it has become an important first source of information. However, considering the pace of advancements and the ever-shrinking timeframe in which initial discoveries are rapidly assimilated into the basic fabric of the science, it is clear that a new treatment is needed. It was tempting simply to update a series that had been so successful. However, this new series took a totally different approach. In deciding to embark upon *Comprehensive Organic Synthesis*, the Editors and Publisher recognized that synthesis stands at the heart of organic chemistry.

The construction of molecules and molecular systems transcends many fields of science. Needs in electronics, agriculture, medicine and textiles, to name but a few, provide a powerful driving force for more effective ways to make known materials and for routes to new materials. Physical and theoretical studies, extrapolations from current knowledge, and serendipity all help to identify the direction in which research should be moving. All of these forces help the synthetic chemist in translating vague notions to specific structures, in executing complex multistep sequences, and in seeking new knowledge to develop new reactions and reagents. The increasing degree of sophistication of the types of problems that need to be addressed require increasingly complex molecular architecture to target better the function of the resulting substances. The ability to make such substances available depends upon the sharpening of our sculptors' tools: the reactions and reagents of synthesis.

The Volume Editors have spent great time and effort in considering the format of the work. The intention is to focus on transformations in the way that synthetic chemists think about their problems. In terms of organic molecules, the work divides into the formation of carbon-carbon bonds, the introduction of heteroatoms, and heteroatom interconversions. Thus, Volumes 1–5 focus mainly on carbon-carbon bond formation, but also include many aspects of the introduction of heteroatoms. Volumes 6–8 focus on interconversion of heteroatoms, but also deal with exchange of carbon-carbon bonds for carbonheteroatom bonds.

The Editors recognize that the assignment of subjects to any particular volume may be arbitrary in part. For example, reactions of enolates can be considered to be additions to C—C π -bonds. However, the vastness of the field leads it to be subdivided into components based upon the nature of the bond-forming process. Some subjects will undoubtedly appear in more than one place.

In attacking a synthetic target, the critical question about the suitability of any method involves selectivity: chemo-, regio-, diastereo- and enantio-selectivity. Both from an educational point-of-view for the reader who wants to learn about a new field, and an experimental viewpoint for the practitioner who seeks a reference source for practical information, an organization of the chapters along the theme of selectivity becomes most informative.

The Editors believe this organization will help emphasize the common threads that underlie many seemingly disparate areas of organic chemistry. The relationships among various transformations becomes clearer and the applicability of transformations across a large number of compound classes becomes apparent. Thus, it is intended that an integration of many specialized areas such as terpenoid, heterocyclic, carbohydrate, nucleic acid chemistry, *etc.* within the more general transformation class will provide an impetus to the consideration of methods to solve problems outside the traditional ones for any specialist.

In general, presentation of topics concentrates on work of the last decade. Reference to earlier work, as necessary and relevant, is made by citing key reviews. All topics in organic synthesis cannot be treated with equal depth within the constraints of any single series. Decisions as to which aspects of a

Preface

topic require greater depth are guided by the topics covered in other recent *Comprehensive* series. This new treatise focuses on being comprehensive in the context of synthetically useful concepts.

The Editors and Publisher believe that *Comprehensive Organic Synthesis* will serve all those who must face the problem of preparing organic compounds. We intend it to be an essential reference work for the experienced practitioner who seeks information to solve a particular problem. At the same time, we must also serve the chemist whose major interest lies outside organic synthesis and therefore is only an occasional practitioner. In addition, the series has an educational role. We hope to instruct experienced investigators who want to learn the essential facts and concepts of an area new to them. We also hope to teach the novice student by providing an authoritative account of an area and by conveying the excitement of the field.

The need for this series was evident from the enthusiastic response from the scientific community in the most meaningful way — their willingness to devote their time to the task. I am deeply indebted to an exceptional board of editors, beginning with my deputy editor-in-chief Ian Fleming, and extending to the entire board — Clayton H. Heathcock, Ryoji Noyori, Steven V. Ley, Leo A. Paquette, Gerald Pattenden, Martin F. Semmelhack, Stuart L. Schreiber and Ekkehard Winterfeldt.

The substance of the work was created by over 250 authors from 15 countries, illustrating the truly international nature of the effort. I thank each and every one for the magnificent effort put forth. Finally, such a work is impossible without a publisher. The continuing commitment of Pergamon Press to serve the scientific community by providing this *Comprehensive* series is commendable. Specific credit goes to Colin Drayton for the critical role he played in allowing us to realize this work and also to Helen McPherson for guiding it through the publishing maze.

A work of this kind, which obviously summarizes accomplishments, may engender in some the feeling that there is little more to achieve. Quite the opposite is the case. In looking back and seeing how far we have come, it becomes only more obvious how very much more we have yet to achieve. The vastness of the problems and opportunities ensures that research in organic synthesis will be vibrant for a very long time to come.

> BARRY M. TROST Palo Alto, California

Contributors to Volume 3

Dr D. C. Billington Residence Offenbach, 33 Rue Cartault, F-92800 Puteaux, Paris, France Dr R. Brettle Department of Chemistry, University of Sheffield, Sheffield S3 7HF, UK Professor D. Caine Department of Chemistry, The University of Alabama, Lloyd Hall, PO Box 870 336, Tuscaloosa, AL 35487, USA Dr D. R. Cheshire Fisons Pharmaceutical Division, Research & Development Laboratories, Bakewell Road, Loughborough, Leicestershire LE11 0RH, UK Dr J. M. Clough ICI Agrochemicals, Jealott's Hill Research Station, Bracknell, Berkshire RG12 6EY, UK Dr D. J. Coveney Avondale Chemical Co., Rathdrum, Co Wicklow, Republic of Ireland Dr P. J. Garratt Department of Chemistry, University College London, 20 Gordon Street, London W1H 0AJ, UK Professor R. E. Gawley Department of Chemistry, University of Miami, Coral Gables, FL 33124, USA Dr G. B. Gill Department of Chemistry, University of Nottingham, University Park, Nottingham NG7 2RD, UK Dr J. R. Hanson School of Molecular Sciences, University of Sussex, Falmer, Brighton, Sussex BN1 9QJ, UK Dr D. C. Harrowven Department of Chemistry, University of Nottingham, University Park, Nottingham NG7 2RD, UK Dr J. M. Klunder Boehringer Ingelheim Pharmaceuticals Inc., PO Box 368, Ridgefield, CT 06877, USA. Dr D. W. Knight Department of Chemistry, University of Nottingham, University Park, Nottingham NG7 2RD, UK Professor A. Krief Departement de Chimie, Facultés Universitaires Notre-Dame de la Paix, Rue de Bruxelles 61, B-5000 Namur, Belgium Dr R. Krishnamurti Loker Hydrocarbon Research Institute, University of Southern California, University Park, Los Angeles, CA 90089, USA Dr J. Mann Department of Chemistry, University of Reading, Whiteknights, PO Box 224, Reading RG6 2AD, UK Dr I. E. Markó Department of Chemistry, University of Sheffield, Sheffield S3 7HF, UK Professor J. A. Marshall Department of Chemistry, University of South Carolina, Columbia, SC 29208, USA Professor G. A. Olah Loker Hydrocarbon Research Institute, University of Southern California, University Park, Los Angeles, CA 90089, USA Professor G. Pattenden Department of Chemistry, University of Nottingham, University Park, Nottingham NG7 2RD, UK Professor G. H. Posner Department of Chemistry, The Johns Hopkins University, Baltimore, MD 21218, USA

Professor G. K. S. Prakash

Loker Hydrocarbon Research Institute, University of Southern California, University Park, Los Angeles, CA 90089, USA

Mrs K. Rein Department of Chemistry, University of Miami, Coral Gables, FL 33124, USA

Professor B. Rickborn Department of Chemistry, University of California, Santa Barbara, CA 93106, USA

Dr G. M. Robertson Research Station, Glaxo Group Research Ltd, Ware, Hertfordshire SG12 0DJ, UK

Professor H. J. Schäfer Organisch-Chemisches Institut der Westfälische Wilhelms-Universität, Orleans-Ring 23, D-4400 Münster, Germany

Professor K. Sonogashira Faculty of Engineering, Osaka City University, Sugimoto, Sumiyoshi, Osaka 558, Japan

Professor J. K. Sutherland Department of Chemistry, University of Manchester, Oxford Road, Manchester M13 9PL, UK

Professor D. F. Taber Department of Chemistry & Biochemistry, University of Delaware, Newark, DE 19716, USA

Professor K. Tamao Department of Synthetic Chemistry, Faculty of Engineering, Kyoto University, Sakyo-ku, Kyoto 606, Japan

Dr D. J. Thompson ICI Specialty Chemicals, Wilmington, DE 19897, USA

Dr D. A. Whiting Department of Chemistry, University of Nottingham, University Park, Nottingham NG7 2RD, UK

х

Abbreviations

The following abbreviations have been used where relevant. All other abbreviations have been defined the first time they occur in a chapter.

Techniques	
CD	circular dichroism
CIDNP	chemically induced dynamic nuclear polarization
CNDO	complete neglect of differential overlap
СТ	charge transfer
GLC	gas-liquid chromatography
НОМО	highest occupied molecular orbital
HPLC	high-performance liquid chromatography
ICR	ion cyclotron resonance
INDO	incomplete neglect of differential overlap
IR	infrared
LCAO	linear combination of atomic orbitals
LUMO	lowest unoccupied molecular orbital
MS	mass spectrometry
NMR	nuclear magnetic resonance
ORD	ontical rotatory dispersion
PE	photoelectron
SCF	self-consistent field
TLC	thin laver chromatography
UV	ultraviolet
Reagents, solven	ts, etc.
Ac	acetyl
acac	acetylacetonate
AIBN	2,2'-azobisisobutyronitrile
Ar	aryl
ATP	adenosire triphosphate
9-BBN	9-borabicyclo[3.3.1]nonyl
9-BBN-H	9-borabicyclo[3.3.1]nonane
BHT	2,6-di-t-butyl-4-methylphenol (butylated hydroxytoluene)
bipy	2,2'-bipyridyl
Bn	benzyl
t-BOC	t-butoxycarbonyl
BSA	N,O-bis(trimethylsilyl)acetamide
BSTFA	N,O-bis(trimethylsilyl)trifluoroacetamide
BTAF	benzyltrimethylammonium fluoride
Bz	benzoyl
CAN	ceric ammonium nitrate
COD	1,5-cyclooctadiene
COT	cyclooctatetraene
Ср	cyclopentadienyl
Cp*	pentamethylcyclopentadienyl
18-crown-6	1,4,7,10,13,16-hexaoxacyclooctadecane
CSA	camphorsulfonic acid
CSI	chlorosulfonyl isocyanate
DABCO	1,4-diazabicyclo[2.2.2]octane
DBA	dibenzylideneacetone
DBN	1,5-diazabicyclo[4.3.0]non-5-ene
DBU	1,8-diazabicyclo[5.4.0]undec-7-ene

xii	Abbreviations
DCC	dicvclohexvlcarbodiimide
DDO	2.3-dichloro-5.6-dicvano-1.4-benzoquinone
DEAC	diethylaluminum chloride
DEAD	diethyl azodicarboxylate
DET	diethyl tartrate (+ or -)
DHP	dihydropyran
DIBAL-H	disobutylaluminum hydride
diglyme	diethylene glycol dimethyl ether
dimsyl Na	sodium methylsulfinylmethide
DIOP	2.3- <i>O</i> -isopropylidene-2.3-dihydroxy-1.4-bis(diphenylphosphino)butane
DIPT	diisopropyl tartrate (+ or –)
DMA	dimethylacetamide
DMAC	dimethylaluminum chloride
DMAD	dimethyl acetylenedicarboxylate
DMAP	4-dimethylaminopyridine
DME	dimethoxyethane
DMF	dimethylformamide
DMI	N.N'-dimethylimidazolone
DMSO	dimethyl sulfoxide
DMTSF	dimethyl(methylthio)sulfonium fluoroborate
DPPB	1.4-bis(diphenylphosphino)butane
DPPE	1.2-bis(diphenylphosphino)ethane
DPPF	1.1'-bis(diphenylphosphino)ferrocene
DPPP	1.3-bis(diphenylphosphino)propane
E ⁺	electrophile
EADC	ethylaluminum dichloride
EDG	electron-donating group
EDTA	ethylenedjaminetetraacetic acid
EEDO	N-ethoxycarbonyl-2-ethoxy-1,2-dihydroquinoline
EWG	electron-withdrawing group
HMPA	hexamethylphosphoric triamide
HOBT	hydroxybenzotriazole
IpcBH ₂	isopinocampheylborane
Ipc ₂ BH	diisopinocampheylborane
K APA	potassium 3-aminopropylamide
K-selectride	potassium tri-s-butylborohydride
LAH	lithium aluminum hydride
LDA	lithium diisopropylamide
LICA	lithium isopropylcyclohexylamide
LITMP	lithium tetramethylpiperidide
L-selectride	lithium tri-s-butylborohydride
LTA	lead tetraacetate
MCPBA	m-chloroperbenzoic acid
MEM	methoxyethoxymethyl
MEM-Cl	β-methoxyethoxymethyl chloride
MMA	methyl methacrylate
MMC	methylmagnesium carbonate
MOM	methoxymethyl
Ms	methanesulfonyl
MSA	methanesulfonic acid
MsCl	methanesulfonyl chloride
MVK	methyl vinyl ketone
NBS	N-bromosuccinimide
NCS	<i>N</i> -chlorosuccinimide

NMO	N-methylmorpholine N-oxide
NMP	N-methyl-2-pyrrolidone
Nu	nucleophile
PPA	polyphosphoric acid
PCC	pyridinium chlorochromate
PDC	pyridinium dichromate
phen	1,10-phenanthroline
Phth	phthaloyl
PPE	polyphosphate ester
PPTS	pyridinium p-toluenesulfonate
Red-Al	sodium bis(methoxyethoxy)aluminum dihydride
SEM	β-trimethylsilylethoxymethyl
Sia2BH	disiamylborane
TAS	tris(diethylamino)sulfonium
TBAF	tetra-n-butylammonium fluoride
TBDMS	t-butyldimethylsilyl
TBDMS-Cl	t-butyldimethylsilyl chloride
TBHP	t-butyl hydroperoxide
TCE	2,2,2-trichloroethanol
TCNE	tetracyanoethylene
TES	triethylsilyl
Tf	triflyl (trifluoromethanesulfonyl)
TFA	trifluoroacetic acid
TFAA	trifluoroacetic anhydride
THF	tetrahydrofuran
THP	tetrahydropyranyl
TIPBS-Cl	2,4,6-triisopropylbenzenesulfonyl chloride
TIPS-Cl	1,3-dichloro-1,1,3,3-tetraisopropyldisiloxane
TMEDA	tetramethylethylenediamine [1,2-bis(dimethylamino)ethane]
TMS	trimethylsilyl
TMS-Cl	trimethylsilyl chloride
TMS-CN	trimethylsilyl cyanide
Tol	tolyl
TosMIC	tosylmethyl isocyanide
TPP	meso-tetraphenylporphyrin
Tr	trityl (triphenylmethyl)
Ts	tosyl (p-toluenesulfonyl)
TTFA	thallium trifluoroacetate
TTN	thallium(III) nitrate

1.1 Alkylations of Enols and Enolates

DRURY CAINE

University of Alabama, Tuscaloosa, AL, USA

1.1.1	INTRODUCTION	1
1.1.2	ALKYLATIONS OF METAL ENOLATES OF KETONES AND ALDEHYDES	3
1.1 1.1 1.1 1.1	 2.1 Regiospecific Alkylations of Unsymmetric Acyclic and Cyclic Ketones 2.2 Stereochemistry of Enolate Alkylations 2.3 Cycloalkylation Reactions of Saturated Ketones 2.4 Alkylations of Metal Enolates of Saturated Aldehydes 	3 12 18 20
1.1.3	ALKYLATIONS OF METAL DIENOLATES OF α , β -UNSATURATED KETONES	21
1.1.4	ALKYLATIONS VIA SILYL ENOL ETHERS AND OTHER ENOL DERIVATIVES	25
1.1.5	ALKYLATIONS OF CARBONYL COMPOUNDS VIA THEIR NITROGEN DERIVATIVES	28
$1.1 \\ 1.1$	 5.1 Regiochemistry and Stereochemistry of Alkylations of Nitrogen Derivatives of Carbonyl Compounds 5.2 Enantioselective Synthesis via Alkylations 	28 35
1.1.6	ALKYLATIONS OF METAL ENOLATES OF CARBOXYLIC ACID DERIVATIVES	39
1.1 1.1 1.1 1.1 1.1 1.1	 5.1 Introduction 5.2 Diastereoselective Alkylations of Exocyclic and Endocyclic Enolates of Carboxylic Acid Derivatives 5.3 Diastereoselective Alkylations of Acyclic Enolates of Carboxylic Acid Derivatives 5.4 Diastereoselective Alkylations of Acyclic Carboxylic Acid Derivatives Containing Chiral Auxiliary Group 5.5 Cycloalkylations of Enolates of Carboxylic Acid Derivatives 5.6 Alkylations of Dienolates of a,β-Unsaturated Carboxylic Acid Derivatives 5.7 Asymmetric Synthesis via Alkylations of Anions of Masked Carboxylic Acid Derivatives 	39 39 42 95 44 48 50 53
1.1.7	ALKYLATIONS OF STABILIZED METAL ENOLATES	54
1.i 1.i	7.1 Alkylations of Monoanions of β -Dicarbonyl and Related Compounds 7.2 γ -Alkylations of Dianions of β -Dicarbonyl Compounds	54 58
1.1.8	REFERENCES	58

1.1.1 INTRODUCTION

The α -alkylation of carbonyl compounds by their conversion into nucleophilic enolates or enolate equivalents and subsequent reaction with electrophilic alkylating agents provides one of the main avenues for regio- and stereo-selective formation of carbon-carbon σ -bonds.¹⁻⁶ Classical approaches to α -alkylation typically involve the deprotonation of compounds containing doubly activated methylene or methine groups and having pK_a values of 13 or below by sodium or potassium alkoxides in protic solvents. Since these conditions lead to monoenolates derived from deprotonation only at the α -site of the substrate, the question of the regioselectivity of C-alkylation does not arise (however, there is competition between C- and O-alkylation in certain cases). In more recent years, dienolates of β -dicarbonyl compounds have been utilized in γ -alkylations with excellent success.

 α -Alkylations of active methylene compounds followed by removal of one of the activating groups to produce monofunctional products is quite useful for large scale synthesis of relatively simple carbonyl compounds. Indeed, if one wishes to use conventional methods of enolate formation in alkylations of

simple unsymmetrical ketones, extraneous methylene blocking groups or activating groups, which are to be removed eventually, must be applied.^{2,4} However, modern synthetic methodology has focused upon α -alkylation of monofunctional carbonyl systems *via* their metal enolates or enolate equivalents produced quantitatively by direct or indirect methods. Since monofunctional carbonyl compounds have pK_a values in the 20–27 range, strong bases in aprotic solvents are required for direct enolate formation by deprotonation. Many of these substrates (aldehydes, carboxylic acid derivatives and certain ketones) contain only one reactive α -site. In these cases, the question of regioselective enolate formation does not arise, although in most cases stereoisomeric (*E*)- or (*Z*)-enolates or mixtures of these species may be produced, depending upon the substrate and reaction conditions.² However, the challenging problems encountered in the total synthesis of steroids, terpenoids or other complex molecules have required the development of methods of regioselective alkylation of unsymmetrical ketones, which are capable of enolization in two directions.

Upon treatment with strong bases unsymmetrical ketones such as 2-methylcyclohexanone (1) can lose a proton to give the tri- or tetra-substituted enolates (2) or (3). Under kinetic control, where the ketone is added slowly to a soluble strong base in an aprotic medium, the enolate mixture composition is determined by the relative rates of proton removal from the alternative α -positions. Because the less-substituted α -position is less hindered, the less-substituted enolate, *e.g.* (2), is usually favored under kinetic conditions. However, when relatively basic Group I sodium or potassium enolates are employed, enolate equilibration, brought about by proton transfer reactions involving the initially formed alkylated ketones and unalkylated enolates, is normally faster than alkylation. As a result, products derived from alkylation of the more highly substituted enolate (3), which is normally the more thermodynamically stable isomer, usually predominate. Rapid proton transfer reactions also account for the formation of di- and even trialkylated products in alkylations involving potassium or sodium enolates.



An illustration of this behavior is provided in equation (1). A 67:33 mixture of the less- and more-substituted potassium enolates was produced upon treatment of 2-methylcyclohexanone with tritylpotassium in DME. However, the major product of alkylation of this mixture with methyl iodide was 2,2-dimethylcyclohexanone and significant amounts of tri- and tetra-methylcyclohexanone were also obtained.⁷



The modern era of enolate alkylations began in 1961 when Stork *et al.*⁸ showed that the less thermodynamically stable lithium 1-enolate (4) of *trans*-10-methyl-2-decalone, produced by lithium-ammonia reduction of the corresponding octalone, could be alkylated regiospecifically with reactive alkylating agents in liquid ammonia. Regiospecificity was not maintained if potassium or sodium was employed in the reduction, or if the enolate (4) was transferred to DMSO prior to alkylation. These conditions gave 3alkyl-substituted products *via* the more thermodynamically stable 2-enolate, as did deprotonation of the parent decalone with strong bases.



The above results led to an explosion of interest in the use of lithium enolates in α -alkylation reactions. It was quickly demonstrated that specific lithium enolates of a variety of unsymmetrical ketones could be trapped with the more reactive alkylating agents in liquid ammonia or aprotic solvents such as DME or THF. Clearly, the ratio of the rate of alkylation to the rate of proton transfer is significantly larger for less basic lithium enolates than for their sodium and potassium counterparts.

In order to exploit the unique behavior of lithium enolates, a wide variety of highly hindered, nonnucleophilic bases such as LDA, LICA, LITMP, lithium hexamethyldisilylamide (LHDS) and lithium tetramethyldiphenyldisilylamide (LTDOS), which kinetically convert unsymmetrical ketones to their less-substituted enolates with high regioselectivity, have been developed.² Lithium *t*-octyl-*t*-butylamide (LOBA) has recently been shown to be a highly useful base for this purpose.⁹ Likewise, a variety of indirect methods for formation of lithium enolates such as lithium-ammonia reduction of enones,¹⁰ conjugate additions of organocuprate reagents to enones¹¹ and the cleavage of silyl enol ethers with organolithium reagents¹² have been developed or expanded. An excellent survey of methods of formation of lithium and other enolates derived from less electropositive metals is found in Volume 2, Chapter 1.4 of this series.¹³

The majority of α -alkylations currently being reported are accomplished via lithium enolates. However, these reactions have several drawbacks, *e.g.* loss of regioselectivity when sterically congested β , β disubstituted enolates are involved or when less reactive alkylating agents are employed, polyalkylation, elimination rather than substitution of the alkylating agent, and poor π -facial diastereoselectivity in many cases. Fortunately, most of these limitations have been overcome by the methodology to be described below.

This chapter will provide coverage of the scope and limitations of alkylations of metal enolates of saturated and unsaturated ketones, aldehydes and carboxylic acid derivatives, together with a discussion of alkylations of various enols and enolate equivalents. Where applicable, the utility of these reactions for the diastereoselective and enantioselective synthesis of α -substituted carbonyl compounds will be described. Inevitably, the coverage of a vast research area such as this will be incomplete and in part will reflect the author's interests. However, it is hoped that most of the useful methods of carbon-carbon σ -bond formation by alkylations of enolates and enols will be included.

1.1.2 ALKYLATIONS OF METAL ENOLATES OF KETONES AND ALDEHYDES

1.1.2.1 Regiospecific Alkylations of Unsymmetrical Acyclic and Cyclic Ketones

There are numerous base-solvent combinations that are capable of quantitatively converting even weakly acidic simple ketones into their enolate anions.¹ However, in order to avoid aldol condensation and unwanted equilibration of enolates of unsymmetrical ketones during enolate formation, it is best to choose conditions under which the ketone, the base and the metal enolate are soluble. Likewise, solutions should be produced when indirect methods of enolate formation are employed. While certain metal cations such as Hg^{2+} form α -metallated ketones, most of the metal cations in Groups I, II and III exist as *O*-metallated tautomers.¹³ For organotin derivatives both the *O*-metallated and *C*-metallated forms probably exist in equilibrium.¹⁴

Metal enolate solutions consist of molecular aggregates (6) such as dimers, trimers and tetramers in equilibrium with monomeric covalently bonded species (7), contact ion pairs (8) and solvent-separated ion pairs (9), as shown in Scheme 1.¹⁵ The nature of the metal cation, the solvent and, to a degree, the structure of the enolate anion itself may significantly influence the extent of association between the anion and the metal cation.¹ In general, the factors that favor loose association, *e.g.* solvent-separated ion pairs, lead to an increase in the nucleophilicity of the enolate toward alkylating agents and also its ability to function as a base, *i.e.* to participate in proton transfer reactions.





Both physical measurements^{15,16} and kinetic studies¹⁷ show that association between metal cations and enolate anions is stronger, *i.e.* the oxygen-metal bond has more covalent character, when small metal cations and less electropositive metals are involved. The degree of aggregation and extent of association of

metal cations and enolate anions is reduced as one goes from monodentate *n*-donor solvents such as diethyl ether and THF to di- and poly-dentate *n*-donor solvents like DME, the higher glymes and crown ethers. These effects are particularly marked in dipolar aprotic solvents such as HMPA, DMSO and DMF, which strongly solvate metal cations by a π -donor mechanism.

The reactivity of metal enolates can be greatly enhanced by adding dipolar aprotic substances to their solutions in ethereal solvents such as THF, where HMPA has become the additive of choice for this purpose.

The geometry of the enolate anion and the degree of substitution on the carbon-carbon double bond may influence the degree of enolate aggregation. Less-substituted metal enolates, which are probably capable of forming larger aggregates, are somewhat less reactive than more-substituted enolates.¹⁸ In the latter species, the increase in nucleophilicity resulting from the presence of an electron-releasing alkyl group is offset by the steric bulk of such groups, which retard alkylation rates. Substituents which are located at positions other than the α -position may influence the extent of enolate aggregation in a particular solvent, or sterically hinder the approach of an alkylating agent. These effects not only influence the rates but the stereochemistry of enolate alkylations (Section 1.1.2.2).

Activated allylic, benzylic and propargylic halides and α -halo esters, as well as methyl and primary halides, are useful reagents for C-alkylation of metal enolates. The order of reactivity of the various halogens is as expected for an S_N2 process, *i.e.* I > Br > Cl. The order of reactivity of a particular halide is usually benzyl > allyl > primary alkyl and branching at the β -carbon reduces reactivity. Branching at the α -carbon reduces reactivity significantly. Thus, secondary halides react with metal enolates slowly and yields of α -alkylation products are usually poor. Tertiary halides undergo elimination almost exclusively upon reaction with enolates. Elimination is also a serious problem when β -vinyl (homoallylic), β -alkynic and β -phenyl halides as well as β -halo ketones, esters or nitriles are employed in alkylations. Alkyl tosylates and benzenesulfonates are also useful reagents for α -alkylation of metal enolates. Their reactivities are similar to those of the corresponding iodides.

O-Alkylation is generally not a problem when lithium and less electropositive metal enolates are involved, unless highly reactive alkylating agents such as dialkyl sulfates, trialkyloxonium salts or α -chloro ethers are employed in solvents of relatively high polarity. As the degree of dissociation between the metal cation and the enolate anion is increased, the anion becomes more capable of exercising its ambident character. The competition between C- and O-alkylation has been traditionally explained in terms of the Pearson–Klopman principle, which states that hard electrophilic reagents should attack the hard oxygen atom of the enolate anion, while soft reagents should attack the soft α -carbon site.^{19,20} However, recent *ab initio* calculations²¹ and experimental gas-phase data²² indicate that free enolate anions react on oxygen irrespective of the hardness or softness of the electrophilic reagent. Coordination of the metal cation with the oxygen must be responsible for reducing its reactivity to the point where C-alkylation is usually preferred in solution. Other data on the variations of C:O-alkylation ratios with the nature of the alkyl halide indicate that it is the ability of the halide to influence the structures of enolate aggregates rather than its 'hardness' or 'softness' which is the controlling factor.²³ This section will be devoted mainly to reactions in which little, if any, O-alkylation is observed.

Enolate equilibration and di- and poly-alkylation are the major side reactions, which lead to reduced yields of desired products in ketone alkylations. These processes occur as a result of equilibration of the starting enolate (or enolate mixture) with the neutral monoalkylation product(s) *via* proton transfer reactions. Polyalkylation may also occur when bases, in addition to the starting enolate, which are capable of deprotonating the monoalkylated ketone are present in the medium. With symmetrical ketones, *e.g.* cyclopentanone and cyclohexanone, the problem of regioselectivity does not arise. However, except under special conditions, polyalkylation occurs to a significant extent during enolate alkylations of more kinetically acidic ketones such as cyclobutanone, cyclopentanone and acyclic ketones, particularly methyl ketones.²⁴ Polyalkylation is also a troublesome side reaction with less acidic ketones such as cyclohexanone.

Scheme 2 shows the results of two studies on the methylation of the lithium enolate of cyclopentanone (10), which was prepared by deprotonation of the ketone with trityllithium in DME²⁵ or by cleavage of the 1-trimethylsiloxycyclopentene with methyllithium in THF.²⁶ A significant quantity of over-alkylation occurred when the enolate was treated with methyl iodide, particularly when DME was employed as the solvent at room temperature. Also, as indicated in Scheme 2, Noyori and coworkers²⁶ showed that by adding 3 equiv. of HMPA to the enolate (10) and reducing the temperature at which the reaction was conducted, the yield of 2-methylcyclopentanone was greatly improved.

As shown in Scheme 3 conversion of the lithium enolate (10) to its complex lithium triethylaluminum enolate prior to alkylation in DME-HMPA at room temperature also significantly reduced the amount of polyalkylation.²⁵ However, the most dramatic results were obtained by addition of dimethylzinc to the



Scheme 2





A comparison of the results of *n*-butylation of the enolate (10) in THF-HMPA with and without the addition of 1 equiv. of dimethylzinc is shown in Scheme 4. Although *n*-butyl iodide is 50-100 times less reactive than methyl iodide, monobutylation occurred with high efficiency in the presence of the additive. It is not clear whether dimethylzinc reacts with (10) to produce a complex lithium dimethylzinc enolate, or whether it influences the reactivity of the enolate in some other way, possibly by changing its state of aggregation and/or the structures of the aggregates.



Alkylation of Carbon

The addition of copper(I) salts to THF-HMPA solutions of the lithium enolate (10) caused a dramatic reduction in the amount of polyalkylation.²⁷ The exact role played by the copper(I) cation is not clear. 2-Benzylcyclopentanone, free of the dibenzylation product, was prepared in 67% yield by alkylation of the 'naked' tetraalkylammonium cyclopentanone enolate generated *in situ* by reaction of trimethylsiloxycyclopentene with BTAF in the presence of benzyl bromide in THF;²⁸ 2-allylcyclopentanone contaminated with $\leq 3\%$ diallylated ketone was obtained in a similar manner using trimethylsiloxycyclopentene, allyl bromide and TAS difluorotrimethylsiliconate in THF.²⁹ The less-substituted tri-*n*-butyltin enolate of 2-allylcyclopentanone, prepared by reductive desulfurization of the corresponding β -keto sulfone with lithium in liquid ammonia followed by addition of tri-*n*-butyltin chloride, was alkylated with allyl bromide to give the 2,5-diallyl ketone in good yield (Scheme 5).³⁰



6

Scheme 6

Lithium,²⁵ lithium triethylaluminum,²⁵ sodium triethylboron,³¹ sodium triethanolamine borate,³¹ potassium triethylboron³² and tri-*n*-butyltin^{25,33} cyclohexanone enolates have been successfully monoalkylated. In Scheme 6 the behavior of the lithium enolate of cyclohexanone (11) and the lithium triethylaluminum enolate upon reaction with methyl iodide is compared. The latter enolate gives better results since no dimethylation products were detected, but clearly the cyclohexanone enolate (11) is much less prone to dialkylation than the cyclopentanone enolate (10). Scheme 6 also provides a comparison of the results of alkylation of the potassium enolate of cyclohexanone, where almost equal amounts of mono- and di-alkylation occurred, with the alkylation of the potassium triethylboron enolate where no polyalkylation occurred. The employment of more covalently bonded enolates offers an advantage in cyclohexanone monoalkylations but not nearly as much as in the cyclopentanone case.

The same conditions which are useful for monoalkylation of enolates of symmetrical ketones also favor regioselective alkylation of specific enolates of unsymmetrical ketones. Since more highly substituted enolates of unsymmetrical ketones are usually the more thermodynamically stable and also undergo alkylation faster than the less-substituted enolates, it is generally possible to successfully alkylate these species.¹⁸ For example, as shown in equation (2), when a (Z):(E) = 74:26 mixture of the lithium 2-enolates of 2-heptanone (12) was benzylated in DME, the 3-benzyl derivative was the only monoalkylation product obtained, although significant amounts of dibenzylated products were produced. However, as shown in equation (3), benzylation of an 84:16 mixture of the lithium 1-(13) and a (Z)/(E)-mixture of the lithium 2-enolates of 2-heptanone (12) gave primarily the internally benzylated product. The equilibrium composition of the structurally isomeric lithium enolates of 2-heptanone in DME at 25 °C is 65% (Z)-2-enolate, 22% (E)-2-enolate and 13% 1-enolate.³⁴ Thus, substantial enolate equilibrium occurred during the reaction.



Equation (4) shows the results of benzylation of an 87:13 mixture of the enolates (13) and (12) in the presence of the dipolar aprotic additive HMPA, where a substantial amount of benzylation occurred via the terminal enolate.¹⁷ Other additives, such as DMF and benzo-14-crown-4, increased the amount of terminal alkylation in comparison with the use of DME alone. The influence of the additive is possibly to prevent the formation of aggregates of the enolate (13) and lithium bromide formed during the reaction. This could cause the terminal enolate to remain as a reactive dimer, which undergoes alkylation more rapidly than equilibration to the internal enolate.¹⁷



Alkylation of Carbon

Deprotonations of unsymmetrical ketones with lithium bases under thermodynamic control usually give the more-substituted enolates with good regioselectivity. However, even better results are provided by various indirect methods of enolate formation, such as cleavage of more-substituted TMS enol ethers with organolithium reagents in ethereal solvents^{12,18,35} or lithium amide in liquid ammonia³⁶ or lithium-ammonia reduction of appropriate α,β -unsaturated ketones. The former procedure provides a 'clean' enolate,³⁷ one free of extraneous bases, which can promote polyalkylation. Several highly regioselective methods of preparing thermodynamic TMS enol ethers of unsymmetrical ketones such as 2-methylcyclohexanone have been reported recently.^{38–40} Alkylations with less reactive alkylating agents such as *n*-butyl iodide provide an important test of regiocontrol in enolate alkylations.¹ An excellent yield of 2-*n*-butyl-2-methylcyclohexanone was obtained when the more highly substituted TMS enol ether of 2-methylcyclohexanone was treated with lithium amide in liquid ammonia–THF and the resulting enolate (3; M = Li) reacted with *n*-butyl iodide (Scheme 7).³⁶



Scheme 7

Regioselective alkylations at C-6 of 2-methylcyclohexanone have been accomplished via the alkylation of thermodynamically unstable trisubstituted lithium, lithium triethanolamine borate,³¹ potassium triethylboron,³² tri-*n*-butyltin¹⁴ and benzyltrimethylammonium enolates (*cf.* 2).²⁸ Alkylation is faster than equilibration for the more reactive alkylating agents. Although enolate equilibration has been shown to compete with butylation using *n*-butyl iodide under certain conditions,^{41,42} butylation of the enolate (2; M = Li) in liquid ammonia–THF gave a mixture of *cis*- and *trans*-2-methyl-6-butylcyclohexanone along with 2-methylcyclohexanone in an 83:17 ratio in 90% yield; no 2,2-dialkylcyclohexanone was obtained in this reaction (Scheme 8).³⁶



 α,β -Dialkylation of enones by conjugate addition-enolate alkylation has been widely used for the synthesis of α,β -dialkyl ketones, particularly cyclohexanones and cyclopentanones. This subject has been recently reviewed by Taylor, who has pointed out the problems associated with tandem, one-pot conjugate addition-enolate alkylation procedures.¹¹ The most frequently used reagents for conjugate additions are lithium dialkylcuprates, and the enolates produced in such reactions are likely to be largely lithium enolates.⁴³ However, copper(I) salts, which may influence lithium enolate reactivity even in low concentration, are undoubtedly present in solution.⁴¹ Conjugate additions usually proceed best in ether, but enolate alkylations are unacceptably slow in this solvent. However, if the reactivity of the medium is adjusted by addition of appropriate cation-complexing agents such as TMEDA^{41,44} or THF/HMPA,⁴⁵ or if the ether is exchanged for a more appropriate solvent for alkylation, *e.g.* DME,⁴⁴ regioselective alkylations can be effected with simple, reactive alkylating agents in many cases. Conjugate additions can also be performed in THF.⁴¹ Examples of the use of one-pot conjugate addition-enolate alkylations for the synthesis of 2,3-dialkylcyclohexanones are shown in Scheme 9.^{41,44}

A useful technique to accomplish overall vicinal dialkylation of enones is to trap the enolate initially formed in the conjugate addition with TMS-Cl, and then regenerate the enolate under suitable conditions for its alkylation. Lithium 1-enolates of 3,5-dialkylcyclohexanones generated from the reaction of the corresponding TMS enol ethers with lithium amide in liquid ammonia–THF can be alkylated efficiently in liquid ammonia–THF, even with an unreactive alkylating agent such as *n*-butyl iodide (Scheme 10).³⁶



2,3-Dialkylcyclopentanones, *e.g.* the prostaglandins, are found widely in nature. An obvious route to the synthesis of these systems is a triply convergent approach involving tandem conjugate addition-enolate alkylation of an appropriate cyclopentenone.^{41,46-48} However, the propensity of cyclopentanone enolates toward equilibration has presented difficulties, which until recently have prevented synthesis of the natural products themselves by this approach.^{47,48} Using cyclopentenone as the substrate, Posner and coworkers found that tandem conjugate addition-enolate alkylations could be accomplished in reasonable yields with relatively small nucleophiles and electrophiles under carefully controlled conditions (Scheme 11).⁴¹ However, the alkylation failed when bulkier nucleophiles and electrophiles bearing the functionality of prostaglandin side chains were employed.

Patterson and Fried found that the clean lithium enolate (14), generated by conjugate addition of the lithium divinylcuprate (15) to cyclopentenone with subsequent trapping of the initial enolate with TMS-Cl and cleavage of the TMS enol ether with lithium amide in liquid ammonia–THF, could be alkylated in a reasonable yield with the (Z)-allylic iodide (16) to give the 11-deoxyprostaglandin derivative (17;



Scheme 12).⁴⁶ However, the sequence failed when applied to a protected 4-hydroxycyclopentenone because of equilibration of the initially formed 1-enolate to the 5-enolate followed by β -elimination of the protected hydroxy substituent.



Scheme 12

Noyori and coworkers⁴⁷ have provided an elegant solution to the problem of enolate equilibration by transmetallation of the enolate obtained from addition of the homochiral phosphine-complexed vinylcopper reagent (19) to the protected homochiral 4-hydroxycyclopentenone (18), with tri-*n*-butyltin chloride, followed by addition of the alkylating agent (16). This sequence gave the PGE₂-type product (20) in 78% yield along with 3% of the C-8 epimer (Scheme 13). Likewise, the propargylic iodide related to (16) provided a 5,6-didehydro-PGE₂ derivative in 82% yield, but use of methyl 7-iodoheptanoate as the alkylating agent provided only a 20% yield of the corresponding PGE derivative. It was found that halide ions must be present in the medium for alkylation to proceed; thus, penta- or hexa-valent tin species may be involved.

An interesting and simple variation on the three-component coupling protocol for prostaglandin synthesis has been recently reported by Noyori and coworkers.²⁶ This variation involved conjugate addition of a reagent formed from a 1:1 mixture of dimethylzinc and the (E)-vinyllithium reagent (21) to the enone (18), followed by alkylation of the enolate intermediate with the propargylic iodide (22) to give





the prostaglandin derivative (23) in 71% overall yield (Scheme 14). The remarkable ability of dimethylzinc to suppress proton transfer reactions of lithium enolates is further illustrated by this experiment.



Scheme 14

In related work, Johnson and Penning showed that the vinylcopper reagent (19) added to a homochiral protected 4,5-dihydroxycyclopent-2-enone to give an enolate intermediate, which upon trapping with the (Z)-allylic iodide (16) in the presence of HMPA gave a 2,3-dialkylated precursor of (-)-PGE₂ methyl ester in good yield.⁴⁸ Apparently, the presence of the oxygen substituent at C-5 of the starting enone reduces the rate of enolate equilibration and allows alkylation to occur at the 2-position.

Regiospecific alkylations have been employed in the synthesis of a large number of alkylated decalones, hydrindanones and tricyclic and steroidal ketones.¹ Regiospecific annulations are often carried out by: (i) regiospecific α -alkylation of a specific enolate with an electrophilic reagent containing latent 2- or 3-ketoalkyl functionality; (ii) transformation of the side chain into a 1,4- or 1,5-diketone; and (iii) formation of a ring by an intramolecular aldol condensation.⁴⁹ (*E*)-3-Trimethylsilyl-2-butenyl iodide⁵⁰ and *t*-butyl γ -iodotiglate⁵¹ are some of the more useful alkylating agents for this purpose. The requisite lithium enolates are often generated by lithium–ammonia reduction of α , β -unsaturated ketones or cleavage of TMS enol ethers with methyllithium. The use of kinetically generated 1-enolates of *trans*-2-decalones in sequences of this type provide important routes to phenanthrene derivatives, which are useful intermediates in steroid total synthesis (Scheme 15).

Base-promoted alkylations of bicyclic ketones such as 1-decalones and 1-hydrindanones give largely nonangular alkylated products. The application of blocking groups at the 2-position provided an early approach to angular alkylations of such systems.^{1,52,53} However, angularly alkylated ketones can be prepared from appropriate lithium enolates generated by indirect methods.^{8,18,54} These enolates exhibit a diastereofacial bias toward the formation of *cis*-fused products. A possible explanation for this is presented in Section 1.1.2.2.



Scheme 15

The focus of this section has been upon uncatalyzed α -alkylations of metal enolates with alkylating agents bearing leaving groups at sp^3 -hybridized carbon atoms. However, in certain cases where conventional alkylations give poor results, palladium(0)-catalyzed regioselective allylations of complex potassium^{55,56} and lithium⁵⁷ triethylboron, zinc,⁵⁸ trialkyltin^{59,60} and other enolates⁵⁸ have been accomplished with relatively inert reagents, *e.g.* allylic acetates. These reactions proceed *via* nucleophilic attack of the metal enolate on a π -allylpalladium complex formed by oxidative addition of the allylic reagent to palladium. The reactions of trialkyltin enolates may involve the intermediacy of 'ate' complexes.⁶⁰

 α -Arylations and α -vinylations of metal enolates also provide regioselective routes for carbon-carbon bond formation. The literature on this subject prior to 1978 has been reviewed.⁶¹ More recent work in this area has involved palladium-catalyzed reactions of trialkyltin enolates with aryl^{62,63} and vinyl bromides.⁶²

1.1.2.2 Stereochemistry of Enolate Alkylations

Existing evidence indicates that C-alkylations of metal enolates with common electrophiles proceeds by an S_N 2-type mechanism; that is, the highest occupied molecular orbital (HOMO) of the enolate attacks the lowest unoccupied molecular orbital (LUMO) of the alkylating agent. Scheme 16 illustrates the principle of stereoelectronic control,^{64,65} which states that the electrophile should approach in a plane perpendicular to the enolate to allow maintenance of maximum orbital overlap in the transition state (24) between the developing C—C bond and the π -orbital of the carbonyl group.





Recent *ab initio* molecular orbital calculations indicate that the developing electrophile–C(1)–C(2) angle is actually larger than 90°.²¹ However, they provide no evidence for the suggestion that the electro-

phile's trajectory is tilted away from perpendicularity to the enolate plane because of the repulsive interaction between the electrophile LUMO and the oxygen atom of the enolate HOMO.⁶⁶

The suggestion that a cyclic six-membered ring transition state of the type (25), in which the metal cation assists the reaction by coordinating with the leaving group seems unlikely.⁶⁷ Such a transition state requires very unfavorable bond angles for an $S_N 2$ displacement.⁶⁸ However, since enolate aggregates are likely to be present, the possibility exists that a transition state (26), related to (24), in which the cation of a second enolate ion-pair assists the reaction by coordinating with the leaving group, is involved.



If the metal enolate contains a center of chirality, diastereoselection may be exhibited in the C—C bond formation process. Evans² has identified three classes of metal enolates in which chirality transfer may occur: (i) endo- and exo-cyclic enolates such as (27) or (28), which contain a chiral center (*) in a ring bonded to the enolate at two points; (ii) acyclic enolates such as (29) or (30), in which the moiety containing the chiral center (*) is bonded to the enolate at only one point; and (iii) chelated enolates such as (31) or (32), in which the chiral center is a part of the chelate ring. (Z)-Endocyclic enolates are also possible for large ring cyclic ketones.

The stereochemical outcome of alkylations can be ascribed to stereoelectronic and steric effects, including the influence of allylic strain⁶⁹ on metal enolate conformations. Metal enolate alkylations are strongly exothermic, and the results of many studies indicate that the transition states closely resemble the reactants.^{1,2,4} Therefore, steric factors within the enolate play a dominant role in determining π -diastereofacial selectivity. This is illustrated in Scheme 17 for alkylation of the chiral lithium enolate of 4-*t*butylcyclohexanone (**33a**), which can take place *via* path A, leading to a chair conformation (**34a**) of the product having the alkyl group axial to the ring and *trans* to the *t*-butyl group. Path B attack leads initially to a twist-boat conformation (**35a**), which may undergo conformational inversion to the chair conformation (**36a**) having the new group equatorial to the ring and *cis* to the *t*-butyl group. Paths A and B (Scheme 17) are normally referred to as axial and equatorial alkylation, respectively.



Scheme 17

In the enolate (33a), the top and bottom π -faces of the molecule present about the same degree of steric hindrance to approach of the alkylating agent. However, if the transition state resembled the alkylation product, the thermodynamically more stable chair conformation (34a) would be expected to develop much faster than the less stable twist-boat conformation; thus, axial alkylation would be strongly

Alkylation of Carbon

preferred. However, this is not the case. House *et al.*⁷⁰ reported that ethylation of (**33a**) with ethyl iodide in DME gave approximately equal amounts of axial and equatorial alkylation products. Similar results were obtained in the alkylation of (**33a**) with methyl iodide.⁷¹ More recent studies on the stereochemistry of alkylation of (**33a**) under carefully controlled conditions and on the alkylation of the corresponding tetraalkylammonium enolate have indicated that there is a small, but apparently real, preference for axial alkylation. For example, axial:equatorial product ratios of 74:26⁷² and 70:30²⁸ have been reported for the reaction of (**33a**) and the corrsponding 'naked' enolate with methyl iodide in THF. If equilibration of the product mixture occurred during alkylation, the axial:equatorial alkylation ratio would be reduced. Perhaps this accounts for the lower ratios obtained in the earlier studies. As expected, alkylations of the lithium enolate (**33a**) were less stereoselective when more reactive alkylating agents than methyl iodide were used.^{70,72} Also, when the counterion was varied, the axial:equatorial ratio was shown to follow the order: Li > Na > K.⁷² However, the cation effect was quite small.

Lithium enolates having α -methyl substituents, such as (33b) and the related species (37b), derived from 5-*t*-butylcyclohexanone, show a somewhat greater stereoselectivity of axial alkylation than the corresponding α -unsubstituted compounds.⁷³ For example, the enolate (37a) gave the products (38a) and (39a) in a 68:32 ratio upon treatment with methyl iodide in THF, while the 2-methyl-substituted derivative (37b) gave an 83:17 mixture of (38b) and (39b) upon reaction with trideuteriomethyl iodide in DME (Scheme 18).⁷³ Similarly, a greater stereoselectivity for axial alkylation has been observed for other α substituted enolates compared with their counterparts lacking α -substituents.^{71,74–78}



U, K' = MC, K' = K' = H	83:17
c: $R^2 = Me, R^1 = R^3 = H$	25:75
d: $R^3 = Me$, $R^1 = R^2 = H$	96:4

Scheme 18

As depicted in (40), House and Umen have proposed that 2-substituted enolates undergo distortion in order to relieve interactions between the R group and the OM substituent and to avoid eclipsing ($A^{1,2}$ -strain)⁶⁹ of the OM group by the 6-quasiequatorial group (hydrogen) and of the R group by the 3-quasiequatorial group (hydrogen). Thus, some rehybridization of the *p*-orbital at C-2 toward an *sp*³-orbital should occur. In such a distorted species, the top π -face of the enolate should be more open to attack by an electrophile than the bottom π -face.⁷³



(40)

Quasiaxial and quasiequatorial substituents at C-6 in conformationally rigid lithium enolates such as (33a) have a very small effect on the stereoselectivity of the reaction.^{71,72} In the case of the lithium 1-enolate of *cis*-3-methyl-5-*t*-butylcyclohexanone (37c), the steric effect of the quasiequatorial methyl group at C-3 and the stereoselectronic preference of axial alkylation oppose each other, and equatorial al-kylation is preferred by 3 to 1. However, in the *trans* isomer (37d), where the C-3 methyl group is quasi-axial, these effects reinforce each other and axial alkylation is significantly favored by 24 to 1 (Scheme 18). Relatively rigid 1-enolates of 2-substituted cyclohexanone derivatives having quasiequatorial substituents at C-6⁷⁷ or C-3^{74-76,78} show an enhanced tendency toward axial alkylation. In such systems, the dihedral angle θ in structure (40) should increase even more with the placement of bulky groups at C-6 and C-3.⁷³ The stereoselectivity of alkylations of sterically congested enolates such as those derived from 2,6-dimethyl-6-phenyl- and 2-methallyl-6-methyl-6-phenyl-cyclohexanone⁷⁷ is probably best accounted for by equatorial alkylation of a conformation of the 1-enolates having the 6-phenyl groups quasiaxial.²

The results of several studies (cf. Scheme 9) show that alkylations of conformationally mobile 1-enolates of 3-alkyl- and 2,3-dialkyl-cyclohexanones give products having the new groups at C-2 *trans* to the groups at C-3 with stereoselectivities in the 75 to 95% range.^{41,44,45,79} Enolates of this type may exist as an equilibrium mixture of conformations (**41**) and (**42**). Conformation (**42**), having the 3-alkyl quasiaxial, is likely to be quite important, particularly when an alkyl group is present at C-2, because $A^{1,2}$ -strain would destabilize conformation (**41**), which has a 3-alkyl group quasiequatorial. Steric interactions involving the 3-alkyl group and the approaching electrophilic reagent appear to be minimized in either of the two possible transition states which lead to the introduction of the new group *trans* to the C-3 substitutent. However, it is somewhat surprising that the stereoselectivity in favor of the *trans* product is apparently somewhat greater for the allylation of the C-2-unsubstituted enolate (**41a**) \leftrightarrow (**42a**) than for the 2-methyl enolate (**41b**) \leftrightarrow (**42b**).^{45,79}



The results of methylations of lithium 1(2)-enolates of $(43a)^{75}$ and $(43b)^{76}$ of *trans*-2-decalones are shown in Scheme 19. An enolate such as (43) may be regarded as a special case of a conformationally rigid 3-alkylcyclohexanone derivative. As in the case of *t*-butylcyclohexanone derivatives such as (37), the presence of a substituent at the α -position of the enolate significantly increases the stereoselectivity in favor of axial alkylation. This may be attributed to a distortion of the enolate, which forces the methyl group at C-1 downward to relieve its interaction with the OLi group and the quasiequatorial 8-methylene group.⁷³ A tricyclic lithium enolate related to (43a) was found to undergo axial alkylation almost exclusively.⁸⁰



A comparison of the data for alkylation of the 10-methyl-2-decalone lithium enolate $(44)^{75}$ with those for enolate (43a) clearly shows that if axial alkylation involves development of a 1,3-interaction with an

axial methyl group, equatorial alkylation is the major pathway (Scheme 20). Likewise, alkylations of lithium 2-enolates of *trans*-2-decalones (**45a**) and (**45b**) gave largely axial and equatorial methylation products, respectively (*cf.* Scheme 20).⁷²



Methylation of the conformationally rigid exocyclic enolate (46) derived from 4-*t*-butylcyclohexyl methyl ketone has been shown to yield an 85:15 mixture of products (47) and (48) derived from equatorial and axial attack of the electrophile, respectively (Scheme 21).⁸¹ In species such as (46) geometric factors associated with a stereoelectronically controlled approach to either π -face of the enolate are very similar. It has been suggested that the interaction of the filled *p*-orbital at C-1 with the vacant symmetrical antibonding orbital of the C(2)—C(3) and C(5)—C(6) bonds causes an increase in electron density on the equatorial π -face of the enolate and favors approach of the electrophile from that side.⁸² However, the more generally accepted argument is that in a reactant-like transition state the axial hydrogen atoms at C-3 and C-5 hinder axial attack, while equatorial attack is relatively unencumbered.⁸¹



Scheme 21

Angular alkylations of the lithium 1(9)-enolate (49) of 1-decalone with methyl iodide in liquid ammonia-ether⁸ or DME⁵⁴ have been shown to yield a mixture of products in which the *cis* isomer (50) is favored by 4:1 or 5:1 over the *trans* isomer (51; Scheme 22). The 10-methyl derivative of (49) also has been reported to give *cis*-9,10-dimethyl-1-decalone exclusively upon methylation.⁴⁵ In systems such as (49) the enolate double bond is exocyclic to the six-membered nonoxygenated ring. Thus, the same factors that control the stereochemistry of alkylation of simple exocyclic enolates such as (46) are likely to be involved. In the axial approach pathway, the electrophile would experience a 1,3-interaction with the axial hydrogen atoms at C-4, C-5 and C-7. Thus, the equatorial approach pathway, which is less sterically hindered, is preferred. The observation that alkylation of (49) with the highly reactive alkylation agent trimethyloxonium 2,4,6-trinitrobenzenesulfonate gave 95% of the *cis*-fused ketone (50), is in keeping with this explanation.⁵⁴ Under these conditions, the transition state should resemble the reactant enolate to an even greater extent than when methyl iodide is employed as the alkylating agent.





Angular alkylations of 1-decalone enolates provide important models for angular alkylations of 18nor-D-homo steroids. The manner in which structural modifications influence *cis:trans* product ratios in alkylations of various enolates of 1-decalones containing blocking groups at C-2 has been thoroughly investigated⁵² and reviewed.¹

Angular alkylations of lithium enolates of hydrindanones with carbonyl groups in the five.⁸³ or sixmembered ring⁸⁴ yield *cis*-fused products with almost complete stereoselectivity. The lithium enolate of bicyclo [2.2.1]heptan-2-one undergoes *exo* alkylation with very high stereoselectivity.^{72,85} The presence of a *syn* methyl group at C-7 reduces the preference for *exo* alkylation, but it is still preferred over *endo* alkylation by about 3:1 unless a 5,6-double bond is also present;⁷² then, *endo* attack is preferred.^{86,87} The expected steric effects control the stereochemistry of alkylation of other bridged bicyclic systems.^{8,88}

The results presented in Schemes 11 to 14, as well as numerous other studies,^{85–91} show that alkylations of 1-enolates of 3-alkyl- and 3-alkenyl-cyclopentanones usually yield *trans*-2,3-disubstituted cyclopentanones with high stereoselectivity. As shown in Scheme 23, the stereoselectivity of alkylation of the 1-enolate of 3-phenylcyclopentanone is dramatically affected by the presence of copper(I) in the medium. Similar results have been obtained for the corresponding 3-methyl-3-phenylcyclopentanone enolates. On the basis of ¹H and ¹³C NMR evidence and on the results of deprotonation of 3-phenylcyclopentanone with strong bases, Posner and Lentz²⁷ have proposed that lithium–arene coordination occurs in the enolates that would be present in the reactions shown in Scheme 23 as well as related cyclohexanone enolates. It was proposed that this complexation directs alkylation at C-2 to occur primarily *cis* to the phenyl ring at C-3 in both lithium and 'copper' enolates, but that the presence of copper(I) in the medium retards equilibration of the 2,3-*cis* product to the more stable 2,3-*trans* product. It should be noted that the extent of diallylation was greatly diminished in the presence of copper(I) species compared with the lithium enolate alone. Alkylation was found to occur *trans* to the phenyl group when a methyl substituent was present at C-2.



Scheme 23

Extraannular chirality transfer has been observed for alkylations of acyclic ketone enolates having chiral β -carbon atoms (Scheme 24).^{92,95} In these reactions, methylation occurs *anti* to the β -dimethylphenylsilyl and β -isopropyl groups with good to excellent diastereoselectivity. Variation in the size of the

Alkylation of Carbon

alkyl group at the β -carbon atom of β -silicon-substituted enolates has indicated that the preference for *anti* attack to silicon is probably electronic in origin. In the transition states for alkylation, *anti* attack of the electrophile may occur on a conformation of the enolate such as (52) with the hydrogen atom eclipsing the double bond and the larger groups staggered,⁹²⁻⁹⁴ or a conformation such as (53) with the more electron-releasing β -substituent perpendicular to the plane of the enolate.⁹⁶ Recent studies by McGarvey and Williams on alkylations of chiral ester enolates indicate that the latter type of transition state is likely to be more favorable.⁹⁷ Further discussion of extraannular and chelate-enforced chirality transfer is provided in Sections 1.1.6.3 and 1.1.6.4.



1.1.2.3 Cycloalkylation Reactions of Saturated Ketones

Cycloalkylation reactions have been utilized for the synthesis of a variety of cyclic compounds including monocyclic, fused-ring, spirocyclic and bridged systems. These reactions require the generation of an enolate in the presence of a suitably disposed leaving group (halide, tosylate, epoxide, *etc.*). Possible modes of cycloalkylation are represented in Scheme 25. If the transition state for *C*-alkylation is strained, *O*-alkylation becomes a competitive process. Traditionally, sodium or potassium enolates were employed for these reactions, but, more recently, lithium enolates, generated by treatment of the substrate with bulky, non-nucleophilic bases, have been generally utilized.

Ketone enolate cycloalkylations may be used to form small to medium and even larger rings under appropriate conditions. Space limitations permit only a few examples of these reactions to be covered here. The literature prior to 1978 has been reviewed.¹

Detailed studies on the modes of cycloalkylation of the terminal lithium enolates of 5-bromo-3,3-dimethyl-2-pentanone (54) and 6-bromo-3,3-dimethyl-2-hexanone (55)^{98,99} have shown that (54) undergoes exclusively *O*-cycloalkylation, while (55) undergoes only *C*-cycloalkylation (Scheme 26). The transition states for the two modes of alkylation are shown in formulas (56) and (57). In the *C*-alkylation, the electrophilic ω -carbon atom must approach the enolate α -carbon perpendicularly along a path colinear with the C—Br bond that undergoes cleavage. This pathway is highly strained when n = 0 or 1 and *O*-alkylation, which can take place *via* attack of the basic *syn* nonbonding electron pair lying in the plane of the enolate, is favored. When n = 2 or larger the *C*-alkylation transition state can be easily attained and this pathway becomes favored.



Scheme 27 illustrates how different conditions of enolate formation may affect the outcome of cycloalkylation reactions.¹⁰⁰ Deprotonation of the bromo ketone (**58**) under equilibrating conditions with potassium *t*-butoxide in *t*-butyl alcohol resulted in *exo* cycloalkylation *via* the more-substituted metal enolate to give the 5,5-fused bicyclic ketone (**59**) in good yield. On the other hand, when (**58**) was treated with LDA in THF under kinetic conditions, deprotonation occurred at the terminal position and *endo* cycloalkylation was faster than enolate equilibration and gave the 5,7-fused ketone (**60**), also in good yield. The conversions of 3(2-tosyloxyethyl)cyclohexanone to bicyclo[2.2.2]octan-2-one¹⁰¹ and of 3(2-tosyloxyethyl)-¹⁰² and 3(2-bromoethyl)-cyclopentanones¹⁰³ to the corresponding bicyclo[2.2.1]heptan-2-ones provide additional examples of cycloalkylations which lead to bridged ring systems.



i, Bu^tOK, Bu^tOH, pentane, 25 °C, 30 min ; ii, LDA, THF, -72 to 65 °C, 2 h

Scheme 27

Exo cycloalkylations have been used to synthesize *cis*-1-decalones. For example, treatment of 2methyl-3(4-tosyloxybutyl)cyclohexanone with sodium *t*-pentylate in benzene gave *cis*-9-methyl-1-decalone (50) in 60% yield.¹⁰⁴ Also, as shown in Scheme 28, conjugate addition-cycloalkylation was employed to synthesize a *cis*-fused decalone related to the sesquiterpene, (\pm)-valerane.⁴¹ Apparently, in these cases, the enolate intermediate adopts a conformation having the 4-bromobutyl side chain quasiaxial, and C—C bond formation occurs *via* equatorial attack to give initially a twist-boat conformation of the product.



In addition to (±)-valerane, a wide variety of other sesquiterpenes, including (±)-ishwarane,¹⁰⁵ (±)-ishwarone,¹⁰⁶ copaene,¹⁰⁷ ylangene,¹⁰⁷ (±)-seychellene^{108,109} (±)-sativene,¹¹⁰ (±)-longifoline,¹¹¹ (±)-copacamphene,¹¹² (±)-damsin,¹¹³ (±)- $\Delta^{9(12)}$ -capnellene,¹¹⁴ (±)-pentalenene,¹¹⁵ (–)- β -vetivone¹¹⁶ and (±)- β -eudesmol¹¹⁷ have been synthesized by pathways involving cycloalkylation of saturated ketone enolates.

1.1.2.4 Alkylations of Metal Enolates of Saturated Aldehydes

Because of their tendency to undergo self (aldol) condensations, Cannizzaro and Tichshenko reactions, the direct alkylation of aldehydes via metal enolates has numerous drawbacks.¹¹⁸ Indeed, poor results have been obtained, even with preformed lithium enolates.¹¹⁸ However, much better yields have been achieved through the use of enolates derived from less or more electropositive metals than lithium. In early experiments in this area, Odic and Perevre showed that tri-n-butyltin aldehyde englates, prepared by reaction of aldehyde enol acetates with tri-n-butyltin methoxide, were C-alkylated in good yields with reactive alkylating agents in THF-HMPA.¹¹⁹ Although these reactions were slower than those of alkali metal enolates and higher temperatures and longer reaction times were required, O-alkylation was not a problem. It was also shown by Jung and Blum that the tri-n-butyltin enolate of acetaldehyde could be alkylated by this method.¹²⁰ Later, van der Gen and coworkers¹²¹ found that potassium enolates of α, α -disubstituted aldehydes, such as 2-methylpropanal and 2-ethylhexanal, prepared by reaction of the substrate with potassium hydride in THF, were rapidly C-alkylated in good yields with reactive alkylating agents at room temperature. When less reactive primary and secondary iodides were used, there was a significant amount of competition between C- and O-alkylation. When aldehydes containing two α protons were employed, mixtures of mono- and di-alkylated products were produced because of rapid proton exchange between the alkylated products and the starting enolates. Aldol condensations probably occur in these reactions, but, because potassium is a relatively poor chelating cation, there is likely to be a mobile equilibrium between the potassium aldolate and the enolate so that the alkylation step can proceed to completion.

Aldehydes that contain only one α -hydrogen atom may be alkylated in reasonable yields with reactive alkylating agents such as methyl iodide, allyl chloride or benzyl chloride in an emulsion of benzene and 50% aqueous sodium hydroxide in the presence of a catalytic amount of a tetra-*n*-butylammonium

salt.¹²² When less reactive alkylation agents were employed or when the aldehyde contained bulky α -substituents, aldol condensations and/or O-alkylations became significant problems.

Aldol condensations of more complex aldehydes are often sufficiently slow to allow successful alkylation reactions. There are numerous examples of aldehyde enolate methylations in the field of natural product synthesis.^{123,124} As shown in Scheme 29, the methylation of a tricyclic aldehyde, which was employed in the synthesis of (\pm) -rimuene, provides an illustrative case.¹²³ As expected for an exocyclic enolate intermediate such as (61), the methyl group was introduced equatorial to the six-membered ring with a high degree of stereoselectivity. α -Alkylated aldehydes may be prepared efficiently by alkylations of enamines, Schiff base anions, hydrazone anions and other methods. A discussion of this methodology is provided in Section 1.1.5.



1.1.3 ALKYLATIONS OF METAL DIENOLATES OF α , β -UNSATURATED KETONES

 α,β -Unsaturated ketones of the general structure (62), which contain α' - and γ -protons undergo deprotonation with strong bases under kinetic control to yield cross-conjugated metal dienolates such as (63).¹ Under thermodynamic conditions, extended metal dienolates such as (64) are produced.¹

A variety of lithium dialkylamide bases can be used to produce cross-conjugated lithium dienolates, which may then be alkylated with even less reactive alkylating agents, *e.g.* propyl iodide, in good to excellent yields without equilibration to the corresponding extended dienolates.¹²⁵ α '-Alkylations of cyclohex-2-enones, certain cyclopent-2-enones, 1(9)-octalin-2-ones and steroidal 4-en-3-ones have been accomplished by this procedure.¹

Stork and Danheiser have developed a highly useful procedure for the synthesis of 4-alkylcyclohex-2enones, which involves α' -alkylations of cross-conjugated lithium dienolates of 3-alkoxycyclohex-2enones, followed by metal hydride reduction of the carbonyl group and hydrolysis (Scheme 30).¹²⁶ Numerous applications of this procedure have been reported.^{1,127,128} Two different alkyl groups may be introduced at the 6-position of a cyclohex-2-enone derivative without difficulty.¹²⁷ While dialkylation is generally not a problem in alkylations of cross-conjugated dienolates of cyclohex-2-enones, it was observed when relatively acidic 3-chlorocyclohex-2-enones were employed.¹²⁹

A comparison of the data shown in Scheme 31 for methylations of cross-conjugated lithium dienolates of 1(9)-octalin-2-ones (**65a**) and (**65b**) with those shown in Scheme 20 for the corresponding decalone 2enolates (**45**) reveals that the tendency for axial β -attack is significantly greater in the former systems.^{66,130} Similar results have been obtained in alkylations of related steroidal cross-conjugated dienolates.¹³¹ Two explanations for these results have been offered: (i) the cross-conjugated dienolate systems can easily adopt a relatively low energy quasi-*cis* conformation with the β -face exposed to electrophilic attack;^{130,131} or (ii) the effect of the 1,9-double bond in species such as (**65**) could alter the trajectory of attack of the electrophile on the HOMO of the dienolate (*cf.* formula **24**).⁶⁶ However, Agami *et al.* showed that the cross-conjugated dienolate (**66**), isomeric with (**65b**), gave exclusively the product of equatorial methylation, *i.e.* α -attack at C-1.¹³⁰ The electronic factor would be expected to increase the degree of axial alkylation of (**66**) relative to the corresponding simple enolate (**44**). Since this was not



Scheme 30





By the use of appropriate dialkylating agents, it is possible to spiroannulate cyclohex-2-enones at the 6-position via kinetic lithium dienolates. This method was used by Stork *et al.*¹³² to prepare the enone (67), a key intermediate for the total synthesis of (\pm) - β -vetivone (Scheme 32). In the cycloalkylation step the cross-conjugated dienolate presumably adopts a conformation with the 5-methyl group quasiaxial to the ring to avoid A^{1,2}-strain and the new C—C bond is formed *trans* to this group.



Scheme 32

Although the reaction failed with cyclopent-2-enone itself, methylations of kinetic lithium dienolates of 3-alkylcyclopent-2-enones have been carried out in acceptable yields.¹³³ Intramolecular α' -alkylations

of 5,6- and 6,6-fused ring bicyclic α , β -unsaturated ketones bearing 2-bromoethyl groups at the angular positions have been reported.¹³⁴ A total synthesis of (±)-clovene has been accomplished using an intramolecular α' -enolate cycloalkylation of 4-(3-chloropropyl)-4-methyl-6-(2-ethylallyl)cyclohex-2-enone to give 5-(2-ethylallyl)-1-methylbicyclo[3.3.1]non-2-en-4-one as a key step.¹³⁵

In general, the thermodynamically stable extended dienolates (64) have been prepared by deprotonation of enones (62) with sodium or potassium alkoxides in protic solvents or with sodium or potassium hydride in aprotic solvents.^{1,2,4,49} Kinetically formed cross-conjugated lithium enolates may be converted into the corresponding extended systems in the presence of excess ketone but in certain cases equilibration is quite slow.¹³⁶ Presumably, because the π -electron density is higher at the α -carbon than the γ carbon, extended dienolates normally react with alkylating agents to produce α -alkyl- β , γ -unsaturated ketones.¹³⁷

If the α -position of an extended dienolate is unsubstituted, dialkylation may be an important reaction or, under certain conditions, the major reaction, particularly when reactive alkylating agents, *e.g.* methyl iodide, are employed.¹³⁸ Equation (5) provides an example of the synthetic utility of the dimethylation process.¹³⁹ Ringold and Malhotra have provided a detailed mechanism to account for the facile dialkylation of these systems.¹³⁸ Briefly, deprotonation by bases present in the medium converts the initially formed α -alkylated β , γ -unsaturated ketone to its dienolate, which then undergoes a second alkylation faster than the β , γ -double bond is isomerized to the α , β -position. When the alkylating agent is of low reactivity or its concentration is limited by dropwise addition, the rate at which the double bond moves into conjugation becomes faster than the second alkylation step. As is described in Section 1.1.5.1, extended lithiated dienamines are the nucleophilic reagents of choice for α -methylation of α , β -unsaturated ketones.¹⁴⁰



With less reactive alkylating agents monoalkylation of extended dienolates at the α -position is feasible. In connection with the total synthesis of steroids, dienolates of bicyclic enones such as (**68a**) and (**68b**), or various derivatives of the saturated carbonyl function of these compounds, have been alkylated with a variety of reagents containing latent 3-ketoalkyl functionality.⁴⁹ The products of these reactions may be converted into tricyclic enones such as (**69a**) and (**69b**). Even diannulating agents which allow elaboration of both the A and B rings of the steroid skeleton have been utilized.⁴⁹



Normally C-alkylation occurs predominantly in reactions of alkali metal extended dienolates with the usual alkylating agents in solvents of low to medium polarity. However, highly reactive benzyloxymethyl halides yield extensive amounts of O-alkylation products;¹⁴¹ when highly polar solvents are used O-alkylation may occur even with unreactive alkylating agents.¹⁴² As expected, the C-alkylation:O-alkylation ratio is influenced by: (i) the polarity of the solvent; (ii) the nature of the alkylating agent; and (iii) the nature of the metal cation.

The stereochemistry of alkylations of extended dienolates of enones such as (70) has been extensively investigated (Scheme 33).^{141,143} In general, the results are similar to those found for the related decalone enolates (43a) and (44), *i.e.* steric factors within the anion play a dominant role. Thus, axial attack is preferred with (70; $R^1 = H$), but equatorial attack is strongly favored when an angular methyl group is present (70; $R^1 = Me$). There is a modest preference for axial alkylation when an angular ethoxycarbonyl

group is present, as in (70; $R^1 = CO_2Et$). This effect has been attributed to the polar nature of the angular ester group, but since an ethoxycarbonyl group is smaller in size than a methyl group, an explanation based upon steric effects alone possibly can apply.



As shown in Scheme 34, a rather profound solvent effect on dienolate alkylation diastereoselectivity has been noted for the steroidal enone (71).¹⁴³ Such large solvent effects have not been documented for other systems. Possible explanations based upon the position of the transition state along the reaction coordinate and/or 'specific' solvation of the dienolate have been advanced to account for preferential axial alkylation in benzene and equatorial alkylation in *t*-butyl alcohol.¹⁴³ However, in view of the fact that the degree of aggregation of the dienolate as well as the structure of the aggregates may be modified considerably in going from one solvent to the other, rationalization of the results is difficult.



Scheme 34

Stork *et al.* have shown that heteroannular extended dienolates such as (73), which contain substituents at both the α - and γ -positions, undergo predominantly equatorial alkylation (Scheme 35).¹⁴⁴ The dienolate (73) was produced by lithium-ammonia reduction of the tricyclic dienone (72) and the product of its alkylation with 1-bromo-3-chloro-2-butene and hydrolysis of the resulting enol ether, *i.e.* (74), was a key intermediate in a short, highly stereoselective synthesis of (±)-adrenosterone. It was pointed out that equatorial alkylation is obtained with dienolates such as (73) and related compounds because a *peri* interaction (Me \leftrightarrow OMe) of the α - and γ -substituents forces the ring A to adopt a half-boat conformation in which the α -face of the π -system is accessible to attack.

 γ -Alkylations of extended dienolates of β -amino- α , β -cyclopentenones,¹⁴⁵ β -amino- α , β -cyclohexenones^{136,146} and related acyclic systems¹⁴⁷ have been reported. Koreeda and coworkers¹⁴⁸ have found that deprotonation of β -alkoxy- α , β -cyclopentenones with LHDS in THF gave dienolates, which underwent regioselective γ -alkylations. However, attempted γ -alkylations of β -alkoxy- or β -thioalkoxy- α , β cyclohexenones were unsuccessful.^{146,149}

As shown in Scheme 36, 2,2-disubstituted-5-alkyl-3(2*H*)-furanones also undergo γ -alkylation via their extended dienolate intermediates (75).¹⁴⁹ Similarly, 4-isopropyl-6-methyl-(2*H*)-pyran-2-one, which may be regarded as a vinylogous 5-alkyl-3(2*H*)-furanone, was deprotonated and alkylated at the methyl group.¹⁵⁰



Scheme 36

The reason γ -alkylations occur in certain systems having heteroatom β -substituents is unclear. The electron density at the γ -carbon of the dienolate may be enhanced through conjugation with the unshared electron pair on the heteroatom. The presence of an exocyclic β , γ -double bond in systems such as (75) seems to be an important structural requirement for γ -alkylation.¹⁴⁹

The reader is referred to two reviews for numerous examples of intramolecular γ -alkylations of ketone dienolates.^{1,2} Lansbury *et al.* have achieved some success in synthesizing γ -alkylated α , β -unsaturated ketones *via* the use of dienolates of γ -benzenesulfonyl enones¹⁵¹ and trilithiated derivatives of α' -benzenesulfonyl enones¹⁵² as nucleophiles.

Potassium hydride in THF is the base-solvent combination of choice for preparing dienolates of α,β unsaturated aldehydes. These enolates are more stable than the corresponding saturated aldehyde enolates.¹²¹ Equation (6) shows that excellent yields of α,α -disubstituted β,γ -unsaturated aldehydes can be obtained by trapping α -substituted aldehyde potassium dienolates with allylic halides.



1.1.4 ALKYLATIONS VIA SILYL ENOL ETHERS AND OTHER ENOL DERIVATIVES

As discussed in Section 1.1.2, C-alkylations of metal enolates are restricted to the use of relatively S_N2 -reactive alkylating agents. Loss of regioselectivity, failure of the reaction, or at least poor yields are observed with alkylating agents that undergo the S_N2 reactions slowly and/or are prone to elimination. A significant advance in C-alkylation chemistry was made when it was shown that silyl enol ethers, and also enol esters and alkyl enol ether derivatives of ketones and aldehydes, are effective nucleophiles for combination with S_N1 -reactive alkylating agents in the presence of Lewis acids.^{12,153–157} The most commonly used Lewis acid catalysts are titanium tetrachloride and zinc bromide. Methylene chloride is used as the solvent. Experimentally, best results are obtained by adding the Lewis acid catalyst to a mixture of the enol derivative and the alkylating agent.^{156,158} TMS triflate is also an effective catalyst for this type of reaction.¹⁵⁹

Tertiary halides, acetates and methyl ethers, aryl-activated secondary halides, prenyl halides and acetates, acetals, ketals, thioacetals, α -chloroalkyl ethers, and, particularly, α -chloroalkyl phenyl sulfides have been widely used as alkylating agents. Such reagents may contain a variety of functional groups including alkenes, esters, halides, ketones, nitro groups and silyl ethers.

As illustrated in Scheme 37, using the data of Reetz *et al.*,¹⁶⁰ the more- and less-substituted TMS enol ethers of 2-methylcyclohexanone have been *t*-butylated with high regiospecificity and in good yields by this method. Paterson has also reported that titanium tetrachloride promoted phenylthioalkylations of the
more- and less-substituted trimethylsilyl enol ethers of a wide variety of unsymmetrical cyclic and acyclic ketones can be accomplished regiospecifically.¹⁵⁶ Loss of regioselectivity was not observed even for rather hindered systems.



As shown in Scheme 38, several primary alkyl-substituted cyclohexanones have been prepared by Lewis acid catalyzed phenylthioalkylation of the TMS enol ether of cyclohexanone followed by reductive removal of a phenylsulfenyl group.¹⁵⁶ The two-step neopentylation sequence is particularly noteworthy. This methodology has been used to prepare numerous α -alkylated cyclic and acyclic ketones. α -Alkylated aldehydes can be produced in a like manner. α -Alkylidenation can also be accomplished by oxidative removal of sulfur.¹⁵⁶ Lee and coworkers have found that TMS triflate-catalyzed reactions of silyl enol ethers of cyclic ketones and aldehydes with saturated and unsaturated 1,1-dimethoxy- ω -trimethylstannanes, followed by addition of titanium tetrachloride, provide novel routes to fused and spirocyclic ring systems.¹⁶¹ Phenylthiomethylstannylations of silyl enol ethers have also been reported.¹⁶²

OSiMe ₃	SPh R Cl conditions	W-2 RaNi acetone-ethanol, 28 °C	O R
R	Conditions	Yield (%)	Yield (%)
Ме	1.1 equiv. TiCl ₄ , CH ₂ Cl ₂ , -23 °C	78	95
Pr ⁿ	1.1 equiv. TiCl ₄ , CH ₂ Cl ₂ , -23 °C	83	90
Pr ⁱ	1.1 equiv. TiCl ₄ , CH ₂ Cl ₂ -23 °C	66	91
Bu ^t	0.02 equiv. $ZnBr_2$, CH ₂ Cl ₂ , 20 °C	78	93

Scheme 38

Studies pertaining to diastereoselectivity in Lewis acid catalyzed alkylations of enol derivatives have been limited. Reetz has reported that t-butylation of 1-trimethylsiloxy-4-t-butylcyclohex-1-ene gave an 85:15 mixture of *cis*- and *trans*-2,4-di-t-butylcyclohexanone, which could result from kinetic equatorial and axial alkylation, respectively. However, equilibration of the products, which would favor formation of the former isomer, was not ruled out. Titanium tetrachloride promoted phenylthiomethylation of the more-substituted TMS enol ether of 1-decalone gave a 4:1 mixture of *cis*- and *trans*-fused 1-decalones.¹⁵⁶ In this case, where equilibration of the product could not occur, the diastereoselectivity was similar to that of methylation of the corresponding lithium enolate (49).^{8,54}

Lewis acid catalyzed intramolecular alkylations of silyl enol ethers containing S_N l-reactive functionality provide useful routes to a variety of carbocyclic systems.^{163,164} Smith *et al.*¹⁶⁵ have employed an intramolecular Mukaiyama reaction of the enol derivative (76) to produce the tetracyclic system (77) (equation 7). This transformation was a key step in their elegant synthesis of jatrophone. The synthesis demonstrated that the molecules involved in the reaction may contain some rather sensitive functionality and that intramolecular formation of large rings is possible. Cycloalkylation of a silyl enol ether has been reported by Magnus and Carter in connection with their research on the antitumor antibiotic esperamicin A (equation 8).¹⁶⁶ They found that upon treatment with titanium chloride the cobalt-complexed propargylic ether (78) gave the cyclization product (79) as a stable compound. Oxidative decomplexation of (79) produced an aromatized tricyclic ketone *via* a diynene intermediate. Lewis acid catalyzed intramolecular cyclizations of enol acetates, containing suitably disposed isopropenyl and isopropylidene groups capable of forming tertiary carbocations, have been employed in a creative manner in the synthesis of several complex natural products.¹⁵⁵



Unlike lithium dienolates, where α -alkylation is the rule in simple systems, Lewis acid catalyzed reactions of extended silyl dienol ethers with S_N 1-reactive alkylating agents may give extensively or exclusively γ -alkylation products. Scheme 39 shows some selected data from the work of Fleming and coworkers who have thoroughly studied reactions of silyl enol ethers such as (80) derived from crotonophenone.¹⁵⁴ These results clearly show that the γ : α -alkylation ratio is increased by: (i) the use of alkylating agents, which give better-stabilized electrophilic species; and (ii) silyl dienol derivatives which contain electron-withdrawing groups on silicon.¹⁵⁴ The effect of a change in the Lewis acid was not very clear cut because overall product yields often changed dramatically. In other work, it was shown that when methyl groups are present at the γ -position of the dienol ether, the γ : α -ratio is reduced.¹⁶⁷



Scheme 39

The position of alkylation of homoannular silyl dienol ethers is apparently dramatically influenced by the nature of the alkylating agent and catalyst. Reetz and coworkers¹⁵⁵ reported that 1-trimethylsiloxy-

Alkylation of Carbon

1,3-cyclohexadiene is *t*-butylated exclusively at the α -position with *t*-butyl acetate/zinc iodide, while Fleming's research group¹⁵⁴ found that phenylthioalkylation of this compound occurred exclusively at the γ -position with 1-chloro-1-phenylthio-*n*-butane/zinc bromide (Scheme 40). The reasons for these contrasting results are unclear.



Scheme 40

Lewis acid catalyzed alkylations of cross-conjugated silyl dienol ethers provide routes to α' -alkylated ketones. A short synthesis of the sesquiterpene (\pm)-*ar*-turmerone has been accomplished using the cross-conjugated TMS dienol ether of mesityl oxide (equation 9).¹⁶⁸



Palladium(0)-catalyzed α -allylations of TMS enol ethers can be carried out cleanly with allylic carbonates.¹⁶⁹ These reactions are highly regioselective, *e.g.* the more- and less-substituted TMS enol derivative of 2-methylcyclohexanone (*cf.* Scheme 37) gave 2-allyl-2-methylcyclohexanone and 2methyl-6-allylcyclohexanone, respectively. Allylations of aldehyde silyl enol ethers occur similarly. Allylations of enol acetates occur with allyl carbonates in the presence of catalytic amounts of palladium(0) complexes and tri-*n*-butyltin methoxide.¹⁶⁹

1.1.5 ALKYLATIONS OF CARBONYL COMPOUNDS VIA THEIR NITROGEN DERIVATIVES

1.1.5.1 Regiochemistry and Stereochemistry of Alkylations of Nitrogen Derivatives of Carbonyl Compounds

Prior to the discoveries that lithium and other less electropositive metal cations were valuable counterions for enolate alkylations, the Stork enamine reaction was introduced to overcome problems such as loss of regioselectivity and polyalkylation that plagued attempts to alkylate sodium or potassium enolates of ketones or aldehydes.^{170–172} Methods of synthesis of enamines by reactions of ketones and aldehydes with secondary amines have been thoroughly reviewed.^{171,172} Enamine alkylations are usually conducted in methanol, dioxane or acetonitrile. Enamines are ambident nucleophiles and C- and N-alkylations are usually competitive. Subsequent hydrolysis of the C-alkylated product (an iminium salt) yields an α -alkylated ketone, while the N-alkylated product (a quaternary ammonium salt) is usually water soluble and relatively inert to hydrolysis.

The monomethylation of the pyrrolidine enamine of β -tetralone shown in Scheme 41 provides a highly successful example of the reaction. Attempted alkylation of the ketone with 1 equiv. of methyl iodide in the presence of a strong base gave almost exclusively the recovered ketone and its dimethylated derivative.



Scheme 41

Although dialkylation was not a problem in the above case, it is frequently a significant side reaction. As illustrated in Scheme 42 for the pyrrolidine enamine of cyclohexanone (81), the initially formed C-alkylated iminium salt (82) can transfer a proton to the starting enamine to yield the enamine of the alkylated product, *e.g.* (83), which can be further alkylated to give the dialkylated iminium salt (85). As a result, the starting ketone (derived from the unalkylated iminium salt 84), the monoalkylated product and the dialkylated product may be isolated after hydrolysis. Although dialkylation is not such a serious problem when higher molecular weight alkylating agents are used, because mono- and di-alkylated ketones can be separated by distillation, the yields in enamine alkylations are usually rather low.



Scheme 42

When unreactive alkylating agents are reacted with enamines, both the C- and N-alkylation processes are essentially irreversible. However, strongly electrophilic reagents such as methyl iodide, allylic halides, benzylic halides, propargylic halides, α -halo ethers and α -halocarbonyl compounds and nitriles can undergo reversible N-alkylation. Thus, higher yields of C-alkylated products are obtained with these reagents. Allylic and propargylic groups are transferred from nitrogen to carbon by intramolecular 3,3-

Alkylation of Carbon

sigmatropic rearrangements, but, in other cases, the process is intermolecular. In some cases, it has been found that the C- to N-alkylation product ratio may be increased by increasing the temperature.¹⁷³

A comparison of the results of the methylation of the pyrrolidine enamine of cyclohexanone (81; Scheme 42) with those shown in Scheme 43 for the methylation of the *n*-butylisobutylamine derivative (86) shows that yields are improved with the latter compound.¹⁷⁴ Presumably, the more sterically hindered amine reduces the amount of *N*-alkylation and retards the rate of proton transfer between the *C*-alkylated iminium salt and the unreacted enamine.



Scheme 43

In contrast to the results of base-catalyzed alkylations of sodium or potassium enolates under thermodynamic control, alkylations of enamines of unsymmetrical ketones occur largely or exclusively at the less-substituted α -position. More-substituted ketone enamines are destabilized relative to the less-substituted isomers by A^{1,2}-strain involving the substituents on the nitrogen atom and at the β -carbon atom. Although in most systems some of the more-substituted enamine is present in equilibrium with the lesssubstituted isomer, the former is less reactive toward *C*-alkylation because steric effects prevent effective overlap of the lone pair of electrons on nitrogen with the carbon–carbon π -bond.

Enamines are relatively unreactive nucleophiles and compared with metal enolates the transition states for their alkylations lie much closer to the products. Thus, the stereoelectronic effect is predominant and axial alkylation of cyclohexanone derivatives is observed.^{175,176} There is such a high propensity for axial alkylation that it obtains even when a 1,3-interaction of the approaching electrophile and a quasiaxial substituent exists. Thus, methylation of the pyrrolidine enamine (**87**), which is expected to exist largely in the conformation (**87**a) having the C-6 side chain quasiaxial, gave largely the *cis*-2,6-disubstituted ketone (**88**) after hydrolysis under nonequilibrating conditions (Scheme 44).¹⁷⁶ It was considered unlikely that the *cis*-2,6-disubstituted product could have arisen by equatorial alkylation of the conformer (**87b**) of the starting enamine, which has the side chain quasiequatorial.

C-Alkylations of hindered aldehyde enamines can be effected with a variety of alkylating agents, but only allylic and benzylic reagents are useful for alkylations of unhindered systems.¹⁷⁷ Acyclic, homoannular and heteroannular dienamines undergo alkylation primarily at the α -positions.^{170,178} As discussed above, reactions of enamines with electrophiles containing sp^3 -hybridized carbon atoms have numerous limitations. On the other hand, enamine reactions with electrophilic alkenes are highly useful and have received wide coverage in the literature.¹⁷¹

In order to overcome some of the limitations of enamine alkylations and consequently those associated with metal enolate alkylations, Stork and Dowd introduced metallated imines as metal enolate equivalents in 1963.¹⁷⁹ These reagents, along with the closely related metallated hydrazones introduced by Stork and Benaim¹⁴⁰ and thoroughly investigated by Corey and Enders¹⁸⁰ and oxime dianions¹⁷² provide the synthetic chemist with the potential of achieving an enormous amount of regiochemical and stereochemical control when synthesizing α -alkylated aldehydes and ketones.^{171,172,181} Additionally, as described in Section 1.1.5.2, through the use of metallated imines and hydrazones derived from certain chiral, nonracemic imines and hydrazine derivatives, asymmetric syntheses of α -alkyl carbonyl compounds can be achieved with high enantioselectivity.



Metallated imines can be formed from imines derived from enolizable carbonyl compounds by deprotonation with Grignard reagents or organolithium reagents, but more recent studies have generally involved the use of lithium dialkylamides, e.g. LDA, as the base. Alternative methods of producing metallated imines, e.g. addition of t-butyllithium to the imine double bond of 2-azadienes,¹⁸² are known.

In their initial work, Stork and Dowd showed that halomagnesium imine alkylations offered the following advantages over enamine alkylations: (i) C-alkylations occurred in high yields with unreactive primary and secondary halides, even those prone to elimination; (ii) aldehyde derivatives were directly C-alkylated in high yields; (iii) alkylations usually occurred at the less-substituted α -position of unsymmetrical systems; and (iv) formation of the halomagnesium salt could be conducted in the presence of the alkylating agent, which allowed *in situ* generation and trapping of unstable species. Examples of these reactions are presented in Scheme 45. Lithiated imines often exhibit similar behavior to halomagnesium species, but the presence of the more electropositive metal cation, *i.e.* Li⁺, allows alkylations to be conducted at much lower temperatures. The success of these reactions depends upon the fact that metallated imines undergo proton transfer reactions much more slowly than the corresponding metal enolates. Alkylations of metallated imines are usually conducted in THF or DME. Hydrolysis of the alkylated imine to the alkylated ketone can be effected with aqueous dilute mineral acids, aqueous acetic acid (recommended for aldimine hydrolysis),¹¹⁸ or, if epimerizations of α -alkyl carbonyl compounds need to be avoided, slightly acidic aqueous buffered solutions.

Whitesell and Whitesell¹⁷² have tabulated some of the many types of electrophilic reagents that C-alkylate metallated imines. These are potent nucleophiles and undergo substitution reactions even with weakly electrophilic species such as epoxides and oxetanes. Lithiated ketimines and aldimines have been frequently used in reactions with alkylating agents containing latent 2-keto (or aldehydo) groups¹⁸³ or 3-keto (or aldehydo) groups.¹⁸⁴

Recent research by Bergbreiter, Newcomb, Meyers and their respective coworkers¹⁸⁵ has shown that a variety of factors, such as the base, the temperature of deprotonation, and the size of the substituent on nitrogen, control the structure of the metallated imine and ultimately the regiochemistry of the alkylation reaction. In contrast to metal enolates, where the more-substituted species is usually the more thermodynamically stable, less-substituted *syn*-metallated ketimines, *e.g.* (89), are the most thermodynamically stable of the possible isomers of unsymmetrical systems. An explanation for the greater stability of *syn* imine anions compared with *anti* imine anions has been presented by Houk, Fraser and coworkers.¹⁸⁶



They suggested that the latter species are destabilized by an electrostatic repulsion between the polar metal-nitrogen bond, which lies in the N-C-C plane and the imine β -carbon atom, which bears a substantial amount of negative charge. Because there is significant carbanion character at the β -carbon, the less-substituted syn-metallated ketimine (89) would be expected to be more stable than the more-substituted syn isomer (90).



Ironically, species such as (89; M = Li) are also the major isomers obtained when unsymmetrical ketimines are deprotonated with LDA at -78 °C.¹⁸⁵ There is a kinetic preference for deprotonation *anti* to the substituent on nitrogen. At very low temperatures, deprotonations with LDA occur at the less-substituted carbon atom *via* the less stable (Z)-imine. Rearrangements then occur to the *syn*, less-substituted imine, which is in turn alkylated. On the other hand, deprotonations conducted at -23 to 0 °C with LDA are faster than imine isomerization and the imine anion mixture composition reflects the (E):(Z) ratio of the starting imine. Anti-syn imine anion isomerization occurs to give predominately the more-substituted *syn*-metallated imine (*cf.* 90), which then undergoes alkylation. These results are summarized in Scheme 46 using the *t*-butylimine of 2-butanone as an example.

The first examples of alkylations at the more-substituted position of an unsymmetrical ketimine were actually reported by Hosomi *et al.*, who carried out deprotonations with alkyllithium reagents at lower temperatures (Scheme 47).¹⁸⁷

Fraser *et al.* have shown that, unlike the corresponding lithium enolates, conformationally rigid lithiated imines such as (91) undergo axial alkylation almost exclusively (Scheme 48).¹⁸⁸ Reaction of (91) with methyl iodide at -78 °C gave 90% of the *syn* axial alkylation product (92). The small quantities of the *anti* axial (93) and *anti* equatorial products (94) obtained were considered to arise from isomerization of (92). Hydrolysis of the alkylated imine mixture gave largely *trans*-2-methyl-4-*t*-butylcyclohexanone. The preference for axial alkylation is so great that it even occurs when a quasiaxial methyl group is present at C-6 on the cyclohexane ring. Thus, further deprotonation and methylation of the mixture of (92)-(94) gave 94% of the 2,6-axial-axial dimethylation product and 6% of the 2,6-axial-equatorial isomer, which again probably arose *via* epimerization of the axial-axial product. Apparently, in these systems,



the stereoelectronic factor favoring axial alkylation is greatly reinforced by a steric effect of the *syn* substituent on nitrogen, which strongly retards equatorial approach of the electrophile. However, as has been pointed out recently, solvation and aggregation effects may play an important role in determining the stereochemistry of metallated imine alkylations.¹⁸⁹ The remarkably high stereoselectivity observed in alkylations of metallated imines greatly enhances their value as metal enolate equivalents.



Scheme 48

As shown in Scheme 49, the cyclohexylimine of 10-methyl-1(9)-octalin-2-one has been methylated at C-3 via the kinetically formed cross-conjugated dienamine anion (95) or monomethylated at C-1 via the thermodynamically more stable extended anion (96).¹⁴⁰ Steroidal and simpler enones have also been monoalkylated at the α -position via their corresponding hetero- or homo-annular extended dienamine anions. Likewise, α -alkylations are the rule for lithiated α , β -unsaturated aldimines.¹⁹⁰ The thermodynamically controlled procedure for the synthesis of α -methyl α , β -unsaturated ketones is a vast improvement over conventional methodology using extended metal dienolates where α , α -dimethylation is a severe

complication (Section 1.1.3). It is successful because the unsaturated lithiated dienimine intermediates undergo proton transfer processes too slowly to allow a significant amount of dimethylation to occur.¹⁴⁰



Scheme 49

Metallated N,N-dialkylhydrazones exhibit similar regioselectivity and stereoselectivity to metallated imines in alkylation reactions, but the former reagents have higher reactivity, and afford higher yields, among other advantages.^{180,191} The diastereoselective and enantioselective synthesis of α -alkylated carbonyl compounds requires mild methods of cleavage of α -alkylated hydrazones. Many of these have been developed by Enders and coworkers¹⁹¹ and they have been recently reviewed by Bergbreiter and Momongan.¹⁹²

Hydrazone deprotonations are normally carried out with lithium dialkylamides or alkyllithium reagents. N,N-Dimethylhydrazones differ from the corresponding imines, which undergo kinetically controlled *anti* deprotonation in that there is little preference for deprotonation *syn* or *anti* to the dimethylamino group on nitrogen in symmetrical systems.¹⁹³

In general, deprotonations of unsymmetrical hydrazones occur with high regioselectivity at the lesssubstituted α -carbon atom. Since metallated hydrazones are highly resistant to proton transfer reactions with neutral ketone hydrazones, regiospecific, *C*-alkylation then occurs at the less-substituted carbon atom. Scheme 50 provides examples of how complex symmetrical¹⁹⁴ or unsymmetrical¹⁹⁵ α, α' -alkylated acetone derivatives may be prepared in one-pot sequences starting with the *N*,*N*-dimethylhydrazone of acetone (97). The conjugate addition of trimethylsilyllithium to α,β -unsaturated hydrazones, followed by trapping of the lithiated hydrazone intermediates has been reported recently.¹⁹⁶ Both hydrazones¹⁸⁰ and imines¹⁹⁷ may be deprotonated at the more-substituted α -carbon if an electron-withdrawing group is located there.

Conformationally rigid lithiated N,N-dimethylhydrazones exhibit axial alkylation preferences exceeding 98%. On the basis of the X-ray crystal structure of lithiated 2-methylcyclohexanone dimethylhydrazone,¹⁹⁸ kinetic studies¹⁹⁹ and other data, Collum, Clardy and coworkers have proposed that the electrophile attacks the anion from the face opposite to an η^4 -coordinated disolvated lithium cation. If the η^4 -complexed lithium cation occupies the 'equatorial' face of the anion, the stereoelectronic factor favoring chair-axial alkylation and the steric factor operate in concert to strongly direct axial alkylation. It was also shown that the lithiated cyclohexanone derivative (**98**) having a methyl group at C-6 yielded mainly *cis*-2,6-dimethylcyclohexanone upon methylation and hydrolysis (Scheme 51).¹⁹⁸ Thus, the stereoselectivity of alkylation of (**98**) was opposite to that originally reported.¹⁸⁰ It appears that, as expected, (**98**) undergoes axial alkylation mainly *via* a conformation with the 6-methyl group quasiaxial to give the *cis*-2,6-disubstituted product.

Cyclic α,β -unsaturated N,N-dimethylhydrazones may be monoalkylated cleanly at the α - or α' -position depending upon the conditions of formation of the metallated unsaturated hydrazone.¹⁹² By using: (i) less reactive bases such as sodium hydride; (ii) additives such as HMPA; and/or (iii) allowing the metallated hydrazone to stand for a period of time before the alkylating agent is added, α -alkylations via the





more thermodynamically stable homoannular or heteroannular linearly conjugated anions are favored.^{140,200} On the other hand, deprotonations with LDA in THF followed immediately by addition of the alkylating agent have provided high yields of α' -monoalkylated products *via* the kinetically formed cross-conjugated anions.^{180,201} α -Alkylations of α,β -unsaturated aldehyde hydrazones containing γ -hydrogen atoms are also known.²⁰²

Unsymmetrical ketoximes or ketoxime methyl ethers undergo rapid deprotonation to give *syn* dianions or monoanions, which react regioselectively with electrophilic reagents.^{203,204} Axial alkylation is observed in conformationally rigid systems. Aldoximes can also be converted to their dianions and alkylated in high yields.²⁰⁵

1.1.5.2 Enantioselective Syntheses via Alkylations

Enamines,¹⁷¹ metallated imines^{171,206,207} and metallated hydrazones^{171,191,206,207} derived from chiral, nonracemic amines have proved to be excellent reagents for the enantioselective synthesis of α -alkylated ketones and aldehydes. Although other optically active systems had been studied,¹⁷¹ the first efficient enantioselective synthesis using a chiral enamine was reported by Whitesell and Felman.²⁰⁸ They found that the cyclohexanone enamine from (+)-*trans*-2,5-dimethylpyrrolidine (99) underwent alkylation and hydrolysis to give 2-substituted cyclohexanones with the (*R*)-configuration in good optical and chemical yields. For the enamine (99) the same face of the cyclohexene ring is shielded from attack irrespective of the conformer which undergoes alkylation.



(99)

Attempted asymmetric alkylations of metallated chiral imines gave poor results unless, as discovered by Meyers *et al.*²⁰⁹ and Whitesell and Whitesell,²¹⁰ a β -ether group was present in the optically active amine component. This allows formation of a five-membered chelate ring involving the metal cation, which imparts rigidity to the system and restricts the number of conformations available for alkylation. The synthesis of (S)-2-ethylcyclohexanone from the imine (100) provides an illustration of the method (Scheme 52). Meyers *et al.* have proposed that alkylation occurs on the top-face of the chelated structure (101).²⁰⁹ However, based upon a proposal by Whitesell and Whitesell²¹⁰ for a related case and experimental data indicating that the lithiated imine exists largely in a *syn* configuration, a better explanation for the 1,4-asymmetric induction is provided by a structure such as (102), which has the chiral center closer to the reaction site and may undergo alkylation from the top-face of the π -system.





(S)-2-Alkyl-2-phenylcycloalkanones have been synthesized in high optical yields using enamines derived from *t*-leucine *t*-butyl ester or valine *t*-butyl ester and 2-phenylcycloalkanones (Scheme 53).²¹¹





Macrocyclic ketones and acyclic ketones are capable of yielding (Z)- or (E)-lithiated imines upon deprotonation. For example, Meyers *et al.*²¹² found that methylation of the kinetically formed (E)-lithiated imine (**103**), derived from cyclododecanone and (S)-phenylalaninol methyl ether, gave (S)-2-methylcyclododecanone in low enantiomeric excess (*ee*), *i.e.* 59%, upon methylation and hydrolysis (Scheme 54). However, refluxing of (**103**) in THF effected its isomerization into the more thermodynamically stable (Z)-isomer (**104**), which upon methylation and hydrolysis gave (R)-2-methylcyclododecanone in 81% *ee* (Scheme 54). Likewise, ethylation and hydrolysis of the acyclic (Z)-lithiated imine (**105**) gave (R)-3methylhexan-4-one in only 3% *ee*, but isomerizations to the more stable (E)-isomer (**106**), followed by ethylation and hydrolysis increased the optical yield of the (R)-ketone to 77% (Scheme 54). (Note: the (Z)/(E) transformation did not lead to a change in configuration of the acyclic ketone product.) Because of problems with (Z)/(E) isomerism and conformational flexibility, metallated aldimines are rather poor reagents for asymmetric syntheses of α -alkylated aldehydes.



Scheme 54

Fortunately, the use of lithiated hydrazones derived from (S)- or (R)-1-amino-2-methoxymethylpyrrolidine (SAMP or RAMP) as nucleophiles for asymmetric alkylations have provided a solution to the problems described above with metallated acyclic ketimines and aldimines.¹⁹¹ Lithiated SAMP or RAMP hydrazones of cyclic ketones are also alkylated in high yields. A major advantage of these chiral hydrazones is that their derivatives of aldehydes, acyclic and cyclic ketones all yield mainly $(E)_{CC-}$, $(Z)_{CN}$ -lithiated species on deprotonation with LDA in ethereal solvents under kinetic control. The $(E)_{CC-}$ configuration obtains as a result of the minimization of steric interactions in the usual closed transition



(107) $R^1 = H$ or alkyl; $R^2 = alkyl$



Scheme 55

state for the deprotonation step.¹⁷¹ These chelated lithiated hydrazones are quite conformationally rigid and they exhibit a high degree of diastereofacial differentiation. The high nucleophlicity of lithiated hydrazones allows alkylations to be performed with a variety of alkylating agents in relatively nonpolar solvents such as diethyl ether at very low temperatures, *e.g.* –110 °C. Such conditions coupled with appropriate hydrazone cleavage methodology^{191,192} can yield α -alkylated aldehydes and acyclic ketones in optical yields usually exceeding 90%. In light of the recent model proposed by Collum, Clardy and coworkers^{198,199} for the stereoselectivity of lithiated hydrazone alkylations, it appears likely that alkylations of (*E*)_{CC}-,(*Z*)_{CN}-SAMP hydrazones may occur *via* a species such as (107), in which the electrophilic reagent attacks from the opposite face of an η^4 -coordinated lithium cation. Considerably lower alkylation enantioselectivities are observed for less rigid (*Z*)_{CC}-,(*E*)_{CN}-SAMP hydrazones, which may be obtained by deprotonation of hydrazones in the presence of HMPA. In this solvent, deprotonation probably proceeds *via* an open transition state.¹⁷¹

Asymmetric synthesis *via* chiral, nonracemic hydrazones have gained wide acceptance in the community of synthetic organic chemists. Some examples of asymmetric alkylations of acyclic,²¹³ saturated²¹⁴ and unsaturated cyclic ketones^{201,215} and aldehydes,²¹⁶ which are applicable in the natural products field, are shown in Scheme 55. Various methods of cleavage of the alkylated hydrazones are also illustrated in Scheme 55.

1.1.6 ALKYLATIONS OF METAL ENOLATES OF CARBOXYLIC ACID DERIVATIVES

1.1.6.1 Introduction

Although carboxylic acids and their derivatives are somewhat weaker carbon acids than aldehydes and ketones, it is generally possible to quantitatively convert them to the corresponding metal enolates with dialkylamide bases, the most popular of which is LDA.^{4,217,218} Thus, monoanions of saturated esters, lactones, nitriles, N,N-dialkylamides and N-alkyllactams and dianions of carboxylic acids and N-unsubstituted amides and lactams are easily prepared in aprotic solvents such as THF and C-alkylated with a variety of simple and functionalized S_N2 -reactive alkylating agents at room temperature or below. When more-hindered systems are involved, the basicity of the metal dialkylamide and the reactivity of the metal enolate can be enhanced by the addition of HMPA. Of course, many of the indirect methods used for the generation of aldehyde and ketone enolates are also applicable to the preparation of enolates of carboxylic acid derivatives (Section 1.1.2.1). *O*-Alkylations or dialkylations at carbon generally are of minimal importance with metal enolates of carboxylic acid derivatives.

In recent years, investigations of the diastereoselectivity and enantioselectivity of alkylations of metal enolates of carboxylic acid derivatives have become one of the most active areas of research in synthetic organic chemistry. Intraannular, extraannular and chelate-enforced intraannular chirality transfer may be involved in determining the stereochemistry of these alkylations.

1.1.6.2 Diasteroselective Alkylations of Exocyclic and Endocyclic Enolates of Carboxylic Acid Derivatives

Intraannular chirality transfer may control alkylations of both exocyclic and endocyclic enolates. Scheme 56 provides examples of 1,2-, 1,3- and 1,4-asymmetric induction in substituted cyclohexylidine enolates.²¹⁹ For compounds such as (**108**) selective *anti* alkylation probably results from axial attack of the alkylating agent on the conformation of the enolate with the methyl substituent quasiaxial. Sterically preferred equatorial attack on a conformation with the ring substituent equatorial seems to provide a reasonable explanation for most of the results of alkylations of compounds (**109**) and (**110**). The proportion of equatorial and axial attack observed in the alkylations of ester (**110a**) is essentially identical to that of the related ketone enolate (**46**). In the case of the acid (**109d**), some sort of chelation phenomenon involving the carboxylate dianion and the methoxy group may cause the ring to prefer a conformation with the methoxy group axial. In such a conformation equatorial alkylation, which would be greatly favored sterically, would give the major product with the carboxylate and methoxy groups *cis* to each other.

Although there are only a few examples of alkylations of carboxylate enolates which are exocyclic to five-membered carbocyclic rings, the usual steric factors seem to control the stereoselectivity of these reactions. Thus, the dienolate (111) underwent reaction with 4-bromo-1-butene *anti* to the methoxymethyl substituents with high diastereoselectivity²²⁰ and enolates derived from norbornane-2-carboxylates (112)



undergo largely *exo* alkylation.²¹⁹ An interesting reversal of the stereoselectivity of alkylations of exocyclic enolates of five-membered ring heterocyclic systems has been observed.^{221,222} For example, the dioxalone-derived enolates $(113)^{222,223}$ and $(114)^{221}$ are alkylated primarily *syn* to the existing substituents, while for the dihydrooxazole- and oxazolidine-derived enolates $(115)^{221}$ and $(116)^{221}$ alkylation occurred *anti* to the substituents. The preference for *syn* attack on enolates of the type (113) and (114) has been attributed to folding of the hetereocyclic ring to avoid electronic repulsion between the enolate π -system and the nonbonding electron pairs on the heteroatoms.^{221,223}



The ester precursors of the enolates (114) and (116) were prepared from reactions of glyceric acid or serine with pivaldehyde. After alkylation, the parent alkylated acids were recovered by hydrolysis. Seebach and coworkers, who have pioneered this type of procedure, have called it 'self-reproduction of chir-



ality.'²²³ As will be described below, self-reproduction of chirality can be accomplished through alkylations of endocyclic as well as exocyclic enolates. It generally entails: (i) production of a ring containing a temporary, auxiliary chiral center by derivatization of an optically active α -hydroxy or α -amino ester; (ii) formation of an enolate by deprotonation at the original asymmetric α -carbon atom; (iii) use of intramolecular chirality transfer to control the stereochemistry of alkylation of the enolate; and (iv) generation of the chiral α -alkylated ester by hydrolysis.

The stereochemistry of alkylations of endocyclic enolates derived from lactones has been reviewed recently² and will be discussed only briefly. β -Substituted β -lactone enolates undergo alkylation *anti* to the substituent with high stereoselectivity, particularly when the β -substituent is a bulky group. Alkylations of β - and γ -substituted γ -lactones are directed almost exclusively to the π -face of the enolate *anti* to the substituent by 1,2- and 1,3-asymmetric induction.²²⁴ As shown in equation (10), the dilithium dianion (117) of (*R*)- β -hydroxy- γ -butyrolactone underwent alkylation *anti* to the alkoxy group, to give the *trans*- α , β -isomer (118) exclusively.²²⁵ γ -Lactone enolates which are *cis*-fused at the β - and γ -positions to fiveor six-membered rings undergo methylation mainly from the convex π -face of the molecule. If the enolate is *trans*-fused at the β - and γ -positions to a six- or seven-membered ring, alkylation occurs *anti* to the γ -ring residue.



Alkylations of β -substituted δ -lactone enolates occur *anti* to the substituent with high diastereoselectivity unless bulky groups are present at the α -position.²²⁴ Then, conformational effects may lead to a reversal of the diastereofacial differentiation. Other δ -lactone enolates are alkylated with poor diastereoselectivity unless they are *cis*-disubstituted at the γ - and δ -positions. Still and Galynker have shown that remote substituents may exert a considerable amount of asymmetric induction in mediumring lactone enolate alkylations.²²⁶ The remote substituent can determine which of the lower energy conformations of the enolate are available for alkylation.²²⁶

Within the past several years, Seebach and coworkers have made extensive use of five-membered ring O,O-, S,O-, O,N- and N,N-acetals of the general structure (119), derived from pivalaldehyde, as protecting groups to preserve both the functionality and the chirality of α -heterosubstituted carboxylic acid derivatives during their conversion to endocyclic enolates and alkylation.^{223,227} Frater *et al.* also made an important original contribution in this field.²²⁸ Alkylation of lithium enolates derived from systems such as (119) occur *anti* to the *t*-butyl group with high diastereoselectivity. Thus, by the choice of the proper diastereomer it is possible to accomplish enantioselective synthesis of α, α -dialkylated α -amino or α -hydroxy acids with overall inversion (from 119a) or retention (from 119b) of configuration. The use of chiral, nonracemic glycine derivatives such as (119; R³ = H) obtained by resolution even allowed the enantioselective synthesis of unbranched α -amino acids.²²⁹ More recently the Seebach research group has extended its studies to the alkylations of endocyclic enolates of six-membered ring O,O-acetals (120) derived from β -hydroxy acids.²³⁰ Highly diastereoselective α -alkylations were accomplished. β -Unsubstituted compounds such as (120a) underwent alkylation predominantly *anti* to the *t*-butyl group, but in

excellent review provides much more thorough coverage of alkylations of heterocyclic systems such as (119) and (120) for 'self-reproduction of chirality.'²²³



By carrying out sequential alkylations of chiral, nonracemic bicyclic lactams, Meyers and coworkers have developed highly efficient syntheses of chiral compounds containing quarternary carbon atoms with high enantiomeric purity.^{231,232} The sequence of steps is illustrated in Scheme 57 using the bicyclic lactam (121). The diasteroselectivity of the first alkylation step was low in this case, but the second step yielded the dialkylated lactam (123), which resulted from *endo* entry of the second alkyl group, with high diasteroselectivity.²³² In a related system where the angular substituent was a phenyl rather than a methyl group, the *endo:exo* ratios were 9:1 and 42:1 for methylation in the first step and benzylation in the second step.²³¹ Other explanations for the preference for *endo* alkylation of enolates such as that derived from (122) have been offered, but most recently Meyers and Wallace proposed that the lone pair electrons on the convex β -face of the molecule perturb the enolate π -system so as to favor *endo* attack by the electrophile.²³³



By Red-Al reduction, acid hydrolysis (to a keto aldehyde) and base-catalyzed intramolecular aldol cyclization, compounds such as (123) have been converted into 4,4-dialkylcyclopentenones in good chemical yields and 99% enantiomeric purity.²³² The Meyers research group has made extensive use of this exciting methodology for the synthesis of a variety of natural products.²³⁴

1.1.6.3 Diastereoselective Alkylations of Acyclic Enolates of Carboxylic Acid Derivatives

Extraannular and chelate-enforced intraannular chirality transfer in enolate alkylations of carboxylic acid derivatives may occur in cases where: (i) the chiral center, *e.g.* a β -carbon atom, is present and remains in the carboxylic portion of the molecule; or (ii) the substrate contains a chiral auxiliary, *e.g.* the alcohol portion of an ester or the amine portion of an amide, which is removed after the alkylation to generate a chiral carboxylic acid. Alkylations of the former type will be discussed in this section. Alkylations of systems containing chiral auxiliaries are described in Section 1.1.6.4.

Fleming and coworkers⁹⁴ and McGarvey and Williams⁹⁷ have shown that acyclic ester enolates, e.g. (124), containing relatively strong electron donor groups such as dimethylphenylsilyl and tri-*n*-butylstan-

nyl at the β -carbon atom, undergo α -alkylation *anti* to such groups with high diastereoselectivity (Scheme 58). These groups presumably do not participate in chelation with the metal cation and enolate oxygen atom. Therefore, a transition state such as (125), in which the β -substituent is aligned perpendicular to the plane of the enolate, seems to adequately account for the observed results.⁹⁷ The ester enolate alkylations reported by Koga and coworkers²³⁵ and Yamamoto and Maruyana²³⁶ are consistent with this explanation.



The results of alkylations of ester enolates such as (126) with β -methyl and β -CH₂OR substituents (which have similar steric requirements) provide additional evidence that alkylation occurs *anti* to the more strongly electron-releasing β -substituent (Scheme 59).⁹⁷ Presumably in (126) a lone pair of electrons on oxygen facilitates electron release by the β -carbon– γ -carbon bond. Thus, an alkylation transition state such as (127) may obtain in such cases. Alkylations of dilithium dianions of 3-benzoylaminobutanoic acid esters have also been shown to give mainly the *anti* 2,3-disubstituted diastereomers.²³⁷





 β -Hydroxy esters may be alkylated via their dilithium dianions.²³⁸ Deprotonation of the hydroxy group by the base prevents its elimination to form an α , β -unsaturated system. Two recent publications by Frater *et al.*²³⁹ and by Seebach *et al.*²⁴⁰ provide summaries of the results of the alkylations of nonracemic

Alkylation of Carbon

 β -hydroxy esters such as (S)-ethyl 3-hydroxybutanoates (128) and (S)-diethyl malate. A rational explanation for the data shown in Scheme 60 for the alkylation of (128) is that a chelated (Z)-enolate, *i.e.* (129), is formed kinetically and then C-alkylation occurs *anti* to the larger group at the β -position. The direct formation of the (Z)-enolate contrasts with the situation for simple esters whose kinetic deprotonations with LDA in THF give largely (E)-isomers.² Presumably, in β -hydroxy systems a chelated β -alkoxy ester is formed initially and its deprotonation at the α -carbon atom leads to the (Z)-enolate. The involvement of chelated species similar to (129), which undergo alkylation *anti* to the alkoxycarbonyl group, also explains the diastereoselectivity observed in malate ester alkylations.²⁴⁰ Such diastereoselective alkylations have found extensive use in the synthesis of optically active natural products. Snieckus and coworkers have shown that alkylations of succinamide dienolates yield largely *threo* dialkylation products.²⁴¹ It is not clear whether extraannular chirality transfer or chelate-enforced chirality transfer account for these results.²



Scheme 60

 α -Substituted N-(9-phenylfluoren-9-yl)amino ketones, e.g. (130), have been found to undergo deprotonation and alkylation at the α' -position to give syn α, α' -disubstituted products (132) with modest diastereoselectivity (Scheme 61). It has been proposed by Lubell and Rapoport that the major alkylation product arises via electrophilic attack on the less-hindered face of a chelated (Z)-enolate such as (131).²⁴²



1.1.6.4 Diastereoselective Alkylations of Acyclic Carboxylic Acid Derivatives Containing Chiral Auxiliary Groups

Alkylations of acyclic enolates containing a collection of chiral auxiliary groups have been used successfully for the asymmetric synthesis of carboxylic acids. The chiral, nonracemic substrates that have been used include amides, imides, esters, imine derivatives of glycinates and acyl derivatives of chiral transition metals. In these systems either extraannular or chelate-enforced intraannular chirality transfer may control the sense of the alkylation step.

In 1978, Larcheveque and coworkers reported modest yields and diastereoselectivities in alkylations of enolates of (-)-ephedrine amides.²⁴³ However, two years later, Evans and Takacs²⁴⁴ and Sonnet and Heath²⁴⁵ reported simultaneously that amides derived from (S)-prolinol were much more suitable substrates for such reactions. Deprotonations of these amides with LDA in the THF gave (Z)-enolates (due to allylic strain that would be associated with (E)-enolate formation) and the stereochemical outcome of the alkylation step was rationalized by assuming that the reagent approached preferentially from the less-hindered *si*-face of a chelated species such as (133; Scheme 62).²⁴⁶ When the hydroxy group of the starting prolinol amide was protected by conversion into various ether derivatives, alkylations of the corresponding lithium enolates were *re*-face selective. Apparently, in these cases steric factors rather than chelation effects controlled the stereoselectivity of the alkylation.^{2,246} It is of interest to note that enolates such as (133) are attached primarily from the *si*-face by terminal epoxides.²⁴⁷



Scheme 62

More recently Katsuki and coworkers have reported that (Z)-enolates of α -alkyl and α -heterosubstituted amides such as (134), derived from pyrrolidine derivatives having a C_2 axis of symmetry, undergo very diastereoselective alkylations with secondary alkyl and other alkylating agents in good to excellent chemical yields (Scheme 63).²⁴⁸ As with prolinol ether amide enolates, it appears that the direction of approach of the alkylating agent to the enolate (134) is controlled mainly by steric factors within the chiral auxiliary, *i.e.* chelation effects seem to be of little importance.

In order to overcome the problems associated with acid hydrolysis of amides of prolinol, the Evans research group has investigated the diastereoselectivity of the alkylation of imides derived from chiral 2oxazolidones.^{2,246} Imide enolates are somewhat less nucleophilic than amide enolates, but they have the advantage that their diastereomeric alkylation products are easily separated and the imide linkage is cleaved with a variety of reagents under mild conditions. As shown in Scheme 64, alkylation of the chelated (Z)-enolate of the propionimide derived from (S)-valinol (135) with benzyl bromide occurred in high chemical yield and with high *si*-face diastereoselectivity.²⁴⁹ In addition to oxazolidones, imidazolidiones have proved to be useful chiral auxiliaries for diastereoselective enolate alkylations.²⁵⁰

While deprotonations of N,N-dialkylamides with lithium dialkylamides in THF yield (Z)-enolates both in the presence and absence of HMPA, simple alkanoate esters give predominantly (Z)-enolates in the presence of HMPA and predominantly (E)-enolates in the absence of the dipolar, aprotic additive.² Presumably, deprotonations occur via an open transition when HMPA is present to complex with the lithium cation and via a closed transition state in THF alone. Helmchen and coworkers have reported that alkylations of esters such as (136)–(138), containing chiral auxiliaries derived from (+)-camphor, afford the opportunity for an exceptional amount of configurational control.²⁵¹ Alkylations of enolates derived from (136)–(138) occur from the face of the enolate anion opposite the benzenesulfonamide group. Since compound (136) is heterochiral with respect to compounds (137) and (138), as far as the ester and sulfonamide portions of the molecules are concerned, their enolates give alkylation products with opposite configurations under the same reaction conditions. Scheme 65 provides an example of how the configuration of the alkylation product was reversed when the configuration of the enolate was changed by the conditions of deprotonation, *i.e.* the (E)-enolate gave largely the (R)-alkylation product and the (Z)-enol-



ate gave mainly the (S)-alkylated system. Esters of the type (136) and (137) derived from benzyl-protected glycolates were found to yield (E)-enolates both in the absence and presence of HMPA. The (E)enolate of a chiral propionate ester derived from 10-sulfonamidocamphor was diastereoselectivity alkylated from the si-face.²⁵²



Chiral auxiliaries may be applied to α -amino acid esters by forming imine derivatives. Enolates from 2-hydroxy-3-pinanone glycinate esters have been alkylated to produce mono-²⁵³ and di-substituted²⁵⁴ α -amino acids in good optical yields after hydrolysis. Recently, McIntosh *et al.*²⁵⁵ reported the results of al-kylations of the enolate (139) derived from the (+)-camphor imine of *t*-butyl glycinate with a variety of



alkylating agents. Selected examples of these reactions are shown in Scheme 66. An electrostatic interaction of the $\pi-\pi$ or Li- π type between the lithium enolate and the alkylating agent was suggested as a possible explanation for the high preference for *re*-face selectivity, which was observed when alkylating agents containing electron rich π -systems, *e.g.* benzyl bromide, were employed.²⁵⁵ Enantioselective alkylations of imines of glycinate esters have been carried out in low optical yields using allyl acetate and palladium(0) catalysts containing chiral phosphine ligands.²⁵⁶ Diastereoselective alkylations of asymmetrically substituted transition metal acyl enolates is an active area of investigation,²⁵⁷ but space does not permit coverage of this subject.



Scheme 66

1.1.6.5 Cycloalkylations of Enolates of Carboxylic Acid Derivatives

Like aldehyde and ketone enolates, enolates of carboxylic acid derivatives containing appropriately located leaving groups undergo cycloalkylations to yield monocyclic or more complex ring systems. Intramolecular alkylations of anions of simple nitriles and protected cyanohydrins have been thoroughly reviewed.^{218,258} Stork *et al.* have uncovered some interesting aspects of the regiochemistry of intramolecular alkylations of nitriles containing *cis*-epoxides.²⁵⁹ They showed that these reactions are subject to two important constraints, *i.e.* (i) the usual preference for nucleophilic attack to occur at the less-substituted carbon atom of the epoxide ring; and (ii) the need for a colinear arrangement of the nucleophile and the leaving group in the transition state for cyclization. In cases where both carbons of the epoxide ring were equally substituted, it was observed that the colinearity requirement always favored formation of the smaller ring. The transformation shown in Scheme 67 was a key step in the synthesis of (\pm)-grandisol, a boll weevil sex attractant.²⁵⁹ As illustrated in structure (**140**) the preferred transition state for the reaction produces stereoselectively the cyclobutane derivative with the 1,2-alkyl substituents *cis* to each other; the lithium α -nitrile anion appears to have a larger effective size than the α -substituent. Cyclopentane derivatives, which would result from attack at the more remote carbon atom of the epoxide ring were not isolated.



The research group of Stork has also shown that decalin or hydrindane ring systems containing angular cyano groups can be prepared by base-promoted intramolecular cyclizations of cyanocyclohexane derivatives containing ketalized 3-keto-4-halo (or 4-tosyloxy) groups or 2-keto-3-halo groups at C-2.²⁶⁰ Cis-fused hydrindanes were formed irrespective of whether the cation present was potassium or lithium. However, as shown in Scheme 68, the nature of the leaving group, the solvent and, especially, the metal cation profoundly affected the *cis:trans* ratio of the decalin products. It was proposed that conditions favoring the *trans* product involved cyclization through a transition state of the anion with the side chain equatorial to the ring. On the other hand, the *cis* product was expected to be obtained in situations where



Scheme 68

the preferred transition state required the side chain to be axial to the ring. Chelation involving the α -nitrile anion, the cation and an oxygen atom of the ketal ring on the side chain may play a role in determining the preferred conformation of the anion for cyclization.

As shown in Scheme 69, Kim and coworkers²⁶¹ have reported highly stereoselective routes to *cis*-1,2dialkylcycloalkanecarboxylates by intramolecular alkylations of tosyloxy esters such as (142). The reaction appears to proceed *via* the 'eclipsed' conformations of the enolate (143), rather than the 'bisected' one that would result from a 180° rotation about the α,β -sigma bond.



Intramolecular alkylations of nitrile-stabilized carbanions have been used to synthesize large rings such as those with 10^{262} and 14 members.²⁶³ Tsuji and coworkers carried out a synthesis of the macrocyclic antibiotic zearalenone by this route.²⁶³ As shown in Scheme 70, conversion of either of the protected cyanohydrins (144) or (145) to the corresponding dianions, resulting from deprotonation at the benzylic positions and α to the nitrile groups, gave the same cyclization product (146) in excellent yields. Dianion formation: (i) provided control of the conformation of the side chain; (ii) protected the ester from nucleophilic attack; and (iii) appeared to increase the rate of the intramolecular cyclization.



There are many examples of intramolecular cyclizations of carboxylate enolates which lead to fusedand bridged-ring systems. The reaction shown in Scheme 71, which was used by Danishefsky *et al.* as a part of their synthesis of (\pm) -quadrone, is illustrative.²⁶⁴



Lewis acid catalyzed reactions of silyl enol derivatives of esters (ketene acetals) with S_N 1-reactive alkylation agents are well known, but space limitations have prevented coverage of this subject here. The reaction shown in equation (11), which was employed by Pattenden and coworkers²⁶⁵ in their synthetic studies on forskolin, provides an example of an intramolecular Mukaiyama reaction of a silyl enol ester derivative.



1.1.6.6 Alkylations of Dienolates of α,β-Unsaturated Carboxylic Acid Derivatives

Except in special cases where there is severe steric hindrance,²⁶⁶ lithium dienolates of all classes of carboxylic acid derivatives undergo almost exclusive α -alkylation with various types of alkylating agents, *e.g.* allyl, benzyl, saturated, *etc.* However, Katzenellenbogen and coworkers,²⁶⁷ and Snieckus and coworkers,²⁶⁸ have shown that γ -alkylation may become the predominant mode of reaction when copper(I) dienolates are involved. This was found to be particularly true for dianionic species derived from α , β -unsaturated carboxylic acids and *N*-alkylamides where the negative charge density on the γ -site would be expected to be higher than for the corresponding monoanionic systems, *e.g.* esters or *N*,*N*-dial-kylamides.

While γ -alkylations did not occur with saturated and benzylic halides, γ -selectivities in the 62–99% range were observed in reactions of dicopper(I) dianions of a variety of α , β -unsaturated acids with allylic halides.²⁶⁷ γ -Unsubstituted allylic halides reacted by an S_N2' mechanism, γ -disubstituted compounds underwent direct S_N2 displacement and γ -monosubstituted systems reacted by both S_N2' and S_N2 pathways. Scheme 72 provides an example of the dramatic reversal in regioselectivity that was observed in the allylation of the dianion (147) of tiglic acid when the metal cation was changed from lithium to copper(I).²⁶⁷ The γ -alkylation product from the latter species was exclusively the (E)-isomer.



Scheme 73 shows the results of alkylations of dilithium and dicopper(I) dianions (148) of senecioic acid *N*-isopropylamide with alkylting agents of various classes.²⁶⁸ A major difference in the behavior of (148; $M = Cu^{I}$) and the corresponding dicopper(I) dianion of senecioic acid itself was that a significant

amount of γ -alkylation occurred with methyl iodide and benzyl bromide. Also, the amide dianion was γ -alkylated with higher (Z)/(E)-stereoselectivity than the acid dianion. The role played by copper(I) in affecting the regiochemistry of alkylations of species such as (147) and (148) is unclear. Possibly C-metallated species are more likely to be involved when copper(I) rather than lithium is employed as the cation.



Scheme 73

Kende and Toder have reported that deconjugative alkylations of (Z)-2-alkenoates give the corresponding (E)-3-enoate derivatives, while (E)-2-alkenoates give (Z)-3-enoate isomers unless the 4-substituent is larger than methyl.²⁶⁹ They suggested that the stereoselectivity of these reactions is probably determined by the conformation of the α , β -unsaturated ester that undergoes deprotonation at C-4.

Birch reductions of arylbenzoic acid derivatives provide a highly useful method of generating enolates of unsaturated carboxylic acid derivatives that can be trapped with a variety of electrophilic reagents. Especially interesting data have been obtained by Schultz *et al.* utilizing arylbenzoic acid derivatives containing chiral auxiliaries derived from L-prolinol.²⁷⁰ The results of alkylations of the lithium amide enolates (150) obtained by reduction of benzoxazepinones (149) are shown in Scheme 74. Enolate (150a) was found to undergo highly diastereoselective β -face attack at the α -position with a variety of alkylating agents. However, γ -alkylation was the major pathway for reaction of enolate (150b). It was proposed that when a methyl group is present at the angular position, the enolate adopts a conformation in which the α -position is shielded from attack on both faces.

Schultz et $al^{.270}$ have also conducted studies on the methylation of the chiral lithium amide enolate (152), which was obtained by lithium-ammonia reduction of the 2-methoxybenzamide (151). This compound is an acyclic counterpart of the benzoxazepinone (149a). Enolate (152) was shown to have the (Z)-configuration by ¹H NMR analysis. As shown in Scheme 75, methylation of (152) in the presence of ammonia gave the amide (153), which resulted from attack of methyl iodide at the α -position and from the underside of the enolate almost exclusively. On the other hand, if the ammonia was removed or if the enolate (152) was protonated and reformed with *n*-butyllithium prior to alkylation, compound (154) resulting from topside attack at the α -position was formed almost exclusively. This dramatic change in diastereoselectivity in the absence of ammonia was attributed to a major change in enolate structure that could result from rotation about the C—N bond and inversion of configuration at nitrogen of the chiral auxiliary. The enolate was considered to exist as a dimer in the presence and absence of ammonia.

Takahashi and coworkers have employed intramolecular reactions of anions of unsaturated protected cyanohydrins to produce unsaturated carbocyclic systems containing 10-, 14- and 16-membered rings which are convertible into natural products.²⁷¹ These reactions occur regiospecifically at the α -position and usually do not involve $(E) \rightarrow (Z)$ isomerization of β , γ -double bonds. The preparation of 2-cyclopen-tadecenone (155), which is convertible into the macrocyclic perfumes muscone and exaltone, is illustrative of this methodology (Scheme 76).



```
Scheme 74
```







Conditions	Ratio	Yield (%)
NH₃–THF, –78 °C	99.6:0.4	85
i, warm to 25 °C with loss of NH ₃		
ii, THF, –78 °C	1:99	80





1.1.6.7 Asymmetric Synthesis via Alkylations of Anions of Masked Carboxylic Acid Derivatives

Over the past 20 years, Meyers and coworkers have developed an elegant methodology for the synthesis of complex carbonyl compounds, which involves alkylations of anions of masked carboxylic acid derivatives. Their earlier work, which involved the use of heterocyclic analogs of imino esters such as 2-alkyldihydro-1,3-oxazines, 2-alkylthiazolines and 2-alkyloxazolines has been thoroughly reviewed.^{5,6,272} Anions of 2-alkyloxazolines turned out to be the most useful reagents for the synthesis of alkylated carboxylic acids and esters, and, by using chiral, nonracemic oxazolines such as (156), derived from commercially available amino alcohols, the research group of Meyers extended this methodology to the asymmetric synthesis of α,α -dialkylcarboxylic acids.²⁷³ It was found that both enantiomers of a carboxylic acid derivative could be produced by starting with selectively substituted oxazolines or reversing the order in which the alkyl groups were introduced.

As shown in Scheme 77, deprotonation of compound (156) with LDA in THF at -78 °C gave a 95:5 (E)/(Z)-mixture of lithiated anions (157) and (158). Upon addition of an alkylating agent at -78 °C or below, the major anion (157) was attacked from the underside with high diastereoselectivity. The highly effective intramolecular chirality transfer was ascribed to the fact that (157) exists as a rigid chelated structure with the lithium cation located on the underside of the molecule and the phenyl group sterically shielding the topside of the molecule. The presence of both the methoxy group and the phenyl were essential for high diastereoselectivity to be observed. Alkylations of (157) with ethyl, propyl or *n*-butyl iodide followed by unmasking of the carboxylic acid by hydrolysis gave the corresponding (S)-2,2-dial-kylacetic acids in 72–78% enantiomeric excess (*ee*) and good chemical yields. The optical purity of the product was reduced because the minor azaenolate (158) gave the corresponding (R)-acids.





Schöllkopf and coworkers have pioneered the development of anions of another type of masked carboxylic acid derivative, *i.e.* bislactim ethers such as (159), derived from (S)-valine and glycine or alanine, for the asymmetric synthesis of amino acids.²⁷⁴ As shown in Scheme 78, compounds such as

Alkylation of Carbon

(159) are deprotonated exclusively at C-5 with *n*-butyllithium or LDA in THF to give endocyclic chiral enolate equivalents such as (160). Alkylations of (160) with saturated primary, allylic and benzylic halides occur *anti* to the 2-isopropyl substituent with high diastereoselectivity. It has been proposed that both coordination of the lithium cation and attack of the alkylating agent occurs on the π -face of the delocalized anion opposite the bulky substituent. When the C-2 substituent was smaller than isopropyl or when systems with C-2 disubstituted were employed, alkylation diastereoselectivities were reduced somewhat. A disadvantage of this method is that after acidic methanolysis of the new bislactim ethers, two amino acid esters which must be separated are produced.



1.1.7 ALKYLATIONS OF STABILIZED METAL ENOLATES

1.1.7.1 Alkylations of Monoanions of β-Dicarbonyl and Related Compounds

Conventional methods of alkylation of stabilized metal enolates such as those derived from β -diesters, β -cyano esters, β -keto esters, β -keto aldehydes, β -diketones, β -keto sulfoxides and β -keto sulfones have been thoroughly discussed elsewhere.^{4,5,6,15} *O*-Alkylation often competes strongly with *C*-alkylation in these systems. Factors which tend to favor *O*-alkylation generally include the use of: (i) substrates that are highly enolic, *e.g.* β -keto aldehydes and cyclic β -diketones; (ii) electropositive metals, *e.g.* potassium, as counterions; (iii) highly dipolar aprotic solvents, *e.g.* HMPA; and (iv) highly reactive alkylating agents, *e.g.* chloromethyl alkyl ethers or secondary alkyl halides. Factors that favor *C*-alkylation include the use of: (i) hetereogeneous conditions; (ii) less electropositive metal cations, *e.g.* lithium; (iii) nonpolar or protic solvents; and (iv) polarizable alkylating agents such as alkyl iodides and allylic, propargylic and benzylic halides. In addition to *O*-alkylation, side reactions such as dialkylation, Claisen condensation, β -keto cleavage and oxidative coupling may compete with *C*-alkylation.

Two methods which reduce or avoid *O*-alkylation and employ heterogeneous reaction conditions include: (i) the heating of crystalline thallium(I) enolates of β -keto esters or β -diketones with primary or secondary alkyl iodides;²⁷⁵ and (ii) the heating of stabilized sodium enolates with methyl- or ethyl-sulfonium salts.²⁷⁶ However, the reproducibility and generality of the former method has been questioned.²⁷⁷

Excellent yields of C-alkylation products have been obtained by the reaction of β -dicarbonyl compounds with alkyl halides in the presence of tetraalkylammonium fluorides²⁷⁸ or salts such as (161).²⁷⁹ The latter method appears to have broad generality. Scheme 79 provides a comparison of the alkylation of acetylacetone with *n*-hexyl iodide in the presence of reagent (161) and *via* conventional methodology. Reagent (161) was assumed to generate quaternary ammonium enolate intermediates from β -dicarbonyl compounds. However, it seems possible that the alkylating agent actually attacks an enol intermediate which is hydrogen-bonded to the pyrrolidone anion. This type of pathway was proposed in cases where quaternary ammonium fluorides were utilized.



Enolates of cyclic 1,3-diketones, such as 2-methyl-1,3-cyclopentanedione and 2-methyl-1,3-cyclohexanedione, are especially prone to O-alkylation because of their 'W' geometry and steric hindrance at carbon. However, C-alkylations of these species can be conducted reasonably successfully in protic solvents, e.g. water.²⁸⁰ Trost and Curran²⁸¹ have reported high yields for the C-alkylation of cyclic 1,3-diketones with allylic acetates in the presence of palladium(0) catalysts and DBU (Scheme 80).



The stereochemical results of methylations of stabilized metal enolates such as (162), derived from 4-tbutylcyclohexanone, and (163) and (164), derived from trans-10-methyl-2-decalone, are shown in Scheme 81. Kuehne and Nelson²⁸² found that the unhindered α -cyano and α -methoxycarbonyl enolates (162a) and (162b) both underwent stereoelectronically controlled axial alkylation preferentially, although not with high diastereoselectivity. Also, the related pair of 3-substituted 2-decalone enolates (163a) and (163b) both underwent mainly equatorial alkylation, presumably because the 10-methyl group provides significant hindrance to axial attack of the electrophile.²⁸² These stereochemical results are entirely consistent with those obtained for the alkylations of related unstabilized enolates discussed in Section 1.1.2.2. However, the 1-substituted 2-decalone enolates (164a) and (164b) were found to undergo alkylation in the opposite sense, *i.e.* the α -cyano species underwent primarily equatorial methylation,²⁸² while the related α -methoxycarbonyl species underwent largely axial methylation²⁸³ (similar results were obtained for alkylations of related pairs of enolates derived from tri- or tetra-cyclic B-keto esters²). Kuehne and Nelson have suggested several factors which may account for the difference in the behavior of (164a) and (164b).²⁸² A major reason may be that in (164b) a peri interaction between the methoxy group of the chelated enolate ring and the B-ring methylene group (C-8), which is not present in (164a), distorts the A-ring so as to open up the top π -face of the enolate to attack.

The formation of cyclic compounds by intramolecular alkylations of stabilized enolates has been widely used in organic synthesis. A recent study of the kinetics of the reaction of diethyl (ω -bromoalkyl)malonates in DMSO using tetramethylammonium hydroxide as the base has shown that relative rates of closure of rings of varying size follow the order: $3 > 5 > 6 > 4 > 7 > 12-21 > 8 > 9 > 11 > 10.^{284}$ These results are consistent with earlier studies in which other base–solvent combinations were employed.⁴ The high rates of closure of three-membered rings allow the formation of a variety of cyclopropane deriva-



tives from active methylene compounds under mild conditions. Quinkert *et al.*²⁸⁵ have shown that basepromoted reactions of chiral malonic esters, prepared from chiral, nonracemic alcohols such as (–)-8phenylmenthol, with 1,4-dihalo-2-butenes lead to intermolecular alkylation followed by S_N2' closure of a three-membered ring to give chiral 2-alkenyl-1,1-cyclopropanedicarboxylates. These products have been shown to be useful building blocks for the synthesis of a variety of natural products.

Ten-membered rings are formed slowly in cycloalkylations because the entropy factor is unfavorable and transannular interactions are severe in the transition state for ring closure. Deslongchamps *et al.*²⁸⁶ have shown that the introduction of double and triple bonds into the carbon chain reduces the importance of these factors by restricting the degrees of freedom of the molecule and partially eliminating transannular interactions. As shown in equation (12), the cyclization of the malonic ester derivative (165) was carried out in good yield without the use of high dilution techniques.

The research group of Trost has shown that stabilized enolates derived from β -diesters and α -phenylsulfonyl esters are excellent nucleophiles for palladium(0)-catalyzed reactions with allylic acetates.²⁸⁷ Since these reactions involve π -allylpalladium intermediates, they proceed with net retention of configuration and, therefore, differ from traditional displacement reactions where inversion is the rule. Moreover, the transition metal imposes a template effect on intramolecular versions of these reactions and



Scheme 82

medium and large rings which are difficult to obtain by traditional methods can be formed easily. The reader is referred to recent reviews for thorough coverage of this important methodology.²⁸⁷

1.1.7.2 γ-Alkylations of Dianions of β-Dicarbonyl Compounds

The synthetic flexibility of β -dicarbonyl compounds was greatly expanded when it was discovered by Hauser and Harris that many of these systems can be converted to their dianions by treatment with 2 equiv. of potassium or sodium amide in liquid ammonia.²⁸⁸ These dianions could then be monoalkylated at the γ -position with high regioselectivity. Early research in this area was devoted to studies on β -diketones and β -keto aldehydes, and this is covered in an excellent review by Harris and Harris.²⁸⁹ It was found that unless activating groups were present, the second proton was removed regioselectively from the less-substituted γ -position of unsymmetrical β -diketones. Of course, β -keto aldehydes are capable of yielding only a single dianion. Scheme 82 provides examples of these reactions. The last reaction in Scheme 82 is particularly interesting since it involves preparation of an angularly alkylated 1-decalone derivative.

Attempted γ -alkylations of β -keto esters by forming their dipotassium dianions with potassium amide and alkylating them in liquid ammonia gave poor results. However, Huckin and Weiler²⁹⁰ found that treatment of methyl acetoacetate with 1 equiv. of sodium hydride, followed by 1 equiv. of n-butyllithium, or 2 equiv. of LDA in THF gave dimetal dienolates (166) which alkylated in good yields with several different alkylating agents including isopropyl iodide at 0-25 °C (Scheme 83).



Scheme 83

The method of Huckin and Weiler has enjoyed widespread use in synthesis. For example, recently reported syntheses of trans-bicyclo[4.3.1]decan-10-one²⁹¹ and a bicyclo[4.4.1]undecan-7-one derivative²⁹² have employed γ -alkylations of β -keto ester dianions as key steps. This methodology has also been extended to the formation and γ -alkylation of dianions of γ , δ -unsaturated β -keto esters.²⁹³

1.1.8 REFERENCES

- 1. D. Caine, in 'Carbon-Carbon Bond Formation', ed. R. L. Augustine, Dekker, New York, 1979, vol. 1, chap. 2, p. 85 and refs. therein. 2. D. A. Evans, in 'Asymmetric Synthesis', ed. J. D. Morrison, Academic Press, New York, 1984, vol. 3, chap.
- 1, p. 1 and refs. therein.
- 3. J. d'Angelo, Tetrahedron, 1976, 32, 2979 and refs. therein.
- 4. H. O. House, in 'Modern Synthetic Reactions', 2nd edn., Benjamin, Menlo Park, CA, 1972, chap. 9, p. 492.
- 5. J. C. Stowell, in 'Carbanions in Organic Synthesis', Wiley, New York, 1979, chap. 6, p. 191.
- 6. W. Carruthers, in 'Some Modern Methods of Organic Synthesis', 3rd edn., Cambridge University Press, New York, 1986, chap. 1, p. 1, 70.
- H. O. House and V. Kramar, J. Org. Chem., 1963, 28, 3362.
 G. Stork, P. Rosen and N. L. Goldman, J. Am. Chem. Soc., 1961, 83, 2965; G. Stork, P. Rosen, N. L. Goldman, R. V. Coombs and J. Tsuji, J. Am. Chem. Soc., 1965, 87, 275.
- 9. E. J. Corey and A. W. Gross, Tetrahedron Lett., 1984, 25, 495.
- 10. D. Caine, Org. React. (N.Y.), 1976, 23, 1.
- 11. R. J. K. Taylor, Synthesis, 1985, 364.
- 12. P. Brownbridge, Synthesis, 1983, 1, 85.
- 13. H. B. Mekelburger, and C. S. Wilcox, in 'Comprehensive Organic Synthesis', ed. B. M. Trost, Pergamon Press, Oxford, 1991, vol. 2, chap. 1.4.
- 14. M. Pereyre, J.-P. Quintard and A. Rohm, in 'Tin in Organic Synthesis', Butterworths, Boston, 1987, chap. 12, p. 286.
- 15. L. M. Jackman and B. C. Lange, Tetrahedron, 1977, 33, 2737.
- 16. H. O. House, A. V. Prabhu and W. V. Phillips, J. Org. Chem., 1976, 41, 1209.
- 17. C. L. Liotta and T. C. Caruso, Tetrahedron Lett., 1985, 26, 1599.
- 18. H. O. House, M. Gall and H. D. Olmstead, J. Org. Chem., 1971, 36, 2361.

- 19. R. G. Pearson, J. Chem. Educ., 1968, 45, 581, 643.
- 20. G. Klopman, J. Am. Chem. Soc., 1968, 90, 223.
- 21. K. N. Houk and M. N. Paddon-Row, J. Am. Chem. Soc., 1986, 108, 2659.
- 22. M. E. Jones, S. R. Kass, J. Filley, R. M. Barkley and G. B. Ellison, J. Am. Chem. Soc., 1985, 107, 109.
- 23. L. M. Jackman and T. J. Dunne, J. Am. Chem. Soc., 1985, 107, 2805.
- J.-M. Conia, Rec. Chem. Prog., 1963, 24, 42.
 P. A. Tardella, Tetrahedron Lett., 1969, 1117.
- 26. Y. Morita, M. Suzuki and R. Noyori, J. Org. Chem., 1989, 54, 1785.
- 27. G. H. Posner and C. M. Lentz, J. Am. Chem. Soc., 1979, 101, 934.
- 28. I. Kuwajima, E. Nakamura and M. Shimizu, J. Am. Chem. Soc., 1982, 104, 1025; I. Kuwajima and E. Nakamura, Acc. Chem. Res., 1985, 18, 181.
- 29. R. Noyori, I. Nishida and J. Sakata, J. Am. Chem. Soc., 1983, 105, 1598.
- 30. M. J. Kurth and M. J. O'Brien, J. Org. Chem., 1985, 50, 3846.
- 31. M. W. Rathke and A. Lindert, Synth. Commun., 1978, 8, 9.
- M. M. Mulliko and A. Elinderi, Synth. Commun., 1970, 0, 9.
 E. Negishi, M. J. Idacavage, F. DiPasquale and A. Silveira, Jr., Tetrahedron Lett., 1979, 845.
 Y. Odic and M. Pereyre, C. R. Hebd. Seances Acad. Sci., Ser. C, 1969, 469.
 H. O. House and B. M. Trost, J. Org. Chem., 1965, 30, 1341.
- 35. G. Stork and P. F. Hudrlik, J. Am. Chem. Soc., 1968, 90, 4462.
- 36. E. S. Binkley and C. H. Heathcock, J. Org. Chem., 1975, 40, 2156.
- 37. G. Stork and J. Singh, J. Am. Chem. Soc., 1974, 96, 6181.
- 38. E. Negishi and S. Chatterjee, Tetrahedron Lett., 1983, 24, 1341.
- 39. M. E. Krafft and R. A. Holton, Tetrahedron Lett., 1983, 24, 1345.
- 40. M. E. Krafft and R. A. Holton, J. Org. Chem., 1984, 49, 3669.
- 41. G. H. Posner, J. J. Sterling, C. E. Whitten, C. M. Lentz and D. J. Brunelle, J. Am. Chem. Soc., 1975, 97, 107.
- 42. I. J. Borowitz, E. W. R. Casper, R. K. Crouch and K. C. Yee, J. Org. Chem., 1972, 37, 3873.
- 43. H. O. House and J. M. Wilkins, J. Org. Chem., 1976, 41, 4031.
- 44. R. M. Coates and L. O. Sandefur, J. Org. Chem., 1974, 39, 275.
- 45. R. K. Boeckman, Jr., J. Org. Chem., 1973, 38, 4450.
- 46. J. W. Patterson, Jr. and J. H. Fried, J. Org. Chem., 1974, 39, 2506.
- 47. M. Suzuki, A. Yanagisawa and R. Noyori, J. Am. Chem. Soc., 1988, 110, 4718.
- 48. C. R. Johnson and T. D. Penning, J. Am. Chem. Soc., 1988, 110, 4726.
- 49. M. E. Jung, Tetrahedron, 1976, 32, 3.
- 50. G. Stork and M. E. Jung, J. Am. Chem. Soc., 1974, 96, 3682.
- 51. P. L. Stotter and K. A. Hill, J. Am. Chem. Soc., 1974, 96, 6524.
- 52. W. S. Johnson, D. S. Allen, Jr., R. R. Hindersinn, G. N. Sausen and R. Pappo, J. Am. Chem. Soc., 1962, 84, 2181
- 53. J. A. Marshall, G. L. Bundy and W. I. Fanta, J. Org. Chem., 1968, 33, 3913.
- H. O. House and B. M. Trost, J. Org. Chem., 1965, 30, 2502.
 E. Negishi, H. Matsushita, S. Chatterjee and R. A. John, J. Org. Chem., 1982, 47, 3188.
- 56. E. Negishi, F.-T. Luo, A. J. Pecora and A. Silveira, Jr., J. Org. Chem., 1983, 48, 2427.
- 57. F.-T. Luo and E. Negishi, Tetrahedron Lett., 1985, 26, 2177.
- 58. E. Negishi and R. A. John, J. Org. Chem., 1983, 48, 4098.
- 59. B. M. Trost and E. Keinan, Tetrahedron Lett., 1980, 21, 2591.
- 60. B. M. Trost and C. R. Self, J. Org. Chem., 1984, 49, 468.
- 61. Ref. 1, p. 152.
- 62. M. Kosugi, I. Hagiwara, T. Sumiya and T. Migita, Bull. Chem. Soc. Jpn., 1984, 57, 242.
- 63. I. Kuwajima and H. Urabe, J. Am. Chem. Soc., 1982, 104, 6831.
- 64. E. J. Corey and R. A. Sneen, J. Am. Chem. Soc., 1956, 78, 6269.
- 65. L. Velluz, J. Valls and G. Nominé, Angew. Chem., Int. Ed. Engl., 1965, 4, 181 and refs. therein.
- 66. C. Agami, J. Levisalles and B. Lo Cicero, Tetrahedron, 1979, 35, 961.
- 67. H. D. Zook and J. A. Miller, J. Org. Chem., 1971, 36, 1112 and refs. therein.
- 68. L. Tenud, S. Farooq, J. Seibl and A. Eschenmoser, Helv. Chim. Acta, 1970, 53, 2059.
- 69. F. Johnson, Chem. Rev., 1968, 68, 375.
- 70. H. O. House, B. A. Tefertiller and H. D. Olmstead, J. Org. Chem., 1968, 33, 935.
- 71. B. J. L. Huff, F. N. Tuller and D. Caine, J. Org. Chem., 1969, 34, 3070.
- 72. G. W. Spears, Ph.D. Dissertation, Columbia University, 1986.
- 73. H. O. House and M. J. Umen, J. Org. Chem., 1973, 38, 1000.
- 74. R. S. Matthews, P. K. Hyer and E. A. Folkers, Chem. Commun., 1970, 38.
- 75. R. S. Matthews, S. J. Girgenti and E. A. Folkers, Chem. Commun., 1970, 708.
- 76. P. T. Lansbury and G. E. DuBois, Tetrahedron Lett., 1972, 3305.
- 77. R. E. Ireland, P. S. Grand, R. E. Dickerson, J. Bordner and D. R. Rydjeski, J. Org. Chem., 1970, 35, 570.
- 78. G. Stork, in 'Proceedings of the 23rd International Congress of Pure and Applied Chemistry', Butterworths, Boston, MA, July 1971, vol. 2, p. 193.
- 79. D. Caine, S. T. Chao and H. A. Smith, Org. Synth. (N.Y.), 1977, 56, 52.
- 80. G. Stork and J. E. McMurry, J. Am. Chem. Soc., 1967, 89, 5464.
- 81. H. O. House and T. M. Bare, J. Org. Chem., 1968, 33, 943.
- 82. J. Klein, Tetrahedron, 1974, 30, 3349.
- 83. H. O. House and C. J. Blankley, J. Org. Chem., 1967, 32, 1741.
- 84. D. Caine, C. J. McCloskey and D. Van Derveer, J. Org. Chem., 1985, 50, 175.
- 85. E. J. Corey, R. Hartmann and P. A. Vatakencherry, J. Am. Chem. Soc., 1962, 84, 2611.
- 86. B. M. Trost, P. R. Bernstein and P. C. Funfschilling, J. Am. Chem. Soc., 1979, 101, 4378.
- 87. P. A. Grieco, T. Takigawa and D. R. Moore, J. Am. Chem. Soc., 1979, 101, 4380.
- 88. T. Rajamannar and K. K. Balasubramanian, Tetrahedron Lett., 1988, 29, 5789.

- 89. W. Oppolzer, K. Bättig and M. Petrzilka, Helv. Chim. Acta, 1978, 61, 1945.
- 90. G. H. Posner, M. J. Chapdelaine and C. M. Lentz, J. Org. Chem., 1979, 44, 3661.
- 91. A. J. Birch and G. S. R. Subba Rao, Tetrahedron Lett., 1967, 2763.
- 92. W. Bernhard, I. Fleming and D. Waterson, J. Chem. Soc., Chem. Commun., 1984, 28.
- 93. I. Fleming and J. J. Lewis, J. Chem. Soc., Chem. Commun., 1985, 149.
- 94. I. Fleming, J. H. M. Hill, D. Parker and D. Waterson, J. Chem. Soc., Chem. Commun., 1985, 318, 95. T. Hayashi, Y. Matsumoto and Y. Ito, Tetrahedron Lett., 1988, 29, 4147; J. Am. Chem. Soc., 1988
- T. Hayashi, Y. Matsumoto and Y. Ito, Tetrahedron Lett., 1988, 29, 4147; J. Am. Chem. Soc., 1988, 110, 5579.
- 96. P. Caramella, N. G. Rondan, M. N. Paddon-Row and K. N. Houk, J. Am. Chem. Soc., 1981, 103, 2438; M. N.
- Paddon-Row, N. G. Rondan and K. N. Houk, J. Am. Chem. Soc., 1982, 104, 7162.
- 97. G. J. McGarvey and J. M. Williams, J. Am. Chem. Soc., 1985, 107, 1435 and refs. therein.
- J. E. Baldwin and L. I. Kruse, J. Chem. Soc., Chem. Commun., 1977, 233.
 H. O. House, W. V. Phillips, T. S. B. Sayer and C.-C. Yau, J. Org. Chem., 1978, 43, 700.
- 100. H. O. House, T. S. B. Sayer and C.-C. Yau, J. Org. Chem., 1978, 43, 2153.
- 101. S. J. Etheredge, J. Org. Chem., 1966, 31, 1990.
 102. J. L. Marshall, Tetrahedron Lett., 1971, 753.
- 103. R. L. Cargill, D. F. Bushey, P. D. Ellis, S. Wolff and W. C. Agosta, J. Org. Chem., 1974, 39, 573.
- 104. J.-M. Conia and F. Rouessac, Tetrahedron, 1961, 16, 45.
- 105. R. B. Kelly, J. Zamecnik and B. A. Beckett, Can. J. Chem., 1972, 50, 3455.
- 106. E. Piers and T. W. Hall, J. Chem. Soc., Chem. Commun., 1977, 880.
- 107. C. H. Heathcock, R. A. Badger and J. W. Patterson, Jr., J. Am. Chem. Soc., 1967, 89, 4133.
- 108. E. Piers, W. de Waal and R. W. Britton, J. Am. Chem. Soc., 1971, 93, 5113.
- 109. K. J. Schmalzl and R. N. Mirrington, Tetrahedron Lett., 1970, 3219.
- 110. J. E. McMurry, J. Am. Chem. Soc., 1968, 90, 6821.
- 111. J. E. McMurry and S. J. Isser, J. Am. Chem. Soc., 1972, 94, 7132.
- 112. J. E. McMurry, J. Org. Chem., 1971, 36, 2826.
- 113. P. A. Grieco, Y. Ohfune and G. Majetich, J. Am. Chem. Soc., 1977, 99, 7393.
- 114. E. Piers and V. Karunaratne, Can. J. Chem., 1984, 62, 629.
- 115. E. Piers and V. Karunaratne, J. Chem. Soc., Chem. Commun., 1984, 959.
- 116. M. Asaoka, K. Takenouchi and H. Takei, Chem. Lett., 1988, 1225.
- 117. G. D. Vite and T. A. Spencer, J. Org. Chem., 1988, 53, 2555.
 118. H. O. House, W. C. Liang and P. D. Weeks, J. Org. Chem., 1974, 39, 3102.
- 119. Y. Odic and M. Pereyre, C. R. Hebd. Seances Acad. Sci., Ser. C, 1970, 270, 100.
- 120. M. E. Jung and R. B. Blum, Tetrahedron Lett., 1977, 3791.
- 121. P. Groenewegen, H. Kallenberg and A. van der Gen, Tetrahedron Lett., 1978, 491.
- 122. H. K. Dietl and K. C. Brannock, Tetrahedron Lett., 1973, 1273.
- 123. R. E. Ireland and L. N. Mander, J. Org. Chem., 1967, 32, 689.
- 124. G. Mehta and K. S. Rao, J. Am. Chem. Soc., 1986, 108, 8015.
- 125. R. A. Lee, C. McAndrews, K. M. Patel and W. Reusch, Tetrahedron Lett., 1973, 965.
- 126. G. Stork and R. L. Danheiser, J. Org. Chem., 1973, 38, 1775.
- 127. S. E. Denmark, K. L. Habermas and G. A. Hite, Helv. Chim. Acta, 1988, 71, 168.
- 128. Y. Zhang, B. O'Connor and E. Negishi, J. Org. Chem., 1988, 53, 5588.
- 129. R. D. Clark and C. H. Heathcock, J. Org. Chem., 1976, 41, 636. 130. C. Agami, M. Chauvin and J. Levisalles, *Tetrahedron Lett.*, 1979, 1855.
- 131. E. Toromanoff and R. Bucourt, Tetrahedron Lett., 1976, 3523; L. Nedelec, J. C. Gasc and R. Bucourt, Tetrahedron, 1974, 30, 3263.
- 132. G. Stork, R. L. Danheiser and B. Ganem, J. Am. Chem. Soc., 1973, 95, 3414.
- 133. B. H. Toder, S. J. Branca, R. K. Dieter and A. B. Smith, III, Synth. Commun., 1975, 5, 435.
- R. L. Cargill and T. E. Jackson, J. Org. Chem., 1973, 38, 2125.
 A. G. Schultz and J. P. Dittami, J. Org. Chem., 1983, 48, 2318.
- 136. J. E. Telschow and W. Reusch, J. Org. Chem., 1975, 40, 862.
- 137. I. Fleming, in 'Frontier Orbitals and Organic Reactions', Wiley, New York, 1976, p. 45.
- H. J. Ringold and S. K. Malhotra, J. Am. Chem. Soc., 1962, 84, 3402.
 W. G. Dauben and A. C. Ashcraft, J. Am. Chem. Soc., 1963, 85, 3673.
- 140. G. Stork and J. Benaim, J. Am. Chem. Soc., 1971, 93, 5938.
- 141. F. H. Bottom and J. F. McQuillin, Tetrahedron Lett., 1967, 1975.
- Z. G. Hajos, R. A. Micheli, D. R. Parrish and E. P. Oliveto, J. Org. Chem., 1967, 32, 3008. 142.
- Y. Nakadaira, J. Hayashi, H. Sato and K. Nakanishi, J. Chem. Soc., Chem. Commun., 1972, 282. 143.
- 144. G. Stork and E. W. Logusch, J. Am. Chem. Soc., 1980, 102, 1218, 1219; G. Stork, J. D. Winkler and C. S. Shiner, J. Am. Chem. Soc., 1982, 104, 3767.
- F. A. Lakhvich, L. G. Lis and A. A. Pap, Zh. Org. Khim., 1988, 24, 1388 (Chem. Abstr., 1989, 110, 114 328h) 145.
- T. A. Bryson and R. B. Gammill, Tetrahedron Lett., 1974, 3963; R. B. Gammill and T. A. Bryson, Synthesis, 146. 1976, 401.
- 147. M. Yoshimoto, N. Ishida and T. Hiraoka, Tetrahedron Lett., 1973, 39.
- 148. M. Koreeda and S. G. Mislankar, J. Am. Chem. Soc., 1983, 105, 7203; M. Koreeda and Y. P. Liang Chen, Tetrahedron Lett., 1981, 22, 15 and refs. therein.
- 149. A. B. Smith, III, P. A. Levenberg, P. J. Jerris, R. M. Scarborough, Jr. and P. M. Wovkulich, J. Am. Chem. Soc., 1981, 103, 1501.
- 150. R. K. Dieter and J. R. Fishpaugh, J. Org. Chem., 1987, 52, 923.
- 151. P. T. Lansbury, R. W. Erwin and D. A. Jeffrey, J. Am. Chem. Soc., 1980, 102, 1602.
- 152. P. T. Lansbury, G. E. Bebernitz, S. C. Maynard and C. J. Spagnuolo, Tetrahedron Lett., 1985, 26, 169.
- 153. T. Mukaiyama, Org. React. (N.Y), 1982, 28, 203.
- 154. I. Fleming, Chem. Soc. Rev., 1981, 10, 83.
- 155. M. T. Reetz, Angew. Chem., Int. Ed. Engl., 1982, 21, 96.

- 156. I. Paterson, Tetrahedron, 1988, 44, 4207.
- 157. T. H. Chan, I. Paterson and J. Pinsonnault, Tetrahedron Lett., 1977, 4183.
- 158. M. T. Reetz, I. Chatziiosifidis, F. Hubner and H. Heimbach, Org. Synth. (N.Y.), 1984, 62, 95.
- 159. S. Murata, M. Suzuki and R. Noyori, J. Am. Chem. Soc., 1980, 102, 3248; Tetrahedron Lett., 1980, 21, 2527.
- 160. M. T. Reetz, W. F. Maier, I. Chatziiosifidis, A. Giannis, H. Heimbach and U. Löwe, Chem. Ber., 1980, 113, 3741.
- 161. T. V. Lee and K. A. Richardson, Tetrahedron Lett., 1985, 26, 3629; T. V. Lee, K. A. Richardson and D. A. Taylor, Tetrahedron Lett., 1986, 27, 5021; T. V. Lee, R. J. Boucher, K. L. Ellis and K. A. Richardson, Tetrahedron Lett., 1988, 29, 685; T. V. Lee, R. J. Boucher and C. J. M. Rockell, Tetrahedron Lett., 1988, 29, 689.
- 162. J. Iqbal and R. Mohan, Tetrahedron Lett., 1989, 30, 239.
- 163. G. S. Cockerill and P. J. Kocienski, J. Chem. Soc., Chem. Commun., 1983, 705 and refs. therein.
- 164. D. L. Kuo and T. Money, Can J. Chem., 1988, 66, 1794 and refs. therein.
- 165. A. B. Smith, III, M. A. Guaciaro, S. R. Schow, P. M. Wovkulich, B. H. Toder and T. W. Hall, J. Am. Chem. Soc., 1981, 103, 219; A. B. Smith, III, in 'Strategies and Tactics in Organic Synthesis', ed. T. Lindberg, Academic Press, New York, 1984, p. 223.
- 166. P. Magnus and P. A. Carter, J. Am. Chem. Soc., 1988, 110, 1626.
- 167. I. Fleming and J. Iqbal, Tetrahedron Lett., 1983, 24, 2913.
- 168. I. Paterson, Tetrahedron Lett., 1979, 1519.
- 169. J. Tsuji, Tetrahedron, 1986, 42, 4361; J. Organomet Chem., 1986, 300, 281.
- 170. G. Stork, A. Brizzolara, H. Landesman, J. Szmuszkovicz and R. Terrell, J. Am. Chem. Soc., 1963, 85, 207.
- 171. P. W. Hickmott, Tetrahedron, 1982, 38, 1975, 3363
- 172. J. K. Whitesell and M. A. Whitesell, Synthesis, 1983, 517.
- M. E. Kuehne and T. Garbacik, J. Org. Chem., 1970, 35, 1555.
 T. J. Curphey, J. C. Hung and C. C. C. Chu, J. Org. Chem., 1975, 40, 607.
- 175. S. Karady, M. Lenfant and R. E. Wolff, Bull. Soc. Chim. Fr., 1965, 2472.
- 176. P. W. Hickmott and N. F. Firrell, J. Chem. Soc., Perkin Trans. 1, 1978, 340.
- 177. T. J. Curphey and J. C. Hung, J. Chem. Soc., Chem. Commun., 1967, 510.
- 178. P. W. Hickmott, Tetrahedron, 1984, 40, 2989.
- G. Stork and S. R. Dowd, J. Am. Chem. Soc., 1963, 85, 2178. 179.
- 180. E. J. Corey and D. Enders, Tetrahedron Lett., 1976, 3, 11; Chem. Ber., 1978, 111, 1337, 1362.
- 181. D. E. Bergbreiter and M. Newcomb, in 'Asymmetric Synthesis', ed. J. D. Morrison, Academic Press, New York, 1983, vol. 2, chap. 9, p. 243 and refs. therein.
- 182. P. A. Wender and M. A. Eissenstat, J. Am. Chem. Soc., 1978, 100, 292; P. A. Wender and J. M. Schaus, J. Org. Chem., 1978, 43, 782.
- 183. D. A. McCrae and L. Dolby, J. Org. Chem., 1977, 42, 1607.
- 184. M. Larchevêque, G. Valette and T. Cuvigny, Synthesis, 1977, 424.
- 185. J. K. Smith, D. E. Bergbreiter and M. Newcomb, J. Am. Chem. Soc., 1983, 105, 4396; J. K. Smith, M. Newcomb, D. E. Bergbreiter, D. R. Williams and A. I. Meyers, Tetrahedron Lett., 1983, 24, 3559.
- 186. K. N. Houk, R. W. Strozier, N. G. Rondan, R. R. Fraser and N. Chuaqui-Offermanns, J. Am. Chem. Soc., 1980, 102, 1426.
- 187. A. Hosomi, Y. Araki and H. Sakurai, J. Am. Chem. Soc., 1982, 104, 2081.
- 188. R. R. Fraser, J. Banville and K. L. Dhawan, J. Am. Chem. Soc., 1978, 100, 7999.
- 189. R. A. Wanat, D. B. Collum, G. Van Duyne, J. C. Clardy and R. T. DePue, J. Am. Chem. Soc., 1986, 108, 3415.
- 190. G. R. Kieczykowski, R. H. Schlessinger and R. B. Sulsky, Tetrahedron Lett., 1976, 597.
- 191. D. Enders, in 'Asymmetric Synthesis', ed. J. D. Morrison, Academic Press, New York, 1984, vol. 3, chap, 4, p. 275 and refs. therein.
- 192. D. E. Bergbreiter and M. Momongan, in 'Comprehensive Organic Synthesis', ed. B. M. Trost, Pergamon Press, Oxford, 1991, vol. 2, chap. 1.17.
- 193. J. W. Ludwig, M. Newcomb and D. E. Bergbreiter, J. Org. Chem., 1980, 45, 4666.
- S. Takano, Y. Shimazaki, M. Takahashi and K. Ogasawara, J. Chem. Soc., Chem. Commun., 1988, 1004. 1**94**.
- 195. M. Yamashita, K. Matsumiya, M. Tanabe and R. Suemitsu, Bull. Chem. Soc. Jpn., 1985, 58, 407.
- 196. P. F. Hudrlik, A. M. Hudrlik, T. Yimenu, M. A. Waugh and G. Nagendrappa, Tetrahedron, 1988, 44, 3791.
- 197. G. R. Kieczykowski, C. S. Pogonowski, J. E. Richman and R. H. Schlessinger, J. Org. Chem., 1977, 42, 175.
 198. D. B. Collum, D. Kahne, S. A. Gut, R. T. DePue, F. Mohamadi, R. A. Wanat, J. C. Clardy and G. Van Duyne, J. Am. Chem. Soc., 1984, 106, 4865.
- 199. R. A. Wanat and D. B. Collum, J. Am. Chem. Soc., 1985, 107, 2078.
- 200. E. J. Corey and H. L. Pearce, J. Am. Chem. Soc., 1979, 101, 5841.
- 201. H. Iio, M. Mondon, K. Okada and T. Tokoroyama, J. Chem. Soc., Chem. Commun., 1987, 358.
- 202. M. Yamashita, K. Matsumiya, K. Nakano and R. Suemitsu, Chem. Lett., 1988, 7, 1215.
- 203. W. G. Kofron and M. K. Yeh, J. Org. Chem., 1976, 41, 439.
- 204. M. E. Jung, P. A. Blair and J. A. Lowe, Tetrahedron Lett., 1976, 1439.
- R. E. Gawley and T. Nagy, *Tetrahedron Lett.*, 1984, 25, 263.
 J. W. ApSimon and T. L. Collier, *Tetrahedron*, 1986, 42, 5157.
- 207. M. Nogradi, in 'Stereoselective Synthesis', VCH, Weinheim, 1987, p. 248.
- 208. J. K. Whitesell and S. W. Felman, J. Org. Chem., 1977, 42, 1663. 209. A. I. Meyers, D. R. Williams and M. Druelinger, J. Am. Chem. Soc., 1976, 98, 3032; A. I. Meyers, D. R.
- Williams, G. W. Erickson, S. White and M. Druelinger, J. Am. Chem. Soc., 1981, 103, 3081.
- 210. J. K. Whitesell and M. A. Whitesell, J. Org. Chem., 1977, 42, 377.
- 211. S. Hashimoto and K. Koga, Tetrahedron Lett., 1978, 573; Chem. Pharm. Bull., 1979, 27, 2760.
- 212. A. I. Meyers, D. R. Williams, S. White and G. W. Erickson, J. Am. Chem. Soc., 1981, 103, 3088
- 213. D. Enders, H. Eichenauer, U. Baus, H. Schubert and K. A. M. Kremer, Tetrahedron, 1984, 40, 1345.
- 214. J. S. Clark and A. B. Holmes, Tetrahedron Lett., 1988, 29, 4333. 215.
- S. I. Pennanen, Acta Chim. Scand., Ser. B., 1981, 35, 555. 216. K. C. Nicolaou, D. P. Papahatjis, D. A. Claremon and R. E. Dolle, III, J. Am. Chem. Soc., 1981, 103, 6967.
- 217. N. Petragnani and M. Yonashiro, Synthesis, 1982, 521.
- 218. S. Arseniyadis, K. S. Kyler and D. S. Watt, Org. React. (N.Y.), 1984, 31, 1.
- 219. A. P. Krapcho and E. A. Dundulis, J. Org. Chem., 1980, 45, 3236.
- 220. F. Plavac and C. H. Heathcock, Tetrahedron Lett., 1979, 2115.
- 221. D. Seebach, J. D. Aebi, M. Gander-Coquoz and R. Naef, Helv. Chim. Acta, 1987, 70, 1194.
- 222. R. Naef and D. Seebach, Angew. Chem., Int. Ed. Engl., 1981, 20, 1030.
- 223. D. Seebach, R. I. Mwinkelried and T. Weber, in 'Modern Synthetic Methods', ed. R. Scheffold, Springer, New York, 1986, p. 148. 224. K. Tomioka, Y.-S. Cho, F. Sato and K. Koga, J. Org. Chem., 1988, 53, 4094; K. Tomioka and K. Koga, in
- 'Asymmetric Synthesis', ed. J. D. Morrison, Academic Press, New York, 1983, vol. 2, chap. 3, p. 201 and refs. therein; K. Tomioka, H. Kawasaki, K. Yasuda and K. Koga, J. Am. Chem. Soc., 1988, 110, 3597
- 225. H.-M. Shieh and G. D. Prestwich, Tetrahedron Lett., 1982, 23, 4643; see also M. Larcheveque and S. Henrot, Tetrahedron, 1987, 43, 2303.
- 226. W. C. Still and I. Galynker, Tetrahedron, 1981, 37, 3981.
- 227. D. Seebach, R. Naef and G. Calderari, Tetrahedron, 1984, 40, 1313; D. Seebach and A. Fadel, Helv. Chim. Acta, 1985, 68, 1243; Th. Weber, R. Aeschimann, Th. Maetzke and D. Seebach, Helv. Chim. Acta, 1986, 69, 1365; D. Seebach, D. D. Miller, St. Müller and T. Weber, Helv. Chim. Acta, 1985, 68, 949; M. Gander-Coquoz and D. Seebach, Helv. Chim. Acta, 1988, 71, 224.
- 228 G. Fráter, U. Müller and W. Günther, Tetrahedron Lett., 1981, 22, 4221.
- 229. R. Fitzi and D. Seebach, Tetrahedron, 1988, 44, 5277.
- 230. J. Zimmerman, D. Seebach and T.-K. Ha, Helv. Chim. Acta, 1988, 71, 1143.
- 231. A. I. Meyers, M. Harre and R. Garland, J. Am. Chem. Soc., 1984, 106, 1146.
- 232. A. I. Meyers and K. Th. Wanner, Tetrahedron Lett., 1985, 26, 2047.
- 233. A. I. Meyers and R. H. Wallace, J. Org. Chem., 1989, 54, 2509.
- 234. A. I. Meyers and B. A. Lefker, Tetrahedron, 1987, 43, 5663; A. I. Meyers and B. A. Lefker, J. Org. Chem., 1986, **51**, 1541.
- K. Tomioka, H. Kawasaki and K. Koga, Tetrahedron Lett., 1985, 26, 3027; H. Kawasaki, K. Tomioka and K. 235. Koga, Tetrahedron Lett., 1985, 26, 3031.
- 236 Y. Yamamoto and K. Maruyama, J. Chem. Soc., Chem. Commun., 1984, 904.
- 237. H. Estermann and D. Seebach, Helv. Chim. Acta, 1988, 71, 1824.
- J. L. Herrmann and R. H. Schlessinger, *Tetrahedron Lett.*, 1973, 2429.
 G. Fráter, U. Müller and W. Günther, *Tetrahedron*, 1984, 40, 1269.
- 240. D. Seebach, J. D. Aebi and D. Wasmuth, Org. Synth., 1985, 63, 109.
- 241. K. K. Mahalanabis, M. Mumtaz and V. Snieckus, Tetrahedron Lett., 1982, 23, 3971, 3975.
- 242. W. D. Lubell and H. Rapoport, J. Am. Chem. Soc., 1988, 110, 7447.
- 243. M. Larcheveque, E. Ignatova and T. Cuvigny, Tetrahedron Lett., 1978, 3961.
- 244. D. A. Evans and J. M. Takacs, Tetrahedron Lett., 1980, 21, 4233.
- 245. P. E. Sonnet and R. R. Heath, J. Org. Chem., 1980, 45, 3137.
- 246. D. A. Evans, Aldrichimica Acta, 1982, 15, 23.
- 247. D. Askin, R. P. Volante, K. M. Ryan, R. A. Reamer and I. Shinkai, Tetrahedron Lett., 1988, 29, 4245.
- Y. Kawanami, Y. Ito, T. Kitagawa, Y. Taniguchi, T. Katsuki and M. Yamaguchi, Tetrahedron Lett., 1984, 25, 248. 857; M. Enomoto, Y. Ito, T. Katsuki and M. Yamaguchi, Tetrahedron Lett., 1985, 26, 1343; S. Ikegami, H. Uchiyama, T. Hayama, T. Katsuki and H. Yamaguchi, Tetrahedron, 1988, 44, 5333.
- 249. D. A. Evans, M. D. Ennis and D. J. Mathre, J. Am. Chem. Soc., 1982, 104, 1737.
- 250. G. Cardillo, A. D'Amico, M. Orena and S. Sandri, J. Org. Chem., 1988, 53, 2354; G. Cardillo, M. Orena, M. Romero and S. Sandri, Tetrahedron, 1989, 45, 1501.
- 251. E. Ade, G. Helmchen and G. Heiligenmann, Tetrahedron Lett., 1980, 21, 1137; R. Schmierer, G. Grotemeier, G. Helmchen and A. Selim, Angew. Chem., Int. Ed. Engl., 1981, 20, 207; G. Helmchen and R. Wierzchowski, Angew. Chem., Int. Ed. Engl., 1984, 23, 60. W. Oppolzer, P. Dudfield, T. Stevenson and T. Godel, Helv. Chim. Acta, 1985, 68, 212.
- 252.
- 253. S. Yamada, T. Oguri and T. Shioiri, J. Chem. Soc., Chem. Commun., 1976, 136.
- 254. J. A. Bajgrowicz, B. Cossec, Ch. Pigiere, R. Jacquier and Ph. Viallefont, Tetrahedron Lett., 1983, 24, 3721.
- 255. J. M. McIntosh, R. K. Leavitt, P. Mishra, K. C. Cassidy, J. E. Drake and R. Chadha, J. Org. Chem., 1988, 53, 1947.
- 256. J.-P. Genet, S. Juge, S. Achi, S. Mallart, J. Ruiz Montes and G. Levif, Tetrahedron, 1988, 44, 5263.
- 257. L. S. Liebeskind and M. E. Welker, Organometallics, 1983, 2, 194; G. J. Baird, J. A. Bandy, S. G. Davies and K. Prout, J. Chem. Soc., Chem. Commun., 1983, 1202; S. G. Davies and J. C. Walker, J. Chem. Soc., Chem. Commun., 1985, 209; C. A. Rusik, T. L. Tonker and J. L. Templeton, J. Am. Chem. Soc., 1986, 108, 4652.
- 258. J. D. Albright, Tetrahedron, 1983, 39, 3207.
- 259. G. Stork, L. D. Cama and D. R. Coulson, J. Am. Chem. Soc., 1974, 96, 5268; G. Stork and J. F. Cohen, J. Am. Chem. Soc., 1974, 96, 5270.
- 260. G. Stork, J. O. Gardner, R. K. Boeckman, Jr. and K. A. Parker, J. Am. Chem. Soc., 1973, 95, 2014; G. Stork and R. K. Boeckman, Jr., J. Am. Chem. Soc., 1973, 95, 2016.
- 261. S. H. Ahn, D. Kim, M. W. Chun and W.-K. Chung, Tetrahedron Lett., 1986, 27, 943; D. Kim and H. S. Kim, J. Org. Chem., 1987, 52, 4633; D. Kim, Y. M. Jang, I. O. Kim and S. W. Park, J. Chem. Soc., Chem. Commun., 1988, 760.
- T. Kitahara and K. Mori, J. Org. Chem., 1984, 49, 3281. 262.
- 263. T. Takahashi, T. Nagashima, H. Ikeda and T. Tsuji, Tetrahedron Lett., 1982, 23, 4361.
- 264. S. J. Danishefsky, K. Vaughan, R. C. Gadwood and K. Tsuzuki, J. Am. Chem. Soc., 1981, 103, 4136.
- 265. J. H. Hutchinson, G. Pattenden and P. L. Meyers, Tetrahedron Lett., 1987, 28, 1313.

- 266. A. S. Kende and J. Chen, J. Am. Chem. Soc., 1985, 107, 7184.
- 267. J. A. Katzenellenbogen and A. L. Crumrine, J. Am. Chem. Soc., 1974, 96, 5662; J. Am. Chem. Soc., 1976, 98, 4925; P. M. Savu and J. A. Katzenellenbogen, J. Org. Chem., 1981, 46, 239.
- 268. M. Majewski, G. B. Mpango, M. T. Thomas, A. Wu and V. Snieckus, J. Org. Chem., 1981, 46, 2029.
- 269. A. S. Kende and B. H. Toder, J. Org. Chem., 1982, 47, 163.
- 270. A. G. Schultz, M. Macielag, P. Sundararaman, A. G. Taveras and M. Welch, J. Am. Chem. Soc., 1988, 110, 7828.
- 271. T. Takahashi, T. Nagashima and T. Tsuji, Tetrahedron Lett., 1981, 22, 1359; T. Takahashi, H. Nemoto and J. Tsuji, Tetrahedron Lett., 1983, 24, 2005, 3485; T. Takahashi, K. Kitamura, H. Nemoto and J. Tsuji, Tetrahedron Lett., 1983, 24, 3489.
- 272. A. I. Meyers, in 'Heterocycles in Organic Synthesis', ed. C. E. Taylor and A. Weissberger, Wiley-Interscience, New York, 1974, chaps. 9 and 10.
 273. K. A. Lutomski and A. I. Meyers, in 'Asymmetric Synthesis', ed. J. D. Morrison, Academic Press, New York,
- 1984, vol. 3, chap. 3, p. 213.
- 274. U. Schöllkopf, W. Hartwig and U. Groth, Angew. Chem., Int. Ed. Engl., 1979, 18, 863; U. Schöllkopf, U. Groth and C. Deng, Angew. Chem., Int., Ed. Engl., 1981, 20, 798; U. Schöllkopf, U. Groth, K.-O. Westphalen and C. Deng, Synthesis, 1981, 969; J. Nozulak and U. Schöllkopf, Synthesis, 1982, 866; U. Schöllkopf and H.-J. Neubauer, Synthesis, 1982, 861; U. Groth and U. Schöllkopf, Synthesis, 1983, 37; U. Schöllkopf, U. Busse, R. Kilger and P. Lehr, Synthesis, 1984, 271; U. Schöllkopf, Pure Appl. Chem., 1983, 55, 1799; U. Schöllkopf, Tetrahedron, 1983, 39, 2085.
- 275. E. C. Taylor, G. H. Hawks, III and A. McKillop, J. Am. Chem. Soc., 1968, 90, 2421.
- 276. M. E. Garst and B. J. McBride, J. Org. Chem., 1983, 48, 1362.
- 277. J. Hooz and J. Smith, J. Org. Chem., 1972, 37, 4200.
- J. H. Clark and J. M. Miller, J. Chem. Soc., Perkin Trans. 1, 1977, 1743.
 T. Shono, S. Kashimura, M. Sawamura and T. Soejima, J. Org. Chem., 1988, 53, 907.
- 280. H. Schick, H. Schwarz, A. Finger and S. Schwarz, Tetrahedron, 1982, 38, 1279.
- 281. B. M. Trost and D. P. Curran, J. Am. Chem. Soc., 1980, 102, 5699.
- 282. M. E. Kuehne and J. A. Nelson, J. Org. Chem., 1970, 35, 161, 171.
- T. A. Spencer, T. D. Weaver, R. M. Villarica, R. J. Friary, J. Posler and M. A. Schwartz, J. Org. Chem., 1968, 283. 33, 712
- 284. M. A. Casadei, C. Galli and L. Mandolini, J. Am. Chem. Soc., 1984, 106, 1051.
- 285. G. Quinkert, H.-G. Schmalz, E. Walzer, S. Gross, T. Kowalczyk-Przewloka, C. Schierloh, G. Dürner, J. W. Bats and H. Kessler, *Liebigs Ann. Chem.*, 1988, 283, and refs. therein. 286. P. Deslongchamps, S. Lamothe and H. S. Lin, *Can. J. Chem.*, 1984, **62**, 2395. 287. B. M. Trost, *J. Organomet. Chem.*, 1986, **300**, 263; *Chemtracts—Org. Chem.*, 1988, **1**, 415.

- 288. C. R. Hauser and T. M. Harris, J. Am. Chem. Soc., 1958, 80, 6360.
- 289. T. M. Harris and C. M. Harris, Org. React. (N.Y.), 1969, 17, 155.
- S. N. Huckin and L. Weiler, J. Am. Chem. Soc., 1974, 96, 1082.
 J. D. Winkler, J. P. Hey and P. G. Williard, Tetrahedron Lett., 1988, 29, 4691.
- 292. R. L. Funk, T. A. Olmstead and M. Parvez, J. Am. Chem. Soc., 1988, 110, 3298.
- 293. J. A. M. van den Goorbergh and A. van der Gen, Recl. Trav. Chim. Pays-Bas, 1983, 102, 393.

1.2 Alkylations of Nitrogen-stabilized Carbanions

ROBERT E. GAWLEY and KATHLEEN REIN

University of Miami, Coral Gables, FL, USA

1.2.1 INTRODUCTION	65
1.2.2 ACHIRAL AND RACEMIC PRECURSORS	68
1.2.2.1 Acyclic Systems 1.2.2.2 Heterocyclic Systems	68 69
1.2.3 HOMOCHIRAL PRECURSORS	72
1.2.3.1 Mechanistic Aspects 1.2.3.2 Acyclic Systems 1.2.3.3 Heterocyclic Systems	72 75 76
1.2.4 REFERENCES	82

1.2.1 INTRODUCTION

The deprotonation of an sp^3 -hydrogen α to nitrogen has been developed into a highly useful synthetic tool in recent years.^{1,2} Although some authors have suggested that this type of reactivity is a reversal of the 'normal' reactivity adjacent to nitrogen,^{2,3} we submit that this notion is inappropriate and should be discontinued. The α -deprotonation of dimethyldodecylamine⁴ and triethylamine⁵ was reported over 20 years ago. In 1984, Ahlbrecht reported that *s*-butylpotassium readily deprotonates *N*-methylpiperidine, *N*-methylpyrrolidine and triethylamine, and that the derived organometallics add readily to aldehydes, ketones and alkyl halides.⁶ An example is shown in Scheme 1.



The above examples notwithstanding, for an efficient and mild deprotonation the nitrogen is best derivatived. Two closely related amine derivatives activate the α -protons: nitrosamines^{1,2,7,8} and amides.² The chemistry of nitrosamine and amide anions and related systems was thoroughly reviewed in 1984,² and so this chapter concentrates on recent developments. Most of these developments are in the area of dipole-stabilized species: amides, amidines and similar compounds. Synthetic applications of these systems have been investigated in two broad categories of substrates: achiral (or racemic) and chiral (non-racemic). In the latter case, the dipole-stabilized organometallic is chiral and nonracemic, affording interesting opportunities for asymmetric synthesis. Indeed, highly stereoselective methods for the prep-

aration of enantiomerically pure compounds have been developed that employ the stereoelectronic requirements of the organometallic to advantage. Thus, this chapter is divided along the following lines: synthetic applications of achiral and racemic systems, followed by synthetic applications of homochiral systems. Within each class, the examples are further categorized as either acyclic or heterocyclic according to the location of the carbanion/metal.

The geometries of amide and nitrosamine anions differ in a fundamental way, and this difference has important implications for synthesis. Although cyclic nitrosamines lose the axial proton *syn* to the nitrosamine oxygen⁹ and alkylate by axial approach of the electrophile,¹⁰ cyclic amides lose the equatorial proton and alkylate equatorially¹¹ to give the thermodynamically less stable product,¹² as illustrated in Scheme 2.



i, BuLi; ii, PhCHO

Scheme 2

Thus the nitrosamine anion is a resonance-stabilized species in which the carbanionic lone pair overlaps the π -system of the nitrosamine. By contrast, the amide anion is a dipole-stabilized species in which the lone pair remains in or near the nodal plane of the amide π -system. Given the similarity of the two systems, this difference in behavior is striking. Although the geometry observed in nitrosamine anion alkylations is similar to that of their isoelectronic counterparts, oxime dianions,^{13,14} it is puzzling that amide anions differ, since they are also isoelectronic, as shown in Figure 1.



Figure 1 Isoelectronic anions

One explanation of the unique stereoelectronic requirements of amide anions ensues from *ab initio* studies conducted by Houk *et al.*,¹² Bach *et al.*,¹⁵ and Bartolotti and Gawley.¹⁶ To explain the equatorial alkylation, Bach argued, on the basis of calculations on a model system, that the rotation of the carbanion lone pair of HCONHCH₂⁻ out of the amide nodal plane and into conjugation with the p-system results in a large increase in energy (16–18 kcal mol⁻¹; 1 cal = 4.18 J). The source of the increase is a destabilizing interaction between the HOMO of the amide and the carbanion lone pair. When mixed, Ψ_3 of the amide is raised 44.6 kcal mol⁻¹, while Ψ_2 is lowered only 27.4 kcal mol⁻¹, resulting in a net increase of 17 kcal mol⁻¹ for the two frontier orbitals.¹⁵ Even so, it is the *less* stable anion geometry (1; Figure 2) that is *syn* to the carbonyl.^{15,16} It is not until the effect of the lithium cation is considered that the syn geometry is seen to be the more stable (Li-2; Figure 2), by 28 kcal mol⁻¹.¹⁶ It therefore seems most appropriate to call the reactive species 'dipole-stabilized organometallics'.



Figure 2 Ab initio structures and relative energies of N-methylformamide anion

The amide deprotonation step involves a prior coordination complex between the amide (or amidine) and the butyllithium base.^{17,18,19} As shown in Scheme 3, formation of a coordination complex between an amide or amidine substrate and butyllithium is observed by IR shortly after mixing, but before the deprotonation. A recent mechanistic study has shown that this coordination complex is between the amide and the *s*-butyllithium aggregate, and that the effect of the coordination is to increase the reactivity of the complex.²⁰ In fact, the aggregate may be activated by adding TMEDA prior to the amide, with a similar increase in reactivity. However, because these studies were conducted in cyclohexane, extrapolation of these observations to ethereal solvents is somewhat problematic.²⁰



Scheme 3

When the α -protons are allylic or benzylic, the selectivity of the deprotonation is attenuated by the competing stereoelectronic requirements of allylic activation and dipole stabilization. Stabilization of a

developing negative charge by resonance delocalization into a double bond and stabilization by an amide dipole which avoids resonance delocalization must be accommodated in the transition state and a balance of both effects must be achieved.²¹ Moreover, the organolithium species must accommodate the amide dipole as well as the allyl system, since reaction with electrophiles at the γ' -position is common. These aspects of reactivity are discussed further in Section 1.2.3.1, in the context of deprotonation and alkylation of homochiral substrates.

1.2.2 ACHIRAL AND RACEMIC PRECURSORS

1.2.2.1 Acyclic Systems

The alkylation of carbons α to nitrogen in acyclic systems has not developed much since the 1984 review.² Although amides may serve as excellent precursors of dipole-stabilized organolithiums, they are quite difficult to remove. For example, the two amide systems shown in Scheme 4 were designed to hinder approach of a nucleophile to the carbonyl carbon in order to preclude addition of the butyllithium base. Metalation on the carbon *syn* to the carbonyl oxygen is then the favored process, and alkylation occurs smoothly. The triisopropylbenzamides, however, could not be cleaved under a variety of acidic, basic, or reductive conditions. The diethylbutanamides behaved similarly in the alkylation step, and could be cleaved in good yield by acid hydrolysis.²²



i, Bu^sLi, TMEDA, Et₂O, -78 °C; ii, MeOD; iii, Me(CH₂)₁₁I; iv, HCl, H₂O, reflux 72 h

Scheme 4

Even though the diethylbutanamides are cleavable, the conditions required are quite harsh: aqueous hydrochloric acid at reflux for 3 d. A much better activating group was introduced in 1980 by Meyers: the formamidines shown in Scheme $5.^{23}$ The decreased reactivity of the C—NR bond relative to the C—O bond toward nucleophilic attack obviates the need for a bulky group to shield the carbonyl carbon. In the initial report, both N-cyclohexyl-, N-n-butyl- and N-t-butyl-formamidines were employed, but subsequent reports have focussed on the latter.



 $R = cyclohexyl, Bu^n, Bu^t$

i, BuLi, THF, -78 °C; ii, BnBr, iii, HCl, H₂O, MeOH

Scheme 5

One aspect of the reactivity of unsymmetrical formamidines such as those shown in Scheme 6 is the selective deprotonation at a benzylic position. This selectivity was used in the illustrated synthesis of nicotine, which also demonstrates the mild conditions for removal of the formamidine activating group by hydrazinolysis.²⁴



i, BuLi, THF, -78 °C; ii, I(CH₂)₃Cl; iii, N₂H₄, AcOH, EtOH, r.t.

Scheme 6

1.2.2.2 Heterocyclic Systems

Because of the difficulty of removing other activating groups, most of the work since the 1984 review² has been done with the formamidine auxiliary. In discussing this chemistry, an important point is that the acidity of the α -protons is considerably enhanced when they are also allylic or benzylic, and their chemistry is also somewhat different. The most important difference is the occurrence of single-electron transfer (SET) processes in the reaction of the lithiated amidines of five-, six- and seven-membered ring saturated heterocycles. For example, reaction of benzophenone with the lithiated piperidine (3), shown in Scheme 7, affords only benzophenone ketyl with no addition.²⁵ Reaction of (3) with alkyl halides results in low yields of alkylation products such as (5), with an oxidation product (6) being produced as well. That radical processes are involved is suggested by the reaction of (3) with hexenyl bromide,²⁶ affording a coupling product (5; R² = cyclopentylmethyl).²⁵ Conformationally locked systems such as (3; R = Bu^t) seem less likely to oxidize, and it was suggested that a conformationally mobile system might undergo ring inversion, thus placing the C—Li bond in an electronically unfavorable axial orientation, thereby encouraging SET.²⁵ Once oxidized, the radical (4) probably produces (6) by disproportionation.



Scheme 7

Two protocols circumvent, or at least minimize, the SET processes discussed above. One is addition of HMPA to the lithiated formamidines prior to addition of the alkyl halides, while the second is transmetalation to a cuprate.²⁵ Heterocycles having the C—Li bond in an allylic or benzylic position do not undergo SET at all.

The following discussion is organized by the type of heterocycle being alkylated, beginning with pyrrolidines. Pyrrolidine formamidines, as their pentynyl cuprate derivatives, may be alkylated in good yield;²⁵ a process for the synthesis of 1-azabicyclic ring systems utilizing this methodology is illustrated in Scheme 8.²⁷ An analogous process was also reported for piperidine formamidines.

Metalated piperidine amides²² and formamidines²⁵ may be converted to α -alkylated piperidines, as illustrated in Scheme 9. Again, the two examples illustrate the superiority of the formamidine moiety in both yield and ease of removal.



i, Bu^tLi, Et₂O-THF, -78 to -20 °C, 1 h; ii, PrC=CCu; iii, I(CH₂)_nCl; iv, KOH, MeOH, H₂O, 55 °C, 8-12 h Scheme 8



i, Bu^sLi, -78 to -25 °C; ii, BuI; iii, HCl, H₂O, reflux, 3 d; iv, PrC≡CCu; v, KOH, MeOH, H₂O, 8-12 h

Scheme 9

An α -substituted piperidine formamidine may be metalated at the α' -position selectively, but the alkylation is not stereoselective, providing a 1:1 mixture of *cis* and *trans* stereoisomers, as illustrated in Scheme 10.²⁵



i, Bu^sLi, −78 to −25 °C; ii, PrC≡CCu; iii, MeI

Scheme 10

When the α -position is allylic, metalation is easier, and the resulting organolithium is not prone to oxidation; γ -substitution then predominates over α -substitution, as shown by the example in Scheme 11.²⁵ The stereoelectronic preference of dipole-stabilized organolithiums for the nodal plane of the amide or

amidine π -system makes possible the directed metalation of bridgehead positions. Two examples are il-



i, BuⁿLi, THF, -78 °C; ii, C₆H₁₁I; iii, NaBH₄, pH 6

Scheme 11

lustrated in Scheme 12. In the first, compound (7) is metalated selectively at the bridgehead methine, in preference to the methyl.²⁸ In the second, the bridgehead methine that is bisbenzylic is preferentially deprotonated.²⁹ The second example is the key step in the synthesis of the anticonvulsant agent MK-801; similar reactions afford access to a number of structural analogs.



i, BuLi, THF, -78 °C; ii, MeI; iii, KOH, HO(CH₂)₂OH or H₂SO₄, EtOH

Scheme 12

In 1981, several papers appeared detailing the alkylation of the 1-position of tetrahydroisoquinolines through the intermediacy of lithiated amides^{30,31} phosphoramides,^{32,33} and formamidines.³⁴ These systems are discussed thoroughly in Beak's 1984 review,² and so we will only provide one illustration of each in Scheme 13.

More recently, an ingenious 'one-pot' procedure for the activation, alkylation and deactivation sequence was reported by Katritzky.³⁵ Reaction of tetrahydroisoquinoline with *n*-butyllithium followed by carbon dioxide affords the lithium carbamate (8), which can be further metalated with *t*-butyllithium and alkylated, as shown in Scheme 14.

Both tetrahydroisoquinoline amides³¹ and formamidines³⁴ may be successively alkylated at the 1-position. The second deprotonation is somewhat less facile, however, since it entails removal of a methine proton. In applying this methodology to the synthesis of naturally occurring tetrahydroisoquinoline alkaloids, a problem is encountered. The presence of oxygen substituents on the aromatic ring destabilizes the benzylic carbanion, and since this effect is felt in the transition state, the acidity of the benzylic protons is lowered. Thus, successive alkylation of *t*-butylformamidines of oxygenated tetrahydroisoquinolines is not possible. To circumvent the decreased acidity, a study of the relative acidity of various *N*substituted formamidines was done³⁶ that resulted in a solution to the problem.³⁷ As shown in Scheme 15, successive alkylation of the methoxymethylphenylformamidine (9) is readily achieved.



i, BuLi, THF, -78 °C; ii, BnCl; iii, HCl, MeOH, H₂O, reflux; iv, MeI; v, Red-Al, C₆H₆; vi, Ph(CH₂)₂Br; vii, KOH, MeOH, H₂O, reflux



i, BuⁿLi, THF; ii, CO₂; iii, Bu^tLi, -20 °C; iv, BuI; v, 2 M HCl

Scheme 14

In 1982, Meyers extended the formamidine methodology to the elaboration of β -carbolines.³⁸ The indole nitrogen of the β -carboline is protected as either its MOM derivative³⁸ or its potassium salt.³⁹ The latter method was used in the short synthesis of (±)-yohimbone illustrated in Scheme 16.⁴⁰

Finally, formamidines may be used to mediate the alkylation of perhydroazepines, as illustrated in Scheme 17.²⁵ As was the case with alkylation of the saturated heterocycles above, transmetalation to a cuprate or addition of HMPA was necessary to preclude SET reactions.

1.2.3 HOMOCHIRAL PRECURSORS

1.2.3.1 Mechanistic Aspects

The vast majority of stereoselective reactions in organic chemistry, especially in asymmetric synthesis, involve diastereofacial differentiation of a trigonal, sp^2 -atom. The stereoelectronic considerations outlined in the introduction to this chapter afford a rare opportunity in stereoselective synthesis: the preparation of a chiral organometallic in which the carbon bearing the metal is stereogenic. As a result of the



i, BuⁿLi, THF, -78 °C; ii, MeI, -100 °C; iii, BnX; iv, N₂H₄, EtOH, H₂O, AcOH, reflux

Scheme 15



i, KH, THF; ii, Bu'Li, -78 °C; iii, ArCH₂Cl

Scheme 16

tendency of the carbon-metal bond of a dipole-stabilized organometallic to orient in the nodal plane of the amide (or amidine) π -system, the carbanion retains sp^3 -geometry, as illustrated above in Figure 2.

When the precursor is achiral, two enantiomeric dipole-stabilized organolithiums may be formed. On the other hand, if the precursor is chiral, the two organolithiums are diastereomeric (Figure 3), and of unequal energy. The chirality center of the substrate may in principle be anywhere in the molecule, but the examples reported to date all have the stereocenter immediately adjacent to one end or the other of the amide group. Thus, the stereogenic carbon bearing the lithium may be influenced by an existing stereocenter either three or five atoms distant.



i, Bu^tLi, THF, -78 to -20 °C; ii, PrC=CCu; iii, PrI; iv, KOH, MeOH, H₂O





X = O, N; * = stereocenter



There are two steps in the overall alkylation process: (i) deprotonation of diastereotopic protons; and (ii) electrophilic attack on the intermediate dipole-stabilized organolithium(s). Either of these processes might, in principle, be the source of the diastereoselectivity observed in the overall process. Ab initio studies suggest that the barrier to inversion for a free dipole-stabilized anion is 16-18 kcal mol⁻¹,¹⁵ and that the *syn* lithiated species is 28 kcal mol⁻¹ more stable than the *anti* (Figure 2).¹⁶ Thus, in the absence of other functionality, the two diastereomers shown in Figure 3 do not invert. Because the stereoelectronic effects of dipole stabilization are felt in the transition state for deprotonation of conformationally locked piperidines, the equatorial proton is removed selectively in both amides^{11,12} and amidines,²⁵ as shown in Scheme 3 above. Note also that prior coordination of the alkyllithium base promotes selective removal of the equatorial proton from one of two chair conformers has been postulated to explain the formation of a single diastereomer of alkylation product, as shown in Scheme 18.⁴²



Figure 4 Chiral auxiliaries for isoquinoline alkylation

Virtually all of the synthetic applications of chiral dipole-stabilized organolithiums reported to date have the C—Li bond in an allylic or benzylic system. The most important consequence of this fact is that the two stereoisomers shown in Figure 3 *can interconvert* by pyramidal inversion. Therefore, the stere-oselectivity of the deprotonation is irrelevant to the stereoselectivity as manifested in the product diastereomer ratio.⁴³ The source of the selectivity in the organolithium alkylation step has not been determined. The available data do not permit a distinction between at least three possibilities: thermo-dynamic control as determined by the equilibrating organolithium diastereomers,^{43,44} kinetic control according to Curtin–Hammett kinetics,⁴⁵ or increased carbon–lithium covalency at low temperature.⁴⁶

The seminal paper on the stereoselective alkylation of chiral dipole-stabilized organolithiums appeared in 1983. In the preliminary communication⁴⁷ and the full paper which followed,⁴⁸ Meyers made the important observation that formamidines lacking an oxygen atom in the auxiliary afford poor selectivity. The mechanistic rationale originally provided⁴⁷ for this observation was later revised.^{49,50} The current picture can be explained by comparison of the structures illustrated in Figure 4. In (a), the dipole-stabilized organolithium is internally chelated by both the nitrogen and the oxygen of the formamidine, an effect which prevents rotation around the bond from the chelating formamidine nitrogen to the stereocenter (*).⁴⁷ In (b), this bidentate chelation is absent and free rotation diminishes the efficient transfer of stereochemical information and poor stereoselection ensues. In (c), free rotation is again precluded, this time by incorporation of the stereocenter into an oxazoline ring.^{42,43,51}

There is another important feature of these systems to keep in mind: the effect of alkylation temperature on the diastereomer ratio. The selectivity observed in the alkylation of the tetrahydroisoquinoline formamidine illustrated in Figure 4(a) (with methyl iodide) increases from 9:1 to 99:1 when the temperature is lowered from -78 to -100 °C.^{47,48} Similarly, the selectivity with the oxazoline illustrated in Figure 4(c) (with methyl iodide) increases from 9:1 at -78 °C to 19:1 at -100 °C.^{43,51}

One last effect is seen in alkylations of chiral isoquinoline derivatives: the effect of oxygen substitution on the aromatic nucleus.⁴³ When electron-donating oxygen substituents are present in the 6- and/or 7-positions of the tetrahydroisoquinoline, the stereoselectivity of the alkylations is increased. This may be interpreted as a decreased tendency of the benzylic carbanion to be delocalized into the ring, and a consequential increase in the importance of dipole stabilization, with its strict stereoelectronic requirements.

1.2.3.2 Acyclic Systems

Whereas acyclic stereoselection in enolate alkylations is now a relatively mature field,⁵² the stereoselective alkylation of acyclic dipole-stabilized systems is virtually undeveloped. The only report to date



i, BunLi, THF; ii, MeI

Alkylation of Carbon

originates from our own laboratories.⁵³ We made a comparison between two systems having the existing stereocenter on either end of the amide/amidine function (*cf.* Figure 3). The pertinent results are shown in Scheme 19. In the first instance, the stereocenter is in a chiral auxiliary appended to *N*-methylbenzyl-amine, and is a case of 1,5-asymmetric induction. Not surprisingly, the selectivity is poor (4:1), although the absolute configuration of the new stereocenter is the same as the well-studied⁴³ tetrahydroisoquino-lyloxazolines (*vide infra*). The second example moves the stereocenter to the other end of the amide system (1,3-asymmetric induction, *cf.* Figure 3), and the selectivity is improved dramatically to $50:1.5^3$

The second example has been used in a stereoselective synthesis of α -alkylbenzylamines, as shown by the example illustrated in Scheme 20.⁵³ Note that this sequence destroys the existing stereocenter. Because the original stereocenter is deliberately destroyed after the creation of a new one, the overall process constitutes a 'self-immolative chirality transfer'.⁵⁴



i, BuⁿLi, -78 °C; ii, MeI, -100 °C; iii, KOH, ether; iv, Pb(OAc)₄, CH₂Cl₂, MeOH; v, HCl, EtOH

Scheme 20

1.2.3.3 Heterocyclic Systems

The following discussion is organized according to the type of heterocycle being alkylated, as follows: dehydropyrrolines and isoindolines, followed by dehydropiperidines, tetrahydroisoquinolines and β -carbolines. As mentioned previously, the alkylation of saturated heterocycles (the acidic protons being neither allylic nor benzylic) is problematic. Either they fail to metalate when the nitrogen contains a



i, BunLi, -78 °C; ii, RX, -100 °C

Scheme 21

chiral formamidine auxiliary,⁵⁵ or they fail to undergo cross-coupling after metalation when the nitrogen contains a chiral oxazoline auxiliary.⁴² In the latter instance, SET processes predominate.

These shortcomings have been circumvented by using a heterocycle containing a double bond, as illustrated by the examples in Scheme 21. As was the case with the achiral systems (cf. Scheme 11), a mixture of regioisomers is obtained, but now the α -substitution product (10) predominates. The amount of γ -substitution product (11) produced depends on the ring size, and may be as high as 30% of the product mixture for the dehydropiperidines.⁵⁵ Nevertheless, the unwanted γ -isomer (11) is destroyed chemoselectively when the mixture is treated with hydrazine to remove the chiral auxiliary. Double bond reduction would then give the α -alkylated saturated heterocycle.

The asymmetric alkylation of Δ^2 -pyrroline has been used in the synthesis of anisomycin in 90% *ee* using a formamidine chiral auxiliary, as shown in Scheme 22.⁵⁶



i, Bu^tLi, THF, -78 °C; ii, ArCH₂Cl, -100 °C; iii, N₂H₄, AcOH, EtOH

Scheme 22

An oxazoline auxiliary has been used to alkylate stereoselectively and regioselectively isoindoline at the α - and α' -positions to produce the C₂-symmetric amine (12), as shown in Scheme 23.⁵⁷



i, BuⁿLi, -78 °C, THF; ii, BnCl; iii, N₂H₄, p-TsOH, EtOH, reflux

Scheme 23

Scheme 24 illustrates the use of a formamidine auxiliary for asymmetric alkylation of dehydropiperidine derivatives in the synthesis of metazocine⁵⁵ and dextrorphan⁵⁸ in 98% and 99% *ee*, respectively.

The enormous number of naturally occurring isoquinoline alkaloids, and their importance biologically,⁵⁹ has stimulated a great deal of effort in their synthesis. The asymmetric alkylation of the 1-position of the tetrahydroisoquinoline nucleus affords an extremely valuable method for the construction of



i, BunLi, -78 °C; ii, ArCH2Cl, -100 °C; iii, N2H4, AcOH

both simple and complex isoquinoline alkaloids by the simple retrosynthesis illustrated in Figure 5. The following discussion presents selected examples of the synthesis of simple isoquinoline alkaloids, followed by a few examples of the use of this method for the synthesis of key intermediates in the preparation of more complex systems.



Figure 5 Retrosynthesis of simple isoquinoline alkaloids

Perhaps the simplest of the isoquinoline alkaloids is salsolidine. It has been synthesized by asymmetric alkylation of 6,7-dimethoxytetrahydroisoquinoline using either a formamidine^{60,61} or oxazoline⁵¹ chiral auxiliary. Scheme 25 illustrates the recently published *Organic Syntheses* preparation of salsolidine on a 5 g scale.⁶² It is of interest to note that, in this and all subsequent examples of asymmetric alkylation of tetrahydroisoquinolines, formamidines derived from L-valine afford (1*S*)-tetrahydroisoquinolines, while oxazolines derived from L-valine afford the (1*R*)-enantiomer. The reason is simply the opposite orienta-



i, Bu'Li, THF, -78 °C; ii, MeI, -100 °C; iii, N₂H₄, AcOH, EtOH

tion of the isopropyl group in the two auxiliaries. In the lithiated formamidine, bidentate chelation by both heteroatoms places the isopropyl on the β -face, whereas it is on the α -face in the oxazolines.

The stereoselective alkylation of tetrahydroisoquinoline pivalamides containing a stereocenter at the 3position was reported in 1987 by Seebach.⁶³ The self-immolative chirality transfer process illustrated in Scheme 26 affords only one diastereomer. Oxidative decarboxylation and reduction afford the alkylation products in $\geq 95\%$ ee.



i, 2 ButLi, THF, -75 °C; ii, MeI; iii, anode, MeOH; iv, NaCNBH3; v, NaAlH4

Scheme 26

The alkylation of the tetrahydroisoquinoline nucleus, mediated by a formamidine auxiliary, with a dimethoxyphenethyl iodide has afforded (S)-homolaudanosine in 95–96% *ee*, as shown in Scheme 27.^{60,61}



Homolaudanosine; 95-96% ee

i, Bu^tLi, THF, -78 °C; ii, Ar(CH₂)₂I, -100 °C; iii, N₂H₄, AcOH; iv, HCO₂Et; v, LiAlH₄

Scheme 27

Reticuline, the biosynthetic precursor of all the morphine alkaloids, was synthesized by a similar approach $(97\% \ ee)$, as shown in Scheme 28.⁶⁴



i, Bu'Li, THF, -78 °C; ii, ArCH2Br, -100 °C; iii, N2H4, AcOH, EtOH; iv, ClCO2Et; v, LiAlH4; vi, H2, Pd/C

Scheme 28

Another benzylisoquinoline alkaloid, laudanosine, is the synthetic precursor of morphinans. It has been synthesized in 95-96% ee by asymmetric alkylation of an isoquinolyloxazoline, as shown in Scheme $29.^{43,65}$ An interesting point here is the discovery of a shortcut for removing the auxiliary and



i, Bu'Li, THF, -78 °C; ii, ArCH2Cl, -100 °C; iii, N2H4•H2O, p-TsOH, EtOH; iv, HCO2COMe; v, LiAlH4





Reframoline; 98-99% ee

O-Methylflavinantine; 94-96% ee

Figure 6 Complex isoquinoline alkaloids synthesized by elaboration of simple benzylsoquinolines. The bond(s) formed after the asymmetric alkylation are bold; the stereocenter created in the asymmetric alkylation is circled

methylating the nitrogen. Treatment of the alkylated isoquinolyloxazoline with acetic formic anhydride removed the oxazoline and formylated the nitrogen in a single operation.⁶⁵

Because benzylisoquinolines have been available synthetically (in racemic form) for decades, there is quite a bit of chemistry known regarding their use as key intermediates in the synthesis of a number of more complex isoquinoline alkaloids. The asymmetric synthesis of benzylisoquinolines has been used to complete total synthesis of representative members of several of these alkaloid classes. As shown in Figure 6, the protoberberine alkaloid norcoralydine,^{60,61} the aporphine alkaloid ocoteine,^{61,66} the isopavine alkaloid reframoline⁶¹ and the morphinan O-methylflavinantine⁶⁵ have been made available in optically active form for the first time (except by isolation or resolution) using this approach.

The asymmetric alkylation of the β -carboline ring system affords access to indole alkaloids. The simplest is tetrahydroharman, whose synthesis is mediated by asymmetric alkylation of a formamidine, as shown in Scheme 30.⁶⁷ In the racemic series (*cf.* Scheme 16), the indole nitrogen may be protected simply by deprotonation with potassium hydride. However, in the chiral series the presence of a potassium ion lowers the selectivity. Thus, the methoxymethyl protecting group was used.



i, BuⁿLi, THF, -78 °C; ii, MeI, -100 °C, iii, N₂H₄, AcOH; iv, H⁺

Scheme 30

The asymmetric alkylation of the β -carboline formamidine was also used in the asymmetric synthesis of deplancheine in 96% *ee* (Scheme 31), a synthesis which served to correct the previously assigned absolute configuration of the natural product.⁶⁸



Deplancheine; 96% ee

i, BuⁿLi, THF, -78 °C; ii, Br(CH₂)₃OBO; iii, N₂H₄, H⁺

Scheme 31

A similar approach was used for the asymmetric synthesis of yohimbone,⁴⁰ a process that uses the stereocenter created by asymmetric alkylation as the control element for the stereoselective formation of two more stereocenters, as shown in Scheme 32.



i, BuⁿLi, THF, -80 °C; ii, ArCH₂Cl, -80 °C; iii, N₂H₄, H⁺

Scheme 32

1.2.4 REFERENCES

- 1. P. Beak and D. B. Reitz, Chem. Rev., 1978, 78, 275.
- 2. P. Beak, W. J. Zajdel and D. B. Reitz, Chem. Rev., 1984, 84, 471.
- 3. D. Seebach, Angew. Chem., Int. Ed. Engl., 1979, 18, 239.
- 4. D. J. Peterson and H. R. Hays, J. Org. Chem., 1965, 30, 1939.
- 5. A. R. Lepley and W. A. Khan, J. Org. Chem., 1966, 31, 2061.
- 6. H. Ahlbrecht and H. Dollinger, Tetrahedron Lett., 1984, 25, 1353.
- 7. D. Seebach and D. Enders, Angew. Chem., Int. Ed. Engl., 1975, 14, 15.

- J. E. Saavedra, Org. Prep. Proced. Int., 1987, 19, 83.
 R. R. Fraser and L. K. Ng, J. Am. Chem. Soc., 1976, 98, 5895.
 R. R. Fraser, T. B. Grindley and S. Passannanti, Can. J. Chem., 1975, 53, 2473.
- 11. D. Seebach, W. Wykypiel, W. Lubosch and H.-O. Kalinowski, Helv. Chim. Acta, 1978, 61, 3100.
- 12. N. G. Rondan, K. N. Houk, P. Beak, W. J. Zajdel, J. Chandrasekhar and P. v. R. Schleyer, J. Org. Chem., 1981, **46**, 4108.
- 13. W. G. Kofron and M. K. Yeh, J. Org. Chem., 1976, 41, 439.
- 14. M. E. Jung, P. A. Blair and J. A. Lowe, Tetrahedron Lett., 1976, 1439.
- 15. R. D. Bach, M. L. Braden and G. J. Wolber, J. Org. Chem., 1983, 48, 1509.
- 16. L. J. Bartolotti and R. E. Gawley, J. Org. Chem., 1989, 54, 2180.
- 17. M. Al-Aseer, P. Beak, D. R. Hay, D. J. Kempf, S. Mills and S. G. Smith, J. Am. Chem. Soc., 1983, 105, 2080. 18. A. I. Meyers, W. F. Rieker and L. M. Fuentes, J. Am. Chem. Soc., 1983, 105, 2082.
- 19. J. J. Fitt and H. W. Gschwend, J. Org. Chem., 1984, 49, 209.
- 20. D. R. Hay, Z. Song, S. G. Smith and P. Beak, J. Am. Chem. Soc., 1988, 110, 8145.

- P. Beak and B. Lee, J. Org. Chem., 1989, 54, 458.
 P. Beak and W. J. Zajdel, J. Am. Chem. Soc., 1984, 106, 1010.
 A. I. Meyers and W. Ten Hoeve, J. Am. Chem. Soc., 1980, 102, 7125.
- A. I. Meyers and J. M. Marra, *Tetrahedron Lett.*, 1985, 26, 5863.
 A. I. Meyers, P. D. Edwards, W. F. Rieker and T. R. Bailey, J. Am. Chem. Soc., 1984, 106, 3270.
- 26. D. Lal, D. Griller, S. Husband and K. U. Ingold, J. Am. Chem. Soc., 1974, 96, 6355.
- 27. P. D. Edwards and A. I. Meyers, Tetrahedron Lett., 1984, 25, 939.
- 28. F. W. Eastwood, D. Gunawardana and G. T. Wernert, Aust. J. Chem., 1982, 35, 2289.
- 29. J. A. Monn and K. C. Rice, Tetrahedron Lett., 1989, 30, 911.
- 30. W. Wykypiel, J.-J. Lohmann and D. Seebach, Helv. Chim. Acta, 1981, 64, 1337.
- 31. J.-J. Lohmann, D. Seebach, M. A. Syfrig and M. Yoshifuji, Angew. Chem., Int. Ed. Engl., 1981, 20, 128.
- 32. D. Seebach and M. Yoshifuji, Helv. Chim. Acta, 1981, 64, 643.
- 33. D. Seebach, J.-J. Lohmann, M. A. Syfrig and M. Yoshifuji, Tetrahedron, 1983, 39, 1963.
- 34. A. I. Meyers, S. Hellring and W. Ten Hoeve, Tetrahedron Lett., 1981, 22, 5115.
- 35. A. R. Katritzky and K. Akutagawa, Tetrahedron, 1986, 42, 2571.
- 36. M. A. Gonzalez and A. I. Meyers, Tetrahedron Lett., 1989, 30, 43.
- 37. M. A. Gonzalez and A. I. Meyers, Tetrahedron Lett., 1989, 30, 47.
- 38. A. I. Meyers and S. Hellring, J. Org. Chem., 1982, 47, 2229. 39. A. I. Meyers and M. F. Loewe, Tetrahedron Lett., 1984, 25, 2641.
- 40. A. I. Meyers, D. B. Miller and F. H. White, J. Am. Chem. Soc., 1988, 110, 4778.
- 41. P. Beak and A. I. Meyers, Acc. Chem. Res., 1986, 19, 356.
- 42. R. E. Gawley, G. C. Hart and L. J. Bartolotti, J. Org. Chem., 1989, 54, 175.

- 43. K. Rein, M. Goicoechea-Pappas, T. V. Anklekar, G. C. Hart, G. A. Smith and R. E. Gawley, J. Am. Chem. Soc., 1989, 111, 2211.
- 44. R. E. Gawley, J. Am. Chem. Soc., 1987, 109, 1265.
- 45. J. I. Seeman, Chem. Rev., 1983, 83, 83.
- 46. J. Heinzer, J. F. M. Oth and D. Seebach, Helv. Chim. Acta, 1985, 68, 1848.
- 47. A. I. Meyers and L. M. Fuentes, J. Am. Chem. Soc., 1983, 105, 117. 48. A. I. Meyers, L. M. Fuentes and Y. Kubota, Tetrahedron, 1984, 40, 1361.
- 49. M. F. Loewe, M. Boes and A. I. Meyers, Tetrahedron Lett., 1985, 26, 3295.
- 50. A. I. Meyers and D. A. Dickman, J. Am. Chem. Soc., 1987, 109, 1263.
- 51. R. E. Gawley, G. C. Hart, M. Goicoechea-Pappas and A. L. Smith, J. Org. Chem., 1986, 51, 3076.
- 52. D. Caine, in 'Comprehensive Organic Synthesis', ed. B. M. Trost, Pergamon Press, Oxford, 1991, vol. 3, chap. 1.1.
- 53. R. E. Gawley, K. Rein and S. R. Chemburkar, J. Org. Chem., 1989, 54, 3002. For the alkylation of chromiumtricarbonyl complexes of N-methylbenzylamine formamidines, see J. Albert and S. G. Davies, Tetrahedron Lett., 1989, 43, 5945.
- 54. E. L. Eliel, in 'Asymmetric Synthesis', ed. J. D. Morrison, Academic Press, New York, 1983, vol. 2, chap. 5, p. 125; see footnote 1, p. 137.
- 55. A. I. Meyers, D. A. Dickman and T. R. Bailey, J. Am. Chem. Soc., 1985, 107, 7974.
- 56. A. I. Meyers and B. Dupre, Heterocycles, 1987, 25, 113.
- 57. R. E. Gawley, S. R. Chemburkar, A. L. Smith and T. V. Anklekar, J. Org. Chem., 1988, 53, 5381.
- 58. A. I. Meyers and T. R. Bailey, J. Org. Chem., 1986, 51, 872.
- 59. J. D. Phillipson, M. F. Roberts and M. H. Zenk (eds.), 'The Chemistry and Biology of Isoquinoline Alkaloids', Springer-Verlag, Berlin, 1985.
- 60. A. I. Meyers, M. Boes and D. A. Dickman, Angew. Chem., Int. Ed. Engl., 1984, 23, 458.
- 61. A. I. Meyers, D. A. Dickman and M. Boes, Tetrahedron, 1987, 43, 5095.
- 62. A. I. Meyers, M. Boes and D. A. Dickman, Org. Synth., 1988, 67, 60.
- 63. I. M. P. Huber and D. Seebach, Helv. Chim. Acta, 1987, 70, 1944.
- 64. A. I. Meyers and J. Guiles, Heterocycles, 1989, 28, 295.
- 65. R. E. Gawley and G. A. Smith, Tetrahedron Lett., 1988, 29, 301.
- 66. D. A. Dickman and A. I. Meyers, Tetrahedron Lett., 1986, 27, 1465.
- 67. M. F. Loewe and A. I. Meyers, Tetrahedron Lett., 1985, 26, 3291.
- 68. A. I. Meyers, T. Sohda and M. F. Loewe, J. Org. Chem., 1986, 51, 3108.

1.3 Alkylations of Sulfur- and Selenium-containing Carbanions

ALAIN KRIEF

Facultés Universitaire Notre Dame de la Paix, Namur, Belgium

1.3.1	INTRODUCTION	86
1.3.2	REACTIONS OF α-THIO- AND α-SELENO-ALKYLLITHIUMS BEARING HYDROGENAND/OR HYDROCARBON SIDE CHAINS ON THEIR CARBANIONIC CENTER	88
1.	3.2.1 Alkylation	88
	1.3.2.1.1 Alkylation of α -thio- and α -seleno-alkyllithiums	88
	1.3.2.1.2 Alkylation of α -thio- and α -seleno-benzylmetals	94
	132.13 Alkylation of α -thio- and α -seleno-allylimetals	95
	$1.32.14$ Alkylation of α -thio- and α -seleno-proparaylic metal derivatives	104
	1.3.2.1.5 Alkylation of α -metallovinyl sulfides and selenides	104
1.:	3.2.2 Synthetically Useful Transformations of Alkylated Sulfides and Selenides	106
	1.3.2.2.1 Synthetic transformations involving the reduction of sulfides and selenides	106
	1.3.2.2.2 Transformations of sulfides and selenides to alkenes	114
	1.3.2.2.3 Transformations of sulfides and selenides to organic halides: the haloalkylation and halopropenylation of alkyl and allyl halides	118
	1.3.2.2.4 Transformation of alkylated vinyl sulfides and selenides to carbonyl compounds	120
1.3.3	REACTIONS OF α -THIO- AND α -SELENO-ALKYLMETALS BEARING FURTHER HETEROATOMIC MOIETIES ON THEIR CARBANIONIC CENTER. APPLICATION TO THE FORMATION OF ACYL ANION EQUIVALENTS	121
1.3	3.3.1 Alkylations of 1-Metallo-1,1-dithioalkanes Bearing Hydrogen and/or Alkyl Groups on their Carbanionic Center and their Synthetic Applications	121
	1.3.3.1.1 Synthetic uses of 1-metallo-1.1-di(alkylthio)alkanes and -di(phenylthio)alkanes	123
	1.3.3.1.2 Synthetic uses of 2-metallo-1.3-dithianes and related derivatives	124
1.3	3.3.2 Alkylations of 1,1-(Dithio)allyl- and 1,1-(Dithio)propargyl-metals and their Synthetic Uses	131
1.3	3.3.3 Alkylations of Acyl Anion Equivalents Containing at Least One Sulfur or Selenium on their Carbanionic Center	134
	1.3.3.3.1 Alkylations of sulfur-containing acyl anion equivalents	134
	1.3.3.3.2 Alkylations of selenium-containing acyl anion equivalents	136
	1.3.3.3.3 Synthesis of aldehydes and ketones from α -heterosubstituted sulfides and selenides and related compounds	141
1.3.4	α -METALLOORTHOTHIO- AND α -METALLOORTHOSELENO-FORMATES, AS PRECURSORS OF ESTERS	144
1.3.5	REACTIONS INVOLVING α-METALLOALKYL SULFOXIDES AND SELENOXIDES	147
1.3	3.5.1 Reactions Involving Sulfarides	147
• ••	1.3.5.1.1 Reactions involving alkyl sulfoxides	147
	1.3.5.1.2 Reactions involving allylic sulfoxides	155
1.3	3.5.2 Alkylations of α -Metallo-alkyl and -vinyl Selenoxides	157
1.3.6	REACTIONS INVOLVING SULFONES	158
1.3	3.6.1 Reactions Involving a-Metalloalkyl Sulfones	158
	1.3.6.1.1 Alkylations of α -metalloalkyl sulfones	158
	1.3.6.1.2 Synthetic applications of alkyl sulfones	160
1.3	$3.6.2$ Reactions Involving α -Metalloallyl Sulfones	168
1.3	3.6.3 Reactions Involving α-Metallovinyl Sulfones	173

12	3.6.4 Reactions Involving α -Metallo- α -heterosubstituted Alkyl Sulfones	174
	1.3.6.4.1 Reactions involving 2-lithio-2-benzenesulfonyltetrahydropyrans	174
	13.6.4.2 Reactions involving α -isocyanoalkyl sulfones	175
	1.3.6.4.3 Reactions involving α -sulfinyl sulfones	176
1.3.7	ALKYLATION OF SULFUR YLIDES	178
1.3.8	ALKYLATION OF ALKYL SULFONATES, SULTONES AND SULFONAMIDES	179
1.3.9	REFERENCES	182

1.3.1 INTRODUCTION

 α -Thio organometallics¹⁻⁴⁴ and their seleno analogs^{1,3-6,10,11,17,18,23,25,45-56} have proved to be extremely useful reagents and they have been used extensively in synthesis over the past 20 years. Upon reaction with compounds bearing an electrophilic carbon atom, they allow the efficient formation of a new carbon-carbon bond to which the heteroatomic moiety is directly linked. The large variety of reactions possible through use of organosulfur and organoselenium reagents with subsequent selective removal of the heteroatomic moiety permits the synthesis of a large variety of compounds.

This chapter is devoted exclusively to the alkylation of the above mentioned organometallic reagents with alkyl halides, epoxides and oxetanes. It includes a large variety of organosulfur- and organoselenium-stabilized carbanions derived from saturated and unsaturated sulfides, selenides, sulfoxides, selenoxides and sulfones as well as those carbanions bearing another heteroatomic moiety. The chapter excludes, however, those organometallics which can be viewed as α -thio and α -seleno enolates.

Although α -thio and α -seleno organometallics can be prepared by a large array of reactions^{4,23,57} which are listed in Scheme 1 and involve: (i) hydrogen-metal exchange, (ii) halogen-metal exchange, (iii) addition of organometallics on α -heterosubstituted carbon-carbon double bonds, and (iv) heteroatom (Sn,Se)-metal exchange, the first reaction has proved to be by far the most convenient due to the availability of the starting material and of a large array of bases.



Scheme 1

Butyllithiums in THF (often without an additive, and in some cases with the help of a complexing agent such as N, N, N', N'-tetramethylethylenediamine (TMEDA),⁵⁸ diazabicyclooctane (DABCO),⁵⁹ hexamethylphosphoramide (HMPA),⁶⁰ and crown ethers) have been used successfully for the metallation of sulfides such as dimethyl sulfide,⁵⁸ methyl phenyl⁵⁹ and primary alkyl phenyl sulfides,^{60,61} cyclopropyl phenyl sulfide,^{17,25,61} as well as allyl,^{14,16,24} benzyl^{14,16} and vinyl sulfides,²⁰ thioacetals,^{20,26,27,30} sulfoxides,^{14,16,32,33} sulfones^{36,37} and sulfoximides (sulfoximines).⁶² However these bases are not able to metallate *s*-alkyl sulfides, due to their low acidity,⁶⁰ or sulfonium salts⁴¹ (except Bu¹Li in some cases) or almost all selenium derivatives^{23,25,52,54} due to their high propensity to react at the sulfur and selenium atoms respectively of these compounds. The last reaction which leads finally in the case of selenides to a butyl selenide and to a novel organometallic, the one possessing the most stabilized carbanionic center, (Scheme 2) is the method of choice^{23,25,46,52,54} for the synthesis of a large variety of α -selenoorganolithium compounds.



Thus, α -selenoalkyllithiums^{23,25,45,47,54,63–75} (see Volume 1, Chapter 2.6), α,α -di(seleno)alkyllithiums^{64,65,76–78} as well as related α -selenobenzyl-,^{72,79,80} α -selenoallyl-⁸¹ and α -selenovinyl-lithiums^{82,83} belonging to the phenyl- and the methyl-seleno series have been successfully synthesized from the corresponding selenoacetals,^{64,74,84–87} selenoorthoesters^{64,65,76} and keteneselenoacetals^{88–90} respectively using *n*-butyllithium in THF or *s*-butyllithium in ether. The Se/Li exchange method has also proved to be particularly efficient for the synthesis of α -thioalkyllithiums including those which possess dialkyl-substituted carbanionic centers from mixed S,Se acetals.^{68,69,73,91–93}

 α -Selenoallyl-,^{81,94-101} α -selenopropargyl-,^{102,103} α -selenoallenyl-¹⁰³ and α -selenobenzylmetals,^{74,80,104-106} 1,1-di(seleno)alkylmetals,^{64,74,76,107,108} 1,1-di(seleno)benzylmetals,⁸⁰ and α -selenovinylmetals¹⁰⁷⁻¹¹¹ have been produced by metallation of the corresponding carbon acid with metalloamides. Although lithium diethylamide, lithium diisobutyramide, lithium diisopropylamide (LDA) or lithium tetramethylpiperidide (LITMP) in THF or THF–HMPA have been used for that purpose, KDA^{80,108} (a 1:1 mixture of LDA and potassium *t*-butoxide) has proved to be by far the most suitable base for this purpose. However the metallation reaction is strictly limited (except rare cases) to the arylseleno series and does not apply to their methylseleno analogs; hence the method employing the Se/Li exchange is preferred. The metallation reaction is the only reasonable route to α -metalloalkyl selenoxides,^{112–115} α -metallovinyl selenoxides,¹⁰⁷ α -metalloalkyl sulfonium salts,⁴¹ α -metalloalkyl selenonium salts^{116,117} and α -metalloalkyl selenones.^{117–119}

Successful alkylation or hydroxyalkylation of the above mentioned species requires that these organometallics, which are potential carbenoids: (i) do not decompose to carbene or to an isomeric organometallic under the conditions required for their reactions with electrophiles; and (ii) that they act as nucleophiles rather than as bases. This has effectively proved to be the case for most of them.

As will be discussed later in this chapter, although the alkylation reaction is in almost all cases reasonably successful with methyl iodide and primary alkyl halides, ethylene oxide and terminal epoxides, it is



less general with s-alkyl halides and α , β -disubstituted epoxides and is exceptional with tertiary alkyl halides¹²⁰ (Scheme 3, entries a and c) and almost unprecedented with tetraalkyl-substituted oxiranes.

The alkylation reaction has proved to be more efficient with the softer organometallics of the series and proceeds *via* an ionic pathway, and *via* radical species with the most substituted alkyl halides.^{120,121} Although alkyl iodides are usually more reactive than the corresponding bromides and chlorides, the use of the former compounds does not seem to be crucial.

Allylation and benzylation reactions usually take place more efficiently than alkylation reactions due to the higher reactivity of the former derivatives. Allylation and benzylation is however difficult with the hardest organometallics of the series such as seleno- or thio-alkyllithiums. Finally, alkyl halides are usually less reactive towards α -thio- and α -seleno-alkyllithiums than aldehydes or ketones (see Scheme 11)^{25,122} which seem to possess a reactivity closer to that of allyl halides (see Scheme 12).¹²³

1.3.2 REACTIONS OF α-THIO- AND α-SELENO-ALKYLLITHIUMS BEARING HYDROGEN AND/OR HYDROCARBON SIDE CHAINS ON THEIR CARBANIONIC CENTER

1.3.2.1 Alkylation

1.3.2.1.1 Alkylation of α -thio- and α -seleno-alkyllithiums

 α -Thioalkyllithiums (even those derived from polymer-bound derivatives)¹²⁴ bearing hydrogen or alkyl groups on their carbanionic center have been alkylated efficiently in the presence of TMEDA,⁵⁸ DABCO^{59,124-126} or HMPA⁶⁰ (usually required in the metallation step) by primary,^{58,59,125} secondary⁵⁹ and allylic halides (Scheme 4, entries a–d; see Scheme 53, entry a). The alkylations are much less successful in the absence of HMPA^{125,127} but addition of copper(I) iodide (CuI)¹²⁵ or CuI-SMe₂ complex (Scheme 4, entry e), which probably leads to the formation of a cuprate intermediate, restores the reactivity. Under these conditions phenylthiomethyl carbanion reacts¹²⁵ in a regioselective manner with geranyl bromide and produces homogeranyl phenyl sulfide with preservation of the (*E*)-stereochemistry of the carbon–carbon double bond (see Scheme 53, entry b).¹²⁵

The latter conditions also proved to be much more successful than those involving lithium derivatives¹²⁸ for the methylation¹²⁹ and the allylation¹²⁷ of phenylthiocyclopropane (Scheme 5), but none of the above mentioned α -thio organometallics can be benzylated in reasonable yield since stilbene, rather than the expected benzylated sulfides, was produced quantitatively, probably *via* benzylidene formation (Scheme 4, entry e).⁶⁰

Use of α -thioalkylmetals whose carbanionic centers are dipole-stabilized, such as α' -lithioalkyl carbamates, ^{130,131} α' -lithioalkyl thioesters, ^{130,131} α -lithioalkyl thiothiazolines^{132,133} and related species¹³⁴ offer interesting advantages over the classical α -alkylthio- and α -phenylthio-alkyllithiums (Scheme 6, entries a–d) since not only is there a large variety of such species, including those bearing two alkyl substituents on the carbanionic center, available by metallation of the corresponding carbon acid (this is not for example feasible for 2-phenylthio-2-propyllithium⁶⁰) but also because they can be alkylated, ^{130–135} allylated^{130–133} and even benzylated^{131–134} in reasonably good yields (Scheme 6, entries a–d).

In fact all these organometallics can be viewed as chelated species which are softer than unchelated ones. In that respect the behavior of the methylthioamidines described in Scheme 6 (entries e and f) is interesting in that the anion derived from the seven-membered ring compound where negative charge can be stabilized by internal chelation is cleanly metallated and alkylated at the SMe group, whereas its lower homolog whose analogous carbanion is expected to be less prone to chelation leads instead to an alkylated enamine.¹³⁵

Haloalkyl phenyl sulfides produce cycloalkyl phenyl sulfides on treatment with base. The synthesis of cyclopropyl phenyl sulfides has attracted particular interest since these compounds can be metallated^{17,25,61} by BuⁿLi in THF and the resulting 1-phenylthiccyclopropyllithium has been used for spiroannelation of various cycloalkanones.^{17,25} Thus, 3-chloro-1-phenylthicpropane leads^{136,137} to phenylthiccyclopropane on reaction with potassium amide in liquid ammonia (Scheme 7, entry a), but attempts to prepare 2-methylcyclopropyl phenyl sulfide from 3-chloro-1-phenylthicbutane by an analogous route failed in the cyclization step.⁶¹ Neither 3-mesyloxy- and 3-tosyloxy-1-phenylthicbutane produce cyclopropane derivatives either on reaction with LDA in THF¹³⁹ (Scheme 7, entry b). Failure in these ring closure reactions has been attributed to inadequate car-



i, BuⁿLi; ii, CuI•SMe₂; iii, CH₂=CHCH₂Br, -78 to 0 °C; iv, MeSO₃F; v, KOH, DMSO; vi, LDMAN, THF, -45 °C; vii, CuI, -78 °C; viii, MeI

banion stabilization by the phenylthio group since closure of the corresponding sulfones (see Section 1.3.6.1.2, Scheme 109) proceeds easily under the same conditions.^{138,139} Addition of methyl iodide to the sulfide shown in Scheme 7 (entry b) gives the 4-ethylbenzenesulfonate, suggesting that a successful ring closure might be achieved if the benzenesulfonate was used instead of the tosylate. Indeed, with the corresponding benzenesulfonate (Scheme 7, entry c) a mixture of isomers (Z:E = 63:37) of 2-methylcy-clopropyl phenyl sulfide is obtained in high yield. Interestingly, the isomer distribution obtained complements that resulting from an alternative route involving 3-chloro-2-methyl-1-phenylthiopropane and potassium amide (Z:E = 17:83; Scheme 7, entry d).⁶¹

The alkylations of α -selenoalkylmetals share some similarities with those of their thio analogs. Thus α -methylseleno- and α -phenylseleno-alkyllithiums, available by cleavage of the corresponding selenoacetals by *n*-butyllithium in THF or *s*-butyllithium in ether, usually react with methyl iodide and primary alkyl halides to produce the corresponding alkylated selenides.^{23,25,52,68,70,71,110,122,127,140-145}

The alkylations of α -selenoalkyllithiums, bearing one or two alkyl substituents on the carbanionic center, with primary alkyl halides take place at low temperatures (-78 °C) in THF^{68,143} but at much higher

Alkylation of Carbon



Scheme 6

temperatures (20 °C for 1 h) when performed in ether. Best results are obtained by use of methylseleno derivatives and when the reaction is performed in THF at the lowest possible temperature (Scheme 8).

The parent compounds, methylselenomethyllithium and phenylselenomethyllithium, are less reactive towards alkylating agents than their higher homologs. They require higher temperatures (-20 °C instead of -78 °C) for a reasonable reaction rate irrespective of whether ether or THF is used.¹⁴¹ Unfortunately they produce, along with the expected alkylated selenides, appreciable amounts of the selenide resulting from substitution of the alkyl halide by the selenolate ion (Scheme 8, entry a). Apparently, under these conditions, the selenoalkyllithiums are in equilibrium with methylene and lithium selenolates. This side reaction can be avoided if the reactions are performed in THF-HMPA,¹⁴⁰ under which conditions alkylation proceeds at temperatures as low as -78 °C (Scheme 8, compare entry b to a). In several cases HMPA aids the alkylation of α -selenoalkyllithiums. For example HMPA has been used¹⁴⁵ for the alkylation of 1-phenylseleno-1-hexyllithium, and also for 1-selenocyclopropyl-^{25,70} and 1-selenocyclobutyl-lithiums^{25,71} (Scheme 8, entries c and e).



 α -Selenoalkyllithiums cannot be alkylated with secondary alkyl halides¹⁴¹ (Scheme 9); instead either an alkene resulting from an elimination reaction or a selenide, derived from substitution of the halide by the selenolate ions formed as mentioned earlier, is produced, depending upon the conditions employed (THF or THF-HMPA; Scheme 9).¹⁴¹

Allylation of α -selenoalkylmetals takes place¹⁴⁰ rather sluggishly and leads to homoallyl selenides in modest yields (Scheme 10, entry a). However the yields are greatly improved by using instead^{123,127} the corresponding cuprate¹⁴⁶ (formed from the organolithium compound and copper iodide-dimethyl sulfide complex at -110 °C; Scheme 10, entries b and c). The reaction must be performed at low temperatures since the novel species decomposes around -30 °C, and instead produces an alkene resulting from coupling of the hydrocarbon framework of two units. When applied to neryl and geranyl halides the reaction occurs regioselectively at their primary carbon and does not affect the original stereochemistry of the carbon-carbon double bond.¹⁴⁷ Unfortunately benzylation of α -selenoalkyllithiums could not be achieved even by use of the corresponding cuprates.¹⁴⁷

As far as chemoselectivity is concerned, α -selenoalkyllithiums react exclusively with the carbonyl carbon of aldehydes or ketones rather than with alkyl halides (Scheme 11).^{25,122}

Allyl bromide and benzaldehyde appear to be equally reactive towards α -selenoalkyllithiums (Scheme 12, entries a and c), whereas α -selenoalkylcuprates exhibit a different behavior with allylation occurring almost exclusively, even when the reaction is performed in the presence of benzaldehyde (Scheme 12, entries b and d).¹²³

The alkylation of a highly functionalized α -selenoalkyllithium has been reported once. In this study the di(methylseleno)acetal resulting from the conjugate addition of 1,1-di(methylseleno)-1-ethyllithium to cyclohexenone, followed by trapping the resulting enolate with trimethylsilyl chloride, led to the corresponding methylated selenide on cleavage of one of the C—Se bonds followed by reaction with methyl iodide (Scheme 8, entry f).¹⁴⁸

 α -Selenoalkyllithiums also react with terminal epoxides^{71,141,149,150} and oxetanes^{141,149} to provide γ and δ -hydroxy selenides in good yields (Scheme 13). The reactions are usually carried out in THF, except in the case of selenomethyllithiums which require the use of HMPA as the cosolvent in order to prevent formation of a by-product resulting from opening of the heterocyclic ring by selenolate ions. Similar observations have already been made for the reaction of alkyl halides (see above).^{23,52,141}

(a)
(b)

$$RSe SeR \xrightarrow{i} [Li SeR] \xrightarrow{ii \text{ or iii}} C_{10}H_{21} SeR + C_{10}H_{21}SeR$$

(ii: R = Ph, 100% (41:59); R = Me, 89% (54:46) (ref. 122)
(iii: R = Ph, 87% (97:3); R = Me, 92% (98:2) (ref. 23)















i, addition of electrophile to organolithium at -100 °C, then -100 to 20 °C; ii, addition of CuI•SMe₂ to organolithium at -78 °C, addition of electrophile, then -78 to -30 °C

52

trace

ii

Bu^t

Me

(d)

Scheme 12

93



R = Ph; R¹ = H; n = 0; R² = C₆H₁₃; solvent = THF- HMPA; 82% R = Ph, Me; R¹ = Me, Et, PhCH₂, C₆H₁₃; n = 0, 1; R² = H, C₆H₁₃; solvent = THF-hexane; 70-75%



i, MeI, NaI, DMF, CaCO₃, 80 °C, 1 h; ii, Bu^tOK, DMSO, 20 °C; iii, Br₂, EtOH-H₂O, 20 °C; iv, MeMgBr, HMPT, 80 °C; v, Br₂, EtOH-H₂O, 20 °C, 3 h; vi, MeI; vii, Bu^tOK, DMSO, 20 °C; viii, CrO₃, H₂SO₄, acetone, 0 °C, 3 h, then 25 °C, 29 h

Scheme 13

1.3.2.1.2 Alkylation of α -thio- and α -seleno-benzyl metals

 α -Thio-¹⁴ and α -seleno-benzylmetals^{17,18,23,25,45,32–56} are more accessible than their alkyl counterparts, probably due to the extra stabilization afforded by the aromatic ring. Also they are usually more nucleophilic towards organo halide derivatives presumably due to their relative softness. α -Thiobenzyllithiums are however prone to rearrange to benzylthiolates *via* a Wittig type rearrangement.^{14,23}

A series of benzyllithiums bearing a phenylthio (Scheme 14, entries a and d, Scheme 15; see Scheme 43),^{14,126,151-155} a thiothiazolino (Scheme 14, entry b)^{132a,133} or a dithiocarbamato^{156,157} group (Scheme 14, entry c; see Scheme 49 and Scheme 50) have been successfully alkylated with primary¹⁵¹⁻¹⁵⁶ and secondary alkyl halides,¹²⁶ benzylated¹⁵⁷ or allylated (see Scheme 43)^{151,153,156,157} without special requirements. Intramolecular alkylation has allowed the synthesis of cyclic compounds, including aromatic lactones of medium and large ring size (14- and 15- but not 12-membered rings)¹⁵³ and cyclopropane derivatives.¹³⁷ Furthermore, benzyl 1-(3-chloropropyl) sulfide cyclizes¹³⁷ on reaction with KNH₂ to 2-phenyltetrahydrothiophene rather than to benzylthiocyclopropane (Scheme 14, entry e).



iv, BuⁿLi, THF, -60 °C, 0.1 h; v, MeI; vi, KNH₂, NH₃-Et₂O

Scheme 14

The case of thiobenzaldehyde dianions, which are available from benzylthiol and two equivalents of *n*-butyllithium–TMEDA complex, merits further comment since alkylation or allylation of these species occurs selectively at the carbanionic site and then a second alkylation can be carried out on the thiolate group (Scheme 16, entry a).^{158,178}

 α -Selenobenzyllithiums, even those bearing a bulky alkyl group at their carbanionic center, react efficiently not only with primary and secondary alkyl halides^{72,74,79,80,104–106} but also with allyl halides,¹⁵⁹ benzyl halides (Scheme 17)^{105,159} and epoxides (Scheme 18)^{106,160} without requiring an additive or the formation of a cuprate intermediate (see above).

1.3.2.1.3 Alkylation of α -thio- and α -seleno-allylmetals

 α -Thio-^{14,16,24} and α -seleno-allylmetals^{17,18,23,46,52,54} display a reactivity profile similar to their benzylic analogs, but they also exhibit a tendency to act as ambident nucleophiles.¹⁴ They can react at either the carbon directly linked to (α -carbon) or that which is two carbons away (γ -carbon) from the



i, 2 BuⁿLi, THF-TMEDA, -80 °C; ii, 2 BuⁿLi, THF, 2 TMEDA

heteroatom itself. The regioselectivity is in general highly dependent upon the location of the metal on the system.

Alkyl,^{151,161,162} aryl (Scheme 19; Scheme 20)^{16,151,161,164,165} and vinyl thioallyllithiums (Scheme 21)^{166,167} react selectively with organic halides at their α -site (α : γ ratio = 95:5 to 80:20) when the reaction is carried out in THF without any additive such as TMEDA, DABCO or HMPA. In contrast to the above mentioned cases, thioallyllithiums in which there is the possibility of lithium chelation at the α -site [such as 2-pyridylthio (Scheme 19; see Scheme 45),^{15,16,168,169} thioimidazoyl (Scheme 19; see



i, MeSeH, TiCl₄; ii, BuⁿLi, THF, -78 °C; iii, PrⁱI, THF, -78 to 0 °C; iv, LDA, THF, -78 °C; v, H₂O₂, THF, 20 °C; vi, Bu^sBr; vii, O₃



The stereochemistries of the products have not yet been established

i, separation of stereoisomers on silica; ii, BuⁿLi; iii, TosCl; iv, BuⁿLi, -78 °C

Scheme 18

Scheme 51)^{16,170,171} and dithiocarbamato (see Scheme 50 and Scheme 51)^{16,156,172–177} moieties] exhibit a dramatic selectivity for α -alkylation (Scheme 19).

The N-methylthioimidazoyl moiety, among the various aromatic and heteroaromatic moieties used in the sulfur series, proved to be the one with the highest propensity to direct the alkylation of the corresponding organometallics (even those bearing an alkyl group in the α -position) to the α -carbon atom (Scheme 19, entry e).^{16,170}

On the other hand, doubly metallated 2-propanethiol (thioacrolein dianion) reacts preferentially (γ : α ratio \approx 70:30, Scheme 16, entries b and d) at its γ -position^{158,178} with alkyl and benzyl chlorides and bromides and with epoxides. The *cis* stereochemistry found in the γ -adduct suggests that the organometallic is internally chelated as shown in Scheme 16. The nature of the metal (Li or K) does not affect the percentages of the alkylated vinyl sulfides,¹⁵⁸ whereas addition of HMPA to the dilithio derivative inverts the regioisomeric ratio obtained (Scheme 16, entry c, compare to entries b and d).¹⁵⁸

There is some evidence that factors other than chelating capability operate in altering the regioselectivity of alkylation. For example a higher α : γ ratio is observed¹⁶ when phenylthio is replaced by a 4-pyridylthio moiety, neither of which can participate in intramolecular chelation. Consequently an explanation



Scheme 19



(c) H DABCO MeI 92 98:2 (ref. l (d) Me None MeI 100 95:5 (ref. l (e) Me DABCO MeI 100 99:1 (ref. l (f) Me HMPA MeI 100 98:2 (ref. l (g) Me [2.2.2] crown ether MeI 80 60:40 (ref. l (h) Me DABCO \checkmark Br 100 87:12 (ref. l		R	Additive	R'X	Yield (%)	Product ratio	
(d) Me None MeI 100 95:5 (ref. 1 (e) Me DABCO MeI 100 99:1 (ref. 1 (f) Me HMPA MeI 100 98:2 (ref. 1 (g) Me [2.2.2] crown ether MeI 80 60:40 (ref. 1 (h) Me DABCO \swarrow Br 100 87:12 (ref. 1	(c)	н	DABCO	MeI	92	98:2	(ref. 161)
(e) Me DABCO MeI 100 99:1 (ref. 1 (f) Me HMPA MeI 100 98:2 (ref. 1 (g) Me [2.2.2] crown ether MeI 80 60:40 (ref. 1 (h) Me DABCO \sim Br 100 87:12 (ref. 1	(d)	Me	None	MeI	100	95:5	(ref. 164)
(f) Me HMPA MeI 100 98:2 (ref. 1 (g) Me [2.2.2] crown ether MeI 80 60:40 (ref. 1 (h) Me DABCO \sim Br 100 87:12 (ref. 1	(e)	Me	DABCO	MeI	100	99 :1	(ref. 164)
(g) Me [2.2.2] crown ether MeI 80 60:40 (ref. 1) (h) Me DABCO \sim Br 100 87:12 (ref. 1) (i) Me DABCO \sim Br 100 87:12 (ref. 1)	(f)	Me	HMPA	MeI	100	98:2	(ref. 164)
(h) Me DABCO \longrightarrow Br 100 87:12 (ref. 1)	(g)	Me	[2.2.2] crown ether	MeI	80	60:40	(ref. 164)
	(h)	Me	DABCO	<i>∕∕∕</i> ^{Br}	100	87:12	(ref. 164)
(1) Me [2.2.2] crown ether 60 50:50 (ref. 1)	(i)	Me	[2.2.2] crown ether	<i>▶</i> Br	6 0	50:50	(ref. 164)



i, Bu^sLi, THF, -78 °C; ii, DME/H₂O (3:1), CaCO₃, reflux 12 h; iii, DME/H₂O (3:1), reflux 12 h

that attributes high alkylation regioselectivity exclusively to intramolecular chelation must be considered tenuous.¹⁶ Furthermore benzyl allyl sulfide is predominantly metallated¹⁷⁹ at the allylic site with *n*-butyllithium–TMEDA complex and provides almost exclusively the α -methylated homolog (Scheme 22, entry a). Surprisingly however the α : γ ratio of alkylation of the related dilithio salt derived from allyl carboxymethyl sulfide, shown in Scheme 22 (entry b) depends upon the nature of the ester.¹⁸⁰

There are several examples where the α : γ ratio of alkylation depends to some extent on the nature of the halide and of the organic residue.^{164,181} In that respect, the softer the electrophile, the higher the percentage of γ -attack observed. Typical examples are displayed in Scheme 23.

In several cases the conditions used are crucial for control of the regiochemistry. The highest percentage of α -attack has been achieved in THF, with *s*-butyllithium being used as the base (Scheme 19; Scheme 20, entry a).^{16,166,181} The presence of an additive such as DABCO,¹⁵¹ TMEDA, crown ethers or HMPA, required in some cases for successful metallation of the allyl sulfide has a detrimental effect on the regiochemical control (Scheme 16; Scheme 19; Scheme 20).^{16,151,158,164,166,178} This result tends to support the theory that 'internal chelation' may play a role in defining regiochemistry. Such cation-coordinating cosolvents appear to partially negate the directive effect: (i) of the heterocyclic ligand (Scheme 24),¹⁶ or (ii) of the internal chelation in dilithiothioacrolein (Scheme 16)^{158,178} or (iii) to disrupt, in the case of phenylthioallyllithiums^{151,164,166} the intimate and/or solvated ion pairs which are expected to react at the α -position to finally produce solvent-separated ion pairs or free ions more inclined to react unselectively at the α - and γ -sites (Scheme 19; Scheme 20).

Phenylthioallylpotassium and phenylthioprenylpotassium in petroleum ether exhibit a higher tendency for methylation at the γ -site compared to their lithium analogs, and addition of THF to the medium tends to increase the γ -selectivity (Scheme 24).¹⁸²

Reactions between α -thioallyllithiums and allyl halides, which usually proceed predominantly *via* $\alpha - \alpha'$ (head-to-head) coupling, have proved to be valuable in synthesis.^{132a,133,165,183–192} They have been used successfully for example in the synthesis of 1,5-dienes such as squalene,^{132a,133,165} (*R*)-(+)-10,11-epoxyfarnesol,¹⁸⁴ (*R*)-(+)- and (*S*)-(-)-2,3-oxidosqualene¹⁸⁴ and mokupalide,¹⁸⁸ dendrolasin,¹⁸⁷ cembranolides,¹⁸⁹ methyl ceriferate¹⁹⁰ and *Cecropia* juvenile hormone (Scheme 25 and Scheme 26).^{191,192}

However, allylation of alkylthioallylcopper compounds, generated in ether from alkylthioallyllithium and CuI at -78 °C (the synthesis of phenylthio analogs fails¹⁶²), affords exclusively^{98,162,163} $\gamma - \gamma'$ (tail-to-tail) coupled products (Scheme 27), whereas alkylthio allylic aluminum and boron 'ate' complexes, readily available from reaction of alkylthioallyllithium compounds with trialkylaluminum⁹⁸ and


i, LDA; ii, Bu^sLi; iii, C₅H₁₁Br; iv, H₃O⁺; v, LDA, -78 to 0 °C; vi, MeI, 0 °C; vii, MCPBA; viii, CaCO₃, benzene, Δ , 24 h

Scheme 22

trialkylboranes respectively, undergo regiocontrolled $\alpha - \gamma'$ (head-to-tail) coupling with allylic halides¹⁸³ (Scheme 28). In general, both the regioselectivity and the yields are higher with Buⁿ-9-BBN-ate complexes than with triethylaluminum⁹⁸ or tri-*n*-butylboronates,^{98,183} and allylic bromides give better results^{98,183} than the corresponding chlorides.

Reactions between α -thicallyllithiums and epoxides have been used in a large number of important syntheses.^{14,193–202} In one example involving an epoxy alcohol,²⁰² the concomitant use of magnesium



i, BuⁿLi, THF, -78 °C; ii, BuⁿLi, THF, -20 °C, RBr, -78 to 0 °C; iii, Li, H₂NEt, -30 °C; iv, Raney Ni, EtOH-THF; v, Li, H₂NEt

Scheme 25



xv, TosCl, DMAP, 0 °C, 3 h; xvi, LDA, THF-hexane, -78 °C; xvii, MeOH, -78 °C; xviii, Li, Et₂NH

Scheme 25 (continued)



alkoxide in HMPA increased both the rate of the reaction and its yield. Most of these reactions involve intramolecular reactions and allow the synthesis of macrocyclic terpenoids from trisubstituted epoxides.^{193-195,198,200,201} Racemic 6,7-epoxygeranyl *t*-butyl sulfide on reaction with BuⁿLi–TMEDA produces two cyclobutyl carbinols possessing the fragranol and grandisol carbon skeletons and a single cyclopentanol of undefined stereochemistry.¹⁹⁸ The latter compound arises from the reaction of the α -lithio allyl sulfide at its γ -site, although it is fully substituted (Scheme 29). Higher homologs react from the less hindered site and provide 10-membered,^{194,195,200} 12-membered²⁰¹ and 14-membered¹⁹³ ring terpenoids (Scheme 30).

Several functionalized α -metalloallyl sulfides have been successfully alkylated or hydroxyalkylated on reaction with organic halides or epoxides. Among them, 1-metallo-1,3-di(thio)propenes²⁰³⁻²⁰⁵ which play the role of β -metallo- α , β -unsaturated aldehydes (--CH=C--CHO; Scheme 31) have proved^{203,206,207} particularly useful for the synthesis of δ -alkoxy- α , β -unsaturated aldehydes; they have been used in^{203,206} the neat synthesis of prostaglandin F_{2 α} from a functionalized oxidocyclopentene derivative (Scheme 31, entry e).

Similar transformations can be achieved from γ -methylthioallyl thiocarbamates¹⁷⁶ or from allyl thiocarbamates¹⁷⁶ (Scheme 32, entries c and d). Both reaction sequences take advantage of the easy sigmatropic rearrangement of allyl thiocarbamates. Furthermore, β -methoxyallyl thiocarbamates have played the role of an acetone enolate (Scheme 32, entry a).¹⁷⁵

As already mentioned, α -metalloallyl vinyl sulfides¹⁶⁶ and their 2-ethoxy analogs¹⁶⁷ have been selectively alkylated, benzylated and allylated at their α -site, then the resulting compounds have been thermally rearranged to γ , δ -unsaturated aldehydes or γ -oxo aldehydes on heating in aqueous DME (Scheme 21).



Scheme 27

Relatively few α -selenoallylmetals have been prepared^{81,95,96,99,101,208,209} compared to their thio analogs (Scheme 33). These species have proved to be powerful nucleophiles which react efficiently with primary,^{81,95,100,101,208,209} secondary^{95,100} and benzylic bromides^{95,208} and with *N*-(bromomethyl)phthalimide–ZnCl₂,⁹⁶ as well as with terminal and α , β -dialkyl-substituted epoxides^{95,100} (Scheme 33). The reactions take place almost instantaneously (<0.2 h) at low temperature (-78 °C) with primary alkyl halides⁹⁵ but require higher temperatures and/or longer reaction times with secondary halides. As for the sulfur analogs α -alkylation prevails (α : γ ratio from 80:20 to 90:10) with allyl phenyl selenides but, in contrast to the 2-thiopyridyl derivative,^{16,168,169} 2-pyridylselenoallyllithium⁹⁵ did not display efficient enhancement of alkylation from the α -site. This has been associated with the presence of diisopropylamine resulting from the metallation of the selenide with LDA (BuⁿLi cannot be used since it leads to allyllithium by Se/Li instead of H/Li exchange)¹⁰¹ which might prevent⁹⁵ lithium chelation with the pyridine nitrogen. Alkylation takes place selectively at the α -position of ω -alkoxy-,⁹⁹ γ -chloro-,⁹⁵ γ -seleno-¹⁰⁰ and γ -seleno, γ -silyl-substituted α -selenoallyllithiums (Scheme 33).¹⁰⁰

1.3.2.1.4 Alkylation of α -thio-^{21,31} and α -seleno-propargylic metal derivatives

Propargylic sulfides^{210,211} and selenides have been metallated^{102,103} with *n*-butyllithium and lithium amide respectively, and allylated²¹¹ or alkylated selectively at the propargylic position (Scheme 34) with primary and secondary alkyl halides.^{102,103,210}

Phenyl propargyl sulfide²¹¹ and selenide^{102,103} are rapidly deprotonated by 2 equiv. of base. The resulting dilithio derivatives react at their allylic rather than at their alkynic carbon centers (Scheme 34, entries c and d). Reduction of the sulfides allows the synthesis of 1,5-enynes (Scheme 34, entry b), whereas oxidation of the selenides leads to α -phenylseleno- α , β -unsaturated carbonyl compounds (Scheme 34, entry d).^{102,103}

1.3.2.1.5 Alkylation of α -metallovinyl sulfides and selenides

Vinyl sulfides²⁰ and selenides can be alkylated efficiently at the α -position in a two step-one pot reaction which involves their α -metallation and further reaction of the resulting anion with alkyl halides.



Side reactions which involve Michael type addition of the metallating agent at the terminus of the alkenic double bond^{27,110,145,212–214} or cleavage of the C—Se bond^{110,145} can be avoided by employing low temperatures and using nonnucleophilic metallating agents in the case of sulfides, and avoiding the use of alkyllithiums for vinyl selenides.

Thus, $aryl^{107,131,166,215}$ and $alkyl^{216}$ vinyl sulfides have been transformed to the corresponding 1-metallovinyl sulfides with LDA in THF alone,^{107,215} or in THF–HMPA,¹⁰⁷ and also with Bu^sLi in THF¹³¹ or in THF–HMPA^{131,166,216} (Scheme 35, entries a–c), and have been subsequentely alkylated with primary alkyl halides and α,ω -dihalides (Scheme 35). However it has not proved possible to extend the reaction to 1-(2-methylpropenyl) phenyl sulfide.

The corresponding 1-(lithiovinyl) sulfides and related derivatives can be produced in THF alone by reaction of ketene thioacetals with lithium naphthalenide (Scheme 36, entry b).^{129,217} This reductive lithiation is however less preferable to deprotonation when alkylation of the anion is planned, since the thiophenoxide anion which is necessarily present, destroys one equivalent of the alkylating agent.²¹⁷ Alternatively 1-metallovinyl sulfides, which have been prepared by Sn/Li exchange⁸² from 1-(stannyl)vinyl phenyl sulfide or by thiophilic addition²¹⁸ of phenyllithium on thioketones, have been methylated with methyl iodide (Scheme 35, entry d).^{82,218}

The alkylations of 2-ethoxy-²¹⁹ and 2-(Z)-alkylthio-1-(alkyl or phenylthio)ethylenes²²⁰ with primary alkyl halides and ethylene oxide have been reported (Scheme 37). The metallations are usually easy, but the alkylation of the former organometallic can be achieved only with primary alkyl halides and on condition that HMPA is used as a cosolvent. Unfortunately the reaction does not take place with benzyl bromide or isopropyl iodide.²¹⁹

1,3-Dienyl sulfides,^{213,221} which have been otherwise efficiently alkylated¹²⁰ at the allylic position using KNH₂ or LDA as the base (Scheme 3, entries a and c) or have added *n*-butyllithium at their δ -site (Scheme 38, entry a),²¹³ have been selectively alkylated at their α -vinylic site by use of a strong metallating base such as BuⁿLi–TMEDA (Scheme 3, entry b) or BuⁿLi–Bu^tOK (Scheme 38, entry b).²¹³



i, BuⁿLi-TMEDA, THF--hexane, -75 °C, 3 h; 5 °C, 2 h

Metallation of various 1-alkynyl sulfides,^{222–224} followed by alkylation of the resulting anion leads exclusively to α -alkylated allenyl sulfides (Scheme 39).

The alkylation of aryl vinyl selenides has been carried out by the usual metallation/alkylation sequence.^{107,108,110,111} Although the metallation of phenyl and *m*-CF₃C₆H₄ vinyl selenide can be achieved with LDA-HMPT,¹¹⁰ or LDA alone,¹⁰⁷ best results have been obtained with KDA in THF in the case of phenyl vinyl selenide,¹⁰⁸ and with LDA for pyridyl vinyl selenide (Scheme 40).¹¹¹ The latter combinations proved valuable for the synthesis of higher homologs bearing one alkyl group in the β -position,^{108,111} especially because metallation and alkylation both occur stereoselectively with retention of configuration. Unfortunately, however, alkylation of those aryl vinyl selenides bearing two alkyl groups in the β -positions has not yet been achieved.^{107,108}

Alternatively, 1-lithio-1-alkenyl phenyl and methyl selenides have been obtained on reduction of the C—Se bond of 1,1-di(seleno)alkenes with alkyllithiums; this reaction is the only one which allows the synthesis of the methylseleno derivatives (Scheme 41).⁸³

Furthermore, allenyl phenyl selenides have been prepared by alkylation of 1-lithioallenyl phenyl selenide, itself produced¹⁰³ on metallation of phenyl 1-propynyl selenide or of allenyl phenyl selenide¹⁰³ (Scheme 42). 1-Lithioallenyl phenyl selenide has poorer nucleophilic properties¹⁰³ than that of α -selenoallyllithiums, since the reaction of the former with 2-phenyl-1-bromoethane gives predominantly styrene, whereas the latter leads to products of substitution in good yield.¹⁰³

1.3.2.2 Synthetically Useful Transformations of Alkylated Sulfides and Selenides

1.3.2.2.1 Synthetic transformations involving the reduction of sulfides and selenides

Alkyl, benzyl, allyl and propargyl sulfides and selenides arising from the alkylation of the corresponding heterosubstituted organometallics have been used in several interesting transformations. These transformations take advantage of the easy reduction of the C—S and of the C—Se bond which can be carried



i, BuⁿLi, THF, -78 to 20 °C; ii, POCl₃, pyridine; iii, H₂, Rh(PPh₃)₃Cl; iv, NaIO₄; v, Et₂NH; vi, Li, Et₂NH, -78 °C

out in both cases with Raney nickel (Scheme 25; Scheme 43; Scheme 44, entry a),^{132a,133,140,153} lithium in ethylamine (Scheme 25, entries a, c and d; Scheme 26; Scheme 34, entry b; Scheme 43),^{140,165,186,187,189,191,192,211} nickel boride (Scheme 25),¹⁸⁸ or lithium naphthalenide (Scheme 15).¹²⁶ Tributyl- or triphenyl-tin hydrides¹⁴⁴ for their part proved particularly efficient for the reduction of selenides (Scheme 44, entry b).^{225,226} Care must be taken in performing such reactions however in the presence of C---C double bonds, to prevent the formation of a five- or a six-membered ring by cyclization of the radical intermediate formed.²²⁷⁻²²⁹

Selective reduction of allyl sulfides implies that no scrambling of the carbon–carbon double bond occurs during the process. Effectively this has proved to be the case especially when lithium in ethylamine is used, and the method has allowed the regio- and stereo-selective synthesis of a large variety of 1,5-dienes including squalene¹⁶⁵ (Scheme 25, entry a),¹⁶⁵ mukapolide (Scheme 25, entry b),¹⁸⁸ dendrolasin (Scheme 25, entry c), the basic nucleus of crassin acetate (Scheme 25, entry d)¹⁸⁹ from γ , γ -dialkylallyl sulfides and allyl halides, and also of 1,5-enynes²¹¹ from propargyl sulfides and allyl halides (Scheme 34, entry b).

Two original syntheses of C_{18} Cecropia juvenile hormones are depicted in Scheme 26,^{191,192} both of which use the strategy presented above. The second synthesis is particularly ingenious since the dihydro-



thiopyran building block allows both the regioselective construction of the polyenic system and the control of the stereochemistry of the (Z)-trisubstituted carbon-carbon double bonds. Reduction of benzyl and allyl sulfides, including allyl 2-pyridyl sulfides has also been performed^{168,169} with CuCl₂-LiAlH₄, in some cases in the presence of lithium methoxide.¹⁶⁹ Sequential alkylation of allyl 2-pyridyl sulfide and reduction with CuCl₂-LiAlH₄ proceeds regioselectively and leads^{15,168} to terminal as well as trisubstituted alkenes (Scheme 45).¹⁶⁹ Under similar conditions γ,γ -dimethylallyl 2-pyridyl sulfide is alkylated or benzylated selectively in the α -position, but finally leads to a mixture of alkenes resulting from the partial scrambling of the carbon-carbon double bond. This set of reactions has been applied to the synthesis of *cis*-bergamotene (Scheme 45, entry b).¹⁶⁹

Otherwise, reduction of allylic dithiocarbamates proved regio- and stereo-selective when carried out with lithium in ethylamine (Scheme 32),¹⁷⁷ but leads to a mixture of alkenic compounds when carried out with Raney nickel.¹⁷⁴

108



i, LDA; ii, R¹X; iii, H₃O⁺, MeOH; iv, Raney Ni, acetone, reflux, 3 h; v, LDA; vi, R¹X; vii, Δ , 3 h; viii, LDA; ix, MeSSMe; x, Hg⁺, aq. MeCN; xi, LDA; xii, MeSSMe; xiii, Δ ; xiv, LDA; xv, R¹X; xvi, Δ ; xvii, Hg⁺, aq. MeCN; xviii, LDA; xix, C₆H₁₃I; xx, Prⁱ(C₆H₁₃)NLi; xxi, THPO(CH₂)₅I;

xxii, Li, H₂NEt; xxiii, 3-hexyne

Scheme 32

Benzyl phenyl sulfides can be sequentially metallated and alkylated. The cleavage of the carbon-sulfur bond in the resulting product leads to a benzyllithium and lithium phenylthiolate.¹²⁶ Alkylation of the former organometallic then produces an alkylbenzene (Scheme 15), but competing alkylation on the thiolate is however observed.¹²⁶

An efficient geminal 'reductive alkylation' of the carbonyl group of aliphatic aldehydes and ketones, including that of cyclic ketones, takes advantage of: (i) the great availability of α -selenoalkyllithiums (themselves readily prepared from the corresponding carbonyl compounds *via* the selenoacetals);⁸⁴ (ii) their alkylation with primary alkyl halides; and (iii) the reduction of the corresponding selenides.¹⁴⁰ A specific example of this process, presented in Scheme 44 (entry a), describes an efficient synthesis of vitamin D.¹⁴⁴ A related reaction involving the reductive allylation of heptanal is recorded in Scheme 10.^{123,127,147,226} The case of aromatic aldehydes and ketones is even more interesting since (i) the related α -selenobenzyllithiums can be alkylated even with *s*-alkyl halides^{74,79,80,105,106} as well as with benzyl bromide,¹⁰⁵ and (ii) because the resulting benzyl selenides on reaction with *n*-butyllithium can be cleaved to produce benzyllithiums. The benzyllithiums in turn can be alkylated even with secondary alkyl





halides. This process therefore allows the geminal dialkylation of the carbonyl group of aromatic aldehydes and ketones, and permits the synthesis of particularly hindered alkylbenzenes (Scheme 17, entry a).^{79,80,230} Selenoacetals derived from aromatic carbonyl compounds as well as benzyl selenides bearing carbon–carbon double bonds in a suitable position possess a high tendency to cyclize leading to aryl cycloalkanes on reaction with tin hydrides or on reaction with *n*-butyllithium, followed by hydrolysis of the resulting alkyllithium (Scheme 46).^{228,231,232} The reaction of tin hydrides with the benzyl selenide shown in Scheme 46 (entry c) leads to a mixture of cyclopentane and cyclohexane derivatives (Scheme 46, entry d).²²⁸ The reaction is much more selective when carried out with BuⁿLi since not only does it provide the five-membered cycle exclusively, but it also produces stereoselectively each of the two stereoisomers depending upon the solvent used.²²⁸



i, Bu^sLi, THF-HMPA, -78 °C; ii, C₈H₁₇Br; iii, HgCl₂, aq. MeCN; iv, Br(CH₂)₄Br; v, HgCl₂, aq. MeCN; vi, Br(CH₂)₃Br; vii, HgCl₂, aq. MeCN; viii, base; ix, MeI; x, TiCl₄, aq. MeCO₂H

Scheme 35



Scheme 36



i, Bu^tLi, THF, -70 °C, 1 h; ii, BuX, THF-HMPA; iii, AcOH, 50%; iv, *p*-TsOH, EtOH, 50 °C; v, LDA, THF, -78 °C, 1 h; vi, MeI



Starting material (*E*): R = Me, 90% (*E*); $R = Bu^{t}$, 85% (*E*) Starting material (*Z*): R = Me, 85% (*Z*); $R = Bu^{t}$, 83% (*Z*)

i, BuⁿLi; ii, MeI; iii, BuⁿLi-Bu^tOK, THF, -65 °C; iv, MeI

Scheme 38







R = Me, 90%; R = Et, 89%; R = Buⁱ, 80%

i, NaNH₂, liq. NH₃; ii, R²Br, THF; iii, LiNEt₂; iv, RX; v, HgCl₂, MeOH

Scheme 39



	Reactants	Temperature (°C)	R	Ar	$R^{1}X$	Yield (%)	Ref.
(a)	LDA, THF-HMPA	-78	Н	Ph	C ₁₀ H ₂₁ Br	70	110
(b)	LDA, THF-HMPA	78	н	m-CF ₃ C ₆ H ₄	BuI	85	107
(c)	KDA, THF	-78	н	Ph	MeI	98	108
	KDA, THF	78	Н	Ph	$C_{10}H_{21}Br$	94	108
(d)	KDA, THF	-78	Bu	Ph	MeI	85ª	108
(e)	LDA, THF	-78	C ₆ H ₁₃	Pyr	MeI	94ª	111

 $^{a}(E/Z)$ stereochemistry retained

Scheme 40



i, BuⁿLi, THF, -78 °C; ii, MeI; iii, Li, EtNH₂, -15 °C, 1 h; iv, Bu₃SnH, AIBN, 90 °C, 3 h; v, Br₂, benzene; vi, Br₂, EtOH, 90 °C, 6 h

Scheme 41

1.3.2.2.2 Transformations of sulfides and selenides to alkenes

Alkyl sulfides and selenides have proved to be valuable precursors of alkenes (Scheme 17, entry b; Scheme 47; Scheme 48; Scheme 49; Scheme 50).^{74,104,157} These transformations can be achieved *via* the corresponding sulfoxides (Scheme 47)^{153,233} and selenoxides (Scheme 17, entry b)^{23,45,47,52,53,63,74,104,234} which collapse to alkenes and sulfenic or seleninic acid respectively on thermolysis. The elimination reaction takes place under particularly mild and neutral conditions with selenides (BuⁱO₂H, Al₂O₃, THF,



i, LDA, THF, -78 °C; ii, 2LDA, THF, -78 °C; iii, Ph(CH₂)₃I, 1 h; iv, 30% aq. H₂O₂, CH₂Cl₂, pyridine

Scheme 42





i, BuⁿLi, ii, C₆H₁₃I; iii, Li/H₂NEt; iv, B(SeMe)₃, CF₃CO₂H, CH₂Cl₂, 5 °C; v, NaHCO₃, EtOH, 80 °C;

vi, 1 BuⁿLi, 0.2 h; vii, 1.5 Br, ; viii, Bu₃SnH, AIBN, toluene, 100 °C; ix, SO₂;

x, 2.2 Bu₃SnH, hv, toluene, 20 °C; xi, NaHCO₃, EtOH, 80 °C

Scheme 44

60 °C, 3 h or H₂O₂, THF, 20 °C),¹⁴² but usually leads to a regioisomeric mixture of alkenes. It is however, particularly useful for the synthesis of 1-alkenes and 1-arylalkenes from the corresponding selenides, themselves available from alkyl halides and selenomethyllithiums or selenobenzyllithiums^{74,104,105} respectively, since, in these examples, there is only one possible regiochemical outcome, and also for the synthesis of 1,3-dienes from allyl halides^{74,127} and α -selenoalkyllithiums (Scheme 10; Scheme 17, entry b) because the presence of an allylic hydrogen in the β -position to the selenoxide directs the elimination towards the original double bond. The overall strategy is particularly useful for the synthesis of terminal alkenes, alkylidene-cyclopropanes and -cyclobutenes from primary alkyl halides and selenomethyllithium,^{52,70} 1-selenocyclopropyllithiums⁷⁰ and 1-selenocyclobutyllithiums⁷¹ respectively (Scheme 48), and for the synthesis of allylidene cyclopropanes from allyl halides and 1-seleno-⁷⁰ or 1-thio-cyclopropyllithiums (Scheme 5, entry a).¹²⁷

The other set of reactions which allow the transformation of sulfides and selenides to alkenes require the intermediary formation of a sulfonium or selenonium salt, and their treatment with a base such as KOH or Bu⁴OK.^{52,70,71,127} Furthermore, conjugated dienes, trienes and polyenes have been produced^{157,172,173} from allyl halides and benzyl and allyl dithiocarbamates respectively, using the set of reactions illustrated in Schemes 48, 49 and 50, and the transformation has been applied *inter alia* to the synthesis of β -parinaric acid (Scheme 50).¹⁷²

The oxidation of allylic sulfides with equivalent amounts of, for example, peroxyacids or sodium periodide leads to allylic sulfoxides.¹⁶ These sulfoxides do not usually eliminate sulfenic acid to provide

116



dienes, but instead produce^{16,170} allyl alcohols via a [2,3] sigmatropic rearrangement when treated with a thiophilic agent such as a secondary amine or a phosphite (Scheme 51).^{235,236}

The rearrangement of allyl selenoxides is even easier; it takes place directly at room temperature when allyl selenides are oxidized with hydrogen peroxide (30% aq. solution in THF)^{45,46,52-54,56,63} or peroxyacids. Allyl selenides produce allyl alcohols, whereas γ -chloro-,⁹⁵ γ -seleno-¹⁰⁰ and γ -seleno- γ -trimethylsilyl analogs¹⁰⁰ lead to α,β -unsaturated aldehydes, ketones and trimethylsilyl ketones respectively (Scheme 33, entries a-e). Therefore the corresponding α -lithium salts play the role of a masked β lithioallyl alcohol (Li-C-C-COH) or a masked β -lithio- α,β -unsaturated carbonyl compound (Li-C-C-C(-O)X) respectively.

Oxidative rearrangements of allylic selenides in the presence of various amine nucleophiles provide synthetic access to a variety of allylic amine derivatives.^{208,209} The stereochemical outcome of these reactions is consistent with a [2,3] sigmatropic rearrangement. The reaction has been applied to the synthesis of β , γ -unsaturated- α -amino acids²⁰⁹ or to their *trans*-alkene isosteres (see for example Scheme 33, entry f). The [2,3] rearrangement of allylic selenoxides has been successfully extended to the propargylic^{102,103} and allenyl¹⁰³ analogs, and provides α -seleno- α , β -unsaturated carbonyl compounds and propargylic alcohols respectively (Scheme 34, entry d; Scheme 42). This process does not occur with sulfur analogs as a consequence of the greater stability of the sulfoxide compared to the rearranged sulfenate.¹⁰²

Finally, α -alkylated allylic dithiocarbamates bearing a terminal C—C bond rearrange smoothly to the allylic dithiocarbamates in which the heteroatomic moiety is terminal (Scheme 52, entry b).¹⁷⁴

As shown earlier, alkenes and allylic alcohols can be prepared from alkyl and allyl sulfides or selenides by a series of reactions which involve alkylation of the corresponding organometallic, followed by oxidation of the resulting sulfides or selenides and further reaction of these species. One can think of the reverse order of reactions involving first the oxidation of the sulfide or selenide followed by alkylation of the resulting oxide. This procedure has in fact been performed in some cases as will be shown in the next section (Section 1.3.5). Of course the second approach offers the advantage of an easier metallation reaction. However, it possesses several disadvantages which have been observed from time to time. For example, alkyl selenoxides are prone to eliminate and are hygroscopic and therefore their metallation, although efficiently achieved, requires tedious manipulation in order to avoid these side reactions.^{46,47,54} As mentioned above, allyl selenoxides are too unstable and have only been postulated in the oxidation of allyl selenides. Their metallation and alkylation have not yet been attempted. Furthermore it has been reported in many instances that selective alkylation at the α -site of allyl sulfides can be



i, 2 MeSeH, $ZnCl_2$ or $TiCl_4$; ii, KDA, THF, -78 °C; iii, $\swarrow n^{Br}$; iv, Bu^nLi , THF, -78 °C; v, -78 °C, 0.5 h; vi, MeOH; vii, 0 °C, 1 h; viii, Bu^nLi , THF; ix, $\square n^{Br}$; x, Bu^nLi , THF, -78 °C, 0.1 h; xi, Bu^nLi , Et_2O , -30 °C, 0.1 h; xii, Bu_3SnH , AIBN, 80 °C

achieved on condition that a substituent is chosen which is suitable for metal chelation. This is not possible with the corresponding sulfoxide.¹⁶ Therefore, at least in these cases, the approaches reported in this section offer definite advantages over related ones.

1.3.2.2.3 Transformations of sulfides and selenides to organic halides: the haloalkylation and halopropenylation of alkyl and allyl halides

Alkyl and allyl sulfides and selenides can be transformed readily into the corresponding alkyl and alkenyl halides respectively. This reaction takes advantage of the easy formation of the corresponding sulfonium and selenonium salts on reaction with alkyl halides,^{125,143} alkyl bromoacetates¹⁴¹ or bromine^{143,237} (Scheme 53). A related process involves the intermediary formation of a selenoxide and its further reaction with hydrochloric or hydrobromic acids.²³⁸

These reactions, coupled with the alkylations of sulfides and selenides, allow the homologation of primary alkyl halides^{124,125,141,143,237,238} and the transformation of allyl halides to homoallyl halides.¹²⁵ A related process, employing 2-methylthio- and 2-allylthio-thiazoline, allows the iodomethylation (ICH₂—) and the iodopropenylation (ICH₂CH—CH—) of alkyl halides (Scheme 54).^{131a,132,171}



i, LICA, DME; ii, RCH₂X, 20 °C, 1 h; iii, DMF; iv, LICA, DME, PrⁱI, 20 °C, 0.3 h; v, heat, 180 °C, 0.2 h

Scheme 47

(a)
(a)

$$R^{1}CH_{2}Br \xrightarrow{i} R^{1}CH_{2}CH_{2}SeR \xrightarrow{H_{17}} R^{1} = C_{8}H_{17}$$

(b)
 $R^{1} = Ph, Me; R^{1} = C_{8}H_{17}, C_{9}H_{19}; 80-88\%$
(c)
 $R^{1} = C_{9}H_{19}; R^{1} = C_{9}H_{19}, C_{9}H_{19}$
 $R^{1} = C_{9}H_{19} \xrightarrow{I} C_{9}H_{19}$
 $R^{1} = C_{9}H_{19} \xrightarrow{I} C_{9}H_{19}$
(ref. 52, 142)
 $R = Ph, 80\%; R = Me, 83\%$
(ref. 52, 70)
 $R^{1} = C_{9}H_{19}$



i, RSeCH₂Li, THF, HMPA, -78 °C to 20 °C; ii, Bu^tO₂H, THF, -60 °C, 3 h; iii, MeI, AgBF₄; iv, Bu^tOK, DMSO, 20 °C; v, $\stackrel{\text{Li}}{\longrightarrow}_{n}$; vi, MeSO₃F; vii, Bu^tOK, DMSO, 20 °C, viii, MeI, AgBF₄; MeSe ix, Bu^tOK, DMSO, 20 °C; x, 180 °C

Scheme 48



i, LDA; ii, R¹CH₂X; iii, MeI, HMPA, Li₂CO₃, LiF

Scheme 49

 γ - and δ -hydroxy selenides which result from the reactions of α -selenoalkyllithiums with epoxides have been transformed to various selenium-free compounds. Thus y-hydroxyalkyl iodides and bromides have been produced in good yields from γ -hydroxyalkyl selenides and methyl iodide or bromine respectively (MeI, NaI, CaCO₃, DMF, 80 °C, 1 h; Br₂, EtOH-H₂O or Et₃N, CH₂Cl₂, 20 °C, 3 h). Under similar conditions δ -hydroxyalkyl selenides instead produce tetrahydrofurans directly.¹⁴⁹ The high propensity of the hydroxyalkyl selenonium salt intermediates to cyclize to a five-membered rather than to a four-membered heterocycle is responsible for these differences. Oxetanes however are available on treatment of the above mentioned y-hydroxyalkyl halides with base (Bu'OK, DMSO, 20 °C or MeMgBr-HMPA, 20 to 80 °C; Scheme 13, entries a and b). The reaction works well with those compounds in which the bromine is attached to a methine group, but increasing the substitution at this carbon dramatically lowers the yield of the heterocycle (Scheme 13, entry b) due to the concomitant formation of homoallyl alcohols.¹⁴⁹ The synthesis of the latter compounds is best achieved by first preparing the γ -hydroxyalkyl selenonium salts (MeI, neat, 20 °C for methylseleno derivatives) then treating them with a base (BuOK, DMSO, 20 °C or 50% aq. KOH, CH₂Cl₂, phase-transfer catalyst).¹⁵⁰ The high regioselectivity observed for this process has been explained by the cyclic transition state depicted in Scheme 13 (entry d). Oxidation of γ -hydroxyalkyl selenides under Sharpless conditions (H₂O₂, THF, 20 °C) produces a mixture of allyl and homoallyl alcohols which suggests that the corresponding selenoxides lead to an indiscriminate elimination reaction.¹⁵⁰ Selective elimination leading to the α,β -unsaturated ketones can however be achieved using Jones' reagent (CrO₃-H₂SO₄, acetone; Scheme 13, entry e).¹⁵⁰

 γ -Hydroxyalkyl selenides derived from α -selenobenzyllithiums and epoxides have proved to be valuable precursors of arylcyclopropanes.¹⁶⁰ The reaction proceeds through the corresponding γ -tosyloxy-benzyllithiums. Although the synthesis of the γ -hydroxyalkyl selenides is not stereoselective, the subsequent steps occur with high stereoselectivity, and have led to the conclusion that the benzyllithiums possess the opposite stereochemistry to that of the selenides and that this stereochemistry is retained in the next step (Scheme 18).¹⁶⁰

1.3.2.2.4 Transformation of alkylated vinyl sulfides and selenides to carbonyl compounds

Vinyl sulfides^{20,82,162,163,166,240} and selenides^{82,83,90,239,241-244} available on alkylation of α -metallovinyl sulfides and selenides^{107,108,110,239} (Scheme 35), themselves available from the corresponding vinyl



β-Parinaric acid

i, LDA, R¹X, -78 °C, 0.5 h; ii, Δ, CHCl₃, 4.5 h; iii, MeI; iv, NaBr, DMF, 50 °C, 24 h; v, HCl, acetone, O₂, 5 °C, 1 week; vi, MeI, LiF, LiCO₃, DMF, 20 °C, 24 h

Scheme 50

sulfides and selenides or from 1,1-diselenoalkenes, (Scheme 41),⁸³ or on γ -alkylation of α -metalloallyl sulfides,^{158,162,163} Scheme 27 and Scheme 52a) have been hydrolyzed to the corresponding carbonyl compounds *inter alia* with acids, mercury(II) or copper(II) salts. Therefore α -metallovinyl sulfides and selenides play the role of acyl anion equivalents in these transformations.

1.3.3 REACTIONS OF α -THIO- AND α -SELENO-ALKYLMETALS BEARING FURTHER HETEROATOMIC MOIETIES ON THEIR CARBANIONIC CENTER. APPLICATION TO THE FORMATION OF ACYL ANION EQUIVALENTS

1.3.3.1 Alkylations of 1-Metallo-1,1-dithioalkanes Bearing Hydrogen and/or Alkyl Groups on Their Carbanionic Center and Their Synthetic Applications

Organometallics bearing two heteroatomic moieties on their carbanionic center have found an important place in organic synthesis since the first report of Arens.²⁴⁵ In particular, they have played a crucial role as formyl or acyl anion equivalents.^{7,8,18-22,26-30}









80 °C; vii, MeI, NaI, CaCO₃, DMF, 80 °C, 5 h; viii, Li CH₂Cl₂, 20 °C, 3 h; x, O₃; xi, HBr, Et₃N, CH₂Cl₂, 20 °C

Scheme 53

1.3.3.1.1 Synthetic uses of 1-metallo-1,1-di(alkylthio)alkanes and -di(phenylthio)alkanes

1,1-Di(ethylthio)alkanes have been alkylated by sequential reaction with metalloamides in liquid ammonia^{245,246} or in HMPA²⁴⁷ followed by addition of alkyl halides. The reaction, which is usually conducted with 2 equiv. of the base is more efficient when carried out with sodium rather than lithium amides²⁴⁶ and, although it works well with all primary alkyl halides, only the parent compound and the one bearing an aryl substituent on the carbanionic center are alkylated with *s*-alkyl halides (Scheme 55, entry a).

Di(phenylthio)methyllithium^{59,126} and di(phenylthio)benzyllithium,^{248,249} which are synthesized from the corresponding carbon acids and *n*-butyllithium in THF, have been monoalkylated with primary but not with secondary alkyl halides. Higher homologs have not found widespread use because of the difficulties encountered in their alkylation.^{59,126,250–252} The successful alkylation of such compounds has been achieved²⁵² by carrying out the reactions in TMEDA (1 equiv.) in hydrocarbons (Scheme 55, entry c). Even the presence of THF with TMEDA almost completely suppresses the alkylation. However, double



alkylation of di(phenylthio)methane is possible²⁵⁰ in the presence of 2 equiv. of alkyl halides and an excess (4 equiv.) of sodium amide in THF. Interestingly the dianion derived from the conjugate addition of lithio tri(phenylthio)orthoformate on cyclohexenone followed by S/Li exchange involving s-butylli-

lithio tri(phenylthio)orthoformate on cyclohexenone followed by S/Li exchange involving *s*-butyllithium, has been methylated selectively at the carbanionic site (Scheme 56, entry a).²⁵³ A further interesting example is the use of 1,1-di(phenylthio)methyllithium to open an epoxide²⁵⁴ followed by a Grob fragmentation process as shown in Scheme 56 (entry b). The reaction of 1,1-di-

lowed by a Grob fragmentation process as shown in Scheme 56 (entry b). The reaction of 1,1-di-(thio)methyllithium with epoxides has also been used^{255,256} in the one-pot synthesis of 1,1-di(thio)cyclopropanes involving an intramolecular cyclization of the γ -tosyloxydi(phenylthio)alkyllithium. This intramolecular alkylation reaction proceeds even more efficiently than its intermolecular version, and allows the synthesis of a large variety of 1,1-di(thio)cyclopropanes from 3-chloro-²⁵⁷ and 3-phenylthio-1,1-di(thio)alkanes²⁵⁸ and *n*-butyllithium in THF (Scheme 57 and Scheme 58).

1,1-Di(methylthio)cyclopropanes have been converted^{255,256} into ketones resulting from the fission of the C(2)—C(3) bond and elimination of dimethyl disulfide on reaction with aqueous trifluoroacetic acid (Scheme 58). Since the epoxides are usually synthesized by oxidation of an alkene, the sequence alkene \rightarrow cyclopropanone \rightarrow thioacetal \rightarrow ketone allows the insertion of a carbonyl group in between the sp^2 carbons of the alkene (Scheme 58).²⁵⁵ Furthermore, reaction of di(phenylthio)cyclopropanes sequentially with lithium dimethylaminonaphthalene, carbonyl compounds and acid allows the synthesis of cyclobutanones by ring expansion of the intermediary 1-(1-hydroxyalkyl)-1-phenylthiocyclopropanes.^{17,25,61,128}

Related reactions involve the alkylation, allylation or benzylation of: (i) α, α -di(thio)alkylmagnesium halides, resulting from the thiophilic addition of Grignard reagents on dithioesters (Scheme 55, entry b);^{259,260} (ii) 2-lithio-1,3-benzodithioles (Scheme 55, entry d);²⁶¹ (iii) 2-aryl-2-lithio-1,3-dithiolanes²⁶² and 2-lithio-1,3-dithianes derived from the 1,2-dimethyl-4,5-di(mercapto)methylbenzene (Scheme 55, entry e).^{263,264}

1.3.3.1.2 Synthetic uses of 2-metallo-1,3-dithianes and related derivatives

2-Lithio-1,3-dithianes, (the Corey-Seebach reagent) have proved by far the most widely used organometallic reagents of this series (Scheme 59).^{27-29,265} This is due to: (i) the easy preparation of 1,3-dithianes from aldehydes and 1,3-propanedithiol; (ii) their metallation reaction which proceeds efficiently with BuⁿLi in THF, irrespective of whether the parent formyl derivative, higher homologs or 1-aryl-substituted derivatives are involved;²⁶⁷⁻²⁷¹ (iii) the high nucleophilicity of all these species towards various electrophiles,^{27,28} including alkyl,^{265,266,270,272-284} allyl,^{266,268,269,285-289} and benzyl halides,²⁶⁶ and arenesulfonates derived from primary alcohols²⁹⁰ as well as with terminal,^{255,266,279,289,291-300} α,β -disubstituted,^{255,299,301} and α,β -unsaturated epoxides²⁵⁵ and with oxiranes;²⁶⁶ and (iv) selective removal of the dithianyl moiety which produces²⁷⁻²⁹ a carbonyl group on reaction with metal ion, alkylative or oxidative hydrolysis^{302,303} such as HgCl₂ alone²⁹¹ or with CaCO₃,^{280,292} HgO,^{268,272,275,283-285,304} HgO– BF₃OEt₂,^{279,286,300,305,306} HgO–aq. HBF₄,³⁰⁷ CdCO₃,^{281,301} NCIS,²⁷⁵ NBS²⁷⁵ alone or with AgNO₃;²⁷⁵ with CuCl₂ and CuO,³⁰⁸ an aroyl peroxide,³⁰⁴ thallium(I) trinitrate³⁰⁹ or bis(trifluoroacetoxy)iodobenzene.³⁰²

This reaction has been used for the synthesis of a large variety of aldehydes and ketones^{20,27,28,278} including: (i) deuterio aldehydes;³¹³ (ii) optically active aldehydes and ketones;^{283,284} (iii) alicyclic (Scheme 60, entry d)²⁸² and cyclic diketones (Scheme 60, entry c),^{274,278,281} including those derived from metacyclophanes,^{267,271} 1,9-dienes and macrocyclic acetylacetone ligands for metal cations;²⁸¹ (iv) silyl and germyl ketones;³¹⁴ and (v) the synthesis of many natural products or their model compounds, notably



Scheme 55



Scheme 55 (continued)



i, (PhS)₃CLi; ii, Bu^sLi; iii, MeI; iv, H₂O; v, (PhS)₂CHLi; vi, BuⁿLi, 4 CF₃CO₂Cu, 20 °C, 3 h

Scheme 56

calcimycin,²⁸⁰ norpyrenophorin, pyrenophorin and vermiculin antibiotics,²⁷⁹ carbohydrates,^{315–317} 2methyllysergic acid,³¹¹ terpenes and terpenoid derivatives^{268,285,318} (including α -turmeron²⁶⁸ (Scheme 61, entry a), *cis*-verbenol³¹⁸ and linaloyl oxide,²⁹⁸), alnusone,^{20,319} rethrolones,^{20,272} prostaglandins²⁷⁶ and analogs,³⁰¹ debromoaplysiatoxin and aplysiatoxin (which are extracted from marine blue green algae),²⁹⁴ (3*R*)- and (3*S*)-2,3-epoxysqualene,^{20,304} 11 α -hydroxyprogesterone,³²⁰ modified steroids,^{20,296} corticosteroids,²⁹⁵ laurencin (a representative of naturally occurring halo compounds possessing a medium-sized cyclic ether skeleton as well as an enyne moiety),³⁰⁶ and the northern part of maytansin.^{291,292}

Reduction of the dithianyl moiety to a methylene group has been achieved²⁰ with Raney nickel, 267,293,305,310 or Na in NH₃ or in ether, 285,305,311 or with LiAlH₄ in the presence of CuCl₂ and ZnCl₂.³¹²

The whole process, which involves the synthesis of the 1,3-dithiane and the reduction of the product resulting from its alkylation, has been used *inter alia* for the synthesis of metacyclophanes (Scheme 60,

126



i, HX; ii, EtSH; iii, BuⁿLi, THF; iv, MeLi, THF-TMEDA; v, HCl; vi, PhSH; vii, BuⁿLi, THF



i, (MeS)₂CHLi, -78 °C, then 0 °C, 15 h ; ii, TosCl; iii, BuⁿLi, THF, -50 °C then 0 °C, 15 h; iv, CF₃CO₂H, H₂O, reflux, 30 h; v, (RS)₂CHLi, THF, -78 °C, then 0 °C, 15 h; vi, TosCl; vii, BuⁿLi, -50 to 0 °C; viii, CF₃CO₂H, H₂O, reflux, 7 h; ix, HCO₂H, H₂O, 100 °C, 20 h

Scheme 58

entries a and b),^{267,270,271} the cannabinoid side chain³⁰⁵ and terpenes such as α - and β -curcumin (Scheme 61, entry b)²⁶⁸ and (20S)-hydroxycholesterol (Scheme 62).²⁹³

The key step in all these transformations is without doubt the reaction of 1-lithio-1,3-dithianes with organic halides and epoxides.²⁷⁻²⁹ The alkylation usually proceeds extremely rapidly with primary alkyl iodides²⁶⁵ (-78 °C, 0.2 h) and with allylic^{265,286} and benzylic halides^{265,267,271} (Scheme 59, entry a) but is much slower with secondary alkyl iodides and bromides. The reaction is best carried out at low temperature in order to obtain good yields by lowering the competitive elimination reaction; it has been found to proceed with inversion of the configuration at the asymmetric carbon when optically active alkyl halides are used.²⁸³

Tertiary and cyclic alkyl halides (cyclohexyl iodide), secondary alkyl chlorides and primary and secondary tosylates cannot be used in these intermolecular alkylations.²⁶⁶ This is also the case with dibromoethane^{28,274} and 2-bromo-2-nitropropane,³²¹ which both lead instead to α,β -dithianylalkanes resulting from the oxidative coupling of the carbanion.



i, BuⁿLi, THF, -78 °C; ii, BuⁿLi, THF, -78 °C, 0.2 h

Scheme 59

Dialkylation of bis-dithianes has proved to be an efficient route to diketones.^{266,270,271,282} This might be carried out by successive monoalkylations, each requiring the metallation of the intermediate bisdithiane) followed by treatment with the alkylating agent (Scheme 63, entry a) or in an operationally simpler alternative by adding 2 equiv. of the base (BuⁿLi) followed by 2 equiv. of the alkylating agent (Scheme 63, entry b). The success of the latter approach depends either on the possibility of generating the dianion directly or more likely on the ability of the excess alkylating agent to survive attack by the second equivalent of base while the first alkylation step on the monoanion is in progress.

Stepwise alkylations have been carried out successfully with primary alkyl halides,²⁸² isopropyl iodide,²⁸² 1-chloro-3-iodopropane²⁷⁸ and benzyl bromide²⁸² (Scheme 63, entry b). Other secondary halides and *t*-butyl halides fail to react.²⁸²

The proof given in support of dianion formation does not however seem to be sufficient.²⁸² Nevertheless the single step process is reasonably efficient when primary alkyl halides are used but none of the dialkylated compound is produced with isopropyl iodide and benzyl bromide. 2-(ω -Haloalkyl)-1,3-dithianes have proved to be valuable precursors of 1,3-dithianes derived from cyclic ketones.²⁷⁸ Halogenated 1,3-dithianes have been in turn prepared from 2-lithio-1,3-dithiane and stoichiometric amounts of ω -chloro- or bromo-alkyl iodides or with an excess of the corresponding dichloride²⁷⁸ (Scheme 64, entry c). Use of ω -bromoalkyldithianes is often impractical because of the ease with which these compounds are transformed into cyclic sulfonium salts.²⁷⁸

Another, but longer, synthesis of such ω -haloalkyldithianes has been achieved by reaction of 1-lithio-1,3-dithianes with epoxides (see below)^{255,256,278,322} followed by tosylation of the corresponding alkoxides and displacement of the tosylate with chloride ion (LiCl, DMF, 30 °C, 6 d; Scheme 64, entries a and b; Scheme 58, entry b).²⁷⁸

Furthermore specifically 2-deuterated-4,6-dimethyl-1,3-dithiane^{273,287,323} and 2-phenyl-4,6-dimethyl-1,3-dithiane^{287,288} produce, on sequential reaction with *n*-butyllithium in THF or in THF—HMPA and with methyl iodide, the corresponding 2-*cis*-4,6-trimethyldithianes in which the 2-methyl group lies in the equatorial position (Scheme 65).

The results displayed in Scheme 65 (entry a) show that, although equatorial abstraction of hydrogen is favored kinetically,²⁸⁷ the ratio of equatorial to axial proton abstraction is considerably less than the ratio of equatorial to axial methylation. Most probably the lithium derivative rapidly reorients itself in such a way as to have the lithium held in the equatorial position by the cooperative effect of the two sulfur atoms.^{287,288,323}

Epoxide^{255,266,278,300} and oxirane²⁶⁶ ring opening by 2-lithio-1,3-dithianes requires the use of low temperatures (-20 to 0 °C), since these organometallics are not sufficiently stable in THF at higher temperatures. Under these conditions the reaction is often slow (1 or 2 h) and requires in some cases 1 or even 2 days to go to completion (Scheme 58, entry b; Scheme 62; Scheme 66; Scheme 67).^{293,296,297}

The reaction usually occurs at the less substituted carbon of the oxirane ring (Scheme 62 and Scheme 67).^{293,297} However regioisomeric dithianyl alcohols are obtained³⁰¹ from 2-lithio-1,3-dithianes and the protected β , γ -epoxy alcohols shown in Scheme 66 (entry b). The observed regioselectivity is very different when methoxymethyl or benzyl protecting groups are used. This has been explained by the different



capacities of these two protecting moieties to form initial coordination complexes with the lithiodithiane reagent.³⁰¹

The reaction of 2-lithio-1,3-dithianes with epoxides has been used in a large number of valuable synthetic transformations including the synthesis of: (i) the corresponding thioacetals of cyclopropanones and of carbonyl compounds resulting from their acid-catalyzed ring opening (Scheme 58, entry b);²⁵⁵ (ii) γ -hydroxy^{297,301} or acetoxy³⁰⁰ aldehydes resulting from the selective hydrolysis of the related thioacetals (Scheme 66); and (iii) α , β -unsaturated aldehydes, including functionalized ones as shown in Scheme 66.³⁰⁰



It has also been shown³²² that the Sn/Li exchange at the 2-position of 2-stannyldithiane takes place within minutes at -78 °C, whereas H/Li metallation does not occur at all at this temperature (Scheme 59, entry b). This reaction, when applied to 2-(ω -oxidoalkyl)-2-stannyl-1,3-dithiane, takes place within a few minutes and the intramolecular nucleophlic reaction, which immediately follows, is usually complete within 0.1 h at -78 °C. The total yield of cyclization products by the tin route has proved to be twice as high as that by direct metallation (Scheme 68).³²²

Finally, although 2-lithio-1,3-dithianes do not usually react with tetrahydrofurans, an intramolecular ring opening of such heterocycles has been once reported and has been applied to the synthesis of a dithiane derived from a functionalized cyclohexanone (Scheme 67).^{297,298}



i, BuⁿLi, THF, -30 °C; ii, RX (X not cited); iii, BuⁿLi; iv, R'X; v, HgCl₂, HgO; vi, 2BuⁿLi, THF, -78 °C; vii, 2 RX

Scheme 63

1.3.3.2 Alkylations of 1,1-(Dithio)allyl- and 1,1-(Dithio)propargyl-metals and their Synthetic Uses

1,1-(Dithio)allyllithiums, which are available on metallation of ketene thioacetals (Scheme 69, entry a; Scheme 70, entry c)³²⁴⁻³²⁷ or 1,1-(dithio)alk-2-ene (Scheme 61, entries b and c)^{268,328-330} and from the addition of alkyllithiums to unsaturated ketene thioacetals) a (Scheme 71, entry a),³³¹ possess a high propensity to be alkylated^{324,326,331} or benzylated^{324,326} at their α -site (Scheme 70, entries a–c; Scheme 71, entry a). The 1,3-dithianyl derivative bearing a phenyl group at the γ -position (the softer site) however exhibits a greater aptitude for reaction at this site (Scheme 70, entries d and e).^{328,329} The α : γ ratio in this case increases by increasing the hardness of the electrophile at both the acid (PhCH₂+ < Me⁺ < Et⁺) or the leaving group (I⁻, Br⁻ < CI⁻ < ⁻OTs < ⁻SO₄Me).³³² The latter effect was found to be greatest in the case of methylation (Scheme 70, entry d).^{328,329}

Allylation of all these species is apparently less selective (Scheme 69, entry a),³²⁵ and among the various thioacetals tested 1,3-dithianes proved to be the ones which lead to the highest α : γ ratio.^{325,326,327,330} The use of HMPA as a cosolvent was shown to have no effect on the regioselectivity of the allylation.³²⁵ However, both exclusive γ -allylation and γ -methylation have been achieved by performing the reaction in the presence of copper(I) iodide–trimethyl phosphite complex (Scheme 69, entries a–e). The reaction is best achieved with allylic chlorides, bromides and phosphate esters and does not occur with acetates, although these compounds usually react with cuprates. In all cases where isomers could be formed with respect to the electrophile, the ratio of $S_N 2':S_N 2$ products was found to be greater than 0.67 (Scheme 69, entries b–e).³²⁵ Related reactions have already been described with allylic sulfides (Scheme 27, see Section 1.3.2.2.4).^{162,163}

On the one hand, thioacetals of α , β -unsaturated ketones resulting from the α -alkylation reaction reported above have been transformed efficiently into the corresponding unsaturated ketones on reaction with mercury(II) chloride and oxide in methanol (Scheme 71, entry a)^{329,331} or on reaction^{326,333} with methyl iodide or methyl fluorosulfonate followed by hydrolysis, with sulfonyl hydroxylamine.³²⁴ On the



i, $\overset{O}{\underset{R}{\longrightarrow}}$; ii, H₃O⁺; iii, TsCl, pyridine; iv, LiCl, DMF; v, BuⁿLi, THF; vi, $\overset{O}{\underset{R}{\longrightarrow}}$ O; vii, H₃O⁺; viii, TsCl, pyridine; ix, LiCl, DMF; x, BuⁿLi; xi, I $\overset{O}{\underset{R}{\longrightarrow}}$ Cl; xii, BuⁿLi, THF; xiii, HgCl₂, HgO, 80 °C, 3 h; xiv, I $\overset{O}{\underset{R}{\longrightarrow}}$ Cl, THF; xv, BuⁿLi, THF

Scheme 64



i, BuⁿLi, THF, -78 °C; ii, MeI; iii, BuⁿLi, solvent, -78 °C; iv, MeI

Scheme 65

other hand, ketene thioacetals resulting from a γ -alkylation reaction lead to carboxylic acids on reaction with mercuric chloride and oxide.³²⁹ Thus the 2-metallo-2-vinyl-1,3-dithianes in the above mentioned transformations have played the role of α , β -unsaturated acyl anion or homoenolate equivalents respectively.

2-Ethynyl-1,3-dithiane when treated with 1.1 equiv. of *n*-butyllithium then with methyl iodide affords³³⁴ only the γ -products as a 1:1 mixture of alkyne and allene, resulting from the direct methylation of the lithioalkyne and of the γ -methylation of 2-ethynyl-2-lithio-1,3-dithiane respectively (Scheme 72,



entry a). Interestingly, the dilithio species produced on reaction with 2.5 equiv. of *n*-butyllithium (0.5 h at -70 °C) leads³³⁴ exclusively to the ethynyl product by selective alkylation at the propargylic site, to the virtual exclusion of the allene (Scheme 72, entry b). 2-Lithio-2-(2'-silylethynyl)-1,3-dithiane has proved to be a valuable reagent which can be methylated³³⁴ or allylated³³⁵ exclusively at its 2-position (Scheme 72, entries c and d). The resulting products have been transformed into the corresponding functionalized ketones on successive treatment with thallium(III) nitrate trihydrate and hydrochloric acid (Scheme 72, entries c and d).³³⁵



1.3.3.3 Alkylations of Acyl Anion Equivalents Containing at Least One Sulfur or Selenium on their Carbanionic Center

1.3.3.3.1 Alkylations of sulfur-containing acyl anion equivalents

Although 2-lithio-1,3-dithianes are easily synthesized and have proved to be valuable nucleophiles towards, *inter alia*, alkyl halides, their use has been limited by the not infrequent difficulty of generating the carbonyl function.³⁰² Thus several functionalized α -thioalkylmetals have been proposed as their alternative.

The methylations of 1,3-dilithio-5,7-dimethyl-2,4,6,8-adamantane³³⁶ and of 1,3,5,7-tetrathiacyclooctane tetraanion³³⁷ are of theoretical value only. 1,3,5-Trithiane and both mono- and 2,4,6-trialkyl-substituted derivatives have been metallated^{28,338,339} with *n*-butyllithium and alkylated^{338,339} with reactive alkylating agents such as primary alkyl iodides or bromides and benzyl bromide (Scheme 73, entries a and b). The metallation of monoalkylated trithianes occurs at one of the two unsubstituted sites and leads eventually to the product having both substituents in the equatorial position.^{273,323} Thus the high equatorial preference for lithium in 2-lithio-1,3-dithiane is also present in 2-lithio-2-methyl-1,3,5-trithiane. Mer-



Scheme 70



i, R¹Li, -80 to 20 °C; ii, MeI; iii, HgCl₂, HgO; iv, LDA; v, MeI

cury(II)-assisted solvolysis of the alkyl trithianes then liberates the most highly substituted carbonyl compounds (Scheme 73, entry b).^{338,339} Thus 1,3,5-trithiane after monoalkylation leads finally to aldehydes, whereas alkylation of lithio-2,4,6-trialkyl-substituted analogs would lead to the corresponding alkyl ketones (Scheme 73, entry b).

Several other acyl anions or potential acyl anion equivalents bearing at least one nonoxidized sulfur atom have since been proposed, and some of them have been alkylated successfully. This is effectively the case for the following metallated compounds: (i) *N*-methylthioformaldine (Scheme 73, entry c);³⁴⁰ (ii) 1,3-oxathianes,^{341,342} and α -trimethylsilylmethyl analogs;³⁴³ (iii) α -methoxythioanisole (Scheme 74, entry a; Scheme 75, entry c);^{344–347} the parent compound also allows³⁴⁵ the synthesis of acetals (on acid-catalyzed methanolysis), of vinyl ethers (on oxidation to the sulfoxide and heating at 120 °C) and carboxylic acids (CrO₃, H₂SO₄, Jones reagent; Scheme 75, entry b) and its α -vinyl-substituted derivatives (Scheme 76);^{347,348} (iv) α -trimethylsilylmethylthioanisole (Scheme 77, entries a-c; Scheme 78, entries e and f),^{252,349-354} and its α -aryl derivative (Scheme 77, entry c);³⁵⁵ (v) methylthiomethyl methyl sulfoxide (Scheme 74, entry b; Scheme 79; Scheme 80, entries b and c),^{356,359-362} and related diethyl (Scheme 74, entries c and d; Scheme 80, entry a),^{358,365,366} or di-*t*-butyl³⁶⁵ derivatives as well as 1,3-dithiane-1-oxide^{367,368} and its 2-alkyl- and 2-phenyl-


i, 1.1 BuⁿLi; ii, MeI; iii, base; iv, RX; v, Tl(NO₃)₃, MeOH; vi, 5% HCl

substituted homologs;³⁶⁸ (vi) methylthiomethyl *N*,*N*-dimethyl dithiocarbamate (Scheme 74, entry e);³⁶⁹ (vii) methylthiomethyl *p*-tolyl sulfone and homologs (Scheme 74, entries f–h; Scheme 80, entry d);^{370,371} (viii) bis(*N*,*N*-dimethyldithiocarbamato)methane;³⁷² and (ix) 4,4-dimethyloxathiolane 3,3-dioxide.³⁷³

1.3.3.3.2 Alkylations of selenium-containing acyl anion equivalents

Related reactions involve 1-metallo-1,1-di(arylseleno)- 64,65,76,78,106,108,122 and 1-lithio-1,1-di(methylseleno)-alkanes (Scheme 81; Scheme 82), 76,78,122 1-lithio-1-phenylseleno-1-phenyltelluromethane (Scheme 78, entry h)³⁷⁴ as well as 1-trimethylsilyl-1-arylselenomethyllithiums (Scheme 78, entries af)^{74,106,353,375} and their 1-aryl homolog^{74,106} and 1-trimethylsilyl-1-methylselenoalkyllithiums (Scheme 78, entry g).⁷⁷ Intramolecular cyclizations of 3-halo-1,1-di(seleno)alkanes^{78,122} and of 3-tosyloxy-1,1,1tri(seleno)alkane⁷⁸ with LDA and *n*-butyllithium respectively lead to the corresponding 1,1-di(seleno)cyclopropanes (Scheme 82, entries b–g). These in turn have proved to be valuable precursors of 1-seleno-1-cyclopropyllithiums (Scheme 8, entry e; Section 1.3.2.1.1).

Interestingly, the synthesis of 1,1-di(phenylseleno)cyclopropane has been performed^{78,122} from 3-chloro-1,1-di(phenylseleno)propane and Bu^tOK (Scheme 82, entry c) or, in a longer sequence, on reaction of 1-(3-chloropropyl)-1-phenylseleno-1-phenyl selenoxide¹¹³ with LDA and further reduction of the resulting 1-phenylseleno-1-phenylselenoxycyclopropane (Scheme 82, entry a). Surprisingly the reaction of Bu^tOK with 3-chloro-1,1-di(methylseleno)propane takes another course and leads to 1,1-di(methylseleno)-1-propene rather than to 1,1-di(methylseleno)cyclopropane.⁷⁸

Most of the organometallics reported above, with the exclusion of 1,1-di(methylseleno)- (Scheme 81, entry e)^{76,78} and α -silyl- α -methylselenoalkyllithiums (Scheme 78, entry e)⁷⁷ have been prepared by metallation of the corresponding carbon acid. The Se/Li exchange method however proved to be the method of choice for the synthesis of almost all the α -selenoalkylmetals described in this section. 64,65.76,78,106,108,122

As far as the further reaction of the above mentioned organometallics with electrophiles is concerned, most of these reagents work well with primary^{341-344,347-352,356-360,362,363,365,366,368-373,375} and secondary^{341-343,349,350,352,358,360,361} alkyl halides, benzyl bromides^{341-344,348,350,352,357,369-373} and allylic bro-

136



mides^{344,347,350,358,370,373} or chlorides.^{347,360,361} Only a few have been reacted with epoxides^{298,344,349,376} and even fewer with trialkyl-substituted halides (Scheme 77, entries f and g).³⁴⁹

Some of the reactions reported above merit further comment. This is especially the case of methylthioformaldine (Scheme 73, entry c)³⁴⁰ and trimethylsilylmethylthioanisole (Scheme 77)^{349,355} which can be alkylated once, but not twice, because of the lack of basic reagents sufficiently reactive to metallate the monoalkylated compound. Alternative routes to trimethylsilyl(*s*-alkyl)thioanisoles however exist.^{254,352} It must be pointed out that the synthesis of fully alkylated seleno analogs also cannot be achieved by metallation reaction,⁷⁷ but takes place by Se/Li exchange from 1-silyl-1,1-di(methylseleno)alkanes and butyllithium and further alkylation of the resulting organometallics (Scheme 78, entry e). Trimethylsilyl(phenylthio)methyllithium is a particularly nucleophilic reagent toward primary and secondary alkyl halides^{349,352,353,355} and epoxides (Scheme 77; Scheme 78; entries c and d).³⁴⁹ Its successful alkylation with trisubstituted epoxides³⁴⁹ and cyclohexyl iodide³⁴⁹ is exceptional (Scheme 77, entries d and e). Alkylation of α -heterosubstituted organometallics with cyclohexyl halides^{350,352} is usually unsuccessful and poor yields result from reaction with *s*-alkyl halides.³⁵³ Allyl halides,³⁵¹ also react efficiently with this organometallic. Most of these reactions have been carried out in THF, even though it has been suggested that better results are obtained in TMEDA–hexane.³⁵⁰

2-Lithio-1,3-oxathiane^{341,342} has proved to be much less reactive than phenylthiomethyllithium (Scheme 74, entry a).³⁴⁴ Although the latter reacts well with alkyl or benzyl bromides and with alkyl iodides without distinction, 2-lithio-1,3-oxathiane works well only with alkyl iodides.³⁴¹

Monoalkylation of the carbanion generated from methyl thiomethyl sulfoxide³⁵⁷ and 1,3-dithiane 1oxide^{368,377} has been reported (Scheme 74, entry b) but reexamination of both reactions^{358,359} proved unsatisfactory under conditions similar to those previously described or slightly modified. Neither the stepwise³⁵⁹ nor the one-pot dialkylation³⁵⁷ was successful at 20 °C, but the latter reaction was apparently achieved by Schill³⁵⁹ by performing it at 50 °C with a slight excess (2.2 equiv.) of *n*-alkyl bromides in



	R	R1	X	i	ii	%	iii	%	Ref.
(a)	Ph	Н	OMe	Bu ^t Li, THF,	MeI	94	_	-	344
				–78 °C	C ₆ H ₁₃ I	84			
					PhCH ₂ Br	88			
					H ₂ C=CHCH ₂ Br	76			
(b)	Me	Н	S(O)Me	NaH, THF,	MeI, 20 to 40 °C	92	HCl, H ₂ O	84	357
				20 °C	Bul	72		91	
	•				PhCH ₂ Br	92		88	
(c)	Et	H, Et	S(O)Et	Bu ⁿ Li or LDA,	EtI, 25 °C, 2 h	>95	HgCl ₂ , HCl,	>80	357
				THF, 2 °C	Pr ⁱ Br		THF, 0 °C, 1 h	>80	
					H ₂ C=CHCH ₂ Br			>80	
(d)	Me	н	SC(S)NMe ₂	Bu ⁿ Li, THF, -50 °C	C ₈ H ₁₇ I, -60 °C, 1 h, 20 °C		aq. HCl, 20 °C	66 overall	369
					PhCH ₂ Br			68 overall	
(e)	Me	Н	SO ₂ Tol	50% aq. NaOH, toluene	C ₁₂ H ₂₅ Br, 20 °C, 4 h	84	Dioxane, H ₂ O, NaOH, <i>hv</i>	72	370
					PhCH ₂ Br , 50 °C, 3 h	88		-	371a
(f)	Me	Н	SO ₂ Tol	2.5 equiv. NaH, DMF	2 C ₁₂ H ₂₅ Br, 20 °C, 48 h		H_2SO_4 , MeOH, Δ , 3 h	90 overall	371a
					2 PhCH ₂ Br, Δ , 3 h			74 overall	371a



i, BuⁿLi, THF, -78 °C (n = 0); ii, LDA, THF, -78 °C (n = 2); iii, RX; iv, cat. TosOH, MeOH, Δ , 2 h; v, CrO₃, H₂SO₄, 0 °C, 1 h, acetone; vi, MCPBA, -5 to 0 °C, toluene, then 40 °C; vii, MCPBA, toluene, 0 °C, then 120 °C (n = 0); viii, Bu^tOK, THF (n = 2)

Scheme 75



i, BuⁿLi, HMPA; ii, R¹X; iii, SiO₂-H₂O₂, 20 °C, 3 h; iv, NaIO₄, dioxane, 20 °C, 3 h; v, MCPBA, CH₂Cl₂, -30 °C, 1 h; vi, SiO₂, hexane; vii, BuⁿLi-TMEDA, THF, 0 °C, 0.5 h; viii, BuI

the presence of an excess of sodium hydride in THF. More recently, this reaction has been successfully used by Evans in a synthesis of the ionophore antibiotic A-23187 (Scheme 79, entry b).³⁶³

Slight variations of the structure of this acyl anion equivalent have led in fact to the discovery of particularly valuable alternative reagents. Among them, ethyl ethylthio sulfoxide can be both monoalkylated^{358,365,366} or sequentially dialkylated³⁵⁹ (Scheme 74, entries c and d). *t*-Butyl methylthio sulfoxide^{359,365} and phenyl methylthio³⁷⁸ analogs have also been monoalkylated but not dialkylated (Scheme 79, entry c; Scheme 80, entry a).

Reaction of methyl methylthio sulfoxide with 1,*n*-dihalo or 1,*n*-(tosyloxy)alkanes in the presence of 2 to 2.8 equiv. of BuⁿLi or KH as a base gave three- to six-membered not seven-membered 1-methylsulfinyl-1-methylthiocycloalkanes (Scheme 80, entries b and c). These cycloalkanes have been subsequently transformed to the corresponding ketones under mild conditions.^{356,360–362,364} The tosyloxy route, which is often inefficient³⁴⁰ with related organometallics has proved to be particularly attractive^{360,364} for the construction of the cyclopropane ring which cannot be produced from 1,2-dibromoethane (Scheme 80, compare entries b and c). This cyclization reaction may involve^{360,361} the formation of an intermediary sulfonium salt (see Scheme 80, entry b), which is metallated and then rearranges to the desired product, *via* a Stevens-type rearrangement. This process allows the high yielding synthesis of diversely substituted (even in the 2-position) cyclobutanones,³⁶⁰ 3-cyclopentenone from (Z)-1,4-dichlorobut-2-ene (Scheme 83, entry a)³⁶⁰ and of (*R*)- and (*S*)-4-hydroxycyclopentenones,³⁶² which are valuable precursors of prostaglandins (Scheme 83, entry c). Particularly impressive is the efficiency of the last mentioned synthesis, which uses tartaric acids and methyl methylthio sulfoxide as the two building blocks. An unusually facile cyclization employing a mesyloxycyclopentane, and leading finally to a bicyclo[3.3.0]octanone (Scheme 79 entry a), a valuable precursor of isocarbacyclin, has also been reported.³⁶⁴

As expected (E)-1,4-dichlorobut-2-ene reacts³⁶⁰ differently from its (2Z)-isomer with methyl methylthiomethyl sulfoxide, and produces a vinylcyclopropane instead of the cyclopentene derivative (Scheme 83, entry b, compare to entry a).

Methyl methylthiomethyl sulfone and methylthiomethyl p-tosyl sulfone have been monoalkylated, benzylated or alkylsilylated with 1.5 equiv. of the electrophile in the presence of aqueous sodium hydroxide and trioctylmethylammonium chloride (TOMAC).^{370,371} Without doubt the organometallic is less reactive than the one derived from the corresponding sulfoxide since the reaction is rather slow with



i, BuⁿLi, TMEDA, 0 °C; ii, MCPBA, -15 °C; iii, H₂O; iv, LiCH(SiMe₃)SPh

allyl and benzyl halides (20 °C, 48 h) and requires heating to 60 °C to go to completion with primary alkyl bromides. The second alkylation, alkynylation and benzylation have been carried out³⁷⁰ with NaH in DMF (1.3 equiv.) at 20 °C for long periods of time (19 to 48 h) or more rapidly with heating (3 h at 60 °C). Cyclo-butane, -pentane and -hexane³⁷¹ as well as cyclopentene derivatives³⁸⁰ have been obtained from the corresponding dibromides. The synthesis of the cyclobutane derivative requires the use of an excess of NaH in DMF^{371a} or BuⁿLi in THF.^{371b} For the other cases, however, phase-transfer conditions can be used (Scheme 80, entry d). Unfortunately, as with methyl methylthio sulfoxide (Scheme 80, entry c), seven-membered rings cannot be synthesized^{371b} (Scheme 80, entry d) by this method.



(a) $R^1 = H; R = Ph; i, LDA, -30 \,^{\circ}C, 0.5 h; ii, C_{10}H_{21}Br, 75\%; iii, H_2O_2, THF$ (ref. 77, 375) (b) $R^1 = H; R = 4-CF_3C_6H_4; i, LDA, -78 \,^{\circ}C, 0.5 h; ii, PhCH_2Br, 83\%; iii, MCPBA, THF, Pr^i_3N, 71\%$ (66:34) (ref. 74, 106)





i, BuⁿLi, THF, -78 °C; ii, C₁₀H₂₁Br; iii, H₂O₂, THF; iv, NaIO₄, EtOH; v, LDA, THF, -78 °C, 0.5 h; vi, MeI

Scheme 78

1.3.3.3.3 Synthesis of aldehydes and ketones from α -heterosubstituted sulfides and selenides and related compounds

Most α -heterosubstituted alkyl sulfides discussed in this section have proved to be much easier to transform to aldehydes or ketones than the corresponding 1,3-dithianes. In particular this is the case of (i) formaldines³⁴⁰ (HgCl₂, HgO, 20 °C, 2 h; Scheme 73, entry c), (ii) α -methoxyalkenyl phenyl sulfide³⁴⁸ (NaIO₄, H₂O, dioxane, 20 °C; Scheme 77) and (iii) the α -silylalkyl phenyl sulfides^{349–351,355} (MCPBA, -15 °C then H₂O; Scheme 77) and α -silylalkyl selenides^{74,77,106,375} (H₂O₂, THF, NaIO₄, MCPBA, 20 °C; Scheme 78, entries a–e). The last mentioned reactions involve the transformation of the α -silyl sulfoxide or selenoxide first produced to the corresponding trimethylsilyl hemithio or hemiseleno acetal *via* a Sila–Pummerer rearrangement. This requires overnight reaction at 20 °C or 0.5 h at 60 °C in the case of sulfur, but proceeds almost instantaneously with the seleno analog. Interestingly, this rearrangement can be sup-



pressed and the elimination reaction which leads to vinylsilane favored if the α -silyl sulfide, selenide or telluride is instead treated with chloramine T (Scheme 78, entries c and d).³⁵³

1,1-Di(seleno)alkanes have been easily transformed to carbonyl compounds on reaction with a large array of reagents including HgCl₂-CaCO₃, CuCl₂·2H₂O-CuO,^{86,243} hydrogen peroxide,⁸⁶ benzene-seleninic anhydride⁸⁶ or clay-supported metal nitrates³⁷⁹ in ways similar to those used with thio analogs. In some cases, an elimination reaction produces vinyl selenides instead of the carbonyl compound. In such an event, the latter derivative can be obtained by further hydrolysis^{243,244} of the vinyl selenide, for example with HgCl₂ in the absence of an added base (Scheme 81).

The transformations of thioacetal monosulfoxides to carbonyl compounds^{357-361,365,366} have been



i, 1 LDA; ii, Br(CH₂)₃Br, 25 °C, 2–10 h; iii, Br(CH₂)_nBr, 2.8 KH, THF, 20 °C, 19–48 h; iv, Bu^sLi; v, 0.5 Tos(CH₂)_nTos; vi, 4.5 M H₂SO₄, 20 °C, 4–15 h (or 20 h, 45 °C if n = 3); vii, 50% aq. NaOH, toluene, TOMAC, 60 °C, 64–144 h, Br(CH₂)_nBr; viii, BuLi; ix, 0.5 Br(CH₂)₃Br, 20 °C, 22 h; x, conc. HCl, MeOH; xi, conc. HCl, dioxane, Δ , 3.5 h

Scheme 80

carried out with hydrochloric, perchloric, or sulfuric acid in water, ether or THF (Scheme 80, entries c and d; Scheme 83, entries a and c) and often in the presence of mercury(II) salts in order to remove the disulfide concomitantly produced,³⁵⁸ or with NBS in the presence of CaCO₃ (Scheme 74, entries b–d; Scheme 79, entries a and c).³⁶⁴ These conditions are sufficiently mild to allow, for example, the almost exclusive formation³⁶⁰ of 3-cyclopentenone from (Z)-1,4-dichlorobut-2-ene without substantial isomerization to the thermodynamically more stable 2-cyclopentenone (Scheme 83, entry a). The synthesis of the more strained cyclobutanone requires slightly more drastic conditions however to go to completion (4.5 M H₂SO₄, 20 °C, 18 h and 45 °C, 5 h).^{371b}

Finally, ketone dimethylthioacetal S,S-dioxides³⁷¹ are more difficult to transform to ketones than their S,S-monoxides³⁵⁷ (Scheme 74, entries f–h). Effective transformation usually requires heating at reflux in the presence of concentrated acid. Moreover, dimethyl thioacetal S,S-dioxides^{371a} derived from aldehydes resist the above mentioned hydrolysis conditions and require, in order to be transformed to the aldehyde, preliminary photochemical cleavage of the C—SO₂ bond^{371a} ($h\nu$ then aq. NaOH; Scheme 74,



i, Buⁱ₂NLi, THF, -78 °C, then -30 °C, 0.2 h; ii, LDA, THF-HMPA; iii, LiTMP, THF-HMPA, -30 °C, then 20 °C, 0.2 h; iv, KDA, THF, -78 °C, 0.2 h; v, HgCl₂, CaCO₃, 20 °C, 2 h; vi, LDA, THF, -78 °C, 0.1 h; vii, BuⁿLi, THF-hexane, -78 °C; viii, LIDBA, THF, -78 °C; ix, LDA, THF, -78 °C; x, H₂O₂, THF

Scheme 81

entry f). Similar conditions can be used for ketones but they are less efficient than acid-catalyzed hydrolysis.^{371a}

In conclusion, the 1,3-dithiane unit still remains among the most valuable acyl anion equivalents, although that involving methyl methylthio sulfoxide or its diethyl analog offers the definite advantages of proceeding with metal hydride or even under phase-transfer catalysis conditions and of producing the carbonyl compound under reasonably mild conditions.

Otherwise α -methoxy allyl sulfides have played the role of an α -methylenated acyl anion³⁴⁷ (Scheme 76, entry a) or a homoenolate dianion³⁴⁸ (Scheme 76, entry b) equivalent.

1.3.4 α -METALLOORTHOTHIO- AND α -METALLOORTHOSELENO-FORMATES AS PRECURSORS OF ESTERS

Metallated orthothioformates and their seleno analogs (available by metallation of orthothio- $^{28,381-383}$ and orthoseleno-formate^{64,65,76-78} with lithium amides or on S/Li or Se/Li exchange between orthothio- 381,382 and orthoseleno-carbonates⁶⁵ react efficiently with methyl iodide^{64,65,381,382,384} as well as with reactive electrophiles such as benzyl chloride and *n*-butyl iodide (Scheme 84, entry a).³⁸³



i, HCl, benzene; ii, RSeH, ZnCl₂; iii, MCPBA; iv, LDA; v, LDA, THF, 0 °C; vi, Bu^tOK, DMSO, 20 °C; vii, TosCl, pyridine, 20 °C; viii, BuⁿLi, THF, -78 to 20 °C

Tri(phenylthio)methyllithium does not react properly with less reactive electrophiles such as *n*-butyl chloride,³⁸² 2-iodopropane (Scheme 84),^{381,382} propylene oxide (Scheme 85)^{381,382} and especially cyclohexene oxide (Scheme 85, entry b)³⁸² owing to its high propensity to decompose into (PhS)₂C and PhSLi (see below).^{381,382,384,385}

1-Lithio-4-methyl-2,6,7-trithiabicyclo[2.2.2]octane, which is far more stable due to its favorable stereochemistry gives the highest alkylation yields.³⁸¹ Tri(phenylseleno)methyllithium^{25,64,65,78} and especially its methylseleno^{25,76–78} analog, although they share similar behavior, have proved to be more stable however than their thio analogs and have provided reasonable yields of alkylated (Scheme 81, entries f and g)^{64,65,76,77} or hydroxyalkylated (Scheme 82, entries e-g)^{25,78} compounds on reaction with alkyl halides and epoxides respectively.

The resulting orthothio- and orthoseleno-esters are valuable precursors of carboxylic esters and acids.^{383,386} The former conversion requires quite drastic conditions (HgCl₂, HgO, MeOH-H₂O, 100 °C, 6 h; Scheme 84, entry b),³⁸³ whereas the same transformation takes place under very mild conditions with the seleno derivatives (aq. H₂O₂, THF, 20 °C, 0.1 h; Scheme 81, entry g).⁷⁶

Tri(thio)- and tri(seleno)-alkyllithiums possess a high propensity to act as carbenoids, and they have been found to add to various nucleophilic alkenes, leading to 1,1-di(thio)-³⁸² and 1,1-di(seleno)-cyclo-propanes (Scheme 85, entry c).^{64,65}



iv,
$$I = \frac{1}{1} = \frac{1}{100} = \frac{1}{100}$$

(a)
i R SPh RCO₂R¹ (ref. 361, 362)
SPh RCO₂R¹ (ref. 361, 362)
RX MeI BuCl PrⁱI

$$\%$$
 95 35 30
(b)
RX MeS S iii RCO₂R¹ (ref. 363)
RX BuⁿI PhCH₂Cl R¹ = Et, 97%;
 $\%$ 78 92 R¹ = Me, 87%
iv Pr S S (ref. 361)
RX = PrⁱI, 64% Pr S S (ref. 361)

i, LiC(SPh)₃; ii,
$$\underset{\text{Li}}{\overset{\text{MeS}}{\xrightarrow{S}}}$$
, -35 °C, 3 h, then 0 °C, 16 h; iii, HgCl₂, HgO, 100 °C, 6 h;
iv, Li $\underset{\text{S}}{\overset{\text{S}}{\xrightarrow{S}}}$, -78 to 20 °C

Scheme 84



1.3.5 REACTIONS INVOLVING α -METALLOALKYL SULFOXIDES AND SELENOXIDES

1.3.5.1 Reactions Involving Sulfoxides

1.3.5.1.1 Reactions involving alkyl sulfoxides

(i) Alkylations of α -metalloalkyl sulfoxides and selenoxides

The reactions between α -metalloalkyl sulfoxides and electrophiles^{8,31-34,387} have been extensively studied. Although alkylations of the sodium or potassium salts of dialkyl sulfoxides are not always very efficient since α, α' -dialkylated sulfoxides^{388,389} are often produced (or stilbene in the case of methyl-sulfinyl carbanion and benzyl bromide³⁹¹), those employing the lithioalkyl aryl sulfoxides work more efficiently with alkyl or allyl halides^{32,34,390,392,393} and with epoxides.³⁹⁴⁻³⁹⁶ Typical examples of these alkylations, allylations and hydroxyalkylations (from epoxides) are illustrated in Scheme 86.

The chirality present at the sulfur atom of, for example, *p*-tolylsulfinyl carbanion leads to the formation of a diastereoisomeric mixture of γ -hydroxy sulfoxides on reaction with prochiral epoxides³⁹⁴ (Scheme 87). This reaction, coupled with the efficient desulfurization of the resulting sulfoxide by Raney nickel, has been used ingeniously³⁹⁴ for the enantioselective synthesis of *trans*-2-methylcyclohexanol from epoxycyclohexane and (+)-*p*-tolylsulfinylmethyllithium (Scheme 87).

Both the metallation of sulfoxides and the reaction of α -lithio sulfoxides in THF are known to be highly stereoselective.^{32,34,390,397-399} It has been found that, on reaction with base, the diastereotopic methylene protons of benzyl methyl sulfoxide exchange at different rates, the relative ratio being





i, BuLi, THF, -78 °C; ii, D₂O; iii, oxidation

Scheme 88

15:1.^{34,390,397,398} Since this discovery the stereochemistry of the H:D exchange, as well as the stereochemical course of deuteration (Scheme 88) and of alkylation (Scheme 88; Scheme 89; Scheme 90; Scheme 91; Scheme 92), a number of acyclic (Scheme 89)^{392,397,400-406} and cyclic sulfoxides (Scheme 90; Scheme 91; Scheme 92)^{390,398,407-413} have been intensively studied and have been the subject of intense controversy.³⁸⁷



Scheme 90



```
i, Bu<sup>n</sup>Li, THF, -78 °C, 0.3 h; ii, MeI, -30 °C; iii, (MeO)<sub>3</sub>PO, 20 °C; iv, (MeO)<sub>3</sub>PO, 10 LiClO<sub>4</sub>;
v, MeI, 20 °C; vi, (MeO)<sub>3</sub>PO, 10 LiClO<sub>4</sub>, 20 °C
```

It was found^{34,397,402,403} (see ref. 34, and ref. 387 for a correction) that for α -deuterated benzyl methyl and benzyl *t*-butyl sulfoxides of known configurations, methyllithium in THF at -78 °C abstracts the *pro-*(*S*) hydrogen of (*S*)-benzyl methyl sulfoxide (1) with a selectivity of 15:1 (Scheme 88 and Scheme 89, entries a), whereas the same reaction performed on the (R)-benzyl *t*-butyl sulfoxide (2; Scheme 88 and Scheme 89, entries b) possessing the same chirality as (1) (it should be noted that the (*R*,*S*) notation appears different in methyl and *t*-butyl sulfoxides because of the definition) leads almost exclusively to the removal of the *pro-*(*R*) hydrogen (selectivity 117:1; note that an incorrect absolute configuration of the (+)-(2) sulfoxide was quoted in the original paper, ref. 397).^{34,387} Finally the *pro-*(*S*) hydrogen atom of (*S*)-benzyl *t*-butyl sulfoxide is abstracted preferentially by a kinetic factor of 1.7:1. As far as alkylation is concerned, the reactions proceed with high stereoselectively, in all cases by inversion of configuration of the originally produced benzyllithiums (Scheme 89); deuteration (or protonation) on the other hand usually occurs with retention with benzyl phenyl sulfoxides and with inversion with benzyl *t*-butyl sulfoxides (Scheme 88).^{34,410}

The stereochemistry of the reaction products is dependent on the nature of the α -sulfinyl carbanion. Thus: (i) its kinetic acidity, controls the stereochemistry of the organometallics initially formed; (ii) its thermodynamic acidity defines the stereochemistry or the conformation of the intermediate organometallic; and (iii) reactivity of the organometallic towards the electrophile controls the stereochemistry of the product. The alkylation reaction has proved to be far more selective than is the metallation. Therefore the contribution of kinetic acidity can be neglected because the carbanion in THF has sufficient time to reorganize into its most stable conformation before it reacts with an electrophile.³⁸⁷ The α -sulfinyl benzyllithiums produced from S_(S)-(1) and S_(R)-(2) should adopt the more stable conformations shown in



Schemes 88 and 89, in which the lithium cation is coordinated to the sulfinyl oxygen and, in the former, with the sulfur lone pair also. Methyl iodide, which is a nonpolar substrate, prefers to react at the more nucleophilic side which is *anti* to the sulfur lone pair. It has been suggested,³⁸⁷ in other words, that the *si* and the *re* faces of the benzyllithium derived from S(s)-(1) are hard and soft reaction centers respectively, whereas the same *si* face has both hard and soft reaction centers; therefore the soft methyl iodide is expected³⁸⁷ to react at the *re* face in the first case and at the *si* face in the second one, which in fact explains the observed products. This type of rationalization also explains³⁸⁷ the stereochemistry of the protonation (deuteration) of these organometallics (for a controversal discussion see ref. 34).

The metallations of rigid molecules such as 4-*t*-butylthiacyclohexane take place more efficiently with hydrogens syn to the S—O bond³⁹⁰ (Scheme 90), and the alkylation with alkyl halides gives products of anti attack with respect to the S—O bond^{390,397,407-411} (Scheme 90; Scheme 91). The last reaction has been purposely used by Marquet⁴⁰⁸⁻⁴¹⁰ in an original synthesis of \pm -biotin and its analogs (Scheme 92).^{409,410}

The overall process takes advantage $also^{409,410}$ of: (i) the selective oxidation of the sulfur in the corresponding sulfide, leading to the predominant formation of the isomer with the S--O bond *cis* to the junction hydrogens, which is the isomer required for the stereoselective alkylation; and (ii) of the easy reduction of the alkylated sulfoxide to the desired sulfide which has been *inter alia* achieved with Ti³⁺ (Scheme 92). The alkylation yield does not exceed 30% if the reaction is carried out with *n*-butyllithium in THF since both starting material and products resulting from β -elimination of the sulfoxide are isolated.⁴⁰⁸⁻⁴¹⁰ This elimination reaction can be suppressed if methyllithium is used instead^{409,410} but the yield remains quite low, even when a large excess of the alkylating agent is added (Scheme 92, entry a).

Alkylation takes place predominantly on the less substituted site of dialkylsulfoxides, 388,407,409,410 and this feature has been used advantageously for the synthesis of biotin analogs bearing a methyl group in the α' -position on the thiophane ring (Scheme 92, entry b).⁴¹⁰

Although reactions with epoxides⁴¹³ proceed with a similar stereochemical outcome to that of the alkylation reaction, the deuteration (D_2O) ,^{397,402,403,406} and carbonation⁴⁰⁵ reactions occur predominantly with opposite stereochemistry.

Alkylation of Carbon

The finding that the stereochemistry of such reactions can be drastically affected as shown in Scheme 93, by: (i) a solvent change, 392,404 (ii) addition of macrocyclic polyethers⁴⁰⁴ (especially if the reaction is carried out in TMEDA), or (iii) addition of lithium salts^{392,406,407} has supported the idea that the stereochemistry may be largely governed by cation-carbanion interactions,^{390,413} and more particularly by the ability of Li⁺ to form a chelate structure involving the carbanionic carbon and the sulfoxide oxygen (Scheme 89; Scheme 91, entry h, structures 3 and 4).^{404,407}

This chelate would undergo electrophilic displacement of Li⁺ with predominantly retention of configuration when the electrophile-donating species can itself act as a chelating agent but with inversion in the absence of this chelating ability.^{406,407} Strong support for these hypotheses has been provided by the experiments with chelating methylating agents such as trimethyl phosphate^{406,407} as compared to nonchelating methyl iodide (Scheme 91, compare entries b to a and entry f to d and e; Scheme 93; similar differences between chelating D₂O and nonchelating proton donors Et₃ND⁺ and D⁺-sponge have been reported⁴⁰⁶).

	$Ph \underbrace{\bigvee_{s=1}^{O}}_{Me} \frac{i, Bu^{n}Li}{ii, see below}$	$Me \xrightarrow{O^{-}}_{H} H$ $Me \xrightarrow{S^{+}}_{Ph} \cdot \cdot$	H Me Ph	
(a)	MeI, THF, 0.1 or 0.5 h; 100%	95 %	5%	(ref. 404)
(b)	MeI, THF, excess HMPA; 100%	90%	10%	(ref. 404)
(c)	MeI, [2.2.2]macrobicyclic polyether, 2 h	83%	17%	(ref. 404)
(d)	MeI, TMEDA, [2.2.2], 1 h	80%	20%	(ref. 404)
(e)	(MeO) ₃ PO, 0 °C	40%	60%	(ref. 406)
(f)	(MeO) ₂ SO ₄ , 0 °C	95%	5%	(ref. 406)

Scheme 93

This interpretation satisfactorily explains all the results reported so far with the exception of those involving the methylations of α -lithiotiepane 1-oxide and thiocane 1-oxide (the seven- and eight-membered homologs of thiacyclohexane)⁴¹³ which provide mainly the compounds having *cis* stereochemistry between the methyl group and the S—O bond (*cis:trans* ratio of 93:7 and 83:17 respectively). The



Scheme 94

reactions of oxiranes with these species are slower, but the stereochemical results are substantially the same as in alkylation with methyl iodide.^{396,413}

Alkylations⁴¹⁴⁻⁴¹⁶ of the dianions derived from β -hydroxyalkyl sulfoxides, shown in Scheme 94 and in Scheme 95, occur stereoselectively.^{415,416}



Scheme 95

The steroselectivity increases by increasing the bulkiness of the R and R¹ groups, and is not affected when the reactions are carried out in the presence of TMEDA or HMPA (2 equiv.).⁴¹⁵ The alkylation is therefore controlled by the stereochemistry of the carbon bearing the hydroxy group rather by that of the sulfinyl group.⁴¹⁶ The preference for *threo* stereochemistry has been explained by the six-membered chair transition states (5 and 6; Scheme 95) produced by the chelation of both oxygens of the hydroxy and sulfinyl groups to the lithium cation. The axial attack of the alkylating agent from the upper side, due to the steric interaction with the axial R group, leads to the selectivity shown in Scheme 95 (entries a–c), whereas equatorial attack on (6) leads to the thermodynamically more stable *threo* derivatives shown in Scheme 95 (entries d–f). In contrast to the reported reversal of the stereochemistry in the reaction of α -lithiobenzyl methyl sulfoxide with methyl iodide and trimethyl phosphate respectively,^{406,407} the reaction of the dianion derived from β -hydroxyalkyl sulfoxide with trimethyl phosphate leads to the same *threo* stereoisomer produced with methyl iodide⁴¹⁶ supporting therefore the chelated structures shown in Scheme 95 (entries c and f), afforded *threo* β -hydroxyalkyl sulfoxides which are useful precursors of optically active *cis*-epoxides.⁴¹⁶ The utility of the above mentioned reactions has been demonstrated in the asymmetric synthesis of (+)- and (-)-cis-7,8-epoxy-2-methyloctadecanes, the female-produced pheromone of the gypsy moth (Scheme 95, entry f).⁴¹⁶

Sulfoxides are known to produce alkenes on heating, by elimination of sulfenic acids.^{8,31,32,233,417-420} This reaction has been applied successfully to a large variety of alkyl sulfoxides and functionalized alkyl sulfoxides. Many of these sulfoxides have been synthesized from simpler sulfoxides by alkylation of the corresponding lithium salts. Thus, methylbenzyl³⁷⁸ (Scheme 47), 3,4-dioxomethylenebenzyl,³⁷⁸ phenylsulfonylmethyl,^{380,421} fluoromethyl,⁴²² chloromethyl,⁴²³ iodomethyl⁴²⁴ and dichloromethyl⁴²⁵ phenyl sulfoxides have been metallated with lithium amides^{378,421-423,425,426} or butyl-lithium³⁷⁸ in THF or in DME³⁷⁸ then alkylated,^{378,422-425} benzylated^{378,421,422,424,425} and allylated (Scheme 96).^{378,380}



Scheme 96

The reactions are usually carried out at 20 °C except with α -halomethyl sulfoxides whose lithium salts are particularly unstable and which have instead been reacted in THF at -78 °C in the presence of HMPA (Scheme 96).⁴²²⁻⁴²⁵ α -Lithiobenzyl phenyl sulfoxide has proved to be the most nucleophilic species among the different functionalized sulfoxides used (Scheme 47).³⁷⁸ Thus, it has been smoothly alkylated, allylated or benzylated in THF and is the only one which has been reported to undergo alkylation with secondary alkyl halides.³⁷⁸ The sluggish alkylation of the organolithium compound derived from di(sulf-inyl) methane with an unactivated alkylating agent dictated the use of HMPA and elevated temperatures, although no complications were encountered.³⁷⁸

(ii) Synthetic uses of alkylated sulfoxides

(a) Synthesis of alkenes and dienes via sulfoxide elimination reactions. The pyrolysis of various sulfoxides takes place between 80 and 150 °C.^{378,421,422,425,426} In some instances the elimination reaction has been carried out in the same pot and the same solvent in which the alkylation reaction has been achieved.³⁷⁸ In other cases trimethyl phosphite, a scavenger of sulfenic acid, has been employed to avoid decomposition and to facilitate isolation of the product.³⁷⁸ Aryl and phenylthio substituents α to the

sulfinyl moiety facilitate the elimination of sulfenic acid whereas a phenyl sulfinyl moiety decelerates elimination.³⁷⁸ The ease of hydrogen abstraction decreases in the order allylic > benzylic > secondary > tertiary. Thus in the absence of conformational restraints there exists the possibility of regioselection. Finally, in several cases (*E*)-alkenic double bonds are produced selectively.³⁷⁸

The method employing the sequential alkylation and elimination of sulfoxides, reported above, has been successfully applied to the synthesis of dienes (Scheme 86, entry c),³⁹⁵ aryl alkenes and dienes (Scheme 47),³⁷⁸ vinyl sulfides,³⁷⁸ α , β -unsaturated sulfoxides,³⁷⁸ vinyl fluorides⁴²² and vinyl chlorides (Scheme 96, entry a),⁴²³ and 1,1-dichloro-1-alkenes (Scheme 96, entry b).⁴²⁵

(b) Reactions employing the reduction of alkyl sulfoxides. Reduction of sulfoxides to sulfides have been successfully carried out with a large variety of reducing agents including TiCl₃,⁴⁰⁸⁻⁴¹⁰ Me₃SiI, NaHSO₃³⁹⁶ and PI₃,⁴²⁷ and they have been used advantageously, as already mentioned, for the synthesis of biotin (Scheme 92, entry a).⁴⁰⁸⁻⁴¹⁰ The PhSO/H exchange can be performed with Raney nickel⁴¹⁸ or with lithium in ethylamine;⁴¹⁴ the latter procedure has allowed the synthesis of linalol and nerolidol (along with 25% of its Z-isomer; Scheme 94).⁴¹⁴ A number of other interesting transformation of sulfoxides can be cited, including: (i) a synthesis of trisubstituted alkenes which takes advantage of the regio-and stereo-selective γ -substitution of allylic sulfoxides with lithium dimethylcuprate;⁴²⁸ and (ii) the transformation of primary alkyl halides into alkanethiols which have a carbon chain longer by one methylene unit than the original halides (Scheme 97).⁴²⁹ The key step in this reaction involves the alkyl-ation of cyclohexanone dimethyl acetal *S*-oxide, followed by acid hydrolysis and reduction of the resulting compound with lithium aluminum hydride (Scheme 97).



i, Et₂NLi, THF, -15 °C; ii, C₁₁H₂₃I, -78 °C, then 20 °C, 2 h; iii, H₃O⁺; iv, LiAlH₄; v, H₃O⁺

Scheme 97

1.3.5.1.2 Reactions involving allylic sulfoxides

A range of diverse substituted allyl phenyl sulfoxides, which are available by addition of dimethyl cuprate to allenyl sulfoxides⁴³⁰ or more generally by alkylation of simpler analogs, 16,235,236,414,431 have been converted into allylic alcohols possessing predominantly (*E*)-stereochemistry (*E*:*Z* ≥ 90:10) following simple admixture with trimethyl phosphite^{235,236} or thiophenolate (MeOH, 60 °C, 7 h; Scheme 98; Scheme 99; for a lower selectivity see Scheme 98, entry a).⁴³¹

Metallations of allyl sulfoxides have been achieved at -78 °C with *n*-butyllithium in THF⁴³¹ or more efficiently with LDA under similar conditions.^{16,235,236} Alkylations with primary alkyl iodides^{16,235,236,431} and allylations with allyl^{235,430} or phenyl²³⁶ bromides proceed in good yield, usually below -20 °C and in less than 6 h. The reaction takes place predominantly at the α -site but there is often competition from γ -alkylation leading to α -alkenyl sulfoxides (Scheme 98; Scheme 99). Changing the solvent, or introducing metal ion complexing agents, such as TMEDA were unsuccessful in changing the α : γ ratios observed in THF.

 α -Alkylation occurs almost exclusively when the γ -carbon is dialkylated (Scheme 99, entry e) or with 1-phenylsulfinyl-2-cycloalkenes (Scheme 98, entries e and f). Apart from these cases the α : γ ratio ranges between 80:20 and 55:45. In general, allyl halides react less selectively than alkyl halides,^{235,236} which in fact lead to the highest α : γ ratio. This competing reaction lowers the overall yield of allyl alcohols resulting from the reaction of allylic sulfoxides with thiophilic reagents. Therefore the route employing: (i) the alkylation of allyl sulfides, (ii) their transformation to the corresponding sulfoxide, and (iii) the rearrangement of the latter to allyl alcohols, which has been reported in Section 1.3.2.2.2, is preferred to that just described due to higher regiocontrol. The first reported set of reactions has been ingeniously applied to the synthesis of a series of cyclopentene-3,5-diols, which are potential precursors of the prostaglandins (Scheme 100).¹⁶⁶

Alkylations of 1-alkenyl aryl sulfoxides have attracted little attention compared to other cases.^{432,433} However it has been found that α -metallation occurs on reaction with LDA and that alkylation, allylation and benzylation occur there as a result (Scheme 101). Interestingly, γ -alkylation resulting from a competitive γ -metallation reaction has not been observed. The last mentioned reaction, coupled with treatment



i, LDA, THF, -78 °C, RX, -60 or -8 °C, 1 or 0.2 h; ii, (MeO)₃P, MeOH, 20 °C, 12 h; iii, (MeO)₃P, MeOH, 20 °C, 1 h; iv, LDA, THF, -78 °C, -50 or -40 °C, 0.5 or 2 h; v, (MeO)₃P, MeOH, 1 h



i, LDA, THF, -60 °C; ii, R³X, -50 or -30 °C, 0.5 to 4 h; iii, (MeO)₃P, MeOH, 12 h, 20 °C



of the α -methylated vinyl sulfoxides with lithium tetramethylpiperidide, provides an original synthesis of allenes (Scheme 101).⁴³³



i, LDA, -78 °C, 0.3 h; ii, RX; iii, LITMP, THF, -100 °C, 0.8 h; iv, NH₄Cl; v, LDA, -78 °C; vi, MeI; vii, LITMP, THF, -100 °C, 0.8 h; viii, NH₄Cl

Scheme 101

1.3.5.2 Alkylations of α -Metallo-alkyl and -vinyl Selenoxides

 α -Lithioalkyl selenoxides, which are available from *in situ* oxidation-metallation of the corresponding alkyl aryl selenides, have been alkylated at -78 °C with methyl iodide and allyl bromides (Scheme 102).^{98,112,113} Unfortunately the utility of this process is limited,¹¹³ as in the case of the corresponding sulfoxides, by the lack of regioselectivity of the subsequent *syn* elimination (Scheme 102, compare entries a and b to entry c) since two alkenes are formed, unless structural features dictate the formation of only a single isomer (Scheme 102). The α -deprotonation of vinyl phenyl selenoxide with LDA has been studied¹⁰⁷ at -78 °C for different time spans. A maximum of 50% of the expected methylation product, α -phenylseleno-2-propene, was isolated after reduction of the intermediate selenoxide with NaI in acetic acid. Deprotonation, even in the presence of methyl iodide, gave poor results.¹⁰⁷



1.3.6 REACTIONS INVOLVING SULFONES

1.3.6.1 Reactions Involving *a*-Metalloalkyl Sulfones

1.3.6.1.1 Alkylations of α -metalloalkyl sulfones

Although the sulfone functional group has been an integral part of organic chemistry for more than a century, it has been only within the last few decades that a more diverse range of chemistry involving this group has been discovered and used, particularly in the total synthesis of various natural products. Various aspects of sulfone chemistry have been reviewed earlier by Magnus³⁶ and by Durst^{37,38} Typically, sulfones exhibit a high degree of chemical and thermal stability. They are stable to oxidizing agents and to acids, but can be reduced to sulfides.^{36–38} The sulfonyl group can be replaced^{36,37} by hydrogen using Raney nickel, lithium in ethylamine or under less severe experimental conditions (6% Na/Hg). On reaction with a base they are usually readily metallated, and when the hydrogen attached to the carbon β to the one bearing the sulfonyl group is sufficiently acidic (*e.g.* α to a carbonyl group or to a C—C double bond) elimination of sulfinic acid occurs leading to a new C—C double bond. Furthermore, the sulfone oxygen atoms show only relatively weak bonding with electrophiles as compared to sulfoxides and selenoxides. These possibilities and this chemistry, as well as the aptitude of some specific sulfones to take part in the Ramberg–Bäcklund reaction⁴³⁴ and extrude SO₂ thermally or photochemically, have led to an increasing use of sulfones in synthesis.^{434,435}

The alkylation of sulfones has played an important role in this development. Most of this work has been performed with aryl sulfones, and in only rare cases have triflones $(RSO_2CF_3)^{436}$ or dialkyl sulfones been employed. $^{437-440}$

Sulfones, including dimethyl sulfones, have been routinely converted into their monoanions by a variety of basic reagents which *inter alia* include: (i) Grignard reagents, $^{439-441}$ (ii) organolithiums, $^{36-38,440,442}$ (iii) potassium *t*-butoxide, 442 (iv) sodium hydride, 391,436 (v) metallo amides, 438,443,444 and (vi) in rare cases, with phenylsodium or phenylpotassium. 442

The presence of another group such as an aryl,^{436,445-447} an allyl,⁴⁴⁸ or an isocyanide,⁴⁴⁹ capable of further stabilization of the carbanionic center allows the use of less strong bases such as 50% sodium

		R SO ₂ CF ₃	-	$\underset{R^{1}}{\overset{SO_{2}CF_{3}}{\bigvee}}$	v Entry (d), 100%	Ph N ₃
Entry	R	$R^{I}X$	Conditions	Yield (%)	(ref. 436)	
(a)	Н	Ph(CH ₂) ₂ Br	i	73		
		PhCH ₂ Br	i	18 (48% dia	alkylation)	
(b)	Н	Pr ⁱ I	ii	60		
(c)	Me	MeI	iii	60		
(d)	Ph	PhCH ₂ Br	iv	70		

hydroxide solution under phase-transfer catalysis,^{445,446,448} sodium hydroxide in DMF (Scheme 103, entries a and b)⁴⁴⁷ or even potassium carbonate in acetonitrile (Scheme 103, entry d).

i, R¹X, NaH, DMF, 25 °C, 10–18 h; ii, R¹X, NaH, glyme, HMPA, reflux, 8 h; iii, R¹X, NaH, glyme, 25 °C, 2.5 h; iv, K₂CO₃, MeCN, 84 °C, 36 h; v, 2 NaH, *p*-TolSO₂N₃, 0 °C, 0.5 h

Scheme 103

 α -Sulfonyl Grignard reagents are not usually alkylated efficiently; for example phenylsulfonylmethylmagnesium bromide does not undergo alkylation⁴⁴¹ with *n*-hexyl iodide or *t*-butyl chloride. However it reacts with benzyl chloride, but leads to the corresponding product in only very modest yield (34%).⁴⁴¹ Similarly α -(*p*-tolylsulfonyl)isopropylmagnesium bromide is alkylated only to the extent of 38% when treated with 3-chloropropyl *p*-toluenesulfonate, but steric hindrance may intervene in this instance (for an intramolecular version of this reaction see Scheme 109).⁴⁴¹

The alkylations of related organolithium derivatives bearing hydrogen or alkyl groups on their carbanionic centers proceed with much better yields, but often involve primary alkyl (Scheme 103; Scheme 104, entries a, b, and d; Scheme 105, entries a and b; Scheme 106, entries a-d; Scheme 107; Scheme 108, entries a and b; Scheme 109),^{13,24,138,436,443,450,451-461} benzyl^{138,436,437,451,454,456,459} or allyl^{138,445,452,454,455,459,460,462} halides or sulfonates as well as 2,3-dihalogeno- and 1,3-dihalogeno-1propenes⁴⁶³ and related dihalides (Scheme 110).⁴⁶³

In one case considerable difficulty was encountered in the alkylation of the anion of methyl phenyl sulfone with *o*-chlorobenzyl bromide since, under a variety of conditions, mixtures containing starting sulfone, mono- and di-alkylated products were obtained.³⁷ Allylation of α -lithio sulfones with allylvinyl dihalides⁴⁶³ has been reported to produce a large amount of the monoallylated compound in addition to some diallylated derivative. The formation of the latter compound can be avoided however if the reaction is carried out in the presence of copper(I) iodide (Scheme 110, entry a).⁴⁶³

Although alkylations of α -metallo sulfones with sodium iodoacetate proceed efficiently (Scheme 105, entry b),⁴⁵⁵ 1:1 mixtures of products are formed on reaction with ethyl iodoacetate in THF (Scheme 110, entry b).⁴⁶¹ The mixtures arise from the expected alkylation reaction and from competitive addition to the carbonyl group of the ester. The product resulting from the alkylation reaction is almost exclusively produced however if the above mentioned reaction is performed in the presence of HMPA (10%) and a trace amount of potassium iodide (Scheme 110, entry c). The resulting β -sulfonyl ester has been further transformed to an α , β -unsaturated ester, taking advantage of the facile elimination of sodium phenyl sulfinate (Scheme 110, entry c).

Alkylations of sulfones with secondary alkyl halides have been described only rarely.^{456,458} They usually proceed with modest yields and require both long reaction times and the presence of HMPA (Scheme 106, entry b). The presence of an aryl group on the carbanionic center softens this site however, and usually favors the alkylation reaction.^{445,461,463}

Mono- or di-alkylation of dimetalloalkyl sulfones has been reported from time to time. Apparently the geminally dilithiated alkyl aryl sulfones are more reactive towards alkyl halides than the monolithiated species. Therefore they can be monoalkylated (Scheme 111, entries a and b)^{443,464} or dialkylated (Scheme 111, entries c and d),^{438,464,466,467} depending upon the conditions employed (*i.e.* excess of alkyl-ating agent and more severe conditions for dialkylation) but success also depends upon the structures of the sulfones and the nature of the alkylating agents. Thus, ferrocenylmethyl phenyl sulfone is efficiently dimethylated or transformed to the corresponding cyclopropane derivative on sequential treatment with 2



 $RX = Ph(CH_2)_2Br$, 58% overall; $RX = PhCH_2Br$, 62% overall $RX = C_5H_{11}C = C(CH_2)_3I$, 72% overall, moth pheromone



Scheme 104

equiv. of *n*-butyllithium and 2 equiv. of methyl iodide or 1 equiv. of 1,2-dichloroethane respectively, but it is monoalkylated with *n*-butyl or benzyl bromide under similar conditions (Scheme 111, entry b).⁴⁶⁵

Reactions of dimethyl,⁴³⁸ and dibenzyl⁴³⁷ sulfones (Scheme 112) and of tetrahydrothiophene 1,1-dioxide (Scheme 107, entry a)⁴⁵³ with 2 equiv. of base, followed by 2 equiv. or more of alkyl halides, take another course since α, α' -dialkylated rather than α, α -dialkylated derivatives are produced.

1.3.6.1.2 Synthetic applications of alkyl sulfones

Alkylation or allylation reactions of α -metalloalkyl sulfones followed by oxidation of the carbon center bearing the sulfonyl group (BuⁿLi and dimethyl disulfide,⁴⁵¹ or a three-fold excess of molybdenum peroxide⁴⁶⁸ or a slight excess of bis(trimethylsilyl) peroxide, BTSP⁴⁶⁹) have led to novel routes to



i, 2 BuⁿLi, -78 °C; ii, RX, -78 °C, 0.5 h, then 25 °C, 2-3 h; iii, Raney Ni; iv, Hg/Na; v, 2.2 BuⁿLi, THF-TMEDA; vi, R¹X; vii, 2.2 BuⁿLi, THF-TMEDA; viii, 1.2 ICH₂CO₂Na; ix, *p*-TsOH, benzene, Δ ; x, 2.5 equiv. Et₃N, benzene, 20 °C

ketones.⁴⁵¹ The first reagent has been used,⁴⁵¹ *inter alia*, in a synthesis of the sex pheromone of Douglas fir tussock moth (Scheme 104, entry a). Furthermore by taking advantage of the fact that one oxygen of BTSP becomes the carbonyl oxygen of the product, it is possible to label specifically this oxygen by using the readily available (TMS¹⁸O)₂.⁴⁶⁹ Reduction of the sulfonyl group, subsequent to alkylation, has been used for the synthesis of 24-hydroxycholesterols (Li/H₂NEt; Scheme 104, entry c),⁴⁴³ (+)-dihydro-compactin (NaHg; Scheme 111, entry a)⁴⁶⁴, sesquifenchene (Li/EtNH₂)⁴⁶², α -santalene (Na/Hg, HMPA),⁴⁷⁰ cyclopropane (Scheme 108, entry a)⁴⁴⁴ or phenylcyclopropane (Scheme 109),¹³⁸ benzocyclobutane (Na/Hg; Scheme 107, entries c and d)⁴⁶⁰ as well as for the coupling of: (i) an alkyl side chain with a functionalized alkyl or alkenyl halide (see for example Scheme 108, entry a);⁴⁶³ and (ii) a functionalized side chain with an alkyl, alkenyl or benzylic halide (Scheme 105, entry a; Scheme 106, entry a; Scheme 107, entries c and d).^{454,456,458,460} In the above mentioned transformations the sulfone has therefore played the role of an alkyl- or functionalized alkyl-metal.

Thus β -, γ - and ω -hydroxy,⁴⁵⁴ β -dioxy- and ω -carboxy-alkyl sulfones have proved to be valuable precursors of alcohols (Scheme 105, entry a),⁴⁵⁴ acetals, aldehydes or ketones (Scheme 106, entries ad)^{456,458} and carboxylic acids,⁴⁷¹ respectively, whereas 1-(ω -alkenyl) halides have been synthesized⁴⁶³ from 1,3-dichloropropene and α -metalloalkenyl sulfones (Scheme 110, entry a). 1-(ω -Alkenyl)-1-sulfonylbenzocyclobutanes are good precursors of quinodimethanes, which are themselves particularly useful reactive intermediates.⁴⁶⁰ Although an internal Diels–Alder reaction leading to the tricyclic compound shown in Scheme 107 (entry c), takes place efficiently on thermolysis of the compound missing the sulfonyl group, the reaction takes another course with the highly electron-deficient diene bearing the sulfonyl moiety which instead leads to the aryl vinyl sulfone shown in Scheme 107 (entry d).

Furthermore, several syntheses of α,β -unsaturated carbonyl compounds have involved the alkylation of suitably functionalized sulfones, followed by the base-promoted elimination of the sulfenate moiety on related alkyl sulfones bearing an oxygen atom at the γ -position (Scheme 106, entries b-d; Scheme 110, entry c; Scheme 105, entry b).^{455,456,458,461} Particularly interesting is the synthesis of nuciferal (Scheme 106, entry d)⁴⁵⁸ which involves sequential alkylation of γ,γ -dioxy sulfones, deblocking of the acetal moiety and sulfenate elimination. In this and related reactions γ,γ -dioxy- α -metalloalkyl sulfone has played the role of a masked β -metallo- α,β -unsaturated carbonyl compound (MC=C-C-C-C), which is not a directly available synthon.



Nonfunctionalized alkenes such as terminal alkenes are not available on similar treatment of, for example, 2-sulfonylalkanes with bases. However, these have been obtained on alkylation of α -metallo sulfones with trimethylsilylmethyl iodide⁴⁵² and subsequent reaction of the resulting β -silyl sulfone with



i, 2 BuⁿLi, THF, -80 °C; ii, excess MeI; iii, BuⁿLi; iv, LiAlH₄, dioxane, Δ; v, BuⁿLi, Et₂O, -78 °C, 1 h;
vi, RBr, -78 °C, then 20 °C, 2 h; vii, hv (254 nm), CH₂Cl₂; viii, -SO₂; ix, MeLi, THF, -78 °C; x, RX, -78 to 20 °C, <1 h; xi, Na/Hg, MeOH, NaHPO₄; xii, 250 °C, 24 h; xiii, 250 °C, c-C₆H₁₂, 20 h; xiv, BuⁱOK, THF, 0 °C, 5 h

fluoride ion (Scheme 106, entry f). A related procedure involving the alkylation,⁴⁵² the allylation,⁴⁵² and the reaction of α -metallo- β -(trimethylsilyl)alkyl phenyl sulfones with epoxides⁴⁷¹ allows the synthesis of 1-alkenes,⁴⁵² 1,4-dienes⁴⁵² and cyclopropyl sulfones⁴⁷¹ respectively.

Extrusion of sulfur dioxide from the products resulting from the alkylation of tetrahydrothiophene 1,1dioxides^{453,472} and the 2-phenylthietane 1,1-dioxides,⁴⁵⁹ using *n*-butyllithium and lithium aluminum hydride⁴⁵³ or photochemically⁴⁵⁹ leads to the synthesis of the corresponding cyclobutenes⁴⁵³ and cyclopropanes,⁴⁵⁹ respectively (Scheme 107, entries a and b).

Internal alkylation takes place on reaction of α -haloalkyl alkyl sulfones with bases and produces episulfones as intermediaries. These intermediates are usually unstable and provide alkenes by thermal extrusion of sulfur dioxide. This reaction, known as the Ramberg–Bäcklund rearrangement, has been widely used in synthesis and has been thoroughly reviewed (Scheme 107, entry e).^{36,38,434,473}



i, Bu^tOK, Bu^tOH, Δ, 6 h; ii, NaNH₂, DME, 60 °C, 2 h; iii, BuⁿLi, THF; iv, R¹X, 0 °C; v, Raney Ni, EtOH; vi, Bu^tOK, DMSO; vii, LDA, THF, -78 to 20 °C

Treatment of γ -halo^{444,474,475} or γ -tosyloxy sulfones¹³⁸ with base results in their conversion to cyclopropyl sulfones in good yields (Scheme 108; Scheme 109).^{35,37,138,450,475} The base/solvent systems that have been used in this reaction include Bu⁴OK/Bu⁴OH,^{442,475} NaNH₂/glyme (but not NaNH₂ in ether),⁴⁴⁴ LDA/THF¹³⁸ and BuⁿLi/THF (Scheme 108, entry a; Scheme 109, entry a).⁴⁷⁴ Treatment of the *exo* bromide shown in Scheme 108 (entry c) with base results in the rapid formation of the sulfonyl quadricyclane,⁴⁷⁶ while the *endo* bromide, whose carbanion cannot assume the suitable semi- ω conformation in the transition state,⁴⁷⁷ does not cyclize under similar or even forcing conditions (Scheme 108, compare entry d to c). Alkylation, benzylation or allylation of the resulting cyclopropyl sulfone has in turn been performed successfully¹³⁸ (Scheme 108, entry a). Its reduction to cyclopropane was achieved¹³⁸ with Raney nickel (Scheme 108, entry a). Cyclopropyl and cyclopentyl phenyl sulfones have been obtained directly⁴⁷⁸ from (phenylsulfonyl)methylenedilithium and 1,2-dichloroethane and 1,4-diiodopentane, and from related ferrocenyl species and dichloroethane (Scheme 111, entry b).⁴⁶⁵ trans-2-Substituted-cyclopropyl sulfones have been produced in up to 80% yield by addition of allylic, propargylic, aryl and



i, PhSNa; ii, NaBH₄; iii, H₂O₂, HOAc; iv, K₂CO₃, MeOH; v, TosCl, pyridine; vi, LDA, THF; vii, Na/Hg; viii, RMgX



Scheme 110

benzyl Grignard reagents to 3-bromo-1-(phenylsulfonyl)-1-propene (Scheme 109, entry b).⁴⁷⁹ Unfortunately however, alkyl Grignard reagents do not lead to the corresponding cyclopropyl sulfones (Scheme 109, entry c).⁴⁷⁹

Cyclobutyl,¹³⁷ cyclopentyl¹³⁷ and cyclohexyl^{137,480} aryl sulfones can also be prepared by the analogous cyclization involving the corresponding halo¹³⁷ or sulfonyloxyalkyl sulfones (Scheme 108, entries b and e).⁴⁸⁰ This reaction has been used as a key step in the stereoselective synthesis of de-A,B-cholestanes



viii, 2 BuⁿLi, THF, 1 h; ix, 2 RX, THF, reflux, 24 h

Scheme 111



(a) R = H; NaNH₂, NH₃ or BuⁿLi, THF

(b) $R = Ph; KNH_2, NH_3$

- $\begin{array}{c} M \\ CS \\ O \\ SS \\ O \\ CS \\ O \\ SS \\ SS \\ O \\ SS \\$
 - 2 BuBr; 44% (ref. 438) 2 PhCH₂Cl; good yield (ref. 437)

Scheme 112

(Scheme 108, entry e). When the possibility of forming rings of different size exists, the product of cyclization is controlled by ring size and not by the acidity of the potential carbanion sites as in sulfides (compare scheme 108, entry b to Scheme 14, entry e).^{137,444}

 α -Metallo-alkyl or -benzyl sulfones^{294,443,460,470,478,481–484} and also triflones⁴⁸³ have been reacted with a large variety of terminal epoxides (Scheme 104, entry c)^{294,443,458,460,478,483,484} and with α , β -di^{470,483} and tri-substituted epoxides⁴⁸³ (Scheme 106, entry e) including allylic epoxides⁴⁸¹ and those which are part of the alkyl side chain of the sulfone (Scheme 113).^{482,485,486} The reactions often require heating (45 to 65 °C) for a prolonged period⁴⁸³ but take place under milder conditions when dilithio derivatives are



i, MeMgI, THF, -70 °C, then 20 °C; ii, cyclization; iii, H_3O^+ ; iv, 2 MeLi, THF; v, R^3Br ; vi, cyclization; vii, H_3O^+ ; viii, NaNH₂, toluene, reflux, 5–10 h; ix, Bu^tOK, HMPA, toluene, 12 °C, 6 h; x, H_2 (catalyzed)

used instead.^{294,380} The sequence has been used⁴⁸¹ in conjunction with reduction of the sulfonyl group on the resulting γ -hydroxysulfone *inter alia* for the synthesis of diastereoisomers of (±)-labda-7,14-dien-13ol⁴⁸⁴ isolated from *Aster spathulifolins* Maxim, debromoplysiatoxin²⁹⁴ and aplysiatoxin²⁹⁴ from α -sulfonyl dianions and of 24-hydroxycholestane derivatives (Scheme 104, entry c).⁴⁴³

Using 3-(phenylsulfonyl)orthopropionate and epoxides, Ghosez *et al.*⁴⁸⁷ have described a practical synthesis of 6-alkyl, 6,6-dialkyl and 5,6-dialkyl α , β -unsaturated- δ -lactones, and this reaction has been used in a synthesis of argentilactone (Scheme 106, entry e).⁴⁸⁷

3,4-Epoxybutyl and 4,5-epoxypentyl sulfones react with about 2 equiv. of methylmagnesium iodide in THF to give 3-phenylsulfonyl-cyclobutanols and -cyclopentanols in good to excellent yields (Scheme 113, entry a).⁴⁸⁶

The cyclization reaction does not apparently take place on the epoxide ring, but instead occurs on the β -oxido alkoxide resulting from the epoxide ring opening.⁴⁸⁶ The synthesis of both γ -sulfonyl-cyclobutanols and -cyclopentanols could also be achieved in 'one pot' sequentially from α -lithioalkyl phenyl sulfones, ω -bromo-1-oxidoalkenes and butyllithium,⁴⁸⁶ or even more expediently and in higher yield from the 1,1-dilithioalkyl sulfones and the same electrophile.⁴⁷⁸ The last reaction, which was found to take place chemoselectively on the epoxide ring has been successfully extended to the synthesis of 3-phenyl-sulfonylcyclohexanol (Scheme 113, entry b).

Application of the above mentioned epoxy sulfone cyclization reaction has been ingeniously used in a synthesis of muscone (Scheme 113, entries c and d)⁴⁸⁵ and also of vitamin D₃ (Scheme 113, entry e).⁴⁸² In the latter case surprisingly the terminal epoxide, when subjected to the intramolecular cyclization gave a mixture of sulfonylcyclopentane and sulfonylcyclohexane derivatives arising respectively from the epoxide ring opening at the most (30%) and the least (50%) substituted carbons.⁴⁸²

1.3.6.2 Reactions Involving α-Metalloallyl Sulfones

 α -Metalloallyl sulfones, which are readily available on metallation of allyl^{202,448,481,488-504,506-508} or in a few cases, of 1-alkenyl⁵⁰⁵ sulfones (Scheme 114), have proved to be the most valuable organometallics of this series. In almost all cases the alkylation,^{488,489,499,505,506} the benzylation,^{488,495,505} and the allylation,^{490,491,493,494,497-502,504,505,507,508} as well as the reaction with epoxides^{202,481} take place selectively at the α -position (Scheme 114; Scheme 115; Scheme 116) even if this site is already substituted by an alkyl group (Scheme 114; Scheme 115, entry c; Scheme 116; for one exception see ref. 496).^{489,503,505}



i, NaH, DMF; ii, RX; iii, desulfonation; iv, PPTS, acetone, NaOMe, MeOH; v, MeLi

Scheme 114

These results are particularly significant since mixtures of regioisomers are usually observed when related sulfoxides or sulfides are used instead (Sections 1.3.2.1.3 and 1.3.5.1.2). In the latter case, as already mentioned in Section 1.3.2.1.3, the presence of a heteroaryl group capable of chelation with the lithium salt is required in order to observe any α -selectivity comparable to that obtained with allyl sulfones.¹⁶

The methylation of the cyclohexenyl sulfone shown in Scheme 116 (entry a) proved most interesting in that equatorial attack predominates.⁴⁸⁹ Such a result is quite surprising since a twist boat conformation leading to (7; Scheme 116, entry a) would be required as an initial product in order to maintain orbital overlap.⁴⁸⁹ The stereochemistry resulting from the methylation of the bicyclic sulfones (9) and (10) is also most unusual (Scheme 116, entries b and c). Not only is the stereochemistry of the alkylation inde-



pendent of the stereochemistry of the sulfone (9), but while the norbornenyl sulfone is mainly alkylated, as expected, from the least hindered *exo* face, the norbornyl system gives⁴⁸⁹ predominantly *endo* attack in a contrasteric approach (Scheme 116, entry b).

Allyl sulfones have been used in a large number of valuable syntheses, but notably in the synthesis of alkenes, dienes and polyenes. Thus α -alkylallyl tolyl sulfones afford terminal alkenes regioselectively when treated sequentially with 2 equiv. of tin hydride and concentrated hydrochloric acid (Scheme 117, entry b).⁵⁰⁹ Allyltin derivatives, which are intermediates in this transformation, can be isolated if the acidic treatment is omitted (Scheme 117, entry a).

Allyl sulfones produce 1,3-dienes directly on sequential reaction with *n*-butyllithium and tri(*n*-butyl)stannylmethyl iodide.⁴⁹² Interestingly, the reaction is completely stereoselective as the (*E*)- or (*Z*)-1,3-dienes are obtained selectively from (*E*)- or (*Z*)-sulfones.⁴⁹² The procedure has been applied⁴⁹² to the synthesis of the major sex pheromone of the red bollworm moth. Ferruginol, which is a precursor of tax-odione, a tumor inhibitory diterpene, has been prepared⁴⁹⁵ by coupling of C₁₀ units between 2-(phe-nylsulfonylmethyl)-1,3,3-trimethylcyclohexane and 3-isopropyl-4-methoxybenzyl bromide followed by desulfonation and acid-catalyzed cyclization. The same α -metallo sulfone also proved to be a valuable precursor of other alicyclic terpenoids, such as deoxytrisporone⁴⁹⁹ and the methyl ester of vitamin A acid.⁵⁰⁰ Allylation of related metalloalkyl sulfones followed by base-assisted desulfonation of the resulting sulfones has been used in several different syntheses of vitamin A^{491,493,503,507} (selected syntheses of vitamin A using allylic sulfones are gathered in Scheme 118 and Scheme 115, entry c) as well as in syntheses of retinoic acid methyl ester,⁵⁰⁸ of the 11-*cis*-locked cyclopentatrienylidene retinals,⁴⁹⁰ and of carotenoids^{498,502} including β-carotene.⁵⁰²

The allylation of α -metalloallyl sulfones followed by reduction of the carbon-sulfur bond, usually with lithium in ethylamine, ^{36,494,501,504,510} or with Na/Hg^{501,506} (the first cited method being the most se-



i, LDA, THF; ii, MeI; iii, BuⁿLi, -78 °C, 0.7 h; iv, MeI, -78 to 20 °C

Scheme 116



i, Bu₃SnH, AIBN, benzene, Δ , 2 h; ii, hv, 20 °C, 10 h; iii, Bu₃SnH, AIBN, Δ , 2 h; iv, conc. HCl, 20 °C, several hours

Scheme 117

lective one), has been used in stereoselective syntheses of squalene (Scheme 25, entry a),⁵⁰⁶ bisgeranyl⁵⁰⁶ and several polyprenols, including (Z,Z)-farnesol (Z,Z,Z)-nerylnerol,⁵⁰⁴ solanesol^{497,501} and a series of betulaprenols.⁴⁹⁴ Related reactions involving allyl sulfones and α , β -unsaturated epoxides have also allowed the syntheses of linalol, (Z)-nerolidol and vitamin A (Scheme 119, entry a).⁴⁸¹

The α -lithicallylic sulfone (11a; Scheme 119, entry b) obtained upon deprotonation with *n*-butyllithium in THF-HMPA of the corresponding carbon acid undergoes smooth coupling with the epoxymagnesio alkoxide (12a) at -78 °C to give the β , γ -dihydroxy sulfone diol (13) in high yield. A facile coupling is also observed when the same epoxide (12a) is reacted with the related lithiated sulfide (11b). In contrast, the epoxy lithium alkoxide (12b) is only slowly attacked by the lithiated sulfide (11b), but not at all by the lithiated sulfone (11a).²⁰²

Stereoselective syntheses of (E,Z)- and (E,E)-conjugated dienes have also been performed, with alkylation of 3-sufolenes as the key step (Scheme 120; Scheme 121).




Although the 3-sulfolene α -carbanion is labile it is possible to synthesize the mono- and the di-alkylated derivatives by performing the metallation in the presence of an alkylating agent.^{511–513} The most successful conditions have used lithium hexamethyldisilazide in THF as the base in the presence of HMPA as the cosolvent.⁵¹¹ The reactions proceed in reasonable yield with primary alkyl iodides, including functionalized ones, and also with benzyl iodides and lead to α -monoalkylated derivatives (Scheme 121, entry a). It is much less efficient with the corresponding bromides and does not occur with *s*-alkyl halides.^{511,512} A remarkable directing effect of the substituent on the double bond of 3-sulfolene has been found in reactions with alkyl iodides^{510,514} or allylic bromides.⁵¹³ Electron-donating groups effect selective substitution at the 2-position while electron-withdrawing groups effect exclusive substitution at the 5-position (Scheme 121, entries c and d).^{510,514} Vitamin D-sulfur dioxide adducts can be alkylated regioselectively at either the 6-position when the adduct is treated with sodium hydride in the presence of an alkyl iodide,⁵¹⁵ or at the 19-position when the adduct is treated under similar conditions but with a stronger and more bulky base such as lithium tetramethylpiperidide.^{515,516}

172



i, I ()n OTHP, (Me₃Si)₂NLi, HMPA; ii, PPTS, MeOH, CH₂Cl₂; iii, Ac₂O, pyridine; iv, NaHCO₃, aq. EtOH, 125 °C; v, MeI, (Me₃Si)₂NLi, HMPA; vi, PPTS, MeOH; vii, K₂CO₃, aq. EtOH, 125 °C; viii, 0.5BuI, 0.5(Me₃Si)₂NLi, HMPA; ix, EtI, THF, HMPA; x, EtI, THF, HMPA, 7 °C

The alkylation of 2-alkyl-substituted sulfolenes takes place with complete regio- and high stereo-selectivity to yield approximately 60% of *trans*-2,5-disubstituted 3-sulfolenes. When 1,4-diiodobutane⁵¹⁷ and 1,3-diiodopropane⁵¹⁷ are used instead, one-pot dialkylation occurs at the 2-position in the first case and leads to a spiro sulfolene, whereas 2,3-dialkylation proceeds in the second case leading to a fused bicyclic product. Interestingly 2,5-dialkylation has been achieved, from 3-bromo(bromomethylpropene), thus allowing the synthesis of bicyclo[3.2.1]sulfones.⁵¹⁷

Sulfolenes are valuable precursors of 1,3-dienes. Thermolytic desulfonylations of 2,5-disubstituted derivatives lead exclusively to (E,Z)-dienes in agreement with the symmetry rules.⁴³⁵ The stereoselectivity of the reaction is dramatically changed however by carrying out the reaction in a protic solvent and in the presence of a base (which probably isomerizes the *trans* sulfolene to the *cis* isomer), since (E,E)-dienes are now obtained exclusively and contrary to the symmetry rules.⁴³⁵ α -Metallo sulfolenes may be viewed therefore as masked 1,3-dienyl anion equivalents ($\neg C = C - C = -C = R$). The above mentioned set of reactions has been used for the regioselective syntheses of 6- and 16-alkyl-substituted vitamin D₃ analogs,^{515,516} as well as *trans*- β -ocimene⁵¹³ and α -farnesene.⁵¹³

Related reactions have been reported by Oppolzer⁴⁵⁷ from the benzosulfolene derivative shown in Scheme 120 which has proved to be a valuable precursor of o-quinodimethane, and by Bloch⁵¹⁸ from the corresponding cyclopentadiene–sulfolene adduct.

1.3.6.3 Reactions Involving *a*-Metallovinyl Sulfones

The alkylation of α -lithiovinyl sulfones, which are available by metallation^{519,520} of vinyl sulfones or by bromine–lithium exchange⁵¹⁹ on α -bromovinyl sulfones, takes place efficiently when the β -carbon carries an aryl, a methyl or a methoxy group (Scheme 122, entries a, b and d). Surprisingly however the alkylation does not proceed with the β -alkoxymethyl-substituted derivatives (Scheme 122, entry c).⁴⁷⁹

The chemistry of sulfoximides (sulfoximines)⁶² is somewhat related to that of sulfones.^{8,35-38} Although several methods are available for the preparation of free sulfoximines from di- and tri-valent sulfur compounds, the methods are of limited applicability when the desired sulfoximines contain labile functional groups. An alternative approach involving alkylating the readily available anion derived from *N*-substituted methylsulfoximides has been described⁵²¹ but preparation of the free sulfoximides has proved



difficult.⁵²² These compounds have in fact been synthesized on reaction of *S*-lithiomethyl-*N*-trimethylsilyl-*S*-phenylsulfoximide with alkyl halides or epoxides, followed by ready desilylation with bis(trimethylsilyl)acetamide (BSA) or (trimethylsilyl)diethylamine (Scheme 123).⁵²²



Scheme 123

1.3.6.4 Reactions Involving α -Metallo- α -heterosubstituted Alkyl Sulfones

Alkylation reactions of α -metallo sulfones bearing another heteroatomic moiety at their α -site have been widely used in synthesis. These reactions include the alkylations of α -isocyanoalkyl sulfones^{449,523,524} α -alkoxyalkyl sulfones^{36,345,525-527} (including 2-benzenesulfonyltetrahydropyrans⁵²⁷) α -haloalkyl⁵²⁸ or α -haloallyl⁵²⁹ sulfones, α -silylalkyl sulfones,⁵³⁰ α -sulfonylalkyl^{380,531-535} and α -sulfonylalkenyl^{536,537} or α -triflylalkyl^{538,539} sulfones. Among these compounds alkoxyalkyl phenyl sulfones (Scheme 75, entries a and d)^{36,345,526} and tosylmethyl isocyanide (TosMIC)^{449,523,524} have been used as precursors of acyl anion equivalents, the latter being the most popular due to the ease with which it can be metallated.

1.3.6.4.1 Reactions involving 2-lithio-2-benzenesulfonyltetrahydropyrans

2-Lithio-2-benzenesulfonyltetrahydropyrans which are readily available on metallation (BuⁿLi, THF, 78 °C) of 2-benzenesulfonyltetrahydropyrans, have been alkylated^{527a,b} with a large variety of organic halides including allyl (Scheme 124, entry a), alkyl and benzyl halides (Scheme 124, entry b) as well as with 3- or 4-(3-tetrahydro-2*H*-pyran-2-yl)oxy- or (*t*-butyldimethylsilyl)oxy-1-haloalkanes (Scheme 124, entry c). Benzenesulfinic acid is usually eliminated spontaneously from the intermediate alkylated products as the reaction warms to room temperature, presumably due to the slightly basic medium and the inherent leaving ability of the benzenesulfonyl group. The *endo* hydrogen is usually removed (Scheme

124, entry a), except with the allylated product, (Scheme 124, entry b) in which case the *exo* hydrogen is taken off selectively. Many of the resulting enol ethers undergo acid-catalyzed cyclization to spiroketals, which are found in many natural products (Scheme 124, entry c). This sequence of reactions has been used for the synthesis of several insect pheromones such as those derived from the olive fly *Dacus oleae* (Scheme 124, entry c).^{527a,b} and the common wasp *Paravespula vulgaris*.^{527a}



2-Lithio-2-benzenesulfonyltetrahydropyrans also add^{527c} to epoxides which contain additional oxygen functionality to permit, after acid hydrolysis, spiroketal formation. The whole process, which is an extremely rapid and efficient entry to hydroxy substituted spiroketals, has allowed the synthesis of a minor component of the insect pheromone derived from the olive fly *Dacus oleae*^{527c} as well as of the C-11 to C-25 fragment of the milbemycins which are potent antiparasitic agents (Scheme 124, entry d). In the latter case the epoxide ring opening was carried out in the presence of titanium isopropoxide. In all of these reactions the lithio sulfones behave as the equivalents of 2-lithiodihydropyrans, but with the added advantage that the initial sulfones are stable, crystalline and easily handled materials.

1.3.6.4.2 Reactions involving *a*-isocyanoalkyl sulfones

TosMIC can be efficiently alkylated with primary alkyl halides, isopropyl iodide and benzyl bromide both to the corresponding mono-⁵²³ or di-alkyl⁴⁴⁹ derivatives using NaH in DMSO or 40% aq. NaOH in CH₂Cl₂^{449,523} and in the presence of Buⁿ₄NI (Scheme 125). The resulting compounds have then been transformed to aldehydes and ketones, including cycloalkanones,⁴⁴⁹ and the method has been successfully applied to the synthesis of optically active 2-methylcyclobutanone from the chiral sulfonylmethyl isocyanide and 1,3-dibromobutane.⁵²⁴



i, aq. NaOH, CH_2Cl_2 , Bu^n_4NI , RX; ii, aq. NaOH, CH_2Cl_2 , Bu^n_4NI , 2RX; iii, 38% HCl; iv, aq. NaOH, CH_2Cl_2 , Bu^n_4NI , R^1I , 0.5 h, then R^2I ; v, 38% HCl

Scheme 125

1.3.6.4.3 Reactions involving α -sulfinyl sulfones

1,1-Disulfones have been increasingly used in synthesis over the last decade. Their alkylation takes place smoothly and has allowed a large variety of synthetically useful reactions (Scheme 127; Scheme 128; Scheme 129).^{380,531,532,535-537} Although most of the alkylations have been carried out *via* metallation of the corresponding carbon acid (see below), some of the alkylations have been performed⁵⁴⁰ on the sodium, potassium or copper salts derived from the ring opening of 1,1-di(benzenesulfonyl)cyclopropane, performed with a variety of nucleophilic reagents (Scheme 126).

These nucleophilic reagents include thiolates, alkoxides, metallo amides, Grignard reagents in the presence of copper salts, and enolates.⁵⁴⁰ The reductive cleavage of the sulfonyl moiety in protic media, which can be readily achieved by lithium phenanthrenide, offers the opportunity to pursue further carbanionic chemistry. Therefore this new conjunctive reagent⁵⁴⁰ can serve as the equivalent of species such as $^{+}CH_2CH_2CH_2^{-}$ and $^{+}CH_2CH_2CH^{2-}$ as well as $^{+}CH_2CH_2C^{3-}$. Furthermore, homoallyl and allyl disulfones have proved valuable precursors of 1,3-dienes⁵³⁷ and (*E*)-alkenes,⁵³⁶ respectively (Scheme 127, entries a and b). The former derivatives have been produced stereoselectively by allylation of 1,1-di(sulfonyl)methane with (*Z*)- or (*E*)-allyl halides and they can be transformed to the corresponding (*Z*,*E*)- or (*Z*,*Z*)-1,3-dienes respectively by a sequence of reactions which involves the reduction of one of the two sulfonyl groups, followed by the elimination of the remaining one (Scheme 127, entry a).⁵³⁷

This route to 1,3-dienes is more efficient than the more direct one employing homoallyl sulfones.⁵³⁷ 1,1-(2-Alkenyl) disulfones display⁵³⁶ only (*E*)-geometry, whereas corresponding monosulfones are difficult to isolate at the same level of stereochemical purity (*i.e. ca.* 20% of the (*Z*)-isomer at equilibrium). Alkylation, which can be readily achieved in the presence of NaH in DMF, takes place predominantly at the α -position, but less selectively than with the corresponding monosulfone (α : γ ratio 90:10 instead of 100:0) probably due to the steric hindrance of the two bulky sulfonyl groups (Scheme 127, entry b). The product resulting from the α -alkylation leads quantitatively to the (*E*)-allylic sulfone on reduction with aluminum amalgam (Scheme 127, entry b)⁵³⁶ or to (*E*)-alkene upon total reduction by lithium in ethylamine.⁵³⁶

1,1-Disulfonyl-cycloalkanes^{534,535} and -cycloalkenes^{380,531} have been synthesized by treatment of ω -halo-alkyl- or -alkenyl 1,1-disulfones with sodium ethoxide in ethanol or from 1,1-di(sulfonyl)methyl-metals and 1,4-dihalo-(2Z)-alkenes, respectively (Scheme 127, entry c; Scheme 128, entry a).

The rate of cyclization of the 1,1-di(phenylsulfonyl) carbanion bearing distal leaving groups has been determined^{534,535} and proved to be very sensitive to the ring size. For iodides in ethanol the ratios for ring size 3:4:5:6:7 are $1:1.1 \times 10^{-5}:1.0 \times 10^{-2}:1.6 \times 10^{-6}:1.6 \times 10^{-6}:7.3 \times 10^{-10}$. The contribution of strain in the product to the enthalpy of activation is slight, whereas the entropy of activation for cyclopropane formation is so favorable as to make this by far the most rapid process observed.^{534,535} 1,1-Di(sulfonyl)cy-



i, PhSNa, DMF; ii, RX; iii, BuⁿMgBr, CuBr•SMe₂, THF, 0-10 °C; iv, MeI, HMPA; v, KCH(CO₂Et)₂; vi, MeI;

vii,
$$(I_n)^{K} R^1$$
, DMF, 90 °C, viii, R^2X

Scheme 126

clopent-3-ene, produced from the allylation of di(sulfonyl)methane with 1,4-dihalo-2-alkenes, has been transformed selectively to cyclopentene and to 3-cyclopentenone derivatives by di-reduction (Na/Hg) or by sequential mono-reduction and oxidation α to the sulfonyl moiety (Scheme 128, entry a). Therefore 1,1-di(sulfonyl)methane plays the role of masked dimetallomethane and formyl dianion equivalents.⁵³¹

1,1-Di(sulfonyl)alkanes have been reacted with allylic epoxides in the presence of polymerically bound transition metal catalysts.⁵³² This reaction has allowed the synthesis of macrolides in a sequence of reactions in which neither the nucleophilic nor the electrophilic centres are unmasked until the substrate encounters an active site on the polymer (Scheme 127, entry d). The sequence is completely regio-selective since it occurs at only one (the terminal alkene) of the three potential active sites, and provides, in around 70% yield, a single 17- or 27-membered ring macrolide from the corresponding substrates.^{39,532}

The alkoxydi(sulfonyl)methane shown in Scheme 128 (entry b) has been used as a useful carbonyl 1,1dipole synthon ($\pm C$ —O). Its alkylation with alkyl halides is best achieved using Cs₂CO₃ in DMF at 50 °C, and its palladium-catalyzed allylation requires the use of phase-transfer conditions (Scheme 128, entry b).⁵³³ The unmasking of the carbonyl group was then achieved by reacting the resulting disulfone with boron trichloride (-78 °C, CH₂Cl₂) followed by the addition of alcohols or amines to provide esters or amides respectively (Scheme 128, entry b).⁵³³

Finally, α -triflyldimethyl sulfone (CF₃SO₂CH₂SO₂Me) is a reagent which allows successive polyalkylations of the CH₂ and CH₃ carbon centers with regiocontrol (Scheme 129). The polyalkylated triflylsulfones then undergo Ramberg-Bäcklund reactions with loss of triflinate anion and extrusion of sulfur dioxide to form alkenes (Scheme 129, entry e).^{538,539}

In synthetic terms the net structural change is equivalent to regiospecific alkylation of tetrametalloethylene (M₂C—CM₂). The monoalkylated derivative is available on alkylation of the α,α -dilithio-

Alkylation of Carbon



i, NaH, DMF, 80 °C, 5 to 15 h; ii, C₆H₁₃I; iii, Al/Hg, THF, H₂O, 20 °C, 1 h; iv, Bu¹OK, THF; v, NaH, DMF, 70 °C, RI; vi, Al/Hg, THF, 0 °C, 3 d; vii, 0.5 M EtONa, EtOH; viii, polymerically bound Pd, THF, 20 °C, 0.3 h

methane derivative (Scheme 129, entry a), whereas the α, α' -dialkylated compounds can be selectively synthesized from the corresponding α, α, α' -trilithio compound (Scheme 129, entry b). Further alkylations of the above mentioned derivatives, as well as one example of alkene formation, are displayed in Scheme 129 (entries c, d and e).

1.3.7 ALKYLATION OF SULFUR YLIDES

Several examples of the alkylation of sulfur ylides are known.^{41,541-544} This reaction is particularly useful for the synthesis of *s*-alkyl diphenylsulfonium salts which cannot otherwise be prepared from diphenyl sulfide, *s*-alkyl iodides and silver tetrafluoroborate, since the latter reaction usually leads to a mixture of regioisomeric sulfonium salts.⁴¹ Diphenylsulfonium methanide on reaction with [¹⁴C]methyl iodide allows the synthesis of [¹⁴C]isopropyl diphenyl sulfonium salt which has been used for the synthesis of [¹⁴C]-2,3-epoxysqualene⁵⁴⁵ (Scheme 130, entry a). This reaction unfortunately is not general



i, (PhO₂S)₂CH₂, NaH, DMF, 20 °C; ii, 6% Na/Hg, NaHPO₄, MeOH, MeCN; iii, 6% Na/Hg, MeOH, MeCN; iv, KN(SiMe₃)₂, THF, MoOPH; v, Cs₂CO₃, Ph(CH₂)₃I, DMF, 50 °C, 12 h; vi, BCl₃, -78 °C; vii, \bigcirc OH or \bigcirc N-H; viii, Ph \bigcirc OAc , (Ph₃P)₄Pd, Bu₄NOH, CH₂Cl₂, H₂O, 20 °C

Scheme 128

and, for example, attempted methylation of diphenyl sulfonium allylide, a more stable ylide, has led only to recovered starting material.⁵⁴⁶

Intramolecular alkylations of ylides appear facile.^{41,543} For example, the cyclizations of diphenyl-3-halo-1-propylsulfonium fluoroborate and 3-halo-1-butylsulfonium fluoroborate proceed⁵⁴³ on reaction with suitable bases at 25 °C via the intermediate ylides (Scheme 130, entry b). The chloro derivatives have proved more efficient than the corresponding iodides.⁵⁴³ Low temperature methylations of five- and six-membered sulfur ylides have been performed,^{542,547} and all three methylations shown in Scheme 130 (entry c) occur with a high degree of stereoselectivity on the thianium salts since a *trans* arrangement of the entering methyl and the *S*-phenyl groups is always obtained. The behavior closely matches that observed with the corresponding thiane-1-oxides (see Section 1.3.5.1.1.i). Thiolanium ylides however appear to react without any significant stereoselectivity (Scheme 130, entry d).⁵⁴⁷ Dimethylsulfoxonium methanide reacts with diazomethane, iodoethane and dimethyl sulfate to form complex mixtures of products.⁴³ This is not the case of aryl(dimethylamino)oxosulfonium isopropanide.⁵⁴⁸ The intramolecular version of this reaction has been used for the synthesis of the corresponding cyclopropyl derivative (Scheme 130, entry e).¹³⁶



i, 2 BuⁿLi, THF, -78 to -55 °C, 0.5 h; ii, RX, -78 to 20 °C; iii, 3.2 BuⁿLi, THF, -78 to -50 °C, 1.5 h; iv, 2.4 RX, -78 to 20 °C; v, K₂CO₃, DMF, EtOTf, 70 °C, 24 h; vi, 2 BuⁿLi, THF, -78 °C; vii, R³X, -78 to 20 °C; viii, 1.2 Bu^tOK, THF, 0 °C, 1 h

Scheme 129

1.3.8 ALKYLATION OF ALKYL SULFONATES, SULTONES AND SULFONAMIDES

Metallation of alkyl α -toluenesulfonates and alkyl alkanesulfonates and subsequent alkylations of the corresponding anions allow the facile synthesis of a range of substituted sulfonates (Scheme 131, entries a and b).⁵⁴⁹ This method offers a useful route to esters of tertiary sulfonic acids which are otherwise difficult to obtain. Similar reactions, performed at low temperature (-78 °C) in order to avoid competing ring opening, permit the alkylations of 1,3-propanesultone with highly reactive alkylating agents such as methyl iodide and benzyl bromide (Scheme 131, entry b).⁵⁵⁰ Less reactive electrophiles such as ethyl bromide or isopropyl iodide react⁵⁵⁰ sluggishly however. The intramolecular version of this reaction permits the synthesis of *t*-butyl cyclopropanesulfonate (Scheme 131, entry d).⁴⁷⁴

Mono-N-substituted alkanesulfonamides react with 2 equiv. of n-butyllithium or LDA generating the corresponding dilithium salts which can then be alkylated chemoselectively on the carbon atom.⁵⁵¹

180



i, LDA, DME, -78 °C; ii, RI; iii, LiCHCl₂; iv, NaH, THF, 25 °C, 12 h; v, Bu'OK, DMSO-THF, 20 °C, 0.1 h; vi, LiCHCl₂, DME, 1 h; vii, MeI; viii, CD₃I; ix, LDA, DME, -78 °C, 0.1 h,; x, NaH, DMSO, 25 °C, 1 h; xi, HBF₄

Scheme 130

Treatment of *N*-methanesulfonyl-1,4-dihydropyridine with *n*-butyllithium, followed by benzyl bromide, leads to the corresponding *N*-1-(2-phenylethyl)sulfonyl-1,4-dihydropyridine in low yield.⁵⁵² The sulfon-amide shown in Scheme 131 (entry c) has proved a valuable *cis*-isoprenoid synthon which allows the two-step C₅-homologation of allyl halides.⁵⁵³ This synthon was used for the remarkable two-step stereo-selective synthesis of nerol from 3-methyl-2-butenyl chloride (Scheme 131, entry c).⁵⁵³ Finally, the α -chloro dicarbanion of 4-(α -chloromethanesulfonyl)morpholine is readily available⁵⁵⁴ on reaction with 2 equiv. of *n*-butyllithium in THF, and it leads to the corresponding dimethyl derivative with no detectable monoalkylated product or starting sulfonamide on methylation.⁵⁵⁴ Intramolecular versions of these reactions allow the low yield synthesis of neopentyl cyclopropanesulfonate (scheme 131, entry d)⁴⁷⁴ and the efficient preparation of cyclopropanesulfomorpholine (scheme 131, entry e).⁴⁷⁴



i, 1.1BuⁿLi, THF, -78 °C, 0.3 h; ii, MeI, -78 °C, 2-12 h, then H₂O; iii, BuⁿLi, THF, -78 °C, 0.1 h;

Cl , -70 to 20 °C; iv, RX, -78 °C, 0.1 h, then -20 °C, 2 h; v, 2BuⁿLi, THF, -70 °C; vi,

vii, 2 equiv. metal/amine or 2e⁻; viii, KH, THF, 20 °C, 5 d; ix, BuⁿLi, THF, 50 °C, then 20 °C, 3 h

Scheme 131

1.3.9 REFERENCES

- ł. D. H. Reid, 'Organic Compounds of Sulphur, Selenium, and Tellurium', in Specialist Periodical Reports, The Chemical Society, London, 1970, vol. 1.
- L. Field, Synthesis, 1972, 101. 2.
- 3. D. H. Reid, 'Organic Compounds of Sulphur, Selenium, and Tellurium', Specialist Periodical Reports, The Chemical Society, London, 1973, vol. 2. 4. B. J. Wakefield, 'The Chemistry of Organolithium Compounds', Pergamon Press, Oxford, 1974.
- 5. D. H. Reid, 'Organic Compounds of Sulphur, Selenium, and Tellurium', Specialist Periodical Reports, The Chemical Society, London, 1975, vol. 3.

- 6. D. R. Hogg, 'Organic Compounds of Sulphur, Selenium, and Tellurium', Specialist Periodical Reports, The Chemical Society, London, 1977, vol. 4.
- S. Oae (ed.) 'Organic Chemistry of Sulfur', Plenum Press, New York, 1977. 7.
- 8. E. Block, Org. Chem. (N.Y.), 1978, 37, 1.
- 9. B. M. Trost, Chem. Rev., 1978, 78, 363
- 10. D. R. Hogg, 'Organic Compounds of Sulphur, Selenium, and Tellurium', Specialist Periodical Reports. The Chemical Society, London, 1979, vol. 5.
- 11. D. R. Hogg, 'Organic Compounds of Sulphur, Selenium, and Tellurium', Specialist Periodical Reports, The Chemical Society, London, 1981, vol. 6.
- 12. E. Vedejs and G. A. Krafft, Tetrahedron, 1982, 38, 2857.
- 13. T.-L. Ho, Tetrahedron, 1985, 41, 3.
- 14. J.-F. Biellmann and J. B. Ducep, Org. React. (N.Y.), 1982, 27, 1.
- 15. T. Mukaiyama, Int. J. Sulfur Chem., Part B, 1972, 7, 173.
- 16. D. A. Evans and G. C. Andrews, Acc. Chem. Res., 1974, 7, 147.
- 17. B. M. Trost, Top. Curr. Chem., 1976, 3, 133.
- 18. D. Seebach and K.-H. Geiss, in 'New Applications of Organometallic Reagents in Organic Synthesis', ed. D. Seyferth, Elsevier, Amsterdam, 1976, p. 1.
- 19. W. Tagaki, in 'Organic Chemistry of Sulfur' ed. S. Oae, Plenum Press, New York, 1977, p. 231.
- 20. B.-T. Gröbel and D. Seebach, Synthesis, 1977, 357.
- 21. G. C. Barrett, in 'Comprehensive Organic Chemistry', ed. D. H. R. Barton and W. D. Ollis, Pergamon Press, Oxford, 1979, vol. 3, p. 33.
- 22. G. C. Barrett, in 'Comprehensive Organic Chemistry' ed. D. H. R. Barton and W. D. Ollis, Pergamon Press, Oxford, 1979, vol. 3, p. 55.
- 23. A. Krief, Tetrahedron, 1980, 36, 2531.
- 24. N. H. Werstiuk, Tetrahedron, 1983, 39, 205.
- 25. A. Krief, Top. Curr. Chem., 1987, 135, 1.
- 26. E. J. Corey, Pure Appl. Chem., 1967, 14, 19.
- 27. D. Seebach, Angew. Chem., Int. Ed. Engl., 1969, 8, 639.
- 28. D. Seebach, Synthesis, 1969, 17.
- 29. O. W. Lever, Jr., Tetrahedron, 1976, 32, 1943.
- 30. T. A. Hase and J. K. Koskimies, Aldrichimica Acta, 1981, 14, 73.
- 31. S. Oae, in 'Organic Chemistry of Sulfur', ed. S. Oae, Plenum Press, New York, 1979, p. 383.
- 32. T. Durst, in 'Comprehensive Organic Chemistry', ed. D. H. R. Barton and W. D. Ollis, Pergamon Press, Oxford, 1979, vol. 3, p. 121. 33. T. Durst, in 'Comprehensive Organic Chemistry', ed. D. H. R. Barton and W. D. Ollis, Pergamon Press,
- Oxford, 1979, vol. 3, p. 157.
- 34. G. Solladié, Synthesis, 1981, 185.
- 35. W. E. Truce, T. C. Klinger and W. W. Brand, in 'Organic Chemistry of Sulfur', ed. S. Oae, Plenum Press, New York, 1977, p. 527.
- 36. P. Magnus, Tetrahedron, 1977, 33, 2019.
- 37. T. Durst, in 'Comprehensive Organic Chemistry', ed. D. H. R. Barton and W. D. Ollis, Pergamon Press, Oxford, 1979, vol. 3, p. 171. 38. T. Durst, in 'Comprehensive Organic Chemistry', ed. D. H. R. Barton and W. D. Ollis, Pergamon Press,
- Oxford, 1979, vol. 3, p. 197.
- 39. B. M. Trost, Chemtracts, Org. Chem., 1988, 1, 415.
- 40. A. W. Johnson, in 'Ylid Chemistry', Academic Press, New York, 1966.
- 41. B. M. Trost and L. S. Melvin, Jr., Org. Chem. (N.Y.), 1975, 31, 1.
- 42. C. J. M. Stirling, in 'Organic Chemistry of Sulfur', ed. S. Oae, Plenum Press, New York, 1977, p. 473. 43. Yu. G. Gololobov, A. N. Nesmeyanov, V. P. Lysenko and I. E. Boldeskul, *Tetrahedron*, 1987, 43, 2609.
- 44. E. T. Kaiser, in 'Organic Chemistry of Sulfur', ed. S. Oae, Plenum Press, New York, 1977, p. 649.
- 45. D. L. J. Clive, Tetrahedron, 1978, 34, 1049.
- 46. H. J. Reich, Org. Chem. (N.Y.), 1978, 5 (part C), 1.
- 47. H. J. Reich, Acc. Chem. Res., 1979, 12, 22.
- 48. P. D. Magnus, in 'Comprehensive Organic Chemistry', ed. D. H. R. Barton and W. D. Ollis, Pergamon Press, Oxford, 1979, vol. 3, p. 491.
- 49. A. Krief, Proc. Int. Symp. Org. Selenium Tellurium Compd., 3rd, 1979, 1981, 11.
- 50. A. Krief, W. Dumont, A. Cravador, J. N. Denis, S. Halazy, L. Hevesi, D. Labar, J. Lucchetti, J. Rémion, M. Sevrin and D. Van Ende, Bull. Soc. Chim. Fr., 1980, 2, 519.
- 51. C. Paulmier, in 'Selenium Reagents and Intermediates in Organic Synthesis', Pergamon Press, Oxford, 1986.
- 52. A. Krief, in 'The Chemistry of Organic Selenium and Tellurium Compounds', ed. S. Patai and Z. Rappoport, Wiley, Chichester, 1987, vol. 2, p. 675. 53. T. G. Back, in 'The Chemistry of Organic Selenium and Tellurium Compounds', ed. S. Patai and Z.
- Rappoport, Wiley, Chichester, 1987, vol. 2, p. 91.
- 54. H. J. Reich, in 'Organoselenium Chemistry', ed. D. Liotta, Wiley-Interscience, New York, 1987, p. 243.
- 55. A. Krief and L. Hevesi, Janssen Chim. Acta, 1984, 2, 3.
- 56. K. C. Nicolaou and N. A. Petasis, in 'Selenium in Natural Products Synthesis', CIS, Philadelphia, 1984.
- 57. D. J. Peterson, Organomet. Chem. Rev., Sect. A, 1972, 7, 295.
- 58. D. J. Peterson, J. Org. Chem., 1967, 32, 1717.
- 59. E. J. Corey and D. Seebach, J. Org. Chem., 1966, 31, 4097.

- 60. T. M. Dolak and T. A. Bryson, Tetrahedron Lett., 1977, 1961.
- 61. B. M. Trost, D. E. Keeley, H. C. Arndt, J. H. Ridby and M. J. Bogdanowicz, J. Am. Chem. Soc., 1977, 99, 3080.
- 62. C. R. Johnson, in 'Comprehensive Organic Chemistry', ed. D. H. R. Barton and W. D. Ollis, Pergamon Press, Oxford, 1979, vol. 3, p. 223.
- 63. K. B. Sharpless, K. M. Gordon, R. F. Lauer, D. W. Patrick, S. P. Singer and M. W. Young, Chem. Scr., 1975, 8.9.
- 64. D. Seebach and N. Peleties, Angew. Chem., Int. Ed. Engl., 1969, 8, 450.
- 65. D. Seebach and N. Peleties, Chem. Ber., 1972, 105, 511.
- 66. W. Dumont, P. Bayet and A. Krief, Angew. Chem., Int. Ed. Engl., 1974, 13, 804.
- 67. D. Van Ende, W. Dumont and A. Krief, Angew. Chem., Int. Ed. Engl., 1975, 14, 700.
- 68. D. Seebach and A. K. Beck, Angew. Chem., Int. Ed. Engl., 1974, 13, 806.
- 69. D. Seebach, N. Meyer and A. K. Beck, Justus Liebigs Ann. Chem., 1977, 846.
- 70. S. Halazy and A. Krief, Tetrahedron Lett., 1979, 4233.
- 71. S. Halazy and A. Krief, Tetrahedron Lett., 1980, 21, 1997.
- 72. A. Krief, W. Dumont, M. Clarembeau, G. Bernard and E. Badaoui, Tetrahedron, 1989, 45, 2005.
- 73. A. Krief, W. Dumont, M. Clarembeau and E. Badaoui, Tetrahedron, 1989, 45, 2023.
- 74. H. J. Reich, F. Chow and S. K. Shah, J. Am. Chem. Soc., 1979, 101, 6638.
- 75. D. Seebach, J. Gabriel and R. Hässig, Helv. Chim. Acta, 1984, 67, 1083.
- D. Van Ende, A. Cravador and A. Krief, J. Organomet. Chem., 1979, 177, 1.
 D. Van Ende, W. Dumont and A. Krief, J. Organomet. Chem., 1978, 149, C10.
- 78. S. Halazy, J. Lucchetti and A. Krief, Tetrahedron Lett., 1978, 3971.
- 79. M. Clarembeau and A. Krief, Tetrahedron Lett., 1986, 27, 1719. 80. M. Clarembeau and A. Krief, Tetrahedron Lett., 1986, 27, 1723.
- 81. L. Hevesi, K. M. Nsunda and M. Renard, Bull. Soc. Chim. Belg., 1985, 94, 1039.
- 82. B.-T. Gröbel and D. Seebach, Chem. Ber., 1977, 110, 867.
- 83. J. N. Denis and A. Krief, Tetrahedron Lett., 1982, 23, 3411.
- 84. M. Clarembeau, A. Cravador, W. Dumont, L. Hevesi, A. Krief, J. Lucchetti and D. Van Ende, Tetrahedron, 1985, 41, 4793.
- 85. N. Petragnani and G. Schill, Chem. Ber., 1970, 103, 2271.
- 86. A. Burton, L. Hevesi, W. Dumont, A. Cravador and A. Krief, Synthesis, 1979, 877.
- 87. R. Dieden and L. Hevesi, Synthesis, 1988, 616.
- 88. B.-T. Gröbel and D. Seebach, Chem. Ber., 1977, 110, 852.
- 89. J. N. Denis and A. Krief, Tetrahedron Lett., 1982, 23, 3407.
- 90. J. N. Denis, S. Desauvage, L. Hevesi and A. Krief, Tetrahedron Lett., 1981, 22, 4009.
- 91. J. N. Denis, W. Dumont and A. Krief, Tetrahedron Lett., 1979, 4111.
- 92. A. J. Anciaux, A. Eman, W. Dumont and A. Krief, Tetrahedron Lett., 1975, 1617.
- W. Dumont and A. Krief, J. Chem. Soc., Chem. Commun., 1980, 673.
 M. R. Binns and R. K. Haynes, J. Org. Chem., 1981, 46, 3790.
- 95. H. J. Reich, J. Org. Chem., 1975, 40, 2570.
- 96. H. J. Reich, M. C. Schroeder and I. L. Reich, Isr. J. Chem., 1984, 24, 157.
- 97. Y. Yamamoto, Y. Saito and K. Maruyama, Tetrahedron Lett., 1982, 23, 4597.
- Y. Yamamoto, H. Yatagai, Y. Saito and K. Maruyama, J. Org. Chem., 1984, 49, 1096. **98**.
- 99. I. Fernández, B. García and J. R. Pedro, Tetrahedron, 1987, 43, 805.
- 100. H. J. Reich, M. C. Clark and W. W. Willis, Jr., J. Org. Chem., 1982, 47, 1618.
- 101. M. Clarembeau and A. Krief, Tetrahedron Lett., 1984, 25, 3629.
- 102. H. J. Reich and S. K. Shah, J. Am. Chem. Soc., 1977, 99, 263.
- 103. H. J. Reich, S. K. Shah, P. M. Gold and R. E. Olson, J. Am. Chem. Soc., 1981, 103, 3112.
- 104. R. H. Mitchell, J. Chem. Soc., Chem. Commun., 1974, 990.
- 105. R. H. Mitchell, Can. J. Chem., 1980, 58, 1398.
- 106. H. J. Reich and S. K. Shah, J. Org. Chem., 1977, 42, 1773.
- H. J. Reich, W. W. Willis, Jr. and P. D. Clark, J. Org. Chem., 1981, 46, 2775.
 S. Raucher and G. A. Koolpe, J. Org. Chem., 1978, 43, 3794.
- 109. H. J. Reich and W. W. Willis, Jr., J. Am. Chem. Soc., 1980, 102, 5967
- 110. M. Sevrin, J. N. Denis and A. Krief, Angew. Chem., Int. Ed. Engl., 1978, 17, 526.
- 111. A. Toshimitsu, H. Owada, K. Terao, S. Uemura and M. Okano, J. Chem. Soc., Perkin Trans. 1, 1985, 373.
- 112. H. J. Reich and S. K. Shah, J. Am. Chem. Soc., 1975, 97, 3250.
- 113. H. J. Reich, S. K. Shah and F. Chow, J. Am. Chem. Soc., 1979, 101, 6648.
- 114. R. C. Gadwood, J. Org. Chem., 1983, 48, 2098.
- 115. P. R. Kanjilal, M. Sarkar, S. K. Patra, S. Ghosh and U. R. Ghatak, J. Org. Chem., 1985, 50, 857.
- W. Dumont, P. Bayet and A. Krief, Angew. Chem., Int. Ed. Engl., 1974, 13, 274. 116.
- 117. A. Krief, W. Dumont, D. Van Ende, S. Halazy, D. Labar, J.-L. Laboureur and T. Q. Le, Heterocycles, 1989, **28**, 1203.
- 118. A. Krief, W. Dumont and J. N. Denis, J. Chem. Soc., Chem. Commun., 1985, 571.
- A. Krief, W. Dumont and J.-L. Laboureur, Tetrahedron Lett., 1988, 29, 3265. 119.
- 120. R. Gräfing, H. D. Verkruijsse and L. Brandsma, J. Chem. Soc., Chem. Commun., 1978, 596.
- 121. F. G. Bordwell and A. H. Clemens, J. Org. Chem., 1981, 46, 1035.
- 122. S. Halazy, Ph. D. Thesis, Facultés Universitaire Notre Dame de la Paix, Namur, 1982.
- 123. B. Demoulin, Mémoire de Licence, Facultés Universitaire Notre Dame de la Paix, Namur, 1985.
- 124. A. Crosby and M. Kato, J. Am. Chem. Soc., 1977, 99, 278.

- 125. E. J. Corey and M. Jautelat, Tetrahedron Lett., 1968, 5787.
- 126. C. G. Screttas and M. Micha-Screttas, J. Org. Chem., 1979, 44, 713.
- 127. M. Clarembeau, J. L. Bertrand and A. Krief, Isr. J. Chem., 1984, 24, 125.
- 128. T. Cohen, W. M. Daniewski and R. B. Weisenfeld, Tetrahedron Lett., 1978, 4665.
- 129. T. Cohen and J. R. Matz, Synth. Commun., 1980, 10, 311.
- P. Beak, B. G. McKinnie and D. B. Reitz, *Tetrahedron Lett.*, 1977, 1839.
 P. Beak and P. D. Becker, J. Org. Chem., 1982, 47, 3855.
- 132. (a) K. Hirai and Y. Kishida, Heterocycles, 1974, 2, 185; (b) K. Hirai and Y. Kishida, Tetrahedron Lett., 1972, 2743.
- 133. K. Hirai, H. Matsuda and Y. Kishida, Tetrahedron Lett., 1971, 4359.
- 134. D. Hoppe, Angew. Chem., Int. Ed. Engl., 1975, 14, 424.
- 135. B. M. Trost, M. Vaultier and M. L. Santiago, J. Am. Chem. Soc., 1980, 102, 7929.
- 136. C. R. Johnson, G. F. Katekar, R. F. Huxol and E. R. Janiga, J. Am. Chem. Soc., 1971, 93, 3771.
- 137. W. E. Truce, K. R. Hollister, L. B. Linday and J. E. Parr, J. Org. Chem., 1968, 33, 43.
- Y.-H. Chang and H. W. Pinnick, J. Org. Chem., 1978, 43, 373.
 C. L. Bumgardner, J. R. Lever and S. T. Purrington, J. Org. Chem., 1980, 45, 748.
- 140. M. Sevrin, D. Van Ende and A. Krief, Tetrahedron Lett., 1976, 2643.
- 141. M. Sevrin, Ph. D. Thesis, Facultés Universitaire Notre Dame de la Paix, Namur, 1980.
- 142. D. Labar, L. Hevesi, W. Dumont and A. Krief, Tetrahedron Lett., 1978, 1141.
- M. Sevrin, W. Dumont, L. Hevesi and A. Krief, Tetrahedron Lett., 1976, 2647.
 M. J. Calverley, Tetrahedron Lett., 1987, 28, 1337.
- 145. S. Raucher and G. A. Koolpe, J. Org. Chem., 1978, 43, 4252.
- 146. G. H. Posner, Org. React. (N.Y.), 1975, 22, 253.
- 147. J. L. Bertrand, Mémoire de Licence, Facultés Universitaire Notre Dame de la Paix, Namur, 1983.
- 148. J. Lucchetti and A. Krief, Tetrahedron Lett., 1981, 22, 1623.
- 149. M. Sevrin and A. Krief, Tetrahedron Lett., 1980, 21, 585.
- 150. M. Sevrin and A. Krief, Tetrahedron Lett., 1978, 187.
- 151. J.-F. Biellmann and J. B. Ducep, Tetrahedron Lett., 1968, 5629.
- 152. H. Ikehira, S. Tanimoto, T. Oida and M. Okano, J. Org. Chem., 1983, 48, 1120.
- 153. T. Takahashi, K. Kasuga and J. Tsuji, Tetrahedron Lett., 1978, 4917.
- 154. U. Folli, D. Larossi and F. Taddei, J. Chem. Soc., Perkin Trans., 2, 1974, 933.
- 155. J.-F. Biellmann and J. L. Schmitt, Tetrahedron Lett., 1973, 4615.
- T. Hayashi and H. Baba, J. Am. Chem. Soc., 1975, 97, 1608.
 T. Hayashi, A. Sakurai and T. Oishi, Chem. Lett., 1977, 1483.
- 158. K.-H. Geiss, D. Seebach and B. Seuring, Chem. Ber., 1977, 110, 1833.
- 159. S. Nonet, Mémoire de Licence, Facultés Universitaire Notre Dame de la Paix, Namur, 1988.
- 160. M. Hobe, unpublished results, Facultés Universitaire Notre Dame de la Paix, Namur, 1989.
- 161. J.-F. Biellmann, J. B. Ducep and J. J. Vicens, Tetrahedron, 1976, 32, 1801.
- 162. K. Oshima, H. Yamamoto and H. Nozaki, Bull. Chem. Soc. Jpn., 1975, 48, 1567.
- 163. K. Oshima, H. Yamamoto and H. Nozaki, J. Am. Chem. Soc., 1973, 95, 7926.
- 164. P. M. Atlani, J.-F. Biellmann, S. Dube and J. J. Vicens, Tetrahedron Lett., 1974, 2665.
- J.-F. Biellmann and J. B. Ducep, *Tetrahedron*, 1971, 27, 5861.
 K. Oshima, H. Takahashi, H. Yamamoto and H. Nozaki, J. Am. Chem. Soc., 1973, 95, 2693.
- 167. K. Oshima, H. Yamamoto and H. Nozaki, J. Am. Chem. Soc., 1973, 95, 4446.
- 168. T. Mukaiyama, K. Narasaka, K. Maekawa and M. Furusato, Bull. Chem. Soc. Jpn., 1971, 44, 2285.
- 169. K. Narasaka, M. Hayashi and T. Mukaiyama, Chem. Lett., 1972, 259.
- T. E. Goodwin, D. G. Ratcliff, D. M. Crowder and N. K. Seitzinger, J. Org. Chem., 1982, 47. 170.
- 171. K. Hirai and Y. Kishida, Org. Synth., 1977, 56, 77.
- 172. T. Hayashi and T. Oishi, Chem. Lett., 1985, 413.
- 173. T. Hayashi, I. Hori and T. Oishi, J. Am. Chem. Soc., 1983, 105, 2909.
- T. Nakai, H. Shiono and M. Okawara, Chem. Lett., 1975, 249.
 T. Nakai, H. Shiono and M. Okawara, Tetrahedron Lett., 1975, 4027
- 176. T. Nakai, H. Shiono and M. Okawara, Tetrahedron Lett., 1974, 3625.
- 177. T. Hayashi and H. Midorikawa, Synthesis, 1975, 100.
- 178. K.-H. Geiss, B. Seuring, R. Pieter and D. Seebach, Angew. Chem., Int. Ed. Engl., 1974, 13, 479.
- 179. J.-F. Biellmann and J. B. Ducep, Tetrahedron Lett., 1971, 33.
- 180. K. Tanaka, M. Terauchi and A. Kaji, Chem. Lett., 1981, 315.
- 181. S. Torii, H. Tanaka and Y. Tomotaki, Chem. Lett., 1974, 1541.
- 182. J. Hartmann, R. Muthukrishnan and M. Schlosser, Helv. Chim. Acta, 1974, 57, 2261.
- Y. Yamamoto, H. Yatagai and K. Maruyama, J. Chem. Soc., Chem. Commun., 1979, 157. 183.
- 184. S. Yamada, N. Oh-hashi and K. Achiwa, Tetrahedron Lett., 1976, 2561.
- 185. E. E. van Tamelen, M. P. Seiler and W. Wierenga, J. Am. Chem. Soc., 1972, 94, 8229,
- 186. L. J. Altman, L. Ash and S. Marson, Synthesis, 1974, 129.
- 187. K. Kondo and M. Matsumoto, Tetrahedron Lett., 1976, 391.
- 188. F. W. Sum and L. Weiler, J. Am. Chem. Soc., 1979, 101, 4401.
- 189. W. G. Dauben, R. K. Saugier and I. Fleischhauer, J. Org. Chem., 1985, 50, 3767.
- 190. M. Kodama, Y. Shiobara, H. Sumitomo, K. Fukuzumi, H. Minami and Y. Miyamoto, Tetrahedron Lett., 1986, 27, 2157.
- 191. (a) E. E. van Tamelen, P. McCurry and U. Huber, Proc. Natl. Acad. Sci. USA, 1971, 68, 1294; (b) K. Mori, M. Ohki and M. Matsui, Tetrahedron, 1974, 30, 715.

- 192. P. L. Stotter and R. E. Hornish, J. Am. Chem. Soc., 1973, 95, 4444.
- 193. M. Kodama, Y. Matsuki and S. Itô, Tetrahedron Lett., 1975, 3065.
- 194. M. Kodama, Y. Matsuki and S. Itô, Tetrahedron Lett., 1976, 1121.
- 195. M. Kodama, S. Yokoo, H. Yamada and S. Itô, Tetrahedron Lett., 1978, 3121.
- 196. M. Kodama, K. Shimada and S. Itô, Tetrahedron Lett., 1977, 2763.
- 197. K. Shimada, M. Kodama and S. Itô, Tetrahedron Lett., 1981, 22, 4275.
- 198. V. Rautenstrauch, J. Chem. Soc., Chem. Commun., 1978, 519.
- 199. M. Hashimoto, T. Kan, M. Yanagiya, H. Shirahama and T. Matsumoto, Tetrahedron Lett., 1987, 5665.
- 200. M. Kodama, T. Takahashi, T. Kojima and S. Itô, Tetrahedron Lett., 1982, 23, 3397.
- 201. M. Kodama, K. Okumura, Y. Kobayashi, T. Tsunoda and S. Itô, Tetrahedron Lett., 1984, 25, 5781.
- J. A. Marshall and R. C. Andrews, J. Org. Chem., 1985, 50, 1602.
 E. J. Corey, B. W. Erickson and R. Noyori, J. Am. Chem. Soc., 1971, 93, 1724.
- 204. T. Cohen, R. E. Gapinski and R. R. Hutchins, J. Org. Chem., 1979, 44, 3599.
- 205. T. Cohen, D. A. Bennett and A. J. Mura, Jr., J. Org. Chem., 1976, 41, 2506.
- 206. E. J. Corey and R. Noyori, Tetrahedron Lett., 1970, 311.
- R. G. Carlson and W. S. Mardis, J. Org. Chem., 1975, 40, 817.
 A. Spaltenstein, P. A. Carpino, F. Miyake and P. B. Hopkins, J. Org. Chem., 1987, 52, 3759.
- 209. R. G. Shea, J. N. Fitzner, J. E. Fankhauser, A. Spaltenstein, P. A. Carpino, D. V. Peevey Pratt, B. J. Tenge and P. B. Hopkins, J. Org. Chem., 1986, 51, 5243.
- 210. R. Mantione and Y. Leroux, C. R. Hebd. Seances Acad. Sci., Ser. C, 1971, 2201.
- 211. E. Negashi, C. L. Rand and K. P. Jadhav, J. Org. Chem., 1981, 46, 5041.
- W. E. Parham and R. F. Motter, J. Am. Chem. Soc., 1959, 81, 2146. 212.
- 213. R. H. Everhardus, R. Gräfing and L. Brandsma, Recl. Trav. Chim. Pays-Bas, 1978, 97, 69.
- 214. T. Kauffmann, H. Ahlers, H. J. Tilhard and A. Woltermann, Angew. Chem., Int. Ed. Engl., 1977, 16, 710.
- 215. B. Harirchian and P. Magnus, J. Chem. Soc., Chem. Commun., 1977, 522.
- 216. K. Oshima, H. Takahashi, H. Yamamoto and H. Nozaki, J. Am. Chem. Soc., 1973, 95, 2693.
- 217. T. Cohen and R. B. Weisenfeld, J. Org. Chem., 1979, 44, 3601.
- 218. E. Schaumann and W. Walter, Chem. Ber., 1974, 107, 3562.
- 219. I. Vlattas, L. Della Vecchia and A. O. Lee, J. Am. Chem. Soc., 1976, 98, 2008.
- 220. R. R. Schmidt and B. Schmid, Tetrahedron Lett., 1977, 3583.
- 221. R. Gräfing and L. Brandsma, Recl. Trav. Chim. Pays-Bas, 1980, 99, 23.
- 222. L. Brandsma, H. E. Wijers and J. F. Arens, Recl. Trav. Chim. Pays-Bas, 1963, 82, 1040.
- 223. R. M. Carlson and J. L. Isidor, Tetrahedron Lett., 1973, 4819.
- 224. R. M. Carlson, R. W. Jones and A. S. Hatcher, Tetrahedron Lett., 1975, 1741.
- 225. D. L. J. Clive, G. J. Chittattu and C. K. Wong, J. Chem. Soc., Chem. Commun., 1978, 41.
- 226. D. L. J. Clive, G. J. Chittattu, V. Farina, W. A. Kiel, S. M. Menchen, C. G. Russell, C. K. Wong and N. J. Curtis, J. Am. Chem. Soc., 1980, 102, 4438.
- 227. A. L. J. Beckwith and P. E. Pigou, Aust. J. Chem., 1986, 39, 77.
- 228. A. Krief and P. Barbeaux, J. Chem. Soc., Chem. Commun., 1987, 1214.
- 229. L. Set, D. R. Cheshire and D. L. J. Clive, J. Chem. Soc., Chem. Commun., 1985, 1205.
- 230. A. Krief, M. Clarembeau and P. Barbeaux, J. Chem. Soc., Chem. Commun., 1986, 457.
- 231. A. Krief, D. Surleraux, W. Dumont, P. Pasau, P. Lecomte and P. Barbeaux, in 'Proceedings of the NATO conference on Strain and its Implication in Organic Chemistry, August 1988'
- P. Barbeaux, unpublished results, Facultés Universitaire Notre Dame de la Paix, Namur, 1989. 232.
- 233. C. A. Kingsbury and D. J. Cram, J. Am. Chem. Soc., 1960, 82, 1810.
- 234. D. N. Jones, D. Mundy and R. D. Whitehouse, J. Chem. Soc., Chem. Commun., 1970, 86.
- 235. D. A. Evans, G. C. Andrews, T. T. Fujimoto and D. Wells, Tetrahedron Lett., 1973, 1385.
- 236. D. A. Evans, G. C. Andrews, T. T. Fujimoto and D. Wells, Tetrahedron Lett., 1973, 1389.
- 237. M. Sevrin and A. Krief, J. Chem. Soc., Chem. Commun., 1980, 656.
- 238. L. Hevesi, M. Sevrin and A. Krief, Tetrahedron Lett., 1976, 2651.
- 239. J. V. Comasseto, J. Organomet. Chem., 1983, 253, 131.
- 240. E. J. Corey and J. I. Shulman, J. Org. Chem., 1970, 35, 777. 241. J. V. Comasseto and N. Petragnani, J. Organomet. Chem., 1978, 152, 295.
- 242. M. Sevrin, W. Dumont and A. Krief, Tetrahedron Lett., 1977, 3835.
- 243. J. Lucchetti and A. Krief, Synth. Commun., 1983, 13, 1153.
- 244. L. Hevesi, J. L. Piquard and H. Wautier, J. Am. Chem. Soc., 1981, 103, 870.
- 245. J. F. Arens, M. Fröling, and A. Fröling, Recl. Trav. Chim. Pays-Bas, 1959, 78, 663.
- 246. A. Fröling and J. F. Arens, Recl. Trav. Chim. Pays-Bas, 1962, 81, 1009.
- 247. H. Normant and T. Cuvigny, Bull. Soc. Chim. Fr., 1965, 1881.
- 248. D. J. Ager, Tetrahedron Lett., 1980, 21, 4763.
- W. E. Truce and R. E. Roberts, J. Org. Chem., 1963, 28, 961.
 G. Schill and C. Merkel, Synthesis, 1975, 387.
- 251. P. Blatcher, J. I. Grayson and S. Warren, J. Chem. Soc., Chem. Commun., 1976, 547.

- D. J. Ager, J. Chem. Soc., Perkin Trans. 1, 1986, 183.
 T. Cohen and L.-C. Yu, J. Org. Chem., 1985, 50, 3266.
 M. F. Semmelhack and J. C. Tomesch, J. Org. Chem., 1977, 42, 2657.
- 255. D. Seebach and M. Braun, Angew. Chem., Int. Ed. Engl., 1972, 11, 49.
- 256. M. Braun and D. Seebach, Chem. Ber., 1976, 109, 669
- 257. C. Feugeas and J. P. Galy, C. R. Hebd. Seances Acad. Sci., Ser. C, 1970, 2157.
- 258. T. Cohen and W. M. Daniewski, Tetrahedron Lett., 1978, 2991

- 259. T. Itoh, Y. Yonekawa, T. Sato and T. Fujisawa, Tetrahedron Lett., 1986, 27, 5405.
- 260. P. Gosselin, S. Masson and A. Thuillier, J. Org. Chem., 1979, 44, 2807.
- 261. S. Ncube, A. Pelter, K. Smith, P. Blatcher and S. Warren, Tetrahedron Lett., 1978, 2345.
- 262. C. Lion, C. R. Hebd. Seances Acad. Sci., Ser. C, 1982, 294, 431.
- 263. K. Mori, H. Hashimoto, Y. Takenaka and T. Takigawa, Synthesis, 1975, 720.
- S. Rozen, I. Shahak and E. D. Bergmann, *Tetrahedron Lett.*, 1972, 1837.
 E. J. Corey and D. Seebach, *Angew. Chem.*, Int. Ed. Engl., 1965, 4, 1075.
- 266. D. Seebach and E. J. Corey, J. Org. Chem., 1975, 40, 231.
- T. A. Hylton and V. Boekelheide, J. Am. Chem. Soc., 1968, 90, 6887.
 H. H. Bokel, A. Hoppmann and P. Weyerstahl, Tetrahedron, 1980, 36, 651.
- R. Sarges, J. Org. Chem., 1975, 40, 1216.
 N. Finch and C. W. Gemenden, J. Org. Chem., 1979, 44, 2804.
- 271. H. W. Geschwend, J. Am. Chem. Soc., 1972, 94, 8430.
- 272. W. D. Woessner and R. A. Ellison, Tetrahedron Lett., 1972, 3735.
- 273. E. L. Eliel, A. G. Abatjoglou and A. A. Hartmann, J. Am. Chem. Soc., 1972, 94, 4786.
- 274. E. J. Corey and D. Seebach, Angew. Chem., Int. Ed. Engl., 1965, 4, 1077.
- 275. E. J. Corey and B. W. Erickson, J. Org. Chem., 1971, 36, 3553.
- 276. R. A. Ellison, E. R. Lukenbach and C.-W. Chiu, Tetrahedron Lett., 1975, 499.
- 277. R. G. Smith, G. D. Daves, Jr. and G. E. Daterman, J. Org. Chem., 1975, 40, 1593.
- 278. D. Seebach, N. R. Jones and E. J. Corey, J. Org. Chem., 1968, 33, 300.
- 279. B. Seuring and D. Seebach, Justus Liebigs Ann. Chem., 1978, 2044.
- 280. Y. Nakahara, A. Fujita, K. Beppu and T. Ogawa, Tetrahedron, 1986, 42, 6465.
- 281. A. H. Alberts and D. J. Cram, J. Am. Chem. Soc., 1979, 101, 3545.
- 282. W. B. Sudweeks and H. S. Broadbent, J. Org. Chem., 1975, 40, 1131.
- 283. D. Seebach, D. Steinmüller and F. Demuth, Angew. Chem., Int. Ed. Engl., 1968, 7, 620.
- 284. D. Seebach and D. Steinmüller, Angew. Chem., Int. Ed. Engl., 1968, 7, 619.
- 285. A. Hoppmann and P. Weyerstahl, Tetrahedron, 1978, 34, 1723.
- 286. D. Tatone, T. C. Dich, R. Nacco and C. Botteghi, J. Org. Chem., 1975, 40, 2987.
- 287. E. L. Eliel, Tetrahedron, 1974, 30, 1503.
- A. G. Abatjoglou, E. L. Eliel and L. F. Kuyper, J. Am. Chem. Soc., 1977, 99, 8262.
 P. F. Jones, M. F. Lappert and A. C. Szary, J. Chem. Soc., Perkin Trans. 1, 1973, 2272.
- 290. D. Seebach and E.-M. Wilka, Synthesis, 1976, 476.
- 291. A. I. Meyers and R. S. Brinkmeyer, Tetrahedron Lett., 1975, 1749.
- 292. E. J. Corey and M. G. Bock, Tetrahedron Lett., 1975, 2643.
- 293. M. Koreeda and N. Koizumi, Tetrahedron Lett., 1978, 1641.
- 294. P. Park, C. A. Broka, B. F. Johnson and Y. Kishi, J. Am. Chem. Soc., 1987, 109, 6205.
- 295. J. B. Jones and R. Grayshan, Can. J. Chem., 1972, 50, 1414.
- 296. J. B. Jones and R. Grayshan, Can. J. Chem., 1972, 50, 1407.
- 297. S. Torii, Y. Matuyama, M. Isihara and K. Uneyama, Chem. Lett., 1973, 947.
- 298. S. Torii, K. Uneyama and M. Isihara, J. Org. Chem., 1974, 39, 3645.
- 299. F. Sher, J. L. Isidor, H. R. Taneja and R. M. Carlson, Tetrahedron Lett., 1973, 577.
- 300. E. Vedejs and P. L. Fuchs, J. Org. Chem., 1971, 36, 366.
- R. Grayshan, C. A. Keal and M. A. Sackville, J. Chem. Res. (S), 1986, 282.
 G. Stork and K. Zhao, Tetrahedron Lett., 1989, 30, 287.
- 303. M. Balogh, A. Cornélis and P. Laszlo, Tetrahedron Lett., 1984, 25, 3313.
- 304. M. A. Abdallah and J. N. Shah, J. Chem. Soc., Perkin Trans. 1, 1975, 888.
- 305. C. G. Pitt, H. H. Seltzman, Y. Sayed, C. E. Twine, Jr. and D. L. Williams, J. Org. Chem., 1979, 44, 677.
- T. Masamune, H. Murase, H. Matsue and A. Murai, Bull. Chem. Soc. Jpn., 1979, 52, 135. 306.
- 307. I. Degani, R. Fochi and V. Regondi, Synthesis, 1981, 51.
- 308. K. Narasaka, T. Sakashita and T. Mukaiyama, Bull. Chem. Soc. Jpn., 1972, 45, 3724.
- 309. Y. Nagao, K. Seno and E. Fujita, Tetrahedron Lett., 1979, 3167.
- 310. G. R. Pettit and E. E. van Tamelen, Org. React., 1962, 12, 356.
- 311. P. Stütz and P. A. Stadler, Helv. Chim. Acta, 1972, 55, 75.
- 312. P. Stütz and P. A. Stadler, Org. Synth., 1977, 56, 8.
- 313. D. Seebach, B. W. Erickson and G. Singh, J. Org. Chem., 1966, 31, 4303.
- 314. A. G. Brook, J. M. Duff, P. F. Jones and N. R. Davis, J. Am. Chem. Soc., 1967, 89, 431.
- 315. A. M. Sépulchre, G. Vass and S. D. Gero, Tetrahedron Lett., 1973, 3619.
- 316. A. M. Sépulchre, G. Lukacs, G. Vass and S. D. Gero, Bull. Soc. Chim. Fr., 1972, 4000.
- 317. A. M. Sépulchre, G. Lukacs, G. Vass and S. D. Gero, C. R. Hebd. Seances Acad. Sci., Ser. C, 1971, 273, 1180.
- 318. C. A. Reece, J. O. Rodin, R. G. Brownlee, W. G. Duncan and R. M. Silverstein, Tetrahedron, 1968, 24, 4249.
- 319. M. F. Semmelhack and L. S. Ryono, J. Am. Chem. Soc., 1975, 97, 3873.
- 320. W. S. Johnson, S. D. Escher and B. W. Metcalf, J. Am. Chem. Soc., 1976, 98, 1039.
- 321. G. A. Russell, M. Jawdosiuk and M. Makosza, J. Am. Chem. Soc., 1979, 101, 2355.
- 322. D. Seebach, I. Willert, A. K. Beck and B.-T. Gröbel, *Helv. Chim. Acta*, 1978, **61**, 2510. 323. E. L. Eliel, A. A. Hartmann and A. G. Abatjoglou, *J. Am. Chem. Soc.*, 1974, **96**, 1807.
- 324. D. Seebach, M. Kolb and B.-T. Gröbel, Tetrahedron Lett., 1974, 3171.
- 325. F. E. Ziegler and C. C. Tam, J. Org. Chem., 1979, 44, 3428.
- E. J. Corey and A. P. Kozikowski, *Tetrahedron Lett.*, 1975, 925.
 A. Closse and R. Huguenin, *Helv. Chim. Acta.*, 1974, 57, 533.
- 328. W. S. Murphy and S. Wattanasin, Tetrahedron Lett., 1979, 1827

- 329. W. S. Murphy and S. Wattanasin, J. Chem. Soc., Perkin Trans. 1, 1980, 2678.
- 330. D. D. Poulter and J. M. Hughes, J. Am. Chem. Soc., 1977, 99, 3830.
- D. Seebach, M. Kolb and B.-T. Gröbel, Angew. Chem., Int. Ed. Engl., 1973, 12, 69. 331.
- W. S. Murphy, R. Boyce and E. A. O'Riordan, Tetrahedron Lett., 1971, 4157. 332.
- 333. M. Fétizon and M. Jurion, J. Chem. Soc., Chem. Commun., 1972, 382.
- 334. N. H. Andersen, A. D. Denniston and D. A. McCrae, J. Org. Chem., 1982, 47, 1145.
- W. S. Johnson, B. Frei and A. S. Gopalan, J. Org. Chem., 1981, 46, 1512. 335.
- 336. K. C. Bank and D. L. Coffen, J. Chem. Soc., Chem. Commun., 1969, 8.
- R. T. Wragg, Tetrahedron Lett., 1969, 4959.
- 338. D. Seebach, E. J. Corey and A. K. Beck, Chem. Ber., 1974, 107, 367.
- 339. M. Schmidt and E. Weissflog, Z. Anorg. Allg. Chem., 1975, 418, 208.
- 340. R. D. Balanson, V. M. Kobal and R. R. Schumaker, J. Org. Chem., 1977, 42, 393.
- 341. K. Fuji, M. Ueda, K. Sumi, K. Kajiwara, E. Fujita, T. Iwashita and I. Miura, J. Org. Chem., 1985, 50, 657.
- 342. K. Fuji, M. Ueda, K. Sumi and E. Fujita, Tetrahedron Lett., 1981, 22, 2005.
- 343. K. Fuji, M. Ueda, K. Sumi and E. Fujita, J. Org. Chem., 1985, 50, 662.
- 344. V. H. Rawal, M. Akiba and M. P. Cava, Synth. Commun., 1984, 14, 1129.
- 345. T. Mandai, K. Hara, T. Nakajima, M. Kawada and J. Otera, Tetrahedron Lett., 1983, 24, 4993.
- 346. T. Mandai, H. Irei, M. Kawada and J. Otera, Tetrahedron Lett., 1984, 25, 2371.
- 347. T. Mandai, M. Takeshita, M. Kawada and J. Otera, Chem. Lett., 1984, 1259.
- 348. T. Mandai, T. Moriyama, Y. Nakayama, K. Sugino, M. Kawada and J. Otera, Tetrahedron Lett., 1984, 25, 5913.
- 349. P. J. Kocienski, Tetrahedron Lett., 1980, 21, 1559.
- 350. D. J. Ager and R. C. Cookson, Tetrahedron Lett., 1980, 21, 1677.
- 351. D. J. Ager, Tetrahedron Lett., 1981, 22, 2803.
- D. J. Ager, J. Chem. Soc., Perkin Trans. 1, 1983, 1131.
 F. Ogura, T. Otsubo, N. Ohira and H. Yamaguchi, Synthesis, 1983, 1006.
- 354. D. J. Ager, Tetrahedron Lett., 1981, 22, 2923.
- 355. D. J. Ager, Tetrahedron Lett., 1980, 21, 4759.
- 356. K. Ogura, M. Yamashita, M. Suzuki and G. Tsuchihashi, Tetrahedron Lett., 1974, 3653.
- K. Ogura and G. Tsuchihashi, Tetrahedron Lett., 1971, 3151. 357.
- 358. J. E. Richman, J. L. Herrmann and R. H. Schlessinger, Tetrahedron Lett., 1973, 3267.
- 359. G. Schill and P. R. Jones, Synthesis, 1974, 117.
- 360. K. Ogura, M. Yamashita, M. Suzuki, S. Furukawa and G. Tsuchihashi, Bull. Chem. Soc. Jpn., 1984, 57, 1637.
- 361. K. Ogura, M. Yamashita, S. Furukawa, M. Suzuki and G. Tsuchihashi, Tetrahedron Lett., 1975, 2767.
- 362. K. Ogura, M. Yamashita and G. Tsuchihashi, Tetrahedron Lett., 1976, 759.
- 363. D. A. Evans, C. E. Sacks, R. A. Whitney and N. G. Mandel, Tetrahedron Lett., 1978, 727.
- 364. Y. Torisawa, H. Okabe and S. Ikegami, J. Chem. Soc., Chem. Commun., 1984, 1602.
- 365. J. A. Marshall and P. G. M. Wuts, Synth. Commun., 1977, 7, 233. 366. J. A. Marshall and P. G. M. Wuts, J. Org. Chem., 1978, 43, 1086.
- 367. R. M. Carlson and P. M. Helquist, J. Org. Chem., 1968, 33, 2596.

- K. A. Carey, O. D. Dailey, Jr. and O. Hernandez, J. Org. Chem., 1976, 41, 3979.
 I. Hori, T. Hayashi and H. Midorikawa, Synthesis, 1974, 705.
 K. Ogura, N. Yahata, K. Hashizume, K. Tsuyama, K. Takahashi and H. Iida, Chem. Lett., 1983, 767.
- 371. (a) K. Ogura, K. Ohtsuki, M. Nakamura, N. Yahata, K. Takahashi and H. Iida, Tetrahedron Lett., 1985, 26, 2455; (b) K. Ogura, M. Suzuki, J. Watanabe, M. Yamashita, H. Iida and G. Tsuchihashi, Chem. Lett., 1982, 813.
- 372. T. Nakai and M. Okawara, Chem. Lett., 1974, 731.
- 373. G. W. Gokel, H. M. Gerdes, D. E. Miles, J. M. Hufnal and G. A. Zerby, Tetrahedron Lett., 1979, 3375.
- C. A. Brandt, J. V. Comasseto, W. Nakamura and N. Petragnani, J. Chem. Res. (S), 1983, 156. 374.
- 375. K. Sachdev and H. S. Sachdev, Tetrahedron Lett., 1976, 4223.
- 376. T. Takeda, S. Naito, K. Ando and T. Fujiwara, Bull. Chem. Soc. Jpn., 1983, 56, 967.
- 377. R. M. Carlson and P. M. Helquist, J. Org. Chem., 1968, 33, 2596.
- 378. B. M. Trost and A. J. Bridges, J. Org. Chem., 1975, 40, 2014.
- 379. P. Laszlo, P. Pennetreau and A. Krief, Tetrahedron Lett., 1986, 27, 3153.
- 380. M. H. Nantz, X. Radisson and P. L. Fuchs, Synth. Commun., 1987, 17, 55.
- 381. D. Seebach, Angew. Chem., Int. Ed. Engl., 1967, 6, 442.
- 382. D. Seebach, Chem. Ber., 1972, 105, 487.
- 383. W. D. Woessner, Chem. Lett., 1976, 43.
- 384. D. Seebach, K.-H. Geiss, A. K. Beck, B. Graf and H. Daum, Chem. Ber., 1972, 105, 3280.
- D. Seebach, Angew. Chem., Int. Ed. Engl., 1967, 6, 443.
 R. A. Ellison, W. D. Woessner and C. C. Williams, J. Org. Chem., 1972, 37, 2757.
- K. Nakamura, M. Higaki, S. Adachi, S. Oka and A. Ohno, J. Org. Chem., 1987, 52, 1414. 387.
- 388. H. Moskowitz, J. Blanc-Guenée and M. Miocque, C. R. Hebd. Seances Acad. Sci., Ser. C, 1968, 267, 898.
- 389. D. E. O'Connor and W. I. Lyness, J. Org. Chem., 1965, 30, 1620.
- 390. S. Bory and A. Marquet, Tetrahedron Lett., 1973, 4155
- 391. E. J. Corey and M. Chaykovsky, J. Am. Chem. Soc., 1965, 87, 1345.
- 392. T. Durst and M. Molin, Tetrahedron Lett., 1975, 63.
- 393. P. Bravo, G. Resnati and F. Viani, Tetrahedron Lett., 1985, 26, 2913.
- 394. G. Tsuchihashi, S. Iriuchijima and M. Ishibashi, Tetrahedron Lett., 1972, 4605.
- 395. E. Guittet and S. Julia, Synth. Commun., 1979, 9, 317

- 396. V. Ceré, C. Paolucci, S. Pollicino, E. Sandri and A. Fava, J. Org. Chem., 1978, 43, 4826. 397. T. Durst, R. Viau and M. R. McClory, J. Am. Chem. 510, 1978, Chem., 1978, 43, 4826.
- T. Durst, R. Viau and M. R. McClory, J. Am. Chem. Soc., 1971, 93, 3077.
- 398. R. Lett, S. Bory, B. Moreau and A. Marquet, Tetrahedron Lett., 1971, 3255.
- 399. A. Rauk, E. Buncel, R. Y. Moir and S. Wolfe, J. Am. Chem. Soc., 1965, 87, 5498.
- 400. T. Durst, R. R. Fraser, M. R. McClory, R. B. Swingle, R. Viau and Y. Y. Wigfield, Can. J. Chem., 1970, 48, 2148.
- 401. K. Nishihata and M. Nishio, J. Chem. Soc., D, 1971, 958.
- 402. K. Nishihata and M. Nishio, J. Chem. Soc., Perkin Trans. 2, 1972, 1730.
- 403. R. Viau and T. Durst, J. Am. Chem. Soc., 1973, 95, 1346.
- 404. J.-F. Biellmann and J. J. Vicens, Tetrahedron Lett., 1974, 2915.
- 405. K. Nishihata and M. Nishio, Tetrahedron Lett., 1976, 1695.
- 406. J.-F. Biellmann and J. J. Vicens, Tetrahedron Lett., 1978, 467
- 407. G. Chassaing, R. Lett and A. Marquet, Tetrahedron Lett., 1978, 471.
- 408. S. Bory, M. J. Luche, B. Moreau, S. Lavielle and A. Marquet, Tetrahedron Lett., 1975, 827.
- 409. S. Lavielle, S. Bory, B. Moreau, M. J. Luche and A. Marquet, J. Am. Chem. Soc., 1978, 100, 1558.
- 410. A. Marquet, Pure Appl. Chem., 1977, 49, 183
- 411. S. Bory, R. Lett, B. Moreau and A. Marquet, Tetrahedron Lett., 1972, 4921.
- 412. R. Lett and A. Marquet, Tetrahedron, 1974, 30, 3379.
- 413. V. Ceré, S. Pollicino, E. Sandri and A. Fava, *Tetrahedron Lett.*, 1978, 5239. 414. P. J. R. Nederlof, M. J. Moolenaar, E. R. de Waard and H. O. Huisman, *Tetrahedron*, 1977, 33, 579.
- 415. R. Tanikaga, K. Hosoya, K. Hamamura and A. Kaji, Tetrahedron Lett., 1987, 28, 3705.
- 416. T. Sato, T. Itoh and T. Fujisawa, Tetrahedron Lett., 1987, 28, 5677.
- 417. C. Walling and L. Bollyky, J. Org. Chem., 1964, 29, 2699.
- 418. S. Wolfe and S. K. Hasan, J. Chem. Soc., Chem. Commun., 1970, 833.
- 419. S. I. Goldberg and M. S. Sahli, J. Org. Chem., 1967, 32, 2059.
- 420. D. W. Emerson and T. J. Korniski, J. Org. Chem., 1969, 34, 4115.
- 421. H. Böhme and B. Clement, Tetrahedron Lett., 1979, 1737.
- 422. V. Reutrakul and V. Rukachaisirikul, Tetrahedron Lett., 1983, 24, 725.
- 423.
- V. Reutrakul and P. Thamnusan, *Tetrahedron Lett.*, 1979, 617. V. Reutrakul, C. Panyachotipun, V. Hahnvajanawong and S. Sotheeswaran, *Tetrahedron Lett.*, 1984, 25, 1825. 424.
- V. Retrakul and K. Herunsalee, Tetrahedron Lett., 1983, 24, 527. 425.
- 426. V. Reutrakul and P. Poochaivatananon, Tetrahedron Lett., 1983, 24, 531.
- 427. J. N. Denis and A. Krief, Tetrahedron Lett., 1979, 3995.
- 428. Y. Masaki, K. Sakuma and K. Kaji, J. Chem. Soc., Chem. Commun., 1980, 434. 429. K. Ogura, S. Furukawa and G. Tsuchihashi, Synthesis, 1976, 202.
- 430. J. Berlan and K. Koosha, J. Organomet. Chem., 1978, 153, 107.
- 431. P. A. Grieco, J. Chem. Soc., Chem. Commun., 1972, 702.
- 432. R. R. Schmidt, H. Speer and B. Schmid, Tetrahedron Lett., 1979, 4277
- 433. G. H. Posner, P. W. Tang and J. P. Mallamo, Tetrahedron Lett., 1978, 3995.
- 434. L. A. Paquette, Acc. Chem. Res., 1968, 1, 209.
- 435. S. D. McGregor and D. M. Lemal, J. Am. Chem. Soc., 1966, 88, 2858.
- 436. J. B. Hendrickson and P. L. Skipper, Tetrahedron, 1976, 32, 1627.
- 437. H. McKennis, Jr., L. B. Turnbull and E. R. Bowman, J. Am. Chem. Soc., 1957, 79, 6342.
- 438. E. M. Kaiser and C. R. Hauser, Tetrahedron Lett., 1967, 3341
- 439. L. Field and J. W. McFarland, J. Am. Chem. Soc., 1953, 75, 5582.
- 440. W. E. Truce and K. R. Buser, J. Am. Chem. Soc., 1954, 76, 3577.
- 441. L. Field, J. R. Holsten and R. D. Clark, J. Am. Chem. Soc., 1959, 81, 2572.
- 442. H. E. Zimmerman and B. S. Thyagarajan, J. Am. Chem. Soc., 1960, 82, 2505.
- 443. P. Koch, Y. Nakatani, B. Luu and G. Ourisson, Bull. Soc. Chim. Fr., Part 2, 1983, 189.
- 444. W. E. Truce and L. B. Lindy, J. Org. Chem., 1961, 26, 1463
- 445. C. G. M. Janssen and E. F. Godefroi, J. Org. Chem., 1982, 47, 3274.
- 446. M. Makosza and A. Tyrala, Synth. Commun., 1986, 16, 419.
- 447. R. R. Regis and A. M. Doweyko, Tetrahedron Lett., 1982, 23, 2539.
- 448. A. Jonczyk and T. Radwan-Pytlewski, J. Org. Chem., 1983, 48, 910.
- 449. O. Possel and A. M. van Leusen, Tetrahedron Lett., 1977, 4229.
- C. J. M. Stirling, Tetrahedron, 1985, 41, 1613. 450.
- Y. Murata, K. Inomata, H. Kinoshita and H. Kotake, Bull. Chem. Soc. Jpn., 1983, 56, 2539. 451.
- 452. P. J. Kocienski, Tetrahedron Lett., 1979, 2649.
- 453. J. M. Photis and L. A. Paquette, J. Am. Chem. Soc., 1974, 96, 4715.
- 454. M. Julia, D. Uguen and A. Callipolitis, Bull. Soc. Chim. Fr., 1976, 519.
- 455. K. Tanaka, K. Ootake, K. Imai, N. Tanaka and A. Kaji, Chem. Lett., 1983, 633.
- 456. M. Julia and B. Badet, Bull. Soc. Chim. Fr., 1975, 1363.
- 457. W. Oppolzer, D. A. Roberts and T. G. C. Bird, Helv. Chim. Acta, 1979, 62, 2017.
- 458. K. Kondo and D. Tunemoto, Tetrahedron Lett., 1975, 1007.
- 459. J. D. Finlay and D. J. H. Smith, Synthesis, 1978, 579
- 460. B. D. Gowland and T. Durst, Can. J. Chem., 1979, 57, 1462.
- 461. M. Julia and B. Badet, Bull. Soc. Chim. Fr., 1976, 525.
- 462. P. A. Grieco and Y. Masaki, J. Org. Chem., 1975, 40, 150.
 463. M. Julia and C. Blasioli, Bull. Soc. Chim. Fr., 1976, 1941.
- 464. J. R. Falck and Y. Yang, Tetrahedron Lett., 1984, 25, 3563.

- 465. J. B. Evans and G. Marr, J. Chem. Soc., Perkin Trans. 1, 1972, 2502.
- 466. J. Vollhardt, H.-J. Gais and K. L. Lukas, Angew. Chem., Int. Ed. Engl., 1985, 24, 610.
- 467. E. M. Kaiser, L. E. Solter, R. A. Schwarz, R. D. Beard and C. R. Hauser, J. Am. Chem. Soc., 1971, 93, 4237.
- 468. R. D. Little and S. O. Myong, Tetrahedron Lett., 1980, 21, 3339.
- 469. J. R. Hwu, J. Org. Chem., 1983, 48, 4432.
- 470. C.-N. Hsiao and H. Shechter, J. Org. Chem., 1988, 53, 2688.
 471. M. Julia and P. Ward, Bull. Soc. Chim. Fr., 1973, 3065.
- 472. L. A. Paquette, T. Wang and C. E. Cottrell, J. Am. Chem. Soc., 1987, 109, 3730.
- 473. L. A. Paquette, Org. React. (N.Y.), 1977, 25, 1.
- 474. W. E. Truce and C. T. Goralski, J. Org. Chem., 1969, 34, 3324.
- 475. A. Ratajczak, F. A. L. Anet and D. J. Cram, J. Am. Chem. Soc., 1967, 89, 2072.
- 476. S. J. Cristol, J. K. Harrington and M. S. Singer, J. Am. Chem. Soc., 1966, 88, 1529.
- 477. A. Nickon and N. H. Werstiuk, J. Am. Chem. Soc., 1967, 89, 3914.
- 478. J. J. Eisch, S. K. Dua and M. Behrooz, J. Org. Chem., 1985, 50, 3674.
 479. J. J. Eisch and J. E. Galle, J. Org. Chem., 1979, 44, 3277.
- 480. H. Nemoto, H. Kurobe, K. Fukumoto and T. Kametani, Chem. Lett., 1985, 259.
- 481. M. Julia and D. Uguen, Bull. Soc. Chim. Fr., 1976, 513.
- 482. H. Nemoto, H. Kurobe, K. Fukumoto and T. Kametani, Heterocycles, 1985, 23, 567.
- 483. S. O. Nwaukwa, S. Lee and P. M. Keehn, Synth. Commun., 1986, 16, 309.
- 484. S. Torii, K. Uneyama, I. Kawahara and M. Kuyama, Chem. Lett., 1978, 455.
- 485. A. Fischli, Q. Branca and J. Daly, Helv. Chim. Acta, 1976, 59, 2443.
- 486. B. Corbel, J. M. Decesare and T. Durst, Can. J. Chem., 1978, 56, 505.
- 487. J. C. Carretero and L. Ghosez, Tetrahedron Lett., 1988, 29, 2059.
- 488. D. Savoia, C. Trombini and A. Umani-Ronchi, J. Chem. Soc., Perkin Trans. 1, 1977, 123.
- 489. B. M. Trost and N. R. Schmuff, J. Am. Chem. Soc., 1985, 107, 396.
- 490. M. Ito, A. Kodama, T. Hiroshima and K. Tsukida, J. Chem. Soc., Perkin Trans. 1, 1986, 905.
- 491. P. S. Manchand, M. Rosenberger, G. Saucy, P. A. Wehrli, H. Wong, L. Chambers, M. P. Ferro and W. Jackson, Helv. Chim. Acta, 1976, 59, 387.
- 492. M. Ochiai, T. Ukita and E. Fujita, Chem. Pharm. Bull., 1983, 31, 1641.
- G. L. Olson, H.-C. Cheung, K. D. Morgan, C. Neukom and G. Saucy, J. Org. Chem., 1976, 41, 3287. 493.
- 494. K. Sato, O. Miyamoto, S. Inoue, F. Furusawa and Y. Matsuhashi, Chem. Lett., 1984, 1105.
- 495. S. Torii, K. Uneyama and K. Hamada, Bull. Chem. Soc. Jpn., 1977, 50, 2503.
- 496. A. Padwa, S. P. Carter, U. Chiacchio and D. N. Kline, Tetrahedron Lett., 1986, 27, 2683.
- Y. Masaki, K. Hashimoto and K. Kaji, Tetrahedron Lett., 1978, 5123. 497.
- 498. A. Fischli and H. Mayer, Helv. Chim. Acta, 1975, 58, 1492.
- 499. K. Uneyama and S. Torii, Tetrahedron Lett., 1976, 443.
- 500. K. Uneyama and S. Torii, Chem. Lett., 1977, 39.
- 501. K. Sato, S. Inoue, A. Onishi, N. Uchida and N. Minowa, J. Chem. Soc., Perkin Trans. 1, 1981, 761.
- 502. A. Fischli and H. Mayer, Helv. Chim. Acta, 1975, 58, 1584.
- 503. A. Fischli, H. Mayer, W. Simon and H.-J. Stoller, Helv. Chim. Acta, 1976, 59, 397.
- 504. K. Sato, O. Miyamoto, S. Inoue, F. Furusawa and Y. Matsuhashi, Chem. Lett., 1983, 725.
- 505. L. A. Paquette and W. A. Kinney, Tetrahedron Lett., 1982, 23, 131.
- 506. P. A. Grieco and Y. Masaki, J. Org. Chem., 1974, 39, 2135.
- M. Julia and D. Arnould, Bull. Soc. Chim. Fr., 1973, 746. 507.
- 508. T. Mandai, Y. Iuchi, K. Suzuki, M. Kawada and J. Otera, Tetrahedron Lett., 1982, 23, 4721.
- 509. Y. Ueno, S. Aoki and M. Okawara, J. Am. Chem. Soc., 1979, 101, 5414.
- 510. H. Takayama, H. Suzuki, T. Nomoto and S. Yamada, Heterocycles, 1986, 24, 303
- 511. S. Yamada, H. Ohsawa, T. Suzuki and H. Takayama, J. Org. Chem., 1986, 51, 4934.
- S. Yamada, H. Ohsawa, T. Suzuki and H. Takayama, Chem. Lett., 1983, 1003. 512.
- 513. T.-S. Chou, H.-H. Tso and L.-J. Chang, J. Chem. Soc., Chem. Commun., 1984, 1323.
- 514. Y.-T. Tao, C.-L. Liu, S.-J. Lee and S.-S. P. Chou, J. Org. Chem., 1986, 51, 4718.
- 515. S. Yamada, T. Suzuki and H. Takayama, *Tetrahedron Lett.*, 1981, 22, 3085. 516. S. Yamada, T. Suzuki, H. Takayama, K. Miyamoto, I. Matsunaga and Y. Nawata, J. Org. Chem., 1983, 48, 3483.
- 517. T.-S. Chou, S.-J. Lee, H.-H. Tso and C.-F. Yu, J. Org. Chem., 1987, 52, 5082.
- 518. R. Bloch and J. Abecassis, Tetrahedron Lett., 1982, 23, 3277.
- 519. J. J. Eisch and J. E. Galle, J. Org. Chem., 1979, 44, 3279.
- 520. N. S. Simpkins, Tetrahedron Lett., 1987, 28, 989.
- 521. C. R. Johnson and C. J. Stark, Jr., J. Org. Chem., 1982, 47, 1193.
- 522. K.-J. Hwang, J. Org. Chem., 1986, 51, 99.
- 523. A. M. van Leusen, R. J. Bouma and O. Possel, Tetrahedron Lett., 1975, 3487.
- 524. D. van Leusen, P. H. F. M. Rouwette and A. M. van Leusen, J. Org. Chem., 1981, 46, 5159.
- 525. J.-M. Beau and P. Sinaÿ, Tetrahedron Lett., 1985, 26, 6189.
- 526. (a) K. Schank, H.-G. Schmitt, F. Schroeder and A. Weber, Justus Liebigs Ann. Chem., 1977, 1116; (b) K. Tanaka, S. Matsui and A. Kaji, Bull. Chem. Soc. Jpn., 1980, 53, 3619.
- (a) S. V. Ley, B. Lygo, F. Sternfeld and A. Wonnacott, Tetrahedron, 1986, 42, 4333; (b) S. V. Ley, B. Lygo 527. and A. Wonnacott Tetrahedron Lett., 1985, 26, 535; (c) C. Greck, P. Grice, S. V. Ley and A. Wonnacott, Tetrahedron Lett., 1986, 27, 5277.
- 528. J. Golinski and M. Makosza, Tetrahedron Lett., 1978, 3495.
- 529. A. Jonczyk and T. Radwan-Pytlewski, Chem. Lett., 1983, 1557.

- 530. J. Vollhardt, H.-J. Gais and K. L. Lukas, Angew. Chem., Int. Ed. Engl., 1985, 24, 696.
- 531. B. M. Trost and M. Shimizu, J. Am. Chem. Soc., 1983, 105, 6757.
- 532. B. M. Trost and R. W. Warner, J. Am. Chem. Soc., 1983, 105, 5940.
- 533. B. M. Trost and P. Quayle, J. Am. Chem. Soc., 1984, 106, 2469.
- 534. F. Benedetti and C. J. M. Stirling, J. Chem. Soc., Chem. Commun., 1983, 1374.
- 535. F. Benedetti and C. J. M. Stirling, J. Chem. Soc., Perkin Trans. 2, 1986, 605. 536. T. Cuvigny, C. Hervé du Penhoat and M. Julia, Bull. Soc. Chim. Fr., Part 2, 1982, 43.
- 537. C. Hervé du Penhoat and M. Julia, Tetrahedron, 1986, 42, 4807.
- 538. J. B. Hendrickson, G. J. Boudreaux and P. S. Palumbo, J. Am. Chem. Soc., 1986, 108, 2358.
- 539. J. B. Hendrickson, G. J. Boudreaux and P. S. Palumbo, Tetrahedron Lett., 1984, 25, 4617.
- 540. B. M. Trost, J. Cossy and J. Burks, J. Am. Chem. Soc., 1983, 105, 1052.
- 541. E. J. Corey, M. Jautelat and W. Oppolzer, Tetrahedron Lett., 1967, 2325. 542. D. M. Roush, E. M. Price, L. K. Templeton, D. H. Templeton and C. H. Heathcock, J. Am. Chem. Soc., 1979, 101, 2971.
- 543. B. M. Trost and M. J. Bogdanowicz, J. Am. Chem. Soc., 1973, 95, 5298.
- 544. B. M. Trost and M. J. Bogdanowicz, J. Am. Chem. Soc., 1971, 93, 3773.
- 545. E. J. Corey, K. Lin and M. Jautelat, J. Am. Chem. Soc., 1968, 90, 2724.
- 546. R. W. La Rochelle, B. M. Trost and L. R. Krepski, J. Org. Chem., 1971, 36, 1126.
- 547. A. Garbesi, Tetrahedron Lett., 1980, 21, 547.
- 548. C. R. Johnson and E. R. Janiga, J. Am. Chem. Soc., 1973, 95, 7692.
- 549. W. E. Truce and D. J. Vrencur, J. Org. Chem., 1970, 35, 1226.
- 550. T. Durst and J. du Manoir, Can. J. Chem., 1969, 47, 1230.
- 551. M. E. Thompson, J. Org. Chem., 1984, 49, 1700.
- 552. D. M. Stout, T. Takaya and A. I. Meyers, J. Org. Chem., 1975, 40, 563. 553. A. M. Moiseenkov, E. V. Polunin and A. V. Semenovsky, Tetrahedron Lett., 1979, 4759.
- 554. W. E. Truce and L. W. Christensen, J. Chem. Soc., Chem. Commun., 1971, 588.

1.4 Alkylations of Other Heteroatom-stabilized Carbanions

DAVID R. CHESHIRE

Fisons Pharmaceutical Division, Loughborough, UK

1.4.1	INTRODUCTION	193
1.4.2	CARBANIONS STABILIZED BY OXYGEN	193
1. 1. 1. 1.	 4.2.1 Preparation by Hydrogen-Metal Exchange 4.2.2 Preparation by Halogen-Metal Exchange 4.2.3 Preparation by Tin-Lithium Exchange 4.2.4 Preparation from α-Sulfides 	194 194 195 196
1.4.3	α-OXYGEN CARBANIONS WITH ADDITIONAL STABILIZATION	196
1. 1. 1. 1.	 4.3.1 Allylic and Benzylic α-Alkoxy Carbanions 4.3.2 α-Alkoxy Carbanions from Protected Cyanohydrins 4.3.3 α-Alkoxy Carbanions Stabilized by Silicon 4.3.4 α-Alkoxy Carbanions Stabilized by Phosphorus 	196 197 198 199
1.4.4	CARBANIONS STABILIZED BY BORON	1 99
1.4.5	CARBANIONS STABILIZED BY SILICON	200
1.4.6	CARBANIONS STABILIZED BY PHOSPHORUS	200
1.4.7	CARBANIONS STABILIZED BY THE HALOGENS	202
1.4.8	CARBANIONS STABILIZED BY ARSENIC, GERMANIUM, TIN, ANTIMONY, LEAD AND BISMUTH	203
1.4.9	REFERENCES	204

1.4.1 INTRODUCTION

The use of heteroatom-substituted carbanions has allowed the chemist to access a broad range of relatively stable and selective reagents capable of a wide range of synthetic transformations.¹⁻³

This chapter deals with the formation of new C—C bonds by alkylation of sp^3 -hybridized carbanions stabilized by a range of heteroatoms, including oxygen, boron, silicon, phosphorus and the halogens. The properties of the same carbanions with additional stabilization provided by an adjacent C—C double bond are also discussed.

1.4.2 CARBANIONS STABILIZED BY OXYGEN

Considering the prevalence of oxygen-containing groups in naturally occurring substances, synthetic methods based upon the reaction of α -alkoxy carbanions would be of considerable value. Such methods have been realized only recently despite theoretical evidence that these carbanions should be relatively stable. The stabilizing effect of the more electronegative first row elements is inductive in nature, whilst

the second row elements stabilize negative charges much more effectively, in part due to their greater polarizability. The first row elements may coordinate to the base, causing an increase in kinetic acidity.² Recently Schleyer *et al.*⁴ have determined theoretically that a major factor contributing to the stabilization of first row heterosubstituted organolithiums is the possibility of bridging between lithium and the heteroatom.

1.4.2.1 Preparation by Hydrogen–Metal Exchange

Early attempts by Ziegler and Gellert⁵ to deprotonate dimethyl ether with *n*-butyllithium gave products apparently derived from the decomposition of methoxymethyllithium. A number of attempts to deprotonate tetrahydrofuran also failed, as the desired metallated species underwent a facile cleavage reaction to give ethylene and the lithium enolate of acetaldehyde.⁶ The successful preparation of the potassium salts of tetrahydrofuran, *t*-butyl methyl ether and dimethyl ether using lithium salt free *t*-butylpotassium was reported by Lehmann and Schlosser.⁷ Corey and Eckrich⁸ have described a method for the preparation of *t*-butoxymethyllithium (1), a synthetic equivalent of the hydroxymethyl anion (Scheme 1). Treatment of the *t*-butyl ethers with anhydrous iron(III) chloride in acetic anhydride affords the acetates.

 $Bu'OCH_3 \xrightarrow{i, ii} [Bu'OCH_2Li] \xrightarrow{iii} Bu'OCH_2CH_2Ph$ (1)

i, Bu^sLi, Bu^tOK, -78 °C; ii, LiBr, 2 equiv., THF; iii, PhCH₂Br

Scheme 1

The research groups of Seebach⁹ and Beak¹⁰ independently developed the use of hindered aryl esters for the preparation of heteroatom-stabilized carbanions. The success achieved in creating anions adjacent to nitrogen via the N,N-dialkylarylamides, and to sulfur via the analogous aryl thioesters led to the preparation of alkyl 2,4,6-triisopropylbenzoates, which have been successfully metallated with a strong base. For example, the lithium reagent (2), prepared as shown in equation (1), can be reacted with a wide range of electrophiles. It is suggested that the stabilization of these anionic carbanions is due to the dipole of the ester.¹¹ The bulky aromatic substituents force the plane of the aromatic ring to be orthogonal to that of the carbonyl, protecting it from the organolithium reagent.



A recent report describes the use of magnesium (2-ethoxy)ethoxide as an additive in reactions to prepare organolithium reagents.¹² The additive moderates the basicity of the prepared anion and prevents unwanted side reactions, for example with the solvent.

1.4.2.2 Preparation by Halogen–Metal Exchange

Schollkopf *et al.*¹³ successfully prepared methoxymethyllithium by the reaction of lithium metal with chloromethyl methyl ether. The Grignard reagents of a variety of α -chlorodialkyl ethers have been prepared.¹⁴ Although thermally unstable, they react with carbonyl compounds. A more recent example is the preparation of 1-ethoxy-1-lithiocyclopropane (3) from 1-bromo-1-ethoxycyclopropane using *t*-butyl-lithium as the metallating agent (Scheme 2).¹⁵ The adducts between (3) and aldehydes or ketones can be rearranged to form cyclobutanes in good yields.



1.4.2.3 Preparation by Tin-Lithium Exchange

The use of organostannanes has considerably expanded the chemistry of α -alkoxy organolithiums.¹⁶ As seen in Section 1.4.2.1, the conditions necessary for the generation of these species by hydrogenlithium exchange are harsh and can result in their destruction. Halogen-metal exchange allows generation of α -alkoxylithiums, but the anions can be contaminated with reactive alkyl halides. These problems can be avoided by the use of tin-lithium transmetallation, producing as the side product the relatively inert tetraorganostannanes. The process is an equilibrium, the position of which is determined by the relative stability (basicity) of the organolithium species. Schollkopf¹⁷ first described the preparation of a number of α -lithio ethers by reaction of *n*-butyllithium with the tetraalkoxystannanes as shown in equation (2).

 $(\text{ROCH}_2)_4\text{Sn} + 4\text{Bu}^{n}\text{Li} \xrightarrow{\text{Et}_2\text{O or THF}} 4\text{LiCH}_2\text{OR} + \text{Bu}^{n}_4\text{Sn} \qquad (2)$

 $\mathbf{R} = \mathbf{Me}$ (60%), Et (70%), Prⁱ (70%), PhCH₂ (65%)

Meyer and Seebach¹⁸ have described the preparation of the 'ate' complex (4) as a nucleophilic hydroxymethylating reagent (Scheme 3). Still¹⁹ improved the potential of the method, approaching the preparation of the required organostannanes by reacting tributylstannyllithium with the appropriate carbonyl compound. The labile nature of the addition products necessitated their protection as the stable ethoxyethyl ethers. The α -lithio ethers were then smoothly generated by treatment of the organostannanes with *n*-butyllithium in THF at -78 °C. Still has demonstrated the use of the overall procedure by synthesizing dendrolasin from furan-3-carbaldehyde and geranyl chloride (Scheme 4).

$$Bu^{n_{3}}SnCH_{2}OH \xrightarrow{i} \left[\begin{array}{c} O \\ Bu^{n_{3}}Sn' \xrightarrow{i} \end{array} \right]^{2} \xrightarrow{ii} RCH_{2}OLi$$
(4)

i, BuⁿLi, 2 equiv., petroleum ether, -20 °C; ii, $RX = n-C_8H_{17}Br$ (45%), PhCH₂Br (45%)



EEO = ethoxyethyl ether

i, Buⁿ₃SnH, LDA, THF, -78 °C; ii, α -chloroethyl ethyl ether, PhNMe₂, CH₂Cl₂, 0 °C; iii, BuⁿLi, THF:glyme, -78 °C; iv, geranyl chloride, -78 °C; v, Li, NH₃, THF, -33 °C

Scheme 4

An exciting new dimension to the chemistry of α -alkoxy organolithium reagents was reported by Still and Sreekumar in 1980.²⁰ Thus, the aldehyde 2-methyl-3-phenylpropanal was treated with tributylstannyllithium and the diastereomeric carbinols were then protected as their methoxymethyl ethers. Each diastereomeric adduct was next treated separately with *n*-butyllithium. The products resulting from trapping with acetone and TMS-CI were isomerically pure, showing that the anions were configurationally stable under the reaction conditions. Retention of configuration was implied by the recovery of starting material when the anions were quenched with tributylstannyl chloride. Thus α -alkoxy organolithiums represented 'a new class of configurationally stable carbanions'. Still reported the preparation of optically active (R)-2-butanol (>95% ee) by chromatographic separation of the diastereomeric Moshers ester derivatives of the addition product between propanal and tributylstannyllithium. Transmetallation of the protected homochiral α -alkoxystannane with *n*-butyllithium (THF, -78 °C) gave the configurationally stable carbanion, which was alkylated with dimethyl sulfate.

A recent report by Chan and Chong²¹ describes the enantioselective reduction of acylstannanes to the α -alkoxy organostannanes by the chiral 2,2'-dihydroxy-1,1'-binaphthyl modified lithium aluminum hydrides. Matteson has also described a possible route to such chiral organostannanes utilizing the chiral α -chloroboronic esters.²²

Gadwood *et al.*¹⁵ have prepared the 1-bromo-1-tributylstannylcyclopropane. The transmetallation proceeds easily in THF at -78 °C. The lower basicity of cyclopropyl carbanions compared with their acyclic analogs explains this ease of metallation. The group of Macdonald and McGarvey²³ has made a thorough investigation of α -alkoxy organolithiums, and estimate that each alkyl substituent destabilized the anion by 3-4 kcal mol⁻¹ (1 cal = 4.18 J). A number of examples demonstrated the utility of tertiary α -alkoxy carbanions for the synthesis of highly substituted carbon centers.

Duchene and Quintard²⁴ have described the preparation of a number of O-ethyl organostannanes, for example by the reaction of ethoxy- α -chloromethyltributylstannane with Grignard reagents. (Diethoxymethyl)tributylstannane has been transmetallated to provide the formyl anion synthetic equivalent (EtO)₂CHLi.²⁵ Another report concerning tin-lithium transmetallation describes the preparation of the configurationally stable α - and β -glycosyllithiums from the α - and β -D-tributylstannylglucopyranosides.²⁶ In addition, α -alkoxy organocopper reagents derived from α -alkoxy organostannanes have been studied.²⁷

1.4.2.4 Preparation from α-Sulfides

Cohen and coworkers²⁸ have described the preparation of α -lithio ethers from α -sulfides by a mechanistically novel route in the presence of lithium (dimethylamino)naphthalenide. The reaction is postulated to proceed through the radical anion produced by single electron transfer (SET) and subsequent homolytic cleavage (loss of thiophenoxide anion) to give a radical. Further reduction of the radical then leads to the α -alkoxy carbanion.

1.4.3 α-OXYGEN CARBANIONS WITH ADDITIONAL STABILIZATION

The introduction of an additional stabilizing influence on the α -alkoxy anion may serve two purposes. Firstly the anions are easier to prepare, generally by hydrogen-lithium exchange. Secondly the effective oxidation state of the α -carbon atom is increased.

1.4.3.1 Allylic and Benzylic α-Alkoxy Carbanions

A number of reviews exist describing the preparation and reactions of allylic heteroatom-stabilized organometallic species.²⁹ Reactions of the ambident anions can occur at either the α - or γ -carbon atoms (with respect to the heteroatom). There is no general rule for predicting the regioselectivity of the reaction, but experimental evidence indicates that alkylation generally occurs at the γ -carbon atom with alkyl or trialkylsilyl allylic ethers.³⁰ Attempts have been made to influence the regioselectivity of the reaction. Thus, Yamamoto *et al.*³¹ have described the use of allylic 'ate' complexes to direct the reaction toward the α -position. The 'ate' complex is formed by reaction of triethylaluminum or trialkylboranes at the least-substituted γ -position, effectively blocking it to further attack by electrophiles.

An example of an oxaallylic carbanion has been described by Oppolzer *et al.*³² In the alkylation of 3-triethylsilyloxypentadienyllithium (5; X = H) the γ -alkylated product predominated. With the anion (5; X = SMe) alkylation occurs exclusively at the γ -carbon atom. The synthetic potential of the anions (5) was demonstrated by the synthesis of bicyclic compounds *via* intramolecular Diels-Alder reactions of the alkylation products (Scheme 5).

Hoppe²⁹ has investigated the regioselectivity of the reaction of oxaallylic anions generated from the N,N-diisopropyl carbamates of a variety of allylic alcohols. With carbonyl compounds reaction was found to take place predominantly at the γ -carbon. The directing influence of the carbamoyloxy group was unfortunately diminished in alkylation reactions.



i, 1-bromo-2,4-pentadiene, THF; ii, KF, MeOH, -10 °C; iii, NaIO₄, aqueous MeOH, then refluxing CCl₄

The rearrangement of 1-(trimethylsilyl)allylic alcohols to 3-(trimethylsilyloxy)allyllithiums in the presence of base is a convenient source of lithium homoenolates.³³ These can be alkylated to yield the corresponding silyl enol ethers.

 α -Alkoxy carbanions generated from benzyl ethers are unstable and undergo the Wittig rearrangement in preference to reaction with electrophiles.³⁴ Yeh has reported, however, that deprotonation of benzyl methyl ether by *n*-butyllithium in the presence of TMEDA leads to the stable benzylic anion, which could then be alkylated.³⁵ Katriztky *et al.*³⁶ have described the successful lithiation of lithium benzyl carbonate generated by the reaction of the lithium benzylalkoxide with carbon dioxide. The chromium tricarbonyl complex of benzylic ethers also allows metallation to occur without rearrangement. A variety of bases can be employed for benzylic deprotonation. Davies *et al.*³⁷ have described the preparation of chromium tricarbonyl complexes of benzyl ethers and their subsequent alkylation chemistry (Scheme 6).



i, Cr(CO)₆, heat; ii, MeOH, HBF₄; iii, BuⁿLi, THF, -40 °C; iv, RX = MeI, EtBr, PrⁱBr, PhCH₂Br;

v, sunlight or air, Et₂O

Scheme 6

1.4.3.2 a-Alkoxy Carbanions from Protected Cyanohydrins

Protected cyanohydrins, when deprotonated by a suitable base, are synthetic equivalents of the acyl anion.³⁸ They display 'umpolung' reactivity as the normally electrophilic carbonyl carbon is transformed into a nucleophile. Stork and Maldonado³⁹ first described the use of such anions for the synthesis of ketones. They found that the cyanohydrins, protected as their ethoxyethyl ethers can be deprotonated with LDA (Scheme 7). After reaction with the alkylating reagent the carbonyl function is restored by treatment with dilute acid and then base. A number of alkylating reagents were investigated, and no problems from competing elimination reactions were noted. The same authors also found similar reactivity in the more stable anions derived from aromatic aldehydes. Deprotonation of the cyanohydrin derivatives of α , β -unsaturated aldehydes gives rise to the ambident allyl anion. Interestingly, Stork found that alkylation occurred at the α -position allowing the synthesis of α , β -unsaturated ketones.



EEO = ethoxyethyl ether

Scheme 7

The alkylation chemistry of a considerable number of O-trimethylsilyl cyanohydrins derived from aryl and heteroaryl aldehydes has been reported by Hunig and coworkers.⁴⁰ The protected cyanohydrins are easily prepared by heating the aldehydes with trimethylsilyl cyanide in the presence of a Lewis acid. The use of dialkyl sulfates and tosylates as alkylating agents was also reported. Hata *et al.*⁴¹ failed to alkylate the anion of the adduct of trimethylsilyl cyanide and acetaldehyde, suggesting that the trimethylsilyl group is incompatible with the more basic anions derived from aliphatic aldehydes. Ficini *et al.*⁴² demonstrated the utility of cyanohydrins in their synthesis of (\pm) -juvabione; a more recent example is the synthesis of (\pm) -trichostatin A.⁴³ Stork and his coworkers have applied the alkylation procedure to the synthesis of small rings,⁴⁴ for example in prostaglandins PGE₁⁴⁵ and PGF_{2α}⁴⁶ as shown in equation (3). The cyanohydrin served as an effective equivalent of the sensitive carbonyl function.



Takahashi, Tsuji and coworkers have demonstrated that the intramolecular alkylation of carbanions derived from protected cyanohydrins can be applied to the synthesis of both large and medium-sized rings. Their first report described the synthesis of cyclohexadecanone.⁴⁷ Slow addition of a THF solution of the protected cyanohydrin of 16-iodohexadecanal to a THF solution of sodium hexamethyldisilazane gave the cyclized product. Treatment with acid and then base gave cyclohexadecanone in 60% overall yield. An advantage of the procedure is that high dilution conditions are not necessary. The method was later extended to the synthesis of several naturally occurring macrocyclic lactones.³⁸ Another development was to employ the cyclization reaction for the synthesis of germacrone, which contains the thermally labile (*E*,*E*)-1,5-cyclodecadiene system, serves to illustrate the strategy (Scheme 8). Deprotonation of (6) and cyclization yielded, after restoration of the carbonyl group, germacrone (7) in 66% yield.



i, NaN(SiMe₃)₂, THF; ii, PTSA, MeOH, 0 °C; iii, aqueous 2% NaOH, Et₂O

Scheme 8

Garcia et al.⁴⁹ have described the synthesis of α -hydroxy- γ -lactones by the alkylation of protected cyanohydrin anions with epoxides.

Eisch and Galle⁵⁰ have described the alkylation with methyl iodide of the configurationally stable anions derived by deprotonation of α,β -epoxyalkylsilanes. The alkylation chemistry is presently restricted to methylation, which is unfortunate as the α,β -epoxyalkylsilanes are easy to prepare and undergo a range of synthetically useful reactions.⁵¹ Unsubstituted α -lithio epoxides derived by tin–lithium transmetallation decompose at -85 °C.

Recently Yoshida *et al.*⁵² have employed silyl-stabilized α -alkoxy organolithium reagents for the synthesis of a variety of carbonyl compounds. Methoxy(trimethylsilyl)methane and methoxybis(trimethylsilyl)methane, when deprotonated with Bu^sLi and BuⁿLi respectively, give anions which can be alkylated with a variety of electrophiles. Electrolysis of a solution of the alkylated product in methanol yields, by virtue of the reduced oxidation potential of ethers α -substituted with silicon, either the dimethyl acetal or in the latter case the orthoester. The mildness of the electrolytic process recommends the method for the preparation of a variety of carbonyl compounds.

1.4.3.4 a-Alkoxy Carbanions Stabilized by Phosphorus

Hata and coworkers⁴¹ have reported the preparation of unsymmetrical ketones employing the adducts of aldehydes with diethyl trimethylsilylphosphite (8). The adducts are easily deprotonated and then alkylated; treatment of the alkylated products with base provides the ketone (Scheme 9).



R = H, Me, Et, Pr^{i} , $n-C_{7}H_{15}$, etc.; R'X = MeI, Etl, $PhCH_{2}Br/I$, etc.

i, Me₃SiOP(OEt)₂, (8), benzene; ii, LDA, THF, -78 °C; iii, R'X

Scheme 9

The anions generated from the adducts of α , β -unsaturated aldehydes and (8) are alkylated exclusively at the γ -position. Evans *et al.*⁵³ have also reported the alkylation of similar oxaallylic anions and the alkylation chemistry of diethyl (2-trimethylsilyl)ethoxymethylphosphonate has been reported.⁵⁴

Sturtz et al.⁵⁵ have described the deprotonation of the allyl bis(dimethylamido)phosphonate (9). Treatment of (9) with two equivalents of base, followed by reaction with an epoxide and hydrolysis, yields the γ -lactone (Scheme 10). Alkylation with alkyl and alkenyl halides occurs at the γ -carbon and hydrolysis yields the saturated and unsaturated carboxylic acids respectively.



i, BuⁿLi, 2 equiv., THF, -50 °C; ii, propylene oxide; iii, H⁺, H₂O

Scheme 10

1.4.4 CARBANIONS STABILIZED BY BORON

The stabilization of a negative charge adjacent to boron has been known for some time.⁵⁶ Zweifel and Arzoumanian⁵⁷ demonstrated the facile base-induced cleavage of a geminal diboroalkane to give the boron-stabilized anion, which was then alkylated. A later approach employed 3,6-dimethylborepane (10) as the hydroborating agent.⁵⁸ After dihydroboration, addition of one equivalent of methyllithium forms the boron-stabilized anion, which upon alkylation and oxidation gives the secondary alcohol (Scheme 11). Matteson⁵⁹ has described the alkylation of the anion derived from the base cleavage of tris(dimethoxyboryl)methane.

Rathke and Kow⁶⁰ first reported the preparation of a boron-stabilized carbanion by direct deprotonation of the carbon acid. They made the important observation that the deprotonation needed a sterically demanding base to prevent its complexation with boron. Thus the anion of *B*-methyl-9-borabicyclo[3.3.1]nonane, prepared by deprotonation with lithium 2,2,6,6-tetramethylpiperidine (LiTMP) in benzene, can be alkylated successfully.

Matteson and coworkers have described the deprotonation and alkylation of bis(1,3,2-dioxaborin-2-yl)methane (11).⁶¹ Deprotonation with LiTMP yields the stabilized anion, which can be alkylated with a variety of primary alkyl halides. Oxidation under basic conditions then unmasks the carbonyl group. Matteson has also described the deprotonation of some substituted boronic esters (12; X = SPh, SiMe₃).^{62,63} The alkylated products derived from (12; X = SPh) can be treated with *N*-chlorosuccinimide under mild basic conditions to yield either the monothioacetals or the acetals.⁶⁴ Boron-stabilized carbanions can also be generated by transmetallation of the organostannanes.⁶⁵

Pelter and Wilson⁶⁶ have described the deprotonation of the air stable dimesitylboranes. The use of two mesityl groups represents a compromise between steric shielding of the boron against the possible complex formation with the base, and easy oxidation of the product. Alkylation of the anion generated



i, MeLi, l equiv., 0 °C; ii, allyl bromide; iii, 3 N aqueous NaOH, 30% H₂O₂

Scheme 11



by deprotonation of dimesitylmethylborane can be effected with a range of primary alkyl halides. The secondary anions derived from ethyl- and propyl-dimesitylborane can also be alkylated, but oxidation is more difficult. Allyldimesitylborane was also deprotonated using mesityllithium. Alkylation with primary halides occurs at the γ -position and the resulting *trans*-vinylboranes can be oxidized to give aldehydes.

1.4.5 CARBANIONS STABILIZED BY SILICON

The preparation and use of silicon-stabilized carbanions is well documented;⁶⁷ the vast majority of reports are concerned with the reaction of these reagents with carbonyl compounds to form alkenes.

Tamao *et al.*⁶⁸ have described the use of (diisopropoxymethyl)silylmethylmagnesium chloride as a reagent for effecting hydroxymethylation of reactive alkyl halides. The alkylation of tris(trimethyl-silyl)methyllithium has also been reported.⁶⁹

There are a number of examples where additional stabilization for the α -silicon anion is derived from the presence of one or more halogen atoms. For example, Larson and Rosario⁷⁰ have described the alkylation of a number of trialkylsilyldichloromethyllithiums; dehydrochlorination yields the *trans*-1-chloro-alkenylsilanes.

Chan and coworkers⁷¹ have investigated the regioselectivity in the alkylation of α -silicon allyllithiums. The alkylation of α -trimethylsilylallyllithium usually occurs at both the α - and γ -carbons, but if the reagent is prepared using Schlosser's base (Bu'OK/BuⁿLi in hexane), γ -alkylation predominates, affording the *trans*-alkenylsilanes. Chan⁷² has also recently described the successful use of metal-chelating substituents on the silyl group to direct the reaction toward the α -position.

1.4.6 CARBANIONS STABILIZED BY PHOSPHORUS

The Wittig reaction of phosphorus ylides derived from phosphonium salts with carbonyl compounds to form alkenes is widely used in synthetic chemistry. The alkylation of phosphorus ylides offers a route to a variety of phosphonium salts which can be further manipulated. Bestman and coworkers⁷³ have been responsible for a large number of reports based upon this and other aspects of phosphorus chemistry. Their work includes many examples of the intramolecular alkylation of ylides derived from phosphonium salts to form cyclic compounds. The salts can be prepared from dihalides either with triphenylphos-

phine or triphenylphosphine methylide. The cyclic phosphonium salt formed can then be eliminated to form an alkene, or reduced or deprotonated to give a new ylide.

Axelrod et al.⁷⁴ have described the synthesis of 1,5-dienes by alkylation of an allylic phosphonium ylide with an allylic bromide. The method was used to synthesize pure all-(E)-squalene from (E,E)-farme-syl bromide and its tributylphosphonium salt. No isomerization occurred during the reaction and reduction of the resulting phosphonium salt with lithium in ethylamine gave squalene in 65% yield.

Alkylation chemistry has also been reported for alkylphosphine oxides and dialkyl phosphonates. The phosphonate carbanions are often preferred as they offer many advantages over the phosphonium ylides.⁷⁵ In particular, the phosphonate group is highly acidifying and the resulting organolithium can be smoothly alkylated.

Warren and coworkers⁷⁶ have described the alkylation of (diphenylphosphinoyl)alkyllithiums with epoxides as an effective means of synthesizing β -(diphenylphosphinoyl) ketones, homoenolate anion equivalents. The treatment of the adducts of lithiated phosphine oxides and epoxides with base to form cyclopropanes was reported by Toscano *et al.*⁷⁷

Pietrusiewicz and coworkers⁷⁸ have developed methods for the synthesis of homochiral phosphine oxides. They play an important role in asymmetric synthesis as chiral ligands for catalysts. The readily prepared (S_p) -ethyl[(menthoxycarbonyl)methyl]phenylphosphine oxide (13) can be deprotonated, alkylated and then decarboxylated to afford the homochiral (R_p) -alkylethylphenylphosphine oxides (Scheme 12). Pietrusiewicz and Zablocka have also described the synthesis of chirally pure α,β -unsaturated phosphine oxides.⁷⁹



i, NaH, THF, 0 °C; ii, RX = CD₃I, EtI, H₂C=CHCH₂Br, PhCH₂Br; iii, LiCl, wet DMSO, 180 °C

Scheme 12

Recently there has been interest in the synthesis of analogs of the naturally occurring mono- and diphosphate group. Corey and Volante⁸⁰ have reported the alkylation of (dimethylphosphoryl)methyllithium with, for example, geranyl, farnesyl and 3-methyl-2-butenyl bromides. The products were assessed for their ability to inhibit the biosynthesis of squalene. McClard and coworkers⁸¹ described the alkylation of the phosphonylphosphinyl anion (14) to yield the 'P—C—P—C' compounds, proven enzyme inhibitors (Scheme 13). The free acid derivatives were obtained by treatment of the alkylated products with bromotrimethylsilane.



 $RX = EtI, Pr^{i}I, Me_{2}C=CHCH_{2}Br, geranyl bromide, PhCH_{2}Br, etc.$

i, NaH, THF, 20 °C; ii, BuⁿLi, -78 °C; iii, RX, -78 °C; iv, H⁺, H₂O

Scheme 13

Alkylation of (diethoxyphosphoryl)methylcopper(I)⁸² and (dialkoxyphosphoryl)trimethylsilylalkyllithiums⁸³ with a variety of alkyl iodides and bromides has been reported by Savignac and coworkers. Tsuge *et al.*⁸⁴ have demonstrated the regioselective dipolar addition of (diethoxyphosphoryl)acetonitrile oxide with monosubstituted alkenes to yield the 3-[(diethoxyphosphoryl)methyl]-2-isoxazolines. The phosphonates can be deprotonated and alkylated in good yields.

Alkylation of Carbon

The incorporation of fluorine into biologically active compounds can greatly enhance their potency; fluoroalkyl phosphonates are isosteres of the natural phosphonate monoesters. The alkylation chemistry of α -fluoro- and α, α -difluoro-alkyl phosphonates has been reported.⁸⁵ Burton and Sprague⁸⁶ have described the alkylation of (diethoxyphosphoryl)difluoromethylzinc bromide with allylic halides; the reaction is catalyzed by copper(I) bromide. The alkylation of (diethoxyphosphoryl)dichloromethyllithium was reported by Normant and coworkers.⁸⁷ The products can be transmetallated and further alkylated.

A method for the preparation of alkenes has been described by Tunemoto and coworkers,⁸⁸ whereby regioselective α -alkylation of γ -substituted allyl phosphonates, followed by reduction with LAH, gives the (*E*)-alkenes (Scheme 14).



i, BuⁿLi, THF, -60 °C; ii, X = Cl, Br, 25 °C; iii, LiAlH₄, Et₂O, 0 °C

Scheme 14

A range of heteroaryl methyl phosphonates have been alkylated with alkyl halides; the anions were generated with sodium amide in liquid ammonia.⁸⁹

1.4.7 CARBANIONS STABILIZED BY THE HALOGENS

Organolithiums bearing a halogen on the α -carbon are termed carbenoids.⁹⁰ Bromine and chlorine are sufficiently acidifying to allow preparation of the halocarbenoids, but fluorine requires additional stabilization, for example by phosphorus (*vide supra*). Lithium halocarbenoids are thermally unstable species whose chemistry must be performed at low temperatures and this feature has hindered their use in preparative synthetic organic chemistry.

Villieras and coworkers⁹¹ have contributed a considerable number of synthetic methods employing lithium halocarbenoids. An early example was the alkylation of dibromomethyllithium and higher homologs (Scheme 15). The anion (15) is easily generated from 1,1-dibromoalkanes by deprotonation with LDA in THF at low temperature. Alkenes can then be generated simply by treating the products with BuⁿLi in Et₂O. Alkylation of dichloromethyllithium with primary alkyl halides followed by reaction of the dichloroalkanes with BuⁿLi represents a method for the preparation of 1-alkynes.⁹²

> $R \xrightarrow{Br} LDA \qquad R \xrightarrow{Br} Li \qquad R'X \qquad R \xrightarrow{Br} R' + LiBr$ Br THF-Et₂O, -90 °C Br 43-93% Br Br

> > R = H, Bu^n , Me_3Si ; $R'X = H_2C=CHCH_2Br$, Bu^nI , $n-C_5H_{11}I$, etc.

Scheme 15

Geminal substituted dibromoalkenes can be prepared by the alkylation of dibromomethyllithium with α -chloroalkyl methyl ethers.⁹³ Deprotonation of the alkylation products results in the elimination of methanol and the formation of the corresponding 1,1-dibromoalkenes. Despite the lower acidity of 1-bromo-1-chloroalkyllithiums relative to the dibromo analogs they exhibit similar nucleophilic properties.⁹⁴ Alkyl dichloroacetates can also be deprotonated with lithium diethylamide and alkylated with a range of alkyl halides.⁹⁵

Cyclopropylidenelithium halocarbenoids are easily prepared by halogen–lithium exchange of the dibromocyclopropanes, and they undergo a variety of synthetically useful reactions including alkylation.⁹⁶ A recent report by Hiyama *et al.*⁹⁷ for example, describes the synthesis of alkyl-substituted 2-cycloheptenones (Scheme 16).

Seyferth and coworkers have prepared 1,1-dichloroallyllithium⁹⁸ and 1,1-difluoroallyllithium.⁹⁹ The former can be alkylated, the reaction occurring at the 'CCl₂' terminus, *i.e.* with α -regioselectivity. The latter reagent, however, is rather unstable and its alkylation chemistry has not yet been explored.



 $RX = Me_2C = CHCH_2Br (46\%) \text{ or } n - C_5H_{11}I (40\%)$

i, BuⁿLi, THF, HMPA, -95 °C; ii, RX; iii, MeOH, K₂CO₃; iv, H⁺, H₂O

Scheme 16

1.4.8 CARBANIONS STABILIZED BY ARSENIC, GERMANIUM, TIN, ANTIMONY, LEAD AND BISMUTH

The chemistry of carbanions stabilized by the heavy main group elements has been extensively investigated by Kauffmann, and a thorough description of his work can be found in a recent review.¹⁰⁰ There are many similarities between the elements. The anions stabilized by the heavy main group elements listed above can generally be alkylated, and in some cases their reaction with carbonyl compounds is a useful alternative to the Wittig reaction. All the stabilized metallomethyl lithium anions (except bismuth) exhibit marked thermal stability.

Among the heavy main group elements arsenic has received the most attention. Arsonium ylides react with carbonyl compounds to yield either the alkenes or epoxides, depending upon the structure of the ylide and the reaction conditions.¹⁰¹ Despite their toxicity, arsenic compounds can be handled safely.

Diphenylarsinomethyllithium (16) can be prepared by transmetallation of bis(diphenylarsino)methane (Scheme 17), or by halogen-lithium exchange of diphenylarsinomethyl iodide. The anion derived by the former method can be alkylated.

 $(Ph_{2}As)_{2}CH_{2} \xrightarrow{Hu^{n}Li, 4 \text{ equiv.}}_{THF} \begin{bmatrix} Ph \\ As \\ Ph \\ Li \end{bmatrix} \xrightarrow{RX} \xrightarrow{Ph} As \\ Ph \\ RX \xrightarrow{Ph}$

RX = $Pr^{n}I$ (93%), $Bu^{n}Br$ (85%), $n-C_{8}H_{17}Br$ (57%)

Scheme 17

Alkylarsino compounds undergo halogenolysis of the metal-carbon bond as do the bismuth and antimony analogs (Scheme 18). Diphenylarsinomethyllithium can also be alkylated with epoxides.



Scheme 18

The major drawback to the use of diphenylarsinomethyllithium is the necessity to use a fourfold excess of *n*-butyllithium in order to prepare it. The arsane oxides offer a good alternative, as the diphenylarsinoyl group is strongly acidifying. Diphenylarsinoylmethane is easily deprotonated with LDA in THF and can be alkylated with alkyl halides and epoxides.

Addition of organolithium reagents to diphenylarsenoethene provides another route to α -arseno anions. This reaction is general to the second and third row elements (Si, P, S, Ge, As and Se) but does not occur with elements from the fourth and fifth rows (Sn, Sb, Te, Pb and Bi). Unlike their phosphorus analogs, triphenylarsonium salts cannot be prepared by reaction of triphenylarsine with an alkyl halide. Their preparation can be achieved by reaction of triphenylarsine with the more electrophilic alkyl triflates. It is also possible to alkylate arsonium ylides.^{101,102}

The organometallic species Ph_nMCH_2Li (17; M = Sn, Pb, n = 3; M = Sb, Bi, n = 2) can be prepared by transmetallation of $(Ph_nM)_2CH_2$ with phenyllithium. In the case of germanium, the anion (17; M = Ge, n = 3) is prepared from the lead-, selenium- or tin-substituted triphenylgermanylmethane. The synthesis of (17) is sometimes possible *via* halogen-metal exchange of the appropriate iodide. Alkylation of the an-

ions, except (17; M = Bi, n = 2), is possible with alkyl halides and with epoxides in preparatively useful yields. With (17; M = Sb) transmetallation with copper is necessary for efficient alkylation.

1.4.9 REFERENCES

- 1. A. Krief, Tetrahedron, 1980, 36, 2531.
- 2. H. W. Gschwend and H. R. Rodriguez, Org. React. (N.Y.), 1979, 26, 1.
- 3. J. C. Stowell, 'Carbanions in Organic Synthesis', Wiley, New York, 1979, p. 81.
- 4. P. v. R. Schleyer, T. Clark, A. J. Kos, G. W. Spitznagel, C. Rohde, D. Arad, K. N. Houk and N. G. Rondan, J. Am. Chem. Soc., 1984, 106, 6467.
- K. Ziegler and H. G. Gellert, Justus Liebigs Ann. Chem., 1950, 567, 185 (Chem. Abstr., 1950, 44, 7757d).
 M. E. Jung and R. B. Blum, Tetrahedron Lett., 1977, 3791 and refs. cited therein.
- 7. R. Lehmann and M. Schlosser, Tetrahedron Lett., 1984, 25, 745.
- 8. E. J. Corey and T. M. Eckrich, Tetrahedron Lett., 1983, 24, 3165.
- 9. R. Schlecker, D. Seebach and W. Lubosch, Helv. Chim. Acta, 1978, 61, 512.
- 10. P. Beak and L. G. Carter, J. Org. Chem., 1981, 46, 2363 and refs. cited therein; P. Beak and D. B. Reitz, Chem. Rev., 1978, 78, 275.
- 11. N. G. Rondan, K. N. Houk, P. Beak, W. J. Zajdel, J. Chandrasekhar and P. v. R. Schleyer, J. Org. Chem., 1981, 46, 4108.
- 12. C. G. Screttas and B. R. Steele, J. Org. Chem., 1989, 54, 1013.
- 13. U. Schöllkopf, H. Kuppers, H. J. Traencker and W. Pitteroff, Justus Liebigs Ann. Chem., 1967, 704, 120 (Chem. Abstr., 1967, 67, 43 843).
- 14. E. Taeger and C. Fiedler, Justus Liebigs Ann. Chem., 1966, 696, 42 (Chem. Abstr., 1966, 65, 19 976) and refs. cited therein; H. Normant and B. Castro, C. R. Hebd. Seances Acad. Sci., 1964, 259, 830 (Chem. Abstr., 1965, 62, 7748a).
- 15. R. C. Gadwood, M. R. Rubino, S. C. Nagarajan and S. T. Michel, J. Org. Chem., 1985, 50, 3255.
- 16. M. Pereyre, J.-P. Quintard and A. Rahm, in 'Tin in Organic Synthesis', Butterworth, Guildford, 1986, p. 149.
- 17. U. Schöllkopf, Method. Org. Chem. (Houben-Weyl), 1970, 13 (1), 87.
- 18. N. Meyer and D. Seebach, Chem. Ber., 1980, 113, 1290; Angew. Chem., Int. Ed. Engl., 1976, 15, 438 (Chem. Abstr., 1980, 93, 45 488).
- 19. W. C. Still, J. Am. Chem. Soc., 1978, 100, 1481.
- 20. W. C. Still and C. Sreekumar, J. Am. Chem. Soc., 1980, 102, 1201.
- P. C.-M. Chan and J. M. Chong, J. Org. Chem., 1988, 53, 5584.
 D. S. Matteson, Acc. Chem. Res., 1988, 21, 294.
- 23. J. S. Sawyer, A. Kucerovy, T. L. Macdonald and G. J. McGarvey, J. Am. Chem. Soc., 1988, 110, 842.
- 24. A. Duchene and J.-P. Quintard, J. Chem. Soc., Chem. Commun., 1987, 29 and refs. cited therein.
- 25. J.-P. Quintard, B. Elissondo and M. Pereyre, J. Organomet. Chem., 1981, 212, C31.
- 26. P. Lesimple, J.-M. Beau and P. Sinaÿ, J. Chem. Soc., Chem. Commun., 1985, 894.
- 27. R. J. Linderman, A. Godfrey and K. Horne, Tetrahedron, 1989, 45, 495 and refs. cited therein.
- 28. T. Cohen and M.-T. Lin, J. Am. Chem. Soc., 1984, 106, 1130; T. Cohen and J. R. Matz, J. Am. Chem. Soc., 1980, 102, 6900.
- 29. D. Hoppe, Angew. Chem., Int. Ed. Engl., 1984, 23, 932; J.-F. Biellmann and J. B. Ducep, Org. React. (N.Y.)., 1982, **27**, 1.
- 30. D. A. Evans, G. C. Andrews and B. Buckwalter, J. Am. Chem. Soc., 1974, 96, 5560; W. C. Still and T. L. Macdonald J. Am. Chem. Soc., 1974, 96, 5561.
- 31. Y. Yamamoto, H. Yatagai, Y. Saito and K. Maruyama, J. Org. Chem., 1984, 49, 1096.
- W. Oppolzer, R. L. Snowden and P. H. Briner, Helv. Chim, Acta, 1981, 64, 2022 and refs. cited therein.
 M. Kato, A. Mori, H. Oshino, J. Enda, K. Kobayashi and I. Kuwajima, J. Am. Chem. Soc., 1984, 106, 1773; see also G. Trimitsis, S. Beers, J. Ridella, M. Carlon, D. Cullin, J. High and D. Brutts, J. Chem. Soc., Chem. Commun., 1984, 1088.
- 34. U. Schöllkopf, Angew. Chem., Int. Ed. Engl., 1970, 9, 763.
- 35. M. K. Yeh, J. Chem. Soc., Perkin Trans. 1, 1981, 1652.
- 36. A. R. Katritzky, W.-Q. Fan and K. Akutagawa, Synthesis, 1987, 415.
- 37. J. Blagg, S. G. Davies, C. L. Goodfellow and K. H. Sutton, J. Chem. Soc., Perkin Trans. 1, 1987, 1805 and refs. cited therein.
- 38. For a review see J. D. Albright, Tetrahedron, 1983, 39, 3207
- 39. G. Stork and L. Maldonado, J. Am. Chem. Soc., 1971, 93, 5286.
- 40. K. Deuchert, U. Hertenstein, S. Hünig and G. Wehner, Chem. Ber., 1979, 112, 2045 (Chem. Abstr., 1979, 91, 107 841).
- 41. M. Sekine, M. Nakajima, A. Kume, A. Hashizume and T. Hata, Bull. Chem. Soc. Jpn., 1982, 55, 224.
- 42. J. Ficini, J. d'Angelo and J. Noire, J. Am. Chem. Soc., 1974, 96, 1213.
- 43. I. Fleming, J. Iqbal and E.-P. Krebs, Tetrahedron, 1983, 39, 841.
- 44. G. Stork, J.-C. Depezay and J. d'Angelo, Tetrahedron Lett., 1975, 389.
- 45. G. Stork and T. Takahashi, J. Am. Chem. Soc., 1977, 99, 1275.
- 46. G. Stork, T. Takahashi, I. Kawamoto and T. Suzuki, J. Am. Chem. Soc., 1978, 100, 8272.
- 47. T. Takahashi, T. Nagashima and J. Tsuji, Tetrahedron Lett., 1981, 22, 1359.
- 48. T. Takahashi, H. Nemoto and J. Tsuji, Tetrahedron Lett., 1983, 24, 2005, 3485; T. Takahashi, K. Kitamura and J. Tsuji, Tetrahedron Lett., 1983, 24, 4695; T. Takahashi, K. Kitamura, H. Nemoto and J. Tsuji, Tetrahedron Lett., 1983, 24, 3489.
- 49. G. A. García, H. Muñoz and J. Tamariz, Synth. Commun., 1983, 13, 569.

- 50. J. J. Eisch and J. E. Galle, J. Organomet. Chem., 1988, 341, 293.
- 51. C. Burford, F. Cooke, G. Roy and P. Magnus, Tetrahedron, 1983, 39, 867.
- 52. J. Yoshida, S. Matsunaga and S. Isoe, Tetrahedron Lett., 1989, 30, 219.
- 53. D. A. Evans, J. M. Takacs and K. M. Hurst, J. Am. Chem. Soc., 1979, 101, 371.
- 54. J. Binder and E. Zbiral, Tetrahedron Lett., 1984, 25, 4213 (Chem. Abstr., 1985, 102, 62 332).
- 55. G. Sturtz, B. Corbel and J.-P. Paugam, Tetrahedron Lett., 1976, 47 (Chem. Abstr., 1976, 84, 121 072).
- 56. For a review, see A. Pelter, Chem. Soc. Rev., 1982, 11, 191.
- 57. G. Zweifel and H. Arzoumanian, Tetrahedron Lett., 1966, 2535.
- 58. G. Zweifel, R. P. Fisher and A. Horng, Synthesis, 1973, 37.
- 59. D. S. Matteson, Synthesis, 1975, 147.
- 60. M. W. Rathke and R. Kow, J. Am. Chem. Soc., 1972, 94, 6854.
- 61. D. S. Matteson and R. J. Moody, Organometallics, 1982, 1, 20.
- 62. D. S. Matteson and K. H. Arne, Organometallics, 1982, 1, 280.
- 63. D. S. Matteson and D. Majumdar, Organometallics, 1983, 2, 230.
- 64. A. Mendoza and D. S. Matteson, J. Chem. Soc., Chem. Commun., 1978, 357.
- 65. D. S. Matteson and J. W. Wilson, Organometallics, 1985, 4, 1690.
- 66. A. Pelter, B. Singaram, L. Williams and J. W. Wilson, Tetrahedron Lett., 1983, 24, 623; A. Pelter, L. Williams and J. W. Wilson, Tetrahedron Lett., 1983, 24, 627; A. Pelter, B. Singaram and J. W. Wilson, Tetrahedron Lett., 1983, 24, 631; see also R. Kow and M. W. Rathke, J. Am. Chem. Soc., 1973, 95, 2715.
- 67. E. W. Colvin, in 'Silicon in Organic Synthesis', Butterworth, Guildford, 1981; L. A. Paquette, Science (Washington, D.C.), 1982, 217, 793.
- 68. K. Tamao, N. Ishida and M. Kumada, J. Org. Chem., 1983, 48, 2120.
- 69. I. Fleming and C. D. Floyd, J. Chem. Soc., Perkin Trans. 1, 1981, 969; D. Seyferth, J. L. Lefferts and R. L. Lambert, Jr., J. Organomet. Chem., 1977, 142, 39.
- 70. G. L. Larson and O. Rosario, J. Organomet. Chem., 1979, 168, 13.
- 71. K. Koumaglo and T. H. Chan, Tetrahedron Lett., 1984, 25, 717.
- 72. R. F. Horvath and T. H. Chan, J. Org. Chem., 1989, 54, 317.
- 73. For a review, see H. J. Bestmann, Bull. Soc. Chim. Fr., 1971, 1619.
- 74. E. H. Axelrod, G. M. Milne and E. E. van Tamelen, J. Am. Chem. Soc., 1970, 92, 2139.
- 75. J. Boutagy and R. Thomas, Chem. Rev., 1974, 74, 87.
- 76. A. Bell, A. H. Davidson, C. Earnshaw, H. K. Norrish, R. S. Torr, B. D. Trowbridge and S. Warren, J. Chem. Soc., Perkin Trans. 1, 1983, 2879; D. Levin and S. Warren, Tetrahedron Lett., 1986, 27, 2265.
- 77. L. Horner, H. Hoffmann and V. G. Toscano, Chem. Ber., 1962, 95, 536 (Chem. Abstr., 1962, 56, 15 537d).
- 78. K. M. Pietrusiewicz, M. Zablocka and J. Monkiewicz, J. Org. Chem., 1984, 49, 1522.
- 79. K. M. Pietrusiewicz and M. Zablocka, Tetrahedron Lett., 1989, 30, 477.
- 80. E. J. Corey and R. P. Volante, J. Am. Chem. Soc., 1976, 98, 1291.
- 81. M. H. B. Stowell, J. F. Witte and R. W. McClard, Tetrahedron Lett., 1989, 30, 411.
- 82. P. Savignac, A. Breque and F. Mathey, Synth. Commun., 1979, 9, 487.
- 83. E. E. Aboujaoude, S. Liétjé, N. Collignon, M. P. Teulade and P. Savignac, Synthesis, 1986, 934.
- 84. O. Tsuge, S. Kanemasa, H. Suga and N. Nakagawa, Bull. Chem. Soc. Jpn., 1987, 60, 2463.
- 85. G. M. Blackburn and M. J. Parratt, J. Chem. Soc., Chem. Commun., 1983, 886; M. Obayashi, E. Ito, K. Matsui and K. Kondo, Tetrahedron Lett., 1982, 23, 2323. 86. D. J. Burton and L. G. Sprague, J. Org. Chem., 1989, 54, 613.
- 87. J. Villieras, A. Reliquet and J. F. Normant, Synthesis, 1978, 27 and refs. cited therein.
- 88. K. Kondo, A. Negishi and D. Tunemoto, Angew. Chem., Int. Ed. Engl., 1974, 13, 407.
- 89. V. Lachkova and M. Kirilov, Justus Liebigs Ann. Chem., 1974, 496 (Chem. Abstr., 1974, 81, 49 741).
- 90. For a review, see H. Siegel, Top. Curr. Chem., 1982, 106, 55.
- 91. J. Villieras, M. Rambaud, B. Kirschleger and R. Tarhouni, Bull. Soc. Chim. Fr., 1985, 837 (Chem. Abstr. 1987, 106, 50 270).
- 92. J. Villieras, P. Perriot and J. F. Normant, Synthesis, 1979, 502; Bull. Soc. Chim. Fr., 1977, 765 (Chem. Abstr., 1978, 88, 152 688).
- 93. P. Perriot, J. F. Normant and J. Villieras, C. R. Hebd. Seances Acad. Sci., Ser. C, 1979, 289, 259 (Chem. Abstr. 1980, 92, 163 509).
- 94. J. Villieras and M. Rambaud, C. R. Hebd. Seances Acad. Sci., Ser. C, 1980, 290, 295 (Chem. Abstr., 1980, 93, 113 860).
- 95. J. Villieras, J. R. Disnar, P. Perriot and J. F. Normant, Synthesis, 1975, 524.
- 96. K. Kitatani, T. Hiyama and H. Nozaki, J. Am. Chem. Soc., 1975, 97, 949.
- 97. T. Hiyama, A. Kanakura, H. Yamamoto and H. Nozaki, Tetrahedron Lett., 1978, 3047.
- 98. D. Seyferth, G. J. Murphy and B. Mauzé, J. Am. Chem. Soc., 1977, 99, 5317.
- 99. D. Seyferth, R. M. Simon, D. J. Sepelak and H. A. Klein, J. Am. Chem. Soc., 1983, 105, 4634.
- 100. For a review, see T. Kauffmann, Angew. Chem., Int. Ed. Engl., 1982, 21, 410.
- 101. W. C. Still and V. J. Novack, J. Am. Chem. Soc., 1981, 103, 1283.

1.5 Alkylations of Nonstabilized Carbanions

JANICE M. KLUNDER* and GARY H. POSNER

Johns Hopkins University, Baltimore, MD, USA

1.5.1 INTRODUCTION	207
1.5.2 ORGANOCOPPER REAGENTS	208
 1.5.2.1 Preparation of the Reagents 1.5.2.2 Nature of the Reagents 1.5.2.2 Catalytic copper compounds [RM + CuX(cat)] 1.5.2.2 Monoorganocopper compounds (RCu) 1.5.2.2.3 Heterocuprates [RCu(Z)Li] 1.5.2.2.4 Homocuprates (R2CuLi) 1.5.2.2.5 Mixed homocuprates (R2CuLi)CuLiCuLi 1.5.2.2.6 Higher-order cuprates [R2Cu(Z)Li₂, Rm + nCumLin (m + n > 2)] 1.5.2.3 Mechanism of the Reaction 1.5.2.4.1 Alkyl halides and sulfonates 1.5.2.4.2 Alkenyl halides and related electrophiles 1.5.2.4.3 Alkynyl halides 1.5.2.4.4 Aryl halides 1.5.2.4.5 Allylic, benzylic and propargylic electrophiles 1.5.2.4.6 Epoxides 1.5.2.4.7 Allylic, the curve of the related electrophiles 1.5.2.4.6 Epoxides 	208 210 210 211 211 211 212 213 215 215 215 217 219 219 219 220 223 225
1.5.2.4.8 Acetals 1.5.2.4.9 β-Lactones	226 227
 1.5.3 NICKEL AND PALLADIUM CATALYSTS 1.5.3.1 Organonickel Catalysis 1.5.3.2 Organopalladium Catalysis 1.5.3.2.1 Aluminum, zinc and zirconium reagents 1.5.3.2.2 Boron reagents 1.5.3.2.3 Tin reagents 1.5.3.2.4 Silicon reagents 	227 228 230 230 231 231 233
1.5.4 REFERENCES	233

1.5.1 INTRODUCTION

The formation of carbon-carbon bonds is truly one of the cornerstones of organic synthesis. While the early organic chemists had the ability to modify the molecules of nature and to degrade them into simpler components, the possibility that complex molecules could be constructed from simpler precursors appeared inconceivable. The development of methodology for the construction of C—C bonds revolutionized that way of thinking, and the field of organic synthesis was born.

Conceptually, perhaps the simplest method of forming a σ -bond between two carbon fragments is the reaction of a carbanion with an electrophilic carbon center. Nature relies heavily upon reactions of carb-

*Current address Boehringer Ingelheim Pharmaceuticals Inc., Ridgefield, CT, USA

anions stabilized by adjacent electron-withdrawing substituents, which can be generated under equilibrium conditions at physiological pH. Similarly stabilized carbanions form the basis of much commonly used synthetic methodology as well, as discussed in Chapters 1.1-1.4 of this volume and also in a number of chapters in Volumes 1, 2 and 4 of the series. However, by a variety of techniques, the chemist is also able to introduce a metal atom at a carbon center that lacks heteroatom stabilization.

The properties and reactivity of the organometallic compounds so produced vary greatly, depending upon the metal counterion.¹ The bonds of carbon to the most electropositive metals, such as sodium and potassium, are predominantly ionic in character. Carbanions of this type are strongly basic, highly reactive and poorly able to discriminate between electrophiles. Also highly reactive are the alkyl and aryl compounds of lithium, magnesium and aluminum, in which the M—C (M = metal) bond is strongly polarized, albeit essentially covalent in nature. By contrast, less electropositive metals form covalent, only slightly polar bonds with carbon, producing organometallic reagents that are less prone to undergo undesired side reactions with functional groups present in the substrate, but that are also correspondingly less reactive. An additional metal catalyst is often necessary in order to achieve suitable reactivity with these reagents.

In this chapter, the substitution reactions of organometallic reagents with organic halides and related electrophiles are reviewed.^{2,3} The major portion of the chapter is devoted to a discussion of organocopper compounds, which first transformed the alkylation of nonstabilized carbanions into a reaction of general synthetic utility. More recently, transition metals other than copper have also found widespread application in such coupling reactions, and developments in this area are outlined in Section 1.5.3.

1.5.2 ORGANOCOPPER REAGENTS

Although organoalkali metal reagents derived from stabilized carbanions can usually be alkylated smoothly in high yields, those derived from weak hydrocarbon acids are prone to undergo numerous competing side reactions (*e.g.* metal-halogen exchange, α -metalation, α - and β -elimination, and homo-coupling to form symmetrical dimers). Neither type of reagent is effective in direct substitution reactions of vinyl or aryl halides. Other organometallics, such as organo-aluminum, -zinc, -cadmium and -magnesium reagents, although often useful for specific reactions, are also susceptible to undesirable side reactions or are unreactive toward important classes of noncarbonyl electrophiles.

That the chemical behavior of traditional organolithium and Grignard reagents can be significantly altered in the presence of copper salts has been known for many years, prominent examples being the Cu¹catalyzed conjugate addition of Grignard reagents to α,β -unsaturated compounds⁴ and the transition metal catalyzed coupling of aryl Grignard reagents with vinyl halides,⁵ both reported by Kharasch in the 1940s. Stoichiometric organocopper compounds also are not new, the oldest members of the class, copper acetylides,⁶ having been reported as early as 1859. Phenylcopper⁷ was prepared by Reich in 1923, and Gilman and coworkers reported preparations of methylcopper⁸ and lithium dimethylcuprate⁹ in 1936 and 1952, respectively. However, it was not until the mid-1960s that the widespread synthetic utility of organocopper reagents was finally appreciated. Their use in coupling reactions with organic electrophiles has been reviewed previously.¹⁰

The past 20 years have witnessed a virtual explosion in research involving organocopper reagents, as reactions employing them have become established in the organic chemist's growing arsenal of consistently reliable methodology. However, the term 'organocopper reagent' itself encompasses a vast array of reagents, possessed of widely divergent chemical behavior. The formulae by which these compounds are represented are indicative of stoichiometry only and are not true reflections of structure, about which information has traditionally been scarce. In addition, factors such as solvent, method of reagent preparation and the nature and purity of the copper source have proven to be critically important in determining the stability and reactivity of the reagents. Not surprisingly, many chemists have tended to view organo-copper chemistry as an art as much as a science. Fortunately, during the past decade a rapidly increasing volume of structural and mechanistic data have become available,¹¹ promising eventually to dispel the mystery surrounding reactions of both stoichiometric and catalytic organocopper reagents.

1.5.2.1 Preparation of the Reagents

Commonly employed organocopper reagents can be divided into five general classes:

- (1) catalytic organocopper compounds [RM + CuX(cat)]
- (2) monoorganocopper compounds (RCu)

(3) homocuprates (R₂CuM, RR'CuM)

(4) heterocuprates [RCu(Z)M]

(5) higher-order cuprates $[R_2Cu(Z)Li_2, R_{m+n}Cu_mLi_n (m+n > 2)].$

In each case, an additional ligand (usually an amine, phosphine, phosphite or sulfide) or other additive (usually a Lewis acid, especially BF_3) may be present, altering the physical and chemical properties of the reagent.

Typically, organocopper reagents are prepared according to one of the general methods outlined in equations (1)–(3) (with or without additional ligands). The choice of the copper salt employed in these reactions is often critical to the outcome. For example, in early investigations, Cu(SCN) and CuI were often used interchangeably, it being assumed that each gave rise to the same organocuprate, according to equation (1). It has since been established, however, that both Cu(SCN) and CuCN, when combined with 2 equiv. of an organolithium reagent RLi, produce, not the expected homocuprate R₂CuLi, but rather the higher-order mixed organocuprates R₂Cu(Z)Li₂ (Z = CN, SCN; equation 3). Even among the copper(I) halides differences are often observed, and debate continues as to whether the use of CuI or CuBr is to be preferred.¹² While copper(I) iodide continues to be most commonly employed for the preparation of dialkylcuprates, copper(I) bromide must be used for the preparation of diarylcuprates.

$$nRM + CuX \longrightarrow R_nCuM_{n-1} + MX$$
 (1)

 $nRM + mRCu \longrightarrow R_{(m+n)}Cu_mM_n$ (2)

$$nRM + CuZ \longrightarrow R_nCu(Z)M_n$$
 (3)

Since the presence of trace impurities (*e.g.* Cu^{II} salts) may have a profound effect on the course of reactions employing organocopper reagents, the purity of the copper source is of critical importance. Copper(I) iodide is usually purified by soxhlet extraction or recrystallization,¹³ followed by *in vacuo* drying, while copper(I) bromide is recrystallized as its dimethyl sulfide complex.¹⁴ If CuBr·SMe₂ is to be used, it is important that the reagent not be dried *in vacuo*, since such handling may significantly alter its composition.¹⁵ Tedious and repetitive recrystallizations of CuBr·SMe₂ can be avoided by simply treating the reaction solution containing CuBr·SMe₂ with copper wire, which serves to reduce any Cu^{II} that might be present.¹⁶ Copper(I) cyanide offers the advantages that it is neither hygroscopic nor light sensitive, it is less easily oxidized than copper(I) halides, and it is less expensive than CuI.

A number of metals (M) have been employed in reactions (1)–(3), including lead, zinc, mercury, boron, aluminum, thallium, zirconium, lithium and magnesium, the latter two being the most commonly used. Catalytic organocopper compounds are best prepared from Grignard reagents, while organolithiums usually provide the most effective stoichiometric organocopper compounds. This requirement for organo-lithium or -magnesium precursors represents a significant synthetic limitation, prohibiting the formation of many reagents that bear functional groups with which the organocopper compound itself, once formed, would be compatible. In an effort to circumvent this difficulty, Behling¹⁷ has prepared higher-order vinyl cuprate reagents by *in situ* transmetalation between Me₂CuCNLi₂ and the corresponding vinylstannane. Similarly, Knochel has reported¹⁸ the successful preparation of functionalized benzylic cuprates by transmetalation of the organozine bromides.

Particularly exciting within this context is recent research in the Reike group,¹⁹ which has demonstrated that organocopper compounds can be prepared directly from organic halides upon treatment with the highly reactive zerovalent copper obtained by lithium naphthalenide reduction of copper(I) halides. Oxidative addition of the reactive copper to alkyl bromides occurs at temperatures as low as -78 °C, while aryl halides react at room temperature or below. Importantly, ester, nitrile, chloride, nitro, epoxide and ketone functionalities are all compatible with this procedure to varying degrees. The method works best in the presence of phosphine ligands, which render the reduced copper more reactive toward oxidative addition, and which also serve to produce a more nucleophilic organocopper species. Moderate to good yields have been obtained in cross-coupling reactions with acid chlorides and alkyl halides.

Several other methods of limited generality have been used to prepare specific types of organocopper compounds. Copper(I) acetylides are routinely prepared from the corresponding alkyne and a copper(I) halide in ammoniacal solution or in HMPA.²⁰ Metalation of compounds with acidic hydrogens has also been accomplished with Cu(OBu¹). Alkenylcopper reagents have been prepared by stereospecific *syn* addition of organocopper reagents to alkynes.²¹ Finally, it should be noted that the Ullmann biaryl synthesis,²² thought to proceed *via in situ* generated arylcopper intermediates, may be considered to be an early predecessor to the Rieke method for the preparation of organocopper reagents.
1.5.2.2 Nature of the Reagents

1.5.2.2.1 Catalytic copper compounds [RM + CuX(cat)]

Copper(I)-catalyzed reactions of organolithiums and Grignard reagents have been reviewed.¹⁰ The term catalytic here refers to the use of copper salts in amounts not exceeding 20 mol %; commonly employed quantities range from 1 to 20%, with 5 and 10% being most typical. While CuCl was an oftenused catalyst in the years following Kharasch and Tawney's orginal report,⁴ its virtual insolubility in ether has led to the more frequent use of CuI, CuBr or Kochi's catalyst, Li₂CuCl₄.²³ Although coppercatalyzed reactions of organometallics are generally believed to proceed by way of *in situ* generated organocopper compounds, little is known about the exact nature of the reactive species involved.

1.5.2.2.2 Monoorganocopper compounds (RCu)

Alkylcopper reagents decompose readily at temperatures above -15 °C, with β -hydrogens and branching at the α -position contributing to decreased stability. By contrast, phenylcopper is stable in an inert at mosphere up to 100 °C, while copper(I) acetylides are even resistant toward hydrolysis. In general, the stability of RCu increases in the order alkyl < aryl \approx alkenyl < alkynyl. Steric hindrance near the copper-carbon bond has a stabilizing influence on aryl- and alkenyl-copper compounds.¹¹ Organocopper reagents are more stable in THF than in ether, and thermal stability is further enhanced in the presence of donor ligands such as sulfides, phosphines and amines. The use of dimethyl sulfide as the solvent in reactions of alkylcopper compounds has recently been recommended.²⁴ Since solvent molecules occupy all available coordination sites on copper, hydride transfer cannot occur and decomposition by β -elimination is effectively minimized. In addition, organocopper reagents form more readily at lower temperatures in dimethyl sulfide than in ethereal solvents, solubility may be increased, and enhanced reactivity has also been observed.

Polynuclear structures have been suggested for most organocopper compounds, which tend to be insoluble in conventional organic solvents. In dimethyl sulfide solution, phenylcopper apparently exists as an equilibrium mixture of trimeric and tetrameric aggregates, while methylcopper is a yellow precipitate, presumably of polymeric composition.²⁴ A recurrent feature of known organocopper crystal structures^{11a} is a central metal atom core, in which copper typically adopts a digonal or trigonal coordination geometry. The formation of 1:1 ArCu-CuBr adducts is not uncommon,²⁵ and the exact synthetic protocol often plays a critical role in determining the nature of the cuprate species isolated.

Particularly interesting bonding patterns have been observed for arylcopper derivatives with nitrogencontaining *ortho* substituents (1-3).²⁶ Tetrameric in the solid state, with the four copper atoms positioned in a butterfly arrangement, these compounds adopt one of the configurations schematically depicted in Scheme 1. As indicated by X-ray crystallography, diastereomer (6) is the ground-state structure for compound (1), while for (2) it may represent an intermediate in the interconversion of enantiomeric configurations (4) and (5), a process that is rapid on the NMR time-scale. The situation is more complex for



Scheme 1

compound (3), which possesses a chiral *ortho* substituent. The ways in which this ligand chirality influences the structures of the derived arylcopper and arylcuprate compounds may have relevance to the design of chiral cuprate reagents for asymmetric synthesis.

1.5.2.2.3 Heterocuprates [RCu(Z)Li]

The enhanced thermal stability of organocopper reagents imparted by the presence of donor molecules, as well as the high value of some R groups, prompted the development of heterocuprates, in which the donor ligand is incorporated into the reagent itself. Prepared according to the general equation (3), the same reagent is obtained regardless of the order of addition. Thus, the lithium anion of the donor ligand can be added to a solution of the alkylcopper compound, or an alkyllithium reagent can be added to a preformed heterocopper compound. Dicyclohexylamido-, dicyclohexylphosphido- and diphenylphosphido-cuprates have been particularly recommended, based on the enhanced thermal stability and reactivity of these reagents relative to other heterocuprates.²⁷ More recently, it has been reported²⁸ that di-*t*-butylphosphidocuprates exhibit significantly improved stability, while maintaining reactivity comparable to that of other phosphidocuprates. In contrast to diphenylphosphidocuprates, for which dimeric structures have been suggested, X-ray analysis indicates that RCuP(Bu^t)₂Li is a monomer, presumably a result of the steric hindrance of the *t*-butyl substituents. In the crystal structure, both phosphorus and lithium exhibit tetrahedral geometries, with solvent molecules occupying three of the coordination sites on lithium (Figure 1).



Figure 1

1.5.2.2.4 Homocuprates (R₂CuLi)

Diorganohomocuprates continue to be the most widely used organocopper reagents. These compounds are both more stable and more nucleophilic than their monoorganic counterparts.²⁹ They are, however, considerably less basic and less nucleophilic than the corresponding alkyllithiums. Like other organocopper compounds, diorganocuprates are very sensitive to heat and oxidants; oxidative decomposition to give the symmetrical organic dimer R—R occurs readily in the presence of oxygen, even at -78 °C. Of the dialkylcuprates, Me₂CuLi is the most stable, showing little decomposition at 0 °C, while secondary and tertiary organocuprates are best prepared at low temperature (-78 °C) in the presence of a stabilizing ligand. Very recently, the immobilization of cuprate reagents on solid supports has been described.³⁰ The dry polymer-bound cuprates are stable for up to two to three weeks at room temperature when stored under an argon atmosphere.

The dimeric structure (7) has long been suggested for Me₂CuLi, based on ¹H NMR, solution X-ray and vapor pressure depression data.³¹ The proposed Cu₂Li₂ metal core with bridging organic substituents is, in fact, evidenced in the crystal structure of Cu₂Li₂(o-C₆H₄CH₂NMe₂)₄ (Figure 2a),³² the only neutral cuprate for which X-ray crystallographic data are available. On the basis of spectroscopic information, a similar structure (Figure 2b) has also been proposed for Cu₂Li₂(p-tolyl)₄·2Et₂O, in which two solvent molecules are required to fill the lithium coordination sites occupied by the o-amino substituents in Figure 2(a). Evidence suggests that, in all of these compounds, the organic moiety is bonded more strongly to copper than to lithium, creating 'nucleophilic' and 'electrophilic' sites within the metal cluster.³³

In the most pivotal structural investigation of organocuprates in recent years, Lipshutz reported^{34a} the startling discovery that the classical Gilman reagent, Me₂CuLi, exists in halide-free THF solution, not as a discrete species, but rather as an equilibrium mixture containing significant amounts of free methyllithium. Attesting to the often crucial role played by lithium salts, the addition of a stoichiometric amount of LiI suppresses the equilibrium, and no methyllithium is detected. Interestingly, no equilibrium is observed when Me₂CuLi is prepared in ether, even if THF is then added to the mixture. However, the





reagent is spectroscopically different from that prepared in THF or THF-ether.³⁵ Clearly, much remains to be learned about organocuprate reagents, but these findings do carry important synthetic implications.

The addition of BF₃·OEt₂ to a solution of an organocopper compound often results in a reagent of enhanced reactivity and/or selectivity.³⁶ That organocopper compounds are compatible with BF₃·OEt₂ at low temperature (*i.e.* transmetalation does not occur) has long been established, and the observed experimental results have been attributed primarily to substrate activation by complexation with the Lewis acid. However, recent spectroscopic and chemical evidence^{34b,c} suggests that, when used in combination with an organocuprate, BF₃·OEt₂ serves not only to activate the substrate, but also to modify the cuprate reagent itself. In particular, one equivalent of alkyllithium is irreversibly sequestered from the R₄Cu₂Li₂ reagent in the form of RLi·BF₃, leaving R₃Cu₂Li as the species predominantly responsible for chemical reactivity. Excess BF₃·OEt₂ must then function in its second, substrate-activating capacity, since R₃Cu₂Li is comparatively unreactive in its absence.

Corey has reported³⁷ the preparation of yet another type of modified cuprate reagent by mixing lithio dimethylcuprate and water in a 3:1 ratio. Of proposed structure (8), this new reagent exhibits enhanced reactivity and stereoselectivity in reactions with enones, effects that are proposed to be due to an increased ability of lithium to coordinate to the substrate. The Corey group has also prepared a rationally designed chiral heterocuprate reagent,³⁸ which undergoes highly enantioselective conjugate addition reactions with cycloalkenones.

1.5.2.2.5 Mixed homocuprates (RR'CuLi)CuLiCuLi

An obvious disadvantage of homocuprates is the fact that one potentially valuable organic ligand is necessarily wasted. Therefore, over the years considerable effort has been devoted to the search for suitable nontransferable (or 'dummy') ligands. A number of heteroatom substituents have been used for this purpose (see Section 1.5.2.2.3), and it should be noted that RCu(SPh)Li allows the successful transfer of secondary and tertiary alkyl groups, a feat not readily accomplished by homocuprates.³⁹ However, heterocuprates, including cyanocuprates, are generally less reactive than homocuprates. Alkynyl ligands have often been successfully employed as 'dummy' ligands; MeOC(Me)₂C=CH offers the advantage that the corresponding monoorganocopper compound is soluble in organic solvents. Most recently, the use of DMSO as a nontransferable ligand has been proposed.⁴⁰ Since product isolation is often complicated by the presence of added ligands, the fact that DMSO can be simply washed away represents a significant advantage.

1.5.2.2.6 Higher-order cuprates $[R_2Cu(Z)Li_2, R_{m+n}Cu_mLi_n (m+n>2)]$

In contrast to earlier reports,⁴¹ recent spectroscopic evidence^{34a} indicates that cuprate reagents in which the stoichiometric ratio of RLi:RCu is >1:1 are not discrete species in ether or THF solution, but rather mixtures of the Gilman reagent, R₂CuLi, and free alkyllithium. The excess alkyllithium present in these mixtures may serve a beneficial role in some cases (*e.g.* converting the by-product RCu back into reactive R₂CuLi), accounting for the improved experimental results that have reportedly been obtained with 'R₃CuLi₂'.⁴² By contrast, the higher-order cuprate Ph₃CuLi₂ is apparently observable in dimethyl sulfide solution.⁴³ Cuprate reagents in which the RLi:RCu ratio is <1:1 have also been described. R₃Cu₂Li is a discrete species in THF, while R₅Cu₃Li₂ is the first soluble cuprate formed upon addition of RLi to RCu in ether.^{42b} The R₅Cu₃Li₂ reagent has proven to be particularly useful for effecting 1,4-conjugate addition to enals,⁴⁴ and it also exhibits better regio- and stereo-selectivity than does the Gilman reagent in S_N2' reactions with allylic carbamates.⁴⁵

Better established are the higher-order cyanocuprates ($R_2CuCNLi_2$) extensively investigated by Lipshutz and coworkers.⁴⁶ These reagents represent a compromise between lower-order cyanocuprates and homocuprates, offering the increased stability of the former, while maintaining reactivity nearly comparable to that of the latter. The mixed gegenion compounds $R_2CuCNLiNa$ and $R_2CuCNMgX$ have also been examined; both are less reactive than the corresponding dilithio cyanocuprates.

Ashby⁴¹ has reported that, like Me₂CuLi, Me₃Cu₂Li is dimeric in THF solution, but that Me₅Cu₃Li₂, which is formed only in ether, is monomeric. Lipshutz^{46c} has presented spectroscopic evidence suggesting that higher-order cyanocuprates (R₂CuCNLi₂) are also dimeric in THF. However, for these reagents, the poorer Lewis basicity of ether apparently promotes higher states of aggregation in that solvent. The spectroscopic investigations conducted by the Lipshutz group have also provided fascinating insight into the dynamic behavior of organocuprates in solution, which stands in sharp contrast to the common perception of the essentially static nature of the reagents.

Like their lower-order homocuprate analogs, higher-order cyanocuprates suffer the disadvantage that only one organic ligand is transferred. Ligand transfer ability decreases in the order Bu > Ph > Me > vinyl, rendering selective vinyl transfer especially difficult. The 2-thienyl (2-Th) group has proven to be a particularly useful 'dummy' ligand for these reagents, allowing even vinyl substituents to be transferred with high selectivity.⁴⁷ The preparation of the mixed cyanocuprate R(2-Th)CuCNLi₂ is most conveniently accomplished by simply adding an alkyllithium reagent to preformed 2-thienylcyanocuprate [(2-Th)CuCNLi], which is stable in THF solution for three weeks or more at room temperature and two months or more at -20 °C.⁴⁸

1.5.2.3 Mechanism of the Reaction

Although much is now known about the coupling of organocopper reagents with organic electrophiles, many details of the reaction mechanism remain incompletely understood. In fact, recent findings indicate that more than one mechanism may be operative, depending upon the nature of the substrate involved (*vide infra*). Several general features of the reaction are clear: (i) the rate law is usually first order in both substrate and organocuprate; (ii) substrate reactivity follows the pattern typical for S_N2 reactions: primary > secondary >> tertiary; (iii) reactions of secondary bromides and tosylates proceed predominantly with inversion of configuration; (iv) alkenyl halides and alkenylcuprates both couple predominantly with retention of configuration; and (v) the reactivity of the transferred group decreases in the order primary > secondary > tertiary. These facts all argue against a process involving free radical intermediates, in agreement with the finding that free radical traps generally have little effect on the course of the reaction.

On the basis of this information, the reaction is now believed to proceed by way of an oxidative addition-reductive elimination pathway, in which the rate-determining oxidative addition step occurs with inversion of substrate configuration by a bimolecular nucleophilic substitution mechanism (Scheme 2). The retention of sp^2 -configuration is also consistent with an oxidative addition process,^{3b,49} and a stereoelectronic rationale for this stereochemical outcome has been proposed.⁵⁰ The intermediate (9) has been the focus of much debate, however, since a Cu^{III} species of this type is expected to be very unstable.^{51,52} In a modification of this mechanism, it has been suggested³¹ that the dimeric structure of the organocuprate reagent plays a critical role in the reaction, allowing the formal 2e⁻ oxidation to take place by way of cooperative single-electron oxidations of both copper atoms in the metal cluster, thereby eliminating the need to generate a high energy Cu^{III} transient intermediate (equation 4).

Inconsistent with both of the above mechanistic proposals is the recent finding that secondary iodides, unlike bromides and sulfonates, undergo nearly complete racemization in coupling reactions with organocuprates.⁵⁴ In addition, bicyclic compounds are produced in reactions of 6-iodocyclooctene with



lithium cuprate reagents (equation 5),⁵⁵ strongly supportive evidence for the intermediacy of free radicals in these reactions. In fact, bicyclic products are also found in reaction mixtures of the alkenic bromide substrate. By contrast, elimination is the only competing pathway observed in reactions of the corresponding p-toluenesulfonate.



An explanation for these results may be found in a comparison of the electrochemical reduction potentials for the various substrates.⁵⁶ Secondary alkyl iodides give two waves at potentials considerably more positive than the single wave exhibited by alkyl bromides, while reductive cleavage of alkyl sulfonates occurs at the S—O bond rather than the C—O bond. Two competitive pathways therefore appear to be operative for reactions of organocuprates: (i) the S_N2 -like mechanism depicted in Scheme 2; and (ii) an alternative mechanism involving single-electron transfer (SET). The observed difference in reactivity order between primary substrates (OTs \ge I > Br > Cl) and secondary substrates (I >> OTs > Br > Cl) is a reflection of this mechanistic duality. Thus, while an energetically favorable electron transfer pathway is available for secondary alkyl iodides, secondary tosylates are constrained to react by the much slower S_N2 -like mechanism. For primary alkyl substrates, on the other hand, the S_N2 -like process is considerably more competitive, reactivity follows the standard order of leaving group ability, and the occurrence of free radical side reactions (*e.g.* cyclization, alkylative cyclization, reduction) is diminished. Even primary iodides give significant amounts of the dehalogenated by-product in reactions with R₂CuCNLi₂, however.⁵⁷

More extensive investigations⁵⁵ have revealed additional complexities in the SET mechanism. Thus, evidence supportive of both free radical and radical anion (S_{RN1}) chain processes has been observed in reactions of alkyl iodides, both pathways presumably being initiated by electron transfer between cuprate and substrate. Although incapable of sustaining free radical chain processes, alkyl bromides do apparently react, at least in part, by an S_{RN1} mechanism. While the intervention of free radical processes in reactions of alkyl bromides would appear to be inconsistent with the predominant stereochemical inversion exhibited by these substrates, Ashby has demonstrated⁵⁸ that electron transfer processes can occur with inversion of configuration.

In summary, while alkyl sulfonates react exclusively by the S_N2 -like mechanism of Scheme 2 (or equation 4), evidence for alternative SET processes in reactions of both alkyl iodides and alkyl bromides has recently accumulated. The relative contributions of S_N2 and SET mechanisms in reactions of alkyl halides is difficult to quantify, however, since the 'normal' substitution product can arguably arise via either pathway.

1.5.2.4 Reactions With Organic Substrates

1.5.2.4.1 Alkyl halides and sulfonates

Primary alkyl halides and sulfonates react readily with organocopper reagents, the order of reactivity being that typical of nucleophilic substitution reactions (OTs > I > Br >> Cl). Since alkene, ester, amide, ketone and nitrile functionalities are all tolerated, the method is of widespread generality, and innumerable synthetic applications have been reported in the literature. The spectacular success enjoyed by organocopper reagents in reactions with unactivated alkyl halides is to be contrasted with the general failure of other transition metal catalyzed cross-coupling reactions of these substrates (see Section 1.5.3), in which β -elimination is usually a predominating side reaction.

Monoorganocopper compounds are generally less reactive than organocuprates and are therefore less often used in reactions with unactivated alkyl halides. Interestingly, however, in one recently reported⁵⁹ reaction, an alkylcopper species derived from the Grignard reagent and one equivalent of CuI coupled satisfactorily at a neopentyl center, while all other reagents failed. Vinylcopper compounds, although not especially reactive, can be alkylated smoothly in the presence of HMPA and a trialkyl phosphite.^{21a} In an application of this procedure, tetrasubstituted alkenes have been prepared in a stereodefined fashion from (2-stannylalkenyl)boranes by sequential transmetalation and alkylation of the borane and stannane moieties.⁶⁰ A similar strategy was utilized in recently reported syntheses of (*E*)- and (*Z*)- γ -bisabolene (10).⁶¹ Alkynyl copper reagents are not effective coupling partners with alkyl halide substrates.



The coupling of organocopper reagents with primary alkyl substrates is often accomplished as effectively under catalytic conditions as it is with stoichiometric organocopper reagents. Whereas a large excess of the stoichiometric reagent is often necessary, the use of copper in catalytic amounts appears to ameliorate this requirement. Alkyl sulfonates are reported to be particularly successful substrates in copper-catalyzed coupling reactions.⁶² Grignard reagents are by far the most frequently employed organometallics in these reactions, but copper-catalyzed alkylations of alkyllithium reagents are not uncommon.^{10c,63} The copper-catalyzed alkylation of a vinyl aluminate has also been reported.⁶⁴ Copper catalysis is known to alter the regioselectivity of reactions of allylic Grignard reagents. Thus, although allylic Grignard reagents ordinarily exhibit a strong preference for reaction at the γ -position, even when that position is the more sterically hindered, under copper catalysis, prenylmagnesium chloride is alkylated at the primary α -position with >99% regioselectivity.⁶⁵

As a rule, stoichiometric cuprate reagents have provided the most consistently successful results in reactions with primary alkyl electrophiles. Diethyl ether is the solvent of choice for reactions of alkyl sulfonates, while reactions of alkyl halides appear to be facilitated by THF. The enhanced basicity of the cuprate reagent in THF may be problematic, however, when racemization of an adjacent chiral center or elimination is a competing side reaction. For example, reactions of serine-derived β -halo esters must be performed in ether, since elimination by-products are the only products isolated when THF is employed as the solvent;⁶⁶ elimination is more problematic with sulfonate than with halide leaving groups.

A number of mixed homocuprates and heterocuprates have been successfully employed in coupling reactions with alkyl halides (see Sections 1.5.2.2.5 and 1.5.2.2.3). The phosphido heterocuprates RCuP(Bu¹)₂Li²⁸ and RCuPCy₂Li^{27b} react readily with primary bromides and iodides, providing coupled products in yields comparable to those obtained with homocuprates. Heterocuprates, of course, offer the advantage that only one equivalent of the organic ligand is required. In a recently reported⁶⁷ synthesis of

Alkylation of Carbon

12-HETE, a new higher-order heterocuprate of the type $RCu(CN)_2NBu_4Li$ was the only reagent to provide satisfactory yields in the coupling of a 1,4-dienylcuprate with an alkyl iodide.

Higher-order cyanocuprates $(R_2CuCNLi_2)^{46}$ have proven to be the most reactive of all organocopper reagents in reactions with alkyl halides.^{68,69} In fact, the reagents are too reactive with primary iodides for the reaction to be of general synthetic utility, since reduction is usually a serious side reaction even at low temperature. Coupling with primary bromides, on the other hand, is facile at -50 °C, and less than 2 equiv. of the cuprate reagent are required. Even primary chlorides, typically very poor substrates, react well with R₂CuCNLi₂, although chemospecific displacement of bromide in the presence of chloride is easily accomplished.

Higher-order cyanocuprates are particularly useful for reactions of secondary alkyl substrates. Although substitution of secondary halides can be successfully effected with lower-order cuprate reagents, results are variable due to competing elimination and reduction side reactions. By contrast, the $R_2CuCNLi_2$ reagents generally provide reproducibly high yields of coupled products in reactions with secondary iodides and bromides. Exceptions are transfers of phenyl and *s*-alkyl, for which the corresponding Gilman reagents are more effective. Secondary chlorides are not reactive with any of the reagents. Interestingly, higher-order cyanocuprates do not react with secondary mesylates and react only sluggishly with secondary tosylates. By contrast, sulfonates are more reactive than halides in reactions with lower-order homocuprates.

Double substitution of geminal dibromides is best accomplished by treatment with an organocuprate, followed by the addition of an alkyl iodide to the crude reaction mixture.⁷⁰ The reaction has found especial application for the synthesis of dimethylcyclopropanes, ubiquitous in natural product chemistry. Excellent yields are often obtained with Me₂CuLi,⁷¹ but with less reactive substrates higher order cyanoand thiocyano-cuprates sometimes provide better results.⁷² Spiro dialkylation of geminal dibromides has also been accomplished.⁷³ Vicinal dibromides, on the other hand, afford alkenes upon treatment with dialkyl cuprates.⁷⁴

The issue of stereochemistry in reactions of organocopper reagents with secondary electrophiles was addressed briefly in Section 1.5.2.3. In general, bromides and sulfonates react predominantly with inversion of configuration, while iodides undergo extensive racemization by a free radical pathway. Iodide (11), with a strong bias for substitution from the *exo* face, was reported⁷⁵ to react with complete retention of configuration (equation 6). The corresponding tosylate, for which an electron transfer pathway is not available, was unreactive. Net retention of configuration has also been observed in reactions of secondary sulfonates, although not in this case by an SET mechanism. In one example, intramolecular carbamate participation was invoked to explain the observed stereochemical results.⁷⁶ Poor stereospecificity in reactions of alkyl sulfonates may also result from a competing double inversion process when I⁻ is present in the reaction mixture.⁷⁷



C-Glycosides have been prepared by the reaction of α -glycosyl halides with dialkylhomocuprates (equation 7).⁷⁸ The reaction proceeded with complete inversion of configuration, despite the fact that α -halo ethers are expected to be especially susceptible to electron transfer processes. Reductive elimination was a serious side reaction, however. With these substrates, higher-order cyanocuprate reagents were surprisingly unreactive, while Grignard reagents gave mixtures containing both α - and β -glycosides, as well as elimination and reduction by-products.



1.5.2.4.2 Alkenyl halides and related electrophiles

Cross-coupling of alkylcopper compounds with alkenyl halides is not generally successful, because the reagents are unstable at the temperatures usually required to effect reaction. Vinyl halides do undergo facile substitution by lithium dialkylcuprate reagents, however. As mentioned in Section 1.5.2.3, alkenic geometry is predominantly retained, but stereochemical scrambling may be observed when an alternative conjugate addition–elimination pathway is available.² In combination with hydrometalation and halometalation methodology, the procedure allows the regio- and stereo-specific preparation of trisubstituted alkenes from alkynic precursors. Methyl substitution is an especially common synthetic operation, due to its application to the synthesis of terpenoid natural products. Substitution reactions of vinyliodonium salts, themselves derived from vinylsilanes, have also been reported.⁷⁹

Cuprate reactions of allenyl halides have been the focus of considerable investigation. The ratio of product resulting by a direct substitution pathway (Scheme 3, path a) to that produced by an S_N2' mechanism (path b) is dependent upon both the nature and structure of the reagent and the substitution pattern of the substrate.⁸⁰ In general, the Gilman reagents, R₂CuLi, afford allenic products exclusively, while RCuCNLi and (RCuBr)MgX·LiBr exhibit highly selective S_N2' reactivity, except when this pathway would be strongly disfavored due to steric hindrance (R = secondary or tertiary alkyl). Unpredictably, phenylcyanocuprate affords the product of direct substitution with high regioselectivity, while (PhCuBr)MgBr·LiBr provides the alkynic S_N2' product with equally high selectivity.



Scheme 3

The stereochemistry of the direct substitution reaction has been the subject of some debate.⁸¹ Most recently, it has been reported^{80b} that reactions of alkylheterocuprates proceed with high *syn* selectivity, while inversion of allenyl configuration, or *anti* selectivity, is observed in reactions of phenylcopper reagents. The degree of selectivity is variable and may be a reflection of product isomerization under the reaction conditions.⁸² Predominant *anti* stereoselectivity (*anti:syn* ratios range from 91:9 to >99:1) is observed also in S_N2' reactions of allenyl halides (see Scheme 3),^{80b,83} a finding that is consistent with the known preference for *anti* substitution of allylic substrates (see Section 1.5.2.4.5). This method for alkyne preparation has found application in the leukotriene area,⁸⁴ and also for the synthesis of alkoxy alkynes.⁸⁵

Vinylcopper reagents, in contrast to their alkyl analogs, have proven to be extremely unreactive toward vinyl halide substrates. However, the desired cross-coupling reaction can be effected in the presence of a palladium catalyst, providing access to stereodefined 1,3-dienes, valuable synthetic intermediates. The procedure has been extensively investigated by Normant and Alexakis,²¹ who have additionally discovered that zinc halide cocatalysts allow the transfer of both vinyl ligands of cuprates derived from Grignard reagents. Likewise, although lithium cuprates transfer only one organic ligand in these zinc-palladium catalyzed reactions, both groups can be successfully transferred if a magnesium halide is also added to the reaction mixture. These reactions are all highly stereospecific, proceeding with virtually exclusive retention of configuration in both components.

In a recent application of similar methodology, α -vinyl acrylates were prepared by palladium-catalyzed cross-coupling of lithium (α -carbalkoxyvinyl)cuprates with vinyl halides.⁸⁶ Use of the dicyclohexylamido group as the nontransferable ligand in this reaction was necessary, since alkyne ligands transferred preferentially over the α -carbalkoxy moiety. It should be noted that (α -carbalkoxyvinyl) cuprates are, in general, extremely unreactive reagents, reacting in the absence of catalysts only with activated halides (allylic, propargylic, acyl, α -halo ethers).⁸⁷ α -Cupriopyrone, a special member of the class, is, in fact, the least nucleophilic organocopper reagent known.⁸⁸

Palladium catalysis is also involved in an exceptionally mild procedure for the stereospecific preparation of enynes.⁸⁹ A catalytic version of the Stephens-Castro coupling,⁹⁰ the reaction presumably involves *in situ* generated copper(I) acetylide intermediates and proceeds with retention of alkenyl configuration. Several experimental modifications have been reported.⁹¹ Notable synthetic applications include ynenol lactone inhibitors of human leukocyte elastase,⁹² an irreversible inhibitor of soybean lipoxygenase,⁹³ and a number of eicosanoid metabolites of arachidonic acid.⁹⁴ Most recently, the reaction has been utilized extensively in synthetic approaches⁹⁵ to the bicyclic enediyne core common to the esperamicin, *e.g.* (12), and calichemicin antitumor agents, compounds that are of considerable biological interest. The reaction sequence depicted in Scheme 4^{95a} is representative of the general approach and is illustrative of the efficiency of the coupling procedure. The coupling of alkynes with allenyl bromides has also been reported.⁹⁶ Palladium-catalyzed coupling reactions of organometallics other than organocopper reagents are discussed in Section 1.5.3.2.



The coupling of organocopper reagents with nonhalide vinyl electrophiles has also been accomplished. Displacement of enol phosphates can be effected in good yields with very reactive dialkylcuprates.⁹⁷ However, less reactive cuprates (*e.g.* Me₂CuLi) give low yields of the coupled product, and hindered enol phosphates are unreactive. Diphenyl phosphates derived from chlorodifluoromethyl aryl ketones do not undergo direct substitution of the enol phosphate moiety, but are rather metalated to give a fluorinated vinylcopper intermediate, which can then be alkylated with allylic halides.⁹⁸ Like enol phosphates, vinyl triflates can be regioselectively prepared from the corresponding ketones, and they have proven to be more reactive than enol phosphates in reactions with cuprates.⁹⁹ Again, retention of alkenyl configuration is the rule, but some isomerization may occur.

Dihydrofuran and dihydropyran substrates react with organocopper reagents to provide homoallylic and bishomoallylic alcohols of (E)-configuration (equation 8).^{100b} This reaction works best with ratios of RLi:CuX well exceeding 2:1; in fact, only 2 mol % of CuCN is required.¹⁰⁰ The stereochemical outcome of the reaction is attributed to the formation of a mixed dihydrofuranylalkylcuprate, followed by migratory insertion with inversion of configuration, a process that is well precedented in organoboron chemistry.

Organometallic substitution of iron-complexed dioxolenes has been reported.¹⁰¹ The reaction proceeds with net inversion of configuration, the result of a two-step addition-elimination pathway. Dialkyl-cuprates, higher order cyanocuprates and Grignard reagents have all been employed, and sequential dis-



placements of both vinyl ether substituents can be accomplished. A single diastereomeric product is produced in reactions of enantiomerically pure dioxolene substrates.¹⁰²

1.5.2.4.3 Alkynyl halides

Like the Stephens–Castro reaction, Cadiot–Chodkiewicz coupling of terminal alkynes with alkynyl halides is presumed to proceed by way of copper(I) acetylide intermediates. The use of preformed copper acetylide reagents is also successful and, in fact, circumvents the problem of homocoupling sometimes encountered in the classical procedure. The reaction is convenient and high yielding, and is compatible with numerous functionalities, including free hydroxy and carboxy groups.¹⁰³ Arylcopper compounds react analogously with alkynyl halides,¹⁰⁴ providing a route to arylalkynes that is complementary to the Stephens–Castro coupling (see Section 1.5.2.4.4).

Metal-halogen exchange is a competing side reaction in the coupling of vinylcopper reagents with alkynyl halides. With monovinylcopper compounds, this problem can be overcome by performing the reaction in a mixture of THF-TMEDA, but lithium divinylcuprates undergo nearly exclusive metal-halogen exchange regardless of the reaction conditions.^{21a} They can, however, be converted to the corresponding vinylcopper compounds by treatment with another equivalent of copper(I) halide prior to the addition of the alkynyl halide substrate, an advantageous procedure in light of the ready availability of divinylcuprates by carbocupration of alkynes.^{21b} High yields of stereoisomerically pure 1,3-enynes can be obtained in this fashion. More recently, Stang and Kitamura have reported the stereospecific preparation of conjugated enynes by direct coupling of vinylcopper compounds with alkynylphenyliodonium tosylates.¹⁰⁵

1.5.2.4.4 Aryl halides

Because aryl ligands are better able than alkyl ligands to adopt the bridging, two-electron three-center bonding pattern characteristic of organocopper reagents (see Section 1.5.2.2),^{11a} metal-halogen exchange predominates in reactions of aryl halides with alkylcopper reagents, especially dialkylcuprates. Therefore, alkyl-aryl coupling is often better accomplished by the alternative reaction of an arylcopper reagent with an alkyl halide. If aryl halide substitution must be effected, RCu·Lig reagents are more successful than R_2CuLi . Alkyl substitution of halopyridines has been accomplished with magnesium dialkylcuprates;¹⁰⁶ copper-free Grignard reagents, lithium dialkylcuprates, and nickel- or palladium-catalyzed coupling procedures were all ineffective for this transformation.

Aryl triflates undergo successful cross-coupling with higher-order cyanocuprates.⁹⁹ Heterocyclic triflates are also reactive, but the yield may be strongly dependent upon the structure of the reagent.¹⁰⁷ Lower-order Gilman reagents, although effective in coupling reactions with vinyl triflates, cause only S—O bond cleavage in reactions with aryl triflates.

The majority of reported reactions of aryl and heteroaryl substrates with organocopper reagents are examples of Stephens-Castro coupling⁹⁰ or the more recent catalytic version of that reaction.⁸⁹ The reaction has found recent application in syntheses of C-(6)-substituted pterins and pteridines,¹⁰⁸ substituted pyridines,¹⁰⁹ and the antitumor antibiotic fredericamycin A,¹¹⁰ to name a few. Aryl iodide can be chemospecifically displaced in the presence of bromide,¹¹⁰ and 2,5-dibromopyridine is regioselectively substituted at the 2-position.¹⁰⁹ Substitution of halobenzenes by propargyl alcohol, followed by oxidative cleavage, provides a convenient route to terminal arylalkynes.¹¹¹ Fused heterocycles are formed in reactions of aryl halides bearing nucleophilic *ortho* substituents.^{90,112}

The classical Ullmann reaction has also attracted renewed interest in recent years. Miyano has prepared optically enriched binaphthyl derivatives by intramolecular biaryl coupling, as shown in equation (9).¹¹³ Diastereoselectivities of up to 70% have been obtained with a binaphthol chiral auxiliary in the linking chain. Intermolecular cross-coupling is best accomplished with preformed arylcopper compounds.¹¹⁴



1.5.2.4.5 Allylic, benzylic and propargylic electrophiles

Allylic, benzylic and propargylic electrophiles are more reactive than the corresponding saturated compounds toward nucleophilic attack, and organocopper reagents of virtually every description, both catalytic and stoichiometric, have been employed successfully in reactions with these activated substrates. In addition to the typical halide and sulfonate leaving groups, ordinarily unreactive ether, ester, sulfone,¹¹⁵ phosphonate,¹¹⁶ carbonate¹¹⁷ and carbamate¹¹⁸ moieties can all be displaced from an allylic or propargylic position. Apart from their obvious synthetic potential, reactions of allylic and propargylic electrophiles are also of considerable mechanistic interest, because of the intriguing regio- and stereochemical questions posed by these substrates. It is primarily these issues that will be addressed in the following section.

Primary allylic halides exhibit a preference for direct substitution (α -attack) in reactions with a variety of organocopper reagents, including lithium cuprates, copper(I) acetylides and catalytic organocopper compounds. Regioselectivity is often quite high, especially with γ , γ -disubstituted substrates, and alkenic stereochemistry is usually retained. For example, in a recently reported synthesis of (15S)-HETE (13) sequential substitution at both primary positions of (Z)-1,4-dichlorobutene was accomplished without loss of the (Z)-alkenic geometry.¹¹⁹ Palladium-catalyzed Stephens-Castro coupling (see Section 1.5.2.4.2) was another key step in the synthesis.



(13)

Secondary allylic halides, on the other hand, exhibit regioselective S_N2' reactivity (γ -attack), particularly when the γ -position is unsubstituted. Thus, although sometimes highly regioselective, reactions with these reagents are not regiospecific.¹²⁰ Regioselectivity is also dependent upon the substitution pattern of the substrate in reactions of allylic esters with lithium cuprates and catalytic organocopper compounds. Again, a preference for nucleophilic attack at the less hindered position is evidenced, and regioisomeric substrates give rise to the same product mixtures.¹²¹ Reactions of acyclic substrates that proceed with allylic rearrangement (γ -attack) often display moderate to good (*E*)-selectivity in formation of the new double bond.

Regioselectivity is less predictable when both termini of the allylic system are comparably substituted. Results can be dependent on the leaving group; oxygen leaving groups appear to favor α -attack more than halides, and chloride is more S_N2 selective than bromide.¹²² Very high S_N2' selectivity has been reported for reactions of some cyclic allylic esters with lithium cuprate reagents,¹²³ but exclusive α -substitution has been observed when γ -attack was disfavored by steric crowding at the δ -position.¹²⁴ Regioselective (*ca.* 9:1) α -attack has also been reported for reactions of (*Z*)-4-hexenolide with both lithium and magnesium cuprates.¹²⁵ As part of research efforts directed toward the total synthesis of ginkgolide B, Corey and Gavai¹²⁶ discovered that highly selective (\geq 98:2) S_N2' displacement of cyclopentenyl pivalate (14) could be effected with a Grignard reagent and 3 mol % of CuCN (equation 10). By contrast, the stoichiometric higher-order lithium cyanocuprate was much less regioselective and favored direct substitution.

A number of organocopper reagents reportedly exhibit $S_N 2'$ selectivity even when steric factors would appear to favor α -attack. Magnesium cuprates, in general, show greater propensity to undergo reaction



with allylic rearrangement than do their lithium counterparts.¹²⁷ The complex organocopper species RCu·MgBrX·LiBr displayed high ($\geq 98:2$) S_N2' selectivity in reactions with 1,4-diacetoxy-2-butene,¹²⁸ but only with R = alkyl; reactions of the phenyl reagent were low yielding and S_N2 selective. Curran has reported¹²⁹ almost exclusive S_N2' stereospecificity, regardless of substitution pattern, in reactions of the bicyclic lactones (15) with MeMgBr-CuBrMe₂S (1:1) (equation 11). One full equivalent of CuBr·SMe₂ was required for high regiospecificity, and CuBr·SMe₂ produced a more selective reagent than did CuI or CuCN. Reactions of the same substrates with lithium cuprates, on the other hand, were controlled by the substrate substitution pattern, but were highly regioselective in favorable cases. This procedure has found recent application in syntheses of the triquinane sesquiterpenes, (\pm)-hypnophilin (16) and (\pm)-coriolin (17).¹³⁰



Copper-catalyzed reactions of organozinc reagents exhibit high S_N2' regiospecificity in reactions with allylic halides, even when steric and/or electronic factors would appear to favor S_N2 attack.¹³¹ The regiospecificity of these copper-catalyzed reactions stands in contrast to nickel- or palladium-catalyzed reactions of organozinc reagents, in which attack occurs regioselectivity at the least hindered position. Very recently, Nakamura¹³² has reported virtually complete diastereofacial selectivity in S_N2' reactions of organocopper reagents with 4-alkoxyallylic chlorides (equation 12). Whereas lithium dialkylcuprates reacted with (18) predominantly by an S_N2 pathway, Bu₂CuZnCl,¹³³ Bu₂CuTi(OPrⁱ)₃, BuCu·BF₃, and Bu₂Zn/cat. CuBr·SMe₂ all provided the γ -substituted products with regioselectivities of $\geq 98:2$ and *anti*-diastereoselectivities of $\geq 99:1$. Reactions of other substrates and reagents were also highly *anti-S*_N2' selective. This procedure promises to find widespread application for the stereoselective construction of acyclic quaternary centers.



Suzuki reported¹³⁴ that copper borates (R₃MeB⁻Cu⁺) exhibit predominant S_N2' reactivity toward allylic halides. A similar 'ate' complex (RF₃B⁻Cu⁺) is also proposed to be the reactive intermediate in the RCu·BF₃ system developed about the same time by Yamamoto and coworkers.³⁶ This latter reagent exhibits remarkable S_N2' regiospecificity in reactions with a variety of allylic substrates, including halides,

Alkylation of Carbon

esters and alcohols. Lithium cuprates¹³⁵ and higher-order cyanocuprates¹³⁶ have also been used in conjunction with BF₃·OEt₂, again with S_N2' regiospecificity. The R₂CuCNLi₂·BF₃ reagent has proven especially useful for S_N2' reactions of γ -oxygenated α,β -unsaturated esters, substrates which are highly susceptible to reduction side reactions. It should be noted, however, that a seemingly similar reagent, RCu-(AlCl)_n, effects highly selective S_N2 substitution ($\alpha:\gamma = 97:3$) of γ -acetoxy- α,β -unsaturated esters.¹³⁷

A mechanistic proposal that accommodates these diverse experimental results is presented in Scheme 5. The result of extensive research efforts in Goering's laboratory, 121c,138 it postulates that coordination of the organocopper reagent to the alkenic π -system is followed by formation of a σ -complex with attachment of copper at the γ -position. Immediate reductive elimination then directly affords the S_N2' product, whereas prior isomerization to a π -complex results in the loss of regiochemical information and leads to the possible formation of both α - and γ -substituted products. Investigations with isotopically labeled, symmetrical substrates confirm that, as required by the proposed mechanism, *all* organocopper reagents exhibit some degree of S_N2' regiospecificity in completely unbiased systems. In agreement with the experimental observations discussed earlier, this S_N2' specificity is barely detectable in reactions of lithium dialkylcuprates, a result attributable to rapid σ - π isomerization. By contrast, isomerization is minimized in reactions of lower-order cyanocuprates RCuCNLi or RCu-BF₃, and virtually exclusive S_N2' substitution obtains. Bäckvall¹³⁹ has invoked a similar mechanistic rationale to explain his finding that regioselectivity in copper-catalyzed reactions of allylic acetates with alkyl Grignard reagents is strongly dependent upon the rate of addition of the organometallic reagent to the reaction mixture.



Scheme 5

The stereochemical outcome of the substitution of allylic substrates by organocopper reagents can also be explained by the mechanism depicted in Scheme $5.^{138b}$ Initial coordination of the organocopper reagent is expected to occur on the less hindered face of the allylic system. In most substrates, this will be the face opposite the leaving group, thus accounting for the predominant *anti* substitution observed in all of the reactions (both $S_N 2$ and $S_N 2'$) discussed thus far. It should be noted, however, that *syn* substitution may prevail when *anti* attack is strongly disfavored by steric factors.^{138b,140}

By contrast, reactions of allylic N-phenylcarbamates with lithium cuprates proceed with very high syn-S_N2' selectivity, regardless of steric considerations.^{118,141} Goering proposes that the reversal in stereoselectivity and the enhanced regiospecificity observed in reactions of carbamate substrates are both consequences of a modified reaction mechanism, in which initial complexation of the copper reagent to nitrogen is followed by intramolecular, syn-facial delivery of copper to the γ -position of the allylic system.¹⁴² A similar mechanism is presumably operative in the substitution of allylic alcohols by RCu-BF₃, a reaction that also proceeds with syn-S_N2' specificity. On the other hand, selective anti-S_N2' substitution of allylic alcohols can be effected by means of the procedure developed by Murahashi,^{143,144} which apparently involves addition of an *in situ* generated heterocuprate to a phosphonium ion intermediate.

Calo has reported¹⁴⁵ very high S_N2' regiospecificity in reactions of benzothiazolyl ethers with RMgX– CuBr. Again, the enhanced regiospecificity is attributed to coordination of the reagent to the ether moiety, and the isolation of coordination complexes lends credence to the proposal. Regioselectivity is completely reversed if the substrate is pretreated with copper(I) bromide prior to addition of the Grignard reagent, however. In that case, the Grignard reagent apparently attacks regioselectively at the least hindered position of a presumed copper(I) halide–substrate complex. The stereospecificity of the reaction has not been investigated. Ammonium salts are apparently also capable of intramolecular participation, as indicated by the $syn-S_N2'$ reaction of (19) with lithium cuprates (equation 13).¹⁶ However, reaction of the regioisometric substrate occurs in an *anti-S_N2* fashion, since in that case the $syn-S_N2'$ pathway is disfavored by steric hindrance of the adjacent silyloxy moiety.



Much of the foregoing discussion regarding allylic substrates applies also to propargylic electrophiles, which exhibit an even greater propensity to undergo reaction with organocopper reagents by an S_N2' mechanism. The preparation of allenes from propargylic substrates by this method has been reviewed elsewhere,¹⁴⁶ and will be discussed only briefly here. Among numerous synthetic applications, the reaction plays a central role in Okamura's synthetic approach to vitamin D¹⁴⁷ and vitamin A¹⁴⁸ analogs, and it has also been utilized for the synthesis of oxabetweenallenes.¹⁴⁹ Propargyl esters and sulfonates are the most commonly employed substrates, but halogen, amide, carbonate, phosphonate, ether and epoxide leaving groups have also been used. Direct conversion of propargyl alcohols to the corresponding allenes can be accomplished by means of the Murahashi procedure.¹⁵⁰ α -Allenyl phosphates and imides also undergo S_N2' substitution by organocopper reagents, in this case providing dienyl products.¹⁸³

Regioselectivity is subject to steric control¹⁵² when dialkylcuprates are employed as the nucleophilic reagents, but the complex organocopper species RCu·MgBrX·LiBr affords the allenic product exclusively, regardless of substrate structure.¹⁵³ In addition, monoorganocopper compounds are less susceptible than cuprate reagents to reduction side reactions. Higher-order cyanocuprates appear to be more effective than Gilman reagents for transfer of phenyl or of secondary or tertiary alkyl groups, although the reverse is true for *n*-alkyl reagents. Anti stereospecificity is observed in all of these reactions,¹⁵⁴ allowing the preparation of enantiomerically enriched allenes from optically active propargylic substrates. In the presence of donor ligands, racemization of the product allene by the cuprate reagent is minimized, and the stereospecificity of the reaction is increased.

In contrast to reactions of all other organocopper reagents, reactions of copper(I) acetylides with propargylic substrates result in regiospecific α -substitution. This procedure for the preparation of 1,4-diynes has proven useful in pheromone synthesis.¹⁵⁵

1.5.2.4.6 Epoxides

Long valued by organic chemists as versatile synthetic intermediates, epoxides have experienced a resurgence in popularity over the past decade as a result of newly developed and highly efficient procedures for their preparation in homochiral form.¹⁵⁶ Owing to the polarity and strain of the three-membered ring, epoxides undergo reaction with a wide variety of reagents, the reactions proceeding in a stereospecific and often highly regioselective fashion.¹⁵⁷ Ring opening by carbon nucleophiles is arguably among the most synthetically useful of epoxide transformations, and organocopper reagents have proven to be especially well suited for this purpose. Not only are they more reactive toward epoxides than are organolithium or organomagnesium compounds, but the side reactions (*e.g.* rearrangement, elimination) that are promoted by the basicity or Lewis acidity of these other organometallics are not usually observed in reactions of organocopper reagents.

The reaction between epoxides and Gilman reagents was first reported in 1970,¹⁵⁸ and in the 20 years since, both lithium and magnesium homocuprates have been widely employed to effect ring opening of reactive mono- and di-substituted epoxide substrates. In reactions of monosubstituted epoxides, comparable results are also obtained with the heterocuprates, RCuCNLi,¹⁵⁹ but yields drop dramatically with less reactive di- and tri-substituted epoxides. In light of their enhanced reactivity toward alkyl halides (see Section 1.5.2.4.1), it is perhaps not surprising that the higher-order cyanocuprates, R₂CuCNLi₂, are also significantly more reactive than lower-order cuprates toward epoxide substrates.^{46,160} Less than 1.5 equiv. of the cuprate reagent are usually sufficient, in contrast to the 2–5 equiv. commonly required in reactions of lithium homocuprates, and even poorly reactive cyclic and trisubstituted epoxides are successfully substituted under mild conditions. Tetrasubstituted epoxides afford only elimination by-products, however.

Alkylation of Carbon

Epoxide opening has also been effectively accomplished with catalytic organocopper compounds.^{161,162} Owing to the easy accessibility of many Grignard reagents, this is often the most convenient procedure by which to effect the desired transformation. In some cases, results are superior to those obtained with stoichiometric reagents.¹⁶³ However, epoxide opening by the halide ion of the Grignard reagent or of the copper halide catalyst is a common side reaction, and at times the halohydrin may become the sole product of the reaction.^{163a,164} In fact, treatment with Li₂CuCl₄ constitutes a preparatively useful procedure for the synthesis of chlorohydrins from the corresponding epoxides.¹⁶⁵ It should be noted that the lithium halide salts present in most preparations of monoorganocopper compounds and homocuprate reagents may also participate in the epoxide opening reaction to produce halohydrin by-products.¹⁶⁶ This side reaction is virtually eliminated with higher-order cyanocuprate reagents, which are both more reactive and can easily be prepared in halide-free form.

Unreactive epoxides^{166,167} and/or reagents¹⁶⁸ undergo successful reaction in the presence of BF₃·OEt₂, which has proven to be uniquely effective among the Lewis acids examined in activating epoxides toward nucleophilic attack by organocuprate reagents. Lithium and magnesium homocuprates and higherorder cyanocuprates have all been used in combination with BF₃·OEt₂ to good effect. The closely related reaction of aziridines with organocopper reagents is also facilitated by BF₃·OEt₂.¹⁶⁹

In general, the rate of epoxide opening by organocuprate reagents is more rapid in diethyl ether than in the more highly coordinating solvent, THF, which interferes with the coordination of the epoxide to Lewis acidic components in the reaction mixture. Also, elimination to form an allylic alcohol is a more problematic side reaction in THF than in ether.¹⁷⁰ On the other hand, organocopper compounds are often only poorly soluble in ether, and mixed solvent systems have proven advantageous in some cases.¹⁷¹

Chemoselectivity in reactions of functionalized epoxides is an issue of growing importance in light of the increasingly widespread use of epoxy alcohols and related derivatives as synthetic building blocks. Epoxy halides undergo reaction with organocopper reagents primarily by way of epoxide opening when both groups are comparably substituted.¹⁷² In contrast, 'hard' nucleophiles, such as organolithium compounds, react with these substrates exclusively at the halo-substituted position.^{172b,173} Steric factors have an overriding influence on chemoselectivity, however, and reactions of unsymmetrically substituted epoxy halides with organocopper reagents occur predominantly at the least hindered position. Thus, a primary iodide can be displaced in the presence of an internal, disubstituted epoxide. Although no substitution at the epoxide center is detected, reaction conditions must be carefully controlled in order to prevent reductive cleavage of the iodide, with concomitant epoxide opening and formation of an allylic alcohol.¹⁷⁴ The regio- and chemo-selective displacement of a primary allylic chloride in the presence of a trisubstituted epoxide has also been reported.¹⁷⁵

Chemoselective reaction of epoxy sulfonates represents a more significant challenge, since the two electrophilic moieties are reported to be of nearly equivalent reactivity toward organocuprate reagents.¹⁷⁶ As expected on the basis of the epoxy halide results just discussed, the displacement of a primary tosylate can be smoothly accomplished in the presence of an adjacent disubstituted epoxide,^{174c,177} and this reaction has been utilized in syntheses of disparlure and of other insect pheromones.¹⁷⁷ On the other hand, a recently reported¹⁷⁸ reaction of a cyclic glucopyranose-derived epoxide with either Me₂CuLi or MeMgCl–CuBr afforded in high yield the desired ring-opened product, leaving intact an exocyclic primary mesylate.

Symmetrically substituted vicinal epoxy mesylates undergo chemoselective epoxide opening by $R_2CuCNLi_2-BF_3\cdot OEt_2$ to provide the corresponding hydroxy sulfonates in good yield.¹⁷⁹ In the absence of BF₃·OEt₂, ring closure occurs *in situ*, and doubly substituted compounds are produced. Similarly, the reaction of homochiral glycidyl tosylate with Ph₂CuCNLi₂ proceeds *via* a hydroxy sulfonate intermediate to afford the substituted epoxide, along with varying amounts of the product of double addition (Scheme 6).¹⁸⁰ In reactions of the same substrate with RMgBr–CuI, ring closure does not occur, and the hydroxy tosylates can be isolated in moderate to good yield. The very high chemoselectivity exhibited by this sterically unbiased epoxy tosylate substrate may be attributable in part to a deactivating effect of the adjacent epoxide moiety on the reactivity of the sulfonate toward S_N2 attack.

Like chemoselectivity, regioselectivity is also governed largely by steric factors, with attack occurring preferentially at the least hindered epoxide center. This generalization holds true even for reactions performed in the presence of BF₃·OEt₂. However, stereoelectronic factors predominate in reactions of cyclic epoxides, which generally proceed predictably in a *trans*-diaxial fashion.

Epoxide substitutents that are capable of stabilizing a positive charge may direct nucleophilic attack to the α -position. For example, styrene oxide exhibits a greater proclivity than other terminal epoxides for substitution at the internal position;^{166,181} regioselectivity is dependent upon the nature of the organocopper reagent. Ring opening of epoxysilanes occurs regiospecifically at the α -position, and this reaction forms the basis for the Hudrlik version of the Peterson alkene synthesis.¹⁸² The use of vinylcuprate



reagents allows the stereospecific synthesis of conjugated dienes (Scheme 7),^{167b} and homochiral allylic alcohols have been prepared by combining kinetic resolution of a silyl-substituted secondary allylic alcohol with Hudrlik–Peterson reaction of the resulting epoxysilane.¹⁸³ It should be noted, however, that epoxysilanes in which silicon bears at least one isopropoxy substituent undergo deoxygenation rather than substitution upon treatment with PrⁱMgBr–CuCN.¹⁸⁴



Scheme 7

2,3-Epoxy alcohols that are branched at C-4 (γ -position) undergo exclusive substitution at C-2,¹⁸⁵ but in the absence of steric constraints reactions are usually nonregioselective. Significant regioselectivity has been observed in reactions of some sterically unbiased γ -oxygenated epoxy alcohols, however. In one case, it was reported that RMgBr–CuI provided a higher ratio of the C(2)-substituted product than did R₂CuLi.¹⁷¹ Recently, Chong has reported¹⁸⁶ that C-2 selectivity is improved when reactions of unbranched epoxy alcohols are conducted in coordinating solvents, while the addition of BF₃·OEt₂ results in enhanced C-3 selectivity. Epoxy ethers exhibit a slight preference for reaction at C-3,¹⁸⁷ but, again, regioselectivity is subject to numerous contributing factors and is not easily predictable. Substitution at C-1 of a 2,3-epoxy alcohol can be accomplished under Payne rearrangement conditions (equation 14).¹⁸⁸ Lithium dialkylcuprates exhibit poor regioselectivity, but less reactive reagents (*e.g.* RCu, RCuCNLi) attack at C-1 with very high selectivity.



Although unsubstituted glycidic esters undergo reaction with organocuprate reagents exclusively at the terminal (C-3) position,¹⁸⁹ predominant C-2 attack is observed in reactions of substituted glycidic esters.^{164,190} Moderate to high C-2 selectivity is also observed in reactions of (*E*)-glycidic acids, but sterically unbiased (*Z*)-glycidic acids react with unexpectedly high C-3 selectivity.¹⁹¹

Vinyl epoxides, like allylic halides and carboxylates, have available both S_N2 and S_N2' reaction pathways. For the most part, regio- and stereo-selectivity follow patterns similar to those discussed in Section

1.5.2.4.5. Although first reported in 1970,¹⁹² the S_N2' addition of organocopper reagents to vinyl epoxides was not further investigated until nearly a decade later. In the past 10 years, research efforts have concentrated primarily on rigid cyclic systems, which exhibit a pronounced preference for *anti-S*_N2' reactivity. The stereospecificity of the reaction has been exploited to introduce steroidal side chains with the proper exocyclic stereochemistry.¹⁹³

Reactions of cycloalkadiene monoepoxides have received considerable attention. In general, cyanocuprates have provided better S_N2' selectivity than lithium homocuprates, and the alternative S_N2 reaction is more competitive with vinyl- or phenyl-cuprates than with alkylcuprate reagents.¹⁹⁴ Reactions of cyclopentadiene monoepoxides with cyanocuprates have found application in prostaglandin synthesis.^{194a} Effective electrophilic α' -alkylation of cyclic enones can be accomplished by S_N2' cuprate addition to the corresponding epoxy enolate,¹⁹⁵ enol phosphate¹⁹⁶ or silyl enol ether.¹⁹⁴

Macrocyclic α -methylenecycloalkylidene epoxides undergo exclusive S_N2' reaction with RMgBr-CuI.¹⁹⁷ By contrast, the corresponding allylic halides and phosphonates provide mixtures of S_N2 and S_N2' products. With ring sizes of 10–14, these substrates exhibit a strong preference for syn- S_N2' reaction from an *exo-trans* conformation. Smaller ring substrates are, of course, constrained to a *cis* conformation. α -Alkenylcyclohexylidene epoxides exhibit a preference for *anti-S_N2'* reaction with lithium homocuprates, but regio- and stereo-selectivity are both subject to steric control.¹⁹⁸

Recently, reactions of acyclic vinyl epoxides have attracted increased attention. In many cases, regioselectivity is strongly dependent upon substrate structure, with (*E*)- and (*Z*)-isomers providing different results.¹⁹⁹ Similarly, stereoselectivity is the result of a complex interplay of steric and conformational effects, but *anti* addition to an *s*-trans conformation appears to be preferred. The regio- and stereo-selective α -alkylation of γ , δ -epoxy- α , β -enoates has been reported recently.²⁰⁰ Again, S_N2' selectivity was greater with cyanocuprates than with the 'harder' homocuprates, but in no case was conjugate addition detected.

1.5.2.4.7 Acyl halides

The reactivity of organic electrophiles toward organocopper reagents follows the general order RCOCl \approx RCHO > tosylate, epoxide > RI > ketone > ester > nitrile. Therefore, the addition of organocuprates to acyl halides occurs cleanly and rapidly at temperatures low enough that the product ketone is essentially inert. Because most other functional groups are also stable to the reaction conditions, halo-, cyano- and carbonyl-substituted ketones can all be prepared from the corresponding acyl chlorides.²⁰¹ Acyl bromides and fluorides are also suitable substrates. Phenylthiocuprates have proven to be more effective than homocuprates in these reactions; fewer equivalents of the reagent are required, and secondary and tertiary alkyl groups can be successfully transferred.^{39a,d} More recently, Rieke has achieved the preparation of functionalized ketones by the coupling of acyl halides with functionalized organocopper reagents.²⁰²

The preparation of α , β -unsaturated ketones by direct acylation of vinylcopper reagents has proven more problematic, since lithium cuprates do add to the product enones.²⁰³ Better results are obtained with the less reactive monovinyl copper compounds in the presence of a palladium catalyst. Alkynic ketones have been prepared by a variation of the Stephens–Castro coupling.²⁰⁴

1.5.2.4.8 Acetals

Widely employed as protecting groups for carbonyl functionalities, acetals do not ordinarily react with organometallic reagents such as organo-lithium, -magnesium or -copper compounds, or lithium cuprates. However, in the presence of a Lewis acid, the carbon–oxygen bond is activated toward nucleophilic attack, and the cleavage of one C—O bond can be achieved. For example, Grignard reagents react with acetals in the presence of TiCl₄ to provide the monosubstituted ethers in high yields.²⁰⁵ The reaction of allylic acetals with alkyl Grignards occurs regiospecifically at the α -position, and diastereomeric ratios of up to 96:4 have been observed in reactions of chiral acetals.²⁰⁶ Cuprate reagents are not stable to TiCl₄, however, and provide only poor yields of the substitution products.

As mentioned in earlier sections (1.5.2.4.5 and 1.5.2.4.6), BF₃·OEt₂ has proven to be uniquely wellsuited for use in combination with organocopper reagents.³⁶ In fact, in the presence of BF₃·OEt₂, acetals react smoothly in ether solution with either RCu or R₂CuLi, the latter being more reactive.²⁰⁷ No reaction occurs in THF, which competes with the substrate for Lewis acid complexation. Mixtures of cyclic and ring-opened substitution products are obtained upon treatment of tetrahydropyranyl ethers with RCu– BF₃·OEt₂, but R₂CuLi–BF₃·OEt₂ affords only the ring-opened products. Similarly, exclusive ring cleavage is also observed in reactions of THP ethers or glycopyranosides with Me₂BBr and R₂CuLi.²⁰⁸

Research efforts have focused primarily on reactions of chiral, nonracemic acetals,²⁰⁵ compounds that have found increasingly widespread application in asymmetric synthesis. Acetals derived from C₂-symmetrical diols provide the best stereocontrol, and the major product is in every case that produced by *anti* nucleophilic displacement of the oxygen adjacent to the axial ring substituent (equation 15). This result has been rationalized on the basis of both steric and electronic considerations, arguments that are supported by the finding that the *meso* (diequatorial) compounds are much less reactive. Diastereomeric ratios are generally highest for six-membered cyclic acetals, which are possessed of greater conformational rigidity than the corresponding five- or seven-membered rings. Reactions of propargylic acetals proceed in a net *anti-S*_N2' fashion, again with cleavage of the C—O bond adjacent to the axial ring substituent. Allylic acetals also undergo regiospecific S_N2' substitution by aryl- and vinyl-copper compounds, but al-kylcopper reagents afford mixtures of S_N2 and S_N2' products.

$$R \xrightarrow[H]{O} O \xrightarrow[H]{O} BF_{3} \circ OEt_{2} \xrightarrow[H]{O} BF_{3} \circ OEt_{2} \xrightarrow[H]{O} H \xrightarrow[H]{O} H$$
(15)

Recently, it was reported that acetals derived from homochiral β -hydroxybutyric acid react with R₃Cu₂Li in the absence of BF₃·OEt₂.²⁰⁹ These reactions proceed with highly diastereoselective *anti* displacement of the carboxylate, and treatment of the product with base liberates a secondary alcohol of high enantiomeric purity.

1.5.2.4.9 β-Lactones

While most organolithium and Grignard reagents attack the carbonyl group of lactone substrates, regiospecific carboxylate displacement can be effected with both stoichiometric and catalytic organocopper reagents.²¹⁰ The reaction provides a convenient method for three-carbon homologation that has proven useful in natural product synthesis.²¹¹ The regiospecific S_N2' reaction of β -ethynyl- β -propiolactone has also been accomplished.²¹² In general, these reactions are best performed in THF, and magnesium cuprates often provide better results than their lithium counterparts. Phenyl, vinyl and allyl transfers are more difficult to achieve than alkyl transfer, and stoichiometric magnesium cuprates have been most successful for these transformations.

Very recently, Vederas has reported²¹³ the extension of this methodology to serine β -lactone substrates (equation 16), providing an important addition to the growing number of procedures for amino acid synthesis. These substrates raise the specter of potentially worrying elimination and racemization side reactions in addition to the possibility of competitive 1,2-addition. In fact, substantial racemization was observed when diprotected amino- β -lactones were employed, but the use of monoprotected substrates (which exist in deprotonated form under the reaction conditions) minimized abstraction of the α -hydrogen. By the same token, racemization following substitution is disfavored by the adjacent carboxylate anion (*cf.* reactions of β -haloserine esters, Section 1.5.2.4.1). Although lithium homocuprates and higher order cyanocuprates provided reasonable yields of the desired products, catalytic magnesium cuprate reagents proved most successful.

$$R^{1}ZN \xrightarrow{O} R^{2}CU' \qquad R^{1}ZN \xrightarrow{R^{2}} CO_{2}H$$
(16)

1.5.3 NICKEL AND PALLADIUM CATALYSTS

The modification of organo-lithium and -magnesium reagents by the addition of catalytic amounts of copper salts was discussed in Section 1.5.2. Other transition metals also have been found to promote the cross-coupling reactions of organometallic reagents with carbon electrophiles, and dramatic advances in this technology over the past decade have served to extend significantly the synthetic utility of cross-coupling reactions for the selective construction of carbon skeletons. Of the transition metals examined,

nickel and palladium have produced the most successful results and have been the most widely used; the following section is limited to a discussion of reactions catalyzed by these two metals.

Given that the catalytic cycles proposed to be operative for reactions employing nickel and palladium catalysts are essentially identical, it is not surprising that the two may be used interchangeably for many reactions. Significant differences in reactivity and selectivity do exist, however, and an effort will be made to point out those differences in the discussion that follows. While meant to cover the important contributions in this area, this account is not intended to be exhaustive. The reader is referred to several excellent reviews²¹⁴ and to other chapters in this volume for additional information on the subject.

1.5.3.1 Organonickel Catalysis

More than 15 years ago, Corriu²¹⁵ and Kumada²¹⁶ independently discovered that organonickel complexes greatly facilitated the reactions of Grignard reagents with alkenyl and aryl halides. Since that time, the reaction has been intensively investigated²¹⁷ and has proven to be quite general for the coupling of $C(sp^2)$ halides with virtually any Grignard reagent. The proposed mechanism⁵³ of the reaction involves initial reaction of a Ni^{II} complex with the Grignard reagent, oxidative addition of the substrate to the resulting Ni⁰ species, and transmetalation of the Grignard reagent. Reductive elimination then liberates the coupled product and regenerates the active catalyst. Reactions of alkenyl halides exhibit predominant retention of alkene geometry, while reactions of alkenyl Grignard reagents are complicated by competing (Z) to (E) isomerization.

In general, nickel complexes bearing phosphine ligands have been the most successful catalysts for these reactions, with bidentate phosphine ligands providing the greatest catalytic activity. The preferred catalyst for a particular reaction is dependent upon the nature of the Grignard reagent and of the halide substrate, with NiCl₂(dppp) (dppp = 1,3-bis(diphenylphosphino)propane) being most effective for simple alkyl and aryl Grignard reagents, NiCl₂(dmpe) (dmpe = 1,2-bis(diphenylphosphino)ethane) for vinylic and allylic Grignard reagents, and NiCl₂(PPh₃)₂ for sterically hindered aryl Grignard reagents and halides. In many cases, the corresponding palladium complexes are also suitable catalysts.

The choice of catalyst is particularly important for reactions of secondary alkyl Grignard reagents. Although primary Grignard reagents couple without rearrangement of the alkyl group, in reactions of secondary Grignard reagents that bear β -hydrogens, β -elimination competes with reductive elimination, resulting in the formation of both reduced and isomerized by-products.²¹⁸ β -Elimination is especially troublesome in reactions with hindered alkenyl halides, since the rate of product-forming reductive elimination is retarded. The catalysts NiCl₂(dppf)²¹⁹ and PdCl₂(dppf)²²⁰ (dppf = 1,1'-bis(diphenylphosphino)ferrocene) minimize these side reactions by accelerating the rate of desired reductive elimination relative to undesired β -elimination. By contrast, alkyl halides that possess β -hydrogens are not usually suitable substrates, regardless of catalyst, because the transmetalation step is slow in comparison to β elimination.

Stereochemistry is an additional factor involved in reactions of secondary Grignard reagents in which the metal is attached to a chiral carbon atom, and this has proven to be a particularly fertile area of research for Hayashi, Kumada and their coworkers.²²¹ The first examples of asymmetric induction in reactions of secondary Grignard reagents employing a homochiral nickel catalyst were reported²²² shortly after the discovery of the original reaction. These early results were obtained using NiCl₂(DIOP) (*cf.* 20) as the chiral catalyst, and the observed enantioselectivities were quite low ($\leq 17\%$). In the years following these initial reports, numerous ligands have been examined, with ferrocenylphosphines (*e.g.* 21, 22) and β -(dimethylamino)alkylphosphines derived from amino acids (23) being the most extensively investigated.

Of the ferrocenylphosphine ligands, (R,S)-PPFA (21) proved to be the most successful, with NiCl₂(PPFA) and PdCl₂(PPFA) producing comparable results. Organozinc reagents were somewhat more selective than the corresponding Grignard reagents in reactions employing PdCl₂(PPFA), but they



(22) (S,R)-PPFOMe



did not react with nickel catalysts. Higher enantioselectivities (up to 94%) were achieved when some of the amino acid derived ligands were used, although with these ligands palladium was less efficient than nickel. Ligands derived from (S)-amino acids generally give rise to the (S)-product, but it has been reported²²³ that this trend is reversed in the presence of ZnX_2 (X = Br, I).

The use of alternative classes of ligands for these reactions has been explored, but with only moderate success.²²⁴ In the most promising example of this work,^{224a,b} a sulfide-containing side chain was appended to the familiar dimethylaminophosphine ligand. The sulfide moiety offers a third site for coordination to the metal, and it was hoped that such intramolecular participation would accelerate the rate of reductive elimination, minimizing the loss of enantiomeric purity resulting from competitive β -elimination. In fact, the enantioselectivity of the homomethphos ligand (24) is comparable to that of *t*-leuphos (23; R = Bu^t).

The asymmetric Grignard cross-coupling reaction has been utilized in syntheses of (R)-(-)-curcumene²²⁵ and of optically active antiinflammatory agents.²²⁶ A particularly successful application of the methodology has been for the preparation of optically active allylsilanes,²²⁷ in which enantioselectivities of up to 95% have been achieved. Interestingly, PdCl₂(PPFA) was the most effective catalyst for silylsubstituted Grignard reagents, with nickel catalysts providing lower yields and selectivities. Most recently, substituted binaphthyls²²⁸ and ternaphthyls²²⁹ have been prepared in high enantiomeric purity by asymmetric biaryl cross-coupling with NiCl₂(PPFOMe) as the catalyst (*e.g.* equation 17). As in the original work with secondary alkyl Grignards, nickel catalysts were more successful than palladium catalysts, but, in this case, complexes containing bidentate ligands (*e.g.* diphosphines and aminophosphines) were unreactive due to steric hindrance.



Asymmetric induction is also observed in nickel-catalyzed reactions of Grignard reagents with allylic alcohols and ethers.²³⁰ Examples of high enantioselectivity have been reported;^{230a} however, the enantioselectivity of the reaction is strongly and unpredictably dependent upon the allylic substrate, the Grignard reagent and the catalyst. In most cases, the best results have been obtained with NiCl₂(chiraphos) (*cf.* **25**) as the homochiral catalyst. Palladium complexes are not suitable catalysts for this reaction. In contrast to transition metal catalyzed reactions of allylic ethers and esters with 'soft', stabilized carbanions, in which the stereochemistry of the starting material is retained in the product, reactions with Grignard reagents proceed with net inversion of configuration.²³¹ There is a slight tendency favoring reaction at the more highly substituted allylic position of unsymmetrical electrophiles, but mixtures of regioisomers are usually obtained.

Vinyl and aryl ethers also react with Grignard reagents under nickel catalysis. Substitution reactions of aryl tetrazolyl ethers are reported²³² to be particularly facile, an effect that may be attributable to the ability of the tetrazolyl moiety to coordinate to magnesium. Reactions of dihydrofuran and dihydropyran substrates with Grignard reagents afford homoallylic and bishomoallylic alcohols, respectively (equation 18).²³³ Retention of alkene geometry is predominantly observed, and by carefully controlling the work-up conditions, stereoselectivities of \geq 98:2 can be achieved.²³⁴ By contrast, reactions of the same substrates with cuprate reagents proceed with inversion of configuration (see Section 1.5.2.4.2, equation 8). As mentioned earlier, β -elimination with resulting substrate reduction is a competing process when Grignard reagents with β -hydrogens are employed. In reactions with dihydrofuran substrates, NiCl₂(dppe) provided the highest ratios of coupled products.

The nickel-catalyzed substitution of vinyl sulfides by Grignard reagents forms the basis of a procedure for the stereoselective synthesis of alkenes by sequential cross-coupling reactions developed by Naso and



coworkers (equation 19).²³⁵ The first step of the sequence is chemospecific, with only the bromide being displaced. Better stereoselectivity is obtained in the initial reaction of (Z)-1-bromo-2-phenylthioethylene when PdCl₂(PPh₃)₂ is employed as the catalyst. Nickel catalysts must be used for the second step, and the particular choice of catalyst can be critical to the stereochemical outcome. In a variation of the method,²³⁶ a dienyl sulfide is stereospecifically prepared by addition of a vinyl cuprate reagent to phenylthioacetylene. Nickel-catalyzed Grignard cross-coupling then affords the desired dienyl product.



Optically active alkenylsulfoximines undergo nickel-catalyzed cross-coupling with organozinc reagents in the presence of an additional magnesium, lithium or zinc salt (equation 20).²³⁷ No reaction occurs in the absence of the cocatalyst, and palladium catalysts are unreactive. The reaction proceeds with 99% retention of alkene geometry, and sulfoximine of \geq 98% enantiomeric purity is liberated. Interestingly, reaction of (26) with Grignard reagents at -20 °C failed to afford the coupled product, but instead resulted in α -metalation.²³⁸ At 0 °C, coupling with a second equivalent of Grignard reagent occurred to provide an α -functionalized alkenyl metal, but as a 1:1 mixture of (*E*)- and (*Z*)-isomers.



1.5.3.2 Organopalladium Catalysis

1.5.3.2.1 Aluminum, zinc and zirconium reagents

While ease of preparation and commercial availability contribute to the popularity of simple Grignard reagents for use in cross-coupling reactions, the scope of the reaction is seriously limited by the poor chemoselectivity of these reagents. In addition, alkenyl Grignard reagents are often isomerized under the reaction conditions, rendering them unsuitable for stereospecific diene synthesis. It is particularly this latter consideration that provided the impetus for an intensive investigation into the use of other organometallics in transition metal catalyzed cross-coupling reactions conducted by Negishi and coworkers.²³⁹ Aluminum and zirconium reagents have received particular attention, since the corresponding alkenyl metals can be prepared in a regio- and stereo-defined fashion utilizing hydrometalation and carbometal-ation procedures.

(E)-Alkenylalanes and (E)-alkenylzirconium derivatives readily undergo nickel-catalyzed cross-coupling with aryl halides to provide the arylalkenes in high yield and with \geq 98% retention of alkene geometry. Ester, nitrile, ether and halide functional groups are compatible with the reaction. Alkenyl-alkenyl cross-coupling also proceeds in moderate to good yield, but homocoupling is a competing side reaction and the stereoisomeric purity of the product is only 95% for (*E*,*E*)-dienes and 90% for (*Z*,*E*)-dienes. Better results are obtained with palladium catalysts (Pd(PPh₃)₄ or PdCl₂(PPh₃)₂), which provide both products in \geq 98% stereoisomeric purity with only trace amounts of homocoupled by-products.

In contrast to β -monosubstituted alkenyl metals, more highly substituted alkenyl metals containing aluminum or zirconium are only poorly reactive. This limitation is overcome by the addition of a catalytic amount of a zinc halide. In general, zinc reagents are far more reactive than the corresponding aluminum or zirconium reagents,²⁴⁰ and their use is particularly advantageous in coupling reactions of aryl, alkynyl and heteroaryl metals. It is additionally noteworthy that, despite the potential problem of β -elimination, cross-coupling of homoallylic and homopropargylic organozinc compounds with alkenyl halides may be successfully accomplished, enabling the selective synthesis of 1,5-dienes and 1,5-enynes.²⁴¹ Palladium-catalyzed coupling reactions of β -, γ -, δ -, ε - and ζ -zinc ketones have also been reported.²⁴² Recently, Negishi has reported²⁴³ the application of palladium-catalyzed cross-coupling reactions of organozinc compounds to the regio- and stereo-specific α -alkenylation and α -arylation of ketones.

1.5.3.2.2 Boron reagents

Like hydroalumination and hydrozirconation, hydroboration of alkynes also provides a convenient and stereospecific route to alkenyl metal reagents. However, initial attempts to achieve palladium-catalyzed cross-coupling of alkenylboranes with alkenyl halides were unsuccessful, due to the poor carbanionic character of these reagents. Later, Suzuki²⁴⁴ discovered that the desired transformation could be effected in the presence of an alkoxide or hydroxide base; weaker bases, such as sodium acetate or triethylamine, were not generally effective. The reaction is suitable for the preparation of (E,E)-, (E,Z)- and (Z,Z)-dienes. Since reactions of alkenylboronates are higher yielding than those of alkenylboranes, the recent availability of (Z)-1-alkenylboronates²⁴⁵ substantially improves the Suzuki method for the preparation of (Z)-alkenes.²⁴⁶ An extension of the methodology to the synthesis of trisubstituted alkenes has also been reported.²⁴⁷

In an approach similar to the Naso alkene synthesis described earlier, Suzuki has reported²⁴⁸ the stereospecific synthesis of alkenes from (E)-(2-bromo-1-alkenyl)dibromoborane by sequential cross-coupling reactions (equation 21). The reaction sequence can be carried out in one pot, without isolation of the intermediate. Transmetalation from boron to palladium occurs only after base has been added.



In addition to alkenyl halides, alkynyl, aryl, allylic and benzylic halides and β -haloenones²⁴⁹ all couple efficiently with alkenylboron reagents in the presence of a palladium catalyst and 2 equiv. of base. The cross-coupling of aryl halides with arylboronic acids, regioselectively prepared by directed *ortho* lithiation methodology, has been investigated by Snieckus and coworkers. Applications to the synthesis of phenanthrols²⁵⁰ and phenanthridines²⁵¹ have been reported most recently. Very recently, Suzuki reported²⁵² the extension of this methodology to include the coupling of alkylboron reagents with vinyl and aryl halides.

It was originally believed that addition of a base to reactions of organoboranes would facilitate the transmetalation step by increasing the carbanionic character of the boron reagent *via* formation of an 'ate' complex. However, the failure of alkenylborates to undergo efficient cross-coupling indicates that the base is not functioning in this manner, and suggests instead the involvement of an alkoxypalladium(II) intermediate, the formation of which is apparently rate limiting. It should be noted, however, that lithium tetraalkylborates do couple successfully.²⁵² Recently, Kishi and coworkers²⁵³ reported that the use of thallium hydroxide had a dramatic rate-enhancing effect on these coupling reactions, allowing application of the methodology to palytoxin synthesis. The high molecular weight substrates involved in that study proved unreactive under the standard reaction conditions.

1.5.3.2.3 Tin reagents

Organotin reagents have proven to be particularly versatile partners in palladium-catalyzed coupling reactions. As impressively demonstrated by Stille and coworkers over the past decade,²⁵⁴ organostannanes of many structural types undergo efficient cross-coupling with an assortment of organic electrophiles, including halides, triflates, acid chlorides and epoxides. Reaction conditions are mild, and a wide

Alkylation of Carbon

variety of functional groups may be present in either component, allowing application of the methodology to the synthesis of sensitive and complex target molecules. Unlike most other organometallics, organotin reagents are easily prepared by a number of different routes, and they are not especially oxygen or moisture sensitive.

Of the electrophiles examined, vinyl triflates have received particular attention.²⁵⁵ These compounds are conveniently prepared from the corresponding ketones, utilizing well-established methodology for regioselective enolate formation.²⁵⁶ Coupling with organostannanes proceeds smoothly in the presence of tetrakis(triphenylphosphine)palladium(0) [Pd(Pd₃P)₄] and a stoichiometric amount of LiCl. No reaction occurs in the absence of LiCl, which apparently allows formation of the catalytically active alkenylchlorobis(triphenylphosphine)palladium(II) complex. Aryl triflates, easily prepared from the corresponding alcohols, are also suitable electrophiles under similar reaction conditions.²⁵⁷ Although successful in THF and other ethereal solvents, the reaction is faster in solvents such as DMF, which can both solubilize LiCl and act as good ligands.²⁵⁸

In reactions of unsymmetrically substituted organotins, unsaturated substituents transfer preferentially over saturated alkyl groups. For synthetic convenience, methyl and butyl groups are usually employed as the nontransferable ligands. In some cases,²⁵⁹ the trimethyltin derivative provides better results due to a decrease in steric crowding, and trimethylorganostannanes offer the additional advantage that the by-products are water soluble. On the other hand, lower toxicity and greater stability are factors favoring the use of tributyltin derivatives.

Few limitations are imposed on the choice of the transferring group. Alkyl, vinyl, allyl and alkynyl compounds all couple effectively with aryl and vinyl triflates. Phenyl and benzyl reagents are unreactive toward vinyl triflates,²⁶⁰ but biaryl coupling is successful. Because reductive elimination occurs more rapidly than β -elimination, organostannanes with β -hydrogens are acceptable coupling partners. However, β -hydrogens attached to an *sp*³-carbon are not usually tolerated in the electrophile, since β -elimination is faster than the rate-determining transmetalation step.²⁶¹ Recent applications of the methodology include syntheses of vitamin D,²⁶² amphimedine²⁶³ and cephalosporin analogs.²⁶⁴

Like the other coupling reactions discussed thus far, palladium-catalyzed coupling reactions of organotin reagents proceed stereospecifically, with the alkene geometries of both vinyltin reagents and vinyl electrophiles being retained in the product. However, the utility of cross-coupling between vinyltins and vinyl triflates for 1,3-diene synthesis is seriously limited by the fact that no method currently exists for the preparation of acyclic vinyl triflates in a stereodefined manner. As mentioned in previous sections, stereospecific 1,3-diene synthesis has been accomplished by palladium-catalyzed coupling of organometallics with stereoisomerically pure vinyl halides, for which synthetic methodology is available. However, initial attempts to achieve the coupling of vinylstannanes with vinyl halides in the presence of Pd(PPh₃)₄ resulted in only modest yields of the desired products.^{239a}

Recently, improved experimental procedures for the cross-coupling of organotin reagents with vinyl iodides have been reported,²⁶⁵ which allow the preparation of (E,E)-, (E,Z)- and (Z,Z)-dienes in high yields and isomeric purities.²⁶⁶ Bis(triphenylphosphine)dichloropalladium [PdCl₂(PPh₃)₂] and di(acetoni-trile)dichloropalladium [PdCl₂(MeCN)₂] proved to be much more reactive than Pd(PPh₃)₄ in these reactions, with PdCl₂(MeCN)₂ in DMF providing the most reactive catalyst. The lower temperatures and shorter reaction times allowed by this system minimized alkene isomerization, which was a problem when less reactive catalysts were used. The coupling of alkynylstannanes with vinyl iodides under the same conditions provided stereoisomerically pure enynes, from which (E,Z)- and (Z,Z)-dienes could be prepared by stereospecific alkyne reduction.^{259b} One limitation of this methodology is the fact that substantial amounts of homocoupled by-products are produced in reactions of silyl-substituted vinyltin reagents, limiting yields of the synthetically useful silyldienes.

While not useful for reactions of vinyl halides, $Pd(PPh_3)_4$ is a suitable catalyst for coupling reactions of organotins with aromatic bromides and iodides; a number of applications to the synthesis of coupled heterocycles have been reported recently.²⁶⁷ Reactions of substituted bromobenzenes with vinyltributylstannane afford the corresponding styrene derivatives in high yield.²⁶⁸ Exclusive displacement of halogen in the presence of a triflate substituent is simply accomplished by performing the reaction in the absence of LiCl, as evidenced in a recently reported synthesis of indole derivatives.²⁶⁹ In the presence of LiCl, the order of electrophilic reactivity is I > Br > OTf when Pd(PPh_3)₄ is employed as the catalyst, but PdCl₂(PPh₃)₂ causes a reversal in this selectivity to I > OTf > Br.²⁵⁷ In general, aromatic chlorides are insufficiently reactive to be suitable substrates for this reaction; however, chloroarylchromiumtricarbonyl complexes do undergo successful palladium-catalyzed cross-coupling with organotin reagents.²⁷⁰

Allylic halides couple effectively with vinyltin reagents to provide the corresponding 1,4-dienes in high yields.²⁷¹ Like nickel- and palladium-catalyzed Grignard reactions with allylic ethers, this reaction also proceeds with net inversion of configuration at the allylic center. Similarly, vinyl epoxides react

stereoselectively with aryl- and vinyl-stannanes to give the anti-substituted allylic alcohols.²⁷² In both cases, reaction occurs regioselectively at the least hindered allylic position of unsymmetrically substituted substrates. In the absence of steric bias, predominant 1,4-addition is observed in reactions of vinyl epoxides, while allylic chlorides apparently afford the products of α - and γ -attack in a 1:1 ratio. Interestingly, regioselectivity is improved when reactions of vinyl epoxides are performed in the presence of 10 equiv. of water (equation 22).



1.5.3.2.4 Silicon reagents

Although low reagent nucleophilicity has proven advantageous in cross-coupling reactions of organostannanes, the extremely low polarity of the carbon-silicon bond has prohibited use of the analogous organosilanes in these reactions. Recently, however, it has been discovered that tris(diethylamino) sulfonium difluorotrimethylsilicate (TASF) renders the compounds more anionic, facilitating the transmetalation step and allowing successful palladium-catalyzed coupling with organic halides.²⁷³ Allylpalladium chloride dimer is the catalyst of choice for these reactions, and the presence of triethyl phosphite as a cocatalyst markedly accelerates the rate of reaction with alkenyl iodides. More recently, it has been reported²⁷⁴ that dimethylfluorosilanes are considerably more reactive than the corresponding trimethylsilanes; advantage may be taken of this fact in order to prepare 1-silyl-1,3-dienes. Although tetrabutylammonium fluoride (TBAF) was not effectual in reactions of trialkylsilanes, it was used successfully with the more reactive dimethylfluorosilane substrates.

ACKNOWLEDGEMENT

JMK gratefully acknowledges financial support from the National Institutes of Health in the form of a postdoctoral fellowship, 1987-1989 (grant GM11540-02).

1.5.4 REFERENCES

- 1. (a) M. E. O'Neill and K. Wade, in 'Comprehensive Organometallic Chemistry', ed. G. Wilkinson, F. G. A. Stone and E. W. Abel, Pergamon Press, Oxford, 1982, vol. 1, p. 1; (b) E. Negishi, 'Organometallics in Organic Synthesis', Wiley, New York, 1980, vol. 1.
- 2. For a recent review of alkylation reactions of transition metal compounds with organic halides, see: F. Naso and G. Marchese, in 'The Chemistry of Functional Groups, Supplement D', ed. S. Patai and Z. Rappoport, Wiley, New York, 1983, p. 1353.
- 3. For general references regarding the use of organometallic reagents in organic synthesis, see: (a) L. S. Hegedus, in 'Organometallic Chemistry Reviews', ed. R. B. King and J. P. Oliver, Elsevier, Amsterdam, 1987, vol. 19, p. 1; (b) J. P. Collman, L. S. Hegedus, J. R. Norton and R. G. Finke, 'Principles and Applications of Organotransition Metal Chemistry', University Science Books, Mill Valley, CA, 1987; (c) D. Applications of Organon alisticin infectal chemistry, University Science Books, Juli Valley, Cr., 1997, C. Seyferth (ed.), 'New Applications of Organometallic Reagents in Organic Synthesis', Elsevier, Amsterdam, 1980; (d) J. C. Stowell, 'Carbanions in Synthesis', Wiley, New York, 1979; see also ref. 1b.
 4. M. S. Kharasch and P. O. Tawney, J. Am. Chem. Soc., 1941, 63, 2308.
- 5. M. S. Kharasch and C. F. Fuchs, J. Am. Chem. Soc., 1943, 65, 504. It should be noted, however, that halides of cobalt and chromium were considerably more successful than copper(I) salts in this reaction.
- R. C. Böttger, Justus Liebigs Ann. Chem., 1859, 109, 351.
 R. Reich, C. R. Hebd. Seances Acad. Sci., 1923, 177, 322.
- H. Gilman and J. M. Straley, Recl. Trav. Chim. Pay-Bas, 1936, 55, 821.
 H. Gilman, R. G. Jones and L. A. Woods, J. Org. Chem., 1952, 17, 1630.
- (a) G. H. Posner, 'An Introduction to Synthesis Using Organocopper Reagents', Wiley, New York, 1980; (b) G. H. Posner, Org. React. (N.Y.), 1975, 22, 253; (c) E. Erdik, Tetrahedron, 1984, 40, 641; (d) A. E. Jukes, Adv. Organomet. Chem., 1974, 12, 215; (e) J. F. Normant, Synthesis, 1972, 63.
- 11. For reviews and leading references, see: (a) G. van Koten and J. G. Noltes, in 'Comprehensive Organometallic Chemistry', ed. G. Wilkinson, F. G. A. Stone and E. W. Abel, Pergamon Press, Oxford, 1982, vol. 2, chap. 14,

p. 709; (b) 'Gmelin Handbook of Inorganic Chemistry: Organocopper Compounds', Springer-Verlag, Berlin, part 1, 1985; part 2, 1983.

- 12. For a systematic comparison of common cuprate precursors, see: S. H. Bertz, C. P. Gibson and G. Dabbagh, Tetrahedron Lett., 1987, 28, 4251.
- 13. G. B. Kauffman and L. A. Teter, Inorg. Synth., 1963, 7, 9.
- 14. Fresh CuBr SMe₂ of high purity can be prepared by reduction of Cu^{II}Br₂ with Na₂SO₃, followed by treatment with dimethyl sulfide. A. B. Theis and C. A. Townsend, Synth. Commun., 1981, 11, 157.
- 15. B. H. Lipshutz, S. Whitney, J. A. Kozlowski and C. M. Breneman, Tetrahedron Lett., 1986, 27, 4273.
- 16. Y. Pan, D. K. Hutchinson, M. H. Nantz and P. L. Fuchs, Tetrahedron, 1989, 45, 467.
- 17. J. R. Behling, K. A. Babiak, J. S. Ng, A. L. Campbell, R. Moretti, M. Koerner and B. H. Lipshutz, J. Am. Chem. Soc., 1988, 110, 2641.
- S. C. Berk, P. Knochel and M. C. P. Yeh, J. Org. Chem., 1988, 53, 5791.
 R. D. Rieke, R. M. Wehmeyer, T.-C. Wu and G. W. Ebert, Tetrahedron, 1989, 45, 443, and refs. cited therein. 20. For a review of the chemistry of copper(I) acetylides, see: A. M. Sladkov and I. R. Gol'ding, Usp. Khim., 1979, 48, 1625; Russ. Chem. Rev. (Engl. Transl.), 1979, 48, 868.
- 21. (a) J. F. Normant, in 'Modern Synthetic Methods', ed. R. Scheffold, Wiley, Chichester, 1983, vol. 3, p. 139, and refs. cited therein; (b) J. F. Normant and A. Alexakis, Synthesis, 1981, 841.
- 22. For reviews, see: (a) P. E. Fanta, Chem. Rev., 1946, 38, 139; (b) P. E. Fanta, Chem. Rev., 1964, 64, 613; (c) P. E. Fanta, Synthesis, 1974, 9.
- 23. M. Tamura and J. K. Kochi, Synthesis, 1971, 303.
- 24. S. H. Bertz and G. Dabbagh, Tetrahedron, 1989, 45, 425.
- 25. E. Wehman, G. van Koten, C. J. M. Erkamp, D. M. Knotter, J. T. B. H. Jastrzebski and C. H. Stam, Organometallics, 1989, 8, 94, and refs. cited therein.
- G. van Koten and J. T. B. H. Jastrzebski, *Tetrahedron*, 1989, 45, 569.
 (a) S. H. Bertz, G. Dabbagh and G. M. Villacorta, J. Am. Chem. Soc., 1982, 104, 5824; (b) S. H. Bertz and G. Dabbagh, J. Org. Chem., 1984, 49, 1119. The authors report that CuBr SMe₂ must be used for the preparation of dicyclohexylphosphidocuprates, since CuI is reduced to Cu⁰.
- 28. S. F. Martin, J. R. Fishpaugh, J. M. Power, D. M. Giolando, R. A. Jones, C. M. Nunn and A. H. Cowley, J. Am. Chem. Soc., 1988, 110, 7226.
- 29. Alkynyl copper reagents are an exception and must be used as the copper(I) acetylides.
- 30. S. Muralidharan and H. Freiser, Inorg. Chem., 1988, 27, 3251.
- 31. R. G. Pearson and C. D. Gregory, J. Am. Chem. Soc., 1976, 98, 4098.
- 32. G. van Koten, J. T. B. H. Jastrzebski, F. Muller and C. H. Stam, J. Am. Chem. Soc., 1985, 107, 697.
- 33. On the basis of molecular orbital calculations, Whangbo has suggested that a structure in which the methyl carbanions are bonded exclusively to copper is a more accurate representation of the true bonding nature of Me₂CuLi. K. R. Stewart, J. R. Lever and M. -H. Whangbo, J. Org. Chem., 1982, 47, 1472. The stability of the bridging, two-electron, three-center bonding pattern increases in the order alkyl < aryl < alkynyl. See ref. 11a. (a) B. H. Lipshutz, J. A. Kozlowski and C. M. Breneman, J. Am. Chem. Soc., 1985, 107, 3197; (b) B. H.
- 34. Lipshutz, E. L. Ellsworth and T. J. Siahaan, J. Am. Chem. Soc., 1989, 111, 1351; (c) B. H. Lipshutz, E. L. Ellsworth and T. J. Siahaan, J. Am. Chem. Soc., 1988, 110, 4834.
- 35. For another discussion of the effect of solvent variation on the structure of cuprate reagents, see: C. Ullenius and B. Christenson, Pure Appl. Chem., 1988, 60, 57.
- 36. For a review, see: Y. Yamamoto, Angew. Chem., Int. Ed. Engl., 1986, 25, 947.
- 37. E. J. Corey, F. J. Hannon and N. W. Boaz, Tetrahedron, 1989, 45, 545.
- 38. E. J. Corey, R. Naef and F. J. Hannon, J. Am. Chem. Soc., 1986, 108, 7114.
- 39. (a) G. H. Posner, C. E. Whitten and J. J. Sterling, J. Am. Chem. Soc., 1973, 95, 7788; (b) E. Piers and I. Nagakura, J. Org. Chem., 1975, 40, 2694; (c) P. A. Grieco, C.-L. J. Wang and G. Majetich, J. Org. Chem., 1976, 41, 726; (d) G. H. Posner and C. E. Whitten, Org. Synth., 1976, 55, 122.
- 40. C. R. Johnson and D. S. Dhanoa, J. Org. Chem., 1987, 52, 1885. These authors previously reported the use of nontransferable sulfonyl ligands. See: C. R. Johnson and D. S. Dhanoa, J. Chem. Soc., Chem. Commun., 1982, 358.
- E. C. Ashby and J. J. Watkins, J. Am. Chem. Soc., 1977, 99, 5312.
- 42. For examples, see: (a) E. C. Ashby and J. J. Lin, J. Org. Chem., 1977, 42, 2805; (b) E. C. Ashby, J. J. Lin and J. J. Watkins, J. Org. Chem., 1977, 42, 1099; (c) Y. Kojima and N. Kato, Tetrahedron Lett., 1980, 21, 4365.
- 43. S. H. Bertz and G. Dabbagh, J. Am. Chem. Soc., 1988, 110, 3668; for a recently reported X-ray crystal structure of a higher-order cuprate prepared in dimethyl sulfide, see M. M. Olmstead and P. P. Power, J. Am. Chem. Soc., 1989, 111, 4135.
- 44. D. L. J. Clive, V. Farina and P. L. Beaulieu, J. Org. Chem., 1982, 47, 2572.
- 45. C. Gallina, Tetrahedron Lett., 1982, 23, 3093.
- 46. For reviews, see: (a) B. H. Lipshutz, Synthesis, 1987, 325; (b) B. H. Lipshutz, R. S. Wilhelm and J. A. Kozlowski, Tetrahedron, 1984, 40, 5005; for spectroscopic investigations of these reagents, see (c) B. H. Lipshutz, J. A. Kozlowski and R. S. Wilhelm, J. Org. Chem., 1984, 49, 3943; (d) S. H. Bertz, J. Am. Chem. Soc., 1990, 112, 4031; (e) B. H. Lipshutz, S. Sharma and E. L. Ellsworth, J. Am. Chem. Soc., 1990, 112, 4032.
- 47. Lower-order heterocuprates containing the 2-thienyl ligand had previously been reported. See: E.-L. Lindstedt and M. Nilsson, Acta Chem. Scand., Ser. B, 1986, 40, 466, and refs. cited therein.
- 48. B. H. Lipshutz, M. Koerner and D. A. Parker, Tetrahedron Lett., 1987, 28, 945.
- 49. J. K. Stille, in 'Modern Synthetic Methods', ed. R. Scheffold, Wiley, Chichester, 1983, vol. 3, p. 1.
- 50. E. J. Corey and N. W. Boaz, Tetrahedron Lett., 1984, 25, 3063.
- 51. Copper(III) complexes that bear strongly σ -donating ligands are known, however. For reviews, see: (a) F. C. Anson, T. J. Collins, T. G. Richmond, B. D. Santarsiero, J. E. Toth and B. G. R. T. Treco, J. Am. Chem. Soc., 1987, 109, 2974; (b) W. Levason and M. D. Spicer, Coord. Chem. Rev., 1987, 76, 45; (c) D. W. Margerum and G. D. Owens, Met. Ions Biol. Syst., 1981, 12, 75.

- 52. Stable Au^{III} intermediates have been identified in analogous reactions of organogold compounds. See ref. 53, p. 386, and refs. therein.
- 53. J. K. Kochi, 'Organometallic Mechanisms and Catalysis', Academic, New York, 1978.
- 54. B. H. Lipshutz, R. S. Wilhelm, J. A. Kozlowski and D. Parker, J. Org. Chem., 1984, 49, 3928.
- 55. E. C. Ashby and D. Coleman, J. Org. Chem., 1987, 52, 4554, and refs. cited therein. For an earlier mechanistic investigation involving these substrates, see: G. H. Posner and J.-S. Ting, Tetrahedron Lett., 1974, 683.
- 56. B. H. Lipshutz, R. S. Wilhelm, S. T. Nugent, R. D. Little and M. M. Baizer, J. Org. Chem., 1983, 48, 3306.
- 57. B. H. Lipshutz and R. S. Wilhelm, J. Am. Chem. Soc., 1982, 104, 4696.
- 58. E. C. Ashby and R. DePriest, J. Am. Chem. Soc., 1982, 104, 6144.
- 59. J. D. White, G. N. Reddy and G. O. Spessard, J. Am. Chem. Soc., 1988, 110, 1624. 60. K.-H. Chu and K. K. Wang, J. Org. Chem., 1986, 51, 767. For another approach to the synthesis of tetrasubstituted alkenes that also relies on alkylation of a vinylcopper intermediate, see: E. Piers and R. T. Skerlj, J. Org. Chem., 1987, 52, 4421.
- 61. E. J. Corey and W. L. Seibel, Tetrahedron Lett., 1986, 27, 905, 909.
- 62. G. Fouquet and M. Schlosser, Angew. Chem., Int. Ed. Engl., 1974, 13, 82.
- 63. A. R. Chamberlin and S. H. Bloom, Tetrahedron Lett., 1984, 25, 4901.
- 64. F. E. Ziegler and K. Mikami, Tetrahedron Lett., 1984, 25, 131.
- 65. F. Derguini, Y. Bessiere and G. Linstrumelle, Synth. Commun., 1981, 11, 859.
- (a) M. Larchevêque and Y. Petit, *Tetrahedron Lett.*, 1984, 3705; (b) J. A. Bajgrowicz, A. El Hallaoui, R. Jacquier, Ch. Pigiere and Ph. Viallefont, *Tetrahedron*, 1985, 41, 1833, and refs. cited therein. 66.
- 67. E. J. Corey, K. Š. Kyler and N. Raju, Tetrahedron Lett., 1984, 25, 5115.
- 68. An exception is (2-furyl)₂CuCNLi₂, which is much less reactive than other higher-order cyanocuprates and which does not displace alkyl iodides or bromides in synthetically useful yields. J. S. Ng, J. R. Behling, A. L. Campbell, D. Nguyen and B. H. Lipshutz, *Tetrahedron Lett.*, 1988, **29**, 3045.
- 69. For a theoretical treatment of the effect of π -accepting ligands (either as part of the reagent or as the solvent) on alkylation reactions of organocuprates, see: L. Hamon and J. Levisalles, Tetrahedron, 1989, 45, 489.
- 70. (a) E. J. Corey and G. H. Posner, J. Am. Chem. Soc., 1968, 90, 5615; (b) K. Kitatani, T. Hiyama and H. Nozaki, J. Am. Chem. Soc., 1976, 98, 2362.
- 71. For a recent example, see: M. Saha, S. Muchmore, D. van der Helm and K. M. Nicholas, J. Org. Chem., 1986, 51, 1960.
- 72. T. Harayama, H. Fukushi, K. Ogawa, T. Aratani, S. Sonehara and F. Yoneda, Chem. Pharm. Bull., 1987, 35, 4977.
- 73. F. Scott, B. G. Mafunda, J. F. Normant and A. Alexakis, Tetrahedron Lett., 1983, 24, 5767.
- 74. G. H. Posner and J.-S. Ting, Synth. Commun., 1973, 3, 281.
- 75. P. Ritterskamp, M. M. Demuth and K. Schaffner, J. Org. Chem., 1984, 49, 1155.
- 76. J. K. Thottathil and J. L. Moniot, *Tetrahedron Lett.*, 1986, 27, 183. This mechanistic rationale has been questioned, however. See: T. Rosen, S. W. Fesik, D. T. W. Chu and A. G. Pernet, *Synthesis*, 1988, 40.
- 77. É. Hebert, Tetrahedron Lett., 1982, 23, 415.
- 78. R. Bihovsky, C. Selick and I. Giusti, J. Org. Chem., 1988, 53, 4026.
- 79. M. Ochiai, K. Sumi, Y. Nagao and E. Fujita, Tetrahedron Lett., 1985, 26, 2351.
- 80. (a) A. M. Caporusso, C. Polizzi and L. Lardicci, J. Org. Chem., 1987, 52, 3920; (b) A. M. Caporusso, C. Polizzi and L. Lardicci, Tetrahedron Lett., 1987, 28, 6073.
- 81. (a) H. H. Mooiweer, C. J. Elsevier, P. Wijkens and P. Vermeer, Tetrahedron Lett., 1985, 26, 65, and refs. cited therein; (b) E. J. Corey and N. W. Boaz, Tetrahedron Lett., 1984, 25, 3059.
- 82. (a) H. Westmijze, I. Nap, J. Meijer, H. Kleijn and P. Vermeer, Recl. Trav. Chim. Pay-Bas, 1983, 102, 154; (b) A. Claesson and L. I. Olsson, J. Chem. Soc., Chem. Commun., 1979, 524.
- For a stereoelectronic explanation, see ref. 81b.
- 84. (a) E. J. Corey and T. M. Eckrich, Tetrahedron Lett., 1984, 25, 2419; (b) N. Jabri, A. Alexakis and J. F. Normant, Tetrahedron Lett., 1982, 23, 1589; (c) E. J. Corey and J. Kang, J. Am. Chem. Soc., 1981, 103, 4618.
- 85. J. M. Oostveen, H. Westmijze and P. Vermeer, J. Org. Chem., 1980, 45, 1158.
- T. Tsuda, T. Yoshida and T. Saegusa, J. Org. Chem., 1988, 53, 607.
 (a) J. P. Marino and R. J. Linderman, J. Org. Chem., 1983, 48, 4621; (b) J. P. Marino and D. M. Floyd, J. Am. Chem. Soc., 1974, 96, 7138.
- 88. G. H. Posner, W. Harrison and D. G. Wettlaufer, J. Org. Chem., 1985, 50, 5041.
- K. Sonogashira, Y. Tohda and N. Hagihara, *Tetrahedron Lett.*, 1975, **50**, 4467.
 (a) C. E. Castro, R. Havlin, V. K. Honwad, A. M. Malte and S. Mojé, J. Am. Chem. Soc., 1969, **91**, 6464, and refs. cited therein; (b) D. C. Owsley and C. E. Castro, Org. Synth, Coll. Vol., 1988, 6, 916. 91. (a) R. Rossi, A. Carpita, M. G. Quirici and M. L. Gaudenzi, Tetrahedron, 1982, 38, 631; (b) V.
- Ratovelomanana and G. Linstrumelle, Synth. Commun., 1981, 11, 917. 92. T. F. Tam, R. W. Spencer, E. M. Thomas, L. J. Copp and A. Krantz, J. Am. Chem. Soc., 1984, 106, 6849.
- 93. E. J. Corey and M. d'Alarcao, Tetrahedron Lett., 1986, 27, 3589.
- 94. (a) D. Guillerm and G. Linstrumelle, Tetrahedron Lett., 1986, 27, 5857; (b) K. C. Nicolaou, C. A. Veale, S. E. Webber and H. Katerinopoulos, J. Am. Chem. Soc., 1985, 107, 7515; (c) K. C. Nicolaou and S. E. Webber, J. Am. Chem. Soc., 1984, 106, 5734.
- 95. (a) P. Magnus, R. T. Lewis and J. C. Huffman, J. Am. Chem. Soc., 1988, 110, 6921; (b) S. L. Schreiber and L. L. Kiessling, J. Am. Chem. Soc., 1988, 110, 631; (c) A. S. Kende and C. A. Smith, Tetrahedron Lett., 1988, 29, 4217; (d) K. Tomioka, H. Fujita and K. Koga, Tetrahedron Lett., 1989, 30, 851.
- 96. T. Jeffrey-Luong and G. Linstrumelle, Synthesis, 1983, 32.
- 97. L. C. Blaszczak, J. D. Winkler and S. O'Kuhn, Tetrahedron Lett., 1976, 49, 4405.
- 98. T. Ishihara, M. Yamana and T. Ando, Tetrahedron Lett., 1983, 24, 5657.
- 99. W. J. Scott and J. E. McMurry, Acc. Chem. Res., 1988, 21, 47, and refs. cited therein.

- 100. (a) T. Fujisawa, Y. Kurita, M. Kawashima and T. Sato, Chem. Lett., 1982, 1641; (b) P. J. Kocienski, S. Wadman and K. Cooper, J. Am. Chem. Soc., 1989, 111, 2363.
- 101. M. Marsi and M. Rosenblum, J. Am. Chem. Soc., 1984, 106, 7264.
- 102. M. Rosenblum, M. M. Turnbull and B. M. Foxman, Organometallics, 1986, 5, 1062.
- 103. D. Villemin, P. Cadiot and M. Kuétegan, Synthesis, 1984, 230.
- 104. (a) W. Verboom, H. Westmijze, H. J. T. Bos and P. Vermeer, Tetrahedron Lett., 1978, 1441; (b) R. Oliver and D. R. M. Walton, Tetrahedron Lett., 1972, 5209.
- 105. P. J. Stang and T. Kitamura, J. Am. Chem. Soc., 1987, 109, 7561.
- 106. T. W. Bell, L.-Y. Hu and S. V. Patel, J. Org. Chem., 1987, 52, 3847.
- 107. K. Hirota, Y. Kitade, Y. Isobe and Y. Maki, Heterocycles, 1987, 26, 355.
- 108. E. C. Taylor and P. S. Ray, J. Org. Chem., 1987, 52, 3997.
- 109. M. J. Robins and P. J. Barr, Tetrahedron Lett., 1981, 22, 421.
- 110. D. L. J. Clive, A. G. Angoh and S. M. Bennett, J. Org. Chem., 1987, 52, 1339.
- 111. N. A. Bumagin, A. B. Ponomaryov and I. P. Beletskaya, Synthesis, 1984, 728.
- 112. For more recent examples, see: (a) G. Batu and R. Stevenson, J. Org. Chem., 1980, 45, 1532; (b) G. E. Schneiders and R. Stevenson, Synth. Commun., 1980, 10, 699.
- 113. S. Miyano, S. Handa, M. Tobita and H. Hashimoto, Bull. Chem. Soc. Jpn., 1986, 59, 235, and refs. cited therein.
- 114. (a) F. E. Ziegler, I. Chliwner, K. W. Fowler, S. J. Kanfer, S. J. Kuo and N. D. Sinha, J. Am. Chem. Soc., 1980, 102, 790; (b) A. S. Kende and D. P. Curran, J. Am. Chem. Soc., 1979, 101, 1857; (c) H. Malmberg and M. Nilsson, Tetrahedron, 1986, 42, 3981.
- 115. Y. Masaki, K. Sakuma and K. Kaji, J. Chem. Soc., Perkin Trans. 1, 1985, 1171, and refs. cited therein.
- 116. S. Araki and Y. Butsugan, J. Chem. Soc., Perkin Trans. 1, 1984, 969.
- 117. C. E. McDonald and R. W. Dugger, Tetrahedron Lett., 1988, 29, 2413.
- C. Gallina and P. G. Ciattini, J. Am. Chem. Soc., 1979, 101, 1035. 118.
- 119. K. C. Nicolaou, T. Ladduwahetty and E. M. Elisseou, J. Chem. Soc., Chem. Commun., 1985, 1580.
- 120. W. Boland and K. Mertes, Synthesis, 1985, 705.
- 121. (a) R. J. Anderson, C. A. Henrick and J. B. Siddall, J. Am. Chem. Soc., 1970, 92, 735; (b) A. Kreft, Tetrahedron Lett., 1977, 1035; (c) C. C. Tseng, S. D. Paisley and H. L. Goering, J. Org. Chem., 1986, 51, 2884.
- 122. S. P. Tanis and D. B. Head, Tetrahedron Lett., 1982, 23, 5509.
- 123. A. J. Pearson and Y.-S. Lai, J. Chem. Soc., Chem. Commun., 1988, 442.
- 124. P. A. Grieco, R. E. Zelle, R. Lis and J. Finn, J. Am. Chem. Soc., 1983, 105, 1403.
- T. Fujisawa, K. Umezu and M. Kawashima, Chem. Lett., 1984, 1795. 125.
- 126. E. J. Corey and A. V. Gavai, Tetrahedron Lett., 1988, 29, 3201.
- 127. For example, see: H.-J. Liu and L.-K. Ho, Can. J. Chem., 1983, 61, 632.
- 128. A. Carpita and R. Rossi, Synthesis, 1982, 469.
- 129. D. P. Curran, M.-H. Chen, D. Leszczweski, R. L. Elliott and D. M. Rakiewicz, J. Org. Chem., 1986, 51, 1612.
- 130. T. L. Fevig, R. L. Elliott and D. P. Curran, J. Am. Chem. Soc., 1988, 110, 5064.
- 131. (a) K. Sekiya and E. Nakamura, Tetrahedron Lett., 1988, 29, 5155; (b) E. Nakamura, S. Aoki, K. Sekiya, H. Oshino and I. Kuwajima, J. Am. Chem. Soc., 1987, 109, 8056; (c) H. Ochiai, Y. Tamaru, K. Tsubaki and Z. Yoshida, J. Org. Chem., 1987, **52**, 4418. 132. E. Nakamura, K. Sekiya, M. Arai and S. Aoki, J. Am. Chem. Soc., 1989, **111**, 3091.
- 133. ' R_2 CuZnCl' and ' R_2 CuTi(OPrⁱ)₃' are prepared by treatment of R_2 CuLi with ZnCl₂ and Ti(OPrⁱ)₃, respectively.
- 134. N. Miyaura, M. Itoh and A. Suzuki, Bull. Chem. Soc. Jpn., 1977, 50, 2199.
- 135. R. W. Rickards and H. Rönneberg, J. Org. Chem., 1984, 49, 572.
- 136. T. Ibuka, T. Nakao, S. Nishii and Y. Yamamoto, J. Am. Chem. Soc., 1986, 108, 7420.
 137. T. Ibuka and H. Minakata, Synth. Commun., 1980, 10, 119.
- 138. (a) T. L. Underiner and H. L. Goering, J. Org. Chem., 1988, 53, 1140, and refs. cited therein; (b) H. L. Goering and S. S. Kantner, J. Org. Chem., 1984, 49, 422.
- 139. J.-E. Bäckvall and M. Sellén, J. Chem. Soc., Chem. Commun., 1987, 827.
- 140. M. H. Beale, J. Chem. Soc., Perkin Trans. 1, 1985, 1183.
- 141. For a recent synthetic application of this reaction, see: A. E. Greene, F. Coelho, J.-P. Deprés and T. J. Brocksom, Tetrahedron Lett., 1988, 29, 5661.
- 142. H. L. Goering, S. S. Kantner and C. C. Tseng, J. Org. Chem., 1983, 48, 715.
- 143. (a) Y. Tanigawa, H. Ohta, A. Sonoda and S.-I. Murahashi, J. Am. Chem. Soc., 1978, 100, 4610; (b) Y. Tanigawa, Y. Fuse and S.-I. Murahashi, Tetrahedron Lett., 1982, 23, 557
- 144. For examples of synthetic applications of this reaction, see: (a) K. Bannai, T. Tanaka, N. Okamura, A. Hazato, S. Sugiura, K. Manabe, K. Tomimori and S. Kurozumi, Tetrahedron Lett., 1986, 27, 6353; (b) C. Fan and B. Cazes, Tetrahedron Lett., 1988, 29, 1701.
- 145. V. Calo, L. Lopez and G. Pesce, J. Chem. Soc., Perkin Trans. 1, 1988, 1301, and refs. cited therein.
- 146. (a) S. Patai (ed.), 'The Chemistry of Ketenes, Allenes, and Related Compounds', Wiley, Chichester, 1980; (b) H. F. Schuster and G. M. Coppola, 'Allenes in Organic Synthesis', Wiley, New York, 1984; (c) S. R. Landor (ed.), 'The Chemistry of the Allenes', Academic Press, London, 1982.
- 147. A. Haces, E. M. G. A. van Kruchten and W. H. Okamura, Isr. J. Chem., 1985, 26, 140, and refs. cited therein.
- 148. G.-Y. Shen, A. R. de Lera, T. C. Norman, A. Haces and W. H. Okamura, Tetrahedron Lett., 1987, 28, 2917, and refs. cited therein.
- 149. J. A. Marshall and S. D. Rothenberger, Tetrahedron Lett., 1986, 27, 4845.
- 150. Y. Tanigawa and S.-I. Murahashi, J. Org. Chem., 1980, 45, 4536.
- 183. (a) W. J. Klaver, M. J. Moolenaar, W. N. Speckamp and H. Hiemstra, Tetrahedron, 1988, 44, 3805; (b) D. Djahanbini, B. Cazes and J. Gore, Tetrahedron, 1984, 40, 3645.
- 152 R. S. Brinkmeyer and T. L. Macdonald, J. Chem. Soc., Chem. Commun., 1978, 876.
- 153. T. L. Macdonald, D. R. Reagan and R. S. Brinkmeyer, J. Org. Chem., 1980, 45, 4740.

- 154. While reactive propargylic substrates probably react by a mechanism analogous to that of allylic substrates, less reactive propargylic ethers apparently undergo reaction by a mechanism involving syn addition across the alkyne, followed by elimination of the leaving group. For an investigation of stereocontrol in these reactions, see: I. Marek, P. Mangeney, A. Alexakis and J. F. Normant, Tetrahedron Lett., 1986, 27, 5499.
- 155. For examples, see: (a) H. R. Buser, P. M. Guerin, M. Toth, G. Szöcs, A. Schmid, W. Francke and H. Arn, Tetrahedron Lett., 1985, 26, 403; (b) J. G. Millar and A. C. Ochlschlager, J. Org. Chem., 1984, 49, 2332. 156. (a) T. Katsuki and K. B. Sharpless, J. Am. Chem. Soc., 1980, 102, 5974; (b) Y. Gao, R. M. Hanson, J. M.
- Klunder, S. Y. Ko, H. Masamune and K. B. Sharpless, J. Am. Chem. Soc., 1987, 109, 5765.
- 157. For recent reviews of epoxide transformations, see: (a) J. G. Smith, Synthesis, 1984, 629; (b) A. S. Rao, S. K. Paknikar and J. G. Kirtane, Tetrahedron, 1983, 39, 2323.
- 158. R. W. Herr, D. M. Wieland and C. R. Johnson, J. Am. Chem. Soc., 1970, 92, 3813.
- (a) R.-D. Acker, Tetrahedron Lett., 1977, 3407; (b) R.-D. Acker, Tetrahedron Lett., 1978, 2399. 159.
- 160. For use in a reiterative approach to all-syn polyols, see: B. H. Lipshutz and J. C. Barton, J. Org. Chem., 1988, 53, 4495.
- 161. C. Huynh, F. Derguini-Boumechal and G. Linstrumelle, Tetrahedron Lett., 1979, 1503.
- Among many recent examples, see: (a) Y. Kobayashi, N. Kato, T. Shimazaki and F. Sato, Tetrahedron Lett., 162. 1988, 29, 6297; (b) K. Mori and M. Kato, Tetrahedron, 1986, 42, 5895.
- 163. For example, see: (a) C. Brockway, P. J. Kocienski and C. Pant, J. Chem. Soc., Perkin Trans. 1, 1984, 875; (b) R. J. Anderson, K. G. Adams, H. R. Chinn and C. A. Henrick, J. Org. Chem., 1980, 45, 2229.
- 164.
- J. Mulzer and O. Lammer, Chem. Ber., 1986, 119, 2178.
 J. A. Ciaccio, K. J. Addess and T. W. Bell, Tetrahedron Lett., 1986, 27, 3697.
- 166. A. Alexakis, D. Jachiet and J. F. Normant, Tetrahedron, 1986, 42, 5607.
- 167. (a) B. H. Lipshutz, D. A. Parker, J. A. Kozlowski and S. L. Nguyen, Tetrahedron Lett., 1984, 25, 5959; (b) A. Alexakis and D. Jachiet, Tetrahedron, 1989, 45, 381, and refs. cited therein.
- 168. J. S. Ng, J. R. Behling, A. L. Campbell, D. Nguyen and B. H. Lipshutz, Tetrahedron Lett., 1988, 29, 3045.
- 169. (a) M. J. Eis and B. Ganem, Tetrahedron Lett., 1985, 26, 1153; (b) A. P. Kozikowski, H. Ishida and K. Isobe, I. Org. Chem., 1979, 44, 2788.
- 170. R. Tanikaga, K. Hosoya and A. Kaji, Synthesis, 1987, 389.
- 171. M. A. Tius and A. H. Fauq, J. Org. Chem., 1983, 48, 4131.
- (a) A. D. C. Schuda, P. H. Mazzocchi, G. Fritz and T. Morgan, Synthesis, 1986, 309; (b) S. P. Tanis and P. M. 172. Herrinton, J. Org. Chem., 1983, 48, 4572.
- 173. (a) G. Adam, R. Zibuck and D. Seebach, J. Am. Chem. Soc., 1987, 109, 6176.
- 174. (a) K. C. Nicolaou, M. E. Duggan and T. Ladduwahetty, Tetrahedron Lett., 1984, 25, 2069; (b) J. G. Millar and E. W. Underhill, J. Org. Chem., 1986, 51, 4726; (c) J. Soulie, M. Lampilas and J. Y. Lallemand, Tetrahedron, 1987, 43, 2701.
- 175. S. P. Tanis, Y.-H. Chuang and D. B. Head, J. Org. Chem., 1988, 53, 4929.
- 176. C. R. Johnson and G. A. Dutra, J. Am. Chem. Soc., 1973, 95, 7777.
- 177. K. Mori and T. Ebata, Tetrahedron, 1986, 42, 3471, and refs. cited therein.
- 178. S. Challenger and G. Procter, Tetrahedron Lett., 1986, 27, 391.
- 179. M. J. Kurth and M. A. Abreo, Tetrahedron Lett., 1987, 28, 5631.
- 180. J. M. Klunder, T. Onami and K. B. Sharpless, J. Org. Chem., 1989, 54, 1295.
- 181. S. Takano, M. Yanase, T. Sugihara and K. Ogasawara, J. Chem. Soc., Chem. Commun., 1988, 1538.
- P. F. Hudrlik, D. Peterson and R. J. Rona, J. Org. Chem., 1975, 40, 2263. 182.
- 183. Y. Kitano, T. Matsumoto and F. Sato, J. Chem. Soc., Chem. Commun., 1986, 1323.
- 184. K. Tamao, E. Nakajo and Y. Ito, J. Org. Chem., 1988, 53, 414.
- 185. For reviews of 2,3-epoxy alcohol transformations, including reactions with organocopper reagents, see: (a) C. H. Behrens and K. B. Sharpless, Aldrichimica Acta, 1983, 16, 67; (b) K. B. Sharpless, C. H. Behrens, T. Katsuki, A. W. M. Lee, V. S. Martin, M. Takatani, S. M. Viti, F. J. Walker and S. S. Woodard, Pure Appl. Chem., 1983, 55, 589.
- 186. J. M. Chong, D. R. Cyr and E. K. Mar, Tetrahedron Lett., 1987, 28, 5009.
- 187. (a) S. Hoagland, Y. Morita, D. L. Bai, H.-P. Märki, K. L. Kees, L. Brown and C. H. Heathcock, J. Org. Chem., 1988, 53, 4730; (b) W. W. McWhorter, Jr., S. H. Kang and Y. Kishi, Tetrahedron Lett., 1983, 24, 2243. For reactions of 3,4-epoxy alcohols and ethers, see: T. Hanamoto, T. Katsuki and M. Yamaguchi, Tetrahedron Lett., 1987, 28, 6195.
- 188. P. C. B. Page, C. M. Rayner and I. O. Sutherland, J. Chem. Soc., Chem. Commun., 1988, 356.
- 189. M. Larchevêque and Y. Petit, Tetrahedron Lett., 1987, 28, 1993.
- 190. (a) Y. Petit, C. Sanner and M. Larchevêque, Synthesis, 1988, 538; (b) J. M. Chong, Tetrahedron, 1989, 45, 623.
- 191. J. M. Chong and K. B. Sharpless, Tetrahedron Lett., 1985, 26, 4683.
- 192. (a) R. J. Anderson, J. Am. Chem. Soc., 1970, 92, 4978; (b) R. W. Herr and C. R. Johnson, J. Am. Chem. Soc., 1970, 92, 4979.
- 193. J. P. Marino and H. Abe, J. Am. Chem. Soc., 1981, 103, 2907. For a recent synthetic application, see: S.-S. Moon, L. M. Stuhmiller and T. C. McMorris, J. Org. Chem., 1989, 54, 26.
- 194. (a) J. P. Marino, R. Fernández de la Pradilla and E. Laborde, J. Org. Chem., 1987, 52, 4898, and refs. therein; (b) M. F. Schlecht, J. Chem. Soc., Chem. Commun., 1982, 1331; (c) J. P. Marino and J. C. Jaén, J. Am. Chem. Soc., 1982, 104, 3165
- 195. P. A. Wender, J. M. Erhardt and L. J. Letendre, J. Am. Chem. Soc., 1981, 103, 2114.
- 196. J. P. Marino and H. Abe, J. Org. Chem., 1981, 46, 5379.
 197. J. A. Marshall and V. H. Audia, J. Org. Chem., 1987, 52, 1106; 1985, 50, 1607.
- 198. F. E. Ziegler and M. A. Cady, J. Org. Chem., 1981, 46, 122.
- 199. (a) J. A. Marshall, J. D. Trometer and D. G. Cleary, Tetrahedron, 1989, 45, 391; (b) J. A. Marshall, J. D. Trometer, B. E. Blough and T. D. Crute, J. Org. Chem., 1988, 53, 4274. 200. T. Ibuka, M. Tanaka, H. Nemoto and Y. Yamamoto, *Tetrahedron*, 1989, 45, 435.

Alkylation of Carbon

- 201. G. H. Posner, C. E. Whitten and P. E. McFarland, J. Am. Chem. Soc., 1972, 94, 5106.
- 202. R. M. Wehmeyer and R. D. Rieke, Tetrahedron Lett., 1988, 29, 4513.
- 203. N. Jabri, A. Alexakis and J. F. Normant, Tetrahedron, 1986, 42, 1369.
- 204. R. F. W. Jackson and R. A. Raphael, Tetrahedron Lett., 1983, 24, 2117.
- 205. T. Mukaiyama and M. Murakami, Synthesis, 1987, 1043, and refs. cited therein.
- 206. S. D. Lindell, J. D. Elliott and W. S. Johnson, Tetrahedron Lett., 1984, 25, 3947.
- 207. (a) J. F. Normant, A. Alexakis, A. Ghribi and P. Mangeney, *Tetrahedron*, 1989, **45**, 507; (b) A. Alexakis, P. Mangeney, A. Ghribi, I. Marek, R. Sedrani, C. Guir and J. F. Normant, *Pure Appl. Chem.*, 1988, **60**, 49.
- 208. (a) Y. Guindon, M. A. Bernstein and P. C. Anderson, Tetrahedron Lett., 1987, 28, 2225; (b) Y. Guindon and P. C. Anderson, Tetrahedron Lett., 1987, 28, 2485.
- 209. S. L. Schreiber and J. Reagan, Tetrahedron Lett., 1986, 27, 2945.
- (a) J. F. Normant, A. Alexakis and G. Cahiez, Tetrahedron Lett., 1980, 21, 935; (b) T. Sato, T. Kawara, M. Kawashima and T. Fujisawa, Chem. Lett., 1980, 571; (c) T. Fujisawa, T. Sato, T. Kawara, M. Kawashima, H. Shimizu and Y. Ito, Tetrahedron Lett., 1980, 21, 2181.
- 211. T. Sato, K. Naruse and T. Fujisawa, Tetrahedron Lett., 1982, 23, 3587, and refs. cited therein.
- 212. T. Fujisawa, N. Okada, M. Takeuchi and T. Sato, Chem. Lett., 1983, 1271.
- 213. L. D. Arnold, J. C. G. Drover and J. C. Vederas, J. Am. Chem. Soc., 1987, 109, 4649.
- (a) R. F. Heck, 'Palladium Reagents in Organic Synthesis', Academic Press, New York, 1985; (b) B. M. Trost and T. R. Verhoeven, in 'Comprehensive Organometallic Chemistry', ed. G. Wilkinson, F. G. A. Stone and E. W. Abel, Pergamon Press, Oxford, 1982, vol. 8, p. 799; (c) P. W. Jolly, in 'Comprehensive Organometallic Chemistry', ed. G. Wilkinson, F. G. A. Stone and E. W. Abel, Pergamon Press, Oxford, 1982, vol. 8, p. 713.
- R. J. P. Corriu and J. P. Masse, J. Chem. Soc., Chem. Commun., 1972, 144.
 K. Tamao, K. Sumitani and M. Kumada, J. Am. Chem. Soc., 1972, 94, 4374.
- 217. For reviews see: (a) K. Tamao and M. Kumada, in 'The Chemistry of the Metal-Carbon Bond', ed. F. R. Hartley, Wiley, New York, 1987, p. 819; (b) M. Kumada, Pure Appl. Chem., 1980, 669.
- 218. K. Tamao, Y. Kiso, K. Sumitani and M. Kumada, J. Am. Chem. Soc., 1972, 94, 9268.
- 219. T. Hayashi, M. Konishi, K. Yokota and M. Kumada, Chem. Lett., 1980, 767.
- 220. T. Hayashi, M. Konishi, Y. Kobori, M. Kumada, T. Higuchi and K. Hirotsu, J. Am. Chem. Soc., 1984, 106, 158.
- 221. For reviews, see: (a) T. Hayashi and M. Kumada, in 'Asymmetric Synthesis', ed. J. D. Morrison, Academic Press, New York, 1985, vol. 5, chap. 5; (b) T. Hayashi, in 'Asymmetric Reactions and Processes in Chemistry', American Chemical Society, Washington, DC, 1982, chap. 12.
- 222. (a) G. Consiglio and C. Botteghi, Helv. Chim. Acta, 1973, 56, 460; (b) Y. Kiso, K. Tamao, N. Miyake, K. Yamamoto and M. Kumada, Tetrahedron Lett., 1974, 3.
- 223. G. A. Cross and R. M. Kellogg, J. Chem. Soc., Chem. Commun., 1987, 1746.
- (a) B. K. Vriesema and R. M. Kellogg, Tetrahedron Lett., 1986, 27, 2049; (b) J. H. Griffin and R. M. Kellogg, J. Org. Chem., 1985, 50, 3261; (c) B. K. Vriesema, M. Lemaire, J. Buter and R. M. Kellogg, J. Org. Chem., 1986, 51, 5169; (d) Chr. Döbler and H. J. Kreuzfeld, J. Organomet. Chem., 1988, 344, 249; (e) H. J. Kreuzfeld, Chr. Döbler and H.-P. Abicht, J. Organomet. Chem., 1987, 336, 287.
- 225. K. Tamao, T. Hayashi, H. Matsumoto, H. Yamamoto and M. Kumada, Tetrahedron Lett., 1979, 23, 2155.
- 226. T. Hayashi, M. Konishi, M. Fukushima, K. Kanehira, T. Hioki and M. Kumada, J. Org. Chem., 1983, 48, 2195.
- 227. T. Hayashi, M. Konishi, Y. Okamoto, K. Kabeta and M. Kumada, J. Org. Chem., 1986, 51, 3772.
- 228. T. Hayashi, K. Hayashizaki, T. Kiyoi and Y. Ito, J. Am. Chem. Soc., 1988, 110, 8153.
- 229. T. Hayashi, K. Hayashizaki and Y. Ito, Tetrahedron Lett., 1989, 30, 215.
- (a) G. Consiglio, O. Piccolo, L. Roncetti and F. Morandini, *Tetrahedron*, 1986, 42, 2043, and refs. cited therein;
 (b) T. Hiyama and N. Wakasa, *Tetrahedron Lett.*, 1985, 26, 3259.
- 231. (a) T. Hayashi, M. Konishi and M. Kumada, J. Chem. Soc., Chem. Commun., 1984, 107; (b) E. Keinan and Z. Roth, J. Org. Chem., 1983, 48, 1769. It has been proposed that this differing stereochemical behavior be used as a system for nucleophile classification.
- 232. R. A. W. Johnstone and W. N. McLean, Tetrahedron Lett., 1988, 29, 5553.
- (a) E. Wenkert, E. L. Michelotti, C. S. Swindell and M. Tingoli, J. Org. Chem., 1984, 49, 4894; (b) E. Wenkert, V. F. Ferreira, E. L. Michelotti and M. Tingoli, J. Org. Chem., 1985, 50, 719.
- 234. S. Wadman, R. Whitby, C. Yeates, P. J. Kocienski and K. Cooper, J. Chem. Soc., Chem. Commun., 1987, 241.
- (a) V. Fiandanese, G. Marchese, G. Mascolo, F. Naso and L. Ronzini, *Tetrahedron Lett.*, 1988, 29, 3705; (b)
 F. Naso, *Pure Appl. Chem.*, 1988, 60, 79, and refs. cited therein.
- 236. V. Fiandanese, G. Marchese, F. Naso, L. Ronzini and D. Rotunno, Tetrahedron Lett., 1989, 30, 243.
- 237. I. Erdelmeier and H.-J. Gais, J. Am. Chem. Soc., 1989, 111, 1125.
- Similar α-metallation of alkenyl sulfoxides with organocuprates and Grignard reagents has been reported. G. H. Posner, P. W. Tang and J. P. Mallamo, *Tetrahedron Lett.*, 1978, 3995.
- (a) E. Negishi, T. Takahashi, S. Baba, D. E. Van Horn and N. Okukado, J. Am. Chem. Soc., 1987, 109, 2393;
 (b) E. Negishi, Acc. Chem. Res., 1982, 15, 340.
- 240. The turnover numbers for reactions of (E)-1-octenyl metals with phenyl iodide in the presence of Pd(PPh₃)₄ in THF at room temperature were 2 (Al), 3 (Zr) and ≥2000 (Zn). See ref. 239a.
- 241. (a) E. Negishi, L. F. Valente and M. Kobayashi, J. Am. Chem. Soc., 1980, 102, 3298; (b) M. Kobayashi and E. Negishi, J. Org. Chem., 1980, 45, 5223.
- 242. Y. Tamaru, H. Ochiai, T. Nakamura and Z. Yoshida, Angew. Chem., Int. Ed. Engl., 1987, 26, 1157.
- 243. E. Negishi and K. Akiyoshi, Chem. Lett., 1987, 1007.
- 244. For reviews, see. (a) A. Suzuki and R. S. Dhillon, Top. Curr. Chem., 1986, 130, 23; (b) A. Suzuki, Top. Curr. Chem., 1983, 112, 67; (c) A. Suzuki, Acc. Chem. Res., 1982, 15, 178.
- 245. (Z)-1-Alkenylboronates are prepared by hydroboration of 1-bromo-1-alkynes with dibromoborane/dimethyl sulfide complex, followed by treatment of the intermediate (Z)-(1-bromo-1-alkenyl)boronates with potassium triisopropoxyborohydride. H. C. Brown and T. Imai, Organometallics, 1984, 3, 1392.

- 246. N. Miyaura, M. Satoh and A. Suzuki, Tetrahedron Lett., 1986, 27, 3745.
- 247. M. Satoh, N. Miyaura and A. Suzuki, Chem. Lett., 1986, 1329.
- 248. S. Hyuga, Y. Chiba, N. Yamashina, S. Hara and A. Suzuki, Chem. Lett., 1987, 1757.
- 249. M. Satoh, T. Ishiyama, N. Miyaura and A. Suzuki, Bull. Chem. Soc. Jpn., 1987, 60, 3471.
- 250. J.-M. Fu, M. J. Sharp and V. Snieckus, Tetrahedron Lett., 1988, 29, 5459.
- 251. M. A. Siddiqui and V. Snieckus, Tetrahedron Lett., 1988, 29, 5463.
- 252. N. Miyaura, T. Ishiyama, H. Sasaki, M. Ishikawa, M. Satoh and A. Suzuki, J. Am. Chem. Soc., 1989, 111, 314.
- 253. J. Uenishi, J.-M. Beau, R. W. Armstrong and Y. Kishi, J. Am. Chem. Soc., 1987, 109, 4756.
- 254. For reviews, see: (a) J. K. Stille, Angew. Chem., Int. Ed. Engl., 1986, 25, 508; (b) J. K. Stille, Pure Appl. Chem., 1985, 57, 1771.
- 255. For a review of the use of vinyl triflates for alkene synthesis, see ref. 99.
- 256. (a) J. E. McMurry and W. J. Scott, Tetrahedron Lett., 1983, 24, 979; (b) M. E. Krafft and R. A. Holton, Tetrahedron Lett., 1983, 24, 1345.
- 257. Bis(triphenylphosphine)dichloropalladium [PdCl₂(Ph₃P)₂] is a particularly effective catalyst for these substrates, especially in reactions with alkyltin reagents. A. M. Echavarren and J. K. Stille, J. Am. Chem. Soc., 1988, 110, 4051.
- 258. W. J. Scott and J. K. Stille, J. Am. Chem. Soc., 1986, 108, 3033.
- 259. (a) Y. Yang, A.-B. Hörnfeldt and S. Gronowitz, Synthesis, 1989, 130; (b) J. K. Stille and J. H. Simpson, J. Am. Chem. Soc., 1987, 109, 2138.
- 260. Arylzinc reagents couple more effectively with vinyl electrophiles. For examples, see: (a) R. McCague, Tetrahedron Lett., 1987, 28, 701; (b) R. B. Miller and M. I. Al-Hassan, J. Org. Chem., 1985, 50, 2121.
- 261. A notable exception to this rule is α -bromo- γ -butyrolactone, which provides high yields of coupled products in reactions with tributylallyltin and tributylacetonyltin. J. H. Simpson and J. K. Stille, J. Org. Chem., 1985, 50, 1759.
- 262. L. Castedo, A. Mouriño and L. A. Sarandeses, Tetrahedron Lett., 1986, 27, 1523.
- 263. A. M. Echavarren and J. K. Stille, J. Am. Chem. Soc., 1988, 110, 4051.
- 264. V. Farina, S. R. Baker and C. Sapino, Jr., Tetrahedron Lett., 1988, 29, 6043.
- 265. J. K. Stille and B. L. Groh, J. Am. Chem. Soc., 1987, 109, 813.
- 266. For recent synthetic applications, see: (a) C. J. Forsyth and J. C. Clardy, J. Am. Chem. Soc., 1988, 110, 5911; (b) C. Chan, P. B. Cox and S. M. Roberts, J. Chem. Soc., Chem. Commun., 1988, 971.
- 267. (a) A. Dondoni, G. Fantin, M. Fogagnolo, A. Medici and P. Pedrini, Synthesis, 1987, 693; (b) Y. Yamamoto, Y. Azuma and H. Mitoh, Synthesis, 1986, 564; (c) T. R. Bailey, Tetrahedron Lett., 1986, 27, 4407.
- D. R. McKean, G. Parrinello, A. F. Renaldo and J. K. Stille, J. Org. Chem., 1987, 52, 422.
 M. E. Krolski, A. F. Renaldo, D. E. Rudisill and J. K. Stille, J. Org. Chem., 1988, 53, 1170.
- 270. W. J. Scott, J. Chem. Soc., Chem. Commun., 1987, 1755.
- 271. F. K. Sheffy, J. P. Godschalx and J. K. Stille, J. Am. Chem. Soc., 1984, 106, 4833.
- 272. (a) A. M. Echavarren, D. R. Tueting and J. K. Stille, J. Am. Chem. Soc., 1988, 110, 4039; (b) D. R. Tueting, A. M. Echavarren and J. K. Stille, Tetrahedron, 1989, 45, 979.
- 273. (a) Y. Hatanaka and T. Hiyama, J. Org. Chem., 1988, 53, 918; (b) Y. Hatanaka and T. Hiyama, Tetrahedron Lett., 1988, **29**, 97.
- 274. Y. Hatanaka and T. Hiyama, J. Org. Chem., 1989, 54, 268.

1.6 Alkylations of Vinyl Carbanions

DAVID W. KNIGHT University of Nottingham, UK

1.6.1	INTRODUCTION	241
1.6.2	ALKYLATIONS USING ALKYL HALIDES AND SULFONATES	242
1.0 1.0 1.0 1.0	 6.2.1 Alkylations of Vinyl and Aryl Grignard Reagents 6.2.2 Alkylations of Vinyl- and Aryl-lithium Species 6.2.3 Alkylations of Heteroatom-substituted Vinyl Carbanions 6.2.4 Alkylations of Allene Carbanions 6.2.5 Alkylations by an S_N2' Process 	242 247 252 256 257
1.6.3	ALKYLATIONS WITH LESS CONVENTIONAL ELECTROPHILES	258
1.6.4	ALKYLATIONS OF VINYL AND ARYL CARBANIONS, RM, WHERE M \neq Li, Mg or Cu	259
1.6.5	ALKYLATIONS OF HETEROAROMATIC CARBANIONS	260
1.6.6	ALKYLATIONS AT sp ² -CENTERS USING EPOXIDES	262
1.6.7	REFERENCES	266

1.6.1 INTRODUCTION

The scope of this chapter is limited to carbon–carbon bond forming reactions between sp^2 -centered carbanions and alkyl halides, alkanesulfonates, epoxides and related carbon centered electrophiles. It does not include reactions between vinylic carbanions and electrophiles such as silyl chlorides, disulfides or other heteroaromatic electrophiles. Some members of this last group of electrophiles are often used to determine the extent to which a particular vinyl carbanion intermediate is formed; having shown that formation has occurred to a large or complete extent, many authors do not then proceed to examine the alkylation reactivity of a particular carbanionic species beyond using methyl iodide. Methyl iodide is such a special electrophile, however, that successful alkylations with it cannot usually be used as an indication that alkylations with higher homologs or with more reactive alkylating agents such as allylic or benzylic halides, will necessarily be so rewarding. Therefore, many potentially useful sp²-carbanionic species are not mentioned in this review, specifically because their behavior with such alkylating reagents has not been reported. In general, the nucleophilicity of vinyl carbanions lies in between similar sp- and sp^3 -centered species, and as such they usually react very well with aldehydes and ketones, and with the simple heteroaromatic electrophiles of the type mentioned above, whereas alkylation reactions give variable yields, depending upon the particular species involved. The high yields normally obtained with reactive electrophiles such as Me₃SiCl and MeI are often not extended to couplings with allylic or benzylic halides, despite the fact that satisfactory results may be obtained using saturated alkyl halides as electrophiles. Presumably in many cases, the carbanion acts as a base rather than a nucleophile and simply abstracts a proton from these more reactive electrophiles.

This review is organized primarily on the basis of the electrophiles used, with a view to allowing easy access to the information. However, within each section, a variety of routes to the same target type are included, as for example in alkylations using simple alkyl halides where both lithium- or magnesium-based carbanions can be used equally well. Both subsections should therefore be consulted.

1.6.2 ALKYLATIONS USING ALKYL HALIDES AND SULFONATES

1.6.2.1 Alkylations of Vinyl and Aryl Grignard Reagents

The most direct method for effecting alkylation of a vinylmagnesium halide is simply to heat the organometallic species with a primary alkyl bromide at 90-100 °C (Scheme 1).¹ The Grignard reagent (a slight excess) is usually generated in tetrahydrofuran (THF) solution, the bromoalkane added and the THF removed by distillation until the required temperature is reached. However, the more reactive allyl halides can be used to alkylate Grignard reagents under milder conditions. Indeed the ease with which allylic halides undergo such couplings can lead to low yields during the attempted preparations of allylic or benzylic magnesium halides. In a nonpolar solvent such as toluene, temperatures of around 100 °C are still required to give good yields of alkylated products such as the phenylpropene (1) (from PhMgBr and 1,3-dichloropropene).² However, similar alkylations of arylmagnesium halides using allylic chlorides or bromides occur at much lower temperatures (15-30 °C) in ether³ or THF,⁴ and also lead to good to excellent yields of allylbenzene derivatives. Despite their generally greater popularity as carbanionic intermediates in alkylation reactions, many sp^2 -centered carbanions having lithium as the countercation do not react efficiently with allylic halides (vide infra); in such cases, transmetallation to the corresponding magnesio species can sometimes obviate this limitation. One such example is the ortho-lithiated benzamide (2), which reacts cleanly with allyl bromide after treatment with magnesium bromide to give the o-allyl homolog (3) in good yield.⁵



Scheme 1



Another series of highly active electrophiles which react smoothly with vinyl and aryl Grignard reagents to give, overall, the products of direct alkylation, are the α -halo ketones; however, this type of reaction is a two-step process which proceeds *via* an intermediate halohydrin arising from initial attack on the ketone group (Scheme 2).⁶ Similarly, vinylmagnesium halides can be transformed in a single flask reaction to α -ethenyl ketones, *e.g.* (4),⁷ whereas vinyllithium species lead to isolable halohydrins, *e.g.* (5), which only rearrange to the α -ethenyl ketone upon reionization using a simple Grignard reagent as base.⁸ The relative dearth of good vinyl cation equivalents lends particular significance to the latter methodology. α -Halo ethers and related species also couple in moderate to good yields with Grignard reagents,⁹ an example being the reactions between 2-chloro-1,3-dithiane and RMgX (Scheme 3) in which the typical umpolung reactivity of the dithiane functionality is returned to normality.¹⁰ The special reactivity of α -halo ethers means that good yields can be obtained by this method, despite the fact that secondary halides generally react poorly as alkylating electrophiles with organomagnesium and organolithium intermediates.



242





An indirect way to effect alkylation at an sp^2 -carbanionic center is by reaction with a trialkylborane (Scheme 4).¹¹ Decomposition of the intermediate borate results in overall alkylation of the original Grignard species by $B \rightarrow C$ alkyl migration; one drawback is that two of the alkyl groups are wasted and hence applications would probably be somewhat limited.



In general, however, alkylations of both organomagnesium and organolithium intermediates (vide infra) more often rely on the presence of a copper salt, either in catalytic or stoichiometric quantities, or another transition metal species.¹² A general method is the alkylation of vinylmagnesium bromides by primary iodides or tosylates (Scheme 5)¹³ in the presence of copper(I) iodide, which proceeds under much milder conditions than those required for the direct alkylation of vinyl Grignards by unactivated alkyl halides.¹ Less reactive alkyl bromides give much poorer yields (ca. 20%) and it is crucial that the copper catalyst is of a high quality. An alternative copper catalyst, Li₂CuCl₄ (vide infra) can also be used, but excess Grignard reagent is required if good yields are to be realized. The addition of copper salts to Grignard reagents gives rise to some useful modulations in the reactivity of the latter. For example, direct coupling between RMgX and 2-bromoethyl acetate is possible, leading to homologs (7),¹⁴ and even bromoethanol (8) can be used to give phenethyl alcohols (9)¹⁴ in this modification of an original Grignard method.¹⁵ A useful source of vinyl magnesiocuprates is by the addition of alkylmagnesium halides to terminal alkynes, in the presence of stoichiometric amounts of a copper(I) salt; the usefulness of these reagents is indicated in Scheme 6 in which is shown an early stage in a synthesis of the insect pheromone trogodermal.¹⁶ However, other authors¹⁷ have indicated that closely related species derived from a monosubstituted alkyne can only be alkylated efficiently by allylic halides or epoxides (see Section 1.6.6). Much the same pattern of reactivity has been found with related lithium divinylcuprates similarly derived from alkynes (vide infra). An alternative catalyst system, CuX-TMEDA, has been shown to be useful for the alkylation of arylmagnesium halides, ArMgX, by primary iodides or benzylic chlorides.¹⁸ Yields are usually around 80% but, as with most of these alkylations, the method fails with a secondary iodide.

$$MgBr + R X \frac{CuI}{80-97\%} R$$

R = Et, $n-C_7H_{15}$; X = I or OTs



Scheme 6

Dilithium tetrachlorocuprate, Li₂CuCl₄, sometimes referred to as the Tamura–Kochi complex, has also found many applications in the alkylation of Grignard reagents in general.¹⁹ Although originally reported to work best with primary alkyl iodides,¹⁹ subsequent research indicated that primary tosylates are the best electrophiles.²⁰ The use of Li₂CuCl₄ allows alkylations of Grignard reagents by α,ω -dibromides leading to monobromides, *e.g.* (10) from PhMgBr, to be effected in moderate to good yields.²¹ Similar couplings of aryllithiums require the presence of TMEDA (*cf.* ref. 18). Li₂CuCl₄ has also been found to be the catalyst of choice for conversions of the dienylmagnesium chloride (11) into the butadiene homologs (12).²²



Palladium and nickel catalysts also occupy an important position in this type of chemistry. The readily prepared palladium complex $[(Ph)Pd^{I}(PPh_{3})_{2}]$ effectively catalyzes the cross-coupling of arylmagnesium halides and alkyl halides, even when the former is sterically hindered, and is reported to be superior to many related nickel-based catalysts.²³ A ferrocene-palladium complex can also be used.²⁴ A combination of a nickel catalyst [Ni(dppp)Cl₂] and t-butyl 2-bromopropionate (rather than the methyl ester)²³ allows direct coupling of the latter with ArMgX to provide respectable yields of 2-arylpropionates (13).25 α -Silylvinylmagnesium bromides (14), prepared in general by Ni-catalyzed addition of an alkyl Grignard to a silylalkyne, undergo in situ, presumably nickel-catalyzed, alkylations by methyl iodide (50%) and allyl bromide, the latter leading to the potentially useful 'skipped' dienes (15).²⁶ The simpler α -silvl species (16) couples efficiently with primary alkyl iodides (but not primary bromides or tosylates) in the presence of a copper(I) salt to give good yields of homologs (17); both α - (16) and β -silvlvinyl Grignards can also be alkylated using a palladium catalyst.²⁷ As expected,^{3,4} alkylations using allylic halides do not require the addition of a catalyst. Alkoxide residues are tolerated in such reactions; thus hydromagnesiation of the silylpropargyl alcohol (18), followed by treatment with iodobutane-CuI leads to the allylic alcohol (19) with 95% retention of stereochemistry.²⁸ Such vinyl Grignard reagents can also be obtained by intramolecular cyclizations; again, homologation using an allylic halide proceeds without recourse to a catalyst (Scheme 7).29







Copper-modified Grignard reagents can also be alkylated using β -lactones, resulting in a useful threecarbon homologation procedure in which attack occurs regioselectively with displacement of carboxylate (Scheme 8).³⁰ In general, reagents prepared using a stoichiometric quantity of copper give higher yields, based on the lactone. The propensity for undergoing conjugate or S_N2' additions imparted upon Grignard reagents by the presence of copper allows this methodology to be extended to ω -ethenyl lactones; thus, reactions between a variety of such species and β -lactone (**20**) afford good to excellent yields of the dienoic acids (**21**; R = CH₂CH, 49%; R = Ph, 84%).³¹ This approach can also be extended to include 5-ethenylbutyrolactone and the corresponding valerolactone (**22**). Once again, good to excellent yields of dienoates (**23**) can be obtained, especially with the former substrate (*i.e.* n = 1).³²



Alkylation of Carbon

These unsaturated lactones are special cases of a much more general theme whereby Grignard reagents can be alkylated by a wide variety of allylic systems, usually in the presence of a nickel- or copper-based catalyst. Arylmagnesium bromides react in an S_N2' fashion with allylic acetates and a nickel catalyst; when the latter is chiral, allylaryls of up to 89% enantiomeric enrichment can be obtained $(24 \rightarrow 25)$.³³ Copper catalysts, such as Li₂CuCl₄, can also be employed in such displacements, but usually *via* a direct S_N2 mechanism.³⁴ Similarly, primary allylic phosphates react with Grignard reagents under copper(I) catalysis exclusively *via* S_N2 displacement; usually no products from allylic rearrangement are observed.³⁵ Chiral allylaryl compounds (25) can also be prepared by alkylations of ArMgX using an allyl phenyl ether and a chiral Ni-phosphine catalyst. Optical yields are fair to good but a more serious problem is poor regioselectivity.³⁶ Much more regioselective are the related reactions with allyl silyl ethers (26); with a ferrocenepalladium(II) catalyst, either isomer of the ether gives very largely the internal alkenes (27) when treated with ArMgX, whereas nickel(II) catalysts lead mainly to the isomeric terminal alkenes (28; Scheme 9).³⁷



Scheme 9

In contrast, a whole range of allylic systems, including allylic alcohols, undergo nickel-catalyzed coupling reactions with aryImagnesium bromide to give, usually exclusively, primary allylaryl compounds uncontaminated by the corresponding conjugated isomers (Scheme 10).³⁸ The reactions proceed via an ω -allylnickel complex and require an excess, typically 4 equiv., of the Grignard reagent in examples of electrophiles with a labile proton (e.g. allyl alcohol). Allylic alcohols can be activated for this type of coupling by conversion to the iminium salts (29); subsequent coupling with Grignard reagents then occurs directly via $S_N 2$ displacement (Scheme 11).³⁹ If, however, the Grignard reagent is treated with copper(I) iodide and HMPA prior to the coupling step, the alternative, $S_N 2'$, products are usually formed. In a related sequence, exclusively $S_N 2'$ products (31) are formed when 2-allyloxybenzothiazoles (30) are reacted with Grignard reagents in the presence of copper(I) bromide.⁴⁰ Finally, the modified aryl Grignard species, PhCu·MgBr₂·LiBr, react in an S_N2' fashion with allenic bromides (e.g. 32) to give very largely alkynes (e.g. 33) contaminated with very little of the corresponding allene.⁴¹ Some less conventional electrophiles have also been used in efficient alkylations of Grignard reagents. For example, benzylamines (e.g. 34) are available from alkylations of PhMgBr using either alkoxymethyl- or phenvlthiomethyl-amines [RXCH2NR2];42 similar products are obtainable from reactions between Grignard reagents or organolithium species and Eschenmoser's salt.⁴³ Finally, Grignard reagents can be efficiently allylated using allyltrialkylammonium salts in the presence of Li₂CuCl₄.⁴

ArMgBr + R X $(PPh_3)_2NiCl_2$ R Ar

X = OH, OMe, OEt, SH or SMe

Scheme 10


1.6.2.2 Alkylations of Vinyl- and Aryl-lithium Species

The simplest possible reaction in this section, that between a vinyllithium and a primary alkyl halide in general works extremely well (Scheme 12).⁴⁵ The corresponding alkyl iodide can equally well be used, but the reaction fails with secondary halides or primary tosylates and is much less satisfactory with alkyl chlorides or when performed in other ethereal solvents. ω -Bromoalkenes can even be obtained from 1, ω dibromoalkanes in good yields;^{18,21,46} virtually no alkene isomerization is observed and therefore both trans- and cis-alkenes are available using this method. 1-Alkylcycloalkenes (e.g. 35) as well as acyclic trisubstituted alkenes can also be prepared in this way;⁴⁶ an alternative strategy is to treat the vinyl bromide with an alkyllithium at -70 °C thus generating the vinyllithium and alkyl bromide by halogen-metal exchange. Upon warming to room temperature, alkylation proceeds smoothly.⁴⁶ However, this latter method is limited by the availability of alkyllithiums. A way around this is to treat the vinyl halide with 2 equiv. of Bu'Li, the extra equivalent of base serving to trap the t-butyl halide produced.⁴⁷ An excellent alternative is to employ tin-lithium exchange reactions between a vinyltrialkylstannane and an alkyllithium, usually MeLi or BuⁿLi, which results in the formation of an essentially inert tetraalkyltin by-product. This method has been used, for example, to generate the useful cyclopentenyllithium (36) which, although it couples efficiently with reactive electrophiles such as allyl bromide and benzyl chloromethyl ether, is reported to react poorly (ca. 30% yields) with primary alkyl iodides.⁴⁸ An extremely useful reagent in this respect is trans-1,2-bis(tri-n-butylstannyl) ethylene, which, upon treatment with BuⁿLi (THF, -78 °C), affords the vinyllithium (37), alkylation of which with a range of alkyl halides is reported to be very efficient.⁴⁹ The initial products, being vinylstannanes, can clearly be further homologated by the same methodology. An example of this type of strategy is to be found in the approach to 5,6-dehydroarachidonic acid outlined in Scheme 13.50 Additional features of note in this scheme are the efficient alkylations of the intermediate vinyllithiums by allylic acetates and the $S_N 2'$ -like reaction⁴¹ with an allenic bromide.



Scheme 12





Further examples of useful vinyllithium intermediates which have been generated using tin-lithium exchange include the substituted butadiene precursors (38),⁵¹ (39),⁵² and (40),⁵² and the cyclohexadiene (41);⁵³ all four intermediates react efficiently (60–86% yields) and directly with a range of primary alkyl bromides or iodides. Such alkylations can also be performed intramolecularly, as for example in a preparation of the hydrindane (43) from the vinylstannane (42).⁵⁴



A very popular and often complementary alternative to the foregoing vinyllithium chemistry is the use of lithium divinyl- or diaryl-cuprate species, R₂CuLi, which also can be alkylated using primary alkyl bromides or iodides but often only in the presence of various additives, typically a stabilizing phosphine ligand and the correct solvent. For example, lithium divinylcuprate is alkylated by *n*-octyl iodide, after treatment with tri-*n*-butyl phosphine, to give 1-decene in 91% yield.⁵⁵ In contrast to the simple aryllithium species, Ph₂CuLi reacts smoothly with a secondary alkyl bromide to give an alkylbenzene with 84–92% inversion of stereochemistry, as expected of an S_N 2 process.⁵⁵ Alkylations of this type generally result in almost complete retention of the original alkene geometry.⁵⁶

Two key differences in reactivity between vinyl- or aryl-lithiums and the corresponding R_2CuLi species are firstly that the cuprates couple efficiently (and more rapidly than with alkyl halides) with both primary and secondary tosylates,^{55,57} which the lithiums do not,⁴⁵ and secondly the general lack of

reactivity between the cuprates and carbonyl functions such as those present in ketones, esters and acetates.^{14,57} Thus, for example, cuprate species react with bromoethyl tosylate (44) to give excellent yields of the bromides (45),¹⁴ and the cuprate (46) reacts smoothly with an acetoxy iodide leading to the acetoxydiene (47).⁵⁸



A major drawback in the use of homocuprates, R₂CuLi, is that one of the alkenyl or aryl groups can often be wasted; clearly, however, this is not a serious problem if the cuprate is derived from a simple organolithium. With reactive electrophiles such as allylic halides or α -halo ethers, both organic ligands are alkylated in the presence of 1 equiv. of HMPA, but with less reactive alkyl halides, at least 3 equiv. of (EtO)₃P are required to stabilize the intermediate cuprate species during the much slower transfer of the second alkenyl ligand.⁵⁹

Further extensions of this general methodology include a useful (E,Z)-diene synthesis (Scheme 14)⁶⁰ and an approach to (Z)-alkenols (Scheme 15).⁶¹ Unfortunately such sequences are ineffective when applied to monosubstituted alkynes, a limitation which also applies to the corresponding alkenylmagnesiocuprates.¹⁷ These intermediates also react smoothly with allylic halides to provide 1,4-dienes, especially when dimethyl sulfide is present as an extra ligand,⁶² which is fortunate as some of the most convenient sources of copper(I) salts having the high purity requirements for this type of chemistry are the sulfide complexes CuX·SMe₂ (X = Br or I).⁶³ One such allylation reaction forms a key step in an approach to α trans- and β -trans-bergamotene (Scheme 16);⁶⁴ an additional noteworthy feature is that the cuprate species does not suffer from β -elimination of the amino function prior to alkylation.



Alternative methods for obtaining these types of cuprates, R_2CuX (where R = alkenyl or aryl), include treatment of RBr with lithium metal and copper(I) iodide under ultrasonication,⁶⁵ reaction between a dialkylvinylborane and CuBr·Me₂S⁶⁶ or between a chlorodivinylborane and methylcopper.⁶⁷ The latter

two intermediates react efficiently with allyl bromide to provide 1,4-dienes, but alkylations by simple alkyl halides require the addition of an extra ligand such as a phosphine or phosphonite, usually in considerable excess. A significant side reaction can be dimerization of the intermediate cuprates. Trialkyl-borons can also be used to alkylate vinyllithiums and related species.⁶⁸

A neat combination of some of the foregoing chemistry is use of the cyclic stannane (48) as an equivalent of the 1,4-pentadienyl dianion (49).⁶⁹ Thus, treatment of the stannane (48) with BuⁿLi, copper(I) iodide-dimethyl sulfide complex and the iodoallene (50) smoothly leads to the dienyne (51) and thence, by a repetition of this sequence but using electrophile (52), to 3-dehydroarachidonic acid methyl ester (53).



A valuable method whereby the wastage of one organic ligand in R₂CuLi species can be avoided is to employ mixed cuprates R¹R²CuLi, where R² is a nontransferable ligand, such as an acetylide, cyanide or benzenethiolate. For example, conversion of the vinylstannane (54) into the corresponding vinyllithium followed by the addition of 1-pentynylcopper gives a mixed cuprate intermediate (55), which is then alkylated by transfer of only the vinyl ligand (Scheme 17).⁷⁰ Similarly, the mixed cuprate (56; cf. ref. 49) can be alkylated using 3-bromomethylfuran to provide the stannane (57), ready for further homologation.⁷¹ A further important development in this area has been the introduction of 'higher order' cuprates in which various stoichiometries of the vinyl- or aryl-lithium are mixed with copper salts.^{12,72} Particularly of note is the ability of the higher order cuprate (58) to couple efficiently with an *s*-alkyl iodide (Scheme 18);⁷³ in contrast, the corresponding lithium divinylcuprate gives only a 23% yield. The analo-



Scheme 17



Scheme 18

gous aryl species, Ph₂Cu(CN)Li₂, is not alkylated efficiently by secondary iodides but does react extremely efficiently with primary chlorides and especially bromides or tosylates⁷⁴ (the simpler species, Ph₂CuLi, also couples efficiently with both primary and secondary bromides and tosylates).^{55–57} An excellent illustration of the power of vinyl- and aryl-cuprates in synthesis is the finding that such intermediates couple smoothly with the halides (**59**; X = Br or I) to provide a racemization-free approach to chiral non-natural α -amino acids (Scheme 19).⁷⁵ As well as the regioselectivity of attack, it is especially noteworthy that no elimination to give dehydroalanine occurs with the lower homologs (**59**; *n* = 1). An excellent and convenient reagent for the preparation of mixed higher-order cuprates (R¹R²Cu(CN)Li₂) is the shelf-stable, commercially available complex (2-thienyl)Cu(CN)Li; the cuprates formed from this species and R'Li react smoothly with primary alkyl iodides with transfer of only R', the 2-thienyl ligand being essentially inert.⁷⁶



Scheme 19

Alkylations of vinyl carbanions can also be carried out intramolecularly to provide three-, four-, fiveand six-membered rings (see also refs. 99 and 100). For example, treatment of the dibromides (**60**) with Bu'Li at low temperature results in selective bromine–lithium exchange at the sp^2 -center; on warming 'Parham-type' cyclization occurs, thus providing a useful route to substituted benzocyclobutanes (**61**).⁷⁷ Similarly, the vinyl iodide (**62**), upon treatment with BuⁿLi (-78 °C \rightarrow +20 °C) smoothly cyclizes to give an 86% yield of the ylidenecyclopentane (**63**), with no scrambling of the alkene stereochemistry.⁷⁸ Similar approaches to related cyclohexane systems also work well.⁷⁸ Another useful method for generating vinyllithium species is the Shapiro reaction (such intermediates can be alkylated reasonably efficiently by primary alkyl bromides);⁷⁹ this methodology can also be applied intramolecularly (Scheme 20).⁸⁰ Presumably, in some cases, the intermolecular version could be improved by the addition of various ligands such as HMPA or (EtO)₃P (vide supra).





Scheme 20

1.6.2.3 Alkylations of Heteroatom-substituted Vinyl Carbanions

The presence of a heteroatom either α or β to an *sp*²-carbanion opens up many additional synthetic possibilities. Furthermore, their presence can often modify the reactivity of the intermediate and allow its generation by direct hydrogen–lithium exchange rather than halogen–lithium exchange, thus contributing an extra degree of simplicity to the reaction sequences. An excellent illustration of these features is the generation of vinyllithium (64) by treatment of methyl vinyl ether with Bu^tLi.⁸¹ Formally an acyl anion equivalent, this intermediate can be alkylated efficiently by primary alkyl iodides or primary allylic bromides, although not by benzylic bromides, which instead react *via* halogen–lithium exchange. The corresponding cuprate, however, does couple smoothly with benzylic bromides as well as with secondary allylic bromides.^{81,82} The vinyllithium (64) can also be generated using Sn–Li exchange and BuⁿLi,⁸³ while the higher homolog (65) can be generated directly only in the presence of the MOM function using BuⁿLi–TMEDA.⁸⁴ Cyclic analogs of the vinyllithium (64) have also proven useful in a variety of synthetic schemes, and can often be alkylated in high yield; for example metallation of 2,3-dihydropyran affords the vinyllithium species (66), which, upon alkylation using a protected ω -iodo alcohol (67), followed by acidification affords good yields of the spiro-acetals (68).⁸⁵

As expected, related sulfur species can also be generated and used to good effect, an example being the vinyl sulfide (69), which can be alkylated with good efficiency (*e.g.* BuⁿI, 68% yield).⁸⁶ Similarly the β -lithio species (70) reacts satisfactorily with, at least, methyl iodide (99%) and allyl bromide (68%). It





should be noted in this context that methyl iodide is a very special electrophile, often more suited to assaying the extent of anion generation rather than to synthetic utility. A much better indicator, and by no means a trivial change, is to examine the reactions of a new carbanion with ethyl or *n*-butyl iodides; good yields from these electrophiles bode well for the general utility of the intermediate. For instance, the α thio enamine (71) is reported to give quantitative yields with both methyl and ethyl iodide;⁸⁷ the latter result is of much more significance. The related sulfones also show some promise as intermediates. Generated from either the acetal (72) by sequential E_2 elimination and deprotonation⁸⁸ or by direct deprotonation of either the (*E*)- or (*Z*)-vinyl sulfone,⁸⁹ the intermediates (73) can be alkylated using BuⁿI (76%), or benzyl bromide (60–77%), thus indicating at least a reasonably wide utility. The homologous acetal (74) has also been reported, although it has only been alkylated with MeI (96%) and with allyl bromide (81%).⁹⁰ Similarly, the vinyl selenide (75) can be obtained and reacts with *n*-octyl bromide to give the expected homolog in 69% yield,⁹¹ indicating a good range of reactivity. These examples are relatively typical in that full ranges of electrophiles are often not examined (or at least not reported), and it is therefore sometimes difficult to judge the likely overall importance of a particular intermediate.



Despite the presence of a potential β -leaving group, the α -lithio acrolein derivative (76)⁹² can be obtained at low temperatures but only condenses efficiently with primary allylic halides after conversion to the corresponding mixed cuprate using PhSCu; direct, rather than S_N2' , attack occurs at least with (E)-1bromo-2-butene to give the functionalized 'skipped' diene (77). With many other intermediates, the problem of $S_N 2 vs. S_N 2'$ attack is often only resolved by experimentation using the desired electrophile. The vinvl ortho ester (78) can be lithiated by halogen-lithium exchange and subsequently alkylated to provide the homologs (79) via a synthetic equivalent of β -lithio acrylate.⁹³ This example is reasonably typical of many such alkylations of lithiovinyl carbanions which work well with methyl iodide (and other reactive and simple electrophiles such as R₃SiCl, PhSSPh or CO₂), but are not so useful with homologous primary iodides. Vinyl carbanions can be generated α to an unprotected ester group in certain cases by Sn-Li exchange. For example, treatment of α,β -unsaturated esters (80) sequentially with BuⁿLi and RX leads to the homologs (81); once again, a typical pattern emerges in that methyl iodide and primary allylic halides afford high yields, whereas 3-chloro-1-iodopropane gives only a 42% yield of the homolog (81; $R^1 = (CH_2)_3Cl$).⁹⁴ Similarly, the β -stannyl analog (82) can be obtained; this intermediate exhibits essentially the same characteristics.⁹⁵ Subsequent conversion of the initial products arising from alkylations of the ester (82) into the allylic alcohol (83) (effectively the β -lithio analog) followed by alkylation results in a stereospecific approach to tetrasubstituted alkenes. The latter alkylations are somewhat more efficient, as primary alkyl iodides afford 65-72% isolated yields.



Vinyl carbanions positioned α to a silicon substituent have also proven to be valuable synthetic intermediates, which in general undergo highly efficient alkylations by primary alkyl iodides or tosylates as well as by allylic halides, especially when a vinylcuprate species is used. Such intermediates (85) can be generated either by the addition of a copper-modified Grignard reagent to silyl alkyne (84)⁹⁶ or by a stereochemically complementary boron–lithium exchange procedure involving vinylboranes (86) derived from the corresponding silylalkyne.⁹⁷ The related vinyllithium species, obtained by halogen–lithium exchange, are also reported to react efficiently with, at least, primary alkyl iodides.⁹⁸ Such alkylations can also be carried out intramolecularly,^{77–80} a rather spectacular example being the elaboration of silylcyclopropenes (88) upon treatment of allylic chloride (87) with BuⁿLi;⁹⁹ the methodology has been extended to the corresponding unsaturated four-, five- and six-membered rings.¹⁰⁰ The isomeric β -lithio silane (89), generated by Sn–Li exchange, shows little tendency to undergo elimination and can be alkylated using BuⁿBr (81%), indicating a reasonably wide utility for this species.¹⁰¹

In general, *ortho*-substituted aryllithiums give variable results in alkylation reactions, and in some cases only after conversion to a cuprate species. For example, the cuprate (90) can be allylated efficiently and then electrolyzed to provide the protected benzoquinone (91) in excellent overall yield.¹⁰² Coupling





reactions with benzyl bromide are also viable when it appears that, unusually, more than 1 equiv. of the organic ligand in the cuprate (90) is alkylated, and direct allylations of the related naphthoquinone-derived species (92) are also successful, but again only after conversion to the corresponding cuprate.



The oxazoline function is one of the most powerful directing groups in metallation chemistry, and allows the direct generation of the ortho lithio species (93).¹⁰³ These have been used with varying success in alkylation reactions; a particularly useful aspect is as precursors to o-allylbenzaldehydes (94) following allylation and reduction-hydrolysis of the oxazoline function.¹⁰⁴ Alkylations with iodomethyltrimethylsilane also work well, leading to quinodimethane precursors (e.g. 95).¹⁰⁵ However, in common with many nucleophiles, alkylations using homoallylic or propargylic halides fail due to competing deprotonation and overall elimination of HX from the electrophile. A solution to this limitation, which can often be applied in the aromatic and heteroaromatic areas, is first to alkylate the sp^2 -center with methyl iodide (usually a very efficient step), then remetallate to generate a (usually) more reactive sp³-carbanion and finally alkylate with an allylic or propargylic halide. This technique is illustrated in Scheme 21, which forms the early stages of the preparation of enediynes suitable for Vollhardttype cobalt-catalyzed cyclizations.¹⁰⁶ The closely related imidazolidine (96), however, requires an excess of alkylating agent, and even then only reacts well with methyl iodide (95%); with n-butyl bromide only a 45% yield is realized, while alkylation with benzyl bromide gives a 30% yield of the dialkylated product (97), a not uncommon feature with this type of reactive halide.¹⁰⁷ In contrast, the dianionic species (98) derived (BuⁿLi-TMEDA) from benzyl alcohol reacts best with *n*-butyl chloride (55%) and gives only traces of product with primary alkyl iodides; presumably in these cases deprotonation of the iodides by the highly basic dianion takes precedence.¹⁰⁸ A related dianionic species (99) derived from benzaldehyde by sequential hemi-aminal formation using N-lithio-N'-methylpiperazine and deprotonation (BuⁿLi) does react reasonably well with BuⁿBr (47%), but only after the addition of copper(I) iodide.¹⁰⁹ The dianion (100), derived from the tosylhydrazone of benzophenone, provides yet another contrast as it does undergo efficient (>80%) alkylation by primary alkyl halides with no detectable reaction at the potentially competing nitrogen center.¹¹⁰ Related ortho-lithiated aryl isocyanides can be similarly useful.¹¹¹

Finally, the useful cyclohexadienyllithium (102; Scheme 22), obtained from the dioxolane (101) by sequential deprotonation, β -elimination and halogen-metal exchange, can be alkylated using ethyl iodide,





and hence probably by higher, primary homologous iodides, to provide a useful route to 3-substituted cyclohexenones.¹¹²



Scheme 22

It is hoped that the foregoing examples provide an indication of the variety of outcomes, both good and bad, which can result from alkylations of this type of sp^2 -carbanion, together with some useful methods for obtaining good yields. Many anionic species are not included in the discussion simply because their alkylation chemistry has not been reported, except sometimes with methyl iodide which, as discussed above, is not always a typical or representative electrophile in this type of chemistry.

1.6.2.4 Alkylations of Allene Carbanions

Allenes can be directly deprotonated using alkyllithiums, and the intermediate lithio allenes can then be alkylated efficiently by primary alkyl bromides or iodides.¹¹³ For example, good yields of homologated allenes can be realized under the conditions specified in Scheme 23 (the brief reaction time appears

to be crucial) when very little of the alkynic isomer is formed. Similarly, the corresponding lithium diallenylcuprates, obtained either by hydrogen- or halogen-lithium exchange can also be alkylated.¹¹⁴ In common with methyl vinyl ether,^{81,82} methoxyallene can be directly deprotonated, although the resulting intermediate (**103**) gives only moderate to good yields of alkylated products upon treatment with a variety of primary alkyl or allylic halides.¹¹⁵ Once the intermediate (**103**) has been alkylated, the products can be again deprotonated to provide the new lithio allene (**104**), and thence the dialkylated products (**105**), following treatment with a primary alkyl iodide (the addition of a copper(I) species is not necessary); these final educts are useful as precursors to α , β -unsaturated ketones, to which they are converted simply by acidic hydrolysis.¹¹⁶ Much the same double alkylating reagents appears to be more limited.¹¹⁷ The related silyloxy species (**108**), generated by rearrangement of the propargylsilane (**107**), can also be alkylated using a range of primary alkyl halides (71–87%) to give the expected homologs, again useful precursors to α , β -unsaturated ketones.¹¹⁸



1.6.2.5 Alkylations by an S_N2' Process

Many of the foregoing alkylations with allylic halides have only been carried out with symmetrical examples of the latter, and therefore it is not clear whether alkylation occurs *via* an $S_N 2$ or $S_N 2'$ process. However, in common with Michael addition chemistry, vinyl and aryl carbanions generally add in a conjugate $S_N 2'$ fashion in the presence of a copper(I) salt, and in the absence of excessive steric demands.^{33,39-41} Thus, in a typical example, lithium diphenylcuprate adds to the allylic acetate (109), derived from the corresponding α -lithio vinylsilane,⁹⁸ to provide a 61% yield of only the (*E*)-vinyl-silane (110);¹¹⁹ with other examples, the regioselectivity can be poorer.¹²⁰ Often such cuprates react with α -bromo ketones to give only poor yields of α -aryl ketones, a common side or even major pathway being halogen-metal exchange. This problem can be overcome by employing the corresponding enamine (*e.g.* 111), which reacts smoothly with Ph₂CuLi, presumably *via* an $S_N 2'$ displacement, to give, in 87% yield, 2-phenylcyclohexanone after hydrolysis¹²¹ (see also refs. 6–8). However, in cases of extreme steric and/ or electronic demands, vinyllithium species will react in an $S_N 2'$ fashion as illustrated in Scheme 24.¹²² The cyclohexene derivative (112) reacts similarly; these are perhaps not typical examples, despite the overall $S_N 2'$ process, since the first step is a Michael-type addition.



1.6.3 ALKYLATIONS WITH LESS CONVENTIONAL ELECTROPHILES

Some examples of less conventional electrophiles which react well with sp^2 -carbanionic centers are grouped together here mainly for reasons of emphasis and because a variety of organometallic species are involved. Aminomethylations of Grignard reagents, for example, can be effected either using an aminomethyl ether^{42,123} or an amino sulfide (Scheme 25).¹²⁴ The latter electrophiles also react well with (Z)-divinylcuprates (113), derived from alkylcuprate additions to acetylene, to provide a useful route to (Z)-allylamines (114).¹²⁵ Methyleneammonium salts (*e.g.* Eschenmoser's salt)⁴³ can also be applied in this way; for example, the dilithiated aromatic intermediate (115) reacts to form the diamino derivative (116).¹²⁶ Two useful extensions of this are that treatment of the precursor of the dianion (115) with 1 equiv. of base results in regioselective metallation of the furan and hence leads to the aminomethylfuran derivative; not unreasonably therefore, treatment of the dianion (115) with 1 equiv. of the electrophile gives the aminomethylphenyl isomer. Although aminoethylation can be achieved by condensations between (usually *N*-tosyl) aziridines and cuprates (113), much better yields are often obtained using mixed higher-order cuprates, R₂Cu(CN)Li₂.¹²⁷ With less reactive cuprates, the addition of BF₃ can be beneficial.



A variety of Grignard reagents also react with ortho esters and acetals leading to acetals and ethers respectively.¹²⁸ A particularly useful extension of this methodology is in its application to alkylations of divinylcuprates (113).¹²⁹ Thus, the ether (117) can be obtained from heptanal dimethylacetal, the appropriate cuprate and BF₃ in 93% yield; similarly acetaldehyde diethylacetal reacts with PhCu·BF₃ to give an equally good yield of the ether (118). With unsaturated acetals, mainly and sometimes exclusively S_N2' addition takes place; thus acrolein diethylacetal and cuprate (113; R = Buⁿ) modified using BF₃ react to give vinyl ether (119) in 66% yield. At the higher oxidation level, reactions with ortho esters provide a useful method for homologation to the (protected) aldehyde level as, for example, in the coupling of PhCu·BF₃ and triethyl orthoformate to give benzaldehyde diethylacetal (92%).



A somewhat unusual reaction is the coupling of phenyllithium-copper(I) iodide mixtures directly with allylic alcohols, in the presence of the salt Ph₃PNMePh⁺I⁻, allowing a usually stereoselective preparation of hydrocarbons (120).¹³⁰ Similar phenyllithium species can be directly alkylated using diazoacetates to give phenylacetates in moderate (*ca.* 50%) yields,¹³¹ the latter usually being inferior to those produced by related boron-based methodology.⁶ Divinylcuprates (113) also add efficiently to vinylphosphonium salts (*e.g.* Schweizer's reagent) to give ylides (121) and thence 'skipped' dienes (122).¹³²



1.6.4 ALKYLATIONS OF VINYL AND ARYL CARBANIONS, RM, WHERE M ≠ Li, Mg or Cu

Some of the most important members in this group are the alkenylalanes (123), which are readily prepared by the addition of DIBAL-H to a terminal alkyne; however, in general, these can only be alkylated in respectable yields following formation of the corresponding 'ate' complexes, often using *n*-butyllithium.¹³³ Unfortunately, even these more reactive intermediates tend to give really good yields only with allylic bromides, although the very reactive chloromethyl methyl ether alkylates the alanes (123) directly, without the need to form an 'ate' complex, to give the expected ethers in 72-80% yield.¹³⁴ 'Ate' complex formation is also necessary in alkylations of the related alanes derived by the zirconium-catalyzed addition of trimethylaluminum to 1-alkynes.¹³³ Much the same is true of more highly substituted derivatives such as the α -silylalane (125)¹³⁵ or the β -stannyl homolog (126),¹³⁶ both of which react well (68-84%) with allylic halides after conversion to an 'ate' complex, whereas reactions with iodo-ethane or -butane are much less efficient. In contrast, the alkenylaluminum (127), the first example of such a species to contain a free carbonyl group, does alkylate directly with a variety of primary and secondary alkenyl bromides,¹³⁷ although $S_N 2 vs$. $S_N 2'$ attack is a problem with appropriate electrophiles such as crotyl bromide. In a rather different reaction type, the allylic acetate (129) couples smoothly with the alane (128) under the influence of a Pd⁰ catalyst to provide only the product (130) of overall S_N ² attack.138



(126)

(127)

(125)



A related set of conditions (Scheme 26) provides an alternative to 'ate' complex formation, at least in alkylations using benzylic or allylic chlorides.^{139,140} The reverse sense of coupling between benzylzinc species and alkenyl halides is also a viable option. In a similar fashion, arylzincs can be coupled readily with allylic halides (Scheme 27).¹⁴⁰ Both alkenylalanes and zinc species can be similarly coupled with allylic acetates, aluminum alkoxides, phosphates or silyl ethers. Again, problems can arise when the electrophiles can readily undergo both S_N2 and S_N2' displacements. In such examples, the ratio of products obtained is more dependent on the nature of the aryl metal and the solvent than on the position or nature of the allylic leaving group. Arylzinc intermediates can also be alkylated by bromoacetates to give phenylacetates, provided a nickel catalyst and phosphine ligand are present;¹⁴¹ the corresponding aryl (and heteroaryl) acetonitriles can be similarly obtained using bromoacetonitrile as electrophile.¹⁴² Phenylboranes can also be alkylated using haloacetonitriles.¹⁴³



The rather unreactive arylcadmium reagents, ArCdCl, do react satisfactorily with both primary and secondary α -bromo esters and with secondary allylic bromides, but curiously they react much more poorly with allyl and benzyl bromide, and α -chloro ethers (*ca.* 40% yields); no coupling occurs using primary alkyl bromides.¹⁴⁴ Finally, the old Wurtz–Fittig methodology, whereby an aryl bromide and an alkyl iodide are coupled in the presence of elemental sodium, can sometimes lead to acceptable yields (40–60%); no rearrangement products are formed, and it is likely that free radicals are not involved.¹⁴⁵

1.6.5 ALKYLATIONS OF HETEROAROMATIC CARBANIONS

Despite the plethora of metallated heteroaromatic intermediates which are available, there is often a dearth of information regarding their alkylation chemistry. Many such intermediates are claimed to be generally useful when the evidence provided only mentions very reactive electrophiles such as CO_2 , MeSSMe, Me₃SiCl or MeI. As mentioned above, MeI however is not a typical alkylating agent and is often more suitable as an assay reagent to determine the extent of metallation. In general, alkylations of heteroaromatics give a wide variety of yields in coupling reactions with primary alkyl bromides or iodides (secondary halides usually give only poor yields); if an electrophile such as ethyl iodide affords a good yield, it is likely that other higher homologs will be similarly effective. Results from more reactive allylic and benzylic halides and α -halo ethers tend to be even more variable, and certainly not always an improvement upon alkylations using saturated analogs as is often the case with enolates. Two serious side reactions can account for this; firstly deprotonation of the allylic or benzylic species by the reactive intermediate, and secondly double alkylation. Typically, but by no means exclusively, such alkylations

are best performed in THF rather than ether, sometimes using HMPA as a cosolvent. Excellent yields are usually obtained with methyl iodide (and also especially aldehydes and ketones), moderate to good yields with higher primary homologs and variable, often poor yields with allylic halides and relatives.¹⁴⁶

Both 2-lithiofuran and 2-lithiothiophene often afford respectable yields with a variety of alkylating reagents. The former furan species can be alkylated by *n*-butyl bromide (77%),¹⁴⁷ by the less reactive diethyl acetal of bromoacetaldehyde (70%)¹⁴⁸ and by allylic bromides generally but not always¹⁴⁸ in high yield.^{146,149} The latter thiophene intermediate tends to give somewhat lower yields (*e.g.* BuⁿBr, 47%; BrCH₂CH(OEt)₂, 40%; PhCH₂Br, 62%),¹⁵⁰ while 2-lithiopyrroles can be even less satisfactory intermediates.¹⁵¹ Much the same is true of the benzo analogs of these simple heterocyclic systems.¹⁴⁶

Conversion of these species to various mixed or higher-order cuprate species (*vide supra*) does not usually improve this situation as the heteroaromatic ligands usually transfer sluggishly from copper; indeed the mixed cuprates R(2-thienyl)Cu(CN)Li₂ alluded to above⁷⁶ are valuable precisely because the thiophene ligand is *not* transferred to an attacking electrophile. However, some other transition metal reagents have proven useful in this area, such as 2-lithiofuran alkylation using allylic or benzylic halides in the presence of Pd(PPh₃)4,¹⁵² and the alkylation of heteroarylzinc chlorides by bromoacetonitrile using a combination of Ni(acac)₂ and a phosphine ligand (Scheme 28).¹⁴²



Scheme 28

The concept that doubly ionized species could be 'more reactive' and hence lead to enhanced yields is often not true, an example being the dianion (131), which follows a typical pattern, reacting exceptionally well with aldehydes and ketones (>90%), poorly with primary alkyl halides (30–40%) except methyl iodide (>95%), and not at all, in the sense of giving alkylated products, with allylic or benzylic halides.¹⁵³ By contrast however the dianion (132) does react well with allyl bromide to give the corresponding dialkylated product in 72% yield.¹⁵⁴



In general, six-membered nitrogen heteroaromatics are not easily converted into carbanionic species,¹⁴⁶ and even if this is achieved, alkylation reactions are often poor.^{146,155} An exception to this is the oxazoline derivative (133), which is metallated smoothly at the 3-position using methyllithium (BuⁿLi or LDA are not suitable); the resulting 3-lithio derivative can then be alkylated in reasonably good yields (*ca*. 55%) by primary alkyl iodides and allylic bromides.^{103,156} The corresponding 3-oxazol-inyl isomer reacts much less effectively. The yields in these types of reactions can often be increased by conducting the reactions intra- rather than inter-molecularly; this is the case in a synthesis of 7-meth-oxymitosene (135) in which the ring system is established by cyclization of an intermediate *N*-chloropropylindolecarboxylate (134) following treatment with LDA (BuⁿLi can be used in related examples).¹⁵⁷



An alternative to this type of chemistry is to react the intermediate lithiated heteroaromatics with trialkylboranes and then induce $B \rightarrow C$ alkyl migration using iodine or N-chlorosuccinimide in the intermediate lithium boronates.¹⁵⁸ With simple trialkylboranes, two of the alkyl groups are wasted, although this can be avoided by using 9-BBN derivatives. A particularly notable feature of this methodology is that it is often more successful with secondary or tertiary alkyl groups, thus circumventing a currently universal limitation on direct sp^2 -alkylation, that of the use of tertiary halides.

1.6.6 ALKYLATIONS AT sp²-CENTERS USING EPOXIDES

Often, though not always, successful alkylations of vinyl carbanions using epoxides rely on the presence of copper, either in catalytic (e.g. CuI) or stoichiometric (e.g. R₂CuLi) amounts.¹² In the older literature, a common method was to form, say, an aryl Grignard species in ether then replace the solvent with higher boiling benzene or toluene prior to addition of the epoxide.¹⁵⁹ A later, systematic study revealed however that THF was also an eminently suitable solvent for this type of reaction,¹⁶⁰ although conversely for similar reactions of R₂CuLi, ether is a superior solvent (vide infra). Problems inherent with this method are usually associated with the inevitable presence of MgX₂ which, being a Lewis acid, can induce either epoxide opening and subsequent rearrangements or halohydrin formation, thus giving rise to sometimes poor yields of the expected products (*i.e.* those formed by attack at the less substituted carbon of the epoxide) possibly contaminated with difficult to separate regioisomers. The more sterically hindered the epoxide, often the more essential is the presence of copper. For example, the conversion (136) to (137) proceeds in only 3% yield in the absence of a copper salt; vinylmagnesium halides behave similarly.¹⁶¹ The related process using Ph₂CuLi is significantly more wasteful as 1 equiv. of the organic ligand is lost,¹⁶² hence the attraction of some of the mixed cuprates described below. This procedure is commonly employed in a wide range of Grignard additions to epoxides, two specific examples being extension of the vinylsilane (138) to give the butyrolactone precursors $(139)^{163}$ and elaboration of the allylsilanes (140) from the homologous Grignard reagent.¹⁶⁴ Copper(I) cyanide can be equally as effective as a copper(I) halide.¹⁶⁵ Similarly, while some aryllithiums are alkylated efficiently by simple epoxides,¹⁶⁶ the addition of a catalytic copper salt is usually necessary if good yields are to be realized.¹⁶⁷



A more recent method for the activation of epoxides, especially in reactions with alkylcoppers and related species, is to add boron trifluoride etherate to the system.¹⁶⁸ Such additions will also facilitate alkyllithium additions to epoxides,¹⁶⁹ two specific examples being homologation of the pyridylindole (141) into the hydroxyethyl derivative (142)¹⁷⁰ and alkylation of the sulfone (73) using propylene oxide, which requires the presence of BF₃ if a good yield, in this example 80%, of the desired product (143) is to be obtained.⁸⁹





The bulk of methodology in this area has however been centered on the use of stoichiometric amounts of a copper reagent. A recently developed procedure for the preparation of 'active' elemental copper, by lithium naphthalide reduction of the CuI–PBu₃ complex, is capable of directly producing alkylcuprate species and could be of particular interest in the formation of the cuprate (144) and related intermediates which cannot be obtained *via* the corresponding organolithium or Grignard species¹⁷¹ (in general, cuprates do not react with esters).^{14,57} Both intermediates (144) and cuprates derived from aryl iodides and this active copper react well with epoxides without the need for further catalysis. Similarly, magnesio-cuprates (145), obtained by copper-mediated additions of Grignard reagents to 1-alkynes, react well with ethylene oxide or monosubstituted epoxides but in general yields of the homologs (146) are improved by prior formation of an 'ate' complex using a lithium acetylide.^{17,172} 1,2-Disubstituted epoxides, however tend to react poorly.



The related (Z)-lithium dialkenylcuprates (147) derived from acetylene itself also react well with epoxides to provide a useful route to (Z)-homoallylic alcohols; the lack of reactivity with esters allows an easy access to lactones (148) by condensations between epoxy esters and this type of cuprate (Scheme 29).¹⁷³ Likewise, the lower homologs (149) and (151), both of which are relatively easy to prepare in optically active forms, can be readily converted into homoallylic and bishomoallylic alcohols (150)¹⁷⁴ and (152)¹⁷⁵ respectively. An ester unit can also be incorporated into the cuprate functions; thus, addition of a mixed lithium cuprate, 'RCuYLi', to ethyl propiolate gives the cuprates (153), which add to epoxides to give unexpectedly the (Z)-crotonates (154).¹⁷⁶ Such isomerization is not uncommon with vinyl carbanions in general, and is obviously a limitation when isomeric mixtures are produced.

The origins of much of this methodology lie in the classical studies of Gilman, which were followed by an extensive investigation into the reaction between lithium diphenylcuprate and epoxides.¹⁷⁷ Whereas cyclohexane epoxide (136) is essentially inert to phenyllithium, Ph₂CuLi reacts well allowing the transformation (136) to (137) to be effected in high yield but using ether rather than THF¹⁶⁰⁻¹⁶² as the solvent and using 2 equiv. of the cuprate. Ph₂CuLi also does not usually attack esters. From the yields re-





ported, there is often little to choose between Ph₂CuLi and a combination of PhMgBr and CuI (see also ref. 186). A wide variety of substituted vinyllithium species also react well with epoxides, after conversion to the corresponding lithium divinylcuprate. For example the dihydropyran derivative (155) couples efficiently with the the epoxy ether (156) to give a good yield of the spiro-acetal precursor (157).¹⁷⁸ Allenyllithiums provide something of a contrast to this general pattern in that they condense efficiently and directly with monosubstituted epoxides,¹⁷⁹ especially in the presence of HMPA,¹⁸⁰ to provide a good route to alcohols (158). In all of these reactions, regioselectivity problems can clearly arise with 1,2-disubstituted epoxides; however, some useful regioselectivities have been discovered¹⁸¹ including 1,2-diol formation from β , γ -epoxy alcohols (159)¹⁸² and β -hydroxy ketone formation from α , β -epoxy ketones (160).¹⁸³



In common with lithioallenes, many lithiated heteroaromatic species react well with unhindered epoxides directly to give respectable yields of the expected homologous alcohols. For example, 2-furyllithium in THF reacts almost quantitatively with propylene oxide^{146,148} but yields are rather lower in ether.¹⁸⁴ Returns are also generally poorer in examples of lithiated heteroaromatics which contain two heteroatoms.¹⁴⁶ In the cases of indoles and pyrroles, the corresponding Grignard reagents formed from the *N*-unsubstituted heterocycles can also be used, although yields are usually only good in the former group, when 3-hydroxyethylindoles are obtained.¹⁵⁹ Limited use has been made of the corresponding lithium diarylcuprates because in general the heteroaryl ligands, when 2-substituted, are very unreactive⁷⁶ (vide *infra*). However, such derivatives of 3-lithiofuran do add efficiently to monosubstituted epoxides, the best stoichiometry found being that indicated in formula (**161**).¹⁸⁵ Indeed, one of the best reagents for the transfer of a vinyl ligand to an epoxide is the mixed cuprate (**162**) in which the 2-thienyl ligand is essentially inert;^{76,186} the simpler species R₂Cu(CN)Li₂ are also particularly useful in this respect, giving exceptionally high yields even with disubstituted epoxides.⁷²

When vinyl epoxides are substrates for this type of reaction, Grignard reagents usually give mixtures of 1,2 (both possibilities) and 1,4 addition products; synthetically useful yields can nevertheless be ob-



tained in some cases. For example, phenylmagnesium bromide in ether reacts with vinyloxirane to give largely (84%) the perhaps least expected product (163), whereas the corresponding 2-naphthyl reagent gives the alcohol (164) as the only isolated product in 58% yield, and 2-thienylmagnesium bromide the 1,4 addition product (165) but only in 26% isolated yield.¹⁸⁷ In contast, and in line with their excellence as Michael nucleophiles, lithium diorganocuprates usually react almost exclusively by the 1,4-pathway, a typical example being Ph₂CuLi in ether–benzene, which with vinyloxirane gives an 85% yield of the conjugate addition product (166), accompanied by only 15% of the alternative product (163).^{72,187} In just the same way, (Z)-lithium divinylcuprates (167) can be used in a stereospecific synthesis of the 'skipped' dienes (168).¹⁸⁸ Similarly, cyclic allylic epoxides can participate in this useful type of reaction which is generally favored in ether rather than THF.¹⁸⁹ Thus, the epoxy enolate (169) undergoes a highly regio-and stereo-selective conversion into the cyclohexenol (170), formally by an umpolung process in which a nucleophile is added α to a ketone function.¹⁹⁰ Likewise, the simpler epoxycyclopentene (171) undergoes predominantly 1,4-attack to give mainly cyclopentenol (172).¹⁹¹



Intramolecular alkylations of epoxides at sp^2 -centers are also possible; thus treatment of the epoxyamide (173) with s-butyllithium affords moderate to good yields of the dihydrobenzofurans (174).¹⁹² However, the homologous system (175) is converted into the tetralin framework (176) rather than to an indan derivative, with the magnesium bromide presumably serving to activate the epoxide.¹⁹³ Under similar conditions, the styrene oxide (177) is converted into the benzocyclobutanol (178) presumably by Lewis acid-induced rearrangement of the epoxide function to the corresponding aldehyde via a benzylic carbonium ion.194



A final group of sp^2 -carbanions which have been found to react well with at least monosubstituted epoxides are the vinylalanes formed either by direct hydroalumination¹⁹⁵ or by zirconium-catalyzed carboalumination¹⁹⁶ of a terminal alkyne. Both species are insufficiently reactive to couple with epoxides and must first be converted into their corresponding 'ate' complexes, usually by treatment with n-butyllithium (Scheme 30).



Scheme 30

1.6.7 REFERENCES

- 1. J. F. Normant, Bull. Soc. Chim. Fr., 1963, 1888.
- 2. L. Bert, C. R. Hebd. Seances Acad. Sci., 1925, 180, 1504; L. Bert and P. Ch. Dorier, C. R. Hebd. Seances Acad. Sci., 1930, 191, 332; A. O. Rogers and R. E. Nelson, J. Am. Chem. Soc., 1936, 58, 1029.
- 3. See, for example, E. B. Hershberg, Helv. Chim. Acta, 1934, 17, 351.
- 4. L. B. Jones and J. P. Foster, J. Org. Chem., 1970, 35, 1777 and refs. therein.
- M. P. Sibi, M. A. J. Miah and V. Snieckus, J. Org. Chem., 1984, 49, 737.
 A. S. Hussey and R. R. Herr, J. Org. Chem., 1959, 24, 843 and refs. therein; R. L. Huang, J. Chem. Soc., 1957, 4089. See also L. Crombie, R. Hardy and D. W. Knight, J. Chem. Soc., Perkin Trans. 1, 1985, 1373. For an alternative strategy, see H. C. Brown and M. M. Rogic, J. Am. Chem. Soc., 1969, 91, 4304; H. C. Brown, H. Nambu and M. M. Rogic, J. Am. Chem. Soc., 1969, 91, 6852.
- 7. D. A. Holt, Tetrahedron Lett., 1981, 22, 2243. See also P. A. Wender, D. A. Holt and S. M. Sieburth, J. Am. Chem. Soc., 1983, 105, 3348.
- 8. P. A. Wender, S. M. Sieburth, J. J. Petraitis and S. K. Singh, Tetrahedron, 1981, 37, 3967.
- 9. M. Noel, J.-C. Combret, Y. Leroux and H. Normant, C. R. Hebd. Seances Acad. Sci., Ser. C, 1969, 268, 1152; A. Reychler, Bull. Soc. Chim. Fr., 1907, 1, 1198.

- 10. C. G. Kruse, A. Wijsman and A. van der Gen, J. Org. Chem., 1979, 44, 1847.
- 11. K. Utimoto, K. Uchida, M. Yamaya and H. Nozaki, Tetrahedron, 1977, 33, 1945.
- 12. For a review of the modification of Grignard reagents by transition metal compounds, see H. Felkin and G. Swierczewski, Tetrahedron, 1975, 31, 2735. For a review of Cu¹ catalysis in reactions of RMgX and RLi, see E. Erdik, Tetrahedron, 1984, 40, 641.
- 13. F. Derguini-Boumechal and G. Linstrumelle, Tetrahedron Lett., 1976, 3225. See also M. Tamura and J. K. Kochi, J. Organomet. Chem., 1972, 42, 205.
- 14. J. F. Normant, J. Villieras and F. Scott, Tetrahedron Lett., 1977, 3263.
- 15. V. Grignard, C. R. Hebd. Seances Acad. Sci., 1905, 141, 44 (J. Chem. Soc. Abstr., 1905, 88 (i), 593).
- 16. T. Sato, K. Naruse and T. Fujisawa, Tetrahedron Lett., 1982, 23, 3587 and refs. therein.
- 17. A. Marfat, P. R. McGuirk and P. M. Helquist, J. Org. Chem., 1979, 44, 3888 and refs. therein.
- 18. K. Onuma and H. Hashimoto, Bull. Chem. Soc. Jpn., 1972, 45, 2582.
- 19. M. Tamura and J. K. Kochi, Synthesis, 1971, 303. See also H. Gilman, R. G. Jones and L. A. Woods, J. Org. Chem., 1952, 17, 1630.
- 20. G. Fouquet and M. Schlosser, Angew. Chem., Int. Ed. Engl., 1974, 13, 82.
- 21. L. Friedman and A. Shani, J. Am. Chem. Soc., 1974, 96, 7101.
- 22. S. Nunomoto, Y. Kawakami and Y. Yamashita, J. Org. Chem., 1983, 48, 1912.
- 23. A. Sekiya and N. Ishikawa, J. Organomet. Chem., 1976, 118, 349. For related alkylations of arylplatinum species by iodoacetates,²⁵ see R. D. O'Sullivan and A. W. Parkins, J. Chem. Soc., Chem. Commun., 1984, 1165.
- 24. P. L. Castle and D. A. Widdowson, Tetrahedron Lett., 1986, 27, 6013.
- 25. T. Amano, K. Yoshikawa, T. Sano, Y. Ohuchi, M. Shiono, M. Ishiguro and Y. Fujita, Synth. Commun., 1986, 16, 499.
- 26. B. B. Snider, M. Karras and R. S. E. Conn, J. Am. Chem. Soc., 1978, 100, 4624.
- 27. C. Huynh and G. Linstrumelle, Tetrahedron Lett., 1979, 1073; F. Sato, H. Watanabe, Y. Tanaka, T. Yamaji and M. Sato, Tetrahedron Lett., 1983, 24, 1041; P. F. Hudrlik, R. H. Schwartz and J. C. Hogan, J. Org. Chem., 1979, 44, 155.
- 28. F. Sato, H. Watanabe, Y. Tanaka and M. Sato, J. Chem. Soc., Chem. Commun., 1982, 1126.
- 29. S. Fujikura, M. Inoue, K. Utimoto and H. Nozaki, Tetrahedron Lett., 1984, 25, 1999
- 30. J. F. Normant, A. Alexakis and G. Cahiez, Tetrahedron Lett., 1980, 21, 935; T. Sato, T. Kawara, M. Kawashima and T. Fujisawa, Chem. Lett., 1980, 571; T. Fujisawa, T. Sato, T. Kawara, M. Kawashima, H. Shimizu and Y. Ito, Tetrahedron Lett., 1980, 21, 2181; T. Fujisawa, T. Sato, T. Kawara and K. Naruse, Chem. Lett., 1980, 1123.
- T. Fujisawa, T. Sato, M. Kawashima and M. Nakagawa, Chem. Lett., 1981, 1307.
 T. Fujisawa, T. Sato, M. Kawashima, K. Naruse and K. Tamai, Tetrahedron Lett., 1982, 23, 3583.
- 33. T. Hiyama and N. Wakasa, Tetrahedron Lett., 1985, 26, 3259.
- 34. See, for example, D. Samain, C. Descoins and A. Commercon, Synthesis, 1978, 388; H. Amri and J. Villieras, Tetrahedron Lett., 1987, 28, 5521.
- 35. M. Bourgain-Commerçon, J. F. Normant and J. Villieras, J. Chem. Res. (S), 1977, 183; see also R. C. Haley, J. A. Miller and H. C. S. Wood, J. Chem. Soc. C, 1969, 264; Y. Kitagawa, S. Hashimoto S. Iemura, H. Yamamoto and H. Nozaki, J. Am. Chem. Soc., 1976, 98, 5030.
- G. Consiglio, F. Morandini and O. Piccolo, J. Chem. Soc., Chem. Commun., 1983, 112.
- 37. T. Hayashi, M. Konishi, K. Yokota and M. Kumada, J. Chem. Soc., Chem. Commun., 1981, 313.
- 38. C. Chuit, H. Felkin, C. Frajerman, G. Rousse and G. Swierczewski, J. Chem. Soc., Chem. Commun., 1968, 1604; H. Felkin and G. Swierczewski, Tetrahedron Lett., 1972, 1433; Tetrahedron, 1975, 31, 2735; E. Wenkert, J. B. Fernandes, E. L. Michelotti and C. S. Swindell, Synthesis, 1983, 701.
- 39. T. Fujisawa, S. Iida, H. Yukizaki and T. Sato, Tetrahedron Lett., 1983, 24, 5745.
- 40. V. Calò, L. Lopez, G. Pesce and A. Calianno, J. Org. Chem., 1982, 47, 4482.
- 41. A. M. Caporusso, C. Polizzi and L. Lardicci, J. Org. Chem., 1987, 52, 3920.
- 42. G. M. Robinson and R. Robinson, J. Chem. Soc., 1923, 532.
- 43. J. L. Roberts, Jr., P. S. Borromeo and C. D. Poulter, Tetrahedron Lett., 1977, 1299.
- 44. A. Hosomi, K. Hoashi, Y. Tominaga, K. Otaka and H. Sakurai, J. Org. Chem., 1987, 52, 2947; J. T. Gupton and W. J. Layman, J. Org. Chem., 1987, 52, 3683 and refs. therein.
- 45. G. Linstrumelle, Tetrahedron Lett., 1974, 3809. See also, for example, J. Boeckmann and G. Schill, Chem. Ber., 1977, 110, 703 and refs. therein.
- J. Millon, R. Lorne and G. Linstrumelle, Synthesis, 1975, 434; J. F. Normant, A. Commercon, G. Cahiez and 46. J. Villieras, C. R. Hebd. Seances Acad. Sci., Ser. C, 1974, 278, 967.
- 47. H. Neumann and D. Seebach, Tetrahedron Lett., 1976, 4839; Chem. Ber., 1978, 111, 2785. For an application, see S. K. Richardson, A. Jeganathan and D. S. Watt, Tetrahedron Lett., 1987, 28, 2335.
- 48. M. Gill, H. P. Bainton and R. W. Rickards, Tetrahedron Lett., 1981, 22, 1437.
- 49. E. J. Corey and R. H. Wollenberg, J. Am. Chem. Soc., 1974, 96, 5581.
- 50. E. J. Corey and T. M. Eckrich, Tetrahedron Lett., 1984, 25, 2419.
- 51. H. J. Reich, K. E. Yelm and I. L. Reich, J. Org. Chem., 1984, 49, 3438.
- E. Piers and H. E. Morton, J. Org. Chem., 1980, 45, 4263.
 E. Piers and H. E. Morton, J. Org. Chem., 1979, 44, 3437.
- 54. E. Piers and H. L. A. Tse, Tetrahedron Lett., 1984, 25, 3155.
- 55. G. M. Whitesides, W. F. Fischer, Jr., J. San Filippo, Jr., R. W. Bashe and H. O. House, J. Am. Chem. Soc., 1969, 91, 4871; E. J. Corey and G. H. Posner, *J. Am. Chem. Soc.*, 89, 3911; 1968, 90, 5615. For a detailed description, see G. Linstrumelle, J. K. Krieger and G. M. Whitesides, *Org. Synth.*, 1976, 55, 103.
- 56. F. Näf and P. Degen, Helv. Chim. Acta, 1971, 54, 1939.
- 57. C. R. Johnson and G. A. Dutra, J. Am. Chem. Soc., 1973, 95, 7777, 7783. See also U. Ravid and R. M. Silverstein, Tetrahedron Lett., 1977, 423.
- 58. G. Cahiez, A. Alexakis and J. F. Normant, Tetrahedron Lett., 1980, 21, 1433.

- 59. A. Alexakis, G. Cahiez and J. F. Normant, Synthesis, 1979, 826.
- 60. A. Alexakis and J. F. Normant, Tetrahedron Lett., 1982, 23, 5151.
- 61. M. Gardette, A. Alexakis and J. F. Normant, Tetrahedron Lett., 1982, 23, 5155.
- 62. G. L. van Mourik and H. J. J. Pabon, Tetrahedron Lett., 1978, 2705.
- 63. H. O. House, C.-Y. Chu, J. M. Wilkins and M. J. Umen, J. Org. Chem., 1975, 40, 1460.
- 64. E. J. Corey, D. E. Cane and L. Libit, J. Am. Chem. Soc., 1971, 93, 7016. 65. J. L. Luche, C. Pétrier, A. L. Gemal and N. Zikra, J. Org. Chem., 1982, 47, 3805.
- 66. H. C. Brown and J. B. Campbell, Jr., J. Org. Chem., 1980, 45, 550; N. Miyaura, T. Yano and A. Suzuki, Bull. Chem. Soc. Jpn., 1980, 53, 1471.
- 67. Y. Yamamoto, H. Yatagai, A. Sonoda and S.-I. Murahashi, J. Chem. Soc., Chem. Commun., 1976, 452.
- 68. N. J. Lalima, Jr. and A. B. Levy, J. Org. Chem., 1978, 43, 1279, and refs. therein.
- 69. E. J. Corey and J. Kang, Tetrahedron Lett., 1982, 23, 1651.
- 70. E. J. Corey, P. Ulrich and J. M. Fitzpatrick, J. Am. Chem. Soc., 1976, 98, 222. See also E. J. Corey, D. M. Floyd and B. H. Lipshutz, J. Org. Chem., 1978, 43, 3418; C. R. Johnson and D. S. Dhanoa, J. Org. Chem., 1987, 52, 1885. For earlier work on the development of mixed cuprates, PhS(R)CuLi, see G. H. Posner, C. E. Whitten and J. J. Sterling, J. Am. Chem. Soc., 1973, 95, 7788; G. H. Posner and C. E. Whitten, Tetrahedron Lett., 1973, 1815; G. H. Posner, D. J. Brunelle and L. Sinoway, Synthesis, 1974, 662.
- 71. W. J. Scott, G. T. Crisp and J. K. Stille, J. Am. Chem. Soc., 1984, 106, 4630. 72. For reviews, see B. H. Lipshutz, R. S. Wilhelm and J. A. Kozlowski, Tetrahedron, 1984, 40, 5005; B. H. Lipshutz, Synthesis, 1987, 325. For earlier reviews of cuprate chemistry in general, see G. H. Posner, Org. React., 1975, 22, 253; J. F. Normant, Pure Appl. Chem., 1978, 50, 709; H. O. House, Acc. Chem. Res., 1976, 9, 59; G. H. Posner, 'An Introduction to Synthesis using Organocopper Reagents', Wiley, New York, 1980.
- 73. B. H. Lipshutz, R. S. Wilhelm and D. M. Floyd, J. Am. Chem. Soc., 1981, 103, 7672.
- B. H. Lipshutz, D. Parker, J. A. Kozlowski and R. D. Miller, J. Org. Chem., 1983, 48, 3334.
 A. Bernardini, A. El Hallaoui, R. Jacquier, Ch. Pigiere, Ph. Viallefont and J. A. Bajgrowicz, Tetrahedron Lett., 1983, 24, 3717; J. A. Bajgrowicz, A. El Hallaoui, R. Jacquier, Ch. Pigiere and Ph. Viallefont, Tetrahedron, 1985, 41, 1833.
- 76. B. H. Lipshutz, M. Koerner and D. A. Parker, Tetrahedron Lett., 1987, 28, 945.
- 77. For example, see C. A. Hergrueter, P. D. Brewer, J. Tagat and P. M. Helquist, Tetrahedron Lett., 1977, 4145, 4573; C. K. Bradsher and K. J. Edgar, J. Org. Chem., 1981, 46, 4600; C. K. Bradsher and D. A. Hunt, J. Org. Chem., 1981, 46, 4608 and refs. therein.
- 78. E. Negishi, Y. Zhang and V. Bagheri, Tetrahedron Lett., 1987, 28, 5793; S. Hyuga, S. Takinami, S. Hara and A. Suzuki, Chem. Lett., 1986, 459; E.-i. Negishi, L. D. Boardman, J. M. Tour, H. Sawada and C. L. Rand, J. Am. Chem. Soc., 1983, **105**, 6344.
- 79. J. E. Stemke, A. R. Chamberlin and F. T. Bond, Tetrahedron Lett., 1976, 2947; J. Org. Chem., 1978, 43, 147; A. R. Chamberlin and F. T. Bond, Synthesis, 1979, 44; T. H. Chan, A. Baldassarre and D. Massuda, Synthesis, 1976, 801; P. C. Traas, H. Boelens and H. J. Takken, Tetrahedron Lett., 1976, 2287. For a review, see R. M. Adlington and A. G. M. Barrett, Acc. Chem. Res., 1983, 16, 55.
- 80. A. R. Chamberlin and S. H. Bloom, Tetrahedron Lett., 1984, 25, 4901.
- 81. J. E. Baldwin, G. A. Höfle and O. W. Lever, Jr., J. Am. Chem. Soc., 1974, 96, 7125; C. G. Chavdarian and C. H. Heathcock, J. Am. Chem. Soc., 1975, 97, 3822.
- 82. R. K. Boeckman, Jr., K. J. Bruza, J. E. Baldwin and O. W. Lever, Jr., J. Chem. Soc., Chem. Commun., 1975, 519.
- 83. J. A. Soderquist and G. J.-H. Hsu, Organometallics, 1982, 1, 830.
- 84. P. G. McDougal and J. G. Rico, J. Org. Chem., 1987, 52, 4817.
- 85. R. Amouroux, Heterocycles, 1984, 22, 1489. See also R. K. Boeckman, Jr. and K. J. Bruza, Tetrahedron Lett., 1977, 4187.
- 86. C. B. B. Ekogha, O. Ruel and S. Julia, Tetrahedron Lett., 1983, 24, 4825.
- 87. L. Duhamel, J. Chauvin and A. Messier, J. Chem. Res. (S), 1982, 48.
- 88. N. S. Simpkins, Tetrahedron Lett., 1987, 28, 989.
- S. W. McCombie, B. B. Shankar, A. K. Ganguly, A. Padwa, W. H. Bullock and A. D. Dyszlewski, Tetrahedron Lett., 1987, 28, 4127.
 C. Najera and M. Yus, Tetrahedron Lett., 1987, 28, 6709.
- T. Kauffmann, Angew. Chem., Int. Ed. Engl., 1982, 21, 410 and refs. therein; M. Sevrin, J. N. Denis and A. Krief, Angew. Chem., Int. Ed. Engl., 1978, 17, 526.
 P. A. Grieco, C.-L. J. Wang and G. Majetich, J. Org. Chem., 1976, 41, 726.
- 93. S. K. Richardson, A. Jeganathan and D. S. Watt, Tetrahedron Lett., 1987, 28, 2335. See also R. R. Schmidt, A. Enhsen and R. Betz, Synthesis, 1985, 160.
- 94. E. Piers and R. T. Skerlj, J. Chem. Soc., Chem. Commun., 1987, 1025.
- E. Piers and R. T. Skerlj, J. Org. Chem., 1987, 52, 4421.
 R. T. Taylor, C. R. Degenhardt, W. P. Melega and L. A. Paquette, Tetrahedron Lett., 1977, 159; M. Obayashi, K. Utimoto and H. Nozaki, Tetrahedron Lett., 1977, 1805; H. Westmijze, J. Meijer and P. Vermeer, Tetrahedron Lett., 1977, 1823.
- K. Uchida, K. Utimoto and H. Nozaki, J. Org. Chem., 1976, 41, 2941; Tetrahedron, 1977, 33, 2987.
 R. B. Miller and G. J. McGarvey, J. Org. Chem., 1979, 44, 4623; R. B. Miller and M. I. Al-Hassan, Tetrahedron Lett., 1983, 24, 2055; R. B. Miller, M. I. Al-Hassan and G. McGarvey, Synth. Commun., 1983, 13, 969.
- 99. A. T. Stoll and E. Negishi, Tetrahedron Lett., 1985, 26, 5671.
- 100. E. Negishi, L. D. Boardman, J. M. Tour, H. Sawada and C. L. Rand, J. Am. Chem. Soc., 1983, 105, 6344.
- 101. R. F. Cunico and J. F. Clayton, J. Org. Chem., 1976, 41, 1480.
 102. B. L. Chenard, M. J. Manning, P. W. Raynolds and J. S. Swenton, J. Org. Chem., 1980, 45, 378.
- 103. For a review, see M. Reuman and A. I. Meyers, Tetrahedron, 1985, 41, 837.

- 104. A. Padwa and A. Ku, J. Am. Chem. Soc., 1978, 100, 2181; A. Padwa and H. Ku J. Org. Chem., 1980, 45, 3756.
- 105. S. Djuric, T. K. Sarkar and P. Magnus, J. Am. Chem. Soc., 1980, 102, 6885.
- 106. J.-C. Clinet, E. Duñach and K. P. C. Vollhardt, J. Am. Chem. Soc., 1983, 105, 6710.
- 107. T. D. Harris and G. P. Roth, J. Org. Chem., 1979, 44, 2004.
- N. Meyer and D. Seebach, *Chem. Ber.*, 1980, 113, 1304.
 D. L. Comins, J. D. Brown and N. B. Mantlo, *Tetrahedron Lett.*, 1982, 23, 3979.
- 110. J. T. Sharp and C. E. D. Skinner, Tetrahedron Lett., 1986, 27, 869.
- 111. H. M. Walborsky and P. Ronman, J. Org. Chem., 1978, 43, 731.
- 112. C. Shih and J. S. Swenton, Tetrahedron Lett., 1981, 22, 4217; J. Org. Chem., 1982, 47, 2825.
- 113. G. Linstrumelle and D. Michelot, J. Chem. Soc., Chem. Commun., 1975, 561; R. Baudouy, F. Delbecq and J. Gore, Tetrahedron Lett., 1979, 937.
- 114. D. Michelot and G. Linstrumelle, Tetrahedron Lett., 1976, 275.
- 115. S. Hoff, L. Brandsma and J. F. Arens, Recl. Trav. Chim. Pays-Bas, 1968, 87, 916; A. Kucerovy, K. Neuenschwander and S. M. Weinreb, Synth. Commun., 1983, 13, 875; Y. Leroux and C. Roman, Tetrahedron Lett., 1973, 2585
- 116. J.-C. Clinet and G. Linstrumelle, Tetrahedron Lett., 1978, 1137.
- 117. B. Corbel, J.-P. Paugam, M. Dreux and P. Savignac, Tetrahedron Lett., 1976, 835.
- 118. R. Matsuoka, Y. Horiguchi and I. Kuwajima, Tetrahedron Lett., 1987, 28, 1299, and refs. therein.
- W. Mychajlowskij and T. H. Chan, Tetrahedron Lett., 1976, 4, 439. 119.
- 120. See, for example, P. Rona, L. Tökes, J. Tremble and P. Crabbé, J. Chem. Soc., Chem. Commun., 1969, 43; R. J. Anderson, C. A. Henrick and J. B. Siddall, J. Am. Chem. Soc., 1970, 92, 735.
- 121. M. W. Rathke and D. Vogiazoglou, J. Org. Chem., 1987, 52, 3697; J. E. Dubois, C. Lion and C. Moulineau, Tetrahedron Lett., 1971, 177; J.-E. Dubois and C. Lion, Tetrahedron, 1975, 31, 1227; C. Wakselman and M. Mondon, Tetrahedron Lett., 1973, 4285; E. J. Corey and G. H. Posner, ref. 55. For a review, see R. M. Magid, Tetrahedron, 1980, 36, 1901.
- 122. P. Knochel and D. Seebach, Tetrahedron Lett., 1981, 22, 3223; 1982, 23, 3897; Helv. Chim. Acta, 1984, 67, 261; A.-T. Hansson and M. Nilsson, Tetrahedron, 1982, 38, 389.
- 123. L. Miginiac and B. Mauzé, Bull. Soc. Chim. Fr., 1968, 2544; H. J. Bestmann, G. Wölfel and K. Mederer, Synthesis, 1987, 848.
- 124. I. E. Pollak, A. D. Trifunac and G. F. Grillot, J. Org. Chem., 1967, 32, 272.
- 125. C. Germon, A. Alexakis and J. F. Normant, Tetrahedron Lett., 1980, 21, 3763.
- 126. G. D. Hartman and W. Halczenko, Tetrahedron Lett., 1987, 28, 3241.
- 127. A. Duréault, I. Tranchepain, C. Greck and J.-C. Depezay, Tetrahedron Lett., 1987, 28, 3341 and refs. therein.
- 128. For a review, see B. A. Trofimov and S. E. Korostova, Usp. Khim., Russ. Chem. Rev. (Engl. Transl.), 1975, 44, 41.
- 129. A. Ghribi, A. Alexakis and J. F. Normant, Tetrahedron Lett., 1984, 25, 3075, 3083.
- 130. Y. Tanigawa, H. Kanamaru, A. Sonoda and S.-I. Murahashi, J. Am. Chem. Soc., 1977, 99, 2361.
- 131. T. Sato and S. Watanabe, J. Chem. Soc., Chem. Commun., 1969, 515.
- 132. See, for example, B. O'Connor and G. Just, J. Org. Chem., 1987, 52, 1801.
- 133. For a summary, see G. Zweifel and J. A. Miller, Org. React. (N. Y.), 1984, 32, 375.
- G. Zweifel and R. A. Lynd, Synthesis, 1976, 816.
 K. Uchida, K. Utimoto and H. Nozaki, J. Org. Chem., 1976, 41, 2215.
- 136. S. Sharma and A. C. Oehlschlager, Tetrahedron Lett., 1986, 27, 6161.
- 137. T. Tsuda, T. Yoshida, T. Kawamoto and T. Saegusa, J. Org. Chem., 1987, 52, 1624.
- 138. H. Matsushita and E. Negishi, J. Chem. Soc., Chem. Commun., 1982, 160.
- 139. E. Negishi, H. Matsushita and N. Okukado, Tetrahedron Lett., 1981, 22, 2715.
- 140. H. Matsushita and E. Negishi, J. Am. Chem. Soc., 1981, 103, 2882; E.-i. Negishi, S. Chatterjee and H. Matsushita, Tetrahedron Lett., 1981, 22, 3737.
- 141. T. Klingstedt and T. Frejd, Organometallics, 1983, 2, 598.
- T. Frejd and T. Klingstedt, Synthesis, 1987, 40.
 H. C. Brown, H. Nambu and M. M. Rogíc, J. Am. Chem. Soc., 1969, 91, 6854.
- 144. P. R. Jones and S. J. Costanzo, J. Org. Chem., 1973, 38, 3189 and refs. therein.
- 145. T. L. Kwa and C. Boelhouwer, Tetrahedron, 1969, 25, 5771 and refs. therein.
- 146. For an excellent and thorough review, see H. W. Gschwend and H. R. Rodriguez, Org. React. (N. Y.), 1979, 26, 1. See also F. Marsais and G. Queguiner, *Tetrahedron*, 1983, 39, 2009. V. Ramanathan and R. Levine, *J. Org. Chem.*, 1962, 27, 1216; S. W. McCombie, B. B. Shankar and A. K.
- 147. Ganguly, Tetrahedron Lett., 1987, 28, 4123.
- 148. M. J. Arco, M. H. Trammell and J. D. White, J. Org. Chem., 1976, 41, 2075.
- N. D. Ly and M. Schlosser, Helv. Chim. Acta, 1977, 60, 2085. 149.
- 150. V. Ramanathan and R. Levine, J. Org. Chem., 1962, 27, 1667.
- 151. See, for example, J. C. Chen, Org. Prep. Proced. Int., 1976, 8, 91.
- 152. A. Pelter, M. Rowlands and G. Clements, Synthesis, 1987, 51 and refs. therein; A. Pelter and M. Rowlands, Tetrahedron Lett., 1987, 28, 1203. 153. C. D. Buttery, D. W. Knight and A. P. Nott, J. Chem. Soc., Perkin Trans. I, 1984, 2839 and refs. therein.
- 154. W. Chen and M. P. Cava, Tetrahedron Lett., 1987, 28, 6025.
- 155. See, for example, G. W. Gribble and M. G. Saulnier, Tetrahedron Lett., 1980, 21, 4137.
- A. I. Meyers and R. A. Gabel, J. Org. Chem., 1982, 47, 2633.
 S. Nakatsuka, O. Asano and T. Goto, Chem. Lett., 1987, 1225.
- 158. I. Akimoto and A. Suzuki, Synthesis, 1979, 146; T. Sotoyama, S. Hara and A. Suzuki, Bull. Chem. Soc. Jpn., 1979, 52, 1865; E. R. Marinelli and A. B. Levy, Tetrahedron Lett., 1979, 2313; A. B. Levy, J. Org. Chem., 1978, 43, 4684; Tetrahedron Lett., 1979, 4021.

- 159. For reviews, see N. G. Gaylord and E. I. Becker, Chem. Rev., 1951, 49, 413; J. G. Smith, Synthesis, 1984, 629; A. S. Rao, S. K. Paknikar and J. G. Kirtane, Tetrahedron, 1983, 39, 2323.
- 160. A. Schaap and J. F. Arens, Recl. Trav. Chim. Pays-Bas, 1968, 87, 1249.
- 161. C. Huynh, F. Derguini-Boumechal and G. Linstrumelle, Tetrahedron Lett., 1979, 1503.
- 162. R.-D. Acker, Tetrahedron Lett., 1977, 3407 and refs. therein.
- 163. I. Matsuda, Chem. Lett., 1978, 773.
- 164. H. Nishiyama, H. Yokoyama, S. Narimatsu and K. Itoh, Tetrahedron Lett., 1982, 23, 1267.
- 165. See, for example, K. Tamao, E. Nakajo and Y. Ito, J. Org. Chem., 1987, 52, 4412.
- 166. See, for example, S. Marburg and R. L. Tolman, J. Heterocycl. Chem., 1980, 17, 1333; R. Stern, J. English, Jr. and H. G. Cassidy, J. Am. Chem. Soc., 1957, 79, 5792.
- 167. J. Eustache, J.-M. Bernardon and B. Shroot, Tetrahedron Lett., 1987, 28, 4681.
- 168. For a review, see Y. Yamamoto, Angew. Chem., Int. Ed. Engl., 1986, 25, 947.
- 169. M. J. Eis, J. E. Wrobel and B. Ganem, J. Am. Chem. Soc., 1984, 106, 3693.
- 170. G. W. Gribble and D. A. Johnson, Tetrahedron Lett., 1987, 28, 5259.
- 171. T.-C. Wu, R. M. Wehmeyer and R. D. Rieke, J. Org. Chem., 1987, 52, 5057.
- 172. A. Marfat, P. R. McGuirk and P. M. Helquist, *Tetrahedron Lett.*, 1978, 1363, 2465, 2973; *J. Org. Chem.*, 1979, 44, 1345; A. Marfat, P. R. McGuirk, R. Kramer and P. Helquist, *J. Am. Chem. Soc.*, 1977, **99**, 253.
- 173. A. Alexakis, G. Cahiez and J. F. Normant, Tetrahedron, 1980, 36, 1961; G. Kunesch, P. Zagatti, J. Y. Lallemand, A. Debal and J.-P. Vigneron, Tetrahedron Lett., 1981, 22, 5271.
- 174. M. Larchêveque and Y. Petit, Tetrahedron Lett., 1987, 28, 1993.
- 175. M. Larchêveque and S. Henrot, Tetrahedron Lett., 1987, 28, 1781.
- 176. J. P. Marino and R. J. Linderman, J. Org. Chem., 1983, 48, 4621.
- 177. R. W. Herr, D. M. Wieland and C. R. Johnson, J. Am. Chem. Soc., 1970, 92, 3813; C. R. Johnson, R. W. Herr and D. M. Wieland, J. Org. Chem., 1973, 38, 4263. See also B. C. Hartman, T. Livinghouse and B. Rickborn, J. Org. Chem., 1973, 38, 4346.
- 178. P. J. Kocienski and C. Yeates, Tetrahedron Lett., 1983, 24, 3905; J. Chem. Soc., Chem. Commun., 1984, 151.
- 179. J.-C. Clinet and G. Linstrumelle, Synthesis, 1981, 875.
- 180. C. Huynh and G. Linstrumelle, J. Chem. Soc., Chem. Commun., 1983, 1133.
- 181. For a review, see K. B. Sharpless, C. H. Behrens, T. Katsuki, A. W. M. Lee, V. S. Martin, M. Takatani, S. M. Viti, F. J. Walker and S. S. Woodard, *Pure Appl. Chem.*, 1983, **55**, 589.
- 182. M. A. Tius and A. H. Fauq, J. Org. Chem., 1983, 48, 4131.
- 183. J. P. Marino and J. C. Jaen, J. Am. Chem. Soc., 1982, 104, 3165; P. A. Wender, J. M. Erhardt and L. J. Letendre, J. Am. Chem. Soc., 1981, 103, 2114.
- 184. D. J. Chadwick, J. Chambers, G. Meakins and R. L. Snowden, J. Chem. Soc., Perkin Trans.1, 1975, 523.
- 185. Y. Kojima, N. Kato and Y. Terada, Tetrahedron Lett., 1979, 4667; Y. Kojima and N. Kato, Tetrahedron Lett., 1980, 21, 4365; Tetrahedron, 1981, 37, 2527.
- 186. B. H. Lipshutz, J. A. Kozlowski, D. A. Parker, S. L. Nguyen and K. E. McCarthy, J. Organomet. Chem., 1985, 285, 437; B. H. Lipshutz and J. A. Kozlowski, J. Org. Chem., 1984, 49, 1147; B. H. Lipshutz, J. A. Kozlowski and R. S. Wilhelm, J. Am. Chem. Soc., 1982, 104, 2305.
- 187. C. B. Rose and C. W. Smith, Jr., J. Chem. Soc., Chem. Commun., 1969, 248; R. J. Anderson, J. Am. Chem. Soc., 1970, 92, 4978; R. W. Herr and C. R. Johnson, J. Am. Chem. Soc., 1970, 92, 4979.
- 188. G. Cahiez, A. Alexakis and J. F. Normant, Synthesis, 1978, 528; Tetrahedron Lett., 1978, 2027.
- 189. M. F. Schlecht, J. Chem. Soc., Chem. Commun., 1982, 1331.
- 190. J. P. Marino and J. C. Jaén, J. Am. Chem. Soc., 1982, 104, 3165.
- 191. J. P. Marino, R. F. de la Pradilla and E. Laborde, J. Org. Chem., 1987, 52, 4898; J. P. Marino and J. S. Farina, J. Org. Chem., 1976, 41, 3213.
- 192. K. Shankaran and V. Snieckus, J. Org. Chem., 1984, 49, 5022.
- 193. K. L. Dhawan, B. D. Gowland and T. Durst, J. Org. Chem., 1980, 45, 922.
- 194. E. Akgun, M. B. Glinski, K. L. Dhawan and T. Durst, J. Org. Chem., 1981, 46, 2730.
- 195. E. Negishi, S. Baba and A. O. King, J. Chem. Soc., Chem. Commun., 1976, 17.
- 196. M. Kobayashi, L. F. Valente, E. Negishi, W. Patterson and A. Silveira, Jr., Synthesis, 1980, 1034.

1.7 Alkylations of Alkynyl Carbanions

PETER J. GARRATT University College London, UK

1.7.1 INTRODUCTION	271
1.7.2 METALLATIONS OF ALKYNES	271
1.7.3 ALKYLATIONS OF ALKYNIDE IONS	272
1.7.3.1 Alkylations with Alkyl Halides and Sulfates 1.7.3.2 Alkylations with Epoxides 1.7.3.3 Other Alkylation Reactions	272 277 281
1.7.4 ELECTROPHILIC SUBSTITUTION OF HALOALKYNES	284
1.7.5 SYNTHETIC APPLICATIONS	286
1.7.5.1 Macrolides 1.7.5.2 Sesquiterpenes 1.7.5.3 Leukotrienes and Prostaglandins 1.7.5.4 Non-natural Products	286 288 289 290
1.7.6 REFERENCES	291

1.7.1 INTRODUCTION

The alkylation of *sp*-carbon can, in principle, involve the alkyne (acetylene) either as the nucleophile or the electrophile. In practice by far the most important process involves the alkyne as nucleophile since the acidity of the alkyne proton ($pK_a \approx 25$) allows the ready formation of alkynide ions. These are excellent nucleophiles and they readily undergo acylation and alkylation with appropriate electrophiles. The recent introduction of palladium-catalyzed reactions, usually involving copper(I) salts but also other cations, has greatly increased the use made of arylation and vinylation reactions.¹ In this chapter only the alkylation of the alkynide ion will be discussed; acylation, vinylation and arylation reactions are discussed elsewhere. The alkylation of alkynide anions is a reaction of considerable synthetic use and has been extensively reviewed.²

1.7.2 METALLATIONS OF ALKYNES

Monometallation of acetylene itself is possible under special conditions, and solutions of LiC=CH and NaC=CH in liquid ammonia, for example, can be prepared by simply adding pieces of metal to boiling ammonia while introducing acetylene. When acetylene is introduced into a solution of BuLi in ether (or THF) at 0 °C a fine suspension of LiC=CLi is formed immediately. Ammonia (or other amines) stabilizes the monolithium species. Thus acetylene with BuLi-TMEDA at low temperatures gives a clear solution of LiC=CH. The preparation of BrMgC=CH is not easy, but can be achieved by adding a solution of EtMgBr to a solution of acetylene in THF kept below 30 °C. Solutions of BrMgC=CMgBr and LiC=CLi are produced by simply introducing acetylene into a solution of EtMgBr in THF (>40 °C) and BuLi in Et₂O (>0 °C) respectively.

Alkylation of Carbon

The low acidity of 1-alkynes means that strong bases must be used to form the alkynide ions and that water is not a suitable solvent; aqueous solutions have a very low concentration of alkynide ions.² Some transition metal alkynides can be prepared by precipitation from aqueous solution because their solubilities are very low.² Suitable solvents for the preparation of alkynide ions must be less acidic than the alkyne, and preferably allow the alkyne and the alkynide ion to remain in solution. Liquid ammonia, te-trahydrofuran, ether and hydrocarbons have all been used, particularly the first, the alkynide anion being readily formed by metal amides. Alkynides of many types have been prepared from various metals. Besides Groups I and III, copper(I), silver, gold(I), zinc, mercury and, more recently, aluminum alkynides have been synthesized. The alkynides of Groups I and II have been principally used as nucleophiles in alkylation reactions, but there are now many examples of other metal alkynides in this role. Palladium-catalyzed reactions, as remarked above, have become increasingly important for the reactions of alkynides of metals other than Groups I and II, but these have not usually involved alkylation.

1.7.3 ALKYLATIONS OF ALKYNIDE IONS

1.7.3.1 Alkylations with Alkyl Halides and Sulfates

The alkynide ion can undergo alkylation with a variety of alkylating reagents, such as haloalkanes and alkyl sulfates, with the formation of a carbon–carbon bond. The alkynide ion is also strongly basic so that elimination reactions may accompany or subvert the substitution reaction. Group I metal alkynides in liquid ammonia give mainly substitution products with primary haloalkanes but secondary and tertiary haloalkanes give mainly elimination products, as do 2-substituted primary haloalkanes (equation 1).

$$R \xrightarrow{R^{1}} M + R^{2} \xrightarrow{R^{3}} X \xrightarrow{R^{2}} R^{2} \xrightarrow{R^{1}} R^{2} \xrightarrow{R^{2}} R^{2} \xrightarrow{R^{2}} K^{2} \xrightarrow{R^{2}} MX \qquad (1)$$

Lithium alkynides in tetrahydrofuran or dioxane often give substitution products with secondary haloalkanes, while alkynide Grignard reagents do not usually react with haloalkanes except in the presence of other metals such as cobalt and copper. Substitution of iodine or bromine for chlorine in the haloalkane often leads to an increased yield of the alkylation product and alkanesulfonates may give greater yields than haloalkanes. Scheme 1 illustrates examples of alkylation of haloalkanes and alkyl sulfates with alkynides of Group I metals.





Scheme 1 (continued)

Alkylations of Grignard reagents usually require the presence of copper or other transition metal salts. Scheme 2 illustrates typical reactions involving Group II metals. Reaction between 1-bromo-2-octyne and the Grignard prepared from propargyl alcohol leads to undeca-2,5-diyn-1-ol in good yield (Scheme 3).¹⁵ Similarly, treatment of 1-iodo-2-octyne with 3-butyn-1-ol in the presence of copper(I) iodide and DBU gave dodeca-3,5-diyn-1-ol in 70% yield (Scheme 3).¹⁶





Scheme 2





Lithium alkynylcuprates react with haloallenes to give similar 'skipped' diacetylenes (see below). The related 'skipped' enynes can be prepared by treatment of (pentadienyl)iron(tricarbonyl) halide complexes with dilithium trialkynylcuprates, the compounds being isolated as the iron(tricarbonyl)(diene) complexes (Scheme 4).¹⁷ Further examples of alkylation reactions of copper alkynides are illustrated in Scheme 5. Reaction between a lithium cyanoalkynecuprate and an iodoallene leads to a 'skipped' diacetylene. This useful reaction has been used by Corey in his synthesis of hybridalactone (Scheme 6).²⁰



Both boron and aluminum alkynides undergo alkylation. Thus, treatment of trialkylboron with lithium acetylide leads to a lithium trialkylethynylboride which, on treatment with iodine²¹ or methanesulfonyl chloride,²² gives the alkyl-substituted acetylene (Scheme 7). The lithium tris(alkylethynyl)boride intermediate can be trapped with an alkylating agent leading to the disubstituted acetylene or the adduct can be oxidized with basic hydrogen peroxide to give the corresponding ketone (Scheme 8).²³

Treatment of lithium alkynides with aluminum trichloride leads to tri(ethynyl)aluminum intermediates, which on treatment with haloalkanes give the corresponding disubstituted acetylenes.²⁴ The reaction is successful with tertiary haloalkanes and a variety of alkynes have been prepared (Scheme 9). An interesting variation of this method has been reported by Trost and Ghadri who reacted a diethylethynylaluminum with an allyl sulfone in the the presence of the Lewis acid aluminum trichloride.²⁵ When the allyl sulfone was part of a six-membered ring, then the alkyne was introduced exclusively in a pseudo axial orientation (Scheme 10).







Scheme 10



Scheme 10 (continued)

Examples of nucleophilic substitution with tin alkynides are also known. Thus, treatment of a masked α -bromo amino acid with tin alkynides in the presence of zinc chloride has been shown to lead to the corresponding ethynyl derivative (Scheme 11).²⁶



1.7.3.2 Alkylations with Epoxides

The reactions between alkynide ions and oxiranes (epoxides) are of considerable synthetic value, and generally provide two functional groups with known relative stereochemistry. Alkynide anions associated with many of the metal cations discussed above have been used, but the lithium alkynides have been by far the most popular, mainly because of their ease of synthesis from alkyllithiums. Examples are shown in Scheme 12; the last example in the scheme may involve rearrangement of the oxirane ring. This is certainly the case in the reactions of a number of hydroxymethyloxirane derivatives with lithium alkynides described by Yamaguchi and Hirao.³³ The reaction gives mixtures of rearranged and unrearranged products with the rearranged material predominating in a number of cases (Scheme 13).







Carlson and Olger have reacted the dilithium salt of propargylic acid with oxiranes and shown that the resulting δ -hydroxycarboxylic acids could be cyclized with solvation to 5,6-dihydro-2-pyrones (Scheme 14).³⁴ The method was subsequently used in a neat synthesis of pestalotin (R = BuⁿCHMe).



Scheme 14

fluoride had to be added as a Lewis acid (Scheme 15).



Scheme 15

In some instances only a low yield of product has been obtained during nucelophilic ring opening of oxiranes with lithium alkynides and it has been shown that in a number of these cases the yields can be greatly improved by the addition of a catalytic amount of trimethylgallium.³⁷ This is illustrated in Scheme 16, where the yields shown in brackets are those without the addition of the trimethylgallium. The improvement in yields by the addition of trimethylgallium may not be surprising in view of the fact that aluminum alkynides have been particularly successful as nucelophiles in additions to oxiranes. Fried and Sih, for example, in the course of a prostaglandin synthesis, prepared a methoxymethylaluminum alkynide which opened an oxirane in a regiospecific manner (Scheme 17).³⁸ The replacement of methyl for methoxy greatly reduced the regiospecificity.





3,4-Epoxycyclopentene (4) undergoes nucelophilic addition with the ethylaluminium alkynide (5) in a mixture of THF, toluene and hexane to give predominantly the isomer (6) in which the alkyne has been introduced at carbon 3, adjacent to the double bond (Scheme 18).³⁹ When the reaction was run in toluene at low temperature, the unexpected 4-hydroxy-4-(1-hexynyl)cyclopentene (7) was obtained, which the authors suggest may arise from an 'aluminum'-catalyzed rearrangement of the epoxide to cyclopent-3-enone, which then undergoes nucleophilic addition.



Scheme 18

Heathcock and coworkers⁴⁰ in the course of their synthesis of vernolepin reacted the oxirane (8) with the diethylaluminum alkynide (9), when regiospecific ring opening of the oxirane occurred, leading to (10; Scheme 19). Interestingly, the stereoisomer (11) did not react under the same conditions.



1.7.3.3 Other Alkylation Reactions

Bhanu and Scheinmann⁴¹ have shown that the dilithium salt of propyne alkylates firstly on the methyl group and then on the alkyne. The process occurs sequentially and two different electrophiles can be used. The reaction gave superior yields when the second electrophile underwent nucleophilic addition rather than substitution (Scheme 20).



Reactions between ω -hydroxyalkynes and ω -bromo- or ω -iodo-carboxylic acids in the presence of two equivalents of butyllithium give rise to ω-hydroxyynoic acids (Scheme 21).⁴² Likewise, dilithiated propargyl alcohol has been found to react with α, ω -dibromoalkanes in liquid ammonia, leading to the corresponding diynediols (Scheme 22).43





A variety of other leaving groups have been used in alkylation reactions of alkynides. Thus, Chiu and Peterson⁴⁴ have treated lithium alkynides with the methyl triflate (12) to give the trimethylsilylmethylalkynes (Scheme 23). The reaction also worked with the corresponding chloride. Johnson and coworkers⁴⁵ had previously used the fluorosulfonate (13) in a similar reaction. In addition, Carling and Holmes⁴⁶ have used triflate as a leaving group in an alkylation step in their synthesis of gloeosporone, a germination self-inhibitor (Scheme 24).







Nitrogen derivatives have been used as leaving groups in reactions with alkynides. Thus, Katritzky and coworkers,⁴⁷ for example, have reacted *N*-substituted benzotriazolines (14) with lithium alkynides, which, after a basic work up, give the alkylated alkyne. This reaction was used to prepare α -amino-alkynes (Scheme 25). Reactions of α -arylamino ethers with lithium alkynides also gave α -amino-alkynes.⁴⁸ Mechanistically, this reaction may be considered to be an acylation since it probably proceeds through the corresponding methaldimine (Scheme 26).



282
283

2-Hydroxybenzothiazole has also been used as a leaving group in allylic displacement reactions with copper alkynides.⁴⁹ Furthermore, the copper alkynides can be used in wet solvents, THF normally being used (Scheme 27).



Trost and Martin⁵⁰ have used the dimethylsulfoxonium ion to provide a leaving group. Thus, a *cis* addition of DMTSF to alkenes first led to adducts of the type (15), which then reacted with lithium alkynides in the presence of diethylaluminum chloride to give the *trans* adduct (Scheme 28). The reaction is presumed to involve the episulfonium ion, which then ensures the observed stereospecificity. The reaction is also regiospecific, the anti-Markovnikov addition being illustrated by the second example in Scheme 28. The adducts can be converted to alkanes with Raney nickel and to alkenes by sulfoxide formation and elimination.



Scheme 28

As was previously remarked, water and its associated base are, respectively, too acidic and insufficiently basic to act as solvent and base for the preparation of alkynide ions. Lissel has shown,⁵¹ however, that with the addition of the crown ether 18-crown-6, alkylation of phenylacetylenes can occur with KOH and an iodoalkane (Scheme 29).



Scheme 29

1,3-Butadiyne has also been alkylated through the lithium alkynide. Thus, Holmes and Jones⁵² treated bis(trimethylsilyl)buta-1,3-diyne with MeLi in the presence of lithium bromide and obtained the monolithium alkynide, which was then alkylated in HMPA (Scheme 30). If the lithium alkynide was complexed with ethylenediamine then DMSO could be used as solvent. In addition, Himbert and Feustel⁵³ prepared the lithium derivative of $1-N_N$ -dialkylbuta-1,3-diyne by treatment of 4-N-dialkyl-1,1,2-trichlorobut-1-en-3-yne with butyllithium. The lithium salt was not isolated but was alkylated to the 1-alkyl-4-N-dialkylbuta-1,3-diyne (Scheme 31).





1.7.4 ELECTROPHILIC SUBSTITUTION OF HALOALKYNES

This process is much less common than nucleophilic substitution by alkynide anions and the actual mechanisms of the reactions in which electrophilic substitution of an *sp*-carbon appears to occur probably do not involve simple substitution. Kende and coworkers,⁵⁴ for example, have reacted tertiary enolate anions with chloroalkynes and obtained the corresponding alkylated products (Scheme 32). These

reactions appear to be addition-elimination reactions, and they only occur with chloroalkynes in which the other alkyne substituent is halo, aryl or a similar substituent capable of stabilizing a negative charge. Primary and secondary enolates gave rise to unidentifiable products except for the anion of dimethyl malonate, which adds twice to dichloroacetylene.



Scheme 32

Giacomelli and Lardicci⁵⁵ have treated bromoalkynes with trialkylaluminum in the presence of bis(*N*-methylsalicylaldimine)nickel and obtained an alkyl-substituted alkyne (Scheme 33). This reaction probably involves: (i) insertion of the nickel into the alkyne-bromine bond; (ii) exchange of alkyl for bromine; and (iii) formation of the alkyne-alkyl bond. Thus, although the process appears to be an electrophilic substitution, it is actually nucleophilic in character.



Treatment of enolate anions derived from β -dicarbonyl compounds with either ethynyl(phenyl)iodonium tetrafluoroborate (16)^{56a} or with ethynyl-lead tetraacetate^{56b} provides a neat and direct synthesis of α -ethynyl-1,3-dicarbonyl compounds. The former reaction probably proceeds via 1,2-hydrogen migration of an alkylidenecarbene intermediate (see Scheme 34).



1.7.5 SYNTHETIC APPLICATIONS

The ethynyl group is an extremely versatile two-carbon synthon and can be transformed to a variety of functions. The process is illustrated by a simple synthesis of the sex attractant disparlure (17) shown in Scheme $35.^{57}$ Here the ethynyl group is first converted to the (Z)-alkene, which is then oxidized with *m*-chloroperbenzoic acid to give the oxirane.



A large variety of natural products and biological materials have been prepared with reaction sequences involving alkylation of *sp*-carbon at some stage. Representative examples of the use of this synthetic manipulation in different classes of compounds are given below, but the examples are in no sense exhaustive.

1.7.5.1 Macrolides

Both the Mukaiyama⁵⁸ and Bestmann⁵⁹ syntheses of the macrocyclic lactone recifeiolide (**18**), isolated from *Cephalosporium recifei*, involve *sp*-carbon alkylation steps (Scheme 36), and Shenvi and Gerlach⁶⁰ have synthesized diplodialides A (**19**) via a route that involved the *sp*-carbon alkylation step shown in Scheme 37.





Brefeldin (20) is a 13-membered lactone isolated from a number of sources and has a wide spectrum of biological activity. Many of the syntheses of this lactone use alkynes as intermediates. Thus, Livinghouse and Stevens⁶¹ used an elegant ring opening of a bicyclo[3.1.0]hexane to give the desired *trans* stereochemistry (Scheme 38), whereas in the synthesis of Kitahara and coworkers,⁶² a more conventional alkynide alkylation was involved (Scheme 39).





In their synthesis of erythronolide B (21), Corey and coworkers⁶³ used two alkylations of an alkynide ion in the preparation of one half of the molecule. One alkylation involved regiospecific ring opening of a resolved oxirane to provide the desired compound with the correct absolute stereochemistry, while the second used a simple methylation (Scheme 40).



1.7.5.2 Sesquiterpenes

Corey et al.⁶⁴ treated an allylic bromide with the THP-protected lithium propynol in their synthesis of (\pm) -sirenin, a sperm attractant of the water mold Allomyces, and the same reaction was also used in a synthesis of (\pm) -sesquicarene (Scheme 41). In addition, Meyers and Bienz⁶⁵ used an alkyne alkylation (22) \rightarrow (23) to prepare the final intermediate for radical cyclization in their synthesis of the unnatural (+) isomer of the sesquiterpene, $\Delta^{9(12)}$ -capnellene (24; Scheme 42).





1.7.5.3 Leukotrienes and Prostaglandins

Some of the first syntheses of the leukotrienes involved alkynes as intermediates, since these are excellent synthons for the stereospecific formation of alkenes. The alkylation with the allyl bromide occurred quite late in the Merck–Frosst synthesis of 5-HETE (Scheme 43).⁶⁶





The bromolithium cuprate analogue of the copper dimethyl sulfide shown in Scheme 42 was also alkylated by the *O*-protected allyl bromide to give a 65% yield of the same product. The Merck-Frosst group⁶⁷ also used a simple alkyne alkylation step in the synthesis of 14,15-dehydro-LTB₄ (Scheme 44).



Scheme 44

In the course of their synthesis of LTB4, Nicolaou and coworkers ring opened the THP-protected hydroxymethyloxirane (Scheme 12, ref. 31) in a regiospecific manner and they used a similar reaction in a later synthesis of (12R)-HETE.⁶⁸ A group at Hoffmann-La Roche⁶⁹ alkylated with a substituted propargyl bromide in a synthesis of (\pm) -LTA4 (Scheme 45).



Scheme 45

A Glaxo research group has used an oxirane ring opening with a dimethylaluminum alkynide, *i.e.* (25) \rightarrow (26), in one of their syntheses of prostaglandin F_{2α} (Scheme 46).⁷⁰ (The alkylation step in the Corey synthesis of hybridal actone was illustrated in Scheme 6.)





1.7.5.4 Non-natural Products

As an example of the use of alkynes in the synthesis of non-natural products, Hubert and Hubert⁷¹ have synthesized the novel bicyclic macrocycle (29) by high dilution treatment of the tris alkynide (27) with the tribromide (28).



Scheme 47

1.7.6 REFERENCES

- 1. L. Cassar, J. Organomet. Chem., 1975, 93, 253; H. A. Dieck and F. R. Heck, J. Organomet. Chem., 1975, 93,
- L. Cassai, J. Organomet. Chem., 1975, 35, 255, 11. K. Dick and P. K. Heck, J. Organomet. Chem., 1975, 95, 259; K. Sonogashira, Y. Tohda and N. Hagihara, Tetrahedron Lett., 1975, 4467.
 (a) K. Friedrich, in 'Chemistry of Functional Groups', 'The Chemistry of Triple-bonded Functional Groups', Supplement C, Part 2, ed. S. Patai and Z. Rappoport, Wiley, Chichester, 1983, p. 1380; (b) V. Jäger, Method. Org. Chem. (Houben-Weyl), 1977, 5 (2a), 408; (c) W. Ziegenbein, in 'Chemistry of Acetylenes', ed. H. G. Viehe, Dekker, New York, 1969, chap. 3; (d) L. Brandsma, 'Preparative Acetylenic Chemistry', 2nd edn., Elevier Acetylenic Chemistry, 2nd edn., 2007, 20 Elsevier, Amsterdam, 1988, and earlier refs. given in these sources.
- 3. D. N. Brattesani and C. H. Heathcock, Synth. Commun., 1973, 3, 245.
- 4. M. Picon, C. R. Hebd. Seances Acad. Sci., 1919, 168, 894.
- 5. A. L. Henne and K. W. Greenlee, J. Am. Chem. Soc., 1945, 67, 484.
- 6. K. N. Campbell and B. K. Campbell, Org. Synth., Coll. Vol., 1963, 4, 117.
- 7. J. Gigg, R. Gigg and C. D. Warren, J. Chem. Soc. (C), 1966, 1882.
- 8. T. F. Rutledge, J. Org. Chem., 1959, 24, 840.
- 9. T. H. Vaughn, G. F. Hennion, R. R. Vogt and J. A. Nieuwland, J. Org. Chem., 1937, 2, 1.
- 10. R. A. Raphael and F. Sondheimer, J. Chem. Soc., 1950, 120.
- 11. J. R. Johnson, A. M. Schwartz and T. L. Jacobs, J. Am. Chem. Soc., 1938, 60, 1882.
- 12. A. S. Pfau, J. Pictet, P. Plattner and B. Susz, Helv. Chim. Acta, 1935, 18, 935.
- 13. M. Winter and F. Gautschi, Helv. Chim. Acta, 1962, 45, 2567.
- 14. E. Müller and A. Huth, Tetrahedron Lett., 1972, 1031, 1035, 4359.
- 15. J. G. Gleason, D. B. Bryan and C. M. Kinzig, Tetrahedron Lett., 1980, 21, 1129.
- 16. R. Zambon and J. Rokach, Tetrahedron Lett., 1983, 24, 999.
- 17. W. A. Donaldson and M. Ramaswamy, Tetrahedron Lett., 1989, 30, 1339.
- P. Kurtz, Justus Liebigs Ann. Chem., 1962, 658, 6.
 J. F. Normant, M. Bourgain and A.-M. Rone, C. R. Hebd. Seances Acad. Sci., Ser. C, 1970, 270, 354.
- 20. E. J. Corey and B. De, J. Am. Chem. Soc., 1984, 106, 2735.
- 21. A. Suzuki, N. Miyaura, S. Abiko, M. Itoh, H. C. Brown, J. A. Sinclair and M. M. Midland, J. Am. Chem. Soc., 1973, 95, 3080; M. M. Midland, J. A. Sinclair and H. C. Brown, J. Org. Chem., 1974, 39, 731.
- 22. M. Naruse, K. Utimoto and H. Nozaki, Tetrahedron Lett., 1973, 1847; Tetrahedron, 1974, 30, 2159.
- 23. A. Pelter, C. R. Harrison and D. Kirkpatrick, J. Chem. Soc., Chem. Comun., 1973, 544.
- 24. E. Negishi and S. Baba, J. Am. Chem. Soc., 1975, 97, 7385
- 25. B. M. Trost and M. R. Ghadiri, J. Am. Chem. Soc., 1986, 108, 1098.
- D. Zhai, W. Zhai and R. M. Williams, J. Am. Chem. Soc., 1988, 110, 2501.
 D. Delorme, Y. Girard and J. Rokach, J. Org. Chem., 1989, 54, 3635.
- 28. F. Bohlmann and R. Krammer, Chem. Ber., 1976, 109, 3362
- 29. G. Boche and J. Bigalke, Tetrahedron Lett., 1984, 25, 955.
- 30. S. Hatakeyama, K. Sakurai, K. Saijo and S. Takano, Tetrahedron Lett., 1985, 26, 1333.
- 31. K. C. Nicolaou, R. E. Zipkin, R. E. Dolle and B. D. Harris, J. Am. Chem. Soc., 1984, 106, 3548.
- 32. D. W. Knight and G. Pattenden, J. Chem. Soc., Perkin Trans. 1, 1975, 641.
- 33. M. Yamaguchi and I. Hirao, J. Chem. Soc., Chem. Commun., 1984, 202.
- 34. R. M. Carlson and A. R. Oyler, Tetrahedron Lett., 1974, 2615.
- 35. M. Yamaguchi and I. Hirao, Tetrahedron Lett., 1983, 24, 391.
- 36. M. Yamaguchi, Y. Nobayashi and I. Hirao, Tetrahedron Lett., 1983, 24, 5121.
- 37. K. Utimoto, C. Lambert, Y. Fukuda, H. Shiragami and H. Nozaki, Tetrahedron Lett., 1984, 25, 5423.
- 38. J. Fried and J. C. Sih, Tetrahedron Lett., 1973, 3899.
- 39. G. A. Crosby and R. A. Stephenson, J. Chem. Soc., Chem. Commun., 1975, 287.
- 40. C. G. Chavdarian, S. L. Woo, R. D. Clark and C. H. Heathcock, Tetrahedron Lett., 1976, 1769.
- 41. S. Bhanu and F. Scheinmann, J. Chem. Soc., Chem. Commun., 1975, 817.
- 42. J. Cossy and J. P. Pete, Tetrahedron Lett., 1986, 27, 573.
- 43. D. E. Ames, A. N. Covell and T. G. Goodburn, J. Chem. Soc., 1965, 4373.
- 44. S. K. Chiu and P. E. Peterson, Tetrahedron Lett., 1980, 21, 4047.
- 45. W. S. Johnson, J. D. Elliot and G. J. Hanson, J. Am. Chem. Soc., 1984, 106, 1138.
- 46. R. W. Carling and A. B. Holmes, Tetrahedron Lett., 1986, 27, 6133.
- 47. A. R. Katritzky, J. K. Gallos and K. Yannakopoulou, Synthesis, 1989, 31.
- 48. J. Barluenga, P. J. Campos and G. Canal, Synthesis, 1989, 33.
- 49. V. Carlò, L. Lopez, G. Marchese and G. Pesce, Tetrahedron Lett., 1979, 3873.
- 50. B. M. Trost and S. J. Martin, J. Am. Chem. Soc., 1984, 106, 4263.
- 51. M. Lissel, Tetrahedron Lett., 1985, 26, 1843.
- 52. A. B. Holmes and G. E. Jones, Tetrahedron Lett., 1980, 21, 3111.
- 53. G. Himbert and M. Feustel, Angew. Chem., Int. Ed. Engl., 1982, 21, 282.
- 54. A. S. Kende, P. Fludzinski, J. H. Hill, W. Swenson and J. C. Clardy, J. Am. Chem. Soc., 1984, 106, 3551. For earlier work, see E. Ott and G. Dittus, Chem. Ber., 1943, 76, 80.
- 55. G. Giacomelli and L. Lardicci, Tetrahedron Lett., 1978, 2831.
- 56. (a) M. Ochiai, T. Iton, Y. Takaoka, Y. Masaki, M. Kunishima, S. Tani and Y. Nagao, J. Chem. Soc., Chem. Commun., 1990, 118; (b) M. G. Moloney, J. T. Pinhey and E. G. Roche, J. Chem. Soc., Perkin Trans. 1, 1989, 333.
- 57. K. Eiter, Angew. Chem., Int. Ed. Engl., 1972, 11, 60.
- 58. K. Narasaka, M. Yamaguchi and T. Mukaiyama, Chem. Lett., 1977, 959.
- 59. H. J. Bestmann and R. Schobert, Angew. Chem., Int. Ed. Engl., 1983, 22, 780.
- 60. A. B. Shenvi and H. Gerlach, Helv. Chim. Acta, 1980, 63, 2426.
- 61. T. Livinghouse and R. V. Stevens, J. Chem. Soc., Chem. Commun., 1978, 754.
- 62. T. Kitahara, K. Mori and M. Matsui, Tetrahedron Lett., 1979, 3021.

- 63. E. J. Corey, E. J. Trybulski, L. S. Melvin, Jr., K. C. Nicolaou, J. A. Secrist, R. Lett, P. W. Sheldrake, J. R. Falck, D. J. Brunelle, M. F. Haslanger, S. Kim and S.-e. Yoo, J. Am. Chem. Soc., 1978, 100, 4618; E. J. Corey, S. Kim, S.-e. Yoo, K. C. Nicolaou, L. S. Melvin, Jr., D. J. Brunelle, J. R. Falck, E. J. Trybulski, R. Lett and P. W. Sheldrake, J. Am. Chem. Soc., 1978, 100, 4620.
- 64. E. J. Corey, K. Achiwa and J. A. Katzenellenbogen, J. Am. Chem. Soc., 1969, 91, 4318. See also, E. J. Corey and K. Achiwa, Tetrahedron Lett., 1970, 2245. 65. A. I. Meyers and S. Bienz, J. Org. Chem., 1990, 55, 791.
- 66. J. Rokach, J. Adams and R. Perry, Tetrahedron Lett., 1983, 24, 5185.
- 67. See J. Rokach, Y. Guindon, R. N. Young, J. Adams and J. G. Atkinson, in 'The Total Synthesis of Natural *Products*', ed. J. W. ApSimon, Wiley, New York, 1988, vol. 7, p. 141. 68. K. C. Nicolaou, J. Y. Ramphal and Y. Abe, *Synthesis*, 1989, 898.
- 69. M. Rosenberger and C. Neukom, J. Am. Chem. Soc., 1980, 102, 5425; M. Rosenberger, C. Neukom and E. R. Aig, J. Am. Chem. Soc., 1983, 105, 3656.
- 70. C. C. Howard, R. F. Newton, D. P. Reynolds, A. H. Wadsworth, D. R. Kelly and S. M. Roberts, J. Chem. Soc., Perkin Trans. 1, 1980, 852; R. J. Cave, C. C. Howard, G. Klinkert, R. F. Newton, D. P. Reynolds, A. H. Wadsworth and S. M. Roberts, J. Chem. Soc., Perkin Trans. 1, 1979, 2954.
- 71. A. J. Hubert and M. Hubert, Tetrahedron Lett., 1966, 5779.

1.8 Friedel–Crafts Alkylations

GEORGE A. OLAH, RAMESH KRISHNAMURTI and G. K. SURYA PRAKASH

University of Southern California, Los Angeles, CA, USA

	2/5
1.8.2 NATURE OF ALKYLATING AGENTS AND CATALYSTS	294
1.8.2.1 Alkylating Agents	294 205
1.0.2.2 Catalysis 18221 Acidic balidas (Lawis acids)	293
1.8222 Metal alkuls and alkorides	296
1.822.3 Acidic oxides and sulfides (acidic chalcogenides), modified zeolites	296
1.8.2.2.4 Acidic cation-exchange resins	296
1.8.2.2.5 Proton acids (Brønsted acids)	297
1.8.2.2.6 Superacids	297
1.8.2.2.7 Lewis superacids	297
1.8.2.2.8 Brønsted–Lewis superacids	29 7
1.8.2.2.9 Solid superacids	297
1.8.2.2.10 Metathetic cation-forming agents	298
1.8.3 ALKYLATION OF ARENES	298
1.8.3.1 Alkylation with Alkyl Halides	299
1.8.3.2 Alkylation with Alkenes	304
1.8.3.3 Alkylation with Alcohols, Ethers, Esters, Epoxides and Lactones	309
1.8.3.4 Alkylation with Di- and Poly-functional Alkylating Agents	317
1.8.3.5 Haloalkylation	320
1.8.3.6 Alkylation with Alkanes	322
1.8.3.7 Cycloalkylation	323
1.8.3.8 Transalkylation, Isomerization, Disproportionation and Dealkylation	327
1.8.4 ALKYLATION OF ALKENES	331
1.8.5 ALKYLATION OF ALKYNES	332
1.8.6 ALKYLATION OF ALKANES	332
1.8.7 REFERENCES	335

1.8.1 INTRODUCTION

In May 1877, Charles Friedel and James Mason Crafts made two communications to the Chemical Society of France of what were to become known as Friedel–Crafts reactions. Realizing the practical importance of their discovery, they lost no time in patenting their findings in France and Great Britain. Friedel and Crafts extended their studies of the catalytic effect of AlCl₃ in processes such asG: (i) reactions of organic halides and unsaturated compounds with aromatic and aliphatic hydrocarbons; (ii) reactions of anhydrides of organic acids with aromatic hydrocarbons; (iii) reactions of oxygen, sulfur, sulfur dioxide, carbon dioxide and phosgene with aromatic hydrocarbons; (iv) cracking of aliphatic and aromatic hydrocarbons; and (v) polymerization of unsaturated hydrocarbons.

Aluminum chloride and related catalysts bring about reactions between aromatic and aliphatic compounds, and inorganic and organic compounds; they also serve in the degradation of aromatic and aliphatic compounds, in substitution and addition reactions and various other reactions, such as polymerizations. Generally, any organic reaction effected by the catalytic action of AlCl₃, or related catalysts, is regarded as a Friedel–Crafts reaction. Many important industrial processes such as the manufacture of high-octane gasoline, ethylbenzene, synthetic rubber, plastics and detergent alkylates are based on Friedel–Crafts chemistry. The scope of Friedel–Crafts reactions is extremely wide, and forms a large part of the more general field of electrophilic reactions. A very large volume of literature has been published and the reader is referred to monographs,^{1,2} comprehensive reviews^{3,4} and an excellent treatise on Friedel–Crafts alkylation chemistry⁵ for more detailed information.

For some time the main emphasis of Friedel–Crafts reactions was chiefly on aromatic compounds. The development of aliphatic Friedel–Crafts chemistry was of minor importance until World War 2, when isomerization of alkanes and cycloalkanes, preparation of high-octane aviation gasoline and synthetic rubber, and polymerization of alkenes achieved considerable importance; these contributed to the growth of aliphatic Friedel–Crafts chemistry.

Despite their seemingly wide variety, Friedel-Crafts reactions can be divided into two general categories: alkylations and acylations. Within these two broad areas, there is considerable diversity. Friedel-Crafts alkylations allow C-C bond formation in both aromatic and aliphatic systems, and are of substantial synthetic and industrial significance.

1.8.2 NATURE OF ALKYLATING AGENTS AND CATALYSTS

1.8.2.1 Alkylating Agents

Some of the earliest-known examples of Friedel–Crafts reactions were alkylations of arenes using alkyl halides. Since then, however, a wide variety of alkylating agents has been used (Table 1); they include alkenes, alkynes, alcohols and various other reagents.

 Table 1
 Most Frequently Used Alkylating Agents in Aromatic Alkylations

Alkynes Alkanes and cycloalkanes Alcohols Mercaptans Esters (of carboxylic and inorganic acids) Thiocyanates	Alkyl halides Alkenes Alkynes Alcohols Esters (of carboxylic and inorganic acids)	Ethers Aldehydes and ketones Alkanes and cycloalkanes Mercaptans Thiocyanates
--	---	---

Alkyl fluorides are the most reactive alkyl halides in aromatic alkylation reactions. The high reactivity of the C—F bond over C—Cl, C—Br and C—I bonds enables the preferential reaction of alkyl fluorides to occur in a system containing mixed halides. However, due to their availability and low cost, alkyl chlorides and bromides are most frequently used as alkylating agents. The reactions that employ alkyl iodides as the alkylating agents are generally accompanied by side reactions and decomposition. The relative ease of alkylation with alkyl halides follows the order: tertiary, benzyl > secondary > primary. In the series of primary alkyl halides, methyl halides are the least reactive. A similar sequence of activity is also observed in alkylations with alcohols, ethers and esters. In alkylation reactions with alkyl halides and alkenes the catalyst is required only in very small quantities. However, with alcohols and related alkylating agents (esters, ethers, etc.) considerably larger quantities of catalyst are necessary because of the complexity of the reaction and interaction of the catalyst with the alkylating agents.

A number of important commercial applications of the Friedel–Crafts reaction are based on the alkylation of aromatic compounds with alkenes, as in the synthesis of styrene from benzene and ethylene, and cumene from benzene and propene. When di- and poly-functional alkylating agents are employed, a wide variety of products such as diarylalkanes, polycyclic arenes, indans, tetralins and alicyclic arenes are formed. In the case of an unsaturated alkyl halide, the reaction with an arene may occur either involving the halide and/or the double bond, depending on the type of catalyst and the reaction conditions. Protic acid catalysts generally favor reaction at the double bond, and metal halides favor reaction of the halide. Rearrangements of alkylating agents catalyzed by metal halides and strong acids have been extensively studied in order to understand the results of alkylations utilizing these reagents.⁵ Particularly, much effort has been expended in determining whether or not these rearrangements occur simultaneous with or consecutive to the alkylation reactions. Reactions involving primary alcohols, ethers and esters give different results depending on whether the Friedel-Crafts catalyst is a strong or a weak acid.

1.8.2.2 Catalysts

Friedel and Crafts themselves observed that aluminum chloride is by no means the only specific catalyst in the Friedel–Crafts reaction. A number of other acidic metal halides could also be employed; however, these were less reactive. The 'strength' or 'coordinating power' of different Lewis acids can vary widely against different Lewis bases. Hence it is extremely difficult to establish a scale of strength of Lewis acids in a manner analogous to that used for Brønsted acids. Despite the difficulties, a number of qualitative orders of reactivity have been proposed.⁶ A comparative study of the activity of various Friedel–Crafts catalysts was performed by Olah and coworkers.⁷ Thus the activity index⁸ (the lowest temperature at which reaction occurs) of a large number of Lewis acid halides was measured using the benzylation reaction as the probe.

1.8.2.2.1 Acidic halides (Lewis acids)

Acidic metal halides, of which AlCl₃ and AlBr₃ are the most frequently used, comprise a large number of Lewis acid catalysts for Friedel-Crafts reactions. Other, frequently used active metal halide catalysts include: BeCl₂, CdCl₂, ZnCl₂, BF₃, BCl₃, BBr₃, GaCl₃, GaBr₃, TiCl₄, TiBr₄, ZrCl₄, SnCl₄, SnBr₄, SbCl₅, SbCl₃, BiCl₃, FeCl₃ and UCl₄. Such Lewis acids possess an electron-deficient central atom, capable of accepting electrons from basic substances. AlCl3 and BF3 (vide infra) are very important Friedel-Crafts catalysts and their applications cover a wide area of Friedel-Crafts-type reactions, including alkylations (both aromatic and aliphatic), different types of cyclizations, isomerizations, polymerizations and many other reactions.² One of the chief advantages in the use of BF₃ as a Friedel-Crafts catalyst is the fact that, being a low-boiling gas (b.p. -101 °C), it is easy to handle and to remove (or recover) from the reaction mixtures. Another advantage is that tarry and undesirable by-products are usually not obtained. Aluminum chloride has been particularly attractive in the industrial area due to its low cost and availability. Olah et al.9 have prepared a new class of Friedel-Crafts catalysts which are triflate derivatives of boron, aluminum and gallium $[M(SO_3CF_3)_3; M = B, Al, Ga]$. These catalysts have been studied in detail for their effectiveness in catalyzing reactions such as alkylation with alkyl halides and isomerization processes. Due to the extreme moisture sensitivity of Lewis acids such as AlCl₃, it is rather difficult to carry out a Friedel-Crafts reaction under strictly anhydrous conditions. Trace amounts of acids (generated by hydrolysis of the metal halides), oxygen and organic halides invariably tend to be present in the reaction medium. However, such impurities, the so-called cocatalysts, have been found to accelerate rather than retard the reaction.¹⁰

Two main groups of cocatalysts can be identified in Friedel–Crafts systems: (i) proton-releasing substances, which consists of all hydroxy compounds (water, alcohol) and proton acids (HCl, H_2SO_4 , H_3PO_4 and other organic and inorganic acids); and (ii) cation-forming substances (cations other than protons), which includes alkyl and acyl halides and a variety of other donor substances (O, S, N, halogen, *etc.* donors). Alkyl and acyl halides form carbocations as the activators. Other positively charged species, such as oxonium, sulfonium and halonium complexes, can also act as cocatalysts in different Friedel–Crafts reactions.

Anhydrous Lewis acids themselves are suitable catalysts for Friedel–Crafts reactions of compounds containing n-donor groups. However, they are generally insufficient as acceptors in reactions of alkenes or alkynes or isomerization of hydrocarbons. These reactions generally require a cocatalyst, such as a hydrogen halide or other cationic species.

Generally, Friedel–Crafts catalysts cannot be reused after aqueous work-up of the reaction mixture, although in some cases they can be recovered by special treatment. Fujiwara and coworkers,¹¹ however, found that lanthanide trihalide compounds (ScCl₃, YCl₃, LaCl₃, CeCl₃, PrCl₃, NdCl₃, SmCl₃, EuCl₃, GdCl₃, TbCl₃, LuCl₃, DyCl₃, HoCl₃, ErCl₃, TmCl₃ and YbCl₃) catalyzed the alkylation of aromatics, and the catalyst could be reused with no apparent loss of catalytic activity.

A Lewis acid catalyst can interact with the reagent containing a functional group having a donor atom with nonbonded pairs of electrons. This gives rise to a positively polarized complex or a carbocationic species, which then reacts with the π -donor substrate (aromatic, alkenic or alkynic hydrocarbons). Though this process can occur under strictly anhydrous conditions, this generally is not the case as impurity, moisture, or other cocatalysts are usually present. In the case of reactions of alkenes and alkynes

it is essential to have a cocatalyst to form a strong conjugate acid or carbocation, which then initiates the reaction. Neat Lewis acids by themselves generally do not initiate Friedel-Crafts alkylation with alkenes. Impurities, physical state and the method of preparation of the catalyst affect both yield and the course of a reaction. For example, although traces of moisture in AlCl₃ may accelerate the effect, other impurities, such as FeCl₃, can decrease the yield.

1.8.2.2.2 Metal alkyls and alkoxides

Alkyl derivatives of metals such as aluminum, boron and zinc are fairly active Friedel–Crafts catalysts. However, hyperconjugative effects result in a lowering of the electron deficiency. In the case of metal alkoxides this effect is even stronger, and, as a result, they are fairly weak Lewis acids. Metal alkyls, such as alkylaluminums, alkylaluminum halides and sesquihalides are also vital components of Ziegler–Natta catalyst systems which sometimes are utilized for Friedel–Crafts-type reactions. For example, alkylations of aromatics with alkenes in the presence of a Ziegler–Natta catalyst such as AlR₃ + TiCl₄ results in lower-chain alkylates. Even alkylaluminum halides and sesquihalides serve as Friedel–Crafts catalysts.

Among metal alkoxides, aluminum phenoxide is one of the most important Friedel-Crafts catalysts in the alkylation of phenol.¹²

1.8.2.2.3 Acidic oxides and sulfides (acidic chalcogenides), modified zeolites

A large variety of solid oxides and sulfides, both natural and synthetic, such as alumina, silica and mixtures of alumina and silica (either natural or synthetic), in which other oxides such as chromia, magnesia, molybdena, thoria, tungstic oxide and zirconia may also be present, as well as certain sulfides of molybdenum, come under this category of catalyst. Many synthetic chalcogenides other than silicaalumina systems are known; these include BeO, Cr₂O₃, P₂O₅, TiO₂, Al₂(SO₄)₃, Al₂O₃, xCr₂O₃, Al₂O₃, Fe₂O₃, Al₂O₃·MnO, Al₂O₃·CoO, Al₂O₃·Mo₂O₃, Al₂O₃·V₂O₃, Cr₂O₃·Fe₂O₃, MoS₂ and MoS₃. In the case of silica-alumina systems, which are the most extensively studied, the acidity could reside either at Lewis acid or Brønsted acid sites. It has been estimated that silica-alumina is quantitatively at least as acidic as 90% sulfuric acid.¹³ Both natural and synthetic zeolites are very important catalysts for many commercial processes. They have good thermal and hydrothermal stabilities, and have the ability to absorb and concentrate hydrocarbons. The well-defined crystal structure enables these zeolites to have pores with one or more discrete diameters approaching molecular dimensions (<1 nm). This property is responsible for their molecular-sieving action and their ability to effect selective transformations. Shapeselective zeolite catalysis is a very powerful means of effecting selective processes. Factors that determine shape selectivity include pore shape (circular, elliptical), intersections of the channels within the zeolite, sizes and shapes of cavities, side pockets and crystal symmetry. Three types of shape selectivities are possible: (i) reactant selectivity — one part of the reacting molecules can pass through the catalyst pores; (ii) product selectivity - only products with proper dimensions can diffuse out from the pores; and (iii) restricted transition state selectivity — only those reactions occur where the transition state 'fits' the internal cavity.

One of the most active areas of zeolite chemistry involves efforts to improve the selectivity of zeolite catalysts. It includes selective deactivation of external acid sites with substances such as amines; replacement of the cationic sites by ion exchange with transition metal ions; and modification of the silica/alumina ratio to achieve optimum hydrophilicity (which increases with decreasing aluminum concentration), acid strength and good thermal, hydrothermal and acid stability. The literature, particularly the patent literature, abounds with numerous examples of these areas and others.¹⁴ Various types of clays are being studied for their ability to catalyze shape-selective reactions. For example, pillared clays can provide discrete pore sizes between 6 and 40 Å. However, they suffer from the disadvantages of catalyst deactivation and thermal instability. Stabilized pillared clay catalysts have been prepared by pillaring the layers with suitable moieties. For example, boron trihalides react with the hydroxy groups on clay and form stable pillars. The resulting catalysts are especially useful for transalkylation reactions.¹⁵

1.8.2.2.4 Acidic cation-exchange resins

Solid acids are widely used to catalyze electrophilic reactions. Sulfonated styrene-divinylbenzene cross-linked polymers are efficient solid acids. These resins, such as Dowex 50, Amberlite IR-112 and

Permutit Q are effective catalysts for the alkylation of phenols with alkenes such as propene, isobutene and 2,5-dimethyl-2-hexene, and alkyl halides and alcohols.¹⁶

Modifications of these resins are also known. For example, aluminum phenoxide bonded to an acidic Dowex resin has been used as a heterogeneous catalyst for alkylation and transalkylation of phenois and polysubstituted phenols, respectively.¹⁷

1.8.2.2.5 Proton acids (Brønsted acids)

Reactions that are catalyzed by metal halide catalysts are also catalyzed by proton acids. The most commonly used Brønsted acids are H_2SO_4 , H_3PO_4 and HF.

1.8.2.2.6 Superacids

Acids which are stronger than 100% sulfuric acid are called superacids.¹⁸ Fluorosulfuric acid (HSO₃F) is one of the strongest Brønsted acids known, with H_0 (Hammett's acidity function) = -15.1. Its acidity is comparable to that of highly concentrated oleum, H₂SO₄·SO₃; however, because of its stability, ease of purification, wide liquid range (m.p. = -89 °C, b.p. = 162 °C) and relatively low viscosity (1.56 cP at 28 °C), and lesser oxidizing ability, it is more convenient to use.

Perfluoroalkanesulfonic acids also show high acidity. The parent trifluoromethanesulfonic acid (triflic acid), CF₃SO₃H, is prepared commercially by electrochemical fluorination of methanesulfonic acid.¹⁹ It has an H_0 value of -14.1. Because it is not a sulfonating agent, its use as an acid catalyst is advantageous. The higher homologs show somewhat decreasing acidities. Perchloric acid and chlorosulfuric acid are also examples of Brønsted superacids.

1.8.2.2.7 Lewis superacids

Any Lewis acid that is stronger than anhydrous AlCl₃ has been arbitrarily defined as a Lewis superacid.²⁰ Examples of such Lewis superacids include SbF₅, NbF₅, BF₃, AsF₅, TaF₅ and BiF₃.

1.8.2.2.8 Brønsted-Lewis superacids

When a suitable Lewis acid halide and proton acid are combined, conjugate Friedel–Crafts acids are formed, which are, indeed, superacids with a wide range of acidity. Anhydrous HF–BF₃ and HCl–AlCl₃ (stable as conjugate pair only under pressure) are widely used examples of such acids. They effectively catalyze hydrocarbon transformations.²⁰ In the early 1960s, much stronger acid systems were prepared²¹ comprising a pentafluoride of a Group V element, particularly SbF₅, and a strong Brønsted acid such as HF, FSO₃H, *etc.* Magic acid (HSO₃F–SbF₅) and fluoroantimonic acid (HF–SbF₅) are two of the best-known examples. The acidity of HF or HSO₃F is increased sharply by adding SbF₅.^{22,23} For example, a composition of 4 mol % SbF₅ in HF has an H_0 value of –19, and with higher SbF₅ concentrations acidities can increase to –22. These very highly acidic systems are utilized for petrochemical transformations such as isomerization of straight chain alkanes, alkane–alkene alkylations, and the like.²⁴ Ternary systems, such as HSO₃F–HF–SbF₅ are also highly efficient superacid catalysts.²¹ Vol'pin and coworkers²⁵ have reported novel, highly active systems derived from aluminum trihalide complexes of acyl and aryl halides, *viz.* RCOX·2AlX₃ (R = alkyl, aryl; X = Br, Cl). These effective electrophilic systems of alkanes and cycloalkanes under mild conditions.

1.8.2.2.9 Solid superacids

Solid or supported catalysts are preferred in petrochemical and chemical industrial processes. Hence, solid superacid catalysts are of substantial interest. Acidic oxides, discussed earlier, are used extensively; however, their relatively low acidity necessitates higher operating temperatures. Even at higher temperatures, they are inefficient in promoting such reactions as the transethylation of benzene with di- and poly-ethylbenzenes. Consequently, new solid acid systems have been developed with considerably higher

acidities than those of acidic oxides. These include acidic zeolites, such as ZSM-5, which combine shape selectivity with high acidity. Graphite-intercalated AlCl₃ is an effective solid Friedel–Crafts catalyst but loses catalytic activity because of ready hydrolysis and leaching of the Lewis acid halide from the graphite.²⁰ Aluminum chloride can also be complexed to sulfonated polystyrene resins, but again the stability of the catalyst is limited. Anhydrous GaCl₃ coated on polystyrene–divinylbenzene copolymer beads forms a stable complex which shows good catalytic activity in a number of organic synthetic reactions. Most notably, the catalyst can be reused several times without losing its activity and it is easily separated from the reaction mixtures.²⁶ Similarly, TiCl₄ supported on polystyrene–divinylbenzene copolymer has been used as a reusable catalyst for many organic reactions including Friedel–Crafts alkylation.²⁷

More stable catalysts are obtained by using fluorinated graphite or fluorinated alumina as backbones and Lewis acid halides, such as SbF_5 , TaF_5 and NbF_5 , which have a relatively low vapor pressure. These Lewis acids are attached to the fluorinated solid supports through fluorine bridging. They show high reactivity in Friedel–Crafts reactions including the isomerization of straight chain alkanes such as *n*-hexane.

Another type of solid superacid is based on perfluorinated resin sulfonic acids, such as the acid form of DuPont's Nafion resin, a copolymer of a perfluorinated epoxide and vinylsulfonic acid, or higher perfluoroalkanesulfonic acids such as perfluorodecanesulfonic acid, CF₃(CF₂)₉SO₃H. Such solid catalysts were found to be very efficient in alkylation of aromatic hydrocarbons and other Friedel–Crafts reactions. A comprehensive review is available on the application of Nafion-H in organic catalysis.²⁸

1.8.2.2.10 Metathetic cation-forming agents²

A number of metathetic agents are capable of forming carbocations from halide precursors. These materials react stoichiometrically with the carbocation precursor, and therefore cannot be considered strictly as catalysts, but initiate Friedel–Crafts-type reactions. Anhydrous silver salts, such as AgClO₄, AgBF₄, AgSbF₆, AgPF₆, AgAsF₆, Ag₃PO₄, *etc.* are representatives of this class of compounds.

1.8.3 ALKYLATION OF ARENES

In the alkylation of aromatic hydrocarbons generally, a hydrogen atom of the aromatic nucleus is replaced by an alkyl group through the interaction of an alkylating agent in the presence of a Friedel– Crafts catalyst. As pointed out earlier, a variety of alkylating agents can be used for this reaction. The reactions in equations (1) to (3) are characteristic of the most commonly employed alkylations of aromatic hydrocarbons.

$$PhH + RX \xrightarrow{AlCl_3} PhR + HX$$
(1)

 $PhH + ROH \xrightarrow{AlCl_3} PhR + H_2O$ (2)

$$PhH + \frac{R}{Ph} \xrightarrow{AlCl_3} \frac{R}{Ph}$$
(3)

The mechanism of alkylation of arenes can be best understood as a carbocationic electrophilic aromatic substitution — a review on this aspect is available.²⁹ The alkylating agent and the catalyst first form an alkyl cation or related polarized complex, which then reacts with the aromatic ring *via* a Wheland intermediate (arenium ion; Scheme 1).

Because of the exhaustive literature on the alkylation of aromatics with a variety of alkylating agents, it is most convenient to discuss the reaction according to various alkylating agents.





1.8.3.1 Alkylation with Alkyl Halides

Methyl and benzyl halides were the earliest alkylating agents for the Friedel–Crafts alkylation of arenes. In fact Friedel and Crafts themselves reported the alkylation of benzene with methyl and benzyl chloride catalyzed by AlCl₃.³⁰ An earlier review by Price³¹ covers the field until 1946, but more extensive treatment of this subject was presented by Olah in 1964 and 1973.³² Further developments up to 1984 have been dealt with by Roberts and Khalaf in their book.⁵ No repetition of the detailed and exhaustive discussion of the topic is, therefore, necessary. Instead, we point out briefly some of the significant work carried out to highlight the most important aspects of the reaction.

Central to the understanding of the Friedel–Crafts alkylation is the nature of the intermediate complexes formed between the catalysts and reactants. Earlier studies³³ on the exchange reactions of alkyl halides with Lewis acids such as AlCl₃ and GaCl₃ led to the postulation of dialkylchloronium ions as the intermediates formed when excess alkyl halide interacts with the Lewis acid. Olah *et al.* in their extensive studies indeed succeeded in preparing and studying dialkylhalonium ions, and the topic was covered in a monograph.^{34a} Olah *et al.*^{34b} also studied alkyl fluoride–antimony fluoride mixtures in solvents such as SO₂ and SO₂ClF, and suggested that they were tight donor–acceptor complexes (*e.g.* MeF→SbF₅ and EtF→SbF₅), which underwent rapid intramolecular fluorine exchange. They also observed that the MeF→SbF₅–SO₂ solution was an extremely powerful methylating agent, capable of reacting with *n*-, π and σ -donor bases as shown in equations (4) to (8). Even the solvents SO₂ and SO₂ClF are *O*-methylated by this system. These reactions showed that MeF→SbF₅ and EtF→SbF₅ can behave like methyl and ethyl hexafluoroantimonates, respectively.

$$MeF \rightarrow SbF_{5} + CO \longrightarrow MeCO SbF_{6} \longrightarrow MeCOOH_{2} SbF_{6} \longrightarrow MeCO_{2}H$$
(4)

$$MeF \rightarrow SbF_5 + = \underbrace{SO_2}_{+} (5)$$

$$MeF + MeF \rightarrow SbF_5 = \begin{bmatrix} H \\ F \\ Me \end{bmatrix}^+ \underbrace{-H^+}_{EtF} EtF \xrightarrow{SbF_5}_{EtF}$$

$$EtF \rightarrow SbF_5 \xrightarrow{EtF} Me_3C^+, etc.$$
(6)

$$SO_2 \xrightarrow{MeF \rightarrow SbF_5} MeO = S = O \overline{SbF_5} \text{ (or } Sb_2F_{11})$$
 (7)

$$SO_2CIF \xrightarrow{MeF \rightarrow SbF_5} MeO = S = O SbF_6 (or Sb_2F_{11})$$
(8)

The first kinetic study of the Friedel–Crafts alkylation reaction was reported by Brown and Grayson in 1953.³⁵ This involved the reactions of some substituted benzyl halides with aromatic compounds in the presence of AlCl₃–PhNO₂ catalyst. The transition state for the rate-determining step (attack of the aromatic component on a polar alkyl halide–AlCl₃ addition compound) was depicted as a σ -complex.

Further studies by Brown and coworkers³⁶ lent additional support to this mechanism and the absence of a free alkyl cation. Olah and coworkers³⁷ have applied the concept of competitive alkylation to the case of naphthalene in order to study both positional and substrate selectivities, and to clarify the nature of kinetically *versus* thermodynamically controlled product composition. They explained the observed results by suggesting that a π -complex, such as (1), was the intermediate involved when highly electrophilic catalysts or strongly basic aromatics were employed, and a σ -complex (as proposed earlier by Brown) was involved in reactions with weakly electrophilic catalysts or less basic aromatics.



Similar explanations were put forth later by Nakane and coworkers³⁸ to explain their observed results. Olah's group³⁹ investigated the Friedel–Crafts alkylation of anisole and the analogous reaction with to luene.³¹ A significantly lower degree of *meta* substitution was observed with anisole. The experimental data suggested that in the alkylation of anisole, in contrast to that of toluene, electrophilic attack by the alkylating agent at the *ortho* and *para* positions is less readily or not at all followed by thermodynamically controlled alkyl and hydrogen shifts, which could lead to increased *meta* substitution. Speranza and coworkers⁴⁰ have recently investigated the gas-phase benzylation of 2,6-dimethylanisole. Besides Friedel–Crafts catalysts, many other catalysts have also been utilized for studies on the benzylation reaction. Notable examples are arene tricarbonylmolybdenum, *i.e.* [ArMo(CO)₃].⁴¹ metal oxides⁴² and calcined sulfate salts of Fe, Zn, Co, Mn and Cu.^{31,43} Recently, Laszlo and Mathy⁴⁴ have studied the utility of K10 montmorillonite as a Friedel–Crafts catalyst by exchange of the interstitial cations with various transition metal ions such as Zn^{II}, Cu^{II} and Zr^{IV}. For example, in the benzylation of benzene using benzyl chloride, excellent conversion of benzyl chloride and good isolated yields (up to 66%) of the monobenzylated product were obtained. The effect of the nature of the metal ion on the reactivity and selectivity of alkylation reactions has also been dealt with.⁵

Alkylations of arenes by ethyl halides under Friedel–Crafts conditions have been carried out by numerous workers. Excellent accounts of the developments made in this area are available from a review,³¹ as well as a recent book.⁵ Equations (9) to (11) illustrate some ethylation reactions.^{37–39}

In 1979, Olah and Meidar⁴⁵ found that alkylation of toluene offered solely a mixture of isopropyltoluenes. No *n*-propyltoluenes were detected. This was explained as proceeding by the initial formation of the Cl-protonated *n*-propyl chloride (2; Scheme 2), which rapidly rearranges to the isopropyl cation, followed by reaction with the aromatic ring.

Masuda and coworkers⁴⁶ have found that both of the alkylating agents (-)-2-chloro-1-phenylpropane [(-)-3] and (+)-1-chloro-2-phenylpropane [(+)-4] react with benzene in the presence of AlCl₃ to give the same product, (-)-1,2-diphenylpropane [(-)-5] in 45-100% *ee*, which was not racemized under the





reaction conditions (Scheme 3). A phenyl π -assisted cation (6) with an unsymmetrical bridging group has been proposed to account for the observed asymmetric induction.



Kebarle and coworkers⁴⁷ have addressed the question of stability of the *t*-butylbenzenium ion obtained by gas-phase reaction of benzene with *t*-butyl cation.

Besides its application in mechanistic studies, the Friedel–Crafts alkylation has also been used to prepare synthetically useful compounds. Roth and Aig^{48} attempted the Friedel–Crafts ethylation of *p*-ethylacetophenone (equation 12) under carefully optimized conditions to obtain the 3,4,5-triethyl derivatives (7) in reasonable yield. They also noted that when less ethyl bromide was employed, the product, surprisingly, was hexaethylbenzene. Compound (7) is a key intermediate in the synthesis of 2,4-diamino-5-(3,4,5-triethylbenzyl)pyrimidine (8), an antibacterial agent.



Highly selective (87%) isopropylation of an aniline at the *para* position has been achieved⁴⁹ by employing a particular mole ratio of the catalyst (aluminum halide) and the aniline (equation 13). Conversion of the substrate in the reaction was 87%.



Zhukov and coworkers⁵⁰ alkylated naphthalene with 1-bromoadamantane in the presence of $ZnCl_2-H_2O$ to afford 2-(1-adamantyl)naphthalene in 90% yield (equation 14).



The Friedel–Crafts benzylation reaction has been employed to prepare polymer-bonded 2-mercaptopyridine and 2-mercaptonitrobenzene derivatives as new reagents for peptide synthesis.⁵¹ Crosslinked polystyrene was subjected to benzylation using the benzyl chloride (9) as the alkylating agent, followed by further manipulation of the polyalkylated polymers. These polymeric reagents were used for the thiolytic cleavage of the 2-(O_2N)C₆H₄S group in peptide synthesis.

Fujiwara and coworkers¹¹ have discovered that anhydrous lanthanoid trichloride salts⁵² are reusable Friedel–Crafts catalysts for the benzylation of benzene. All of the lanthanoid trichlorides offer fairly good yields of benzylated product (equation 15). The yields do not drop if the recovered catalyst is used for a second time, or in some cases even a third time, for the benzylation of benzene.

Asymmetric Friedel–Crafts alkylations, besides affording interesting mechanistic problems, also offer valuable synthetic utility. For example, reaction of benzene with optically active 2-chloro-1-phenyl-propane or 1-chloro-2-phenylpropane in the presence of AlCl₃ affords good yield (60%) of alkylated product with up to quantitative optical yields (see Scheme 3).⁴⁶



M = Sc, Y, La, Ce, Pr, Nd, Sm, Eu, Gd, Tb, Dy, Ho, Er, Tm, Yb, Lu

Optically active 2-arylglycine esters, such as (10), have been prepared⁵³ by Friedel-Crafts alkylation of alkoxybenzenes and naphthalenes with the chiral cation of the bislactim ether of cyclo-(L-Val-Gly). For example, when the chloride (12), prepared from the bislactim ether (11), was allowed to react with 1,4-diethoxybenzene in the presence of tin tetrachloride as the Lewis acid (equation 16), the adduct (13) was obtained in 65% yield. Acid hydrolysis of (13) furnished the previously unknown (R)- α -arylglycine methyl ester (10). The active species in the alkylation step has been proposed to be the ion pair (14). Unactivated arenes were found not to react under these conditions. 2-Arylglycines such as (10) are of pharmacological interest due to their potential biological activity and as components of drugs such as semisynthetic penicillins, ampicillins and amoxycillins.



A practical application⁵⁴ of the benzylation reaction is the AlCl₃-catalyzed Friedel–Crafts alkylation of a styrene–ethylene–butene–styrene block copolymer with benzyl chloride to afford modified thermoplastic alkene elastomers. Another application⁵⁵ of Friedel–Crafts-type regiospecific alkylation of arenes with cycloalkyl fluorides derived from sugars (equation 17) constitutes a facile methodology to prepare C-arylglycosides, some of which are potent antitumor agents. The glycosyl fluorides are activated by the novel Lewis acid system, Cp_2ZrCl_2 -AgClO4.⁵⁶



Alkylation of benzene by ω -chloroalkanoic acids Cl(CH₂)_nCO₂H and their methyl esters and nitriles has also been studied.⁵⁷ It was found that introduction of electron-acceptor substituents such as —CO₂H, —CO₂Me and —CN into *n*-alkyl chlorides decreased the extent of isomerization of the alkyl chain below that of the corresponding 1-chloroalkanes, as seen from the examples in equations (18) and (19).

Besides these more recent examples, many instances of the use of benzylation reactions to prepare a variety of substituted aromatic compounds are known.⁵ Some examples are shown in equations (20)⁵⁸ and (21).^{58b}



1.8.3.2 Alkylation with Alkenes

The alkylation of arenes with alkenes such as ethylene and propene are of great commercial interest. Ethylbenzene and isopropylbenzene (cumene), products of the Friedel–Crafts alkylation of benzene with ethylene and propene, respectively, are two of the most important petrochemical raw materials.⁵⁹ Roberts and Khalaf⁵ have followed the developments made in this vast field up to the early part of this decade. This is evident from the large number of references quoted, most of which describe efforts to evaluate conditions for optimal production in the presence of various catalyst systems.

Reports of kinetic studies of positional and substrate selectivities for Friedel-Crafts ethylation and propylation reactions are available in the literature. Thus ethylene was used to alkylate aromatics such as toluene,⁶⁰ naphthalene⁶¹ and biphenyl⁶² using a variety of catalysts. Ethylation and propylation of aromatics such as naphthalene and biphenyl can give rise to varying mixtures of isomeric alkylated products depending on the nature of the catalyst and the severity of the reaction conditions. For example, in the isopropylation of naphthalene nonisomerizing catalysts such as BF₃-H₃PO₄, HF, BF₃·Et₂O or P₂O₅ gave higher 1-isomer: 2-isomer ratios than the isomerizing catalysts.⁶³ Mechanistic studies on the alkylation of arenes with alkenes, particularly with propene, have been carried out using many techniques, such as standard kinetic methods and ${}^{14}C$ - and ${}^{2}H$ -labeling methods. These aspects are well documented.⁵ More recent kinetic and mechanistic studies include: kinetic modeling of the liquid-phase alkylation of toluene with n-butenes;⁶⁴ thermodynamic analysis of the alkylation of phenols with branched alkenes;⁶⁵ kinetics of benzene alkylation with 1-decene in the presence of soluble quantities of AlCl₃ catalyst complex;⁶⁶ isomer distribution studies on the Friedel-Crafts alkylation of toluene with cycloalkenes;⁶⁷ a mathematical model for the homogeneous alkylation of benzene by ethylene in the presence of an aluminum chloride catalyst;68 and kinetics of cycloalkylation of phenol and cresols by cyclohexene and cyclopentene using H₂SO₄ as catalyst.⁶⁹

Friedel-Crafts alkylation of aromatics with a wide variety of alkenes has led to many useful applications in organic synthesis. Particularly, much effort has been devoted to choosing appropriate catalysts and reaction conditions to produce predominantly or selectively one product isomer. This is especially true of alkylations with both linear and branched C₄, C₅ and higher alkenes as the alkylating agents. Other factors such as steric and electronic interactions could also lead to regioselective formation of products as seen in equations (22) to (26).⁷⁰⁻⁷⁴



The effect of the acidity of polyfunctional zeolite catalysts on their activity in benzene alkylation by propene has been examined.^{75a} Crystalline zeolites exist with a variety of characteristic pore and channel sizes. In the case of ZSM-5 zeolite, the interconnected channels formed by 10-membered rings of oxygen atoms allows certain benzene derivatives to fit rather closely, diffuse into the pores, and, after undergoing a reaction, diffuse out. This is the origin of the high *para* selectivity in reactions such as alkylation of toluene by methanol catalyzed by zeolites like ZSM-5 and related modifications.^{75b,c}

Olah⁷⁶ has found perfluoroalkanesulfonic acids, such as perfluorodecanesulfonic acid (PDSA, $C_{10}F_{21}SO_3H$), to be very effective catalysts for regioselective isopropylation of naphthalene (equation 27).



Controlled *ortho* alkylation of phenol by 1-hexene to produce the mono- or di-alkylated product has been achieved by using aluminum phenolate as the catalyst under carefully optimized reaction conditions.⁷⁷ Sterically hindered phenols such as (15) are of particular interest as antioxidants (Scheme 4).



72% (in product mixture) mole ratio of phenol:1-hexene:catalyst = 1:2.7:0.3

Scheme 4

Shudo and coworkers⁷⁸ have reported that O,O-diprotonated nitroalkenes (16) behave as novel electrophilic species which can efficiently alkylate aromatics such as benzene, anisole, chlorobenzene, naphthalene, *etc.* under extremely mild conditions (equation 28). The reaction enables the synthesis of α -arylated ketones (17) which are difficult to synthesize by conventional Friedel–Crafts reactions.



Crotonic acid has been used as a carboxyalkylating agent in its reaction with benzimidazolin-2-ones (equation 29).⁷⁹ It must be emphasized, however, that when long chain linear alkenes are used as alkylating agents, one frequently encounters a mixture of products containing rearranged forms of the alkyl group. With isobutene as the alkylating agent, the products are found to be exclusively *t*-butylated derivatives. A number of isomerically pure *t*-butylated aromatics have been thus prepared. The regioselectivity is also appreciably enhanced because of the greater steric factors operative. Alkylations with branched C₅ and higher alkenes, in general, are accompanied by side reactions,⁸⁰ such as reorientation, disproportionation, fragmentation, polymerization, hydride transfer and skeletal rearrangement of the alkylating group. However, use of moderated reaction conditions, including a milder catalyst, can afford one product selectively. Examples illustrating the synthetic utility of alkylations with isobutene and higher branched alkenes are shown in equations (30) to (34).^{81–85}

Alkylation of phenols with isobutene has also been accomplished⁸⁶ using a heterogeneous catalyst system comprising an aluminum phenoxide bonded to a solid polymeric resin having acidic functional





groups. A strongly acidic ion-exchanger has been employed⁸⁷ as the catalyst for the Friedel–Crafts alkylation of phenols with disobutene to afford p-t-octylphenol. Such alkylations in heterogeneous media are of great importance due to the ease of separation of the product from the catalyst.

The *t*-butyl group has been extensively utilized as a positional protective group in both separation and synthesis problems. This is due to the ease with which *t*-butyl groups can be introduced (by alkylation) and removed (by dealkylation). Tashiro's review⁸⁸ covers this topic very well. Examples to illustrate the utility of the *t*-butyl group for the selective preparations of phenolic compounds (equations 35 and 36),







1,2-di- and 1,2,3-tri-substituted benzenes (equations 37 and 38), diarylalkanes (equation 39), 2-monoand 2,2'-di-substituted biphenyls (equations 40 and 41), [2,2]metacyclophanes (equation 42) and carbazoles (equation 43) are shown.

Alkylation of arenes with cycloalkenes (cycloalkylation) is also well documented in the literature. A variety of Lewis and Brønsted acids have been used as catalysts for these reactions. These reactions share many similarities with the alkylation reactions of arenes with other alkylating agents such as alkenes. The selectivity of the cycloalkylation reaction depends upon the conditions and catalyst. A few examples of cycloalkylation reactions are presented in equations (44) to (46).^{89–91}



1.8.3.3 Alkylation with Alcohols, Ethers, Esters, Epoxides and Lactones

The alkylations of arenes by alcohols, ethers and esters, as well as with epoxides and lactones, are of considerable interest, and constitute a significant part of the field of Friedel–Crafts alkylations. Some of the earliest examples are the alkylations of arenes with primary alcohols,^{92a} esters^{92b} and ethers.^{92c} Subsequent work over the years has revealed that these alkylations are often accompanied by various side



reactions, such as isomerization, fragmentation and dealkylation, together with side-chain and positional rearrangements.

Early studies have shown that, compared to alkylations with alkyl halides or secondary alcohols, alkylations with primary alcohols require larger amounts of the Lewis acid catalyst (*ca.* 2 mol for 1 mol of alcohol) and longer heating at higher temperatures. The relative ease of alkylation with alcohols follows the order: benzyl, allyl, tertiary > secondary > primary > methyl. Besides using catalysts such as BF₃ and AlCl₃, many milder catalysts have been used for the alkylation of aromatics with alcohols, ethers and esters. These include weakly acidic metal halides, such as AlCl₃–MeNO₂, FeCl₃, TiCl₄, SnCl₄ and ZnCl₂; protic acids such as H₃PO₄, HF, H₂SO₄ and polyphosphoric acid; solid superacids such as Nafion-H; and inorganic acidic oxides such as P₂O₅, alumina, natural and synthetic aluminosilicates and zeolites. These milder catalysts afford solely or mostly the rearranged secondary alkylates.

A careful choice of catalyst and reaction conditions is necessary to obtain selective Friedel–Crafts alkylation. The following examples illustrate reactions with alcohols as the alkylating agent (equations 47 to 51).^{44,93-96}

The *t*-butylated naphthalene derivative (18; equation 49) is of interest as an intermediate⁹⁵ for the synthesis of fungicides. Tetrahydronaphthalene derivatives, such as (20), have been evaluated⁹⁶ as antifertility agents. Sugita *et al.*⁹⁷ have studied the AlCl₃-catalyzed Friedel–Crafts alkylation of benzene with 1-phenyl-2-propanol and 2-phenyl-1-propanol in the presence of additives, such as CuCl₂, Cu₂Cl₂ and decalin. Highly regioselective formation of 1,1-diphenylpropane was observed with Cu^I or Cu^{II} chloride as the additive. Some pertinent results of synthetic value are shown in Scheme 5. The addition of decalin diminished the alkylation reaction to give the reduction product 1-phenylpropane.





Nutaitis and Gribble⁹⁸ have found that arenes react with NaBH₄ in CF₃CO₂H to afford 1,1,1-trifluoro-2,2-diarylethanes, which were otherwise difficult to prepare. They proposed a Friedel–Crafts-type (Baeyer) condensation to give an intermediate carbinol (21; equation 52), which further alkylates benzene to give the observed products. The reaction with sterically congested arenes such as mesitylene and durene stops at the carbinol stage (equation 53).

Isopropylation of toluene by isopropyl halo- and alkyl-sulfonates (equation 54) has been performed by Olah *et al.*⁹⁹ A variety of catalysts, such as AlCl₃, AlCl₃-MeNO₂ and Nafion-H were employed, and the isomer distribution (*o*, *m*, *p*) in the product was determined. Sartori *et al.*¹⁰⁰ have recently reported an unusual Friedel-Crafts alkylation of lithium phenolates with ethyl pyruvate in the presence of AlCl₃ to afford α -(2-hydroxyphenyl)ethyl lactates (22), which are the precursors of 3-methyl-2,3-dihydrobenzo-furan-3-ols (23; Scheme 6).



Scheme 6

Optically active 2-arylalkanoic acid esters (24) have been prepared (equation 55)¹⁰¹ by the Friedel-Crafts alkylation of arenes with optically active esters in the presence of AlCl₃. Piccolo and coworkers¹⁰² have also found that alkylation of benzene with optically pure (S)-methyl 2-(chlorosulfonyloxy)- or 2-(mesyloxy)-propionate, in the presence of AlCl₃, gave (S)-methyl 2-phenylpropionate in good chemical (50–80%) and excellent optical yield (>97% *ee*, as determined by rotation), with inversion of configuration at the attacking carbon atom. In view of the ready availability of optically pure lactic acid derivatives, this reaction offers an attractive general method of preparing optically active compounds by further elaboration of the phenyl and ester groups in the alkylated product (25; R¹C*H(Ar)Y, R¹ = Me; Y = CO₂Me, CO₂Et).

$$R^{3} * CO_{2}R^{2} + ArH \xrightarrow{AlCl_{3}} Ar * CO_{2}R^{2}$$

$$R^{1}, R^{2} = alkyl; \qquad R^{1}, R^{2} = alkyl$$

$$R^{3} = Cl, Br, CISO_{3} \qquad (24)$$

Steroidal alkenol esters have been employed as alkylating agents to introduce aryl groups (equation 56).¹⁰³



Examples of Friedel–Crafts alkylations of arenes with acyclic ethers are rare in the literature. With cyclic ethers (epoxides, oxetanes, *etc.*), however, a number of interesting alkylation reactions are known. With epoxides alkylation proceeds with ring opening to give β -hydroxyarylalkanes, which, depending on catalyst type and reaction temperatures, can react further with aromatics to produce 1,2-diarylalkanes. Milstein¹⁰⁴ found that benzene, for example, was alkylated by 1,2-epoxypropane in the presence of AlCl₃ to give a 64% yield of 2-phenyl-1-propanol (equation 57); the isomeric 1-phenyl-2-propanol was observed only in trace amounts. This was rationalized by suggesting that the AlCl₃-coordinated epoxide complex initially undergoes ring opening to give the most stable ion pair (**26**) before attacking the benzene (equation 58).



Suga et al.¹⁰⁵ studied the SnCl₄- or AlCl₃-catalyzed Friedel-Crafts alkylation of benzene with (R)-(+)-1,2-epoxybutane (27) to investigate the stereochemistry of the ring-opening process. The primary products were (R)-(-)-(28), with almost 100% optical purity, and (S)-(-)-(29), with 55-56% optical purity (Scheme 7).

Under the influence of mild acid catalysts, such as MgBr₂, ZnCl₂ and BF₃·Et₂O, epoxides are initially isomerized to an aldehyde which then alkylates the arene to afford 1,1-diarylated products (Scheme 8). With strong Lewis acids, however, an S_N 2-type ring opening of the activated epoxide by the reactive aromatic component occurs to give the normal Friedel–Crafts alkylation product. Thus, in those cases where ring opening of the epoxide gives a more stable secondary or tertiary carbocation, good yields of diarylated products are obtained. Some examples using anisole as the arene are shown in equations (59) and (60).¹⁰⁶



major product

(59)

The alkylation of arenes by an oxetane affords the carbinol via a ring-opening reaction as seen in equation (61).¹⁰⁷ Suga and coworkers¹⁰⁸ have examined the stereochemistry of the ring-opening process by carrying out the alkylations of benzene with (+)-2-methyloxetane in the presence of AlCl₃, SnCl₄ or TiCl₄.



Friedel–Crafts alkylations of arenes with higher cyclic ethers such as tetrahydrofurans and pyrans are often accompanied by many side reactions such as intramolecular alkylation, condensation and further transformations of the initial alkylated product. However, it is possible to obtain one product selectively by choosing the proper reaction conditions. An example is shown in equation (62).¹⁰⁹



The ease with which cyclic ethers alkylate arenes is in the order¹¹⁰

 $\triangle > \bigcirc > \bigcirc > \bigcirc > \bigcirc$

It also decreases with the degree of branching,¹¹¹ as in



Brauman et al.¹¹² have alkylated benzene with (S)-(+)-2-methyltetrahydrofuran to give (R)-(-)-4-phenyl-1-pentanol. The reaction occurred with at least 35% inversion of configuration (equation 63). This result has been explained by an S_N 2-type attack of benzene on a cyclic ion pair such as in (30). Studies on stereospecific Friedel-Crafts reactions by Sugita and coworkers¹¹³ have also lent support to the suggestion that stereospecifity might be general for Friedel-Crafts reactions of cyclic compounds.





Aziridines also are capable of alkylating arenes to form β -arylethylamines. Milstein's work¹¹⁴ on the alkylation of arenes with 2-methylaziridine revealed the formation of two isomeric products (equation 64). This was explained by the occurrence of two competing pathways, (a) and (b), as shown in Scheme 9.





The Friedel-Crafts alkylation of arenes with lactones provides a convenient method for the synthesis of arylalkanoic acids. However, depending on the nature of the arene, catalyst, temperature and reaction time, the resulting arylalkanoic acids can undergo partial or complete cyclization to the corresponding cyclic ketones (Scheme 10). Equations (65) to (68) show examples of Friedel-Crafts alkylations using lactones as the alkylating agents.¹¹⁵⁻¹¹⁸





Brauman *et al.*¹¹⁹ have examined the mechanism of the alkylation of lactones by reacting optically active γ -valerolactone with benzene in the presence of AlCl₃. They found the reaction to proceed with nearly 50% net inversion of configuration (equation 69). The mechanism put forth to account for this result is similar to that proposed for the observed stereochemistry of the Friedel–Crafts alkylation of (S)-(+)-2-methyltetrahydrofuran¹¹² (vide supra).



Sultones have been shown to alkylate arenes in the presence of $AlCl_3$ (equation 70)¹²⁰ to give aryl-alkenesulfonates.



1.8.3.4 Alkylation with Di- and Poly-functional Alkylating Agents

In 1884, Friedel and Crafts reported that the alkylation of benzene or toluene with dichloromethane in the presence of AlCl₃ afforded the corresponding diarylmethanes, besides other products derived from further alkylation and other transformations.¹²¹ The products from the reaction with benzene are shown

in equation (71). Subsequent work on the alkylation reactions with toluene, other substituted arenes and naphthalene describes the efforts to understand the formation of a number of alkylated products derived from competing processes, such as isomerization and dealkylation.



1,2-Dihaloethanes alkylate arenes under Friedel–Crafts catalysis by Lewis acids to afford 1,2-diarylated ethanes. Dolgov and Larin¹²² suggested that the reaction proceeds *via* the intermediate monoarylated haloethane (Scheme 11) which affords the final product *via* a phenonium-type intermediate.



Scheme 11

The reaction of 1,2-dihalo-2-phenylethane with arenes in the presence of TiCl₄ afforded a number of 1,1-diaryl-2-haloethanes in 40–73% yield (equation 72).¹²³

$$X = Cl, Br; Ar = Ph, p-MeC_6H_4, p-PhOC_6H_4, etc.$$

Friedel–Crafts alkylations with higher dihaloalkanes generally give a mixture of products derived from processes such as isomerization, multiple addition and dehydrogenation (caused by AlCl₃), as seen from the examples shown in equations $(73)^{124}$ and $(74)^{.125}$



Ransley¹²⁶ has examined in detail the AlCl₃-catalyzed alkylation of benzene with a series of α, ω -dichloroalkanes, and suggested possible mechanisms for the formation of the various phenylated products. With 2,4-dihalohexanes as the alkylating agents for benzene, Gelin *et al.*¹²⁷ found that about 90% of the products consisted of 1,4-dimethyltetralin and compounds derived from it. The alkylation of 1,3-dimethylbenzene by 2,5-dichloro-2,5-dimethylhexane in the presence of AlCl₃ has been reported¹²⁸ to give a moderate yield of the tetrahydronaphthalene derivatives, as in equation (75).

Stereospecific (ranging from 15 to 60%) alkylation of benzene with optically active 1,2-, 1,3-, 1,4- and 1,5-dihaloalkanes in the presence of AlCl₃ has also been reported.¹²⁹ In the case of 1,3- and 1,5-dihaloal-


kanes, alkylation with inversion of configuration at the secondary halogen positions occurred predominantly.

Branched dihaloalkanes in general give a mixture of products resulting from processes such as isomerization and cycloalkylation. These processes in turn depend on the structure of the dihaloalkanes, the strength of the catalyst and the severity of the reaction conditions.

Other polyhaloalkanes, such as dichloromethane, chloroform and carbon tetrachloride, have also been used as alkylating agents. The examples in equations (76) to (79) serve to illustrate this aspect.¹³⁰⁻¹³²



(ref. 132) (79)

1.8.3.5 Haloalkylation

Haloalkylation of arenes by unsymmetrical or mixed halides occurs when one of the halogen atoms has a greater reactivity towards the Lewis acid catalyst. It is then possible to isolate the comparatively less reactive haloalkylated products. An example is the reaction of chloroform with benzene in the presence of AlCl₃. The major product (90%) is dichlorodiphenylmethane (equation 80).¹³³

$$+ CHCl_3 \xrightarrow{AlCl_3, r.t.} \left(\swarrow_2 \right)_2 CCl_2$$
(80)

major product

Haloalkylations have also been carried out with branched, unsymmetrical simple di- and poly-halides, straight-chain, unsymmetrical simple di- and poly-halides, and mixed di- and poly-halides. Schmerling *et al.*¹³³ have studied the Friedel–Crafts reactions of benzene with 1,3-dichloro-3-methylbutane in the presence of AlCl₃ and obtained a 28–29% yield of 1-chloro-3-methyl-3-phenylbutane (equation 81).



In the case of dihalides containing two different halogens, alkylation with an arene is influenced by factors such as reactivity and steric accessibility of the halogens with respect to the Lewis acid catalyst. Bugrova and Tsukervanik¹³⁴ have demonstrated this by carrying out the four reactions in equations (82) to (85). Higher reactivity of Br over Cl is seen in equations (82) and (83), and steric effects are demonstrated in equations (84) and (85).



Olah and Kuhn¹³⁵ found that fluorochloro-, fluorobromo- and fluoroiodo-alkanes are effective chloro-, bromo- and iodo-alkylating agents, respectively, in Friedel–Crafts alkylations of arenes in the presence of boron halide catalysts (equation 86). Boron trihalides catalyze reactions of only the C—F bonds, but not of the C—Cl, C—Br or C—I bonds. The order of reactivity of the catalysts was found to be BI₃ > BBr₃ > BCl₃ > BF₃, and that of the carbon–halogen bonds C—F > C—Cl > C—Br > C—I.

ArH +
$$F(CH_2)_n X \xrightarrow{BX_3} Ar(CH_2)_n X$$
 (86)
 $n > 1; X = Cl, Br, I$

Allyl halides have been used to effect selective haloalkylations catalyzed by proton acids. For example, 3-chloropropylene reacts with benzene and toluene in the presence of H_2SO_4 to give haloalkylated products almost exclusively (equation 87).¹³⁶ The same result is obtained when ZnCl₂ is used as the catalyst.¹³⁷



Selective haloalkylation of aromatic compounds has also been achieved by employing alkyl haloalkyl ethers as the haloalkylating agents, as seen from equations (88)¹³⁸ and (89).¹³⁹ However, the use of chloromethyl ethers in chloromethylation (equation 89) is discouraged due to the carcinogenic nature of the ether substrate. As an alternative, chloromethylation reactions can also be carried out using paraformal-dehyde-HCl in the presence of ZnCl₂ or SnCl₄ as catalyst.²



Unsaturated alkyl halides, alcohols and ethers can also act as alkylating agents to give products derived by alkylation at one or more reactive sites. Tsukervanik and Yuldashev¹⁴⁰ examined the reaction of vinyl chloride with toluene and anisole in the presence of AlCl₃ by varying the ratio of reactants. For example, with anisole (equation 90) two products were formed. The diarylated product resulted from reaction of anisole with both reactive centers in vinyl chloride.



The nature of the catalyst also influences the course of the reaction. α -Substituted vinyl halides react more selectively with arenes to give diarylated products.

Allyl alcohols and ethers also react with arenes in the presence of Lewis and Brønsted acid catalysts to give alkylated products. The nature of the product depends upon the nature of the Lewis acid. However, in most cases the products are mixtures of regioisomeric allylated arenes.

Dienes also can alkylate arenes to give products which are generally cyclic. For example, isoprene reacts with 1,2,4-trimethylbenzene in the presence of H_2SO_4 to give an indane derivative (equation 91).¹⁴¹ A large number of phenols have been used for alkylations with dienes. The products are either allylated phenols, or products derived by further ring closure of the phenolic oxygen with the side chain derived from the diene.



1.8.3.6 Alkylation with Alkanes

Friedel–Crafts alkylation of arenes with alkanes was studied extensively by Olah and coworkers^{142a} as part of the general investigation of electrophilic reactions at single bonds. Benzene was alkylated with C_1-C_5 alkanes, alkane–alkene, and alkane–alkylbenzene mixtures in the presence of anhydrous HF–SbF5 (HSbF₆). Products arising from alkylation by the alkanes as well as side reactions, such as isomerization and disproportionation, were observed. In the presence of added alkene, greater amounts of both alkane and alkene alkylation products resulted. When a *t*-alkylbenzene was present, products deriving from both transalkylation and alkylation by the alkanes were obtained. Kröger and coworkers^{142b} studied the alkylation of benzene with cyclohexane in the presence of catalysts such as HBr–AlBr₃, HCl–AlCl₃ and HF– SbF₅, with isobutane as the promoter, under varying temperatures and reaction times. Four representative groups of hydrocarbons were found including cycloalkylbenzenes, substituted indanes or tetralins, C₁–C₆ alkylbenzenes and isomeric dicycloalkyls. Their formation was explained by a competition between alkylation. Phenylcycloalkyl cations and phenylalkyl cations were proposed as intermediates. Mechanisms for the formation of complex product mixtures were also discussed.

Miethchen *et al.*¹⁴³ have carried out ultrasound-induced alkylation of benzene with 2,2,4-trimethylpentane in the presence of various highly acidic catalysts such as CF₃SO₃H–SbF₅, H[B(SO₃CF₃)₄], AlBr₃, HF-TaF₅, HF-NbF₅ and HF-TiF₄ (equation 92). The effect of temperature and duration of the reaction on the yield and composition of *t*-butylbenzene and di-*t*-butylbenzenes was also determined.



Laszlo and coworkers¹⁴⁴ achieved the direct arylation of adamantane by employing FeCl₃-doped K10 montmorillonite (equation 93). Depending upon the amounts of adamantane, FeCl₃, the catalyst and the reaction time, varying proportions of mono- and di-arylated adamantanes were obtained. The results were taken to support the intermediacy of a 1-adamantyl cation.



1.8.3.7 Cycloalkylation

A Friedel–Crafts cycloalkylation reaction occurs intramolecularly when an aryl-substituted monofunctional compound cyclizes to attach a new ring to the aromatic nucleus. In many instances, this process occurs during the course of the reaction between a difunctional alkylating agent and an aromatic compound — the intermediate arylated monofunctional compound subsequently undergoes cycloalkylation. However, only those cycloalkylation reactions involving intramolecular ring closures will be discussed at length here.

Arylalkyl halides undergo cycloalkylation reactions in the presence of a variety of Lewis acids. Depending upon the nature of the Lewis acid, varying amounts of cycloalkylated and other products can be obtained. Equations (94) to (96) show some cycloalkylation reactions of arylhaloalkanes.¹⁴⁵



Yang and Young¹⁴⁶ have observed a novel Friedel–Crafts metallocyclization of halomethyl(arylphosphine)platinum(II) complexes (**30**) in the presence of AgBF₄ and a triarylphosphine to afford cationic metallacyclic species (**32**; equation 97).



Alkylation of Carbon

Cycloalkylations with primary and secondary phenylalkyl chlorides are often accompanied by rearrangements. The ease of formation of a six-membered ring (tetralin derivative) is much greater than that of a five-membered ring (indane derivative), based on entropy and strain factors. Khalaf and Roberts¹⁴⁷ have performed stereochemical studies on cycloalkylation reactions to determine the effect of stereochemistry and steric factors in the formation of six-membered rings.

Aryl esters such as those of α -halopropionic and α -halobutyric acids undergo cycloalkylation reactions, probably *via* preliminary Fries rearrangement to β -halohydroxy ketones, giving the corresponding substituted indanones (equation 98).¹⁴⁸



Similarly, arylalkylamides, haloalkylarylamines, haloalkylaryl ketones and haloalkyl ethers undergo Friedel–Crafts cycloalkylation to give six- or five-membered ring closure products. Representative examples of these reactions are shown in equations (99) to (103).^{150–154} An excellent review by Brad-sher¹⁴⁹ on the cycloalkylation of arylcarbonyl compounds is available.



Charpentier-Morize and coworkers¹⁵⁵ have developed conditions for the cycloalkylation of γ -aryl- α -trifluoromethyl ketones selectively to either the cycloalkene (33) or the cycloalkyl chloride (34; Scheme 12) by varying the catalyst.

Friedel-Crafts Alkylations



Scheme 12

Cycloalkylations of arylalkanols and related systems are very well known in the literature. Many of the earlier examples were fraught with ambiguities in terms of product identification. However, Khalaf and Roberts¹⁵⁶ have carefully investigated an extensive series of variously substituted arylalkanols to help clear the ambiguities and provide more insight into the mechanism. Some examples from their study are shown in equations (104) to (106).



The cycloalkylation reactions of suitable arylalkyl carbinols leading to tricyclic diterpenoids have been examined for their synthetic and mechanistic significance,¹⁵⁷ and the syntheses of some octahydrophenanthrenes, such as in equation (107), have been reported.



Khalaf and coworkers¹⁵⁸ have described a facile synthesis of substituted acenaphthenes (equation 108), and Sharma *et al.*¹⁵⁹ have reported a synthesis of the precursor for 8-methylbenzo[j]fluoranthene using the cycloalkylation of an arylalkylcarbinol precursor (equation 109).

Khalaf and coworkers¹⁶⁰ recently reported an interesting case of carbonyl retardation of cycloalkylation of the enone (35) in the presence of AlCl₃. The isomeric enone (36), however, reacted to give the perinaphthenones (37) and (38) (Scheme 13). Numerous other examples of cycloalkylation reactions of aryl-substituted enones exist in the literature.

The cycloalkylations of arylalkyl epoxides were studied in detail by Taylor and coworkers,¹⁶¹ in view of their applications¹⁶² in natural product synthesis. Equation (110)^{161a} shows an example of this reaction.



Superacids such as HF-SbF₅ have been used to cyclize alkyl phenyl ketones to the corresponding tetralone derivatives in good yield, as shown by the example in equation (111).¹⁶³



Multiple intramolecular ring closures of aryl-substituted unsaturated long chain alcohols, acids, acid chlorides and ethers in the presence of Friedel–Crafts catalysts have been extensively employed to synthesize polynuclear hydroaromatic hydrocarbons and polycyclic ketones. This is illustrated by two examples shown in equations (112)¹⁶⁴ and (113).¹⁶⁵ The application of stereospecific cycloalkylations in approaching the synthesis of complex organic molecules has been reviewed by Barclay.¹⁶⁶



As pointed out earlier, cycloalkylation reactions can also result from the reaction of arenes with bifunctional alkylating agents or in cases where bifunctional intermediates are involved. An example of the former is shown in equation (114),¹⁶⁷ where a chiral, lactone-substituted alkene cyclizes with the arene to produce a chiral tetrahydronaphthalene derivative.



1.8.3.8 Transalkylation, Isomerization, Disproportionation and Dealkylation

The general process of transfer of alkyl groups between aromatic rings is called transalkylation. When an alkylbenzene is converted into a mixture of benzene and dialkylbenzene which may further afford a trialkylbenzene, the process is termed disproportionation. Dealkylation of an alkylbenzene is a process where an alkyl group is removed to form the corresponding alkane. The isomerization of di- or polyalkylbenzenes refers to the process where the alkyl groups on the ring change their relative orientations. Such a process can occur either in an inter- or an intra-molecular fashion depending on the nature of the alkyl groups on the ring.

Transalkylation of alkylbenzenes, polyalkylbenzenes and other arenes can be brought about by a variety of catalysts including Lewis acids, Brønsted acids and various zeolites and silicates with or without being doped with various transition metals or their oxides. There has been a particularly explosive growth in the volume of literature pertaining to the use of various natural and modified zeolites. Recent developments include the study and applications of shape-selective catalysis by zeolites. Much of the work is patented, and largely applies to industrial processes.

Friedel–Crafts alkylation reactions are, in general, accompanied by isomerization processes. Olah *et al.*¹⁶⁸ reported the results of the water-promoted, AlCl₃-catalyzed isomerization of o-, m- and p-di-t-butylbenzene. No *ortho* isomer was present in the equilibrium mixture. The isomerization of o-di-t-butylbenzene was very rapid largely due to relief of steric strain. In these and other related sterically hindered arenes, intramolecular isomerization and not dealkylation was observed. Isomerization of diand mono-methylnaphthalenes, catalyzed by HF–BF₃, was also reported.¹⁶⁹ Isomerization of *n*-alkyltoluenes and -xylenes, catalyzed by AlCl₃ at room temperature, afforded chiefly *m*-*n*-alkyltoluenes and *m*-*n*-alkylxylenes, respectively.¹⁷⁰ The process leading to the *meta* isomer has a lower energy than the other processes.

Due to the greater industrial importance of benzene and xylenes than toluene, a large number of studies have concentrated on the transalkylation and disproportionation of these two arenes.

Camacho et $al.^{171}$ have found that during the alkylation of benzene with ethylene in the presence of AlCl₃ as the catalyst, both alkylation and transalkylation (of the produced ethylbenzene) occur simulta-

neously. They also found that the ethylene absorption rate was almost constant when the concentrations of polyalkylates higher than diethylbenzene (formed as in equation 115) were negligible.



Among alkylarenes, methylarenes are unique in that they may undergo intramolecular isomerizations without any observable transalkylations. The other alkylarenes such as ethyl-, *n*-propyl- and *n*-butyl-benzenes, however, undergo isomerization and disproportionation reactions without rearrangements in the alkyl side chains. The mobilities of the alkyl groups in alkylbenzenes were measured using ¹⁴C-labeling techniques,¹⁷² and showed the ethyl group to have higher mobility than other *n*-alkyl groups. However, recent studies¹⁷³ on the transalkylations of *p*-ethylalkylbenzenes with ¹⁴C-labeled benzene, using AlBr₃ as catalyst, revealed no marked difference in the mobilities of the ethyl and the other alkyl groups. Reactivity-selectivity relationships in the transalkylation of monoalkylbenzenes revealed an inverse sense for electrophilic reagents.¹⁷⁴ The mechanism and product distribution during the transalkylations of toluene¹⁷⁵ and alkyl and dialkylarenes¹⁷⁶ catalyzed by ZSM-5 zeolites have been discussed. The significance of shape-selective catalysis occurring in these processes was also pointed out. Again, it must be emphasized that ZSM-5 exhibits high *para* selectivity, and this property has been utilized for the production of aromatics such as *p*-xylene and *p*-ethyltoluene.

Buchanan *et al.*¹⁷⁷ conducted model studies on the hydrocracking of coal by studying in detail the action of SbCl₃ melts with (and in some cases without) AlCl₃ on α,ω -diphenylalkanes. They observed selective sp^2-sp^3 bond cleavage for the four classes of diphenylalkanes studied, as shown in equations (116) to (118). Notably, 1,3-diphenylpropane and 1,4-diphenylbutane gave only the indan and tetralin derivatives, respectively.



Besides the transarylalkylation reactions described above (equation 116), debenzhydrylation and transbenzhydrylation reactions are also well known, and these have been extensively reviewed.¹⁷⁸ Applications of the benzhydryl group as a protective group have also been discussed.

Transalkylation reactions of simple aromatics such as benzene and toluene with various coals have proved to be a valuable means of investigating the composition and structural features of coal.¹⁷⁹

Wu and Leu,¹⁸⁰ and Bursian *et al.*¹⁸¹ have carried out disproportionation and transalkylation of toluene with pseudocumene, using various modifications of mordenite zeolite catalysts. The disproportionation

of o- and p-cresol, and the transmethylation between isomeric xylenols and phenol have been examined using a chromia-alumina catalyst.¹⁸² It was found that p-cresol underwent disproportionation more readily than o-cresol, whereas m-cresol did not undergo this reaction. Also 2,6-, 2,4- and 3,4-xylenols underwent transalkylation reactions with phenol, whereas 3,5-xylenol did not. Mechanistic interpretations for these observations were also given. Transalkylation of phenol with ethyl- and isopropyl-benzene and ethyltoluene has also been reported.¹⁸³

Trans-t-butylation and de-t-butylation reactions are facile processes, and form the basis of many applications of the t-butyl group as a positional protective group. This aspect was well reviewed by Tashiro,⁸⁸ and is also discussed in Section 1.8.3.2. Tashiro and coworkers¹⁸⁴ have more recently applied the trans-tbutylation and de-t-butylation reactions to prepare metacyclophanes (**39**) and related compounds.



(39) R = Me, OH, F

Olah *et al.*¹⁸⁵ showed Nafion-H to be a very efficient de-*t*-butylation agent for aromatic compounds, and Rosevear and Wilshire¹⁸⁶ have applied the trans-*t*-butylation reaction to the synthesis of a variety of ultraviolet absorbers of the general structure (40), as shown in Scheme 14.



Transbenzylation reactions are also used on occasion for deprotection as seen from the reaction shown in equation (119).¹⁸⁷

In 1937, Ipatieff and Pines¹⁸⁸ found that the ease of dealkylation of alkylbenzenes in the presence of AlCl₃ at 65–80 °C to give the corresponding alkane decreased in the order of the side chains: *t*-butyl > *s*-butyl > isopropyl. Ethyl and methyl groups were not cleaved.

From their studies on dealkylation of alkylbenzenes with AlCl₃ at 100 °C, Roberts *et al.*¹⁸⁹ found that, in general, the catalyst and experimental conditions required to produce dealkylation are forceful enough to effect several other reactions, such as disproportionation, isomerization, rearrangement of the side chain and fragmentation. The latter refers to a process in which the alkylbenzene affords alkanes and alkylbenzenes with fewer carbons than those in the original alkyl side chain. Formation of alkanes in the dealkylation process can take place even in the absence of an added hydride donor agent. In this case, a second molecule of the alkylbenzene acts as the hydride donor.



Due to the ease of dealkylation of *t*-alkylbenzenes, such as *t*-butylbenzene, some interesting applications of this process in both the synthetic and mechanistic areas have resulted. The intermediacy of the *t*-butyl cation was hinted to account for the observed products. Thus, Knight *et al.*¹⁹⁰ obtained excellent yields of pivalic acid by dealkylating *t*-butylbenzene and 1,3-dimethyl-5-*t*-butylbenzene in the presence of carbon monoxide and BF₃·H₂O as the catalyst (equation 120). They also found that H₃PO₄, H₂SO₄ and MeSO₃H were ineffective for the dealkylation–carbonylation of *t*-butylbenzene.



Brouwer¹⁹¹ and Olah *et al.*¹⁹² obtained direct evidence for the *t*-butyl cation intermediate by ¹H NMR studies of *t*-butylbenzene in superacid solutions at low temperature. Fărcasiu and Schlosberg¹⁹³ demonstrated that, in the presence of carbon monoxide, the *t*-alkyl cation generated *in situ* from the dealkylation of a tertiary alkyl arene resulted in acylation of the dealkylated arene, as shown in Scheme 15.



Scheme 15

Olson and Haag¹⁹⁴ observed a high selectivity for xylene isomerization versus disproportionation in the presence of ZSM-5 zeolite, and ascribed it to a transition state selective process rather than shape-selective diffusion. They were also able to produce p-xylene in high selectivity (up to 80%) from toluene disproportionation by using suitable modifications of the ZSM-5 catalyst. This selectivity was said to

arise from shape-selective diffusion due to the unique pore size of the catalyst. Ratnasamy *et al.*¹⁹⁵ reported that shape-selective zeolites isomerized *m*-xylene to *p*- and *o*-xylene with higher selectivities.

Balaban et al.¹⁹⁶ have reported the catalytic automerization of $[1^{-13}C]$ - or $[3^{-13}C]$ -phenanthrene in the presence of a 1:1 mixture of AlCl₃-NaCl. They found that $[1^{-13}C]$ phenanthrene transferred part of the label to C-3, but $[3^{-13}C]$ phenanthrene transferred the label to two positions, *i.e.* mainly C-1 and to a lesser extent C-4, as shown in Scheme 16. Possible reaction mechanisms to account for these observations were also proposed.



Scheme 16

1.8.4 ALKYLATION OF ALKENES

Ipatieff and coworkers¹⁹⁷ carried out the first alkylation with alkenes and branched and normal chain alkanes (except methane and ethane) in the presence of AlCl₃ as the catalyst. The sulfuric acid catalyzed alkylation reaction of arenes and isoalkanes, developed in 1938, is a still widely used industrial process¹⁹⁸ to produce alkylates with high octane numbers. For synthetic applications, however, Friedel–Crafts-type alkylations of alkenes and alkanes have limited value since they tend to give mixtures of products, including oligomers of alkenes.¹⁹⁹

Oligomerization of aliphatic alkenes such as ethylene and propene, and aromatic alkenes, such as styrene, indene and their derivatives, catalyzed by a variety of Lewis acids, such as AlCl₃, BF₃, AlBr₃, and Brønsted acids, such as H₃PO₄, is a well-studied subject.¹⁹⁹ These reactions are important, particularly in the industrial arena.

Olah and coworkers,^{200a} and Mayr and Striepe^{200b} discussed the scope and limitations of aliphatic Friedel-Crafts alkylations. In particular, they considered factors that would favor reactions of the type shown in equation (121), where an alkene is alkylated by an alkyl halide.²⁰¹ They reasoned that formation of the 1:1 addition products (42) can be expected, if (41) reacts faster with the alkene than (42), otherwise higher addition products will be formed. Mayr²⁰² suggested that the relative dissociation rates of (41) and (42) induced by the Lewis acid should reflect their relative rates of addition to a common alkene. Furthermore, it was assumed that the solvolysis rates in 80% ethanol were proportional to the Lewis acid induced dissociation constants. A few examples where good yields of alkylated (addition) products were obtained are shown in equations (122) and (123).





1.8.5 ALKYLATION OF ALKYNES

The benzylation reaction of 3-hexyne with benzyl chloride was reported to give a mixture of *cis* and *trans* addition compounds and 2,3-diethylindenone (equation 124).²⁰³ It was found that in high polarity solvents the yield of the *trans* addition compound and of the indenone increased. An ionic mechanism was suggested to account for the *trans* addition compound, and a more covalent complex or ion pair to account for the *cis* isomer.



1.8.6 ALKYLATION OF ALKANES

The electrophilic alkylation of alkanes using superacidic catalysts, allowing elimination or minimization of any equilibrium with related alkenes, has been successfully studied.^{204,205} These alkylations demonstrate the difference between conventional acid-catalyzed alkylations, where alkenes always play a key role in the reactions, and superacidic chemistry which allows carbocationic reactions with the alkanes themselves.

The nonexistence of alkane–alkene equilibrium in superacid medium has been elegantly demonstrated by the behavior of isobutane in deuterated superacid medium (DSO₃F–SbF₅ or DF–SbF₅).²⁰⁶ Isobutane at low temperature undergoes hydrogen–deuterium exchange only at the methine position through the involvement of a three-center bound pentacoordinate carbonium ion (Scheme 17).



Scheme 17

On the other hand, Otvos *et al.*²⁰⁷ showed that acid treatment (D₂SO₄) of isobutane effects hydrogendeuterium exchange at the methyl groups, involving isobutene as an intermediate. They suggested that under the reaction conditions a small amount of *t*-butyl cation is formed in an oxidative step which deprotonates to isobutene. The reversible protonation (deuteration) of isobutene is responsible for the H–D exchange on the methyl hydrogens, whereas tertiary hydrogen is involved in intermolecular hydride transfer from unlabeled isobutane (at the CH position, see Scheme 18).

$$Bu'H \xrightarrow{D_2SO_4} Bu'^+ \xrightarrow{-H^+} \downarrow \longrightarrow CH_2D \xrightarrow{-H^+} (CD_3)_3C^+ \xrightarrow{Bu'H} (CD_3)_3CH$$

Scheme 18

Olah and coworkers^{34,208} have observed that alkyl fluoride–SbF₅ complexes, such as MeF·SbF₅ and EtF·SbF₅, or stable prepared alkylcarbenium hexafluoroantimonate salts directly alkylate alkanes (equations 125 and 126).

$$Bu^{t}H + EtF \rightarrow SbF_{5} \longrightarrow Bu^{t}$$
 (125)

$$Bu'H + Pr'H^+ SbF_6^- \longrightarrow Bu' + HSbF_6$$
 (126)

The pathway of these alkylations was clearly demonstrated by Olah *et al.*²⁰⁹ from their extensive work on the alkylation of the lower alkanes by stable carbocations under superacidic, stable ion conditions. They found that the order of reactivity of C—C and C—H bonds reflected their σ -donor abilities and was in the order: tertiary C—H > C—C > secondary C—H >> primary C—H, although various specific factors, such as steric hindrance, can influence the relative rates.

Olah and his group²¹⁰ also investigated the direct ethylation of methane with ethylene using ¹³Clabeled methane over solid superacid catalysts, such as TaF₅-AlF₃, TaF₅ and SbF₅-graphite. A high selectivity (up to 96% of label content of C₃ fraction) for monolabeled propane (¹³CC₂H₈) was observed. This clearly indicated a direct electrophilic attack of the ethyl cation on methane *via* a pentacoordinated carbonium ion, as in Scheme 19.

$$H_{2}C = CH_{2} \xrightarrow{H^{+}} CH_{3}CH_{2} \xrightarrow{H^{+}} CH_{3}CH_{2} \xrightarrow{H^{+}} CH_{3}CH_{2} \xrightarrow{H^{+}} CH_{3}CH_{2}^{+} \xrightarrow{H^{+}} CH_{3}CH_{2}^{+13}CH_{3}$$

Scheme 19

Similarly, Siskin^{211a} found that when ethylene was allowed to react with ethane in a flow system, only n-butane was obtained. This was explained by the direct alkylation of ethane by ethyl cation through a pentacoordinated carbonium ion (equation 127). The absence of a reaction between ethyl cation and ethylene was explained by the fact that no rearranged alkylated product (isobutane) was observed.

$$H_2C = CH_2 \xrightarrow{H^+} Et^+ \xrightarrow{C_2H_6} \begin{bmatrix} H \\ Et \\ Et \end{bmatrix}$$
(127)

Alkylations of alkanes, such as propane, isobutane and *n*-butane, by *t*-butyl or *s*-butyl carbenium ions are always accompanied by competing hydride transfer and rearrangement processes. In general, once an incipient carbocation is produced, further alkylations and carbocation rearrangements can continue until the cation is quenched. This process constitutes a potentially powerful method of converting lower alkanes into higher alkylates in the gasoline range.^{211b} Typically, these reactions are performed by first preparing the alkylcarbenium hexafluoroantimonates from the corresponding alkyl halides and SbF₅ in sulfuryl chloride fluoride as the solvent, and treating them in the same solvent with alkanes. For example, propylation of propane by the isopropyl cation gave a significant amount (26% of the C₆ fraction) of the primary alkylation product (equation 128). The C₆ isomer distribution, 2-methylpentane (28%), 3-methylpentane (14%) and *n*-hexane (32%), was very far from the thermodynamic equilibrium. This suggests that not only the isopropyl, but also the *n*-propyl cation was involved as intermediate (as shown by ¹³C(2)–¹³C(1) scrambling in the stable ion²¹²) as illustrated in Scheme 20.



Scheme 20

Lower alkanes such as methane and ethane have been polycondensed in superacid solutions at 50 °C to yield higher alkanes.²¹³ Subsequently Roberts and Calihan²¹⁴ were able to prepare oligomeric and polymeric alkanes with molecular weights of up to 700 by direct condensation of C₁ to C₄ alkanes over the superacid FSO₃H–SbF₅ at room temperature. Alkylation of adamantane with lower alkenes (ethylene, propene and butenes) catalyzed by CF₃SO₃H or CF₃SO₃H–B(O₃SCF₃)₃ was studied recently by Olah *et al.*²¹⁵ A number of potential processes²¹⁶ were developed to use natural gas instead of pure methane in the alkylation of methane by alkenes. When natural gas is dehydrogenated, the C₂–C₄ alkanes it contains are converted into alkenes. The resulting methane–alkene mixture can then, without separation, be passed through a superacid catalyst resulting in exothermic alkylation condensation.

Isomerizations of alkanes catalyzed by superacidic media are very important reactions, particularly in the industrial arena. Conversion of linear C_5 - C_8 alkanes to their branched counterparts substantially improves their combustion properties. The isomerization of *n*-butane to isobutane is of great importance because isobutane reacts under mildly acidic conditions with alkenes to give highly branched hydrocarbons in the gasoline range.²¹⁷ Various difficulties are encountered in handling liquid superacids and in separating the product from the catalyst. Solid superacids, such as SbF₅-intercalated graphite, SbF₅-SiO₂-TiO₂ and TiO₂-ZrO₂-SbF₅ have proved to be very efficient catalysts for the isomerization of alkanes and resulted in easy product separation.

The isomerization of a large number of isomeric tricyclic C_{10} hydrocarbons under Lewis acid catalysis gives the unusually stable isomer adamantane.²¹⁸ This unusual reaction was originally developed by Schleyer.²¹⁹ The method since has been adopted in the synthesis of diamantane and triamantane from the appropriate C_{14} and C_{18} precursors.²²⁰ Olah *et al.*²²¹ extended this reaction by isomerizing a series of $C_{4n + 6}H_{4n + 12}$ (n = 1-3) polycyclic precursors to diamondoid hydrocarbons using superacidic catalysts^{221a} as well as the NaBH₄–CF₃SO₃H reagent, which is used as a reagent for mild reductive isomerizations.^{221b} Paquette's synthesis²²² of 1,16-dimethyldodecahedrane (44) from the alkene (43) involved a superacid-catalyzed isomerization (equation 129).



Vol'pin *et al.*²⁵ found that complexes of the type RCOX·2AlX₃ (termed aprotic organic superacids)²⁵ are extremely reactive towards alkanes, and catalyze a variety of hydrocarbon transformations. These complexes were prepared simply by mixing 1 mol of distilled acyl halide (MeCOX, PrⁿCOX, PhCOX, *etc.*; X = Br, Cl) with 2 mol of sublimed AlX₃ (X = Br, Cl) at room temperature. The liquid RCOX·2AlBr₃ and solid RCOX·2AlCl₃ complexes were stable at room temperature under anhydrous conditions. The superacidic systems appear attractive in terms of their availability, nontoxicity and ease of use. Facile coupling of C₅-C₆ cycloalkanes was observed in the presence of these catalysts. For example, high yields of isomeric mixtures of dimethyldecalins were formed when cyclohexane or methylcyclopentane were employed in excess (12:1) relative to RCOX·2AlBr₃ (Scheme 21). With a

lower cycloalkane:complex ratio, acylation of the cycloalkane was observed. Mechanisms for such alkylation reactions, as well as applications of these complexes to catalyze efficiently other processes such as cracking, isomerization and ionic bromination of linear and cyclic alkanes, have been discussed.





1.8.7 REFERENCES

- G. A. Olah (ed.), 'Friedel-Crafts and Related Reactions', Wiley-Interscience, New York, vols. 1-4, 1963-65.
 G. A. Olah (ed.), 'Friedel-Crafts Chemistry', Wiley, New York, 1973.
- 3. G. Kranzlein, 'Aluminum Chloride in der Örganischen Chemie', 3rd edn., Verlag Chemie, Berlin, 1939.
- 4. C. A. Thomas, 'Anhydrous Aluminum Chloride in Organic Chemistry', Reinhold, New York, 1961.
- 5. R. M. Roberts and A. A. Khalaf, 'Friedel-Crafts Alkylation Chemistry. A Century of Discovery', Dekker, New York, 1984.
- G. A. Olah, G. K. S. Prakash and J. Sommer, in 'Superacids', Wiley, New York, 1985, p. 24. 6.
- G. A. Olah, S. Kobayashi and M. Tashiro, J. Am. Chem. Soc., 1972, 94, 7448. 7.
- 8. T. Asaoka, C. Shimasaki, K. Taki, M. Funayama, M. Sakano and Y. Kamimura, Yuki Gosei Kagaku Kyokai Shi, 1969, 27, 783.
- G. A. Olah, O. Farooq, S. M. F. Farnia and J. A. Olah, J. Am. Chem. Soc., 1988, 110, 2560.
- 10. C. D. Nenitzescu and E. I. P. Cantuniari, Chem. Ber., 1933, 66, 1097.
- 11. N. Mine, Y. Fujiwara and H. Taniguchi, Chem. Lett., 1986, 357
- 12. A. J. Kolka, J. P. Napolitano and G. G. Ecke, J. Org. Chem., 1956, 21, 712.
- 13. R. Roe, in 'Organic Reactions', ed. R. Adams, Wiley, New York, 1948, vol. 5, p. 200.
- 14. E. W. Valyocsik, D. H. Olson and P. G. Rodewald (Mobil Oil Corp.), Eur. Pat. 193 282 (1986) (Chem. Abstr., 1986, 105, 228 915).
- 15. M. P. Atkins (British Petroleum Co.), Eur. Pat. 130 055 (1985) (Chem. Abstr., 1985, 102, 120 777).
- 16. B. Loev and J. T. Massengale, J. Org. Chem., 1957, 22, 988.
- 17. S. W. Holmes, E. A. Burt and D. E. Goins (Ethyl Corp.), US Pat. 4 628 127 (1986) (Chem. Abstr., 1987, 106, 104 218).
- 18. G. A. Olah, G. K. S. Prakash and J. Sommer, Science (Washington, D.C.), 1979, 206, 13.
- 19. (a) R. N. Haszeldine and J. M. Kidd, J. Chem. Soc., 1954, 4228; (b) T. Gramstad and R. N. Haszeldine, J. Chem. Soc., 1956, 173.
- 20. D. A. McCaulay and A. P. Lien, J. Am. Chem. Soc., 1952, 74, 6246.
- 21. Ref. 6, chap. 2.
- 22. M. Kilpatrick and F. E. Luborsky, J. Am. Chem. Soc., 1954, 76, 5863.
- 23. R. J. Gillespie and T. E. Peel, J. Am. Chem. Soc., 1973, 95, 5173.
- 24. G. A. Olah and J. Sommer, Recherches, 1979, 10, 624.
- 25. M. Vol'pin, I. Akhrem and A. Orlinkov, Nouv. J. Chim., 1989, 112, 4057.
- 26. R. Ruicheng, J. Shuojian and S. Ji, J. Macromol. Sci., Chem., 1987, A24, 669.
- R. Ran, S. Jiang and J. Shen, Yingyong Huaxue, 1985, 2, 29 (Chem. Abstr., 1985, 103, 27 939).
 G. A. Olah, P. S. Iyer and G. K. S. Prakash, Synthesis, 1986, 513.
- 29. H. R. Kaenel, SLZ, Schweiz. Lab.-Z., 1984, 41, 364.
- 30. C. Friedel and J. M. Crafts, Ann. Chim. (Paris), 1884, 1, 449.
- 31. C. C. Price, in 'Organic Reactions', Wiley, New York, 1946, vol. 3, p. 2.
- 32. Ref. 1, vol. 1, p. 623.
- 33. (a) F. Fairbrother, Trans. Faraday Soc., 1941, 37, 763; J. Chem. Soc., 1945, 503; W. J. Jacober and C. A. Kraus, J. Am. Chem. Soc., 1949, 71, 2405, 2409; M. F. A. Dove and D. B. Sowerby, in 'Halogen Chemistry', ed. V. Gutmann, Academic Press, New York, 1967, vol. 1, chap. 2, p. 41; (b) S. U. Choi and J. E. Willard, J. Am. Chem. Soc., 1965, 87, 3072; O. C. Kwun and S. U. Choi, J. Phys. Chem., 1968, 72, 3148; (c) F. P. DeHaan, H. C. Brown, D. C. Conway and M. G. Gibby, J. Am. Chem. Soc., 1954, 91, 4854; F. P. DeHaan, M. G. Gibby and D. R. Aebersold, J. Am. Chem. Soc., 1969, 91, 4860.
- 34. (a) G. A. Olah, 'Halonium Ions', Wiley-Interscience, New York, 1975; p. 8; (b) G. A. Olah, J. R. DeMember, R. H. Schlosberg and Y. Halpern, J. Am. Chem. Soc., 1972, 94, 156.
- 35. H. C. Brown and M. Grayson, J. Am. Chem. Soc., 1953, 75, 6285.
- 36. H. C. Brown and H. Jungk, J. Am. Chem. Soc., 1955, 77, 5584.
- 37. G. A. Olah and J. A. Olah, J. Am. Chem. Soc., 1976, 98, 1839.
- 38. A. Takematsu, K. Sugita and R. Nakane, Bull. Chem. Soc. Jpn., 1978, 51, 2082.
- 39. G. A. Olah, J. A. Olah and T. Ohyama, J. Am. Chem. Soc., 1984, 106, 5284.
- 40. G. Angelini, C. Sparapani and M. Speranza, Gazz. Chim. Ital., 1986, 116, 479.
- 41. J. F. White and M. F. Farona, J. Organomet. Chem., 1973, 63, 329.
- 42. K. Arata, N. Azumi and H. Sawamura, Bull. Chem. Soc. Jpn., 1975, 48, 2944.

Alkylation of Carbon

- 43. K. Tanabe, T. Yamagatsu and T. Takeshita, J. Res. Inst. Catal., Hokkaido Univ., 1965, 12, 230 (Chem. Abstr., 1965, 63, 8145); K. Arata, T. Takeshita and K. Tanabe, Shokubai, 1966, 8, 226; T. Takeshita, K. Arata, T. Sano and K. Tanabe, Kogyo Kagaku Zasshi, 1966, 69, 916 (Chem. Abstr., 1967, 66, 45 989).
- 44. P. Laszlo and A. Mathy, Helv. Chim. Acta, 1987, 70, 577.
- 45. G. A. Olah and D. Meidar, Nouv. J. Chim., 1979, 3, 269.
- 46. S. Masuda, T. Nakajima and S. Suga, Bull. Chem. Soc. Jpn., 1983, 56, 1089.
- 47. D. K. Sen Sharma, S. Ikuta and P. Kebarle, Can. J. Chem., 1982, 60, 2325.
- 48. B. Roth and E. R. Aig, J. Med. Chem., 1987, 30, 1998.
- 49. F. Baardman, R. van Helden and M. J. De Nie-Sarink (Shell Internationale Research Maatschappij B. V.), Eur. Pat. 79 093 (1983) (Chem. Abstr., 1983, 99, 121 982).
- 50. A. G. Zhukov, V. A. Topchii and A. G. Yurchenko, Vestn. Kiev. Politekh. Inst., [Ser.] Khim. Mashinostr. Tekhnol., 1986, 23, 16 (Chem. Abstr., 1988, 108, 37 350).
- 51. M. Stern, A. Warshawsky and M. Fridkin, Pept., Proc. Eur. Pept. Symp., 16th, 1980 (1981), 99 (Chem. Abstr., 1983, 98, 72 709).
- 52. G. A. Olah, S. Kobayashi and M. Tashiro, J. Am. Chem. Soc., 1972, 92, 7448.
- 53. U. Schöllkopf, S. Grüttner, R. Anderskewitz, E. Egert and M. Dyrbusch, Angew. Chem., Int. Ed. Engl., 1987, 26, 683.
- 54. S. Matsumoto and H. Oshima (NEC Corp.), Jpn. Pat. 62 34 902 (1987) (Chem. Abstr., 1987, 107, 41 492).
- T. Matsumoto, M. Katsuki and K. Suzuki, Tetrahedron Lett., 1989, 30, 833.
- 56. T. Matsumoto, M. Maeta, K. Suzuki and G. Tsuchihashi, Tetrahedron Lett., 1988, 29, 3567, 3571, 3575.
- 57. L. I. Zakharkin and E. V. Anikina, Izv. Akad. Nauk SSSR, Ser. Khim., 1987, 367.
- 58. (a) Kh. S. Tadzhimukhamedov, A. R. Abdurasuleva, K. N. Akhmedov, E. R. Riskibaev and N. U. Yusupov, J. Org. Chem. USSR (Engl. Transl.), 1975, 11, 1655; (b) G. A. Olah and J. A. Olah, J. Org. Chem., 1967, 32, 1612.
- 59. L. F. Hatch and S. Matar, in 'From Hydrocarbons to Petrochemicals', Gulf Publishing Co., Houston, 1981, p.
- 60. A. K. Askerov, T. P. Kamysheva, S. I. Sadykh-Zade, I. G. Ismailzade and I. M. Mamedov, Azerb. Khim. Zh., 1964, 31 (Chem. Abstr., 1965, 63, 6888).
- T. Morimoto and I. Hashimoto, Jpn. Pat. 7 436 659 (1974) (Chem. Abstr., 1974, 81, 49 470). 61.
- 62. D. B. Priddy, Ind. Eng. Chem., Prod. Res. Dev., 1969, 8, 239.
- 63. (a) H. I. Enos, Jr., Ger. Pat. 936 089 (1955) (Chem. Abstr., 1959, 53, 6189); (b) H. M. Friedman and A. L. Nelson, J. Org. Chem., 1969, 34, 3217; (c) J. A. Vesely, US Pat. 3 637 884 (1972) (Chem. Abstr., 1972, 76, 72 309).
- 64. R. Prasad, A. R. Gupta and K. K. Bhattacharya, Front. Chem. React. Eng., (Proc. Int. Chem. React. Eng. Conf.), 1984, 1, 531.
- 65. T. N. Nesterova, S. P. Verevkin, T. N. Malova and V. A. Pil'shchikov, Zh. Prikl. Khim., 1985, 58, 827 (Chem. Abstr., 1985, 103, 159 918).
- 66. B. R. Serebryakov and V. N. Filin, Khim. Prom-st., 1985, 70 (Chem. Abstr., 1985, 103, 5721).
- 67. P. K. Singh, K. Rajeswari and K. Ranganayakulu, Natl. Acad. Sci. Lett. (India), 1983, 6, 321.
- N. G. Bineeva, G. A. Reitman, V. V. Lobkina and L. V. Arakelova, Khim. Prom-st., 1982, 520.
 E. P. Babin, N. A. Dzhafarova, V. M. Farzaliev and M. A. Allakhverdiev, Zh. Prikl. Khim., 1985, 58, 424 (Chem. Abstr., 1985, 103, 21 965).
- 70. S. V. Zavgorodnii, G. K. Bereznitskii and D. A. Pisanenko, Zh. Org. Khim., 1972, 8, 1484.
- 71. G. A. Olah, S. H. Flood, S. J. Kuhn, M. E. Moffatt and N. A. Overchuck, J. Am. Chem. Soc., 1964, 86, 1046.
- 72. E. S. Mortikov, V. Andreev, I. I. Lishchiner, I. E. Pakhomova, D. Z. Levin, M. P. Chekesova, N. F. Kononov, A. E. Gol'dovskii and Kh. M. Minachev, Zh. Prikl. Khim., 1977, 50, 1867 (Chem. Abstr., 1977, 87, 184 105).
- 73. B. V. Ioffe and B. V. Stolyarov, Dokl. Akad. Nauk SSSR, 1965, 161, 1339.
- 74. V. E. Kryuchkova and S. V. Zavgorodnii, Izv. Vyssh. Uchebn. Zaved., Khim. Khim. Tekhnol., 1968, 11, 1353 (Chem. Abstr., 1969, 71, 21 789).
- 75. (a) I. I. Lishchiner, V. A. Plakhotnik, D. Z. Levin, E. S. Mortikov and V. F. Il'in, React. Kinet. Catal. Lett., 1983, 23, 261 (Chem. Abstr., 1983, 101, 22 696); (b) S. M. Csicsery, in 'Zeolite Chemistry and Catalysis', ed. J. A. Rabo, American Chemical Society, Washington DC, chap. 12; W. W. Kaeding, C. Chu, L. B. Young, B. Weinstein and S. A. Butter, J. Catal., 1981, 67, 159; (c) N. Y. Chen, W. W. Kaeding and F. G. Dwyer, J. Am. Chem. Soc., 1979, 101, 6783.
- (a) G. A. Olah (Produits Chimiques Ugine Kuhlmann), Can. Pat. 1 152 111 (1983) (Chem. Abstr., 1984, 100, 76. 34 291); (b) G. A. Olah (Produits Chimiques Ugine Kuhlmann), Fr. Pat. 2 498 177 (1982) (Chem. Abstr., 1983, 98, 16 438).
- 77. M. Yu. Belik, Ya. B. Kozlikovskii and V. A. Koshchii, Ukr. Khim. Zh. (Russ. Ed.), 1986, 52, 1090.
- 78. K. Okabe, T. Ohwada, T. Ohta and K. Shudo, J. Org. Chem., 1989, 54, 733.
- 79. Ch. Sh. Kadyrov and S. S. Khalikov, Dokl. Akad. Nauk Uzb. SSR, 1983, 31 (Chem. Abstr., 1983, 99, 70 627).
- 80. A. A. Khalaf and R. M. Roberts, Rev. Roum. Chim., 1985, 30, 507.
- V. V. Voronenkov, V. I. Yulina and E. A. Lazurin, Neftekhimiya, 1971, 11, 857; Pet. Chem. USSR (Engl. Transl.), 1971, 11, 242 (Chem. Abstr., 1972, 76, 152 849). 81.
- 82. R. G. Ismailov, R. S. Alimardanov and E. Kh. Mustafaev, Dokl. Akad. Nauk Az. SSR, 1971, 27, 27 (Chem. Abstr., 1971, 75, 118 059).
- 83. G. Bertholon and R. Perrin, C. R. Hebd. Seances Acad. Sci., Ser. C., 1969, 268, 1413 (Chem. Abstr., 1969, 71, 21 786).
- 84. R. G. Ismailov, G. M. Mamedaliev, Yu. I. Tokarev and T. I. Skameikina, Dokl. Akad. Nauk. Az. SSR, 1969, 25, 17 (Chem. Abstr., 1970, 73, 25 024).
- 85. F. H. Howell (Ciba-Geigy A. G.), Eur. Pat. 186 627 (1986) (Chem. Abstr., 1987, 106, 34 611).
- 86. S. W. Holmes, E. A. Burt and D. E. Goins (Ethyl Corp.), US Pat. 4 628 127 (1986) (Chem. Abstr., 1987, 106, 104 218).

- 87. H. Alfs, W. Boexkes and E. Vangermain (Chemische Werke Huels A.-G.) Ger. Pat. 3 151 693 (1983) (Chem. Abstr., 1983, 99, 139 496).
- 88. M. Tashiro, Synthesis, 1979, 921.
- 89. A. K. Mirzaeva, N. G. Sidorova and A. F. Pankratova, Nauchn. Tr. Tashk. Gos. Univ. im. V. I. Lenina, 1974, 462, 35 (Chem. Abstr., 1986, 84, 30 546).
- C. C. Price and M. Meister, J. Am. Chem. Soc., 1939, 61, 1595.
 V. N. Ipatieff, H. R. Appell and H. Pines, J. Am. Chem. Soc., 1950, 72, 4260.
- 92. (a) I. P. Tsukervanik, J. Gen. Chem. USSR, 1938, 8, 1512; (b) E. Bowden, J. Am. Chem. Soc., 1938, 60, 645; (c) J. F. Norris and J. N. Ingraham, J. Am. Chem. Soc. 1938, 60, 1421.
- 93. B. V. Ioffe and Tsan-Hsi Yang, Zh. Obshch. Khim., 1963, 33, 2196.
- 94. L. M. Kozlova and I. Romadane, Neftekhimiya, 1970, 10, 845 (Chem. Abstr., 1971, 74, 64 007).
- 95. M. Maue and A. Shimada (Yoshitomi Pharmaceutical Industries, Ltd.), Jpn. Pat. 62 185 056 (1987) (Chem. Abstr., 1988, 108, 186 331).
- 96. Council of Scientific and Industrial Research (India), Indian Pat. 150 678 (1982) (Chem. Abstr., 1983, 99, 158 238).
- 97. K. Ichikawa, K. Chano, M. Inoue and T. Sugita, Bull. Chem. Soc. Jpn., 1982, 55, 3039.
- 98. C. F. Nutaitis and G. W. Gribble, Synthesis, 1985, 756.
- 99. (a) G. A. Olah and J. Nishimura, J. Am. Chem. Soc., 1974, 96, 2214; (b) G. A. Olah, J. Nishimura and Y. Yamada, J. Org. Chem., 1974, 39, 2430.
- 100. G. Casiraghi, G. Sartori, G. Casnati and F. Bigi, J. Chem. Soc., Perkin Trans. 1, 1983, 1649.
- 101. G. Fritz, M. Eggersdorfer and H. Siegel (BAŠF A.-G.), Ger. Pat. 3 404 336 (1985) (Chem. Abstr., 1986, 104, 148 526).
- 102. O. Piccolo, F. Spreafico, G. Visentin and E. Valoti, J. Org. Chem., 1985, 50, 3945.
- 103. M. Stefanovic, M. Dermanovic, B. Solaja, V. Dermanovic and R. Prewo, J. Serb. Chem Soc., 1987, 52, 549 (Chem. Abstr., 1988, 109, 170 705v)
- 104. N. Milstein, J. Heterocycl. Chem., 1968, 5, 337
- 105. T. Nakajima, Y. Nakamoto and S. Suga, Bull. Chem. Soc. Jpn., 1975, 48, 960.
- M. Inoue, T. Sugita and K. Ichikawa, Bull. Chem. Soc. Jpn., 1978, 51, 174.
 S. Searles, J. Am. Chem. Soc., 1954, 76, 2313.
- 108. M. Segi, M. Takebe, S. Masuda, T. Nakajima and S. Suga, Bull. Chem. Soc. Jpn., 1982, 55, 167.
- 109. I. P. Tsukervanik and Ch. Sh. Kadyrov, Dokl. Akad. Nauk. Uzb. SSR, 1958, 31 (Chem. Abstr., 1959, 53, 16 083).
- 110. J. Colonge and P. Rochas, Bull. Soc. Chim. Fr., 1948, 818, 822, 825, 827.
- 111. Ch. Sh. Kadyrov and I. P. Tsukervanik, Dokl. Akad. Nauk Uzb. SSR, 1960, 27 (Chem. Abstr., 1963, 58, 2420).
- 112. J. I. Brauman and A. Solladié-Cavallo, J. Chem. Soc., Chem. Commun., 1968, 1124.
- 113. T. Nakajima, S. Suga, T. Sugita and K. Ichikawa, Bull. Chem. Soc. Jpn., 1967, 40, 2980.
- 114. N. Milstein, J. Heterocycl. Chem., 1968, 5, 339.
- 115. C. A. Kerr and I. D. Rae, Aust. J. Chem., 1978, 31, 341.
- 116. Ch. Sh. Kadyrov and D. Z. Laipanov, Zh. Org. Khim., 1966, 2, 1272.
- 117. J. F. McKenna and F. J. Sowa, J. Am. Chem. Soc., 1937, 59, 1204.
- 118. C. C. Price and J. M. Ciskowski, J. Am. Chem. Soc., 1938, 60, 2499.
- J. I. Brauman and A. J. Pandell, J. Am. Chem. Soc., 1967, 89, 5421.
 W. E. Truce and F. D. Hoerger, J. Am. Chem. Soc., 1954, 76, 5357.
- 121. C. Friedel and J. M. Crafts, Bull. Soc. Chim. Fr., 1884, 41, 322.
- 122. B. N. Dolgov and N. A. Larin, J. Gen. Chem. USSR, 1950, 20, 450.
- 123. S. A. Vartanyan, A. G. Vartanyan and E. A. Araratyan, Arm. Khim. Zh., 1972, 25, 948 (Chem. Abstr., 1973, 78, 83 924).
- 124. D. L. Ransley, J. Org. Chem., 1966, 31, 3595.
- 125. K. M. Shadmanov, Uzb. Khim. Zh., 1963, 7, 57 (Chem. Abstr., 1963, 59, 2732).
- 126. D. L. Ransley, J. Org. Chem., 1968, 33, 1517.
- 127. R. Gelin, B. Chantegrel and S. Gelin, C. R. Hebd. Seances Acad. Sci., Ser. C, 1970, 270, 1123 (Chem. Abstr., 1970, 73, 14 037).
- 128. D. D. Gverdtsiteli and Sh. Sh. Barabadze, Izv. Akad. Nauk Gruz. SSR, Ser. Khim., 1987, 13, 232 (Chem. Abstr., 1988, 109, 6188).
- 129. S. Masuda, M. Segi, T. Nakajima and S. Suga, J. Chem. Soc., Chem. Commun., 1980, 86.
- 130. W. F. Beckert and J. U. Lowe, Jr., J. Org. Chem., 1967, 32, 582.
- 131. E. T. McBee, E. P. Wesseler, D. L. Crain, R. Hurnaus and T. Hodgins, J. Org. Chem., 1972, 37, 683.
- 132. V. V. Korshak and K. K. Samplavskaya, Sb. Statei Obshch. Khim., 1953, 2, 1020 (Chem. Abstr., 1955, 49, 8207).
- 133. L. Schmerling, R. W. Welch and J. P. West, J. Am. Chem. Soc., 1956, 78, 5406.
- 134. L. V. Bugrova and I. P. Tsukervanik, Zh. Org. Khim., 1965, 1, 714.
- 135. G. A. Olah and S. J. Kuhn, J. Org. Chem., 1964, 29, 2317.
- 136. Y. Monteils, Bull. Soc. Chim. Fr., 1951, 637.
- 137. C. D. Nenitzescu and D. A. Isacescu, Chem. Ber., 1933, 66, 1100.
- 138. I. P. Tsukervanik, V. A. Ermokhina and A. R. Abdurasuleva, USSR Pat. 351 822 (1972) (Chem. Abstr., 1973, 78, 29 418).
- 139. S. Kajigaeshi, T. Kadowaki, A. Nishida, S. Fujisaki and M. Noguchi, Synthesis, 1984, 335.
- 140. I. P. Tsukervanik and Kh. Yu. Yuldashev, Zh. Obshch. Khim., 1961, 31, 858.
- 141. (a) R. C. Bansal, J. R. Mattox, E. J. Eisenbraun, P. W. K. Flanagan and A. B. Carel, J. Org. Chem., 1966, 31, 2716; (b) E. J. Eisenbraun, R. C. Bansal, J. R. Mattox and P. W. K. Flanagan, Prepr., Div. Pet. Chem., Am. Chem. Soc., 1965, 10, 157.

- 142. (a) G. A. Olah, P. Schilling, J. S. Staral, Y. Halpern and J. A. Olah, J. Am. Chem. Soc., 1975, 97, 6807; (b) R. Miethchen, S. Steege and C.-F. Kröger, J. Prakt. Chem., 1983, 325, 823, and refs. therein; R. Miethchen, C. Röhse and C.-F. Kröger, Z. Chem., 1984, 24, 145.
- R. Miethchen, K. Kohlheim and A. Müller, Z. Chem., 1986, 26, 168, and refs. therein. 143
- 144. S. Chalais, A. Cornélis, A. Gerstmans, W. Kolodziejski, P. Laszlo, A. Mathy and P. Métra, Helv. Chim. Acta, 1985, 68, 1196.
- 145. A. A. Khalaf and R. M. Roberts, J. Org. Chem., 1966, 31, 89.
- 146. Z.-u. Yang and G. B. Young, J. Chem. Soc., Dalton Trans., 1984, 9, 2019.
- 147. A. A. Khalaf and R. M. Roberts, J. Org. Chem., 1973, 38, 1388.
- 148. K. von Auwers and E. Hilliger, Chem. Ber., 1916, 49, 2410.
 149. C. K. Bradsher, Chem. Rev., 1987, 87, 1277.
- 150. F. Mayer, L. van Zütphen and H. Phillips, Chem. Ber., 1927, 60, 858.
- 151. J. von Braun, K. Heider and E. Müller, Chem. Ber., 1917, 50, 1637.
- 152. R. T. Arnold and E. Rondestvedt, J. Am. Chem. Soc., 1945, 67, 1265.
- 153. R. W. Layer and I. R. MacGregor, J. Org. Chem., 1956, 21, 1120.
- 154. P. Maitte, Ann. Chim. (Paris), 1954, 9, 431.
- 155. D. Bonnet-Delpon, M. Charpentier-Morize and R. Jacquot, J. Org. Chem., 1988, 53, 759.
- 156. (a) A. A. Khalaf and R. M. Roberts, J. Org. Chem., 1972, 37, 4227; 1973, 38, 1388; 1969, 34, 3571; 1971, 36, 1040; (b) A. A. Khalaf, Rev. Roum. Chim. (Bucharest), 1974, 19, 1373.
- 157. B. R. Davis, S. J. Johnson and P. D. Woodgate, Aust. J. Chem., 1987, 40, 1283. 158. A. A. Khalaf, R. A. Kabli, M. S. T. Mekki, A. M. Kaddah and A. B. Ghawgi, Indian J. Chem., Sect. B, 1987, 26, 220.
- 159. K. S. Sharma, K. L. Taneja and S. M. Mukherji, Indian J. Chem., Sect. B, 1982, 21, 572.
- 160. A. M. El-Khawaga, A. M. Abdel-Wahab, M. F. Elzohry and A. A. Khalaf, Rev. Roum. Chim., 1985, 30, 599.
- 161. (a) S. K. Taylor, G. H. Hockerman, G. L. Karrick, S. B. Lyle and S. B. Schramm, J. Org. Chem., 1983, 48, 2449; (b) S. K. Taylor, M. E. Davisson, B. R. Hissom, Jr., S. L. Brown, H. A. Pristach, S. B. Schramm and S. M. Harvey, J. Org. Chem., 1987, 52, 425; (c) S. K. Taylor, C. L. Blankespoor, S. M. Harvey and L. J. Richardson, J. Org. Chem., 1988, 53, 3309.
- 162. (a) S. P. Tanis and P. M. Herrinton, J. Org. Chem., 1983, 48, 4572; (b) S. P. Tanis and J. W. Raggon, J. Org. Chem., 1987, 52, 819; (c) R. H. Burnell and J.-M. Dufour, Can. J. Chem., 1987, 65, 21.
- 163. N. Yoneda, Y. Takahashi and A. Suzuki, Chem. Lett., 1978, 231.
- 164. M. F. Ansell and M. E. Selleck, J. Chem. Soc., 1956, 1238.
- 165. M. F. Ansell and S. S. Brown, J. Chem. Soc., 1958, 3956.
- 166. L. R. C. Barclay, in 'Friedel-Crafts and Related Reactions', ed. G. A. Olah, Wiley-Interscience, New York, 1964, vol. 2, chap. 22,
- 167. G. Suzukamo and Y. Sakito (Sumitomo Chemical Co., Ltd.), Eur. Pat. 71 006 (1983) (Chem. Abstr., 1983, 99, 122 073).
- 168. G. A. Olah, C. G. Carlson and J. C. Lapierre, J. Org. Chem., 1964, 29, 2687.
- 169. G. Suld and A. P. Stuart, J. Org. Chem., 1964, 29, 2939.
- 170. R. M. Roberts, A. A. Khalaf and J. E. Douglass, J. Org. Chem., 1964, 29, 1511.
- 171. F. Camacho, R. Fagundo, A. Brito and F. Diaz, Chem. Eng. Res. Des., 1986, 64, 273 (Chem. Abstr., 1986, 105, 135 880u).
- V. G. Lipovich, A. V. Vysotskii and V. V. Chenets, Neftekhimiya, 1971, 11, 838. 172.
- 173. V. V. Chenets, O. R. Sergeeva and V. G. Lipovich, Zh. Org. Khim., 1983, 19, 805.
- 174. O. R. Sergeeva, V. V. Chenets and V. G. Lipovich, Zh. Org. Khim., 1983, 19, 395.
- 175. N. R. Meshram, S. B. Kulkarni and P. Ratnasamy, J. Chem. Technol. Biotechnol., 1984, 34A, 119.
- 176. D. S. Santilli, J. Catal., 1986, 99, 327.
- 177. A. C. Buchanan, III, A. S. Dworkin and G. P. Smith, J. Am. Chem. Soc., 1983, 105, 2843.
- 178. P. Buckus, Russ. Chem. Rev. (Engl. Transl.), 1983, 52, 1203.
- 179. (a) M. Farcasiu, T. R. Forbus and R. B. LaPierre, Prepr. Div. Pet. Chem., Am. Chem. Soc., 1983, 28, 279; (b) B. M. Benjamin, E. C. Douglas and D. M. Canonico, *Fuel*, 1984, **63**, 888; (c) R. M. Roberts and K. M. Sweeney, *Fuel*, 1985, **64**, 321; (d) B. M. Benjamin and E. C. Douglas, *Prepr. Pap.* — Am. Chem. Soc., Div. *Fuel Chem.*, 1985, **30**, 99; (e) M. Farcasiu, E. J. Y. Scott and R. B. LaPierre, *Prepr. Pap.* — Am. Chem. Soc., Div. Fuel Chem., 1985, 30, 427; (f) B. M. Benjamin, E. C. Douglas, P. M. Hershberger and J. W. Gohdes, Fuel, 1985, 64, 1340.
- 180. J. C. Wu and L.-J. Leu, Appl. Catal., 1983, 7, 283.
- 181. N. R. Bursian, Yu. A. Shavandin, E. M. Tat'yanina and N. P. Shimuk, Zh. Prikl. Khim., 1984, 57, 1244.
- 182. K. K. Tiwari and P. N. Mukherjee, Indian J. Technol., 1984, 22, 295.
- 183. D. A. Sibarov, T. G. Dokuchaeva and O. N. Mukhin, Zh. Prikl. Khim., 1985, 58, 2685.
- 184. T. Yamato, T. Arimura and M. Tashiro, J. Chem. Soc., Perkin Trans. 1, 1987, 1.
- 185. G. A. Olah, G. K. S. Prakash, P. S. Iyer, M. Tashiro and T. Yamato, J. Org. Chem., 1987, 52, 1881.
- 186. J. Rosevear and J. F. K. Wilshire, Aust. J. Chem., 1985, 38, 1163.
- 187. M. Tashiro and G. Fukata, J. Org. Chem., 1977, 42, 1208
- 188. V. N. Ipatieff and H. Pines, J. Am. Chem. Soc., 1937, 59, 56.
 189. R. M. Roberts, E. K. Baylis and G. J. Fonken, J. Am. Chem. Soc., 1963, 85, 3454.
- 190. H. M. Knight, J. T. Kelly and J. R. King, J. Org. Chem., 1963, 28, 1218.
- 191. D. M. Brouwer, Recl. Trav. Chim. Pays-Bas, 1968, 87, 210.
 192. G. A. Olah, R. H. Schlosberg, R. D. Porter, Y. K. Mo, D. P. Kelly and G. D. Mateescu, J. Am. Chem. Soc., 1972, 94, 2034
- 193. D. Fărcasiu and R. H. Schlosberg, J. Org. Chem., 1982, 47, 151.
- 194. D. H. Olson and W. O. Haag, ACS Symp. Ser., 1984, 248, 275 (Chem. Abstr., 1984, 101, 54 237).
- 195. P. Ratnasamy, G. P. Babu, A. J. Chandwadkar and S. B. Kulkarni, Zeolites, 1986, 6, 98.
- 196. A. T. Balaban, M. D. Gheorghiu, A. Schiketanz and A. Necula, J. Am. Chem. Soc., 1989, 111, 734.

- 197. V. N. Ipatieff, A. V. Grosse, H. Pines and V. I. Komarewsky, J. Am. Chem. Soc., 1936, 58, 913.
- 198. S. F. Birch, A. E. Dunstan, F. A. Fidler, F. B. Pim and T. Tait, J. Inst. Pet. Technol., 1938, 24, 303.
- 199. J. P. Kennedy and E. Maréchal, in 'Carbocationic Polymerization', Wiley-Interscience, New York, 1982, p.82.
- 200. (a) G. A. Olah, S. J. Kuhn and D. G. Barnes, J. Org. Chem., 1964, 29, 2685; (b) H. Mayr and W. Striepe, J. Org. Chem., 1983, 48, 1159.
- 201. L. Schmerling, J. Am. Chem. Soc., 1945, 67, 1152; 1946, 68, 1650.
- 202. H. Mayr, Angew. Chem., Int. Ed. Engl., 1981, 20, 184.
- 203. H. Martens and G. Hoornaert, Tetrahedron Lett., 1970, 1821.
- 204. G. A. Olah, O. Farooq and G. K. S. Prakash, in 'Homogeneous Systems for the Activation and Functionalization of Alkanes', ed. C. L. Hill, Wiley-Interscience, New York, 1989, chap 2.
- G. A. Olah, 'Carbocations and Electrophilic Reactions', Verlag Chemie, Wiley. New York, 1974, p. 108.
 G. A. Olah, Y. Halpern, J. Shen and Y. K. Mo, J. Am. Chem. Soc., 1971, 93, 1251.
- 207. J. W. Otvos, D. P. Stevenson, C. D. Wagner and O. Beeck, J. Am. Chem. Soc., 1951, 73, 5741.
- 208. G. A. Olah, J. R. DeMember and R. H. Schlosberg, J. Am. Chem. Soc., 1969, 91, 2112.
- 209. (a) G. A. Olah, Y. K. Mo and J. A. Olah, J. Am. Chem. Soc., 1973, 95, 4939; (b) G. A. Olah, J. R. DeMember and J. Shen, J. Am. Chem. Soc., 1973, 95, 4952.
- G. A. Olah, J. D. Felberg and K. Lammertsma, J. Am. Chem. Soc., 1983, 105, 6529.
 (a) M. Siskin, J. Am. Chem. Soc., 1976, 98, 5413; (b) M. Siskin, R. H. Schlosberg and W. P. Kocsi, ACS Symp. Ser., 1977, 55, 77.
- 212. G. A. Olah and A. M. White, J. Am. Chem. Soc., 1969, 91, 5801.
- 213. G. A. Olah and R. H. Schlosberg, J. Am. Chem. Soc., 1968, 90, 2726.
- 214. D. T. Roberts, Jr. and L. E. Calihan, J. Macromol. Sci., Chem., 1973, A7, 1629.
- 215. G. A. Olah, O. Farooq, V. V. Krishnamurthy, G. K. S. Prakash and K. Laali, J. Am. Chem. Soc., 1985, 107, 7541.
- 216. G. A. Olah, Eur. Pat. 73 673 (1983) (Chem. Abstr., 1983, 98, 218 647).
- 217. Ref. 6, p. 254.
- 218. R. C. Fort, Jr., in 'Adamantane, the Chemistry of Diamond Molecules', ed. P. G. Gassman, Dekker, New York, 1976, p. 35.
- 219. P. v. R. Schleyer, J. Am. Chem. Soc., 1957, 79, 3292.
- 220. (a) C. A. Cupas, V. Z. Williams, Jr., P. v. R. Schleyer and D. Trecker, J. Am. Chem. Soc., 1965, 87, 617; (b) D. Faulston, R. A. Gledinning, D. E. Johnston and M. A. McKervey, Tetrahedron Lett., 1971, 1671; (c) F. S. Hollowood, M. A. McKervey, R. Hamilton and J. J. Rooney, J. Org. Chem., 1980, 45, 4954.
- 221. (a) O. Farooq, S. M. F. Farnia, M. Stephenson and G. A. Olah, J. Org. Chem., 1988, 53, 2840; (b) G. A. Olah, A. Wu, O. Farooq and G. K. S. Prakash, J. Org. Chem., 1989, 54, 1450.
- 222. L. A. Paquette and A. Balogh, J. Am. Chem. Soc., 1982, 104, 774.

1.9 Polyene Cyclizations

JAMES K. SUTHERLAND University of Manchester, UK

1.9.1 INTRODUCTION	341
1.9.2 INITIATION OF CYCLIZATION	342
1.9.3 PROPAGATION OF CYCLIZATION	343
1.9.4 TERMINATION OF CYCLIZATION	345
1.9.5 MONOCYCLIZATIONS	347
1.9.5.1 Five-membered Ring Formation 1.9.5.2 Six-membered Ring Formation 1.9.5.3 Seven-membered Ring Formation	347 349 357
1.9.6 BICYCLIZATIONS	359
1.9.6.1 Bicyclo[4.3.0] Ring Formation 1.9.6.2 Bicyclo[4.4.0] Ring Formation	359 360
1.9.7 TRI- AND TETRA-CYCLIZATIONS	362
1.9.8 SYNTHESIS OF STEROIDS AND RELATED COMPOUNDS 1.9.8.1 Aromatic Steroids 1.9.8.2 Nonaromatic Steroids and Related Compounds	366 366 369
1.9.9 MECHANISM	374
1.9.10 SUMMARY	375
1.9.11 REFERENCES	375

1.9.1 INTRODUCTION

Much of the early work on polyalkene cyclizations derives from the classical structural investigations on terpenes¹ and this, together with speculation concerning their biosynthesis and stereochemistry, culminated in the Stork-Eschenmoser hypothesis of $1955.^2$ This postulated that polyalkenes could react in defined conformations which, in combination with antiperiplanar addition to the double bonds, allowed prediction of the relative stereochemistry of the cyclization products (Scheme 1). The attractiveness of these ideas as the basis of a general synthetic method is clear, not least because it suggests methods for forming C—C bonds between highly substituted atoms where organometallic methods are often subject to severe steric hindrance. To establish the method required the development of (a) suitable initiators for cyclization, (b) methods for the stereoselective synthesis of alkenes (Volume 1, Chapters 3.1 and 3.2) and (c) termination of cyclization by a single mechanism. Johnson's group has made major contributions to the development of polyalkene cyclization as a viable synthetic method; their earlier work has been reviewed³ as has that of other groups.⁴ There is also an excellent general review by Bartlett.⁵

In this chapter emphasis will be placed on preparatively useful procedures, the general 'rules' which pertain in polyene cyclization, and the circumstances under which the Stork-Eschenmoser hypothesis is valid. The success, or otherwise, of polyalkene cyclization depends on a complex interplay between the



Scheme 1

method of initiation, the nucleophilicity of the double bonds involved, and the mechanism of termination. Because of this, it is rather artificial to consider these separately but still useful in a pedagogic sense. Since the mechanism of polyalkene cyclization is by no means firmly established (indeed, there is unlikely to be a unique mechanism) the process will be represented as involving 'simple' carbocations; this does not exclude concerted and part-concerted mechanisms or those involving bridged cations or π complexes. Formulae representing racemates are drawn showing only one enantiomer. Where a single enantiomer is represented, it is described as chiral.

1.9.2 INITIATION OF CYCLIZATION

Cyclization is initiated by formation of a putative cation either by electrophilic addition to a double bond or by ionization, usually from an sp^3 -hybridized carbon. Protonic and Lewis acids have been the most frequently used electrophiles but successful monocyclizations have been reported using HOBr,6 MeCOBF₄,⁷ MeOCH₂BF₄,⁸ Br_2-AgBF_4 , ⁹ PhSBF₄, ¹⁰ PhSeBF₄, ¹¹ Tl(ClO₄)₃, ¹² Hg(OAc)₂, ¹³ $Hg(OCOCF_3)_2^{14}$ and $HgTf_2$ (with and without PhNMe₂). The $Hg(OCOCF_3)_2$ reagents have been used successfully for the bicyclization of famesol derivatives to drimnanes (Section 1.9.6.2). The Semenovskii group has shown¹⁴ that, of the electrophiles tested, $Hg(OCOCF_{3})_2$ is the only reagent which effects antiperiplanar addition to the initiating double bond; whether this is due to the bridged mercuricinium ion maintaining stereochemical integrity or a concerted cyclization is not known. Cyclization with H2SO4 gave a 1:1 mixture of equatorial and axial CD₃, establishing that it is a nonconcerted reaction (Scheme 2). This contrasts with the cyclization of 2,6-octadiene¹⁵ (where a less-stable secondary cation would be the equivalent intermediate) having the characteristics of concerted protonation-ring formation (Scheme 3). PhSOMe-BF₃ and N-phenylselenylsuccinimide-BF₃ have also been established as initiators of bicyclizations (Section 1.9.6.2). Understandably, the electrophile-induced cyclizations are most efficient when the initial reaction occurs at the most nucleophilic double bond. However, these initiators have not been tested with a sufficient range of substrates to allow generalizations about their individual effectiveness.

Ionization methods for production of the initiating cation have been more generally tested and shown to be ineffective at inducing cyclization at primary and secondary centers where the ion does not have additional stabilization. Acid-catalyzed cyclization of tertiary derivatives has been more successful. From a preparative viewpoint, biomimetic epoxide cleavage as a means of cyclization has been disappointing, though much useful mechanistic and structural information has been gleaned from this work, in particular that of the van Tamelen group.^{4a} Certain methylcyclohexene epoxides give acceptable yields in monocyclizations^{4d} but are not useful for bi- and tri-cyclizations. The most efficient general initiators of cyclization are cations which are stabilized by a heteroatom, a double bond or both. The allylic cations (1), (2), (3) and (4) introduced by the Johnson group and the acyclimonium ion (5) used by Speckamp



Scheme 3

(Volume 2, Chapter 4.5) have been tested with a wide variety of substrates and shown to be efficient at inducing mono- and poly-cyclizations. The Johnson group has also used oxonium ions (derived from acetal cleavage) as initiators, in particular for preparing chiral cyclized products. Imonium¹⁶ and acycloxallyl cations (6) are also useful initiators;¹⁷ with the latter, carbon bond formation does not occur at the O-terminus. It is worth noting that the cyclopentenyl and cyclohexenyl cations with two hydrogens at the terminus are not particularly good initiators and that the trialkylated ions are preferentially attacked at the least-alkylated terminus when the choice is open. To summarize, monocyclizations can be achieved using a variety of electrophilic reagents or ionization methods, carbabi- or carbatri-cyclizations are achievable in good yield when preparing drimnane terpene derivatives; otherwise initiation of cyclization by stabilized cations is the preferred strategy.



1.9.3 PROPAGATION OF CYCLIZATION

There are a host of examples of carbamonocyclizations, many bicyclizations, a substantial number of tricyclizations and a few tetracyclizations. Where cyclohexane rings are being formed, the overwhelming number of cases can be interpreted as forming via a chair-like transition state. Where ring junctions are formed in bi- and tri-cyclizations (Scheme 4) most examples are of (E)-alkenes cyclizing to form trans-fused rings giving the thermodynamically stable product. The corollary of the Stork-Eschenmoser hypothesis that a cis ring junction would form from the (Z)-alkene has been tested infrequently (Section 1.9.6.2). Where the comparison has been carried out, it is clear that the extent to which the hypothesis is upheld is dependent on the initiator and the nucleophilicity of the double bonds participating. Where a double bond is 5.6 to the initiating center and electronically unbiased or substituted at C-5, then 6-endo cyclization is almost invariably favored over the 5-exo mode (Scheme 5); the cyclization of (7) is a rare exception to this rule.¹⁸ When there is additional substitution at C-6 polarizing the double bond, then 5-

Alkylation of Carbon

exo cyclization is preferred; the products of cyclization can be complicated by the cyclopentylmethyl cation undergoing rearrangement following cyclization. A double bond 6,7 to the initiating center leads to 6-exo cyclization unless this contravenes the Markovnikov rule, in which case 7-endo cyclization ensues (Scheme 5). With 4,5 double bonds, 5-endo cyclization is seldom observed, in accord with Baldwin's rules.





Allenyl and alkynyl functions can also participate in cyclization, usually in the formation of the terminal ring. Alkynyl groups 5,6 to the cationic center cyclize 6-*endo* when the terminal group is hydrogen or silyl (Scheme 6) but dialkylalkynes have a kinetic preference for 5-*exo* cyclization though under certain conditions the 5-*exo* cations can rearrange to the thermodynamically more stable 6-*endo* ion. The homologous 6,7-alkynes behave in a parallel manner, *i.e.* terminal alkynes give seven-membered rings while dialkylalkynes cyclize 6-*exo* to cyclohexenes. An allenyl group 4,5,6 relative to the initiating center with at least one terminal alkyl group induces five-membered ring formation, whereas with a CH₂ terminus both five- and six-membered ring formation has been noted (Scheme 6). The reluctance of 4,5-alkynes to cyclize in either 4-*exo* or 5-*endo* modes is illustrated by the cyclizations to the bridged bicyclics (Scheme 7).¹⁹

When substituents are bonded to the sp^3 -hybridized carbons between reacting double bonds then, for six-membered ring formation, the model with the substituent adopting an equatorial disposition in a



Scheme 7

chair-like transition state has strong predictive value; however, there are occasions when an axial disposition is adopted for understandable reasons.

1.9.4 TERMINATION OF CYCLIZATION

The ideal for a synthetically useful polyalkene cyclization would be termination by one mechanism giving a single product. Termination can be achieved by elimination and/or attack by an internal or external nucleophile. Proton elimination can be regioselective or random. The cyclization of geraniol and its derivatives has been extensively investigated, in particular by the groups of Schinz²⁰ and Semenovskii.²¹ Both (Z)- and (E)-isomers cyclize initially to the α -isomers when an electron-withdrawing group is conjugated to the second double bond (Scheme 8). It is suggested that the reduced nucleophilicity of the double bond leads to a concerted cyclization-elimination being favored over the two-step process. Where the nucleophilicity of the double bond is not reduced then (E)-isomers give the α -series and (Z)isomers the β -compounds, consistent with elimination of an axial proton to give the most stable alkene attainable. This rule does not apply when there is oxygen functionality in the vicinity of the second double bond. Geraniol and its derivatives (8) cyclize initially to α -isomers. Utilization of the oxygen function as an internal base for proton removal has been invoked; this would require conformational inversion placing CH₂OR axial. The contrary behavior of lavandulol (9) illustrates that some subtle features are involved. Together with the allogeraniol (10) cyclization, the one common feature appears to be that elimination remote from the gem-dimethyl group is preferred. Another rule which seems to have some predictive value is that bicyclo[4.4.0] decenes (both cis and trans) formed from the C-2 cation favor the $\Delta^{2,3}$ - rather than $\Delta^{1,2}$ -double bond. However, in many cases, eliminations are not regioselective and the device of R_3Si elimination introduced by Fleming²² is becoming increasingly popular as a method for preparing compounds of defined regiochemistry. Other terminating groups which remove any ambiguity concerning the structure of the product are alkynes, allenes, enols and enol derivatives. Aryl and heteroaryl rings, particularly if they are electron rich, are good terminators, but when appropriately substituted can give mixtures of positional isomers.

Proton elimination is the most common termination mode for tertiary cations and also for secondary cations when the cyclizing reagent is a Lewis acid. With protonic acids, nucleophilic attack is often observed and is usually stereoselective; in many cases, due to the electron deficiency of the double bonds involved, nucleophilic attack may be concerted with ring formation. Cyclization of the acids (11) is in



accord with this view.²³ In the few cases where a tertiary center undergoes intermolecular nucleophilic attack, epimeric mixtures have been obtained. However, intramolecular nucleophilic attack by heteroatoms can be stereospecific and efficient, as in the cyclization of geranylacetone (12).²⁴ The lack of predictability when alkenes terminate cyclizations has led to the development of terminators where the products can be more precisely anticipated.



1.9.5 MONOCYCLIZATIONS

1.9.5.1 Five-membered Ring Formation

Examples of acyclic compounds being cyclized to cyclopentanes are sparse, but there are significantly more where an annulation is involved. 5-*Exo* cyclization of alkenes electronically biased to achieve this mode was used successfully, in particular by the Landsbury group.²⁵ Use of the chlorovinyl functionality gave acetylcyclopentanes with a preference for the *cis* ring junction with conformationally mobile substrates (13). When a conformationally rigid precursor (14) was used then predominant formation of a *trans* ring junction resulted; similar results were reported for the propargyl terminating group. Stereochemically constrained dienes,²⁶ *e.g.* (15), can be cyclized under mild conditions with high efficiency. Limitation of the degrees of rotational freedom of a diene is likely to increase the rate of cyclization provided that the double bonds are suitably aligned.



Anti-Markovnikov 5-exo cyclizations of geraniol derivatives (16) have been reported with Tl(ClO₄)₃;¹² however, a more normal 6-endo cyclization followed by ring contraction could account for the products. The most reliable prescription for cyclopentane annulations is to use allylic ion initiators with appropriate terminators. In Scheme 9 the *cis* stereochemistry of the ring junctions arises in two ways; for the 4-substituted cyclohexenone derivatives, the axial attack of the nucleophilic side chain on the allylic ion requires this stereochemistry. In the 2-substituted cyclohexenones the immediate product of cyclization is an enol ester which is hydrolyzed, giving the thermodynamically more stable *cis* ring junction. Other points to note are that stereochemical control in example 3 (Scheme 9) is consistent with the Prⁱ group being equatorial and the isopropylidene being oriented '*endo*'. The subsidiary rearrangement seen in example 4 (Scheme 9) does not occur with the unmethylated compound. In example 6 (Scheme 9) there is a substantial difference in yield between the methyl and nor series. There are few examples of 5-endo





cyclizations; the conversion of the menthadiene $(17)^{32}$ to the camphor derivative is both 5-*endo* and 5-*exo*. The silylenol ether (18) cyclizes to cyclopentanone only when the allyl terminus is dimethylated;³³ with monomethyl derivatives the major mode of cyclization is 7-*endo* (Section 1.9.5.3). The Cookson group has shown (Scheme 10) that 4-en-l-als can be cyclized to cyclopentanones;³⁴ labeling and other evidence support the mechanism shown.



1.9.5.2 Six-membered Ring Formation

This is by far the most common mode of monocyclization. Some aspects of the cyclization of monoterpenes have been described (Sections 1.9.6.3 and 1.9.6.4). Proton-induced cyclizations are successful when an initial tertiary cation is formed, then participation of electron-deficient and other double bonds occurs provided the Markovnikov rule is not violated. The rare examples of cyclization induced by protonation to form a notional secondary cation probably do not involve such a species, but rather ring formation is concerted with cyclization. Such cyclizations do not occur when the ring-forming double bond is electron deficient. When the initiating double bond is tetraalkylated then intriguing differences from the trialkyl system appear (Scheme 11). It is known that in the proton-induced cyclization of geranylacetone the stereochemical distinction between the two methyls of the isopropylidene group is lost. However, the homolog (19) cyclizes stereoselectively,³⁵ implying either a concerted protonation cyclization or, less likely, that the axial orientation of the methyl group is favored in a two-step cyclization. Lewis acid induced cyclization of ψ -irone (20) takes a different stereochemical course, giving (21), while the (Z)-trisubstituted double bond isomer gives the product with the pendant groups *trans.*³⁶ The proton-induced cyclization of (20) follows a different path,⁴⁰ giving (22); applying similar conditions of cyclization to ψ -ionone gives α -ionone. The divergent stereochemical behavior of (19) and (20) requires

both reactions to be concerted and the Lewis acid-carbon bond of the cyclization product leading to (21) to be protonated with inversion of configuration. Alternatively, the cyclization of (20) can be stepwise, due to the electron-deficiency of the ring-forming double bond, and the transition state with methyl equatorial and BF₃ axial favored. With TiCl₄ ψ -ionone cyclizes in yet another fashion to give (23).³⁷



Scheme 11

In Scheme 12 some annulations leading to decalins are shown. The formation of *cis* isomers (examples 1 and 2, Scheme 12) accords with a fully-concerted mechanism, while the cyclization of both the α - and β -isomers of the acids (example 3, Scheme 12) to the *trans* compound supports stepwise formation of the cyclohexyl cation followed by concerted cyclization–elimination due to the electron-deficient double bond. The cyclization of the acetonyl β -isomer (example 4, Scheme 12) to the *trans* isomer implies a stepwise mechanism, which is perhaps surprising in the light of the first two results, since the ring-forming double bond is not electron deficient. There appears to be an unfavorable steric interaction between the angular methyl and two axial hydrogens in ring B (24). This may make the concerted reaction energetically unfavorable compared to the stepwise process. Cyclization of the isomer (example 5, Scheme 12) takes a different course and must involve some type of concerted process to give the unexpected stereochemistry found. A chair–chair transition state is disfavored due to the similarity with (24). A twist-boat ring B conformation suffers from similar interactions with the angular methyl. Diequatorial addition to the double bond in a chair–chair conformation has little precedent. However, a more plausible model





Scheme 12



which effectively leads to diequatorial attack has ring A as a twist-boat (25) which minimizes interactions with the angular methyl and improves orbital overlap at the reacting centers. If this model is correct, it would imply that concerted cyclization of a 1,6-diene is energetically more favorable than the 1,5-diene concerted reaction. It should be noted that the *cis*-decalin conformations involved in the annulations are inverted compared to those obtaining in bicyclizations. The Julia group has observed remarkable stereochemical control on annulation stereochemistry by the chloromercuri substituent (examples 6 and 7, Scheme 12); similar control is found in the cyclization to hydrindanes. The dienes (from dehydration) do not exhibit this effect. The β -keto ester terminator has been successfully used in annulations (example 8, Scheme 12) as has the alkynyl group to form the cyclohexanone ring (example 1, Scheme 17).

Other electrophiles have been applied mainly to geraniol derivatives (Section 1.9.2). Product stereochemistry is as anticipated except for initial addition to the double bond where, of the reagents examined, mercury salts appear to be unique in maintaining the stereochemical integrity of the double bond. Hg(OAc)₂ has the useful property of initiating cyclization at a vinyl group in the presence of a trisubstituted double bond (26);⁴² this accords with the relative rates of reaction established for differently substituted alkenes.^{43b} The reversible formation of the mercuricinium ion may also play its part. Phenylselenation provides a rare example⁴⁴ of initial attack at an 'internal' double bond to give (27).

The initiation of cyclization by ionization is best achieved using stabilized ions. The prototypical cyclization of allyl derivatives is the conversion of linalool to terpineol (Scheme 13), which the Arigoni group⁴⁵ has shown, in an elegant investigation, cyclizes with inversion of configuration. The Johnson group has examined a number of allylic alcohols which can be used as initiators (Scheme 14). Where ring junction stereochemistry is discernible in annulations (examples 1 and 5, Scheme 14) then cis stereochemistry, consistent with axial attack of the double bond, is observed. Terminal allenyl and alkynyl groups undergo endo cyclization, ultimately forming a cyclohexanone ring (example 4, Scheme 14).

Table 1 Isomer Ratios Formed from Regioisomeric Allylic Alcohols



Scheme 13

Polyene Cyclizations



Scheme 14



The anisyl ring is an efficient terminator (example 5, Scheme 14) and reacts mainly *para* to the methoxy, but *ca.* 15% of *ortho* reaction is also found. The exocyclic allylic alcohol cyclizes predominantly to a *trans*-decalin (example 6, Scheme 14); this parallels the stereochemical behavior of cyclohexyl cations. Where the putative allylic cation is unsymmetrical and cyclization to a six-membered ring can occur at either terminus, then reaction at the least-alkylated position is preferred (example 7, Scheme 14). Table 1 clearly establishes that a common allylic cation is not formed from the structurally isomeric alcohols, and reveals a tendency for bond formation at the least-hindered center.⁵⁰ Additionally there is a preference for bonding remote from the leaving group; however, unlike the linalool cyclization, chirality at the tertiary alcohol center does not lead to the formation of chiral products.⁵¹ Little is known of the stereochemical fate of substituents on the cyclohexenyl ring; the diol (example 8, Scheme 14) cyclizes but under the reaction conditions dehydrates to a diene product. Prediction is complicated by the inability of octalins (**28**) to adopt a half-chair–chair conformation other than with the bridgehead substituent axial; with the substituent equatorial one of the rings must adopt a twist-boat conformation. Thus, to achieve efficient overlap at the reacting centers, it is likely that the cyclohexene ring is in a twist-boat conformation.

In spirocyclizations using the allylic initiators, the stereochemical question arises as to whether the initiator double bond shows a preference for the equatorial or axial position relative to the ring being formed. The Harding group has shown that the equatorial disposition is favored, especially when a 1,2disubstituted alkene is cyclized, *e.g.* (29).^{4c} As expected, when there is a choice between seven-membered ring formation or spirocyclization to a six-membered ring, the latter is observed (30).⁶⁰



Enones are precursors of oxyallyl cations, either by enol ester formation and protonation as developed by the Harding group, or by coordination with Lewis acids. In Scheme 15 some typical cyclizations are illustrated. Under certain circumstances (example 4, Scheme 15) secondary rearrangement can occur by hydride migration. The vinyl compound (example 2, Scheme 15) does not give a rearrangement product, suggesting that the hydride migrations are not concerted. It is notable that cyclization of this enone in the absence of acetic anhydride leads to termination predominantly by elimination. In example 5 (Scheme 15) stereochemical control is exerted at four centers; in the cyclization a chair transition state with the pendant methyl group equatorial and the double bond oriented *exo* accounts for the relative stereochemistry at three centers. The fourth center is established by stereoselective alkylation of the immediate cyclization product, presumably a titanium enolate. Scheme 16 shows how oxonium ions derived from aldehydes or their acetals can induce cyclization. In certain cases (examples 1 and 4, Scheme 16) the stereochemistry found is consistent with a catalyzed 'ene' reaction. The oxidative cyclizations reported by the Corey group (example 5, Scheme 16) may involve oxonium ion intermediates. The successful cyclization⁶¹ of the epoxide (**31**) suggests that benzylic ions could be useful initiators. Another ion which has not been widely examined is that derived from the ketene dithioacetal group (**32**).⁶²







́н


Ionization methods leading to unstabilized cations can give preparatively useful yields when the center is tertiary. In Scheme 17 a variety of examples is shown. Cyclohexene epoxides (example 2, Scheme 17) are generally useful initiators for monocyclizations. The cyclopropyl ketone is effective with aryl termination (example 4, Scheme 17), but in example 5 (Scheme 17) the ion obtained on cyclization is quenched by intramolecular attack of the enol.





1.9.5.3 Seven-membered Ring Formation

The 7-endo cyclization has been little investigated and can be complicated by secondary reactions. Some examples are shown in Scheme 18. The more stringent conditions required and the lower yield obtained in example 1 (Scheme 18) compared with the lower homolog (11) illustrate the difficulties associated with seven-membered ring formation. A common complicating factor (examples 2 and 3, Scheme 18) is further 5-endo cyclization of the ion (33) from monocyclization. This does not occur with the alkynyl compound (example 4, Scheme 18) nor when the double bond is endocyclic to the newly formed ring (examples 6 and 7, Scheme 18). Cyclization in the 7-exo mode (example 5, Scheme 18) is also accompanied by bicyclization and the Johnson group has made use of this in an elegant synthesis of longifolene. The use of recently developed terminators might prevent the second cyclizations occurring.







68

3







69

7 33

6



Ο

Scheme 18





1.9.6 BICYCLIZATIONS

1.9.6.1 Bicyclo[4.3.0] Ring Formation

The Johnson group has used the tetramethylallyl cation as initiator for the synthesis of *trans*-hydrindanes (Scheme 19) using a variety of different terminating groups. The chiral acetal (34) has been



Scheme 19

Alkylation of Carbon

cyclized⁷² in an enantioselective manner to the *trans*-hydrindanes (35) and (36). This accords with the stereochemical model in which the C—O bond next to the axial methyl in the acetal ring breaks and the double bond attacks from the opposite face; the driving force for this selection is postulated to be relief of the axial methyl-hydrogen interaction in the acetal ring. There are two chair-chair conformations (37 \rightarrow 35) and (38 \rightarrow 36) which have this feature, and it is not immediately obvious why (37) is favored over (38).



1.9.6.2 Bicyclo[4.4.0] Ring Formation

The majority of bicyclizations described lead to trans-decalin derivatives and the most extensive results relate to the cyclization of farmesol derivatives and the synthesis of estrone-related compounds (see Section 1.9.8.1). Early work was concerned with testing the validity of the Stork-Eschenmoser hypothesis in accounting for ring junction stereochemistry, and showed that both (Z)- and (E)-isomers gave trans-decalins when the terminating double bond was electron deficient due to conjugation with carboxvlic acid or ester functions, e.g. $(39 \rightarrow 40)$ ² The cyclization was shown to be a stepwise process via a common intermediate formed from (Z)- or (E)-alkenes. Subsequently, the Semenovskii group has shown that alkenes with internal (Z)-double bonds can be cyclized to cis-decalins in acceptable yields (Scheme 20). Other cases where good yields have been reported for *cis*-decalins are from the acetal (example 3, Scheme 20), the acylimonium ion initiated cyclization (example 5, Scheme 20) and with the estrone precursor (example 4, Scheme 20). However, there are a significant number of cases where varying proportions of *trans*-decalins are formed. As a route to *cis*-decalins this approach must be treated with some caution, and careful planning in terms of initiator, nucleophilicity of participating double bonds and termination is required to ensure success. The annulation approach (see Scheme 14) to cis-decalins is more predictable. There is no such reservation in applying the bicyclization route to the synthesis of transdecalins. Scheme 21 shows a variety of farnesyl derivatives which have been cyclized to give useful synthetic intermediates. Where the stereochemical integrity of the initiating double bonds needs to be reflected in the product then, as for monocyclization, $Hg(OCOCF_3)_2$ is the reagent of choice (example 2, Scheme 21). Phenyl-selenyl and -sulfenyl cations offer promise in inducing bicyclizations (examples 4 and 5, Scheme 21), whereas epoxide ring opening and bromonium ion formations have given disappointing results in terms of yield. Undoubtedly the substituted allylic cations and oxonium ions are the most generally effective initiators of bicyclizations. In Scheme 22 some results obtained in these cyclizations are listed. In example 2 (Scheme 22) and example 5 (Scheme 22) bicyclization around a tetraalkylalkene is shown to be possible. The sulfur-stabilized cation, derived from the ketene dithioacetal (example 6, Scheme 22) shows promise as an initiator, as does the acylium ion (example 4, Scheme 22). In the latter case, the vinylsilane terminator is used and the chair-chair transition state geometry is adopted where the bulkier group is equatorial. Use of a chiral acetal allows enantioselective cyclization (example 3, Scheme 21). The preference for one series is in accord with the explanation developed for enantioselection in (34).







Scheme 21 (continued)

1.9.7 TRI- AND TETRA-CYCLIZATIONS

Most cyclizations in this category were designed to form steroids and related compounds and are discussed in Section 1.9.8. Early attempts to cyclize geranylgeraniol derivatives were not promising and it was not until the late 1960s that the Johnson group demonstrated that substantial yields of tri- and tetracyclic compounds could be prepared by Lewis acid induced cyclization of acetals, e.g. (41) \rightarrow (42)⁸⁹ and (43) \rightarrow (44);⁹⁰ mixtures of isomers were obtained due to initiation and termination not being specific, but the carbon skeletons were shown to be *trans-anti-trans-(anti)*. Later the same group showed that the





Alkylation of Carbon

tetramethylallyl initiating group was also effective (Scheme 23) in forming 6,6,6 and 6,6,7 ring systems. It is not surprising that two diastereoisomers are formed in example 2 (Scheme 23) since the initiating group is remote from the chiral terminating function and unable to induce enantioselections in the formation of the first two rings. The alkyne (example 3, Scheme 23) cyclized as anticipated with five-membered ring formation being favored. However, the acylimonium ion induced cyclization⁷⁷ of (45) generates the azasteroid skeleton (46), perhaps due to rearrangement of an intermediate cyclopentylidene cation. Polyalkene cyclizations to form triterpenes have also been examined (Scheme 24). The van Tamelen group has investigated epoxyalkene cyclizations (examples 1 and 2, Scheme 24) extensively, with yields varying from moderate to poor; however, true assessment of the method is difficult since the cyclization precursors were, in a number of cases, mixtures of diastereoisomers. The Ireland group assessed a number of initiating groups (examples 3, 4 and 5, Scheme 24) again with yields varying from moderate to poor. In all of the Scheme 24 examples substantial amounts of bicyclization products were



Scheme 23







obtained. This was also observed by the Corey group⁹⁶ in the cyclization of the ester (47). The cyclization of the furan $(48)^{97}$ shows diastereoselection in favor of the isomer with the sulfonyl group axial. This can be rationalized using Bartlett's suggestion⁵ that only bicyclizations are concerted (Section 1.9.10). The diastereoselection is then not determined by a chair-like conformation for the incipient ring c; instead it is dependent on the relative rates at which conformers form the enantiomeric bicyclic ring systems (49). In particular, the rates of cyclization of the rotamers about the bond indicated will influence diastereoselection. A similar explanation accounts for the formation of two diastereoisomers from cyclization of the alcohol (example 2, Scheme 23); the chiral unit is too distant to have any significant influence in determining the absolute configuration adopted by rings A and B. The tricyclization of the furan (50) must involve independent monocyclization and bicyclization and suggests that this might be a strategy worth exploring further.⁹⁸



1.9.8 SYNTHESIS OF STEROIDS AND RELATED COMPOUNDS

1.9.8.1 Aromatic Steroids

The Johnson group's approach to estrone (Scheme 25)⁹⁹ has been much developed since variants of the basic cyclization have been used to elicit further stereochemical and mechanistic information. Groen



i,HCO₂H; ii, HOBr, KOH; iii, BF₃

Polyene Cyclizations

and Zeelen¹⁰⁰ examined the diastereoselection obtained when a methyl group was introduced at varying positions on the methylenes of the carbon chain (Table 2). The results agree with a product-like chairchair transition state where methyls at C-7 or C-12 favor the equatorial orientation. The methyl at C-6 is found only in the axial orientation, presumably to relieve an unfavorable equatorial methyl interaction with the vinylic methyl group. Axial disposition of the methyl at C-11 is also preferred, presumably to avoid the steric repulsion which would ensue with C-1 if the methyl were equatorial. Cyclization of the thiophene analogs (examples 1 and 2, Scheme 26) gives analogous results. With the dimethyl compound

Table 2 Stereochemistry of Cyclization of Monomethyl Derivatives





Scheme 26

Alkylation of Carbon

(example 3, Scheme 26), the diaxial dimethyl compound is the major product but appreciable amounts of the product with *cis* ring fusion are formed by a stepwise mechanism. A *cis* product is also formed from the (Z)-alkene (example 4, Scheme 26) with the methyl adopting an equatorial orientation. Diastereoselection is also found when a methyl is introduced at C-1 (example 5, Scheme 26), leading to preferred attack on the least-hindered face of the allyl cation. An azasteroid has been prepared by the Speckamp group¹⁰⁴ in excellent yield by cyclization of the amide (**51**), while the Ziegler group¹⁰⁵ has described the novel cyclization of the aldehyde (**52**). A comparison of different initiators by the Sutherland group (Scheme 27) showed that the epoxide (example 1, Scheme 27) was unsatisfactory, giving mainly monocyclization, but that both the enone (example 2, Scheme 27) and the derived methylcyclohexenol (example 3, Scheme 27) could be cyclized in acceptable yields.











106

3



1.9.8.2 Nonaromatic Steroids and Related Compounds

In the first successful production of tetracycles by polyalkene cyclization the Johnson group used oxonium ion initiation to convert an acetal to a p-homosteroid (43). The yield was moderate but trans-antitrans-anti stereochemistry was demonstrated. Subsequently the steroid skeleton was prepared using the allylsilane terminator (example 1, Scheme 28) and, more recently, the yield has been almost doubled by stabilization of the ion notionally produced by bicyclization (example 2, Scheme 28); perhaps two consecutive concerted bicyclizations are involved. The methylcyclohexenyl cation was also an effective trigger for cyclization to D-homosteroids (Scheme 29) when the terminating group was methylene (example 1, Scheme 29) or trimethylsilylalkynyl (example 3, Scheme 29). With the isopropylidene terminator (example 2, Scheme 29) the steroid skeleton was formed, but the cation then underwent hydride and methyl migration. These migrations could be suppressed by stabilization of the terminal cation with aryl groups (example 4, Scheme 29) of which α -naphthyl was the most effective. The *p*-methoxystyryl group was ineffective, probably due to the independent formation of the highly stabilized p-methoxybenzyl cation by protonation of the *p*-methoxystyryl group. Clearly if the terminating cation is formed more readily or competitively with the initiating cation, successful cyclizations are unlikely. Cyclization of the alkyne (example 5, Scheme 29) forms the steroid skeleton without rearrangement. Trapping of the vinylic cation with ethylene carbonate prevents ring expansion to the D-homosteroid. When the terminal group was vinyl⁴⁰ then control over the c-D ring junction stereochemistry was lost; *cis* isomer was formed by a proton elimination-monocyclization sequence. The intervention of this mechanism can be detected by use of O-deuterated acids, since it leads to deuterium incorporation (Scheme 30).



Scheme 28

The dimethylcyclohexenyl cation has been used to prepare D-homosteroid (example 1, Scheme 31) and steroid skeletons (example 2, Scheme 31). One solution to the problem of preparing steroids without an alkyl substituent at C-4 is the use of a phenyldimethylsilyl substituent at one terminus of the allyl cation. Carbon bond formation occurs remote from the silyl substituent, which is then eliminated by protodesilylation *in situ* (Scheme 32). Reduction of the number of alkyl substituents on the allylic cation has a dramatic effect on yield (example 4, Scheme 31), suggesting that useful yields are obtained only when the allylic cations have three or four carbon substituents.

The initiator which has been studied most extensively (Scheme 33)¹¹⁷⁻¹²⁵ is the dimethylcyclopentenyl cation, which is generated from the corresponding alcohol by reaction with protonic or Lewis acids. An





early testosterone synthesis was based on the cyclization (example 1, Scheme 33); ozonolysis and aldol cyclization of the product formed the steroid skeleton. 19-Norsteroids could also be prepared by this route (example 2, Scheme 33). Much time and ingenuity have been spent in deriving direct methods for forming the *trans*-c,D ring fusion of steroids. Besides the stereochemistry of the ring junction, the question of the orientation of pendant group on the cyclopentane also arises. The alkynyl group in various forms is an efficient terminator but up to 20% of the *cis*-c,D isomer (13 α) is formed. The styryl (example 6, Scheme 33), fluorovinyl (example 7, Scheme 33), and allylsilyl (examples 8 and 9, Scheme 33) lead to significantly lesser amounts of 13 α -isomer but roughly equivalent quantities of 17 α - and 17 β -isomers. This is not a major drawback as 20-ketones can be isomerized to mixtures rich in the 17 β natural isomers at later stages. This lack of selectivity is to be contrasted with mono- and bi-cyclizations using these terminating groups where there is a distinct preference for the α -isomers.

As in the estrone cyclization (Scheme 25) substituents introduced on to the methylene chains can induce diastereoselection. The Johnson group cyclized a precursor with a methyl group at pro-C-11 to give only the 11α -isomer (example 1, Scheme 34). The origin of the diastereoselection is likely to lie with the unfavorable interaction between the vinylic methyl and the axially oriented 11β -methyl (53) excluding cyclization in this conformation; cyclization of the compound lacking the vinylic methyl exhibits no diastereoselection. When the methyl group was replaced by hydroxyl (example 2, Scheme 34), the yield and rate of reaction were both reduced, presumably due to the inductive effect of the hydroxyl reducing the double bond nucleophilicity; however, complete diastereoselection was observed and using chiral alcohol an enantiospecific synthesis of 11-hydroxyprogesterone was achieved. This must be regarded as the





Polyene Cyclizations

major achievement of polyalkene cyclization up to the present. With a benzyloxy at pro-C-7 diastereoselection in favor of the axial 7α -ether was found (example 3, Scheme 34). This diastereoselection was improved in the cyclization of the normethyl secondary alcohol in contrast to the pro-C-11-substituted series. These results also contrast with the diastereoselections found in the closely related estrone cyclizations (Scheme 26) where the pro-C-11 methyl favors axial orientation; the absence of the 1,3 methyl-methyl interaction in the estrone precursor may offer a rationalization. The differing selectivities for the pro-C-7 compounds are more difficult to understand. Recently the Johnson group has used the device of two possible consecutive bicyclizations (example 4, Scheme 34) and achieved impressive yields and diastereoselectivity; the effectiveness of this approach is apparent when compared with the relatively poor yields obtained when the ion from bicyclization is not stabilized (example 5, Scheme 34). The Brunke steroid synthesis¹³⁰ has as its basis the cyclization of the tetraene (54) to the tetracycle (55). Presumably the first cyclization forms the allylic cation (56), which undergoes equatorial attack to form the tetracyclic ion. Equatorial attack is preferred over the more usual axial addition to allylic ions since the latter would require ring c to adopt a twist-boat conformation. This form of 'remote' stereochemical control might be capable of further exploitation. Van Tamelen and Hwu¹³¹ have described a progesterone synthesis based on cyclization of a cyclohexene epoxide (57).



Scheme 34





1.9.9 MECHANISM

In 1983 Bartlett wrote an excellent and well-balanced account discussing possible mechanisms for polyalkene cyclizations. In it is summarized the powerful evidence that structural and electronic features in the incipient second ring can influence both the rate and the stereochemical outcome of a reaction. In one case Borčić et al.¹³² found that the diene (58) solvolyzed nine times faster than the alkene (59), and that there were dramatic differences in the enthalpies (ΔH^{\ddagger}) and entropies ($-\Delta S^{\ddagger}$) of activation. In his article⁵ Bartlett asks 'is there concertedness beyond the second ring?', and argues persuasively that the likely answer is 'no', due to the entropic debt which would result in 'freezing' additional bonds. To account for the various effects in bicyclizations there are three distinct options - a concerted reaction, a concerted cyclization to a π -complex, or a stepwise process in which the cyclizations are reversible (Scheme 35). The geometric restrictions on the first two options are similar and distinction between them is difficult; thermodynamic arguments militate against the retrocyclization mechanism. Another mechanistic question arises from the strong tendency of cyclohexyl cations to undergo equatorial attack, particularly when produced by the cyclization route. This raises the possibility of two types of cyclohexyl cation separated by a small energy barrier, one arising from C---C hyperconjugation, the other from C--H hyperconjugation, with the empty p-orbital; the former would lead to equatorial attack. Recent work suggests that there could be two methylcyclohexyl cations.¹³³





(59)



1.9.10 SUMMARY

Polyalkene cyclization is a powerful synthetic method for the preparation of alicyclic compounds containing certain combinations of five-, six- and seven-membered rings. The Stork-Eschenmoser hypothesis can be adopted as a guiding principle in synthetic planning but it is not universally applicable. However, using stabilized cations as initiators and participating double bonds of appropriate nucleophilicity then the notion of concerted bicyclization via chair-chair transition states has good predictive value. It is the synthesis of cis-fused rings from (Z)-alkenes that requires the most careful planning. For tri- and tetra-cyclizations the formations of the third and fourth rings are governed by the rules for the cyclization of cycloalkyl cations. In future synthetic planning it may be advantageous to plan a tricyclization as a bicyclization followed by a monocyclization, and a tetracyclization as two consecutive bicyclizations, *i.e.* the initial bicyclization produces a cation which is itself a good initiator of cyclization. This will require a delicate 'tuning' of reactivity since initiation at the second center would be unproductive; indeed the approach of generating the second cyclization initiator from the terminator of the first cyclization in a separate chemical step may be more productive.

1.9.11 REFERENCES

- 1. J. Simonson and W. C. J. Ross (eds.), 'The Terpenes', Cambridge University Press, Cambridge, vols. 1-7.
- G. Stork and A. W. Burgstahler, J. Am. Chem. Soc., 1955, 77, 5068; A. Eschenmoser, L. Ruzicka, O. Jeger and D. Arigoni, Helv. Chim. Acta, 1955, 38, 1890.
- 3. W. S. Johnson, Bioorg. Chem., 1976, 5, 51.
- (a) E. E. van Tamelen, Acc. Chem. Res., 1968, 1, 111; (b) P. T. Lansbury, Acc. Chem. Res., 1972, 5, 311; (c) K. E. Harding, Bioorg. Chem., 1973, 2, 248; (d) J. K. Sutherland, Chem. Soc. Rev., 1980, 9, 265; (e) W. N. Speckamp, Recl. Trav. Chim. Pays-Bas, 1981, 100, 345.
- 5. P. A. Bartlett, in 'Asymmetric Synthesis', ed. J. D. Morrison, Academic Press, New York, 1984, vol. 3, p. 341.
- 6. E. E. van Tamelen and E. J. Hessler, Chem. Commun., 1966, 411.
- 7. V. A. Smit, A. V. Semenovsky, V. F. Kucherov, T. N. Chernova, M. Z. Krimer and O. V. Lubinskaya, Tetrahedron Lett., 1971, 3101.
- 8. V. A. Smit and A. V. Semenovsky, Tetrahedron Lett., 1965, 3651.
- 9. T. R. Hoye and M. J. Kurth, J. Org. Chem., 1978, 43, 3693.
- 10. F. Rouessac and H. Zamarlik, Tetrahedron Lett., 1981, 22, 2643.
- 11. W. P. Jackson, S. V. Ley and A. J. Whittle, J. Chem. Soc., Chem. Commun., 1980, 1173.
- 12. Y. Yamada, H. Sanjoh and K. Iguchi, J. Chem. Soc., Chem. Commun., 1976, 997.
- 13. M. Julia and J.-D. Fourneron, Tetrahedron, 1976, 32, 1113.
- M. Kurbanov, A. V. Semenovsky, V. A. Smit, L. V. Shmelev and V. F. Kurcherov, Tetrahedron Lett., 1972, 2175; M. T. Mustafaeva, V. A. Smit, A. V. Semenovsky and V. F. Kucherov, Izv. Akad. Nauk SSSR, Ser. Khim., 1973, 1151 (Chem. Abstr., 1973, 79, 77 795b).
- 15. H. E. Ulery and J. H. Richards, J. Am. Chem. Soc., 1964, 86, 3113.

- 16. G. Demailly and G. Solladié, J. Org. Chem., 1981, 46, 3102.
- 17. J. L. Cooper and K. E. Harding, Tetrahedron Lett., 1977, 3321.
- 18. E.-J. Brunke, F.-J. Hammerschmidt and H. Struwe, Tetrahedron Lett., 1981, 22, 5259.
- 19. L. G. Kozar, R. D. Clark and C. H. Heathcock, J. Org. Chem., 1977, 42, 1386.
- 20. H. Kappeler, H. Schinz, A. Eschenmoser and D. Stauffacher, Helv. Chim. Acta, 1954, 37, 957; H. Grütter and H. Schinz, Helv. Chim. Acta, 1952, 35, 1656; H. Grütter, H. R. Vogt and H. Schinz, Helv. Chim. Acta, 1954, 37, 1791.
- 21. A. V. Semenovsky, V. A. Smit, T. N. Chernova and V. F. Kucherov, Izv. Akad. Nauk SSSR, Ser. Khim., 1965, 1068 (Chem. Abstr., 1965, 63, 8407f).
- 22. I. Fleming, A. Pearce and R. L. Snowden, J. Chem. Soc., Chem. Commun., 1976, 182.
- 23. A. J. Frey, A. Nechvatal, P. A. Stadler and A. Eschenmoser, Helv. Chim. Acta, 1957, 40, 1373.
- 24. V. A. Smit, A. V. Semenovsky and V. F. Kucherov, Izv. Akad. Nauk SSSR, Ser. Khim., 1959, 1848 (Chem. Abstr. 1960, 54, 8887f).
- 25. P. T. Lansbury, T. R. Demmin, G. E. DuBois and V. R. Haddon, J. Am. Chem. Soc., 1975, 97, 394; P. T. Lansbury, P. C. Briggs, T. R. Demmin and G. E. DuBois, J. Am. Chem. Soc., 1971, 93, 1311.
- 26. B. Tomita and Y. Hirose, Tetrahedron Lett., 1970, 143.
- 27. K. E. Harding, P. M. Puckett and J. L. Cooper, Bioorg. Chem., 1978, 7, 221.
- 28. F.-H. Köster and H. Wolf, Tetrahedron Lett., 1981, 22, 3937.
- 29. J. A. Amupitan, E. G. Scovell and J. K. Sutherland, J. Chem. Soc., Perkin Trans. 1, 1983, 755.
- 30. D. Schinzer, J. Steffen and S. Sólyom, J. Chem. Soc., Chem. Commun., 1986, 829.
- 31. D. Schinzer, S. Sólyom and M. Becker, Tetrahedron Lett., 1985, 26, 1831.
- 32. G. L. Hodgson, D. F. MacSweeney and T. Money, Chem. Commun., 1971, 766.
- 33. S. Hashimoto, A. Itoh, Y. Kitagawa, H. Yamamoto and H. Nozaki, J. Am. Chem. Soc., 1977, 99, 4192.
- 34. R. C. Cookson and S. A. Smith, J. Chem. Soc., Chem. Commun., 1979, 145.
- 35. I. G. Mursakulov, A. V. Semenovsky and V. A. Smit, Izv. Akad. Nauk SSSR, Ser. Khim., 1967, 384 (Chem. Abstr., 1968, 68, 29 873b).
- 36. Y.-R. Naves, Bull. Soc. Chim. Fr., 1954, 321, 667.
- 37. G. Büchi, E. Koller and C. W. Perry, J. Am. Chem. Soc., 1964, 86, 5646.
- 38. R. P. Linstead, A. F. Millidge and A. L. Walpole, J. Chem. Soc., 1937, 1140.
- A. Eschenmoser, H. Schinz, P. A. Stadler and G. Stork, *Helv. Chim. Acta*, 1957, 40, 2191.
 K. E. Harding, E. J. Leopold, A. M. Hudrlik and W. S. Johnson, *J. Am. Chem. Soc.*, 1974, 96, 2540.
- 41. A. V. Semenovsky, V. A. Smit and V. F. Kucherov, Dokl. Akad. Nauk SSSR, Ser. Khim., 1965, 160, 1097 (Chem. Abstr., 1965, 62, 14 733h).
- 42. M. Julia and J.-D. Fourneron, Bull. Soc. Chim. Fr., 1975, 770.
- 43. (a) R. W. Skeean, G. L. Trammell and J. D. White, Tetrahedron Lett., 1976, 525; (b) D. J. Nelson, P. J. Cooper and J. M. Coerver, Tetrahedron Lett., 1987, 28, 943.
- 44. W. P. Jackson, S. V. Ley and J. A. Morton, J. Chem. Soc., Chem. Commun., 1980, 1028.
- 45. S. Godtfredsen, J. P. Obrecht and D. Arigoni, Chimia, 1977, 31, 62.
- 46. W. S. Johnson, W. H. Lunn and K. Fitzi, J. Am. Chem. Soc., 1964, 86, 1972; W. S. Johnson and K. E. Harding, J. Org. Chem., 1967, 32, 478.
- 47. W. S. Johnson, M. F. Semmelhack, M. U. S. Sultanbawa and L. A. Dolak, J. Am. Chem. Soc., 1968, 90, 2994.
- 48. K. E. Harding, J. L. Cooper and P. M. Puckett, J. Am. Chem. Soc., 1978, 100, 993.
- 49. B. R. Davis and S. J. Johnson, J. Chem. Soc., Chem. Commun., 1978, 614.
- 50. K. E. Harding, R. C. Ligon, T.-C. Wu and L. Rodé, J. Am. Chem. Soc., 1972, 94, 6245.
- 51. E.-J. Brunke, F.-J. Hammerschmidt and H. Struwe, Tetrahedron, 1981, 37, 1033.
- 52. A. B. Kazi and J. K. Sutherland, Tetrahedron Lett., 1985, 26, 2921.
- 53. J. A. Amupitan, E. Huq, M. Mellor, E. G. Scovell and J. K. Sutherland, J. Chem. Soc., Perkin Trans. 1, 1983, 751
- 54. T. Tokoroyama, M. Tsukamoto, T. Asada and H. Iio, Tetrahedron Lett., 1987, 28, 6645.
- 55. J. A. Marshall and P. G. M. Wuts, J. Org. Chem., 1977, 42, 1794.
- T. K. Sarkar and N. H. Andersen, Tetrahedron Lett., 1978, 3513. 56.
- 57. A. Mann, C. Nativi and M. Taddei, Tetrahedron Lett., 1988, 29, 3247.
- 58. A. Itoh, K. Oshima and H. Nozaki, Tetrahedron Lett., 1979, 1783.
- E. J. Corey and D. L. Boger, Tetrahedron Lett., 1978, 2461.
 Y. Kitagawa, S. Hashimoto, S. Iemura, H. Yamamoto and H. Nozaki, J. Am. Chem. Soc., 1976, 98, 5030.
- 61. D. N. Jones and M. R. Peel, J. Chem. Soc., Chem. Commun., 1986, 216.
- 62. N. H. Andersen, Y. Yamamoto and A. D. Denniston, Tetrahedron Lett., 1975, 4547.
- 63. R. E. Ireland, P. Bey, K.-F. Cheng, R. J. Czarny, J.-F. Moser and R. I. Trust, J. Org. Chem., 1975, 40, 1000.
- G. Stork and M. Gregson, J. Am. Chem. Soc., 1969, 91, 2373. 64.
- 65. C. Daesslé, H. Favre and H. Schinz, Helv. Chim. Acta, 1957, 40, 2278.
- 66. F. E. Brot, W. S. Johnson, B. E. Ratcliffe and G. D. Stelling, Bioorg. Chem., 1977, 6, 257.
- 67. R. A. Volkmann, G. C. Andrews and W. S. Johnson, J. Am. Chem. Soc., 1975, 97, 4777.
- 68. K. E. Harding, J. L. Cooper, P. M. Puckett and J. D. Ryan, J. Org. Chem., 1978, 43, 4363.
- 69. P. T. Lansbury and A. K. Serelis, Tetrahedron Lett., 1978, 1909.
- 70. W. S. Johnson and L. A. Bunes, J. Am. Chem. Soc., 1976, 98, 5597; M. B. Gravestock, W. S. Johnson, B. E. McCarry, R. J. Parry and B. E. Ratcliffe, J. Am. Chem. Soc., 1978, 100, 4274.
- 71. A. V. Semenovsky and M. M. Emel'yanov, Izv. Akad. Nauk SSSR, Ser. Khim., 1979, 479 (Chem. Abstr., 1979, 90, 203 742z).
- W. S. Johnson, J. D. Elliott and G. J. Hanson, J. Am. Chem. Soc., 1984, 106, 1138. 72.
- V. F. Kucherov, S. L. Portnova, A. V. Semenovsky and V. A. Smit, Dokl. Akad. Nauk SSSR, Ser. Khim., 1965, 73. 160, 1097 (Chem. Abstr., 1965, 62, 14 733h).
- 74. S. Torii, K. Uneyama, I. Kawahara and M. Kuyama, Chem. Lett., 1978, 455.

- 75. A. van der Gen, K. Wiedhaup, J. J. Swoboda, H. C. Dunathan and W. S. Johnson, J. Am. Chem. Soc., 1973, 95, 2656.
- 76. M. B. Groen, B. Hindriksen and F. J. Zeelen, Recl. Trav. Chim. Pays-Bas, 1982, 101, 148.
- J. Dijkink and W. N. Speckamp, Tetrahedron, 1978, 34, 173.
 M. Nishizawa, H. Nishide and Y. Hayashi, J. Chem. Soc., Chem. Commun., 1984, 467.
- 79. E. J. Corey, M. A. Tius and J. Das, J. Am. Chem. Soc., 1980, 102, 1742.
- 80. G. E. Muntyan, M. Kurbanov, V. A. Smit, A. V. Semenovsky and V. F. Kucherov, Izv. Akad. Nauk SSSR, Ser. Khim., 1973, 633 (Chem. Abstr., 1973, 79, 18 869f).
- 81. E. D. Edstrom and T. Livinghouse, J. Org. Chem., 1987, 52, 949.
- 82. D. Nasipuri and G. Das, J. Chem. Soc., Perkin Trans. 1, 1979, 2776.
- W. S. Johnson, N. P. Jensen, J. Hooz and E. J. Leopold, J. Am. Chem. Soc., 1968, 90, 5872.
 R. E. Ireland, M. I. Dawson, J. Bordner and R. E. Dickerson, J. Am. Chem. Soc., 1970, 92, 2568.
- 85. W. S. Johnson, C. A. Harbert, B. E. Ratcliffe and R. D. Stipanovic, J. Am. Chem. Soc., 1976, 98, 6188.
- 86. S. D. Burke, J. O. Saunders, J. A. Oplinger and C. W. Murtiashaw, Tetrahedron Lett., 1985, 26, 1131.
- 87. R. E. Ireland, T. C. McKenzie and R. I. Trust, J. Org. Chem., 1975, 40, 1007.
- 88. R. S. Brinkmeyer, Tetrahedron Lett., 1979, 207.
- 89. G. D. Abrams, W. R. Bartlett, V. A. Fung and W. S. Johnson, Bioorg. Chem., 1971, 1, 243.
- 90. W. S. Johnson, K. Weidhaup, S. F. Brady and G. L. Olson, J. Am. Chem. Soc., 1974, 96, 3979.
- 91. W. S. Johnson and T. K. Schaaf, Chem. Commun., 1969, 611.
- 92. G. D. Prestwich and J. N. Labovitz, J. Am. Chem. Soc., 1974, 96, 7103.
- W. S. Johnson, M. B. Gravestock, R. J. Parry, R. F. Myers, T. A. Bryson and D. H. Miles, J. Am. Chem. Soc., 93. 1971. 93. 4330.
- 94. E. E. van Tamelen and R. J. Anderson, J. Am. Chem. Soc., 1972, 94, 8225.
- 95. E. E. van Tamelen, R. A. Holton, R. E. Hopla and W. E. Konz, J. Am. Chem. Soc., 1972, 94, 8228.
- 96. E. J. Corey and R. M. Burk, Tetrahedron Lett., 1987, 28, 6413.
- 97. M. Nishizawa, H. Yamada and Y. Hayashi, Tetrahedron Lett., 1986, 27, 187.
- 98. U. C. Pandey, P. Sarmah and R. P. Sharma, Tetrahedron, 1984, 40, 3739.
- 99. P. A. Bartlett and W. S. Johnson, J. Am. Chem. Soc., 1973, 95, 7501.
- 100. M. B. Groen and F. J. Zeelen, Recl. Trav. Chim. Pays-Bas, 1978, 97, 301; 1979, 98, 32; M. B. Groen and F. J. Zeelen, J. Org. Chem., 1978, 43, 1961; A. I. A. Broess, N. P. van Vliet and F. J. Zeelen, J. Chem. Res. (S), 1981, 20.
- 101. A. A. Macco, J. M. G. Driessen-Engels, M. L. M. Pennings, J. W. de Haan and H. M. Buck, J. Chem. Soc., Chem. Commun., 1978, 1103.
- 102. A. A. Macco, R. J. de Brouwer, P. M. M. Nossin, E. F. Godefroi and H. M. Buck, J. Org. Chem., 1978, 43, 1591.
- 103. M. B. Groen and F. J. Zeelen, Recl. Trav. Chim. Pay-Bas, 1979, 98, 239.
- 104. J. Dijkink and W. N. Speckamp, Tetrahedron Lett., 1977, 935.
- 105. F. E. Ziegler and T.-F. Wang, J. Am. Chem. Soc., 1984, 106, 718.
- 106. J. A. Amupitan, R. L. Beddoes, O. S. Mills and J. K. Sutherland, J. Chem. Soc., Perkin Trans. 1, 1983, 759.
- 107. W. S. Johnson, Y.-Q. Chen and M. S. Kellog, J. Am. Chem. Soc., 1983, 105, 6653.
- 108. W. S. Johnson, S. J. Telfer, S. Cheng and U. Schubert, J. Am. Chem. Soc., 1987, 109, 2517.
- 109. R. L. Carney and W. S. Johnson, J. Am. Chem. Soc., 1974, 96, 2549.
- 110. W. S. Johnson, T. M. Yarnell, R. F. Meyers and D. R. Morton, Tetrahedron Lett., 1978, 2549.
- 111. K. A. Parker and W. S. Johnson, J. Am. Chem. Soc., 1974, 96, 2556
- 112. M. E. Garst, Y.-F. Cheung and W. S. Johnson, J. Am. Chem. Soc., 1979, 101, 4404.
- W. S. Johnson, B. E. McCarry, R. L. Markezich and S. G. Boots, J. Am. Chem. Soc., 1980, 102, 352. 113.
- W. R. Bartlett and W. S. Johnson, Bioorg. Chem., 1975, 4, 342. 114.
- 115. P. M. Bishop, J. R. Pearson and J. K. Sutherland, J. Chem. Soc., Chem. Commun., 1983, 123.
- 116. J. A. M. Peters, T. A. P. Posthumus, N. P. van Vliet, F. J. Zeelen and W. S. Johnson, J. Org. Chem., 1980, 45, 2208.
- 117. W. S. Johnson, T. Li, C. A. Harbert, W. R. Bartlett, T. R. Herrin, B. Staskun and D. H. Rich, J. Am. Chem. Soc., 1970, 92, 4461.
- 118. W. S. Johnson, M. B. Gravestock and B. E. McCarry, J. Am. Chem. Soc., 1971, 93, 4332.
- W. S. Johnson, W. F. Huffman and S. G. Boots, Recl. Trav. Chim. Pays-Bas, 1979, 98, 125. 119.
- 120. W. S. Johnson, L. R. Hughes and J. L. Carlson, J. Am. Chem. Soc., 1979, 101, 1281.
- 121. W. S. Johnson, G. W. Daub, T. A. Lyle and M. Niwa, J. Am. Chem. Soc., 1980, 102, 7800.
- W. S. Johnson, L. R. Hughes, L. A. Kloek, T. Niem and A. Shenvi, J. Am. Chem. Soc., 1979, 101, 1279. 122.
- 123. R. Schmid, P. L. Huesmann and W. S. Johnson, J. Am. Chem. Soc., 1980, 102, 5122.
- 124. L. R. Hughes, R. Schmid and W. S. Johnson, Bioorg. Chem., 1979, 8, 513.
- 125. W. S. Johnson, C. Newton and S. D. Lindell, Tetrahedron Lett., 1986, 27, 6027.
- W. S. Johnson and G. E. DuBois, J. Am. Chem. Soc., 1976, 98, 1038. 126.
- W. S. Johnson, R. S. Brinkmeyer, V. M. Kapoor and T. M. Yarnell, J. Am. Chem. Soc., 1977, 99, 8341. 127.
- W. S. Johnson, S. D. Lindell and J. Steele, J. Am. Chem. Soc., 1987, 109, 5852.
 W. S. Johnson, D. Berner, D. J. Dumas, P. J. R. Nederlof and J. Welch, J. Am. Chem. Soc., 1982, 104, 3508.
 E.-J. Brunke, C.-H. Kappey, N. Müller and H. Wolf, Chem. Ber., 1980, 113, 2714.
- 131. E. E. van Tamelen and J. R. Hwu, J. Am. Chem. Soc., 1983, 105, 2490.
- 132. O. Kronja, E. Polla and S. Borčić, J. Chem. Soc., Chem. Commun., 1983, 1044.
- 133. R. P. Kirchen, K. Ranganayakulu and T. S. Sorensen, J. Am. Chem. Soc., 1987, 109, 7811.

1.10 Transannular Electrophilic Cyclizations

DAVID C. HARROWVEN and GERALD PATTENDEN

University of Nottingham, UK

1.10.1 INTRODUCTION	379
1.10.2 CYCLOOCTENES	380
 1.10.2.1 Bicyclo[3.3.0]octane and Bicyclo[3.2.1]octane Ring Formation 1.10.2.2 Caged Compounds 1.10.2.3 Endocyclic 1,3- and 1,4-Cyclooctadienes 1.10.2.4 Synthesis of Natural Products 	380 382 383 384
1.10.3 CONFORMATIONAL ARRANGEMENTS OF 1,5-DIENES IN MEDIUM RINGS	386
1.10.4 CYCLONONA-1,5-DIENES	386
1.10.5 CYCLODECENES	388
 1.10.5.1 Relevance to the Biosynthesis of Polycyclic Sesquiterpenes 1.10.5.2 Cyclodecadienes 1.10.5.2.1 Cyclodeca-1 5-dienes 1.10.5.2.2 Other cyclodecadiene systems 1.10.5.3 Cyclodecenols 1.10.5.4 Cyclodecadiene Monoepoxides 	388 389 389 390 393 393 393
1.10.6 CYCLOUNDECENES	398
1.10.6.1 Humulene 1.10.6.2 Humulene Epoxides 1.10.6.2.1 Humulene-1,2-epoxide 1.10.6.2.2 Humulene-4,5-epoxide 1.10.6.2.3 Humulene-8,9-epoxide	399 402 402 404 404
1.10.7 OTHER CYCLOALKENES	406
1.10.8 REFERENCES	407

1.10.1 INTRODUCTION

It is now almost 40 years since Cope and Prelog first described the phenomenon of 'transannulation' reactions during their independent studies of electrophilic additions to medium ring cycloalkenes.^{1,2} Since that time transannular reactions have been encountered in an enormous variety of carbo- and hete-ro-cyclic medium ring compounds having a wide array of differing functionalities. Indeed, the chemistry of these fascinating processes has now become part of the day-to-day armoury of the modern synthetic chemist.³

Transannular reactions are defined as those reactions which lead to the formation of a covalent bond between atoms on opposite sides of a ring compound. They occur frequently in medium, *i.e.* 8-, 9-, 10- and 11-membered, rings where favored conformations allow opposite sides of the rings to come close to each other for bond formation. Probably the most commonly encountered transannulation processes are

those involving H-transfer reactions,⁴ from, for example, ring epoxides, *i.e.* (1) \rightarrow (2; equation 1),⁵ and those involving ring closure reactions, *i.e.* (3) \rightarrow (4; equation 2)⁶ from cyclic polyenes in the presence of electrophilic reagents, *i.e.* conditions which favor carbonium ion formation. Transannular ring closure reactions *via* intermediate carbanions,⁷ carbenes⁸ and radicals,⁹ and *via* pericyclic processes,¹⁰ are also known. So too are several reactions involving nitrogen,¹¹ sulfur,¹² and oxygen¹³ heterocyclic ring formation.



The chemistry of transannular cyclizations involving 1,5-diene units in medium ring carbocycles has been studied with particular vigor in recent years because of the relevance of this chemistry to the biogenetic origins of cyclic terpenes.¹⁴ Indeed, these fundamental studies have now led to a number of novel 'biogenetically inspired' syntheses of a large range of complex ring-fused natural products of both biosynthetic significance and biological importance.¹⁵

This chapter summarizes those transannular reactions in medium ring carbocycles which result in carbon-carbon bond formation *via* alkylation of carbon, in the presence of electrophilic reagents. It is therefore a logical extension of two other chapters in this volume, covering Friedel-Crafts Alkylations (Chapter 1.8), and Polyene Cyclizations (Chapter 1.9). The discussion of these processes is organized according to the size of the cycloalkene from which transannular cyclization is effected.

1.10.2 CYCLOOCTENES

1.10.2.1 Bicyclo[3.3.0]octane and Bicyclo[3.2.1]octane Ring Formation

Following on from earlier work on transannular H-transfer reactions in cyclooctenes (*i.e.* equation 1), Cope and collaborators^{1,5} carried out extensive studies of the solvolyses of a wide range of substituted cyclooctenes in a search for further unusual transannular processes. Thus, both the tosylate (**5a**) and the brosylate (**5b**) of cyclooct-4-en-1-ol were shown to produce the bicyclo[3.3.0]octanes (**6**) and the bicyclo[3.3.0]octane (**7**) on solvolysis and acetolysis respectively (equation 3).¹⁶ The bicyclic molecules (**6**) and (**7**) are formed from (**5**) by transannular interaction in the cyclooctenyl carbonium ion (**8**), followed by quenching of the resulting carbonium ion (**9**) with a nucleophilic counterion (to form **6**), or deprotonation (to form **7**). In the case of acetolysis of (**5a**), a small amount of the bicyclo[3.2.1]octane (**10**) was produced concurrently *via* the rearranged carbonium ion (**11**; Scheme 1). Interestingly, the corresponding isomeric cyclooct-3-en-1-ol brosylate (**12**; equation 4) and the homologous brosylate (**14**; equation 5)



under the same conditions undergo transannulation to the bicyclic products (13) and (15), respectively. $^{16a,17-19}$



Transannular cyclization in cycloocta-1,5-diene (16; equations 6 and 7) can be brought about by a wide variety of electrophilic reagents.¹⁸⁻³² These reactions lead to a range of functionalized bicyclo[3.3.0]octanes (Tables 1 and 2), many of which have found widespread use in general synthesis.³³ Although the use of mineral acids favors the formation of bicyclo[3.3.0]octanes, other electrophilic agents



 Table 1
 Transannular Cyclization of (16; Equation 6)



instead lead to bicyclo[3.2.1]octanes, i.e. (21), or to products resulting from hydride shifts, i.e. (18), and/or Wagner-Meerwein rearrangements, *i.e.* (21; see Scheme 2).



 Table 2
 Transannular Cyclization of (16; Equation 7)



The high degree of selectivity observed in the aforementioned transannulation process can be rationalized in terms of Baldwin's rules,³⁴ and the possible low energy conformation available to the ring.^{35,36} This situation becomes complicated however in the case of 1,5-dimethylcycloocta-1,5-diene (22), since transannulation first leads to the more stable bicyclo[4.2.0]octane carbonium ion (23), which then undergoes a series of Wagner-Meerwein rearrangements (Scheme 3) before producing either the bicyclo[3.3.0]octane (24) or the bicyclo[3.2.1]octane (25), as shown in Table 3.3^{7-39} Surprisingly the monoepoxide derived from 1,5-cyclooctadiene affords only a small yield of bicyclo[3.3.0]octane-2,6diol on transannular cyclization.3b,40

 Table 3
 Transannular Cyclization and Rearrangement of (22; Scheme 3)

Reagents	N	Yield (%)		Ref.
		(24)	(25)	
BF3·OEt2	Н	60		37
TsOH HCO2H/HCIO4	OTs OCHO	16	70 62	37 38
AcOH/HClO4	OAc	32	38	38

1.10.2.2 Caged Compounds

Numerous examples of transannulation reactions amongst caged compounds, which can be considered as conformationally restricted medium rings, have been studied, and some representative examples are



collected in Scheme 4.^{41–44} Generally, such reactions lead to the products of 'crossed' rather than 'parallel' cyclization, *i.e.* fused 5,5-ring products predominate over fused 6,4-rings.⁴⁵



1.10.2.3 Endocyclic 1,3- and 1,4-Cyclooctadienes

Transannulation reactions involving cyclooctenes are not restricted to endocyclic alkenes or to those systems which require a 1,5-relationship of the reacting centers. Thus, the exocyclic polyenes (26; equation 8) and (27; equation 9) take part in some novel transannulation processes when treated with bromonium ion.^{46,47} and both cycloocta-1,3-diene and cycloocta-1,4-diene undergo transannulation reactions under favorable conditions producing the bicycles (28; equation 10) and (29; equation 11), respectively, in very acceptable yields.^{48,49} Perhaps even more novel is the transannular cyclization of 5-methylenecyclooctanecarbaldehyde (30) in water, leading to the unstable bicyclo[3.3.0]octanecarbaldehyde (32); this reaction is best formulated as proceeding via the corresponding enol (31) under acid catalysis (equation 12).⁵⁰



1.10.2.4 Synthesis of Natural Products

Transannulation of the mesylate (33), derived from 4,8-dimethylcycloocta-4-en-1-ol, using sodium carbonate in aqueous dioxane has provided the bicyclo[3.3.0]octanol (34), a central precursor to the monoterpene iridomyrmecin (35; equation 13).⁵¹ In studies of the biomimetic synthesis of the natural triquinanes capnellene (37; equation 14) and pentalenene (40; equation 15) Pattenden *et al.* have shown that both molecules can be produced from appropriate cycloocta-1,5-diene precursors, *i.e.* (36) and (39) [or indeed their corresponding positional isomers (38) and (41), respectively] by treatment with boron trifluoride.^{52,53} Mehta *et al.* have described an alternative transannulation approach to the triquinane unit found in pentalenene, ⁵⁴ *i.e.* (42) \rightarrow (43; equation 16), and also to the ring system (44; equation 17) found

in the diterpene laurenene (45).⁵⁵ The key feature of Chatterjee's approach to the sesquiterpene isocomene (47) is the transannular cyclization of the alkene epoxide (46) in the presence of TFA in CH₂Cl₂ at -78 °C (equation 18); however, doubt has been cast on these results.⁵⁶



(46)

0

(47)

1.10.3 CONFORMATIONAL ARRANGEMENTS OF 1,5-DIENES IN MEDIUM RINGS

Three distinct conformational arrangements for a 1,5-diene set within a medium ring have been identified, and these are shown in formulae (48)–(50).^{3c} Conformation (48) is referred to as 'crossed', abbreviated to C, whereas the conformation (49) is 'parallel', abbreviated to T. Formula (50) indicates an extended conformation. Models show that the C-conformation (48) can be adopted by (E,E)- and (Z,E)-1,5-dienes in medium rings, whereas the T-conformation (49) is possible for (E,E)-, (Z,E)- and (Z,Z)isomers. The extended conformation (50) is available only to (Z,Z)-1,5-dienes. The conformations adopted by an isomeric 1,5-diene in a medium ring are crucial in determining the stereochemical outcome of any transannular cyclization product. These principles have been amply demonstrated during studies of the stereospecificities in the cyclizations of 9-, 10- and 11-membered cyclic 1,5-dienes, summarized below.⁵⁷



1.10.4 CYCLONONA-1,5-DIENES

Transannular cyclization of the (Z,Z)-isomer (51) of cyclonona-1,5-diene in the presence of a variety of electrophilic reagents provides a useful synthesis of a range of substituted *cis*-hydrindane derivatives (53; Table 4).⁵⁸⁻⁶¹ In each of these cases the 1,5-diene unit in (51) is reacting in the T-conformation shown in (52).



 Table 4
 Synthesis of cis-Hydrindane Derivatives (53) from 1,5-Diene (51)

Reagents	Ε	N	Yield (%)	Ref.
Br2/AcOH Hg(OAc)2/NaBH4 Br2/CCl4 PhSeCl/AcOH NBS/H2O NBS/MeOH	Br H Br SePh Br Br	OAc OH Br OAc OH OMe	71 68 68 44 90	58, 59 58, 60 58 60 59 59 59

Several (Z,E)-cyclonona-1,5-dienes of the type (55; equation 19) have been prepared via rearrangement of dibromobicyclo[6.1.0]nonenes (54) in the presence of silver salts.⁶² On treatment with aqueous NBS these molecules undergo transannular cyclization, following initial attack on the (Z)-disubstituted double bonds, producing the corresponding *trans*-hydrindanes (56).⁶³ In related studies, Sutherland and his coworkers have also shown that similar treatment of (57; equation 20) leads to the bromo ketone (58),⁶⁴ and that when the epoxide (59; equation 21) is simply crystallized, it undergoes a unique transannular cyclization, producing a mixture of (60) and (61).^{3c,65}

Some of the most remarkable transannular cyclizations have been uncovered during the treatment of natural caryophyllene (63) with mineral acids.^{66–68} Thus, treatment of (63) with concentrated sulfuric acid has been shown to lead to a mixture of clovene (62), caryolan-1-ol (64) and neoclovene (65;



Scheme 5). Similarly, isocaryophyllene (66) undergoes transannular cyclization in mineral acid to (65) and to the tricyclic alkene (67; Scheme 6) in high yield.



Scheme 5



1.10.5 CYCLODECENES

1.10.5.1 Relevance to the Biosynthesis of Polycyclic Sesquiterpenes

Transannular cyclizations involving carbonium ions generated from cyclodeca- and cycloundeca-polyenes have been the subject of extensive research, largely as a consequence of their relevance to our understanding of the biosynthesis of a wide range of polycyclic sesquiterpenes.¹⁴ Thus, macrocyclization of farnesyl pyrophosphate (68) first produces both the 10-ring (69) and the 11-ring (70) carbonium ion intermediate precursors to germacrene (71) and humulene (72), respectively (Scheme 7). According to biogenetic speculation, transannulation involving the cyclodeca-1,5-diene system in germacrene (71) then produces the eudesmane (73) and guaiane (74) groups of natural sesquiterpenes.¹⁴ It is generally accepted that the eremophilane (75) and the pseudoguianane (76) carbon skeletons are derived in nature by rearrangement of eudesmane and guaiane derivatives, respectively (see Scheme 8). In like manner, and using a series of regio- and stereo-selective transannulation reactions, sometimes in tandem with carbo-



nium ion rearrangements, the 11-ring humulene (72) gives rise to the caryophyllane (77), protoilludane (78), hirsutane (79), capnellane (80), pentalenane (81) and other families of sesquiterpenes (Scheme 9).¹⁴



Scheme 9

1.10.5.2 Cyclodecadienes

1.10.5.2.1 Cyclodeca-1,5-dienes

Although some detailed and fundamental studies of transannulation reactions involving cyclodeca-1,5diene itself have been made, the main body of published work in this area has been associated with transannular reactions of germacrenes (71). These reactions generally proceed in both regio- and stereo-selective fashion leading to bicyclo[4.4.0]decanes of defined stereochemistry. This stereochemistry is determined largely by the geometrical configurations of the double bonds in the cyclodecadienes. Thus, the (E,E)-isomer (82; equation 22) of a cyclodeca-1,5-diene undergoes electrophilic cyclization producing the *trans*-decalin (83), whereas the corresponding (E,Z)-isomer (84; equation 23) leads to the *cis*-decalin (85).



In a detailed study of transannular cyclizations of all the possible isomers of naturally occurring hedycaryol (86) in the presence of formic or tosic acid, Itô *et al.* have been able to rationalize the reaction pathways on the basis of the preferred conformations, *i.e.* crossed conformation (CC, CT) or parallel conformations (TT, TC), of the reacting dienes (Scheme 10).^{69,70}

A selection of examples (equations 24-31) demonstrate the considerable scope for transannular reactions of cyclodeca-1,5-dienes in the synthesis, including biomimetic synthesis, of a wide range of isomerically pure, substituted bicyclo[4.4.0]decanes.⁷¹⁻⁸⁶



1.10.5.2.2 Other cyclodecadiene systems

Treatment of the bicyclogermacrene (89; equation 32), containing a cyclopropane ring, with dilute sulfuric acid leads to the aromadendrene (90; cf. guaiane 74), together with the bicyclo[4.4.0]decadiene



Scheme 10

(91).⁸⁷ Interestingly, when the 1,5-vinylcyclopropane (92; equation 33) is treated with acetic acid under reflux for six days it produces largely the tricyclic acetate (93; 58%).⁸⁸

Geometrical isomers of cyclodeca-1,6-dienes show a similar reactivity profile to the corresponding 1,5-dienes in their transannular cyclizations to bicyclo[4.4.0]decanes, *i.e.* (Z,Z)-isomers lead to *cis*-decalins, whereas (E,E)-isomers produce *trans*-decalin products.⁸⁹⁻⁹¹ Some pertinent examples are shown in equations (34-37).



The cyclodecatetraene (94), incorporating an allene unit, undergoes transannulation accompanied by rearrangement in the presence of HgSO₄, producing a mixture of the bicyclo[4.4.0]decenol (95) and the vinylcyclopropane (96; Scheme 11).⁹²


1.10.5.3 Cyclodecenols

Very early on in the development of the principles of transannulation, Cope and coworkers showed that the tosylates (97; equation 38) and (100; equation 39), derived from (Z)- and (E)-cyclodec-5-enol, underwent smooth transannulation in the presence of acetic acid, leading to mixtures of the isomeric bi-cyclo[4.4.0]decenes (98), (99) and (101) together with the cyclodecenol (102).⁹³ Later work by Goering *et al.* demonstrated that the rate of solvolysis of the *p*-nitrobenzoate derivative of (E)-cyclodec-5-enol was such that chiral starting materials led to chiral products, (*e.g.* 103 \rightarrow 104; equation 40).⁹⁴ Similar to

n

Ö



Scheme 11

the observation made with geometrically defined cyclodeca-1,5-dienes, described earlier, the (Z)-cyclodecene (105; equation 41), corresponding to (97), produced the *cis*-decanol (106) on solvolysis in acetone at 100 $^{\circ}$ C.⁹⁵



Marshall and Huffman have employed a neat combination of the aforementioned principles to access the hydroazulene system (109) by solvolysis of (107) in the presence of aqueous dioxane; the reaction (107) \rightarrow (109; equation 42) proceeds *via* the incipient allylic carbonium ion intermediate (108).⁹⁶ Indeed, subsequent to this work, a wide range of cyclodecenols (many naturally occurring) have been used to synthesize both hydroazulene (guaiane) and bicyclo[4.4.0]decane ring systems, and a selection is given in equations (43-48).⁹⁷⁻¹⁰²





Transannulations involving cyclodec-5-ynols have also been described, and they lead largely to bicyclo[4.4.0]decanones (equations 49–53).^{103,104}



1.10.5.4 Cyclodecadiene Monoepoxides

Transannular cyclizations of cyclodecadiene monoepoxides have been studied in exhaustive detail, particularly amongst natural germacranolides, because of their relevance to biogenetic models for the synthesis of eudesmane- and guaiane-type sesquiterpenes.^{105–115} A full discussion of this topic is beyond the scope of this chapter, but the examples collected in equations (54–64) provide the reader with an in-

dication of the scope for this transannulation strategy in synthesis. The specificity of cyclization to either the [3,5]- or [4,4]-bicyclodecane systems can be predicted largely on the basis of the stability of the carbonium ion resulting from initial epoxide ring opening. In some instances rearrangement of the product resulting from transannulation of certain cyclodecene epoxides has been found to be a significant problem, e.g. (110) \rightarrow (111; Scheme 12).¹¹⁶





1.10.6 CYCLOUNDECENES

As summarized in the above discussion (Section 1.10.5.1) and in Scheme 9, the 11-ring triene humulene (72) is a central intermediate in the biosynthesis of a very wide range of polycyclic sesquiterpenes. Consequently, studies of the chemistry, the stereochemistry and the conformation of humulene, together with its use in biomimetic synthesis have been the subjects of considerable research. Indeed, studies of transannulation reactions amongst cyclononadecenes have been dominated by humulene, and this is reflected in this discussion. The humulene ring system is numbered here according to the IUPAC recommendations, and this numbering is shown on formula (112). It should be noted that some authors, e.g. Shirahama *et al.*, use an alternative numbering scheme in their publications.



1.10.6.1 Humulene

More than 20 years ago Sutherland *et al.*¹¹⁷ first showed that when humulene (72) was treated with *N*bromosuccinimide in aqueous acetone it underwent a remarkable transannulation, leading to the tricyclic bromohydrin (113) in 25% yield. Hydrolysis of (113a) then led to tricyclohumuladiol (113b), which was later found as a constituent of hop oil, and two synthetic steps converted (113b) into the cyclononadiene caryophyllene (63; equation 65).



Later work by Parker et al.¹¹⁸ demonstrated that extended treatment of humulene with aqueous sulfuric acid in acetone led largely to the bicyclo[5.3.0]deca-2,10-diene (115), via initial formation of humulol

(114). Other products which were isolated from this study were the related fused 5,7-ring molecules (117)–(119), together with α -caryophyllene alcohol (116; Scheme 13).



Scheme 13

In extensive studies of oxymercuration reactions of humulene, Matsumoto and Shirahama have shown that, whereas oxymercuration with Hg(OAc)₂ in THF-H₂O followed by demercuration leads largely to the tricyclic ether (120; equation 66),¹¹⁹ use of Hg(NO₃)₂ in AcOH-H₂O instead produces mainly the isomeric ether (121).¹²⁰ Interestingly, when humulene was treated with Hg(NO₃)₂ followed by aqueous HBr, chromatography led to the two bromomercury compounds (122) and (123) in a combined yield of 52%.¹²¹ The mercury compounds were then elaborated, in several steps, to the novel cyclooctenol (124) which was beautifully set-up for a further transannulation reaction leading to the hydroxypentalenene (125); several synthetic steps then led to pentalenic acid (126; equation 67).¹²² In further extensive studies Matsumoto and Shirahama and their collaborators have also shown how the 10α -alcohol (127), derived from (123), can be used to elaborate pentalenolactones G and H, (128) and (129) respectively (Scheme 14),¹²³ and how the ether (121) can be converted into pentalenene, pentalenolactone E (130) and pentalenolactone F (131; Scheme 15.124 These studies are therefore based on the biogentic speculation alluded to earlier whereby the 11-ring humulene is converted into the angular fused pentalenane 5,5,5-ring system in sequence, via the intermediate fused 5,8-ring system (132; Scheme 16). In related work Pattenden and Teague⁵³ have also described a total synthesis of pentalenene (40), based on the transannulation reaction shown earlier, *i.e.* (39) or (41) \rightarrow (40; Scheme 15). Other extensive studies of the transannulation chemistry of the tricyclic ether (134) by the same Japanese group, led by Matsumoto













HO H O



steps

(127)



Scheme 14





and Shirahama, have led to the synthesis of natural sterpurene $(133)^{125}$ and of hirsutene (135; equation 68).¹²⁶



1.10.6.2 Humulene Epoxides

The three monoepoxides of humulene, *i.e.* (136), (150) and (162), occur in nature, 127 and there is every reason to assume that each epoxide is involved separately in transannulation processes leading to various fused-ring systems found amongst the sesquiterpenes. All three humulene monoepoxides have been synthesized and their individual chemistry with Lewis acids has been examined in an effort to effect their biogenetic conversions to natural terpenes.

1.10.6.2.1 Humulene-1,2-epoxide

The $\Delta^{1,2}$ -double bond in humulene (72) has been found to be the most reactive towards electrophiles, and hence the chemistry of the corresponding 1,2-epoxide (136) was the first to be studied. Thus, early work of McKervey and Wright¹²⁸ demonstrated that when (136) was treated with 20% H₂SO₄ in acetone, it produced the tricyclohumuladiol (113b), described earlier by Sutherland *et al.*¹¹⁷ Subsequently Takahashi and his coworkers¹²⁹ showed that the diol (113b) could be produced in 95% yield from humulene-1,2-epoxide when the latter is treated with aqueous H₂SO₄ in acetone at 0 °C instead of at room temperature. Continuation of this reaction at room temperature for 7 h, however, gave rise to the new [6.3.0.0^{2,4}]undec-9-enol (138) in *ca.* 25% yield. Even later work by Shirahama *et al.*¹³⁰ showed that when (136) was treated with trimethylsilyl triflate (TMSOTf) at -75 °C the alkene (139) corresponding to (138) could be produced cleanly, in one step in 80% yield (Scheme 17).

Contemporaneous studies of the conformational behavior of humulene (72) by Shirahama *et al.*^[31] using empirical force calculations have suggested that, although the molecule can assume any of the four stable conformations (140), (141), (142) and (143), abbreviated to CT, CC, TT and TC respectively, (see Section 1.10.3), only the CT- and the CC-conformers (140) and (141) are involved in biosynthetic transannulation reactions. Thus, Shirahama *et al.* have suggested that the transannulation reaction leading from (136) to (139) in the presence of TMSOTf proceeds *via* the CT-conformer (137) of humulene-1,2-epoxide. Interestingly, when (139) was converted into the epoxide (144), treatment with TMSOTf trig-





(137)

(136)









10

(138)

(139)

Scheme 17













TT (142)



steps

steps (139)

(146)



(144)









H



(147)

н^н



^{'''}OSiMe₃

OH

gered a series of carbonium ion rearrangements, *i.e.* (145) \rightarrow (146) \rightarrow (147) leading to the intermediate (148), which could be transformed into naturally occurring capnellene (149; Scheme 18).¹³²

1.10.6.2.2 Humulene-4,5-epoxide

The biosynthesis of the hirsutane, protoilludane, marasmane, illudane and africane families of sesquiterpenes are all thought to occur *via* initial protonation of the $\Delta^{4,5}$ -double bond in humulene (*cf.* Scheme 9). Attempts to mimic these reactions in the laboratory were frustrated for many years due to the inaccessibility of the 4,5-epoxide (150) of humulene. An indirect synthesis of humulene-4,5-epoxide permitted Roberts and coworkers¹³³ to study the cyclization of (150) in the presence of BF₃·OEt₂ (equation 69). This study led to the formation of the two alcohols (152a) and (153), which had the same tricyclic carbon framework to that found in naturally occurring africanol (154) and the related ester (152b).¹³⁴



Sometime later, Matsumoto and Shirahama¹³⁵ demonstrated that when humulene-4,5-epoxide (150) was treated with $BF_3 \cdot OEt_2$ in HOAc at -50 °C, the only product isolated was the bicyclohumulenediol (156), in 70% yield. The formation of (156) was rationalized as occurring *via* the CC-conformation (155; equation 70) of the epoxide, whereas the production of the africanol system (152) occurs *via* the corresponding CT-conformation (151) of (150).



Appropriate manipulations of the tricycle (152a) resulting from transannulation of humulene-4,5epoxide permitted Roberts and Shirahama and their respective collaborators to access africanol (154) and 8-oxysenoxyn-4-en-3-one (157).¹³⁶ Shirahama *et al.*¹³⁷ have also described the *in vitro* conversion of africanol (154) to dactylol (158), found in the sea hare *Aplsysia dactyloma*, via the novel and unusual cyclopropane-sliding reaction, summarized in Scheme 19.





i, POCl₃/pyridine; ii, MCPBA + chromatography

Scheme 19

Finally, transannular cyclization involving the $\Delta^{8,9}$ - rather than the $\Delta^{1,2}$ -double bond in humulene-4,5epoxide (150) can be achieved by first converting (150) to the corresponding aldehyde (159). Treatment of (159) with TMSOTf is then shown to lead to the fused 5,7-ring system (160) in good yield (equation 71).¹³⁸



1.10.6.2.3 Humulene-8,9-epoxide

In contrast to the fascinating transannulation reactions associated with humulene-1,2- and humulene-4,5-epoxides, humulene-8,9-epoxide (161) undergoes transannulation and rearrangement in the presence of SnCl₄ leading only to the bicyclic alcohol (162), related to the rearrangement product (115) of humulene itself (Scheme 20).¹³⁹



Scheme 20

Formic acid catalyzed transannulation of the 8,9-epoxide (163), derived from natural zerumbone, has been shown to lead to a mixture of the bicyclo[6.3.0]undecane (164) and the hydroazulene (165; equation 72).¹⁴⁰



1.10.7 OTHER CYCLOALKENES

The transannular cyclizations described in Sections 1.10.1-1.10.6 are unique to medium-sized ring systems. Although transannular electrophilic processes are known with smaller rings, *i.e.* 3- to 7-rings, many of these processes can best be reviewed in terms of nonclassical carbonium ion phenomena. Some selected examples are collected in equations (73-77).¹⁴¹⁻¹⁴⁵



Although transannular electrophilic cyclizations involving large rings have not been studied in any great detail, the available information suggests that these reactions occur only with difficulty. Thus, the cembranoid diterpene sarcophine (166) produces only the oxepine derivative (167; equation 78) on transannulation in acid,¹⁴⁶ and all attempts to synthesize the taxane ring system (169) via transannular cyclization of the 12-ring precursor (168; equation 79) have so far failed.¹⁴⁷ Interestingly however, when the 14-membered diterpene ovatodiolide (170) is treated with hot methanolic hydrochloric acid¹⁴⁸ or with *p*-toluenesulfonic acid (PTSA)¹⁴⁹ it produces the product (171; equation 80). In addition, the tobacco constituent (172) is reported to produce the tetracyclic ether (173; equation 81) upon heating with PTSA in benzene.¹⁵⁰











(167)

(79)

(80)



(170)



CO₂H



1.10.8 REFERENCES

- 1. A. C. Cope, S. W. Fenton and C. F. Spencer, J. Am. Chem. Soc., 1952, 74, 5884.
- 2. V. Prelog and K. Schenker, Helv. Chim. Acta, 1952, 35, 2044.
- (a) V. Prelog and J. G. Traynham in 'Molecular Rearrangement', ed. P. de Mayo, Wiley, New York, 1963, p. 593;
 (b) A. C. Cope, M. M. Martin and M. A. McKervey, *Q. Rev., Chem. Soc.*, 1966, 20, 119;
 (c) J. K. Sutherland, *Tetrahedron*, 1974, 30, 1651;
 (d) G. Haufe and M. Mühlstädt, *Z. Chem.*, 1979, 19, 170;
 (e) G. Haufe and G. Mann, 'Chemistry of Alicyclic Compounds: Structure and Chemical Transformations', Elsevier, Amsterdam, 1989.
- (a) H. C. Chiang, M. C. Woods, Y. Nakadaira and K. Nakanishi, J. Chem. Soc., Chem. Commun., 1967, 1201;
 (b) E. N. Marvell, J. Seubert, D. Sturmer and W. Federici, J. Org. Chem., 1970, 35, 396; (c) T. Kobayashi, M.

Kurono, H. Sato and K. Nakanishi, J. Am. Chem. Soc., 1972, 94, 2863; (d) R. J. Hargrove and P. J. Stang, Tetrahedron, 1976, 32, 37; (e) A. P. Davis, G. J. Hughes, P. R. Lowndes, C. M. Robbins, E. J. Thomas and G. H. Whitham, J. Chem. Soc., Perkin Trans. 1, 1981, 1934; (f) G. Nagendrappa, Tetrahedron, 1982, 38, 2429; (g) M. Luyten and R. Keese, Helv. Chim. Acta, 1984, 67, 2242; (h) B. S. Bandodakar and G. Nagendrappa, Tetrahedron Lett., 1989, 30, 7461.

- (a) A. C. Cope, A. H. Keough, P. E. Peterson, H. E. Simmons and G. W. Wood, J. Am. Chem. Soc., 1957, 79, 3900;
 (b) A. C. Cope and B. C. Anderson, J. Am. Chem. Soc., 1957, 79, 3892;
 (c) A. C. Cope and A. Fournier, J. Am. Chem. Soc., 1957, 74, 3896;
 (d) A. C. Cope and R. W. Gleason, J. Am. Chem. Soc., 1962, 84, 1928.
- 6. G. Haufe, M. Mühlstädt and J. Graefe, Monatsh. Chem., 1977, 108, 199.
- (a) E. Wenkert, J. Am. Chem. Soc., 1962, 84, 98; (b) S. Swaminathan, R. K. Natarajan, S. Ramachandran and S. K. Sankarappa, J. Org. Chem., 1966, 31, 656; (c) J. P. Kutney, E. Piers and R. T. Brown, J. Am. Chem. Soc., 1970, 92, 1700; (d) J. P. Kutney, R. T. Brown, E. Piers and J. R. Hadfield, J. Am. Chem. Soc., 1970, 92, 1708; (e) J. K. Crandall, R. D. Huntington and G. L. Brunner, J. Org. Chem., 1972, 37, 2911; (f) R. Balasubramanian, K. Rajagopalan, K. G. Srinivasan and S. Swaminathan, Tetrahedron Lett., 1977, 2607; (g) E. W. Warnhoff and V. Srinivasan, Can. J. Chem., 1977, 55, 1629; (h) R. Balasubramanian, S. Chandrasekhar, K. Rajagopalan and S. Swaminathan, Tetrahedron, 1978, 34, 1561; (i) K. V. Geetha, K. Rajagopalan and S. Swaminathan, Tetrahedron, 1978, 34, 2201; (j) K. V. Geetha, K. Rajagopalan and S. Swaminathan, Indian J. Chem., Sect. B, 1978, 16, 322; (k) N. Raju, K. Rajagopalan, S. Swaminathan and J. N. Shoolery, Tetrahedron Lett., 1980, 21, 1577; (l) G. Buchbauer, J. Gabmeier, E. Haslinger, W. Robien and H. Steindl, Helv. Chim. Acta, 1985, 68, 231; (m) M. N. Deshpande, S. Wehrli, M. Jawdosiuk, J. T. Guy, Jr., D. W. Bennett, J. M. Cook, M. R. Depp and U. Weiss, J. Org. Chem., 1986, 51, 2436; (n) E. Carceller, A. Moyano, F. Serratosa, M. Font-Altaba and X. Solans, J. Chem. Soc., Perkin Trans. 1, 1986, 2055.
- Moyano, F. Serratosa, M. Font-Altaba and X. Solans, J. Chem. Soc., Perkin Trans. 1, 1986, 2055.
 (a) S. S. Hecht, J. Chem. Educ., 1971, 48, 340; (b) P. W. Concannon and J. Ciabattoni, J. Am. Chem. Soc., 1973, 95, 3284; (c) J. Casanova, B. Waegell, G. Koukoua and V. Toure, J. Org. Chem., 1979, 44, 3976; (d) J. A. Marshall, J. C. Peterson and L. Lebioda, J. Am. Chem. Soc., 1983, 105, 6515; (e) I. D. Reingold and J. Drake, Tetrahedron Lett., 1989, 30, 1921.
- (a) R. Dowbenko, J. Am. Chem. Soc., 1964, 29, 946; (b) T. W. Sam and J. K. Sutherland, J. Chem. Soc., Chem. Commun., 1971, 970; (c) P. Y. Johnson and N. A. Priest, J. Am. Chem. Soc., 1974, 96, 5618; (d) P. Y. Johnson and M. A. Priest, Tetrahedron Lett., 1975, 3133; (e) M. E. Kuehne and R. E. Damon, J. Org. Chem., 1977, 42, 1825; (f) E. Malamidou-Xenikaki and N. Alexandrou, Tetrahedron Lett., 1982, 23, 3957; (g) J. D. Winkler and V. Sridar, J. Am. Chem. Soc., 1986, 108, 1708; (h) Y. Tobe, T. Iseki, K. Kakiuchi and Y. Odaira, Tetrahedron Lett., 1984, 25, 3895; (i) T. Sato, K. Nishiyama, A. Morita and Y. Iitaki, Bull. Chem. Soc. Jpn., 1985, 58, 2366; (j) A. J. Bloodworth, D. Crich and T. Melvin, J. Chem. Soc., Chem. Commun., 1987, 786; (k) F. MacCorquodale and J. C. Walton, J. Chem. Soc., Chem. Commun., 1987, 1456.
- For Diels-Ålder reactions see (a) K. Baettig, C. Dallaire, R. Pitteloud and P. Deslongchamps, Tetrahedron Lett., 1987, 28, 5259; (c) G. Bérubé and P. Deslongchamps, Tetrahedron Lett., 1987, 28, 5253; (c) G. Bérubé and P. Deslongchamps, Tetrahedron Lett., 1987, 28, 5253; (c) G. Bérubé and P. Deslongchamps, Tetrahedron Lett., 1987, 28, 5255; (d) T. Takahashi, K. Shimizu, T. Doi and J. Tsuji, J. Am. Chem. Soc., 1988, 110, 2674; (e) S. Lamothe, A. Ndibwami and P. Deslongchamps, Tetrahedron Lett., 1988, 29, 1639; (f) S. Lamothe, A. Ndibwami and P. Deslongchamps, Tetrahedron Lett., 1988, 29, 1641; (g) A. Marinier and P. Deslongchamps, Tetrahedron Lett., 1988, 29, 6215; (h) A. Marinier, K. Baettig, C. Dallaire, R. Pitteloud and P. Deslongchamps, Can. J. Chem., 1989, 67, 1609; for ene reactions see (i) P. A. Wender and J. C. Hubbs, J. Org. Chem., 1980, 45, 365; (j) P. A. Wender and L. J. Letendre, J. Org. Chem., 1980, 45, 367; (k) P. A. Wender and S. L. Eck, Tetrahedron Lett., 1982, 23, 1871; for Wittig rearrangements see (l) J. A. Marshall, in 'Comprehensive Organic Synthesis', ed. B. M. Trost, Pergamon Press, Oxford, 1991, vol. 3, chap. 3.11; (l) J. A. Marshall and J. Lebreton, J. Am. Chem. Soc., 1988, 110, 2925; (m) D. N. Sarma, N. C. Barua and R. P. Sharma, Chem. Ind. (London), 1985, 167 (Nazarov); (n) W. T. Borden, I. L. Reich, L. A. Sharpe, R. B. Weinberg and H. J. Reich, J. Org. Chem., 1975, 40, 2438 (photochemical [2 + 2]); (o) G. L. Lange and M. Bosch, Tetrahedron Lett., 1971, 13, 315; (Paterno Büchi); (p) T. Tsuji and S. Nishida, Tetrahedron Lett., 1981, 22, 867 (biradical cycloaddition); (r) R. Singh and G. Just, Tetrahedron Lett., 1990, 31, 185 (electrocyclization); for selected metal carbonyl mediated reactions see (s) A. Brookes, J. Howard, S. A. R. Knox, F. G. A. Stone and P. Woodward, J. Chem. Soc., Chem. Commun., 1973, 587; (t) L. A. Paquette, S. V. Ley, M. J. Broadhurst, D. Truesdell, J. Fayos and J. C. Clardy, Tetrahedron Lett., 1973, 2943; (u) R. B. King and C. A. Harmon, J. Organomet.
- (a) N. J. Leonard and T. Sato, J. Org. Chem, 1969, 34, 1066; (b) P. W. Jeffs and J. D. Scharver, J. Am. Chem. Soc., 1976, 98, 4301; (c) S. R. Wilson and R. A. Sawicki, J. Chem. Soc., Chem. Commun., 1977, 431; (d) M. E. Christy, P. S. Anderson, S. F. Britcher, C. D. Colton, B. E. Evans, D. C. Remy and E. L. Engelhardt, J. Org. Chem., 1979, 44, 3117; (e) Y. L. Chow, K. S. Pillay and H. Richard, Can. J. Chem., 1979, 57, 2923; (f) T. Nomoto and H. Takayama, J. Chem. Soc., Chem. Commun., 1982, 1113; (g) S. Karady, E. G. Corley, N. L. Abramson and L. M. Weinstock, Tetrahedron Lett., 1989, 30, 2191.
- (a) T. Komeno, M. Kishi and K. Nabeyama, Tetrahedron, 1971, 27, 1503; (b) G. Haufe and M. Mühlstädt, J. Prakt. Chem., 1981, 323, 89.
- (a) S. Moon and L. Haynes, J. Org. Chem., 1966, 31, 3067; (b) A. C. Cope, M. A. McKervey and N. M. Weinshenker, J. Am. Chem. Soc., 1967, 89, 2932; (c) M. A. Eakin, J. Martin and W. Parker, J. Chem. Soc., Chem. Commun., 1967, 955; (d) P. Brun, M. Pally and B. Waegell, Tetrahedron Lett., 1970, 331; (e) M. L. Mihailovic, V. Andrejevic, M. Jakovljevic, D. Jeremic, A. Stojiljkovic and R. E. Partch, J. Chem. Soc., Chem. Commun., 1970, 854; (f) T. Koga and S. Kawashima, Chem. Pharm. Bull., 1972, 20, 21; (g) P. Brun, M. Pally and B. Waegell, Tetrahedron Lett., 1970, 331; (h) P. Brun and B. Waegell, Soc. Chim. Fr., 1972, 1825; (i) D. W. Balogh and L. A. Paquette, J. Org. Chem., 1980, 45, 3038; (j) M. Ishitsuka, T. Kusumi and H. Kakisawa, Chem. Lett., 1982, 23, 3179; (k) M. Ishitsuka, T. Kusumi and H. Kakisawa, Chem. Lett., 1982, 24, 3179; (k) M. Ishitsuka, T. Kusumi and H. Kakisawa, Chem. Lett., 1982, 24, 3179; (k) M. Ishitsuka, T. Kusumi and H. Kakisawa, Chem. Lett., 1982, 24, 3179; (k) M. Ishitsuka, T. Kusumi and H. Kakisawa, Chem. Lett., 1982, 24, 3179; (k) M. Ishitsuka, T. Kusumi and H. Kakisawa, Chem. Lett., 1982, 24, 3179; (k) M. Ishitsuka, T. Kusumi and H. Kakisawa, Chem. Lett., 1982, 24, 3179; (k) M. Ishitsuka, T. Kusumi and H. Kakisawa, Chem. Lett., 1982, 24, 3179; (k) M. Ishitsuka, T. Kusumi and H. Kakisawa, Chem. Lett., 1982, 24, 3179; (k) M. Ishitsuka, T. Kusumi and H. Kakisawa, Chem. Lett., 1982, 24, 3179; (k) M. Ishitsuka, T. Kusumi and H. Kakisawa, Chem. Lett., 1982, 24, 3179; (k) M. Ishitsuka, T. Kusumi and H. Kakisawa, Chem. Lett., 1982, 24, 3179; (k) M. Ishitsuka, T. Kusumi and H. Kakisawa, Chem. Lett., 1982, 24, 3179; (k) M. Ishitsuka, T. Kusumi and H. Kakisawa, Chem. Lett., 1982, 24, 3179; (k) M. Ishitsuka, T. Kusumi and H. Kakisawa, Chem. Lett., 1982, 24, 3179; (k) M. Ishitsuka, T. Kusumi and H. Kakisawa, Chem. Lett., 1982, 24, 3179; (k) M. Ishitsuka, T. Kusumi and H. Kakisawa, Chem. Lett., 1982, 24, 3179; (k) M. Ishitsuka, T. Kusumi and H. Kakisawa, Chem. Lett., 1982, 24, 3179; (k) M.

1983, 999; (1) M. K. Baynham and J. R. Hanson, J. Chem. Soc., Perkin Trans. 1, 1984, 2859; (m) G. Haufe, Tetrahedron Lett., 1984, 25, 4365; (n) D. Czarkie, S. Carmely, A. Groweiss and Y. Kashman, Tetrahedron, 1985, 41, 1049; (o) M. S. Jelenick and T. A. Bryson, J. Org. Chem., 1986, 51, 802; (p) M. F. Schlecht and H. Kim, Tetrahedron Lett., 1986, 27, 4889; (q) H. Kim and M. F. Schlecht, Tetrahedron Lett., 1987, 28, 5229

- 14. 'Terpenoids and Steroids', Specialist Periodical Reports, Royal Society of Chemistry, London, 1971-1982, vols. 1–11.
- 15. 'Natural Product Reports', Royal Society of Chemistry, London, 1984-1990, vols. 1-7.
- 16. (a) A. C. Cope and P. E. Peterson, J. Am. Chem. Soc., 1959, 81, 1643; (b) A. C. Cope, J. M. Grisar and P. E. Peterson, J. Am. Chem. Soc., 1960, 82, 4299; (c) A. C. Cope, S. Moon and P. E. Peterson, J. Am. Chem. Soc., 1959, **81**, 1650; (d) S. Moon and C. R. Ganz, J. Org. Chem., 1970, **35**, 1241. A. C. Cope, D. L. Nealy, P. Scheiner and G. W. Wood, J. Am. Chem. Soc., 1965, **87**, 3130.
- 17.
- 18. M. Hanack and W. Kaiser, Angew. Chem., 1964, 76, 572.
- 19. K. B. Wiberg and J. G. Pfeiffer, J. Am. Chem. Soc., 1970, 92, 553.
- 20. J. R. Norell, J. Org. Chem., 1972, 37, 1971.
- Y. Souma, H. Sano and J. Iyoda, J. Org. Chem., 1973, 38, 2016. 21.
- D. R. Kell and F. J. McQuillin, J. Chem. Soc., Perkin Trans. 1, 1972, 2096. 22.
- 23. D. Wittenberg, Ger. Pat. 1 (1964) 167 824 (Chem. Abstr., 1964, 61, 1776e).
- (a) I. Tabushi, K. Fujita and R. Oda, J. Org. Chem., 1970, 35, 2376; (b) I. Tabushi, K. Fujita and R. Oda, Tetrahedron Lett., 1967, 3755; (c) I. Tabushi, K. Fujita and R. Oda, Tetrahedron Lett., 1967, 3815.
- M. Julia and E. Colomer, An. Quim., 1971, 67, 199.
- 26. S. Uemura, S. Fukuzawa, A. Toshimitsu, M. Okano, H. Tezuka and S. Sawada, J. Org. Chem., 1983, 48, 270.
- 27. A. Toshimitsu, S. Uemura and M. Okano, J. Chem. Soc., Chem. Commun., 1982, 87.
- 28. G. Haufe, G. Alvernhe and A. Laurent, Tetrahedron Lett., 1986, 27, 4449.
- 29. (a) S. Uemura, A. Onoe and M. Okano, Bull. Chem. Soc. Jpn., 1977, 50, 1078; (b) S. Uemura, A. Onoe and M. Okano, J. Chem. Soc., Chem. Commun., 1975, 210.
- 30. Y. Yamada, A. Shibata, K. Iguchi and H. Sanjoh, Tetrahedron Lett., 1977, 2407.
- 31. N. A. Maly, H. Menapace and M. F. Farona, J. Catal., 1973, 29, 182.
- 32. (a) E. A. Shokoka, S. I. Khromov and B. A. Kazanskii, Neftekhimiya, 1961, 1, 353 (Chem. Abstr., 1962, 57, 14 959h); (b) T. S. Cantrell, J. Org. Chem., 1967, 32, 1669; (c) T. S. Cantrell and B. L. Strasser, J. Org. Chem., 1971, 36, 670; (d) P. M. Henry, M. Davies, G. Ferguson, S. Phillips and R. Restivo, J. Chem. Soc., Chem. Commun., 1974, 112; (e) H. Fritz, C. D. Weis and T. Winkler, Helv. Chim. Acta, 1975, 58, 1345.
- 33. L. A. Paquette, Top. Curr. Chem., 1984, 119, 1.
- 34. J. E. Baldwin, J. Chem. Soc., Chem. Commun., 1976, 734.
- 35. E. G. Scovell and J. K. Sutherland, J. Chem. Soc., Chem. Commun., 1977, 275.
- 36. (a) W. Baker, R. Banks, D. R. Lyon and F. G. Mann, J. Chem. Soc., 1945, 27; (b) J. D. Roberts, J. Am. Chem. Soc., 1950, 72, 3300; (c) E. W. Randall and L. E. Sutton, J. Chem. Soc., 1958, 1266; (d) B. S. Green, M. Lahav and G. M. J. Schmidt, J. Chem. Soc. B, 1971, 1552; (e) R. K. MacKenzie, D. D. MacNicol, H. H. Mills, R. A. Raphael, F. B. Wilson and J. A. Zabkiewicz, J. Chem. Soc., Perkin Trans. 2, 1972, 1632; (f) D. Montecalvo, M. St-Jacques and R. Wasylishen, J. Am. Chem. Soc., 1973, 95, 2023; (g) F. A. L. Anet and L. Kozerski, J. Am. Chem. Soc., 1973, 95, 3407; (h) N. L. Allinger and J. T. Sprague, Tetrahedron, 1975, 31, 21; (i) N. L. Allinger, J. F. Viskocil, Jr., U. Burkert and Y. Yuh, *Tetrahedron*, 1976, **32**, 33; (j) O. Ermer, J. Am. Chem. Soc., 1976, **98**, 3964; (k) F. A. L. Anet and I. Yavari, J. Am. Chem. Soc., 1977, **99**, 6986; (l) H.-D. Martin, M. Kunze, H.-D. Beckhaus, R. Walsh and R. Gleiter, *Tetrahedron Lett.*, 1979, 3069; (m) F. A. L. Anet, N. R. Easton, Jr. and I. Yavari, Org. Magn. Reson., 1979, 12, 299.
- 37. J. K. Whitesell, R. S. Matthews and P. A. Solomon, Tetrahedron Lett., 1976, 1549.
- 38. G. Haufe, A. Wolf and K. Schulze, Tetrahedron, 1986, 42, 4719.
- 39. For additional support of this mechanism see K. Hayano, Y. Ohfune, H. Shirahama and T. Matsumoto, Helv. Chim. Acta, 1981, 64, 1347.
- 40. J. M. McIntosh, Can. J. Chem., 1972, 50, 2152.
- 41. (a) T. Sasaki, K. Kanematsu and A. Kondo, J. Org. Chem., 1974, 39, 2246; (b) T. Sasaki, K. Kanematsu and A. Kondo, J. Org. Chem., 1975, 40, 1642.
- 42. G. R. Underwood and B. Ramamoorthy, Tetrahedron Lett., 1970, 4125.
- 43. G. Klein and L. A. Paquette, Tetrahedron Lett., 1976, 2419.
- 44. (a) H. Stetter and J. Gärtner, Chem. Ber., 1966, 99, 925; (b) D. Helmlinger and G. Ourisson, Tetrahedron, 1969, 25, 4895; (c) N. C. Yang and J. Libman, J. Am. Chem. Soc., 1972, 94, 9228; 1973, 95, 4473; (d) T. Sasaki, K. Kanematsu and A. Kondo, J. Chem. Soc., Perkin Trans. 1, 1976, 2516.
- 45. E. Osawa, K. Aigami and Y. Inamoto, Tetrahedron, 1978, 34, 509.
- 46. (a) K. H. Baggaley, W. H. Evans, S. H. Graham, D. A. Jonas and D. H. Jones, Tetrahedron, 1968, 24, 3445; (b) R. Bishop, J. Chem. Soc., Perkin Trans. 1, 1974, 2364.
- 47. F. N. Stepanov, V. D. Sukhoverkhov, V. F. Baklan and A. G. Yurchenko, Zh. Org. Khim., 1970, 6, 884.
- 48. A. R. Allan and M. S. Baird, Tetrahedron Lett., 1976, 3831.
- 49. K. K. Joshi, R. H. B. Mais, F. Nyman, P. G. Owston and A. M. Wood, J. Chem. Soc. A, 1968, 318.
- 50. S. H. Graham and D. A. Jonas, J. Chem. Soc., Chem. Commun., 1968, 1091.
- 51. R. S. Matthews and J. K. Whitesell, J. Org. Chem., 1975, 40, 3312.
- 52. (a) A. M. Birch and G. Pattenden, Tetrahedron Lett., 1982, 23, 991; (b) A. M. Birch and G. Pattenden, J. Chem. Soc., Perkin Trans. 1, 1983, 1913.
- 53. (a) G. Pattenden and S. J. Teague, Tetrahedron Lett., 1984, 25, 3021; (b) G. Pattenden and S. J. Teague, Tetrahedron, 1987, 43, 5637. 54. (a) G. Mehta and K. S. Rao, J. Am. Chem. Soc., 1986, 108, 8015; (b) G. Mehta and K. S. Rao, Tetrahedron
- Lett., 1984, 25, 3481; (c) G. Mehta and K. S. Rao, J. Chem. Soc., Chem. Commun., 1985, 1464.
- 55. G. Mehta and K. S. Rao, J. Org. Chem., 1988, 53, 425.
- S. Chatterjee, J. Chem. Soc., Chem. Commun., 1979, 620; but see J. Cornforth, Tetrahedron Lett., 1980, 21, 56. 709.

- For some further discussion see (a) K. Tori, I. Horibe, K. Kuriyama, H. Tada and K. Takeda, J. Chem. Soc., Chem. Commun., 1971, 1393; (b) K. Tori, I. Horibe, M. Minato and K. Takeda, Tetrahedron Lett., 1971, 4355; (c) K. Tori, I. Horibe, Y. Tamura and H. Tada, J. Chem. Soc., Chem. Commun., 1973, 620; (d) G. Buemi, F. Zuccarello and G. Favini, J. Mol. Struct., 1974, 21, 41; (e) W. Messerotti, U. M. Pagnoni, R. Trave, R. Zanasi, G. D. Andreetti, G. Bocelli and P. Sgarabotto, J. Chem. Soc., Perkin Trans. 2, 1978, 217; (f) H. Shirahama, E. Osawa and T. Matsumoto, Tetrahedron Lett., 1979, 2245. 58. G. Nagendrappa and D. Devaprabhakara, Tetrahedron Lett., 1970, 4687.
- 59. G. Haufe, M. Mühlstädt and J. Graefe, Monatsh. Chem., 1977, 108, 1431.
- D. L. J. Clive, G. J. Chittattu and C. K. Wong, J. Chem. Soc., Chem. Commun., 1978, 441.
 (a) S. N. Moorthy, D. Devaprabhakara and K. G. Das, Tetrahedron Lett., 1975, 257; (b) Puttaraja, D. S. S. Gowda and W. L. Duax, J. Cryst. Spectrosc. Res., 1984, 14, 13; (c) D. Devaprabhakara, J. Indian Chem. Soc., 1974, 51, 913.
- 62. D. Duffin and J. K. Sutherland, J. Chem. Soc., Chem. Commun., 1970, 626.
- 63. D. Duffin and J. K. Sutherland, J. Chem. Soc., Chem. Commun., 1970, 627.
- 64. W. Heggie and J. K. Sutherland, J. Chem. Soc., Chem. Commun., 1974, 596.
- 65. W. Heggie and J. K. Sutherland, J. Chem. Soc., Chem. Commun., 1972, 957.
- 66. (a) For leading references see A. Nickon, Perfumery Essent. Oil Rec., 1954, 45, 149; (b) D. H. R. Barton and P. de Mayo, Q. Rev., Chem. Soc., 1957, 11, 189.
- (a) W. Parker, R. A. Raphael and J. S. Roberts, J. Chem. Soc. C, 1969, 2634; (b) T. F. W. McKillop, J. Martin, W. Parker, J. S. Roberts and J. R. Stevenson, J. Chem. Soc. C, 1971, 3375.
- 68. (a) G. Ferguson, D. M. Hawley, T. F. W. McKillop, J. Martin, W. Parker and P. Doyle, J. Chem. Soc., Chem. Commun., 1967, 1123; (b) R. I. Crane, C. R. Eck, W. Parker, A. B. Penrose, T. F. W. McKillop, D. M. Hawley and J. M. Robertson, J. Chem. Soc., Chem. Commun., 1972, 385; (c) D. Baines, C. Eck and W. Parker, Tetrahedron Lett., 1973, 3933; (d) A. V. Tkachev, Y. V. Gatilov, I. K. Korobeinicheva, Z. V. Dubovenko and V. A. Pentegova, Khim. Prir. Soedin., 1983, 164; (Chem. Abstr., 1983, 99, 105 534z).
- 69. M. Kodama, S. Yokoo, Y. Matsuki and S. Itô, Tetrahedron Lett., 1979, 1687.
- (a) R. V. H. Jones and M. D. Sutherland, J. Chem. Soc., Chem. Commun., 1968, 1229; (b) M. Kodama, Y. Matsuki and S. Itô, Tetrahedron Lett., 1976, 1121; (c) M. Kodama, S. Yokoo, H. Yamada and S. Itô, Tetrahedron Lett., 1978, 3121.
- 71. E. D. Brown, M. D. Solomon, J. K. Sutherland and A. Torre, J. Chem. Soc., Chem. Commun., 1967, 111.
- 72. E. D. Brown, T. W. Sam, J. K. Sutherland and A. Torre, J. Chem. Soc., Perkin Trans. 1, 1975, 2326.
- 73. E. Tsankova, I. Ognyanov and T. Norin, Tetrahedron, 1980, 36, 669.
- 74. (a) M. Niwa, A. Nishiyama, M. Iguchi and S. Yamamura, Bull. Chem. Soc. Jpn., 1975, 48, 2930; (b) M. Iguchi, M. Niwa and S. Yamamura, Tetrahedron Lett., 1973, 4367.
- 75. T. C. Jain and J. E. McCloskey, Tetrahedron Lett., 1971, 1415.
- 76. (a) T. C. Jain and J. E. McCloskey, Tetrahedron Lett., 1969, 2917; (b) T. C. Jain and J. E. McCloskey, Tetrahedron, 1975, 31, 2211.
- 77. R. W. Doskotch and F. S. El-Feraly, J. Org. Chem., 1970, 35, 1928. 78. (a) T. C. Jain and C. M. Banks, Can. J. Chem., 1980, 58, 447; (b) T. C. Jain, C. M. Banks and J. E. McCloskey, Experientia, 1969, 25, 906.
- 79. (a) H. Tada and K. Takeda, J. Chem. Soc., Chem. Commun., 1971, 1391; (b) R. W. Doskotch, C. D. Hufford and F. S. El-Feraly, J. Org. Chem., 1972, 37, 2740; (c) T. C. Jain and J. E. McCloskey, Tetrahedron Lett., 1972, 5139; (d) H. Tada and K. Takeda, Chem. Pharm. Bull., 1976, 24, 667; (e) T. C. Jain, C. M. Banks and J. E. McCloskey, Tetrahedron, 1979, 35, 885; (f) T. Takahashi, H. Nemoto, Y. Kanda, J. Tsuji, Y. Fukazawa, T. Okajima and Y. Fujise, Tetrahedron, 1987, 43, 5499.
- 80. C. B. Rose and S. K. Taylor, J. Heterocycl. Chem., 1979, 16, 705.
- 81. J. G. Traynham, G. R. Franzen, G. A. Knesel and D. J. Northington, Jr., J. Org. Chem., 1967, 32, 3285.
- 82. H. J. Günther, V. Jäger and P. S. Skell, Monatsh. Chem., 1977, 108, 2539.
- 83. (a) C. Ganter and B. Keller-Wojtkiewicz, Helv. Chim. Acta, 1971, 54, 183; (b) J. G. Traynham and H. H. Hsieh, J. Org. Chem., 1973, 38, 868; (c) see also ref. 59.
- W. Brenner, P. Heimbach, K.-J. Pioner and F. Thömel, Justus Liebigs Ann. Chem., 1973, 1882. 84.
- 85. J. R. Williams, J. F. Callahan and J. F. Blount, J. Org. Chem., 1981, 46, 2665.
- 86. M. Iguchi, M. Niwa and S. Yamamura, Tetrahedron Lett., 1973, 1687.
- 87. K. Nishimura, N. Shinoda and Y. Hirose, Tetrahedron Lett., 1969, 3097.
- 88. P. S. Wharton, R. A. Kretchmer, R. J. Kilian and T. Oda, J. Org. Chem., 1974, 39, 3755.
- 89. R. M. Gipson, H. W. Guin, S. H. Simonsen, C. G. Skinner and W. Shive, J. Am. Chem. Soc., 1966, 88, 5366.
- 90. A. G. González, A. Galindo, H. Mansilla and A. Gutiérrez, J. Chem. Soc., Perkin Trans. 1, 1982, 881.
- 91. A. Shani, Isr. J. Chem., 1975, 13, 35.
- 92. (a) R. W. Thies, P.-K. Hong, R. Buswell and J. L. Boop, J. Org. Chem., 1975, 40, 585; (b) R. W. Thies, P.-K. Hong and R. Buswell, J. Chem. Soc., Chem. Commun., 1972, 1091.
- 93. (a) A. C. Cope, R. J. Cotter and G. G. Roller, J. Am. Chem. Soc., 1955, 77, 3594; (b) H. L. Goering, H. H. Espy and W. D. Closson, J. Am. Chem. Soc., 1959, 81, 329.
- 94. H. L. Goering and R. F. Myers, J. Am. Chem. Soc., 1969, 91, 3386.
- 95. H. L. Goering and W. D. Closson, J. Am. Chem. Soc., 1961, 83, 3511
- 96. (a) J. A. Marshall and W. F. Huffman, J. Am. Chem. Soc., 1970, 92, 6358; (b) J. A. Marshall, Synthesis, 1972, 517; (c) J. A. Marshall, W. F. Huffman and J. A. Ruth, J. Am. Chem. Soc., 1972, 94, 4691.
- 97. A. G. González, A. Galindo and H. Mansilla, Tetrahedron, 1980, 36, 2015.
- 98. F. Bellesia, U. M. Pagnoni and R. Trave, J. Chem. Soc., Chem. Commun., 1976, 34.
- 99. K. Toma, T. Murae and T. Takahashi, Chem. Lett., 1982, 551.
- 100. M. Iguchi, M. Niwa and S. Yamamura, J. Chem. Soc., Chem. Commun., 1971, 974.
- 101. M. Kodama, K. Shimada, T. Takahashi, C. Kabuto and S. Itô, Tetrahedron Lett., 1981, 22, 4271.
- (a) D. F. Wiemer, J. Meinwald, G. D. Prestwich, B. A. Solheim and J. C. Clardy, J. Org. Chem., 1980, 45, 102. 191; (b) Y. Kashman and A. Groweiss, J. Org. Chem., 1980, 45, 3814.

- 103. (a) M. Hanack, C. E. Harding and J.-L. Derocque, Chem. Ber., 1972, 105, 421; (b) B. Rao and L. Weiler, Tetrahedron Lett., 1971, 927; (c) R. J. Balf, B. Rao and L. Weiler, Can. J. Chem., 1971, 49, 3135.
- 104. M. J. Chandy and M. Hanack, Tetrahedron Lett., 1977, 4377. 105. S. K. Taylor and C. B. Rose, J. Org. Chem., 1977, 42, 2175.
- 106. S. K. Taylor, G. L. Lilley, K. J. Lilley and P. A. McCoy, J. Org. Chem., 1981, 46, 2709.
- W. Dittmann and F. Stürzenhofecker, Justus Liebigs Ann. Chem., 1965, 688, 57.
 S. Sokoloff and R. Segal, Tetrahedron, 1977, 33, 2837.
- 109. R. W. Doskotch, S. L. Keely, Jr. and C. D. Hufford, J. Chem. Soc., Chem. Commun., 1972, 1137.
- 110. (a) E. Tsankova, Tetrahedron Lett., 1982, 23, 771; (b) E. Tsankova, S. Cerrini and D. Lamba, Tetrahedron, 1983, 39, 3397.
- 111. E. D. Brown, J. K. Sutherland and T. W. Sam, J. Chem. Soc., Perkin Trans. 1, 1975, 2332.
- S. K. Talapatra, A. Patra and B. Talapatra, J. Chem. Soc., Chem. Commun., 1970, 1534. 112.
- 113. M. Niwa, M. Iguchi and S. Yamamura, Tetrahedron Lett., 1975, 3661.
- 114. E. D. Brown, T. W. Sam and J. K. Sutherland, Tetrahedron Lett., 1969, 5025.
- 115. (a) T. R. Govindachari, B. S. Joshi and V. N. Kamat, Tetrahedron, 1965, 21, 1509; (b) H. Hikino, C. Konno, T. Nagashima, T. Kohama and T. Takemoto, Tetrahedron Lett., 1971, 337; (c) M. Iguchi, M. Niwa and S. Yamamura, J. Chem. Soc., Chem. Commun., 1972, 689; (d) T. W. Sam and J. K. Sutherland, J. Chem. Soc., Perkin Trans. 1, 1975, 2336; (e) M. Niwa, M. Iguchi and S. Yamamura, Bull. Chem. Soc. Jpn., 1976, 49, 3137; (f) H. Hikino, C. Konno, T. Nagashima, T. Kohama and T. Takemoto, Chem. Pharm. Bull., 1977, 25, 6; (g) M. Niwa, M. Iguchi and S. Yamamura, Tetrahedron Lett., 1978, 4043; (h) J. de Pascual Teresa, M. S. González, M. C. Caballero, T. Parra and I. S. Bellido, Tetrahedron Lett., 1987, 28, 821; (i) S. Kuwahara and K. Mori, Tetrahedron Lett., 1989, 30, 7447.
- 116. M. Niwa, M. Iguchi and S. Yamamura, Tetrahedron Lett., 1979, 4291; see also refs. 111 and 114.
- 117. J. M. Greenwood, M. D. Solomon, J. K. Sutherland and A. Torre, J. Chem. Soc. C, 1968, 3004.
- 118. D. Baines, J. Forrester and W. Parker, J. Chem. Soc., Perkin Trans. 1, 1974, 1598.
- 119. S. Misumi, Y. Ohfune, A. Furusaki, H. Shirahama and T. Matsumoto, Tetrahedron Lett., 1976, 2865.
- 120. S. Misumi, T. Ohtsuka, Y. Ohfune, K. Sugita, H. Shirahama and T. Matsumoto, Tetrahedron Lett., 1979, 31.
- 121. S. Misumi, T. Ohtsuka, H. Hashimoto, Y. Ohfune, H. Shirahama and T. Matsumoto, Tetrahedron Lett., 1979, 35.
- K. Sakai, T. Ohtsuka, S. Misumi, H. Shirahama and T. Matsumoto, Chem. Lett., 1981, 355. 122.
- 123. T. Ohtsuka, H. Shirahama and T. Matsumoto, Chem. Lett., 1984, 1923.
- 124. T. Ohtsuka, H. Shirahama and T. Matsumoto, Tetrahedron Lett., 1983, 24, 3851.
- 125. Y. Murata, T. Ohtsuka, H. Shirahama and T. Matsumoto, Tetrahedron Lett., 1981, 22, 4313.
- 126. S. Misumi, H. Matsushima, H. Shirahama and T. Matsumoto, Chem. Lett., 1982, 855; K. Hayano, Y. Ohfune, H. Shirahama and T. Matsumoto, Helv. Chim. Acta, 1981, 64, 1347.
- 127. N. P. Damodaran and Sukh Dev, Tetrahedron, 1968, 24, 4113, 4123; A. S. Gupta and Sukh Dev, Tetrahedron, 1971, **27**, 635.
- 128. M. A. McKervey and J. R. Wright, J. Chem. Soc., Chem. Commun., 1970, 117. 129. M. Namikawa, T. Murae and T. Takahashi, Bull. Chem. Soc. Jpn., 1978, 51, 3616.
- 130. H. Shirahama, S. Murata, T. Fujita, B. R. Chhabra, R. Noyori and T. Matsumoto, Bull. Chem. Soc. Jpn., 1982, 55, 2691.
- 131. H. Shirahama, E. Osawa and T. Matsumoto, J. Am. Chem. Soc., 1980, 102, 3208.
- 132. T. Fujita, T. Ohtsuka, H. Shirahama and T. Matsumoto, Tetrahedron Lett., 1982, 23, 4091.
- 133. A. Sattar, J. Forrester, M. Moir, J. S. Roberts and W. Parker, Tetrahedron Lett., 1976, 1403.
- 134. J. A. Mlotkiewicz, J. Murray-Rust, P. Murray-Rust, W. Parker, F. G. Riddell, J. S. Roberts and A. Sattar, Tetrahedron Lett., 1979, 3887.
- H. Shirahama, K. Hayano, Y. Kanemoto, S. Misumi, T. Ohtsuka, N. Hashiba, A. Furusaki, S. Murata, R. 135. Noyori and T. Matsumoto, Tetrahedron Lett., 1980, 21, 4835.
- 136. I. Bryson, J. A. Mlotkiewicz and J. S. Roberts, Tetrahedron Lett., 1979, 3891.
- 137. K. Hayasaka, T. Ohtsuka, H. Shirahama and T. Matsumoto, Tetrahedron Lett., 1985, 26, 873.
- 138. H. Shirahama, K. Hayano, G. S. Arora, T. Ohtsuka, Y. Murata and T. Matsumoto, Chem. Lett., 1982, 1417.
- 139. I. Bryson, J. S. Roberts and A. Sattar, Tetrahedron Lett., 1980, 21, 201.
- 140. H. W. D. Matthes, B. Luu and G. Ourisson, Tetrahedron, 1982, 38, 3129.
- 141. T. Ishiguro, Y. Kondo and T. Takemoto, Tetrahedron, 1975, 31, 305.
- 142. S. Winstein and L. de Vries, in 'American Chemical Society, 15th National Organic Chemistry Symposium', Rochester, NY, June 1957, Abstracts.
- 143. W. R. Nes and J. A. Steele, J. Org. Chem., 1957, 22, 1457.
- 144. S. N. Suryawanshi and U. R. Nayak, Indian J. Chem., Sect. B, 1979, 18, 190.
- 145. K. B. Wiberg and A. J. Ashe, III, J. Am. Chem. Soc., 1968, 90, 63.
- 146. D. Czarkie, S. Carmely, A. Groweiss and Y. Kashman, Tetrahedron, 1985, 41, 1049.
- C. B. Jackson and G. Pattenden, Tetrahedron Lett., 1985, 26, 3393; M. J. Begley, C. B. Jackson and G. 147. Pattenden,. Tetrahedron Lett., 1985, 26, 3397.
- 148. P. S. Manchand and J. F. Blount, J. Org. Chem., 1977, 42, 3824.
- 149. R. Toubiana, M.-J. Toubiana, A. T. McPhail, R. W. Miller and K. Tori, J. Chem. Soc., Perkin Trans. 2, 1976, 1881.
- 150. R. L. Rowland, A. Rodgman, J. N. Schumacher, D. L. Roberts, L. C. Cook and W. E. Walker, Jr., J. Org. Chem., 1964, 29, 16; W. G. Dauben, J. P. Hubbell, P. Oberhansli and W. E. Thiessen, J. Org. Chem., 1979, 44, 669.

2.1 Coupling Reactions Between *sp*³ Carbon Centers

DAVID C. BILLINGTON

Institut de Recherches Servier, Paris, France

2.1.1 INTRODUCTION AND SCOPE	413
2.1.2 THE WURTZ REACTION AND VARIANTS	414
2.1.2.1 The Classical Wurtz Reaction	414
2.1.2.2 Organomagnesium-derived Reagents	415
2.1.2.3 Organolithium-derived Reagents	419
2.1.2.4 Other Organometallic Reagents	420
2.1.2.5 Formation of Cyclic Systems	422
2.1.3 π-ALLYLNICKEL HALIDES	423
2.1.3.1 Preparation, Structure and Experimental Techniques	423
2.1.3.2 Reactivity Patterns	424
2.1.3.2.1 Chemoselectivity	424
2.1.3.2.2 Regioselectivity	426
2.1.3.2.3 Stereoselectivity	426
2.1.3.3 Coupling with sp ³ Centers	426
2.1.3.3.1 Alkyl halides	426
2.1.3.3.2 Allyl halides	428
2.1.3.3.3 Formation of cyclic systems	429
2.1.4 REFERENCES	432

2.1.1 INTRODUCTION AND SCOPE

The classical method of coupling two sp^3 -carbon centers is the Wurtz reaction (equation 1; M = Na). This chapter covers the generalized transformation shown in equation (1), where R and R¹ may be the same (dimerization) or different (cross-coupling) and where R and R¹ may represent a single molecule (intramolecular cyclization). In synthetic terms the most valuable type of coupling is the cross-coupling reaction. This reaction also presents the most problems, as the possibility of halogen-metal exchange and disproportionation of the reagents may lead to homo-coupled products (R—R and R¹—R¹ in equation 1), in addition to the desired cross-coupled product. As dimerization reactions are in general of less synthetic value, this treatment of the Wurtz-type coupling concentrates on cross-coupling reactions for each reagent considered, with less emphasis on dimerization reactions. The uses of coupling reactions for the synthesis of cyclic systems are treated separately in Section 2.1.2.5.

$$\mathbf{R} - \mathbf{X} + \mathbf{R}^{1} - \mathbf{X} \xrightarrow{\mathbf{M}} \mathbf{R} - \mathbf{R} + \mathbf{R}^{1} - \mathbf{R}^{1} + \mathbf{R} - \mathbf{R}^{1}$$
(1)

 $R = sp^3$ center, X = halogen, M = metal

Severe limitations on the usefulness of the classical Wurtz reaction in the production of cross-coupled products have led to the development of many more generally useful variants. In particular, the use of copper catalysis and of stoichiometric organocuprate species have proved very valuable. The reactions of π -allylnickel halides with sp^3 halides is also represented by equation (1), and the uses of these reagents are treated separately. In order to provide a balanced view of the value of π -allylnickel halides, some additional reactions with centers other than sp^3 are described.

2.1.2 THE WURTZ REACTION AND VARIANTS

2.1.2.1 The Classical Wurtz Reaction

The Wurtz reaction involves the coupling of alkyl halides using sodium metal according to equation (1). The reaction is severely limited in scope as demonstrated by the following observations. (i) In the synthesis of noncyclic systems (dimerization or cross-coupling) reasonable yields require the use of primary alkyl halides, with iodides giving the best results. Secondary halides give very poor results. (ii) Cross-coupling reactions give approximately statistical ratios of cross-coupled:dimeric products, due to the high reactivity of the organosodium intermediates involved. Thus, in the synthesis of 3-methyluntria-contane (4; equation 2)¹ coupling of the iodide (1) with 1-iodooctadecane (2) under standard Wurtz conditions gave a mixture of (3) from dimerization of (1), (4) from the desired cross-coupling, and (5) from dimerization of (2), in the ratio 21:50:29. The classical reaction is however of value in the preparation of cyclic systems, notably [2.2]phanes and bicyclobutanes and in the preparation of symmetric dimers.



ratio of products (3):(4):(5) = 21:50:29

The synthesis of [2.2]phanes has been reviewed.^{2.3} The Wurtz reaction provides one of the most successful approaches to this class of compounds, by ring-closure reactions. The reaction may proceed either intra- or inter-molecularly (as shown in equation 3), with intramolecular couplings giving better yields. The most efficient procedure⁴ involves the reaction of the halo compounds with sodium metal in THF at -78 °C in the presence of catalytic quantities of tetraphenylethylene (TPE). Under these conditions the sodium metal dissolves as the disodium species, allowing a homogeneous reaction to take place, leading to better yields of coupled products. The modest yields of these products are due to competing oligomerization/cyclization reactions. This is demonstrated by the product distribution obtained on treatment of the dibromide (6) with sodium/TPE in THF at -80 °C (equation 4).² A homologous series of products is obtained under these conditions, the approximate yields of the metacyclophanes being 61%, 15%, 3.5%, 9%, 7%, 1%, 1%, 1%, 0.5%, for n = 2, 3, 4, 5, 6, 7, 8, 9, 10, respectively. Organolithium species also give acceptable yields of [2.2]phanes, from the treatment of suitable dibromides with phenyllithium.

The classical Wurtz reaction is also of value in the preparation of small cyclic systems such as bicyclo[1.1.0]butane and other cyclopropane derivatives. Treatment of the 1,3-dihalide (7) with sodium metal in refluxing dioxane,⁵ for example, gives a high yield (78–90%) of bicyclobutane (8; equation 5). Although sodium may be used for the synthesis of other small rings, *e.g.* cyclobutane from dibromobutane, in general other metals (particularly zinc and potassium) give higher yields (equation 6).



2.1.2.2 Organomagnesium-derived Reagents

Direct coupling of alkyl halides via conversion of one reactant to the Grignard reagent has limited synthetic utility. Simple Grignard reagents couple in rather low yields with tertiary alkyl halides,⁶ improved yields being obtained by running the coupling reaction in a nonpolar solvent such as hexane or heptane. This method gives reasonable results for the preparation of highly branched hydrocarbons such as neopentane and hexamethylethane.⁷ Examples of the use of this reaction are given in Scheme 1.⁸ Grignard reagents derived from activated halides such as benzyl or related halides couple readily to give alkylbenzenes. *n*-Pentylbenzene and *n*-propylbenzene for example, may both be prepared in high yield by this procedure.⁹ Intramolecular uncatalyzed Grignard coupling reactions are of value in the preparation of small ring cyclic systems (see Section 2.1.2.5).

In general, the cross-coupling of Grignard reagents with alkyl halides becomes much more synthetically useful in the presence of catalytic quantities of other metals, particularly copper species. Treatment of Grignard reagents with simple copper salts such as CuBr or CuI, or more complex species such as Li₂CuCl₄, leads to coupling with alkyl halides *via* transient organocopper species (equation 7). A very generalized order of reactivity of electrophiles toward organocopper species is as follows: acid chlorides > aldehydes > tosylates/epoxides > iodides > bromides > chlorides > ketones > esters > nitriles >> alkenes. In terms of coupling of *sp*³ carbon centers, the approximate order of reactivity of alkyl halides is: allylic/benzylic > primary > secondary >> tertiary. In general, primary alkyl halides have been most widely used as sources of the Grignard reagent. Following the pioneering studies of Tamura and Kochi,¹⁰



widespread use has been made of the THF-soluble dilithium tetrachlorocuprate as a catalyst for the coupling of alkyl halides via their Grignard reagents.

$$R^{1}-X \xrightarrow{Mg} R^{1}-MgX + R^{2}-X \xrightarrow{Cu^{1}} R^{1}-R^{2}$$
(7)
X = halogen; Cu¹ = e.g. CuI or Li₂CuCl₄; R¹ and R² = alkyl sp³ carbon

Under suitable conditions, the yields of homodimeric products $(R^1-R^1 \text{ and } R^2-R^2)$ are negligibly small and high yields of cross-coupled products are obtained (equation 7). These reactions are normally performed in THF, at low temperatures (-20 to 0 °C) and require only catalytic quantities of the readily prepared copper species. The reaction characteristics may be summarized by the following general observations.¹¹⁻²² (i) Coupling proceeds very readily with primary halides; iodides and bromides giving good yields and chlorides giving lower yields. (ii) Secondary iodides also give good yields; secondary bromides and chlorides do not react. (iii) Activated halides, such as allylic or benzylic bromides or chlorides, give good yields for both primary and secondary substrates. (iv) Tertiary halides generally give very poor results, with disproportionation of the reagents predominating. (v) The degree of substitution of the Grignard reagent is not very important in determining reactivity. From the examples in equations (8)–(18) it can be seen that high yields of the desired cross-coupled products are normally obtained





Coupling Reactions

and that a number of functional groups are inert in the reaction. Functionality that is tolerated includes alkynes, ethers, alcohols protected as THP or TBDMS ethers, epoxides, furans and unprotected alcohols and carboxylic acids, as their chloromagnesium salts. These observations have allowed the use of this catalyzed Grignard coupling in the preparation of a range of natural products, intermediates towards some of which are shown in equations (8)–(18). Other copper species used for the catalysis of Grignard couplings include copper(I) chloride, bromide and iodide²³ and some recent examples are given in equations (17) and (18).



The dimerization of sp^3 centers via conversion of the alkyl halide to a Grignard reagent, followed by homo-coupling is promoted by a number of metals. Treatment of a Grignard reagent with a copper(I) species such as tetrakis[iodo(tri-*n*-butylphosphine)copper(I)],²⁴ followed by oxidation of the resulting ate complex with either oxygen or copper(II) salts, lead to dimerization (equation 19). 'Soluble silver' in THF may be prepared from silver nitrate and ethylmagnesium bromide in THF. The silver reagent is stable at greater than molar concentrations in THF for prolonged periods at 0 °C. Grignard reagents undergo efficient homo-coupling at room temperature in the presence of this silver catalyst.¹⁰ Crosscouplings give approximately equimolar quantities of the two possible homo-coupled products together with the desired cross-coupled product (*cf.* equation 1). Disproportionation of the reagent becomes a major pathway for secondary and tertiary halides.^{10,25,26} The special case of intramolecular homo-coupling via this method is treated in Section 2.1.2.5.



The *in situ* synthesis of organoboranes *via* reaction of alkyl halides with magnesium in the presence of diborane can also be used to prepare coupled products (equations 20 and 21).²⁷ Oxidation of the reaction mixture with alkaline silver nitrate leads to good yields of dimeric products. The reaction is successful for primary and secondary halides. A related reaction is the coupling of secondary alkyl halides in the presence of catalytic quantities of thallium salts.²⁸ This procedure fails for primary alkyl halides and gives modest yields for secondary alkyl halides (equation 22).

2.1.2.3 Organolithium-derived Reagents

The noncatalyzed coupling between organolithium reagents and organic halides leads to mixtures of hydrocarbons similar to the classical Wurtz reaction. The strongly basic and weakly nucleophilic nature of simple organolithium species leads to products derived from halogen-metal exchange, α -metalation and β -elimination, in addition to the desired coupling reactions. These factors mean that, in general, simple organolithium reagents are not useful intermediates for Wurtz-type couplings. Exceptions to this generalization include: the reaction of dihalides with *t*-butyllithium, to give cyclic systems (see Section 2.1.2.5);^{29,30} the coupling of *t*-butyllithium with benzylic halides, giving highly branched systems (equation 23);³¹ and the coupling of charge-delocalized alkyllithiums (*e.g.* benzyl- or allyl-lithiums) with organic halides (equation 24).^{32,33} Several reviews concerning halogen-metal exchange reactions and the coupling behavior of organolithiums are available.³⁴⁻³⁶



Reaction of an organolithium species with a copper(I) species at low temperature generates an organocopper reagent, which may have one of several different constitutions depending on the stoichiometry, exact reaction conditions, additives present, *etc.* Many types of organocopper reagent have proven very valuable in organic synthesis, and the use of these reagents is described in detail elsewhere in this work. In terms of the coupling of sp^3 carbon centers, the following general observations may be made about lithium organocopper species. (i) For catalytic reactions involving copper(I) intermediates, in general, Grignard reagents are preferred over organolithium species due to ease of preparation, and higher yields obtained in coupling reactions (see Section 2.1.2.2). (ii) For stoichiometric copper reagents, R_2CuLi reagents are more reactive, and give better results in coupling reactions than simple RCu species. (iii) A generalized order of reactivity towards lithium diorganocopper reagents is acid chlorides > aldehydes > tosylates/epoxides > iodides > bromides > chlorides > ketones > esters > nitriles >> alkenes. (iv) The relative reactivity of alkyl halides towards lithium diorganocopper reagents is allylic/benzylic > primary > secondary >> tertiary. (v) The kinetics of coupling are roughly first order in both cuprate reagent and alkyl halide. (vi) The cuprate reagent may be derived from a primary, secondary or tertiary halide.

Many reviews dealing with the chemistry of organocopper species have appeared, $^{37-42}$ and a comprehensive listing of organocopper substitution reactions reported prior to 1975 together with a detailed discussion of the factors influencing the reactions is available.³⁸ Many natural product syntheses have used cuprate couplings to prepare key intermediates and some representative examples of both stoichiometric and catalytic reactions are given in equations (25)–(29).^{43–47} For a general introduction to the field of organocopper chemistry the reader is recommended to consult the excellent book by Posner.⁴² Most organocopper reagents are extremely sensitive to heat and to oxidants, decomposing to the dimeric products of homo-coupling. Many organocopper species have been dimerized by heating or simple oxidation with oxygen.³⁸ Little use has been made synthetically of this method of coupling *sp*³ carbon centers.



Coupling Reactions



2.1.2.4 Other Organometallic Reagents

Many other metals, in addition to magnesium, lithium and copper, promote Wurtz-type couplings. In most cases cross-couplings are not possible and the methods are only of value in producing homocoupled dimers. Reagents which do permit cross-coupling reactions are normally very limited in either the type of halide used for preparation of the metal derivative, or the halide to be cross-coupled, or both.

Organozinc species show some valuable facets for cross-coupling reactions. Primary iodo esters may be converted into organozinc species by direct treatment with a zinc-copper couple.⁴⁸ In the presence of catalytic copper cyanide these organozinc reagents couple efficiently with allylic halides and tosylates, unfortunately with low regioselectivity (equation 30).⁴⁹ Perfluoroalkyl-magnesium and -lithium halide species decompose readily into perfluoroalkenes and metal halides and are of little value for the introduction of perfluoroalkene substituents. In contrast, perfluoroalkylzinc species are much more stable and in the presence of ultrasound and a palladium catalyst, couple with allylic halides showing high δ -regioselectivity (equation 31).⁵⁰



Organozinc reagents prepared from allylic bromides couple regioselectively in a δ -selective manner with *N*-chloromethyl-*N*-methylformamide, to give unsaturated tertiary formamides (equation 32).⁵¹ Treatment of these formamides with butyllithium followed by hydrolysis removes the formyl group, giving secondary amines. In a total synthesis of retronecic acid, coupling of the allylic bromo ester (11) with the secondary chloride (12) was the only convenient way of obtaining the desired cross-coupled product

(13), albeit in low yield (equation 33).⁵² Cross-coupling of allylic with benzylic halides may be achieved by oxidation of a mixture of the halides and triethylborane in THF with oxygen.⁵³



Iodofluoroacetate copper reagents can be prepared directly from the halide using copper powder in DMSO and the reagents couple with a range of electrophilic centers, including primary iodides and allylic bromides (equation 34).⁵⁴ Benzylic halides couple with bromoacetonitrile in the presence of metallic nickel in refluxing glyme (85 °C).⁵⁵ Organometallic species derived from methyllithium and manganese, cobalt or iron are all less suitable for cross-coupling reactions than the analogous species (R₂CuLi) derived from methyllithium and copper salts.⁵⁶



Methyltitanium(IV) chlorides have been used for the methylation of tertiary alkyl halides, with good results. The reagents are prepared by treatment of dimethylzinc species (available from methyl iodide and zinc metal) with TiCl₄ and similar results are obtained by using dimethylzinc in the presence of catalytic quantities of TiCl₄.^{57,58} The methylations are chemoselective, with esters and alkenes remaining unaffected and coupling only occurs at tertiary centers, primary and secondary alkyl halides being inert (equations 35 and 36). The method also works well in systems prone to skeletal rearrangements.



Benzylic and allylic halides dimerize (homo-couple) readily in the presence of a number of metal species. Synthetically useful yields of homo-coupled products have been obtained using chlorotris(triphenylphosphine)cobalt(I),⁵⁹ nickel(0) complexes generated *in situ*,⁶⁰ Te²⁻ species,⁶¹ VCl₃/LAH,⁶² CrCl₃/LAH⁶³ and TiCl₃ or TiCl₄/LAH.⁶⁴ Representative examples are given in equations (37) and (38). Treatment of δ -bromocrotonates with zinc(0) in DMSO leads to dimerization.⁶⁵



2.1.2.5 Formation of Cyclic Systems

The classical Wurtz-type coupling leads to cyclic systems when applied to suitable dihalides. This intramolecular version of the Wurtz reaction is more efficient when metals other than sodium, and in particular zinc or Li/Hg amalgam, are used. A comprehensive review of the older literature on this intramolecular reaction is available.⁶⁶ The Wurtz coupling has proved particularly valuable for the synthesis of strained small ring systems. Bicyclobutane (8), for example, can be obtained in 80–90% yield by treatment of 1-bromo-3-chlorocyclobutane (7) with sodium at high temperatures (equation 5).⁵ Other examples of the synthesis of strained cyclopropane-containing systems are given in equations (39) and (40).^{67,68} The synthesis of cyclopropanes by reductive coupling reactions has also been reviewed.⁶⁹

Cyclobutanes may also be prepared by this type of Wurtz reaction, but the approach is not successful for carbocyclic ring sizes greater than four. Studies of the reagents able to generate cyclobutane from 1,4-dihalo substrates have indicated that the use of Li/Hg amalgam in dioxane at high temperature gives the best yield.⁷⁰



Treatment of primary α,ω -haloalkanes with *t*-butyllithium at low temperature leads to efficient production of cycloalkanes, provided at least one of the halides is an iodide.^{29,30} This method provides very good yields of three-, four- and five-membered rings, but produces only minor amounts of cyclic products for 1,6-dihalides. Examples of this approach are given in equations (41) and (42).



A valuable modification to the coupling of alkylmagnesium halides involves the treatment of di-Grignard reagents with soluble silver salts, leading to cyclization. This method gives high yields of four-, five- and six-membered rings, but fails for larger ring sizes.⁷¹ Examples of this synthesis are given in equations (43) and (44). An interesting application of organocopper-type Wurtz coupling uses the novel 'cyclocuprate' (14),⁷² which reacts with *gem*-dihalides to give spiro-annulation products (equation 45).



2.1.3 π-ALLYLNICKEL HALIDES

In recent years transition metal π -allyl complexes have emerged as efficient reagents for the introduction of allyl units into organic substrates.^{73–78} The main advantage of the transition metal π -allyl complexes over normal allyl-magnesium, -lithium, -zinc, *etc.* systems is their much higher chemoselectivity. Two classes of reagents have proved particularly valuable, the π -allyl complexes of palladium^{75,78} and π -allylnickel halides.^{73,74,76,77} These complexes are complementary, since π -allylpalladium complexes react with electron-rich centers (*e.g.* stabilized anions) and π -allylnickel halides react with electron-poor centers, (*e.g.* alkyl halides and carbonyl groups). The high reactivity of π -allylnickel halides towards alkyl halides, coupled with their chemoselectivity and the use of heteroatom-substituted complexes make these reagents valuable alternatives to the normal Wurtz-type reagents.

2.1.3.1 Preparation, Structure and Experimental Techniques

The dimeric π -allylnickel halide complexes are most conveniently prepared from allylic halides and nickel(0) species, such as Ni(COD)₂ and Ni(CO)₄.⁷³ Although other allylic systems have been used,⁷⁹ the most common procedure involves treatment of an allylic bromide with Ni(CO)₄ at 50–70 °C.⁸⁰ This gives rise to the moderately air-sensitive dimeric π -allyl species (equation 46). In polar coordinating solvents, (DMF, HMPA, *N*-methylpyrrolidone, *etc.*) these dimeric species are converted into highly reactive monomeric species (equation 47).⁸¹



Coupling Reactions

The following general protocol describes the preparation and use of the complexes. Treatment of an allylic halide with Ni(CO)₄ in benzene at 50 °C, or Ni(COD)₂ at *ca*. 0 °C generates the dimeric complex. Solvent removal, followed by addition of a polar coordinating solvent and the substrate leads to cross-coupling of the reactive monomeric allylnickel species with the substrate. Excellent practical details are available for these reactions and their synthetic uses have been reviewed,⁷³ together with the structure and bonding of the complexes.

2.1.3.2 Reactivity Patterns

2.1.3.2.1 Chemoselectivity

The π -allylnickel halides show quite different reactivity profiles to the more normal organo-lithium, -sodium or -copper species used in Wurtz-type couplings. In particular, many functional groups which react rapidly with the conventional organometallic reagents are inert to π -allylnickel halides. Functional groups which are either inert to the complexes or react very slowly under forcing conditions include acid chlorides, esters, ethers, nitriles, alkynic protons, alkenes, alcohols, aryl, vinyl and alkyl chlorides, allylic ethers and acetals. Functional groups which react under forcing conditions (*i.e.* above 55 °C leading to extensive thermal decomposition of the complexes) include simple ketones and α , β -unsaturated ketones, with only 1,2-addition being observed (even in the presence of CuI). The complexes react slowly, with temperatures of 40–50 °C required for synthetic utility, with aldehydes, cyclic (and some acyclic) ketones, allylic chlorides and, in one report, epoxides. Examples of these reactions are given in equations (48)–(51).



In contrast to this low reactivity towards many electron-poor centers, the complexes react at or below room temperature with primary and secondary alkyl iodides, bromides and allyl, vinyl and aryl bromides and iodides, giving cross-coupled products. Phenyl ketones react to give homoallylic alcohols, whereas quinones give allyl-substituted quinones or hydroquinones. 1,2-Diketones give α -ketohomoallylic alcohols (even in the presence of excess complex, reaction occurs at only one carbonyl group) and 2-pyridyl carboxylates give β , γ -unsaturated ketones as the major products. Examples illustrating this reactivity profile are given in equations (52)–(59).⁷³ This chemoselectivity allows the reaction of the complexes with substrates containing more than one functional group, *e.g.* secondary iodide and secondary alcohol or iodide and chloride in the same molecule (equations 59–61). It is also possible to prepare a range of complexes containing functional groups; thus the coupling reaction both forms a C—C bond and intro-



425



duces a functional group, possibly in a masked form, into the product. Examples of substituted complexes which have been prepared successfully and used in synthesis include structures (15)-(19).⁷³

2.1.3.2.2 Regioselectivity

Substituted π -allylnickel complexes which could give products derived from reaction either at a primary or a secondary/tertiary terminus almost invariably react only at the primary center (equations 62 and 63).⁸²



2.1.3.2.3 Stereoselectivity

Coupling of a π -allylnickel halide with an alkyl halide regenerates the double bond originally present in the allyl halide, in addition to forming a new carbon-carbon bond. For trisubstituted double bonds, the products are normally mixtures of (*E*)- and (*Z*)-alkenes, either in a close to 1:1 ratio or with the (*E*)isomer in excess. In certain favorable cases^{83,84} (*E*):(*Z*) ratios of up to 98:2 have been achieved, albeit at the expense of rather low yields of coupled products (see Scheme 2). In general, highly coordinating solvents (HMPA, DMF) give high yields but rather low stereoselection, whereas solvents of lower coordinating power give better stereoselection, coupled with lower overall yields.^{83,84} Optically active halides couple with the reagents to give racemic products.⁷⁴

2.1.3.3 Coupling with sp³ Carbon Centers

2.1.3.3.1 Alkyl halides

In natural product synthesis, the 1,1-dimethylallyl complex (20) has been used to introduce a 5-carbon unit into a number of complex carbon skeletons. Examples of the use of this complex include the prep-



aration of α -santalene (21) and epi- β -santalene (22),⁸⁵ campherenone (23) and epi-campherenone (24),⁸⁶ and desmosterol (25)⁸⁷ (equations 64-68). It is interesting that more traditional approaches to α -santalene via coupling with Grignard reagents give only very low yields, compared to the excellent yield obtained with the nickel complex.⁸⁸





2.1.3.3.2 Allyl halides

Allylic halides may be coupled in the presence of Ni(CO)4 to give 1,5-dienes (Scheme 3).⁸⁹ This approach is very attractive for the synthesis of terpenes by sequential addition of isoprene units. Reaction of 1,1-dimethylallylnickel bromide (20), for example, with the allylic halide shown in Scheme 3a gives the geranyl carbon skeleton.⁹⁰ Unfortunately this approach is complicated by the fact that the π allylnickel halide complex can undergo ligand exchange with the substrate allyl halide,⁹¹ giving a new complex (Scheme 3b); thus homo-coupling products derived from both of the allyl units result (Scheme 3b). The relative proportions of the homo- and cross-coupled products depend on the relative rates of ligand exchange and the rates of reaction of each complex with the allylic halides present. These reactions normally generate all possible 1,5-diene products in the expected statistical distribution; for example in Scheme 3b the ratio of the three 1,5-diene products is $C_6:C_7:C_8 = 25:52:25$. For simple complexes, the ratio of these products is not altered significantly by changes in solvent, addition of competing ligands or changes in the leaving group in the allylic substrate.⁷³ This approach has found application for certain complexes which undergo ligand exchange reactions only slowly. The 1,1-dimethylallyl complex (20) for example, reacts with a range of allylic bromo ethers and esters giving the geranyl derivatives (26), (27) and (28), cross-coupled products, in reasonable yields.⁹⁰ Only small quantities of homo-coupled products were observed in these reactions (equation 69). In contrast, the reaction of the bromo ester (29) with the complex (20) gave 70% of (30), the product of homo-coupling (*i.e.* dimerization of the substrate) as shown in equation (70).







The substituted π -allylnickel bromide complexes (31) and (32) have been reacted with prenyl bromide to give geranyl ethers and with geranyl bromide to give farnesyl ethers (equation 71)⁹¹; (E):(Z) ratios of up to 93:7 were observed in these cross-couplings. The substituted complex (33) has been used to introduce the isoprene unit into a range of terpenoid skeletons;⁸² thus reaction with prenyl bromide leads to myrcene and with geranyl bromide β -farnesene is formed, both reactions giving (E):(Z) mixtures. β -Sinensal has been prepared from the allylic chloride (34), by coupling with the π -allylnickel bromide complex (35; equation 72). The desired cross-coupled material (50% yield) was accompanied by lesser quantities (20% and 26%) of the homo-coupled products. The (E):(Z) ratio of the desired product was 93:7 in DMF and this ratio could be improved to 98:2 in THF, although the chemical yield fell to 23% under these conditions.



2.1.3.3.3 Formation of cyclic systems

Two allylic halides in the same molecule may, under high dilution conditions, be coupled using Ni(CO)₄ to give a cyclic 1,5-diene. As each allyl group can react to one of two positions, three possible cyclic products may result. In Table 1 this corresponds to ring sizes of n + 2, n + 4 and n + 6. In the generalized example shown, this corresponds to secondary-secondary, primary-secondary, and primary-primary coupling respectively. The relative proportion of each product for a given value of n is deter-
Coupling Reactions

mined by the normal rules governing ring closure effects, together with the preference for primary-primary coupling shown by the π -allylnickel halide intermediates. The accepted normal rates of ring closure reactions are 3 < 4 < 5 > 6 > 7 > 8 < 9 < 10 < 11 < 12, etc. up to 18, with the relative rate differences becoming smaller as the ring size increases. From the results in Table 1⁹² it can be seen that for n = 6, 8and 12, the ring size obtained is governed by the primary-primary bias of the coupling reaction as the largest available ring size predominates, rather than formation of a mixture of ring sizes due to roughly similar rates of closure, as would be predicted. For n = 2 and n = 4, however, the normal preference for formation of six-membered rings overcomes this bias and six-membered rings are formed by primarysecondary and secondary-secondary closures, respectively. The observed high yields for ring sizes > 12, coupled with the fact that regardless of starting geometry of the allyl halides the products are >95% (E), (E)-isomers, has led to the use of this method for the synthesis of macrocyclic natural products. Somewhat different results are obtained for exocyclic alkenes⁹³ and these are summarized in Table 2. It can be seen that this method is of no value for 12-membered rings. A cyclization with one endocyclic and one exocyclic alkenic bond gives intermediate results (equation 73). Related to these results is the interesting trimerization of (36) giving (37; Scheme 4). Corey et al. showed that (36), (38) and (39) added to (36) all give (37) on treatment with Ni(CO)4.94 This behavior is very interesting, as it requires (39) to undergo intermolecular reaction with (36), followed by ring closure to the nine-membered ring (37),





Value of n	Possible ring size of product	Observed ring size of product	Yield (%)	
2	4/6/8	6	42	
4	6/8/10	6	a	
6	8/10/12	12	59	
8	10/12/14	14	74	
12	14/16/18	18	84	

"Yield unknown.



Table 2

Value of n	Possible ring size	Observed ring size	Yield (%)	
6 8 10 14	8/10/12 10/12/14 12/14/16 16/18/20	14 16 20	37 81 40	

rather than intramolecular coupling to give a six-membered ring system. Indeed, treatment of (**39**) with excess Ni(CO)₄ gives only 10% of the six-membered ring derived from intramolecular coupling. Two 14-membered macrocyclic natural products have been prepared using this methodology, *i.e.* cembrene (**40**; equation 74)^{95,96} and casbene (**41**; equation 75)^{97,98} and the 11-membered macrocycle humulene (**43**; equation 76) has been prepared *via* its geometric isomer (**42**) obtained from a bisallylic bromide precursor.^{99,100} In none of these cases were the reactions as clean or as high yielding as the simple model cyclizations would predict. The preference for formation of six-membered rings in this intramolecular coupling has been exploited in a synthesis of elemol (**44**; Scheme 5).¹⁰¹ The stability of esters to the π -allylnickel reagents has allowed the extension of this method to the synthesis of macrocyclic lactones.¹⁰²





Scheme 5

2.1.4 REFERENCES

- 1. A. C. Cope, E. P. Burrows, M. E. Derieg, S. Moon and W.-D. Wirth, J. Am. Chem. Soc., 1965, 87, 5452.
- F. Vögtle and P. Neumann, Synthesis, 1973, 85. 2.
- 3. K. Burri and W. Jenny, Helv. Chim. Acta, 1967, 50, 1978.
- 4. E. Müller and G. Röscheisen, Chem. Ber., 1957, 90, 543.
- G. M. Lampman and J. C. Aumiller, Org. Synth., 1971, 51, 55. 5.
- 6. H. Soroos and H. B. Willis, J. Am. Chem. Soc., 1941, 63, 881; D. Bryce-Smith and E. T. Blues, Org. Synth.,
- 1967, 47, 113. 7. F. C. Whitmore and G. H. Fleming, J. Am. Chem. Soc., 1933, 55, 3803; G. Calingaert, H. Soroos, V. Itnizda and H. Shapiro, J. Am. Chem. Soc., 1944, 66, 1389. C. F. Hobbs and W. C. Hammann, J. Org. Chem., 1970, 35, 4188.
- 8.
- H. Gilman and J. Robinson, Org. Synth., Coll. Vol., 1943, 2, 47; H. Gilman and W. E. Catlin, Org. Synth., 9. Coll. Vol., 1941, 1, 471.
- 10. M. Tamura and J. Kochi, Synthesis, 1971, 303.
- S. P. Tanis, Tetrahedron Lett., 1982, 23, 3115. 11.
- 12. S. P. Tanis and D. B. Head, Tetrahedron Lett., 1982, 23, 5509.
- 13. S. P. Tanis and D. B. Head, Tetrahedron Lett., 1984, 25, 4451.
- T. A. Baer and R. L. Carney, Tetrahedron Lett., 1976, 4697. 14.
- L. Pop, I. Oprean, A. Barabas and F. Hodosan, J. Prakt. Chem., 1986, 328, 867. 15.
- 16. S. V. Trivedi and V. R. Mamdapur, Indian J. Chem., Sect. B, 1986, 25, 176.
- 17. P. Perrin, F. Aubert, J. P. Lellouche and J.-P. Beaucourt, Tetrahedron Lett., 1986, 27, 6193.
- 18. O. P. Vig, M. L. Sharma and N. Vohra, Indian J. Chem., Sect. B, 1986, 25, 1052.
- S. P. Tanis and P. M. Herrinton, J. Org. Chem., 1983, 48, 4572. 19.
- 20. F. Derguini-Boumechal, R. Lorne and G. Linstrumelle, Tetrahedron Lett., 1977, 1181.
- C. Descoins and C. A. Henrick, Tetrahedron Lett., 1972, 2999.
 R. Baker, P. M. Winton and R. W. Turner, Tetrahedron Lett., 1980, 21, 1175.
- 23. Y.-S. Lee, L. del Valle and G. L. Larson, Synth. Commun., 1987, 17, 385.
- G. M. Whitesides, J. SanFilippo, Jr., C. P. Casey and E. J. Panek, J. Am. Chem. Soc., 1967, 89, 5302. 24.
- M. Tamura and J. K. Kochi, Bull. Chem. Soc. Jpn., 1972, 45, 1120. G. M. Whitesides, D. E. Bergbreiter and P. E. Kendall, J. Am. Chem. Soc., 1974, 96, 2806. 25.
- 26. 27. S. W. Breuer and F. A. Broster, Tetrahedron Lett., 1972, 2193.
- 28.
- A. McKillop, L. F. Elsom and E. C. Taylor, J. Am. Chem. Soc., 1968, 90, 2423. 29. W. F. Bailey and R. P. Gagnier, Tetrahedron Lett., 1982, 23, 5123.
- 30. W. F. Bailey, R. P. Gagnier and J. J. Patricia, J. Org. Chem., 1984, 49, 2098.

- 31. A. R. Bassindale, C. Eaborn and D. R. M. Walton, J. Chem. Soc. C, 1969, 2505.
- 32. L. H. Sommer and W. D. Korte, J. Org. Chem., 1970, 35, 22.
- 33. W. D. Korte, K. Cripe and R. Cooke, J. Org. Chem., 1974, 39, 1168.
- 34. B. J. Wakefield, 'The Chemistry of Organolithium Compounds', Pergamon Press, Oxford, 1974.
- 35. B. J. Wakefield, in 'Comprehensive Organic Chemistry', ed. D. H. R. Barton and W. D. Ollis, Pergamon Press, Oxford, 1979, vol. 3, p. 943.
- 36. U. Schöllkopf, Methoden Org. Chem. (Houben-Weyl), 1970, 13, 1.
- 37. G. H. Posner, Org. React. (N.Y.), 1972, 19, 1.
- 38. G. H. Posner, Org. React. (N.Y.), 1975, 22, 253.
- J. F. Normant, Synthesis, 1972, 63.
- 40. J. F. Normant, Pure Appl. Chem., 1978, 50, 709.
- 41. J. F. Normant, J. Organomet. Chem. Libr., 1976, 1, 219.
- 42. G. H. Posner, 'An Introduction to Synthesis Using Organocopper Compounds', Wiley, Chichester, 1980.
- 43. J. N. Labovitz, C. A. Henrick and V. L. Corbin, *Tetrahedron Lett.*, 1975, 4209; L. D. Rosenblum, R. J. Anderson and C. A. Henrick, *Tetrahedron Lett.*, 1976, 419.
- 44. E. J. Corey and G. H. Posner, J. Am. Chem. Soc., 1968, 90, 5615.
- 45. J.-P. Gorlier, L. Hamon, J. Levisalles and J. Wagnon, J. Chem. Soc., Chem. Commun., 1973, 88.
- 46. G. H. Posner, Ph. D. Thesis, Harvard University, 1968 (Diss. Abstr., 1968, 29, 1613B).
- 47. G. M. Whitesides, W. F. Fischer, Jr., J. SanFilippo, Jr., R. W. Bashe and H. O. House, J. Am. Chem. Soc., 1969, 91, 487.
- 48. Y. Tamaru, H. Ochiai, T. Nakamura, K. Tsubaki and Z. Yoshida, Tetrahedron Lett., 1985, 26, 5559.
- 49. H. Ochiai, Y. Tamaru, K. Tsubaki and Z. Yoshida, J. Org. Chem., 1987, 52, 4418.
- 50. T. Kitazume and N. Ishikawa, J. Am. Chem. Soc., 1985, 107, 5186.
- 51. G. Courtois, D. Mesnard, J. R. Mahoungou and L. Miginiac, Bull. Soc. Chim. Fr., 1986, 449.
- 52. F. Ameer, S. E. Drewes, R. Hoole, P. T. Kaye and A. T. Pitchford, J. Chem. Soc., Perkin Trans. 1, 1985, 2713.
- 53. A. Suzuki, S. Nozawa, M. Harada, M. Itoh, H. C. Brown and M. M. Midland, J. Am. Chem. Soc., 1971, 93, 1508.
- 54. T. Taguchi, O. Kitagawa, T. Morikawa, T. Nishiwaki, H. Uehara, H. Endo and Y. Kobayashi, Tetrahedron Lett., 1986, 27, 6103.
- 55. S. Inaba and R. D. Rieke, Synthesis, 1984, 842.
- 56. E. J. Corey and G. H. Posner, Tetrahedron Lett., 1970, 315.
- 57. M. T. Reetz, J. Westermann and R. Steinbach, Synthesis, 1980, 900, 901.
- 58. M. T. Reetz, B. Wenderoth, R. Peter, R. Steinbach and J. Westermann, J. Chem. Soc., Chem. Commun., 1980, 1202.
- 59. Y. Yamada and D. Momose, Chem. Lett., 1981, 1277; D. Momose, K. Iguchi, T. Sugiyama and Y. Yamada, Tetrahedron Lett., 1983, 24, 921.
- 60. M. Iyoda, M. Sakaitani, H. Otsuka and M. Oda, Chem. Lett., 1985, 127.
- 61. D. L. J. Clive, P. C. Anderson, N. Moss and A. Singh, J. Org. Chem., 1982, 47, 1641.
- 62. T.-L. Ho and G. A. Olah, Synthesis, 1977, 170.
- 63. Y. Okude, T. Hiyama and H. Nozaki, Tetrahedron Lett., 1977, 3829.
- 64. G. A. Olah and G. K. S. Prakash, Synthesis, 1976, 607.
- 65. F. Orsini, F. Pelizzoni and G. Ricca, Synth. Commun., 1982, 12, 1147.
- 66. E. E. Royals, 'Advanced Organic Chemistry', Prentice-Hall, Englewood Cliffs, NJ, 1954.
- 67. S. J. Cristol and B. B. Jarvis, J. Am. Chem. Soc., 1967, 89, 401.
- 68. T. Clark and M. A. McKervey, in 'Comprehensive Organic Chemistry', ed. D. H. R. Barton and W. D. Ollis, Pergamon Press, Oxford, 1979, vol. 1, p. 37.
- 69. R. Kh. Freidlina, A. A. Kamyshova and E. Ts. Chukovskaya, Russ. Chem. Rev. (Engl. Transl.), 1982, 51, 368.
- 70. D. S. Connor and E. R. Wilson, Tetrahedron Lett, 1967, 4925.
- 71. G. M. Whitesides and F. D. Gutowski, J. Org. Chem., 1976, 41, 2882.
- 72. F. Scott, B. G. Mafunda, J. F. Normant and A. Alexakis, Tetrahedron Lett., 1983, 24, 5767.
- 73. D. C. Billington, Chem. Soc. Rev., 1985, 14, 93.
- 74. M. F. Semmelhack, Org. React. (N.Y.), 1972, 19, 115.
- 75. L. S. Hegedus, J. Organomet. Chem. Libr., 1976, 1, 329.
- 76. R. Baker, Chem. Rev., 1973, 73, 487.
- 77. G. P. Chiusoli and G. Salerno, Adv. Organomet. Chem., 1979, 17, 195.
- 78. B. M. Trost, Acc. Chem. Res., 1980, 13, 385.
- 79. M. F. Semmelhack and S. J. Brickner, J. Am. Chem. Soc., 1981, 103, 3945; N. L. Bauld, Tetrahedron Lett., 1962, 859; M. F. Semmelhack, A. Yamashita, J. C. Tomesch and K. Hirotsu, J. Am. Chem. Soc., 1978, 100, 5565.
- 80. M. F. Semmelhack and P. M. Helquist, Org. Synth, 1972, 52, 115
- 81. D. Walter and G. Wilke, Angew. Chem., Int. Ed. Engl., 1966, 5, 897.
- L. S. Hegedus and S. Varaprath, Organometallics, 1982, 1, 259; M. Onaka, T. Goto and T. Mukaiyama, Chem. Lett., 1979, 1483; K. Sato, S. Inoue and K. Saito, J. Chem. Soc., Chem. Commun., 1972, 953; K. Sato, S. Inoue, Y. Takagi and S. Morii, Bull. Chem. Soc. Jpn., 1976, 49, 3351.
- 83. K. Sato, S. Inoue and K. Watanabe, J. Chem. Soc., Perkin Trans. 1, 1981, 2411.
- K. Sato, S. Inoue and K. Saito, J. Chem. Soc., Perkin Trans. 1, 1973, 2289.
 E. J. Corey and M. F. Semmelhack, J. Am. Chem. Soc., 1967, 89, 2755.
- 86. G. L. Hodgson, D. F. MacSweeney, R. W. Mills and T. Money, J. Chem. Soc., Chem. Commun., 1973, 235.
- 87. S. K. Dasgupta, D. R. Crump and M. Gut, J. Org. Chem., 1974, 39, 1658.
- 88. E. J. Corey, S. W. Chow and R. A. Scherrer, J. Am. Chem. Soc., 1957, 79, 5773.
- 89. I. D. Webb and G. T. Borcherdt, J. Am. Chem. Soc., 1951, 73, 2654.
- 90. K. Sato, S. Inoue, S. Ota and Y. Fujita, J. Org. Chem., 1972, 37, 462.

- K. Sato, S. Inoue and S. Morii, Chem. Lett., 1975, 747.
 E. J. Corey and E. K. W. Wat, J. Am. Chem. Soc., 1967, 89, 2757.
 E. J. Corey and P. M. Helquist, Tetrahedron Lett., 1975, 4091.
- 94. E. J. Corey and M. F. Semmelhack, Tetrahedron Lett., 1966, 6237.
- 95. W. G. Dauben, G. H. Beasley, M. D. Broadhurst, B. Muller, D. J. Peppard, P. Pesnelle and C. Suter, J. Am. Chem. Soc., 1974, 96, 4724. 96. W. G. Dauben, G. H. Beasley, M. D. Broadhurst, B. Muller, D. J. Peppard, P. Pesnelle and C. Suter, J. Am.
- Chem. Soc., 1975, 97, 4973.
- 97. L. Crombie, G. Kneen and G. Pattenden, J. Chem. Soc., Chem. Commun., 1976, 66.
- 98. L. Crombie, G. Kneen, G. Pattenden and D. Whybrow, J. Chem. Soc., Perkin Trans. 1, 1980, 1711. 99. E. J. Corey and E. Hamanaka, J. Am. Chem. Soc., 1967, 89, 2758.
- 100. E. J. Corey and E. Hamanaka, J. Am. Chem. Soc., 1964, 86, 1641.
- 101. E. J. Corey and E. A. Broger, Tetrahedron Lett., 1969, 1779.
- 102. E. J. Corey and H. A. Kirst, J. Am. Chem. Soc., 1972, 94, 667.

2.2 Coupling Reactions Between *sp*³ and *sp*² Carbon Centers

KOHEI TAMAO

Kyoto University, Japan

2.2.1	INTRODUCTION	436
2.2.2	COUPLING BETWEEN C(sp ³)-ORGANOMETALLICS AND ALKENYL HALIDES	436
2.2	2.2.1 Primary Alkyllithium and Alkylmagnesium Reagents	436
	2.2.2.1.1 Fundamental aspects of alkene stereochemistry	436
	2.2.2.1.2 Alkenyl chlorides	437
	2.2.2.1.3 Alkenyl bromides	437
2	2.2.2.1.4 Alkenyl todides	439
2.4	2.2.2 Secondary Arkyi Grighara Reagents	440
2.2	2.2.5 Ternary Anyi Originara Reagents	442
	2.2.2.4.1 Primary and secondary alkylainc reagents	442
	2.2.2.4.2 Reformatsky and related reagents	443
	2.2.2.4.3 Perfluoroalkylzinc reagents	444
2.2.3	COUPLING OF C(sp ³)-ORGANOMETALLICS WITH ALKENYL O, S, Se AND TE COMPOUNDS	444
2.2	2.3.1 Enol Ethers, Enol Carboxylates, Enol Phosphates and Enol Sulfonates	444
2.2	2.3.2 Alkenyl Sulfides, Sulfones, Selenides and Tellurides	446
2.2.4	COUPLING BETWEEN C(sp ³)-ORGANOMETALLICS AND DIFUNCTIONAL ALKENES	448
2.2.5	COUPLING BETWEEN C(sp ³)-ORGANOMETALLICS AND AROMATIC HALIDES	450
2.2	2.5.1 Coupling of Primary Alkyl Grignard Reagents with Aromatic Halides	450
2.2	2.5.2 Coupling of Secondary and Tertiary Alkyl Grignard Reagents with Aromatic Halides	452
2.2	2.5.3 Organozinc and Organotin Reagents	452
	2.2.5.3.1 Primary alkylzinc and alkyltin reagents	453
	2.2.5.3.2 Reformalsky and related reagents	454
2.2.6	COUPLING OF C(sp ³)-ORGANOMETALLICS WITH ARYL O, S AND Se COMPOUNDS	455
2.2.7	COUPLING OF C(sp ³)-ORGANOMETALLICS WITH POLYFUNCTIONAL AROMATICS	457
2.2.8	COUPLING OF C(sp ³)-ORGANOMETALLICS WITH HETEROAROMATIC HALIDES AND SULFIDES	459
2.2	2.8.1 Furans and Thiophenes	459
2.2	2.8.2 Pyridines and Quinolines	460
2.2	2.8.3 Heteroaromatic Compounds containing more than Two Heteroatoms	461
2.2.9	COUPLING OF C(sp ³)-ORGANOMETALLICS WITH ACYL CHLORIDES AND RELATED COMPOUNDS	463
2.2.10	COUPLING OF C(sp ²)-ORGANOMETALLICS WITH C(sp ³) HALIDES AND RELATED COMPOUNDS	464
2.2	10.1 Coupling between C(sp ²)-Organometallics and Primary Alkyl Halides	464
2.2	1.10.2 Coupling between C(sp ²)-Organometallics and Secondary Alkyl Halides	466
2.2	10.3 Coupling between C(sp ²)-Organometallics and C(sp ³)-Oxygen Compounds	466
2,2	1.10.4 Coupling of C(sp ⁴)-Organometallics with Allylic Halides and Related Compounds	467
2.2.11	REFERENCES	477

2.2.1 INTRODUCTION

Selective cross-coupling reactions between $C(sp^3)$ and $C(sp^2)$ centers had been one of the most difficult tasks in carbon-carbon bond synthesis until the early 1970s, when it was first reported that iron,^{1,2} nickel,^{3,4} palladium⁵ and copper^{2,6} catalysts are extremely effective for cross-coupling of Grignard reagents with organic halides. Now, nearly 20 years later, transition metal catalyzed cross-coupling has become the reaction of first choice for this purpose.

This section describes the cross-coupling reactions between organometallic reagents and organic halides and related reactions catalyzed by transition metal complexes or salts (<10 mol %). The following two types of reaction (equations 1 and 2) are treated, where R = primary, secondary or tertiary alkyl, alicyclic, allyl or functionalized alkyl groups; R' = alkenyl, aryl or heteroaryl groups; M = Li, Mg, Cu, Zn, B, Al, Si, Sn, Cd, Hg, Ti or Zr; X = F, Cl, Br, I, OH, OR, OCOR, OTs, OTf, OP(O)(OR)₂, SH, SR, SO₂R, SR₂⁺, SeR or NR₃⁺; and catalyst = Mn, Fe, Ru, Co, Ni, Pd or Cu complex or salt. Among the catalysts listed above, Fe, Ni and Pd catalysts seem to be most effective for the coupling $R(sp^3)$ —M + $R'(sp^2)$ —X, while the coupling $R(sp^3)$ —X + $R'(sp^2)$ —M is catalyzed best by Cu complexes and the allylic $R(sp^3)$ —X + R'(sp²)—M by Cu, Ni or Pd complexes. Whilst coupling reactions are achieved efficiently with primary alkyl organometallics in the reaction shown by equation (1) and with primary alkyl halides in the reaction shown by equation (2), irrespective of the presence or absence of β -hydrogens, there are some limitations on the secondary and tertiary alkyl organometallics and halides. Thus, coupling reactions of β -H-bearing secondary and tertiary alkyl metallic reagents with $C(sp^2)$ halides are frequently accompanied by reduction of halides or by alkyl group isomerization owing to β -H elimination in alkyl-transition metal intermediates, especially in Ni- or Pd-catalyzed reactions; these problems, however, may be suppressed by the appropriate choice of ligand or metallic reagent. The reaction occurring in equation (2) may also be complicated with β -H-bearing secondary alkyl halides owing to the competing β -H elimination. Not yet achieved are the coupling reactions between tertiary alkyl organometallics and aromatic halides and between tertiary alkyl halides and $C(sp^2)$ -organometallics.

$$R(sp^3) - M + R'(sp^2) - X \xrightarrow{\text{catalyst}} R(sp^3) - R'(sp^2) + M - X$$
(1)

$$R(sp^3) - X + R'(sp^2) - M \xrightarrow{\text{catalyst}} R(sp^3) - R'(sp^2) + M - X$$
(2)

The literature has been surveyed up to 1988. Some recent reviews relevant to this section are as follows: Ni-,⁷⁻¹² Pd-,^{7,11-17} and Cu-catalyzed^{7,18} coupling reactions. Catalytic asymmetric cross-coupling reactions are not treated in this section, since the subject has recently been well reviewed.¹⁹

2.2.2 COUPLING BETWEEN C(sp³)-ORGANOMETALLICS AND ALKENYL HALIDES

2.2.2.1 Primary Alkyllithium and Alkylmagnesium Reagents

2.2.2.1.1 Fundamental aspects of alkene stereochemistry

The coupling reactions between $C(sp^3)$ -organometallics and alkenyl halides provide synthetically useful methods for the stereoselective synthesis of di-, tri- and tetra-substituted alkenes, since the reactions proceed with retention of alkene geometry of the stereodefined alkenyl halides in most cases. The most fundamental stereochemical aspects are summarized first. Thus, (E)- and (Z)-bromopropenes or bromostyrenes couple with the methyl Grignard reagent to form the methylated products in higher than 90% yields and with almost complete retention of configuration in the presence of Fe(DBM)3 (DBM = dibenzoylmethanato),²⁰ NiCl₂(DPPP)²¹ or Pd(PPh₃) $_{4}^{5}$ (equations 3 and 4). Similar stereospecific coupling

$$R \longrightarrow Br + MeMgBr \xrightarrow{Fe, Ni or Pd} R$$

$$R \longrightarrow Fe, Ni or Pd \qquad (3)$$

$$R \longrightarrow Fe, Ni or Pd \qquad (4)$$

R _____

occurs with methyllithium in the presence of $Pd(PPh_3)_4^{22,23}$ or $RuCl(PPh_3)_3^{22}$ in refluxing benzene. Silver-catalyzed coupling of (Z)-bromopropene with MeMgBr proceeds with loss of stereochemistry.²⁴ The most useful catalysts seem to be Ni, Pd, or Fe complexes.

2.2.2.1.2 Alkenyl chlorides

Couplings between alkenyl chlorides and alkyl Grignard reagents are catalyzed best by nickel complexes,^{25,26} as exemplified by the synthesis of terminal alkenes,²⁵ disubstituted alkenes,²⁷ cyclic alkenes²⁵ and a silylmethyldiene (equations 5–8),²⁸ while Pd(PPh₃)₄ has also been claimed to be useful for a few cases.²⁶



2.2.2.1.3 Alkenyl bromides

In addition to Ni catalysts, Fe and Pd complexes are also effective for couplings involving alkenyl bromides. Although organolithium reagents are not generally useful for transition metal catalyzed coupling reactions, moderate yields are achieved in the coupling of *n*-butyllithium with (E)- and (Z)-bromostyrene in the presence of Pd(PPh₃)₄ in benzene, the solvent being chosen to minimize side reactions such as metalation (equation 9).²²

Coupling reactions of alkyl Grignard reagents lacking β -H substitution are exemplified by the stereoselective synthesis of a trideuteriomethylated alkene (equation 10)²⁹ and of allylsilanes (equations 11 and 12).^{30,31} Cyclic bromoalkenes also undergo coupling smoothly (equation 13).³¹



Coupling Reactions

Coupling reactions of β -H-bearing alkyl Grignard reagents with parent vinyl bromide, (Z)- and (E)-alkenyl bromides are catalyzed by Fe,^{2,20} Ni³² and Pd³³ respectively (equations 14–16). Similar Pd-catalyzed alkylations can be applied to 1- and 2-bromoethenylsilanes (equations 17 and 18).³⁴ β -Branched primary alkyl Grignard reagents also undergo Fe-³⁵ or Pd- catalyzed³⁶ coupling smoothly (equations 19 and 20).



Interestingly, there seems to be a striking difference in reactivity between (E)- and (Z)-isomers of 1bromoalkenes. (E)-Isomers are more reactive than the (Z)-isomers in the Pd-catalyzed reaction.³⁷ The reverse order is found in the Fe-catalyzed coupling reaction with β -H-bearing primary alkyl Grignard reagents,²⁰ but with the methyl and *s*-and *t*-alkyl Grignard reagents (E)-isomers seem to be more reactive. Therefore, coupling reactions of a geometrically isomeric mixture with molarly deficient alkyl Grignard

reagent can afford isomerically pure disubstituted alkenes, *i.e.* (E)-alkenes with Pd catalyst and (Z)-alkenes with Fe catalyst (equations 21 and 22).



Axially chiral, optically active alkenyl bromides couple with methyllithium³⁸ or with alkyl Grignard reagents³⁹ with retention of configuration (96% retention) in the presence of NiCl₂(DPPE) or Fe(DBM)₃; the reaction may be useful for the synthesis of optically active liquid crystals (equation 23). Allenyl bromides also undergo coupling with alkyl Grignard reagents under the catalysis by Ni(mesal)₂ (mesal = N-methylsalicylaldiminato); coupling of optically active allenyl bromides proceeds with retention, but with substantial loss of stereochemistry (equation 24).^{40,41}



2.2.2.1.4 Alkenyl iodides

In addition to Ni and Pd catalysts, Li₂CuCl₄ is also an effective catalyst for the coupling of alkenyl iodides with Grignard reagents.⁴² Since the stereochemistry of alkenyl iodides is also retained, the coupling reactions are useful for the stereoselective synthesis of disubstituted alkenes,^{36,43} trisubstituted alkenes,⁴⁴ allyl alcohols⁴⁵ and tetrasubstituted alkenes (equations 25–29).⁴⁶





2.2.2.2 Secondary Alkyl Grignard Reagents

The cyclohexyl Grignard reagent couples smoothly with vinyl chloride in the presence of NiCl₂(DPPP) (equation 30).²⁵ Alkyl group isomerization from secondary to primary is frequently observed in the coupling of secondary alkyl Grignard reagents with alkenyl halides catalyzed by Ni or Pd complexes; the extent of the isomerization is strongly dependent upon the nature of ligand and metal in the catalyst as well as the organic halides (see also Section 2.2.5.2). Thus, in the coupling of (*E*)-bromostyrene with the *s*-butyl Grignard reagent (equation 31), while NiCl₂(DPPP) or PdCl₂(DPPP) as catalysts induce the isomerization to some extent, no isomerization is observed with PdCl₂(DPPF) at all, only the *s*-butylated product being formed in almost quantitative yields and with retention of the alkene stereo-chemistry.⁴⁷ The catalyst PdCl₂(DPPF) is also effective for similar coupling of less reactive alkenyl bromides, such as 2-bromopropene, without alkyl isomerization (equation 32).⁴⁷

Cross-coupling reactions between the isopropyl Grignard reagent and alkenyl bromides or iodides are also catalyzed by Fe^{20} or Cu^{42} complexes, no alkyl group isomerization being observed (equations 33





and 34). Here again, (Z)-1-bromoalkenes seem to be more reactive than the (E)-isomers in the Fe-catalyzed reactions.²⁰ In the cross-coupling reaction of (E)- and (Z)-1-bromo-2-phenylthioethylenes with secondary alkyl Grignard reagents, however, the extent of isomerization is less with NiCl₂(DPPE) than with PdCl₂(DPPF), no isomerization being observed with Fe(DPM)₃ (DPM = dipivaloylmethanato); stereoselectivities are higher than 97% in all cases (equation 35; see also Section 2.2.4).^{48,49}



2.2.2.3 Tertiary Alkyl Grignard Reagents

There have been only three reports on cross-coupling of t-butylmagnesium chloride with alkenyl halides. An Fe-catalyzed reaction with (Z)-1-bromopropene affords a mixture of (Z)- and (E)-isomers of t-butylated product with some loss of stereochemistry (equation 36).²⁰ In the presence of NiCl₂(DPPF), (E)-bromostyrene couples with Bu'MgCl without alkyl group isomerization and stereochemical scrambling, while alkyl group isomerization is induced by NiCl₂(DPPP) as catalyst (equation 37); the NiCl₂(DPPF)-catalyzed t-butylation, however, seems to be restricted only to a reactive alkenyl halide such as (E)-bromostyrene.⁵⁰ Trisubstituted alkenyl iodides undergo coupling with Bu'MgCl in the presence of Li₂CuCl₄ with retention of configuration (equation 38).⁴²



NiCl₂(DPPP) 59% 78:22

$$\begin{array}{ccccccccccccc} Bu^n & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & \\ & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\$$

2.2.2.4 Organozinc and Organotin Reagents

Coupling reactions of alkylzinc and alkyltin reagents with alkenyl halides are catalyzed efficiently by Pd and, in some cases, by Ni complexes.

2.2.2.4.1 Primary and secondary alkylzinc reagents

The advantages of zinc reagents over Grignard reagents are that yields of coupled products are generally higher and a wider degree of functionality can be tolerated. It should be noted that coupling of β -Hbearing alkylzinc reagents, as well as β -H-lacking alkylzinc reagents, with alkenyl bromides or iodides are catalyzed even by unidentate phosphine-ligated Pd complexes such as Pd(PPh₃)₄, as illustrated in equation (39).⁵¹ *n*-Butyl-,⁵¹ benzyl-,⁵² 2-phenylethyl-,⁵³ homoallyl-^{51,54} and homopropargyl-zinc^{51,55} reagents all have been shown to couple with trisubstituted alkenyl iodides and β -bromo- α , β -unsaturated carboxylates with retention of the alkene stereochemistry, as exemplified by the stereoselective synthesis of terpenoid skeletons (equations 40 and 41).



Secondary alkylzinc reagents also couple with a trisubstituted alkenyl iodide under similar conditions, but the coupling is accompanied by alkyl group isomerization from secondary to primary (sec:prim = 60:40);⁵¹ no alkyl group isomerization has been observed in the Pd-catalyzed coupling of a *s*-butylzinc reagent with an (*E*)-bromovinylboron derivative (equation 42).⁵⁶ A cyclopropylzinc reagent undergoes a facile coupling with (*E*)-iodoalkenes catalyzed by Pd(PPh₃)₄ (equation 43).⁵⁷



2.2.2.4.2 Reformatsky and related reagents

Cross-coupling reactions of the Reformatsky reagent with vinyl, 2-propenyl and styryl bromides are catalyzed by Ni(PPh₃)₄ or Pd(PPh₃)₄ (10 mol %) in HMPA as an essential solvent to form β , γ -unsaturated esters in moderate yields.⁵⁸ While the (*E*)-stereochemistry of alkenyl bromides is almost retained, stereochemical scrambling is often observed with (*Z*)-isomers (equations 44 and 45). It may be mentioned here that similar, but stoichiometric, Ni-promoted couplings of lithium ester enolates with alkenyl bromides are also known.^{59,60} Zinc reagents derivable *in situ* from ω -iodoalkyl ketones with Zn/Cu also undergo Pd-catalyzed coupling with iodoalkenes (equation 46).⁶¹



 γ -Stannyl- α , β -unsaturated esters have been shown to couple with an alkenyl bromide at the γ -position selectively, but in low yield (equation 47).⁶² α -Sulfonylalkylstannanes also undergo the Pd-catalyzed coupling with alkenyl iodides (equation 48).⁶³



2.2.2.4.3 Perfluoroalkylzinc reagents

Perfluoroalkylzinc iodides, generated in situ from the iodides and zinc powder under ultrasound irradiation, couple smoothly with alkenyl bromides in the presence of $Pd(PPh_3)_4$ as catalyst, linear and branched perfluoroalkyl groups being coupled with retention of the (E)-alkene stereochemistry (equation 49).⁶⁴



2.2.3 COUPLING OF C(sp³)-ORGANOMETALLICS WITH ALKENYL O, S, Se AND TE COMPOUNDS

Only primary alkyl-metal reagents have been used in these cases, except in the instance of the Fe-catalyzed coupling of alkenyl sulfones.

2.2.3.1 Enol Ethers, Enol Carboxylates, Enol Phosphates and Enol Sulfonates

Enol ethers (equation 50),⁶⁵ dihydrofurans (equation 51),^{66,67} dihydropyrans (equation 52)^{65,68} and benzofuran (equation 53)⁶⁵ react with alkyl Grignard reagents in the presence of Ni catalysts to form the cross-coupling products through cleavage of $C(sp^2)$ -oxygen bonds. Benzene is used as a suitable solvent for the coupling of enol ethers. NiCl₂(PPh₃)₂ is used mostly for alkyl Grignard reagents lacking β -H substitution, while NiCl₂(DPPP) is suitable for β -H-bearing alkyl Grignard reagents. Only primary alkyl Grignard reagents can be used in all cases. The stereochemistry present in the cyclic ethers is highly retained, especially in those cases where the oxygen-bearing carbon contains an exocyclic substituent (equations 51 and 52), thus providing stereoselective synthesis of trisubstituted alkenes with a hydroxy group on the side chain.





Cross-coupling of enol silvl ethers with primary alkyl Grignard reagents is also catalyzed by Ni complexes, as exemplified by the conversion of an enol silvl ether into an allylsilane (equation 54)⁶⁹ and by stereoselective alkylation (equation 55).⁷⁰



Enol phosphates couple with primary alkyl Grignard reagents in the presence of Ni complexes. While NiBr₂ is a suitable catalyst for trimethylsilylmethylation (equation 56),⁷¹ NiCl₂(DPPP) should be used for β -H-bearing alkyl Grignard reagents (equation 57)^{71,72} Enol phosphates also undergo Pd(PPh₃)₄-catalyzed coupling reactions with trimethyl- or triethyl-aluminum with retention of the enol stereochemistry (equation 58)⁷³ and with trialkylmanganates, R₃MnLi.⁷⁴



Enol triflates couple with the allyl or benzyl Grignard reagent in the presence of Li₂MnCl₄ as a catalyst to give coupling products in high yields (equation 59).⁷⁴ Enol triflates can be alkylated with alkylzinc⁶¹

and alkyltin reagents⁷⁵ under catalysis by Pd(PPh₃)₄ with β -H-bearing alkyl groups being introduced; in the latter case LiCl is required as an essential additive (equations 60 and 61).



Diketene, a cyclic enol carboxylate, reacts with Grignard reagents in the presence of Ni^{76,77} or Co⁷⁸ salts or complexes to form 3-methylenealkanoic acids; in addition to methyl, trimethylsilylmethyl (equation 62)⁷⁶ or benzyl groups and β -H-bearing primary alkyl groups can be introduced if NiCl₂(DPPP)⁷⁷ or CoI₂ (equation 63)⁷⁸ are used as a catalyst at low temperatures.



2.2.3.2 Alkenyl Sulfides, Sulfones, Selenides and Tellurides

Nickel complexes, mostly NiCl₂(PPh₃)₂ and NiCl₂(DPPP), are effective catalysts for cross-coupling reactions between alkyl Grignard reagents and alkenyl–S and alkenyl–Se compounds. Although the sulfide leaving groups may be alkylthio or arylthio groups, aryl–S bonds (see Section 2.2.6) are also cleaved competitively in coupling of alkenyl phenyl sulfides, thus requiring excess amounts of Grignard reagents. The coupling reaction proceeds with retention (>94%) of alkene geometry (equations 64 and 65).^{79,80} The coupling reaction of alkenyl sulfides has been applied to the stereoselective synthesis of di-⁸¹ and tri-substituted alkenes (equation 66),^{82,83} cyclic alkenes (equation 67),⁸⁴ 1,3-dienes (equations 68–70)^{85–88} and 1,3,5-trienes (equation 71);⁸⁹ in the case of (Z)-silyldienyl sulfides, NiCl₂(DPPM)₂ (DPPM = Ph₂PCH₂PPh₂) seems to be the most suitable catalyst.⁸⁷

Ph,	; +	Bu ⁿ MgBr	$NiCl_2(PPh_3)_2(3\%)$	Ph,	(64)
SMe SMe			THF, reflux 44%	(E) 98%	(04)
EPh	+	Bu ⁿ MgBr	NiCl ₂ (PPh ₃) ₂ or NiCl ₂ (DPPP) Et ₂ O, reflux	Bu ⁿ Ph (Z) 94% E = S 45% (Z) 94% E = Se 76%	(65)



2,3-Dihydrothiopyran undergoes a ring-opening coupling with primary alkyl Grignard reagents catalyzed by NiCl₂(DPPP) in benzene to form, after S-methylation, a methylthio-containing (Z)-alkene stereoselectively (equation 72).⁹⁰ Thiophene, selenophene and tellurophene also undergo similar coupling reactions to form symmetrical (Z,Z)-1,3-diene derivatives (equation 73),^{91,92} furan being almost inert under similar conditions. 1-Benzothiophene reacts with 1 equiv. of the methyl Grignard reagent in the presence of NiCl₂(DPPP) as a catalyst (50 mol %) to form an o-[(Z)-1-propenyl]phenyl sulfide derivative stereoselectively. The sequence occurs via preferential cleavage of the S-alkenyl bond rather than the S-aryl bond, the latter also being cleaved with excess amounts of the Grignard reagent to form a double methylation product (equation 74, see also Section 2.2.6).⁹⁰

Alkenyl sulfones are methylated with MeMgX in the presence of Ni(acac)₂, the stereoselectivity being dependent upon the magnesium salt in the order MgCl > MgBr > MgI (equation 75).⁹³ Coupling of (*E*)- and (*Z*)-alkenyl sulfones with primary alkyl (except methyl) Grignard reagent occurs stereospecifically





in the presence of $Fe(acac)_3$ as a catalyst to afford trisubstituted alkenes (equation 76).⁹⁴ Fe catalysts, rather than Ni, afford better results in the coupling of secondary alkyl Grignard reagents with alkenyl sulfones; the coupling is, however, accompanied by reduction and alkyl group isomerization (equation 77).⁹⁵



2.2.4 COUPLING BETWEEN C(sp³)-ORGANOMETALLICS AND DIFUNCTIONAL ALKENES

1,1-Dichloroethylene undergoes a Ni-catalyzed double coupling reaction with the benzyl Grignard reagent (equation 78), but a complex mixture of products may result with β -H-bearing alkyl Grignard reagents.²⁵ Stereoselective, stepwise coupling of 1,1-dichloroalkenes can be achieved with a β -H-bearing primary alkylzinc reagent in the presence of PdCl₂(DPPB), followed by reaction with an alkyl Grignard reagent catalyzed by NiCl₂(DPPP) (equation 79); the higher reactivity of the chlorine atom *trans* to the 2-substituent may be noted (*cf.* Section 2.2.2.1.1).⁹⁶ 1,1-Disulfide analogs (ketene dithioacetals) also undergo stereoselective, stepwise coupling with primary alkyl Grignard reagents in the presence of Ni complex (equation 80).⁹⁷ These two reactions may be useful for the stereoselective synthesis of trisubstituted alkenes.



The predictabilities of Ni-catalyzed Grignard reactions involving 1,2-dichloroethylenes are not simple. Thus, whilst with the benzyl Grignard reagent bibenzyl is formed predominantly, double coupling products are obtained in moderate yields with β -H-bearing primary alkyl or cyclohexyl Grignard reagents, together with homocoupling of the Grignard reagent; only (*E*)-dicyclohexylethylene is formed both from (*E*)- and (*Z*)-dichloroethylenes (equation 81).²⁵ On the other hand, in the presence of excess amounts of (*E*)- or (*Z*)-dichloroethylene, selective monoalkylation by alkyl Grignard reagents may be realized in the presence of Ni(PPh₃)₄ as a catalyst to afford (*E*)- or (*Z*)-1-chloro-1-alkenes stereoselectively (equation 82).⁹⁸



A chemoselective monoalkylation of two unlike leaving groups on alkenes can more easily be attained based on the general reactivity order Br > Cl > SR > OR in Ni-catalyzed reactions. Grignard coupling reactions of α -bromoenol ethers catalyzed by NiCl₂(DPPP) generally stop cleanly at the monoalkylation stage to provide new routes to various enol ethers, primary alkyl, including β -H-bearing alkyl, Grignard reagents being employed (equations 83 and 84).^{99,100} (*E*)- and (*Z*)-1-Bromo-2-phenylthioethylenes can be transformed into disubstituted alkenes stereoselectively [(*E*) > 99%, (*Z*) > 98%] by stepwise alkyl Grignard coupling in a one-pot manner; whilst the first step with the (*Z*)-isomer may be catalyzed best by PdCl₂(PPh₃)₂, the first step with the (*E*)-isomer and the second step with both isomers are catalyzed most efficiently by NiCl₂(DPPE) (equation 85).¹⁰¹ In the first step, secondary alkyl Grignard reagents may be used without alkyl group isomerization (*cf*. Section 2.2.2.2; equation 86).⁴⁸ In Pd-catalyzed reactions of trimethylaluminum, the reactivity is in the order alkenyl–OP(O)(OPh)₂ > –SPh (equation 87).⁷³

449



2.2.5 COUPLING BETWEEN C(sp³)-ORGANOMETALLICS AND AROMATIC HALIDES

Cross-coupling reactions of alkyl-metal reagents with aromatic halides are catalyzed efficiently by Ni or Pd complexes.

2.2.5.1 Coupling of Primary Alkyl Grignard Reagents with Aromatic Halides

Primary alkyl Grignard reagents, regardless of the presence or the absence of β -hydrogens, couple with aromatic halides most efficiently in the presence of NiCl₂L₂ as catalyst, NiCl₂(DPPP) being most active and of general use.^{25,26} The catalytic activity of the Ni complexes, however, depends strongly upon not only the nature of the ligands (L), but also on the combination of Grignard reagent and organic halide, as will be mentioned at appropriate positions. While Pd catalysts are usually used for coupling of β -H-lacking alkyl (Me or PhCH₂) Grignard reagents with aromatic bromides and iodides,^{102,103} PdCl₂(DPPF) is also effective for coupling of β -H-bearing alkyl Grignard reagents with aryl bromides.³³

The reactivity order of halides is generally ArI > ArBr > ArCl >> ArF for Ni-catalyzed reactions and ArI > ArBr >> ArCl, ArF for Pd-catalyzed reactions. In the Ni-catalyzed reactions, chlorides usually give the most satisfactory results, since they exhibit a reasonable reactivity and give little side product; notably even fluorobenzene undergoes coupling (equation 88).²⁵ For couplings with bromobenzene, PdCl₂(DPPF) seems to be more effective (equation 89).³³ Generally, an electron-withdrawing substituent on the aromatic ring accelerates the coupling and an alkyl substituent deactivates the halide; methoxy group, however, seems to be an activating substituent (equation 90).²⁵ Dichlorobenzenes²⁵ and trichlorobenzenes¹⁰⁴ undergo dialkylation smoothly in the presence of NiCl₂(DPPP); even *o*-dialkylation is possible (equation 91, see Section 2.2.7 for monocoupling of polyfunctional compounds).^{25,26} Coupling reactions of alkyl Grignard reagents, lacking β -H substitution, with sterically hindered aromatic halides are also catalyzed by Ni complexes,^{25,105} but NiCl₂(PEt₃)₂ is better than NiCl₂(DPPP) in some cases (equations 92 and 93).²⁵ It should be noted that a reaction of sterically hindered mesityl bromide with BuⁿMgBr catalyzed by NiCl₂(DPPP) results in the formation of reduction product instead of coupling.²⁵ Coupling of the allyl Grignard reagent with bromobenzene is catalyzed by the complexes NiCl₂(DMPE) (DMPE = Me₂PCH₂CH₂PMe₂) or NiCl₂(PEt₃)₂ that contains more electron-donating ligands than NiCl₂(DPPP).²⁵



A β -branched, optically active primary alkyl Grignard reagent has been shown to undergo coupling with chlorobenzene without loss of optical activity in the presence of NiCl₂(DPPP), while about 10% racemization is induced when NiCl₂(PPh₃)₂ or NiCl₂(DMPE) are used as catalysts, possibly due to β -H elimination-addition sequences (equation 94).¹⁰⁶



Coupling Reactions

Di-Grignard reagents couple with *m*-dichlorobenzene to form *m*-cyclophanes in one step but in up to 20% yields depending on the chain lengths; oxamethylene bridges can also be introduced (equations 95 and 96).¹⁰⁷



2.2.5.2 Coupling of Secondary and Tertiary Alkyl Grignard Reagents with Aromatic Halides

The cyclopropyl Grignard reagent undergoes coupling with bromobenzene in the presence of NiCl₂(DMPE) rather than NiCl₂(DPPP), but in only moderate yields (equation 97).²⁵ Coupling reactions of the cyclohexyl Grignard reagent are catalyzed by NiCl₂(DPPP) (equation 98)²⁵ or less effectively by a Pd complex.¹⁰⁸



ortho 31%; meta 73%; para 63%

Alkyl group isomerization from secondary to primary occurs frequently in the Ni- or Pd-catalyzed coupling reactions of secondary alkyl Grignard reagents with aromatic halides, as described in the coupling reactions with alkenyl halides (Section 2.2.2.2). The isomerization is strongly dependent upon the nature of the phosphine ligands^{33,106,109,110} and of the substituents on the aromatic ring;¹¹¹ while NiCl₂(DMPE) induces the isomerization most readily,¹⁰⁶ with PdCl₂(DPPF) the isomerization is completely suppressed (equation 99).³³ The 1-phenylethyl Grignard reagent undergoes NiCl₂(DPPP)-catalyzed coupling with chlorobenzene without isomerization.¹⁰⁹

The reaction of the *t*-butyl Grignard reagent with chlorobenzene in the presence of NiCl₂(DMPE) results in the formation of isobutylbenzene in low yields through tertiary to primary group isomerization; no reaction whatsoever is observed with NiCl₂(DPPP) (equation 100).

2.2.5.3 Organozinc and Organotin Reagents

Couplings of alkylzinc and alkyltin reagents with aromatic halides, mostly bromides, are catalyzed by Pd or Ni complexes; toleration of functional groups such as CN, COR, CO₂R, NO₂, Cl and F is the most outstanding feature.



2.2.5.3.1 Primary alkylzinc and alkyltin reagents

Benzyl-, phenylethyl-, homoallyl-, homopropargyl- and silylmethyl-zinc reagents undergo coupling with aryl halides in the presence of Ni or Pd complexes (equations 101-103).^{31,53,103} Methyl-,¹¹² hydroxymethyl-,¹¹³ methoxymethyl-¹¹³ and allyltin^{114,115} compounds also couple with aryl bromides under the catalysis of Pd—PPh₃ complexes: the first two reagents require HMPA as solvent and in the last allylation the unusual reactivity order ArI < ArBr > ArCl has been noted (equations 104-106). With an α -ethoxyallyltin compound, the coupling occurs selectively at the γ -position, together with a low stereoselectivity at the resulting vinyl ether group (equation 107).¹¹⁵



453



Chloroarene chromium tricarbonyl complexes undergo Pd-catalyzed coupling with tetrabutyltin, the butylated arenes being freed from chromium by oxidation with iodine; activation of the otherwise inert C—Cl bond by the electron-withdrawing $Cr(CO)_3$ group may be noted (equation 108).¹¹⁶



2.2.5.3.2 Reformatsky and related reagents

Couplings between Reformatsky reagents and aromatic halides are catalyzed by $Ni(PPh_3)_4^{117}$ or $Pd(PPh_3)_4^{,117,118}$ with DMF, NMP or HMPA being used as the essential cosolvents. While the Ni-catalyzed reaction can be applied to aromatic chlorides, bromides and iodides, the Pd-catalyzed reaction is useful only for iodides. Compatibility of the free carboxylic acid may especially be noted (equation 109).

Enol silvl ethers undergo Pd-catalyzed coupling with aromatic bromides in the presence of tributyltin fluoride, which converts the enol silvl ethers into the stannyl ethers or α -stannyl ketones regarded as real active species; chemoselective α -arylation of terminal ketones is possible (equation 110).¹¹⁹



Cyanomethyltributyltin is a reagent for Pd-catalyzed cyanomethylation of aryl bromides with the Me, MeO or Cl substituent, but not with the MeCO, CN or NO₂ substituent (equation 111).¹²⁰ Dicyanomethylation of aryl iodides can be achieved with a sodium salt of malononitrile in the presence of Pd complexes (equation 112).¹²¹



2.2.6 COUPLING OF C(*sp*³)-ORGANOMETALLICS WITH ARYL O, S AND Se COMPOUNDS

Aryl phosphates undergo Ni-catalyzed coupling with primary alkyl Grignard reagents and triethylaluminum (equations 113–115).¹²² Arene triflates^{123,124} or homologous perfluoroalkanesulfonates¹²⁴ couple with primary alkyltin,¹²³ benzylzinc¹²⁴ and ester-containing alkylzinc (ester homoenolate) reagents¹²⁴ in the presence of Pd complexes as catalysts and LiCl as an essential additive (equations 116 and 117); compatibility of the free OH group in the coupling of alkyltin compounds is worthy of note. Allylation with allyltin compounds is accompanied substantially by 1-propenylation *via* alkene isomerization.¹²³ Siloxycyclopropanes serve as aldehyde, ketone and ester homoenolate equivalents in Pd-catalyzed coupling reactions with aryl triflates to form β -aryl carbonyl compounds (equations 118–120).¹²⁵





Nickel-catalyzed Grignard coupling reactions can be applied to arenethiols, ¹²⁶ sulfides, ^{79,90,126} sulfoxides, ¹²⁶ sulfones¹²⁶ and selenides.⁸⁰ Whilst the methyl Grignard reagent couples efficiently with all of them, β -H-bearing alkyl Grignard reagents give rather low yields of coupling products even in the presence of active NiCl₂(DPPP); the reactivity order is PhSeMe >> PhCl > PhSMe (equations 121 and 122).⁸⁰ Similarly, while benzothiophene may be converted into o-(Z)-1-propenyltoluene (see Section 2.2.3.2) and thioanthrene into o-xylene by the Ni-catalyzed methylation, dibenzothiophene gives a monomethylation product mainly, even in the presence of 2 equiv. of MeMgI (equation 123).⁹⁰

PhX + MeMgBr
$$\xrightarrow{\text{NiCl}_2(\text{PPh}_3)_2}$$
 PhMe (121)
X = SH, 64%; SMe, 97%; SOPh, 77%; SO₂Me, 97%
PhSeMe + BuⁿMgBr $\frac{\text{NiCl}_2(\text{DPPP})}{\text{Et}_2\text{O}, \text{ reflux}}$ PhBuⁿ (122)

95%



2.2.7 COUPLING OF C(sp³)-ORGANOMETALLICS WITH POLYFUNCTIONAL AROMATICS

Chemoselective couplings of polyfunctional aromatic compounds may be achieved by Ni- or Pd-catalyzed coupling with alkyl Grignard reagents. Whereas Grignard coupling reactions of aromatic polyhalides catalyzed by active NiCl₂(DPPP) or NiCl₂(DPPE) tend to result in the complete alkylation of all the halogen atoms present (see Section 2.2.5.1), a highly selective monoalkylation of dichlorobenzenes and trichlorobenzenes can be achieved with 1 equiv. of primary alkyl Grignard reagents in the presence of [Ni(triphos)Cl]PF₆¹²⁷ (triphos = PhP(CH₂CH₂PPh₂)₂) or Ni(acac)₂¹²⁸ as catalyst (equations 124 and 125). Dibromobenzenes and 1,3,5-tribromobenzene can also be monoalkylated with primary alkyl Grignard reagents or with benzylzinc halides in the presence of PdCl₂(DPPF) or PdCl₂(DPPB) as catalyst (equation 126).^{129,130}



Certain substituents can activate one halogen atom specifically in polyhalogen compounds, as exemplified by monoalkylation directed by methyl,¹²⁷ methoxy,^{127,131} and oxazolinyl groups (equations 127–130).¹³² A stepwise Grignard coupling of two unlike leaving groups is also possible based on the reactivity order Cl > SR (equation 131).¹³³



Monoalkylation of 2,3-dichloro-1,4-naphthoquinone is achieved by Pd-catalyzed reaction with tetraalkyltins or, less efficiently, with alkylzirconium complexes (equation 132).¹³⁴



2.2.8 COUPLING OF C(sp³)-ORGANOMETALLICS WITH HETEROAROMATIC HALIDES AND SULFIDES

Nickel- or palladium-catalyzed coupling reactions of alkyl Grignard or zinc reagents can be applied to heteroaromatic halides and sulfides. The characteristic features are, therefore, based on those described in the preceding sections and hence details are not repeated in this section. Aspects are summarized by the types of heteroaromatic compounds; NiCl₂(DPPP) is used as catalyst, unless stated otherwise hereafter.

2.2.8.1 Furans and Thiophenes

A chlorobenzofuran¹³⁵ and a 2-furyl sulfide¹³⁶ have been shown to couple with primary alkyl Grignard reagents (equations 133 and 134). 2-Bromothiophene undergoes a halogen-metal exchange reaction mainly with a basic primary alkyl Grignard reagent such as BuⁿMgBr to give only trace amounts of coupling products, but it couples smoothly with less basic Grignards such as (TMS)CH₂MgCl (equation 135).¹³⁷ 3-Bromothiophene shows no such problem and undergoes the Ni-catalyzed coupling with primary and secondary alkyl (Me, Buⁿ, PhCH₂, (TMS)CH₂, cyclopentyl) Grignard reagents (equation 136);^{137,138} the Pd-catalyzed coupling with a silylmethylzinc reagent is also possible (equation 137).³¹ One-step double coupling¹³⁷ and stepwise coupling¹³⁹ reactions of 3,4-dibromothiophene are possible under the catalysis by NiCl₂(DPPP) and PdCl₂(DPPF), respectively (equations 138 and 139).





2.2.8.2 Pyridines and Quinolines

2-Chloro- and 2-bromo-pyridines undergo Ni-catalyzed coupling reactions with primary alkyl Grignard reagents (equation 140);^{137,140,141} 3-halopyridines are less reactive and an '*ortho*' methyl substituent retards the coupling substantially.¹⁴⁰ Couplings of bromo- and iodo-pyridines with alkylzinc reagents seem to be catalyzed by Pd-complexes (equations 141 and 142).^{142,143} 2-Pyridyl sulfides¹³⁶ and triflate¹²⁵ also undergo Ni-catalyzed Grignard coupling and Pd-catalyzed coupling with siloxycyclopropanes, respectively (equations 143 and 144).



2,6-Dichloropyridine undergoes Ni-catalyzed double coupling reactions readily with primary alkyl Grignard reagents¹³⁷ and with di-Grignard reagents¹⁰⁷ as exemplified by the one-step synthesis of muscopyridine (equation 145). Pd-catalyzed monocoupling reactions occur selectively at the 2-position in 2,3-, 2,5- and 2,6-dichloropyridines and hence unsymmetrical dialkylpyridines are readily available (equation 146).¹³⁰





i, Me(CH₂)₁₀MgBr, PdCl₂(DPPB), Et₂O, reflux; ii, MeMgI, NiCl₂(DPPP), Et₂O, reflux

2-Chloro- and 3-bromo-quinolines and 2- and 4-chloroisoquinolines undergo Ni-catalyzed coupling with primary alkyl, allyl and cyclohexyl Grignard reagents; selective monoalkylation in 4,5-dichloroiso-quinoline may be noted (equations 147–150).^{137,144}



2.2.8.3 Heteroaromatic Compounds containing more than Two Heteroatoms

2-Chloro-¹⁴⁵ and 2-methylthio-benzothiazoles¹³⁶ undergo Ni-catalyzed coupling with alkyl Grignard reagents and the isopropyl group may also be introduced with ease (equations 151 and 152). 2-Methyl-thiobenzoimidazole is inert under similar conditions.¹³⁶ 2-Chloropyridazines,¹⁴⁶ 2-chloro-, 4-chloro-, 2,4-dichloro-, 4,6-dichloro- and 2,4,6-trichloro-pyrimidines¹⁴⁷ and 2-methylthiopyrimidines¹³⁶ are readily alkylated by primary alkyl Grignard reagents (equations 153–155). The Reformatsky reagent and homologs couple with 2- and 4-halopyrimidines in the presence of Pd complexes (equation 156).^{143,148} Lower reactivities of halogens β to nitrogen are noted in all of these cases.





6-Chloro-¹⁴⁹ and 6-methylthio-purine¹⁵⁰ derivatives undergo Ni-catalyzed coupling reactions with primary alkyl Grignard reagents; the reactions have been applied to the synthesis of 6-alkylpurine nucleosides (equation 157). Similarly, 8-alkyl- and 8-allyl-adenosine derivatives are obtained by Pd-catalyzed Grignard coupling reactions of 8-bromoadenosines (equation 158).^{151,152}



2.2.9 COUPLING OF C(sp³)-ORGANOMETALLICS WITH ACYL CHLORIDES AND RELATED COMPOUNDS

This section describes ketone synthesis by transition metal catalyzed cross-coupling of alkyl metal reagents with acyl chlorides and related compounds. Carboxylic acids react with excess amounts of alkyl Grignard reagents in the presence of a Ni catalyst to form unsymmetrical ketones (equation 159).¹⁵³ Acyl chlorides readily couple with primary alkyl Grignard reagents in the presence of Fe(acac)₃ as catalyst even in the presence of functional groups, such as ester and cyano groups, susceptible to Grignard reagents (equation 160).^{154,155} Acyl chlorides also undergo Pd-catalyzed coupling with primary and secondary alkylzinc reagents (equations 161 and 162),^{156,157} with the Reformatsky reagents (equation 163)¹⁵⁸ and homologs (equation 164),^{157,159} with alkyltins (equation 165)^{160–162} and with dialkylmercurials;¹⁶³ coupling of allyl- and benzyl-tins with acyl chloride is also catalyzed by a Rh complex (equation 166).¹⁶⁴ Acyl chlorides and acid anhydrides also couple with tetraalkylaluminates in the presence of CuCl (equation 167).¹⁶⁵ While chloroformates undergo Pd-catalyzed coupling with alkylzinc reagents to form esters,¹⁵⁶ a chlorothioformate (carbonochloridothioate) is transformed into unsymmetrical ketones by a stepwise Grignard coupling; the first step is catalyzed by Ni and the second step by Fe complexes, secondary alkyl Grignard reagents being applicable in the second step (equation 168).¹⁶⁶ Finally, an imidoyl chloride couples with primary alkyl Grignard reagents in the presence of PdCl₂(DPPF) (equation 169).¹⁶⁷





THF 52%

2.2.10 COUPLING OF C(sp²)-ORGANOMETALLICS WITH C(sp³) HALIDES AND RELATED COMPOUNDS

Coupling reactions between $C(sp^2)$ -organometallics, mostly Grignard reagents, and $C(sp^3)$ halides and related compounds are usually achieved with Cu catalysts. Primary alkyl halides react smoothly, secondary alkyls are applicable in only a few cases, and no successful coupling has been reported for tertiary alkyl halides. Allylic halides and related compounds are treated separately, since there are important regiochemical problems.

2.2.10.1 Coupling between C(sp²)-Organometallics and Primary Alkyl Halides

Primary alkyl bromides and iodides and benzyl chlorides smoothly couple with aryl Grignard reagents in the presence of Li₂CuCl₄¹⁶⁸ or CuCl together with TMEDA (equation 170).¹⁶⁹ Cross-coupling of aryl Grignard reagents with primary alkyl iodides are also reported to be catalyzed by PdCl₂(DPPF).¹⁷⁰ In the presence of NiCl₂(DPPP), 3-butenyl bromide couples with the phenyl Grignard reagent to form mainly 3-phenyl-1-butene *via* primary to secondary isomerization (equation 171).¹⁷¹ Bromohydrin undergoes Cu-catalyzed coupling with aryl Grignard reagents to give arylethanols (equation 172).¹⁷² α,ω -Dibromoalkanes afford monoarylation products selectively with aryl Grignard reagents in the presence of Li₂CuCl₄ (equation 173),¹⁷³ while double coupling is also possible.^{174,175}

Vinyl, 1-alkenyl and 2-alkenyl Grignard reagents couple with alkyl bromides or iodides in the presence of Cul^{34,176,177} or AgNO₃²⁴ the configuration of the alkenyl Grignard reagent is essentially





retained (equations 174–177). Alkenylboranes couple with benzyl bromides in the presence of $Pd(PPh_3)_4$ and NaOH also with retention of configuration (equation 178).¹⁷⁸ The 2-butadienyl Grignard reagent undergoes Cu-catalyzed couplings with alkyl bromides and iodides even in the presence of functional groups such as ester and cyano groups (equation 179); it may be noted that aromatic iodides are inert under these conditions (equation 180).¹⁷⁹




Arylzinc reagents couple with ethyl bromoacetate¹⁸⁰ and with bromoacetonitrile¹⁸¹ in the presence of Ni complexes (equation 181).



2.2.10.2 Coupling between C(sp²)-Organometallics and Secondary Alkyl Halides

Coupling reactions between vinyl Grignard reagents and 2-oxy secondary alkyl bromides are reported to be catalyzed by Cu¹⁸² or Fe salt (equations 182 and 183).¹⁸³



2.2.10.3 Coupling between C(sp²)-Organometallics and C(sp³)-Oxygen Compounds

Primary alkyl tosylates are more reactive than halides in the Cu-catalyzed coupling with Grignard reagents.¹⁸⁴ Aryl,¹⁸⁴ alkenyl,¹⁷⁶ silylalkenyl³⁴ and 2-butadienyl¹⁸⁵ Grignard reagents have been used successfully for the coupling in the presence of CuI or Li₂CuCl₄; the coupling of optically active tosylates¹⁸⁶ and ditosylates¹⁸⁷ having chiral center(s) at the 2-position may also be noted (equations 184–186).

Ring opening of epoxides with Grignard reagents is greatly promoted by Cu catalysts;¹⁸⁸ terminal epoxides and epoxycyclohexane are readily cleaved by aryl,^{188,189} and alkenyl^{188,190,191} Grignard reagents (equations 187–189) Oxetane also undergoes similar Cu-catalyzed coupling with phenyllithium¹⁹² and phenyl Grignard reagent (equation 190).¹⁸⁸ β-Propiolactone reacts with the phenyl Grignard reagent in the presence of CuBr to form β-phenylpropionic acid via cleavage of the $C(sp^3)$ —O bond (equation 191).¹⁹³





2.2.10.4 Coupling of C(sp²)-Organometallics with Allylic Halides and Related Compounds

The most fundamental and important problem in coupling of allylic compounds with organometallic reagents is the regio- and stereo-control, as shown by the following general scheme (equation 192).¹⁹⁴



Allylic halides, alcohols, ethers, acetates, lactones, phosphates, epoxides, sulfides, sulfonium salts, selenides and ammonium salts undergo transition metal catalyzed coupling reactions with $C(sp^2)$ —Li, —Mg, —B, —Al, —Sn, —Zr, —Cd and —Hg reagents. Table 1 summarizes the allylic leaving groups, alkenyl and aryl metallic reagents, catalytically active metals and references and Table 2 the regio- and stereo-chemical aspects.

Allylic leaving group	C(sp ²)-metallics ^a	Catalyst metal	Ref.
Cl, Br	<i>B</i> X₂	Pd Cu	178, 195, 205 196, 197
Cl, Br	AlR ₂	Pd	198200
Cl, Br	SnR ₃	Pd	201
Cl, Br	SiF5 ²⁻	Pd	203
Cl, Br	ZrCp ₂ Cl	Pd	202
Cl, Br	HgCl	Pd	204
Cl, Br Cl, Br Cl, Br Cl, Br OH	ArMgX ArSnR3 ArHgCl RCOSnR3 ArMgX	Ni, Pd Pd Pd Pd Ni	219 201, 208 218, 220 216 80, 221
OR	BX,	Pd	206
OR OSiR ₃	ArMgX ArMgX	Ni Ni, Pd	80, 221 219
OCOR	MgX	Cu	212, 213
OCOR	AlR ₂	Pd	199, 210
OCOR	SnR ₃	Pd	211, 215
OCOR OCOR OCOR OCOR OCOR	ArMgX ArZnX ArCdX ArAlR ₂ ArSnR ₃	Cu Pd Pd Pd Pd	209, 212, 213 199, 210 199 199 215
OCO ₂ Me	SnR ₃	Pd	1 99
OP(O)(OR) ₂	MgX	Cu	223
OP(O)(OR) ₂	AlR ₂	Pd	215
OP(O)(OR) ₂ OP(O)(OR) ₂	ArMgX ArSnR ₃	Cu Pd	223 215
Epoxide	Li	Cu	214
Epoxide	MgX	Cu	214
Epoxide	BX2	Pd	207
Epoxide SH SR	ArSnR3 ArMgX ArMgX	Pd Ni Ni	211 80 224
$SR_2^+ X^-$	MgX	Cu	225
$sR_2^+ x^-$	ArMgX	Cu	225
SeR	ArMgX	Ni	79
$NR_3^+ X^-$	MgX	Cu	226
$NR_3^+ X^-$	ArMgX	Cu	226

Table 1 Coupling of $C(sp^2)$ -Organometallics with Allylic Halides and Related Compounds

Allyl–X	$C(sp^2)-M$	Catalyst	Ref.
Alkenylmetallics			
Cl			
t t			
25 75	G		016
23 73 90 10	Sn Zr	Pd(DBA)2/PPh3 [PdCl(C3H3)]2/PPh3	202
8 92	Zr	$[PdCl(C_3H_5)]_2/MA^a$	202
\searrow			
CI			
t t			
0 100	5:		202
25 75	Sn	Pd(DBA) ₂ /PPh ₃	203
C1			
K I			
0 100	Al	Pd(PPh ₃) ₄	198-200
0 100	Sn	PdCl ₂ (MeCN) ₂ /PPh ₃	201
MeO ₂ C Cl			
† †			
0 100	Sn	PdCl ₂ (MeCN) ₂ /PPh ₃	201
NC			
t t			
0 100	Sn	PdCl ₂ (MeCN) ₂ /PPh ₃	201
Ph OPh			
¥ ¥			
0 100	[B]	Pd(PPh ₃) ₄	206
OSiMe ₃			
R			
ĪĪ			
0 100	[Al]	Pd(PPh ₃) ₄	199
OAc			
 R			
+ +			
0 100	[A1]	$Pd(PPh_3)_4$	199

Table 2 Regioselectivity in Coupling of $C(sp^2)$ -metallics with Allylic Halides and Related Compounds

Allyl–X	C(sp ²)–M	Catalyst	Ref.
R^1 OP(O)(O) R^2	Et) ₂		
0 100 0 100	[Mg] [A1]	CuBr Pd(PPh ₃) ₄	223 199
ŠMe ₂ Br	-		
35 65 +	[Mg]	CuBr	225
NEt ₃ Br ⁻			
0 100	[Mg]	CuBr	226
26 74	В	Pd(DBA) ₂	207
100 0 100 0	Li Mg	CuBr CuBr	214 214
	Sn	PdCl ₂ (MeCN) ₂	211
† †			
100 0	Mg	CuI	213
Arylmetallics			
75 25 18 82	Mg Mg	NiCl ₂ (DPPF) PdCl ₂ (DPPF)	219 219



Table 2 (continued)				
Allyl-X		$C(sp^2)-M$	Catalyst	Ref.
\checkmark	OSiEt ₃			
t	1			
88	12	Mg	NiCl ₂ (DPPF)	219
33	67	Mg	NiCl ₂ (PPh ₃) ₂	219
4	96	Mg	PdCl ₂ (DPPF)	219
10	90	Mg	Pd(PPh ₃) ₄	219
	N			
4	4			
81	19	Mg	NiCl ₂ (DPPF)	219
9 Dhs		Mg	Paci ₂ (DPPP)	219
	Olvie			
Ī	Ť			
0	100	Mg	$NiCl_2(PPh_3)_2$	222
\checkmark	∕ ^{OAc}			
t	1			
52	48	[Mg]	Pd(PPh ₃) ₄	1 99
23	77	[Zn]	Pd(PPh ₃) ₄	199
23	77	[Cd]	Pd(PPh ₃) ₄	199
38	62	[Al]	Pd(PPh ₃) ₄	199
54	46	[Zr]	Pd(PPh ₃) ₄	199
	OAc			
R				
Ť	Ī			
0	100	[M g]	Li ₂ CuCl ₄	209
\checkmark	∕ ^{SMe}			
t	t			
32	68	[Mg]	$NiCl_2(PPh_3)_2$	224
\searrow	SMe			
t t	t			
36	64	[M g]	NiCl ₂ (PPh ₃) ₂	224
		-		

Allyl-X	$C(sp^2)-M$	Catalyst	Ref.
SMe ₂	Br ⁻		
5 95 SePh	[Mg]	CuBr	225
i i 39 61	[Mg] Br	NiCl ₂ (PPh ₃) ₂	79
	[Mg]	CuBr	226
$\frac{1}{98} \frac{1}{2}$	[Sn]	PdCl ₂ (MeCN) ₂	211
Ŏ	(n = 1, 2 or 3)		
100 0 OEt OEt	[Mg]	CuI	214, 215
$\begin{array}{ccc} 100 & 0 \\ (E/Z = 42/58) \end{array}$	[M g]	CuBr	217

 Table 2 (continued)

^a MA = maleic anhydride.

Some other characteristic features, in focus on the stereochemical aspects, are described hereafter. Allylic chlorides and bromides couple with alkenyl–BR₂,^{178,195–197} –AIR₂,^{198–200} –SnR₃,²⁰¹ –ZrCp₂Cl,²⁰² –SiF₅^{2-,203} and –HgCl²⁰⁴ compounds in the presence of Pd or Cu catalysts to form 1,4-dienes with retention of stereochemistry of alkenylmetals (equations 193–200); in the Si and Hg cases, only allylic chlorides having a terminal alkene react with the complete allylic transposition. An intramolecular coupling of alkenyl–Al reagents the reactivity of allylic substrates depends on the leaving group in the order Cl, OAc > OAlMe₂ > OP(O)(OEt)₂ >> OSiR₃.¹⁹⁹ In contrast, the reverse order OPh >> OAc, Cl is found in the Pd-catalyzed coupling of alkenyl–B reagents, the alkene geometry being retained (equation 202); the reaction condition may be compared with that in the reaction occurring in equation (193), where the hydroxide ion is needed as the essential additive.²⁰⁶ The stereochemistry of alkenyl–B reagents is retained also in the Pd-catalyzed coupling with epoxides.²⁰⁷



Remarkably, alkene stereochemistry in allylic chlorides is completely retained in the Pd-catalyzed coupling with alkenyl-Al (equations 195 and 196)²⁰⁰ and aryl-Sn reagents (equation 203),²⁰¹ whereas stereochemical scrambling is also encountered in some cases (equation 204).²⁰⁸ Retention of alkene stereochemistry is also observed in the Cu-catalyzed Grignard coupling of allylic acetates (equation 205).²⁰⁹

The Pd-catalyzed coupling reactions of cyclic allyl chlorides,²⁰¹ carboxylates²¹⁰ and epoxides²¹¹ with alkenyl–Al, alkenyl–Sn and aryl–Sn reagents proceed with inversion of configuration of the allylic sp^3 carbon centers (equations 206–208).



Some other synthetically interesting reactions are listed below. Thus, vinyl lactones undergo ringopening and coupling with Grignard reagents in the presence of CuI to form β , γ -, γ , δ - or δ , ε -unsaturated carboxylic acids (equation 209),^{212,213} and isoprene monoepoxide couples with alkenyl–Grignard reagents²¹⁴ or alkenyl–Sn reagents²¹¹ to form terpenoid allylic alcohols (equation 210). Pd-catalyzed coupling of 1-ethoxyvinyl–²¹⁵ and acyl–Sn²¹⁶ reagents with allylic acetates, carbonates, phosphates or chlorides forms unsaturated vinyl ethers and ketones, respectively (equations 211 and 212). The formation of alkenyl ethers is also attained by Cu-catalyzed Grignard coupling with an allylic acetal (equation 213).²¹⁷ Finally, allyluridines can be obtained by Pd-catalyzed allylation of the corresponding mercury intermediates (equation 214).²¹⁸



2.2.11 REFERENCES

- 1. M. Tamura and J. K. Kochi, J. Am. Chem. Soc., 1971, 93, 1487.
- 2. M. Tamura and J. K. Kochi, Synthesis, 1971, 303.
- 3. K. Tamao, K. Sumitani and M. Kumada, J. Am. Chem. Soc., 1972, 94, 4374.
- 4. R. J. P. Corriu and J. P. Masse, J. Chem. Soc., Chem. Commun., 1972, 144.
- 5. M. Yamamura, I. Moritani and S.-I. Murahashi, J. Organomet. Chem., 1975, 91, C39.
- 6. M. Tamura and J. K. Kochi, J. Am. Chem. Soc., 1971, 93, 1485.
- C. L. Raston and G. Salem, in 'The Chemistry of the Metal-Carbon Bond', ed. F. R. Hartley and S. Patai, 7. Wiley, New York, 1987, vol. 4, p. 159.
- 8. M. Kumada, Pure Appl. Chem., 1980, 52, 669.
- 9. P. W. Jolly, in 'Comprehensive Organometallic Chemistry', ed. G. Wilkinson, F. G. A. Stone and E. W. Abel, Pergamon Press, Oxford, 1982, vol. 8, p. 713.
- 10. K. Tamao and M. Kumada, in 'The Chemistry of the Metal-Carbon Bond', ed. F. R. Hartley and S. Patai, Wiley, New York, 1987, vol. 4, p. 819.
- 11. F. Naso and G. Marchese, in 'The Chemistry of Functional Groups, Supplement D: The Chemistry of Halides, Pseudohalides and Azides', ed. S. Patai and Z. Rappoport, Wiley, New York, 1983, p. 1353.
- 12. E. Negishi, Acc. Chem. Res., 1982, 15, 340.
- 13. B. M. Trost and T. R. Verhoeven, in 'Comprehensive Organometallic Chemistry', ed. G. Wilkinson, F. G. A. Stone and E. W. Abel, Pergamon Press, Oxford, 1982, vol. 8, p. 799.
- 14. I. P. Beletskaya, J. Organomet. Chem., 1983, 250, 551.
- 15. R. F. Heck, 'Palladium Reagents in Organic Syntheses', Academic Press, New York, 1985.
- 16. J. K. Stille, Angew. Chem., Int. Ed. Engl., 1986, 25, 508.
- 17. W. J. Scott and J. E. McMurry, Acc. Chem. Res., 1988, 21, 47.
- 18. E. Erdik, Tetrahedron, 1984, 40, 641.
- 19. T. Hayashi and M. Kumada, in 'Asymmetric Synthesis', ed. J. D. Morrison, Academic Press, New York, 1985, vol. 5, p. 147.
- 20. R. S. Smith and J. K. Kochi, J. Org. Chem., 1976, 41, 502.
- 21. K. Tamao, M. Zembayashi, Y. Kiso and M. Kumada, J. Organomet. Chem., 1973, 55, C91.
- 22. S.-I. Murahashi, M. Yamamura, K. Yanagisawa, N. Mita and K. Kondo, J. Org. Chem., 1979, 44, 2408.
- 23. S. Murahashi, T. Naota and Y. Tanigawa, Org. Synth., 1984, 62, 39.
- 24. M. Tamura and J. K. Kochi, J. Am. Chem. Soc., 1971, 93, 1483.
- 25. K. Tamao, K. Sumitani, Y. Kiso, M. Zembayashi, A. Fujioka, S. Kodama, I. Nakajima, A. Minato and M. Kumada, Bull. Chem. Soc. Jpn., 1976, 49, 1958.
 M. Kumada, K. Tamao and K. Sumitani, Org. Synth., 1978, 58, 127 or Org. Synth. Coll. Vol., 1988, 6, 409.
- 27. V. Ratovelomanana and G. Linstrumelle, Synth. Commun., 1984, 14, 179.
- 28. A. Hosomi, M. Saito and H. Sakurai, Tetrahedron Lett., 1979, 429.
- 29. G. Consiglio, D. A. von Bézard, F. Morandini and P. Pino, Helv. Chim. Acta, 1978, 61, 1703.
- 30. T. Hayashi, K. Kabeta, I. Hamachi and M. Kumada, Tetrahedron Lett., 1983, 24, 2865.
- 31. K. Tamao, N. Ishida and M. Kumada, J. Org. Chem., 1983, 48, 2120.
- 32. R. W. Hoffmann and B. Landmann, Tetrahedron Lett., 1983, 24, 3209.
- 33. T. Hayashi, M. Konishi and M. Kumada, Tetrahedron Lett., 1979, 1871.
- 34. C. Huynh and G. Linstrumelle, Tetrahedron Lett., 1979, 1073.
- 35. H. G. Richey, Jr., C. W. Wilkins, Jr., B. S. Brown and R. E. Moore, Tetrahedron Lett., 1976, 723.
- 36. K. Kondo and S. Murahashi, Tetrahedron Lett., 1979, 1237.
- 37. R. Rossi and A. Carpita, Tetrahedron Lett., 1986, 27, 2529.
- 38. H. M. Walborsky and R. B. Banks, J. Org. Chem., 1981, 46, 5074.
- 39. G. Solladié and R. G. Zimmermann, Tetrahedron Lett., 1984, 25, 5769.
- 40. A. M. Caporusso, L. Lardicci and F. Da Settimo, Tetrahedron Lett., 1986, 27, 1067.
- 41. A. M. Caporusso, F. Da Settimo and L. Lardicci, Tetrahedron Lett., 1985, 26, 5101.
- 42. A. Commercon, J. F. Normant and J. Villieras, J. Organomet. Chem., 1977, 128, 1.
- 43. H. P. Dang and G. Linstrumelle, Tetrahedron Lett., 1978, 191.
- 44. E. Negishi, F.-T. Luo and C. L. Rand, Tetrahedron Lett., 1982, 23, 27.
- 45. J. G. Duboudin, B. Jousseaume and A. Bonakdar, J. Organomet. Chem., 1979, 168, 227.
- M. Malacria and K. P. C. Vollhardt, J. Org. Chem., 1984, 49, 5010.
 T. Hayashi, M. Konishi, Y. Kobori, M. Kumada, T. Higuchi and K. Hirotsu, J. Am. Chem. Soc., 1984, 106, 158.
- 48. V. Fiandanese, G. Marchese, F. Naso and L. Ronzini, J. Chem. Soc., Chem. Commun., 1982, 647.
- 49. V. Fiandanese, G. Miccoli, F. Naso and L. Ronzini, J. Organomet. Chem., 1986, 312, 343.
- 50. T. Hayashi, M. Konishi, K. Yokota and M. Kumada, Chem. Lett., 1980, 767.
- 51. E. Negishi, L. F. Valente and M. Kobayashi, J. Am. Chem. Soc., 1980, 102, 3298.
- 52. E. Negishi, H. Matsushita and N. Okukado, Tetrahedron Lett., 1981, 22, 2715.
- 53. E. Negishi, H. Matsushita, M. Kobayashi and C. L. Rand, Tetrahedron Lett., 1983, 24, 3823.
- 54. M. Kobayashi and E. Negishi, J. Org. Chem., 1980, 45, 5223.
- 55. C. L. Rand, D. E. Van Horn, M. W. Moore and E. Negishi, J. Org. Chem., 1981, 46, 4093.
- 56. S. Hyuga, N. Yamashina, S. Hara and A. Suzuki, Chem. Lett., 1988, 809.
- 57. E. Piers, M. Jean and P. S. Marrs, Tetrahedron Lett., 1987, 28, 5075.
- 58. J.-F. Fauvarque and A. Jutand, J. Organomet. Chem., 1981, 209, 109.
- 59. A. A. Millard and M. W. Rathke, J. Am. Chem. Soc., 1977, 99, 4833.
- 60. P. Albaugh-Robertson and J. A. Katzenellenbogen, J. Org. Chem., 1983, 48, 5288.
- Y. Tamaru, H. Ochiai, T. Nakamura and Z. Yoshida, Angew. Chem., Int. Ed. Engl., 1987, 26, 1157.
 Y. Yamamoto, S. Hatsuya and J. Yamada, J. Chem. Soc., Chem. Commun., 1988, 86.
- 63. H. Takayama and T. Suzuki, J. Chem. Soc., Chem. Commun., 1988, 1044.

- 64. T. Kitazume and N. Ishikawa, J. Am. Chem. Soc., 1985, 107, 5186.
- 65. E. Wenkert, E. L. Michelotti, C. S. Swindell and M. Tingoli, J. Org. Chem., 1984, 49, 4894.
- 66. S. Wadman, R. Whitby, C. Yeates, P. J. Kocienski and K. Cooper, J. Chem. Soc., Chem. Commun., 1987, 241,
- 67. R. Whitby, C. Yeates, P. J. Kocienski and G. Costello, J. Chem. Soc., Chem. Commun., 1987, 429.
- 68. P. J. Kocienski, N. J. Dixon and S. Wadman, Tetrahedron Lett., 1988, 29, 2353.
- 69. T. Hayashi, Y. Katsuro and M. Kumada, Tetrahedron Lett., 1980, 21, 3915.
- 70. G. D. Crouse and L. A. Paquette, J. Org. Chem., 1981, 46, 4272.
- 71. T. Hayashi, T. Fujiwa, Y. Okamoto, Y. Katsuro and M. Kumada, Synthesis, 1981, 1001.
- 72. C. Sahlberg, A. Quader and A. Claesson, Tetrahedron Lett., 1983, 24, 5137.
- 73. K. Takai, M. Sato, K. Oshima and H. Nozaki, Bull. Chem. Soc. Jpn., 1984, 57, 108. 74. K. Fugami, K. Oshima and K. Utimoto, Chem. Lett., 1987, 2203.
- 75. W. J. Scott and J. K. Stille, J. Am. Chem. Soc., 1986, 108, 3033.
- 76. K. Itoh, T. Yogo and Y. Ishii, Chem. Lett., 1977, 103.
- 77. Y. Abe, M. Sato, H. Goto, R. Sugawara, E. Takahashi and T. Kato, *Chem. Pharm. Bull.*, 1983, **31**, 4346. 78. T. Fujisawa, T. Sato, Y. Gotoh, M. Kawashima and T. Kawara, *Bull. Chem. Soc. Jpn.*, 1982, **55**, 3555.
- 79. H. Okamura, M. Miura and H. Takei, Tetrahedron Lett., 1979, 43.
- 80. H. Okamura, M. Miura, K. Kosugi and H. Takei, Tetrahedron Lett., 1980, 21, 87.
- 81. E. Wenkert, J. B. Fernandes, E. L. Michelotti and C. S. Swindell, Synthesis, 1983, 701.
- 82. H. Takei, H. Sugimura, M. Miura and H. Okamura, Chem. Lett., 1980, 1209.
- 83. B. M. Trost and P. L. Ornstein, J. Org. Chem., 1982, 47, 748.
- 84. B. M. Trost and A. C. Lavoie, J. Am. Chem. Soc., 1983, 105, 5075.
- 85. Y. Ikeda, J. Ukai, N. Ikeda and H. Yamamoto, Tetrahedron Lett., 1984, 25, 5177.
- 86. J. Ukai, Y. Ikeda, N. Ikeda and H. Yamamoto, Tetrahedron Lett., 1984, 25, 5173.
- 87. V. Fiandanese, G. Marchese, G. Mascolo, F. Naso and L. Ronzini, Tetrahedron Lett., 1988, 29, 3705.
- 88. K. Furuta, Y. Ikeda, N. Meguriya, N. Ikeda and H. Yamamoto, Bull. Chem. Soc. Jpn., 1984, 57, 2781.
- 89. T. Ishiyama, N. Miyaura and A. Suzuki, Chem. Lett., 1987, 25.
- 90. M. Tiecco, M. Tingoli and E. Wenkert, J. Org. Chem., 1985, 50, 3828.
- 91. E. Wenkert, T. W. Ferreira and E. L. Michelotti, J. Chem. Soc., Chem. Commun., 1979, 637.
- 92. E. Wenkert, M. H. Leftin and E. L. Michelotti, J. Chem. Soc., Chem. Commun., 1984, 617.
- 93. J.-L. Fabre, M. Julia and J.-N. Verpeaux, Tetrahedron Lett., 1982, 23, 2469.
- 94. J.-L. Fabre, M. Julia and J.-N. Verpeaux, Bull. Soc. Chim. Fr., 1985, 772.
- 95. E. Alvarez, T. Cuvigny, C. Hervé du Penhoat and M. Julia, Tetrahedron, 1988, 44, 111, 119.
- 96. A. Minato, K. Suzuki and K. Tamao, J. Am. Chem. Soc., 1987, 109, 1257.
- 97. E. Wenkert and T. W. Ferreira, J. Chem. Soc., Chem. Commun., 1982, 840.
- 98. V. Ratovelomanana and G. Linstrumelle, Tetrahedron Lett., 1981, 22, 315.
- 99. K. Tamao, M. Zembayashi and M. Kumada, Chem. Lett., 1976, 1237.
- K. Tamao, M. Zembayashi and M. Kumada, Chem. Lett., 1976, 1239. 100.
- V. Fiandanese, G. Marchese, F. Naso and L. Ronzini, J. Chem. Soc., Perkin Trans. 1, 1985, 1115. 101.
- 102. J.-F. Fauvarque and A. Jutand, Bull. Soc. Chim. Fr., 1976, 765.
- 103. E. Negishi, A. O. King and N. Okukado, J. Org. Chem., 1977, 42, 1821.
- 104. K. C. Eapen and C. Tamborski, J. Org. Chem., 1988, 53, 5564.
- Y. Ito, M. Nakatsuka and T. Saegusa, J. Am. Chem. Soc., 1981, 103, 476. 105.
- 106. K. Tamao, Y. Kiso, K. Sumitani and M. Kumada, J. Am. Chem. Soc., 1972, 94, 9268.
- 107. K. Tamao, S. Kodama, T. Nakatsuka, Y. Kiso and M. Kumada, J. Am. Chem. Soc., 1975, 97, 4405.
- 108. A. Sekiya and N. Ishikawa, J. Organomet. Chem., 1976, 118, 349.
- Y. Kiso, Ph.D Thesis, Kyoto University, 1974. 109.
- 110. G. Consiglio, F. Morandini and O. Piccolo, Tetrahedron, 1983, 39, 2699.
- 111. Y. Kiso, K. Tamao and M. Kumada, J. Organomet. Chem., 1973, 50, C12.
- 112. D. Milstein and J. K. Stille, J. Am. Chem. Soc., 1979, 101, 4992.
- 113. M. Kosugi, T. Sumiya, T. Ogata, H. Sano and T. Migita, Chem. Lett., 1984, 1225.
- 114. M. Kosugi, K. Sasazawa, Y. Shimizu and T. Migita, Chem. Lett., 1977, 301.
- 115. J.-P. Quintard, B. Elissondo and M. Pereyre, J. Org. Chem., 1983, 48, 1559.
- 116. W. J. Scott, J. Chem. Soc., Chem. Commun., 1987, 1755.
- J.-F. Fauvarque and A. Jutand, J. Organomet. Chem., 1979, 177, 273.
 J.-F. Fauvarque and A. Jutand, J. Organomet. Chem., 1977, 132, C17.
- 119. I. Kuwajima and H. Urabe, J. Am. Chem. Soc., 1982, 104, 6831.
- 120. M. Kosugi, M. Ishiguro, Y. Negishi, H. Sano and T. Migita, Chem. Lett., 1984, 1511.
- M. Uno, K. Seto and S. Takahashi, J. Chem. Soc., Chem. Commun., 1984, 932.
 T. Hayashi, Y. Katsuro, Y. Okamoto and M. Kumada, Tetrahedron Lett., 1981, 22, 4449.
- 123. A. M. Echavarren and J. K. Stille, J. Am. Chem. Soc., 1987, 109, 5478.
- 124. Q.-Y. Chen and Y.-B. He, Tetrahedron Lett., 1987, 28, 2387.
- S. Aoki, T. Fujimura, E. Nakamura and I. Kuwajima, J. Am. Chem. Soc., 1988, 110, 3296. 125.
- 126. E. Wenkert, T. W. Ferreira and E. L. Michelotti, J. Chem. Soc., Chem. Commun., 1979, 637.
- 127. G. S. Reddy and W. Tam, Organometallics, 1984, 3, 630.
- 128. K. C. Eapen, S. S. Dua and C. Tamborski, J. Org. Chem., 1984, 49, 478.
- 129. A. Minato, K. Tamao, T. Hayashi, K. Suzuki and M. Kumada, Tetrahedron Lett., 1980, 21, 845.
- 130. A. Minato, K. Tamao, K. Suzuki and M. Kumada, J. Org. Chem., to be submitted.
- 131. A. Minato and S. Ozasa, unpublished results, 1988.
- 132. L. N. Pridgen, J. Org. Chem., 1982, 47, 4319.
- 133. M. Tiecco, L. Testaferri, M. Tingoli, D. Chianelli and E. Wenkert, Tetrahedron Lett., 1982, 23, 4629.
- 134. W. G. Peet and W. Tam, J. Chem. Soc., Chem. Commun., 1983, 853.
- 135. D. E. Bergstrom and P. A. Reddy, J. Heterocycl. Chem., 1983, 20, 469.
- 136. H. Takei, M. Miura, H. Sugimura and H. Okamura, Chem. Lett., 1979, 1447.

- 137. K. Tamao, S. Kodama, I. Nakajima, M. Kumada, A. Minato and K. Suzuki, Tetrahedron, 1982, 38, 3347.
- 138. C. V. Pham, H. B. Mark, Jr. and H. Zimmer, Synth. Commun., 1986, 16, 689.
- 139. A. Minato, K. Tamao, K. Suzuki and M. Kumada, Tetrahedron Lett., 1980, 21, 4057.
- 140. L. N. Pridgen, J. Heterocycl. Chem., 1975, 12, 443.
- 141. O. Piccolo and T. Martinengo, Synth. Commun., 1981, 11, 497.
- E. Negishi, F.-T. Luo, R. Frisbee and H. Matsushita, *Heterocycles*, 1982, 18, 117.
 T. Sakamoto, S. Nishimura, Y. Kondo and H. Yamanaka, *Synthesis*, 1988, 485.
- 144. E. Thorsett and F. R. Stermitz, J. Heterocycl. Chem., 1973, 10, 243.
- 145. F. Babudri, S. Florio, L. Ronzini and M. Aresta, Tetrahedron, 1983, 39, 1515.
- 146. A. Ohsawa, Y. Abe and H. Igeta, Chem. Pharm. Bull., 1978, 26, 2550.
- 147. H. Yamanaka, K. Edo, F. Shoji, S. Konno, T. Sakamoto and M. Mizugami, Chem. Pharm. Bull., 1978, 26, 2160.
- 148. H. Yamanaka, M. An-naka, Y. Kondo and T. Sakamoto, Chem. Pharm. Bull., 1985, 33, 4309.
- 149. D. E. Bergstrom and P. Anantha Reday, Tetrahedron Lett., 1982, 23, 4191.
- 150. H. Sugimura and H. Takei, Bull. Chem. Soc. Jpn., 1985, 58, 664.
- 151. D. E. Bergstrom, Nucleosides and Nucleotides, 1982, 1, 1.
- 152. N. Công-Danh, J.-P. Beaucourt and L. Pichat, Tetrahedron Lett., 1979, 3159.
- 153. V. Fiandanese, G. Marchese and L. Ronzini, Tetrahedron Lett., 1983, 24, 3677
- 154. V. Fiandanese, G. Marchese, V. Martina and L. Ronzini, Tetrahedron Lett., 1984, 25, 4805.
- 155. C. Cardellicchio, V. Fiandanese, G. Marchese and L. Ronzini, Tetrahedron Lett., 1987, 28, 2053.
- 156. E. Negishi, V. Bagheri, S. Chatterjee, F.-T. Luo, J. A. Miller and A. T. Stoll, Tetrahedron Lett., 1983, 24, 5181.
- 157. Y. Tamaru, H. Ochiai, F. Sanda and Z. Yoshida, Tetrahedron Lett., 1985, 26, 5529.
- T. Sato, T. Itoh and T. Fujisawa, Chem. Lett., 1982, 1559.
 Y. Tamaru, H. Ochiai, T. Nakamura and Z. Yoshida, Org. Synth., 1988, 67, 98.
- 160. M. Kosugi, Y. Shimizu and T. Migita, Chem. Lett., 1977, 1423.
- 161. D. Milstein and J. K. Stille, J. Am. Chem. Soc., 1978, 100, 3636.
- 162. D. Milstein and J. K. Stille, J. Org. Chem., 1979, 44, 1613.
- 163. K. Takagi, T. Okamoto, Y. Sakakibara, A. Ohno, S. Oka and N. Hayama, Chem. Lett., 1975, 951.
- 164. M. Kosugi, Y. Shimizu and T. Migita, J. Organomet. Chem., 1977, 129, C36.
- 165. F. Sato, H. Kodama, Y. Tomuro and M. Sato, Chem. Lett., 1979, 623.
- 166. C. Cardellicchio, V. Fiandanese, G. Marchese and L. Ronzini, Tetrahedron Lett., 1985, 26, 3595.
- 167. Y. Ito, M. Inoue and M. Murakami, Tetrahedron Lett., 1988, 29, 5379.
- 168. J. Novak and C. A. Salemink, Synthesis, 1983, 597.
- 169. K. Onuma and H. Hashimoto, Bull. Chem. Soc. Jpn., 1972, 45, 2582.
- 170. P. L. Castle and D. A. Widdowson, Tetrahedron Lett., 1986, 27, 6013.
- 171. M. Zembayashi, K. Tamao and M. Kumada, Tetrahedron Lett., 1975, 1719.
- 172. J. F. Normant, J. Villieras and F. Scott, Tetrahedron Lett., 1977, 3263.
- 173. L. Friedman and A. Shani, J. Am. Chem. Soc., 1974, 96, 7101.
- 174. M. Kadkhodayan, T. Singleton and F. J. Heldrich, Synth. Commun., 1984, 14, 707.
- 175. J. Nishimura, N. Yamada, Y. Horiguchi, E. Ueda, A. Ohbayashi and A. Oku, Bull. Chem. Soc. Jpn., 1986, 59, 2035.
- 176. F. Derguini-Boumechal and G. Linstrumelle, Tetrahedron Lett., 1976, 3225.
- 177. F. Sato, H. Watanabe, Y. Tanaka, T. Yamaji and M. Sato, Tetrahedron Lett., 1983, 24, 1041.
- 178. N. Miyaura, T. Yano and A. Suzuki, Tetrahedron Lett., 1980, 21, 2865.
- S. Nunomoto, Y. Kawakami and Y. Yamashita, J. Org. Chem., 1983, 48, 1912.
 T. Klingstedt and T. Frejd, Organometallics, 1983, 2, 598.
- 181. T. Frejd and T. Klingstedt, Synthesis, 1987, 40.
- 182. J. F. Normant, T. Mulamba, F. Scott, A. Alexakis and G. Cahiez, Tetrahedron Lett., 1978, 3711.
- 183. U. H. Brinker and L. König, Chem. Ber., 1983, 116, 882.
- 184. G. Fouquet and M. Schlosser, Angew. Chem., Int. Ed. Engl., 1974, 13, 82.
- 185. K. J. Shea and P. Q. Pham, Tetrahedron Lett., 1983, 24, 1003.
- 186. J. Leder, H. Fujioka and Y. Kishi, Tetrahedron Lett., 1983, 24, 1463.
- 187. J. Yamashita, M. Minagawa, A. Sonobe and S. Ohashi, Chem. Lett., 1982, 1409.
- 188. C. Huynh, F. Derguini-Boumechal and G. Linstrumelle, Tetrahedron Lett., 1979, 1503.
- 189. P. Boudjouk, J. S. Kiely and R. Sooriyakumaran, J. Organomet. Chem., 1981, 221, 33.
- 190. L. D. Martin and J. K. Stille, J. Org. Chem., 1982, 47, 3630.
- 191. H. Nishiyama, H. Yokoyama, S. Narimatsu and K. Itoh, Tetrahedron Lett., 1982, 23, 1267.
- 192. J. Millon and G. Linstrumelle, Tetrahedron Lett., 1976, 1095.
- 193. J. F. Normant, A. Alexakis and G. Cahiez, Tetrahedron Lett., 1980, 21, 935.
- 194. R. M. Magid, Tetrahedron, 1980, 36, 1901
- 195. H. Yatagai, Bull. Chem. Soc. Jpn., 1980, 53, 1670.
- M. Hoshi, Y. Masuda and A. Arase, Bull. Chem. Soc. Jpn., 1983, 56, 2855.
 M. Hoshi, Y. Masuda, Y. Nunokawa and A. Arase, Chem. Lett., 1984, 1029.
- 198. E. Negishi and H. Matsushita, Org. Synth., 1984, 62, 31.
- 199. E. Negishi, S. Chatterjee and H. Matsushita, Tetrahedron Lett., 1981, 22, 3737.
- 200. H. Matsushita and E. Negishi, J. Am. Chem. Soc., 1981, 103, 2882.
- 201. F. K. Sheffy, J. P. Godschalx and J. K. Stille, J. Am. Chem. Soc., 1984, 106, 4833.
- 202. Y. Hayashi, M. Riediker, J. S. Temple and J. Schwartz, Tetrahedron Lett., 1981, 22, 2629.
- 203. J. Yoshida, K. Tamao, H. Yamamoto, T. Kakui, T. Uchida and M. Kumada, Organometallics, 1982, 1, 542.
- 204. R. C. Larock, J. C. Bernhardt and R. J. Driggs, J. Organomet. Chem., 1978, 156, 45.
- 205. N. Miyaura, H. Suginome and A. Suzuki, Tetrahedron Lett., 1984, 25, 761
- 206. N. Miyaura, K. Yamada, H. Suginome and A. Suzuki, J. Am. Chem. Soc., 1985, 107, 972.

- 207. N. Miyaura, Y. Tanabe, H. Suginome and A. Suzuki, J. Organomet. Chem., 1982, 233, C13.
- 208. S. Katsumura, S. Fujiwara and S. Isoe, Tetrahedron Lett., 1987, 28, 1191.
- 209. S. Suzuki, M. Shiono and Y. Fujita, Synthesis, 1983, 804.
- 210. H. Matsushita and E. Negishi, J. Chem. Soc., Chem. Commun., 1982, 160.
- A. M. Echavarren, D. R. Tueting and J. K. Stille, J. Am. Chem. Soc., 1988, 110, 4039.
 T. Sato, M. Takeuchi, T. Itoh, M. Kawashima and T. Fujisawa, Tetrahedron Lett., 1981, 22, 1817.
- T. Sato, M. Takeuchi, T. Hon, M. Kawashima and T. Fujisawa, Tetrahedron Lett., 1961, 22, 1817.
 T. Fujisawa, T. Sato, M. Kawashima, K. Naruse and K. Tamai, Tetrahedron Lett., 1982, 23, 3583.
- 213. G. Cahiez, A. Alexakis and J. F. Normant, Synthesis, 1978, 528.
- 215. M. Kosugi, K. Ohashi, K. Akuzawa, T. Kawazoe, H. Sano and T. Migita, Chem. Lett., 1987, 1237.
- 215. M. Kosugi, H. Naka, S. Harada, H. Sano and T. Migita, Chem. Lett., 1987, 1257. 216. M. Kosugi, H. Naka, S. Harada, H. Sano and T. Migita, Chem. Lett., 1987, 1371.
- 210. M. Rosugi, H. Naka, S. Harada, H. Salo and T. Higha, Chem. Lett., 1967, 1971. 217. J. F. Normant, A. Commercon, M. Bourgain and J. Villieras, Tetrahedron Lett., 1975, 3833.
- 218. J. L. Ruth and D. E. Bergstrom, J. Org. Chem., 1978, 43, 2870.
- T. Hayashi, M. Konishi, K.-i. Yokota and M. Kumada, J. Chem. Soc., Chem. Commun., 1981, 313.
- 220. R. F. Heck, J. Am. Chem. Soc., 1968, 90, 5531.
- 221. C. Chuit, H. Felkin, C. Frajerman, G. Rousse and G. Swierczewski, J. Organomet. Chem., 1977, 127, 371.
- 222. H. Sugimura and H. Takei, Chem. Lett., 1984, 1505.
- 223. M. Bourgain-Commercon, J. F. Normant and J. Villieras, J. Chem. Res. (S) 1977, 183: J. Chem. Res. (M), 1977, 2101.
- 224. H. Okamura and H. Takei, Tetrahedron Lett., 1979, 3425.
- 225. Y. Gendreau, J. F. Normant and J. Villieras, J. Organomet. Chem., 1977, 142, 1.
- 226. M. Commercon-Bourgain, J. F. Normant and J. Villieras, C. R. Hebd. Seances Acad. Sci., Ser. C, 1977, 285, 211.

2.3 Coupling Reactions Between *sp*² Carbon Centers

DAVID W. KNIGHT

University of Nottingham, UK

2.3.1 INTRODUCTION	481
2.3.2 COUPLING REACTIONS INVOLVING ALKENES	482
2.3.2.1 Alkene–Alkene Dimerization 2.3.2.2 Crossed Alkene–Alkene Coupling 2.3.2.3 Allene Synthesis by sp ² –sp ² Coupling 2.3.2.4 Alkene–Aryl Coupling 2.3.2.5 Alkene–Heteroaryl Coupling	482 484 491 492 492
2.3.3 ARYL-ARYL COUPLING REACTIONS	499
2.3.3.1 Aryl–Aryl Dimerization 2.3.3.2 Crossed Aryl–Aryl Coupling 2.3.3.3 Intramolecular Aryl–Aryl Coupling	499 501 505
2.3.4 COUPLING REACTIONS INVOLVING HETEROARYLS	509
2.3.4.1 Heteroaryl-Heteroaryl Coupling 2.3.4.2 Aryl-Heteroaryl Coupling	509 512
2.3.5 REFERENCES	516

2.3.1 INTRODUCTION

The scope of this chapter is largely confined to the direct coupling reactions of sp^2 -carbons as outlined in equation (1), where both X and Y are groups or atoms other than hydrogen. In retrosynthetic terms, such processes are usually best regarded as 'illogical' and indeed prior to the early 1970s, with the exception of the classical Ullmann biphenyl synthesis, few examples of such reactions were in general use. During the past 15 years, however, enormous progress has been made in this area, particularly since the introduction of zerovalent palladium and nickel catalysts for carrying out the coupling reactions.

Two important groups of reactions of the type outlined in equation (1) are not included in this chapter, as they are reviewed elsewhere in Comprehensive Organic Synthesis. Firstly, Heck-type processes (Y = H; equation 1) are discussed in Volume 4, Chapter 4.3. A second way to effect the overall process indi-



X, Y \neq H

cated in equation (1) is by a tandem Michael addition-elimination sequence, such as the addition of a vinylcuprate species to methyl β -bromoacrylate; such methods are also covered in Volume 4.

The chapter has been organized so as to give rapid access to information regarding a particular structural type. This inevitably requires the inclusion of extensive cross-referencing as, not surprisingly, many of the coupling methods can and have been applied to targets belonging to more than one of these groups.

2.3.2 COUPLING REACTIONS INVOLVING ALKENES

2.3.2.1 Alkene–Alkene Dimerization

In common with the well-established Ullmann biphenyl synthesis, many methods for alkene dimerization feature the use of a copper species, either formed in situ using a copper salt (and often of an undefined nature) or by using a preformed vinylcuprate and a coupling reagent. For example, iodofumarates (1) can be converted into tetraesters (2) simply by using activated copper powder, 1 although later work has revealed that copper(I) triflate and saturated aqueous ammonia in acetone is a more effective reagent mixture.² The couplings are highly stereoselective and work almost equally well with iodomaleates. A general method for coupling vinyl halides consists of conversion to the corresponding lithium divinvlcuprate followed by treatment with molecular oxygen (Scheme 1).³ The method is general and usually high yielding and is also effective in coupling mixed vinylmagnesiocuprate species. The latter type of intermediate (e.g. 3) also undergoes efficient dimerization when treated with dilithium tetrachlorocuprate (Li₂CuCl₄ from CuCl₂ + 2LiCl) resulting, in this particular example, in an expedient and stereospecific approach to bis(allylsilanes) (4).⁴ The conversion (3) \rightarrow (4) represents a general type of oxidative coupling which is also applicable to a wide range of vinyllithium and vinyl Grignard species, and which can be effected using CuCl or CuCl₂ in addition to Li₂CuCl₄, as well as by a variety of cobalt, manganese and chromium salts.⁵ Symmetrical 1,3-butadienes can also be obtained from vinylcopper species and the corresponding vinylsilver(I) salts by a simple thermal decomposition process; in general both the yields and stereochemical retentions are high, even in the preparations of (Z,Z)-1,3butadienes.⁶ Silver salts can also be used to carry out dimerizations of vinyl Grignard reagents, in conjunction with a nitrogen-containing oxidizing agent such as LiNO₃, MeNO₃ or NO₂.⁷ A wide range of simple transition metal halides have also been examined for their suitability as mediators of such coupling reactions. One of the best turned out to be palladium(II) chloride, which at the time it was described was certainly a signal for many of the later spectacular advances in sp^2 -sp²-coupling reactions. Two





Scheme 1



other reagents that have been used in this type of Grignard dimerization are $K_4[Ni_2(CN)_6]^8$ and, rather unexpectedly, thionyl chloride.⁹

Alkenylboranes have also been found to be valuable precursors of 1,3-butadienes and, once again, copper reagents play a crucial role in these procedures. For example, hydroboration of an alkyne using monochloroborane (Scheme 2) leads to a divinylborane, and thence to an (E,E)-1,3-butadiene (5) following treatment with 3 equiv. of methylcopper.¹⁰ Monoalkenyldibromoborane species can be coupled in a similar fashion and both sequences are stereospecific; two limitations, however, are the potential lack of regioselectivity in the first step with unsymmetrical alkynes and the possibility of side reactions between the hydroborating reagent and other functional groups contained in the substituents R¹ and R². Both (*E*)-and (*Z*)-vinylboranes (6) can be coupled by sequential treatment with sodium methoxide and copper(I) bromide–dimethyl sulfide complex, thus providing a useful alternative approach to (*E,E*)-dienes (5; R² = H) as well as to (*Z,Z*)-butadienes (7).¹¹ Both procedures are reported to be stereospecific. A rather older procedure can be used to prepare (*E,Z*)-butadienes (8); thus, hydroboration of a terminal alkyne using thexylborane first affords an (*E*)-alkenyldithexylborane, which upon treatment with iodine and sodium hydroxide is then converted selectively into the (*E,Z*)-isomer (8).¹² Similarly, the (*E,Z*)-isomers of tetra-substituted butadienes (5; R¹ = R²) can be prepared from symmetrical internal alkynes.



A closely related sequence which has been used for the specific preparation of (E,E)-butadienes (10) from terminal alkynes involves hydroalumination, usually using DIBAL-H, followed by coupling of the intermediate vinylalane (9) using copper(I) chloride (*vide supra*).¹³ In much the same manner, hydrozirconation of a terminal alkyne and transmetallation using copper(I) iodide also leads smoothly to butadienes (10).¹⁴ A rather different source of copper, hydrated copper(II) nitrate, is very effective for the rapid and efficient coupling of vinylstannanes (11), leading to butadienes (10).¹⁵ (Z,Z)-Butadienes (7) can also be produced from (Z)-vinylstannanes by this method, which will also tolerate a substituent α to the tin functionality. A final example of the use of both copper(I) and silver(I) salts is as reagents for the coupling of (E)-alkenyl pentafluorosilicates to give symmetrical (E,E)-butadienes (5) in moderate to good yields; in contrast, reactions with copper(II) salts give only monomeric products by protonolysis of the C—Si bond.¹⁶



The direct dimerization of an alkenyl halide offers a potentially attractive alternative to many of the above two-step procedures. Such a transformation, $(12 \rightarrow 10)$, can be achieved using bis(1,5-cyclooctadiene)nickel(0) in the presence of triphenylphosphine.¹⁷ This mild method tolerates the presence of sensitive functional groups such as acrylic esters^{1,2} and has been improved since the original report by the introduction of more convenient catalyst systems such as triphenyl-¹⁸ and trialkyl-phosphine–nickel(0)¹⁹

species and an Ni⁰ reagent prepared by reduction of nickel(II) chloride using zinc in the presence of potassium iodide and/or thiourea.²⁰ An even simpler method for effecting such dimerizations is to employ 5 mol % Pd(OAc)₂ as catalyst in the presence of triphenylphosphine and potassium carbonate.²¹ This methodology looks particularly attractive for various intramolecular dimerizations, such as in the conversion of the bis(vinyl bromide) (13) into the cyclopentane (14) in 95% yield. Simple vinylic chlorides have also been coupled using a catalyst system composed of platinum chloride and cesium fluoride in the presence of Et₄NSnCl₃ in aqueous DMF.²²



Vinylmercury species (15) are very efficiently dimerized to dienes (16) upon treatment with palladium chloride and lithium chloride $[Li_2PdCl_4]$.²³ In addition to the high yields, a further attraction is the wide range of functionality which can be tolerated (this type of coupling does not appear to be too seriously affected by steric hindrance); this includes both α - and β -alkenyl and -alkynyl groups as well as aryl and β -acetoxy functions. An important disadvantage, however, along with the requirement for HMPA, is the need to use stoichiometric quantities of expensive PdCl₂; this can be overcome in many examples by using a rhodium species, [Rh₂(CO)₄Cl₂]·LiCl, which is only required in catalytic quantities.²⁴ A most unusual turnaround occurs with these couplings when they are carried out in nonpolar benzene using 0.5 equiv. of PdCl₂, in that the 'head to tail' product is obtained instead (*e.g.* 17 \rightarrow 18), thereby adding a useful facet to such vinylmercury chemistry.²⁵



2.3.2.2 Crossed Alkene–Alkene Coupling

This type of coupling reaction has only become a really viable option in synthesis during the past 15 years or so, with the advent of a number of nickel and palladium complexes, especially zerovalent species. The direct coupling of alkenyl and aryl Grignard reagents with alkenyl and aryl halides, catalyzed by various nickel complexes, has been extensively studied by Kumada and his coworkers in particular, with the result that some of the most suitable catalysts discovered, such as [NiCl₂(PPh₃)₂] and [NiCl₂(DPPP)], are sometimes referred to as Kumada- or Tamao–Kumada-type catalysts, as is the zero-valent species [Ni(PPh₃)₄]. These catalysts are particularly effective in conjugated diene synthesis of the type indicated in equation (2).²⁶ In general, the reactions proceed efficiently and under mild conditions, although with a lack of stereospecificity in some cases, even though vinyl Grignard reagents tend to be relatively less reactive in such reactions than the corresponding aryl species. Given protection against the Grignard reagent, a wide range of functional groups can be tolerated and in sterically hindered examples it is usually best to utilize the more hindered Grignard reagent, rather than to couple a relatively unhindered vinyl Grignard with a crowded alkenyl halide. In those examples with similar steric demands, the

ordering can still be important. For example, coupling between the allylsilane (19) and Grignard reagent (20) produces a 75% yield of the useful butadiene (21); coupling in the opposite sense leads to a lower (56%) yield.²⁷ Vinyl Grignard reagents obtained by Ni(acac)-catalyzed additions of RMgX to silylalkynes can also be coupled directly to vinyl bromides to provide silylbutadienes (*e.g.* 22, from vinyl bromide) in moderate yields; the second step is presumably also nickel catalyzed.²⁸ Isomeric silanes, as well as a variety of other butadienes, can be obtained by nickel-catalyzed displacements of thiol groups using vinyl Grignards (*e.g.* 23 \rightarrow 24).²⁹ Such reactions are quite general and have been successfully applied, for example, in styrene syntheses (Section 2.3.2.4).



A major advance in this area came with the discovery that vinyl Grignard species could be smoothly coupled with vinyl halides using tetrakis(triphenylphosphine)palladium(0) as catalyst (Scheme 3).30 Typical yields for such coupling reactions are 75-87% and stereochemical retention is normally excellent in both components, facts which often render this method more useful than a number of alternatives, such as couplings between various vinylcuprates and vinyl iodides (vide infra). As will be seen from the subsequent discussion, Pd^0 catalysts, especially [Pd(PPh_3)4], are undoubtedly the most important in this whole area of sp^2 - sp^2 -coupling reactions. As an example, the reaction shown in Scheme 3 can also be carried out using Cu¹ catalysts, but yields are generally around 50% and the method cannot be applied to vinylmagnesium bromide itself, due to polymer formation.³¹ However, preformed vinylmagnesiocuprates can be coupled to vinyl halides using 5 mol % of [Pd(PPh₃)₄] or [Pd(PO₃)₄]⁴ as catalyst in good yields (50-70%) and with high stereochemical retention.³² A specific example is the formation of the (Z,Z)-diene (27) from the magnesiocuprate (25) and the iodide (26); two notable features of this particular transformation are the highly selective production of the (Z,Z)-isomer and the high degree of substitution present in both components.³³ A further important discovery^{33,34} is that these and related reactions also benefit from the addition of a zinc salt, typically zinc chloride, an observation first reported by Negishi and coworkers (vide infra) during their studies on vinylaluminum coupling reactions. Not surprisingly perhaps, lithium divinylcuprates (e.g. 28) can also be coupled with vinyl iodides (29) using $[Pd(PPh_3)_4]$ and $ZnCl_2$ as catalysts.³⁴ This route to (E,Z)-butadienes (30) also displays a further useful feature, *i.e.* couplings with the dijiodide (31) proceed as expected in the usual ethereal solvents to give, for example, the dienyl iodide (32), whereas when HMPA is added alkylation of the vinyl carbanion occurs to give the alternative product, the homologated vinyl iodide (33). While this methodology can also be used to prepare (Z,Z)-butadienes, one major disadvantage is that one of the vinyl groups in the cuprate (28) is wasted, clearly a serious problem in cases when this function is valuable. This problem can be overcome by treating a divinyl species (34) with ZnCl₂; transmetallation then ensues to give two intermediates, (35) and (36), which both couple with a vinyl halide under the influence of a Pd^0 catalyst.³⁵ Such palladium(0) catalysts, especially [Pd(PPh₃)₄], are also very useful for effecting couplings between organolithiums and vinyl halides, although this type of reaction has been used mostly with examples of aryl- and heteroaryl-lithium species, as well as alkyllithiums, and is discussed more fully in the relevant sections which follow.³⁶ In general, such couplings involving alkenyllithiums cannot be carried out directly using Pd⁰ or related nickel catalysts. However, the discovery of the value of zinc chloride in Pd⁰catalyzed sp^2 -couplings has subsequently led to the development of what is perhaps the method of choice for coupling alkenyllithiums with alkenyl halides, which proceeds via transmetallation with zinc chloride followed by $[Pd(PPh_3)_4]^{-37}$ or $[Pd(DBA)_2]^{-2}PPh_3$ -catalyzed³⁸ coupling with an (E)- or (Z)-alkenyl halide. Yields and preservation of stereochemical integrity are generally very good and the method

(Scheme 4) is also applicable to aryl halides and to allenyllithium reagents (vide infra). One drawback is that an excess of the vinylzinc species is usually required if good yields are to be obtained. The method is not limited to hetero substituted alkenyllithiums and, for example, has subsequently been used to prepare homologs, (38) and (39), from the alkenylzinc bromide (37) and the corresponding vinyl iodides, in 50-70% yields.³⁹ One of the first generally useful methods in this area of crossed couplings was between alkenylaluminum reagents and vinyl halides, catalyzed by either nickel or palladium salts and ZnCl₂. The couplings are essentially stereospecific, except in examples of nickel-catalyzed reactions of (Z)-vinyl iodides, where this usually falls to ca. 90% and they sometimes fail when the aluminum species carries bulky substituents.⁴⁰ A specific example (Scheme 5) has been described in detail⁴¹ and, overall, the procedure complements many of the approaches developed by Normant and his coworkers³¹⁻³⁵ involving (Z)-alkenyl organometallics generated by stereospecific additions to terminal alkynes. In this present Negishi procedure, (E)-alkenylaluminum reagents are obtained by stereospecific hydroalumination of 1alkynes, usually using DIBAL-H. As a stoichiometric amount of ZnCl₂ is employed, this method is clearly closely related to the foregoing examples (Scheme 4).³⁷⁻³⁹ Zirconium species, which are prepared by hydrozirconation of terminal alkynes using Schwartz's reagent [Cp₂Zr(H)Cl], react in a similar fashion to that shown in Scheme 5 and are sometimes preferable to the corresponding alkenylaluminum species.⁴² A number of functional groups can be tolerated in the vinyl halide partner, including acetals (OTHP), ethers, ketones and esters.⁴³ As would be expected from the foregoing results, mixed aluminum-zirconium species (41), obtained from 1-alkynes (40), also undergo smooth Pd⁰- or Ni⁰-catalyzed cross-coupling reactions with alkenyl bromides, but only following the addition of ZnCl₂ (or CdCl₂).⁴⁴ In examples where the alkenylaluminum intermediate is highly hindered, it can sometimes be advantageous to convert it to the corresponding vinyl iodide (using I_2) and then couple this with a (less-hindered) organometallic.



486



A useful series of coupling reactions with polychloroethylenes has been developed, based upon the foregoing type of chemistry. For example, 1 equiv. only of the vinyl Grignard reagents (42) can be added to (Z)-1,2-dichloroethylene to provide the (E,Z)-chlorobutadienes (43) in good yield.⁴⁵ The method also works for (E)-1,2-dichloroethylene, which can be coupled similarly to (E)-alkenylaluminums,⁴⁶ both routes leading to (E,E)-1-chlorobutadienes (44). The latter aluminum species can also be coupled with 1,1-dichlorethylene (45), using a Pd⁰ catalyst at ambient temperature, leading to the useful 2-chlorobutadienes (46) in good yield.⁴⁷

A further significant advance in the general area of sp^2-sp^2 -coupling has been the finding that enol triflates can be coupled directly with lithium divinylcuprates (in addition to many related species) to provide butadienes in high yields and with good stereoselectivity (*ca.* 90% in the example shown in Scheme 6).⁴⁸ Enol triflates are readily prepared from the corresponding ketones, and are usually stable and easy to handle, and so the subsequent discovery that such intermediates could be smoothly coupled with vinylstannanes, under the catalytic influence of the ubiquitous [Pd(PPh_3)4], has enhanced their utility even further.⁴⁹ This latter protocol is of considerable merit, being highly efficient (80–100% yields) and stereoselective, as well as applicable to a wide range of substrates, both of which can be heavily substituted. The specific example shown in Scheme 7 represents a simple and efficient synthesis of the furanosesquiterpene pleraplysillin 1. Many further examples have been described in full,⁵⁰ including a simple approach to 1-silylbutadienes (Scheme 8), and the elaboration of an advanced model precursor of anthramycin (Scheme 9).⁵¹ In this latter example, the isolated yield was 78%, in contrast to a direct

Heck-type coupling using ethyl acrylate, which afforded a lower 40% return. Such couplings can also be carried out intramolecularly, thus providing a novel disconnection in the retrosynthetic analysis of a wide range of targets, which is especially attractive because the required precursors are often highly accessible. Some generalized examples are the conversions shown in Scheme $10^{52.53}$ for the formation of five- and six-membered rings; an application in the natural products area is in a neat synthesis of a dolastane diterpene in which the key cyclization (*cf.* Scheme 10) was effected in 81% yield by *in situ* formation of the vinyl triflate from the corresponding ketone (LDA, THF, HMPA, PhN(SO₂CF₃)₂) followed by the addition of [Pd(PPh₃)₄] (catalytic) and work-up after only 5 min at 30 °C.⁵³ In general, such cyclizations are faster when carried out in acetonitrile rather than in THF. In a similar fashion, vinylstannanes can be coupled with vinyl iodides, both inter- and intra-molecularly. For example, a recently reported approach to the natural fatty acid 13-HODE (the hydroxy acid corresponding to ester **49**) had as a key step a palladium-catalyzed reaction between the vinyl iodide (**47**) and the stannane (**48**).⁵⁴ A notable feature with this and many related processes is the tolerance of function groups such as carboxylic esters, the presence of which would be precluded in much of the more well-established carbanion chemistry. The



Scheme 9

intramolecular version of this reaction type, $e.g. (50) \rightarrow (51)$, is also catalyzed by a palladium(II) species, and is notable in this example for producing a 55% yield of a rather strained bicyclo[3.3.0]octadiene (51).⁵⁵



In a similar manner to vinyl triflates (Scheme 6),⁴⁸ vinylmercury chlorides also undergo direct coupling with vinylcuprates leading to butadienes in *ca*. 60–66% yield, but only when a higher order cuprate, R_3CuLi_2 , is used.⁵⁶

A final class of intermediates which are valuable in the synthesis of conjugated dienes are alkenylboranes and alkenylboronates; indeed such species were employed in some of the first reasonably satisfactory diene syntheses.⁵⁷ As with many of the foregoing reaction types, the advent of [Pd(PPh₃)₄] as a coupling catalyst has considerably transformed this area of chemistry. Much of the progress has been due to contributions made by Suzuki and coworkers, to the extent that such reactions involving alkenylboron species are often referred to as Suzuki couplings. In essence, the reactions consist of bond formation between an alkenylborane and an alkenyl halide in the presence of both [Pd(PPh₃)₄] and a base such as a sodium alkoxide (Scheme 11).⁵⁸ (E,E)-Butadienes can generally be obtained stereochemically pure by this method, but similar couplings involving (Z)-alkenyl halides often lead to isomeric mixtures, although in some cases even (Z,Z)-butadienes (53) have been obtained from (Z)-alkenylboranes (52), but in relatively low yields.^{59,63,64} With more weakly basic conditions, head to tail coupling tends to occur. Under similar conditions, (E)-alkenylboronic acids couple smoothly and stereospecifically with alkenyl iodides (Scheme 12).⁶⁰ As the alkenylboronic acids are readily obtained by hydroboration of terminal alkynes, this method is especially suitable for the synthesis of (E,E)- or (E,Z)-dienes. A third class of intermediates are the alkenylboronates. All three species have been compared for their suitability in the synthesis of the diene-containing insect pheromone bombykol and its three geometric isomers.⁶¹ The choice of method can often be important; for example, coupling the boronic acid (54) and the bromide (55) provides an 82% yield of the target (56), whereas an alternative using the vinyl iodide (57) and the borane (58) produces only a 50% yield. However, the latter type of coupling has been used successfully in a scaled-up preparation of leukotriene B₄ (Scheme 13).⁶² Despite these successful applications,⁶³ most of the methods are still not satisfactory for the elaboration of (Z,Z)-conjugated dienes as, in general,

yields are rather poor in such preparations. Perhaps the best way to gain access to this class of compound is by coupling alkenylboronates with alkenyl bromides, using the usual Suzuki-type conditions (Scheme 14).⁶⁴ This methodology is now viable as the required (Z)-alkenylboronates can be readily obtained from 1-alkynes via the corresponding 1-bromo-1-alkenylboronate. The method can also be used to prepare (E,Z)-dienes from (E)-alkenyl bromides and has also been extended to the stereospecific elaboration of more highly substituted dienes, e.g. (59) and (60), by using 1,2-disubstituted alkenylboronates.⁶⁵ Related (Z,Z)-1-bromo-1,3-butadienes can be obtained using a haloboration-hydroboration sequence involving two alkynes; further elaboration of these initial products by, for example, halogen-lithium exchange and homologation provides yet another route to (Z,Z)-butadienes.⁶⁶





2.3.2.3 Allene Synthesis by sp²-sp² Coupling

Not surprisingly, many of the methods which can be used in the synthesis of conjugated dienes can also be applied to dimerization of allenes. Thus, the bromoallenes (61) can be smoothly dimerized, for example, by treatment with 2 equiv. of copper(I) chloride in DMF at ambient temperature to give good yields of the diallenes (62).^{1,2,67}



The aforementioned method for coupling alkenylzinc halides with vinyl halides^{37–39} is also applicable to the bromoallenes (**61**) (the corresponding bromoalkyne can also be used), which react smoothly with both 1-alkenylzinc chlorides and phenylzinc chloride in the presence of $[Pd(PPh_3)_4]$ to give the homologs (**63**) in good yields.⁶⁸ By studying such reactions with chiral haloallenes, a duality of mechanism has been found.⁶⁹ Iodoallenes react with ArZnCl-Pd⁰ with retention of configuration suggesting that oxidative insertion by palladium into the C—I bond is the key step, whereas the corresponding chloro- and bromo-allenes react with moderate levels of inversion implying the participation of an *anti-S*_N2 like reaction followed by a [1,3] Pd shift with inversion. A direct S_N2-type inversion is observed when chiral bromoallenes, *e.g.* (**64**), are treated with the Lipshutz higher order cyanocuprates resulting in the generation of essentially enantiomerically pure homologs, *e.g.* (**65**), in 70–80% yields.⁷⁰ Other reagents such as 'PhCuMgBr₂' give poorer optical yields and large amounts of the corresponding alkynic product.



Aryl Grignard reagents also coupled satisfactorily with bromoallenes, using $[Pd(PPh_3)_4]$ as catalyst,³⁰ to give good yields of arylallenes (**63**; $R^4 = Ar$),⁷¹ whereas couplings in the opposite sense between allenyllithiums derived from allenyl ethers or allenylsilanes and aryl iodides can also be achieved,⁷² again using Pd⁰ catalysts. This is in contrast to similar reactions with vinyllithiums,^{36–39} which usually require

prior transmetallation to the corresponding alkenylzinc species before successful couplings can be achieved (for an example using an allenylzinc intermediate, see ref. 38).

2.3.2.4 Alkene–Aryl Coupling

Many of the methods outlined in Section 2.3.2.2 (alkene-alkene coupling) are equally suited to alkene-aryl coupling reactions simply by either substituting an aryl halide for the alkenyl halide or conversely using an aryl organometallic in place of the corresponding alkenyl species. For example, arylmagnesium halides can be coupled to alkenyl halides using a variety of nickel(II) species as catalysts (Scheme 15).²⁶ In general, due to the relatively lower reactivity of vinylmagnesium halides, it is preferable to carry out these coupling reactions as shown, rather than the converse of reacting a vinylmagnesium species with an aryl halide. An exception to this could be with examples where the alkenyl function is highly substituted as, in general, it is better to use the Grignard reagent derived from the more sterically crowded partner in this type of coupling. A recent specific example of the utility of this general method is the elaboration of 2-arylallylamines (68) from arylmagnesium bromides (66; Ar = m- and p-MeOC₆H₄, -FC₆H₄ and -ClC₆H₄) and the vinyl chloride (67).⁷³ An additional key feature in the success of this method is the use of the Magnus stabase function (Me₂Si(CH₂)₂SiMe₂) to protect the amino group. Using these conditions, any Grignard reagents also couple in a regiospecific manner with β -halovinyl ethers (e.g. Scheme 16; silyl enol ethers⁷⁶ can also be used).⁷⁴ It should be noted that a variety of vinylic ether and sulfide groups can also be directly displaced by aryl Grignard and other organometallic species using nickel catalysts, For example, enol ethers (e.g. 69) and phenylmagnesium bromide (70) react together in the presence of the usual type of nickel catalyst to give substituted styrenes (e.g. 71) in >70% yields.⁷⁵ However, such coupling reactions are limited to relatively simple substrates as the reactions are rather sensitive to steric effects; enolates and enamines fail to react in a similar fashion. However, silvloxy groups in simple O-silvl enolates, e.g. (72), can be displaced by phenyl groups in the same way (PhMgBr, [NiCl₂(PPh₃)₂]) to give styrenes, e.g. (73), with a high degree (92%) of stereochemical retention,⁷⁶ while a method related to this and previously applied to butadiene synthesis (vide supra) consists of coupling lithium diarylcuprates with enol triflates $(74 \rightarrow 75)$.⁴⁸ In much the same way, enol phosphate groups can also be displaced (e.g. $76 \rightarrow 77$) using an aryl Grignard reagent and [NiCl₂(DPPE)] as catalyst.⁷⁷ This type of coupling can also be carried out in the reverse sense by reacting aryl phosphates (78), which are readily available from the corresponding phenol, with alkenylaluminums and using [Ni(acac)₂] as catalyst.^{78,94}

The sulfur function in vinyl sulfides can be displaced in the same manner, using Ni or Pd catalysts, and the methodology has been extended to tandem couplings using 1-bromo-2-phenylthioethene (Scheme 17).⁷⁹ The regioselectivity of the first step is exactly as that observed with the corresponding β -bromovinyl ether (Scheme 16).⁷⁴ In just the same way the 1,1-disubstituted alkene (**79**) can be converted into





the derivatives (80);²⁹ although the whole sequence can be carried out in one pot, overall yields are higher if the intermediate vinyl sulfide is isolated. A recent application of this methodology has been in the homologation of the pyrrolizidine sulfide (81) to the phenyl-substituted alkaloid (82) using PhMgBr-[NiCl₂(PPh₃)₂], usually the catalyst of choice for this type of displacement.⁸⁰ Vinyl sulfones⁸¹ and vinyl selenides⁸² also undergo such displacement reactions with aryl Grignard reagents; with the former substrates, the reactions proceed with complete retention of stereochemistry using either nickel or iron catalysts, while the latter reactions require the use of excess (*e.g.* 2.5 equiv.) aryl Grignard reagent and use Ni¹¹-phosphine complexes as catalysts. Both sequences afford good to excellent yields, at least for the relatively simple examples studied (Scheme 18).





The sulfide displacement method has also been carried out successfully in the opposite sense (cf. Scheme 17), wherein aryl sulfides are reacted with vinylmagnesium halides, as well as with a variety of other Grignard reagents, using a nickel catalyst.⁸³ In examples with bis-sulfides, a high degree of regio-selectivity is observed if the sulfur groups are differentially substituted, the less-hindered reacting much faster (Scheme 19).

Ph



Scheme 19

Despite the great preponderance of nickel(II) catalysts used in the foregoing methods, aryl Grignard reagents and alkenyl halides have been successfully coupled using other catalysts, most notably examples of iron(III) complexes, including FeCl₃,⁸⁴ Fe(acac)₃⁸⁴ and Fe(DBM)₃ [tris(dibenzoylmethido) iron(III)].⁸⁵ Yields are often not as good as with the nickel catalysts, but the relative cheapness of these reagents could be an advantage.

An entirely different method for effecting, overall, an sp^2-sp^2 -coupling of aryl Grignard reagents is by reaction of the latter with pyridazine N-oxide (Scheme 20).⁸⁶ In the simple cases studied, yields of enynes were 52–77%. The possibility of applying this approach to more highly substituted substrates does not appear to have been examined.



Aryllithium species can also be efficiently coupled with alkenyl halides (Scheme 21) but using a palladium(0) catalyst (usually [Pd(PPh₃)₄]) or a related rhodium species; in general, nickel catalysts are not suitable for couplings involving organolithiums.³⁶ The reactions have a broad scope, within the limitation of providing protection for groups sensitive to organolithium species, and are generally very efficient and essentially stereospecific (*cf.* Scheme 15).²⁶ A way in which the requirement for an expensive catalyst can be avoided is to couple an alkenyl halide with a lithium diarylcuprate directly (Scheme 21).⁸⁷ The addition of various ligands, such as LiBr or Buⁿ₃P, and the correct choice of solvent are all necessary considerations if high yields are to be obtained using this method. In view of its greater simplicity, the ArLi–Pd⁰ combination is probably the method of choice. However, the latter method has been put to good use in the homologation of arylchromium complexes (**83**) to substituted styrene complexes (**84**), which are useful as benzofuran precursors.⁸⁸



A combination of the foregoing methods has been used to obtain (Z)-styrenes (86) by the alternative approach of employing an alkenyl organometallic, initially a divinylcuprate (85) derived by addition of R_2CuLi to acetylene.⁸⁹ In contrast to the foregoing methods involving aryl organometallics, the aryl ring in the product (86) can be substituted by a wider variety of functional groups, including Br, OMe, NO₂ and CO₂Me. This procedure is also useful for the synthesis of conjugated dienes^{34,35} and is notable for its

stereospecificity and efficiency, in that both alkenyl groups in the precursor (85) are utilized. However, there can sometimes be problems in forming this latter species cleanly and avoiding further addition of acetylene to give a bis(dienyl)cuprate species. Not surprisingly in view of the foregoing, alkenylzinc species,³⁷⁻³⁹ directly generated by H/Li exchange and transmetallation (ZnCl₂), undergo smooth coupling with aryl iodides using [Pd(PPh₃)₄] as catalyst. Conversely, arylzinc chlorides can also be linked to alkenyl bromides (Scheme 22), again using a Pd⁰ catalyst.^{49,90} A notable feature of this example is the highly substituted nature of the alkenyl bromide. Usually, it is better to convert the more crowded partner into the initial organometallic species,²⁶ although in this case the success of the scheme probably reflects the higher reactivity of arylzinc species.⁴⁰





Both Ni⁰ and Pd⁰ catalysts induce the coupling of alkenylaluminum reagents to aryl halides (Scheme 23),^{40,91} an approach which has been employed in an alternative synthesis, in a converse manner, of tetrasubstituted alkenes of the type shown in Scheme 22.⁹² One limitation with this approach is the lack of flexibility in the available stereochemistries of the alkenylaluminum species; usually only the (*E*)-isomer is readily available. In just the same way, alkenylzirconium species, obtained either by hydrozirconation or the addition of R₃Al–Cl₂ZrCp₂ to a terminal or symmetrical internal alkyne, can also be coupled with aryl iodides using either Ni⁰ or Pd⁰ catalysts,^{43,44} to give styrene derivatives in generally excellent yields. With hindered examples, the addition of zinc chloride is essential if good yields are to be obtained and in extreme cases, it has been recommended⁴⁴ that the best tactic may be to convert a hindered alkenylaluminum species into the corresponding alkenyl iodide (I₂) and then couple this with an organometallic species.





Palladium(0)-catalyzed coupling of alkenyl halides with vinylstannanes^{54,55} also constitutes a useful approach to styrenes (Scheme 24);^{93a} a particularly attractive feature of this method is the toleration of a wide variety of substituents on the aromatic ring which include methoxy, nitro and bromide as well as unprotected carbonyls, acetate and 1,3-dione functions. One small limitation is that the reaction fails when free amino groups are present. β -Silylvinylstannanes can also be used, leading to β -silyl-styrenes.^{93b} A valuable extension to this method is the finding that aryl triflates, *e.g.* (87), can also be coupled to vinylstannanes to give generally excellent yields of arylalkenes (*e.g.* 88).⁹⁴ Closely related to similar couplings of vinyl triflates^{48–53} and of arylphosphates,⁷⁸ such reactions are mild (dioxane or DMF, <100 °C) and widely applicable, although the stereochemical integrity of the vinylstannane is not always preserved. An alternative approach to less readily available (Z)-arylalkenes is by palladium-catalyzed coupling reactions between aryl bromides and (E)-vinylsilanes,⁹⁵ which proceed with overall inversion (Scheme 25).⁹⁶

In the same manner that various alkenylboron species can be coupled with alkenyl halides,⁵⁷⁻⁶⁵ so too can aryl halides be homologated to styrenes (Scheme 26).^{60,97} In general, such reactions require a Pd⁰ catalyst and the presence of a base, such as sodium ethoxide, in order to achieve this type of head to head coupling.⁵⁹ In sharp contrast, mainly head to tail products are formed when a weak base, such as triethyl-





amine, is used in place of ethoxide in such reactions with alkenylboronates (Scheme 27).⁹⁸ A particularly attractive route to (Z)-styrenes, as well as to arylbutadienes, is by reaction between an aryl halide and a (Z)-alkenylboronate (cf. Scheme 14);^{64,65} similar reactions using (Z)-alkenylboranes are generally much less satisfactory. Using standard Suzuki-type conditions ([Pd(PPh₃)₄], NaOH), the β -ethoxyboronates (90) derived from the corresponding ethoxyalkyne (89) and catecholborane, also couple smoothly with aryl iodides to give excellent yields of substituted styrenes (91).⁹⁹



R = n-alkyl; $R^1 = OR^2$, NR^2_2 , X

As in some examples of the synthesis of conjugated dienes,⁵⁶ arylmercuric chlorides (92) react directly with vinylcuprates (93) (other stoichiometries are less satisfactory) to give substituted styrenes (94).²³ A number of relatively sensitive functions, such as CO₂Me, can be present on the aryl ring. An alternative method for the direct coupling of alkenyl iodides to mercurals (92) is to heat the two together in HMPA containing lithium chloride and Wilkinson's catalyst [RhCl(PPh₃)₃]; unfortunately, (Z)-1-alkenyl iodides give largely (*e.g.* 3:1) the (*E*)-substituted styrene.^{23,100}



Another useful approach to styrenes via sp^2-sp^2 -coupling reactions, which is beyond the scope of this section, is Meerwein-type arylations in which aryldiazonium salts undergo usually copper(I)- or palladium-catalyzed couplings with electron deficient alkenes, indicating a radical-based mechanism.¹⁰¹ The method is also useful for the synthesis of stilbenes from unsubstituted styrenes¹⁰² or β -silylstyrenes.¹⁰³

2.3.2.5 Alkene–Heteroaryl Coupling

Not surprisingly, much of the methodology suitable for alkene-aryl coupling is also applicable to the elaboration of conjugated alkenyl heteroaromatic systems. In many cases, both types of reaction, either between an alkenyl organometallic and an aryl halide or *vice versa*, are possible, with the optimum pathway often dependent on the relative substrate availability. Although many of the examples involve only the simplest of heteroaromatic compounds, such as 2-furyl- or 2-thienyl-lithium or the corresponding bromides, the mildness of the procedures and their generality in other areas strongly suggests that they should be applicable to the synthesis of many, more complex, targets.

2-Furyl- and 2-thienyl-lithium reagents undergo smooth Pd⁰-catalyzed coupling with alkenyl bromides (Scheme 28); yields are generally excellent, as is the degree of stereochemical retention.³⁶ The method would certainly not be applicable to the corresponding 3-lithio species which undergo ring opening at well below the reaction temperature. 2-Thienyllithium can also be coupled to the vinyl chloride (67) using a nickel catalyst, [NiCl₂(DPPP)], to give the vinylthiophene (95) (cf. 67 \rightarrow 68).⁷³ In related work, access to the 2- and 3-vinylthiophenes, (96) and (97), has been shown to be possible by coupling a Grignard species with the appropriate bromide or, in the latter 3-substituted case, specifically by reaction between 3-bromothiophene and the alkenylmagnesium bromide in the presence of the palladium catalyst [PdCl₂(DPPB)].¹⁰⁴



As mentioned in the foregoing sections, some of the most suitable intermediates for sp^2-sp^2 -coupling reactions are alkenyl- or aryl-zinc species. In general, these species tend to be relatively the most reactive in such Pd- or Ni-catalyzed reactions and often lead to the highest yields.⁴⁰ For example, 2-furylzinc chloride reacts smoothly and efficiently with a range of alkenyl iodides (Scheme 29)¹⁰⁵ and this method has been used to prepare the Lophotoxin intermediate (98) in 70% yield.¹⁰⁶ Such coupling reactions can also be catalyzed by [Pd(PPh₃)₄] and can be carried out in either sense; thus the thienyldihydronaph-thalene (99) has been obtained both from 2-thienylzinc bromide or from the corresponding alkenylzinc

bromide and 2-iodothiophene.³⁹ Similarly, a variety of fluorinated vinylzinc chlorides can be reacted with a range of iodinated heteroaromatics to give the expected vinyl derivatives (*e.g.* **100**, from 2-iodopyridine and CF₂CHZnCl).¹⁰⁷ Vinylpyridines, *e.g.* **(101)**, can also be obtained from alkenylalanes (prepared from terminal alkynes using Me₃Al–Cl₂ZrCp₂)⁴⁰ and bromopyridines using a Pd⁰ catalyst.⁴⁰



A variety of methods have been examined for the preparation of the vinyl ether homolog (102) of the corresponding 4-bromo-3-iodoindole.¹⁰⁸ Clearly, relatively unreactive species were required which would not displace the 4-bromo atom. The best approach was found to be a Negishi-type Ni⁰-catalyzed coupling with an alkenylzirconium species obtained by hydrozirconation of ethoxyacetylene using Schwartz's reagent.⁴² Other procedures such as Pd-catalyzed coupling reactions with alkenylboranes⁹⁹ or Heck-type reactions were less successful. However, related organoboronates have been found to be very suited as intermediates for the elaboration of trisubstituted alkenes *via* sp^2 - sp^2 -coupling reactions.⁶⁵ For example, formation of an 'ate' complex using 2-thienyllithium and the boronate (103), derived from 1-bromo-1-hexyne, followed by base-induced migration leads to the vinylboronate (104), and, thence to the alkene (105) following Pd⁰-catalyzed coupling^{59,64,98,99} with iodobenzene (Scheme 30). The overall yield in this particular case was 94% of material with >97% isomeric purity. Related isomers can be obtained by using an iodothiophene in the last step. Coupling reactions in the reverse sense, between pyridylboranes and vinyl bromides, also catalyzed by zerovalent palladium species, have been used to obtain a series of vinylpyridines.¹⁰⁹ Heteroaromatic mercurals can also be coupled directly to alkenyl halides, but using Wilkinson's catalyst.¹⁰⁰



Scheme 30

2.3.3 ARYL-ARYL COUPLING REACTIONS

2.3.3.1 Aryl-Aryl Dimerization

Within the confines of equation (1), the most extensively studied and exploited methodology for the elaboration of symmetrical biaryls is the classical Ullmann reaction,¹¹⁰ which is usually defined as the copper-mediated coupling of aryl halides (equation 3). The types of copper used in the older literature are copper powder itself, copper bronze¹¹¹ or copper(I) oxide, although this latter reagent is often more suited¹¹² to the Ullmann biaryl ether synthesis from phenols and aryl halides. The most common solvents employed are pyridine, quinoline, DMF and tetramethylurea (for reactions requiring higher temperatures than DMF¹¹³); temperatures of up to 260 °C have been used with success in examples involving robust substrates and such couplings can also be effected without the use of a solvent in sealed tubes. Rather more recent methods for obtaining activated copper reagents include pretreatment of the copper powder by EDTA¹¹⁴ and generation of copper powder by zinc reduction of copper(II) sulfate¹¹⁵ or by reduction of copper(I) iodide using potassium naphthalenide.¹¹⁶ These methods, especially the latter, produce highly activated material which can allow Ullmann couplings to be conducted at lower temperatures. One of the mildest reagents discovered is copper(I) triflate in association with 5% aqueous ammonia in acetone or acetone-acetonitrile mixtures.^{2,117,118} For example, 2-iodonitrobenzene (**106**) is converted into biphenyl (107) at 20 °C in just 5 mins. Unfortunately, other substrates often take an alternative pathway consisting of replacement of the halide by NH₂ or OH; thus methyl 2-iodobenzoate is transformed largely into methyl salicylate.



The mechanism of Ullmann couplings most likely proceeds via discrete arylcopper species^{1,110} and it is therefore not surprising that preformed arylcopper species can be used in such reactions, for which DMF rather than pyridine or quinoline is often the preferred solvent.¹¹⁹ Arylcopper reagents can be generated by direct lithiation followed by Li/Cu exchange as exemplified by the facile synthesis of a tetramethoxybiphenyl (Scheme 31).^{120,140} This type of approach is especially useful for sterically crowded cases, where the classical Ullmann approach (equation 3) fails,¹²¹ and also for the elaboration of fluorinated biphenyls using pentafluorophenylcopper.¹²² Such couplings are not limited to Li-Cu systems; oxidative couplings of both aryllithiums and arylmagnesium halides can be achieved using a variety of copper, cobalt, chromium, iron and manganese salts.^{5,7,128} Silver salts, in the presence of a nitrogen-containing oxidizer, such as LiNO3 or MeNO3, are particularly effective in dimerizations of aryl Grignards,⁷ which can also be carried out using thallium(I) bromide.¹²³ This latter method is relatively mild and high yielding although some drawbacks, beyond the need to protect Grignard-sensitive groups, include failure in attempted couplings of ortho-substituted Grignards and reagent toxicity, as excess of the thallium salt is usually required. Direct dimerizations of electron rich arenes, such as examples containing alkoxy groups, can be smoothly carried out using thallium tris(trifluoroacetate) (TTFA) in carbon tetrachloride or acetonitrile containing boron trifluoride etherate.¹²⁴ In general, classical Ullmann-type dimerizations are less satisfactory with electron rich substrates; a further bonus of the TTFA method is that halogen atoms are preserved (e.g. Scheme 32), allowing further manipulation of the initial biphenyl. Palladium(II) acetate can also be used as a catalyst in this type of coupling,¹²⁵ which is also especially suitable for the synthesis of binaphthyls and in intramolecular coupling reactions (Section 2.3.3.3).¹⁴⁰

Many substituents are tolerated during classical Ullmann couplings. Nitro (e.g. $106 \rightarrow 107$) and alkoxycarbonyl functions strongly activate the reactions when these groups are positioned ortho to the reaction site, but not when they are meta or para. Alkyl and alkoxy groups can activate from all positions, although with polysubstituted examples, excessive electron donation can be deleterious.¹²⁴ Ullmann couplings are not especially sensitive to steric effects, but are usually inhibited by the presence in either reactant of free OH, CO₂H or NH₂ groups. Fortunately, these can all be readily masked (and sub-



sequently easily deprotected) by trimethylsilylation¹²⁶ or, in the case of phenolic functions, by mesylation.¹²⁷

A major advance on the traditional Ullmann method (equation 3) was made in the early 1970s, when Semmelhack and his coworkers discovered that Ni⁰ species could be used in place of elemental copper.¹²⁸ The original complex used was bis(1,5-cyclooctadiene)nickel(0) which, as an indication of the much greater mildness of this and related procedures, affords an 82% yield of biphenyl from bromobenzene at 50 °C in DMF (Scheme 33). The limitations associated with nickel-based approaches are similar to the Ullmann reaction in that free hydroxy groups (OH, CO₂H) prevent coupling, although the former method does sometimes suffer more from the steric effects of ortho substituents, two of which usually prevent reaction entirely. The toleration of methoxy, methoxymethyl, ester and especially formyl groups renders this general approach more attractive than ones based on aryllithium or Grignard intermediates (vide supra). Subsequent developments have essentially been aimed at discovering alternative nickel catalysts to [Ni(COD)2] which do not require dry box techniques, etc. Successful examples include triphenylphosphine–Ni⁰ complexes generated by zinc reduction of the corresponding nickel(II) chloride,¹⁸ a process which can be catalytic in nickel¹²⁹ and which is promoted by the addition of various salts (e.g. KI,¹²⁹ NaI¹³⁰) and of 2,2'-bipyridine¹³⁰ to such an extent that aryl chlorides can be coupled without excessive heating (65-80 °C), and aryl iodides coupled at room temperature.¹²⁹ Similar species derived by reduction of bis(trialkylphosphine)nickel(II) dichlorides¹⁹ or from the related system (NiBr₂-Zn-HMPT-KI)¹³¹ also lead to high yields of symmetrical biaryls under mild conditions. A full report from Semmelhack's research group indicates that [Ni(PPh₃)₄] is a good reagent both for this type of coupling and especially for intramolecular versions.¹³² Aryl triflates can also be dimerized using a nickel(0) catalyst similar to some of the foregoing examples generated in situ from nickel(II) chloride and zinc powder, in the presence of triphenylphosphine and sodium iodide.¹³³ A relatively large excess of the latter reagents is required and rates are considerably enhanced by sonication. Yields are generally above 80% with ca. 10% of the deoxygenated arene being formed as a by-product.



Scheme 33

In just the same way that vinylstannanes can be dimerized, $(11) \rightarrow (10)$, aryltri-*n*-butylstannanes react rapidly (10–30 min) with hydrated copper(II) nitrate in THF at 20 °C to afford high yields of biaryls,¹⁵ although once again, *ortho* and especially bis-*ortho* substituents inhibit the coupling.

Biaryls can also be derived (70–90%) from mercury salts, ArHgX, using methodology very similar to a normal Ullmann coupling (copper powder, pyridine, reflux), except that a catalytic amount of PdCl₂ is also required.^{23,134} Less basic solvents do not lead to such good yields, but otherwise the reaction charac-

teristics are very similar to the Ullmann method except that free amino groups can be present. Related procedures can also be used to extrude elemental mercury from diarylmercurals, Ar₂Hg, to give bia-ryls.^{23,110} Excellent yields of biaryls can also be obtained from dimerizations of arylmercuric salts using a rhodium catalyst, [Rh(CO)₂Cl]₂LiCl, or stoichiometric amounts of Li₂PdCl₄.²⁴

Palladium, in the form of palladium(II) acetate, has also been used to catalyze biaryl formation directly from aryl iodides (R₃N; 100 °C), especially *p*-NO₂ and *p*-Cl derivatives. As usual, *ortho* substituents severely hinder this type of coupling.¹³⁵ Related reductive couplings of aryl halides have been achieved using hydrazine and a Pd–Hg catalyst,¹³⁶ electrochemically generated Pd⁰ catalysts,¹³⁷ or a palladium on carbon catalyst in the presence of aqueous sodium formate, sodium hydroxide and, crucially, a catalytic amount of a surfactant.¹³⁸ The first two procedures look to be particularly selective and efficient, while the latter, rather different, method is not so efficient but does look amenable to large scale work.

Symmetrical biaryls can of course also be prepared using methods designed for the elaboration of unsymmetrical biaryls, by a judicious choice of substrate.

2.3.3.2 Crossed Aryl-Aryl Coupling

The standard Ullmann methodology can be applied to examples where sufficient differences exist between the two reactants that the desired crossed product is formed most rapidly. Usually, electronic factors are the most useful in this respect as in the example shown (Scheme 34), wherein one component is extremely electron deficient at the site of coupling.¹¹⁰ Two such coupling reactions can also be viable as shown by a preparation of the tetraphenyl (109) from the biphenyl (108); despite a lowish yield (43%), this is still a rapid and efficient route to such compounds.¹¹⁰ Quite complex biphenyls can be made in this way, given a sufficient electronic and reactivity (I > Br > Cl) difference. Thus, a good combination is the iodide of the more electron rich component, which will presumably form an arylcopper species more rapidly, and the bromide of the more electron deficient partner (Scheme 35).¹²⁷ This and the following example further exemplify the relative insensitivity of such couplings to steric crowding. An additional tactic for such reactions is to use a large excess of unactivated copper powder and a reaction temperature which is lower than that required for self-coupling of the less reactive component. The excess copper results in much shorter reaction times (Scheme 36),¹³⁹ even with such crowded substrates. Despite these successes, an obvious drawback with such reactions is the often very high temperature required. This can be circumvented by using the method of Ziegler and coworkers in which a preformed arylcopper, having the copper atom ligated both by triethyl phosphite and the lone pair of a substitutent atom, couples at below ambient temperature with an o-iodobenzaldehyde imine; an example (Scheme 37) is an approach to steganone.¹⁴⁰ In addition to the low temperature, once again steric hindrance is not a problem, which is certainly not the case with many of the related transition metal catalyzed processes using Ni⁰ or Pd⁰ catalysts, for example. The method is also very useful for the dimerization of o-bromobenzaldehydes, and in general appears to be the method of choice for obtaining this type of highly substituted biphenyl, given that a suitable ligand is present in the ortho position.

A less general but related method for obtaining 2,6-dinitrobiphenyls consists of metallation of 2,6-dinitrobenzene using copper(I) *t*-butoxide in a mixture of pyridine and DME followed by coupling with an




Scheme 37

iodoarene (Scheme 38).¹⁴¹ Yields are generally in the range 50–95%. As would be expected from the foregoing discussion¹⁴⁰ (cf. Scheme 21), lithium diarylcuprates in general couple with iodoarenes to give good yields, at least in sterically unencumbered examples of unsymmetrical biaryls, given an appropriate additive (e.g. LiBr or Buⁿ₃P) and solvent.⁸⁷



The by now well-established nickel-catalyzed procedure for coupling Grignard reagents with alkenyl and aryl halides (equation 2 and Scheme 15) is also highly suitable for biaryl synthesis, except in examples where both of the substrates have ortho substituents. In cases where one component does carry ortho substituents, it is preferable to generate the Grignard from this component, rather than the converse (Scheme 39).²⁶ In addition to the Tamao-Kumada catalysts, [NiCl₂(PR₃)₂], a wide range of other nickel catalysts can also be used^{142,143} and sequential replacement of two halogens in a dihaloarene is also possible.¹⁴³ Unsymmetrical 1,1'-binaphthyls can be prepared similarly in a procedure which is greatly facilitated by the application of ultrasound.¹⁴⁴ Full details of a related coupling (not sp^2-sp^2) using the Tamao-Kumada method have been given which will be of use to those wishing to use these procedures.¹⁴⁵ The same type of catalyst can also be used to induce the displacement of aryl methoxy groups (cf. 69 \rightarrow 71) by Grignard reagents.⁷⁵ As expected, this type of coupling (Scheme 40) can suffer from steric effects. Sulfide groups can be displaced similarly^{29,79} by arylmagnesium bromides; a particularly attractive feature of this chemistry is its application to terphenyl synthesis by sequential introduction of two different aryl groups to a chlorophenyl alkyl sulfide (Scheme 41).¹⁴⁶ Presumably a wide variety of combinations of both alkenyl^{29,79} and aryl groups could be introduced by this methodology, which is also applicable to both o- and m-chlorophenyl sulfides.



Often, reactions catalyzed by nickel species can also be effected using closely related palladium compounds; this is certainly the case with cross-coupling reactions between aryl Grignard reagents and aryl halides, which can be effected using a variety of palladium catalysts.¹⁴⁷ These reactions often show greater selectivity than the related nickel catalysts in, for example, couplings between ArMgX and 4bromo-1-chlorobenzene, where the more reactive bromo substituent is displaced. A more recent study has shown that this approach is particularly suitable for the elaboration of polyoxygenated biphenyls.¹⁴⁸ An important feature of the chemistry of aryl Grignard reagents, which should be noted when contemplating the use of this type of coupling, is that such species are capable of directly displacing methoxy groups adjacent to electron-withdrawing functions; indeed such reactions constitute a rapid and simple entry into a variety of substituted biphenyls and binaphthyls (Scheme 42).¹⁴⁹ In general, poorer yields are obtained using the corresponding aryllithium species.

The Negishi method for coupling alkenyl- or aryl-zinc halides to sp^2 -centers^{37–40,90} can equally well be used to obtain biphenyls (Scheme 43).¹⁵⁰ This excellent method, in which the arylzinc species is usually generated by transmetallation of the corresponding aryllithium, has been described in graphic detail.¹⁵¹ Related coupling reactions can also be achieved by using an arylstannane in place of an arylzinc species.¹⁵² A special case of this type of reaction is useful for the elaboration of phenylaryls by transfer



Scheme 42

of one (only) phenyl ligand from tetraphenyltin to an aryl halide using a palladium(II) catalyst.¹⁵³ Of potentially more general use is the finding that, under the influence of a Pd⁰ catalyst, an aryl group can be transferred selectively from an aryltrialkylstannane to an aryl triflate; yields of biaryls are generally high, at least in the relatively simple cases so far studied.⁹⁴



Scheme 43

The Suzuki method, using alkenylboronic acids as intermediates, $^{60-63,97}$ can also be applied with great effect to the synthesis of biaryls (Scheme 44).¹⁵⁴ Yields are generally excellent and either aryl bromides or iodides (but not chlorides) can be used. Methoxy, methoxycarbonyl and nitro functions at least do not interfere and the method seems to be relatively insensitive to steric factors; furthermore catalyst levels as low as 3 mol % are adequate. A useful combination for the elaboration of substituted biaryls and higher homologs is the *o*-lithiation methodology, due mainly to Snieckus and coworkers, followed by Suzuki-type coupling (Scheme 45).¹⁵⁵ A good method for formation of the intermediate boronic acid species proceeds via sequential *o*-silylation (TMS-CI) and *ipso*-boro-desilylation using boron tribromide. A simple hydrolytic work-up using 5% aqueous HCl then delivers the boronic acid.



Scheme 45

Although not strictly belonging to the class of reactions indicated in equation (1), a number of other approaches to biaryls are worth noting as these can be very useful in some circumstances. For example, reactions proceeding via S_{RN}1 mechanisms can be especially useful for the introduction of aryl groups at positions adjacent to either amino¹⁵⁶ or hydroxy¹⁵⁷ functions (Scheme 46). The latter type of transformation can also be effected using triphenylbismuth carbonate¹⁵⁸ or other pentavalent bismuth species.¹⁵⁹ Aryllead triacetates [ArPb(OAc)₃] are also useful for aromatic arylation reactions, especially with electron rich substrates.¹⁶⁰ A classical method for biaryl synthesis, again somewhat outside the scope of this chapter, consists of basification of an aryldiazonium salt in the presence of a second aryl component, often benzene itself. Variously referred to as the Gomberg, Gomberg-Bachmann or Gomberg-Bachmann-Hey process,¹⁶¹ this and related free radical type methods¹⁶² give only moderate yields of biaryls, but at least they are generally simple reactions to carry out. Also closely related is the Meerwein arylation procedure,¹⁰¹ wherein aryldiazonium species are coupled with aryls usually using copper(II) salts as catalysts. In general, the reactions work well only with electron poor substrates. A more recent development with the Gomberg reaction is the possibility of using phase transfer conditions resulting in considerable increases in yields relative to the classical methods.¹⁶³ The method is, however, still largely limited to the elaboration of relatively simple 2- and 4-monosubstituted biaryls. Related alternatives include the photolysis of aryl iodides in an aromatic solvent¹⁶⁴ or similarly of arylthallium trifluoroacetates; this latter method often delivers yields in excess of 80%, although it is rather limited to relatively simple products.¹⁶⁵ Many of the above methods are perhaps more suited to intramolecular biaryl formation.



Scheme 46

2.3.3.3 Intramolecular Aryl–Aryl Coupling

The classical Ullmann procedure¹¹⁰ has often been applied to ring closure reactions by formation of an aryl-aryl bond. In general, yields are much higher when electron-withdrawing groups are positioned ortho to the halogen atoms, which should be iodine rather than bromine (Scheme 47). Similar conditions (Cu bronze, reflux in DMF) can also be used to close a four-membered ring in respectable yield (Scheme 48). As with the intermolecular version, it is often preferable to use one of the usual copper reagents rather than copper(I) oxide.¹⁶⁶ At the other end of the scale, both 12- and 24-membered rings have been formed from substituted 1,1'-binaphthyls using this methodology, although in rather poor yields (ca. 10%).¹⁶⁷ Identical types of substrates can be cyclized under somewhat milder conditions by using conditions developed by Semmelhack and his coworkers, in which [Ni(PPh₃)₄] is used in place of copper powder.¹³² This methodology has been elegantly illustrated in the natural product area by syntheses of the alnusone precursor (110) and the erythrina alkaloid skeleton (111; Scheme 49), although it is not suitable for syntheses in the steganone area (vide infra), possibly indicating a sensitivity to excessive steric crowding. Copper-mediated intramolecular coupling reactions of Grignard reagents⁵ have also found occasional use in this area, as illustrated by a preparation of the 14π -aromatic system (112; Scheme 50).¹⁶⁸ The much lower temperatures required in this case are a particularly attractive feature. Similarly, aryllithium species can be coupled intramolecularly using copper salts; the particular example shown (Scheme 51) affords a 53% yield of the dimeric product, accompanied by only 3% of the monomeric dibenzocyclobutane.⁵ More recently, palladium(II) catalysis has been found to be particularly effective for the ring closure of a monobromo precursor (Scheme 52).¹⁶⁹ The product (113), an advanced precursor of the unusual isoquinoline alkaloid ancistrocladine, is formed with reasonable stereoselectivity being accompanied by ca. 25% of the alternative diastereoisomer.





Scheme 52

O

In spite of the foregoing methods however, many of the most useful ring closure reactions of this type tend to involve either radical or aryl cationic intermediates. Although beyond the formal scope of this review, a brief survey is included here for the sake of completeness. A classical method is the Pschorr ring closure (Scheme 53), essentially an intramolecular version of the Gomberg-Bachmann-Hey process,^{161,162} which can be triggered using the usual Gomberg conditions (*i.e.* basification) or by Meerwein-type arylation procedures by treatment with Cu^I salts.¹⁷⁰ The poor yields which usually result from Pschorr closures have meant that the method does not enjoy a particularly prominent position in synthesis. However, one area where the reaction is particularly useful is in phenanthrene synthesis, especially when carried out under modified conditions. For example, when sodium iodide rather than a copper reagent is used, respectable yields can be realized (Scheme 54),¹⁷¹ while a further improvement is to position a phenylsulfonyloxy function para to the site of attack (Scheme 54; $R = OSO_2Ph$).¹⁷² A related photochemical cyclization of aryl iodides, the intramolecular version of methodology discussed above,¹⁶⁴ is also effective both for phenanthrene synthesis (Scheme 55)¹⁷³ and also for the elaboration of some alkaloid ring systems (Scheme 56).¹⁷⁴ Closely related to this is the overall oxidative ring closure between two unsubstituted positions by photolysis in the presence of iodine; the particular example shown (Scheme 57)¹⁷⁵ constitutes a key step in a synthesis of the natural alkaloid cryptopleurine. Indeed, some of the most useful methods in this area, especially for the elaboration of polymethoxylated natural products having at least one oxygen function para to the site of coupling (cf. Scheme 57), involve direct cyclization of substrates having no substituents at the two carbon sites to be coupled. In this respect, a combination of thallium(III) trifluoroacetate (TTFA) and boron trifluoride etherate is especially useful and has been used, for example, to prepare the biphenyl (114) from the unsubstituted precursor, in 81% yield, and aporphine (115), in 46% yield.¹²⁴ Similar nonphenolic oxidative couplings using vanadium(V)



Scheme 53



R = H





Scheme 55



n = 1, 57%; n = 2, 25%

Scheme 56



Scheme 57

oxyfluoride generally produce lower yields than TTFA.¹⁷⁶ Subsequently, ruthenium(IV) tetrakis(trifluoroacetate) (RUTFA) has been found to be especially useful for carrying out such couplings in the steganane area.¹⁷⁷ The reagent is prepared from ruthenium dioxide, TFA-TFAA and boron trifluoride etherate and is said to be generally superior to TTFA; a particularly impressive example is a preparation of the highly crowded steganane (116) in 90–95% isolated yield using RUTFA at ambient temperature. The reagent is also useful for aporphine synthesis (cf. 115).



2.3.4 COUPLING REACTIONS INVOLVING HETEROARYLS

2.3.4.1 Heteroaryl–Heteroaryl Coupling

Not surprisingly, many of the methods described above are equally well-suited to the coupling of heteroaromatic intermediates. The Ullmann method¹¹⁰ can be applied successfully to the dimerization of haloheteroaromatics (e.g. 117 \rightarrow 118) and to some cross-coupling reactions, where the reactivity differences¹³⁹ are sufficiently large (e.g. 119 \rightarrow 120). In general, this method is, however, less satisfactory when applied to five-membered heteroaromatics, although the intramolecular version has been used to prepare derivatives of this group (e.g. Scheme 58). Intermolecular dimerizations of furans and thiophenes are more efficiently carried out via the corresponding chloromercury derivatives, when the addition of catalytic amounts of palladium(II) chloride to a typical Ullmann system is particularly beneficial (Scheme 59).^{23,134} The corresponding heteroaryllithium species can also be dimerized by treatment with a variety of transition metal salts.⁵ The intermediacy of carbanions and the rather harsh Ullmann conditions can be avoided by direct coupling of heteroaryl halides using a variety of nickel(0) species. Although the earlier Semmelhack reagent, [Ni(COD)2], does not appear to be particularly suitable,¹²⁸ bis(trialkylphosphine)nickel(0) species can give rise to excellent yields (e.g. Scheme 60).¹⁹ In this particular case, an anionic intermediate would normally be precluded due to facile ring opening. This approach is also successful with similar coupling reactions of halothiophenes but fails with halopyridines. However, both of the latter substrate types can be dimerized using catalytic nickel salts in the presence of triphenylphosphine and a reducing metal such as zinc or magnesium.¹³⁰ An arylnickel(I) complex may well be involved. Such a system has been found to be superior to Ullmann-type methods in a synthesis of the mushroom toxin orellanine (Scheme 61).¹⁷⁸ An excellent alternative is the system NaH-Bu'ONa-Ni(OAc)₂-PPh₃ which gives high yields of homocoupled products from chloro- or bromo-pyridines and -quinolines under mild conditions (DME, 30-60 °C, 1.5-3.5 h).¹⁷⁹ It has been established for some time that the parent of this class, 2,2'-bipyridyl, can be prepared directly from pyridine by oxidative coupling using Raney nickel;¹⁸⁰ much better yields are obtained by heating 2-bromopyridine or simple alkyl derivatives with stoichiometric quantities of Raney nickel in toluene.^{181,196} Electrochemically generated palladium(0) species can also be used for the very efficient homocouplings of bromopyridines.¹³⁷





Scheme 61

OMe

Unsymmetrical bipyridyls can be readily obtained, although in variable yields (14–75%), by direct coupling reactions between pyridyl Grignard reagents and pyridyl sulfoxides.¹⁸² A more versatile type of reaction of heterocyclic Grignard reagents is their palladium(0)-catalyzed coupling reaction with heteroaryl halides.^{183,206} As well as being useful for the elaboration of a wide variety of unsymmetrical dimers, a whole range of oligomers can also be prepared using this methodology by, for example, sequential coupling with a heterocyclic dihalide, in which the two halogens have different reactivities (Scheme 62). A particularly good alternative to this procedure is the closely related Negishi method, wherein arylzinc species (derived from the corresponding aryllithium or aryl Grignard) undergo Pd⁰-catalyzed coupling reactions with aryl halides (Scheme 63).¹⁸⁴ The method is applicable to both 2- and 3-pyridylzinc chlorides. Even more general, given the availability of the starting materials, is the corresponding coupling reactions between pyridylstannanes and bromopyridines (Scheme 64).¹⁸⁵ Also applicable to quinolines and presumably many related systems, the method has one significant drawback, that of the requirement for lengthy heating (*ca.* 12 h) in xylene. Chloropyridines are inert, which should allow some useful regioselective couplings to be carried out; furthermore two groups can be added at once, as in the conversion of the distannylpyridine (**121**) into the tripyridylnicotelline (**122**). It is prob-

able that this type of coupling between a bromoaryl and an aryltin derivative will be very widely applicable and perhaps the method of choice in very many cases. For example, the 2-stannyloxazole (**123**) has been coupled with a wide variety of heteroaromatic bromides (2- and 3-thienyl, -furyl, -pyridyl and -quinonyl, 2-bromothiazole) to give excellent yields of the expected derivatives (**124**),¹⁸⁶ while a whole range of symmetrical and unsymmetrical thiazoles have been similarly prepared (Scheme 65).¹⁸⁷ However, another excellent route to unsymmetrical heterobiaryls, which also relies on a palladium(0) catalyst, is the Suzuki method,^{58–66,97–99,154,155} wherein a heteroarylboronic acid is coupled with a heteroaryl halide.^{188,212} Sequential additions of two aryl ligands to trialkoxyboranes can be used to obtain boronate species, which then collapse to give biaryls upon treatment with bromine or NBS; the methodology affords a variety of yields and is generally most suited to heteroaryl-heteroaryl coupling.¹⁸⁹

Although beyond the scope of this section, some related approaches to these compounds are worth mentioning. 2-Substituted pyridines and quinolines can be obtained in two steps from the parent compounds following nucleophilic attack by an aryllithium (*e.g.* 2-thienyllithium) and then oxidation.¹⁹⁰ Intramolecular oxidative coupling reactions between the 2-positions of two indole nuclei can be achieved using DDQ, in the presence of a trace of tosic acid.¹⁹¹ A very efficient route to a variety of unsymmetri-





cally substituted bithiophenes is by photolysis¹⁶⁴ of an iodothiophene in the presence of an excess of an α -unsubstituted thiophene.^{192,213} Finally, a wide variety of Gomberg-Bachmann-Hey processes have been used to prepare biheterocyclic systems.¹⁶²

2.3.4.2 Aryl–Heteroaryl Coupling

Almost all of the foregoing methods which can be used to prepare unsymmetrical biaryls or biheteroaryls have been, or probably could be, applied to the elaboration of targets belonging to this section. Given sufficient reactivity differences¹³⁹ between the two components, Ullmann-type coupling¹¹⁰ can be one of the simplest and most efficient methods in this area. 2-Substituted thiophenes are particularly suitable adducts (Scheme 66),¹⁹³ whereas 2-furylcoppers are not such good substrates,¹⁹⁴ although the convenience could sometimes outweigh the relative inefficiency (Scheme 67).¹²⁶ Good yields (50-80%) of 2-arylpyridines (125) can also be obtained by the Ullmann approach from 2-pyridylcopper species and aryl halides, but using rather different conditions consisting of heating the reactants together with triphenylphosphine in toluene, rather than the more usual basic solvent.¹⁹⁵ In general, 2-arylpyridines (125) are among the most accessible of this group as they can be obtained by direct nucleophilic attack by aryllithiums or Grignard reagents at the 2-position of a pyridine nucleus, followed by a facile oxidation back to the aromatic level. For example, reaction between phenyllithium and 2,2'-bipyridyl in ether affords a 54% yield of the 2-phenyl derivative (126) after oxidation.¹⁹⁶ Similarly, reaction with 2-lithiopyridine leads to the expected terpyridine.¹⁹⁷ More highly substituted pyridines can be obtained from similar reactions with stannylated pyridinium salts (Scheme 68).¹⁹⁸ The bulk of the tin substituent directs the incoming nucleophile and hence the tricyclohexylstannyl derivative is preferred. The intermediate stannyldihydropyridine can be easily acylated using an acyl chloride and a palladium catalyst and finally oxidation, using o-chloranil, regenerates the pyridine nucleus. Similarly, pyridine N-oxides can be converted to 2-arylpyridines (125) by reaction with arylmagnesium bromides followed by acetylation (Ac₂O) and spontaneous elimination of acetic acid.¹⁹⁹ Alkoxysulfonyl groups in the 2-position of a pyridine nucleus can be displaced by aryllithiums, leading directly to 2-arylpyridines (125) in good to excellent yields;²⁰⁰ halopyridines can undergo similar couplings.²⁰¹ A related approach to 4-arylpyridines relies on activation of the 4-position by an adjacent oxazoline function;¹⁴⁹ attack by phenylmagnesium bromide, or better phenyllithium, leads to a dihydropyridine and thence to the corresponding pyridine (127), following oxidation.²⁰² Presumably, other aryl- and heteroaryl-lithiums could also be used. Activation of the usually nucleophilic 3-position of an indole to nucleophilic substitution can be achieved by formation of the iodonium species (128) which upon reaction with phenyllithium, in the presence of boron trifluoride etherate, leads to 3-phenylindole (129) in 62% yield.²⁰³

In common with much of the foregoing chemistry, some of the most flexible and versatile methods in this area are based on nickel- or palladium-catalyzed coupling reactions between organometallic species



Scheme 66



and aryl halides. Particularly useful in this respect are nickel(II)-catalyzed couplings involving arylmagnesium halides;²⁶ for example, Grignard reagent (130) can be coupled to 3-bromothiophene to give the arylthiophene (131) in 67% yield.²⁰⁴ Double couplings can similarly be carried out, as in the case of a preparation of the diarylthiophene (132) from the corresponding 2,5-dibromothiophene and 2 equiv. of an arylmagnesium bromide, which requires only 1 mol % of the catalyst [Ni(DPPP)Cl₂].²⁰⁵ It is even possible to prepare unsymmetrical derivatives of thiophenes (132) by sequential coupling reactions with different Grignard reagents, but using a palladium catalyst in the first step. In addition to bromothiophenes, the methodology can also be extended to chloropyridines and various homologs, such as chloroquinolines,²⁰⁶ and to the elaboration of nucleoside analogs (133 and 134).²⁰⁷ The nickel-catalyzed displacement of alkylthio groups from sp^2 -centers using Grignard reagents^{29,79,83,146} has also been used to prepare purine analogs under closely similar conditions, (135 \rightarrow 136);²⁰⁸ the method can also be applied to highly efficient preparations of 4-phenylpyridines (137) and 2-phenylindole (138) from the respective 4- and 2-methylthio derivatives.

Equally useful palladium-catalyzed coupling reactions between aryl halides and aryl organometallics have been developed, in which the latter component is usually either a trialkylstannane derivative or the corresponding arylzinc chloride species. In contrast to the foregoing nickel-mediated couplings with





arylmagnesium halides, the palladium-catalyzed processes often work best when the organometallic component is derived from the heteroaromatic substrate which is then reacted with an aryl halide, usually the bromide or iodide. A study in which compounds (139)-(144) were prepared in generally excellent yields from the corresponding heteroaryltrimethylstannane and iodobenzene serves to exemplify the potential of this methodology.²⁰⁹ In all these examples, [PdCl₂(PPh₃)₂] was used as the catalyst; in a related study, coupling between a rather unstable 2-trimethylstannylbenzofuran and a protected iodoresorcinol has been achieved using [Pd(PPh₃)₄], leading to the arylbenzofuran (145).²¹⁰ In this example, coupling reactions with the tin derivative of the benzofuran were found to be more suitable than either use of the corresponding zinc bromide (vide infra) or coupling in the reverse sense, between the 2-bromobenzofuran and an arylzinc bromide. Particularly notable in this type of coupling is the tolerance of a wide range of functional groups, especially esters, acetates, nitriles and chlorines. Further examples of this approach are provided by coupling reactions between the 2-stannyloxazole (123) and a range of substituted bromobenzenes, including the 4-F, 4-Me, 4-SPh and 2-CN derivatives.¹⁸⁶ Also, using $[Pd(PPh_3)_4]$ as catalyst, the process is relatively mild (reflux, C₆H₆, 12–24 h) and usually very efficient (80-100% yields), even when a 2-substituent is present. The related procedure wherein arylstananes are coupled with any triflates⁹⁴ has been applied to an any heteroary example, the alkaloid dubamine (146), which was obtained in 79% yield from the corresponding 2-quinolyltriflate and 5-(trimethylstannyl)-1,3benzodioxole ([Pd(PPh₃)₄], LiCl, dioxane, 98 °C, 68 h); it seems highly probable that this approach will find wide application in the elaboration of substituted heteroaromatics.

Despite some of the foregoing observations,²¹⁰ Negishi-type coupling reactions between furylzinc chlorides and bromobenzenes catalyzed by [Pd(PPh₃)₄] have been found to be valuable for the preparation of arylfurans (Scheme 69).²¹¹ Also applicable to the corresponding thiophenes, a particularly interesting aspect of this chemistry is the coupling of the bis(zinc chloride) (147) with 1,4-dibromobenzene,





which gives rise to a series of potentially useful polymers. One or both of the zinc groups in the dianion (147) can be replaced by monobromobenzene simply by using 1 or 2 equiv. of the latter.



The Suzuki boronic acid method^{58-66,97-99,154,155,188} is also likely to enjoy many applications in this area; one example is a brief and efficient synthesis of 5-arylnicotinates (Scheme 70).²¹² One limitation of this approach, however, is its sensitivity to substituents *ortho* to the site of coupling.





A simple route to the phenylthienyl ketones and aldehydes (148) and (149) consists of photolysis of the corresponding bromothienylcarbonyl compounds in benzene,²¹³ in a similar fashion to the preparation of phenylbenzenes from iodobenzenes¹⁶⁴ and of bithiophenes from iodothiophenes.¹⁹²



2.3.5 REFERENCES

- 1. T. Cohen and T. Poeth, J. Am. Chem. Soc., 1972, 94, 4363.
- 2. T. Cohen and I. Cristea, J. Org. Chem., 1975, 40, 3649.
- R. B. Banks and H. M. Walborsky, J. Am. Chem. Soc., 1976, 98, 3733; G. M. Whitesides, J. San Filippo, Jr., C. P. Casey and E. J. Panek, J. Am. Chem. Soc., 1967, 89, 5302; T. Kauffmann, G. Beissner, E. Köppelmann, D. Kuhlmann, A. Schott and H. Schrecken, Angew. Chem., Int. Ed. Engl., 1968, 7, 131; T. Kauffmann, G. Beissner, H. Berg, E. Köppelmann, J. Legler and M. Schönfelder, Angew. Chem., Int. Ed. Engl., 1968, 7, 540; T. Kauffmann, D. Kuhlmann, W. Sahm and H. Schrecken, Angew. Chem., Int. Ed. Engl., 1968, 7, 541.
- 4. J. P. Foulon, M. Bourgain-Commerçon and J. F. Normant, Tetrahedron, 1986, 42, 1399.
- 5. For a review, see T. Kauffmann, Angew. Chem., Int. Ed. Engl., 1974, 13, 291; H. Felkin and G. Swierczewski, Tetrahedron, 1975, 31, 2735; for vinyllithium coupling using Mn^{II} salts, see G. Cahiez, D. Bernard and J. F. Normant, Synthesis, 1976, 245; J. Organomet. Chem., 1976, 113, 99.
- 6. G. M. Whitesides, C. P. Casey and J. K. Krieger, J. Am. Chem. Soc., 1971, 93, 1379.
- 7. M. Tamura and J. K. Kochi, Bull. Chem. Soc. Jpn., 1971, 44, 3063; 1972, 45, 1120.
- 8. I. Hashimoto, M. Ryang and S. Tsutsumi, Tetrahedron Lett., 1970, 4567 and refs. therein.
- 9. A. Uchida, T. Nakazawa, I. Kondo, N. Iwata and S. Matsuda, J. Org. Chem., 1972, 37, 3749.
- 10. Y. Yamamoto, H. Yatagai, K. Maruyama, A. Sonoda and S.-I. Murahashi, J. Am. Chem. Soc., 1977, 99, 5652; H. C. Brown and J. B. Campbell, Jr., J. Org. Chem., 1980, 45, 389.
- 11. J. B. Campbell, Jr. and H. C. Brown, J. Org. Chem., 1980, 45, 549.
- 12. G. Zweifel, N. L. Polston and C. C. Whitney, J. Am. Chem. Soc., 1968, 90, 6243; for an application in insect pheromone synthesis see G. Giacomelli, L. Lardicci, C. Bertucci and A. M. Caporusso, Tetrahedron, 1978, 34, 2015.
- 13. G. Zweifel and R. L. Miller, J. Am. Chem. Soc., 1970, 92, 6678; for a review of alkenylaluminum chemistry, see K. Maruoka and H. Yamamoto, Angew. Chem., Int. Ed. Engl., 1985, 24, 668.
- 14. M. Yoshifuji, M. J. Loots and J. Schwartz, Tetrahedron Lett., 1977, 1303.
- 15. S. Ghosal, G. P. Luke and K. S. Kyler, J. Org. Chem., 1987, 52, 4296.
- 16. K. Tamao, H. Matsumoto, T. Kakui and M. Kumada, Tetrahedron Lett., 1979, 1137; J. Yoshida, K. Tamao, T. Kakui and M. Kumada, Tetrahedron Lett., 1979, 1141.
- 17. M. F. Semmelhack, P. M. Helquist and J. D. Gorzynski, J. Am. Chem. Soc., 1972, 94, 9234.
- 18. A. S. Kende, L. S. Liebeskind and D. M. Braitsch, Tetrahedron Lett., 1975, 3375.
- K. Takagi, N. Hayama and K. Sasaki, Bull. Chem. Soc. Jpn., 1984, 57, 1887.
 K. Takagi and N. Hayama, Chem. Lett., 1983, 637.
- 21. R. Grigg, P. Stevenson and T. Worakun, J. Chem. Soc., Chem. Commun., 1984, 1073; 1985, 971.
- 22. F. N. Jones, J. Org. Chem., 1967, 32, 1667.
- R. C. Larock, J. Org. Chem., 1976, 41, 2241; for reviews of organomercury chemistry see R. C. Larock, Tetrahedron, 1982, 38, 1713; R. C. Larock, 'Organomercury Compounds in Organic Synthesis', Springer-Verlag, Heidelberg, 1985.
- 24. R. C. Larock and J. C. Bernhardt, J. Org. Chem., 1976, 41, 2241; 1977, 42, 1680; K. Takagi, N. Hayama, T. Okamoto, Y. Sakakibara and S. Oka, Bull. Chem. Soc. Jpn., 1977, 50, 2741; for related dimerizations of vinylstannanes see S. Kanemoto, S. Matsubara, K. Oshima, K. Utimoto and H. Nozaki, Chem. Lett., 1987, 5.
- R. C. Larock and B. Riefling, J. Org. Chem., 1978, 43, 1468.
 K. Tamao, K. Sumitani, Y. Kiso, M. Zembayashi, A. Fujioka, S. Kodama, I. Nakajima, A. Minato and M. Minato, K. Sumitani, Y. Kiso, and M. Minato, N. Minato, Kumada, Bull. Chem. Soc. Jpn., 1976, 49, 1958; K. Tamao, A. Minato, N. Miyake, T. Matsuda, Y. Kiso and M. Kumada, Chem. Lett., 1975, 133; K. Tamao, K. Sumitani and M. Kumada, J. Am. Chem. Soc., 1972, 94, 4374; K. Tamao, Y. Kiso, K. Sumitani and M. Kumada, J. Am. Chem. Soc., 1972, 94, 9268; R. J. P. Corriu and J. P. Masse, J. Chem. Soc., Chem. Commun., 1972, 144; M. Kumada, Pure Appl. Chem., 1980, 52, 669.
- 27. A. Hosomi, K. Otaka and H. Sakurai, Tetrahedron Lett., 1986, 27, 2881.
- 28. B. B. Snider, M. Karras and R. S. E. Conn, J. Am. Chem. Soc., 1978, 100, 4624.
- V. Fiandanese, G. Marchese, F. Naso and L. Ronzini, Synthesis, 1987, 1034; J. Chem. Soc., Perkin Trans. 1, 1985, 1115; V. Fiandanese, G. Miccoli, F. Naso and L. Ronzini, J. Organomet. Chem., 1986, 312, 343. 29.
- 30. H. P. Dang and G. Linstrumelle, Tetrahedron Lett., 1978, 191; D. Michelot, Synthesis, 1983, 130; for a review of Pd^o-catalyzed couplings see J. K. Stille, Angew. Chem. Int. Ed. Engl., 1986, 25, 508.
- A. Commercon, J. F. Normant and J. Villieras, J. Organomet. Chem., 1977, 128, 1.
 N. Jabri, A. Alexakis and J. F. Normant, Tetrahedron Lett., 1982, 23, 1589; for a review of vinylcuprate chemistry see J. F. Normant and A. Alexakis, Synthesis, 1981, 841.
- 33. N. Jabri, A. Alexakis and J. F. Normant, Bull. Soc. Chim. Fr., Part 2, 1983, 332.
- 34. N. Jabri, A. Alexakis and J. F. Normant, Bull. Soc. Chim. Fr., Part 2, 1983, 321; Tetrahedron Lett., 1981, 22, 959; see also ref. 89.
- 35. M. Gardette, N. Jabri, A. Alexakis and J. F. Normant, Tetrahedron, 1984, 40, 2741.
- 36. S.-I. Murahashi, M. Yamamura, K. Yanagisawa, N. Mita and K. Kondo, J. Org. Chem., 1979, 44, 2408, and refs. therein.
- 37. E. Negishi and F.-T. Luo, J. Org. Chem., 1983, 48, 1560.
- 38. C. E. Russell and L. S. Hegedus, J. Am. Chem. Soc., 1983, 105, 943.
- 39. T. L. Gilchrist and R. J. Summersell, Tetrahedron Lett., 1987, 28, 1469.
- S. Baba and E. Negishi, J. Am. Chem. Soc., 1976, 98, 6729; for a summary, see E. Negishi, Pure Appl. Chem., 1981, 53, 2333; Acc. Chem. Res., 1982, 15, 340; for a review of alkenylaluminum chemistry see G. Zweifel and J. A. Miller, Org. React. (N.Y.), 1984, 32, 375.
- 41. E. Negishi, T. Takahashi and S. Baba, Org. Synth., 1988, 66, 60.
- N. Okukado, D. E. Van Horn, W. L. Klima and E. Negishi, *Tetrahedron Lett.*, 1978, 1027.
 E. Negishi and D. E. Van Horn, *J. Am. Chem. Soc.*, 1977, 99, 3168.
- 44. E. Negishi, N. Okukado, A. O. King, D. E. Van Horn and B. I. Spiegel, J. Am. Chem. Soc., 1978, 100, 2254.

- 45. V. Ratovelomanana and G. Linstrumelle, *Tetrahedron Lett.*, 1981, 22, 315; V. Ratovelomanana, G. Linstrumelle, and J. F. Normant, *Tetrahedron Lett.*, 1985, 26, 2575; K. Tamao, M. Zembayashi, Y. Kiso and M. Kumada, J. Organomet. Chem., 1973, 55, C91; G. Consiglio and C. Botteghi, Helv. Chim. Acta, 1973, 56, 460.
- 46. V. Ratovelomanana and G. Linstrumelle, Tetrahedron Lett., 1984, 25, 6001.
- 47. V. Ratovelomanana, A. Hammoud and G. Linstrumelle, *Tetrahedron Lett.*, 1987, 28, 1649. 48. J. E. McMurry and W. J. Scott, *Tetrahedron Lett.*, 1980, 21, 4313.
- 49. W. J. Scott, G. T. Crisp and J. K. Stille, J. Am. Chem. Soc., 1984, 106, 4630; for a review of vinylstannane chemistry see J. K. Stille, Pure Appl. Chem., 1985, 57, 1771.
- 50. W. J. Scott and J. K. Stille, J. Am. Chem. Soc., 1986, 108, 3033.
- 51. M. R. Peña and J. K. Stille, Tetrahedron Lett., 1987, 28, 6573.
- 52. E. Piers, R. W. Friesen and B. A. Keay, J. Chem. Soc., Chem. Commun., 1985, 809.
- 53. E. Piers and R. W. Friesen, J. Org. Chem., 1986, 51, 3405.
- 54. C. Chan, P. B. Cox and S. M. Roberts, J. Chem. Soc., Chem. Commun., 1988, 971.
- 55. E. Piers and R. T. Skerlj, J. Chem. Soc., Chem. Commun., 1987, 1025.
- 56. R. C. Larock and D. R. Leach, Tetrahedron Lett., 1981, 22, 3435.
- 57. E. Negishi and T. Yoshida, J. Chem. Soc., Chem. Commun., 1973, 606; E. Negishi, G. Lew and T. Yoshida, J. Chem. Soc., Chem. Commun., 1973, 874; G. Zweifel and N. L. Polston, J. Am. Chem. Soc., 1970, 92, 4068.
- 58. N. Miyaura, K. Yamada and A. Suzuki, Tetrahedron Lett., 1979, 3437; A. Suzuki, Acc. Chem. Res., 1982, 15, 178; R. Rossi, A. Carpita and M. G. Quirici, Tetrahedron, 1981, 37, 2617.
- 59. N. Miyaura, H. Suginome and A. Suzuki, Tetrahedron Lett., 1981, 22, 127.
- 60. G. Cassani, P. Massardo and P. Piccardi, Tetrahedron Lett., 1983, 24, 2513.
- 61. N. Miyaura, H. Suginome and A. Suzuki, Tetrahedron, 1983, 39, 3271; F. Bjoerkling, T. Norin, C. R. Unelius and R. B. Miller, J. Org. Chem., 1987, 52, 292.
- 62. Y. Kobayashi, T. Shimazaki and F. Sato, Tetrahedron Lett., 1987, 28, 5849.
- 63. N. Miyaura, K. Yamada, H. Suginome and A. Suzuki, J. Am. Chem. Soc., 1985, 107, 972.
- 64. N. Miyaura, M. Satoh and A. Suzuki, Tetrahedron Lett., 1986, 27, 3745; T. Ishiyama, N. Miyaura and A. Suzuki, Chem. Lett., 1987, 25.
- 65. M. Satoh, N. Miyaura and A. Suzuki, Chem. Lett., 1986, 1329
- 66. S. Hyuga, S. Takinami, S. Hara and A. Suzuki, Chem. Lett., 1986, 459.
- 67. F. Toda and Y. Takehira, J. Chem. Soc., Chem. Commun., 1975, 174; for a recent review of allene chemistry see D. J. Pasto, Tetrahedron, 1984, 40, 2805
- 68. K. Ruitenberg, H. Kleijn, C. J. Elsevier, J. Meijer and P. Vermeer, Tetrahedron Lett., 1981, 22, 1451.
- 69. C. J. Elsevier and P. Vermeer, J. Org. Chem., 1985, 50, 3042.
 70. A. M. Caporusso, C. Polizzi and L. Lardicci, Tetrahedron Lett., 1987, 28, 6073; for a review of higher order cuprates see B. H. Lipshutz, R. S. Wilhelm and J. A. Kozlowski, Tetrahedron, 1984, 40, 5005.
- T. Jeffery-Luong and G. Linstrumelle, Tetrahedron Lett., 1980, 21, 5019.
 D. Michelot, J.-C. Clinet and G. Linstrumelle, Synth. Commun., 1982, 12, 739; T. Jeffery-Luong and G. Linstrumelle, Synthesis, 1982, 738; P. Pappalardo, E. Ehlinger and P. Magnus, Tetrahedron Lett., 1982, 23, 309.
- 73. T. M. Bargar, J. R. McCowan, J. R. McCarthy and E. R. Wagner, J. Org. Chem., 1987, 52, 678.
- K. Tamao, M. Zembayashi and M. Kumada, Chem. Lett., 1976, 1237, 1239.
 E. Wenkert, E. L. Michelotti and C. S. Swindell, J. Am. Chem. Soc., 1979, 101, 2246; E. Wenkert, E. L. Michelotti, C. S. Swindell and M. Tingoli, J. Org. Chem., 1984, 49, 4894.
- 76. T. Hayashi, Y. Katsuro and M. Kumada, Tetrahedron Lett., 1980, 21, 3915.
- C. Sahlberg, A. Quader and A. Claesson, Tetrahedron Lett., 1983, 24, 5137.
 T. Hayashi, Y. Katsuro, Y. Okamoto and M. Kumada, Tetrahedron Lett., 1981, 22, 4449.
- 79. V. Fiandanese, G. Marchese, F. Naso and L. Ronzini, J. Chem. Soc., Chem. Commun., 1982, 647; H. Okamura, M. Miura and H. Takei, Tetrahedron Lett., 1979, 43.
- W. H. Pearson, J. E. Celebuski, Y.-F. Poon, B. R. Dixon and J. H. Glans, *Tetrahedron Lett.*, 1986, 27, 6301.
 J.-L. Fabre, M. Julia and J.-N. Verpeaux, *Tetrahedron Lett.*, 1982, 23, 2469.
 H. Okamura, M. Miura, K. Kosugi and H. Takei, *Tetrahedron Lett.*, 1980, 21, 87.

- 83. M. Tiecco, L. Testaferri, M. Tingoli and E. Wenkert, Tetrahedron, 1983, 39, 2289.
- 84. M. Tamura and J. K. Kochi, J. Am. Chem. Soc., 1971, 93, 1487; S. M. Neumann and J. K. Kochi, J. Org. Chem., 1975, 40, 599; T. Amano, T. Ota, K. Yoshikawa, T. Sano, Y. Ohuchi, F. Sato, M. Shiono and Y. Fujita, Bull. Chem. Soc. Jpn., 1986, 59, 1656.
- 85. G. A. Molander, B. J. Rahn, D. C. Shubert and S. E. Bonde, Tetrahedron Lett., 1983, 24, 5449.
- 86. L. Crombie, N. A. Kerton and G. Pattenden, J. Chem. Soc., Perkin Trans. 1, 1979, 2136 and refs. therein.
- 87. G. M. Whitesides, W. F. Fischer, Jr., J. San Filippo, Jr., R. W. Bashe and H. O. House, J. Am. Chem. Soc., 1969, **91**, 4871.
- 88. P. J. Beswick and D. A. Widdowson, Synthesis, 1985, 492.
- 89. N. Jabri, A. Alexakis and J. F. Normant, Tetrahedron Lett., 1981, 22, 3851; S. Nunomoto, Y. Kawakami and Y. Yamashita, J. Org. Chem., 1983, 48, 1912.
- 90. R. B. Miller and M. I. Al-Hassan, J. Org. Chem., 1985, 50, 2121 and refs. therein.
- 91. E. Negishi and S. Baba, J. Chem. Soc., Chem. Commun., 1976, 596.
- 92. M. I. Al-Hassan, Synthesis, 1987, 816.
- 93. (a) D. R. McKean, G. Parrinello, A. F. Renaldo and J. K. Stille, J. Org. Chem., 1987, 52, 422 and refs. therein; (b) T. N. Mitchell, R. Wickenkamp, A. Amamria, R. Dicke and U. Schneider, J. Org. Chem., 1987, 52, 4868.
- 94. A. M. Echavarren and J. K. Stille, J. Am. Chem. Soc., 1987, 109, 5478.
- 95. A. Hallberg and C. Westerlund, Chem. Lett., 1982, 1993.
- 96. K. Kikukawa, K. Ikenaga, F. Wada and T. Matsuda, Tetrahedron Lett., 1984, 25, 5789.
- 97. N. Miyaura and A. Suzuki, J. Chem. Soc., Chem. Commun., 1979, 866.

- 98. N. Miyaura and A. Suzuki, J. Organomet. Chem., 1981, 213, C53.
- 99. N. Miyaura, K. Maeda, H. Suginome and A. Suzuki, J. Org. Chem., 1982, 47, 2117; M. Satoh, N. Miyaura and A. Suzuki, Synthesis, 1987, 373.
- 100. R. C. Larock, K. Narayanan and S. S. Hershberger, J. Org. Chem., 1983, 48, 4377.
- 101. For a review see C. S. Rondestvedt, Jr., Org. React. (N. Y.), 1960, 11, 189; for a recent example see S. J. F. MacDonald, T. C. McKenzie and W. D. Hassen, J. Chem. Soc., Chem. Commun., 1987, 1528.
 102. For a review of stilbene synthesis see K. B. Becker, Synthesis, 1983, 341.
- 103. K. Kikukawa, K. Ikenaga, F. Wada and T. Matsuda, Chem. Lett., 1983, 1337.
 104. A. Minato, K. Suzuki, K. Tamao and M. Kumada, Tetrahedron Lett., 1984, 25, 83.
- 105. E. Negishi, F.-T. Luo, R. Frisbee and H. Matsushita, Heterocycles, 1982, 18, 117.
- 106. M. A. Tius and S. Trehan, J. Org. Chem., 1986, 51, 765.
- 107. J. P. Gillet, R. Sauvêtre and J. F. Normant, Synthesis, 1986, 538.
- L. S. Hegedus, J. L. Toro, W. H. Miles and P. J. Harrington, J. Org. Chem., 1987, 52, 3319.
 M. Ishikura, M. Kamada, T. Ohta and M. Terashima, *Heterocycles*, 1984, 22, 2475.
- 110. For reviews see P. E. Fanta, Chem. Rev., 1946, 38, 139; 1964, 64, 613; R. G. R. Bacon and H. A. O. Hill, Q.
- Rev., Chem. Soc., 1965, 19, 95; P. E. Fanta, Synthesis, 1974, 9; A. E. Jukes, Adv. Organomet. Chem., 1974, 12, 215; for general overviews of biaryl synthesis see M. Sainsbury, Tetrahedron, 1980, 36, 3327 and ref. 130.
- 111. For a preparation of activated Cu bronze see R. C. Fuson and E. A. Cleveland, Org. Synth., Coll. Vol., 1955, 3, 339; A. I. Vogel, in 'A Textbook of Practical Organic Chemistry', 3rd edn., Wiley, New York, 1957, p. 192.
- 112. R. G. R. Bacon and O. J. Stewart, J. Chem. Soc. C, 1965, 4953.
- 113. A. Lüttringhaus and H. W. Dirksen, Angew. Chem., Int. Ed. Engl., 1964, 3, 260.
- 114. A. H. Lewin, M. J. Zovko, W. H. Rosewater and T. Cohen, J. Chem. Soc., Chem. Commun., 1967, 80.
- 115. P. H. Gore and G. K. Hughes, J. Chem. Soc., 1959, 1615.
- 116. R. D. Rieke and L. D. Rhyne, J. Org. Chem., 1979, 44, 3445; H. Matsumoto, S.-I. Inaba and R. D. Rieke, J. Org. Chem., 1983, 48, 840.
- 117. T. Cohen and J. G. Tirpak, Tetrahedron Lett., 1975, 143.
- B. J. Hathaway, D. G. Holah and J. D. Postlethwaite, J. Chem. Soc., 1961, 3215. 118.
- 119. M. Nilsson and O. Wennerström, Tetrahedron Lett., 1968, 3307; M. Nilsson and O. Wennerström, Acta Chem. Scand., 1970, 24, 482.
- C. Bjorklund, M. Nilsson and O. Wennerström, Acta Chem. Scand., 1970, 24, 3599; G. Lindsten, O. Wennerström and R. Isaksson, J. Org. Chem., 1987, 52, 547. 120.
- 121. G. Büchi, D. H. Klaubert, R. C. Shank, S. M. Weinreb and G. N. Wogan, J. Org. Chem., 1971, 36, 1143.
- 122. A. Cairncross and W. A. Sheppard, J. Am. Chem. Soc., 1968, 90, 2186.
- A. McKillop, L. F. Elsom and E. C. Taylor, J. Am. Chem. Soc., 1968, 90, 2423; A. McKillop, L. F. Elsom and E. C. Taylor, Tetrahedron, 1970, 26, 4041; L. F. Elsom, A. McKillop and E. C. Taylor, Org. Synth., 1976, 55, 48.
- 124. A. McKillop, A. G. Turrell and E. C. Taylor, J. Org. Chem., 1977, 42, 764; E. C. Taylor, J. G. Andrade and A. McKillop, J. Chem. Soc., Chem. Commun., 1977, 538; A. McKillop, A. G. Turrell, D. W. Young and E. C. Taylor, J. Am. Chem. Soc., 1980, 102, 6504.
- 125. A. D. Ryabov, S. A. Deiko, A. K. Yatsimirsky and I. V. Berezin, Tetrahedron Lett., 1981, 22, 3793.
- 126. F. D. King and D. R. M. Walton, Synthesis, 1976, 40.
- 127. I. Lantos and B. Loev, Tetrahedron Lett., 1975, 2011; for an alternative strategy see T.-I. Ho, C.-S. Shu, M. K. Yeh and F.-C. Chen, Synthesis, 1987, 795.
- 128. M. F. Semmelhack, P. M. Helquist and L. D. Jones, J. Am. Chem. Soc., 1971, 93, 5908. 129. M. Zembayashi, K. Tamao, J. Yoshida and M. Kumada, Tetrahedron Lett., 1977, 4089.
- 130. I. Colon and D. R. Kelsey, J. Org. Chem., 1986, 51, 2627
- 131. K. Takagi, N. Hayama and S. Inokawa, Chem. Lett., 1979, 917; K. Takagi, N. Hayama and S. Inokawa, Bull. Chem. Soc. Jpn., 1980, 53, 3691.
- 132. M. F. Semmelhack, P. M. Helquist, L. D. Jones, L. Keller, L. Mendelson, L. S. Ryono, J. G. Smith and R. D. Stauffer, J. Am. Chem. Soc., 1981, 103, 6460; S. Brandt, A. Marfat and P. Helquist, Tetrahedron Lett., 1979, 2193.
- J. Yamashita, Y. Inoue, T. Kondo and H. Hashimoto, Chem. Lett., 1986, 407.
 R. A. Kretchmer and R. Glowinski, J. Org. Chem., 1976, 41, 2661.
- 135. F. R. S. Clark, R. O. C. Norman and C. B. Thomas, J. Chem. Soc., Perkin Trans. 1, 1975, 121.
- 136. R. Nakajima, Y. Shintani and T. Hara, Bull. Chem. Soc. Jpn., 1980, 53, 1767.
- 137. S. Torii, H. Tanaka and K. Morisaki, Tetrahedron Lett., 1985, 26, 1655.
- 138. P. Bamfield and P. M. Quan, Synthesis, 1978, 537.
- 139. E. Brown and J.-P. Robin, Tetrahedron Lett., 1977, 2015; 1978, 3613.
- 140. F. E. Ziegler, K. W. Fowler and S. J. Kanfer, J. Am. Chem. Soc., 1976, 98, 8282; F. E. Ziegler, K. W. Fowler and N. D. Sinha, Tetrahedron Lett., 1978, 2767; F. E. Ziegler, I. Chliwner, K. W. Fowler, S. J. Kanfer, S. J. Kuo and N. D. Sinha, J. Am. Chem. Soc., 1980, 102, 790; A. S. Kende and D. P. Curran, J. Am. Chem. Soc., 1979, 101, 1857.
- 141. J. Cornforth, A. F. Sierakowski and T. W. Wallace, J. Chem. Soc., Chem. Commun., 1979, 294; J. Cornforth, A. F. Sierakowski and T. W. Wallace, J. Chem. Soc., Perkin Trans. 1, 1982, 2299.
- 142. E. Ibuka, S. Ozasa, Y. Fujioka, M. Okada and K. Terada, Bull. Chem. Soc. Jpn., 1980, 53, 821.
- 143. L. N. Pridgen, J. Org. Chem., 1982, 47, 4319 and refs. therein.
- 144. S. Miyano, S.-i. Okada, T. Suzuki, S. Handa and H. Hashimoto, Bull. Chem. Soc. Jpn., 1986, 59, 2044.
- M. Kumada, K. Tamao and K. Sumitani, Org. Synth., 1978, 58, 127.
 M. Tiecco, L. Testaferri, M. Tingoli, D. Chianelli and E. Wenkert, Tetrahedron Lett., 1982, 23, 4629; E. Wenkert and T. W. Ferreira, J. Chem. Soc., Chem. Commun., 1982, 840.
- 147. A. Sekiya and N. Ishikawa, J. Organomet. Chem., 1976, 118, 349; 1977, 125, 281.

- 148. D. A. Widdowson and Y.-Z. Zhang, Tetrahedron, 1986, 42, 2111.
- 149. R. C. Fuson and F. W. Wassmundt, J. Am. Chem. Soc., 1956, 78, 5409; J. E. Rice, H. C. Shih, N. Hussain and E. J. LaVoie, J. Org. Chem., 1987, 52, 849; A. I. Meyers and K. A. Lutomski, Synthesis, 1983, 105; D. J. Cram, J. A. Bryant and K. M. Doxsee, Chem. Lett., 1987, 19; for a review see M. Reuman and A. I. Meyers, Tetrahedron, 1985, 41, 837; A. Risaliti, S. Bozzini and A. Stener, Tetrahedron, 1969, 25, 143; for further applications of the Meyers oxazoline method see J. A. Findlay, A. Daljeet, P. J. Murray and R. N. Rej, Can. J. Chem., 1987, 65, 427; A. D. Patten, N. H. Nguyen and S. J. Danishefsky, J. Org. Chem., 1988, 53, 1003.
- 150. E. Negishi, A. O. King and N. Okukado, J. Org. Chem., 1977, 42, 1821.
- 151. E. Negishi, T. Takahashi and A. O. King, Org. Synth., 1988, 66, 67.
- 152. A. N. Kashin, I. G. Bumagina, N. A. Bumagin, I. P. Beletskaya and O. A. Reutov, Izv. Akad. Nauk SSSR, Ser. Khim., 1980, 479.
- 153. D. Milstein and J. K. Stille, J. Am. Chem. Soc., 1979, 101, 4992.
- 154. N. Miyaura, T. Yanagi and A. Suzuki, Synth. Commun., 1981, 11, 513; R. B. Miller and S. Dugar, Organometallics, 1984, 3, 1261; M. E. Jung and Y. H. Jung, Tetrahedron Lett., 1988, 29, 2517
- 155. M. J. Sharp, W. Cheng and V. Snieckus, Tetrahedron Lett., 1987, 28, 5093 and refs. therein; W. Cheng and V. Snieckus, Tetrahedron Lett., 1987, 28, 5097; M. J. Sharp and V. Snieckus, Tetrahedron Lett., 1985, 26, 5997.
- 156. A. B. Pierini, M. T. Baumgartner and R. Rossi, Tetrahedron Lett., 1987, 28, 4653 and refs. therein.
- 157. R. Beugelmans, M. Bois-Choussy and Q. Tang, Tetrahedron Lett., 1988, 29, 1705 and refs. therein.
- 158. D. H. R. Barton, D. J. Lester, W. B. Motherwell and M. T. B. Papoula, J. Chem. Soc., Chem. Commun., 1980, 246.
- 159. D. H. R. Barton, N. Y. Bhatnagar, J.-P. Finet and W. B. Motherwell, Tetrahedron, 1986, 42, 3111.
- 160. H. C. Bell, J. R. Kalman, G. L. May, J. T. Pinhey and S. Sternhell, Aust. J. Chem., 1979, 32, 1531.
- 161. M. Gomberg and W. E. Bachmann, Org. Synth., Coll. Vol., 1932, 1, 113; W. E. Bachmann and R. A. Hoffman, Org. React. (N.Y.), 1944, 2, 224; O. C. Dermer and M. T. Edmison, Chem. Rev., 1957, 57, 77; J. I. G. Cadogan, J. Chem. Soc., 1962, 4257.
- 162. G. Vernin, H. J.-M. Dou and J. Metzger, Bull. Soc. Chim. Fr., 1972, 1173; D. H. Hey and M. J. Perkins, Org. Synth. Coll. Vol., 1973, 5, 51; D. R. Augood and G. H. Williams, Chem. Rev., 1957, 57, 123.
- 163. S. H. Korzeniowski, L. Blum and G. W. Gokel, Tetrahedron Lett., 1977, 1871.
- W. Wolf and N. Kharasch, J. Org. Chem., 1965, 30, 2493; for a review see R. K. Sharma and N. Kharasch, 164. Angew. Chem., Int. Ed. Engl., 1968, 7, 36.
- 165. E. C. Taylor, F. Kienzle and A. McKillop, J. Am. Chem. Soc., 1970, 92, 6088.
- 166. W. C. Lothrop, J. Am. Chem. Soc., 1941, 63, 1187; J. C. Salfeld and E. Baume, Tetrahedron Lett., 1966, 3365; J. Hine and K. Ahn, J. Org. Chem., 1987, 52, 2083.
- S. Miyano, S. Handa, M. Tobita and H. Hashimoto, Bull. Chem. Soc. Jpn., 1986, 59, 235 and refs. therein.
 R. H. Mitchell, B. N. Ghose and M. E. Williams, Can. J. Chem., 1977, 55, 210.
- 169. G. Bringmann, J. R. Jansen and H.-P. Rink, Angew. Chem., Int. Ed. Engl., 1986, 25, 913; D. E. Ames and A. Opalko, Tetrahedron, 1984, 40, 1919.
- 170. For a review see P. H. Leake, Chem. Rev., 1956, 56, 27; for an electrochemical procedure see R. M. Elofson and F. F. Gadallah, J. Org. Chem., 1971, 36, 1769 and refs. therein.
- 171. B. Chauncy and E. Gellert, Aust. J. Chem., 1969, 22, 993; 1970, 23, 2503.
- R. I. Duclos, Jr., J. S. Tung and H. Rapoport, J. Org. Chem., 1984, 49, 5243.
 S. M. Kupchan and H. C. Wormser, J. Org. Chem., 1965, 30, 3792; B. S. Thyagarajan, N. Kharasch, H. B. Lewis and W. Wolf, J. Chem. Soc., Chem. Commun., 1967, 614.
- 174. P. W. Jeffs and J. F. Hansen, J. Am. Chem. Soc., 1967, 89, 2798.
- 175. H. Iida, Y. Watanabe, M. Tanaka and C. Kibayashi, J. Org. Chem., 1984, 49, 2412.
- S. M. Kupchan, O. P. Dhingra, C.-M. Kim and V. Kameswaran, J. Org. Chem., 1978, 43, 2521.
 Y. Landais and J.-P. Robin, Tetrahedron Lett., 1986, 27, 1785; Y. Landais, A. Lebrun and J.-P. Robin, Tetrahedron Lett., 1986, 27, 5377; Y. Landais, D. Rambault and J.-P. Robin, Tetrahedron Lett., 1987, 28, 543; Y. Landais, A. Lebrun, V. Lenain and J.-P. Robin, Tetrahedron Lett., 1987, 28, 5161.
- 178. M. Tiecco, M. Tingoli, L. Testaferri, D. Chianelli and E. Wenkert, Tetrahedron, 1986, 42, 1475; E. V. Dehmlow and H.-J. Schulz, Tetrahedron Lett., 1985, 26, 4903; M. Tiecco, L. Testaferri, M. Tingoli, D. Chianelli and M. Montanucci, Synthesis, 1984, 736.
- 179. R. Vanderesse, M. Lourak, Y. Fort and P. Caubere, Tetrahedron Lett., 1986, 27, 5483; for an examination of various 2,2'-bipyridyl preparations see G. R. Newkome, W. E. Puckett, G. E. Kiefer, V. K. Gupta, Y. Xia, M. Coreil and M. A. Hackney, J. Org. Chem., 1982, 47, 4116.
- 180. W. H. F. Sasse, Org. Synth. Coll. Vol., 1973, 5, 102.
- 181. T. Rode and E. Breitmaier, Synthesis, 1987, 574.
- 182. N. Furukawa, T. Shibutani and H. Fujihara, Tetrahedron Lett., 1987, 28, 5845.
- 183. A. Minato, K. Suzuki, K. Tamao and M. Kumada, J. Chem. Soc., Chem. Commun., 1984, 511.
- 184. A. S. Bell, D. A. Roberts and K. S. Ruddock, Synthesis, 1987, 843.
- 185. Y. Yamamoto, Y. Azuma and H. Mitoh, Synthesis, 1986, 564.
- 186. A. Dondoni, G. Fantin, M. Fogagnolo, A. Medici and P. Pedrini, Synthesis, 1987, 693; A. Dondoni, M. Fogagnolo, G. Fantin, A. Medici and P. Pedrini, Tetrahedron Lett., 1986, 27, 5269.
- 187. A. Dondoni, M. Fogagnolo, A. Medici and E. Negrini, Synthesis, 1987, 185.
- 188. S. Gronowitz and K. Lawitz, Chem. Scr., 1984, 24, 5 and refs. therein; W. J. Thompson and J. Gaudino, J. Org. Chem., 1984, 49, 5237.
- 189. G. M. Davies, P. S. Davies, W. E. Paget and J. M. Wardleworth, Tetrahedron Lett., 1976, 795; A. Pelter, H. Williamson and G. M. Davies, *Tetrahedron Lett.*, 1984, **25**, 453. 190. H. Gilman and D. A. Shirley, J. Am. Chem. Soc., 1949, **71**, 1870.
- 191. S. M. Weinreb, R. S. Garigipati and J. A. Gainor, Heterocycles, 1984, 21, 309; R. P. Joyce, J. A. Gainor and S. M. Weinreb, J. Org. Chem., 1987, 52, 1177; J. Bergman and B. Pelcman, Tetrahedron Lett., 1987, 28, 4441.
- 192. M. D'Auria, A. De Mico, F. D'Onofrio and G. Piancatelli, J. Org. Chem., 1987, 52, 5243.

- 193. M. Nilsson and C. Ullenius, Acta Chem. Scand., 1970, 24, 2379.
- 194. C. Ullenius, Acta Chem. Scand., 1972, 26, 3383.
- 195. H. Malmberg and M. Nilsson, Tetrahedron, 1986, 42, 3981.
- 196. T. Kauffmann, J. König and A. Woltermann, Chem. Ber., 1976, 109, 3864.
- 197. For a review of dihydropyridine synthesis by nucleophilic attack on a pyridine nucleus see D. M. Stout and A. I. Meyers, Chem. Rev., 1982, 82, 223.
- 198. D. L. Comins and N. B. Mantlo, Tetrahedron Lett., 1987, 28, 759 and refs. therein; D. L. Comins and S. O'Connor, Tetrahedron Lett., 1987, 28, 1843; D. L. Comins and J. D. Brown, Tetrahedron Lett., 1986, 27, 4549; M. Natsume, I. Utsunomiya, K. Yamaguchi and S. Sakai, Tetrahedron, 1985, 41, 2115.
- 199. T. Kato and H. Yamanaka, J. Org. Chem., 1965, 30, 910; T. J. Van Bergen and R. M. Kellogg, J. Org. Chem., 1971, 36, 1705.
- 200. S. Hanessian and M. Kagotani, Synthesis, 1987, 409.
- 201. T. W. Bell, L.-Y. Hu and S. V. Patel, J. Org. Chem., 1987, 52, 3847; J. F. Bunnett and R. E. Zahler, Chem. Rev., 1951, 49, 273.
- 202. A. I. Meyers and R. A. Gabel, J. Org. Chem., 1982, 47, 2633.
 203. R. M. Moriarty, Y. Y. Ku, M. Sultana and A. Tuncay, Tetrahedron Lett., 1987, 28, 3071.
- 204. G. D. Hartman, W. Halczenko and B. T. Phillips, J. Org. Chem., 1986, 51, 142.
- 205. A. Minato, K. Tamao, K. Suzuki and M. Kumada, Tetrahedron Lett., 1980, 21, 4017.
- 206. K. Tamao, S. Kodama, I. Nakajima, M. Kumada, A. Minato and K. Suzuki, Tetrahedron, 1982, 38, 3347.
- 207. D. E. Bergstrom and P. A. Reday, Tetrahedron Lett., 1982, 23, 4191; N. Công-Danh, J.-P. Beaucourt and L. Pichat, Tetrahedron Lett., 1979, 3159.
- 208. E. Wenkert, J. M. Hanna, Jr., M. H. Leftin, E. L. Michelotti, K. T. Potts and D. Usifer, J. Org. Chem., 1985, 50, 1125; H. Sugimura and H. Takei, Bull. Chem. Soc. Jpn., 1985, 58, 664.
- 209. T. R. Bailey, *Tetrahedron Lett.*, 1986, 27, 4407 and refs. therein. 210. J. M. Clough, I. S. Mann and D. A. Widdowson, *Tetrahedron Lett.*, 1987, 28, 2645.
- 211. A. Pelter, M. Rowlands and G. Clements, Synthesis, 1987, 51; A. Pelter and M. Rowlands, Tetrahedron Lett., 1987, 28, 1203; A. Pelter, M. Rowlands and I. H. Jenkins, Tetrahedron Lett., 1987, 28, 5213.
- 212. W. J. Thompson and J. Gaudino, J. Org. Chem., 1984, 49, 5237.
- 213. R. Antonioletti, M. D'Auria, F. D'Onofrio, G. Piancatelli and A. Scettri, J. Chem. Soc., Perkin Trans. 1, 1986, 1755.

2.4 Coupling Reactions Between *sp*² and *sp* Carbon Centers

KENKICHI SONOGASHIRA

Osaka City University, Japan

2.4.1	INTRODUCTION	521
2.4.2	CROSS-COUPLING REACTIONS BETWEEN ORGANOMETALLIC REAGENTS AND VINYL, ARYL AND ALKYNYL HALIDES AND RELATED COMPOUNDS	522
2.4 2.4	1.2.1 Organocopper Reagents 1.2.2 Alkynylboranes	522 523
2.4.3	PALLADIUM- AND NICKEL-CATALYZED CROSS-COUPLING REACTIONS	523
2.4 2.4 2.4 2.4 2.4 2.4	 A.1 Alkynylzinc Chlorides A.2 Alkynylmagnesium Halides A.3 Alkynyltin Reagents A Reactions of 1-Alkenyl Metals with 1-Alkynyl Halides A.3.5 Reactions of Terminal Alkynes with sp²-Carbon Halides 2.4.3.5.1 General aspects 2.4.3.5.2 Synthesis of terminal alkynes 2.4.3.5.3 Stereospecific synthesis 2.4.3.5.4 Phase-transfer catalysis 2.4.3.5.5 Syntheses of Aromatic and Heterorings via sp²-sp coupling reactions 3.6 Palladium-catalyzed Vinylation of Alkynic Iodides 	524 527 529 529 530 530 531 539 539 541 544
2.4.4	APPLICATIONS TO THE SYNTHESIS OF NATURAL PRODUCTS	545
2.4.5	REFERENCES	548

2.4.1 INTRODUCTION

A σ -bond between sp^2 - and sp-carbon centers can be formed either by vinylation of a 1-alkyne or an organometallic alkynyl, or by alkynylation of a vinyl organometallic reagent.¹⁻³ Several copper(I) alkynides undergo successful coupling reactions with halogen compounds in which the halogen is linked to an sp^2 -carbon center (vinylic, allenic, aromatic and heteroaromatic carbon). In all cases, however, a stoichiometric amount of a copper(I) salt appears to be necessary, which makes the synthesis less attractive for performance on a large scale.⁴⁻⁶

During the past 15 years, enormous advances have been made in the area of transition metal mediated reactions in organic synthesis. Two important transition metal mediated reactions are available for the coupling of alkynes with the halogen-bearing sp^2 -carbon compounds mentioned above. These are: (i) reaction of the free alkyne with the halide in the presence of a catalytic amount of Pd⁰ or Pd^{II} complexes, copper(I) iodide and an excess of an amine; here the amine is not only used as the solvent but also as a scavenger for the hydrogen halides generated during the reaction (equation 1);⁷ and (ii) reaction of metallated alkynes, preferably alkynylzinc halides, with the halogen compound in the presence of catalytic amounts of Pd^{II} or Pd⁰ compounds (equation 2).^{8,9} As usual, the reactivity order with respect to the

halogen bound to the sp^2 -carbon is I > Br > Cl² and with respect to the sp^2 -center is vinylic > allenic > heteroaromatic > aromatic.



2.4.2 CROSS-COUPLING REACTIONS BETWEEN ORGANOMETALLIC REAGENTS AND VINYL, ARYL AND ALKYNYL HALIDES AND RELATED COMPOUNDS

2.4.2.1 Organocopper Reagents

Organocopper and organolithium alkylcuprate reagents are finding ever-increasing usage in organic synthesis. Normally, these reagents are produced from the reaction between an organolithium or Grignard reagent and an appropriate copper(I) salt. Accordingly, the types of functional groups that can be tolerated are rather limited.^{4–6} Rieke and his coworkers¹⁰ have reported the preparation of a highly reactive zerovalent copper slurry, which readily undergoes oxidative addition to aryl, vinyl, alkynyl and alkyl halides under mild conditions. The corresponding organocopper reagents are stable and they undergo the usual types of homo- and cross-coupling reactions. Moreover, these reagents can be prepared in the presence of a variety of functional groups such as nitro, nitrile, ester and ketone.

The reaction of aryl halides with copper(I) alkynides is known as the Castro reaction (equation 3).⁴ The reaction has proved to be particularly important in the synthesis of a wide range of tolan and heteroaromatic alkynes. Vinyl and allenic halides¹¹⁻¹³ can also be used and several reviews of the reaction have been published.^{5,6,14-16}

$$Ar - Cu + XAr' - View - Ar' - Ar' - (3)$$

Aryl and alkenyl copper compounds also couple reasonably well with haloalkynes, leading to conjugated enynes (equations 4 and 5).^{17,18} Normant *et al.* have reported that TMEDA is essential to promote the clean coupling of alkenylcopper reagents with 1-halo-1-alkynes (equation 5).^{18,19} Alkenylcopper(I) derivatives undergo substitution with retention of configuration about the C—C double bond, leading to conjugated and functionalized enynes. However, Brown and his coworkers have pointed out that the TMEDA is not necessary and is, in fact, detrimental to the desired process.²⁰ Thus, alkenylcopper intermediates, which are readily generated from alkenylboron derivatives of 9-BBN, undergo coupling with 1-haloalkynes to provide stereodefined conjugated enynes of high isomeric purity and in 90–98% yields (Scheme 1). The same coupling reaction can be catalyzed using bis (acetylacetonate)copper.²¹

More recently, the stereoselective formation of 1,3-enynes has been reported, *via* a new carbon–carbon bond-forming reaction, involving the direct coupling of alkenyliodinium tosylates (1) with alkenylcopper(I) reagents.²² A wide variety of alkynylcopper reagents (2) are easily made by addition of alkylcop-



Coupling Reactions Between sp² and sp Carbon Centers



Scheme 1

per reagents to terminal alkynes. The reaction is highly selective and occurs with complete retention of the alkenyl geometry. Furthermore, this stereospecific high-yielding route to conjugated enynes has been utilized in the synthesis of insect pheromones and other interesting natural products. What makes the method particularly attractive, as demonstrated by Scheme 2, is that by simple choice of the order of addition of the alkenylcopper reagent to the alkyne, either geometric isomer around the C---C double bond in the 1,3-enyne product may be obtained in pure form. This simple process, complementing the Pd-Cucatalyzed sp^2 -sp carbon couplings, will be of considerable synthetic utility (see Section 2.4.3.5).



2.4.2.2 Alkynylboranes

The reactions of alkenyl- and alkynyl-boranes can provide efficient and selective methods for the coupling between two unlike boron-bound groups. General, highly stereoselective (>99%) syntheses of conjugated *trans*-enynes can be obtained by treatment of alkenylalkynylboranes (4), derived from the alkenylborane (3) and alkynyllithiums, with iodine (Scherne 3).²³ In addition, the alkynyl-9-BBN·THF complexes (5) undergo a facile reaction with enol alkyl ethers of β -keto aldehydes in hexane at room temperature to provide conjugated enynones (6) in excellent yields (Scherne 4).²⁴

The (Z)-alkynylhaloalkenes (7) can be synthesized in relatively good yields from 1-alkynes in a regioand stereo-selective manner, following haloboration with Br-9-BBN and subsequent treatment with lithium alkynides and iodine (Scheme 5).^{25,26}

2.4.3 PALLADIUM- AND NICKEL-CATALYZED CROSS-COUPLING REACTIONS

Transmetallations from main group metals to organometallic species produced by oxidative addition processes are becoming increasingly important in sp^2-sp carbon-coupling reactions (Scheme 6). In contrast to sp^2-sp^2 or sp^3-sp^2 coupling reactions, the product yields in Ni-catalyzed sp^2-sp coupling processes such as aryl-alkynyl and alkenyl-akynyl have been substantially lower than those realized with Pd catalysts, presumably because the alkynyl starting compound and/or product tends to react with the Ni catalyst to afford polymers and cyclic oligomers.



R = R' = alkyl, cycloalkyl; X = Br, I

Scheme 5

A species such as (8), formed *via* transmetallation, often undergoes reductive elimination to complete a very useful coupling reaction. Thus, magnesium,²⁷ boron, zinc and tin are effective. The major asset of this method is the complete retention of the configuration in the alkene, resulting in products of exceptional isomeric purity. The halides involved include aromatic, alkenyl and allenyl. Several reviews of this method have been published.^{8,9,28-34}

2.4.3.1 Alkynylzinc Chlorides

The Pd-catalyzed reactions of alkynylzinc chlorides, which are readily obtainable from zinc chloride, with alkenyl iodides or bromides provide the corresponding terminal or internal enynes in high yield and with high (>97%) stereospecificity (Scheme 7).³⁵ Aryl bromides or iodides can also be used to produce arylalkynes (equation 6).^{36,37}

The interesting stereospecific Pd-catalyzed cross-coupling reactions of 1-alkynylzinc chlorides (9) with a diastereoisomeric mixture of alkenyl halides (10) have been described. The (E)-bromoalkene reacts preferentially in these reactions to afford good yields of (E)-enynes (11), (12) and (13) having very



i, oxidative addition; ii, transmetallation; iii, reductive elimination

Scheme 6



$$R^1 = H, Bu^n, CO_2Me; R^2 = H, Me, Et, Bu^n; R^3 = H, Bu^n, n-C_5H_{11}; X = Br, I$$

i, Catalyst PdL_n (L = PPh_3), 0–25 °C, THF

Scheme 7

$$R = H$$
, alkyl, aryl; $X = I$ or Br

high stereoisomeric purities (equations 7, 8 and 9). $^{38-40}$ The enynes can easily be selectively transformed into the corresponding dienes, which may be used to prepare stereodefined polyunsaturated natural products.

Several fluorinated enynes (14) have been prepared by the reaction of alkynylzinc chlorides with fluoroalkenyl iodides (equation 10),⁴¹ and the Pd-catalyzed cross-coupling reactions of aryl fluoroalkanesulfonates with organozinc reagents have been reported (equation 11).⁴²

A general and convenient route to conjugated diynes, especially terminal diynes from the corresponding chloroenynes, has been developed.⁴³ Thus, readily available (*E*)-iodochloroethylene is first reacted with alkynylzinc derivatives in the presence of a Pd-phosphine complex catalyst to produce (*E*)-chloroenynes (15) in high yield. The chloroenynes (15) can then be readily converted into the corresponding



sodiodiynes (16), which can be transformed into terminal diynes (17), silylated diynes (18) and internal diynes (19; Scheme 8).



2.4.3.2 Alkynylmagnesium Halides

The palladium-catalyzed cross-coupling reactions of vinyl iodides with alkynylmagnesium halides have been found to occur with retention of configuration (>97%; equation 12).⁴⁴ Using this procedure, the dithienylenyne (**20**), which is a natural antifungal and nematicidal agent, can be prepared by a combination of Pd-catalyzed cross-couplings (Scheme 9).⁴⁵



i, Pd(PPh₃)₄, THF, benzene; ii, KOH, MeOH; iii, EtMgBr, THF; iv, CH₂=CHBr, Pd(PPh₃)₄, benzene

527

The Ni-catalyzed cross-coupling reaction of alkynyl Grignard reagents (21) with *trans*-dichloroethylene (22) has been applied to a simple procedure for the preparation of the protected form (23) of an extremely unstable synthon, hexadiynene (24). Separation of the diastereomers (23a from 23b) is facile since the former is an oil and the latter is a solid (Scheme 10).⁴⁶



i, Ni(dppp)Cl₂, dppp = Ph₂P(CH₂)₃PPh₂; ii, NaH-catalyst, MeOH/dry THF

Scheme 10

Recently, a mechanism for the cross-coupling reaction between phenyl iodide and methylmagnesium iodide, catalyzed by *cis*- and *trans*-[(PEt₂Ph)₂Pd(Ph)(I)], was proposed (Scheme 11).⁴⁷ The toluene produced from *cis*-(**25**) is thought to result *via* an intramolecular reductive elimination process without predissociation of the phosphine ligand.



Scheme 11

2.4.3.3 Alkynyltin Reagents

The reactivity of tin-carbon bonds allows the creation of new $sp-sp^2$ carbon to carbon bonds through substitution reactions. General reviews of the organotin route for carbon-carbon bond formation are available.²⁸⁻³¹ With Pd catalysis, alkynyl groups are more readily transferred than alkyl groups to the aryl or vinyl iodides to afford $sp-sp^2$ carbon bonds (equations 13 and 14).^{48,49}



 $R = Bu^t$, Bu^n , $SiMe_3$

Extensions to sp^2-sp cross-coupling reactions with vinyl triflates can be successfully achieved, as exemplified by the synthesis of the enyne (26). It is noteworthy that this reaction is compatible with the presence of alkynylsilanes, emphasizing the different behavior of organosilicon and organotin derivatives (equation 15).⁵⁰



Finally, imidoyl chloride can be transformed into the corresponding ketimines via palladium complex catalyzed cross-coupling reactions with alkynyltins (equation 16).^{51,52}

$$R^{1} = NR^{2} + R^{3} = SnBu_{3} = \frac{(PPh_{3})_{2}PdCl_{2}(4\%), EtPh, 70 °C}{69-84\%} R^{1} = NR^{2} (16)$$

 $R^{1} = Ph, 2$ -thienyl, Et, Cl; $R^{2} = p$ -tolyl, Buⁿ, p-C₆H₄CO₂Et; $R^{3} = Bu^{n}$, SiMe₃, EtC=C

2.4.3.4 Reactions of 1-Alkenyl Metals with 1-Alkynyl Halides

Using (*E*)-methylalkenylalanes, obtained by highly stereo- and regio-selective Zr-catalyzed carboalumination of alkynes, the stereochemically defined enynes (27) can be synthesized by Pd-catalyzed cross-coupling reactions with 1-haloalkynes (Scheme 12).⁵³ Although the mechanism of the Zr-catalyzed carboalumination is not very clear, certain aspects have been clarified,⁵⁴ and general reviews have been published.^{9,55,56}



i, Me₃Al-Cl₂ZrCp₂, 50 °C, 6 h; ii, H₂O for the case of R = Me; Buⁱ₂AlH for the case of R = H; iii, ZnCl₂, Pd(PPh₃)₄ (5 mol %), 20-25 °C

Scheme 12

Cross-coupling reactions between 1-alkynyl halides and 1-alkenylboranes, which are readily available *via* hydroboration of alkynes, can also be catalyzed by the Pd-phosphine complex (Scheme 13).^{57,58}



i, Pd(PPh₃)₄, benzene, NaOMe/MeOH, 80 °C, 2 h

Scheme 13

2.4.3.5 Reactions of Terminal Alkynes with sp²-Carbon Halides

2.4.3.5.1 General aspects

Terminal alkynes can be coupled directly to aryl and to vinyl halides in the presence of a palladium catalyst and a base. The mechanism of this reaction appears to involve oxidative addition of the sp^2 halide to palladium(0), followed by alkynylation of the intermediate organopalladium halide and reductive elimination of the disubstituted alkyne (equations 17 and 18).^{7,59,60}

Different conditions have been employed for this reaction, depending on the reactivity of the halide, the alkynes and the base used. Copper(I) iodide is a particularly effective cocatalyst, allowing the reactions to occur at room temperature.⁷ Additional examples of this type of coupling reaction are given in Table 1.⁶¹⁻⁸³

$$2 \text{ ArX} + = \frac{(PPh_{3})_{2}PdCl_{2}-CuI, Et_{2}NH, r.t., 3-6 h}{60-85\%} \text{ Ar} = R$$
(17)

Ar = Ph, 2-pyridyl

530



531

R = H, Ph; R' = Ph, CH_2OH

Aryl triflates, like aryl halides, can undergo Pd-catalyzed coupling reactions with terminal alkynes (Scheme 14),⁶⁷ and bromoallenes react with terminal alkynes^{84,85} or acetylene⁸⁶ under similar conditions to form allenynes (Schemes 15 and 16).



 $R = Ph, Me_3Si; R_f = CF_3, H(CF_2)_2O(CF_2)_2$

i, (PPh₃)₂PdCl₂, Et₃N/DMF, 90 °C, 3-17 h

Scheme 14







2.4.3.5.2 Synthesis of terminal alkynes

The sp^2 -sp coupling reaction can be extended to the synthesis of terminal alkynes by use of protected alkynes such as trimethylsilylacetylene (**28**; TMSA) or 2-methyl-3-butyn-2-ol (**29**), followed by subsequent removal of the protecting group (Schemes 17, 18 and 20).^{61,87,88} Thus, commercially available TMSA (**28**) reacts with anyl bromides or iodides in the presence of a palladium complex and copper(I) iodide, followed by treatment with dilute aqueous potassium hydroxide in methanol or a F⁻ source, such

Alkyne RC≡CH	Halide	Catalyst ^b	Conditions	Product	Yield (%)	Ref.
R = H	I I	A	r.t., 6 h	PhPh	85	7
	Ph Br	Α	r.t., 6 h	Ph	95	7
	N Br	A	r.t., 6 h		60	7
$\mathbf{R} = \mathbf{SiMe}_3$	Br — NO ₂	Α'	r.t., 4 h		92	61
	Br — COBu ^t	Α	r.t.	Bu ^t CO		62
	Br — Br	Α'	50 °C	RR	77	61
	NH ₂	Α	30 °C	R NH ₂	68	63
		Α'	r.t.	R	85	61

Table 1	Formation of Disubstituted Alkynes by	the Reaction of sp ²	Halides or Their Related (Compounds with Terminal Alkynes

Table 1 (continued)							
Alkyne RC≡CH	Halide	Catalyst ^b	Conditions	Product	Yield (%)	Ref.	
R = SiMe ₃		Α'	100 °C, ^a 20 h		76	64	
	Br N Br	Α'	5 °C, 1 h, r.t., 1 h	$Br \longrightarrow R$	74	65	
	Cr(CO) ₃	Α'	20 °C, 18 h	R	85	66	
	TFO	A'/DMF		Ph————R	87	67	
		D	r.t., 5 h	CI	82	68	
	NPh Cl	В	80 °C, 2 h	NPh	78	69	

Table 1 (continued)							
Alkyne RC≡CH	Halide	Catalyst ^b	Conditions	Product	Yield (%)	Ref.	34
R = SiMe3	OAc Br		80 °C, ^a 3 h		47	70, 71	
		A	50 °C, 2 h	AcO	80	72	Coupling
PhC=CCu	<i>p</i> -MeOC ₆ H ₄ Br	G	>20	<i>p</i> -MeOC ₆ H ₄ C=CR	98	73	Reacti
Ph-====	Ph N Br Ph N Br	A'/DMSO	r.t., 6 h	Ph N Ph	74	74	ons
Ph	Cl Br	A	r.t., 4 h	NEt ₂ R	37	75	
$R = CH_2CH_2OCOCH_2Pr^i$	ISI	F	80 °C, 6 h	I S R	50	76	





^a In sealed tube. ^bA = (PPh₃)₂PdCl₂, CuI/Et₂NH; A' = (PPh₃)₂PdCl₂, CuI/Et₃N; B = Pd(OAc)₂, PPh₃, CuI/Et₃N; C = (PPh₃)PdCl₂, CuI/Me₂NH; D = Pd(PPh₃)₄, CuI/BuⁿNH₂; E = Pd(PPh₃)₄/Et₃N; F = Pd(PPh₃)₄, CuI/Et₃N; G = PhPd(PPh₃)₂I, Bu₄N⁺T/HMPA; Tf = CF₃SO₂; TMT = dimethoxytrityI.

as KF-crown ether in DME or tris(dimethylamino)sulfonium trimethyldifluorosilicate (TASF), to give arylalkynes.^{61,69,89,90}



i, (PPh₃)₂PdCl₂-CuI cat., Et₃N or pyridine, r.t. to 60 °C; ii, 1N aq. KOH/MeOH, r.t., 1 h

Scheme 17



 $R = NO_2$, CN, COMe, CO_2Me , OPr^n ; X = Cl, Br, I

i, PhPdI(PPh₃)₂, CuI/NEt₃/C₆H₆, r.t. to 80 °C, 10 min to 2 h; ii, 10 equiv. MnO₂/5 equiv. KOH/C₆H₆, r.t., 10 min

Scheme 18

Terminal arylalkynes can be prepared by oxidation-decarbonylation of 3-arylpropargyl alcohols using manganese dioxide in the presence of alkali. The corresponding arylpropargyl alcohols are available by palladium-catalysed cross-coupling of aryl halides with commercial propargyl alcohol. The yield of the second step can be improved by the addition of a phase-transfer catalyst (18-crown-6; Scheme 18).⁸⁸

The coupling reaction between an aromatic halide and TMSA (28) in the presence of a Pd⁰ catalyst generated *in situ*, followed by treatment with potassium carbonate in methanol, provides a simple approach to various ethynylated aromatic derivatives (Scheme 19).⁶⁹ By this method even hindered, electron-rich arylalkynes, such as (30), can be prepared from the corresponding aryl iodide by refluxing for 2 d in triethylamine.⁸⁹



 $R = o_{-}, m_{-} \text{ or } p_{-}CHO, m_{-} \text{ or } p_{-}CO_{2}Me, m_{-}CF_{3}, 2-F_{-}5-NO_{2}, 2-F_{-}5-NH_{2}, m_{-}CF_{3}, m_{-}CF_{$



i, (PPh₃)₂Pd(OAc)₂, Et₃N, reflux for 2-20 h; ii, K₂CO₃, MeOH, 25 °C

Scheme 19
Using the above procedure, a wide variety of aromatic and heteroaromatic polyethynyl derivatives have been prepared, e.g. (31), $^{92}(32)^{90}$ and (33). 91,93





(32)

The more easily available starting material (29), as a protected alkyne, can condense with aryl halides, followed by treatment with sodium hydroxide and toluene also to provide terminal alkynes (Scheme 20),^{87,94}



i, (PPh₃)₂PdCl₂-CuI, Et₂NH; ii, NaOH/toluene, reflux 2 h

Scheme 20

In the presence of TASF as a reagent for cleavage of the Si—C bond and allylpalladium dimer as a coupling catalyst, alkynylsilanes react with aryl and vinyl halides to give the corresponding coupled products in high yield and in a stereospecific manner (Scheme 21).⁹⁵ In a special case, piperidine and formic

acid can also cleave the Si—C bond (Scheme 22).⁹⁶ This technique has been extended to a one-pot synthesis of conjugated dienynes or arylalkynes by a palladium-mediated three-component cross-coupling reaction, as shown in Scheme 23.⁹⁷ Thus, the Pd-catalyzed reaction of the silylstannylacetylene, first with an alkenyl iodide (34) and secondly with another sp^2 iodide in the presence of newly added TASF, affords the conjugated dienyne (35) or arylenyne (36) with high stereospecificity.

2.4.3.5.3 Stereospecific synthesis

The coupling reaction between sp^2 - and sp-carbon centers is stereospecific with (Z)- and (E)-vinylic bromides. For example, (Z)-methyl-3-bromoacrylate reacts with 1-hexyne in the presence of a Pd catalyst to form the (Z)-enzyne as the only product (Scheme 24).⁹⁸ This technique has also been applied in the stereoselective synthesis of 1-chloroenynes from (Z)-dichloroethylene (Scheme 25).⁹⁹

Trimethylsilylacetylene affords the 2:1 coupled endiyne product (37) in a stereospecific manner on coupling with geometrical isomers of 1,2-dichloroethylene (Scheme 26). In the absence of copper(I) iodide, no reaction occurs.¹⁰⁰

2.4.3.5.4 Phase-transfer catalysis

The technique of phase-transfer catalysis is useful for a variety of interesting metal-catalyzed reactions.^{101,102} For the synthesis of 1,3-enynes, the procedure based on the Pd-catalyzed reaction of alkenyl halides with 1-alkynes under phase-transfer conditions offers distinct advantages, in terms of single procedure and cost, over methods involving the use of organometallic compounds or of amines as solvents. Moreover, this procedure can tolerate the presence of functional groups such as hydroxy groups. Better yields are obtained when the coupling reactions are carried out under phase-transfer conditions, at room temperature, employing benzyltriethylammonium chloride (1 mol %) as the phase-transfer agent, a large excess of 2.5 N aqueous sodium hydroxide as the base, and a mixture of tetrakis (triphenylphosphine)palladium (1 mol %) and copper(I) iodide (0.5–2 mol %) as catalyst. Using these conditions, a variety of alkenyl halides (**38**) react with aliphatic 1-alkynes (**39**; Scheme 27).^{103,104}

The Pd-catalyzed coupling reactions of alkenyl bromides with heterocyclic alkynes (40) under the above phase-transfer conditions have been employed to prepare a large number of heterocyclic alkyne derivatives, including some naturally occurring compounds (Scheme 28).⁴⁵ The experimental conditions



 $R = Ph, n-C_5H_{11}, CH_2OH$

i, TASF, 2.5 mol % of $[(\eta^3-C_3H_5)PdCl]_2$, THF, r.t.





i, Pd(OAc)₂(PPh₃)₂, piperidine, HCO₂H/DMF, 60 °C, 8 h

Scheme 22



 $\begin{aligned} R^{1} = H, (CH_{2})_{4} (cyclo), & AcOCH_{2}, Ph, Pr^{n}, n-C_{6}H_{13}; R^{2} = H, Bu^{n}, (CH_{2})_{4} (cyclo); R^{3} = H, n-C_{6}H_{13}, n-C_{8}H_{17}; \\ R^{4} = H, n-C_{6}H_{13}, n-C_{8}H_{17}; R^{5} = H, Bu^{n}; R^{6} = H, Pr^{n}, n-C_{6}H_{13}; Ar = 1-naphthyl, 2-thienyl \\ R^{4} = H, n-C_{6}H_{13}, n-C_{8}H_{17}; R^{5} = H, Bu^{n}; R^{6} = H, Pr^{n}, n-C_{6}H_{13}; Ar = 1-naphthyl, 2-thienyl \\ R^{4} = H, n-C_{6}H_{13}, n-C_{8}H_{17}; R^{5} = H, Bu^{n}; R^{6} = H, Pr^{n}, n-C_{6}H_{13}; Ar = 1-naphthyl, 2-thienyl \\ R^{4} = H, n-C_{6}H_{13}, n-C_{8}H_{17}; R^{5} = H, Bu^{n}; R^{6} = H, Pr^{n}, n-C_{6}H_{13}; Ar = 1-naphthyl, 2-thienyl \\ R^{4} = H, n-C_{6}H_{13}, n-C_{8}H_{17}; R^{5} = H, Bu^{n}; R^{6} = H, Pr^{n}, n-C_{6}H_{13}; Ar = 1-naphthyl, 2-thienyl \\ R^{4} = H, n-C_{6}H_{13}, n-C_{8}H_{17}; R^{5} = H, Bu^{n}; R^{6} = H, Pr^{n}, n-C_{6}H_{13}; Ar = 1-naphthyl, 2-thienyl \\ R^{4} = H, n-C_{6}H_{13}, n-C_{8}H_{17}; R^{5} = H, Bu^{n}; R^{6} = H, Pr^{n}, n-C_{6}H_{13}; Ar = 1-naphthyl, 2-thienyl \\ R^{4} = H, n-C_{6}H_{13}, n-C_{8}H_{17}; R^{5} = H, R^{4} = H, R^{4$

i, 1 mol % of Pd(PPh₃)₄, 2.4 mol of TASF, THF, 50 °C; ii, I(R⁵)C=CR⁴R⁶, TASF; iii, ArI, TASF





- $R = cyclohexyl, Ph, Bu^n; R' = H, Me$
- i, Pd(OAc)₂ + 2PPh₃, NEt₃, r.t., 40 h

Scheme 24



>99% isomeric purity

 $R = n-C_5H_{11}$, CH₂OTHP, CH₂OAc, CH₂SMe

i, Pd(PPh₃)₄-CuI cat., n-butylamine, r.t., 5 h

Scheme 25



i, Pd(PPh₃)₄-CuI cat., n-butylamine, benzene, r.t., 1 h





 $R = H, Bu^{n}, n-C_{8}H_{17}; R' = H, Me; R'' = n-C_{5}H_{11}, CH_{2}OH, n-C_{6}H_{13}, (CH_{2})_{2}OH, (CH_{2})_{8}OH$

i, Pd(PPh₃)₄-CuI cat., 10% aq. NaOH, benzene, BzEt₃⁺ Cl⁻





X = I, Br;



2-furyl, 2-pyridyl;

R' = 2-thienyl, HO(CH₂)₂, HOCH₂, 2-furyl, 2-pyridyl, Ph, (EtO)₂CH

i, Pd(PPh₃)₄, CuI, C₆H₆, BzEt₃N Cl⁻, aq. 2.5 N NaOH

Scheme 28

of the procedure allow the direct preparation of alkynic derivatives starting from 1-alkynyltrimethylsilanes (41) and heteroaryl halides (Scheme 29).⁴⁵ An efficient one-pot synthesis of arylalkynes can be achieved by application of the phase-transfer reaction using commercially available 2-methyl-3-butyn-2ol (42) as the protected alkyne starting material (Scheme 30).¹⁰⁵ Using a similar method, oligo(thienylalkyne)s (43) have been synthesized in a stepwise manner.^{105,106}

2.4.3.5.5 Syntheses of aromatic and heterorings via sp²-sp coupling reactions

Alkynic intermediates derived from sp^2-sp coupling reactions can be applied to the syntheses of a variety of aromatic and heterocyclic molecules.¹⁰⁷⁻¹¹⁶ For example, a one-step synthesis of the aminoindolizines (44a-e) and the 5-aza analog (44f) from α -haloaza aromatics has been reported as outlined in Scheme 31.¹⁰⁷ The mechanism of this reaction appears to involve the initial formation of pyridyl-acetylene (45), followed by ring closure via the intermediate (46; Scheme 32).¹⁰⁷

Coupling Reactions





(42)

R = 2-thienyl, 2-pyridyl, nitrophenyl; R' = phenyl, 2-thienyl, 5-(2,3-benzothienyl)i, (PPh₃)₄Pd-CuI cat., BzEt₃N⁺Cl⁻, 5.5 N NaOH/benzene, r.t.

Scheme 30



b: $NR_2 = N(CH_2)_5$ **b**: $NR_2 = N, O$ **c**: $NR_2 = NEt_2$

i, (PPh₃)₂PdCl₂-CuI cat., 70-80 °C, 3-16 h





Scheme 32



Although the stoichiometric ethynylation of (47) with copper(I) phenylacetylide gives the product (48) of a normal double ethynylation, the Pd-catalyzed coupling reaction affords the fluorene derivative (49) instead. The reaction leading to (49) can be rationalized as proceeding via the σ -fluorenyl-Pd complex (50; Scheme 33).¹¹⁶



(49) $R = NO_2$, H

i, (PPh₃)₂PdCl₂-CuI, Et₃N, 80-85 °C, 3 h; ii, pyridine, 110 °C, 24 h



Scheme 33

Pyridopyrimidines and pyrimidines fused with five- or six-membered heterocycles can be prepared *via* the ring closure of intermediate ethynylpyrimidines, which are themselves obtained by the Pd-catalyzed cross-coupling of halopyrimidines with alkynic compounds (Scheme 34).¹⁰⁸⁻¹¹⁰

o-Alkynylanilines such as (51), which are made by Pd-catalyzed coupling reactions of alkynes with o-haloaniline precursors, are easily cyclized to the indoles in the presence of base (Scheme 35).^{111,112} Similar methods can be employed in the synthesis of isocoumarins,¹¹³ thienopyridines,¹¹⁷ furopy-ridines¹¹⁷ and the marine alkaloid asptamine.¹¹⁸



i, (PPh₃)₂PdCl₂-CuI cat., Et₃N, reflux 24 h; ii, Et₃N, EtOH, 120 °C, in a sealed tube, 12 h



i, (PPh₃)₂PdCl₂ cat., Et₃N, 100 °C; ii, NaOEt; iii, H₂O

Scheme 35

2.4.3.6 Palladium-catalyzed Vinylation of Alkynic Iodides

Heck-type vinylations of alkynic halides proceed only under solid-liquid phase-transfer conditions. These mild conditions allow the reaction to be performed with thermally unstable substrates. Thus, methyl (E)-enynoates can be obtained from the reaction of 1-iodoalkynes with methyl 2-propenoate in DMF at room temperature in the presence of potassium carbonate, tetrabutylammonium chloride and a catalytic amount of palladium acetate (Scheme 36).¹¹⁹



 $R = n-C_6H_{13}$, Bu^n , $SiMe_3$, Ph, $n-C_5H_{11}$; R'= OMe, Me

i, Pd(OAc)₂ cat., K₂CO₃ or Na₂CO₃, Buⁿ₄NCl/DMF, r.t.

Scheme 36

2.4.4 APPLICATIONS TO THE SYNTHESIS OF NATURAL PRODUCTS

As described already, Pd-catalyzed sp^2-sp coupling reactions are extremely useful in the synthesis of natural products. For example, the novel antitumor esperamycins have strong DNA binding/damaging properties which are related to the strained enediyne bridge present in their structures. A plausible bio-synthetic pathway to the bicyclic core of the esperamycins could involve intramolecular [4 + 2] cycload-dition of a conjugated polyenyne (54).¹²⁰ This polyenyne precursor (54) has been assembled by palladium-catalyzed stepwise coupling reactions of two different alkyne units to *cis*-dichloroethylene (52),⁹⁹ leading to (54), which was then further elaborated to (55), which showed a striking similarity to the esperamycin bicyclic core (Scheme 37).





Scheme 37

Enynes of the structural type (57) have been used as key intermediates in the total synthesis of vitamin D. The enyne (57) can be obtained by a simple synthesis based on Pd-catalyzed coupling of the ketoenol triflate (56) and an alkynic compound containing the vitamin D A-ring fragment (Scheme 38).¹²¹

The insecticidally active natural isobutylamides, such as dehydropipercide (60), and their synthons have been synthesized by Pd-catalyzed coupling reactions. For example, the aryl iodide (58) can couple in the dark with the alkynic amide (59) in the presence of a Pd-CuI catalyst in triethylamine to give dehydropipercide (60).

Engues of the type (62), which are similar to those occurring in Achillea tomentosa, are made by a similar coupling reaction between vinyl bromide (61) and the alkyne (59). The stereoisomers of (62) can be separated by HPLC to give pure (E)- and (Z)-isomers in 24% and 19% yields, respectively (Scheme 39).¹²²



i, (PPh₃)₂PdCl₂ (2 mol %), Et₃N (3.4 equiv.) in DMF, 75 °C, 4 h



i, (PPh₃)₂PdCl₂-CuI cat., Et₃N

Scheme 39

Palladium-catalyzed coupling reactions have been utilized in the total synthesis of ginkgolide B, which has been extracted from the ginkgo tree and is a most active anti-PAF (platelet-activating factor) agent. The important intermediate (65) is obtained from the reaction between the enol triflate (63) and the al-kynic OBO ester (64; Scheme 40).¹²³



i, Pd(PPh₃)₄ (5.7 mol %), CuI (0.5 equiv.), n-propylamine (2.3 equiv.), in benzene, 16 °C, 4 h

Scheme 40

Despite the multitude of functional groups present and the insolubility of the nucleosides in inert solvents, palladium-catalyzed coupling reactions have been used in their synthesis. Thus, the coupling of terminal alkynes with iodinated uracyl nucleosides proceeds in high yield in the presence of a Pd⁰–Cu¹ catalyst in warm triethylamine, and several of the products obtained by this route have shown useful antiviral activity (Scheme 41).¹²⁴



		R		R'
	а	Et	a	Et
	b	Pr ⁿ	b	Pr ⁿ
0	с	Bu ⁿ	с	Bu ⁿ
Tol = -U	d	$n-C_5H_{11}$	d -	$n-C_5H_{11}$
	e	Bu ^t	e	Bu ^t
	f	Ph	f	Ph
0-	g	SiMe ₃		
THP = ~~~~	h	Н	h	Н
	i	CH ₂ OTHP		
	j	CH ₂ OH	j	CH ₂ OH
O	k	CH ₂ CH ₂ OTHP		
$Ts = -\hat{S} - \langle \rangle$	- 1	CH ₂ CH ₂ OTol		
Ö	m	CH ₂ CH ₂ OH	m	CH ₂ CH ₂ OH
	n	CH ₂ CH ₂ OTs	n	CH ₂ CH ₂ OTs
			0	CH=CH ₂
	р	CH ₂ CH ₂ CH ₂ OTol		
			q	CH ₂ CH ₂ CH ₂ OH

2.4.5 REFERENCES

- 1. L. I. Simondi, in 'The Chemistry of Functional Groups, Supplement C: The Chemistry of Triple-bonded Functional Groups', ed. P. Patai and Z. Rappoport, Wiley, New York, 1983, p. 510. L. Brandsma, 'Preparative Acetylenic Chemistry', 2nd edn., Elsevier, Amsterdam, 1988, chap. 10.
- 2.
- 3. J. Mathieu and J. Weil-Raynal, 'Introduction of a Carbon Chain or an Aromatic Ring', Thieme, Stuttgart, 1975, pp. 491, 501.
- R. D. Stephens and C. E. Castro, J. Org. Chem., 1963, 28, 3313.
 G. H. Posner, Org. React. (N.Y.), 1975, 22, 253.
- 6. G. H. Posner, 'An Introduction to Synthesis Using Organocopper Reagents', Wiley, New York, 1980.
- 7. K. Sonogashira, Y. Tohda and N. Hagihara, Tetrahedron Lett., 1975, 4467.
- 8. R. F. Heck, in 'Palladium Reagents in Organic Syntheses', Academic Press, London, 1985, p. 306.
- 9. E. Negishi, Acc. Chem. Res., 1982, 15, 340.
- G. W. Ebert and R. D. Rieke, J. Org. Chem., 1988, 53, 4482.
 J. Burdon, P. L. Coe, C. R. Marsh and J. C. Tatlow, J. Chem. Soc., Chem. Commun., 1967, 1259.
- 12. P. D. Landor, S. R. Landor and J. P. Leighton, Tetrahedron Lett., 1973, 1019.
- 13. G. Struve and S. Seltzer, J. Org. Chem., 1982, 47, 2109.
- 14. A. E. Jukes, Adv. Organomet. Chem., 1974, 12, 215.
- 15. J. Lindley, Tetrahedron, 1984, 40, 1433.
- 16. T. Kauffmann, Angew. Chem., Int. Ed. Engl., 1974, 13, 291.
- 17. R. Oliver and D. R. M. Walton, Tetrahedron Lett., 1972, 5209.
- 18. J. F. Normant, A. Commerçon and J. Villiéras, Tetrahedron Lett., 1975, 1465.
- 19. A. Commerçon, J. F. Normant and J. Villiéras, Tetrahedron, 1980, 36, 1215.
- . Brown and G. A. Molander, J. Org. Chem., 1981, 46, 645 20. H.C
- 21. M. Hoshi, Y. Masuda and A. Arase, Bull. Chem. Soc. Jpn., 1983, 56, 2855.
- 22. P. J. Stang and T. Kitamura, J. Am. Chem. Soc., 1987, 109, 7561.
- E. Negishi, G. Lew and T. Yoshida, J. Chem. Soc., Chem. Commun., 1973, 874.
 G. A. Molander and H. C. Brown, J. Org. Chem., 1977, 42, 3106.
- 25. S. Hara, Y. Satoh, H. Ishiguro and A. Suzuki, Tetrahedron Lett., 1983, 24, 735.
- 26. A. Suzuki, Pure Appl. Chem., 1986, 459.
- 27. K. Tamao, K. Sumitani and M. Kumada, J. Am. Chem. Soc., 1972, 94, 4374.
- 28. J. K. Stille. Angew. Chem., Int. Ed. Engl., 1986, 25, 508.
- 29. I. P. Beletskaya, J. Organomet. Chem., 1983, 250, 551.
- 30. M. Pereyre, J.-P. Quintard and A. Rahm, 'Tin in Organic Synthesis', Butterworths, London, 1987, p. 185.
- 31. J. K. Stille, Pure Appl. Chem., 1985, 57, 1771.
- 32. M. Kumada, Pure Appl. Chem., 1980, 52, 669.
- 33. B. M. Trost and T. R. Verhoeven, in 'Comprehensive Organometallic Chemistry', ed. G. Wilkinson, F. G. A. Stone and E. W. Abel, Pergamon Press, Oxford, 1982, vol. 8, p. 799.
- 34. J. Tsuji, 'Organic Synthesis with Palladium Compounds', Springer-Verlag, Berlin, 1980.
- 35. A. O. King, N. Okukado and E. Negishi, J. Chem. Soc., Chem. Commun., 1977, 683.
- 36. A. O. King, E. Negishi, F. J. Villani, Jr. and A. Silveira, Jr., J. Org. Chem., 1978, 43, 358.
- 37. A. O. King and E. Negishi, J. Org. Chem., 1978, 43, 358; see also ref. 9, p. 342.
- 38. B. P. Andreini, A. Carpita and R. Rossi, Tetrahedron Lett., 1988, 29, 2239.
- 39. A. Carpita and R. Rossi, Tetrahedron Lett., 1986, 27, 4351.
- 40. B. P. Andreini, A. Carpita and R. Rossi, Tetrahedron Lett., 1986, 27, 5533.
- 41. F. Tellier, R. Sauvêtre and J.-F. Normant, Tetrahedron Lett., 1986, 27, 3147.
- 42. Q.-Y. Chen and Y.-B. He, Tetrahedron Lett., 1987, 28, 2387.
- 43. E. Negishi, N. Okukado, S. F. Lovich and T.-T. Luo, J. Org. Chem., 1984, 49, 2629.
- 44. H. P. Dang and G. Linstrumelle, Tetrahedron Lett., 1978, 191.
- 45. R. Rossi, A. Carpita and A. Lezzi, Tetrahedron, 1984, 40, 2773.
- 46. J. A. Walker, S. P. Bitler and F. Wudl, J. Org. Chem., 1984, 49, 4733.
- 47. F. Ozawa, K. Kurihara, M. Fujimori, T. Hidaka, T. Toyoshima and A. Yamamoto, Organometallics, 1989, 8, 180.
- N. A. Bumagin, I. G. Bumagina and I. P. Beletskaya, Dokl. Akad. Nauk SSSR, 1983, 272, 1384. 48.
- 49. D. E. Rudisill, L. A. Castonguay and J. K. Stille, Tetrahedron Lett., 1988, 29, 1509.
- 50. W. J. Scott, G. T. Crisp and J. K. Stille, J. Am. Chem. Soc., 1984, 106, 4630.
- 51. T. Kobayashi, T. Sakakura and M. Tanaka, Tetrahedron Lett., 1985, 26, 3463.
- 52. Y. Ito, M. Inoue and M. Murakami, Tetrahedron Lett., 1988, 29, 5379
- 53. E. Negishi, N. Okukado, A. O. King, D. E. Van Horn and B. I. Spiegel, J. Am. Chem. Soc., 1978, 100, 2254.
- 54. E. Negishi, T. Takahashi, S. Baba, D. E. Van Horn and N. Okukado, J. Am. Chem. Soc., 1987, 109, 2393.
- 55. E. Negishi and T. Takahashi, Aldrichimica Acta, 1985, 18, 31.
- E. Negishi and H. Takahashi, J. Synth. Org. Chem., Jpn., 1989, 47, 2.
 N. Miyaura, K. Yamada and A. Suzuki, Tetrahedron Lett., 1979, 3437.
- 58. N. Miyaura, K. Yamada, H. Suginome and A. Suzuki, J. Am. Chem. Soc., 1985, 107, 972.
- 59. L. Cassar, J. Organomet. Chem., 1975, 93, 253.
- H. A. Dieck and F. R. Heck, J. Organomet. Chem., 1975, 93, 253.
 S. Takahashi, Y. Kuroyama, K. Sonogashira and N. Hagihara, Synthesis, 1980, 627.
- 62. F. Schaub and R. Schneider, Ger. Pat. 3 430 833 (1985) (Chem. Abstr., 1985, 103, 123 483n).
- 63. T. Kurihara, M. Fujiki and H. Tabei, Jpn. Pat. 61 27 934 (1961) (Chem. Abstr., 1986, 105, 78 677g).
- 64. T. Sakamoto, Y. Kondo and H. Yamanaka, Chem. Pharm. Bull., 1985, 33, 626. 65. J. W. Tilley and S. Zawoiski, J. Org. Chem., 1988, 53, 386.
- 66. D. Villemin and E. J. Schigeko, J. Organomet. Chem., 1985, 293, C10.
- 67. Q.-Y. Chen and Z.-Y. Yang, Tetrahedron Lett., 1986, 27, 1171.

- 68. V. Ratovelomanana, A. Hammoud and G. Linstrumelle, Tetrahedron Lett., 1987, 28, 1649.
- 69. W. B. Austin, N. Bilow, W. J. Kelleghan and K. S. Y. Lau, J. Org. Chem., 1981, 46, 2280.
- 70. G. C. Nwokogu, Tetrahedron Lett., 1984, 25, 3263.
- 71. G. C. Nwokogu, J. Org. Chem., 1985, 50, 3900.
- 72. M. J. Robins and P. J. Barr, J. Org. Chem., 1983, 48, 1854.
- 73. N. A. Burnagin, I. O. Kalinovski and I. P. Beletskaya, Izv. Akad. Nauk SSSR, Ser. Khim., 1981, 2836 (Chem. Abstr., 1982, 96, 85 133a).
- 74. D. E. Ames and M. I. Brohi, J. Chem. Soc., Perkin Trans. 1, 1980, 1384.
- 75. D. E. Ames and D. Bull, Tetrahedron, 1982, 38, 383.
- 76. G. T. Crisp, Synth. Commun., 1989, 19, 307.
- 77. M. J. Chapdelaine, P. J. Warwick and A. Shaw, J. Org. Chem., 1989, 54, 1218.
- 78. K. Kikukawa, A. Abe, F. Wada and T. Matsuda, Bull. Chem. Soc. Jpn., 1983, 56, 961.
- 79. K. A. Cruickshank and D. L. Stockwell, Tetrahedron Lett., 1988, 29, 5221.
- 80. K. Sanechika, T. Yamamoto and A. Yamamoto, Bull. Chem. Soc. Jpn., 1984, 57, 752.
- 81. A. Kasahara, T. Izumi, I. Shimizu, M. Satou and T. Katou, Bull. Chem. Soc. Jpn., 1982, 55, 2434.
- A. Kasahara, T. Izumi and T. Katou, Chem. Lett., 1979, 1373.
 A. Kasahara, T. Izumi and I. Shimizu, Chem. Lett., 1979, 1119.
- 84. K. Ruitenberg, H. Kleijn, C. J. Elsevier, J. Meijer and P. Vermeer, Tetrahedron Lett., 1981, 22, 1451.
- 85. T. Jeffery-Luong and G. Linstrumelle, Synthesis, 1983, 32.
- 86. G. Märkl, P. Attenberger and J. Kellner, Tetrahedron Lett., 1988, 29, 3651.
- 87. D. E. Ames, D. Bull and C. Takundwa, Synthesis, 1981, 364.
- 88. N. A. Bumagin, A. B. Ponomaryov and I. P. Beletskaya, Synthesis, 1984, 728.
- 89. M. E. Garst and B. J. McBride, J. Org. Chem., 1989, 54, 249.
- 90. R. Diercks, J. C. Armstrong, R. Boese and K. P. C. Vollhardt, Angew. Chem., Int. Ed. Engl., 1986, 25, 268.
- 91. L. Blanco, H. E. Helson, M. Hirthammer, H. Mestdagh, S. Spyroudis and K. P. C. Vollhardt, Angew. Chem., Int. Ed. Engl., 1987, 26, 1246.
- 92. R. Diercks and K. P. C. Vollhardt, Angew. Chem., 1986, 98, 268.
- 93. H. E. Helson, K. P. C. Vollhardt and Z.-Y. Yang, Angew. Chem., Int. Ed. Engl., 1985, 24, 114.
- 94. S. A. Nye and K. T. Potts, Synthesis, 1988, 375.
- 95. Y. Hatanaka and T. Hiyama, J. Org. Chem., 1988, 53, 920.
- 96. A. Arcadi, S. Cacchi and F. Marinelli, Tetrahedron Lett., 1986, 27, 6397.
- 97. Y. Hatanaka, K. Matsui and T. Hiyama, Tetrahedron Lett., 1989, 30, 2403.
- 98. J. R. Weir, B. A. Patel and R. F. Heck, J. Org. Chem., 1980, 45, 4926.
- **99**. V. Ratovelomanana and G. Linstrumelle, Tetrahedron Lett., 1981, 22, 315.
- 100. K. P. C. Vollhardt and L. S. Winn, Tetrahedron Lett., 1985, 26, 709.
- 101. L. Cassar, Ann. N. Y. Acad. Sci., 1980, 333, 208.
- 102. H. Alper, Adv. Organomet. Chem., 1981, 19, 183.
- 103. R. Rossi, A. Carpita, M. G. Quirici and M. L. Gaudenzi, Tetrahedron, 1982, 38, 631.
- 104. R. Rossi and A. Carpita, Tetrahedron, 1983, 39, 287.
- 105. A. Carpita, A. Lessi and R. Rossi, Synthesis, 1984, 571.
- 106. S. Kotani, M. Ono, H. Yamaguchi, K. Shiina and K. Sonogashira, in 'Abstracts of 35th Symposium in Organometallic Chemistry', Osaka, Japan, 1988, p. 190. 107. A. Ohsawa, Y. Abe and H. Igeta, *Chem. Lett.* 1979, 241.
- 108. T. Sakamoto, Y. Kondo and H. Yamanaka, Chem. Pharm. Bull., 1982, 30, 2410.
- 109. T. Sakamoto, Y. Kondo and H. Yamanaka, Chem. Pharm. Bull., 1982, 30, 2417.
- T. Sakamoto, Y. Kondo and H. Yamanaka, *Chem. Pharm. Bull.*, 1986, 34, 2362.
 T. Sakamoto, Y. Kondo and H. Yamanaka, *Heterocycles*, 1986, 24, 31.
- 112. A. Tischler and T. J. Lanza, Tetrahedron Lett., 1986, 27, 1653.
- T. Sakamoto, M. An-naka, Y. Kondo and H. Yamanaka, Chem. Pharm. Bull., 1986, 34, 2754.
 T. Sakamoto, M. Shiraiwa, Y. Kondo and H. Yamanaka, Synthesis, 1983, 312.
- 115. G. Just and R. Singh, Tetrahedron Lett., 1987, 28, 5981.
- 116. T. K. Dougherty and K. S. Y. Lau, J. Am. Chem. Soc., 1983, 48, 5273.
- 117. T. Sakamoto, Y. Kondo, R. Watanabe and H. Yamanaka, Chem. Pharm. Bull., 1986, 34, 2719.
- 118. T. Sakamoto, N. Miura, Y. Kondo and H. Yamanaka, Chem. Pharm. Bull., 1986, 34, 2760.
- 119. T. Jeffery, Synthesis, 1987, 70.
- 120. S. L. Schreiber and L. L. Kiessling, J. Am. Chem. Soc., 1988, 110, 631.
- 121. L. Castedo, A. Mouriño and L. A. Sarandeses, Tetrahedron Lett., 1986, 27, 1523.
- 122. L. Crombie, M. A. Horsham and R. J. Blade, Tetrahedron Lett., 1987, 28, 4879.
- 123. E. J. Corey, M.-C. Kang, M. C. Desai, A. K. Ghosh and I. N. Houpis, J. Am. Chem Soc., 1988, 110, 649.
- 124. M. J. Robins and P. J. Barr, J. Org. Chem., 1983, 48, 1854.

2.5 Coupling Reactions Between *sp* Carbon Centers

KENKICHI SONOGASHIRA

Osaka City University, Japan

2.5.1 INTRODUCTION	551
2.5.2 OXIDATIVE HOMOCOUPLING REACTIONS OF TERMINAL ALKYNES	552
2.5.3 CHODKIEWICZ-CADIOT REACTION	553
2.5.4 COUPLING OF ORGANOMETALLIC ALKYNIDES AND 1-HALOALKYNES	553
2.5.5 OXIDATIVE COUPLING REACTIONS OF ORGANOMETALLIC ALKYNIDES	554
2.5.6 APPLICATIONS	555
 2.5.6.1 Linear Polyalkynes 2.5.6.2 Cyclic Alkynes 2.5.6.3 Polymer Synthesis 2.5.6.3.1 Synthesis of organic polymers 2.5.6.3.2 Synthesis of organometallic polymers 2.5.6.4 Natural Product Synthesis 2.5.6.5 Phase-transfer Catalysts 2.5.6.6 Copper-Polymer Complexes as Catalysts for Oxidative Coupling Reactions 	555 556 557 557 557 558 559 559
2.5.7 REFERENCES	560

2.5.1 INTRODUCTION

Some of the most useful synthetic transformations of terminal alkynes involve intermolecular and intramolecular homo- and cross-coupling reactions between their *sp*-carbon centers, leading to butadiyne or polyyne derivatives. The two most widely used and practical systems are: (i) oxidative homocoupling reactions, *i.e.* Glaser and Eglington reactions; and (ii) heterocoupling reactions, *i.e.* Chodkiewicz-Cadiot coupling of a terminal alkyne with a haloalkyne.

Other methods for the synthesis of butadiynes and polyalkynes using organometallic alkynide derivatives are less widely used. Reviews describing the above two coupling methods have been published.¹⁻⁶ Oxidative couplings of terminal alkynes by a copper(I) catalyst or a copper(II) reagent (Glaser or Eglington reaction) are the best methods of preparing symmetrically substituted butadiyne or polyyne derivatives (equation 1). Unsymmetrically substituted butadiyne derivatives can be prepared by Chodkiewicz–Cadiot type couplings (equations 2 and 3).

These methods have been applied to the syntheses of a wide variety of alkynic compounds.³ Recently there have been a number of reports describing the syntheses of natural products by oxidative coupling

$$2 R \xrightarrow{Cu^+ \text{ or } Cu^{2+}, O_2} R \xrightarrow{R} R$$
(1)



reactions of terminal alkynes, followed by stereo- or regio-selective addition of hydrogen or organometallic reagents (see Section 2.5.6.4 and Chapter 2.4).

2.5.2 OXIDATIVE HOMOCOUPLING REACTIONS OF TERMINAL ALKYNES

The oxidative coupling of terminal alkynes by copper salts, discovered in 1869 by Glaser, has evolved to the modified method reported in 1962 by Hay.⁷ In the Hay procedure, oxygen is passed through a solution of the alkyne and a catalytic amount of a copper(I) salt in a complex-forming solvent, such as pyridine and TMEDA. Although the oxidative coupling by Cu¹ salt catalysts in suitable amines has wide scope,^{2–6} it is less successful for less acidic terminal alkynes, such as alkyl- or silyl-alkynes.

An essential step in the above process is the reversible formation of the alkynic anion, $R-C=C^-$, which is less acidic and less reactive than that of active alkynes such as phenylacetylene or conjugated polyynes. It has been found that the coupling rate can be accelerated⁶ when a small amount of DBU is added to the solution in pyridine (equation 4). The addition of the stronger base facilitates the proton removal from the alkyne. If the product or reactants are oxygen sensitive, it may be necessary to shorten the reaction time.⁶ Coupling in a nitrogen atmosphere by using a large excess of the Hay reagent has also been recommended.⁸

2 Et
$$\longrightarrow$$
 + O₂ $\xrightarrow{\text{CuCl, DBU, pyridine}}$ Et \longrightarrow Et + H₂O (4)
30-35 °C
-90%

Another modification of the coupling conditions for alkylalkynes has been reported by Knol *et al.*⁹ Thus, the oxidative polymerization of 1,8-nonadiyne was carried out using a catalyst complex prepared by reacting a solution of TMEDA and CuCl in *o*-dichlorobenzene with oxygen gas at 1 atm. It proved to be necessary to dry the reagents as well as the monomer before use, and to add some molecular sieve to the reaction mixture in order to remove the water generated during the reaction.

A study of the kinetics of coupling has established that the reaction rate decreases with increasing water content and that it is enhanced by increasing the concentration and the basicity of the ligand.¹⁰ The formation of water may occur in reversible reaction steps from alkynic end groups and a dimeric copper(II) complex (1), followed by intramolecular oxidative coupling of two complexed alkynic groups (Scheme 1). Reviews summarizing the mechanism of the oxidative couplings have been published.^{3,5}



Scheme 1

Coupling by copper(II) acetate in pyridine is also a useful procedure for active alkynes such as phenylacetylene and conjugated polyynes. This method is also a useful procedure for making cyclic alkynes.

2.5.3 CHODKIEWICZ-CADIOT REACTION

The heterocoupling of a terminal alkyne with a 1-bromoalkyne in the presence of an aliphatic amine and a catalytic amount of a copper(I) salt affords unsymmetrically substituted diynes (2; equation 5). This useful reaction, discovered by Chodkiewicz and Cadiot,¹¹ can be employed advantageously for the synthesis of several polyunsaturated systems.



Generally the bromoalkyne RC=CBr is introduced dropwise to a mixture of the alkyne RC=CH, ethylamine, methanol or ethanol, in the presence of a catalytic amount of copper(I) chloride, and a small amount of NH₂OH·HCl. The reaction is usually exothermic and efficient cooling is required during the introduction of the bromoalkyne. The copper(I) alkynyl (equation 6) is assumed to be the reactive intermediate, and the copper(I) ion generated in equation (7) can be employed in equation (6).

The formation of the symmetrical diyne (4), generated as in equation (8), can be suppressed by maintaining the concentration of the bromoalkyne (3). This side reaction is particularly serious in the case of less acidic alkynes such as alkylalkynes. The reducing agent, $NH_2OH \cdot HCl$, is used to reduce the copper(II) ion.

2.5.4 COUPLING OF ORGANOMETALLIC ALKYNIDES WITH 1-HALOALKYNES

The coupling reaction between alkyne Grignard derivatives and 1-haloalkynes has been investigated in some detail. Copper(I) and cobalt(I) salts are used as catalysts (equation 9; $M = MgBr)^{12,13}$ and the most effective alkynides appear to be organocopper reagents. Such reagents can be used in the presence of a range of functional groups (*e.g.* OH, equation 10;¹⁴ CO₂H, equation 11¹⁵).



A useful and convenient synthesis of 1-trimethylsilyl-1,3-diynes (7) by the reaction of copper(I) alkynides (5) with bromoalkynes (6) has been reported by Zweifel and coworkers (Scheme 2).¹⁶

A highly reactive copper slurry, prepared by the reduction of $CuI \cdot PEt_3$ with 1 equiv. of lithium naphthalide, undergoes rapid oxidative addition to alkynyl halides under mild conditions, and induces their homocoupling when oxygen is bubbled through the mixture.¹⁷



R = n-hexyl, cyclohexyl, 1-cyclohexenyl, *t*-butyl, THP-O(CH₂)₉

Scheme 2

2.5.5 OXIDATIVE COUPLING REACTIONS OF ORGANOMETALLIC ALKYNIDES

Conjugated diverse may be prepared by the reaction of organometallic alkynides with iodine or other oxidants (equation 12).^{18,19}

$$2 R - M - M = MgX R - R$$
(12)

Alkynyllithium reagents add readily to metal carbonyls, leading to thermally unstable anionic alkynic metal carbonylates (8), which give mainly the 1,3-diyne, by treatment with iodine-methanol in THF (Scheme 3).^{20,21}



The reactions of alkynylborates can provide an efficient and selective method of coupling two unlike boron-bound groups. Both symmetrical and unsymmetrical conjugated dignes can be synthesized *via* iodination of the borates prepared by utilizing two cyclohexyl or siamyl groups as blocking groups (equation 13).²²⁻²⁴ These intramolecular coupling reactions of organoborons provide viable alternatives to the more conventional cross-coupling procedures.



 $R^1 = n$ -hexyl; $R^2 = Et$, Bu^t , Ph

Conjugated diynes can also be prepared by decarbonylation of the corresponding diethynyl ketones using (PPh₃)₃RhCl in boiling xylene (equation 14).²⁵ High temperatures may be needed to produce the intermediate dialkynyl-rhodium complex.

Alkynylpalladium reagents can be useful intermediates in the synthesis of diynes by catalytic coupling reactions since the dialkynyl-palladium complex $(PPh_3)_2Pd(C = CR)_2$ (9) has been shown to easily produce the corresponding diyne on reductive elimination.²⁶

Reactions of benzene solutions of arylalkynes with 1 equiv. of chloroacetone and 2 equiv. of Et₃N, using a mixture of (PPh₃)₄Pd and CuI as a catalyst, afford 1,4-diarylbutadiynes in very good yields. Under similar reaction conditions aliphatic 1-alkynes yield a mixture of symmetrically disubstituted 1,4-dialkynyl-1,3-butadiynes and 3-alkyl-4(1-alkynyl)-hexa-1,5-diyn-3-enes (**10**; equations 15 and 16).²⁷ This method may represent a good alternative, in nonpolar organic solution, to the Glaser reaction.



 $R = Bu^n$, *n*-hexyl, Pr^n

Ito and his coworkers have reported that palladium-catalyzed oxidative coupling reactions of Grignard reagents in the presence of N-substituted isocyanide dichloride afford diynes (equation 17). Isocyanide dichloride may serve as a reoxidant of the palladium catalyst in this sequence via a catalytic cycle.²⁸ In addition Kiji and his coworkers have described the oxidative coupling of phenylacetylene by a Pd–Cu catalyst in the presence of 4-iodo-(3H)-phenothiazin-3-one.²⁹

$$RN = \begin{pmatrix} Cl \\ + 2R'MgBr \\ Cl \end{pmatrix} \xrightarrow{PdCl_2(dppf) \text{ or } Pd(PPh_3)_4} R'-R' + R-N \equiv \overline{C}$$
(17)

R = Ph, cyclohexyl; $R^1 = PhC \equiv C$, $Me_2(MeO)CC \equiv C$; dppf = 1,1-bis(diphenylphosphino)ferrocene

2.5.6 APPLICATIONS

2.5.6.1 Linear Polyalkynes

Many stabilized higher polyynes terminated with bulky groups such as *t*-butyl or trialkylsilyl can be derived from alkyne coupling reactions. Thus polyyne chain extensions using $Et_3Si(C=C)_mH$ (m = 1, 2, 3, 6) in mixed oxidative couplings using the modified Hay method,³⁰ have been reported by Walton and coworkers (Scheme 4).³¹ Another polyyne chain extension has been used in the synthesis of the unsymmetrically substituted octatetraynediamines (13) *via* the reaction of the lithium alkynide (11) with per-chlorobutenyne (12) (Scheme 5).³²

 $Et_{3}Si(C=C)_{2}H + Et_{3}SiC=CH \longrightarrow Et_{3}Si(C=C)_{3}SiEt_{3} \longrightarrow Et_{3}Si(C=C)_{3}H \longrightarrow Et_{3}Si(C=C)_{6}H \longrightarrow Et_{3}Si(C=C)_{12}SiEt_{3} \longrightarrow H(C=C)_{12}H$



2.5.6.2 Cyclic Alkynes

The oxidative coupling of α,ω -diethynyl derivatives by the Eglington method, which can be performed under homogeneous high dilution conditions, is suitable to obtain cyclic diynes. A general review of cyclic alkynes is available.^{33–35} A wide variety of dehydroannulenes have been synthesized by oxidative coupling reactions, *e.g.* the oxidative coupling reaction of the α,ω -diethynyl compound (14) to give the cyclic polyyne (15) (which could be separated into *meso* and racemic diastereomers), followed by baseinduced prototropic rearrangement to provide the novel cyclic polyenepolyyne (16; Scheme 6).³⁶



Scheme 7

Toda and coworkers have found drastic differences in the reaction behavior of racemic and optically active 3,6-di-*t*-butylocta-1,4,7-triyne-3,6-diol (17) upon oxidative coupling by the Eglington method.^{37,38} Thus, the noncentrosymmetric cyclic dimers, (18a) and (18b), can be obtained from the racemic monomer, but the corresponding optically active monomer does not lead to any of the cyclic dimer. In the latter case the optically active polymer (19) is obtained as the sole isolable product (Scheme 7).

Recently oxidative coupling reactions of α,ω -diethynyl compounds have been applied in the synthesis of cyclophanes, which are interesting functional host molecules.^{39–42} Whitlock and coworkers,⁴³ for example, have reported the synthesis of the water-soluble cyclophane (21), designed as an esterase mimic by the Eglington reaction of (20; Scheme 8).



2.5.6.3 Polymer Synthesis

Since the yield in the oxidative coupling is practically quantitative, the oxidative polymerization of α,ω -diethynyl monomers would be expected to yield high molecular weight, linear polymers, as shown in equation (18). Hay has reported that almost any diethynyl monomer, even organometallic monomers, can be polymerized to high molecular weight polymers in the presence of a soluble amine complex catalyst of a copper(I) salt (Hay modification).^{44,45}

$$n = Y = + 1/2 O_2 = Y = \frac{1}{n} + nH_2O \quad (18)$$

2.5.6.3.1 Synthesis of organic polymers

m-Diethynylbenzene gives a pale yellow polymer in quantitative yields when treated by the Hay procedure. The polymer is soluble in solvents such as chlorobenzene and nitrobenzene above 100 °C.^{45,46} Recent commercial developments of organic solids from alkynic precursors date from the discovery of the above soluble butadiynylenephenylene polymer. The molecular weight and solubility of the polymer can be controlled by using a mixture of *m*-diethynylbenzene and phenylacetylene in the polymerization. The oligomers produced can then be further processed at high temperatures to form carbon films and fibers.⁴⁶

In order to obtain high molecular weight polymers by the oxidative coupling of α,ω -diethynylalkanes as described in Section 2.5.2, higher temperatures (to increase the polymer solubility), optimized concentrations of the reagents, and addition of molecular sieves (to remove the water generated during the polymerization) have all been shown to lead to best results.⁹

2.5.6.3.2 Synthesis of organometallic polymers

The Hay modification can be applied to organometallic diethynyl compounds such as Group IV alkynes.⁴⁵ Parnell and coworkers,⁴⁷ for example, have successfully used the procedure to carry out the oxidative coupling of 1,3-bis(dimethylethynyl)disiloxane (22). However, diethynyldimethylsilane (23) did not give the polymer, as cleavage of the silicon-ethynyl bond occurred instead.

An interesting class of polymers, e.g. (25) and (27), containing conjugated alkynic groups and σ bonded transition metal atoms in the main chain has been reported.^{48,49} Oxidative coupling reactions of



dialkynyl complexes are used as one of three methods for preparing the polymers. Polymerization using the Hay modification affords a high molecular weight product, e.g. $M = 1.5 \times 10^5$ (25b). However, there is a severe restriction on this method since it cannot be applied to monomer complexes such as alkynyl-nickel complexes which are unstable to the oxidant.

A platinum-polyyne polymer, containing diethynylsilane groups in the main chain, can be synthesized by the Hay reaction conditions after appropriate modification to protect the cleavage of Si—C bonds. The characteristics of the synthesis are as follows: (a) since the Si—C=C bond may be cleaved by uncomplexed copper(II) chloride, a large excess of TMEDA, which is added to catch free copper(II) ion, may retard the formation of the oxygen complex; (b) a homogeneous catalyst solution is prepared by oxygen-bubbling for many hours to complete the formation of the oxygen complex in methylene chloride in the presence of molecular sieves; (c) polymerization is carried out in methylene chloride over the molecular sieves using 3.5 equiv. of the above reagent; (d) to optimize the molecular weight of the polymer the coupling is monitored by GPC, and it requires 7 h. Under the above conditions a white polymer film (**28**; $M = 77\ 000$) can be obtained.⁵⁰



2.5.6.4 Natural Product Synthesis

There is a very wide variety of natural products which, with their biological precursors, have structures based on straight-chain fatty acids, containing conjugated alkenic and/or alkynic linkages. Many of these natural polyalkynes have been prepared from alkynic precursors using oxidative couplings. For example, Nicolaou and his coworkers have reported the stepwise, stereocontrolled total synthesis of endiandric acids, utilizing the alkadiynic diol (**30**) as a key intermediate, which was readily available from *trans*-pent-2-en-4-yn-1-ol (**29**) by Glaser coupling.⁵¹ Similarly the biomimetic precursor (**33**) of endiandric acids A–G was prepared by coupling (**31**) (1 equiv.) with the more plentiful (**32**) (5 equiv.) in pyridine-methanol (1:1), containing Cu(OAc)₂ (2 equiv.; Eglington method).⁵² A further example is the coupling of the ω -alkynic synthon (**35**) with the bromoalkyne (**34**), by the Chodkiewicz–Cadiot method, to give anacyclin (**36**), an insecticide found in *Anacyclus pyrethrum*.⁵³





2.5.6.5 Phase-transfer Catalysts

Phase-transfer catalysis is a useful procedure for a variety of interesting metal-catalyzed reactions.^{54,55} However, only one example of this approach has been reported for the synthesis of diynes by the sp-sp carbon coupling reaction. Thus vinylic dibromides derived from aromatic aldehydes have been shown to react with carbon monoxide, in the presence of zerovalent palladium compounds as the metal catalyst, and under phase-transfer conditions in a two-phase system (benzene, 5 M NaOH), to give the corresponding diynes in reasonable yields (equations 21 and 22).⁵⁶

$$Ar Br Br 54-74\% CO, BzEt3NClPd(diphos)2/NaOHAr Br 54-74\% Ar (21)$$

Ar = Ph, p-MeC₆H₄, p-ClC₆H₄, p-MeOC₆H₄



2.5.6.6 Copper-Polymer Complexes as Catalysts for Oxidative Coupling Reactions

The oxidative coupling reaction of terminal alkynes is critically dependent on the water concentration in the reaction mixture (see Section 2.5.2). Since water is produced during the reaction, careful elimination of it may be required. Challa and Meinders⁵⁷ have demonstrated that the polymer catalyst derived from copper(II) chloride and either N,N-dimethylbenzylamine or N,N-dimethylaminomethylated atactic polystyrene (**37**) provides an extra protection of the catalytic copper complexes against water in the coupling reaction of phenylacetylene (equation 23), resulting in a higher reaction rate than the low molecular weight catalyst.

$$Ph \longrightarrow + 1/2 O_2 \longrightarrow Ph \longrightarrow Ph$$
(23)



2.5.7 REFERENCES

- 1. G. Eglington and W. McCrae, in 'Advances in Organic Chemistry, Method and Results', ed. R. A. Raphael, E. C. Taylor and H. Wynberg, Wiley, New York, 1963, vol. 4, p. 225.
- T. F. Rutledge, 'Acetylenic Compounds', Reinhold, New York, 1968, chap. 6.
- 3. P. Cadiot and W. Chodkiewicz, in 'Chemistry of Acetylenes', ed. H. G. Viehe, Dekker, New York, 1969, chap. 9.
- 4. W. G. Nigh, in 'Oxidation in Organic Chemistry, Part B' ed. W. S. Trahanovsky, Academic Press, New York, 1973, p. 1.
- 5. L. I. Šimondi, in 'The Chemistry of Functional Groups, Supplement C: The Chemistry of Triple-bonded Functional Groups', ed. P. Patai and Z. Rappoport, Wiley, New York, 1983, p. 510.
- L. Brandsma, 'Preparative Acetylenic Chemistry', 2nd edn., Elsevier, Amsterdam, 1988, chap. 10. 6.
- 7. A. S. Hay, J. Org. Chem., 1962, 27, 3320.
- 8. S. Takahashi, E. Murata, K. Sonogashira and N. Hagihara, J. Polym. Sci., Polym. Chem. Ed., 1980, 18, 661.
- 9. K. E. Knol, L. W. Van Horssen, G. Challa and E. E. Havinga, Polym. Commun., 1985, 26, 71.
- 10. H. J. Kevelam, K. P. De Jong, H. C. Meinders and G. Challa, Makromol. Chem., 1975, 176, 1369.
- 11. W. Chodkiewicz and P. Cadiot, C. R. Hebd. Seances Acad. Sci., 1955, 241, 1055.
- 12. H. K. Black, D. H. S. Horn and B. C. L. Weedon, J. Chem. Soc., 1954, 1704.
- 13. H. H. Schlubach and V. Franzen, Justus Liebigs Ann. Chem., 1951, 572, 116.
- 14. R. F. Curtis and J. A. Taylor, J. Chem. Soc. C, 1971, 186.
- 15. M. Rösner and G. Köbrich, Angew. Chem., Int. Ed. Engl., 1975, 14, 708.
- 16. J. A. Miller and G. Zweifel, Synthesis, 1983, 128.
- 17. G. W. Ebert and R. D. Rieke, J. Org. Chem., 1988, 53, 4482.
- 18. J. B. Armitage, E. R. H. Jones and M. C. Whiting, J. Chem. Soc., 1952, 2014.
- 19. J. B. Armitage, E. Entwhistle, E. R. H. Jones and M. C. Whiting, J. Chem. Soc., 1954, 147. 20. I. Rhee, N. Mizuta, M. Ryang and S. Tsutsumi, Bull. Chem. Soc. Jpn., 1968, 41, 1417.
- 21. I. Rhee, M. Ryang and S. Tsutsumi, Tetrahedron Lett., 1969, 4593.
- 22. A. Pelter, K. Smith and M. Tabata, J. Chem. Soc., Chem Commun., 1975, 857.
- 23. A. Pelter, R. Hughes, K. Smith and M. Tabata, Tetrahedron Lett., 1976, 4385.
- 24. J. A. Sinclair and H. C. Brown, J. Org. Chem., 1976, 41, 1078.
- 25. E. Müller and A. Segnitz, Justus Liebigs Ann. Chem., 1973, 1583.
- 26. K. Sonogashira, Y. Tohda and N. Hagihara, Tetrahedron Lett., 1975, 4467.
- 27. R. Rossi, A. Carpita and C. Bigelli, Tetrahedron Lett., 1985, 26, 523.
- 28. Y. Ito, M. Inoue and M. Murakami, Tetrahedron Lett., 1988, 29, 5379.
- 29. H. Konishi, C. Matsumura, T. Okano and J. Kiji, J. Organomet. Chem., 1989, 364, 245.
- 30. G. E. Jones, D. A. Kendrick and A. B. Holmes, Org. Synth., 1987, 65, 52.
- 31. R. Eastmond, T. R. Johnson and D. R. M. Walton, Tetrahedron, 1972, 28, 4601.
- 32. D. Faul and G. Himbert, Chem. Ber., 1988, 121, 1367.
- 33. A. Krebs, in 'Chemistry of Acetylenes', ed. H. G. Viehe, Dekker, New York, 1969.
- 34. M. Nakagawa, in 'The Chemistry of Functional Groups, The Chemistry of Triple-bonded Functional Groups', ed. P. Patai, Wiley, New York, 1978, p. 635. 35. S. Misumi and T. Kaneda, in 'The Chemistry of Functional Groups, The Chemistry of Triple-bonded
- Functional Groups, ed. P. Patai, Wiley, New York, 1978, p. 635.
- 36. T. Nomoto, K. Fukui and M. Nakagawa, Bull. Chem. Soc. Jpn., 1976, 49, 305.
- F. Toda, J. Okada and K. Mori, Angew. Chem., Int. Ed. Engl., 1988, 27, 859.
 F. Toda, J. Okada and T. C. W. Mak, Chem. Lett., 1988, 1829.
- 39. J.-M. Lehn, F. Schmidt and J.-P. Vigneron, Tetrahedron Lett., 1988, 29, 5255.
- 40. B. J. Whitlock, E. T. Jarvi and H. W. Whitlock, Jr., J. Org. Chem., 1981, 46, 1832.
- 41. E. T. Jarvi and H. W. Whitlock, Jr., J. Am. Chem. Soc., 1982, 104, 7196.
- F. Diederich, Angew. Chem., Int. Ed. Engl., 1988, 27, 362.
 B. J. Whitlock and H. W. Whitlock, Jr., Tetrahedron Lett., 1988, 29, 6047.
- 44. A. S. Hay, J. Org. Chem., 1960, 25, 1275.
- 45. A. S. Hay, J. Polym. Sci., Part A, 1969, 7, 1625.
- T. X. Neenan and G. M. Whitesides, J. Org. Chem., 1988, 53, 2489. 46.
- 47. D. R. Parnell and D. P. Macaione, J. Polym. Sci., Polym. Chem. Ed., 1973, 11, 1107.

- 48. N. Hagihara, K. Sonogashira and S. Takahashi, in 'Advances in Polymer Science', ed. H.-J. Cantow et al., Springer, Berlin, 1981, vol. 41, p. 149. 49. S. Takahashi, H. Morimoto, E. Murata, S. Kataoka, K. Sonogashira and N. Hagihara, J. Polym. Sci., Polym.
- Chem. Ed., 1982, 20, 565.
- 50. T. Matsumoto, S. Kotani, H. Yamaguchi. K. Shiina and K. Sonogashira, Preprint of Tokyo Seminar on Macromolecular Metal Complexes, 1987, Tokyo, 207. 51. K. C. Nicolaou, N. A. Petasis, R. E. Zipkin and J. Uenishi, J. Am. Chem. Soc., 1982, 104, 5555.
- 52. K. C. Nicolaou, R. E. Zipkin and N. A. Petasis, J. Am. Chem. Soc., 1982, 104, 5558.
- 53. L. Crombie, A. J. W. Hobbs and M. A. Horsham, Tetrahedron Lett., 1987, 28, 4875.
- L. Cassar, Ann. N. Y. Acad. Sci., 1980, 333, 208.
 H. Alper, Adv. Organomet. Chem., 1981, 19, 183.
- 56. V. Galamb, M. Gopal and H. Alper, Organometallics, 1983, 2, 801.
- 57. G. Challa and H. C. Meinders, J. Mol. Catal., 1977, 3, 185.

2.6 Pinacol Coupling Reactions

GRAEME M. ROBERTSON Glaxo Group Research, Ware, UK

2.6.1 INTRODUCTION	563
2.6.2 INTERMOLECULAR COUPLINGS	564
2.6.2.1 Aromatic Carbonyl Compounds 2.6.2.2 Aliphatic Carbonyl Compounds	564 570
2.6.3 INTRAMOLECULAR COUPLINGS	572
2.6.4 α,β-UNSATURATED CARBONYL COMPOUNDS	577
2.6.5 IMINE COUPLINGS	579
2.6.5.1 Intermolecular Couplings 2.6.5.2 Intramolecular Couplings	579 581
2.6.6 THIOCARBONYL COUPLINGS	582
2.6.7 McMURRY-TYPE COUPLINGS	583
2.6.7.1 Intermolecular Couplings 2.6.7.2 Intramolecular Couplings	583 588
2.6.8 MIXED COUPLINGS	595
2.6.9 MISCELLANEOUS COUPLINGS	597
2.6.10 CARBONYL-ALKENE CYCLIZATIONS	598
2.6.11 CARBONYL-ALKYNE CYCLIZATIONS	602
2.6.12 CARBONYL-ALLENE CYCLIZATIONS	605
2.6.13 REFERENCES	606

2.6.1 INTRODUCTION

The reductive coupling of carbonyl compounds, especially ketones, to give pinacols is an important method for the formation of vicinally functionalized C—C bonds.¹ In addition, the related coupling of carbonyls to give alkenes, the McMurry reaction, provides a complementary route for C—C bond elaboration.¹¹⁷

The intermolecular pinacolic coupling of carbonyl compounds was first described in 1859^2 and, although the classical reagents lacked selectivity, it has since been refined into a mild and selective method for the synthesis of 1,2-diols. Coupling can be initiated photochemically, electrochemically, or with a range of metal reducing agents. A number of mechanisms have been proposed³ and these can be generalized into two basic types, as shown in Scheme 1. Coupling is propagated by single-electron reduction of the carbonyl group to form a ketyl radical anion (1), which either undergoes radical-radical coupling (route a), or is further reduced to the corresponding dianion (2) and then nucleophilically attacks a second carbonyl group (route b), so leading to pinacol formation.



Scheme 1

By far the majority of pinacolic couplings occur via radical-radical coupling and generally afford a mixture of (\pm) (*like*) and *meso* (unlike) diols.⁴ An alternative and often competing reaction to pinacolization is reduction to the corresponding alcohol.⁵ Indeed, selective pinacol formation is critically dependent on many factors, including the reductant used, the solvent⁶ and the reaction pH.^{30,31,79} The traditionally used reducing agents such as alkali metals⁷ or metal amalgams⁸ suffered from lack of selectivity and/or sluggish reaction with only low to modest yields. Moreover electrochemical^{9,10} and photochemical¹¹ methods were largely limited to aryl (activated) systems, as was the extensively used Gomberg–Bachman procedure, using Mg/MgI₂.¹² The first major improvement in reaction methodology was the use of aluminum amalgam in dichloromethane or THF (equation 1).¹³ The use of these solvents greatly facilitates product isolation, thus improving yields to 40–60%.



i, Al(Hg), CH₂Cl₂ (or THF), reflux, 40-60%

More recently the introduction of low-valent transition metal and lanthanoid based reducing systems, especially those based on titanium, has provided dramatic advances in efficiency and selectivity. It is now possible to select appropriate conditions for efficient coupling of all types of carbonyl compounds, often with high chemo-, regio- and stereo-selectivity. Moreover, imino- and thio-carbonyl derivatives are also coupled via pinacolic methodology. The coupling of imines to 1,2-diamines is particularly effective, with excellent control of vicinal stereochemistry.

2.6.2 INTERMOLECULAR COUPLINGS

2.6.2.1 Aromatic Carbonyl Compounds

The relative ease of pinacolization is primarily determined by the reduction potential of the carbonyl group involved. Many reductants are therefore selective for aromatic and other electronically activated systems. Moreover, as a result of this ready reduction, pinacolization of such carbonyls can be effected by either anionic or radical routes. For example, treatment of aromatic aldehydes or ketones with Mg/TMSCl in HMPA promotes pinacolization *via* formation of an α -silyloxy carbanion^{14,15} and nucleophilic attack on a second carbonyl group (equation 2). Furthermore, with benzaldehyde the reaction is stereodirecting with a preference for *lk*-coupling.⁴ Whilst an alternate coupling method using the milder

conditions of Zn/TMSCl in THF proceeds via radical-radical coupling,¹⁶ interestingly ultrasonication of this reaction increases yields by up to 50%. In a similar manner reaction of aromatic aldehydes with hexamethyldisilane and a fluoride catalyst also gives 1,2-diarylethanediols in moderate to good yields, after hydrolytic work-up (equation 3).¹⁷



i, Mg, Me₃SiCl, HMPA, 95 °C; ii, Zn, Me₃SiCl, THF, 0-81% (stirring), 56-85% (ultrasonication)



i, (Me₃Si)₂, Bu₄NF, HMPA, 61-95%; ii, (Me₃Si)₂, CsF, HMPA, 26-100%

One-electron reductants promoting radical-mediated pinacolic coupling are more common however. Early examples were the use of organometallic species derived from vanadium(II)¹⁸ or chromium(II) salts (equation 4)^{18,19} which convert aromatic aldehydes to mixtures of pinacols and the corresponding monoalkyl ethers. Vanadium salts also effect the coupling of capto-datively activated carbonyls such as (3) to provide a convenient route to tartaric acid derivatives in good yields (70–80%; equation 5).²⁰ In contrast, chromium and europium salts give only reduction to the corresponding alcohols.

$$2 PhCHO \xrightarrow{i} Ph OH Ph OEt Ph OH Ph OH (4) 69\% 11\%$$

i, Cr(ClO₄)₂, HCl, EtOH, r.t.



i, V(ClO₄)₂, 1 M HClO₄, r.t., 10 h

Treatment of aromatic aldehydes with $Fe(CO)_5$ or $Fe_3(CO)_{12}$ in pyridine (equation 6) gives the corresponding pinacols in good yield, together with some reduction.²¹ Alternative iron-based reducing systems such as BuLi/FeCl₃ or BuLi/Fe₄S₄(SPh)₄ also reductively couple aromatic aldehydes and ketones to 1,2-diols (equation 7).²² The former reagent, however, also produces significant amounts of the corresponding alcohol.

Of more general application are the titanium-based reagents first introduced by Mukaiyama²³ and by Trylik (equation 8).²⁴ Mukaiyama demonstrated that TiCl₄/Zn selectively coupled both aromatic or aliphatic aldehydes and ketones to either pinacols or alkenes by judicious choice of the appropriate reaction



conditions.²⁵ In contrast, the Trylik reagent, TiCl₃/Mg, gave alkenes with aromatic systems, but coupled aliphatic carbonyls to pinacols in moderate yields. Stable soluble Ti^{II} species can be prepared from TiCl₄ and Buⁱ₂Te in an inert solvent.²⁶ In DME this reagent promotes selective *lk*-coupling of benzaldehyde to the corresponding (±)-pinacol (equation 9). More recently Seebach reported that the TiCl₄/BuLi system also promotes selective *lk*-coupling of aromatic aldehydes (equation 10).²⁷ The stereoselectivity is, however, dependent on the nature of the aryl substituent. Thus when R = H, 4-Me, 4-Cl or 2-Me the *l*-pinacol is formed selectively, but when R = 2-Me both *l*- and *u*-isomers are formed.⁴ Reduction of Cp₂TiCl₂ with Bu^sMgCl also produces a titanium species exhibiting high *l*-(*threo*)-selectivity (equation 11).²⁸ In addition, this reagent is also capable of coupling both aromatic and α , β-unsaturated aldehydes.^{28,29} Although most titanium-mediated pinacolic couplings employ Ti⁰ or Ti^{II} as the reductant, Ti^{III} has also been used. The reducing power of Ti^{III} redox systems is strongly pH dependent. Thus whilst highly activated carbonyl compounds such as 2- or 4-acetylpyridines are coupled to their respective pinacols by treatment with aqueous acidic TiCl₃ (equation 12),³⁰ less reactive species like 3-acetylpyridine or benzaldehyde require the more strongly reducing basic TiCl₃ (equation 13).³¹ A similar effect is observed in the electro-³² and photo-pinacolization³³ of pyridinyl ketones, where the 3-isomers react more slowly.



i, TiCl₄, Buⁱ₂Te, DME, r.t., 99%



i, BuLi, TiCl₄, Et₂O, 50-92%



ds = 100:1

i, Cp2TiCl2, BuSMgCl, THF, -78 °C to r.t., 96%



i, AcOH, aq. TiCl₃, r.t., 72%



i, 30% NaOH, aq. TiCl₃, MeOH, pH 10-12, r.t., 73%

Chemoselective pinacolization has recently been established with the advent of lanthanoid metal based coupling reagents. Imamoto *et al.* employed low-valent cerium as a mild selective reagent (equation 14) suitable for both aromatic and aliphatic compounds.³⁴ Under these conditions carboxyl, cyano and vinyl halide groups are unaffected. This is in sharp contrast to the relatively low chemoselectivity shown by most titanium-based reagents. Similarly Kagan and coworkers have used samarium diiodide as an alternative chemoselective reductant for the coupling of either aromatic or aliphatic carbonyls (equation 15). With aromatic aldehydes this reagent exhibits complete selectivity for the carbonyl group in the presence of either carboxyl, cyano or nitro moieties.³⁵ Recently Fujiwara demonstrated that aromatic aldehydes and ketones can be coupled to pinacols using 0.5 equiv. of ytterbium metal (equation 16).³⁶ The reaction proceeds *via* the formation of an intermediate ytterbium species (4) which acts as a nucleophile toward a second carbonyl group.¹⁹⁰

Photo-¹¹ and especially electro-chemical⁹ pinacolizations of aromatic systems are well established routes to 1,2-diols. Furthermore, stereoselective pinacol formation is possible *via* both routes through control of the reaction media. For example, photopinacolization requires the use of an external hydrogen donor, usually a secondary alcohol or an amine,³⁷ and stereoselection can be influenced through changes in these hydrogen donors. Thus under neutral conditions approximately 1:1 (\pm):*meso* mixtures are



i, Cel₂, THF, 0 °C to r.t., 91%



i, 2 SmI₂, THF, 66-95%

567



i, Yb (0.5 equiv.), THF, HMPA, 58-99%



formed (equation 17), but under basic conditions *lk*-coupling, to give the (\pm) -pinacol, is favored due to increased charge interactions.³⁸ Amines are more efficient hydrogen donors than alcohols and therefore favor *meso* pinacol formation *via ul*-coupling of neutral ketyl radicals.³⁹ Recently homochiral amines have been studied as a method for chiral induction.⁴⁰ Optical yields of up to 23% have been obtained using the amine (5) *via* formation of complexes of the type (6).

$$\begin{array}{c} O \\ Ph \end{array} \qquad \begin{array}{c} i \text{ or ii} \\ Ph \end{array} \qquad \begin{array}{c} Ph \\ Ph \end{array} \qquad \begin{array}{c} O \\ Ph \end{array} \qquad P$$

Electropinacolization can be a highly stereoselective and in some cases a stereospecific reaction. The degree of selectivity has been ascribed to both solution and cathode surface effects. For example, the electropinacolization of hydroxybenzaldehyde in an aqueous basic medium gives the meso hydrobenzoin (7) stereospecifically (equation 18).⁴¹ The related coupling of benzoin (8; X = OH) is also stereospecific, giving only one of six possible diastereomeric products, the threo isomer (9) in 50% yield via an lk-coupling.⁴² However when X = Me, that is in the absence of an additional hydrogen-bonding site, two diastereomers are formed. In agreement with photopinacolic couplings, Stocker and Jenevein demonstrated⁴³ that at basic pH *lk*-coupling to give (\pm) -products is again favored (equation 20). A major drawback of electropinacolizations carried out in aqueous media is competing reduction to the corresponding alcohol. Selectivity for pinacol formation is much improved in aprotic media. Under these conditions coupling occurs between neutral radicals, and by the use of conditions that minimize protonation and hydrogen-bonding, (±):meso ratios of up to 19:1 can be obtained (equation 21).44.45 Tetraalkylammonium salts have proven particularly useful in promoting high yielding selective pinacolizations under aprotic conditions, giving almost quantitative yields of pinacols (equation 22).46 These salts not only lower the reaction potential, a feature that permits the electropinacolic coupling of aliphatic ketones,⁵⁷ but also facilitate very high (\pm) :meso ratios of up to 50:1.⁴⁷ Increased yields have also been obtained using added bivalent transition metal cations, such as Cr²⁺, Mn²⁺, Fe²⁺, Co²⁺ or Zn²⁺. Product distribution is dependent on the Lewis acidity and complexing power of the cation used.⁴⁸ The logical extension of using homochiral coelectrolytes to promote asymmetric couplings has also been investigated but with



i, Hg, 2 M NaOH, or Hg, 4 M NaOH, EtOH, -1.6 to -2.0 V

only limited success. Thus Seebach⁴⁹ has obtained optical yields of *ca*. 6% using the amine (5; equation 23), whilst van Tilborg⁵⁰ has obtained optical yields of up to 21% with the β -hydroxyamine salt (10).



i,	acidic	Hg, aq. AcOH, LiCl, 60%	1.2:1
ii,	neutral	Hg, aq. NH4OH, aq. NH4OAc, 46%	1.1:1
iii,	basic	Hg, 2M KOAc, 61%	2.8:1









i, Hg, MeCN,
$$-1.8$$
 V, Ph $\stackrel{\text{OH}}{\longrightarrow}$ $\stackrel{\text{NMe}_3I^-}{\longrightarrow}$, 98%



optical yield %



(21)

2.6.2.2 Aliphatic Carbonyl Compounds

The synthesis of pinacols from aliphatic carbonyl compounds requires the use of more efficient redox systems than does the corresponding reductive coupling of aromatic systems. Until the early 1970s there were few general efficient methods available. The classical alkali or alkaline metal amalgam reagents are not very selective and are of little use with complex substrates. The first general method for the intermolecular coupling of aliphatic aldehydes was reported in 1975⁵¹ and employs TMSCl/Li as the reductant (equation 24). The use of TMSCl is essential for the *in situ* trapping of the intermediate alkoxides. Titanium(II)-based pinacolic coupling reagents are of more general use, but unfortunately both Mukaiyama's reagent, TiCl₄/Zn,²³ and Trylik's reagent, TiCl₃/Mg,²⁴ offer only limited scope. A systematic analysis of titanium-based reagents by Corey and his coworkers^{52–54} led to the development of three generally applicable reagents. Thus TiCl₄/Mg(Hg) efficiently promotes intermolecular couplings, whilst CpTiCl₃/LiAlH₄ or the hexamethylbenzene complex of (Cl₂AlCl₂)Ti are more suitable for intramolecular examples.⁵³ The nature and yield of product are highly dependent on the type of titanium(II) species used. The relative efficacy of a selection of these reagents for the coupling of cyclohexanone is shown in Table 1.



 Table 1
 Relative Efficacy of Ti^{II} Reagents at the Pinacolic Coupling of Cyclohexanone



Ti ^{II} system	Yield (%)
Mg(Hg)/TiCl4	93
Mg/TiCl3	45
Zn/TiCl4	24
LiAlH4/TiCl3	(Alkene)

Both aldehydes and ketones are efficiently coupled (80-95%) by TiCl4/Mg(Hg), which is also suitable for unsymmetrical couplings (equation 25). For low molecular weight aldehydes CpTiCl₃/LiAlH₄ is the reagent of choice. Recently Wiedmann has introduced graphite-suspended magnesium, C8Mg, and titanium, C₈Ti, as simple to use, universally applicable reagents.⁵⁵ These reagents couple both aromatic and aliphatic aldehydes or ketones efficiently in good yields (48-93%) with a high tolerance for other reducible substituents. Unlike many other reducing agents these reagents are also effective at both inter- and intra-molecular coupling. In addition C₈Ti also promotes McMurry-type coupling of carbonyls to alkenes¹²⁹ and the related cyclization of keto esters. Other highly chemoselective reagents based on lowvalent lanthanoid metals^{1h} have also been applied to aromatic and aliphatic couplings.^{34,35} Thus, low-valent cerium (CeI2) couples cycloalkanones in good yields (64-95%), although competing reduction can be a problem.³⁴ Organocerium reagents (RCeI) have also been used;⁵⁶ however these reagents also give Grignard-type products and reduction in addition to pinacols. Samarium diiodide, an efficient reagent for stereoselective intramolecular couplings, also promotes intermolecular couplings in high yields (80-95%).³⁵ Recently Kariv-Miller and Mahachi reported a potentially useful method for the electropinacolization of aliphatic carboyl compounds through the use of dimethylpyrrolidinium tetrafluoroborate (DMPBF₄) as an electrode potential reducing catalyst (equation 26).⁵⁷ In the absence of DMPBF₄ only reduction is observed, but using DMPBF4 selective pinacolization occurs.

Although a range of chemoselective pinacolic coupling methods is becoming available the problem of stereoselective coupling of aliphatic carbonyls is less well defined. This is in contrast to couplings of aromatic systems, where both pinacolic coupling to 1,2-diols and reductive (McMurry-type) pinacolic couplings to alkenes can be highly selective. Controlling the stereochemistry of the diol formed from aliphatic carbonyls remains a problem. For example, Mundy *et al.* have shown that the coupling of homochiral



i, Hg, DMF, aq. diglyme, -2.9 V, 76%; ii, Hg, DMF, aq. diglyme, -2.7 V, DMPBF₄, 95%

methyl cyclohexanones gives a mixture of stereoisomeric products whose distribution differs when using Al(Hg) or the strongly coordinating TiCl4/Mg(Hg) reagent (Scheme 2).⁵⁸ Stereoselective *lk*-coupling of aliphatic aldehydes to (\pm)-pinacols with up to 100% selectivity (typically 70–96%) has however been achieved *via* photoinduced formation of 2-stanna-1,3-dioxolanes.⁵⁹ Formation of the cyclic organometal-lic intermediate (**11**) controls the stereochemical outcome of the reaction (equation 27). Unfortunately this method is not suitable for ketones since the transition state becomes too sterically hindered. A convenient method for stereoselective coupling of ketones is to link them *via* a chain containing a removeable heteroatom. Such intramolecular cyclizations are then equivalent to stereoselective intermolecular couplings (*e.g.* equation 40).^{74,75} Highly *lk*-selective hydrodimerization of α , β -unsaturated aldehydes is possible using the Cp₂TiCl₂/Bu^sMgCl reagent (equation 28).²⁸ Again metal templating accounts for the observed stereoselection, with the complex (**12**) postulated as the stereocontrolling intermediate.

In some instances the carbonyls in pinacolic coupling reactions exhibit chirality (enantiomer) recognition, that is the ability of an enantiomer to recognize a molecule of like chirality and react exclusively



i, Bu₈Sn₃, hv, C₆H₆; ii, AcCl, 70%

(27)



with it or its mirror image. For pinacolic couplings a reaction between two like ketones, (R) + (R) or (S) + (S), giving a homodiol is energetically preferred over reaction between opposite enantiomers to give heterodiols. Thus Tuoboul and Dana have shown that electropinacolization of racemic (13) yields exclusively the *trans-threo-trans* diol (14) in an enantioselective manner *via* an *lk*-type coupling (equation 29).^{60,61} As part of a program directed towards the synthesis of dodecahedrane, Paquette and coworkers investigated the hydrodimerization of both homochiral and racemic ketone (15) promoted by mercury amalgam (equation 30).⁶² In each case pinacolic coupling occurred only between molecules of the same absolute configuration to give homodiols. Chirality recognition has also been observed with (-)-carvone using the TiCl₄/Mg reagent⁶³ and with homochiral and racemic camphor using lithium in THF or liquid ammonia.⁶⁴



2.6.3 INTRAMOLECULAR COUPLINGS

Intramolecular pinacolic coupling reactions are a powerful and selective method for ring construction, enabling the synthesis of a variety of structures ranging from small-ring 1,2-diols to macrocyclic systems.⁶⁵ The success of these reactions over such a large range of ring sizes is largely due to the templating effect offered by coordination with the low-valent metal reductants employed, especially titanium. Such templating overcomes the angle strain invoked in synthesis of small rings and the entropic factors associated with macrocycle formation. The reaction can even be applied to the synthesis of cyclopropanediols (equation 31),⁶⁶ despite their high sensitivity towards acid or base. The classical methods for pinacolic coupling are generally not very effective in the intramolecular domain;⁵³ e.g. compare the efficacy of the Mg/MgI₂ reagent⁶⁷ with that of Corey's TiCl₄/Mg(Hg) reagent⁵³ in the formation of cyclobutane-1,2-diols (equation 32).⁶⁸

Corey and coworkers have developed a number of reagents (equations 33 and 34) for intramolecular pinacolic couplings during their studies on the synthesis of gibberellic acid $(17)^{52,69}$ via cyclization of the keto aldehyde (16). These reagents include Mg(Hg)/Me₂SiCl₂,⁵² TiCl₄/Mg(Hg), and





i, Hg, aq. MeOH, pH 3.7, -1.25 V (vs. SCE), 80%



CpTiCl₃/LiAlH₄.^{53,69} Titanium-based reagents have also been employed in a new approach to propellane systems, including the quadrone ring skeleton (equation 35).⁷⁰ More recently Takeshita *et al.* demonstrated the chemoselectivity of such reagents in a formal synthesis of cuparene (18; equation 36).⁷¹ Similar chemoselectivity is observed in pinacolic cyclizations promoted by TMSCl/Zn (equation 37).⁷²



i, Mg(Hg), Me₂SiCl₂, THF, 75%; ii, TiCl₄, Mg(Hg), THF, 90%



(16)

i, TiCl₃, K, THF, 55% or TiCl₃, LiAlH₄, THF, 40-45%



i, TiCl₃, Zn, pyridine, THF, reflux, 69%



i, Me₃SiCl, Zn, 2,6-lutidine, THF, reflux, 75%

A more versatile reducing agent is samarium diiodide, which promotes chemoselective cyclizations of functionalized keto aldehydes in a stereodefined manner to form 2,3-dihydrocyclopentane carboxylate derivatives in good yields and with diastereoselectivities of up to 200:1 (equation 38).⁷³ The reaction proceeds via selective one-electron reduction of the aldehyde component and subsequent nucleophilic attack on the ketone moiety. Stereochemical control is established by chelation of the developing diol (19) with Sm³⁺ which thereby selectively furnishes *cis* diols (equation 39). The stereoselective *ul*-cyclization of 1,5-diketones to *cis* cyclopentane-1,2-diols using TiCl₄/Zn has been used to prepare stereodefined sterically hindered acyclic 1,2-diols when a removable heteroatom, such as sulfur⁷⁴ or selenium,⁷⁵ is included in the linking chain (equation 40).



i, 2 SmI₂, MeOH, THF, -78 °C, 44%



i, TiCl₄, Zn, THF, 0 °C; ii, Raney nickel, EtOH, reflux, 51-81%

Cyclopentane-1,2-diols have also been prepared by cyclization of aromatic diketones *via* electrochemical pinacolization (equation 41).⁷⁶ In contrast to chemically induced reactions, electropinacolizations involve an anionic addition process and not a ketyl radical coupling.⁷⁷ Substituents on the aromatic ring can have a profound effect on the stereoselectivity of the reaction (equation 42),⁷⁸ as can the pH of the reaction medium.⁷⁹ In strongly acidic media *cis* 1,2-diols are formed preferentially *via* coupling of neutral radicals. In basic media a radical anion or anionic attack mechanism is operative and *trans* diols are favored due to charge interactions. In the medium pH range the two mechanisms are in competition and mixtures of *cis* and *trans* diols are formed. As would be expected, increasing steric hindrance decreases the efficacy of electropinacolization.⁸⁰

Larger ring size carbocyclic 1,2-diols can be synthesized via extension of the methods used for cyclopentane formation (equations 43-45). Thus, cyclizations of 1,6-dicarbonyls give the corresponding cis



i, Mg cathode, NaI, pyridine, 20%; ii, Al(Hg), EtOH, 50%



i, Hg, -2.0 V, 2M aq. (or ethanolic) KOH, r.t.

Yield (%)

R	cis	trans	ratio
Н	42	47	0.9
Me	79	3	26.3

cyclohexane-1,2-diols via ul-selective pinacolizations.^{53,73} Intramolecular pinacolic coupling is a particularly efficient method for the construction of cyclooctanes via cyclization of 1,8-dialdehydes. Takeshita and coworkers for example have used this route in their syntheses of the fusicoccin (**20**; equation 46)⁸¹ and the ophiobolins (**21**) and (**22**; equation 47).⁸² Pinacolic cyclizations are not restricted to the formation of small- and medium-sized rings. Through the use of high dilution techniques macrocycle synthesis, via C—C bond formation, is highly effective. Thus, nickel-promoted coupling of the diketone (**23**) gives the meso diol (**24**; equation 48) stereoselectively,⁸³ whilst the (+)-antipode (**26**) of the macrocyclic antibiotic (*S*,*S*)-(–)-grahamimycin A was recently synthesized using the chemoselective pinacolization of the homochiral dialdehyde (**25**; equation 49).⁸⁴



i, TiCl₄, Mg(Hg), THF, 32%



i, TiCl₄, Mg(Hg), THF, 43%



i, 2 SmI₂, THF, 47%




i, TiCl₄, Zn, pyridine, THF, 0 °C, 84%





(21) $R^1 = CO_2H$ (22) $R^1 = CH_2OH$ (46)

i, TiCl₄, Zn, THF, reflux, 96%



i, Ni(OAc)₂, pyridine, DMF, H₂O, 1,2-diaminobenzene, 120 °C, 43%



2.6.4 α,β-UNSATURATED CARBONYL COMPOUNDS

The reductive coupling of α,β -unsaturated carbonyl compounds can lead to three types of dimerization products; pinacols (1,2-coupling), 1,6-diones (1,4-coupling), and γ -hydroxycarbonyl compounds (mixed coupling). These couplings are further complicated by the ability of the initially formed dimerization products to undergo further reactions and complex product mixtures can sometimes result.^{85,86}

In electrochemically initiated reactions coupling at the β -carbon to give 1,6-diones is preferred; even 2,2,6,6-tetramethyl-4-hepten-3-one couples selectively *via* this mode.⁸⁷ Pinacol formation is favored when β -coupling is sterically hindered. Thus polyenones, such as retinal, are hydrodimerized to the corresponding pinacol in the presence of efficient hydrogen donors (equation 50).⁸⁸ In the absence of added hydrogen donors the yield of retinal pinacol is much reduced, even though α - and β -ionone are efficiently coupled under these conditions (equation 51).⁸⁹



i, Hg, MeCN, Bu₄NClO₄, CH₂(CO₂Et)₂; ii, Hg, MeCN, Bu₄NOAc, 1 M AcOH



i, Hg, MeCN, Bu₄NOAc, 1M AcOH, 46%

Product distribution in couplings promoted by low-valent metals is dependent on the Lewis acidity/complexing power of the metal employed, and either 1,2- or 1,4-coupling is possible. Thus, cyclohexenone is selectively coupled at the β -position using a TMSCl, TiCl₄, Mg, HMPA reagent system (equation 52),⁹⁰ whilst methyl vinyl ketone couples *via* the 1,2-mode to give the corresponding pinacol with the less coordinating TiCl₄, Mg, Bu^tOH reagent system (equation 53).⁹¹ Unsaturated carbonyls are



i, Me₃SiCl, Mg, TiCl₄, HMPA, 80 °C; ii, HCl, aq. EtOH, 65%



i, Mg, TiCl₄, Bu^tOH, 25%

Coupling Reactions

also coupled to pinacols under acidic conditions using low-valent zinc as the radical initiator (equation 54).^{16,92} Indeed efficient cross-coupling is possible with this reagent. Organomanganese reagents are normally associated with oxidative radical coupling processes. Organomanganese(II) compounds however can effect β -dimerization of enones (equation 55), although generally a mixture of conjugate addition and hydrodimerization results.⁹³



i, PrⁱMn, THF, -30 °C to r.t., 89%; ii, PrⁱMnMgCl, THF, -30 °C to r.t., 80%; iii, BuMgCl, 5% MnCl₂, THF, r.t., 77%

Generally the 1,2-coupling of α , β -unsaturated carbonyls is not highly stereoselective and generates a mixture of stereoisomeric products. A recently developed low-valent titanium species derived from Cp₂TiCl₂/Bu^sMgI overcomes this problem and is very effective at promoting such pinacolizations with high *lk-threo*-selectivity (equation 56).²⁸ Intramolecular coupling of enones is also possible and by using



i, Cp₂TiCl₂, Bu^sMgCl, THF, 98%



i, Hg, Et₄NCl, aq. MeCN, -1.80 V, 81%



i, Hg/Pt, Bu₄NBF₄, DMF, -2.35 V

linked cyclohexenones provides a stereoselective method for perhydrophenanthraene synthesis, suitable for application to steroidal systems (equation 57).⁹⁴ In a related manner, selective 1,6-coupling of the *cis* decalin dione (27) gives the tricycle (28), which then undergoes aldol cyclization to give the tetracycle (29; equation 58).⁹⁵

2.6.5 IMINE COUPLINGS

2.6.5.1 Intermolecular Couplings

The reductive couplings of imines show some significant differences from those of their carbonyl counterparts.⁹⁶ Thus in the coupling of *N*-acylimines promoted by low-valent metals the product 1,2-diamines are frequently formed with higher (\pm) -selectivity than are similar 1,2-diols (equation 59).⁹⁷ Smith and coworkers have shown that this phenomenon results from equilibration of dimeric diamines (**31**) and the precursor monomeric radical anions (**30**), to give the thermodynamically preferred (\pm) -isomer, rather than from selective *lk*-coupling (equation 60).⁹⁸ Coupling of *N*-benzylimines also shows some selectivity for the (\pm) -amine.⁹⁹

$$Ph \qquad N \qquad Ph \qquad i \qquad Ph \qquad NHPh \qquad (59)$$

$$Ph \qquad NHPh \qquad (59)$$

$$(\pm):meso = 15:1$$

i, Na, THF, 45-50 °C, 99%

$$Ph \stackrel{\bullet}{\longrightarrow} \overline{N}, Ph \stackrel{\bullet}{\longrightarrow} Ph \stackrel{\overline{N}}{\longrightarrow} Ph \qquad (60)$$

$$(30) \qquad (31)$$

In general, N-alkylimines show little reactivity with McMurry-type titanium reagents.¹²⁵ Mangeney however has demonstrated that symmetrical vicinal (\pm)-diamines can be prepared using low-valent titanium species generated from TiCl4/Mg(Hg) (equation 61).¹⁰⁰ Furthermore with this reagent coupling is not restricted to aryl (activated) imines. N-Alkylimines derived from aldehydes or ketones are efficiently coupled using a Pb/Al bimetal redox system (equation 62);¹⁰¹ while unsubstituted 1,2-diamines can be prepared from either N-(trimethylsilyl)imines or nitriles using the niobium reagent NbCl4(THF)₂ (equation 63).¹⁰² Coupling proceeds with excellent *lk*-diastereoselectivity with (\pm):*meso* ratios typically 16:1

i, HgCl₂, Mg, THF, 0 °C to r.t., 40-75%

$$Ar \xrightarrow{R} Ph \qquad i \qquad Ar \xrightarrow{R} N \xrightarrow{Ph} Ph \qquad Ar \xrightarrow{R} N \xrightarrow{Ph} Ph \qquad (62)$$

i, PbBr₂, Al, TFA, THF, r.t., 62–84% (aldimines) i, PbBr₂, Al, AlBr₃, THF, r.t., 63–90% (ketimines) to 24:1. Aldimines can also be coupled via an aminative reduction of aldehydes using either VCl3¹⁰³ or McMurry-type conditions¹⁰⁴ to give 1,2-diarylethylenediamines (Scheme 3). Electroreductive dimerizations of N-benzylimines generally give (\pm) :meso ratios of ca. 1:1.¹⁰⁵ Modest chiral induction has been observed using chiral coelectrolytes (equation 64).¹⁰⁶ In contrast to metal-promoted dimerizations, photopinacolic reactions of N-arylimines show a high preference for the meso product via ul-coupling of neutral ketyl-type radicals (equation 65).¹⁰⁷ For N-alkylimines, however, the ratio is dependent on the nature of the alkyl substituent (equation 66).¹⁰⁸

$$R \bigvee NSiMe_{3}$$
or
$$i,ii$$
RCN + Bu_{3}SnH
$$RCN + Bu_{3}SnH$$

 $X = \bigvee^{l} (NEt_2)_2$ ArCHO X = TiClTi⁰ NR $(\pm):meso = 1:1$ VCl₃, LiNR₂, 14–54% TiCl₄, LiNR₂, 23-81% Scheme 3 √HBn (64) NBn NHBn de 5.3% (+)-(S) Me₃ I⁻, EtOH aq., MeOAc, 70% i, Hg, Ph (-)-(R)NHAr NHAr i, hv, EtOH, 65-95%

i, NbCl₄(THF)₂; ii, KOH or KF, 40-73%

(65)



i, hv, EtOH, 50-70%

2.6.5.2 Intramolecular Couplings

Diastereoselective reaction of diimines with sodium in ether leads to the formation of macrocyclic ethers of the type (32; equation 67) and (33; equation 68) in modest yields.¹⁰⁹ However, there are no reports of examples using the templating effect of low-valent transition metals, as used to good effect for intramolecular carbonyl couplings. Electrochemical reduction of imines has been used to produce a variety of bicyclic structures *via* cyclization–aromatization (equation 69) or transannular cyclization (equation 70).¹¹⁰



i, Na, Et₂O, reflux, 6–16%



i, Hg, -2.70 V, Bu₄NNO₃, DMF, 60%



i, Hg, -2.6 to -2.7 V, Bu₄NClO₄, glyme, 94%

2.6.6 THIOCARBONYL COUPLINGS

The reductive coupling of thiocarbonyl compounds to give alkenes can be performed under a range of conditions.¹¹¹ This process, the Gatterman reaction,¹¹² in general requires the presence of conjugated electron-withdrawing groups to facilitate reaction. The traditionally used metal initiators such as copper or iron powder or zinc dust were initially replaced by more efficient initiators such as antimony, bismuth, dehydrogenated Raney nickel or silver.^{111,113} Thus, for example, thioacetophenone can be coupled to give the *trans*-stilbene (**34**) by Raney nickel (18% yield)¹¹³ or bismuth (35% yield),¹¹⁴ but not by copper (equation 71). These initiators have in turn been superseded by even more efficient reductants, including copper powder in DMSO,¹¹⁵ FeCl₃/NaBHEt₃,¹¹⁶ TiCl₃/K,¹²⁵ and MgC₈ (Scheme 4).⁵⁵



Scheme 4

Interestingly, there are no examples of the synthesis of thiopinacols by the coupling of thioketyl radicals. Moreover the longer C—S bond allows for the possibility of alternate reaction pathways and both episulfides and 1,2-dithietanes have been postulated as reaction intermediates.¹¹¹ Finally, certain thiocarbonyls can also undergo thermally induced alkene formation (equation 72).¹¹⁷



i, P₂S₅, C₆H₆, reflux; ii, 100 °C, 2 h, 50-70%

2.6.7 McMURRY-TYPE COUPLINGS

Although the reduction of carbonyls to alcohols or hydrocarbons is well documented, no general method was available for the reductive coupling of carbonyl compounds directly to alkenes prior to the use of low-valent transition metals. The reductive coupling of carbonyl compounds by low-valent transition metals is now an important method for C-C bond formation, and has been widely reviewed.¹¹⁸

2.6.7.1 Intermolecular Couplings

The reductive pinacolic coupling reaction was introduced by Sharpless, who demonstrated that alkenes could be synthesized from aldehydes or ketones by treatment with tungsten-based reagents derived from WCl₆/BuLi (Scheme 5).¹¹⁹ Aryl aldehydes are more efficiently coupled using the WCl₆/LiAlH₄ system developed by Fujiwara (equation 73).¹²⁰ Electrochemically generated tungsten species have also been used (equation 74).¹²¹ Active metal slurries prepared from MoCl₅/LiAlH₄, ZrCl₄/LiAlH₄, and VCl₃/LiAlH₄ also promote coupling but with only modest to poor yields (8-33%).¹²² Low-valent niobium generated from NbCl₅/NaAlH₄ or NbCl₅/LiAlH₄ is more effective and reductively couples both aromatic aldehydes and ketones stereoselectively (equation 75).¹²³ Interestingly the reaction selectively gives (E)-alkenes from aldehydes but (Z)-alkenes from ketones.

0

.

The most widely used systems are those employing low-valent titanium species. The reductive coupling of carbonyl compounds by titanium reagents was independently discovered in 1973 by Trylik²⁴ and Mukaiyama,²³ and has since been extensively developed by McMurry and his coworkers. 118d, 124-126 The McMurry reaction, as it has become known, has proved to be a valuable synthetic tool for the construction of C-C bonds.

There are in fact 10 titanium-based reagents commonly used for the reductive coupling of carbonyls to alkenes; these are $TiCl_3/Mg$,²⁴ $TiCl_4/Zn$,²³ $TiCl_3/LiAlH_4$,¹²⁴ $TiCl_3/K$ or /Li,^{125,127} $TiCl_2/Zn$,¹²⁸ $TiCl_3/Zn/Cu$,¹²⁵ $TiCl_4/Al/AlCl_3$,⁵³ $TiCl_4/Mg(Hg)$,⁵³ $CpTiCl_3/LiAlH_4$,⁵³ and $TiCl_3/CgK$.^{55,129} By far the most widely used reagent is the McMurry reagent, TiCl₃/LiAlH₄. However more reliable and consistent results are obtained using potassium or lithium metal for generation of the titanium(0) species, 125,127

Coupling Reactions

whilst TiCl₃/Zn/Cu is the reagent of choice for intramolecular variants.¹²⁵ The nature of the titanium species formed is dependent on the method employed. Thus the TiCl₃/LiAlH₄ system can produce Ti⁰ or Ti¹ depending on the molar ratio of TiCl₃ to LiAlH₄, whilst the reactive species generated from either the TiCl₃/alkali metal or Zn/Cu systems is Ti^{0.125} The mechanism outlined in Scheme 6 has been proposed,¹³⁰ and proceeds through an intermediate titanium pinacolate. Indeed pinacols can be isolated from these reductive couplings after short reaction times. It has also been shown that 1,2-diols are deoxygenated to alkenes by Ti⁰ reagents.¹²⁵ The stereochemistry of the alkene formed from aliphatic carbonyls is controlled by the rate-determining deoxygenation step. When the energy difference between the (*E*)- and (*Z*)-alkene exceeds 4–5 kcal mol⁻¹ (1 kcal = 4.18 kJ) the (*E*)-isomer is formed selectively (Scheme 7).¹³¹ Alkyl aryl ketones however can couple reductively to give predominately the (*Z*)-isomer (equation 76).¹³² This phenomenon has been rationalized as arising from π -complex formation (**35**) between the phenyl rings and Ti⁰.¹³² Increased steric hindrance favors the (*E*)-isomer whilst *para* electron-withdrawing groups promote pinacol formation.



Scheme 6



Scheme 7



i, TiCl₄, Zn, pyridine, THF, reflux, 44-81%



(35)

In general the McMurry reaction has been limited to substrates lacking other reducible functional groups.¹²⁵ More recently, and contrary to earlier results, various ester¹³³ and haloaryl groups¹³⁴ have proven to be compatible with the reaction conditions. Indeed such selectivity was observed in a new synthesis of the antitumor agent tamoxifen (**36**) using modified McMurry conditions (equation 77).¹³⁵



i, TiCl₄, Zn, THF, reflux, 54%; ii, Me₂NH, EtOH, 75 °C, 83%

The intermolecular McMurry reaction has proven a particularly valuable route to alkenic hydrocarbons. This is especially true for polyenes and sterically hindered alkenes. For example, β -carotene (37; equation 78) can be prepared in 85% yield by reductive dimerization of retinal.^{124,136} Isorenieratene (38; equation 78) is similarly synthesized in 96% yield,¹³⁷ and the aldehyde (39; equation 79) has been ela-



borated into perhydrosqualene via coupling followed by hydrogenation.¹³⁸ Highly conjugated alicyclic systems such as ω, ω' -diazulenylpolyenes (40; equation 80)¹³⁹ and di-2-thienylpolyenes (41; equation 81)¹⁴⁰ have also been prepared.



i, TiCl₃, Zn, THF, reflux, 63-71%

Strained and sterically hindered alkenes are of special interest in physical organic chemistry. By their very nature these structures are difficult to synthesize and the McMurry reaction represents the only viable route to many such compounds (see Table 2). Torsionally distorted alkenes, capable of exhibiting optical activity without a chiral center, have also been prepared.¹⁵¹ If the reaction is carried out in a chiral solvent the (\pm) isomer is formed selectively (equation 82).



Unsymmetrical alkenes can be prepared by mixed intermolecular reactions if one of the components, commonly acetone, is used in excess (equation 83).¹⁵² As the isopropyl group is a common subunit of many terpenes this method provides a valuable route for its introduction. Pattenden and Robertson used such a reaction followed by a Grob-type fragmentation in their preparation of the daucenone (42) from the readily enolized ketone (43).¹⁵³ The bicycle (42) was used as an intermediate for the synthesis of the diterpene (\pm)-isoamijiol (44; equation 84). Mixed couplings are not restricted to acetone, and almost any carbonyl may be used. For example, Paquette *et al.* employed the aldehyde (45)¹⁵⁴ in a synthesis of (\pm)- α -vetispirene (46; equation 85).¹⁵⁵ More complex mixed couplings are also possible. For example, the aldehydes (47) and (48) are coupled efficiently to the stilbene (49), which in turn is converted to phenan-threne alkaloids such as atherosperminine and thalictuberine (equation 86).¹⁵⁶



i, TiCl₃, LiAlH₄, acetone (4 equiv.), reflux, 63%



i, TiCl₃, Li, acetone, DME, reflux, 76%; ii, HF, THF, reflux, 62%

Substrate	Product	Yield (%)	Ref.
Pr ⁱ = 0	Pr ⁱ Pr ⁱ Pr ⁱ	6–12	141, 142
o		13	143
DO		85–87	124, 144, 145
	R	28–85	146
° C		55–77	147, 148
R	R	53–95	149, 150
SiMe₃ +		iMe ₃	(85)

Table 2	Preparation of Sterically Hindered Alkenes by the McMurry Coupling Reaction

i, TiCl₃, Li, DME, reflux, 50-60%

(46)

сно

(45)



i, TiCl₃, Li, DME, reflux, 55-67%

When one of the reactants in a mixed coupling is a diaryl ketone an excess of the second component is not always necessary (Scheme 8),^{125,152} since diaryl ketones are readily reduced to dianions, which then react selectively with unreduced aliphatic carbonyls.



2.6.7.2 Intramolecular Couplings

The intramolecular McMurry reaction has been used to prepare cycloalkenes of ring size 3 through 16 and up to 22, in good yields (46–95%).^{157,159} The method is quite general and is applicable to the intramolecular coupling of dialdehydes, keto aldehydes or diketones. Through utilization of the templating effect conferred by deoxygenation on a titanium surface the reaction is highly efficient and is not affected by ring size effects in contrast to the related acyloin and Thorpe–Zeigler cyclizations. The versatility of the intramolecular McMurry reaction is demonstrated by the variety of ring types that have been synthesized. These range from highly strained cyclopropenes (equation 87)¹⁵⁷ and cyclobutenes¹⁵⁸ through medium-sized rings,^{157,159} and on up to macrocycles such as (**50**; equation 88).¹⁵⁹ Although a number of titanium reagents have been employed for intramolecular couplings, the TiCl₃/Zn/Cu system¹²⁵ generally gives the best yields and represents the reagent of choice for the synthesis of medium ring and macrocyclic alkenes.

A variety of natural products have been synthesized via application of reductive pinacolic couplings. An early example is the cyclization of the keto aldehyde (51) to the alkene (52; equation 89); a precursor to the terpene, cuparene (18).^{127a} A modified McMurry procedure provides a novel route to steroids such as estrone methyl ether, via chemoselective cyclization of the keto aldehyde (53; equation 90).¹⁶⁰ The



i, TiCl₃, LiAlH₄, THF, reflux R = Me, 46%; R = Et, 40%



i, TiCl₃, Zn/Cu, DME, reflux, 83%

critical importance of the type of titanium reagent used is emphasized in Clive's recent synthesis of (+)compactin and (+)-mevinolin (equations 91 and 92). Model studies¹⁶¹ demonstrated that coupling of the keto aldehyde (54) was efficiently promoted by McMurry's TiCl₃/Zn/Cu reagent (equation 91). For the total syntheses however cyclization of the key keto aldehyde (55; equation 92) required the use of the Weidmann reagent TiCl₃/C₈K in a specific molar ratio with respect to (55).¹⁶² The use of any other low-



i, TiCl₃, K, THF, reflux, 55%



(55)

i, TiCl₃, C₈K, DME, r.t. to reflux, 85%

Coupling Reactions

valent titanium reagent or departure from the ratio keto aldehyde:TiCl₃:C₈K of 1:17:34 gave much reduced yields. Transannular coupling of the diketone (56) is a key step in a short photochemical route to (\pm) -hirsutene (equation 93),¹⁶³ whilst regioselective coupling of the triketone (57) provides a new route to zizaene sesquiterpenes such as (\pm) -isokhusimone (58; equation 94).¹⁶⁴



McMurry and coworkers have synthesized a range of medium and macrocyclic terpenes using the TiCl₃/Zn/Cu system.¹⁶⁵⁻¹⁶⁸ Of particular note are the syntheses of (\pm) -humulene (**59**) and (\pm) -flexibilene (**60**) which provide a show-case for transition metal based methodology (equation 95).¹⁶⁵ In a related fashion Jackson and Pattenden synthesized verticillene (**62**), a putative biogenetic precursor to the taxane alkaloids, by selective 1,2-coupling of the 1,12-dialdehyde (**61**; equation 96)¹⁶⁹ Several macrocyclic crown ethers of the type (**63**; equation 97) which act as enzyme catalytic models, have been prepared using titanium(0) or titanium(I) reagents.¹⁷⁰ Reductive coupling of the related dialdehyde (**64**) provides cyclophanes of the type (**65**; equation 98), which can be further elaborated to phenanthrenes, of which cannithrene II (**66**) is representative.¹⁷¹



(65)

(64)

i, TiCl₃, Zn/Cu, THF, reflux, 57-78%; ii, Cu^{II} decanoate, I₂, Et₂O, hv, 450 W, 51-78%



A wide range of alkenic compounds exhibiting interesting properties associated with their π -systems have been elaborated *via* the reductive coupling carbonyls. For example, Marshall and coworkers have studied the intramolecular McMurry reaction as a route to betweenanenes, a class of conformationally

Coupling Reactions

flexible bicyclic alkenes (equation 99).¹⁷² Nakazaki *et al.* have used a similar approach to optically active doubly bridged allenes of the type (67; equation 100).¹⁷³ A class of compounds with interesting intracyclic electronic properties are compounds such as (68) and (69) (equation 101). McMurry and coworkers have prepared these alkenes *via* cyclization of the appropriate diketones (70)¹⁷⁴ and (71)¹⁷⁵ respectively. It is worth noting that these cyclizations overcome considerable strain associated with distortion of the cyclohexane rings to boat topographies during coupling.¹⁷⁵ The alkene (68) exhibits an intracyclic π -sys-





i, TiCl₃, Zn/Cu, DME, reflux, 30%



 Table 3 Preparation of Cyclic Polyenes by the McMurry Coupling Reaction

Table 3 (continued)				
Substrate	Product	Yield (%) Ref		
CHO CHO		50 184		

tem capable of forming a stable square planar d^{10} -complex with silver.¹⁷⁴ The in-out bicycloalkane (72) was recently synthesized by McMurry and Hodge via the coupling of the keto aldehyde (73) followed by hydrogenation. Protonation of (72) gives a stable μ -hydrido-bridged carbocation (equation 102).¹⁷⁶

A wide range of cyclic polyenes, such as annulenes, porphycenes, cyclophanes and ferrocenophanes, have been prepared using McMurry methodology, and these are summarized in Table 3.

Cyclization of 1,5-diketones linked by a methylthiomethyl chain gives dihydrothiophenes (74), which upon oxidation and elimination provide stereodefined 1,3-dienes (equation 103).¹⁸⁶ Thus, meso diketones give cis, cis 1,4-disubstituted dienes whilst (±)-diketones give the *trans, cis* isomers.



43

185

i, TiCl₄, Zn, THF, 66-86%

Finally an interesting system for the synthesis of alkenes from carbonyls based on TMSCl/Zn has been developed by Motherwell *et al.*,¹⁸⁷ and is compatible with esters and halogen groups (equation 104). The reaction does not however proceed via a ketyl radical coupling but rather via an organozinc carbeniod (75) which is trapped by a second carbonyl molecule to give an intermediate epoxide (76). The intermediate epoxide is then deoxygenated to the product alkene (Scheme 9).



i, Me₃SiCl, Zn, THF, -50 °C, 85%

HO OHC



2.6.8 MIXED COUPLINGS

This section will deal with pinacolic couplings of fundamentally different C—X (X = O or NHR) groups, that is those with different reduction potentials. Examples involving similar C-X groups, such as aldehyde-ketone couplings have already been discussed under the relevant sections and will only be mentioned briefly.¹⁸⁸ A number of efficient methods have been devised for the cross-coupling of carbonyl compounds to mixed pinacols. This type of reaction is especially facile for the cross-coupling of aromatic and aliphatic ketones. Aromatic carbonyls especially diaryl ketones are much more readily reduced than their aliphatic counterparts. Indeed chemoselective reduction of aromatic systems in the presence of aliphatic carbonyls allows for efficient cross-coupling via nucleophilic attack of the aryl anion/radical.^{125,190} Clerici and coworkers have extensively studied such reactions of aryl ketones (77) with saturated and α , β -unsaturated carbonyl compounds, promoted by aqueous TiCl₃ (Scheme 10).¹⁸⁹ Under these conditions chemoselective formation of capto-datively stabilized radicals such as (78) is followed by subsequent trapping by the aliphatic carbonyl group. The use of an excess of the aliphatic carbonyl compound increases the ratio of mixed pinacol to dimer up to a maximum of 6:1. These reactions provide the first examples of radical addition to saturated carbonyls such as acetone and proceed via activation of the saturated carbonyl by coordination with titanium to give the electrophilic intermediate (79) followed by combination with the ketyl radical (78; equation 105). Fujiwara has published a potentially important method for the coupling of diaryl ketones with electrophiles, including ketones, nitriles, epoxides and CO₂, mediated by ytterbium metal (equation 106).¹⁹⁰





Scheme 10



Mixed coupling of ketones and α , β -unsaturated ketones can be profoundly affected by the pH of the solution. Thus the Wieland-Miesher ketone (80) undergoes intramolecular coupling to give the cyclopropanol (81) under basic conditions;¹⁹¹ whilst in neutral media 1,2-hydrodimerization occurs leading to the pinacol (82; equation 107).¹⁹²



i, Li, NH₃, THF, -78 °C, 87%; ii, Hg, MeOH, 0.1M KCl (1:1), -1.7 V, 71%

The mixed coupling of aliphatic compounds is also well established. Through the use of an excess (usually 10-fold) of one of the reactants cross-coupling of aliphatic carbonyls is highly effective.¹⁹³ This protocol was first developed by Corey⁵³ and then applied to reductive pinacolic couplings by McMurry.¹²⁵ Li *et al.* applied this method to the synthesis of the alkaloid isoharringtonine (**83**; equation 108).¹⁹⁴



i, CpTiCl₃, LiAlH₄, THF, MeO₂CCHO

Early work on the electropinacolic coupling of aldehydes with imines was not promising since, as the reaction is reversible, complex product mixtures are formed.¹⁹⁵ However, recent developments in mixed coupling reactions have established some highly efficient and selective methods for the coupling of aldehydes or ketones with oximes or imines.^{72,196} Thus stereoselective nonreversible coupling of ketones with oximes using TMSCI/Zn proceeds in excellent yield to give compounds of the type (**84**; equation 109),⁷² and a highly versatile and selective method for the intermolecular coupling of imines with aldehydes or ketones uses a niobium(III) reagent providing a diastereoselective route to α -aminoalcohols



i, Me₃SiCl, Zn, 2,6-lutidine, THF, reflux, 84%

(equation 110).¹⁹⁶ Chemoselective formation of the metallaaziridine (85) in the latter reaction is followed by a highly efficient reaction with the carbonyl compound, even with only a 1.5 molar excess of the imine; no doubt this excellent method will be well exploited in the near future.



- 2

ds = 3:1 to 83:1

i, NbCl₃ (DME), THF; ii, R³COR⁴, THF, 33-97%

 $\begin{array}{c}
 R^{1} \\
 N \\
 R^{2} \\
 (85)
\end{array}$

2.6.9 MISCELLANEOUS COUPLINGS

There are a number of alternative methods for the synthesis of 1,2-diols and their derivatives that do not proceed *via* the coupling of two carbonyl groups. Ketyl radicals can be generated by the alkylative deoxygenation of carboxylic acids, and in the presence of TiCl₃ coupled to give 1,2-diols (equation 111).¹⁹⁷ However, yields are limited (22–33%) and complex product mixtures are often formed.

$$\begin{array}{c} O \\ R^{1} \\ O \\ O \\ H \end{array} \xrightarrow{i} R^{1} \\ R^{2} \\ R^{2} \\ R^{2} \\ R^{2} \\ R^{1} \\ R^{2} \\ R$$

i, TiCl₃, R²Li, DME, -78 °C to r.t., 22-33%

The formation of pinacols *via* the dimerization of alcohols is not feasible because of disproportionation or oxidation of the intermediate radical. Alcohols do however couple with ketyl radicals in photochemically initiated cross-pinacolization reactions (equation 112).¹⁹⁸ Product distribution is highly dependent on the light intensity. Photochemical cross-pinacolization has been used for a one-step conversion of osulose compounds to the corresponding hydroxymethyl branched derivatives (equation 113).¹⁹⁹ Alcohols protected as silyl ethers are efficiently coupled under free-radical conditions using di-*t*-butyl peroxide, to give, after hydrolysis, the corresponding pinacols in moderate to good yields (26–89%; equation 114).²⁰⁰ Electroreductive cross-coupling of diaryl ketones with aldehyde-derived enol acetates yields unsymmetrical pinacols (equation 115) *via* nucleophilic attack onto the enol double bond.²⁰¹

_ .

i, hv, MeOH



i, MeOH, hv, 450 W

Yield (%)

$$\alpha$$
-OH β -OH
 $R^1 = R^2 = R^3 = Me$ 20 22
 $R^1 = H, R^2 = Et, R^3 = Ph$ 30 25



 $(\pm):meso = 1:1$

i, (Bu^tO)₂, 145 °C, N₂ pressure; ii, ROH, 26-89%



i, C-electrode, RCH=CHOAc, Et₄NTs, DMF, H₂O, 200 mA, 45-76%

The reduction of dialkyl acetals of aromatic aldehydes or ketones with a titanium(II) reagent gives pinacol ethers or the corresponding alkene in high yield (46–100%; equation 116).²⁰² Acetals derived from aliphatic aldehydes or ketones are reduced to the corresponding ethers.



i, TiCl₄, LiAlH₄, THF, r.t., 100%

2.6.10 CARBONYL-ALKENE CYCLIZATIONS

In addition to coupling with heteroalkenes (C=O, C=N and C=S) carbonyls also undergo intramolecular reaction with nonconjugated C=C bonds. Together with the related reactions with alkynes and allenes these cyclizations provide regio- and often stereo-selective routes to five- and six-membered rings via C-C bond formation with net retention of functionality.^{9c,203} Although there are no intermolecular examples, the intramolecular variant is a powerful method for the synthesis of polycyclic systems. Thus, electroreductive cyclization of ω -unsaturated alkenic ketones gives cyclic tertiary alcohols of the type (86) regio- and stereo-selectively (equation 117).^{204,205,222} Cyclization occurs through formation of the ketyl radical anion and *exo* radical addition to the alkene to form the intermediate (87). The developing charge interactions formed between the alkyl radical and the oxyanion in (87), within the electrode double layer, control the stereochemical integrity of the reaction. The reaction is completely regio- and stereo-specific in contrast to similar chemically induced cyclizations with either TiCl₄/Mg,²⁰⁴ or Na/Bu^tOH/*N*-ethylpyrrolidine,²⁰⁶ both of which are stereorandom. A related cyclization of the ketoal-kene (88) with TMSCl/Zn shows a 5:1 preference for the *trans* isomer (89; equation 118).⁷²



i, Me₃SiCl, Zn, 2,6-lutidine, THF, reflux, 75%

The cyclizations of ketyl radicals onto double bonds occurs in an antiperiplanar fashion as demonstrated by the respective cyclizations of the epimers (90) and (91).²⁰⁷ In ketone (90) the ether moiety is suitably disposed to act as a leaving group allowing *endo* cyclization to the alkene (92), whilst the epimeric ether (91) gives the cyclopentane (93) via the *exo* mode (equations 119 and 120). Stereochemical control can also be imparted by use of a chelating auxiliary as in the samarium-mediated cyclization of the ketone (94) to the spirolactone (95; equation 121).^{73,208} The lactone moiety of (94) functions as a Lewis base enabling chelation of the Sm³⁺ ion and thus stereoselective cyclization (equation 122). Photochemical radical generation can also be stereoselective, thus the ketones (96) and (97) give their respective bicyclic tertiary alcohols in good yields (equation 123).²⁰⁹ A short synthesis of the monoterpenoid (\pm)-actinidine (98) has been accomplished using a photoreductive cyclization of the ketone (99; equation 124).²¹⁰

The radical cyclization of aldehydes onto alkenes was first demonstrated by Hutchinson and coworkers using a TMSCl/Mg reducing agent (equation 125).²¹¹ The cyclization of the aldehyde (100) to loganin tetraacetate (101) serves as a model reaction for studying the mechanism of the biological conversion of loganin to secologanin. In a related fashion cyclization of the (stannyloxy)alkyl radical gener-



i, Na, THF, 64%



i, Na, THF, 70%



i, 2 SmI₂, Bu^tOH, THF, -78 °C to r.t., 87%





i, hv, HMPA, n = 1, 81%; n = 2, 76%



i, hv, Et₃N, MeCN, 50%

ated from aldehyde (102) gives the alcohol (103; equation 126).²¹² Further exploitation of the cyclization of aldehyde-derived radicals should provide a general route to polycyclic secondary alcohols.

Transannular cyclization of ketoalkenes was first reported in 1965. Treatment of the conformationally restricted ketone (104) with sodium in moist ether gave the alcohol (105; equation 127).²¹³ Similarly the ketoalkene (106) transannulated in 73% yield by exposure to sodium in refluxing propanol (equation 128).²¹⁴ Conformational restriction is not a prerequisite for transannular reaction; thus the caryophyllene (107) undergoes cyclization to the alcohol (108) with lithium in liquid ammonia (equation 129).²¹⁵ Transannulation across a nine-membered ring has also been observed upon treatment of ketone (109; equation 130) with samarium diiodide, *via* cyclization of the ketoalkene (110).²¹⁶ Of more practical importance, the electrochemical transannulation of the cyclooct-4-en-1-one gives the bicyclo[3.3.0]octanol (111; equation 131).²¹⁷







i, Bu₃SnH, AIBN, C_6H_6 , reflux, 90%



i, Na, moist ether









i, Li, NH3 , 30%



i, C-electrode, MeOH, dioxane, Et₄NTs, -2.70 V (vs. SCE), 69%

2.6.11 CARBONYL-ALKYNE CYCLIZATIONS

Cyclizations of ketoalkynes constitute an important method for the synthesis of ring junction allylic alcohols such as those in the polycyclic systems of gibberellins, capnellenols, and amijiols. Thus reductive cyclization of ketones (112; equation 132) and (113; equation 133) form the c/p rings of the gibberellic acid^{218–220} and provide a one-step alternative to the corresponding pinacolic coupling/alkenation of the keto aldehyde (16).⁶⁹ Ketoalkynes have also been cyclized *via* electroreduction,^{221,222} photoreduction,²²³ and using the sodium naphthalene radical anion^{224,225} or TMSCl/Zn (equations 134–139).⁷² It should be noted that the enone (114) is cyclized by sodium naphthalene radical anion, but is 1,4-reduced under protic conditions using Na/NH₃ (equation 138).^{224,225} Treatment of the homochiral ketone (115) with Na/NH₃ furnishes the hydrindane (116), a potential synthon for optically active steroids and terpenes (equation 140).²²⁶

The capnellene and hirsutene marine sesquiterpenes are ideal candidates for radical cyclizations and both have been elaborated *via* carbonyl-alkyne cyclizations (equations 141-143). Thus treatment of the ketone (117) with the sodium naphthalene radical anion gives the triquinane (118). Subsequent allylic



i, Li, NH₃, Et₂O, NH₄SO₄, ~50%

oxidation and inversion gives (\pm) - $\Delta^{9(12)}$ -capnellene-8 β ,10 α -diol (119).²²⁷ Similar photoreduction of the ketone (120) gives the tertiary alcohol (121) from the α -epimer, whilst the β -isomer of (120) is reduced to the alcohol (122).²²⁸ Conversion of the tertiary alcohol in (121) to a methyl group completes the synthesis of hirsutene. A particularly elegent tandem radical cyclization of the aldehyde (123) with SmI₂ provides the alcohol (124), a key intermediate in the synthesis of corriolin and hypnophilin stereoselectively (equation 143).²²⁹ A synthesis of the dolastane diterpene, (\pm)-isoamijiol (44; equation 144), utilizing only seven C—C bond-forming reactions, of which four involve radical intermediates, is completed by stereospecific reductive cyclization of the ketoalkyne (125) to the alcohol (126) followed by allylic



i, Li, NH₃, NH₄SO₄, 50% or K, NH₃, THF, NH₄SO₄, 60-70%



i, C-electrode, Et₄NTs, DMF, 75%



i, Hg, DMPBF₄, DMF, -2.70 V (vs. SCE), 85%



i, hv, Et₃N, MeCN, 58%



i, Na, C₁₀H₈, THF, 69-72%

Coupling Reactions



i, Na, C₁₀H₈, THF, 20–25%; ii, Na, NH₃, THF, 94%



i, Me₃SiCl, Zn, 2,6-lutidine, THF, reflux, 77%



i, Na, NH₃, (NH₄)₂SO₄ , ~90%

oxidation.¹⁵³ The stereochemical integrity of the unusual *trans* ring junction is controlled by steric interactions of the alkynic side chain and the azulenone nucleus. This cyclization has also been effected using homochiral (**125**) to furnish (+)-isoamijiol.²³⁰



i, Na, C₁₀H₈, THF, 30%; ii, SeO₂, Bu^tO₂H, CH₂Cl₂, 40%; iii, MsCl, Et₃N, CH₂Cl₂; KO₂, DMSO, DMF, -5 °C to r.t., 40%



i, Na, C₁₀H₈, THF, 41%; ii, SeO₂, Bu^tO₂H

2.6.12 CARBONYL-ALLENE CYCLIZATIONS

Reductive cyclization of ketoallenes (127) and (128) either electrochemically, or by brief exposure to the sodium naphthalene radical anion, gives the tertiary alcohols (129) and (130) respectively, via regioand stereo-specific reactions in an *exo* mode (equations 145 and 146)²³¹ The stereochemistry at the ring junction in (130) is *cis* with a *trans* relationship between the pendant vinyl group and the tertiary alcohol, in a similar fashion to ketoalkene cyclizations. Best yields are obtained electrochemically under nonenolizing conditions. In contrast, prolonged exposure to excess sodium naphthalene radical anion is not regioselective; thus ketone (131) gives a mixture of *exo* and *endo* double bond isomers (equation 147).²³²



i, Hg, -2.43 V (vs. Ag/AgI₂), Et₄NTs, DMF, 41%; ii, Na, C₁₀H₈, THF, 28%



i, Hg, -2.43 V (vs. Ag/AgI₂), ET₄NTs, DMF, 37%; ii, Na, C₁₀H₈, THF, 28%



i, Na, C₁₀H₈, THF, r.t., 5 h, 46%

2.6.13 REFERENCES

- (a) R. C. Fuson, Rec. Chem. Prog., 1951, 12, 1 (Chem. Abstr., 1951, 45, 10187c); (b) H. O. House, in 'Modern Synthetic Reactions', 2nd edn., Benjamin, New York, 1972, p. 167; (c) H. Kropf and J. Thiem, 'Houben-Weyl, Methoden der Organischen Chemie', ed. E. Müller, Thieme-Verlag, Stuttgart, 1973, vol. 6/1b/2, p. 1458; (d) T. Mukaiyama, Angew. Chem., Int. Ed. Engl., 1977, 16, 817; (e) P. Sobota and B. Jerzowska-Trzebiatowska, Coord. Chem. Rev., 1978, 26, 71; (f) T.-L. Ho, Synthesis, 1979, 1; (g) Y.-H. Lai, Org. Prep. Proced. Int., 1980, 12, 361; (h) B. E. Kahn and R. D. Rieke, Chem. Rev., 1988, 88, 733; (i) J.-M. Pons and M. Santelli, Tetrahedron, 1988, 44, 4295.
- 2. R. Fittig, Ann., 1859, 110, 23.
- 3. See references 1b, 1g, 37, 40 and 189 and refs. therein.
- 4. For an explanation of like (lk) versus unlike (ul) relative topicity and like (l) versus unlike (u) relative configuration see D. Seebach and V. Prelog, Angew. Chem., Int. Ed. Engl., 1982, 21, 654.
- 5. For a detailed analysis and review of such reductions see J. W. Huffman, in 'Comprehensive Organic Synthesis', ed. B. M. Trost, Pergamon Press, Oxford, 1991, vol. 8, chap. 1.4.
- M. Hulce and T. LaVaute, Tetrahedron Lett., 1988, 29, 525; A. Clerici and O. Porta, J. Org. Chem., 1987, 52, 5099.
- G. Majerus, E. Yax and G. Ourisson, Bull. Soc. Chim. Fr., 1967, 4143; A. Coulombeau and A. Rassat, Chem. Commun., 1968, 1587; J. W. Huffman and J. T. Charles, J. Am. Chem. Soc., 1968, 90, 6486; H. O. House, W. C. McDaniel, R. F. Sieloff and D. Vanderveer, J. Org. Chem., 1979, 43, 4316.
- H. L. Bradlow and C. A. Van der Werf, J. Am. Chem. Soc., 1947, 69, 1254; M. R. Kegelman and E. V. Brown, J. Am. Chem. Soc., 1953, 75, 4649; R. Adams and E. W. Adams, Org. Synth., Coll. Vol., 1941, 1, 459; D. G. Botteron and G. Wood, J. Org. Chem., 1965, 30, 3871; F. Munoz-Madrid and J. Pascual, An. Quim., 1978, 74, 1270.
- 9. (a) M. M. Baizer and J. P. Petrovich, Prog. Phys. Org. Chem., 1970, 7, 189; (b) L. G. Feoktistov and H. Lund, in 'Organic Electrochemistry', ed. M. M. Baizer, Dekker, New York, 1973, chap. 8; (c) H. J. Schäfer, Angew. Chem., Int. Ed. Engl., 1981, 20, 911.
- For early refs. see M. J. Allen, J. A. Siragusa and W. G. Pierson, J. Chem. Soc., 1960, 1045; M. J. Allen, M. Cohen, W. G. Pierson and J. A. Siragusa, J. Chem. Soc., 1961, 757; J. H. Stocker and R. M. Jenevein, Collect. Czech. Chem. Commun., 1971, 36, 925 and refs. therein.
- A. Schönberg, 'Preparative Organic Photochemistry', Springer-Verlag, New York, 1968, p. 203; S. G. Cohen, A. Parola and G. H. Parsons, Jr., Chem. Rev., 1973, 73, 141.
- M. Gomberg and W. E. Bachmann, J. Am. Chem. Soc., 1927, 49, 236; for a review see M. D. Rausch, W. E. McEwen and J. Kleinberg, Chem. Rev., 1957, 57, 417; for a modified method see F. F. Ebetino and E. D. Amstutz, J. Org. Chem., 1963, 28, 3249.
- 13. A. A. P. Schreibmann, Tetrahedron Lett., 1970, 4271.
- 14. T. H. Chan and E. Vinokur, Tetrahedron Lett., 1972, 75.
- 15. For a related example see ref. 51; for pinacolic couplings promoted by trimethylsilyl radical see A. G. Beaumont, R. W. Bott, C. Eaborn and R. A. Jackson, J. Organomet. Chem., 1966, 6, 671.
- 16. J.-H. So, M.-K. Park and P. Boudjouk, J. Org. Chem., 1988, 53, 5871.
- 17. T. Hiyama, M. Obayashi, I. Mori and H. Nozaki, J. Org. Chem., 1983, 48, 912; for related examples using triphenylsilyllithium see D. Wittenberg, T.-C. Wu and H. Gilman, J. Org. Chem., 1959, 24, 1349; H. Gilman and G. D. Lichtenwalter, J. Am. Chem. Soc., 1958, 80, 607.
- 18. J. B. Conant and H. B. Cutter, J. Am. Chem. Soc., 1926, 48, 1016.
- 19. D. D. Davis and W. B. Bigelow, J. Am. Chem. Soc., 1970, 92, 5127.
- N. Katsaros, E. Vrachnou-Astra, J. Konstantatos and C. F. Stassinopoulou, Tetrahedron Lett., 1979, 4319; J. Konstantatos, E. Vrachnou-Astra, N. Katsaros and D. Katakis, J. Am. Chem. Soc., 1980, 102, 3035.

- 21. K. Ito, S. Nakanishi and Y. Otsuji, Chem. Lett., 1980, 1141.
- 22. H. Inoue, M. Suzuki and N. Fujimoto, J. Org. Chem., 1979, 44, 1722.
- 23. T. Mukaiyama, T. Sato and J. Hanna, Chem. Lett., 1973, 1041.
- 24. S. Tyrlik and I. Wolochowicz, Bull. Soc. Chim. Fr., 1973, 2147.
- 25. For a detailed discussion of the reductive coupling of carbonyl compounds to alkenes see Section 2.6.7.
- 26. H. Suzuki, H. Manabe, R. Enokiya and Y. Hanazaki, Chem. Lett., 1986, 1339.
- 27. H. G. Raubenheimer and D. Seebach, Chimica, 1986, 40, 12.
- 28. Y. Handa and J. Inanaga, Tetrahedron Lett., 1987, 28, 5717.
- 29. T. L. Chen, T. H. Chan and A. Shaver, J. Organomet. Chem., 1984, 268, C1.
- 30. A. Clerici and O. Porta, Tetrahedron, 1982, 38, 1293. 31. A. Clerici and O. Porta, Tetrahedron Lett., 1982, 23, 3517.
- 32. M. J. Allen and H. Cohen, J. Electrochem. Soc., 1959, 106, 451; M. J. Allen, Can. J. Chem., 1959, 37, 257; J. F. Rusling and P. Zuman, J. Org. Chem., 1981, 46, 1906.
- 33. W. L. Bencze, C. A. Burckhardt and W. L. Yost, J. Org. Chem., 1962, 27, 2865; W. L. Bencze and M. J. Allen, J. Am. Chem. Soc., 1959, 81, 4015.
- 34. T. Imamoto, T. Kusumoto, Y. Hatanaka and M. Yokoyama, Tetrahedron Lett., 1982, 23, 1353.
- 35. J.-L. Namy, J. Souppe and H. B. Kagan, Tetrahedron Lett., 1983, 24, 765; for intramolecular examples of samarium diiodide-promoted pinacolic couplings see ref. 72.
- 36. Z. Hou, K. Takamine, Y. Fujiwara and H. Taniguchi, Chem. Lett., 1987, 2061; aromatic ketones can be cross-coupled with aliphatic ketones using this method, see ref. 190.
- 37. For examples see S. G. Cohen and J. I. Cohen, J. Am. Chem. Soc., 1967, 89, 164; S. G. Cohen and R. J. Baumgarten, J. Am. Chem. Soc., 1965, 87, 2996; R. S. Davidson, Chem. Commun., 1966, 575.
- 38. J. H. Štocker and D. H. Kern, J. Org. Chem., 1968, 33, 291.
- J. H. Stocker and D. H. Kern, J. Org. Chem., 1968, 33, 1270.
 D. Seebach and H. Daum, J. Am. Chem. Soc., 1971, 93, 2795; this homochiral amine hes also been employed in electrochemical and alkali metal induced pinacolic couplings, see ref. 49 and D. Seebach, H.-A. Oei and H. Daum, Chem. Ber., 1977, 110, 2316.
- 41. J. Grimshaw and J. S. Ramsey, J. Chem. Soc. C, 1966, 653.
- 42. R. E. Juday and W. L. Waters, Tetrahedron Lett., 1975, 4321
- 43. J. H. Stocker and R. M. Jenevein, J. Org. Chem., 1968, 33, 294.
- 44. A. Bewick and H. P. Cleghorn, J. Chem. Soc., Perkin Trans. 2, 1973, 1410.
- 45. A. Bewick and D. J. Brown, J. Chem. Soc., Perkin Trans. 2, 1977, 99.
- 46. W. J. M. van Tilborg and C. J. Smit, Tetrahedron Lett., 1977, 3651.
- 47. W. J. M. van Tilborg and C. J. Smit, Recl. Trav. Chim. Pays-Bas, 1978, 97, 89; for an example where the meso isomer is favored see V. J. Puglisi, G. L. Clapper and D. H. Evans, Anal. Chem., 1969, 41, 279.
- 48. F. Fournier and M. Fournier, Can. J. Chem., 1986, 64, 881.
- 49. D. Seebach and H.-A. Oei, Angew. Chem., Int. Ed. Engl., 1975, 14, 634.
- 50. W. J. M. van Tilborg and C. J. Smit, Recl. Trav. Chim. Pays-Bas, 1979, 98, 532.
- 51. V. Rautenstrauch, Synthesis, 1975, 787; see also ref. 14.
- 52. E. J. Corey and R. L. Carney, J. Am. Chem. Soc., 1971, 93, 7318.
- 53. E. J. Corey, R. L. Danheiser and S. Chandrasekaran, J. Org. Chem., 1976, 41, 260; for intramolecular applications including a synthesis of gibberellic acid see ref. 69.
- 54. For application of this method in the total synthesis of (±)-perhydrohistrionicotoxin see E. J. Corey, J. F. Arnett and G. N. Widiger, J. Am. Chem. Soc., 1975, 97, 430.
- 55. A. Fürstner, R. Csuk, C. Rohrer and H. Weidmann, J. Chem. Soc., Perkin Trans. 1, 1988, 1729.
- 56. S. Fukuzawa, T. Fujinami and S. Sakai, J. Organomet. Chem., 1986, 299, 179.
- 57. E. Kariv-Miller and T. J. Mahachi, J. Org. Chem., 1986, 51, 1041.
- 58. B. P. Mundy, D. R. Bruss, Y. Kim, R. D. Larsen and R. J. Warnet, Tetrahedron Lett., 1985, 26, 3927, and refs. therein.
- C. Grugel, W. P. Neumann, J. Sauer and P. Seifert, *Tetrahedron Lett.*, 1978, 2847.
 E. Touboul and G. Dana, J. Org. Chem., 1979, 44, 1397.
- 61. For an earlier example see J. Fried and N. A. Abraham, Tetrahedron Lett., 1964, 1879.
- 62. L. A. Paquette, I. Itoh and W. B. Farnham, J. Am. Chem. Soc., 1975, 97, 7280.
- 63. J.-M. Pons and M. Santelli, Tetrahedron Lett., 1986, 27, 4153.
- V. Rautenstrauch, B. Willhalm, W. Thornmen and U. Burgar, Helv. Chim. Acta, 1981, 64, 2109; S. K. Pradhan, K. R. Thakker and A. T. McPhail, Tetrahedron Lett., 1987, 28, 1813; J. W. Huffman, R. H. Wallace 64. and W. T. Pennington, Tetrahedron Lett., 1988, 29, 2527; see also ref. 5.
- 65. For reviews covering intramolecular examples see refs. 1g and 1i.
 66. R. Le Goaller and J.-L. Pierre, Can. J. Chem., 1977, 55, 757; see also T. J. Curphey, C. W. Amelotti, T. P. Layloff, R. L. McCartney and J. H. Williams, J. Am. Chem. Soc., 1969, 91, 2817; for a related chemically induced example see R. Le Goaller, M. Rougier, C. Zimero and P. Arnaud, Tetrahedron Lett., 1969, 4193.
- 67. G. W. Griffin and R. B. Hager, J. Org. Chem., 1963, 28, 599.
- Cyclobutane-1,2-diols have also been produced under Clemmensen-type conditions, see E. Wenkert and J. E. 68. Yoder, J. Org. Chem., 1970, 35, 2986.
- 69. E. J. Corey, R. L. Danheiser, S. Chandrasekaran, P. Siret, G. E. Keck and J.-L. Gras, J. Am. Chem. Soc., 1978, 100, 8031, 8034, and refs. therein; for an in depth analysis of this approach see R. L. Danheiser, 'The Total Synthesis of Gibberellic Acid', in 'Strategies and Tactics in Organic Synthesis', Academic Press, London, 1984, p. 22; for an alternative approach see ref. 218.
- 70. P. E. Eaton, P. G. Jobe and K. Nyi, J. Am. Chem. Soc., 1980, 102, 6636.
- 71. H. Takeshita, A. Mori and S. Nakamura, Bull. Chem. Soc. Jpn., 1984, 57, 3152; for a related example see N. Kato and H. Takeshita, Bull. Chem. Soc. Jpn., 1985, 58, 1574. 72. E. J. Corey and S. G. Pyne, Tetrahedron Lett., 1983, 24, 2821.
- 73. G. A. Molander and C. Kenny, J. Org. Chem., 1988, 53, 2132.

Coupling Reactions

- 74. J. Nakayama, S. Yamaoka and M. Hoshino, Tetrahedron Lett., 1988, 29, 1161; 1987, 28, 1799.
- 75. J. Nakayama, F. Murai, M. Hoshino and A. Ishii, Tetrahedron Lett., 1988, 29, 1399.
- 76. W. D. Hoffman, W. E. McEwen and J. Kleinberg, Tetrahedron, 1959, 5, 293.
- 77. F. Ammar, C. P. Andrieux and J. M. Saveant, J. Electroanal. Chem. Interfacial Electrochem., 1974, 53, 407 (Chem. Abstr., 1974, 81, 130 180s); C. P. Andrieux, J. M. Saveant and D. Tessier, J. Electroanal. Chem. Interfacial Electrochem., 1975, 63, 429 (Chem. Abstr., 1976, 84, 42 901j). 78. R. N. Gourley and J. Grimshaw, J. Chem. Soc. C, 1968, 2388.
- 79. T. Nonaka and M. Asai, Bull. Chem. Soc. Jpn., 1978, 51, 2976, and refs. therein.
- 80. E. J. Moriconi, F. T. Wallenberger, L. P. Kuhn and W. F. O'Connor, J. Org. Chem., 1957, 22, 1651.
- N. Kato, K. Nakanishi and H. Takeshita, Bull. Chem. Soc. Jpn., 1986, 59, 1109.
 N. Kato, H. Takeshita, H. Kataoka, S. Ohbuchi and S. Tanaka, J. Chem. Soc., Perkin Trans. 1, 1989, 165.
- 83. D. St. C. Black, H. Blatt, C. H. Bos Vanderzalm and A. J. Liepa, Aust. J. Chem., 1983, 36, 1133.
- 84. D. Ghiringhelli, Tetrahedron Lett., 1983, 24, 287.
- 85. D. Miller, L. Mandell and R. A. Day, Jr., J. Org. Chem., 1971, 36, 1683; J.-M. Pons and M. Santelli, Tetrahedron Lett., 1988, 29, 3679.
- 86. J.-M. Pons and M. Santelli, Tetrahedron Lett., 1988, 29, 3263.
- 87. K. W. Bowers, R. W. Giese, J. Grimshaw, H. O. House, N. H. Kolodny, K. Kronberger and D. K. Roe, J. Am. Chem. Soc., 1970, 92, 2783.
- L. A. Powell and R. M. Wightman, J. Am. Chem. Soc., 1979, 101, 4412. 88.
- 89. R. E. Sioda, B. Terem, J. H. P. Utley and B. C. L. Weedon, J. Chem. Soc., Perkin Trans. 1, 1976, 561.
- 90. J. Dunoguès, R. Calas, M. Bolourtchian, C. Biran and N. Duffaut, J. Organomet. Chem., 1973, 57, 55.
- 91. J.-M. Pons, J.-P. Zahra and M. Santelli, Tetrahedron Lett., 1981, 22, 3965.
- 92. J. Wiemann, M. R. Monot, G. Dana and J. Chuche, Bull. Soc. Chim. Fr., 1967, 3293.
- 93. G. Cahiez and M. Alami, Tetrahedron Lett., 1986, 27, 569.
- 94. L. Mandell, R. F. Daley and R. A. Day, Jr., J. Org. Chem., 1976, 41, 4087.
- 95. P. Margaretha and P. Tissot, Helv. Chim. Acta, 1982, 65, 1949.
- 96. For early examples of imine couplings see W. E. Bachmann, J. Am. Chem. Soc., 1931, 53, 2672, and refs. therein.
- 97. J. J. Eisch, D. D. Kaska and C. J. Peterson, J. Org. Chem., 1966, 31, 453.
- 98. J. G. Smith and I. Ho, J. Org. Chem., 1972, 37, 653; see also J. G. Smith and C. D. Veach, Can. J. Chem., 1966, 44, 2497.
- 99. K. N. Mehrotra and B. P. Giri, Synthesis, 1977, 489.
- 100. P. Mangeney, T. Tejero, A. Alexakis, F. Grosjean and J. F. Normant, Synthesis, 1988, 255; for application of the product in the resolution of aldehydes see P. Mangeney, A. Alexakis and J. F. Normant, Tetrahedron Lett., 1988, 29, 2677.
- 101. H. Tanaka, H. Dhimane, H. Fujita, Y. Ikemoto and S. Torii, Tetrahedron Lett., 1987, 29, 3811.
- 102. E. J. Roskamp and S. F. Pedersen, J. Am. Chem. Soc., 1987, 109, 3152.
- 103. R. Imwinkelried and D. Seebach, Helv. Chim. Acta, 1984, 67, 1496.
- 104. C. Betschart and D. Seebach, Helv. Chim. Acta, 1987, 70, 2215; C. Betschart, B. Schmidt and D. Seebach, Helv. Chim. Acta, 1988, 71, 1999.
- 105. L. Horner and D. H. Skaletz, Justus Liebigs Ann. Chem., 1975, 1210.
- 106. L. Horner and D. H. Skaletz, Tetrahedron Lett., 1970, 3679.
- 107. A. Padwa, W. Bergmark and D. Pashayan, J. Am. Chem. Soc., 1969, 91, 2653.
- 108. P. Beak and C. R. Payet, J. Org. Chem., 1970, 35, 3281.
- 109. R. Jaunin and R. Holl, Chimica, 1957, 11, 312; R. Jaunin and R. Holl, Helv. Chim. Acta, 1958, 41, 1783; R. Jaunin and J.-P. Magnenat, Helv. Chim. Acta, 1959, 42, 328.
- 110. R. W. Koch and R. E. Dessy, J. Org. Chem., 1982, 47, 4452.
- 111. For early reviews see (a) A. Schönberg and A. Wagner, Methoden Org. Chem. (Houben-Weyl), 1954, 9, 699;
 (b) E. Campaigne, Chem. Rev., 1946, 39, 50; (c) A. Schönberg, in 'Sammlung Chemischer and Chemischtechnischer Vortage', ed. F. Ahrens, Enke, Stuttgart, 1933, p. 1; E. Campaigne, in 'The Chemistry of the Crebergh' of S. Potter Willey Work, 1966, 504. of the Carbonyl Group', ed. S. Patai, Wiley, New York, 1966, p. 944.
- 112. L. Gatterman, Ber., 1895, 28, 2869; A. Schönberg, O. Schütz and S. Nickel, Ber., 1928, 61B, 2175.
- 113. J. K. Cline, E. Campaigne and J. W. Spies, J. Am. Chem. Soc., 1944, 66, 1136.
- 114. W. B. Ried, Ph D Thesis, Indiana University, 1946.
- 115. J. Baran and P. Laszlo, Tetrahedron Lett., 1985, 26, 5135.
- 116. H. Alper, S. Ripley and T. L. Prince, J. Org. Chem., 1983, 48, 250.
- 117. F. Amdt and P. Nachtwey, Ber., 1923, 56, 2406; F. Ardnt, P. Nactwey and J. Pusch, Ber., 1925, 58, 1644; A. Schönberg and W. Asker, Chem. Rev., 1945, 37, 1.
- 118. For reviews on the McMurry reaction see (a) P. C. Auderset, T. C. C. Gartenmann and E. R. F. Gesing, Kontakte (Darmstadt), 1985, 14 (Chem. Abstr., 1985, 104, 185 615m); (b) P. Welzel, Nachr. Chem. Tech. Lab., 1983, 31, 814 (Chem. Abstr., 1983, 99, 193 959v); (c) P. Finocchiaro, E. Libertini and A. Recca, Chim. Ind. (Milan), 1982, 64, 644 (Chem. Abstr., 1983, 98, 15 992x); (d) J. E. McMurry, Acc. Chem. Res., 1974, 7, 281; 1983, 16, 405.
- 119. K. B. Sharpless, M. A. Umbreit, M. T. Nieh and T. C. Flood, J. Am. Chem. Soc., 1972, 94, 6538; for an example of an alkyl ketone coupling using $W_2(OPr^i)_6(py)_2$ see T. P. Blatchford, M. H. Chisholm, K. Folting and J. C. Huffman, J. Chem. Soc., Chem. Commun., 1984, 1295.
- Y. Fujiwara, R. Ishikawa, F. Akiyama and S. Teranishi, J. Org. Chem., 1978, 43, 2477.
 M. Petit, A. Mortreux and F. Petit, J. Chem. Soc., Chem. Commun., 1984, 341.
- 122. R. Dams, M. Malinowski and H. J. Geise, Bull. Soc. Chim. Belg., 1982, 91, 149.
- 123. M. Sato and K. Oshima, Chem. Lett., 1982, 157.
- 124. J. E. McMurry and M. P. Fleming, J. Am. Chem. Soc., 1974, 96, 4708.
- 125. J. E. McMurry, M. P. Fleming, K. L. Kees and L. R. Krepski, J. Org. Chem., 1978, 43, 3255.

- 126. McMurry has developed a titanium-mediated keto ester coupling (cf. acyloin condensation see R. Brettle, in 'Comprehensive Organic Synthesis', ed. B. M. Trost, Pergamon Press, Oxford, 1991, vol. 3, chap. 2.7); J. E. McMurry and D. D. Miller, J. Am. Chem. Soc., 1983, 105, 1660.
- (a) J. E. McMurry and M. P. Fleming, J. Org. Chem., 1976, 41, 896; (b) see also S. Nishida and F. Kataoka, J. 127. Org. Chem., 1978, 43, 1612.
- 128. D. Lenoir, Synthesis, 1977, 553
- 129. A. Fürstner and H. Weidmann, Synthesis, 1987, 1071.
- 130. R. Dams, M. Malinowski, I. Westdrop and H. J. Geise, J. Org. Chem., 1982, 47, 248.
- 131. D. Lenoir and H. Burghard, J. Chem. Soc. (S), 1980, 396.
- 132. J. Leimner and P. Weyerstahl, Chem. Ber., 1982, 115, 3697.
- 133. L. Castedo, J. M. Saá, R. Suau and G. Tojo, J. Org. Chem., 1981, 46, 4292.
- 134. W. H. Richardson, Synth. Commun., 1981, 11, 895.
- 135. P. L. Coe and C. E. Scriven, J. Chem. Soc., Perkin Trans. 1, 1986, 475.
- A. Ishida and T. Mukaiyama, Chem. Lett., 1976, 1127.
 S. Akiyama, S.-i. Nakatsuji, S. Eda, M. Kataoka and M. Nakagawa, Tetrahedron Lett., 1979, 2813; for a related example on the coupling of citral see L. Barlow and G. Pattenden, J. Chem. Soc., Perkin Trans. 1, 1976, 1029.
- 138. T. Mandai, H. Yamaguchi, K. Nishikawa, M. Kawada and J. Otera, Tetrahedron Lett., 1981, 22, 763.
- 139. S. Hünig and B. Ort, Liebigs Ann. Chem., 1984, 1905.
- 140. J. Nakayama, S. Murabayashi and M. Hoshino, Heterocycles, 1986, 24, 2639.
- 141. D. S. Bomse and T. H. Morton, Tetrahedron Lett., 1975, 781; R. F. Langler and T. T. Tidwell, Tetrahedron Lett., 1975, 777; M. Bruch, Y. M. Jun, A. E. Luedtke, M. Schneider and J. W. Timberlake, J. Org. Chem., 1986, **51**, 2969.
- 142. For other examples of the synthesis of hindered alkenes see ref. 127a; P. D. Mollere, K. N. Houk, D. S. Bomse and T. H. Morton, J. Am. Chem. Soc., 1976, 98, 4732; G. A. Olah and G. K. S. Prakash, J. Org. Chem., 1977, 42, 580; D. Lenoir, Chem. Ber., 1978, 111, 411; D. Lenoir, D. Malwitz and B. Meyer, Tetrahedron Lett., 1984, 25, 2965; J. Janssen and W. Lüttke, Chem. Ber., 1982, 115, 1234.
- 143. G. Böhrer and R. Knorr, Tetrahedron Lett., 1984, 25, 3675; G. R. Newkome and J. M. Roper, J. Org. Chem., 1979, 44, 502; for related examples see S. Nishida and F. Kataoka, J. Org. Chem., 1978, 43, 1612; J. W. Timberlake and Y. M. Jun, J. Org. Chem., 1979, 44, 4729
- 144. M. P. Fleming and J. E. McMurry, Org. Synth., 1981, 60, 113.
- 145. G. A. Olah, G. K. S. Prakash and G. Liang, Synthesis, 1976, 318.
- 146. S. Madhava Reddy, M. Duraisamy and H. M. Walborsky, J. Org. Chem., 1986, 51, 2361.
- 147. H.-Fr. Grützmacher and W. Husemann, Tetrahedron Lett., 1985, 26, 2431.
- 148. For related examples see ref. 127a; R. Willem, H. Pepermans, K. Hallenga, M. Gielen, R. Dams and H. J. Geise, J. Org. Chem., 1983, 48, 1890; for the synthesis of sterically hindered alkylarylstilbenes see refs. 130, 131 and F. A. Bottino, P. Finocchiaro, E. Libertini, A. Reale and A. Recca, J. Chem. Soc., Perkin Trans. 2, 1982, 77; B. G. James, G. Pattenden and L. Barlow, J. Chem. Soc., Perkin Trans. 1, 1976, 1466; W. Adam, N. Carballeira, C.-C. Cheng, K. Sakanishi and R. Gleiter, J. Org. Chem., 1979, 44, 851.
- 149. D. Lenoir and P. Lemmen, Chem. Ber., 1980, 113, 3112; P. Lemmen and D. Lenoir, Chem. Ber., 1984, 117, 2300.
- 150. For a related example see H. Schwager and G. Wilke, *Chem. Ber.*, 1987, 120, 79.
 151. B. Feringa and H. Wynberg, *J. Am. Chem. Soc.*, 1977, 99, 602.
- 152. J. E. McMurry and L. R. Krepski, J. Org. Chem., 1976, 41, 3929.
- 153. M. J. Begley, G. Pattenden and G. M. Robertson, J. Chem. Soc., Perkin Trans. 1, 1988, 1085.
- 154. For hydrodimerization of this aldehyde see L. A. Paquette, G. J. Wells and G. Wickham, J. Org. Chem., 1984, **49**, 3618.
- 155. T.-H. Yan and L. A. Paquette, Tetrahedron Lett., 1982, 23, 3227.
- 156. J. A. Seijas, A. R. de Lera, M. C. Villaverde and L. Castedo, J. Chem. Soc., Chem. Commun., 1985, 839.
- 157. A. L. Baumstark, E. J. H. Bechara and M. J. Semigran, *Tetrahedron Lett.*, 1976, 3265. 158. A. L. Baumstark, C. J. McCloskey and K. E. Witt, *J. Org. Chem.*, 1978, **43**, 3609.
- 159. J. E. McMurry and K. L. Kees, J. Org. Chem., 1977, 42, 2655.
- 160. F. E. Zeigler and H. Lim, J. Org. Chem., 1982, 47, 5229
- 161. P. C. Anderson, D. L. J. Clive and C. F. Evans, Tetrahedron Lett., 1983, 24, 1373.
- 162. D. L. J. Clive, K. S. K. Murthy, A. G. H. Wee, J. S. Prasad, G. V. J. da Silva, M. Majewski, P. C. Anderson, R. D. Haugen and L. D. Heerze, J. Am. Chem. Soc., 1988, 110, 6914
- 163. B. W. Disanayaka and A. C. Weedon, J. Org. Chem., 1987, 52, 2905; for another example of transannular coupling of a 1,5-cyclooctanedione see P. J. Okarma and J. J. Caringi, Org. Prep. Proced. Int., 1985, 17, 212.
- 164. Y.-J. Wu and D. J. Burnell, Tetrahedron Lett., 1988, 29, 4369.
- 165. Synthesis of humulene and flexibilene see J. E. McMurry, J. R. Matz and K. L. Kees, Tetrahedron, 1987, 43, 5489.
- 166. Synthesis of bicyclogermacrene and lepidozene see J. E. McMurry and G. K. Bosch, Tetrahedron Lett., 1985, 26, 2167.
- 167. Synthesis of helminthogermacrene and β -elemene see J. E. McMurry and P. Kočovský, Tetrahedron Lett., 1985, 26, 2171.
- 168. Synthesis of bicyclogermacrene, lepidozene and casbene see J. E. McMurry and G. K. Bosch, J. Org. Chem., 1987, **52**, 4885.
- 169. C. B. Jackson and G. Pattenden, Tetrahedron Lett., 1985, 26, 3393.
- 170. J. Tirado-Rives, M. A. Oliver, F. R. Fronczek and R. D. Gandour, J. Org. Chem., 1984, 49, 1627.
- 171. I. Ben, L. Castedo, J. M. Saá, J. A. Seijas, R. Suau and G. Tojo, J. Org. Chem., 1985, 50, 2236.
- 172. J. A. Marshall, R. E. Bierenbaum and K.-H. Chung, Tetrahedron Lett., 1979, 2081; J. A. Marshall, T. H. Black and R. L. Shone, Tetrahedron Lett., 1979, 4737; J. A. Marshall, M. Constantino and T. H. Black, Synth. Commun., 1980, 10, 689; J. A. Marshall and T. H. Black, J. Am. Chem. Soc., 1980, 102, 7581.

- 173. M. Nakazaki, K. Yamamoto, M. Maeda, O. Sato and T. Tsutsui, J. Org. Chem., 1982, 47, 1435. 174. J. E. McMurry, G. J. Haley, J. R. Matz, J. C. Clardy and J. Mitchell, J. Am. Chem. Soc., 1986, 108, 515.
- 175. J. E. McMurry, G. J. Haley, J. R. Matz, J. C. Clardy, G. Van Duyne, R. Gleiter, W. Schäfer and D. H. White, J. Am. Chem. Soc., 1986, **108**, 2932.
- 176. J. E. McMurry and C. N. Hodge, J. Am. Chem. Soc., 1984, 106, 6450.
- 177. L. A. Paquette, J. M. Gardlik, K. J. McCullough and Y. Hanzawa, J. Am. Chem. Soc., 1983, 105, 7644.
- 178. G. Märkl and W. Burger, Angew. Chem., Int. Ed. Engl., 1984, 23, 894.
- 179. E. Vogel, M. Köcher, H. Schmickler and J. Lex, Angew. Chem., Int. Ed. Engl., 1986, 25, 257; E. Vogel, M. Sicken, P. Röhrig, H. Schmickler, J. Lex and O. Ermer, Angew. Chem., Int. Ed. Engl., 1988, 27, 411. 180. D. Tanner, O. Wennerström and E. Vogel, Tetrahedron Lett., 1982, 23, 1221. 181. K. Yamamoto, M. Shibutani, S. Kuroda, E. Ejiri and J. Ojima, Tetrahedron Lett., 1986, 27, 975.

- 182. K. Yamamoto, S. Kuroda, Y. Nozawa, S. Fujita and J. Ojima, J. Chem. Soc., Chem. Commun., 1987, 199.
- 183. D. Tanner and O. Wennerström, Tetrahedron Lett., 1981, 22, 2313.
- 184. D. Tanner, O. Wennerström, U. Norinder, K. Müllen and R. Trinks, Tetrahedron, 1986, 42, 4499.
- 185. A. Kasahara and T. Izumi, Chem. Lett., 1978, 21; see also A. Kasahara, T. Izumi and I. Shimizu, Chem. Lett., 1979, 1119.
- 186. H. Machida, J. Nakayama and M. Hoshino, Heterocycles, 1985, 23, 215; J. Nakayama, H. Machida, R. Saito, K. Akimoto and M. Hoshino, Chem. Lett., 1985, 1173.
- 187. A. K. Banerjee, M. C. Sulbaran de Carrasco, C. S. V. Frydrych-Houge and W. B. Motherwell, J. Chem. Soc., Chem. Commun., 1986, 1803.
- 188. For mixed aromatic ketone couplings see M. J. Allen, W. G. Pierson and J. A. Siragusa, J. Chem. Soc., 1961, 2081; for mixed aliphatic carbonyl couplings see ref. 53; for mixed intramolecular couplings see refs. 52, 53, 69, 71 and 73; for mixed α , β -unsaturated ketone couplings see ref. 92; for mixed McMurry-type couplings see refs. 125, 146, 152, 153, 155 and 156; for mixed intramolecular McMurry-type couplings see refs. 127a, 160-162, 165-168 and 176.
- 189. For coupling with saturated ketones see A. Clerici and O. Porta, J. Org. Chem., 1982, 47, 2852; A. Clerici and O. Porta, Synth. Commun., 1988, 18, 2281; A. Clerici, O. Porta and P. Zago, Tetrahedron, 1986, 42, 561; for coupling with α,β-unsaturated systems see A. Clerici and O. Porta, J. Org. Chem., 1983, 48, 1690. Z. Hou, K. Takamine, O. Aoki, H. Shiraishi, Y. Fujiwara and H. Taniguchi, J. Chem. Soc., Chem. Commun.,
- 190. 1988, 668; Z. Hou, K. Takamine, O. Aoki, H. Shiraishi, Y. Fujiwara and H. Taniguchi, J. Org. Chem., 1988, 53, 6077; see also ref. 36.
- 191. W. Reusch, K. Grimm, J. E. Karoglan, J. Martin, K. P. Subrahamanian, Y.-C. Toong, P. S. Venkataramani, J. D. Yordy and P. Zoutendam, J. Am. Chem. Soc., 1977, 99, 1953; J. R. Christensen and W. Reusch, J. Org. Chem., 1983, 48, 3741.
- 192. L. Mandell, H. Hamilton and R. A. Day, Jr., J. Org. Chem., 1980, 45, 1710.
- 193. Even with equimolar amounts of each ketone true statistical distribution is not observed, see B. P. Mundy, R. Srinivasa, Y. Kim, T. Dolph and R. J. Warnet, J. Org. Chem., 1982, 47, 1657.
- 194. Y. Li, Y. Li and W. Huang, K'o Hsueh T' ung Pao (Foreign Lang. Ed.), 1984, 29, 1131 (Chem. Abstr., 1985, 102, 132 319v); see also Y. Li, Y. Li, X. Pan and W. Haung, Hua Hsueh Shih Chieh, 1985, 7, 164 (Chem. Abstr., 1986, 104, 167 959x).
- 195. L. Horner and D. H. Skaletz, Justus Liebigs Ann. Chem., 1975, 1210.
- 196. E. J. Roskamp and S. F. Pedersen, J. Am. Chem. Soc., 1987, 109, 6551; for use in imine hydrodimerization see ref. 102.
- 197. E. H. Axelrod, J. Chem. Soc. D, 1970, 451.
- 198. M. B. Rubin, Tetrahedron Lett., 1982, 23, 4615; see also P. Yates, Pure Appl. Chem., 1968, 16, 93.
- 199. P. M. Collins, V. R. N. Munasinghe and N. N. Oparaeche, J. Chem. Soc., Perkin Trans. 1, 1977, 2423.
- 200. D. C. Owsley, J. J. Bloomfield and R. D. Kelman, J. Org. Chem., 1979, 44, 295.
- 201. T. Shono, H. Ohmizu and S. Kawakami, Tetrahedron Lett., 1979, 4091.
- 202. H. Ishikawa and T. Mukaiyama, Bull. Chem. Soc. Jpn., 1978, 51, 2059.
- 203. D. J. Hart, Science (Washington D.C.), 1984, 223, 883; M. Ramaiah, Tetrahedron, 1987, 43, 3541; B. Giese, in 'Radicals in Organic Synthesis: Formation of Carbon-Carbon Bonds', Pergamon Press, Oxford, 1986, p. 185.
- 204. T. Shono, I. Nishiguchi, H. Ohmizu and M. Mitani, J. Am. Chem. Soc., 1978, 100, 545.
- 205. E. Kariv-Miller and T. J. Mahachi, J. Org. Chem., 1986, 51, 1041.
- A. F. Sowinski and G. M. Whitesides, J. Org. Chem., 1979, 44, 2369. 206.
- 207. M. Bertrand, P. Teisseire and G. Pélerin, Nouv. J. Chim., 1983, 7, 61.
- 208. G. A. Molander and C. Kenny, Tetrahedron Lett., 1987, 28, 4367.
- 209. D. Belotti, J. Cossy, J. R. Pete and C. Portella, Tetrahedron Lett., 1985, 26, 4591.
- 210. J. Cossy and D. Belotti, Tetrahedron Lett., 1988, 29, 6113
- 211. T. Ikeda, S. Yue and C. R. Hutchinson, J. Org. Chem., 1985, 50, 5193.
- 212. A. L. J. Beckwith and D. H. Roberts, J. Am. Chem. Soc., 1986, 108, 5893.
- 213. M. Eakin, J. Martin and W. Parker, Chem. Commun., 1965, 206.
- 214. P. K. Jadhav and U. R. Nayak, Indian J. Chem. Sect. B, 1978, 16, 1047.
- 215. J. M. Greenwood, I. H. Qureshi and J. K. Sutherland, J. Chem. Soc., 1965, 3154.
- 216. R. A. Holton and A. D. Williams, J. Org. Chem., 1988, 53, 5981.
- 217. T. Shono and M. Mitani, J. Am. Chem. Soc., 1971, 93, 5284.
- 218. G. Stork, S. Malhotra, H. Thompson and M. Uchibayashi, J. Am. Chem. Soc., 1965, 87, 1148.
- 219. R. B. Miller, Synth. Commun., 1972, 2, 273.
- 220. G. Stork, R. K. Boeckmann, Jr., D. F. Taber, W. C. Still and J. Singh, J. Am. Chem. Soc., 1979, 101, 7107.
- 221. T. Shono, I. Nishiguchi and H. Omizu, Chem. Lett., 1976, 1233.
- 222. J. E. Swartz, T. J. Mahachi and E. Kariv-Miller, J. Am. Chem. Soc., 1988, 110, 3622.
- 223. J. Cossy, D. Belotti and J. P. Pete, Tetrahedron Lett., 1987, 28, 4545.
- 224. S. K. Pradhan, T. V. Radhakrishnan and R. Subramanian, J. Org. Chem., 1976, 41, 1943.

- 225. S. K. Pradhan, S. R. Kadam, J. N. Kolhe, T. V. Radhakrishnan, S. V. Sohani and V. B. Thaker, J. Org. Chem., 1981, 46, 2622. 226. M. E. Jung and G. L. Hatfield, Tetrahedron Lett., 1983, 24, 3175.

- 227. G. Pattenden and S. J. Teague, J. Chem. Soc., Perkin Trans. 1, 1988, 1077.
 228. J. Cossy, D. Belotti and J. P. Pete, Tetrahedron Lett., 1987, 28, 4547.
 229. T. L. Fevig, R. L. Elliott and D. P. Curran, J. Am. Chem. Soc., 1988, 110, 5064.
 230. G. Mehta and N. Krishnamurthy, Tetrahedron Lett., 1987, 28, 5945.
- 231. G. Pattenden and G. M. Robertson, Tetrahedron, 1985, 41, 4001.
- 232. J. K. Crandall and M. Mualla, Tetrahedron Lett., 1986, 27, 2243.
2.7 Acyloin Coupling Reactions

ROGER BRETTLE

University of Sheffield, UK

2.7.1	THE REDUCTION OF CARBOXYLIC ESTERS BY ALKALI METALS	613
2.7.2 2.7 2.7 2.7	NECESSARY REACTION CONDITIONS FOR THE ACYLOIN COUPLING REACTIONS OF ESTERS 2.1 Heterogeneous Conditions 2.2 Homogeneous Conditions 2.3 Reaction Variables	614 614 615 615
2.7.3 2.7 2.7 2.7 2.7	COMPETING BASE-CATALYZED REACTIONS AND THEIR SUPPRESSION. TRAPPING WITH TRIMETHYLSILYL CHLORIDE (3.1) Side Reactions (3.2) Trapping with Trimethylsilyl Chloride (3.3) Preferred Reaction Conditions for an Attempted Acyloin Coupling Reaction	615 615 616 617
2.7.4	ACYLOIN COUPLING REACTIONS WITH ACYL CHLORIDES AND ACID ANHYDRIDES	617
2.7.5	REDUCTIVE COUPLING OF THIOESTERS	618
2.7.6 2.7 2.7	APPLICATIONS TO SPECIFIC SYSTEMS 2.6.1 Acyclic Systems 2.7.6.1.1 Symmetrical condensations 2.7.6.1.2 Mixed condensations 2.7.6.2.1 Carbocyclic systems 2.7.6.2.2 Heterocyclic systems 2.7.6.2.2 Heterocyclic systems	618 618 620 620 620 629
2.7.7	COUPLING REACTIONS BETWEEN ESTERS AND KETONES	630
2.7.8	REFERENCES	631

2.7.1 THE REDUCTION OF CARBOXYLIC ESTERS BY ALKALI METALS

Under certain circumstances the reduction of a carboxylic ester by an alkali metal leads to an enediolate. Protonation then gives an acyloin, an α -hydroxy ketone, in which two atoms which were originally the carbon atoms of the ester carbonyl groups are joined by a new single carbon–carbon bond. This coupling reaction, in both its intermolecular mode (Scheme 1) and, particularly, in its intramolecular mode (Scheme 2), is a useful synthetic operation.¹ It is, however, only one of a number of processes which can occur as a result of the reduction of a carboxylic ester by an alkali metal. For example, whereas the use of molten sodium dispersed in refluxing toluene converts a simple ester into the acyloin, the use of an excess of sodium in ethanol converts a simple methyl or ethyl ester into the corresponding primary alcohol (*i.e.* the Bouveault–Blanc reduction). A different reduction occurs when, for example, hindered esters are reduced by lithium in ethylamine² or sodium in HMPA in the presence of 2-methylpropan-2-ol.³ Acyloin formation does not occur, and instead the alkoxy portion of the ester is deoxygenated (equation 1).

If the ester group is conjugated with a π -electron system, alternative reduction processes can occur. Benzoate esters are reduced to the 1,4-dihydrobenzoates by sodium in the system NH₃/THF/H₂O.⁴ For



 α , β -alkenic esters both simple reduction at the carbon–carbon double bond and hydrodimerization at the β -position have been observed,¹ as a result of the action of alkali metals under particular conditions.

A further reductive process is often observed in the case of hindered vicinal diesters, where fragmentation occurs;⁵ the cleaved products, still diesters, have sometimes then, in a separate step, been submitted to the acyloin coupling reaction. A typical sequence from the diester (1) is shown in Scheme $3.^6$



i, Na, toluene, Me₃SiCl, reflux; ii, MeOH

Scheme 3

In addition, many nonreductive side reactions occur. The choice of suitable reagents and conditions for an acyloin coupling reaction is still based on analogy and empiricism.

2.7.2 NECESSARY REACTION CONDITIONS FOR THE ACYLOIN COUPLING REACTIONS OF ESTERS

Two sets of conditions have commonly been employed in order to effect the acyloin coupling of esters. In each case it is important that both the condensation and the work-up should be performed in the absence of moisture and of oxygen.

2.7.2.1 Heterogeneous Conditions

The solvent is usually refluxing toluene or xylene, though occasionally benzene or ether has been used. The substitution of DME, THF or dioxane offers no advantages. At the higher temperatures obtaining in refluxing toluene (b.p. 111 °C) or xylene, the alkali metal is molten. The reaction is a surface reaction and it is essential that the metal is finely dispersed, for example by efficient stirring. A slight excess of the metal is normally used.

2.7.2.2 Homogeneous Conditions

The solvent is liquid ammonia, at its boiling point -33 °C. The exact amount of the metal is used. In particular cases one or other set of conditions may give the better yield. In investigating a new acyloin coupling reaction the first choice would be heterogeneous conditions.

2.7.2.3 Reaction Variables

The initial interaction between an alkali metal and an ester leads, by single electron transfer, to the alkali metal derivative of a radical anion. Radical anions from nonphenolic esters normally fragment, to give the carbonylate anion and a radical derived from the alkyl group. It has been pointed out² that this makes the acyloin coupling a curiosity and difficult to rationalize. The alkali metal is usually sodium. The choice has an effect on the course of the reaction, partly due to the potential at which the initial single electron transfer step occurs, and partly due to differences in the nature and stability of the alkali metal derivative of the radical anion under the prevailing conditions. No systematic study of this aspect of the reaction has ever been undertaken. At the higher temperatures of the heterogeneous conditions, the reaction of the radical anion must be a surface reaction. At the lower temperatures used in the homogeneous conditions the free radical anion is formed, but only undergoes the competing fragmentation slowly, so that the acyloin coupling is favored. The use of a sodium–potassium alloy (with its lower melting point relative to the pure metals) allows the heterogeneous conditions to be used at lower temperatures. This type of procedure has been found to be particularly successful for thiophene-containing systems (Scheme 4).⁷



i, Na/K, xylene, 55-60 °C; ii, H₃O⁺

Scheme 4

The products of an acyloin coupling reaction are an enediolate dianion and alkoxide ions. In the presence of the alkoxide the enediolate may be oxidized to the corresponding diketone by molecular oxygen.⁸ It is therefore important that, even when the coupling is complete, all reaction mixtures should be worked up in an oxygen-free atmosphere so long as the mixture remains alkaline. For reactions performed in liquid ammonia, all the ammonia should be allowed to evaporate before the reaction is worked up.

2.7.3 COMPETING BASE-CATALYZED REACTIONS AND THEIR SUPPRESSION. TRAPPING WITH TRIMETHYLSILYL CHLORIDE

2.7.3.1 Side Reactions

Low yields in acyloin coupling reactions in the past were often the consequence of competing basecatalyzed processes, although the exact nature of the by-products, especially those of high molecular mass, has often not been established. The conditions which lead to Claisen or Dieckmann condensations are quite similar to those used in the acyloin coupling reaction and sometimes such processes predominate. For example, an attempt to prepare the cyclic acyloin from either the (\pm) - or *meso*-form of diethyl 3,4-diphenylhexanedioate, using sodium in benzene, gave only the Dieckmann product, 2-ethoxycarbonyl-3,4-diphenylcyclopentan-1-one,⁹ and other similar cases have been collected.¹⁰ Other base-catalyzed reactions have intervened. An attempt to prepare 2-hydroxycyclopentan-1-one, from dimethyl pentane-1,5-dioate, gave, instead, a C₁₀ product derived by an aldol condensation,¹ and an attempt to effect the acyloin coupling reaction with the diester (2) led only to the products of base-catalyzed elimination (equation 2).¹ Problems like these can now be overcome by carrying out the coupling reaction in the presence of the trapping agent TMS-Cl or by adding it before the work-up.¹¹



2.7.3.2 Trapping with Trimethylsilyl Chloride

In the presence of TMS-Cl the enediolate dianion and, importantly, the alkoxide ions, are trapped as their neutral silyl ethers (Scheme 5). This leads to much improved yields of the coupled product; the acyloin is isolated in the form of its silyl enediol ether (3). Work-up is much easier. It is only necessary to filter the solution, evaporate the solvent, and isolate the product by distillation or chromatography. The TMS-Cl should be purified by distillation from calcium hydride, under a nitrogen or argon atmosphere, before use. A convenient procedure when using an organic solvent is to add the ester and the TMS-Cl together, dropwise, to the alkali metal finely dispersed in the solvent, at a rate sufficient to maintain the reaction. An explosion has been reported where this procedure was not followed.¹ For a reaction conducted in liquid ammonia¹ the TMS-Cl is added at the end of the reaction and after all the ammonia has been allowed to evaporate. Particularly in cases where sodium–potassium alloy has been used, a pyrophoric residue may have formed, so that the filtration must be carried out under an inert atmosphere.



The kinetic product of an acyloin coupling is the enediolate, which may be formed in (Z)- and (E)forms, and is accompanied by some of the thermodynamic product, the corresponding α -hydroxy enolate. Each of these dianions is trapped by the TMS-Cl. The products, and the proportions in which they are formed, for the reactions of ethyl ethanoate¹² and diethyl hexane-1,6-dioate¹³ are shown in Scheme 6. Use of dichlorodimethylsilane in conjunction with the reduction of ethyl ethanoate¹² gave the *acyclic*





products shown in Scheme 7. All the silvlated products can be converted into the corresponding acyloin by hydrolysis with dilute aqueous acid, but better yields are obtained by treatment with methanol which has been completely deoxygenated by the passage of a stream of nitrogen.^{5a,14} It is normally advantageous to prepare the acyloin as it is required, since most acyloins give dimers, through hemiacetal formation, on standing. Moreover several reactions, for example the oxidation to the α -diketone, can be performed on the silyl enediol ether without prior isolation of the acyloin.



Scheme 7

Although ethyl 3-methylbutanoate on reduction with sodium in the presence of TMS-Cl in benzene or ether gives the expected silvl enediol ether (4), in DME or THF the product is the noncoupled product (5).15



2.7.3.3 Preferred Reaction Conditions for an Attempted Acyloin Coupling Reaction

The preferred procedure at the present time would thus be to react an ester (or diester) in a heterogeneous system consisting of a highly dispersed alkali metal, probably sodium, in an organic solvent such as toluene, in the presence of trimethylsilyl chloride and under an inert atmosphere, and to decompose the isolated silvlated products to the required acyloin by treatment with oxygen-free methanol.

2.7.4 ACYLOIN COUPLING REACTIONS WITH ACYL CHLORIDES AND ACID ANHYDRIDES

In principle the acyloin coupling of acyl chlorides, in which the enediolate dianion is trapped by unreacted acyl chloride, might be an effective alternative to trapping with TMS-Cl, but the method has rarely been used. Good yields of the enediol diesters (6) were obtained with several fatty acid chlorides using sodium in ether, and the enediol diesters could easily be hydrolyzed to give the acyloins.¹⁶

The coupling of aroyl chlorides to give 1,2-diaroyloxy-1,2-diarylethylenes has been known since 1865. The best reagent, from many which have been tried, is activated copper powder (Scheme 8).¹⁷ It does not couple aliphatic acyl chlorides. The acyloin coupling of terephthaloyl chloride under either heterogeneous or homogeneous conditions, using sodium, has been shown to give a para-linked copolymer containing benzoin and benzil residues.¹⁸



Scheme 8

For the modern TMS-Cl trapping technique it has been shown that esters give considerably better yields than the corresponding acyl chlorides or acid anhydrides.^{11,19}

2.7.5 REDUCTIVE COUPLING OF THIOESTERS

Nicolaou has recently shown that reductive coupling of thioesters can be accomplished using sodium naphthalenide; the resulting dianion can then be trapped, for example with an alkyl halide (Scheme 9). One of the six examples cited by Nicolaou²⁰ is shown in Scheme 10, where the tricyclic bisthiolactone (7) is reduced to the tetracyclic compound (8). A related coupling can be achieved photochemically. Ultraviolet irradiation of (7) for a short period, gives the stable 1,2-dithietane, dithiatropazine (9), together with a little of the tetracyclic alkene (10). Longer irradiation of (7) or the thermal decomposition of (9) also leads to (10).²¹



i, Na⁺, C₁₀H₈, THF, -78 °C; ii, excess MeI, -78 to 0 °C; iii, UV light, toluene, room temperature, 15 min; iv, heat alone, or in xylene

Scheme 10

2.7.6 APPLICATIONS TO SPECIFIC SYSTEMS

2.7.6.1 Acyclic Systems

2.7.6.1.1 Symmetrical condensations

Early applications of the acyloin coupling reaction to the esters of aliphatic monocarbocyclic acids, before the introduction of the TMS-Cl trapping technique, gave variable yields and the separation of the pure acyloins from by-products was often troublesome. Nevertheless a satisfactory preparation of 5-hydroxyoctan-4-one in 65–70% yield, by the use of sodium dispersed in xylene, is described in Organic Syntheses.²² Yields in the 50–75% range were recorded²² for other acyloins prepared in the same way, including 4-hydroxy-2,2,5,5-tetramethylhexan-3-one (pivaloin). Even higher yields were recorded for the esters of fatty acids.¹ It was shown that a nonconjugated double bond and a nonconjugated, nonterminal triple bond in an aliphatic ester were unaffected by these simple coupling conditions.¹ Other successful acyloin coupling reactions¹ using the earlier procedures were those on methyl phenylacetate and ethyl 3-thienylpropionate, containing aromatic residues, and alicyclic esters such as methyl cyclohexanecarboxylate and ethyl adamantane-l-carboxylate.

The introduction of the trimethylsilyl chloride trapping technique¹¹ led to improved yields in the case of simple aliphatic esters. The initial silylated products are easily isolated and can be converted into the acyloins simply and in high yield. For simple aliphatic esters the yields are in the range 56–92%. Use of trimethylsilyl esters, rather than simple alkyl esters, leads to faster reactions, but lower yields.^{11,15} Substituted esters which have been successfully used in the newer procedure include ethyl 2-ethylhexanoate (83%),²³ ethyl trimethylsilylacetate (90%),¹¹ ethyl 3-trimethylsilylpropionate (65%),¹¹ ethyl phenylacetate (48%),¹¹ ethyl 3-phenylpropionate (79%))¹¹ and 2-(2-methoxycarbonylethyl)-2-methyl-1,3-dioxolane derived from levulinic acid (65%).¹¹ In the case of ethyl adamantane-1-carboxylate the yield using the newer procedure is reported to be inferior to that using the earlier procedure.



 $X = OEt (47\%); SEt (69\%); NEt_2 (62\%)$

Scheme 11

Because TMS-Cl scavenges basic products, the coupling reaction in its presence can be successfully conducted on esters containing a potential leaving group in the β -position, without any elimination occurring (Scheme 11).¹¹ However the reaction fails for β -halogen-substituted esters, due to reactions consequent on an initial reductive dehalogenation.¹¹ Reduction of either ethyl or trimethylsilyl benzoate with sodium only gives <15% of benzoin;¹ the trimethylsilyl enedicl ether is formed in ~40% yield if trapping with TMS-Cl is employed.¹¹ However, an 86% yield is reported for the corresponding trimethylsilyl enedicl ether when trimethylsilyl, but not ethyl, 4-methylbenzoate is used.¹¹ Reduction of ethyl benzoate with sodium in liquid ammonia gives a 50% yield of benzoin;¹ if water is added before the sodium, reduction of the benzene ring occurs (see Section 2.7.1).⁴ Reduction of ethyl benzoate with sodium naph-thalenide in THF gives benzoin in 86% yield,²⁴ but much lower yields of the acyloins are produced by the reduction of the esters (11) and (12) with lithium naphthalenide in THF.²⁵



Reaction of methyl or ethyl benzoate with the potassium–graphite intercalation compound C₈K, in DME or THF under argon at 25 °C, followed by quenching with water and air oxidation, gives a 30% yield of phenanthraquinone (equation 3).²⁶ The conversion of benzoin into phenanthraquinone by C₈K at the layer edge of the intercalate, with loss of molecular hydrogen, had been described earlier.²⁷ The overall process (equation 3) has been described by its discoverers as the hyperacyloin condensation. The conversion of methyl 4-alkylbenzoates into 3,6-dialkylphenanthraquinone has also been reported.²⁶



(3)

2.7.6.1.2 Mixed condensations

Very few intermolecular mixed acyloin coupling reactions using two different monocarboxylic esters have been attempted.¹ Inevitably, mixtures result, and the yields are low, especially when one component is a benzoate. No attempts to use the TMS-Cl trapping technique seem to have been made.

2.7.6.2 Cyclic Acyloins

2.7.6.2.1 Carbocyclic systems

(i) Small rings

A single example of the formation of a cyclopropane on reduction of a malonate has been reported (equation 4);¹ most of the products of the reduction were acyclic, and the cyclopropane (13) is not a derivative of an acyloin.



The intramolecular acyloin coupling reaction, on the other hand, is a useful method for constructing cyclobutane rings, although by no means all vicinal esters can be used successfully in this way. Diethyl succinate is not converted into 2-hydroxycyclobutanone (succinoin) by sodium and refluxing toluene alone, but using trapping with trimethylsilyl chloride a 78% yield of the corresponding trimethylsilyl enediol ether can be obtained.^{1,11} The procedure is described in detail in *Organic Syntheses*.¹⁴ Although dimethyl tetramethylsuccinate is not reduced by sodium in refluxing toluene or sodium in liquid ammonia, treatment with sodium–potassium alloy in xylene gives the acyloin in 35% yield. It is better, however, to carry out the reduction with sodium in toluene in the presence of TMS-Cl, and then hydrolyze the trapped product with aqueous acid, when the overall yield rises to 60%. Other simple substituted succinates have been reduced by this preferred procedure.¹ Dimethyl (+)-(*R*)-2-methylsuccinate and dimethyl (-)-(*S*)-2-methylsuccinate both give the corresponding chiral trimethylsilyl cyclobutenediol ethers,²⁸ and the (±)- and *meso*-forms of dimethyl dideuteriosuccinate give deuterium-labeled products of known stereochemistry (Scheme 12).²⁹ Diethyl (±)-2,3-di-*t*-butylsuccinate is reduced by sodium in toluene, in



Scheme 12

the absence of TMS-Cl, to give the *trans*-di-*t*-butylsuccinoin in 90% yield,¹ but the corresponding *meso* form does not react,¹ presumably because of the severe steric interaction between the two adjacent *t*-butyl groups which would be involved during the coupling step. The successful acyloin coupling of simple succinates using sodium in THF in the presence of TMS-Cl has also been reported.¹⁵

Other vicinal diesters which have been used successfully in the acyloin coupling reaction, using TMS-Cl trapping, are shown in Figure 1.^{1,30,31}



Figure 1 Vicinal diesters (in addition to those discussed in the text) which undergo the acyloin coupling reaction. Yields (in brackets), where quoted, and literature citations are given

The derivatives of dimethyl *trans*-cyclohexane-1,2-dicarboxylate (14–17) give trimethylsilyl enediol ethers which are thermally labile and undergo an electrocyclic ring opening to give a derivative of *cis*-*cis*-octa-1,3-diene.¹ The acyloin coupling can be accomplished using sodium-potassium alloy at room temperature; in refluxing toluene the monocyclic product is formed (Scheme 13).¹ The corresponding products from dimethyl *cis*-cyclohexanedicarboxylate are thermally stable at the boiling point of tol-

uene.¹ With the medium-sized ring systems (20; n = 11, 12 or 13) a mixture of the *cis*- and *trans*-diesters was used. Owing to the larger ring size both the *cis*- and *trans*-trimethylsilyl cyclobutadienol ethers underwent thermal ring opening, at 140 °C, and hydrolysis then gave the medium ring α -diketones (Scheme 14).¹



i, Na/K, Me₃SiCl, inert solvent, 20 °C; ii, Δ ; iii, Na, Me₃SiCl, toluene, reflux

Scheme 13



i, Na, xylene, Me₃SiCl; ii, 140 °C; iii, H₃O⁺

Scheme 14

The diester (18) gives the trimethylsilyl enediol ether, a propellane, in 88% yield, using the preferred TMS-Cl trapping procedure,¹ although sodium-potassium alloy in benzene gives a 76% yield of the acyloin directly.¹

The diester (19) with sodium and TMS-Cl in an inert solvent gives the trimethylsilyl enediol ether, another propellane, in only 24–40%, together with 58-40% of the silylated compound (21), formed by reductive cleavage of the central bond (19; Scheme 15).¹



i, Na, Me₃SiCl, inert solvent

A combination of the alternative pathways illustrated in Schemes 13 and 15 explains why the derivative of dimethyl *trans*-cyclohexane-1,2-dicarboxylate (22) fails to give any of the silylated coupled enediol; even at 25 °C, using sodium-potassium alloy in benzene, thermal rearrangement to an octa-1,3diene occurs, whereas use of sodium in liquid ammonia, at -78 °C, cleaves the bond joining the two functionalized carbon atoms, leading to dimethyl 2,7-dimethyloctane-1,8-dioate.¹



The vicinal diester (1) is one of a number of complex vicinal diesters investigated by Paquette and his coworkers, 6,32 which does not undergo the acyloin coupling reaction but rather undergoes bond cleavage as shown in Scheme 3. Other vicinal diesters which have been reported *not* to undergo the acyloin coupling reaction are shown in Figure 2.^{1,33–35}



Figure 2 Vicinal diesters (in addition to those discussed in the text) which fail to undergo the acyloin coupling reaction. References are given in brackets

The acyloin condensation has also been used to construct five-membered rings. Very high yields are obtained using trapping with TMS-Cl after carbon–carbon bond formation following reduction with sodium in toluene. The reaction is also reported to succeed using THF as the solvent.¹⁵ Diethyl glutarate, labeled with ¹³C at the 2- and 4-positions, gives the cyclic trimethylsilyl enediol ether in 94% yield.³⁶ Similarly 2-substituted glutarates, *e.g.* (23),³⁷ and 3,3-disubstituted glutarates^{1,38} give excellent yields of the corresponding cyclic enediol silyl ethers, including spiro ring systems formed from 1,1-bis(methoxy-carbonylmethyl)cycloalkenes^{1,11} and the tetraester (24).^{1,11} A further spiroannelation³⁹ is illustrated in equation (5). Acyloin coupling has also been applied to the elaboration of fused systems incorporating a



Figure 3 Diesters (in addition to those discussed in the text) which give six-membered carbocyclic acyloins on reduction. Yields (in brackets), where quoted, and literature citations are given

five-membered ring. Earlier work, not using TMS-Cl trapping, has been summarized¹ and includes several examples using seco-steroidal diesters. A high-yielding example is shown in equation (6). A few examples using the more modern procedures have also been reported,^{1,11} including the one shown in equation (7).

Two 1,3-diesters which are reported *not* to undergo acyloin coupling are the unsaturated ester $(25)^{37}$ and the triester (26).¹

Six-membered rings can be made by the acyloin coupling reaction. Dimethyl 2,2,5,5-tetramethylhexane-1,6-dioate gives the corresponding acyloin in 91% yield on reduction with sodium and xylene,¹ but for dimethyl hexane-1,6-dioate itself a comparable yield is only achieved using a TMS-Cl trapping technique.¹ Some other diesters which give six-membered acyloins are shown in Figure 3. In the case of the diesters (27), (28) and (29), TMS-Cl trapping was used and the yields refer to the trimethylsilyl enediol ethers. A key step in the synthesis of corannulene was the acyloin coupling reaction shown in equation (8).¹ It was necessary to avoid any excess of sodium and to operate at the temperature of an acetone/solid carbon dioxide bath, to prevent reduction of the ester function in the product. Some work has been reported on the reduction by alkali metals of ester lactones in the diterpene field. Complex mixtures of products result. In the example illustrated in Scheme 16 products which are the cyclic hemiacetals of sixmembered ring acyloins were isolated.¹ Two six-membered ring trimethylsilyl enediol ethers are produced in the acyloin coupling reaction of dimethyl 5-methylenenonane-1,9-dioate (30), as shown in equation (9). Their genesis must involve transannular bond formation after the initial coupling to give a nine-membered ring intermediate.⁴⁰



Attempts to bring about the acyloin coupling reaction with the diester (31) were unsuccessful.⁴¹ The failure can be attributed to the severe nonbonded interactions between the two ester groups in the cisoid conformation required for coupling to occur.



Coupling Reactions

Examples of the formation of seven-membered tilligs by the acyloin coupling reaction are collected in Figure 4. The reduction of the ester lactone (32) to give an acyloin hemiacetal (*cf.* Scheme 16) in very low yield was reported as part of a synthetic route to colchicitle.¹ In both these reductions of ester lactones, TMS-Cl trapping was not employed, and base-catalyzed β -elimination of the lactone oxygen would have been a competing process.



Figure 4 Diesters which give seven-membered carbocyclic acyloins on reduction. Yields (in brackets), where quoted, and literature citations are given

(ii) Medium rings

The yields in the Dieckmann and Thorpe-Ziegler cyclizations drop to almost zero for the production of nine- to twelve-membered rings. Fortunately the yields in the acyloin coupling reactions leading to the same ring sizes are quite good, making this procedure uniquely effective for the synthesis of medium-sized carbocyclic rings.



Scheme 17

The yields obtained in the preparation of medium-sized ring trimethylsilyl enediol ethers using TMS-Cl trapping and a high dilution strategy are shown in Table 1.⁴² With rapid addition of the diester considerable dimer formation occurs (Scheme 17). The acyloin coupling reaction was earlier used without TMS-Cl trapping to prepare many eight-, nine-, ten- and eleven-membered cyclic acyloins directly, with yields in the range 24–97%.¹ For example (+)-dimethyl 2-methylnonanoate was converted into the (–)form of the corresponding acyloin in 33% yield. Dimethyl decane-1,10-dioate behaves like a simple monocarboxylic ester in not undergoing the acyloin coupling reaction with sodium and TMS-Cl when

Acyloin Coupling Reactions

the solvent is THF.¹⁵ Medium-sized ring acyloins in which the ring is fused to a smaller ring have been prepared by applying the acyloin coupling reaction to the diesters (**33**) and (**34**). The product from (**33**) was obtained in 75% yield using sodium in xylene,¹ but the product from (**34**) was only obtained in a reasonable yield, of 36%, by using sodium in dioxane.¹ No coupling product could be obtained from the diastereomeric ester (**35**) under a variety of conditions.¹ In more complex systems a coupling reaction leading to an eight-membered ring⁶ is shown in Scheme 3, and the reduction of the diester (**36**) provides a further example.⁶ On the other hand the diester (**37**) failed to take part in the acyloin coupling reaction;⁴³ the failure was attributed to the severe nonbonded interactions which would arise as the two carbon atoms, which would have to be linked together, align themselves for the reaction. A polycyclic system incorporating eight- and nine-membered trimethylsilyl enediol ether rings is formed by the reaction shown in equation (10).¹ As part of a program leading to the synthesis of betweenanenes and triannulenes, Marshall has used the acyloin coupling reaction to prepare trimethylsilyl enediol ethers possessing two fused medium-sized rings with a *trans* double bond common to both rings (Scheme 18).⁴⁴

	Using I	high Dilution		
Substance	Yield (%)	Substance		Yield (%)
MeO2C(CH2)6CO2Me MeO2C(CH2)7CO2Me MeO2C(CH2)8CO2Me	72-85 68 58-69	MeO2C(CH2)9CO MeO2C(CH2)10CO	2Me 12Me	48 68
(33)	CO ₂ Me	CO_2Me CO_2Me (34)	CO_2Me CO_2Me (35)	
MeO ₂ C	CO ₂ Me	MeO ₂ C	CO ₂ Me	
EtO ₂ C —	$\sim - CO_2 Et$ $\sim - \frac{Na, tol}{Me_3 S}$ = 40%	uene iCl	OSiMe ₃ OSiMe ₃	(10)
$MeO_2C(CH_2)_n$	$(CH_2)_{10}$ $(CH_2)_n CO_2 Me$ or 4	(CH ₂)	$(CH_2)_{10}$ $(CH_2)_n$ $OSiMe_3$	\supset

Table 1 Yields of Trimethylsilyl Enediol Ethers in the Acyloin Coupling Reactions of α, ω -DiestersUsing High Dilution



Scheme 18

(iii) Large Rings

Even without the use of TMS-Cl trapping the acyloin coupling reaction was reported to give cyclic acyloins from dimethyl or diethyl esters of unsubstituted α,ω -diacids containing from 12 to 20 carbon atoms in yields of 64–96%.¹ Using TMS-Cl trapping, the cyclic trimethylsilyl enediol ethers are formed in similar yields;¹ for example, the cyclic trimethylsilyl enediol ethers containing 22 and 42 atoms in the ring are formed in *ca.* 90% yield, using sodium and toluene in the presence of TMS-Cl under argon.⁴⁵ An early route to paracyclophanes¹ completed the large ring by an acyloin coupling reaction using diesters containing two separated cyclohexane rings (equation 11); several of the yields were low, but moderate yields were obtained in other cases.¹ The presence of a (Z)-alkenic or an alkyne linkage near the center of a long chain α,ω -diester allows cyclic acyloins containing double or triple bonds to be constructed in good yield.¹ Similarly, a centrally placed ketone function, protected as a cyclic ethylene ketal, allows the synthesis of a large ring ketoacyloin.¹ The acyloin coupling reaction has also been performed with long chain α,ω -diesters with one or more alkyl branches.¹ An example of the synthesis of a 14-membered acyloin ring fused to a thiophene ring is illustrated in Scheme 4.



The acyloin coupling reaction, at high dilution, of the *cis* and *trans* forms of 1,9-bis(3-methoxycarbonylpropyl)cyclooctadecane was the key step in the synthesis of homeomorphic isomers of bicyclo[8.8.8]hexacosane.¹ The reaction of the *cis* isomer to give an acyloin, which was converted into the *in*, *in*-isomer of the hydrocarbon, is shown in equation (12).



The first claim to have prepared a catenane was based on the use of the acyloin coupling reaction to prepare both the interlocking rings.⁴⁶ A pentadeuteriotetratriacontane was first prepared by chemical modification of the large ring acyloin obtained from diethyl tetratriacontane-1,34-dioate. A second acyloin coupling reaction in the presence of the deuterated hydrocarbon using diethyl tetratriacontane-1,34-dioate for the second time was then undertaken in the hope that some of the newly formed ring would interlock with the first ring.⁴⁶

Other large ring acyloins have been prepared in connection with the preparation of rotaxanes, the acyloins forming the 'wheel' on the 'axle'. Using a slow addition technique dimethyl octacosane-1,28-dioate gave the acyloin in 78% yield.⁴⁷ Lower yields were earlier reported for the cyclization of diethyl pentacosane-1,25-dioate and diethyl hexacosane-1,26-dioate in the presence of molecules which might have led to rotaxanes, although no rotaxanes were detected.¹ The preparations of 30- and 32-membered cyclic acyloins, and of a mixture containing *every* cyclic acyloin with 14–42 carbon atoms in the ring, have also been reported.¹

(iv) Paracyclophanes

Early work on the use of the acyloin coupling reaction in the synthesis of paracyclophanes has been summarized by Smith.⁴⁸ The intramolecular coupling of diesters of the type (38),¹ (39)¹ and (40),⁴⁹ using sodium and xylene under conditions of high dilution, is generally successful, but some failures, *e.g.* with (38) and (39); *n* and/or m = 1, have been reported, largely attributable, in the absence of TMS-Cl trapping, to base-catalyzed side reactions. The only example of the use of TMS-Cl trapping seems to be the one shown in Scheme 19.⁵⁰



i, Na, toluene, Me₃SiCl; ii, H₃O⁺

Scheme 19

2.7.6.2.2 Heterocyclic systems

The acyloin coupling reaction has been used to prepare cyclic acyloins with ring sizes from six to 23, in which one of the ring atoms is a nitrogen atom bearing an alkyl, cycloalkyl, or aryl substituent, as shown in equation (13). The use of TMS-Cl trapping may be advisable for the synthesis of azacyloheptane derivatives,^{1,51} but in most cases good yields have been reported without the use of TMS-Cl, particularly with diesters possessing no hydrogen atoms α to the ester groups. The yield of the acyloin from the diester (41; R = Et) was only 10%, but its tetramethyl derivative (42) gave the acyloin in 83% yield. However, using TMS-Cl, trapping the diester (41; R = Me) gave the trimethylsilyl enediol ether in 74% yield. Similarly the lactone ester (43) gave yields of the acyloin (44) of 35–62%, but with TMS-Cl trapping, followed by hydrolysis, the yield was improved to 68%.⁵² The diester (45) underwent the acyloin coupling reaction in the presence of highly dispersed sodium in xylene in 56% yield, and the product was converted into a 3-aza[10]paracyclophane.⁵³ The related diesters (46) and (47; n = 1 or 2) all failed to give an acyloin, even in the presence of TMS-Cl.⁵⁴



The acyloin coupling reaction has also been used to prepare cyclic acyloins in which one of the ring atoms is an oxygen, sulfur or silicon atom.¹ For seven- or eight-membered rings the use of TMS-Cl trapping may give improved yields.¹ An example is shown in Scheme 20.¹



Various acyloins which are 2,5-bridged thiophenophane derivatives have been prepared by the intramolecular coupling of the appropriate diesters; a typical example is given in Scheme 4. Attempts to effect the intramolecular acyloin coupling reaction with the benzoate-type diester (48) were unsuccessful under several standard conditions.⁵⁵



2.7.7 COUPLING REACTIONS BETWEEN ESTERS AND KETONES

Under conditions which have been used for the acyloin coupling reaction without TMS-Cl trapping, certain δ -keto esters, instead of undergoing intramolecular ester coupling to give diketoacyloins, give products derived by intramolecular coupling of the two carbonyl carbon atoms, and a five-membered carbocyclic ring is formed. In the few cases studied a mixture containing a range of functional groups was formed on reduction with sodium in ether–liquid ammonia or sodium naphthalenide in THF, and the method was not an attractive one.¹

More recently a much superior and more generally applicable method for the intramolecular coupling of keto esters has been reported by McMurry.⁵⁶ The procedure leads to an enol ether, which can be isolated, but is normally hydrolyzed to a ketone (Scheme 21). McMurray used a heterogeneous coupling reagent formed by the action of lithium aluminum hydride on titanium(III) chloride in the presence of triethylamine in DME under argon. Recently titanium on graphite, produced by the action of the potassium–graphite intercalate on titanium(III) chloride, has been shown to give a higher yield in at least one case; no tertiary amine is necessary with this reagent, which was used in THF under argon employing a high dilution technique.⁵⁷ Table 2 lists the known examples of these coupling procedures; ring sizes of from four to 14 atoms have been prepared in this way. In the case of the keto ester (51), containing an



i, TiCl₃, LiAlH₄, Et₃N; ii, H₃O⁺

Scheme 21

(E)-double bond, the product is the cyclic ketone (52) containing a (Z)-double bond.⁵⁸ The isomerization of the double bond is thought to be connected uniquely with the highly strained fused system which is being constructed, and the phenomenon is unlikely to be a general one.

 Table 2
 Yields of Ketones from the Intramolecular Coupling of Keto Esters by the McMurry Method

Compound	R	R ¹	n	Ring size formed	Isolated yield (%)	Ref.
(50) (50) (50) (50) (50) (50) (51) (49) (49) (49) (49) (50)	Bu ^t H H Me H H H Et Et Et Me H	Et Et Et Et Et Et Et Me Me Et	1 2 3 3 4 4 5 6 8 9 10 11 11	4 5 6 6 7 8 9 9 10 11 12 13 14	57 75 91 80 77 82 52 50 38 56 45 63 60 54	56 56 57 56 57 56 56 56 56 56 56 56 56
R		– O ₂ R ⁱ		CO ₂ Et		
	(50)		(51)		(52)	

2.7.8 REFERENCES

- 1. K. T. Finley, Chem. Rev., 1964, 64, 573; J. J. Bloomfield, D. C. Owsley and J. M. Nelke, Org. React. (N.Y.), 1976, 23, 259.
- 2. R. B. Boar, L. Joukhadar, J. F. McGhie, S. C. Misra, A. G. M. Barrett, D. H. R. Barton and P. A. Prokopiou, J. Chem. Soc., Chem. Commun., 1978, 68; A. G. M. Barrett, P. A. Prokopiou, D. H. R. Barton, R. B. Boar and J. F. McGhie, J. Chem. Soc., Chem. Commun., 1979, 1173; A. G. M. Barrett, C. R. A. Godfrey, D. M. Hollinshead, P. A. Prokopiou, D. H. R. Barton, R. B. Boar, L. Joukhadar, J. F. McGhie and S. C. Misra, J. Chem. Soc., Perkin Trans. 1, 1981, 1501.
- H. Deshayes and J. P. Pete, J. Chem. Soc., Chem. Commun., 1978, 567.
 P. W. Rabideau, D. M. Wetzel and D. M. Young, J. Org. Chem., 1984, 49, 1544.
- 5. (a) J. J. Bloomfield, R. A. Martin and J. M. Nelke, J. Chem. Soc., Chem. Commun., 1972, 96; (b) P. G. Gassman and X. Creary, J. Chem. Soc., Chem. Commun., 1972, 1214.
- 6. R. Bartetzko, R. Gleiter, J. L. Muthard and L. A. Paquette, J. Am. Chem. Soc., 1978, 100, 5589.
- Ya. L. Gol'dfarb, S. Z. Taits and L. I. Belen'kii, Tetrahedron, 1963, 19, 1851.
- 8. M. Stoll and J. Hulstkamp, Helv. Chim. Acta, 1947, 30, 1815.
- 9. H. A. Weidlich, Chem. Ber., 1938, 71, 1601.
- 10. J. P. Schaefer and J. J. Bloomfield, Org. React. (N.Y.), 1967, 15, 1.
- 11. K. Rühlmann, Synthesis, 1971, 236.
- 12. C. M. Cookson and G. H. Whitham, J. Chem. Soc., Perkin Trans. 1, 1975, 806.

- 13. S. R. Wilson, M. E. Walters and B. Orbaugh, J. Org. Chem., 1976, 41, 378.
- 14. J. J. Bloomfield and J. M. Nelke, Org. Synth., 1977, 57, 1.
- 15. I. Kuwajima, N. Minami, T. Abe and T. Sato, Bull. Chem. Soc. Jpn., 1978, 51, 2391.
- 16. A. W. Ralston and W. M. Selby, J. Am. Chem. Soc., 1939, 61, 1019.
- 17. T.-C. Wu and R. D. Rieke, J. Org. Chem., 1988, 53, 2381.
- C. Nallaiah, J. Polym. Sci., Polym. Chem. Ed., 1984, 22, 3107.
 F. M. F. Chen, Diss. Abstr. Int. B., 1970, 31, 1150 (Chem. Abstr., 1971, 75, 63 229).
- 20. K. C. Nicolaou, C.-K. Hwang, M. E. Duggan, K. Bal. Reddy, B. E. Marron and D. G. McGarry, J. Am. Chem. Soc., 1986, 108, 6800.
- K. C. Nicolaou, C.-K. Hwang, M. E. Duggan and P. J. Carroll, J. Am. Chem. Soc., 1987, 109, 3801.
- 22. J. M. Snell and S. M. McElvain, Org. Synth., Coll. Vol., 1943, 2, 114.
- 23. A. Olszanowski and K. Alejski, Przem. Chem., 1981, 60, 537 (Chem. Abstr., 1982, 97, 38 456).
- 24. M. Vora and N. Holy, J. Org. Chem., 1975, 40, 3144. 25. R. G. H. Kirrstetter and U. Vagt, Chem. Ber., 1981, 114, 630.
- D. Tamarkin and M. Rabinovitz, J. Org. Chem., 1987, 52, 3472. 26.
- 27. D. Tamarkin, D. Benny and M. Rabinovitz, Angew. Chem., Int. Ed. Engl., 1984, 23, 642.
- 28. J. R. Salaün and B. Karkour, Tetrahedron Lett., 1987, 28, 4669.
- R. E. K. Winter and M. L. Honig, J. Am. Chem. Soc., 1971, 93, 4616.
 R. D. Miller, D. L. Dolce and V. Y. Merritt, J. Org. Chem., 1976, 41, 1221.
- 31. G. McMullen, G. Sedelmeier, R. Hildebrand, H. Fritz and H. Prinzbach, Tetrahedron Lett., 1979, 3847. 32. (a) L. A. Paquette, M. J. Wyvratt, O. Schallner, D. F. Schneider, W. J. Begley and R. M. Blankenship, J. Am. Chem. Soc., 1976, 98, 6744; L. A. Paquette, M. J. Wyvratt, O. Schallner, J. L. Muthard, W. J. Begley, R. M. Blankenship and D. W. Balogh, J. Org. Chem., 1979, 44, 3616; (b) D. W. Balogh, L. A. Paquette, P. Engel and J. F. Blount, J. Am. Chem. Soc., 1981, 103, 226.
- 33. D. McNeil, B. R. Vogt, J. J. Sudol, S. Theodoropoulos and E. Hedaya, J. Am. Chem. Soc., 1974, 96, 4673; L. A. Paquette, M. J. Wyvratt, H. C. Berk and R. E. Moerck, J. Am. Chem. Soc., 1978, 100, 5845.
- 34. E. Nakamura and I. Kuwajima, J. Am. Chem. Soc., 1977, 99, 961.
- V. Batroff and W. Flitsch, *Liebigs Ann. Chem.*, 1987, 621.
 H. Thies, H. Halim and H. Schwarz, *Chem. Ber.*, 1983, 116, 2015.
- 37. R. C. Cookson and S. A. Smith, J. Chem. Soc., Perkin Trans. 1, 1979, 2447.
- 38. P. Eilbracht and P. Dahler, Liebigs Ann. Chem., 1979, 1890.
- 39. T. Ibuka, K. Hayashi, M. Minakata, Y. Ito and Y. Inubushi, Can. J. Chem., 1979, 57, 1579; T. Ibuka, T. Aoyagi and F. Yoneda, J. Chem. Soc., Chem. Commun., 1985, 1452; T. Ibuka, T. Aoyagi and Y. Yamamoto, Chem. Pharm. Bull., 1986, 34, 2417.
- 40. P. Y. Johnson and M. A. Priest, J. Am. Chem. Soc., 1974, 96, 5618.
- 41. R. Römer, J. Harnisch, A. Röder, A. Schöffer, G. Szeimies, G. Germain and J. M. Arrieta, Chem. Ber., 1984, 117, 925.
- 42. J. J. Bloomfield, D. C. Owsley, C. Ainsworth and R. E. Robertson, J. Org. Chem., 1975, 40, 393.
- 43. L. A. Paquette, J. W. Fischer, A. R. Browne and C. W. Doecke, J. Am. Chem. Soc., 1985, 107, 686.
- 44. J. A. Marshall and M. Lewellyn, J. Am. Chem. Soc., 1977, 99, 3508; J. A. Marshall, J. C. Peterson and L. Lebioda, J. Am. Chem. Soc., 1983, 105, 6515; J. A. Marshall, J. C. Peterson and L. Lebioda, J. Am. Chem. Soc., 1984, 106, 6006.
- 45. P. Ashkenazi, J. Kettenring, S. Migdal, A. L. Gutman and D. Ginsburg, Helv. Chim. Acta, 1985, 68, 2033.
- 46. E. Wasserman, J. Am. Chem. Soc., 1960, 82, 4433; E. Wasserman, Sci. Am., 1962, 207 (5), 94.
- 47. G. Schill, W. Beckmann and W. Vetter, Chem. Ber., 1980, 113, 941.
- 48. B. H. Smith, 'Bridged Aromatic Compounds', Academic Press, New York, 1964, p. 27.
- 49. M. Nakazaki, K. Yamamoto, S. Isoe and M. Kobayashi, J. Org. Chem., 1979, 44, 2160.
- 50. M. B. Rubin and S. Welner, J. Org. Chem., 1980, 45, 1847.
- 51. P. Y. Johnson, I. Jacobs and D. J. Kerkman, J. Org. Chem., 1975, 40, 2710; N. Finch, L. Blanchard and L. H. Werner, J. Org. Chem., 1977, 42, 3933.
- 52. P. Y. Johnson and D. J. Kerkman, J. Org. Chem., 1976, 41, 1768.
- 53. G.-S. Wu, L. C. Martinelli, C. D. Blanton, Jr. and R. H. Cox, J. Heterocycl. Chem., 1977, 14, 11.
- 54. C. S. Yi, L. C. Martinelli, C. D. Blanton, Jr. and K. H. Co., J. Helefolytt. Ch.
 55. R. M. Acheson and G. C. M. Lee, J. Chem. Res. (S), 1986, 380.
 56. J. E. McMurry and D. D. Miller, J. Am. Chem. Soc., 1983, 105, 1660.
 57. A. Fürstener, P. Contennation of the content of the conten

- 57. A. Fürstner, R. Csuk, C. Rohrer and H. Weidmann, J. Chem. Soc., Perkin Trans. 1, 1988, 1729.
- 58. J. E. McMurry and D. D. Miller, Tetrahedron Lett., 1983, 24, 1885.

2.8 Kolbe Reactions

HANS J. SCHÄFER

Westfälische Wilhelms Universität, Münster, FRG

2.8.1	INTRODUCTION AND HISTORICAL BACKGROUND	633
2.8.2	REACTION CONDITIONS FOR THE KOLBE ELECTROLYSIS	634
2.8.3	MECHANISM OF THE KOLBE ELECTROLYSIS	636
2.8.4	SYMMETRICAL COUPLING REACTIONS OF CARBOXYLIC ACIDS	638
2.8	8.4.1 Experimental Procedure for Symmetrical Kolbe Electrolysis — Preparation of Dimethyl Hexadecanedioate	642
2.8.5	CROSS-COUPLING REACTIONS OF DIFFERENT CARBOXYLIC ACIDS	642
2.8	3.5.1 Experimental Procedure for Mixed Kolbe Electrolysis — Preparation of 11-Hexadecynoic Acid	646
2.8.6	ADDITION OF KOLBE RADICALS TO DOUBLE BONDS	646
2.8.7	THE NONKOLBE ELECTROLYSIS	649
2.8	8.7.1 Experimental Procedure for NonKolbe Electrolysis — Preparation of Methyl 2-(t-2-Phenylsulfonyl-3-cyclopentent-1-yl) acetate	654
2.8.8	CONCLUSIONS	654
2.8.9	REFERENCES	654

2.8.1 INTRODUCTION AND HISTORICAL BACKGROUND

By anodic decarboxylation carboxylic acids can be converted simply and in large variety into radicals. The combination of these radicals to form symmetrical dimers or unsymmetrical coupling products is termed Kolbe electrolysis (Scheme 1, path a). The radicals can also be added to double bonds to afford additive monomers or dimers, and in an intramolecular version can lead to five-membered heterocycles and carbocycles (Scheme 1, path b). The intermediate radical can be further oxidized to a carbenium ion (Scheme 1, path c). This oxidation is favored by electron-donating substituents at the α -carbon of the carboxylic acid, a basic electrolyte, graphite as anode material and salt additives, *e.g.* sodium perchlorate. The carbocations lead to products that are formed by solvolysis, elimination, fragmentation or rearrangement. This pathway of anodic decarboxylation is frequently called nonKolbe electrolysis.

Faraday probably experienced in 1834 the first example of a radical dimerization by anodic decarboxylation, when he noticed an inflammable gas whilst studying the conductivity of acetate.¹ However, it took another 15 years until Kolbe in 1849 recognized the nature and dimensions of this conversion.² This way the name of the reaction was coined. In 1855 Wurtz obtained unsymmetrical coupling products, when he coelectrolyzed two different carboxylates.³ Crum-Brown and Walker later opened a simple entry to α, ω -dicarboxylic esters, useful products for the synthesis of 1,*n*-difunctionalized alkanes or medium- and large-sized rings, when they electrolyzed half esters of 1,*n*-diacids.⁴ In the 1950s and 1960s Weedon *et al.* extensively applied the Kolbe electrolysis to the synthesis of rare fatty acids and to acyclic natural products.⁵



Scheme 1

Major contributions to the mechanism of the Kolbe electrolysis have been made by Conway *et al.*,⁶ who mainly used electroanalytical techniques, and by Eberson⁷ and Utley,⁸ who drew their conclusions from the dependence of product structures on reaction conditions.

A large number of reactions have been published in the last 140 years on the Kolbe and nonKolbe electrolysis. Many of them have been compiled in reviews^{5,6,9–15} and books.^{8,16–23}

2.8.2 REACTION CONDITIONS FOR THE KOLBE ELECTROLYSIS

The pathway leading to radical products (Kolbe electrolysis) and/or cationic intermediates (nonKolbe electrolysis) is determined mainly by the substituents in the α -position of the carboxylic acid and some experimental factors.

The electrolysis products of different carboxylates have been compared with the ionization potentials of the intermediate radicals.^{7c} From this it appeared that alkyl radicals with gas-phase ionization potentials smaller than 8 eV mainly lead to carbenium ions. Accordingly, α -substituents such as carboxy, cyano or hydrogen support the radical pathway, whilst alkyl, cycloalkyl, chloro, bromo, amino, alkoxy, hydroxy, acyloxy or aryl more or less favor the route to carbenium ions. Besides electronic effects, the oxidation seems also to be influenced by steric factors.²⁴ Bulky substituents diminish the extent of coupling. The main experimental factors that affect the yield in the Kolbe electrolysis are the current density, the pH of the electrolyte, ionic additives, the solvent and the anode material.

High current densities, together with high carboxylate concentrations favor the formation of dimers. This results from a high radical concentration at the electrode surface that supports the dimerization. Furthermore, at higher current densities a critical potential of 2.4 V (*versus* the normal hydrogen electrode) is reached, 25a which seems to be a prerequisite for the Kolbe electrolyis, because at and above this potential the oxygen evolution and solvent oxidation is effectively suppressed. $^{6.26,27}$

A weakly acidic medium favors the Kolbe electrolysis. Therefore the carboxylic acid is in most cases neutralized to an extent of only 2 to 5%. In an undivided cell the concentration of the carboxylate remains constant until near the end of the electrolysis, because when the carboxylate is converted at the anode it is continously regenerated from the carboxylic acid by the base formed at the cathode. The endpoint of the electrolysis is recognizable by a steep rise of the pH in the electrolyte. This procedure is called the salt-deficit method. The degree of neutralization can be as low as 0.5%. Low cell voltages, in spite of low conductivity, are obtained in these cases by using cells with a very small electrode distance (0.1 to 2 mm),^{25b-28} as shown in the flow-through cell in Figure 1. Sometimes the carboxylic acid must be converted totally into the carboxylate to increase its solubility in aqueous solution or to achieve a uniform discharge in mixed Kolbe electrolysis, when the acidities of the two acids are too different. A mercury cathode is then applied, and the electrolyte stays neutral, because the alkali metal cations are discharged as amalgams.²⁹ In such cases a divided cell with a permselective ion-exchange membrane may also be suitable.

Temperature has only a minor effect on the Kolbe electrolysis. In the Kolbe coupling, temperatures higher than 50 °C should be avoided because they favor unwanted side reactions such as disproportionation or the esterification of the carboxylic acid.³⁰

Ionic additives to the electrolyte can influence the Kolbe electrolysis in a negative way. Anions other than the carboxylate should be excluded, because they hinder the formation of a carboxylate layer at the anode, that seems to be a prerequisite for the decarboxylation. In the electrolysis of phenyl acetate the coupling to dibenzyl is totally suppressed when sodium perchlorate is present in concentrations of 5×10^{-3} mol l⁻¹; benzyl methyl ether, the nonKolbe product, is formed instead.²⁴ This shift from the radical



Figure 1 (i) Flow-through cell and accessories: (a) cell; (b) electrical connection to regulated d.c. power supply;
(c) thermometer; (d) heat exchanger; (e) vessel for gas outlet, sample withdrawal and temperature control; (f) circulation pump. (ii) Exploded view of the flow-through cell: (a) brass lid (10 mm); (b) platinum anode (0.01 × 30 × 60 mm); (c) silicon seal (0.5 mm); (d) electrolyte in/out; (e) V4A-steel base (cathode); (f) V4-A steel base (side view); (g,h) electrical connections

to the carbenium ion pathway can be rationalized by a partial obstruction of the anode surface by coadsorption of perchlorate. That way the radical concentration at the electrode is lowered, which disfavors the bimolecular dimerization and supports an electron transfer from the radical to the electrode to form a carbenium ion.

Whilst cations such as Fe^{2+} , Co^{2+} , Mn^{2+} and Pb^{2+} lower the yield in the Kolbe electrolysis,¹¹ alkali and alkylammonium ions have no effect³¹ and are therefore preferably used as counterions for the carboxy-late.

Methanol is the solvent of choice for the Kolbe electrolysis.^{5,32} The following electrolytes with methanol as solvent have been used: methanol/sodium carboxylate, methanol/sodium methoxide/carboxylic acid,^{33,34} methanol/water/sodium hydroxide/carboxylic acid,³⁵ methanol/triethylamine/pyridine/carboxylic acid.³⁶ An increasing amount of water often decreases the yield of dimer.

Dimethylformamide is also a useful solvent;³⁷ its relatively low oxidation potential,³⁸ however, prevents its frequent application. The ternary system water/methanol/dimethylformamide has been systematically studied, when the electrolyses of ω -acetoxy- or ω -acetamido-carboxylic acids were optimized.^{25b} Wet acetonitrile has also been used as solvent.³⁹ The influence of various solvents on the ratio of Kolbe

Solvent	Bicyclohexyl:carbenium ion product ^b
H ₂ O/MeOH (30% v/v) ^c	0.97:1
MeOH ^c	1.76:1
MeCN ^d	2.26:1
HCONMe ₂ ^d	4.32:1

Table 1	Anodic Decarbox	vlation ^a of C	vclohexanecarbox	ylic Acid in	Different Solvents
---------	-----------------	---------------------------	------------------	--------------	--------------------

^aAcid (0.78 M), neutralized to 25%; platinum anode, 0.25 A cm⁻². ^bAlcohol, ester, ether and acetamide according to solvent. ^cSodium salt. ^dTetrabutylammonium salt.

to nonKolbe products is shown in Table 1.40

Other electrolytes that have been applied, are dimethyl sulfoxide/sodium hydride,⁴¹ glycol methyl ether/water/potassium carbonate⁴² or acetic acid/potassium carboxylate.¹¹

As anode material, smooth platinum, as foil or net, is most widely used.^{25b,26} In nonaqueous solvents, additionally, platinized titanium, gold, hard graphite⁴³ and ruthenium dioxide on titanium⁴⁴ have been employed. To keep the amount of the costly platinum low, the foils have been glued to a graphite support.²⁷ With vitreous or baked carbon in protic solvents (methanol, ethanol, water) dimer yields nearly comparable to those at platinum have been obtained.³⁶

Coupling Reactions

The nature of the cathode material is uncritical in the Kolbe electrolysis. The discharge of protons from the carboxylic acid is in most cases the only cathode reaction, which allows the convenient electrolysis in an undivided cell. For the electrolysis of unsaturated carboxylic acids, however, platinum as cathode material should be avoided, since cathodic hydrogenation can occur; in this case a steel cathode should be used instead.

In summary, good yields and selectivities in the Kolbe electrolysis are favored by the following experimental conditions: an undivided beaker-type cell is sufficient for most cases, it should be equipped with a smooth platinum anode of small size (area of 1 to 10 cm^2) to assure a high current density (0.25 A cm⁻² and higher) and a platinum, steel or nickel cathode in close distance; the current is provided by a regulated d.c. power supply (Figure 2); methanol is preferred as solvent, and the concentration of the carboxylic acid should be as high as possible. It is neutralized to an extent of 2 to 10% with sodium methoxide. A double-walled cell should be used to maintain temperatures between 10 to 45 °C by external cooling.



Figure 2 Beaker-type cell: (a) double-walled cell; (b) platinum sheet or grid on teflon support; (c) electrode leads (steel rods) to d.c. power supply; (d) thermometer; (e) magnetic stirring bar; (f) teflon stopper

2.8.3 MECHANISM OF THE KOLBE ELECTROLYSIS

In the past, widely differing mechanisms have been proposed for the Kolbe electrolysis. The free-radical theory, which assumed acyloxy radicals as intermediates, was suggested by Crum-Brown and Walker.⁴⁵ According to Glasstone and Hickling⁴⁶ in aqueous solution hydroxyl radicals and hydrogen peroxide were formed, which converted the carboxylic acids into dimers and alcohols or esters, the socalled Hofer-Moest products.⁴⁷ Schall⁴⁸ and Fichter¹⁶ proposed that the carboxylates are oxidatively coupled to diacyl peroxides, that subsequently decomposed. These mechanistic proposals have been reviewed.^{49,50} The presently accepted mechanistic schemes have been reviewed,^{6-8,51-54} and the general scheme summarized in Scheme 2 is assumed.

Current potential curves exhibit a critical potential at 2.1 to 2.4 V versus the normal hydrogen electrode. At this critical potential the coverage of the electrode with carboxylate increases sharply.⁵⁵ From electroanalytical data^{55,36} it is deduced that at the critical potential, and higher, a rigid layer of adsorbed acyloxy or alkyl radicals is formed on top of the metal^{7c,56,57} or metal oxide,⁵⁰ which inhibits oxygen evo-lution or solvent oxidation and promotes the Kolbe electrolysis.^{6,7a,b,58} The oxide formation and the adsorption of intermediates during the Kolbe electrolysis have also been followed by modulated specular reflectance spectroscopy.⁵⁹ Examination of the electrode surface by polaromicrotribometry has indicated that at the critical potential a hydrophobic film is formed that seems to inhibit the oxygen discharge.⁶⁰ The rate-determining step for the Kolbe reaction is most probably an irreversible single electron transfer



alcohols, ethers, alkenes, esters

Scheme 2

from the carboxylate with simultaneous bond-breaking, leading to the alkyl radical and carbon dioxide.⁷ The electroanalytical data have been interpreted in favor of a mechanism where adsorbed radicals couple.^{6,61} On the other hand, the stereochemistry of the products and the regioselectivity of the coupling reaction indicates that adsorption of alkyl radicals plays a minor role. This is indicated by the dimers of cyanoalkyl radicals (1),⁷ which can combine at carbon and/or nitrogen. Here the same ratios of carbon to carbon and carbon to nitrogen coupling products are obtained, irrespective of whether (1) is generated by photolytic or thermal decomposition of the corresponding azonitrile, by persulfate oxidation or electrolysis of α -cyanoacetic acid in different solvents. However, when azonitrile is adsorbed on silica the decomposition yields only the carbon to carbon coupling product. The radical (4; Scheme 3) can add to butadiene to yield 1,1-, 1,3- and 3,3-additive dimers. The reaction gives a further indication that adsorption is not involved. The same product ratio is found, independent of the formation of (4) either by anodic or persulfate oxidation of methyl malonate (2) or reductive cleavage of the hydroperoxide (3).⁶²



Carboxylic acids, that bear an asymmetric center at C-2, lose all their optical activity in the Kolbe electrolysis.^{63,64} This racemization can be interpreted as the result of a free radical or a fast equilibrium during desorption and adsorption at the electrode. A more detailed picture has been obtained by looking at the diastereoselectivity of the coupling of the cyclohexane- and cyclohexene-carboxylic acids (5)-(9).^{65,66} The saturated acids (5) and (6) couple randomly, which indicated that the intermediate radicals in these cases are not significantly adsorbed. With the unsaturated acids (7)-(9) the ratio of dimers was not totally statistical, reflecting somewhat the different stabilities of the products. This result can be interpreted to be due to a more product-like transition state that indicates some adsorption of the unsaturated radical.

The intermediate radical can be further oxidized to form a carbenium ion, that undergoes solvolysis, rearrangement and elimination. This conversion is strongly influenced by the structure of the carboxylic acid, the electrolyte, the presence of foreign anions and the anode material (see Section 2.8.2). Besides



the dimerization or addition to double bonds (Sections 2.8.4–2.8.6) disproportionation products of the radicals are also found. The ratio of disproportionation to coupling seems to be higher in the electrolysis than in the homogeneous reaction.⁶⁷

2.8.4 SYMMETRICAL COUPLING REACTIONS OF CARBOXYLIC ACIDS

The alkyl groups of two identical carboxylic acids can be coupled to symmetrical dimers in the presence of a fair number of functional groups (equation 1). Since free radicals are the reactive intermediates, polar substituents need not be protected. This saves the steps for protection and deprotection that are necessary in such cases when electrophilic or nucleophilic C—C bond-forming reactions are involved. Furthermore, carboxylic acids are available in a wide variety from natural or petrochemical sources, or can be readily prepared from a large variety of precursors. Compared to chemical methods for the construction of symmetrical compounds, such as nucleophilic substitution or addition, decomposition of azo compounds or of diacyl peroxides, these advantages make the Kolbe electrolysis the method of choice for the synthesis of symmetrical target molecules. No other chemical method is available that allows the decarboxylative dimerization of carboxylic acids.

$$R^1$$
, R^2 , R^3 = H, alkyl, arylalkyl
 R^1 , R^2 = H; R^3 = CO₂Me, (CH₂)_nX (X = COR, CO₂R, n ≥ 1; X = OAc, NHAc, Hal, n ≥ 4)

A large number of different Kolbe dimerizations has been compiled.^{8,10,19,20} Without attempting a complete coverage of the published reactions, the scope and limitations of the Kolbe electrolysis to afford symmetrical compounds are indicated in Table 2. In general, only the substituents at C-2 are critical for the yield of dimer. Electron-attracting groups (cyano, alkoxycarbonyl, carbonyl substituents) or hydrogen favor the radical coupling. Electron-donating substituents (two or three alkyl groups at C-2, phenyl, vinyl, halo, hydroxy or amino groups) on the other hand, more or less shift the reaction towards nonKolbe products. The experimental conditions that favor the radical coupling, namely high current density, slightly acidic electrolyte, platinum as anode material and no anions other than the carboxylate, have been treated in Section 2.8.2. The dimerization of two identical carboxylic acids has been used for the efficient synthesis of symmetrical natural products. Thus, the Kolbe electrolysis of (10) is the key step in a $(+)-\alpha$ -onocerin synthesis,⁸⁹ and by dimerization of (11) a pentacyclosqualene has been prepared.⁹⁰ An alkane with two quaternary carbon atoms has been obtained from (12),⁹¹ and 2,6,10,15,19,23-hexamethyltetracontane was accessible from (13).⁹²

For the preparation of long chain alkanes from fatty acids it is useful to extract the electrolyte continously with a high-boiling nonpolar solvent,⁹³ e.g. 2-methylheptane. Cyclopropanecarboxylic acids in some cases have been dimerized, e.g. $(14)^{94}$ or (15);³⁴ in other cases the radical is further oxidized to a carbocation which then undergoes ring opening and solvolysis to allylic compounds. Specifically substituted 1,2-diphenylethanes have been prepared from phenylacetic acids (Table 2, entries 22 and 23). A variety of 2,3-disubstituted succinic acids and their derivatives have become accessible from malonic

Entry	Substituents in R ¹	$\frac{R^1R^2R^3CCO_2}{R^2}$	<i>R</i> ³	Yield [®] of coupling product (%)	Ref.
1 2 3 4	H Alkyl (C_1-C_{17}) Alkyl RO ₂ C(CH ₂)n n = 3-15 P = Ma Et	H H Aikyl H	H H H H	93 30–90 0–26 40–95	5, 8 5, 8 8 5
5 6 7 8 9	Alkyl Alkyl Alkyl Alkyl Alkyl AlkylCO MeCO(CHa)r	EtO2C Alkyl CN CONH2 Alkyl, H	H EtO2C H, Alkyl H, Alkyl Alkyl, H	20-85 15-35 30-60 5-55 32-40	68, 69 68, 70 71 72 73
11 12a 12b 13	n = 1 n = 5 F, Br, Cl, I, OH, NH ₂ F F X(CH ₂)n	H F F H	H F Cl H	70 63 0 93 43	74 75 5, 8, 76 56b, 77 78
14 15	X = F, n > 3 X = Cl, n = 1, 3, 4, 5, 7, 8, 9, 11 X = Br, n = 4-11, 14 $RO_2C(CF_2)n$ n = 0-3, R = Me, Et $H(CF_2)n$	F	F F	45–70 40–80 54–71 High yield 40–86	79 79, 80, 81 30a, 79 82 83
16 17	n = 3, 5 CF ₃ (C ₂ F ₅) ₂ C MeCO ₂ (CH ₂) _n	H H	H H	75 70–83	84 25b
18 19	n = 2-3 MeCONH(CH ₂) ₄ Phthaloyt-NCH(CO ₂ H)CH ₂	H H	H H	2439 32	25b, 85 86
20	x	н	н		
21	$X = p - OMe \text{ to } X = F_5$ Ph	Ph	н	<1–74 25	24 87
22	\rightarrow	н	н	45	87
23		н	Н	39	88

 Table 2
 Symmetrical Coupling Reactions of Selected Carboxylic Acids

*Electrolyses are normally performed at a platinum anode in water, methanol, ethanol or ethanol/water.





half esters, nitriles or amides (Table 2, entries 5-8). The anodic coupling of (16) has been used as part of a semibullvalene synthesis.⁹⁵

The dimerization of half esters to 1,*n*-diesters is also called Brown–Walker electrolysis. Thereby valuable intermediates for the synthesis of medium-sized rings or 1,*n*-difunctionalized compounds can be prepared (Table 2, entry 4). This reaction is also of industrial interest since in this way sebacic acid can be prepared from adipic acid half ester.⁹⁶ This process has been scaled-up in Germany,⁹⁷ the USSR⁹⁸ and Japan,⁹⁹ and yields as high as 93% have been reported.¹⁰⁰ Reaction conditions and yields for the coupling of other half esters have also been studied in detail.⁹⁸ 1,*n*-Polyethylene- or polydifluoromethylene-dicarboxylic acids are reported to be formed by electrolysis of azelaic acid¹⁰¹ or perfluoroglutaric acid.¹⁰² Ketocarboxylic acids can be coupled to 1,4-, 1,6- or 1,14-diketones (Table 2, entries 9 and 10). Aldehydes must be dimerized in the form of their acetals to obtain good yields, as has been shown for (17)¹⁰³ and (18).¹⁰⁴ The arrow on (10)–(18) indicates the location of dimerization, along with the yield and reaction conditions.

Carboxylic acids that carry a halide, hydroxy or amino group at C-2 generally cannot be dimerized (Table 2, entry 11). In the case of trifluoroacetic acid, however, high yields of hexafluoroethane are reported (Table 2, entry 12). When the halide substituent is more distant from the carboxyl group the dimer yields are in general good (Table 2, entry 13). The coupling of fluorocarboxylic acids has been investigated intensively (Table 2, entries 14–16) because of the interesting properties of the fluorohydrocarbons produced.

Hydroxy- and amino acids can be dimerized in good to moderate yields, provided the substituents are not at C-2 or C-3 and are additionally protected against oxidation by acylation (Table 2, entries 17–19). 2-Alkenoic acids cannot be dimerized, but lead to a severe passivation of the anode due to the formation of polymer films.¹⁰⁵ 3- and 4-alkenoic acids can be dimerized in moderate yields, when they are neutralized with tributyl- or triethyl-amine.¹⁰⁵ 3-Alkenoic acids, obtained from Stobbe condensations, can be dimerized in 14–30% yield; major side products arise from further oxidation of the allyl radical to its cation and subsequent solvolysis.¹⁰⁶ The 3-alkenoic acids dimerize to three isomeric 1,5-dienes (Scheme 4), that arise by 1,1'-, 1,3'- and 3,3'-coupling of the intermediate allyl radical.¹⁰⁷ By steric shielding of the 3position the proportion of 3-coupling decreases. Depending on the bulkiness of the 3-substituent, the relative yield of the 1,1'-dimer can vary from 52 to 73% (Table 3). The configuration of the double bond is retained to a high degree.^{107a}

In the case of 6-alkenoic acids the intermediate 5-hexenyl radical cyclizes to a cyclopentylmethyl radical in a 5-exo-trig cyclization (Scheme 5; see also Section 2.8.6)^{107a,108,109} In some cases double bond migration has been observed. This can be avoided, however, when the electrolyte is prevented from becoming basic by using a mercury cathode. (Z)-4-Enoic acids isomerize partially to the corresponding (E)-isomers. The products from methyl- and deuterium-labeled carboxylic acids indicate that this isomerization occurs via a reversible ring closure to cyclopropylcarbinyl radicals. (Z)-Enoic acids where the double bond is more distant from the carboxylic acid group retain totally their configuration.¹⁰⁹ A key step in a perhydrophenanthrene synthesis involves the dimerization of (**19**) by Kolbe electrolysis.¹¹⁰





Carboxylic acid [®]	1,1'	Product distribution (%) 1,3'	3,3'	Yield (%)
Me(CH ₂) ₇ CH—CHCH ₂ CO ₂ ⁻ Me ₂ CHCH—CHCH ₂ CO ₂ ⁻ Me ₃ CCH—CHCH ₂ CO ₂ ⁻	52 59 60	39 41 40	9	67 79 15
	60	40		45
$Me_2CH = CHCH_2CO_2^{-}$	65	-	36	42
	76	24	_	29

 Table 3
 Yields in the Kolbe Dimerization of 3-Alkenoic Acids^{107b}

 $^{a}10-150$ mmol were dissolved in 25-100 mL methanol, neutralized to 8-50% with triethylamine and electrolyzed in an undivided cell at platinum electrodes at 400-800 mA cm⁻² with change of polarity of the electrodes until pH = 8 was reached.



(19)

In unsaturated carboxylic acids, where the oxidation of the double bond is facilitated by alkyl substitution, the double bond rather than the carboxylic group is oxidized, and a lactone is formed (Scheme 6).¹¹¹



Scheme 6

Other alkenoic acids that have been coupled without change of the double bond configuration are oleic acid (24% yield of dimer), elaidic acid (44%)¹¹² and erucic acid.¹¹³

2.8.4.1 Experimental Procedure for Symmetrical Kolbe Electrolysis — Preparation of Dimethyl Hexadecanedioate

A solution of 55 g (0.27 mol) of methyl azelate in 350 mL methanol, neutralized to the extent of 6% with KOH in methanol, is electrolyzed in a flow-through cell (Figure 1) at 35–40 °C at a cell voltage of 40 V between a platinum anode (area = 17 cm^2) and a steel cathode at a current density of 200 mA cm⁻² until 1.2 F m⁻¹ were consumed. The methanol is then evaporated, and the residue is dissolved in 500 mL ether and extracted with a saturated aqueous NaHCO₃ solution (3 × 100 mL) and subsequently washed with water. After drying (MgSO₄) and removal of the ether, fractionated distillation affords at b.p. 107–113 °C/0.02 Torr, 34.14 g (80%) dimethyl hexadecanedioate.¹¹³ For the preparation of dimethyl decanedioate from methyl adipate see ref. 114.

2.8.5 CROSS-COUPLING REACTIONS OF DIFFERENT CARBOXYLIC ACIDS

Cross-coupling reactions of two carboxylates with different alkyl groups by anodic decarboxylation (mixed Kolbe electrolysis) is an electrochemical method that allows the synthesis of unsymmetrical compounds (Scheme 7).

$$4R^{1}CO_{2}H + 4R^{2}CO_{2}H - \frac{-e^{-}}{-CO_{2}} + 4R^{1} + 4R^{2} + R^{2} - R^{1} + 2R^{1} - R^{2} + R^{2} - R^{2}$$

Scheme 7

A disadvantage of this reaction is the statistical coupling of the intermediate radicals. This results in the additional formation of two symmetrical dimers as major side products. However, by taking the less costly acid in excess the number of products is lowered to two, the unsymmetrical product and one symmetrical dimer. When the chain length of the two acids is properly chosen, the two products can usually be separated by distillation. Furthermore, the more costly acid is incorporated to a major extent into the mixed dimer. The calculated yields for mixed couplings are listed in Table 4, and the experimental results for the coupling of methyl azelate with pentanoate are shown in Table 5.¹¹³

When one of the two acids is used in excess and the acidity of the two acids differs strongly, the saltdeficit method should be used with caution, as one of the acids might be preferentially converted into the carboxylate. In such a case it is useful to neutralize both acids completely.

For the selective preparation of unsymmetrical coupling products on a small scale, the photolysis of peroxides is a favorable alternative to the mixed Kolbe electrolysis. By photolysis of unsymmetrical diacyl peroxides at -60 to -70 °C in the solid state two different alkyl groups can be joined selectively.¹¹⁵

R^1CO_2H	Ratio	R ² CO ₂ H	Yield of $R^1 - R^2 (\%)^{\mu}$
1 1 1 1		1 2 4 6 8	50 66 80 86 89

Table 4	Calculated Theoretical	Yields for Mi	xed Coupling	Reactions Products	$R^{1}R^{2}$
---------	------------------------	---------------	--------------	---------------------------	--------------

*Calculated according to Yield (%) = 100n/(1 + n) with 1:n as ratio of R¹CO₂H:R²CO₂H.

Table 5	Yield of Methy	vl Dodecanoate in	the Cross-couplin	g Reactions of Meth	vl Azelate and Valerate ¹¹³
				- A reaction of the	

Rat	io	
HO ₂ C(CH ₂) ₇ CO ₂ Me	$Me(CH_2)_3CO_2H$	Yield of MeO ₂ C(CH ₂) ₁₀ Me in MeOH (%)
1	2	56
1	4	72
1	8	83

Problems with passivation, that is, the coverage of the anode with insulating films which increase the cell voltage, or with the competitive oxidation to carbenium ions, are often less pronounced in crossed coupling.^{113,116}

The mixed Kolbe electrolysis has turned out to be a widely used synthetic method, despite its disadvantage that at least one symmetrical dimer is formed as major side product, which has to be separated and which consumes a greater part of the coacid. Crossed Kolbe electrolyses have been used for the efficient synthesis of rare fatty acids and related compounds, of pheromones and of chiral building blocks. Weedon *et al.* during the 1950s and 1960s did much of the pioneering work in these type of syntheses.⁵ Selected examples from the numerous reactions that have been published are compiled in Tables 6 and 7 and in the formulae (20)–(39). In some of the formulae the end-product of several transformations is shown; the bond formed in the mixed Kolbe electrolysis is marked by an arrow and the yield is indicated. Further examples of crossed couplings can be found in refs. 8 and 12.

 Table 6
 Cross-coupling Reactions by Kolbe Electrolysis of Unsubstituted (A) with Substituted Carboxylic Acids (B)

Carbo	oxylic acids	Yield of cross-coupling	Ref.
A—CO ₂ H	B—CO2H	product (A—B) (%)	
Me(CH ₂) ₆ CO ₂ H Me(CH ₂) ₁₄ CO ₂ H Me(CH ₂) ₅ CO ₂ H CD ₃ CO ₂ H CD ₃ CO ₂ H MeCO ₂ H Me(CH ₂) ₃ CO ₂ H Me(CH ₂) ₃ CO ₂ H Me(CH ₂) ₅ CO ₂ H Me(CH ₂) ₂ CO ₂ H Me(CH ₂) ₂ CO ₂ H Methylhexadecanoic acid Me(CH ₂) ₈ CO ₂ H	$\begin{array}{c} MeO_2C(CH_2)_4C=C(CH_2)_4CO_2H\\ Me_2CHCO_2H\\ (E)-Me(CH_2)_7CH=CH(CH_2)_7CO_2H\\ CF_3CO_2H\\ CF_3CF_2CO_2H\\ HC=C(CH_2)_8CO_2H\\ Ph(CH_2)_2CO_2H\\ MeO_2C(CH_2)_7CO_2H\\ MeO_2C(CH_2)_7CO_2H\\ MeO_2CCH(OCOMe)CO_2H\\ MeO_2CCH(OCOMe)CO_2H\\ MeO_2C(CH_2)_4CO_2H\\ MeO_2C(CH_2)_4CO_2H\\ MeO_2C(CH_2)_4CO_2H\\ \end{array}$	23 40 37 68 75 36 44 69 61 29 62 42 38	117 118 118 119 120 121 113 113 122 123 124 125
Me(CH ₂) ₂ CO ₂ H	MeO ₂ CCH ₂ CHMeCH ₂ CO ₂ H	30	126
CO ₂ H	MeO ₂ C(CH ₂) ₁₁ CO ₂ H	30	127
Bu ^l (CH ₂) ₂ CHMeCH ₂ CO ₂ H	MeO ₂ CCH ₂ CHEtCH ₂ CO ₂ H	15	128
Me(CH ₂) ₁₀ CO ₂ H	MeO ₂ CCH(OMe)CH ₂ CO ₂ H	21	129
$Me(CH_2)7CHMeCH_2CO_2H$ $Me(CH_2)_mCO_2H$ $m = 8-16, 18$ $Me_2CHCH_2CO_2H$	MeO ₂ C(CH ₂) ₇ CO ₂ H MeO ₂ C(CH ₂) _n CO ₂ H n = 4, 7, 8, 12–20 Et ₂ C(CO ₂ Et)CO ₂ H	30 38	130 131 132

Carboxylic acids A—CO2H	В—СО₂Н	Yield of cross-coupling product (A—B) (%)	Ref.
CH2—CH(CH2)2CO2H Me2CHCH—CHCH2CO2H	$MeO_2C(CH_2)_{10}CO_2H$ $MeO_2C(CH_2)_nCO_2H$ $n = 1, 2, 4$	68 Good	133 134a
(Z)-MeO ₂ C(CH ₂) ₂ CH \rightarrow CH(CH ₂) ₂ CO ₂ H CH ₂ \rightarrow CH(CH ₂) ₈ CO ₂ H RC \rightarrow C(CH ₂) ₃ CO ₂ H	$\begin{array}{c} Me(CH_2)_2CO_2H\\ MeO_2C(CH_2)_2CO_2H\\ MeO_2C(CH_2)_nCO_2H\\ MeO_2C(CH_2)_nCO_2H\\ \end{array}$	30 52 45–59	134b 134c 135
K = Et, FI, Bu (Z)-Me(CH ₂) ₇ CH—CH(CH ₂) ₇ CO ₂ H CF ₃ CF ₂ OCF ₂ CO ₂ H	n = 2, 3, 4, 6 MeO ₂ C(CH ₂) ₂ CO ₂ H MeO ₂ C(CF ₂) _n CO ₂ H n = 1-3	15–20 24–34	136 137
CF3OCF2CF2CO2H CF3CO2H Br(CH2)10CO2H	$MeO_{2}C(CF_{2})_{n}CO_{2}H$ $n = 1-3$ $MeO_{2}CCH_{2}CO_{2}H$ $MeO_{2}C(CH_{2})_{7}CO_{2}H$	24-34 46 50	137 138 113
Br(CH2)10CO2H Br(CH2)10CO2H AcO(CH2)4CO2H Me3SiCH2CO2H	$EtO_2C(CH_2)_8CO_2H$ MeO_2C(CH_2)_4CO_2H EtO_2C(CH_2)_{10}CO_2H MeO_2C(CH_2)_nCO_2H	54 37 27–32 71–76	30b 140, 141 142
MeO ₂ C(CH ₂) ₁₁ CO ₂ H MeO ₂ C(CH ₂) ₁₀ CO ₂ H MeO ₂ C(CF ₂) _n CO ₂ H	n = 4, 7 MeO ₂ C(CH ₂) ₄ CO ₂ H MeO ₂ C(CH ₂) ₄ CO ₂ H MeO ₂ C(CF ₂) _m CO ₂ H	25-38	143 144 82
n = m = 1-4 MeCO(CH ₂) ₂ CO ₂ H MeCO(CH ₂) ₈ CO ₂ H	MeCO(CH ₂) ₄ CO ₂ H MeO ₂ C(CH ₂) ₄ CO ₂ H	31 32	145 146

 Table 7 Cross-coupling Reactions by Kolbe Electrolysis of Substituted Carboxylic Acids (A) with Substituted Carboxylic Acids (B)

Tables 6 and 7 contain the syntheses of esters of trialkylacetic acids, specifically deuterated carboxylic acids, of branched acids, of enoic and ynoic acids, of ω -halo-, acetoxy- or trimethylsilyl-carboxylic acids, of perfluorinated 1,*n*-diacids, of keto esters and of diketones.

Crossed coupling has also been applied to the extension of the carbon chain in fatty acids. Extension by two carbon atoms is achieved with succinate half esters,^{120,147–149} whereby separation problems can be simplified by using the benzyl half ester.^{148a} Expansion by the propane unit has been accomplished with methyl glutarate,¹⁵⁰ by the isoprene unit with ethyl 3-methyladipate,¹⁵¹ by four carbons with methyl adipate,¹¹² by the pentane unit with methyl pimelate,¹⁵² and by six carbon atoms with methyl suberate.¹⁵³

A variety of pheromones have been prepared in few steps using the crossed coupling reaction. Muscalure (20), the pheromone of the housefly, becomes available from the natural pool of unsaturated fatty acids, namely oleic or erucic acid.¹⁵⁴⁻¹⁵⁶ The synergist of muscalure, (Z)-11-heneicosene (21), was similarly prepared.¹⁵⁵ Methyl (Z)-4-octenedioate, which is accessible from (Z,Z)-1,5-cyclooctadiene by partial ozonolysis, is a valuable starting acid to prepare (22),^{134b} a precursor of the cabbage looper pheromone, or disparlure (23),¹⁵⁷ the sex attractant of the gypsy moth. The optically active *Trogoderma* pheromones (E)- and (Z)-(24) have been synthesized using 4-pentynoic acid and (S)-citronellic acid as precursor from the chiral pool.¹⁵⁸ Alkatrienes (25), which are sex attractants of certain Lepidoptera. become accessible in a one step synthesis using linolenic acid from the pool of polyunsaturated fatty acids.¹⁵⁹ Compound (26), the pheromone of the German cockroach Blattela germanica, was obtained from 3-methyl heneicosanoic acid.¹⁶⁰ The bromide (29), which is an important intermediate for the synthesis of natural α -tocopherol has been synthesized starting from (R)-(+)-citronellic acid.¹⁶¹ The coelectrolysis of (Z)-4-nonenoic acid with methyl glutarate provided an alternative approach to looplure (31),¹⁶² the pheromone of the Cabbage looper. Similarly the pheromone of the fruit pest insect Dacus cucurbitae (32)¹⁶³ and of the false codling moth (33)¹⁶⁴ became accessible by crossed coupling. Brevicomin (34), the sex attractant of the western pine beetle, became available in a short route by mixed Kolbe electrolysis of (E)-3-hexenoic acid with levulinic acid.¹⁶⁵ Besides (23), a number of other compounds have been prepared by successive cross couplings. (±)-Tuberculostearic acid (27) has been obtained in two consecutive Kolbe electrolyses from 2-methylsuccinate.¹⁶⁶ 10,16-Dimethylicosanoic acid (28) was synthesized in three subsequent Kolbe electrolyses.¹⁴⁷ The alkane (30) with three quaternary carbon atoms was prepared in two successive crossed couplings.⁹¹ Pure (E)- or (Z)-configurated unsaturated pheromones are accessible by mixed coupling with 5-alkynoic acids and subsequent selective hydrogenation (Table 7).¹³⁵ The mixed Kolbe electrolysis of protected L- or D-glutamic acids provides access to dicarba analogs of cystine peptides (35).¹⁶⁷ Enantiomerically pure β-hydroxybutyric acid derivatives have been converted into the useful chiral building blocks (36)-(38).¹⁶⁸ Chiral γ -lactones (39) could be obtained from (R)-3-cyclohexene-1-carboxylic acid.¹⁶⁹



2.8.5.1 Experimental Procedure for Mixed Kolbe Electrolysis — Preparation of 11-Hexadecynoic Acid

In a beaker-type cell (Figure 2) equipped with two platinum electrodes (area 10 cm^2 , thickness 0.05 mm, supported on a teflon frame), a solution of 14.3 mmol 5-decynoic acid and 86 mmol methyl suberate in 40 mL methanol, neutralized to 5% with KOH, was electrolyzed at 22 °C and a current density of 100 mA cm⁻², provided from a regulated d.c. power supply, until a pH of 8–9 was reached. After evaporation of the methanol the crude product was refluxed with equimolar amounts of KOH in 100–200 mL methanol for 24 h; thereafter the mixture was evaporated to dryness and extracted with petroleum ether. The remaining salt was dissolved in water, acidified with 15% HCl, and continuously extracted with petroleum ether. The precipitated tetradecanoic dicarboxylic acid (72%, m.p. 126 °C) was filtered off, and after evaporation of the petroleum ether 8.43 mmol (59%) 11-hexadecynoic acid was isolated by bulb to bulb distillation; m.p. (from pentane) 40–41 °C.¹³⁵

2.8.6 ADDITION OF KOLBE RADICALS TO DOUBLE BONDS

The radicals generated by anodic decarboxylation of carboxylic acids can be intercepted by alkenes that are present in the electrolyte. The adduct radical can couple with the radical generated from the carboxylate to form an additive monomer I (Scheme 8 path a), it can dimerize to form an additive dimer II (path b), it can be further oxidized to a cation, which reacts with a nucleophile to form III (path c), or it can disproportionate (path d).



Scheme 8

As the reactive intermediates in electrolyses are confined to narrow reaction layers in front of the electrode, the radical concentration is much higher there than in homogeneous radical reactions. Therefore the propagation step of a polymerization is suppressed and the termination step leading to products I and II predominates. The products I and II appear to be exclusive products of electrolysis.

The high radical concentration at the anode, on the other hand, favors the dimerization of the radical generated from the carboxylate, which leads in the case of less reactive alkenes, like cyclohexene or 2-methylpropene (Table 8, entry 11), to low yields of adducts and the predominant formation of Kolbe dimers. With reactive alkenes (butadiene, isoprene, styrene) the yields of adducts are in general good (Table 8, entries 1–10). However, in the case of unreactive ethylene high adduct yields have also been claimed (Table 8, entry 12). Higher yields are found in the reaction of nucleophilic radicals with electrophilic alkenes (Table 8, entries 17–22) or *vice versa* (Table 8, entries 23–27). The ratio of additive monomer I Scheme 8 to additive dimer II can be controlled to some extent by the current density. High current densities aid the formation of I, low ones that of II (Table 8, entries 4 and 5). These results can be interpreted according to Scheme 8. When high current densities prevail, which correspond to high radical concentrations in the reaction layer, the alkene can trap only some of the radicals generated from the carboxylate (Kolbe radicals). This leads to their preferred combination with the primary adduct leading

to I and their dimerization to the Kolbe dimer. At low current densities, on the other hand, the major proportion of the Kolbe radicals are trapped by the alkene, which leads to a preferential formation of II.

In the addition of Kolbe radicals to 1,3-dienes allyl radicals are intermediates, that can then couple to form 1,1'-, 1,3'- and 3,3'-dimers (Scheme 9). In general, the reactivity of the allyl radical is two to three times higher at C-1 than at C-3, so that the 1,1'-dimer predominates.¹⁷⁰ Besides the examples given in Table 8, further additions of Kolbe radicals to dienes can be found in refs. 33, 179 and 191.



Scheme 9

The addition of the 5-pentanoate radical from methyl adipate to butadiene has been intensively investigated, because in this way long chain 1,*n*-diacids are easily accessible; a total yield of 96% has been claimed for this reaction (Table 8, entry 7). Different Kolbe radicals from acetic acid, monochloroacetic acid, trichloroacetic acid, oxalic acid, methyl adipate and methyl glutarate have been added to ethylene, propylene, fluoroalkenes and dimethyl maleate.¹⁷⁷ In this detailed study the influence of current density, alkene type and alkene concentration on the product yields and product ratios have been discussed.

In some cases reactive alkenes can be polymerized with Kolbe radicals as initiators, *e.g.* styrene,¹⁷⁶ acrylonitrile,¹⁷⁶ vinyl acetate,¹⁹² methyl acrylate,¹⁹² acrylic acid,¹⁹³ acrylamide¹⁹⁴ or vinyl chloride.¹⁹³ The addition of Kolbe radicals to pyridine,¹⁹⁵ benzotrifluoride or benzonitrile¹⁹⁶ affords only low yields.

The addition of alkyl radicals to less reactive alkenes becomes more efficient if the addition is conducted intramolecularly. The Kolbe electrolysis of $\Delta^{6,7-}$ and $\Delta^{7,8-}$ unsaturated carboxylates produces fiveand six-membered rings in a 5- or 6-*exo-trig*-cyclization.¹⁰⁸ Such a cyclization was first observed by Weedon¹⁰⁷ and later studied in more detail.¹⁰⁹ By intramolecular addition of appropriate precursors, 3substituted tetrahydrofurans (Scheme 10; **40**, 41%),¹⁹⁷ pyrrolidines (**41**, 53%),¹⁹⁸ and cyclopentanes (**42**, 74%), and (**43**, 75%)¹⁹⁹ are obtained. In the reactions leading to cyclopentanes the extent of cyclization



Coupling Reactions

Entry	Carboxylic acid	Alkene	Yield of additive monomer (%)	Yield of additive dimer (%)	Ref.
1 2 3	MeO2CCH2CO2H MeO2CCH2CO2H MeO2CCH2CO2H MeO2CCH2CO2H	Styrene Butadiene Isoprene	4 17 1	38ª 18 43	170 170 170
4	EtO2CCO2H	Butadiene ($i = 0.025 \text{ A cm}^{-2}$)	4	66	170, 171
5	EtO ₂ CCO ₂ H	Butadiene ($i = 0.66 \text{ A cm}^{-2}$)	15	8	170
07		Isoprene Butadiana	8 49	39 48	172 173
8a	HOCCOH	Butadiene	40 —	20	172, 173
		O 			
<u>еь</u> -	HOLCCOL	OMe	62	34	175
80	NO2CCO2N	OMe	02	54	175
9	MeO ₂ CCH ₂ CO ₂ H	α -Methylstyrene	_	46 ^b	107ь
10	MeCO ₂ H	Styrene	-	15	176
11	EtO ₂ CCO ₂ H	Isobutene	- 70		170
12	MeCO2H ClaCCO2H	CH2-CH2	≈/0		179
14	HOCCOTH		25	46	170
15	MeO ₂ C(CH ₂) ₄ CO ₂ H	CH2=CH2	15	10	177
16	MeCO ₂ H	Bu ⁴ CH—CH ₂	46	_	179
17	MeCO ₂ H	CH2-CMeCHO	—	80	180
18	MeCO ₂ H	CH2—CHCO2Et	_	70	180
		∠CO₂Me			
10	M-00-11	l z	01 bisb		101 102
19	MeCO2H	CO ₂ Me	21—nign		101, 102
20	PhCH ₂ CO ₂ H	CO ₂ Me	24	_	183
21	MeCO ₂ H	$ \begin{array}{c} O \\ N-R \\ O \\ R = Et, Ph \end{array} $	80 –8 8	_	182, 183
22	$RCO_2H.R = Me. Et. Bu^i$	PhSO ₂ CH=CH ₂	37-69		184
			0 0		
23	CF ₃ CO ₂ H	MeC(OAc)—CHCO2C8H17		I ₁₇ 185	
			64		
24	MeO ₂ CCH ₂ CO ₂ H	CH2=CHOEt		35	186
23 26	CF3CU2H CF3CO2H		1. <u>8</u> ª	40	18/
20			10		100
27	CF3CO9H	SO	31	_	189
~ '	01,00211	, SU ₂	5.		. 07
28	CECONH	CHamCHCOaMe	5	50	187
29	CF ₃ CO ₂ H	CH2—CHCO2H	ı 7	28	138
30	CF ₃ CO ₂ H	CH2-CHCF3		40	177
31	CF ₃ CO ₂ H	CHF-CF2		30	177

Table 8 Addition of Kolbe Radicals to Alkenes
Entry	Carboxylic acid	Alkene	Yield of additive monomer (%)	Yield of additive dimer (%)	Ref.
		CO ₂ R	30	10	177
32	CF ₃ CO ₂ H	CO ₂ R	41	27	189
		R = Me, Et			
33	CF3CO2H	$\langle \rangle _{n}$ CO ₂ Et	4-42		190
		<i>n</i> = 0–3			

Table 8 Addition of Kolbe Radicals to Alkenes (continued)

*Additional 16% disproportionation product. ^b6% Product type IV, Scheme 8.

can be considerably increased by lowering the current density (300 mA cm⁻², ratio of cyclized:uncyclized product = 32:35; 10 mA cm⁻² ratio = 74:8). The intramolecular cyclization has also been applied to the stereoselective synthesis of a prostaglandine precursor (equation 2).²⁰⁰ The bonds formed in the Kolbe radical reactions are shown in (40)–(43).



$$\mathbf{R} = \mathbf{Me}, (\mathbf{CH}_2)_2 \mathbf{CO}_2 \mathbf{Me}$$

Compared to chemical radical cyclizations,²⁰¹ the ring closures *via* Kolbe radicals have the advantage that here two carbon–carbon bonds can be joined in one step, whilst in the majority of chemical alternatives only one carbon–carbon bond is formed. Furthermore electrolysis avoids the toxic tributyltin hydride, that is usually needed as initiator or reagent in chemical radical cyclizations.

2.8.7 THE NONKOLBE ELECTROLYSIS

Besides the aforementioned Kolbe dimers, alcohols, esters or ethers can become the major products in the electrolysis of carboxylic acids. These results have suggested that in anodic decarboxylation the intermediate radicals were further oxidized to carbocations that yielded solvolysis and elimination products.²⁰² This part of the anodic decarboxylation, which leads to carbonium ions, is frequently called nonKolbe electrolysis. Applications of the nonKolbe electrolysis to synthesis and to mechanistic investigation of carbocations are summarized in refs. 8, 19, 20 and 23.

The oxidation of carboxylates to radicals (Scheme 2) has to be extended by the additional steps, shown in Scheme 11. The radical is further oxidized to a carbocation, which can solvolyze to ethers, alcohols or esters, can lose a proton to form an alkene or can undergo rearrangement or solvolysis prior to those reactions.

The formation of alcohols or ethers is favored by low current densities,^{8,24} porous graphite as anode material,^{203-205,206b} a solvent such as water/pyridine²⁰⁶ instead of methanol, a pH of the electrolyte above 7 and anions, such as bicarbonate, sulfate or perchlorate, as additives.²⁴ Electron-donating substituents at C-2 of the carboxylic acid favor the oxidation of the intermediate radical. Thus α -alkyl, α -cycloal-kyl^{207,208} α -chloro,²⁰⁹ α -bromo, α -amino, α -alkoxy, α -hydroxy, α -acyloxy and α , α -diphenyl substituents more or less promote the formation of nonKolbe products.

The nonKolbe electrolysis can lead to complex product mixtures, especially when there are equilibrating carbenium ions of about equal energy involved. On the other hand, the conversion of carboxylic acids to ethers, acetals, esters or alkenes can become very selective when the intermediate carbocation is





 Table 9
 Preparation of Ethers, Esters and Alcohols by NonKolbe Electrolysis of Carboxylates

Entry	Carboxylate	Conditions	Product	Yield (%)	Ref.
1	Crco ₂ H	MeOH, Pt	Доме	35–40	202c
	exo or endo				
2	CO ₂ H	МеОН	ОМе	100	66
3	$CO_{2}Et$ $R^{1}C - CO_{2}H$ $^{1}NHCOMe$ $R^{1} = H, Me, Et, Pr^{1}, PhCH_{2}$	R ² OH R ²	$CO_{2}Et$ $R^{1}C - OR^{2}$ $HCOMe$ $= Me, Et, Pr^{i}, MeCO$	7996	210c
	ОН		●OH		
4	HO ₂ C ^{""} N COMe	MeOH, Pt	MeO [~] N COMe	97	210ъ
5	Me(CH ₂) _n CH(SPh)CO ₂ H	MeOH, LiClO ₄ , Pt	$Me(CH_2)_nCH(OMe)_2$	72–98	211
6	$n = 2, 7$ $()_n$ CO_2H COR^1	R ² OH	$()_{n}$ N OR ² COR ¹	75–98	212
	$n = 1, 2; R^{1} = H, Me, Ph, OEt$ $R^{2} = Me, Et, H, Pr^{i}$				
7	$ \begin{array}{c} \mathbf{RO} \\ \mathbf{H} \\ \mathbf{H} \\ \mathbf{H} \\ \mathbf{CO}_2 \mathbf{H} \\ \mathbf{N} \\ \mathbf{H} \end{array} $	AcOH, NaOAc, MeC	RO H H OAc	84	213

Kolbe Reactions

stabilized, and either solvolysis or elimination can be favored. This is substantiated below with some selected examples. Suitably substituted carboxylates can be converted into ethers or acetals (Table 9).

Both *exo-* and *endo-*norbornane-2-carboxylic acids are converted to the *exo-*2-methyl ethers, indicating that a bridged norbornyl cation is the intermediate (Table 9, entry 1). 2-Cyclohexene-1-carboxylic acid yields quantitatively 2-cyclohexen-1-yl methyl ether (Table 9, entry 2). Substituted *N*-acetylaminomalonic half esters are transformed in high yields into the corresponding aminoacetals (Table 9, entry 3). 4-Hydroxy-L-proline is converted into the corresponding 2-methoxypyrrolidine (Table 9, entry 4), which can be used to prepare optically active 2-substituted pyrrolidines.^{210c} α -Phenylthio carboxylates are converted with loss of one carbon to dimethyl acetals (Table 9, entry 5). Cyclic acetals of β -ketocarboxylic acids react to form 2-methoxy-1,4-dioxacycloalkanes, that subsequently can be converted to 1,4-dioxenes. 4-Acetoxyazetidinone can be prepared by nonKolbe electrolysis of the corresponding carboxylic acid (Table 9, entry 7); it can serve as a valuable intermediate for the synthesis of β -lactams.

Entry	Carboxylic acid	Reaction condition	Product	Yield (%)	Ref.
1	O (CH ₂) ₄ Br CO ₂ H	Н ₂ О, КОН	O (CH ₂) ₄ B	^{ir} 50	73b, 214
2	$R^{1} = Pr^{i}; R^{2} = Me, MeO; R$	MeOH, NaOMe ³ = H	R^2 R^3	60–70	215
	$R^1 = Me(CH_2)_4; R^2 = R^3 =$	ОМе			
3	O = O = O = O = O = O = O = O = O = O =	C anode, NaOMe, MeOH		91	216
4	CO ₂ H	MeCN, EtOH, C anode	\langle	76	217
5	CO ₂ H	90% pyridine/H ₂ O	\langle	15	218
6	MeO ₂ C O CO ₂ H	MeOH, NaOMe, Pt anode	MeO ₂ C O	50	219

 Table 10
 Conversion of Carboxylic Acids into Alkenes

Coupling Reactions

Carboxylic acids can be transformed into alkenes when they contain a leaving group like H (Scheme 12), SiMe₃, SPh or CO₂H in the β -position. The alkene is formed by an *E*1-elimination from the intermediate carbocation. Some examples are summarized in Table 10. The decarboxylative elimination of 1,4-cyclohexadiene-6-carboxylic acids (Table 10, entry 2) is part of a useful method for the alkylation of aromatic compounds. This involves first a reductive alkylation using a Birch reduction, which is then fol-

 Table 11
 Rearrangements by NonKolbe Electrolysis

Entry	Carboxylic acid	Reaction conditions	Product	Yield (%)	Ref.
1	CO_2H exo or endo	MeOH, Et ₃ N	OMe	56	202c
2	CO ₂ Me CO ₂ H	AcOH, Bu ^t OH, Et ₃ N	MeO ₂ C OAc	56	220
3	$R^{1} = R^{2} = Me H$	MeOH, NaOMe	HO R^2 H R^2 H CO_2 Me R^1	87	221
4	$ \begin{array}{c} O \\ O \\$	MeOH, NaOMe	MeO O O O H	49–58	222
5	(CH ₂) _n C OH	Pt anode, MeCN	$(CH_2)_{n+1}$ C = O		
		C anode, pyridine/H ₂ C	n = 4 n = 5 n = 9 n = 11	54–63 54–63 82 82	202c 202c 223 223
6	H OH CO ₂ H	C anode, DMF		56	224

Kolbe Reactions

lowed by a nonKolbe electrolysis. 1,4-Cyclohexadienes are obtained from β -trimethylsilylcarboxylic acids (Table 10, entry 4); the latter are prepared by a Diels-Alder reaction of dienes with β -trimethylsilylacrylic acid or its derivatives. Vicinal dicarboxylates can be converted into alkenes by bisdecarboxylation (Table 10, entries 5 and 6). The reaction can be combined with a [4 + 2] or [2 + 2] cycloaddition, *e.g.* with maleic anhydride, whereby polycyclic hydrocarbons can be prepared in few steps.



Scheme 12

The carbocations generated by nonKolbe electrolysis can rearrange by alkyl, phenyl or oxygen migration. Some examples are given in Table 11. The carbocations generated from *exo-* or *endo-5-*norbornene-2-carboxylic acid react by double bond participation and subsequent rearrangement to 3-methoxynortricyclene (Table 11, entry 1); the same compounds were also obtained by solvolysis.²²⁵ The product in Table 11 (entry 2) has been used in a methyl (\pm)-jasmonate synthesis. Stereospecifically substituted cyclopentanes can be prepared by decarboxylation of oxabicycloheptanecarboxylic acids (Table 11, entries 3 and 4). A one-carbon ring extension of cyclic ketones is possible by first converting the ketones to β -hydroxycarboxylic acids by way of a Reformatzky reaction, and then subjecting these to a nonKolbe electrolysis (Table 11, entries 5 and 6).

The nonKolbe electrolysis of carboxylic acids can be used for a selective fragmentation, when the carbocation formed initially is stabilized in the γ -position by a hydroxy or trimethylsilyl group. This way the reaction can be used for a four-carbon ring extension (Table 12, entries 1 and 2). Furthermore it can be applied for the stereospecific construction of *cis*- or *trans*-disubstituted cyclopentenes from 6-hydroxynorbornane-2-carboxylic acids (Table 12, entries 3 and 4).

Entry	Carboxylic acid HO ₂ C H OH	Reaction conditions	Product	Yield	Yield (%) Ref.	
1		МеОН	o o	44 35	226 223	
2	CO ₂ H CO ₂ H CO ₂ H SiMe ₃	C anode, MeCN, EtOH		85	15	
3	ОСТНР	C anode, MeCN, EtOH	CHO	38	227	
4	O CO ₂ H	MeOH, C anode	SO ₂ Ph	63	227	

 Table 12
 Fragmentation of Carboxylic Acids by NonKolbe Electrolysis

2.8.7.1 Experimental Procedure for NonKolbe Electrolysis --- Preparation of Methyl 2-(t-2-Phenylsulfonyl-3-cyclopenten-r-1-yl)acetate

A solution of 0.5 g (1.69 mmol) 6-oxo-c-3-phenylsulfonylbicyclo[2.2.1]heptane-t-2-carboxylic acid in methanol (50 mL), neutralized to 100% with 1 M NaOH in methanol, was electrolyzed at graphite electrodes (anode area 15 cm²) in an undivided cell (Figure 2) with a current density of 100 mA cm⁻² at 0 °C until consumption of 4 F mol⁻¹. The solution was then evaporated to dryness, treated with a saturated NaCl solution and extracted five times with CH₂Cl₂. After drying (MgSO₄) and removal of the CH₂Cl₂, 0.3 g (1.07 mmol, 63%) methyl 2-(t-2-phenylsulfonyl-3-cyclopenten-r-1-yl) acetate was obtained as a colorless oil by column filtration (silica gel, petroleum ether: ether = 2:1). Acidification of the aqueous phase with concentrated HCl to pH 1 and threefold extraction with CH₂Cl₂ afforded 0.04 g (9%) starting acid.227

2.8.8 CONCLUSIONS

Carboxylic acids can be converted by anodic decarboxylation into radicals and/or carbocations. The reaction conditions are simple; an undivided beaker-type cell as reaction vessel, controlled current supplied from an inexpensive d.c. power supply and methanol as solvent are in most cases sufficient. A scale-up is fairly easy and the yields are in general good. By the radical pathway 1,n-diesters, -diketones, -dienes and -dihalides, chiral intermediates for synthesis, pheromones and unusual fatty acids are accessible in just a few steps. The addition of the intermediate radicals to double bonds affords additive dimers, whereby four building units, two alkyl radicals from the carboxylates and two alkenes, can be coupled in one step. Five-membered hetero- or carbo-cyclic compounds can be prepared by intramolecular addition starting from unsaturated carboxylic acids.

The cationic pathway allows the conversion of carboxylic acids into ethers and acetals. The decarboxylative elimination of vicinal diacids or β-silylcarboxylic acids in combination with cycloaddition reactions leads efficiently to cyclobutenes or cyclohexadienes. Cationic rearrangements or fragmentations initiated by anodic decarboxylation allow the synthesis of specifically substituted cyclopentanes and ring extensions by one or four carbon atoms.

2.8.9 REFERENCES

- 1. M. Faraday, Pogg. Ann., 1834, 33, 438.
- 2. H. Kolbe, Liebigs Ann. Chem., 1849, 69, 257; 1869, 113, 125; J. Prakt. Chem., 1874, 10, 89; 1875, 11, 24.
- 3. A. Wurtz, Ann. Chim. Phys., 1855, 44, 291.
- 4. A. C. Brown and J. Walker, Liebigs Ann. Chem., 1891, 261, 107.
- 5. B. C. L. Weedon, Adv. Org. Chem., 1960, 1, 1.
- A. K. Vijh and B. E. Conway, Chem. Rev., 1967, 67, 623.
 (a) L. Eberson, Electrochim. Acta, 1967, 12, 1473; (b) L. Eberson, S. Gränse and B. Olofsson, Acta Chem. 7. Scand., 1968, 22, 2462; (c) L. Eberson, Acta Chem. Scand., 1963, 17, 2004.
- 8. J. H. P. Utley, in 'Technique of Electroorganic Synthesis', ed. N. L. Weinberg, Wiley, New York, 1974, vol. 5, part 1, p. 793.
- 9. A. Hickling, Q. Rev., Chem. Soc., 1949, 3, 95.
- 10. N. L. Weinberg and H. R. Weinberg, Chem. Rev., 1968, 68, 449.
- G. E. Svadkovskaya and S. A. Voitkevich, Russ. Chem. Rev. (Engl. Transl.), 1960, 29, 161.
 H. J. Schäfer, Chem. Phys. Lipids, 1979, 24, 321.
 M. M. Baizer, Tetrahedron, 1984, 40, 935.

- 14. H. J. Schäfer, in 'Recent Advances in Electroorganic Synthesis, Proceedings of the 1st International Symposium on Electroorganic Synthesis', ed. S. Torii, Kodansha Tokyo, 1987, p. 3.
- 15. H. J. Schäfer, in 'DECHEMA-Monogr., 1985, 112, 399.
- 16. F. Fichter, in 'Organische Elektrochemie', Steinkopff, Dresden, 1942, p. 17.
- 17. M. J. Allen, in 'Organic Electrode Processes', Reinhold, New York, 1958.
- 18. C. J. Brockmann, in 'Electroorganic Chemistry', Wiley, New York, 1926, p. 23.
- 19. L. Eberson, in 'Chemistry of Carboxylic Acids and Esters', ed. S. Patai, Interscience, New York, 1969, p. 53.
- 20. L. Eberson and J. H. P. Utley, in 'Organic Electrochemistry', ed. M. M. Baizer, 2nd edn., Dekker, New York, 1983, p. 435.
- 21. S. D. Ross, M. Finkelstein and E. F. Rudd, in 'Anodic Oxidation', Academic Press, New York, 1975, p. 134 and p. 156.
- 22. L. Eberson and K. Nyberg, in 'Encyclopedia of Electrochemistry of the Elements', ed. A. J. Bard and H. Lund, Dekker, New York, 1978, vol. 12, p. 261.
- 23. S. Torii, in 'Electroorganic Syntheses, Methods and Applications, Part 1: Oxidation', VCH Publishers, Deerfield Beach, 1985, p. 51.
- 24. J. P. Coleman, R. Lines, J. H. P. Utley and B. C. L. Weedon, J. Chem. Soc., Perkin Trans. 2, 1974, 1064.

- 25. (a) T. Dickinson and W. F. K. Wynne-Jones, Trans. Faraday Soc., 1962, 58, 382, 388, 400; (b) J. Haufe and F. Beck, Chem.-Ing.-Tech., 1970, 42, 170.
- 26. Ref. 19, p. 58.
- 27. Ref. 8, p. 887.
- 28. (a) F. Beck, Electrochim. Acta, 1973, 18, 359; (b) F. Wenisch, H. Nohe, H. Hannebaum, R. K. Horn, M. Stroezel and D. Degner, AlChE Symp. Ser, 1979, 75, 14 (Cnem. Abstr., 1979, 91, 65 193); (c) BASF, Ger. Pat. 2 014 985 (197 0) (Chem. Abstr., 1972, 76, 3402h).
- 29. N. Dinh-Nguyen, Acta Chem. Scand., 1958, 12, 585.
- 30. (a) R. G. Woolford, Can. J. Chem., 1962, 40, 1846; (b) R. G. Woolford, W. Arbic and A. Rosser, Can. J. Chem., 1964, 42, 1788.
- 31. M. Y. Fioshin, Y. B. Vasil'ev and E. G. Gaginkina, Dokl. Akad. Nauk SSSR, 1960, 135, 909 (Chem. Abstr., 1962, 56, 2273d).
- 32. S. Swann, Jr., Tech. Org. Chem., 2nd Edn., 1956, 2, 385.
- 33. R. F. Garwood, Naser-ud-Din, C. J. Scott and B. C. L. Weedon, J. Chem. Soc., Perkin Trans. 1, 1973, 2714.
- 34. L. B. Rodewald and M. C. Lewis, Tetrahedron, 1971, 27, 5273.
- 35. L. S. Kanevskii and Y. B. Vasil'ev, Elektrokhimiya, 1976, 12, 1487 (Chem. Abstr., 1977, 86, 80 805).
- 36. M. P. J. Brennan and R. Brettle, J. Chem. Soc., Perkin Trans. 1, 1973, 257.
- 37. L. Rand and A. F. Mohar, J. Org. Chem., 1965, 30, 3156, 3885.
- 38. (a) E. F. Rudd, M. Finkelstein, and S. D. Ross, J. Org. Chem., 1972, 37, 1763; (b) G. Bélanger, C. Lamarre, and A. K. Vijh, J. Electrochem. Soc., 1975, 122, 46.
- 39. D. L. Muck and E. R. Wilson, J. Electrochem. Soc., 1970, 117, 1358.
- 40. G. E. Hawkes, J. H. P. Utley and G. B. Yates, J. Chem. Soc., Perkin Trans. 2, 1976, 1709.
- 41. A. Laurent, E. Laurent and M. Thomalla, C. R. Hebd. Seances Acad. Sci., Ser. C, 1972, 274, 1537.
- V. G. Gurjar, J. Appl. Electrochem., 1978, 8, 207 (Chem. Abstr., 1978, 89, 82 038t).
 F. Beck, in 'Elektroorganische Chemie', VCH, Weinheim, 1974, table 28, p. 229.
- 44. (a) C. Iwakura, F. Goto and H. Tamura, Denki Kagaku Oyobi Kogyo Butsuri Kagaku, 1980, 48, 21 (Chem. Abstr., 1980, 93, 15 655m); (b) L. A. Mirkind and G. L. Al'bertinskii, Elektrokhimiya, 1980, 16, 911 (Chem. Abstr., 1980, 93, 83 496); (c) L. A. Mirkind, G. L. Al'bertinskii and A. G. Kornienko, Elektrokhimiya, 1983, 19, 122 (Chem. Abstr., 1983, 98, 115 648).
- 45. O. J. Walker and J. Weiss, Trans. Faraday Soc., 1935, 31, 1011 and earlier papers.
- 46. (a) S. Glasstone and A. Hickling, Chem. Rev., 1939, 25, 407; (b) A. Hickling, Quart. Rev. Chem. Soc., 1949, 3, 95.
- 47. H. Hofer and M. Moest, Justus Liebigs Ann. Chem., 1902, 323, 284.
- 48. C. Schall, Z. Elektrochem., 1896, 3, 83.
- 49. Ref. 8, p. 796.
- 50. P. R. Nadebaum and T. Z. Fahidy, Elektrochim. Acta, 1972, 17, 1659.
- 51. Ref. 21, p. 139.
- 52. Ref. 20, p. 442. 53. Ref. 8, p. 886.
- 54. Ref. 43, p. 227.
- 55. G. P. Girina, M. Y. Fioshin and V. E. Kazarinov, Elektrokhimiya, 1965, 1, 478 (Chem. Abstr., 1965, 63, 9453h).
- 56. (a) B. E. Conway and M. Dzieciuch, Proc. Chem. Soc., London, 1962, 121 (Chem. Abstr., 1963, 61, 308b); (b) B. E. Conway and M. Dzieciuch, Can. J. Chem., 1963, 41, 38, 55.
- 57. (a) B. E. Conway, and M. Dzieciuch, Can. J. Chem., 1963, 41, 21; (b) S. N. Shukla and O. J. Walker, Trans. Faraday Soc., 1931, 27, 722.
- 58. R. M. Cervino, W. E. Triaca and A. J. Arvia, J. Electroanal. Chem. Interfacial Electrochem., 1984, 172, 255.
- 59. A. Bewick and D. J. Brown, Electrochim. Acta, 1976, 21, 979.
- 60. F. Bruno and J. E. Dubois, Electrochim. Acta, 1972, 17, 1161.
- 61. B. E. Conway and A. K. Vijh, Electrochim. Acta, 1967, 12, 102.
- 62. R. Pistorius, Ph.D. Thesis, University of Göttingen, 1974
- 63. L. Eberson and G. Ryde-Petterson, Acta Chem. Scand., 1973, 27, 1159.
- 64. L. Eberson, K. Nyberg and R. Servin, Acta Chem. Scand., Ser. B, 1976, 30, 906.
- 65. G. E. Hawkes, J. H. P. Utley and G. B. Yates, J. Chem. Soc., Perkin Trans. 2, 1976, 1709.
- 66. J. H. P. Utley and G. B. Yates, J. Chem. Soc., Perkin Trans. 2, 1978, 395.
- 67. Ref. 6, p. 629.
- 68. L. Eberson, Acta Chem. Scand., 1959, 13, 40.
- 69. L. Eberson and B. Sandberg, Acta Chem. Scand., 1966, 20, 739.
- 70. L. Eberson, Acta Chem. Scand., 1960, 14, 641.
- 71. L. Eberson and S. Nilsson, Acta Chem. Scand., 1968, 22, 2453.
- 72. L. Eberson, Acta Chem. Scand., 1963, 17, 1196.
- 73. (a) D. Lelandais and M. Chkir, Tetrahedron Lett., 1974, 36, 3113; (b) M. Chkir, D. Lelandais and C. Bacquet, Can. J. Chem., 1981, 59, 945.
- 74. S. Motoki and T. Odaka, Nippon Kagaku Zasshi, 1955, 76, 930.
- 75. J. Tsuji, M. Kaito, T. Yamada and T. Mandai, Bull Chem. Soc. Jpn., 1978, 51, 1915.
- 76. R. G. Woolford, J. Soong and W. S. Lin, Can. J. Chem., 1967, 45, 1837.
- 77. B. E. Conway and M. Dzieciuch, Can. J. Chem., 1963, 41, 21.
- J. P. Waefler and P. Tissot, *Electrochim. Acta*, 1978, 23, 899.
 F. L. M. Pattison, J. B. Stothers and R. G. Woolford, J. Am. Chem. Soc., 1956, 78, 2255.
- 80. M. Y. Fioshin, A. P. Tomilov, I. A. Avrutskaya, L. I. Kazakova, N. T. Eskin and G. A. Gromova, Zh. Vses. Khim. Ova., 1963, 8, 600 (Chem. Abstr., 1964, 60, 3992h).
- 81. K. Saotome, H. Komoto and T. Yamazaki, Bull. Chem. Soc. Jpn., 1966, 39, 480.

- 82. V. V. Berenblit, E. S. Panitkova, D. S. Rondarev, V. P. Sass and S. V. Sokolov, Zh. Prikl. Khim., 1974, 47, 2427 (Chem. Abstr., 1975, 82, 78 455).
- 83. A. I. Levin, O. N. Chechina and S. V. Sokolov, Zh. Obshch. Khim., 1965, 35, 1778 (Chem. Abstr., 1966, 64, 1943a).
- 84. P. L. Coe, S. F. Sellers, J. C. Tatlow, H. C. Fielding and G. Whittaker, J. Chem. Soc., Perkin Trans. 1, 1983, 1957.
- 85. Y. Kobayashi, T. Chiba, K. Yokota and Y. Takata, Hokkaido Daigaku Kogakubu Kenkyu Hokoku, 1979, 45 (Chem. Abstr., 1980, 92, 163 547).
- 86. K. Mori, Nippon Kagaku Zasshi, 1961, 82, 1375 (Chem. Abstr., 1962, 57, 14 929e).
- 87. L. Rand and A. F. Mohar, J. Org. Chem., 1965, 30, 3885.
- 88. P. J. Bunyan and D. H. Hey, J. Chem. Soc., 1962, 1360.
- 89. G. Stork, A. Meisels and J. E. Davies, J. Am. Chem. Soc., 1963, 85, 3419.
- 90. E. J. Corey and R. R. Sauers, J. Am. Chem. Soc., 1959, 81, 1739.
- 91. N. Rabjohn and G. W. Flasch, Jr., J. Org. Chem., 1981, 46, 4082.
- S. Kobayashi, Jpn. Pat. 7 686 401 (1976) (Chem. Abstr., 1977, 86, 120 766b). 92.
- 93. L. Zhang, X. Pan, M. Zhao, H. Tang and B. Xu, Hua Hsueh Tung Pao, 1988, 41 (Chem. Abstr., 1988, 108, 212 438x).
- 94. K. Kimura, S. Horie, I. Minato and Y. Odaira, Chem. Lett., 1973, 1209; see also T. D. Binns, R. Brettle and G. B. Cox, J. Chem. Soc. C, 1968, 584.
- 95. H. Quast and J. Christ, Liebigs Ann. Chem., 1984, 1180.
- 96. For a good summary, see ref. 43, p. 233.
- 97. (a) F. Beck, Electrochim. Acta, 1973, 18, 359; (b) F. Beck, W. Himmele, J. Haufe and A. Brunold (BASF), Ger. Pat. 1 643 693 (1976) (Chem. Abstr., 1977, 86, 54 997).
- 98. (a) Y. B. Vassil'ev, E. P. Kovsman and G. N. Freidlin, Electrochim. Acta, 1982, 27, 953; (b) Y. B. Vassil'ev, E. P. Kovsman and G. N. Freidlin, Electrochim. Acta, 1982, 27, 937.
- 99. (a) K. Yamataka, Y. Matsuoka and T. Isoya (Asahi Chem.), Ger. Pat. 2 830 144 (1979) (Chem. Abstr., 1979, 90, 112 142n); (b) K. Yamataka, T. Isoya and C. Kawamura (Asahi Chem.), Ger. Pat. 3 019 537 (1980) (Chem. Abstr., 1981, 94, 73 666x).
- 100. D. Degner, Top. Curr. Chem., 1988, 148, 11, 25.
- 101. J. M. Mersereau (Uniroyal), Ger. Pat. 2 022 341 (1970) (Chem. Abstr., 1971, 74, 54 404f); see also A. Gozlan, G. Agam, S. Vardi and A. Zilkha, Eur. Polym. J., 1984, 20, 759, 1199 (Chem. Abstr., 1984, 101, 152 443p; 1985, 102, 114 066f); for oligomers see A. A. Adamov, E. P. Kovsman, G. N. Freidlin and G. A. Tarkhanov, Elektrokhimiya, 1975, 11, 1773 (Chem. Abstr., 1976, 84, 58 558). 102. M. S. Toy, J. Electrochem. Soc., 1967, 114, 1042.
- 103. G. Cauquis and B. Haemmerlé, Bull. Soc. Chim. Fr., 1970, 183.
- 104. A. Wohl and H. Schweitzer, Chem. Ber., 1906, 39, 890.
- 105. J. Knolle and H. J. Schäfer, Electrochim Acta, 1978, 23, 5.
- 106. A. K. Yadav, A. Jain and R. A. Misra, Electrochim. Acta, 1982, 27, 535.
- 107. (a) R. F. Garwood, Naser-ud-Din, C. J. Scott and B. C. L. Weedon, J. Chem. Soc., Perkin Trans. 1, 1973, 2714; (b) J. Knolle, Ph.D. Thesis, University of Münster, 1976.
- 108. J. E. Baldwin, J. Chem. Soc., Chem. Commun, 1976, 734.
- 109. M. Huhtasaari, H. J. Schäfer and H. Luftmann, Acta Chem. Scand., Ser. B, 1983, 37, 537.
- 110. L. Mandell, R. F. Daley and R. A. Day, Jr., J. Org. Chem., 1976, 41, 4087.
- C. Adams, N. Jacobsen and J. H. P. Utley, J. Chem. Soc., Perkin Trans. 2, 1978, 1071.
 D. G. Bounds, R. P. Linstead and B. C. L. Weedon, J. Chem. Soc., 1953, 2393.
- 113. A. Weiper, Ph.D. Thesis, University of Münster, 1991.
- 114. D. A. White, Org. Synth., 1981, 60, 1 (Chem. Abstr., 1982, 96, 51 775w).
- 115. (a) M. Feldhues and H. J. Schäfer, Tetrahedron, 1985, 41, 4195, 4213; 1986, 42, 1285; (b) R. Lomölder and H. J. Schäfer, Angew. Chem., 1987, 99, 1282; Angew. Chem., Int. Ed. Engl., 1987, 26, 1253.
- 116. P. F. Levy, J. E. Sanderson and L. K. Cheng, J. Electrochem. Soc., 1984, 131, 773.
- 117. B. W. Baker, R. W. Kierstead, R. P. Linstead and B. C. L. Weedon, J. Chem. Soc., 1954, 1804.
- 118. I. I. Krasavtsev, Ukr. Khim. Zh., 1980, 46, 776 (Chem. Abstr., 1980, 93, 139 974j).
- 119. R. N. Renaud and D. E. Sullivan, Can. J. Chem., 1972, 50, 3084.
- Naser-ud-Din, Pak. J. Sci. Ind. Res., 1976, 19, 132 (Chem. Abstr., 1978, 88, 120 537).
 F. F. Rawlings, G. W. Thiessen, T. G. Lee, M. J. Murray, Jr. and J. L. Seager, J. Electrochem. Technol., 1964, 2, 217 (Chem. Abstr., 1964, 61, 9172).
- 122. R. Brettle and D. W. Latham, J. Chem. Soc. C, 1968, 906.
- 123. S. Motoki and T. Odaka, Nippon Kagaku Zasshi, 1956, 77, 163.
- W. Fuchs and E. Barnetzky, Fette, Seifen, Anstrichm, 1955, 57, 675. 124.
- 125. W. S. Greaves, R. P. Linstead, B. R. Shephard, S. L. S. Thomas and B. C. L. Weedon, J. Chem. Soc., 1950, 3326.
- 126. S. Ställberg-Stenhagen, Ark. Kemi, 1951, 2, 95.
- K. Mislow and I. V. Steinberg, J. Am. Chem. Soc., 1955, 77, 3807.
 M. Romañuk, L. Streinz and F. Sorm, Collect. Czech. Chem. Commun., 1972, 37, 1755.
- 129. G. Odham, B. Pettersson and E. Stenhagen, Acta Chem. Scand., Ser. B., 1974, 28, 36.
- R. P. Linstead, J. C. Lunt and B. C. L. Weedon, J. Chem. Soc., 1951, 1130.
 (a) G. A. Tember, Z. I. Getmanskaya and P. R. Nichikova, Zh. Prikl. Khim., 1974, 47, 477 (Chem. Abstr., 1974, 80, 120 168j); (b) B. F. Taikov, E. M. Novakovskii, V. P. Zhelkovskaya, V. N. Shadrova and P. K. Shcherbik, Khim. Tverd. Topl. (Moscow), 1981, 61 (Chem. Abstr., 1982, 96, 51 767v); (c) Okamura Oil Mill, Jpn. Pat. 59 200 778 (Chem. Abstr., 1985, 102, 228 323t). L. Eberson, J. Org. Chem., 1962, 27, 3706.
- 132.
- 133. Y. Okida (Okamura Oil Mill), Jpn. Pat. 61 147 890 (Chem. Abstr., 1987, 106, 127 947).

- 134. (a) M. Takahashi, K. Osawa, J. Ueda and K. Okada, Yakugaku Zasshi, 1976, 96, 137 (Chem. Abstr., 1976, 84, 135 228k); (b) U. Jensen, Ph.D. Thesis, University of Münster, 1981; (c) Naser-ud-Din, Pak. J. Sci. Ind. Res., 1976, 19, 132 (Chem. Abstr., 1978, 88, 120 571).
- W. Seidel and H. J. Schäfer, Chem. Ber., 1980, 113, 3898. 135.
- 136. Ref. 9, Table 6.11.
- V. V. Berenblit, A. Y. Zapevalov, E. S. Panitkova, V. S. Plashkin, D. S. Rondarev, V. P. Sass and S. V. Sokolov, Zh. Org. Khim., 1979, 15, 1417 (Chem. Abstr., 1979, 91, 192 779). 137.
- R. N. Renaud and P. J. Champagne, Can. J. Chem., 1975, 53, 529. 138.
- Y. Suhara and S. Miyazaki, Bull. Chem. Soc. Jpn., 1969, 42, 3022. 139.
- 140. V. M. Andreev, S. G. Polyakova, Z. S. Khrustova, V. I. Bazhulina and V. V. Smirnova, USSR Pat. 789 509 (1980) (Chem. Abstr., 1981, 95, 6513c).
- 141. V. M. Andreev, S. G. Polyakova, V. I. Bazhulina, V. V. Smirnova and V. P. Gorbunkova, Zh. Org. Khim., 1981, 17, 86 (Chem. Abstr., 1981, 94, 208 308e).
- H. J. Schäfer and D. Hermeling, unpublished results.
 Asahi Glass Co., Jpn. Pat. 57 200 576 (1982) (Chem. Abstr., 1983, 98, 206 529).
- Asahi Chem. Industry Co., Jpn. Pat. 58 161 784 (198?) (Chem. Abstr., 1984, 100, 128 859). 144.
- 145. M. Stoll, Helv. Chim. Acta, 1951, 34, 1817.
- 146. S. Motoki and Y. Yamada, Nippon Kagaku Zasshi, 1960, 81, 665 (Chem. Abstr., 1962, 56, 396).
- 147. R. P. Linstead, B. R. Shephard, B. C. L. Weedon and L. C. Lunt, J. Chem. Soc., 1953, 1538.
- 148. (a) R. P. Linstead, B. C. L. Weedon and B. Wladislaw, J. Chem. Soc., 1955, 1097; (b) D. G. Bounds, R. P. Linstead and B. C. L. Weedon, J. Chem. Soc., 1954, 4219.
- 149. K. Kihira, A. K. Batta, E. H. Mosbach and G. Salen, J. Lipid Res., 1979, 20, 421 (Chem. Abstr., 1979, 91, 175 606).
- 150. (a) B. W. Baker, R. P. Linstead and B. C. L. Weedon, J. Chem. Soc., 1955, 2218; (b) M. Takahashi (Sanei Chem. Ind.), Jpn. Pat. 76 113 835 (1976) (Chem. Abstr., 1977, 86, 155 359).
- 151. R. P. Linstead, J. C. Lunt, B. C. L. Weedon and B. R. Shephard, J. Chem. Soc., 1952, 3621.
- 152. M. Takahashi, (Sanei Chem. Ind.), Jpn. Pat. 76 113 836 (1976) (Chem. Abstr., 1977, 86, 139 611).
- D. G. Bounds, R. P. Linstead and B. C. L. Weedon, J. Chem. Soc., 1954, 448.
 G. W. Gribble, J. K. Sanstead and J. W. Sullivan, J. Chem. Soc., Chem. Commun., 1973, 735.
- 155. W. Seidel, Ph.D. Thesis, University of Münster, 1975
- 156. A. K. Yadav and P. Tissot, Helv. Chim. Acta, 1984, 67, 1698.
- 157. H. Klünenberg and H. J. Schäfer, Angew. Chem., 1978, 90, 48; Angew. Chem., Int. Ed. Engl., 1978, 17, 47.
- 158. U. Jensen and H. J. Schäfer, Chem. Ber., 1981, 114, 292.
- 159. H. J. Bestmann, K. Roth, K. Michaelis, O. Vostrowsky, H. J. Schäfer and R. Michaelis, Liebigs Ann. Chem., 1987, 417.
- 160. W. Seidel and H. J. Schäfer, Chem. Ber., 1980, 113, 451; see also U. Jensen-Korte and H. J. Schäfer, Liebigs Ann. Chem., 1982, 1532.
- 161. J. Takahashi, K. Mori and M. Matsui, Agric. Biol. Chem., 1979, 43, 1605 (Chem. Abstr., 1979, 91, 174 752).
- 162. W. Seidel, J. Knolle and H. J. Schäfer, Chem. Ber., 1977, 110, 3544.
- 163. J. Knolle, Ph.D. Thesis, University of Münster, 1976.
- 164. C. Wittenbrink, Ph.D. Thesis, University of Münster, 1977.
- 165. J. Knolle and H. J. Schäfer, Angew. Chem., 1975, 87, 777; Angew. Chem., Int. Ed. Engl., 1975, 14, 758.
- 166. (a) R. P. Linstead, J. C. Lunt and B. C. L. Weedon, J. Chem. Soc., 1950, 3331; (b) R. P. Linstead, J. C. Lunt and B. C. L. Weedon, J. Chem. Soc., 1951, 1130.
- R. F. Nutt, R. G. Strachan, D. F. Veber and F. W. Holly, J. Org. Chem., 1980, 45, 3078. 167.
- 168. D. Seebach and P. Renaud, Helv. Chim. Acta, 1985, 68, 2342.
- 169. O. Ceder and H. G. Nilsson, Acta Chem. Scand., Ser. B, 1977, 31, 189.
- 170. H. J. Schäfer and R. Pistorius, Angew. Chem., 1972, 84, 893; Angew. Chem., Int. Ed. Engl., 1972, 11, 841.
- 171. R. V. Lindsey, Jr. and M. L. Peterson, J. Am. Chem. Soc., 1959, 81, 2073.
- 172. M. Y. Fioshin, A. I. Kamneva, L. A. Mirkind, L. A. Salmin and A. G. Kornienko, Neftekhimiya, 1962, 2, 557 (Chem. Abstr., 1963, 58, 11 205b).
- 173. M. Y. Fioshin, A. I. Kamneva, L. A. Mirkind, A. G. Kornienko and L. A. Salmin, Khim. Prom. (Moscow), 1966, 42, 804 (Chem. Abstr., 1967, 66, 43 109x).
- 174. M. Y. Fioshin, L. A. Mirkind, L. A. Salmin and A. G. Kornienko, Zh. Obshch. Khim., 1965, 10, 238 (Chem. Abstr., 1965, 63, 15 858c).
- V. P. Nechiporenko, K. G. Bogoslovskii, N. A. Novikov, N. I. Guigina and L. A. Mirkind, Zh. Org. Khim., 175. 1984, 20, 1799 (Chem. Abstr., 1985, 102, 61 763).
- 176.
- S. Goldschmidt and E. Stöckl, Chem. Ber., 1952, 85, 630. Y. B. Vassil'ev, V. S. Bagotzky, E. P. Kovsman, V. A. Grinberg, L. S. Kanevsky and V. R. Polishchyuk, 177. Electrochim. Acta, 1982, 27, 919.
- 178. Y. B. Vassil'ev, B. M. Lotvin and V. A. Grinberg, Elektrokhimiya, 1981, 17, 1252 (Chem. Abstr., 1981, 95, 211 885).
- 179. W. B. Smith and Y. Hyon Yuh, Tetrahedron, 1968, 24, 1163.
- 180. M. Chkir and D. Lelandais, J. Chem. Soc., Chem. Commun., 1971, 1369.
- 181. K. G. Karapetyan, A. A. Bezzubov, L. S. Kanevskii, A. M. Skundin and Y. B. Vassil'ev, Elektrokhimiya, 1976, 12, 1623 (Chem. Abstr., 1977, 86, 62 705t).
- 182. R. N. Renaud and P. J. Champagne, Can. J. Chem., 1979, 57, 990.
- 183. K. G. Bogoslovskii, L. A. Mirkind and N. B. Kondrikov, Zh. Vses. Khim. Ova., 1986, 31, 470 (Chem. Abstr., 1986, 105, 215 678z).
- 184. P. J. Champagne and R. N. Renaud, Can. J. Chem., 1980, 58, 1101.
- 185. K. Uneyama and K. Ueda, Chem. Lett., 1988, 853.
- 186. L. Stork and H. J. Schäfer, unpublished results.

- 187. C. J. Brookes, P. L. Coe, D. M. Owen, A. E. Pedler and J. C. Tatlow, J. Chem. Soc., Chem. Commun., 1974, 323; see also C. J. Brookes, P. L. Coe, A. E. Pedler and J. C. Tatlow, J. Chem. Soc., Perkin Trans. 1, 1978, 202
- 188. N. Müller, J. Org. Chem., 1983, 48, 1370.
- 189. R. N. Renaud, P. J. Champagne and M. Savard, Can. J. Chem., 1979, 57, 2617.
- 190. R. N. Renaud, C. J. Stephens and D. Bérubé, Can. J. Chem., 1982, 60, 1687.
- 191. W. B. Smith and H. G. Gilde, J. Am. Chem. Soc., 1959, 81, 5325; 1961, 83, 1355.
- 192. W. B. Smith and H. G. Gilde, J. Am. Chem. Soc., 1960, 82, 659.
- 193. W. B. Smith and D. T. Manning, J. Polym. Sci., 1962, 59, S45 (Chem. Abstr., 1962, 57, 10 027).
- 194. (a) Z. Ogumi, Z. Takehara and S. Yoshizawa, Bull. Chem. Soc. Jpn., 1976, 49, 841, 2883; (b) S. Yoshizawa, Z. Takehara, Z. Ogumi and C. Najai, Denki Kagaku Oyobi Kogyo Butsuri Kagaku, 1972, 40, 724 (Chem. Abstr., 1973, 79, 19 194u).
- 195. J. H. P. Utley and R. J. Holman, Electrochim. Acta, 1976, 21, 987.
- 196. V. A. Grinberg, V. P. Polishchuk, L. S. Kanevskii and L. S. German, Tezisy Dokl.-Vses. Soveshch. Polyarogr., 7th, 1978, 53 (Chem. Abstr., 1980, 93, 157 798).
- 197. H. J. Schäfer, M. Huhtasaari and L. Becking, Angew. Chem., 1984, 96, 995; Angew. Chem., Int. Ed. Engl., 1984, 23, 980.
- 198. L. Becking and H. J. Schäfer, Tetrahedron Lett., 1988, 29, 2797.
- 199. G. Dralle, Ph.D. Thesis, University of Münster, 1990.
- 200. L. Becking and H. J. Schäfer, Tetrahedron Lett., 1988, 29, 2801.
- 201. (a) B. Giese, in 'Radicals in Organic Synthesis', Pergamon Press, Oxford, 1986, p. 141; (b) D. P. Curran, Synthesis, 1988, 417, 489.
- 202. (a) C. Walling, in 'Free Radicals in Solution', Wiley, New York, 1957, p. 581; (b) C. G. Overberger and P. Kabasakalian, J. Am. Chem. Soc., 1957, 79, 3182; (c) E. J. Corey, N. L. Bauld, R. T. La Londe, J. Casanova, Jr. and E. T. Kaiser, J. Am. Chem. Soc., 1960, 82, 2645.
- 203. W. J. Koehl, Jr., J. Am. Chem. Soc., 1964, 86, 4686.
- 204. N. Sato, T. Sekine and K. Sugino, J. Electrochem. Soc., 1968, 115, 242.
- 205. K. Kase, N. Sato and T. Sekine, Denki Kagaku Oyobi Kogyo Butsuri Kagaku, 1983, 51, 749 (Chem. Abstr., 1984, 100, 120 310f).
- 206. (a) P. G. Gassman and F. V. Zalar, J. Am. Chem. Soc., 1966, 88, 2252; (b) S. D. Ross and M. Finkelstein, J. Org. Chem., 1969, 34, 2923.
- 207. J. G. Traynham, E. E. Green and R. L. Frye, J. Org. Chem., 1970, 35, 3611.
- 208. J. A. Van Zorge, J. Strating and H. Wynberg, Recl. Trav. Chim. Pays-Bas, 1970, 89, 781.
- 209. P. C. Arora and R. G. Woolford, Can. J. Chem., 1971, 49, 2681.
- 210. (a) P. Renaud and D. Seebach, Helv. Chim. Acta, 1986, 69, 1704; (b) P. Renaud and D. Seebach, Synthesis, 1986, 424; (c) T. Iwasaki, H. Horikawa, K. Matsumoto and M. Miyoshi, Bull. Chem. Soc. Jpn., 1979, **52**, 826. J. Nokami, M. Kawada, R. Okawara, S. Torii and H. Tanaka, Tetrahedron Lett., 1979, 1045.
- 211.
- T. Iwasaki, H. Horikawa, K. Matsumoto and M. Miyoshi, J. Org. Chem., 1979, 44, 1552. 212.
- 213. M. Mori, K. Kagechika, K. Tohjima and M. Shibasaki, Tetrahedron Lett., 1988, 29, 1409.
- 214. D. Lelandais and M. Chkir, C.R. Hebd. Seances Acad. Sci., Ser. C., 1975, 281, 731.
- 215. J. Slobbe, J. Chem. Soc., Chem. Commun., 1977, 82.
- 216. T. Iwasaki, H. Horikawa, K. Matsumoto and M. Miyoshi, Tetrahedron Lett., 1978, 4799.
- 217. D. Hermeling and H. J. Schäfer, Chem. Ber., 1988, 121, 1151
- 218. H. H. Westberg and H. J. Dauben, Jr., Tetrahedron Lett., 1968, 5123.
- 219. R. C. Kelly and I. Schletter, J. Am. Chem. Soc., 1973, 95, 7156.
- 220. S. Torii, H. Tanaka, and T. Mandai, J. Org. Chem., 1975, 40, 2221.
- T. Akiyama, T. Fujii, H. Ishiwari, T. Imagawa and M. Kawanisi, Tetrahedron Lett., 1978, 2165; 1979, 1691; 221. J. Org. Chem., 1980, 45, 2005.
- 222. U. Müller, Ph.D. Thesis, University of Münster, 1986.
- 223. M. Bitenc, Ph.D. Thesis, University of Münster, 1987.
- 224. L. Rand and C. S. Rao, J. Org. Chem., 1968, 33, 2704.
- 225. S. Winstein and D. Trifan, J. Am. Chem. Soc., 1952, 74, 1147, 1154.
- 226. P. S. Wharton, G. A. Hiegel and R. V. Coombs, J. Org. Chem., 1963, 28, 3217.
- 227. R. Michaelis, U. Müller and H. J. Schäfer, Angew. Chem., 1987, 99, 1049; Angew. Chem., Int. Ed. Engl., 1987, 26, 1026.

2.9 Oxidative Coupling of Phenols and Phenol Ethers

DONALD A. WHITING University of Nottingham, UK

2.9.1 INTRODUCTION	659
2.9.1.1 General	659
2.9.1.2 Mechanistic Possibilities	660
2.9.1.2.1 Coupling of aryloxy radicals	660
2.9.1.2.2 Radical substitution	661
2.9.1.2.3 Heterolytic substitution	661
2.9.1.2.4 Concerted coupling and electron transfer	661
2.9.1.2.5 Postoxidative coupling	662
2.9.1.2.6 Radical cation reactions	662
2.9.1.3 Product Types and Chapter Organization	663
2.9.2 SYNTHESIS OF BIARYLS AND RELATIVES	663
2.9.2.1 Phenol-Phenol C-C Coupling	663
2.9.2.2 Biaryls Formed in Phenol Ether Couplings	668
2.9.2.3 Biaryls Formed by Aryl Radical Insertion	677
2.9.3 SYNTHESIS OF SPIRODIENONES	679
2.9.3.1 Spirodienones via Phenol–Phenol C—C Coupling	679
2.9.3.2 Cyclohexadienones via Phenol Ether Coupling	683
2.9.3.3 Spirodienones via Aryl Radical Insertion	686
2.9.4 FORMATION OF CARBON-OXYGEN BONDS	686
2.9.4.1 Synthesis of Aryl Ethers	686
2.9.4.2 Spiroethers	688
2.9.4.3 Aryl Alkyl Ethers via C—O Coupling	690
2.9.5 C—C COUPLING THROUGH ARYL CONJUGATED DOUBLE BONDS	692
2.9.6 MISCELLANEOUS OXIDATIVE COUPLINGS	696
2.9.6.1 Couplines involving Cycloadditions	696
2.9.6.2 Coupling at Benzylic Carbon	698
2.9.6.3 C-N Bond Formation	699
2.9.7 REFERENCES	700

2.9.1 INTRODUCTION

2.9.1.1 General

The ready oxidation of phenols to dimeric products has been well known for more than a century. The invaluable survey of the area by Musso¹ lists over 20 papers published before 1900 on, for example: the

Coupling Reactions

formation of the dilactone (1) from gallic acid² and its derivatives (1871);³ the production of both parapara and ortho-ortho coupled dimers from 1-naphthol (1873);⁴ the synthesis of the diphenoquinone (2).⁵ in 97% yield, from 2,5-dimethoxyphenol (1878); and the preparation of 2,2'-dihydroxybiphenyl from phenol itself (1878).⁶ Reagents used in such early work include iron(III) salts, potassium ferricyanide, oxygen and the halogens; electrochemical methods were also known, and enzyme-catalyzed reactions were reported in, for example, the oxidative dimerization of eugenol (1896).⁷ Extensive chemical investigations into this major reaction continued this century and further momentum was gained from the recognition of the role of oxidative coupling in biogenesis, signalled by the important and influential papers of Barton and Cohen.⁸ and of Erdtman and Wachtmeister.⁹ Subsequent biosynthetic investigations have confirmed the significance of the title reaction in the in vivo formation of many aromatic natural products including alkaloids (Battersby¹ estimated that such coupling was involved in ca. 10% of known alkaloids), lignans, lignin, tannins, and plant and insect pigments. Many biomimetic studies of oxidative phenolic coupling were made, either for synthetic ends or to demonstrate possibilities in biological processes. Low chemical yields were often reported in such work, particularly in certain well-known alkaloid cases, and attention was turned to new reagents, e.g. manganese(III), vanadium(V) and thallium(III), which generally proved more effective. Further, oxidants such as vanadium(V) were found to effect coupling of phenol ethers, and interest was revived in the anodic oxidation of such aromatic substrates. The oxidation of phenol ethers has proved a valuable synthetic process, often more predictable and higher yielding than the corresponding phenol oxidations, and this reaction has been employed in recent years in a number of notable natural product syntheses.

A number of good reviews of the oxidation of phenols have appeared;^{1,10–19} the most recent covers the literature to mid-1980. These offer historical perspective and coverage of the biosynthetic aspects of the area, both inappropriate in the present article. This chapter focuses entirely on synthetic uses of the reaction and sets out: (i) to survey the main types of reaction stemming from oxidation of phenols and phenol ethers; and (ii) to offer selected examples of reactions showing satisfactory yield and regioselectivity.



2.9.1.2 Mechanistic Possibilities

A wide variety of mechanistic pathways^{16,20} present themselves for consideration, and since awareness of these ramifications aids synthetic planning they are outlined here. The major variables are as follows. (i) Substrate — either an un-ionized phenol or a phenolate anion may be oxidized, leading to different intermediates. Also the products after coupling may be new substrates for oxidation. (ii) Reagent — both homogeneous and heterogeneous reagents are employed covering a range of oxidation potential. Metal complexes are often implicated, which sometimes appear to be precipitated. Both inner-sphere and outersphere electron transfer processes can be involved. (iii) Pathway — overall two electrons and two protons are removed in a typical oxidation, and various sequences are thus possible. Two single-electron steps may occur, or one two-electron step, leading to radical or ionic processes. Species with carbonmetal or oxygen-metal bonds may intervene. The route may involve oxidation of one phenolic nucleus to quinone or quinone methide followed by a nonoxidative coupling step.

The main reaction pathways thus available for this process are summarized in the following schemes. For an excellent full discussion see ref. 20.

2.9.1.2.1 Coupling of aryloxy radicals

Scheme 1 illustrates this mechanism, which is the most generally accepted and widely discussed, for the *para-para* self-coupling of a simple phenol. Oxidation to the radical (3) may proceed from the phenol or phenolate anion according to pH, *etc*. The formation of such radicals is well attested, for example

by ESR,²¹⁻²³ and it has been shown²⁴ that the subsequent dimerization fits a diffusion-controlled model. The coupling step $(3) \rightarrow (4)$ may be reversible, and in the absence of acid the enolization of (4) to produce biphenyl may be slow and rate determining.



Scheme 1

2.9.1.2.2 Radical substitution

While the above mechanism certainly operates in many systems, it is not universal. If each radical is surrounded by unoxidized phenol or phenolate anion molecules, then aromatic substitution becomes possible, and this becomes more likely in intramolecular cases where two units of different reduction potential are held in proximity. Scheme 2 shows the likely pathway with radical insertion into a phenoxide and the second one-electron transfer following coupling, and such a mechanism has been shown to operate in the oxidation of 2,3,4'-trihydroxybenzophenone to 2,6-dihydroxyxanthone (ref. 20, p. 114).



2.9.1.2.3 Heterolytic substitution

Scheme 3 shows that a second one-electron oxidation of an aryloxy radical yields an aryloxonium species, which could lead to a coupled product by electrophilic substitution. Although a less probable reaction course because of the energetics for two successive one-electron transfers, it cannot be discounted, since such transfers have been observed by polarography. Also phenoxy cations have been isolated as fluoroborate salts, and react rapidly with nucleophiles. Alternatively a single two-electron transfer to a suitable metal ion can be envisaged (Scheme 3), also leading to a cation. In this case no radical intermediates are involved. 2,3-Dichloro-2,6-dicyanobenzoquinone is thought to effect phenolic coupling by hydride transfer to an aryloxonium species.²⁵

2.9.1.2.4 Concerted coupling and electron transfer

Scheme 4 shows the ionic substitution of a phenol by a metal phenolate compound, by a concerted mechanism. Although similar to Scheme 3, a phenoxonium ion is avoided, with probable energetic advantages: thallium(III) oxidations of phenols may follow such a path.



Scheme 4

2.9.1.2.5 Postoxidative coupling

In some instances it has been demonstrated that oxidation is complete before coupling occurs, the second step thus being nonoxidative. Such cases (Scheme 5) involve oxidation to a quinone or a quinone methide whose further reactions may involve nucleophilic addition, *e.g.* by phenol, or cycloaddition. Examples are given below.



2.9.1.2.6 Radical cation reactions

One-electron oxidation of an aryl ether, for example at the anode, gives rise to a radical cation whose fate may be radical coupling (shown as dimerization) or substitution into a neutral phenol ether; both paths are shown in Scheme 6 and converge to a biaryl product. Other products are possible if coupling at a quaternary center takes place. Mechanisms of this type must operate for the important couplings of phenol ethers with phenol ethers, and phenols with phenol ethers; possibly they should not be neglected for phenol-phenol coupling also, in certain cases.



2.9.1.3 Product Types and Chapter Organization

Inter- or intra-molecular carbon-carbon coupling of phenols at two unsubstituted ring carbons leads to the corresponding biaryl. The product may then be further oxidized, for example to a diphenoquinone. Oxygen-unsubstituted ring carbon coupling gives an aryl ether. If coupling involves a substituted ring site then a cyclohexadienone results, which in intramolecular cases will be a spirodienone. Further reactions, for example with nucleophiles, may ensue.

In phenols with *ortho* or *para* unsaturated side chains, coupling reactions may occur outside the aryl ring, and may form new carbon-carbon or carbon-oxygen bonds. Benzylic radicals may also be generated, perhaps by hydrogen atom abstraction, and lead to coupling at this position. Finally the products arising from quinones, quinone ketals, or quinone methides may dominate.

The survey of reactions presented here is divided first into the main product types above. Thus Sections 2.9.2 and 2.9.3 deal with the synthesis of biaryls and spirodienones respectively, by C—C bond formation. Section 2.9.4 covers the preparation of ethers through C—O bond making. The coupling of phenols through side chain sites is dealt with in Section 2.9.5. Section 2.9.6 contains a miscellany; post-oxidative cycloaddition reactions, coupling through benzyl radicals, and some examples of C—N bond forming. Within sections, material is divided firstly by reactant type and secondly by reagent type.

2.9.2 SYNTHESIS OF BIARYLS AND RELATIVES

2.9.2.1 Phenol–Phenol C—C Coupling

This is the best known and most quoted chemistry in this area. Oxidation of a phenol as in Scheme 7, with loss of one proton and one electron, provides an aryloxy radical, which may dimerize: three modes of C---C union are possible, and two modes of C---O bonding. Spin density is greatest, in general, in the *para* position, but coupling is reversible (before intermediate dienone phenolization) and product ratios depend on pH, temperature, concentration and oxidant, for a given phenol.²⁶ Mixtures of isomeric dimers may, and often do, result. If different phenols are present then potential products are numerous. A survey of the literature indicates that reasonable chemospecificity is most likely observed either in cases of phe-

Coupling Reactions

nol self-coupling where one or two of the ortho/para sites are substituted, or in intramolecular examples. The intermolecular coupling of different phenols can also be successful, but often involves a polyhydroxy compound which can be oxidized to a quinone. The coupling step is then nonoxidative, involving addition of phenol to quinone. The feasibility of this reaction is shown by the example in Scheme 8; the initial product is oxidized to a new quinone under the reaction conditions.²⁷ o, o-Disubstituted phenols (5) give excellent yields of para-para coupled products, e.g. the formation of (2). Essentially quantitative yields of the diphenoquinone (6), with R = various alkyl groups, were obtained from the corresponding phenol (5) by refluxing in benzene with silver carbonate-celite;²⁸ conversion of (5; $R = Bu^{t}$) to (6; R =But) was also quantitative, and complete in a few minutes at room temperature with hydrogen hexacyanoferrate(III) in methanol;²⁹ and (6; R = Me) was prepared in 92% yield using triphenylbismuth carbonate.³⁰ Many other oxidants have been investigated,³¹ and for (5; R = Me) the biphenol (7) could be obtained with a silver nitrate-potassium persulfate reagent (27%),³¹ or manganese(III) acetylacetonate (45%).³² Phenol yielded the 4.4'-dihydroxybiphenyl (7; R = H; 34%) on oxidation with vanadium tetrachloride,³³ together with the 2,4'-isomer (17%). Vanadium oxytrichloride oxidized 1-naphthol to the para-para coupled binaphthyl (8; R = H; 56%). The tetrahydroxybinaphthyl (8; R = OH), a metabolite of Daldinia concentrica, was synthesized³⁴ by treatment of 1,8-dihydroxynaphthalene with potassium ferricyanide.



Para-substituted phenols may still couple through the para position and products may be formed in good yield, especially if stable. Thus p-cresol gives Pummerer's ketone (9; 63%) on treatment with silver

carbonate;²⁶ only the o-p coupled intermediate can cyclize to (9), which is then not subject to phenolic oxidation. In contrast p-cresol with iron(III) chloride gives at least 10 products including (9), and the biphenyl and terphenyl products of o-o coupling (10) and (11).³⁵ Simple o-o linking can be the major pathway, e.g. alkaline potassium ferricyanide oxidation of totarol (12; R = Me) in benzene-water afforded podototarin (13; R = Me) in 43% yield; and ethylmacrophyllate (13; $R = CO_2Et$) was the major product from similar oxidation of (12; $R = CO_2Et$).³⁶ An aqueous alkali-organic solvent two-phase system is usually favored for ferricyanide oxidation since the products may be separated from the oxidant. The naphthol (14) dimerizes quantitatively on heating (215 °C) in air to the bisnaphthol (15; R = H), which contains the gossypol nucleus. O-Methylation of (15; R = H) gave apogossypol hexamethyl ether (15; R = Me)³⁷ The allylguaiacol (16) was oxidized quantitatively at a platinum anode to the biphenol (17), using an alkaline electrolyte,³⁸ although to very different products in neutral media (vide infra). The trimethyltetralol (18) gave the dimer (19) in 46% yield using a copper(II)-amine complex.³⁹ The oxidation of a similar chiral reactant (20) has been examined from the stereochemical viewpoint.⁴⁰ The (\pm) -tetrahydroisoquinoline (20) on oxidation with potassium ferricyanide, or with oxygen over platinum, provided three stereoisomers of (21); a meso (R,S) form, and two separable rotamers of the (S,S)/(R,R)racemate. However, on electrooxidation at a graphite anode only one stereomer was formed (69%), the rotamer shown (21) of the (S,S)/(R,R) form. Separate oxidations of the (S)- and (R)-enantiomers of (20) afforded only the (S,S) (62%) and the (R,R) (66%) isomers, both as a single rotamer.



2-Naphthol can successfully be dimerized oxidatively, selectively through the o-site to give (22) using copper(II)-amine complexes $(70\%)^{39}$ or manganese(III) acetylacetonate $(69\%)^{32}$ and o-o coupling is the major pathway (90%) on ferricyanide oxidation of the trisubstituted phenol (23) to the orthodiphenoquinone (24).⁴¹ In this context it is of interest that a compound obtained (74%) on ferricyanide oxidation of





the 2,2'-dihydroxybiphenyl (25) proved to be the oxepinobenzofuran $(26)^{42}$ rather than the benzoxete (27) as originally formulated.⁴³ The intermediate orthodiphenoquinone (28) rearranges as shown. Another case of o-o dimerization is that of the anodic oxidation (graphite anode, acetonitrile-water, 0.3 V) of the sodium salt of the *p*-hydroxybenzyltetrahydroisoquinoline (29), which selectively gave product (30; 45%).⁴⁴ The *N*-ethoxycarbonyl substituent is necessary in this example to prevent oxidation at nitrogen; with NH or NMe analogs fragmentation supersedes coupling, as pictured in (31), leading to initial products (32) and (33).



The question of cross-coupling of different phenols was alluded to above. An interesting example⁴⁵ of selectivity is shown in Scheme 9; the major product, shown, forms a relatively stable radical anion under the reaction conditions. To what extent this thermodynamic factor controls product ratios is not clear.



Few such cases have been described and it does not seem possible to make reliable predictions of such cross-coupling reactions on the basis of present knowledge.



Intermolecular coupling has been displayed in the foregoing reactions. Intramolecular cases are of interest, potentially offering greater product selectivity, and many have been investigated as models of biogenetic significance. A high proportion of these proceed in poor yield and have limited synthetic value. However there are some rewarding examples. Thus laudanosoline methiodide (**34**) underwent preferential o-p linking to (**35**; 62%), with aqueous iron(III) chloride at room temperature.⁴⁶ Dihydropiceatannol (**36**) was also selectively oxidized to a 9,10-dihydrophenanthrene, isolated (66%) as its tetraacetate (**37**) after reductive acetylation.⁴⁷ It is possible but unproven that in both these cases an o-benzoquinone intermediate is generated. Such a pathway appears very likely in the following reaction⁴⁸ in which the tetrahydroisoquinoline (**38**) was treated with diphenyl selenoxide in methanol at ambient temperature to afford in 80% yield the coupled product (**39**), *N*-trifluoroacetylwilsonirine. This reagent cleanly oxidizes catechols to o-quinones via an intermediate seleno acetal (**40**). The arylethyltetrahydroisoquinoline (**41**) is similarly intracoupled,⁴⁸ forming a new seven-membered ring in product (**42**).

Finally in this section it is worth noting that the oxidative coupling of phenols may be extended to form interesting polymers. Thus electrooxidative polymerization of phenol using basal-plane pyrolytic graphite electrodes gives thin black polymer films which are electrically conducting.⁴⁹



2.9.2.2 Biaryls Formed in Phenol Ether Couplings

This section covers the union of two aryl moieties, one of which is a fully alkylated phenol, and the other is either a free phenol or a phenol ether. Clearly, radical dimerization is not operative in such reactions. An early example is to be found in the work of Fitcher and Dietrich $(1924)^{50}$ who showed that 3,3',4,4'-tetramethoxybiphenyl was among the products of electrolytic oxidation of veratrole in sulfuric acid using a lead dioxide anode; the biaryl was formed in about 20% yield based on reacted veratrole.

The potential of such syntheses was not realized for several decades. (The following account is not historical.)

The intermolecular examples of synthetic value are self-couplings, *e.g.* formation of the dimer (43) from benzylsesamol,⁵¹ in 85% yield using vanadium oxytrifluoride; preparation of the biaryl (44; 95%), from 4-methylveratrole, employing iron(III) chloride supported on silica;⁵² and synthesis of 4,4'-dimeth-oxybiphenyl (69%) from anisole by oxidation with thallium(III) trifluoroacetate in the presence of catalytic palladium(II) acetate.⁵³ This approach has been used in a natural product synthesis. The dimers (45) and (46) were prepared from appropriate derivatives of gallic acid, and transformed into schizandrin C (47) and an isomer respectively.^{54,55}

Recently the conditions have been defined under which trimerization of some phenol ethers is viable; these require anodic oxidation in a flow cell with porous electrodes, using acetonitrile or trifluoroacetic acid-methylene chloride.⁵⁶ Catechol derivative (48) and 1,2-methylenedioxybenzene were thus converted into trimers (49) and (50) respectively, in *ca.* 30% yield.









OMe

OMe

(44)

(45)

(46)

(47)





(48)



(50)

The intramolecular versions of this reaction have been widely exploited particularly for the synthesis of alkaloids and lignans. New six-, seven- and eight-membered rings are readily formed and there are some cases involving macrocycles.⁵⁷

Starting with six-membered ring formation, it has been shown that (\pm) -laudanosine (**51**; $R^1 = R^2 = Me$) can be oxidized with vanadium oxytrifluoride in trifluoroacetic acid-fluorosulfonic acid to (\pm) -glaucine (**52**; $R^1 = R^2 = Me$; 43%).^{57a} The same oxidant also effected phenol-phenol ether coupling in this alkaloid series, with *N*-trifluoroacetylcodamine (**51**; $R^1 = COCF_3$, $R^2 = H$) yielding the aporphine (\pm) -*N*-trifluoroacetylwilsonirine (**52**; $R^1 = COCF_3$, $R^2 = H$). The benzyltetrahydroisoquinoline (**53**) was intracoupled to (\pm) -ocoteine (= thalicmine; **54**) in 46% yield employing thallium trifluoroacetate at -40 °C, with catalytic boron trifluoride.⁵⁸

To exemplify phenol-phenol ether coupling the benzylisoquinoline (55) was oxidized with a range of reagents, to form the novel quinonoid oxaporphine (56) with generation of the quinone methide after coupling; best yields were obtained with vanadium oxytrifluoride in trifluoroacetic acid (59%) and molybdenum oxytetrachloride (62%).⁵⁹ The former reagent was also very effective in converting the (non-natural) system (57) to (58; 70%); no further product oxidation was observed here, possibly because of reduced thermodynamic drive.⁶⁰

Returning to phenol ether-phenol ether coupling, synthetic septicine (59) gave (\pm) -tylophorine (60) on treatment with thallium trifluoroacetate, and the same reagent converted synthetic julandine (61) to (\pm) -cryptopleurine (62; 69%).⁶¹ In another synthesis of tylophorine the lactam (63) was transformed with va-





nadium oxytrifluoride to (64; 59%); reduction with diborane (85%) completed the sequence to tylophorine (60).⁶²

Electrooxidation is also viable for phenol ether-phenol ether coupling and has been thoroughly investigated. For example a range of compounds (65) with R = H or Me, $Z = CH_2$ or O, n = 1 or 2, reacted at the anode to give fair yields of the corresponding biaryls (66),⁶³ together with spirodienone products (see Section 2.9.3.2).





Aryl coupling to a benzylic site has also been observed: the monophenol (67) yielded the aryltetralin (68; 55%), with thallium trifluoroacetate-boron trifluoride.⁶⁴ Probably oxidation to quinone methide precedes the ring closure. Separate oxidation and cyclization steps were employed in the synthesis of (\pm) -thaliphorphine acetate. (\pm) -Codamine (69) underwent Wessely oxidation with lead tetraacetate to the acetoxycyclohexadienone (70), which closed in acetic anhydride-acid to (\pm) -thaliphorphine acetate (71), albeit in modest overall yield (14%).⁶⁵



It is worth noting an alternative strategy for intracoupling to biaryls,⁶⁶ in which the quinone monoketal (72) undergoes addition by a suitable carbanion (73); the resulting tertiary alcohols, *e.g.* (74) and (75), could be cyclized with acid (conditions of low nucleophilicity), to biaryls (76; 77%), and (77; 61%), respectively. The new C—C bond forms *meta* to methoxy of the new aryl ring, thus complementing the oxidative approach. The formation of a five-membered ring is worthy of note.

Closure to seven-membered rings has been demonstrated a number of times. Thus the 1,3-bisarylpropane (78) was oxidized anodically to the bridged biaryl (79; 72%). In a thorough study⁶⁷ of the effects of





variations in conditions and in substrate structure on the competition between cyclization and dimerization it was concluded that intermolecular processes were favored at low anode potential, and conditions were given for dimerization of the 1,2-bisarylethane (80), to (81; 61%). It was concluded that, in the full range of examples studied, both cation radical couplings, and electrophilic substitution by cation radicals on unoxidized rings, were operating. Anodic oxidation was used to prepare homoglaucine (83) from homolaudanosine (82; 34%).⁶⁸

Thallium trifluoroacetate has proved an excellent reagent for similar oxidations, *e.g.* of (84) and (85) to the bridged biaryls (86) and (87), 80% and 81%, respectively,^{58,69} and this reagent was selected for a crucial step in a total synthesis of (\pm) -steganone.⁷⁰ Thus the (*E*)-2-benzylcinnamate (88) provided the bis-





benzocycloheptatriene (89; 81%), at -18 °C in trifluoroacetic acid. Reduction of the methoxycarbonyl function and Simmons–Smith methylation then generated the cyclopropane (90; R = CH₂OH) from which (±)-steganone (91) could be generated *via* ring expansion with the correct biaryl twist, relative to lactone fusion. Alternatively the cyclopropane (92) could be cyclized (45%) to (90; R = CO₂Me). Interestingly it was not possible to induce biaryl formation with an eight-membered ring by treatment of (93) with thallium trifluoroacetate.^{70b} Much synthetic interest has been shown in steganone and its relatives, induced by the anticancer properties of this group of lignans.

Although the bisarylbutene (93) proved unamenable to oxidative cyclization, eight-membered rings have been constructed successfully using substrates with saturated interaryl links. Several relatives of steganone have been prepared in this way. Thus (\pm)-steganacin was synthesized⁷¹ using intracoupling of the malonate-derived bisaryl butane (94) to (95), in 45% yield, using vanadium oxytrifluoride at 25 °C,









and (\pm)-isostegane (96) was obtained⁷² in good yields (65–70%) by oxidation of the (\pm)-bisbenzylbutyrolactone (97), employing the same reagent at 45 °C. (–)-Isostegane (96) was similarly prepared⁷³ (61%) from (+)-deoxypodorhizone (97) (from enantiospecific synthesis), and converted into (+)-stegane and (+)-steganacin (98), the enantiomer of natural (–)-steganacin. This sequence demonstrated absolute stereochemistry in the sequence.⁷³ A related, but not natural, steganane (99) was obtained⁷⁴ from natural dimethylmatairesinol (100), effected by thallium tristrifluoroacetate in methylene chloride at room temperature; the reaction was accelerated by boron trifluoride and yields up to 90% were recorded.



Some simpler lignans of the dibenzocyclooctadiene group have been prepared by related couplings, *e.g.* (\pm)-deoxyschizandrin (**101**; *ca.* 50%) and (\pm)-wuweizisu-C (**102**; yield unstated) from precursors (**103**) and (**104**), using low temperature vanadium oxytrifluoride oxidations^{75,76} A reinvestigation⁷⁷ of the last two syntheses, comparing various reagents, concluded that iron(III) reagents were most effective although the yields reported were only fair, 32% and 46% for (**101**) and (**102**), utilizing iron(III) perchlorate hexahydrate in acetonitrile. Finally the successful formation is worth noting of the eight-membered nitrogen heterocycles (**105**) and (**106**) by anodic oxidations (platinum anode) of the precursor amides (**107**; 45%) and (**108**; 60%).⁷⁸



A few coupling reactions to afford macrocycles have been uncovered.⁷⁹ The ester (109), on treatment with thallium tristrifluoroacetate for 3 min at room temperature, yielded the 11-membered lactone (110; 14%), via para-para linking. Longer (10 min) reaction of the related ester (111), with water quenching, afforded the 13-membered lactone (112), by way of coupling, para to a methoxylated site, and loss of a benzylic hydrogen to intermediate (113), which was rearomatized in allylic hydrolysis. In studies on anodic oxidations of substrates (78), cyclization to biaryls with bridges as long as n = 16, *i.e.* 20-membered rings, have been observed, although in low yield.⁸⁰

In concluding this section it is worth noting that in these nonphenolic aryl couplings, the new bond is usually formed between sites *para* to ether oxygen, in accord with electron density calculations on methoxybenzene cation radicals, and with their ESR spectra.



2.9.2.3 Biaryls Formed by Aryl Radical Insertion

In the introduction it was pointed out that phenolic coupling may involve addition of a carbon radical to a phenolate anion, and nonphenolic coupling may proceed by way of radical cation attack on an unoxidized aromatic unit. A closely related group of transformations involve generation of an aryl radical which then inserts into a second aryl unit. A few examples are given here, since the process can supplement the direct oxidation approach. A representative reaction is the photochemical transformation,⁸¹ under basic conditions (sodamide-dimethylformamide), of the bromophenol (114) into (±)-domesticine (115; 31%), with formation of a six-membered ring. The first synthesis⁸² of a naphthylisoquinoline alkaloid, O-methyltetradehydrotriphyophylline (116) used irradiation (254 nm) of bromide (117) as the key step, forming the pentacycle (118; 15%) for conversion to (116). An aryl radical may also be generated by reductive cleavage of halide at the cathode; the bromobenzoylaniline (119) gave the biaryl (120) in this way.⁸³ In another example⁸⁴ the bromobenzovlnaphthylamine (121) afforded the arylnaphthalene (122; 44%), by closure and rearrangement through the spiro radical (123). Seven-membered rings are amenable to formation in such reactions; irradiation of the iodobenzylamine (124) in dilute acid gave the dibenzoazepine (125) in 57% yield,85 and photolysis in dilute alkali of the bromophenol (126) yielded the homoaporphine (127)⁸⁶ albeit in only 55%. Two examples of intracoupling to eight-membered rings complete this brief survey; photolysis of the iodide $(128)^{87}$ and the bromide $(129)^{85}$ gave the dibenzoazacyclooctadienes (130; 21%) and (131; 22%). The former is an apogalanthamine analog.



,,,,,,,,,

Ph

Ń

-

0





(117)





(118)



(119)









MeO

MeO



(123)









(125)







2.9.3 SYNTHESIS OF SPIRODIENONES

2.9.3.1 Spirodienones via Phenol-Phenol C-C Coupling

If, in an intramolecular phenolic coupling process, one of the preferred coupling sites (most usually *para* to OH) is already substituted, then a spirodienone product may form, and may be favored over coupling at another unsubstituted site. In practice good yields of spirodienones have often been isolated in such cases. Spirodienones are prone to undergo dienone-phenol rearrangement (see Chapter 3.5, Whiting, this volume) and occasionally reorganize under the reaction conditions; it is not always clear from products whether or not a dienone has intervened.

The clearest examples are suitably substituted 1,3-diarylpropanes. The 3,4'-dihydroxy compound (132) on treatment at -78 °C in ether with vanadium oxytrichloride yielded (75%) the tricyclic dienone (133),^{RR} and the methoxylated relatives (134) and (135) both cyclized in parallel fashion to (136) and (137), in 67% yield, using an iron(III) complex reagent.⁸⁹ The amide (138) formed the spiro-linked benzazepine (139; 50%) on oxidation with vanadium oxytrichloride,⁹⁰ showing that a seven-membered ring may form. Generation of five-membered rings is also satisfactory, as evidenced by the synthesis of picrolichenic acid (140) by manganese dioxide oxidation of the aryl ester (141), in 25% (max.) yield.⁹¹

These couplings are of major importance in the alkaloid area, where they may be biomimetic, and there has been a considerable quantity of work on the 1-benzyl- and 1-phenylethyl-tetrahydroquinoline alkaloids, which has been reviewed.^{1,17,19}

One major preoccupation was the realization of a laboratory analogy for the intracoupling of relatives of reticuline (142) to the morphine group via p-o (4a-3') linking. In various studies p-p coupling (4a-7') and o-p (8-7') of (±)-reticuline were achieved, leading to (±)-isosalutaridine (143) and (±)-isoboldine





(144) respectively, in reactions of considerable biosynthetic interest, but little preparative value. The long-sought p-o linking (4a-3') was eventually realized using thallium tristrifluoroacetate as oxidant at low temperature; in this way the N-ethoxycarbonyl relative (145) provided the spirodienone (146) in 23% yield.⁹² Lithium aluminum hydride reduction gave alcohol (147), which closed in acid to (±)-thebaine (148).



The (\pm) -benzyltetrahydroisoquinoline (149), with 3',7- rather than 4',7-dihydroxylation as in (142), was coupled⁹³ using potassium ferricyanide to the spiroketone (150), as a pair of diastereoisomers (up to 50% yield). One stereoisomer, on reduction of carbonyl and aqueous acid treatment, was converted to the new spirodienone (151), the alkaloid orientalone.



In the 1-arylethyltetrahydroisoquinoline series, the 6',7-dihydroxy example (152) underwent o-p coupling (49%) with ferricyanide;⁹⁴ the product, (153), was rearranged into (±)-multifloramine (see Chapter 3.5, this volume). Similarly the dimethoxy analog (154) was oxidized to (±)-(155; two diastereomers), in 36% at a graphite anode,⁹⁵ and in 68% with vanadium oxytrifluoride;⁶⁸ one of the stereoisomers was kreysiginone. With a 6,6'-dihydroxy pattern, as in (156), p-p coupling was induced by the last named reagent to provide the spiro tetracycle (157).⁶⁸



In a biomimetic synthesis of erysodienone (158), the bis(arylethyl)amine (159) was oxidized with alkaline ferricyanide to give the desired natural product in 35% yield.⁹⁶ The reaction probably involves p-p coupling, oxidation to the diphenoquinone (160), and intra-Michael addition; in support similar oxidation⁹⁷ of the bridged biaryl (161) also gave erysodienone, 80%. The key intermediate (161), both in *in vitro* and *in vivo* formation of (158), has also been prepared by oxidative coupling of the benzylisoquinoline (162) by vanadium oxytrifluoride at -10 °C. The product dienone (163; 40%) fragmented on base deacylation, and reduction of the product imine gave (161).⁹⁸

The arylethylbenzylamine derivative (164) with one blocked *para* position was smoothly oxidized with alkaline ferricyanide to the spirodienone (165),⁹⁹ related to the Amaryllidaceae alkaloid system.

In recent work Kende and his coworkers¹⁰⁰ have explored intramolecular oxidative coupling between phenolate and enolate, rather than between two phenolates. This proved a viable process for certain carbonyl enolates, and for nitro-stabilized anions. Thus potassium ferricyanide oxidations of the indandione



derivative (166) and 4-(ω -nitropentyl)phenol (167) gave the product spirodienones (168) and (169) in 88% and 83% yields respectively, while reaction of the coumarin (170) with dipotassium hexachloro-iridate effected cyclization to (171; 33%).

For recent examples of intramolecular phenolic coupling to spirodienones, see refs. 164 and 165.





2.9.3.2 Cyclohexadienones via Phenol Ether Coupling

A number of interesting and useful reactions involve cyclization, forming six- to eight-membered rings, between the two aryl units separated by a short chain. Thus the bisarylethane (172) gave, on anodic oxidation, the tricyclic dienone (173) in 90% yield, arising from intracoupling followed by 1,2-shift.¹⁰¹ Similarly the analog (174) afforded dienone (175; 98%), using silica-supported iron(III) chloride as oxidant.⁵² The 1,2-shift was not observed with the example (176) with monooxygenated aryls, which gave the steroidal tricycle (177) at the anode (22%) or on reaction with vanadium oxytrifluoride (30%). With suitably substituted diarylpropanes formation of a spirodienone is feasible. Thus vanadium oxytrichloride treatment of (178) led to tricyclic spirodienone (179; 70%);⁹⁰ only 18% of the dibenzocycloheptadiene alternative (180) was formed. The parallel monophenolic coupling $(181) \rightarrow (179)^{90}$ was also effected at -78 °C with the same reagent (97% yield); also, both thallium and silver trifluoroacetates induced the coupling. Such monophenolic coupling, using electrooxidation, was employed for transformation¹⁰² of the biarylpropane (182) into the spirodienone (183; 80%); the latter was subjected to cyclopropanation. followed by fragmentation and ring expansion to the tropolone (184), a key intermediate in a colchicine synthesis. Another example involves monophenolic coupling in the acetal (185) using bistrifluoroacetyloxyiodobenzene;¹⁰³ the product (186), only obtained in 13% yield, was converted on to 6a-epipretazettine.

An original biomimetic synthesis¹⁰⁴ of the Amaryllidaceae alkaloid narwedine involved oxidation of the arylethylbenzylamine (**187**) with lithium tetrachloropalladate to generate the arylpalladium species (**188**); the course of further oxidation with thallium tristrifluoroacetate could be controlled to lead either to the bisspirodienone complex (**189**; 1'-1" coupling) or to narwedine (**190**; 51%; 1'-2" coupling). Seven-membered rings were also formed in the monophenolic oxidation of (**191**; vanadium oxytrifluoride-thallium tristrifluoroacetate, 88%)^{57b} and in the anodic oxidation (62%) of (**192**),¹⁰⁵ in both cases yielding dienone (**193**). Deacylation of (**193**) leads to spontaneous Michael addition, forming (±)-oxocrinine (**194**).¹⁰⁵





(190)

(191) R = OH(192) R = OMe

(189)


Benzyltetrahydroisoquinolines have been thoroughly investigated for reasons discussed above. (\pm) -Laudanosine (195) was oxidized at a platinum anode to yield methylflavinantine (196) in 52% yield¹⁰⁶ (4a-7' coupling), and a similar reaction proceeds with a number of relatives including 6',8-dimethoxy-laudanosine (197), which provided protostephanone (198; 35%).¹⁰⁷ An enantioselective synthesis of (*R*)-laudanosine enabled preparation of (*R*)-methylflavinantine by anodic oxidation,¹⁰⁸ with reported yields 70–90%. Various analogs of the benzylisoquinoline system have been examined. Among these, oxidative coupling of the benzyltetralin (199) is of interest;⁶³ varying the conditions of anodic oxidation, *i.e.* solvent, temperature and pH, it was found possible to obtain either the dienone (200; 4a-7' coupling, 71%) or the biaryl (201; 7'-8 coupling, 45%). The benzyl lactone (202) was electrooxidized to the spirolactone





(203),⁷⁸ in parallel with the last two cases, but the isomeric lactone (204) afforded a rearranged spirolactone (205),¹⁰⁹ both with vanadium(V) reagents (59%) and at the anode (33%). The arylethylisoquinoline (206) also afforded a similarly rearranged spirodienone (207; 64%), on intracoupling, using vanadium oxytrifluoride.⁶⁸

2.9.3.3 Spirodienones via Aryl Radical Insertion

Two examples suffice to illustrate this reaction type, both involving 1-benzyltetrahydroisoquinolines. The 8-bromo structure (208) was photolyzed under alkaline conditions to (\pm)-mecambrine (209), in 10% yield,⁸¹ and the 7'-bromo relative (210) afforded dienone (211), 34% isolated, on the way to synthesis of (\pm)-boldine.¹¹⁰



2.9.4 FORMATION OF CARBON-OXYGEN BONDS

2.9.4.1 Synthesis of Aryl Ethers

An early example¹¹¹ was the surprisingly selective oxidation of methoxythymol (212) with alkaline ferricyanide to libocedrol (213; 49%); further cross-coupling of the latter with more monomer using the same reagent yielded (46%) the trimeric libocedroxythymoquinone (214; 46%). Lophocerine (215) also formed a dimer, isopilocereine (216; 32%), on exposure to ferricyanide, together with traces of the

trimeric pilocerine (217).¹¹² Magnocurarine methiodide (218) was similarly dimerized (15%) to the model (219) for certain dimeric benzylisoquinoline alkalotds;¹¹³ and in the same series, phenol (220)⁴⁴ was coupled at a graphite anode to a product which was transformed by *O*-protection, reduction, and deprotection to (\pm)-dauricine (221), 8% overall. Yet another benzylisoquinoline (222) gave the unusual macrocyclic dehydrodimer (223) on treatment with diphenyl selenoxide⁴⁸ followed by permethylation.







(218)



(220) $R^1 = OMe$, $R^2 = H$, $R^3 = CO_2Et$ (222) $R^1 = R^2 = OH$, $R^3 = COCF_3$



A number of intramolecular variations are documented, some of biogenetic interest. The benzophenone (224) was cyclized in good yield (83%) to 2,6-dihydroxyxanthone (225) with ferricyanide, ¹¹⁴ and the synthetic depside (226) yielded diploicin (227) on manganese dioxide oxidation.¹¹⁵ Finally the diarylheptanoid (228) afforded the macrocyclic ether (229) on reaction with thallium tristrifluoroacetate; similar ethers, *e.g.* acerogenin, occur naturally.¹¹⁶



2.9.4.2 Spiroethers

Ring systems containing cyclohexadienones spiro-linked to a cyclic ether can be constructed by oxidative C—O bond connection in the intramolecular mode. Five-membered ring formation is well documented, and the reactions may be efficient. For example, the linked bisnaphthols (230), (231) and (232) formed spirodienones (233),¹¹⁷ (234)¹¹⁷ and (235)¹¹⁸ on treatment with sodium hypochlorite (*ca.* 60% yield), alkaline ferricyanide (60%), and dichlorodicyanobenzoquinone ('quantitative'), respectively. A well-known reaction in this area is the oxidation of griseophenone A (236) to dehydrogriseofulvin (237),¹¹⁹ which has been induced with various reagents, potassium ferricyanide, lead dioxide and manganese dioxide, and quantitative yields have been reported. Copper(II) salt–amine complexes are also effective¹²⁰ and, using (–)- α -phenylethylamine, dehydrogriseofulvin with a small enantiomeric excess was obtained. The oxygen atom participating in C—O bond formation may derive from a function other than phenol. Thus the arylpropanoic acids (238) and (239) gave the spirodienones (240) and (241) at a platinum anode, in modest yield, while the acid (242) yielded (243; 40%), with buffered *N*-bromosuccinimide.¹²¹ In view of the disparity between oxidation potentials of the interacting functions it is unlikely that these reactions involve radical–radical coupling. The same is true for the oxidative cyclization of phenolic oximes, *e.g.* (244) to the spirooxazoline (245; 62%), using manganese(III) acetylacetonate.¹²² The reaction, a model for the biosynthesis of aerothionin, probably proceeds *via* an incipient phenoxonium ion (see Scheme 3).



Few examples, in this class, of phenol-phenol coupling generating six-membered rings, are available: the bisarylethane (246) gave (247) only in 11% yield.¹²³ However biomimetic oxidation of phenolic

Coupling Reactions

acids offers several high yielding examples, *e.g.* the formation of geodoxin (248) from geodin (249),¹²⁴ employing lead dioxide, 88% yield; and the oxidation of the acid (250) to dehydrogriseofulvoxin (251), 85% with manganese dioxide.¹¹⁴ A striking case of successful intramolecular coupling¹²⁵ is the formation of the 17-membered cyclic ether (252) from the phenolic tripeptide (253) on oxidation with thallium trinitrate in methanol. Presumably a dibromospirodienone is formed, which suffers allylic displacement of bromine by methanol. Dienone ether (252) was transformed into the cyclic bisaryl ether peptide (254), an aminopeptidase inhibitor.



2.9.4.3 Aryl Alkyl Ethers via C-O Coupling

Two useful groups of related reactions may be discerned here, both of which are observed for propenyl-phenols. The first type is illustrated by the oxidation of isoeugenol (255) with aqueous iron(III) chloride, ¹²⁶ which yields the *threo*- and *erythro*-stereomers (55%) of the hydroxy ether (256), presumably by O—C_β linking of the initial phenoxyl radical, and addition of water to the resulting intermediate quinone methide. A similar reaction¹²⁷ of the phenol (257), but using silver oxide, provided (258), isolated (62%) as a monomethyl ether; *threo*-(258) was (\pm)-virolin. Cross-coupling is also possible in an efficient manner; thus silver oxide treatment of a mixture of the propenylphenol (255) and the allylphenol (259; R = Me) yielded¹²⁷ an intermediate quinone methide which was reduced *in situ* and methylated to afford the neolignan (260), overall in 80% yield. Cross-coupling of (259) with (Z)-isoeugenol at a platinum anode gave stereoisomers of the ether (261; 34%).¹²⁸ Related examples are known.^{129–131} It has been demonstrated¹²⁷ that oxidation of the allylphenol (259) with silver oxide in the absence of other phenols or nucleophiles yielded, in solution, the relatively stable quinone methide (262); addition of nucleophiles, *e.g. p*-methoxyphenol, proceeded smoothly to give ether (263), 71% overall.



If, in this type of reaction, a suitable catechol component is present, then the major products may be benzodioxanes, *e.g.* silver oxide treatment (benzene, r.t.) of a mixture of propenylphenol (255) and allylphenol (259; R = H), gave the (±)-product (264)¹³² in 'good yield', which on methylation formed the neolignan eusiderin. Coumarinolignans have been synthesized in a parallel way, *e.g.* cross-coupling¹³³ of fraxetin (265) with coniferyl alcohol (266) using silver oxide or horseradish peroxidase, leads to both cleomiscosin A (267) and its regioisomer (268), cleomiscosin B. Other members of this class, *i.e.* propacin, aquillochin, and daphneticin were prepared in parallel fashion. The interesting antihepatotoxic flavonolignan silybin (269) was obtained from (2*R*,3*R*)-dihydroquerctin (270) and coniferyl alcohol; again the regioisomer, isosilybin (271), was also produced.¹³⁴ These natural products occur as mixtures of stereoisomers, all (2*R*,3*R*) in the chromanone unit but differing in the geometry of the benzodioxane section. The same coupling may be effected enzymically,¹³⁵ and peroxidases induce coupling in other vinyl catechols.¹³⁶



(271)

2.9.5 C-C COUPLING THROUGH ARYL CONJUGATED DOUBLE BONDS

In structures which contain an o- or p-hydroxystyryl subunit, the radical from phenolic oxidation is delocalized through the aromatic ring and the side chain (272), and couplings may thus involve the β -carbon. The charted reactions which fall into this class can be subdivided into three sets, with products derived from: (i) C_β—C_β coupling; (ii) C_β—C_{AR} coupling; and (iii) both modes of coupling.

In the first category the best known cases are those in which a hydroxycinnamic acid is dimerized oxidatively to a bisaryldilactone. A range of examples are known.¹³⁷ Typical reactions are the oxidation of sinapic acid (273) to the dilactone (274), in 65% yield using oxygen-iron(III) chloride,¹³⁸ and in 60% yield using a high acid concentration at a glossy carbon anode,¹³⁹ and the oxidation of ferulic acid (275) to 'dehydrodiferulic acid' (276), with iron(III) chloride–air (22%)¹⁴⁰ and with a iron(III) chloride–dimethylformamide reagent.⁸⁹ A range of nonphenolic cinnamic acids have been oxidized with thallium tristrifluoroacetate–boron trifluoride to dilactones analogous to (274);¹⁴¹ a nonradical mechanism is preferred whose key step is shown in cipher (277). If a *p*-hydroxycinnamate ester is employed, rather than acid, then C_{β} — C_{β} coupling can be followed by trapping of quinone methide intermediates with water, as in the formation of the tetrahydrofuran (278) from methyl dibromoferulate (279), with iron(III) chloride in aqueous acetone.¹⁴² One stereoisomer of the diester (278) was obtained, albeit in modest yield; further reduction steps gave the lignan (±)-veraguensin. *p*-Hydroxycinnamyl alcohols can be oxidized to bisaryldioxabicyclo[3.3.0]octanes, *e.g.* syringin (280) is both hydrolyzed and oxidized by crude almond emulsion to (±)-syringaresinol (281), in up to 68% yield.¹⁴³ A cross-coupling between ferulic acid and coniferyl alcohol was used to synthesize, in low yield but in one step, the (±)-natural lactone (282).¹⁴⁴ Propenylphenols may also be induced to undergo C_β—C_β coupling, *e.g.* the phenol (283) on electrooxidation in methanol¹²⁸ afforded the cyclic ether (284; 47%), as well as some primary coupled product (285; 16%).

The formation of C_{β} — C_{AR} bonds is illustrated by the generation of 'dehydrodiisoeugenol' (**286**) from (*E*)-isoeugenol (**257**), enzymically with a mushroom juice¹⁴⁵ (45%) yield, with horseradish peroxidase (65%),¹³⁰ with iron(III) chloride¹⁴⁶ (53%), by photooxidation (20%)¹²⁹ and by (-)- α -phenylethylamine-copper(II) nitrate (21%). In the last case a low enantiomeric excess was recorded.^{120b} Coniferyl alcohol can be oxidized to 'dehydrodiconiferyl alcohol' (**287**) with various enzyme oxidases.¹⁴⁷⁻¹⁴⁹ The *cis*





(285)

stereochemistry, for the dihydrobenzofuran ring substituents, was specifically formed in one case.¹⁴⁸ When horseradish peroxidase was used in a reaction mixture saturated with β -cyclodextrin, it is claimed¹⁴⁹ that the product was formed enantiospecifically. Sensitized photooxidation of methyl ferulate¹⁵⁰ afforded a related 2-aryldihydrobenzofuran (**288**).¹⁵⁰

An unusual C_{β} — C_{AR} coupling was observed on anodic oxidation, in acetic and trifluoroacetic acids, of a mixture of (*E*)-isosafrole (**289**) and the allylphenol (**290**).¹⁵¹ Various interesting products were characterized, among them the cyclohexadienone (**291**). Although only formed in 4% yield, this product was readily converted by successive alkali and acid reactions, into the neolignan derivative denudatin A (**292**).



Intramolecular C_{β} — C_{AR} couplings have been observed in the anodic oxidation of a series of 4-hydroxy-2-vinylbiphenyls, (e.g. 293),¹⁵² which afforded the corresponding spirodienones (e.g. 294; 92% from 293). Another C_{β} — C_{AR} linkage resulted from the brief oxidation of the dihydrocinnamyl cinnamate (295) with thallium tristrifluoroacetate-catalytic boron trifluoride (water-quench), when the eightmembered lactone (296) was formed.79

There has been much interest in the synthesis of lignan lactones of the aryl tetralin type, induced by the antitumour activity of podophyllotoxin relatives, and some short oxidative routes into the series have been pursued. In a typical reaction, in which both C_{β} — C_{β} and C_{β} — C_{AR} couplings occur, the cinnamyl cinnamate (297) was treated for 30 s with chromium trioxide-fluoroboric acid in acetonitrile.¹⁵³ Three tetracyclic products were obtained, the lactones (298; 19%) and (299; 7%), and (±)-isopodophyllotoxone

















(297)



(299)



(300)



(300; 16.3%), which was epimerized (80%) to the *cis* ring fused natural isomer (\pm)-picropodophyllone. The same reagent converted *O*-methylisoeugenol (3,4-dimethoxyphenyl-1-propene) directly into the tetralone (301; 16%), which on deoxygenation gave the lignan (\pm)-galbulin.¹⁵⁴

2.9.6 MISCELLANEOUS OXIDATIVE COUPLINGS

2.9.6.1 Couplings Involving Cycloadditions

The anodic coupling of (E)-isosafrole (289) and the allylphenol (290), in acidic medium was discussed above (see Section 2.9.5). The reaction took a different course under less acidic conditions (methanolacetic acid) as illustrated in Scheme 10.¹⁵¹ Two single-electron transfers from the phenol lead to the phenoxonium cation (302), which underwent [3 + 2] cycloaddition with the electron-rich alkene; demethylation of the product cation gave the neolignan (303; 81%), which occurs naturally in an *Aniba* species. (E)-Isosafrole could be replaced by a protected 4-hydroxy-3-methoxyphenylpropene, to form analogous compounds. Using (Z)-isosafrole in the reaction, and adding trifluoroacetic acid to the medium, led to isolation of two products, the stereoisomer (304) of the *Aniba* neolignan (303), and futoene (305; 15%), another unusual neolignan. As shown in Scheme 11, the former product arose from *exo* [3 + 2] cycloaddition, while futoene probably was formed by acid-catalyzed rearrangement of the initial *endo* cycloadduct.¹⁵¹



Scheme 11

The electrooxidations of allyl- and propenyl-phenols may follow even more different pathways leading to a rich diversity of products. At a carbon anode in methanol-lithium perchlorate, ferulic acid formed the tricyclic diacid (**306**; *cf.* Section 2.9.5), in a concentration-dependent process¹³⁹ probably involving [4 + 2] dimerization of the primary oxidation product, the dienone acetal (**307**).

The allyl phenol (259; R = Me) at a platinum anode in methanol-sodium methoxide, gave a C—O coupled major product (see Section 2.9.4.3), but in aqueous methanol-sodium bicarbonate the adduct (308; 23%) was formed, by a cycloaddition mechanism whose course is not entirely clear, with a trace of the natural dione asatone (309).¹³¹ The constitution of asatone suggests that it originates in [4 + 2] dimerization of the *o*-quinone acetal (310), and, in support, electrolysis of (259; R = Me) in neutral methanol gave acetal (310) which spontaneously dimerized at room temperature to asatone.¹⁵⁵ Further heating together of the quinone acetal (310) and asatone (309) gave the natural sesquilignans heterotropantrione (311) and isoheterotropantrione (312) by a further Diels-Alder process.¹⁵⁶ Bisdemethoxyasatone (313) can be prepared from 4-allyl-2-methoxyphenol (16), and on irradiation in hexane afforded demethoxy-isoasatone (314) by [2 + 2] cycloaddition in the mode indicated.³⁸



(313)

(314)

The methylenedioxypropenylphenol (283) undergoes another form of oxidative dimerization on treatment with palladium(II) chloride in aqueous methanol with sodium acetate. As shown in Scheme 12,

Coupling Reactions

 C_{β} — C_{β} coupling occurs, probably *via* the first-formed palladium phenolate (**315**) to give the bisquinone methide (**316**), and the latter spontaneously undergoes intramolecular Diels–Alder reaction to the natural lignan carpanone (**317**) in 46% yield, with stereocontrol at five chiral centers.^{157a} High yields, up to 94%, have been recorded using oxygen as oxidant with a metal(II)–salen complex as catalyst, *e.g.* cobalt(II) salen.^{157b} A low yield of carpanone was also obtained in electrooxidation.¹²⁸



Scheme 12

Another case of oxidative coupling involving [4 + 2] cycloaddition is that of α -tocopherol (318). In alkaline ferricyanide the dehydrodimer (319) is formed, *ca.* 60%, by the sequence indicated in Scheme 13.¹⁵⁸ The chemistry was elucidated in the model coupling (320) \rightarrow (321).¹⁵⁹



2.9.6.2 Coupling at Benzylic Carbon

A minor variant in the oxidative coupling process stems from generation of a benzylic radical and coupling at that site. The mechanistic course remains obscure, although it seems plausible to postulate primary oxidation at phenolic oxygen followed by hydrogen migration from a benzylic site. In practice excellent yields may be obtained, as illustrated by the oxidation of *p*-cresols (322) by silver oxide on Celite in refluxing dry benzene, forming the bisquinonemethides (323), *e.g.* with $R^1 = R^2 = Me$, 93% yield.²⁸ In a biomimetic conversion emodin anthrone (324) was oxidized in pyridine by oxygen to form the benzylically coupled dimer (325); on further oxidation conventional phenol coupling and dehydro-

genation lead through protohypericin (327) to the natural pigment hypericin (328). Penicillopsin (326) also reacted, on exposure to oxygen, by benzylic coupling to give protohypericin and then hypericin, in 50% overall yield.¹⁶⁰ As a final example, the lactone (204), the anodic oxidation (acidic medium) of which was discussed in Section 2.9.3.2, could be alternatively oxidized in neutral acetonitrile to the bridged lactone (329). Although the yield was only 20%, the reaction represents a novel C_{AR} —C benzylic bond formation.¹⁰⁹



2.9.6.3 C-N Bond Formation

Coupling between aromatic amines and either phenols or other aromatic amino compounds is known and it is important in the production of azo polymers, and in the formation of azo dyes. Most of this work is outside the scope of this chapter; a number of reviews are cited in ref. 20. One or two examples will suffice here to illustrate the area. Thus anthranilic acid (330) forms phenazine-1,6-dicarboxylic acid (331) on treatment with manganese dioxide or lead dioxide in dry benzene or chloroform.¹⁶¹ The mechanistic details are obscure. Excellent yields could be obtained in intramolecular cases,¹⁶² e.g. amine (332)





afforded phenazine (333) quantitatively on oxidation with iron(III) chloride. These studies comprise models for bacterial phenazine pigments. Lastly N-C (phenolic) linking is exemplified in the anodic conversion of the phenylethyltetrahydroisoquinoline (334) to the tetracyclic ammonium salt (335).¹⁶³

2.9.7 REFERENCES

- 1. W. I. Taylor and A. R. Battersby (eds.), 'Oxidative Coupling of Phenols', Arnold, London, 1967.
- V. Griessmayer, Justus Liebigs Ann. Chem., 1871, 160, 55. 2.
- 3. P. Ernst and C. Zwenger, Justus Liebigs Ann. Chem., 1871, 159, 32.
- 4. H. Dianin, Chem. Ber., 1873, 6, 1252.
- A. W. Hofmann, Chem. Ber., 1878, 11, 335. 5
- 6. H. Kammerer and E. Benzinger, Chem. Ber., 1878, 11, 557.
- E. Bourquelot, C.R. Hebd. Seances Acad. Sci., Ser. C, 1896, 123, 317. 7.
- 8. D. H. R. Barton and T. Cohen, Festschr. Prof. Dr. Arthur Stoll Siebzigsten Geburstag, 1957, 117.
- 9. H. Erdtman and C. A. Wachtmeister, Festschr. Prof. Dr. Arthur Stoll Siebzigsten Geburstag, 1957, 144. 10. B. S. Thyagarajan, Chem. Rev., 1958, 58, 439.
- 11. J. R. Lewis, Chem. Ind. (London), 1962, 159; 1964, 1672.
- 12. H. Musso, Angew. Chem., Int. Ed. Engl., 1963, 2, 723.
- 13. A. B. Turner, Q. Rev., Chem. Soc., 1964, 18, 347.
- 14. A. I. Scott, Q. Rev., Chem. Soc., 1965, 19, 1.
- 15. C. H. Hassall and A. E. Scott, in 'Chemistry of Phenolic Compounds', ed. W. D. Ollis, Pergamon Press, Oxford, 1961, p. 119.
- 16. W. A. Waters, 'Mechanisms of Oxidation of Organic Compounds', Methuen, London, 1964.
- 17. T. Kametani and K. Fukumoto, Synthesis, 1972, 657.
- 18. M. Sainsbury, Tetrahedron, 1980, 36, 3327.
- 19. O. P. Dhingra, in 'Oxidation in Organic Chemistry', Part D, ed. W. S. Trahanovsky, Academic Press, New York, 1982, p. 207.
- 20. P. D. McDonald and G. A. Hamilton, in 'Oxidation in Organic Chemistry', Part B, ed. W. S. Trahanovsky, Academic Press, New York, 1973, p. 97.
- 21. T. J. Stone and W. A. Waters, J. Chem. Soc., 1964, 213, 4302.
- 22. E. Müller, H. Eggensperger, A. Rieker, K. Scheffler, H.-D. Spanagel, H. B. Stegmann and B. Teissier, Tetrahedron, 1965, 21, 227.
- 23. E. Müller, A. Rieker, K. Scheffler and A. Moosmayer, Angew. Chem., Int. Ed. Engl., 1966, 5, 6.
- 24. L. R. Mahoney and S. A. Weiner, J. Am. Chem. Soc., 1972, 94, 585, 1412.
- 25. J. W. A. Findlay, P. Gupta and J. R. Lewis, J. Chem. Soc. C, 1969, 2761.
- 26. R. A. Anderson, D. T. Dagleish, D. C. Nonhebel and P. L. Pauson, J. Chem. Res. (S), 1977, 12.
- 27. H. Musso, U. V. Gizycki, U. I. Záhorszky and D. Bormann, Justus Liebigs Ann. Chem., 1964, 676, 10.
- 28. V. Balogh, M. Fétizon and M. Golfier, J. Org. Chem., 1971, 36, 1339.
- 29. L. Taimr and J. Pospišil, Tetrahedron Lett., 1971, 2809.
- 30. D. H. R. Barton, J.-C. Blazejewski, B. Charpiot and W. B. Motherwell, J. Chem. Soc., Chem. Commun., 1981, 503.
- 31. R. G. R. Bacon and A. R. Izzat, J. Chem. Soc. C, 1966, 791.
- 32. M. J. S. Dewar and T. Nakaya, J. Am. Chem. Soc., 1968, 90, 7134.
- 33. W. L. Carrick, G. L. Karapinka and G. T. Kwiatkowski, J. Org. Chem., 1969, 34, 2388.

- D. C. Allport and J. D. Bu'Lock, J. Chem. Soc., 1960, 654.
 C.-L. Chen, W. J. Connors and W. M. Shinker, J. Org. Chem., 1969, 34, 2966.
 (a) C. P. Falshaw, A. W. Johnson and T. J. King, J. Chem. Soc., 1963, 2422; (b) A. C. Day, Chem. Ind. (London), 1963, 1760.
- 37. J. D. Edwards, Jr. and J. L. Cashaw, J. Am. Chem. Soc., 1957, 79, 2283.
- 38. M. Iguchi, A. Nishiyama, Y. Terada and S. Yamamura, Chem. Lett., 1978, 451.
- B. Feringa and H. Wynberg, Tetrahedron Lett., 1977, 4447.
 J. M. Bobbitt, I. Noguchi, H. Yagi and K. H. Weisgraber, J. Am. Chem. Soc., 1971, 93, 3551.
- 41. D. Schulte-Frohlinde and F. Erhardt, Justus Liebigs Ann. Chem., 1964, 671, 92.
- 42. H. Meier, H.-P. Schneider, A. Rieker and P. B. Hitchcock, Angew. Chem., Int. Ed. Engl., 1978, 17, 121.
- 43. E. Müller, R. Mayer, B. Narr, A. Rieker and K. Scheffler, Jusius Liebigs Ann. Chem., 1961, 645, 25.
- 44. J. M. Bobbitt and R. C. Hallcher, J. Chem. Soc., Chem. Commun., 1971, 543.
- 45. A. C. Waiss, Jr., J. A. Kuhnle, J. J. Windle and A. K. Wiersema, Tetrahedron Lett., 1966, 6251.
- 46. B. Franck, G. Blaschke and G. Schlingloff, Angew. Chem., Int. Ed. Engl., 1964, 3, 192.

- 47. H. Erdtman and A. Ronlan, Acta Chem. Scand., 1969, 23, 249.
- 48. J. P. Marino and A. Schwartz, Tetrahedron Lett., 1979, 3253.
- 49. N. Oyama, T. Ohsaka, T. Hirokawa, and T. Suzuki, J. Chem. Soc., Chem. Commun., 1987, 1133.
- 50. F. Fitcher and W. Deitrich, Helv. Chim. Acta, 1924, 7, 131.
- 51. F. R. Hewgill, Tetrahedron, 1978, 34, 1595.
- 52. T. C. Jempty, L. L. Miller and Y. Mazur, J. Org. Chem., 1980, 45, 749. 53. A. K. Yatsimirsky, S. A. Deiko and A. D. Ryabov, Tetrahedron, 1983, 39, 2381.
- 54. J. Xie, J. Zhou and C. Zhang, K'o Hsueh T'ung Pao (Foreign Lang. Ed.), 1983, 28, 430 (Chem. Abstr., 1984, 101, 6915).
- 55.
- J. Xie, J. Zhou, C. Zhang, J. Yang and J. Chen, Sci. Sin., Ser. B. (Engl. Ed.), 1983, 26, 1291. V. Le Berre, L. Angély, N. Simonet-Guéguen and J. Simonet, J. Chem. Soc., Chem. Commun., 1987, 984. 56.
- (a) S. M. Kupchan, A. J. Liepa, V. Kameswaran and R. F. Bryan, J. Am. Chem. Soc., 1973, 95, 6861; (b) S. M. 57. Kupchan, O. P. Dhingra and C.-K. Kim, J. Org. Chem., 1978, 43, 4076.
- 58. E. C. Taylor, J. G. Andrade and A. McKillop, J. Chem. Soc., Chem. Commun., 1977, 538.
- 59. S. M. Kupchan and A. J. Liepa, J. Am. Chem. Soc., 1973, 95, 4062.
- 60. S. F. Dyke and P. Warren, Tetrahedron, 1979, 35, 2555.
- 61. R. B. Herbert, J. Chem. Soc., Chem. Commun., 1978, 794.
- 62. A. J. Liepa and R. E. Summons, J. Chem. Soc., Chem. Commun., 1977, 826.
- U. Palmquist, A. Nilsson, T. Pettersson and A. Ronlán, J. Org. Chem., 1979, 44, 196.
 (a) A. S. Kende and P. S. Rutledge, Synth. Commun., 1978, 8, 245; (b) A. S. Kende, L. S. Liebeskind, J. E. Mills, P. S. Rutledge and D. P. Curran, J. Am. Chem. Soc., 1977, 99, 7082.
- 65. (a) O. Hoshino, H. Hara, M. Ogawa and B. Umezawa, J. Chem. Soc., Chem. Commun., 1975, 306; (b) Y. Kai, N. Yasuoka, N. Kasai, T. Minami, K. Yamataka, Y. Ohshiro and T. Agawa, J. Chem. Soc., Chem. Commun., 1971, 1532.
- 66. D. A. Evans, P. A. Cain and R. Y. Wong, J. Am. Chem. Soc., 1977, 99, 7083.
- 67. V. D. Parker and A. Ronlán, J. Am. Chem. Soc., 1975, 97, 4714.

- S. M. Kupchan, O. P. Dhingra, C.-K. Kim and V. Kameswaran, J. Org. Chem., 1978, 43, 2521.
 A. McKillop, A. G. Turrell and E. C. Taylor, J. Org. Chem., 1977, 42, 764.
 (a) P. Magnus, J. Schultz and T. Gallagher, J. Chem. Soc., Chem. Commun., 1984, 1179; (b) P. Magnus, J. Schultz and T. Gallagher, J. Am. Chem. Soc., 1985, 107, 4984.
- 71. A. S. Kende and L. S. Liebeskind, J. Am. Chem. Soc., 1976, 98, 267.
- R. E. Damon, R. H. Schlessinger and J. F. Blount, J. Org. Chem., 1976, 41, 3772.
 (a) K. Tomioka, T. Ishiguro and K. Koga, J. Chem. Soc., Chem. Commun., 1979, 652; (b) K. Tomioka, T. Ishiguro, Y. Iitaki and K. Koga, Tetrahedron, 1984, 40, 1303.
- 74. (a) R. C. Cambie, M. G. Dunlop, P. S. Rutledge and P. D. Woodgate, Synth. Commun., 1980, 10, 827; (b) G. R. Clark, P. A. Craw, P. S. Rutledge, R. C. Cambie and P. D. Woodgate, Aust. J. Chem., 1984, 37, 1775. 75. T. Biftu, B. G. Hazra and R. Stevenson, J. Chem. Soc., Perkin Trans. 1, 1979, 2276.
- 76. G. E. Schneiders and R. Stevenson, J. Org. Chem., 1981, 46, 2969.
- 77. T. Takeya, T. Okubo, S. Nishida and S. Tobinaga, Chem. Pharm. Bull., 1985, 33, 3599.
- 78. M. Sainsbury and J. Wyatt, J. Chem. Soc., Perkin Trans. 1, 1976, 661.
- 79. S. Nishiyama and S. Yamamura, Chem. Lett., 1981, 1511.
- (a) A. Ronlán and V. D. Parker, J. Org. Chem., 1974, 39, 1014; (b) A. Nilsson, U. Palmquist, A. Ronlán and V. D. Parker, J. Am. Chem. Soc., 1975, 97, 3540; (c) U. Palmquist, A. Nilsson, V. D. Parker and A. Ronlán, J. Am. Chem. Soc., 1976, 98, 2571.
- Z. Horii, S. Uchida, Y. Nakashita, E. Tsuchida and C. Iwata, Chem. Pharm. Bull., 1974, 22, 583. 81.
- 82. G. Bringmann and J. R. Jansen, Tetrahedron Lett., 1984, 25, 2537.
- 83. J. Grimshaw, R. Hamilton and J. Trocha-Grimshaw, J. Chem. Soc., Perkin Trans. 1, 1982, 229.
- 84. W. J. Begley and J. Grimshaw, J. Chem. Soc., Perkin Trans. 1, 1977, 2324.
- P. W. Jeffs, J. F. Hansen and G. A. Brine, J. Org. Chem., 1975, 40, 2883.
 T. Kametani, Y. Satoh and K. Fukumoto, Tetrahedron, 1973, 29, 2027.
- 87. M. Kihara and S. Kobayashi, Chem. Pharm. Bull., 1978, 26, 155.
- 88. M. A. Schwartz, R. A. Holton and S. W. Scott, J. Am. Chem. Soc., 1969, 91, 2800.
- 89. S. Tobinaga and E. Kotani, J. Am. Chem. Soc., 1972, 94, 309.
- 90. M. A. Schwartz, B. F. Rose, R. A. Holton, S. W. Scott and B. Vishnuvajjala, J. Am. Chem. Soc., 1977, 99, 2571.
- 91. T. A. Davidson and A. I. Scott, J. Chem. Soc., 1961, 4075.
- 92. M. A. Schwartz and I. S. Mami, J. Am. Chem. Soc., 1975, 97, 1239.
- 93. (a) A. H. Jackson and J. A. Martin, J. Chem. Soc., Chem. Commun., 1965, 142, 420; (b) M. Shamma and W. A. Siusarchyk, J. Chem. Soc., Chem. Commun., 1965, 528; (c) A. H. Jackson and J. A. Martin, J. Chem. Soc. C, 1966, 2222.
- 94. A. R. Battersby, R. B. Bradbury, R. B. Herbert, M. H. G. Monro and R. Ramage, J. Chem. Soc., Perkin Trans. 1, 1974, 1394.
- 95. J. M. Bobbitt, I. Noguchi, R. S. Ware, K. N. Chiong and S. J. Huang, J. Org. Chem., 1975, 40, 2924.
- 96. J. E. Gervay, F. McCapra, T. Money, G. M. Sharma and A. I. Scott, J. Chem. Soc., Chem. Commun., 1966, 142
- 97. D. H. R. Barton, R. James, G. W. Kirby, D. W. Turner, and D. A. Widdowson, J. Chem. Soc. C, 1968, 1529.
- 98. S. M. Kupchan, C.-K. Kim and J. T. Lynn, J. Chem. Soc., Chem. Commun., 1976, 86.
- 99. R. A. Abramovitch and S. Takahashi, Chem. Ind. (London), 1963, 1039.
- (a) A. S. Kende, F. H. Ebetino and T. Ohta, *Tetrahedron Lett.*, 1985, 26, 3063; (b) A. S. Kende and K. Koch, *Tetrahedron Lett.*, 1986, 27, 6051; (c) A. Leboff, A.-C. Carbonelle, J.-P. Alazard, C. Thal and A. S. Kende, *Tetrahedron Lett.*, 1987, 28, 4163; (d) A. S. Kende, K. Koch and C. A. Smith, J. Am. Chem. Soc., 1988, 110, 2210.

- 101. L. L. Miller, R. F. Stewart, J. P. Gillespie, V. Ramachandran, Y. H. So and F. R. Stermitz, J. Org. Chem., 1978, 43, 1580.
- 102. E. Kotani, F. Miyazaki and S. Tobinaga, J. Chem. Soc., Chem. Commun., 1974, 300.
- 103. J. D. White, W. K. M. Chong and K. Thirring, J. Org. Chem., 1983, 48, 2300.
- 104. R. A. Holton, M. P. Sibi and W. S. Murphy, J. Am. Chem. Soc., 1988, 110, 314.
- 105. E. Kotani, N. Takeuchi and S. Tobinaga, J. Chem. Soc., Chem. Commun., 1973, 550.
- 106. L. L. Miller, F. R. Stermitz and J. R. Falck, J. Am. Chem. Soc., 1971, 93, 5941.
- 107. (a) L. L. Miller, F. R. Stermitz and J. R. Falck, J. Am. Chem. Soc., 1973, 95, 2651; (b) J. R. Falck, L. L. Miller and F. R. Stermitz, Tetrahedron, 1974, 30, 931.
- 108. R. E. Gawley and G. A. Smith, Tetrahedron Lett., 1988, 29, 301.
- 109. I. W. Elliott, Jr., J. Org. Chem., 1977, 42, 1090
- 110. S. M. Kupchan, C.-K. Kim and K. Miyano, J. Chem. Soc., Chem. Commun., 1976, 91.
- 111. E. Zavarin, J. Org. Chem., 1958, 23, 1198.
- 112. (a) J. M. Bobbitt, R. Ebermann and M. Schubert, Tetrahedron Lett., 1963, 575; (b) B. Franck and G. Blaschke, Tetrahedron Lett., 1963, 569.
- 113. B. Franck and G. Blaschke, Justus Liebigs Ann. Chem., 1963, 668, 145.
- 114. (a) J. R. Lewis and B. H. Warrington, J. Chem. Soc., 1964, 5074; (b) J. R. Lewis and J. A. Vickers, Chem. Ind. (London), 1963, 779.
- 115. C. J. Brown, D. E. Clark, W. D. Ollis and P. L. Veal, Proc. Chem. Soc. London, 1960, 393; see also ref. 11.
- 116. D. A. Whiting and A. F. Wood, J. Chem. Soc., Perkin Trans. 1, 1980, 623.
- 117. E. A. Shearing and S. Smiles, J. Chem. Soc., 1937, 1931.
- 118. I. G. C. Coutts, D. J. Humphreys and K. Schofield, J. Chem. Soc. C, 1969, 1982.
- 119. (a) A. C. Day, J. Nabney and A. I. Scott, J. Chem. Soc., 1961, 4067; (b) D. Taub, C. H. Kuo, H. L. Slates and N. L. Wendler, Tetrahedron, 1963, 19, 1.
- 120. B. Feringa and H. Wynberg, Bioorg. Chem., 1978, 7, 397.
- 121. A. I. Scott, P. A. Dodson, F. McCapra and M. B. Meyers, J. Am. Chem. Soc., 1963, 85, 3702.
- 122. A. R. Forrester, R. H. Thomson and S.-O. Woo, J. Chem. Soc., Perkin Trans. 1, 1975, 2340.
- 123. D. H. R. Barton, Y. L. Chow, A. Cox and G. W. Kirby, J. Chem. Soc., 1965, 3571. 124. (a) R. F. Curtis, C. H. Hassall, D. W. Jones and T. W. Williams, J. Chem. Soc., 1960, 4838; (b) C. H. Hassall and J. R. Lewis, J. Chem. Soc., 1961, 2312.
- 125. S. Nishiyama, Y. Suzuki and S. Yamamura, Tetrahedron Lett., 1988, 29, 559.
- A. F. A. Wallis, Aust. J. Chem., 1973, 26, 585.
 A. Zanarotti, J. Chem. Res. (S), 1983, 306.
- 128. A. Nishiyama, H. Eto, Y. Terada, M. Iguchi and S. Yamamura, Chem. Pharm. Bull., 1983, 31, 2834.
- 129. K. Eskins, C. Glass, W. Rohwedder, R. Kleiman and J. Sloneker, Tetrahedron Lett., 1972, 861.
- K. V. Sarkanen and A. F. A. Wallis, J. Chem. Soc., Perkin Trans. 1, 1973, 1869.
 M. Iguchi, A. Nishiyama, Y. Terada and S. Yamamura, Tetrahedron Lett., 1977, 4511.
- 132. L. Merlini and A. Zanarotti, Tetrahedron Lett., 1975, 3621.
- 133. L.-J. Lin and G. A. Cordell, J. Chem. Soc., Chem. Commun., 1984, 160.
- 134. (a) L. Merlini, A. Zanarotti, A. Pelter, M. P. Rochefort and R. Hänsel, J. Chem. Soc., Chem. Commun., 1979, 695; (b) L. Merlini, A. Zanarotti, A. Pelter, M. P. Rochefort and R. Hänsel, J. Chem. Soc., Perkin Trans. 1, 1980, 775.
- 135. R. Schrall and H. Becker, Planta Med., 1977, 32, 27.
- 136. D. M. X. Donnelly, F. G. Murphy, J. Polonski and T. Prangé, J. Chem. Soc., Perkin Trans. 1, 1987, 2719.
- 137. R. S. Ward, Chem. Soc. Rev., 1982, 11, 75.
- 138. R. Ahmed, M. Lehrer and R. Stevenson, Tetrahedron, 1973, 29, 3753.
- 139. M. Iguchi, A. Nishiyama, H. Eto, Y. Terada and S. Yamamura, Chem. Lett., 1979, 1397; Chem. Pharm. Bull., 1983, 31, 2845.
- 140. N. J. Cartwright and R. D. Haworth, J. Chem. Soc., 1944, 535.
- 141. (a) E. C. Taylor, J. G. Andrade, G. J. H. Rall and A. McKillop, Tetrahedron Lett., 1978, 3623; (b) E. C. Taylor, J. G. Andrade, G. J. H. Rall, K. Steliou, G. E. Jagdamann, Jr. and A. McKillop, J. Org. Chem., 1981, 46, 3078.
- 142. R. Ahmed, F. G. Schreiber, R. Stevenson, J. R. Williams and H. M. Yeo, Tetrahedron, 1976, 32, 1339.
- 143. (a) E. E. Dickey, J. Org. Chem., 1958, 23, 179; (b) K. Freudenberg and H. Dietrich, Chem. Ber., 1953, 86, 4.
 144. R. Cooper, E. C. Levy and D. Lavie, J. Chem. Soc., Chem. Commun., 1977, 794.
- 145. H. Cousin and H. Herissey, C. R. Hebd. Seances, Acad. Sci., 1908, 147, 247; J. Pharm. Chim., 1908, 28, 93; Bull. Soc. Chim. Fr., 1908, 3, 1070.
- 146. Y. H. Kuo and S. T. Lin, Experientia, 1983, 39, 991
- 147. K. Freudenberg and H. H. Hübner, Chem. Ber., 1952, 85, 1181.
- 148. C. J. Sih, P. R. Ravikumar, F.-C. Huang, C. Buckner and H. W. Whitlock, Jr., J. Am. Chem. Soc., 1976, 98, 5412.
- 149. A. Ichihara, H. Kawagishi and S. Sakamura, Tennen Yuki Kagobutsu Toronkai Koen Yoshishu, 24th, 1981, 490 (Chem. Abstr., 96, 138 887h).
- 150. Y. H. Kuo, P. C. Kuo and S. T. Lin, Proc. Natl. Sci. Counc. Repub. China, Part B, 1983, 7, 28.
- 151. Y. Shizuri and S. Yamamura, Tetrahedron Lett., 1983, 24, 5011.
- G. W. Morrow and J. S. Swenton, *Tetrahedron Lett.*, 1987, 28, 5445.
 T. Takeya, Y. Akabane, E. Kotani and S. Tobinaga, *Chem. Pharm. Bull.*, 1984, 32, 31.
- 154. T. Takeya, E. Kotani and S. Tobinaga, J. Chem. Soc., Chem. Commun., 1983, 98.
- 155. A. Nishiyama, H. Eto, Y. Terada, M. Iguchi and S. Yamamura, Chem. Pharm. Bull., 1983, 31, 2820.
- 156. A. Nishiyama, H. Etoh, M. Iguchi and S. Yamamura, Agric. Biol. Chem., 1985, 49, 1197.
- 157. (a) O. L. Chapman, M. R. Engel, J. P. Springer and J. C. Clardy, J. Am. Chem. Soc., 1971, 93, 6696; (b) M. Matsumoto and K. Kuroda, Tetrahedron Lett., 1981, 22, 4437.
- 158. D. R. Nelan and C. D. Robeson, J. Am. Chem. Soc., 1962, 84, 2963.

- P. Schudel, H. Mayer, J. Metzger, R. Rüegg and O. Isler, Helv. Chim. Acta, 1963, 46, 636.
 H. Brockmann, Fortschr. Chem. Org. Naturst., 1957, 14, 141.
 L. R. Morgan, Jr. and C. C. Aubert, J. Org. Chem., 1962, 27, 4092.
 F. G. Holliman, B. A. Jeffery and D. J. H. Brock, Tetrahedron, 1963, 19, 1841.
 A. Najafi and M. Sainsbury, Heterocycles, 1977, 6, 459.
 M. A. Schwartz and P. T. K. Pham, J. Org. Chem., 1988, 53, 2318.
 G. Blaskó and G. A. Cordell, Heterocycles, 1988, 27, 445.

3.1 Wagner–Meerwein Rearrangements

JAMES R. HANSON University of Sussex, Brighton, UK

JES	718
ses of the Tetracyclic Diterpenoids	715
netic Syntheses	714
Expansion Reactions	713
ses Based on Camphor	710
S OF THE USE OF THE WAGNER-MEERWEIN REARRANGEN	ENT 710
ECTRONIC FEATURES OF THE WAGNER-MEERWEIN REAR	RANGEMENT IN 709
MEERWEIN PATHWAYS IN BICYCLIC SYSTEMS	706
IN OF THE WAGNER-MEERWEIN REARRANGEMENT	706
CTION	705

3.1.1 INTRODUCTION

Since the initial studies by Wagner at the turn of the century, the Wagner-Meerwein rearrangement has been the subject of almost continuous investigation.¹ The reaction was initially examined in terms of the rearrangement of α -pinene (1) to bornyl chloride (2; Scheme 1), and of camphene (3) via camphene hydrochloride (4) to isobornyl chloride (5).



Scheme 1

The ionic nature of the rearrangement was first recognized by Meerwein in 1922.² These ideas replaced some earlier suggestions by Ruzicka.³ However, in 1938 Bartlett questioned the intermediacy of the discrete ions proposed by Meerwein.⁴ In the following year Wilson, in a remarkable study for the period, suggested the possibility of a mesomeric delocalized carbocationic intermediate.⁵ The subsequent controversy between Brown and Winstein in the 1950s and 1960s on the role of classical or nonclassical ions in the norbornyl series meant that the rearrangement and the nature of the intermediates became one of the most widely studied reactions. It is, however, beyond the scope of this chapter to review these mechanistic arguments.⁶

3.1.2 DEFINITION OF THE WAGNER-MEERWEIN REARRANGEMENT

The Wagner-Meerwein rearrangement involves the generation of a carbocation on a bicyclic system followed by the 1,2-shift of an adjacent skeletal C--C bond to generate a new carbocation.^{7,8} The reaction is thus accompanied by a modification of the bicyclic skeleton. Although the Wagner-Meerwein rearrangement strictly involves the migration of a skeletal bond, the term is sometimes widened to embrace the migration of a substituent such as a methyl group as in the related Nametkin rearrangement. The latter may be exemplified by the conversion of camphenilol (6) to santene (7; Scheme 2)





Since its discovery in the context of the reactions of the monoterpenoids, examples of the rearrangement have been found in many other systems. Therefore, for the purposes of this chapter, which is concerned with the synthetic utility of the Wagner-Meerwein type of rearrangement, some reactions involving substituents migrating will be included as well as examples of skeletal rearrangement in polycyclic systems.

3.1.3 WAGNER-MEERWEIN PATHWAYS IN BICYCLIC SYSTEMS

The branched chain of the terpenoids and the consequent number of tertiary centers that these natural products contain lead to many cationic rearrangements. Not unexpectedly the majority of the examples



Scheme 3

of the Wagner-Meerwein rearrangement come within this area,^{9,10} and it is also here that its synthetic utility has been exploited. In the first instance, in order to assess the value of the reaction, it will be help-ful to examine the consequences of the rearrangement in terms of the reactions of a simple bicy-clo[2.2.1]heptyl (norbornyl) carbocation (8; Scheme 3).

These examples illustrate both the utility and the limitations of the rearrangement as part of a synthetic strategy. These may be summarized in the following terms. Firstly, there are many powerful methods with well-established regiospecificity that lead to the bridged ring system of defined stereochemistry, for example the Diels-Alder strategy (Scheme 4).



Scheme 4

Such reactions may form part of a convergent synthesis and by their very nature place a functional group adjacent to the bridge. Secondly, the rearrangement may lead to the possibility of placing a functional group in a relatively inaccessible position. For example the rearrangement of the epoxide (9) affords the primary alcohol (10; Scheme 5).



Scheme 5

However, the existence of many competing pathways as illustrated by Scheme 3 implies significant limitations on the generality of a strategy based on a particular Wagner-Meerwein rearrangement. Indeed the need to build features into a molecule which will drive the rearrangement along a specific pathway is a significant constraint. The tendency is for the carbocation to rearrange to the most stable structure. Thus 1,2-shifts tend to afford five- or six-membered rings rather than three- or four-membered rings. An example is given by the the rearrangement of the cyclobutanone (11), obtained from geronic acid, to the bicycloheptenone (12).¹¹ Secondly a bridge carbon bearing a positive charge or a double



bond is unstable. On the other hand conjugative or hyperconjugative stabilization may favor a particular carbocation. Furthermore a carbocation may collapse by fragmentation rather than by rearrangement as exemplified by the formation of α -terpineol (13) and limonene (14) from α - and β -pinene in the presence of aqueous mineral acid.¹²

The dehydration of fenchyl alcohol (15) demonstrates the many structural possibilities that a Wagner-Meerwein rearrangement may afford (see Scheme 6).¹³ The initial carbocation might in principle rearrange by the 1,2-shift of any one of six groups indicated on (16). Shift 1 would produce a bridgehead





(23)

(22)

carbocation, whereas shifts 3 or 6 would form less stable rings. On the other hand shifts 2 and 4 are feasible and produce α -fenchene (17) and ε -fenchene (18).

The ion (19) plays an important role in the formation of ε -fenchene; however, the reaction is more complex. In addition to the α - and ε -fenchenes, small amounts of cyclofenchene (20), β -fenchene (23), γ -fenchene (22) and δ -fenchene (21) are also formed. Cyclofenchene (20) represents the key to the formation of these substances (see Scheme 7). It may arise by the loss of a proton from the carbocation (24). Reprotonation of the cyclopropane ring may lead to a new carbocation from which the alkenes may be derived.

This intricacy, whilst revealing the limitations of a Wagner-Meerwein reaction, also indicates another aspect of the rearrangement as part of a synthetic strategy. The cyclopropyl system exemplified by cyclofenchene (20) may be accessed by a range of synthetic routes, for example the addition of diazo ketones across alkenes, and the resultant cyclopropane may then be subjected to the conditions of the Wagner-Meerwein rearrangement to afford a more stable ring system.

3.1.4 STEREOELECTRONIC FEATURES OF THE WAGNER-MEERWEIN REARRANGEMENT IN SYNTHESIS

The course of the Wagner-Meerwein rearrangement is markedly influenced by stereoelectronic effects.^{14,15} In particular for a concerted Wagner-Meerwein rearrangement to occur the leaving and migrating groups should be antiperiplanar — the so-called sp^3 -alignment factor. The rigid polycyclic triterpenoid and steroidal skeleta often provide a framework for a cascade of Wagner-Meerwein 1,2-shifts in which antiperiplanar groups, typically axially oriented hydrogen atoms and alkyl groups, migrate. These backbone rearrangements have been explored in many systems and are exemplified by the conversion of friedelin (25) to olean-13(18)-ene (26).¹⁶ Many other examples in which the reaction is initiated by the cleavage of an epoxide or the protonation of a tertiary alcohol are found in the steroid series (*e.g.* a C-5 alcohol, in the Westphalen rearrangement).



For a nonconcerted rearrangement to a carbocationic site to occur, the migrating group and the receptor *p*-orbital should ideally be in a plane. These features have been explored in the acetolysis of *exo*-twistbrendan-2-ol brosylate (27). sp^3 -Alignment clearly favors the migration of bond 'a', whilst sp^2 -alignment favors the migration of bond 'b'. Solvolysis in acetic acid buffered with potassium acetate gave products consistent with sp^3 bond alignment being the dominant factor.

The importance of stereoelectronic factors in a synthetic context have been examined as part of a program directed at the synthesis of quadrone (28). A series of model 4,3,2-propellanols (see Scheme 8) were examined and shown to undergo concerted rearrangements involving initial 1,2-alkyl shift of the best-aligned cyclobutyl bond.¹⁷ However, a number of the reactions were then accompanied by secondary rearrangements, again revealing limitations in the use of the rearrangement in a synthetic strategy. Nevertheless this approach has led to a successful synthesis of quadrone (28).¹⁸



3.1.5 EXAMPLES OF THE USE OF THE WAGNER-MEERWEIN REARRANGEMENT

3.1.5.1 Syntheses Based on Camphor

Since α -pinene is a readily available monoterpene, it forms the source of many monoterpenes, including those of the bornane skeleton through a Wagner-Meerwein rearrangement.¹⁹ Camphor (**29**) itself has formed the starting material for a number of syntheses in which Wagner-Meerwein rearrangements have played an important part (for an excellent detailed review see ref. 20). The propensity of this skeleton to undergo Wagner-Meerwein rearrangements enables substituents to be introduced at C-3, C-5, C-8, C-9 and C-10. An example of this is shown in the formation of camphor-10-sulfonic acid (**30**; Scheme 9).²⁰ A variety of other substituents may then be introduced in place of the sulfonic acid group.





Functionalization at C-9 (see Scheme 10) in camphor illustrates not only the complexity of the pathways when the possibility of Nametkin (2,3-methyl shift) and 2,6-hydride shifts are taken into account, but also one of the solutions to the simplification of the rearrangement.²¹ The product is a mixture of (+)-and (-)-camphor-9-sulfonic acids, *i.e.* the pathway allows for racemization. Indeed camphor itself can be racemized in concentrated sulfuric acid by a similar pathway involving both Wagner-Meerwein and



Nametkin rearrangements. A method of avoiding racemization was to use (+)-3-*endo*-bromocamphor (31), when only the (+)-3-*endo*-bromocamphor-9-sulfonic acid (32) was formed.²² This problem of racemization has arisen in other situations, for example in syntheses in the alkene series.²³



i, N₂H₄; ii, HgO, heat; iii, NaI, DMSO; iv, (Me₂C=CHCH₂NiBr)₂, DMF

The C-8-substituted derivatives of camphor are formed by a route from 3,3-dibromocamphor (33) outlined in Scheme 11.²⁴ This involves a series of Wagner-Meerwein rearrangements.

These variously substituted camphor derivatives have in turn been used in a number of syntheses of natural products, some of which also involve Wagner-Meerwein rearrangements in their later stages. Examples drawn from the syntheses of (+)- α -santalene (34),²⁵ (-)- β -santalene (35),²⁶ (-)-copacamphene $(36)^{26}$ and (-)-sativene $(37)^{26}$ are given in Schemes 12-14.



i, NaI, HMPA; ii, (CH₂OH)₂, H⁺; iii, (Me₂C=CHCH₂NiBr)₂, DMF; iv, Me₂CO, HCl; v, Na, PrⁱOH; vi, LiAl(OMe)₃H, THF; vii, TsCl, pyridine

Scheme 13



i, ClC₆H₄CO₃H, benzene; ii, Bu'OK, Bu'OH; iii, SOCl₂, pyridine; iv, H₂, Pt; v, LiAlH₄; vi, MeSO₂Cl, pyridine

Scheme 14

A number of aspects of longifolene chemistry may be understood in terms of Wagner-Meerwein rearrangements that have their parallel in the camphene series.²⁷ However, these rearrangements are often of a more deep-seated character, as in the conversion of longifolene (**38**) to isolongifolene (**39**), and have consequently found less application in a purely synthetic context. The intense activity that has surrounded sesquiterpenoid synthesis has involved many other examples of Wagner-Meerwein rearrangements.²⁸



3.1.5.2 Ring Expansion Reactions

The ready availability of four-membered rings of defined stereochemistry from enone-alkene photocycloadditions has been the origin of several syntheses which employ Wagner-Meerwein rearrangements in subsequent steps. This is exemplified by a neat and very short synthesis of α -caryophyllene alcohol (40) from cyclopentene and 3-methylcyclohexenone which was described some years ago (see Scheme 15).²⁹ The rearrangement in 40% sulfuric acid proceeded remarkably smoothly. A short synthesis of isocomene (41; Scheme 16) also illustrates this strategy.³⁰



i, hv; ii, Ph₃PCH₂; iii, TsOH, benzene

The defined geometry of these steps has permitted the synthesis of some cycloheptane acids related to terpenoid natural products. The key reaction was the rearrangement of a suitably substituted bicyclo[4.2.0]octane. Thus the photoaddition of ethylene to 3-methylcyclohexenone gave the ketone (42), which was converted to the alcohol (43). On treatment with HgO and HBF₄ this gave the unstable hydroxy aldehyde (44), which was readily oxidized to the dicarboxylic acid (45; Scheme 17).³¹ Ring expansion methodology was also used in an approach to the synthesis of the trichothecenes (see Scheme 18).³²



i, 1,3-dithiane, BuLi; ii, HgO, HBF₄, aq. THF; iii, CrO₃, H₂SO₄







Scheme 18

Whereas this synthesis places an oxygen function on a bridge, an oxygen function of this type may also be the source of a ring contraction as exemplified by the conversion of (46) to (47).³³



3.1.5.3 Biomimetic Syntheses

A number of biomimetic syntheses have included Wagner-Meerwein rearrangements.³⁴ A chemical conversion of humulene (48) to sterpurene (50) involved an interesting series of Wagner-Meerwein rearrangements (see Scheme 19).³⁵ Humulene (48) was converted to the cyclooctenol (51) and thence to the bromide (52) via the protoilludyl cation (49). Treatment of (52) with silver acetate in acetic acid gave racemic sterpurene (50). In contrast the epimeric bromide (53) gave (54).



Scheme 19

3.1.5.4 Syntheses of the Tetracyclic Diterpenoids

A possible biogenetic relationship between the pimaradienes and the tetra- and penta-carbocyclic diterpenoids was suggested by Wenkert.³⁶ Implicit in this scheme (see Scheme 20) are a number of rearrangements. There have been many investigations aimed at simulating aspects of this scheme. In particular the beyerene-kaurene rearrangement takes place quite readily. Thus solvolysis of the beyeran-16 β -yl toluene-*p*-sulfonate (**55**) affords a mixture of kaurene (**57**), isokaurene (**58**), kauranol (**59**), and beyer-15-ene (**56**; Scheme 21).³⁷

An allied rearrangement of the epoxide (60) has been used in partial syntheses in both the kaurene and gibberellin series to place oxygen functions at the relatively inaccessible C-14 position (*e.g.* 61).³⁸ This type of rearrangement was also used in the final stages of one of the early gibberellin syntheses.^{39,40}

The reverse rearrangement, involving the conversion of the kaurene skeleton to that of the beyerene series, is found in the steviol-isosteviol and gibberellin-8,13-isogibberellin rearrangements (see Scheme 22).⁴¹ Apart from their mechanistic interest, these rearrangements were used in partial syntheses directed at structural correlations.

Prior to the advent of simple methods for radical deoxygenation, a double Wagner-Meerwein rearrangement (Scheme 23) was used as a method for removing a bridgehead hydroxy group in this series.⁴² The success of this sequence indicates the subtle balance of stabilities in this series.

Interrelationships between the kaurene and atiserene series have also been established by a Wagner-Meerwein rearrangement. For example treatment of isokaurene epoxide (62) with acid gave atisiran-15-one (63), possibly through the sequence shown in Scheme $24.^{43}$







Scheme 20











Scheme 21

BF3•Et2O

{













Scheme 22

н⁺



Scheme 24

The biologically important diterpenoid aphidicolin (64) and its relatives stemodin and maritimol have been the targets of several synthetic studies employing Wagner-Meerwein rearrangements at key steps in the construction of the carbon skeleton. One such strategy involves the solvolytic rearrangement of suitably oriented bicyclo[2.2.2]octenyl methanesulfonates exemplified by Scheme 25.⁴⁴ A synthesis of maritimol involved a similar [2.2.2] \rightarrow [3.2.1] skeletal rearrangement of (65) to (66).⁴⁵ The rearrangement was induced by treatment of the alcohol with excess toluene-*p*-sulfonyl chloride in pyridine.



In conclusion, the Wagner-Meerwein rearrangement can be a useful component of a synthetic strategy provided its stereoelectronic requirements can be met and the possible pathways are limited. It has been of particular value in the synthesis of bridged ring systems in which the strategy requires the modification of the skeletal framework.



Scheme 25

3.1.6 REFERENCES

- 1. G. Wagner and W. Brickner, Chem. Ber., 1899, 32, 2307.
- 2. H. Meerwein and K. Van Emster, Chem. Ber., 1922, 55, 2500.
- 3. L. Ruzicka, Helv. Chim. Acta, 1918, 1, 110.
- P. D. Bartlett and I. Pöckel, J. Am. Chem. Soc., 1938, 60, 1585.
 T. P. Nevell, E. de Salas and C. L. Wilson, J. Chem. Soc., 1939, 1188.
- 6. H. C. Brown, Acc. Chem. Res., 1973, 6, 377; 1983, 16, 432; S. Winstein, Q. Rev., Chem. Soc., 1969, 23, 141; G. A. Olah, Acc. Chem. Res., 1976, 9, 41; G. A. Olah, G. K. S. Prakash and M. Saunders, Acc. Chem. Res., 1983, 16, 440; W. Kirmse, Acc. Chem. Res., 1986, 19, 36.
- 7. J. A. Berson, in 'Molecular Rearrangements', ed. P. de Mayo, Wiley-Interscience, New York, 1963, vol. 1, p. 111.
- 8. W. F. Erman, 'Chemistry of the Monoterpenes', Dekker, New York, 1985.
- 9. P. de Mayo, 'Mono and Sesquiterpenoids', Interscience, New York, 1959, chap. 4.
- 10. J. F. King and P. de Mayo, in 'Molecular Rearrangements', ed. P. de Mayo, Wiley-Interscience, New York, 1964, vol. 2, p. 771.
- 11. J. J. Beereboom, J. Org. Chem., 1965, 30, 4230.
- 12. See ref. 8, p. 965 et seq. for a summary.
- T. S. Sorensen, Acc. Chem. Res., 1976, 9, 257, and refs. therein.
 A. Nickon and R. C. Weglein, J. Am. Chem. Soc., 1975, 97, 1271.
- 15. P. Kočovský, F. Tureček, V. Langer, J. Podlahová and J. Podlaha, J. Org. Chem., 1986, 51, 4888, and refs. therein.
- E. J. Corey and J. J. Ursprung, J. Am. Chem. Soc., 1956, 78, 5041.
 A. B. Smith, III, B. A. Wexler, C.-Y. Tu and J. P. Konopelski, J. Am. Chem. Soc., 1985, 107, 1308.
- 18. A. B. Smith, III and J. P. Konopelski, J. Org. Chem., 1984, 49, 4094.
- 19. D. H. Grayson, Nat. Prod. Rep., 1988, 5, 419, and previous reports.
- 20. T. Money, Nat. Prod. Rep., 1985, 2, 253.
- A. M. T. Finch, Jr. and W. R. Vaughan, J. Am. Chem. Soc., 1969, 91, 1416. 21.
- 22. P. Cachia, N. Darby, C. R. Eck and T. Money, J. Chem. Soc., Perkin Trans. 1, 1976, 359.
- 23. J. E. Baldwin and T. C. Barden, J. Am. Chem. Soc., 1983, 105, 6656.
- 24. W. M. Dadson and T. Money, J. Chem. Soc., Chem. Commun., 1982, 112.
- E. J. Corey and M. F. Semmelhack, J. Am. Chem. Soc., 1967, 89, 2755.
 C. R. Eck, G. L. Hodgson, D. F. MacSweeney, R. W. Mills and T. Money, J. Chem. Soc., Perkin Trans. 1, 1974, 1938.
- 27. S. Dev, Fortschr. Chem. Org. Naturst., 1981, 40, 49.
- 28. M. Vandewalle and P. de Clercq, Tetrahedron, 1985, 41, 1767.
- 29. E. J. Corey and S. Nozoe, J. Am. Chem. Soc., 1964, 86, 1652.
- 30. M. C. Pirrung, J. Am. Chem. Soc., 1981, 103, 82.
- 31. B. C. Ranu and D. C. Sarkar, J. Chem. Soc., Chem. Commun., 1988, 245.
- 32. J. D. White, T. Matsui and J. A. Thomas, J. Org. Chem., 1981, 46, 3376. 33. J. W. Wilt, V. A. Curtis, L. N. Congson and R. Palmer, J. Org. Chem., 1984, 49, 2937.
- 34. R. M. Coates, Fortschr. Chem. Org. Naturst., 1976, 33, 73.

- Y. Murata, T. Ohtsuka, H. Shirahama and T. Matsumoto, *Tetrahedron Lett.*, 1981, 22, 4313.
 E. Wenkert, *Chem. Ind. (London)*, 1955, 282.
- 37. R. R. Sobti and S. Dev, Tetrahedron Lett., 1966, 3939; R. A. Appleton, P. A. Gunn and R. McCrindle, J. Chem. Soc. C, 1970, 1148. 38. J. R. Hanson, Tetrahedron, 1967, 23, 793.

- K. Mori, M. Shiozaki, N. Itaya, M. Matsui and Y. Sumiki, *Tetrahedron*, 1969, 25, 1293.
 For a review see: L. N. Mander, Acc. Chem. Res., 1983, 16, 48; Nat. Prod. Rep., 1988, 5, 541.
- 41. J. R. Hanson, 'The Tetracyclic Diterpenoids', Pergamon Press, Oxford, 1968; more recent refs. to this area are given in J. R. Hanson, Nat. Prod. Rep., 1989, 6, 347, and previous reports.
- B. E. Cross, J. R. Hanson and R. N. Speake, J. Chem. Soc., 1965, 3555.
 J. MacMillan and E. R. H. Walker, J. Chem. Soc., Perkin Trans. 1, 1972, 1274.
- 44. D. Bravetti, R. M. Bettolo and A. Lupi, Helv. Chim. Acta, 1982, 65, 371.
- 45. E. E. van Tamelen, J. G. Carlson, R. K. Russell and S. R. Zawacky, J. Am. Chem. Soc., 1981, 103, 4615.

3.2 The Pinacol Rearrangement

BRUCE RICKBORN

University of California, Santa Barbara, CA, USA

3.2.1	INTRODUCTION	721
3.2.2	MECHANISTIC FEATURES	722
3.2.3	APPLICATIONS	726
3.2.4	REFERENCES	731

3.2.1 INTRODUCTION

The acid-catalyzed reaction of vicinal diols, which results in dehydration and the migration of an alkyl or aryl group, or a hydrogen atom, to form an aldehyde or ketone is known as the Pinacol rearrangement.¹ The name is derived from the material used in the earliest recorded example. Although the structures of both the starting material and product were unknown when the experiment was published in 1860, Fittig² found that treatment of pinacol (1) with sulfuric acid gave pinacolone (2; equation 1).



All classes of vicinal diols (primary, secondary, tertiary, alkyl- or aryl-substituted) will undergo the pinacol rearrangement, and many acids and solvents have been used for this purpose. Various procedural modifications have been introduced over the years for particular glycols, but sulfuric acid remains the most commonly employed catalyst. The use of 25% H₂SO₄, as recommended in the procedure of Adams,³ affords pinacolone in essentially quantitative yield. In some instances better results are obtained when cold concentrated acid is used as the solvent. The choice of reagents and conditions is important, and can completely alter the course of the reaction. For example, pinacol also serves as the starting material for the synthesis of 2,3-dimethylbutadiene, formed by slow distillation of a mixture of the diol and catalytic HBr.⁴

The pinacol rearrangement and the acid-catalyzed rearrangement of epoxides (see this volume, Chapter 3.3) may be closely related, depending upon the substrates, reagents and conditions employed. Epoxides have on occasion been isolated from diols under pinacol rearrangement conditions,⁵ and they are implicated as reactive intermediates in many other instances. Similarly, the treatment of an epoxide with aqueous acid may cause opening to the vicinal diol either prior to, or in competition with, rearrangement. The two substrates (diol and epoxide) are equivalent for most purposes if the equilibrium shown in Scheme 1 is established more rapidly than rearrangement can occur. The use of strong acids and substrates which bear cation-stabilizing groups will encourage the merger of the two processes shown in Scheme 1 via a common carbenium ion intermediate. Conversely, mild acids and good nucleophiles will favor covalent bond-forming reactions, *e.g.* the formation of a halohydrin upon treatment of an epoxide

Rearrangement Reactions

with HX. These covalently bonded materials may be stable and isolable, or may serve as reactive intermediates which lead to rearranged products under the conditions used for their formation. The rearrangement step itself can be regarded as the interaction of an intramolecular nucleophile (the migrating group and its bonding electron pair) with an adjacent electrophilic center. The latter may range from a formal carbocation to a carbon which bears a leaving group suitable for S_N2 displacement. These distinctions can be helpful in designing experiments to take advantage of the versatility of epoxides as starting materials.



Scheme 1

Other methods of generating β -hydroxy carbocations lead to similar or identical products. The reaction of β -amino alcohols with nitrous acid, one variant of the semipinacol rearrangement (or Tiffeneau–Demjanov ring expansion if used for this purpose), is a familiar example.⁶ Another is the so-called α -ketol rearrangement, in which the hydroxy and carbonyl groups of an acyloin exchange positions with concurrent migration of a substituent.⁶ Many of the general features described here also apply to these related reactions (see this volume, Chapter 3.4).

3.2.2 MECHANISTIC FEATURES

Given the range of substrates and reagents that have been used in pinacol rearrangements, it is not surprising that very few mechanistic conclusions are general for all diols under all conditions. Indeed, the same substrate may give different products when subjected to apparently similar conditions. A classic illustration is provided by the diol (3), which gives the ketone (4) when treated with concentrated sulfuric acid, but mostly the aldehyde (5) when 40% acid is used (equation 2). Close mechanistic examination of this reaction with ¹⁴C-labeled material showed that the kinetically controlled ratio of (4) to (5) does depend to some extent on the solvent and acid employed, but is further complicated by the conversion of the aldehyde (5) to the ketone (4) under the reaction conditions.


Other detailed mechanistic examples are described in two reviews by Collins.^{6,7} Although published two decades ago, these require little updating in content, and appear to be unchallenged in general mechanistic conclusions, which can be summarized as follows: (i) pinacol rearrangements are exclusively intramolecular; (ii) the course of the pinacol rearrangement usually depends on which OH group is most easily removed, as predicted by consideration of the stability of the intermediate carbenium ion; (iii) the group which preferentially migrates is that which is better able to stabilize a positive charge, although complications may arise due to stereochemical and conformational effects, rapid equilibration of intermediate carbenium ions or interconversion of products; and (iv) either inversion or racemization may be observed at the migration terminus; racemization may occur either at the stage of an intermediate cation or by further reaction of the product.

It appears that the stereochemistry at the migrating center has not been explicitly examined in a pinacol rearrangement but, as seen in related 1,2-migrations, retention of configuration is expected.

Pinacol rearrangements are effectively irreversible in the sense that carbonyl products have not been shown to revert to diols. However, the products may be in equilibrium with the same cationic intermediates that are generated from the diol. This feature emerges clearly under strongly acid conditions, where the 1,2-carbonyl transposition of aldehydes and ketones is observed (*cf.* equation 2). An especially pertinent example comes from the work of Fry,⁸ who found that pinacolone itself undergoes the ¹⁴C-label redistribution shown in equation (3). Benzpinacolone, isotopically labeled at the carbonyl carbon, gives material with the label transposed as shown in equation (4).⁹ These and related observations can be explained by invoking two or more carbonium ion intermediates connected by 1,2-migrations. The label exchange in equation (4) requires oxygen migration, presumably *via* an epoxide (or protonated epoxide) intermediate.



Many other 1,2-carbonyl transpositions with skeletal rearrangement are known.⁶ In general, aldehydes react more rapidly than ketones; the rearrangement of a ketone to form an aldehyde appears to be unknown, but ketones can be converted to other ketones. A few examples from the early work of Venus-Danilova¹⁰ are illustrated. Thus cyclobutane- and cyclopentane-carbaldehyde both gave the simple ring-expanded ketones as shown in equations (5) and (6), upon treatment with concentrated sulfuric acid.



A more complex sequence, such as that suggested in Scheme 2, must be invoked to explain the formation of the ring-contracted methyl cyclopentyl ketone from cyclohexanecarbaldehyde.¹⁰

In general, 1,2-carbonyl transpositions require more vigorous conditions (stronger acid, higher temperatures) than pinacol rearrangements in which the same carbenium ion intermediates may be generated. However, harsher conditions than actually required have frequently been used to carry out pinacol rearrangements, and care must be taken to avoid over-interpretation of the mechanism from examination of the products.

The evidence for carbocation intermediates in the pinacol rearrangement is compelling for some substrates and conditions. Bunton¹¹ found that pinacol itself that was recovered from ¹⁸O-enriched aqueous sulfuric acid had incorporated an appreciable amount of solvent oxygen. Under these conditions the reaction goes to completion, *i.e.* the incorporation cannot be due to reversible formation of diol from pina-



Scheme 2

colone. A common carbenium ion intermediate was proposed for both solvent incorporation and rearrangement. Similarly, Collins¹² found that the *threo* and *erythro* isomers of a triaryl-substituted diol, when treated with sulfuric acid in aqueous ethanol, were interconverted more rapidly than either isomer rearranged.

An early report that the *threo* and *erythro* isomers of 1,2-diphenyl-1,2-di(1-naphthyl)ethanediol gave two different products (a ketone and an unidentified substance)¹³ could not be substantiated by other workers,¹⁴ who also found that the *threo* and *erythro* isomers of 1,2-bis(2-chlorophenyl)-1,2-diphenylethanediol both gave the same ketone (phenyl migration).¹⁴ Yields were not recorded, precluding substantive mechanistic interpretation.

The cis (6) and trans (7) isomers of 1,2-dimethyl-1,2-cyclohexanediol have been examined by several research groups, but a clear understanding of this system did not emerge until modern analytical tools were employed. The very thorough study of Bunton and Carr¹⁵ provides evidence for a carbenium ion intermediate common to both isomers. In ca. 1–3 M aqueous perchloric acid at 60 °C the isomeric diols are partially interconverted, and solvent oxygen is incorporated. The ring-contracted ketone (8) is the major product from both diols, but a small amount of 2,2-dimethylcyclohexanone (9) is also formed (equation 7). A carbenium ion intermediate with sufficient lifetime to undergo chair-chair equilibration would lead to identical product ratios from both diol isomers. The slightly different ratios which were found (from (6), (8):(9) = 93:7; from (7), (8):(9) = 97:3) presumably reflect a small stereochemical memory effect in one or both of the isomers, but the initial stereochemistry of the center which bears the departing hydroxy group clearly does not dominate the stereochemical course of these reactions. Larger differences in reactivity and product composition are found for the five-membered ring analogs.¹⁶



Earlier studies¹⁷ of these diols, with one exception¹⁸ which was later shown^{17b} to be erroneous, differ only marginally in stereochemical conclusions, probably because the minor isomer could not be detected by the analytical methods used.

Another aspect of stereoselectivity has been probed by Berti and coworkers,¹⁹ who used the *t*-butyl group to control the conformational populations of the diols depicted in equations (8) to (11). The results are informative, although atypical conditions ($BF_3 \cdot OEt_2$ in benzene) for pinacol rearrangement were used, and the generality of the observations is unknown. The products of all four reactions are ascribed to initial loss of the benzylic OH group, with formation of a planar carbenium ion at C-1. The close similarity of the ratios of products for equations (8) and (9), or equations (10) and (11), respectively, supports the view that the departing OH group has at most a very slight influence on the outcome. Formal back-side (equation 8) and frontside (equation 9) hydride migrations are involved in the formation of (11) as the exclusive ketone from (10) and (13), while (15) is the only ketone formed from the axial 2-hydroxy substrates (equations 10 and 11). The formation of the single stereoisomer of aldehyde (12) in all four reactions is in keeping with ring contraction *via* the planar carbenium ion held in the chair conformation of the starting materials.



Carbenium ion intermediates are implicated not only for diols in which one or both carbinol centers are tertiary, but also in less-substituted substrates. Alexander and Dittmer²⁰ found that the meso and (\pm) isomers of 2,3-butanediol both gave 2-butanone and isobutyraldehyde when refluxed in 60% phosphoric acid, but the ratios were different, ruling out exclusive reaction via a single common intermediate. More recently, Herlihy²¹ has reported the results of a thorough kinetic analysis of the rearrangement of optically active 1,2-propanediol in aqueous perchloric acid. The rate of racemization exceeded the rate of rearrangement, in keeping with a mechanism involving a carbenium ion intermediate. The rate of exchange of ¹⁸O-enriched substrate was also examined, and found to be half as fast as racemization; a significant memory effect, e.g. shielding of one face of the cation by the departing water molecule, can be used to rationalize this rather S_N2-like behavior. Arguments against the involvement of an epoxide intermediate were presented, although rigorous exclusion of any role for epoxide is difficult. The ratio of racemization/rearrangement was found to be dependent on the acid concentration, an effect attributed to the influence of acidity on partitioning of the carbocation. The mechanistic conclusions for this system parallel those reached earlier¹¹ for pinacol itself. The nearly exclusive product from 1,2-propanediol is propanal (which does not survive the reaction conditions), but interestingly, a small amount of acetone is also formed.

Rearrangement Reactions

Although inversion at the migration terminus has been observed in several pinacol rearrangements, the dramatic rate effects (anchimeric assistance) sometimes associated with neighboring group participation have not been observed or documented. The choice of a proper model to establish a baseline rate for comparison remains a problem. Alcohols are not suitable for this purpose, since diols are significantly less reactive than their monohydroxy analogs, presumably because of the inductive effect of the second OH group.^{11,21}

Most detailed studies of pinacol rearrangements support the view that carbenium ion formation precedes migration, but occasionally stereoselectivity is observed when not anticipated from this perspective. The results of three separate labeling and stereochemical experiments carried out by Collins *et al.*²² are incorporated in the depiction (equation 12) of the rearrangement of diol (17) to the ketone (19). The course of this reaction can be explained by considering the bridged ion (18) as an intermediate, even though the tertiary and benzylic nature of the corresponding classical ion makes it a poor candidate for nonclassical participation.



Aqueous or concentrated acids are the most commonly used catalysts, but several other acidic materials can cause pinacol rearrangement, as noted in a recent review of the dehydration reactions of diols.²³

Few generalizations can be made about the 'migratory aptitudes' of substituents in pinacol rearrangements, but some observations are worth noting. Phenyl and hydrogen appear to be similar in rate of migration, and hydrogen is some 20 times faster than methyl. The *t*-butyl group has been reported to shift much more rapidly than other simple alkyl groups (*t*-butyl:ethyl:methyl = >4000:17:1).²⁴ Cyclopropyl substituents migrate more readily than simple alkyl groups, and do so without ring opening, as illustrated in equation (13).²⁵



 $\mathbf{R} = \mathbf{M}\mathbf{e}, \mathbf{P}\mathbf{r}^{i}, \mathbf{cyclopropyl}$

3.2.3 APPLICATIONS

In order to maximize the synthetic utility of pinacol rearrangements, it is necessary to exclude competing processes, and control both regio- and stereo-selectivity. The two factors which influence regioselectivity — which OH is lost, and which adjacent group migrates — are at least marginally predictable and, in favorable circumstances, controllable. Control of stereoselectivity is more problematic, as noted above. Because of these complications and the difficulties associated with the preparation of complex diols, the pinacol rearrangement has not been widely used in multistep syntheses, in spite of its capacity for generating some unusual structural features.

Both ring expansions and contractions are well documented. Ring size can strongly influence the course of the reaction. For example, $Vogel^{26}$ found that the diol (20), obtained by reductive dimerization of cyclobutanone, cleanly rearranged to the ketone (21) on warming in dilute sulfuric acid; (21) was subsequently used to prepare the novel pentalene (22; equation 14).



The symmetrical diols derived from cyclopentanone²⁷ and cyclohexanone²⁸ can similarly be converted to ring-expanded ketones in good yield. The diols from reductive coupling of cycloheptanone and cyclooctanone give mainly the corresponding dienes in aqueous acid,²⁸ especially when heated, but Christol²⁹ found that pinacol rearrangement is strongly favored even for these materials when cold concentrated sulfuric acid is used as the solvent.

Results which follow the mechanistic generalizations given above have been obtained from examination of similar unsymmetrical diols. The formation of (24) as the major product from (23) can be explained by more rapid development of the carbenium ion in the five-membered ring (equation 15).



Other unsymmetrical diols, prepared (inefficiently) by the coupling of mixtures of ketones, have been studied.²⁸⁻³⁰ The 'preferred cation' argument is also found to be applicable to mixed medium-ring analogs, *i.e.* the kinetically favored pinacol rearrangement product is that predicted by consideration of the relative stabilities of the two possible, initially formed, carbocations.³⁰ Earlier work²⁸ with these compounds may have given misleading results due to product instability. To illustrate, Mundy³⁰ observed that (**24**) is converted to (**25**) as the temperature is increased.

The *cis*- and *trans*-hydrindandiols (26) are both converted exclusively to the spiro ketone (27) on brief treatment with cold concentrated sulfuric acid (equation 16).³¹ Plausible alternative products, the epoxide and spiro[5.3]nonanone, could not be ruled out as intermediates, since both were prepared by independent routes and found to give (27) under the reaction conditions.



The pinacol rearrangements of several cyclobutanediols, mostly induced by BF₃·OEt₂, have been examined by Conia.³² The reactions of symmetrically substituted materials occurred with high specificity and yield. Both isomers (*cis* and *trans*) gave the same product in instances where this question was examined. As shown in equation (17), for the substituents R = H, alkyl or phenyl, only ring contraction was observed, while equally regiospecific formation of the cyclobutanone product was found for the substituents R =allyl or vinyl.



The strained four-membered ring has been used in a different way by Kuwajima.³³ Bis(trimethylsilyl)succinoin (28) when treated with benzaldehyde in the presence of TiCl₄ afforded the intermediate (29), which gave the ring-expanded product (30) on treatment with trifluoroacetic acid (equation 18). An

analogous reaction of (28) with ketals, catalyzed by BF₃·OEt₂, led to good yields of 2,2-disubstituted-1,3-cyclopentanediones. These reactions illustrate the facile migration of an acyl group.



An unusual observation was made when diol (31) was treated with $BF_3 \cdot OEt_2$. The ring-enlarged terpene karahanaenone (32) was formed (along with a double-bond positional isomer); the authors³⁴ reported that other catalysts gave little or no ring-expanded product (equation 19).



When one of the diol centers bears a vinyl group, a preference for initial formation of an allylic carbenium ion is anticipated. Oppolzer and coworkers³⁵ made use of this feature to develop a cyclohexenone 1,2-carbonyl transposition reaction, illustrated by the sequence shown in equation (20). Oxidation of (33) with Pb(OAc)₄ was used to introduce the acetoxy group in (34); subsequent reduction or addition of an organometallic reagent (*e.g.* R'Li) followed by hydroysis gave the diol (35). The pinacol rearrangement was best catalyzed by *p*-toluenesulfonic acid in refluxing benzene, which also caused migration of the double bond into conjugation. Moderate overall yields of the transposed cyclohexenone (36) were obtained.



Dana *et al.*³⁶ examined some similarly unsaturated acyclic diols under pinacol rearrangement conditions. With few exceptions these gave rather complex mixtures of products, and it appears that the synthetic utility of this method is limited.

Some diols of heterocycles have been subjected to pinacol rearrangement. Mundy³⁷ found that (37) gave only (38) when treated with cold concentrated sulfuric acid (equation 21). It was noted that (38) is *not* the product expected from initial cation stability arguments based on anticipated heteroatom dipole effects. The basis for this selectivity remains unclear. An attempt to detect alternative products at short reaction times failed to shed light on this question.



Diols obtained by osmium tetroxide treatment of porphyrins constitute some of the more complex structures which have been subjected to pinacol rearrangement.³⁸ These reactions left the polycyclic framework intact, and the oxochlorin products were formed by rearrangements which followed normal migratory aptitudes (*e.g.* H, Et > Me).

The diol (39) gave (40; 49%) when treated with $ZnCl_2$ in acetic anhydride (equation 22).³⁹ Interestingly, the analogous epoxide (41) gave the alternative expanded heterocycle azepinone (42) when subjected to BF₃ OEt₂ in benzene (equation 23).



The distinction between the pinacol and semipinacol rearrangements is blurred in instances where the starting material is a diol, but steps are taken to form a derivative in order to control the regiochemistry of the reaction. For example, it is sometimes possible to take advantage of differential alcohol reactivity (primary > secondary > tertiary) to form a specific derivative, such that solvolysis will lead to migration to the less-substituted center.^{7b} Corey *et al.*⁴⁰ used this approach to effect the ring expansion of (43) to (44) (note the preference for vinyl over alkyl migration), a key step in a synthesis of the sesquiterpene longifolene (equation 24).



Migration to a primary center has also been effected, as shown by the conversion of (45) to (46; equation 25).⁴¹ Conditions for these reactions are quite mild, consisting of solvolysis in tetrahydro-furan/LiClO₄, with CaCO₃ present to neutralize the *p*-toluenesulfonic acid that is formed.



Rearrangement Reactions

Tsuchihashi and coworkers used methanesulfonate esters in a novel variant.⁴² An unusual feature is the use of alanes (Et₃Al or Et₂AlCl) in CH₂Cl₂ solvent at low temperatures to promote the rearrangement. Aryl and vinyl groups migrated more rapidly than alkyl substituents, and the more acidic Et₂AlCl was needed for good results with the latter. Some noteworthy features of this procedure are: (i) migration of vinyl groups with retention of (E,Z)-configuration; (ii) clean inversion of configuration at the migration terminus; and (iii) the formation of rather sensitive optically active materials, as illustrated in equation (26).^{42b} The authors invoke a 'push-pull' mechanism involving initial formation of an aluminum alkoxide by reaction at the free hydroxy group.



A similar reaction of vicinal diol monoacetates promoted by alanes has been reported.⁴³ Higher temperatures are needed with these substrates, and in some instances the rearrangement product (ketone) was alkylated (or alkynylated) by the alane promoter. The use of Et₂AlSPh avoided this complication.

Bicyclic systems have provided an interesting framework for examination of stereochemical preferences in the rearrangements of diol monoethers. Monti *et al.*⁴⁴ found that the *anti* (double bond and hydroxy group) isomer of the [2.2.2] bicyclic substrate (49) gave (50) stereospecifically, whereas the *syn* isomer gave a mixture of products (equation 27).



Surprisingly, the [3.2.2] bicyclic analogs gave contrasting results, with the *syn* isomer (51; equation 28) rearranging cleanly to (52), whereas the *anti* isomer gave a mixture derived from migration of both the saturated and unsaturated two-carbon bridges.⁴⁵ The remarkable net retention at the migration terminus in the formation of (52) has been ascribed to conformational effects, which also caused the *syn* substrate to react more rapidly that its *anti* counterpart.



Although its utility appears to be limited, Mukaiyama *et al.*⁴⁶ found that the chloropyrimidinium fluorosulfate (53) can effect both functionalization and subsequent pinacol rearrangement in a one-pot procedure. Unsymmetrical secondary-tertiary diols give the products expected of derivatization at the secondary hydroxy group. A major side reaction is elimination.



3.2.4 REFERENCES

- 1. Recent Chem. Abstr. subject indexes list this topic under the heading 'Rearrangement, pinacol', it has also been called the 'pinacol-pinacolone rearrangement', or (rarely) the 'pinacolone rearrangement'. 2. R. Fittig, Justus Liebigs Ann. Chem., 1860, 114, 54. The structure of pinacol (hydrate) was a matter of
- controversy; pinacolone was clearly formed in this work, as judged by descriptions of the odor, b.p., and C, H analysis (the analysis gave correct percentages, but the resultant formula was in error because atomic weights C = 6 and O = 8 were used).
- 3. R. Adams and E. W. Adams, Org. Synth., 1925, 5, 87.
- 4. C. F. R. H. Allen and A. Bell, Org. Synth., Coll. Vol., 1955, 3, 312.
- 5. (a) An interesting and sometimes overlooked example is found in very early work of McKenzie et al. The diol formed from addition of MeMgI to (-)-benzoin gave racemic 1,2-diphenyl-1-propanone on treatment with conc. H₂SO₄, whereas dilute acid treatment afforded optically active (-)-epoxide. The latter in turn gave the racemic ketone on further acid-catalyzed rearrangement. Racemization probably occurs by achiral enol formation after rearrangement to the ketone in this instance. See A. McKenzie, R. Roger and G. O. Wills, J. Chem. Soc., 1926, 779; (b) H. J. Gebhart, Jr. and K. H. Adams, J. Am. Chem. Soc., 1954, 76, 3925; (c) K. Matsumoto, Tetrahedron, 1968, 24, 6851; (d) S. Wold, Acta Chem. Scand., 1969, 23, 1266; (e) Y. Pocker and B. P. Ronald, J. Am. Chem. Soc., 1970, 92, 3385.
- (a) C. J. Collins and J. F. Eastham, in 'The Chemistry of Functional Groups; The Carbonyl Group', ed. S. Patai, Wiley-Interscience, New York, 1966, chap. 15. This excellent review, which reflects the authors' mechanistic interests, covers several topics in addition to the pinacol rearrangement (e.g. semipinacol, α -ketol, etc.); (b) the 'semipinacol' terminology was first used by Tiffeneau around the turn of the century to describe rearrangement with migration towards the secondary carbinol center of a secondary/tertiary glycol. The name is now more commonly used to describe many variants involving leaving groups other than OH.
- 7. C. J. Collins, Q. Rev., Chem. Soc., 1960, 14, 357.
- T. S. Rothrock and A. Fry, J. Am. Chem. Soc., 1958, 80, 4349.
 A. Fry, W. L. Carrick and C. T. Adams, J. Am. Chem. Soc., 1958, 80, 4743.
- E. D. Venus-Danilova, Zh. Obshch. Khim., 1936, 6, 697, 917, 1757; E. D. Venus-Danilova and V. F. Kazimirova, Zh. Obshch. Khim., 1938, 8, 1438. For a more recent discussion of carbonyl transposition reactions see refs. 6-9; P. Bauer and J. E. Dubois, J. Am. Chem. Soc., 1976, 98, 6999; and H. D. Zook, W. E. Smith and J. L. Greene, J. Am. Chem. Soc., 1957, 79, 4436.
- C. A. Bunton, T. Hadwick, D. R. Llewellyn and Y. Pocker, J. Chem. Soc., 1958, 403.
 C. J. Collins, W. T. Rainey, W. B. Smith and I. A. Kaye, J. Am. Chem. Soc., 1959, 81, 460.
 W. E. Bachmann and R. V. Shankland, J. Am. Chem. Soc., 1929, 51, 306.
- 14. E. Bergmann and W. Schuchardt, Justus Liebigs Ann. Chem., 1931, 487, 225.
- 15. C. A. Bunton and M. D. Carr, J. Chem. Soc., 1963, 5854.
- 16. C. A. Bunton and M. D. Carr, J. Chem. Soc., 1963, 5861.
- 17. (a) S. Nametkin and N. Delektorsky, Chem. Ber., 1924, 57, 583. (b) H. Meerwein, Justus Liebigs Ann. Chem., 1939, **542**, 123.
- 18. P. D. Bartlett and I. Poeckel, J. Am. Chem. Soc., 1937, 59, 820. Although these authors were misled, apparently by a coincidence of m.p. values (see ref. 17b), the stereochemical and conformational concepts discussed in this paper are especially noteworthy for the period.
- 19. P. L. Barili, G. Berti, B. Macchia, F. Macchia and L. Monti, J. Chem. Soc. (C), 1970, 1168.
- 20. E. R. Alexander and D. C. Dittmer, J. Am. Chem. Soc., 1951, 73, 1665; the 2,3-epoxybutane isomers were also examined under the same conditions, and gave results which paralleled those of the related (by anti opening) diols.
- 21. K. P. Herlihy, Aust. J. Chem., 1981, 34, 107; 1982, 35, 2221.
- 22. C. J. Collins, Z. K. Cheema, R. G. Werth and B. M. Benjamin, J. Am. Chem. Soc., 1964, 86, 4913.
- 23. M. Bartok and A. Molnar, in 'The Chemistry of Functional Groups. Supplement E: The Chemistry of Ethers, Crown Ethers, Hydroxyl Groups, and their Sulfur Analogues. Part 2', ed. S. Patai, Wiley-Interscience, New York, 1980.
- M. Stiles and R. P. Mayer, J. Am. Chem. Soc., 1959, 81, 1497.
 T. Shono, K. Fujita, S. Kumai, T. Watanabe and I. Nishiguchi, Tetrahedron Lett., 1972, 3249.
- E. Vogel, Chem. Ber., 1952, 85, 25.
 D. J. Cram and H. Steinberg, J. Am. Chem. Soc., 1954, 76, 2753.
- 28. R. D. Sands and D. G. Botteron, J. Org. Chem., 1963, 28, 2690, and refs. therein; D. G. Botteron and G. Wood, J. Org. Chem., 1965, 30, 3871.
- 29. H. Christol, A. P. Krapcho and F. Pietrasanta, Bull. Soc. Chim. Fr., 1969, 4059.
- 30. B. P. Mundy and R. Srinivasa, Tetrahedron Lett., 1979, 2671.
- 31. B. P. Mundy, R. Srinivasa, R. D. Otzenberger and A. R. DeBernardis, Tetrahedron Lett., 1979, 2673.
- 32. J.-P. Barnier and J.-M. Conia, Bull. Soc. Chim. Fr., 1976, 285.
 - 33. E. Nakamura and I. Kuwajima, J. Am. Chem. Soc., 1977, 99, 961; R. A. Martinez, P. N. Rao and H. K. Kim, Synth. Commun., 1989, 19, 373.
 - 34. V. Bhushan and S. Chandrasekaran, Chem. Lett., 1982, 1537.
 - 35. W. Oppolzer, T. Sarkar and K. K. Mahalanabis, Helv. Chim. Acta, 1976, 59, 2012.
 - 36. G. Dana, J. Gharbi-Benarous and S. L. T. Thuan, *Can. J. Chem.*, 1980, **58**, 1451. 37. B. P. Mundy, Y. Kim and R. J. Warnet, *Heterocycles*, 1983, **20**, 1727.

 - 38. C. K. Chang and C. Sotirious, J. Heterocycl. Chem., 1985, 22, 1739.
 - 39. Y. Nagai and H. Uno, Chem. Pharm. Bull., 1979, 27, 2056.
 - 40. E. J. Corey, M. Ohno, R. B. Mitra and P. A. Vatakencherry, J. Am. Chem. Soc., 1964, 86, 478.
 - 41. W. Tubiana and B. Waegell, Angew. Chem., 1972, 84, 642; Angew. Chem., Int. Ed. Engl., 1972, 11, 640.

- 42. (a) K. Suzuki, E. Katayama and G. Tsuchihashi, *Tetrahedron Lett.*, 1983, 24, 4997; (b) K. Suzuki, E. Katayama and G. Tsuchihashi, *Tetrahedron Lett.*, 1984, 25, 1817; (c) G. Tsuchihashi, K. Tomooka and K. Suzuki, *Tetrahedron Lett.*, 1984, 25, 4253.
- 43. S. Kanemoto, Y. Morizawa, K. Oshima and H. Nozaki, Bull. Chem. Soc. Jpn., 1982, 55, 3941.
 44. S. A. Monti, S.-C. Chen, Y.-L. Yang, S.-S. Yuan and O. P. Bourgeois, J. Org. Chem., 1978, 43, 4062.
 45. T. Uyehara, J. Yamada and T. Kato, Tetrahedron Lett., 1985, 26, 5069.
 46. T. Mukaiyama, Y. Echigo and M. Shiono, Chem. Lett., 1977, 179.

3.3 Acid-catalyzed Rearrangements of Epoxides

BRUCE RICKBORN University of California, Santa Barbara, CA, USA

3.3.1 INTRODUCTION	733
3.3.2 STEREOCHEMISTRY OF EPOXIDE OPENING	733
3.3.3 PROTIC ACID CATALYZED REARRANGEMENTS	734
3.3.4 BF3-INDUCED REARRANGEMENTS	741
3.3.5 MAGNESIUM HALIDE CATALYSIS	754
3.3.6 LITHIUM SALT CATALYSIS	760
3.3.6.1 LiClO4 Catalysis 3.3.6.2 Lithium Halide Catalysis	761 763
3.3.7 OTHER CATALYSTS	769
3.3.8 REFERENCES	771

3.3.1 INTRODUCTION

The parallels between acid-catalyzed reactions of epoxides and the pinacol rearrangement are noted in Chapter 3.2 of this volume. Epoxides react more or less readily with acids, but they are at least moderately stable to all but the strongest of bases. It is difficult to find a persuasive instance of epoxide opening, even under basic conditions, that does not involve some probable form of electrophilic assistance, and a case can be made for the view that electrophile bonding/coordination to oxygen is required in order to cleave an epoxide. This electrophilic assistance may be as mild as hydrogen bonding by water in the opening of epoxides with aqueous amines, or as strong as the interactions that occur with concentrated H_2SO_4 or powerful Lewis acids.

3.3.2 STEREOCHEMISTRY OF EPOXIDE OPENING

The stereochemical outcome is known for many epoxide-opening reactions. Examination of the effects of substituents and conditions on stereochemistry can provide insight into the mechanisms of the rearrangements which are the focus of this section. The words *syn* and *anti* are used here to describe mechanistic features, with the terms *cis* and *trans* retained as structural descriptors. The vast majority of epoxide-opening reactions occur with *anti* stereospecificity, *i.e.* backside displacement of the cleaved C—O bond.¹ Anti preference dominates sufficiently to warrant calling this the 'normal' stereochemical outcome.

The stereoelectronic aspects of *anti* opening have been explored in studies of cyclohexene oxides, *e.g.* (1; equation 1), including steroidal epoxides.² For the few reagents which have been examined in detail,

it has been found that an antiperiplanar array of the attacking nucleophile (N in equation 1), the two carbons of the oxirane ring and the epoxy oxygen (bonded to the electrophile E) is very strongly preferred. This antiperiplanar array is feasible with concurrent formation of either a chair (2) or twist boat (4) cyclohexane skeleton. The appreciable difference in $\Delta\Delta G^{\ddagger}$, which typically favors opening via (2) over (4), suggests that a significant portion of the chair versus twist boat energy difference is experienced in the transition states, *i.e.* these reactions are thought to have product-like transition states.



The chair antiperiplanar mode leads initially to the *trans* diaxial product (2), which typically will equilibrate *via* chair-chair interconversion with the *trans* diequatorial conformer (3). The twist boat antiperiplanar initial product (4) will commonly undergo rotation to form the more stable chair conformer (5). Thus both the upper and lower pathways of equation (1) can result in *trans* diequatorial products. Structures (3) and (5) are enantiomeric; these pathways can be distinguished through the use of an appropriate stereochemical marker (substituent).

Strong preference for chair over twist boat development appears to be general, and is often referred to as the rule of diaxial opening or the Furst–Plattner rule. Twist boat processes can be encouraged by suitable substitution,³ but in the absence of such special features there are no known examples of cyclohexene oxides which give appreciable amounts of twist boat opened products (sometimes called diequatorial-opening products). The lone apparent exception⁴ has been shown⁵ to be an artifact caused by an impurity.

Although kinetic control typically favors the *trans* diaxial product, the *trans* diequatorial product is commonly thermodynamically preferred. If the epoxide can be regenerated due to reversibility under particular reaction conditions, a small amount of competing twist boat opening can in principle allow eventual product equilibration. An analogy is found in the classical steroid diaxial to diequatorial dibromide interconversion that is believed to occur *via* reversible formation of a bromonium ion intermediate that slowly drains to the thermodynamically favored product through the kinetically disfavored antiperiplanar twist boat intermediate.⁶

The preference for *anti* opening of epoxides by reagents which function both as an electrophile and a nucleophile is critical to the selectivity observed in some of the rearrangements elaborated below.

Syn (abnormal) opening⁷ of epoxides appears to be closely associated with the formation of carbenium ion intermediates, and thus is typically observed when substituents, reagents or conditions favor the formation of carbocations. Increasing solvent polarity and acid strength can lead to syn opening. Most of the documented examples involve phenyl-substituted epoxides,⁸ although an alkyl⁹ or alkynyl¹⁰ group at a tertiary center can be sufficient. Syn openings of bicyclic alkene oxides have also been observed.¹¹ The isolation of syn-opening products, e.g. the cis-diol from 1-phenylcyclohexene oxide, requires the use of a solvent, such as water, which is sufficiently nucleophilic to trap the intermediate carbenium ion before pinacol-like rearrangement can occur.

Apparent *syn* opening may also be the result of epimerization after normal *anti* opening. Examples have been identified in the cyanide opening of an anhydro sugar¹² and in the opening of steroidal epoxides by hydrogen fluoride.¹³

3.3.3 PROTIC ACID CATALYZED REARRANGEMENTS

The conjugate base of a protic acid is often sufficiently nucleophilic to prevent rearrangements in epoxide-opening reactions. As a consequence, most studies aimed at examining rearrangements have utilized aprotic Lewis acids, especially BF₃. Many such studies have compared protic and aprotic acids;

examples in this section are restricted to those in which the protic acid proved superior, or was the sole reagent used.

Acidic treatment of epoxides has provided a wealth of unusual rearrangements, many of which involve participation of a remote substituent with the incipient positive center. Conversely, the oxirane ring itself may function as a neighboring group, although this phenomenon has been little studied. Richey and Kinsman¹⁴ examined the solvolyses of substrates such as (6; equation 2), and concluded that conversion to (7) was probably accelerated by the oxirane. Similar participation may be involved in the acid-catalyzed rearrangement of (8) to (9; equation 3).¹⁵ On the other hand, Hornback¹⁶ concluded that there was no anchimeric assistance in the acetolyses of the *cis*- and *trans*-tosylates (10; equation 4). Both isomers gave mixtures of products derived mainly from acetolysis of the epoxide ring of inverted acetates (11).



Transannular hydride shifts, first detected by Cope and coworkers¹⁷ in solvolyses of cyclooctene oxide, have subsequently been found in a number of related systems, *e.g.* cyclooctadiene monoepoxides,¹⁸ *exo*-bicyclo[3.3.1]non-2-ene epoxide¹⁹ and 1-oxaspiro[2.6]nonane.²⁰ In general these reactions do not involve skeletal rearrangements, and they will not be discussed in detail.

Hydroxy groups which are β to the oxirane ring readily participate in opening if the geometrical features are favorable. This, the Payne rearrangement, serves to interconvert two epoxy alcohols with inversion of the central carbon. In the simplest case, that of 2,3-epoxypropan-1-o1, the reaction constitutes a mechanism for racemization. Although usually carried out under basic conditions if the rearranged epoxide is to be isolated, the reaction may also be facile under acidic conditions, as illustrated by an example from the steroid literature (equation 5).²¹ In this instance, Payne rearrangement effects the interconversion of (12) and (13), and normal *anti* opening of the latter by solvent then gives the formal *syn*-opening (from 12) product, triol (14).



When the hydroxy group is further removed from the oxirane it may still participate in opening, but these reactions have been studied mainly under basic²² conditions or with Lewis acids, as described in the next section.

Remote double bonds often participate in epoxide openings, sometimes with extensive skeletal changes. *exo*-Norbornadiene oxide (15; equation 6) provides an interesting example. Meinwald *et al.*²³ found that this very acid-sensitive material rearranged simply on standing to give the *endo*-aldehyde (17), presumably *via* the intermediate nortricyclanol ion (16).



Murray and coworkers²⁴ extended their studies of the novel oxidant dimethyldioxirane (19) to the reaction of quadricyclane (18; equation 7). The epoxide (15) was the nearly exclusive monooxidation product when the acetone solution of the oxidant was carefully dried, but in the presence of water a considerable amount of (17) was formed.



Competition between an alkene and an ether oxygen for participation in epoxide opening is evident in the reactions of trichothecenes (illustrated by the skeletal structure 20; equation 8). In aqueous acid, participation of the ether oxygen is favored and (21) is formed, probably due to preferred cleavage at the tertiary center under the more electrophilic conditions. Under neutral or basic conditions, the diminished role of intermolecular electrophile (water) results in unusual scission of the primary epoxide bond, with formation of (22).²⁵



Detailed mechanistic studies of epoxide reactions in aqueous acid led Pocker *et al.* to conclude that protonated epoxide is an intermediate, distinguishable from the ring-opened carbocation (the latter leads to pinacol rearrangement),²⁶ and more recently a neutral solution zwitterion intermediate has been proposed for reactions in solutions of high ionic strength.²⁷ Solvent hydrogen bonding of the zwitterion would appear to be equivalent to calling upon water as the electrophile in 'spontaneous' openings such as that shown in equation (8).

An interesting approach to the pyrrolizidine ring system was developed by Glass *et al.*,²⁸ who found that transannular displacement of the epoxide (23; equation 9) could be effected simply by heating in ethanol.

Cyclopropanes can also participate in epoxide openings, much like double bonds. The work of Ohloff and Giersch²⁹ with carene epoxides illustrates this behavior, *e.g.* (25; equation 10) is converted to (26) and related carbenium ion derived products, and (27; equation 11) similarly forms (28).



When the carbamazepin epoxide (29; equation 12) is treated with *p*-toluenesulfonic acid in chloroform solvent in the presence of methanethiol, the dithioacetal (30) is formed.³⁰ Apparently, rearrangement to ring-contracted aldehyde occurs more rapidly than normal thiol opening of the epoxide.



Attempted peroxy acid epoxidation of the bicyclic ketone (31; equation 13) gave the lactone (33), instead of several possible rational alternatives.³¹ The epoxide (32) was implicated as an intermediate when it was independently synthesized from the epoxy alcohol, and shown to give (33) on treatment with aqueous acid.³² A mechanism involving scission of the acyl bridgehead bond via the hydrated 1,1-diol form of the ketone was proposed to account for the formation of this unexpected product. The rearrangement of the isolongifolene derivative (34; equation 14) appears to be mechanistically related. The product (35) is formed by brief treatment with dilute HClO₄ in dioxane as a mixture of isomers believed to arise by acid-catalyzed epimerization of the carbinol center.³³





Hart and coworkers³⁴ used trifluoroacetic acid to catalyze some deep-seated epoxide rearrangements, illustrated by the examples in equations (15) and (16). An unusual mechanism involving scission of the epoxide C—C bond was suggested for the conversion of (38) to (39), in order to accommodate deute-rium-labeling results.



HBr in HOAc has been used to effect the rearrangement/aromatization of the androstenone epoxide (40). The use of DBr/DOAc and ¹³C NMR analysis led to the conclusion that the reaction takes place via the phenonium ion (41) shown in equation (17).³⁵



One of the few (intentional) applications of HF to epoxide rearrangement is found in an informative study by Audouin and Levisalles,³⁶ who isolated fluorohydrin and diene products on mild treatment of (43; equation 18); these proposed intermediates were converted to the dione (44) when resubjected to the reaction conditions.

Some α -substituted epoxides exhibit predictable and synthetically useful behavior when subjected to acids. Others are thermally unstable, and appear to rearrange by different mechanisms in the presence and absence of acids.

Lewis acid induced reactions of α -carbonyl epoxides have been extensively studied, but protic acid reactions in general have not found favor. An exception is shown in equation (19). The protic acid mixture shown was found to be superior to BF₃ etherate for the desired rearrangement.³⁷

In the Darzens' aldehyde synthesis the α -substituent is a carboxyl(ate) group, which is lost during or subsequent to rearrangement. The normal mode of reaction of glycidic acids generates the new carbonyl group at the point of the original carboxyl group attachment, as shown in equation (20). However, when



the α -cation is stabilized by phenyl substitution as in (49), the product (50), derived from abnormal α -opening, may dominate (equation 21), depending upon the β -substituents.³⁸



Epoxides derived from vinyl halides and vinyl acetates tend to rearrange readily on heating. McDonald and his coworkers uncovered several mechanistically intriguing examples of thermal rearrangements involving α -chloro epoxides. In some instances the epoxides could not be isolated, but presumably are reactive intermediates, *e.g.* in the peroxy acid reaction of chlorostilbenes.³⁹ A common intermediate was proposed to account for the formation of the same (chlorine-migrated) product from both geometrical isomers of (51; equation 22).



The *cis* and *trans* isomers of (53) both gave (54) as the kinetically controlled product (equation 23).⁴⁰ These results, and those obtained in a study of (+)-2-chloronorbornene *exo*-oxide,⁴¹ led the authors to conclude that the thermal reaction occurs by way of intermediate α -ketocarbenium-chloride ion pairs; preferential axial attack results in the formation of (54) as shown.

The role of acid in some of these rearrangements is not obvious. Kirrmann and coworkers⁴² determined that thermal and acid-catalyzed reactions of α -chloro epoxides give different products in some instances. Again, preferred migration of chlorine was observed in thermal reactions.

Different mechanisms for thermal and acid-catalyzed rearrangements of steroidal α -acetoxy epoxides have also been proposed.⁴³ On heating (55; equation 24), intramolecular migration of the acetate group occurs, with stereospecific inversion of the cleaved epoxide center leading to (56). Treatment of (55)



with acid results in formation of the more stable equatorial acetate (57). From rate studies, it was demonstrated that (56) is not an intermediate in the acid-catalyzed formation of (57). Although (56) does rearrange to (57) under the acidic reaction conditions, it does so much more slowly than the direct conversion of (55) to (57). The acyclic analog (58) gave the aldehyde (59) both thermally and under acid catalysis (equation 25). Several other reagents and conditions were examined in an effort to clarify the mechanisms of various reactions of (58), but some uncertainty remains.⁴⁴



Stevens and Pillai⁴⁵ prepared and isolated the azirane (60; equation 26), a rare example of an α -amino epoxide. This remarkably stable material required vigorous heating in *o*-dichlorobenzene to effect the ring expansion to (61); the remainder was converted to polymeric material.



Yamamoto et al.⁴⁶ pointed out potential utility as an 'acyl anion' equivalent for the reaction shown in equation (27). The reaction of (62) with *m*-chloroperbenzoic acid was carried out in refluxing CCl₄. The presumed intermediate epoxide (63) rearranged directly under these conditions to give the product (64) in excellent yield.

More remote substituents may also participate in epoxide opening/rearrangement. Christol and coworkers examined several 5,(6)-substituted-2,3-norbornene *exo*-oxides under acidic conditions, and found that certain 5-*endo* substituents played a part in oxirane opening. For example, oxa rings were formed in reactions of substrates bearing 5-*endo*-methoxycarbonyl or -hydroxymethyl groups.⁴⁷ A novel 1,4-migration of chloride was also detected (equation 28).⁴⁸



Macchia *et al.*⁴⁹ observed that the proportion of *syn* opening of 1-benzylcyclohexene oxide increased as the nucleophilicity of the acid/solvent decreased, and became the major pathway when trichloroacetic acid in methylene chloride was employed (equation 29). The relative amount of *syn* opening was greater for (67) than for 1-methylcyclohexene oxide under the same conditions. The authors concluded that a phenonium ion intermediate, *i.e.* (68), was generated, even though no product involving phenyl migration was detected.



3.3.4 BF₃-INDUCED REARRANGEMENTS

Scores of epoxides have been subjected to BF₃ (mostly as the etherate), and this appears to be the most widely used Lewis acid for rearrangement. The BF₃ is often consumed or altered in the course of these reactions, and is thus a reagent rather than a catalyst, although less than an equivalent is effective in some instances. Steroidal and terpene epoxides have been favorite substrates for study. In spite of the very extensive literature in this area, it is difficult to make product predictions with confidence. Some processes are clean and occur in high yield; many others give low yields and multicomponent mixtures. There is no single recommended standard set of conditions, and many solvent, temperature and time combinations have been described. Products range from simple hydride shift carbonyl isomers to structures formed by multistep rearrangement. Epoxides may also be converted, with or without skeletal rearrangement, to unsaturated alcohols,⁵⁰ dienes,^{50,51} dimers or oligomers⁵² or (very commonly) fluorohydrins.⁵³ To complicate matters further, the products of several reactions have been shown to rearrange to other materials on longer exposure. Given this context, it is not surprising to find that BF₃-induced rearrangements of epoxides are not commonly used in multistep rational syntheses.

On the other hand, no epoxide has been reported to be immune to reaction with BF₃, although an interesting example of inertness was noted by Crandall and Machleder.⁵⁴ Allene epoxides are usually too sensitive to acids to be isolable from peroxy acid epoxidations, but the bulky *t*-butyl groups on (**70**; equation 30) make it exceptionally unreactive. Several hours of reflux with BF₃ in ether were required to effect the conversion to (**71**).



Rearrangement Reactions

Steric effects are presumably also responsible for the formation of pivalaldehyde (73) when (72) is treated with ether-free BF₃ in CCl₄ (equation 31).⁵⁵ Although both isomers (*E* and *Z*) of (72) form pivalaldehyde as the major product under these conditions, a sharp distinction between the isomers was noted in ether solvent. The (*E*)-isomer was recovered unchanged from conditions that with the (*Z*)-isomer gave fluorohydrin in good yield.⁵⁵

$$\begin{array}{cccccc} & & & & & & \\ & & & & \\ & & & & \\ Bu^{t} & & & & \\ & & & & \\ & & & & \\ & & & \\ & & & \\ & & & & \\ & & & \\ & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\ & & & & \\$$

Mechanistic details are rarely obvious in BF₃-epoxide reactions, and various views ranging from concerted rearrangement to free carbenium ion intermediate processes have been put forward. The bulk of evidence, especially that obtained in elegant experiments carried out in the laboratory of Coxon and coworkers in New Zealand, supports a general mechanism involving carbenium ions which retain modest stereochemical memories. Coxon has introduced the perspective of relatively slow epoxide opening to give a carbenium ion which is then subject to rapid migration of adjacent groups, at rates which can be somewhat faster than conformer equilibration in the cleaved complex. Preferential direction of rotation to give a conformer which is stereoelectronically suitable for a 1,2-shift is dictated by steric interactions between the OBF₃ moiety and groups on the adjacent carbocation center. Rotation towards the smaller of these groups correctly predicts the results obtained with simple aliphatic epoxides. For example, by use of deuterium labeling with corrections for isotope effects, it was found that HA migrates 1.4 times more readily than H_B in 1,2-epoxyoctane (74; equation 32).⁵⁶ This outcome was attributed to preferred rotation, as shown in the initial conformer (75), towards the smaller group (H versus alkyl) at the carbenium ion center. As a conformer with suitable orbital overlap features arises, (76), migration of this bond (H_A) must take place at a rate greater than or equal to that for continued rotation, in order to account for the formation of (77) as the major product. Similar results were obtained with epoxide (78; equation 33), which exhibited a preference of $H_A/H_B = 1.9$ for the formation of the aldehyde (79).⁵⁷



This mechanistic view can be used to rationalize other epoxide rearrangements. For example, the observation⁵⁵ that BF₃ treatment of (Z)-2,3-epoxybutane gives only 2-butanone (equation 34) is nicely accommodated. The (E)-isomer gives some isobutyraldehyde (2-butanone:aldehyde = 3:1), which again may be rationalized by this model. Preferred rotation (as in 82) leads to conformer (83), in which the poorer intrinsic migratory aptitude of the methyl group (compared to H) prevents migration from being much faster than further rotation (equation 35).



The important study of Berti *et al.*⁵⁸ (also discussed in the context of the pinacol rearrangement: see equations 8 to 11 of Chapter 3.2 in this volume) included BF₃-induced reactions of the *cis*- and *trans*-epoxides (**85**; equation 36) and (**91**; equation 37), respectively. These very informative reactions show that, at least under the particular reaction conditions used in this work (benzene as solvent), the Coxon mechanism must be expanded to include an appreciable antiperiplanar geometrical feature. Unlike the pinacol rearrangements of the related diols, which gave only ketone (**89**) and aldehyde (**90**) under the same conditions, epoxide (**85**) gives, in addition to these same products, a significant amount of aldehyde (**87**). This appears to require the involvement of the twist boat conformer (**86**), which is the expected intermediate if the shown starting material conformer opens at the tertiary benzylic center with antiperiplanar constraints. Subsequent rotation of (**86**) to the chair conformer (**88**) allows formation of the ketone (**89**) and the aldehyde (**90**).



This view is reinforced by the reaction of isomer (91; equation 37). In this case antiperiplanar opening would occur directly to give a chair (92), which can then produce ketone (93) and aldehyde (90). The striking feature here is the preponderance of aldehyde relative to ketone, not expected based on migratory aptitude considerations, and quite different from the pinacol rearrangement of the related diols. Whatever the cause of this stereocontrol, it signifies fairly strong preference for migration to the back side of the cleaved epoxide center. It should be noted that some possible antiperiplanar-opening intermediates, *e.g. trans*-fluorohydrins, will not serve to rationalize this intriguing behavior.

The formation of aldehydes from 1,1-disubstituted epoxides has occasionally found use in synthesis, although simpler aldehydes in particular tend to form dioxolane dimers by BF₃-induced reaction with epoxide.⁵⁷ Hill *et al.*⁵⁹ converted the epoxide (**94**), which had been prepared from a β -ionone derivative, into luciferin aldehyde (**95**) by treatment with cold BF₃ etherate (equation 38).

Coxon and coworkers⁶⁰ found a small but consistent bias favoring the product involving inversion of the cleaved center when eight epoxides (four pairs of α - and β -isomers) derived from *exo*-methylene steroids were subjected to BF₃ etherate in benzene at room temperature. A carbenium ion intermediate was



proposed, and appears to be required to account for the nearly identical mixtures of epimeric aldehydes obtained from the stereoisomeric pairs of epoxides. This result is not due to equilibration, which would strongly favor the equatorial aldehyde. One isomeric pair is used to illustrate these features (equation 39).



The 1,1-disubstituted epoxide (100; equation 40) undergoes ring expansion rather than hydride migration, to afford materials such as (101) and related products.⁶¹ It is conceivable that an aldehyde was generated in this instance, which then rearranged to the isolated products. This kind of behavior is seen with (102; equation 41) which upon brief treatment with BF₃ afforded aldehyde (103) and ketone (104) in nearly equal amounts. When resubjected to the reaction conditions the aldehyde (103) gave a new ketone (105). Compound (103) was used as a precursor to the hydrocarbon cuparene, which has the structure of (103) modified by reduction of CHO to Me; in spite of the modest yield, this proved a convenient approach to a structure with two adjacent quaternary carbons.⁶²

Some unsymmetrical tetrasubstituted alkenes (and hence epoxides) have become more available since the advent of McMurry low-valent Ti reductive-coupling procedures and related reactions, but relatively little use has been made of this in epoxide rearrangement studies. An exception involves the cycloheptyl derivative (106; equation 42) which was found to rearrange cleanly to ketone (107).⁶³

Rearrangement of a tetrasubstituted epoxide is a key step (equation 43) in a recent approach to the taxane ring system.⁶⁴



Berti et $al.^{65}$ found that the tetrasubstituted epoxide derivative (110; equation 44) of the triterpene hopane gave an efficient rearrangement to the ketone (111).



745

The phenyl substituent appears to be a major factor leading to the interesting results displayed in equation (45).⁶⁶ The timing needed to obtain the excellent yield of (113) is noteworthy, and indicates just how easy it can be to miss interesting chemistry associated with such rapid reactions.



Disubstituted epoxides tend to give mixtures of products when treated with BF₃. An interesting exception is found in the novel approach to the noradamantane ring developed by Majerski and Hamersak.⁶⁷ The epoxide (**115**; equation 46) gives an excellent yield of this ring system with other acids as well, but BF₃ in HOAc offers the added advantage of directly differentiating the two alcohol functions in the formation of (**116**).



The β -cation-stabilizing influence of the trimethylsilyl group probably assists in the rearrangement of (117) to (118; equation 47).⁶⁸



Reactions of BF₃ and epoxides bearing an α -electron-withdrawing group (EWG) have been extensively studied. Often the course of such reactions can be predicted by assuming formation of a carbenium ion β to the EWG (avoidance of positive charge adjacent to the EWG dipole). When this occurs, facile migration of the EWG may take place.

House and Reif examined certain heavily substituted acyclic keto epoxides, and found diverse behavior as illustrated by equations (48) to (51).⁶⁹

The reaction of (119; equation 48) follows what may be regarded as normal behavior, *i.e.* the product is that predicted by β -cleavage followed by migration of the acyl group. Rearrangement of (121; equation 49) can be similarly explained. The formation of (124; equation 50) from (123) is most expediently regarded as involving β -cleavage, but with aryl migration occurring more rapidly than benzoyl migra-

$$\begin{array}{c|cccc} O & & O \\ Ph & & \hline BF_3 & Ph \\ \hline Et_2O & CHO \end{array}$$
(48)
(119) (120)



tion. This order of migratory aptitude appears to be in keeping with observations in other rearrangements. The (Z)-isomer (125; equation 51) gave an unexpected product, (126). The formation of this aldehyde requires cleavage of the epoxide α to the EWG. These early examples served to frame some of the important mechanistic questions.

Kagan *et al.*⁷⁰ examined a series of 3-phenylglycidic esters (EWG = ethoxycarbonyl) and observed migratory preferences which parallel those developed in other studies. However, one important caveat arose from examination of variations in acid, solvent and temperature: the course of the reactions proved to be highly dependent on these variables, leading Kagan to disavow predictability based on structural features alone.

Nonetheless, acyl epoxides and related materials have given useful results when subjected to BF₃. The interesting spiro diketone (**128**; equation 52) for example, is readily prepared by BF₃ etherate treatment of the epoxide (**127**).⁷¹



When the initially formed rearrangement product is a β -keto aldehyde, facile loss of the formyl group is sometimes observed (retro-aldol). An interesting example is found in the reactions of the quinone derivative (129; equation 53) Rearrangement accompanied by deformylation occurred under a variety of acidic conditions, but the BF₃-induced reaction was especially efficient.⁷²



Rearrangement Reactions

An example from the terpene literature⁷³ is shown in equation (54). The corresponding epoxy alcohol in this instance gave a complex mixture from which no product could be isolated. The formation of a second carbonyl group may help to stabilize the product via complex formation, as discussed by House.⁶⁹



McDonald and Hill⁷⁴ found that α -cleavage is a common feature of aromatic epoxynitriles such as (133), which gave (134) in excellent yield (equation 55). In simpler systems the nitrile group typically leads to β -cleavage, a feature that has been used for the synthesis of fluoro ketones as shown by the example in equation (56).⁷⁵



Other EWG which migrate readily upon β -cleavage of an epoxide include the phosphonate ester⁷⁶ (equation 57), sulfoxide (equation 58)⁷⁷ and sulfone (equation 59).⁷⁸



Bach and Domagala have demonstrated that the acyl migrations which occur when diastereomers (145; equation 60) and (147; equation 61) are treated with BF_3^{79} (and in the case of (145), the analogous

thermal⁸⁰ reaction) are stereospecific, in each instance taking place with complete inversion at the migration terminus, yielding the enantiomers (146) and (148), respectively. It was noted that such behavior is more in keeping with a concerted rearrangement than a carbenium ion process. The contrast to the modest levels of inversion discussed previously is appreciable, but it is not clear if this signals a fundamental change in mechanism or some feature of the carbonyl group which allows especially favorable interaction with the developing positive center. Bach suggested that the carbonyl group may have a conformational requirement for migration, and explored this idea with the systems shown below.81 It was found that fluorohydrin formation occurred very rapidly with (149; equation 62) and (152; equation 63), although none was detected with (154; equation 64) under conditions where rearrangement to (155) took place. The fluorohydrins (150) and (153) both rearranged to diones rapidly in refluxing benzene containing BF3. It was thus established that the fluorohydrins are viable but not necessarily required precursors to the dione product. The authors suggested that fluorohydrin formation occurred because acyl migration was geometrically retarded, and that the fluorohydrin in an appropriate conformation could serve as the immediate precursor to the dione. If the fluorohydrins are formed by normal anti opening of the epoxides, one would expect retention at the migration terminus in these reactions, an interesting possibility that remains to be tested on a suitably designed substrate.



Zwanenburg *et al.* examined keto epoxides which contained an acid-sensitive diazomethyl group, *e.g.* (156; equation 65), and found conditions for selective initial reaction at both sites. BF₃ treatment gave the heterocycle (157), presumably by epoxide rearrangement followed by a carbene reaction with the

carbonyl group.⁸² Pd(OAc)₂ catalysis was used to retain the epoxide function while carrying out typical carbenoid reactions, illustrated by the formation of (158).⁸³



A subtle substituent effect is evident in the reactions of (159; equation 66); when the amine (R = alkyl) is subjected to acid-induced rearrangement it gives the hydride migration product (160), whereas the amide (R = acyl) reacts almost exclusively by ring contraction.⁸⁴



The epoxy ketal (162; equation 67) gave a modest yield of the furan (163) along with several other unidentified products.⁸⁵



An effort to use an adjacent hydroxy group to force the conversion of a [4.4.0] derivative (164; equation 68) to the [5.3.0] ring system was thwarted by an alternative high yield rearrangement to the [4.3.1] bicyclic product (165).⁸⁶



Another example of a rational synthetic plan going awry occurred with (166; equation 69). In this instance the desired reaction was simple methyl migration followed by loss of a proton, but instead

contraction of ring \land took place. A high yield of oxetane (167) was isolated in this example. The stereochemistry of this product was attributed to a mechanism involving opening and reclosure of ring B.⁸⁷



The epoxy oxygen reappears in the furanoid rings of (169) and (170) when (168) is subjected to typical rearrangement conditions (equation 70). The stereochemical relationships of these products support a stepwise carbenium ion process.⁸⁸



The remarkable, highly symmetrical compound (173) is generated when the triepoxide (172; equation 71) is treated with BF₃ etherate.⁸⁹



Remote hydroxy groups participate in various epoxide openings. Coxon *et al.*⁹⁰ observed the formation of cyclic products in the peroxy acid epoxidation of some unsaturated alcohols, and examined the process in greater detail using BF₃ in ether. These reactions exhibited a ring size preference: 5 > 6 > 7. The reactions of *cis*- and *trans*-5,6-epoxyheptan-1-ol and *cis*- and *trans*-4,5-epoxyhexan-1-ol appeared to be stereospecific, with inversion of the cleaved center as expected for opening with participation of the hydroxy group. In contrast, the reactions of *cis*- and *trans*-3,4-epoxypentan-1-ol were not stereospecific, in spite of the apparently favored formation of five-membered ring products. Some of these features are illustrated in equations (72) and (73).



A novel mechanism involving an orthoester intermediate was proposed to account for the stereochemical features of reactions like that shown in equation (74), in which the relative configuration of the epoxide is retained.⁹¹



Suitably positioned carbonyl groups may cause ring closure, leading to either five- or six-membered ring lactones. BF₃ etherate proved to be more selective than protic acids in a study of unsaturated epoxides, giving exclusively the larger ring size, as illustrated in equation (75).⁹²



Among the best-known epoxide rearrangements are those which utilize remote double bonds in biomimetic reactions.⁹³ Two recent syntheses of aphidocolin illustrate variants of this concept. The application described by Van Tamelen *et al.*⁹⁴ is outlined in equation (76), while the Tanis *et al.*⁹⁵ alternative is shown in equation (77). These reactions tend to be highly stereoselective. Efforts are usually made to avoid nucleophilic gegenions which can disrupt the desired reaction cascade; a wide variety of catalysts in addition to BF₃ have been explored. In the conversion of (**185**) to (**186**), for example, an unusual 2:1 ratio of BF₃·OEt₂ and triethylamine in a three component solvent was found to give the best yield.



Conacher and Gunstone⁹⁶ reported the novel conversion of the diene monoepoxide methyl vernolate (**187**; equation 78) to the cyclopropyl keto isomer (**188**). Later Italian work⁹⁷ supported this conclusion, and furnished additional examples of cyclopropane formation.

Excellent yields of bi- or poly-cyclic materials may be obtained if the starting material contains significant portions of the desired product structure. Thus the isogermacrone epoxide (189; equation 79) is transformed into (190) by brief treatment with BF₃ etherate, while longer contact causes further rearrangements.⁹⁸ Similarly, epoxide (191; equation 80) with two six-membered rings already in place,

752



affords a nearly quantitative yield of the hydrophenanthrene derivative (192), isolated after hydrolytic work-up as a mixture of the two double-bond isomers.⁹⁹



Cyclopropane participation occurs in the reaction of (193; equation 81) which rearranges exclusively to the aldehyde (194), although the latter is unstable to the reaction conditions. The *endo* (cyclopropyl) isomer was also examined in this study,¹⁰⁰ and gave mostly polymeric product.



Epoxides may be generated as reactive intermediates when unsaturated materials are treated with BF₃/trifluoroperacetic acid, a powerful oxidation/rearrangement mixture developed by Hart.¹⁰¹ The products from numerous alkenes and aromatic substrates, to the extent that comparisons can be made, are those expected from epoxide/carbenium ion rearrangements.

Although infrequently used, sodium cyanotrihydroborate in the presence of BF₃ has been shown to reduce epoxides with interesting regio- and stereo-chemical results.¹⁰² For example, 1-methylcyclohexene oxide is converted in high yield to *cis*-2-methylcyclohexanol contaminated with only small amounts of the *trans* and 1-methylcyclohexanol isomers.

Acyloins are sometimes formed in moderate yield when 1,2-disubstituted epoxides are treated with BF₃ in dimethyl sulfoxide solvent.¹⁰³ A curious report¹⁰⁴ of highly temperature dependent nonoxidative rearrangement of 2,3-epoxycyclododecanone indicates that a 1,2-dione is formed at 85 °C, whereas a 1,3-dione is the product at 150 °C. All of these applications require heating, since the Lewis acidity of BF₃ is strongly attenuated in dimethyl sulfoxide.¹⁰⁵

3.3.5 MAGNESIUM HALIDE CATALYSIS

French work over 60 years ago first led to the recognition that skeletal rearrangement can take place when epoxides are treated with Grignard reagents. By the 1940s the involvement of halohydrin salt intermediates was considered proven, and general stereochemical features were understood, *i.e.* preference for a *trans* relationship between the migrating group and the departing halide.¹⁰⁶ Geissman and Akawie¹⁰⁷ provided additional evidence relating cyclic halohydrin stereochemistry with the course of rearrangement. This work reinforced the view that a *trans* relationship was preferred for rearrangement, but also showed that the same rearrangement product could be formed (in low yield) from the 'wrong' halohydrin stereoisomer. Subsequent studies support the view that epoxide/Grignard and related reactions rarely provide clearcut mechanistic features, and that mixtures of products not explicable by a single pathway are commonly encountered.

Halohydrins are often the major products isolated (after hydrolysis) from epoxide/Grignard reagent mixtures. The magnesiohalide salt of the halohydrin (196; equation 82) is assumed to be the species present in solution, formed by reaction of the MgX₂ that is present from Schlenk equilibrium. Anti opening of the epoxide (195) is also usually assumed; there is experimental evidence in several instances to support this conclusion. The alternative (syn opening) halohydrin has rarely been ruled out, however, especially as a minor product. The magnesiohalide salts appear to be reasonably stable in ether solutions. Unlike the lithium salt analogs discussed in the next section, (re)closure of (196) to the epoxide does not appear to be thermodynamically favorable, and is known to be relatively slow in instances where this question has been addressed.



Rearrangements of halohydrin salts usually require prolonged treatment of ether solutions at ambient temperature or heating in a less polar solvent, but there are some interesting exceptions. The 'standard' conditions consist of formation of the magnesiohalide salt in ether, followed by solvent replacement by benzene and short term reflux. House¹⁰⁸ found that residual ether can significantly influence the course of a reaction, and since this variable appears not to have been carefully controlled in most experiments, comparisons may have limited significance.

The reaction of cyclohexene oxide with MeMgX was reexamined in 1969, in work which clearly established the importance of the halide.¹⁰⁹ Standard conditions (1 h at 80 °C) were employed with 1.4 equiv. of the organometallic, to give the yields listed under equation (83). Only minor amounts of the 'normal' displacement product (200) are formed from the chloride and bromide, and none from the iodide. The iodide and bromide give extensive rearrangement, with the bromide being more selective in the sense that product (198) is expected on the basis of stereoelectronic considerations (backside displacement of halide) from the *trans*-halohydrin. The unusual product from this perspective is (199). It must arise either from the *cis*-halohydrin or a process which is not subject to the same stereoelectronic controls, *e.g. via* a carbenium ion.



Treatment of (197) with MgX₂ (X = I or Br) in ether gave high yields of the corresponding *trans*-halohydrins, with no sign of the *cis* isomers. When heated as in the Grignard reagent experiments, cyclopentanecarbaldehyde and cyclohexanone were formed in ratios of *ca*. 2:1 for the iodide, and 9:1 for the bromide. Based on these observations, the authors suggested that (199), or more precisely its precursor cyclohexanone, arises from the carbenium ion generated by loss of halide from the halohydrin salt, and that this process is more favorable with X = I than Br. House¹⁰⁸ had earlier reached the same conclusion (competing concerted and carbenium ion pathways) from a study of the reactions of *cis*- and *trans*-stilbene oxide with MgBr₂ in refluxing benzene. The *cis*-epoxide (**201**; equation 84) gave diphenylacetaldehyde (**202**) with traces of deoxybenzoin (**204**), while the *trans*-epoxide (**203**; equation 85) gave a mixture of (**202**) and (**204**). The significance of these observations follows from the rearrangements of the corresponding independently prepared bromohydrins (**205**; equation 86) and (**206**; equation 87). When treated with RMgBr to form the magnesiobromide salts and then heated as before, *threo*-(**205**) gave only (**202**), while *erythro*-(**206**) gave only (**204**).



The cleaner reactions of the halohydrins show that these, if formed by *anti* opening and not subsequently epimerized, cannot account for the product mixtures from the epoxides. House proposed a competing carbenium ion route for both epoxides, leading to the normal pinacol rearrangement (carbenium ion) product, the aldehyde (202).

It is by no means obvious why these acyclic halohydrins should give mutually exclusive products. Simple stereoelectronic and steric considerations do not provide explanations. House suggested that the OMgX substituent might have abnormally large spatial requirements. A similar observation was made by Curtin and Meislich, who proposed large differences in solvation for two alternative transition states to rationalize an even more subtle distinction (migration of Ph *versus p*-ClC₆H₄ in two diastereomers).¹¹⁰ These suggestions appear reasonable especially when aggregation effects are taken into consideration.

Similar experiments with *cis*- and *trans*-2,3-epoxybutane and the corresponding bromohydrins all gave a single product, 2-butanone.¹¹¹ Thus the unusual factors in the aryl-substituted materials failed to appear in the simpler aliphatic materials. Here the product can be rationalized simply by assuming that hydride migration occurs more readily than methyl migration. Of course, this may be true for both concerted and carbenium ion processes, and the existence of a dual mechanism may be hidden in this example. The cyclohexene oxide results cited above suggest that this is plausible, and perhaps probable.

In all of these reactions halohydrin salts were implicated as the reactive species, since no rearrangement occurred when the halohydrins were treated with MgBr₂.

House also examined analogous reactions of 2-bromo-3-pentanol (207; equation 88) and 3-bromo-2pentanol (210; equation 89), presumed to be mixtures of diastereomers obtained by NaBH₄ reduction of the corresponding bromo ketones. Each system gave some of the oxygen-shifted ketone, as shown in equations (88) and (89), presumably *via* the epoxides, with reopening leading to the opposite regioisomer. This work is significant in showing that such interconversions can occur, even though epoxide formation in these systems must be slower than rearrangement. Other examples of oxygen shifts which appear to require formation of an epoxide intermediate are known.¹¹²



Cleavage of an epoxide by MgX_2 is only one of several ways by which halohydrins can be made. Other significant methods are reduction of halo ketones, addition of HOX to alkenes, bromination of alcohols and addition of RMgX or RLi to halo ketones. Some offer the advantage of controlling the regiochemistry of the halohydrin. Because of this diversity of approaches, MgX_2 -catalyzed rearrangements of halohydrins have been studied more extensively than analogous reactions of epoxides.

The addition of Grignard reagents to α -halo ketones can give a variety of products, including enolates generated either by dehalogenation or deprotonation.¹¹³ Substitution of the halide by the organometallic R group has also been known for over half a century. Tiffeneau was the first to recognize that this product resulted from rearrangement of an initially formed halohydrin salt.¹¹⁴ The halohydrin configuration that produces this skeletally unrearranged material is apparently formed preferentially because of addition to the carbonyl face away from the halide. However, these reactions rarely give pure products or high yields. An angular methylation application developed by Sisti and Vitale¹¹⁵ illustrates the stereochemical features (equation 90).



Simpler α -chloro ketones have also been examined,¹¹⁵ and exhibit negligible selectivity. The mixtures of products shown in equations (91) and (92) may be the result of formation of both *cis*- and *trans*-halo-hydrin isomers in the addition step or, less likely with the chloride, a carbenium ion mechanism. The effect of varying the halide in such reactions has not been systematically investigated.



Sisti has also explored the potential utility of halohydrin rearrangements for ring expansions. A few examples are outlined. Ring-expanded α -phenyl ketones of product ring size six to nine were prepared in 60–80% yields by the sequence shown in equation (93).¹¹⁶



The reactions of bromohydrins shown in equation (94) to (96) are especially informative. No cycloheptanone was obtained from the primary halide (222; equation 94),¹¹⁷ whereas the secondary (225; equation 95)¹¹⁸ and tertiary (227; equation 96)¹¹⁹ halides both gave good yields of ring-expanded products. These results reinforce the observation made earlier by Geissman¹⁰⁷ that facile rearrangement is associated with ease of carbenium ion formation. It would thus be misleading to view the rearrangement process as a simple S_N2 -like displacement, in spite of the apparent operation of stereoelectronic control features. More complex halohydrins have also been examined; these tend to give mixtures of products.¹²⁰



Enhanced reactivity associated with a tertiary center appears to be a factor in the observations made by Naqvi *et al.*¹²¹ on treatment of 1-methylcyclohexene oxide (**229**; equation 97) with MgBr₂. At 0 °C, conditions which give only halohydrin with cyclohexene oxide, (**229**) was converted into the aldehyde (**230**). Conversely, when (**229**) was added to a solution of MgBr₂ at 60 °C, only ketone rearrangement products were isolated. The modest yields prevent mechanistic conclusions.

A striking example of enhanced reactivity is found in the vitamin A related studies of Rosenberger et $al.^{122}$ The example shown (equation 98) and several other analogous high yield reactions almost certainly proceed via the stabilized carbocation (tertiary and dienyl in this instance). Higher temperatures were detrimental to these applications, probably because of product instability.

Matsuda and Sugishita¹²³ found that cyclooctatetraene monoepoxide (233; equation 99) gave only skeletally rearranged products, *e.g.* (234), when treated with Grignard reagents. In an effort to isolate the presumed intermediate cycloheptatrienecarbaldehyde, (233) was subjected to a catalytic amount of MgBr₂, but this resulted in the formation of phenylacetaldehyde.

The stereochemistry of the product obtained when β -2,3-epoxycholestane is treated with MeMgI is in keeping with a mechanism involving initial diaxial opening to form an iodohydrin salt (equation 100), followed by rearrangement with backside displacement of the iodide *via* a twist boat conformer. Addition of MeMgI to the A-norsteroid aldehyde formed in this manner would then result in the two alcohols (epimeric at the carbinol center), which were isolated.¹²⁴



Although indirectly related to the present topic, a common alternative method of inducing rearrangement of halohydrins utilizes a soluble silver salt. It is interesting that the bromohydrin which is regioisomeric to the intermediate iodohydrin proposed in equation (100), when treated with AgNO₃ in refluxing ethanol, delivered the analogous aldehyde (of unknown stereochemistry) in protected form (equation 101).¹²⁵


An unusual (for MgX₂) rearrangement of an epoxide to an allylic alcohol has been described in the terpene literature.¹²⁶

Several epoxides substituted by α -EWG or other substituents have been treated with RMgX and/or MgX₂, generally to give products derived from initial β -cleavage. These substituents include chloride,¹²⁷ nitrile¹²⁸ and chloride/ester combined. The latter easily prepared substrates offer an attractive route to pyruvates and 1,2-diones. Chloro epoxy esters such as (**241**; equation 102) react with Grignard reagents at low temperature at the ester group, without opening the epoxide, to furnish epoxy ketones, *e.g.* (**242**), in modest yield.¹²⁹ This outcome is somewhat unusual, since ketones are rarely isolated in significant amounts from ester/Grignard reagent reactions; it appears that the electronegative substituents improve the stability of the initially formed intermediate. Subsequent treatment of (**242**) with MgI₂ gave the interesting iododione (**243**; equation 103) which could be reduced to the dione (**244**) by treatment with aqueous NaHSO₃.¹³⁰ In this sequence, facile loss of chloride from the iodohydrin (salt) intermediate prevents rearrangement from occurring.



Attempts to prepare β -trimethylsilyl epoxides by closure of halohydrins resulted instead in the formation of aldehydes or ketones.¹³¹ Epoxides are thought to be involved as intermediates, and the addition of MgBr₂ improved the overall conversion illustrated in equation (104). The potent β -cation-stabilizing capacity of β -trimethylsilyl groups was invoked to explain the high reactivity of the intermediate epoxides. Only hydride migration products were found. Application to tertiary halide halohydrins, which would require carbon migration to effect rearrangement, has apparently not been done.



 α -Trimethylsilyl epoxides have been more extensively studied, particularly by Hudrlik and coworkers. Reactions have been carried out with both MgBr₂¹³² and MgI₂.¹³³ The products obtained arise from nucleophilic attack at the carbon adjacent to the silicon.^{132,134} Depending upon the other substituents, the resulting bromohydrin (salt) may be stable and isolable, or rearrangement may occur to form β -keto-(aldehydo)silanes, or enol trimethylsilyl ethers derived from these carbonyl primary products. These features are illustrated in equations (105) to (107).



$$Me_{3}Si \xrightarrow{O} SiMe_{3} \xrightarrow{MgBr_{2}} Me_{3}Si \xrightarrow{O} SiMe_{3}$$
(107)
(253) $1.5 h \\ 90\%$ (254)

The stereochemistry shown in the bromohydrin (249; equation 105) has been proven (in analogs) by isolation and stereospecific conversion to vinyl bromide.¹³⁵ The reaction of (251; equation 106) to give exclusively ring-contracted product also supports the intermediacy of a bromohydrin (salt), and suggests the importance of normal stereoelectronic factors in the rearrangement. The preferential migration of a trimethysilyl group (over H) in equation (107) is noteworthy. The ketosilanes (and *in situ* Grignard reagent trapped aldehyde analogs¹³⁵) are useful synthons.

3.3.6 LITHIUM SALT CATALYSIS

An attempt¹³⁶ to repeat, with presumably a minor modification, an established procedure¹³⁷ for the conversion of cyclohexene oxide to the cyclopropane (255; equation 108) gave instead the ring-contracted isomer (256). This reaction is fairly general and constitutes a convenient method for the preparation of acrylate esters, including some not easily made in other ways.



The key difference between the literature procedure¹³⁷ and the acrylate-forming reaction proved to be the use of a phosphonium bromide and BuLi to form the ylide in the latter. This gave a combination of ingredients (LiBr, ylide and/or R₃PO) and conditions (benzene solvent, reflux) which caused rapid rearrangement of epoxides to carbonyl compounds. Subsequent study provided the first unequivocal evidence¹³⁸ that LiBr as well as some other lithium salts can cause epoxide rearrangements.

In contrast to Grignard reagent/epoxide reactions in which MgX_2 (X = Br, I) is clearly responsible for many skeletal or hydride shift rearrangements, organolithium reactions usually do not exhibit these features, although some exceptions are known.¹³⁹ Heeren *et al.*¹⁴⁰ found that LiI altered the (complex) product mixture from treatment of cyclohexene oxide with MeLi in refluxing ether. One of the products (1-cyclopentylethanol) formed in the presence of LiI requires skeletal rearrangement. This material was not produced in the reaction with MeLi prepared from MeBr, *i.e.* LiBr did not cause similar rearrangement. We now recognize that this reflects the relative rates of competing reactions, and the dampening effect of ethereal solvents on the rearrangement process.

A systematic study¹⁴¹ of epoxide/lithium salt reactions established the following general features: (i) LiI¹³⁸ is more reactive than LiBr. LiCl does not cause rearrangements, perhaps due to insolubility; (ii) LiBr is insoluble in benzene, and does not cause rearrangement unless a solubilizing agent is present (in controlled amounts). Phosphorus ylides (equation 108), tri-n-butylphosphine oxide and hexamethylphosphoramide (HMPA) will function as solubilizers; and (iii) LiClO₄ is insoluble in benzene but is partially solubilized by epoxides. Substrates susceptible to carbenium ion formation rearrange readily. Solubilizers and other Lewis bases diminish the rates of LiClO₄ reactions.

The effects of epoxide structure on rates and product distributions indicate that LiClO₄ reactions occur by a carbenium ion mechanism. Conversely, the LiBr-catalyzed reactions involve bromohydrin salts as precursors to rearrangement products. These are sharp distinctions, and provide the cleanest examples of the two extreme mechanisms for epoxide–carbonyl rearrangement.

3.3.6.1 LiClO₄ Catalysis

Certain epoxides rearrange readily in refluxing benzene in the presence of solid LiClO₄. When any epoxide (reactive or not) is added, the salt slowly disappears from the bottom of the flask, and is redeposited as a ring at the surface of the boiling liquid.¹⁴¹ This phenomenon is presumably associated with a strong Lewis acid/base interaction between the lithium cation and ether oxygen. Pocker and Buchholz¹⁴² carried out detailed studies of solutions of LiClO₄ in diethyl ether (up to 6 M solutions can be made), which exhibited various phenomena verifying a strong interaction. Solid mono- and di-etherate phases were also identified. It was noted that hydrated LiClO₄ has very low solubility in ether, in contrast to the anhydrous material. Hydration is also expected to diminish catalytic activity, although this has not been specifically demonstrated. A relatively simple procedure for purifying and drying LiClO₄ (vacuum, 160 °C, 8 h) is described in this work.¹⁴²

The evidence that LiClO₄ catalysis involves carbenium ion intermediates arises from consideration of the structures that undergo reaction and the products that are formed. For example, all epoxides with a tertiary oxirane center rearrange readily, and the products are those expected from tertiary carbenium ion formation.¹⁴¹ Thus 1-methylcyclohexene oxide gives 80% of 2-methylcyclohexanone (hydride migration) and 20% of the ring-contracted product, 1-methylcyclopentanecarbaldehyde. The products from 1,2-dimethylcyclohexene oxide are identical in structure and proportion to those found by Bunton and Carr¹⁴³ in the pinacol rearrangement of 1,2-dimethyl-1,2-cyclohexanediol (see equation 7 of Chapter 3.2 in this volume). This is a remarkable outcome given the differences in conditions (particularly solvent, benzene *versus* aqueous perchloric acid), and strongly supports a similar mechanism for both processes. Presumably ions present or generated in the nonpolar medium exist as ion pairs or higher aggregates, but they appear to react in traditional ways.

Facile rearrangement is observed with 1,1-disubstituted epoxides and LiClO₄. For example, (257; equation 109) gives the aldehyde (258) exclusively.¹⁴¹



Although cyclopentene oxide was unreactive and cyclohexene oxide reacted very slowly to give mostly nonvolatile products, some other 1,2-disubstituted epoxides will rearrange, but at a diminished rate. Cycloheptene oxide provides an interesting illustration; treatment with LiClO₄ (equation 110) affords mainly the ring-contracted aldehyde along with some ketone formed by hydride migration.¹⁴¹



Norbornene oxide (equation 111) rearranges with apparent nonclassical ion involvement, to give a similar ratio of aldehyde (261) and norbornanone (262).¹⁴¹



The reaction of styrene oxide is noteworthy, not because phenylacetaldehyde (only) is formed, but because this reactive product appears to be reasonably stable to the reaction conditions.¹⁴¹ This illustrates one of the advantages of using a catalyst which is only slightly soluble and contains an especially nonnucleophilic, nonbasic gegenion. Much of the work that has been done using BF₃ to catalyze epoxide rearrangements might benefit from reexamination with LiClO₄.

Trost and Bogdanowicz¹⁴⁴ examined several acidic catalysts for the rearrangement of oxaspiropentanes to cyclobutanones, and found the LiClO₄ method to be the most generally useful for this transformation. The reaction of (**263**), which is converted quantitatively to the ketones as shown in equation (112), illustrates a tendency for migration with inversion at the cleaved center, reminiscent of some BF₃-induced reactions discussed previously.



Synthetic utility depends upon compatibility with other functional groups, and although LiClO₄ has not been extensively tested in this regard, the method appears to be suitable for those groups examined to date. Thus a remote alkene does not interfere (equation 113).¹⁴⁴



Allylic epoxides seem to behave normally. Stereospecific hydride migrations were observed in the McMurry *et al.* synthesis of eremophilone.¹⁴⁵ Treatment of the isomeric epoxides (**268**) and (**270**) under standard conditions resulted in the exclusive formation of the rearranged ketones, as illustrated in equations (114) and (115).



Details are lacking, but the amide group apparently does not interfere with LiClO₄-catalyzed rearrangement of compound (159; equation 66), which gave the same product as BF₃ catalysis.⁸⁴

Two interesting applications show that LiClO₄ is compatible with at least some acetal groups. The reaction shown in equation (116) was used in the Corey *et al.* synthesis of aphidicolin.¹⁴⁶

Burge *et al.*¹⁴⁷ reported that (274; equation 117) is cleanly converted to ketone (275) by LiClO₄ under the standard refluxing benzene conditions. Apparently the ketal group helps to prevent the small amount of ring contraction that accompanies analogous ketone formation in the parent system.¹⁴¹

Kennedy and Buse¹⁴⁸ used a LiClO₄-LiClO₃ eutectic mixture as a molten salt phase in a GLC column at 280 °C for rearrangement of a few model epoxides. Those with tertiary centers gave lower product selectivity than in the typical solution reaction, but the harsher conditions did cause cyclohexene oxide to rearrange, giving cyclopentanecarbaldehyde as the major product.



3.3.6.2 Lithium Halide Catalysis

Both LiBr and LiI have been widely used to effect epoxide–carbonyl rearrangements. These salts differ significantly in reactivity,¹³⁸ although no systematic comparison has been made. LiBr in benzene (or toluene) solvent requires a solubilizer (unlike LiClO₄, for which the epoxide itself is sufficient). LiI does not require a solubilizer, and has been used in several solvents, most often CH_2Cl_2 ; the possible role that solubilizer or solvent may play in LiI reactions has not been addressed. LiI is often employed as a hydrate, but occasionally 'anhydrous' salt is specified, although the purity of this difficult to dry material may be questioned.

Various addends have been used to bring LiBr into solution. Hexamethylphosphoramide (HMPA) has been most widely used, although its reputation as a carcinogen provides interest in identifying alternatives. Magnusson *et al.*¹⁴⁹ found that tetramethylurea (TMU) is a suitable replacement for some highly oxygenated epoxides derived from sugars. It is not yet clear what role the addend plays in the determination of products. Magnusson found that TMU and HMPA gave essentially the same product mixtures, whereas some differences were observed in an earlier comparison of HMPA and Buⁿ₃PO.^{138,141}

Only products which can be attributed to bromohydrin (salt) intermediates are formed when epoxides are treated with LiBr (solubilizer) in refluxing benzene. Any apparent exceptions to this generalization are believed to be due to interconversion of halohydrin stereoisomers, as discussed below. The bromohydrin is formed by traditional *anti* opening (antiperiplanar, probably nearly exclusively *trans* diaxial). It has been established¹⁴¹ that a model epoxide (1-methylcyclohexene oxide) reacts by a process that is kinetically first order in a 1:1 complex of LiBr-HMPA, but the state of aggregation [*n* in the formula (LiBr·HMPA)_n] is not known.¹⁵⁰ The nature of the epoxide–(LiBr·HMPA)_n interaction that allows bromide to attack (exclusively) in the *anti* mode is not known, but must require fairly extensive ion pair (aggregate) reorganization.

Solid complexes of defined stoichiometry have been prepared for all the lithium halides with HMPA.¹⁵¹ For LiBr, both [LiBr(HMPA)₂] and [LiBr(HMPA)₄] have been obtained as solids of defined m.p., but the 1:1 complex, the kinetically active species for epoxide rearrangement, has not been isolated. The rate of epoxide loss and solubility of LiBr increased proportionately with added solubilizer (HMPA), to a maximum rate at a 1:1 ratio of addend:LiBr.¹⁴¹ Additional HMPA beyond this ratio caused the rate to decrease even though all the LiBr remained in solution. At an addend:LiBr ratio of 2:1, the reaction effectively ceased. These observations allow the conclusions that [LiBr(HMPA)₂] is more stable than the reactive 1:1 complex in benzene, and that only the latter is kinetically competent.¹⁴¹

l-Methylcyclohexene oxide (equation 118) provides a useful model system for mechanistic discussion. It is important that no 2-methylcyclohexanone is formed, since this rules out carbenium ion pathways. The major product is derived from the bromohydrin (276), formed by bromide attack at the tertiary center (followed by chair-chair interconversion), but care must be taken to avoid overinterpretation of this observation. Thus, if the bromohydrins are rapidly interconverting *via* the epoxide, the product distribution would be determined not only by the equilibrium ratio of (276) and (277), but also by the respective rate constants for rearrangement to (230) and (215). Although cyclohexene bromohydrin is immediately converted to the epoxide by treatment with BuⁿLi in benzene,¹³⁸ the possible effect of HMPA on the bromohydrin(salt)/epoxide equilibrium is not known. The rearrangement rates would be Br^t (276) > Br^s

(277) if parallels with MgX₂-catalyzed halohydrin–carbonyl rates can be drawn. The rates for the final rearrangement steps rather than the rates of initial bromohydrin formation may thus control the product ratio.



Rate constants were obtained¹⁴¹ for rearrangement of a variety of epoxides with LiBr-HMPA in refluxing benzene, and an inverse relationship between degree of epoxide substitution and reactivity is evident. Terminal alkene epoxides are the most reactive, followed by 1,2-disubstituted, which are in turn more reactive than 1,1-disubstituted, while trisubstituted and tetrasubstituted epoxides react even more slowly. While this appears to parallel S_N2 reactivity, the expected ease of bromide attack is not reflected in the products of many of these reactions. For example, the terminal epoxide (**278**; equation 119) is among the most reactive examined but it gives a mixture of products, the major one requiring attack of bromide at the secondary center. Both rates and product ratios thus appear to be associated with more subtle features of the actual rearrangement step, as suggested above.



The reaction of cyclohexene oxide with LiBr HMPA in refluxing benzene leads exclusively to cyclopentanecarbaldehyde, but, like other enolizable aldehydes, the product is not indefinitely stable to the reaction conditions. The rearrangement of 1,2-dimethylcyclohexene oxide, although much slower, gives only the ketone (**218**) as shown in equation (120). This result is especially difficult to rationalize by any mechanism other than one requiring a bromohydrin intermediate.



Styrene oxide reacted rapidly but gave no volatile material, perhaps because of aldol polymerization of phenylacetaldehyde (the probable initial product). Norbornene oxide gave no detectable reaction, either because *endo* attack of bromide is especially unfavorable, or because the bromohydrin salt has no geometrically accessible alternative other than to return to epoxide.

Cyclopentene oxide is very unreactive ($k \le 10^{-3}k_{cyclohexene oxide}$), an observation clearly not in accord with expected S_N2 reactivity; it gives cyclopentanone as the only observed product. This ketone cannot be formed by antiperiplanar hydride displacement of bromide from the *trans*-bromohydrin, whereas the *cis*-bromohydrin could easily adopt the necessary conformation for hydride migration. Almost certainly the latter is formed by slow, perhaps rate-determining S_N2 attack by bromide on the initially formed *trans*-bromohydrin salt, as outlined in equation (121). This secondary reaction may be more important with LiI than with LiBr, although this has not been proven (see also equation 83 and discussion of MgX₂-catalyzed reactions).



LiBr solubilized by Buⁿ₃PO (or HMPA) was found¹⁵² to catalyze the formation of 2-oxazolidones from organic isocyanates and terminal alkene epoxides. The epoxide substituent appears at the 5-position of the product as shown in equation (122). This outcome is in keeping with rapid trapping by the isocyanate of the halohydrin salt formed by attack of bromide at the primary center. This interpretation requires that oxazolidone formation be faster than epoxide rearrangement; data are not available to confirm this point.



Rearrangements of epoxides catalyzed by LiBr·HMPA or LiI have figured in a number of natural product syntheses in which ring expansions to cyclopentanones are required. Both salts were earlier shown to effect the especially facile rearrangements of the small ring spiro epoxides which serve as models for these processes. Thus (289; equation 123) was found by Salaun and Conia¹⁵³ to rearrange rapidly and in excellent yield to cyclobutanone, simply on exposure to commercial hydrated LiI.



Conia and coworkers¹⁵⁴ later examined the rearrangements of acylated derivatives such as (291; equation 124) which undergo analogous ring expansion to give unstable diones. Rearrangement of (291) was effected by LiBr in CCl₄, reflecting not only high reactivity but perhaps also the ability of the keto epoxide to solubilize the salt.



Leriverend and Leriverend¹⁵⁵ used LiI hydrate in refluxing CH_2Cl_2 to examine the conversion of the unsubstituted homolog (**293**; equation 125) into cyclopentanone. Analogs with methyl substituents on the cyclobutane ring reacted more slowly (up to 75 h as opposed to 6 h for **293**) and, interestingly, exhibited a strong preference for migration of the more-substituted carbon. Thus (**294**; equation 126) gave exclusively (**295**).¹⁵⁵ This kind of selectivity appears in a number of spiro epoxide rearrangements discussed below, although usually smaller in magnitude.





Halazy and Krief¹⁵⁶ showed that substituents on the epoxide ring are tolerated, and analogous migration selectivity was exhibited by substrates having one substituent on the cyclobutane and one on the epoxide (equation 127). Substrates with more than two substituents at the pertinent positions have apparently not been examined.



Trost and Latimer's approach to gibberellins¹⁵⁷ made use of the rearrangement shown in equation (128). Migration of the more-substituted center is evident in the conversion of (**298**) to (**299**; 65%), and even more so in the isomer of (**298**), epimeric at the epoxide center, which gave (**299**) in even higher yield (equation 128).



Related selectivity is also evident in the reaction of (300; equation 129), used as an undetermined mixture of epimers at the epoxide center. Ketones (301) and (302) were formed in a 9:1 ratio.¹⁵⁸ The use of THF solvent for this and some other anhydrous LiI-catalyzed reactions is worth noting, since ethers have been shown to inhibit reactions of LiBr.



High migration selectivity is again demonstrated in the efficient conversion of (303; equation 130) to (304), a key step in a synthesis of a pentalenolactone G.¹⁵⁹



Only one exception to this otherwise general selectivity feature has been reported. In prostacyclin synthon studies, Hart and Comte¹⁶⁰ separately examined the two epimeric epoxides (**305**) and (**308**). Both gave only modest selectivity when treated with LiI in THF, with (**305**) showing the normal preference for migration of the more highly substituted center. In a slower reaction, (**308**) rearranged with the opposite selectivity, giving mainly the product (**307**) formed by migration of the methylene carbon (equation 131). This 'abnormal' behavior may be associated with an adverse steric interaction between the R group and the iodine in the conformation of the halohydrin needed for migration of the more-substituted center.



Ring contraction from cyclobutyl to cyclopropyl will also occur if other structural features permit; thus conversion of an epoxide ring to a carbonyl group more than compensates for any free energy cost associated with the change in carbocyclic ring size. Garin's work¹⁶¹ nicely proves the stereochemical features of this kind of rearrangement, and provides convincing evidence for the intermediacy of a halohydrin. Halohydrins formally derived from cyclobutene are known¹⁶² to undergo facile ring contraction. The slow conversion of the initially formed *endo* aldehyde (**311**) to the more stable *exo*-(**312**) form is typical behavior of enolizable aldehydes under LiX epoxide rearrangement conditions (equation 132). Garin also made the interesting observation that lithium thiocyanate could be used in place of LiI to effect this epoxide rearrangement.¹⁶¹ There is no record in the literature of LiSCN causing the rearrangement of any other epoxide, so generality remains doubtful.



Lin and coworkers¹⁶³ used LiBr in acetonitrile to carry out the conversion of the *trans*-diepoxide (**313**) to (**314**); in this instance the enol(ate) generated after the initial aldehyde formation causes aromatization and opening of the other epoxide in an elimination-type reaction (equation 133). The use of BF₃ in ether allowed isolation of the dialdehyde (**315**). It is worth noting that the more acidic conditions cause epoxide rearrangement to occur more rapidly than enolization reactions of this sensitive substrate.

Banks and Ziffer¹⁶⁴ used LiBr in both acetonitrile and in benzene/HMPA in attempts to cause carbon migration (ring contraction) of epoxides such as (**316**; equation 134), but in all instances only ketones formed by hydride migration were obtained. The authors noted the lability caused by the tertiary benzylic center, and suggested that this could account for the failure of normal stereoelectronic features with these substrates. Ring contraction was also thwarted in substrates lacking a suitable hydride for migration; instead, relatively slow elimination with loss of a proton from the exocyclic methyl group occurred.

The work of Magnusson and his colleagues¹⁶⁵ with epoxides derived from allylic alcohols has provided important mechanistic insights as well as useful synthetic procedures. These reactions require the



use of stoichiometric or greater amounts of LiBr-HMPA to obtain high yields, presumably because the water that is formed inhibits the catalytic power of the salt. The formation of the two isomeric aldehydes (320) and (321) from both epoxides (318) and (322) was shown to occur by interconversion of the bromohydrin salts (319) and (323), as depicted in equation (135). The fact that the two epoxides give different ratios of the two aldehydes conversely shows that this halohydrin salt interconversion is not rapid compared to rearrangement. Since the interconversion of (319) and (323) appears to require only proton (and lithium ion) exchange, this is a surprising observation, and suggests that the mechanism may be more complex than depicted here. These intriguing mechanistic points were demonstrated by carrying out the reactions with specifically deuterated substrates.



Similar bromohydrin interconversions were demonstrated in sugar-based epoxide rearrangements.¹⁴⁹ The unsaturated aldehydes formed in these reactions are useful for further elaboration, as shown in a recent synthesis of botryodiplodin.¹⁶⁶

A one-step homologation/ring expansion has been reported,¹⁶⁷ in which a ketone is treated with CH₂I₂ and Li metal. The epoxides that could conceivably be generated from the presumed iodohydrin salts

formed in this process generally do not ring-expand with other lithium salts. Thus if this reaction involves intermediate iodohydrin salts, rearrangement must take place more rapidly than closure to epoxide. Acyclic aldehydes apparently give isolable homologated epoxides from similar treatment.

Kennedy and Buse¹⁴⁸ examined the rearrangement of a series of epoxides on molten salt (LiBr + RbBr) at 280 °C. At this temperature the epoxides also undergo thermal rearrangement, and it is difficult to separate out the effects of the salt-catalyzed process. In general, these reactions are much less selective than the 80 °C LiBr HMPA solution analogs, and some of the products appear to require formation of the alternative halohydrin stereoisomer or carbenium ion intermediates.

Another interesting LiBr variant was recently discovered by Suga and Miyake,¹⁶⁸ who reported that LiBr coated on alumina (LiCl and MgBr₂ were also effective, as was LiBr on silica gel) is an effective catalyst for use either in the gas phase or as a solid added to refluxing toluene, affording 80% of cyclopentanecarbaldehyde from cyclohexene oxide. This reagent (LiBr–alumina) promises to be useful, especially for the formation of large amounts of volatile aldehydes by a continuous stream gas phase process.

There are many different variations for epoxide-carbonyl rearrangements that utilize lithium salts. Unfortunately, there are very few controlled comparisons of the sort needed to determine the best choice of LiX, solvent, addend and conditions for a particular application.

3.3.7 OTHER CATALYSTS

Most common Lewis acids have at one time or another been examined as prospective catalysts for epoxide rearrangements. Relatively few cause skeletal changes, although formal hydride migration leading to ketones is often observed.

Epoxides have occasionally been used in Friedel–Crafts reactions, and some interesting stereochemical observations have been made in this context. Quite unlike secondary alcohols which give almost fully racemized product, it has been shown¹⁶⁹ that optically pure propylene oxide with AlCl₃ and benzene gives optically pure 2-phenyl-1-propanol with inversion of configuration at the cleaved center. AlBr₃ leads to much lower levels of optical purity; it was demonstrated that both starting material and product are optically stable to the reaction conditions, and therefore partial racemization is intrinsic to the mechanism with AlBr₃. It is nonetheless clear from these and other results that even powerful Lewis acids do not assure reaction *via* simple planar carbenium ions.

Strong Lewis acids are often chosen for use with epoxides when the goal is intramolecular addition to an alkene or aromatic system. In an informative study, Sutherland *et al.*¹⁷⁰ compared the effect of different catalysts on the ratio of cyclization (**325**) to ring contraction (**326**) for the system shown in equation (136). A positive correlation between Lewis acid strength and cyclization was noted. Thus AlCl₃ was reasonably effective in this regard (ratio (**325**):(**326**) = 3:1) but it was inferior to TiCl₄ which gave only cyclized product (**55%**). Other catalysts (ratio) are: FeCl₃ (1.1); SnCl₄ (0.6); BF₃·Et₂O (0.8); and ZnCl₂ (0.3).



AlCl₃ in ether has been used to explore intramolecular cyclization of some germacrones, and gives a [5.3.0] bicyclic derivative in one instance, illustrated in equation (137).¹⁷¹



(327)

An organoaluminum compound in hydrocarbon solvent caused rearrangement of an acid-sensitive diene monoepoxide, but a change to ether solvent mediated this (undesired) reaction.¹⁷²

The modest Lewis acid properties of $Al(OPr^i)_3$ are needed for the fragmentation reactions of epoxides derived from homoallylic alcohols described recently by Waddell and Ross.¹⁷³ Carbonyl groups in the initial fragmentation products are reduced by $Al(OPr^i)_3$ to give the materials shown in equations (138) and (139).



Al(OPrⁱ)₃ was less satisfactory than Ti(OPrⁱ)₄ for the hydroxy-influenced cyclizations examined by Sharpless *et al.*¹⁷⁴ The relative stereochemistry of the hydroxy group and epoxide proved critical for this process, leading to the suggestion that an intramolecular metal alkoxide was the active catalyst (equation 140).



SnCl₄ was used with epoxide (**335**), and resulted in a deep-seated rearrangement. The mechanism proposed by Scovell and Sutherland¹⁷⁵ is shown in equation (141).



FeCl₃ is infrequently used to initiate epoxide rearrangements. It has been shown to give chlorohydrins rapidly with a range of oxiranes, simply by brief mixing in ether followed by an aqueous wash.¹⁷⁶ Other rarely used reagents for rearrangement are SbCl₅,¹⁷⁷ Me₂SO + $Pr^{n}I$,¹⁷⁸ various clays,¹⁷⁹ [Rh₂Cl₂(CO)₄]¹⁸⁰ and [Mo(CO)₆],¹⁸¹ each of which has either demonstrated or potential Lewis acid properties. Probably not in this category but worthy of note as a useful synthetic method is the Pd⁰-catalyzed conversion of

2,3-epoxy ketones to 1,3-diones.¹⁸² Some epoxides are rearranged to allylic alcohols by diborane, with subsequent reaction (addition, oxidation) typically leading to diols.¹⁸³

An epoxide has been identified as an intermediate in the copper(I) triflate-catalyzed reaction outlined in equation (142).¹⁸⁴



The zinc halides have a mixed reputation in the field of epoxide chemistry. ZnI₂ is recommended as a catalyst for epoxide opening, without rearrangement, in the addition of thiosilanes¹⁸⁵ and selenosilanes.¹⁸⁶ Conversely, rearrangements catalyzed by zinc salts have been clearly documented for several epoxides. Sutherland's comparison¹⁷⁰ suggests that ZnCl₂ should be rated on the weaker end of the strong acid scale.

Terpene oxides have been favorite substrates for ZnBr₂-catalyzed reactions, probably because early work indicated that certain reactions afforded high yields of unusual materials. Indeed some of these claims have held up well to later analytical scrutiny. For example, α -pinene oxide (342; equation 143) yields essentially pure (343; 88%) upon brief treatment with ZnBr₂ (catalytic amount) in refluxing benzene.¹⁸⁷



More typically, mixtures of products are formed. Terpene epoxides that have been subjected to ZnBr₂catalyzed rearrangement include the oxiranes derived from limonene,¹⁸⁸ carvomenthene,¹⁸⁸ other pinenes,¹⁸⁹ 2,3-carvene,¹⁹⁰ pulegone¹⁹¹ and piperitone.¹⁹¹ ZnCl₂ is rarely employed, but has been compared with AlCl₃ in a study involving several epoxides.¹⁹² In most instances there appears to be no evidence for special advantages of zinc halides compared with other more commonly used Lewis acids.

3.3.8 REFERENCES

- 1. J. G. Buchanan and H. Z. Sable, Sel. Org. Transform., 1972, 2, 1 (a review of stereoselective epoxide cleavages); G. H. Posner and D. Z. Rogers, J. Am. Chem. Soc., 1977, 99, 8214 (alumina-catalyzed opening of several epoxides by a variety of nucleophiles is described); G. Bellucci, G. Berti, G. Ingrosso and E. Mastrorilli, J. Org. Chem., 1980, 45, 299 (hydrolyses promoted by rabbit liver microsomes); R. E. Parker and N. S. Isaacs, Chem. Rev., 1959, 59, 737.
- 2. D. H. R. Barton, J. Chem. Soc., 1953, 1027; diaxial opening was shown to be preferred under both acidic and basic conditions.
- 3. D. K. Murphy, R. L. Alumbaugh and B. Rickborn, J. Am. Chem. Soc., 1969, 91, 2649.
- 4. H. B. Henbest, M. Smith and A. Thomas, J. Chem. Soc., 1958, 3293.
- 5. B. Rickborn and D. K. Murphy, J. Org. Chem., 1969, 34, 3209.
- 6. For a related example of diaxial to diequatorial rearrangement which illustrates the expected transposition of the vicinal substituents see S. Hanessian and A. P. A. Staub, Carbohydr. Res., 1970, 14, 424.
- 7. A. A. Akhrem, A. M. Moiseenkov and V. N. Dobrynin, Usp. Khim., 1968, 37, 1025.
- 8. C. Dupin and J. F. Dupin, Bull. Soc. Chim. Fr., 1970, 249 (optically active styrene oxide); H. Rivière, Bull. Soc. Chim. Fr., 1964, 97 (1-phenylcycloheptene oxide); M. Imuta and H. Ziffer, J. Am. Chem. Soc., 1979, 101, 3990 (indene oxide); C. Battistini, P. Crotti, D. Damiani and F. Macchia, J. Org. Chem., 1979, 44, 1643
- (1-phenyl-4-t-butylcyclohexene oxides); M. Chini, P. Crotti and F. Macchia, J. Org. Chem., 1989, 54, 3930. 9. P. L. Barili, G. Bellucci, B. Macchia, F. Macchia and G. Parmigiani, Gazz. Chim. Ital., 1971, 101, 300 (1-methyl-4-*t*-butylcyclohexene oxides). 10. C. Battistini, P. Crotti and F. Macchia, J. Org. Chem., 1981, 46, 434 (1-ethynylcyclohexene oxide).

- 11. F. Plénat, G. Renard and H. Christol, Bull. Soc. Chim. Fr., Part 2, 1980, 125 (bicyclo[2.2.2]octene oxides).
- 12. N. R. Williams, Chem. Commun., 1967, 1012.
- 13. M. Neeman and J. S. O'Grodnick, Tetrahedron Lett., 1971, 4847.
- 14. H. G. Richey, Jr. and D. V. Kinsman, Tetrahedron Lett., 1969, 2505.
- 15. W. V. McConnell and W. H. Moore, J. Org. Chem., 1963, 28, 822; a similar rearrangement caused by BF3 has also been reported by these workers: J. Org. Chem., 1965, 30, 3480. 16. J. M. Hornback, J. Org. Chem., 1973, 38, 4122.
- 17. A. C. Cope, J. M. Grisar and P. E. Peterson, J. Am. Chem. Soc., 1959, 81, 1640.
- 18. N. Heap, G. E. Green and G. H. Whitham, J. Chem. Soc. C, 1969, 160; J. M. McIntosh, Can. J. Chem., 1972, 50, 2152.
- R. A. Appleton, J. R. Dixon, J. M. Evans and S. H. Graham, Tetrahedron, 1967, 23, 805; E. N. Marvell, J. 19. Seubert, D. Sturmer and W. Federici, J. Org. Chem., 1970, 35, 396.
- 20. L. H. Schwartz, M. Feil, A. J. Kascheres, K. Kaufmann and A. M. Levine, Tetrahedron Lett., 1967, 3785.
- 21. G. A. Morrison and J. B. Wilkinson, Tetrahedron Lett., 1975, 2713.
- 22. T. Masamune, M. Ono, S. Sato and A. Murai, Tetrahedron Lett., 1978, 371; T. G. Waddell, Tetrahedron Lett., 1985, 26, 6277.
- 23. J. Meinwald, S. S. Labana, L. L. Labana and G. H. Wahl, Jr., Tetrahedron Lett., 1965, 1789.
- 24. R. W. Murray, M. K. Pillay and R. Jeyaraman, J. Org. Chem., 1988, 53, 3007.
- 25. J. F. Grove, J. Chem. Soc., Perkin Trans. 1, 1986, 647; J. Gutzwiller, R. M. Mauli, H. P. Sigg and C. Tamm, Helv. Chim. Acta, 1964, 47, 2234; B. Jarvis, personal communication.
- 26. Y. Pocker and B. P. Ronald, J. Am. Chem. Soc., 1980, 102, 5311.
- 27. Y. Pocker, B. P. Ronald and K. W. Anderson, J. Am. Chem. Soc., 1988, 110, 6492.
- 28. R. S. Glass, D. R. Deardorff and L. H. Gains, Tetrahedron Lett., 1978, 2965.
- 29. G. Ohloff and W. Giersch, Helv. Chim. Acta, 1968, 51, 1328; analogous ring-opened products were observed when ZnBr₂ was employed: B. C. Clark, Jr., T. C. Chafin, P. L. Lee and G. L. K. Hunter, J. Org. Chem., 1978, 43, 519; J. A. Donnelly and J. G. Hoey, J. Chem. Soc., Perkin Trans. 1, 1975, 2364 and refs. therein.
- 30. T. Scholl and H. J. Roth, Arch. Pharm. (Weinheim.) Ger. 1985, 318, 634.
- J. Meinwald, M. C. Seidel and B. C. Cadoff, J. Am. Chem. Soc., 1958, 80, 6303.
 J. Meinwald and B. C. Cadoff, J. Org. Chem., 1962, 27, 1539.
- 33. J. S. Yadav, D. G. Patil, R. R. Krishna, H. P. S. Chawla and S. Dev, Tetrahedron, 1982, 38, 1003.
- 34. H. Hart and E. M. Shih, J. Org. Chem., 1975, 40, 1128; H. Hart, S.-M. Chen and S. Lee, J. Org. Chem., 1980, 45, 2096.
- 35. J. R. Hanson and M. Siverns, J. Chem. Soc., Perkin Trans. 1, 1975, 1110; J. R. Hanson and P. B. Reese, Tetrahedron Lett., 1983, 24, 3405.
- 36. M. Audouin and J. Levisalles, Bull. Soc. Chim. Fr., 1985, 1280.
- 37. A. Barco, S. Benetti, G. P. Pollini and R. Taddia, Synthesis, 1975, 104.
- 38. S. P. Singh and J. Kagan, J. Org. Chem., 1970, 35, 2203.
- 39. R. N. McDonald and P. A. Schwab, J. Am. Chem. Soc., 1963, 85, 4004.
- 40. R. N. McDonald and T. E. Tabor, J. Am. Chem. Soc., 1967, 89, 6573.
- 41. R. N. McDonald and R. N. Steppel, J. Am. Chem. Soc., 1970, 92, 5664.
- 42. A. Kirrmann and R. Nouri-Bimorghi, Bull. Soc. Chim. Fr., 1968, 3213.
- 43. K. L. Williamson, J. I. Coburn and M. F. Herr, J. Org. Chem., 1967, 32, 3934.
- 44. J. J. Riehl, P. Casara and A. Fougerousse, C. R. Hebd. Seances Acad. Sci., Ser. C, 1974, 279, 79.
- 45. C. L. Stevens and P. M. Pillai, J. Org. Chem., 1972, 37, 173.
- 46. M. Yamamoto, K. Suzuki, S. Tanaka and K. Yamada, Bull. Chem. Soc. Jpn., 1987, 60, 1523.
- H. Christol, J. Coste and F. Plénat, Bull. Soc. Chim. Fr., 1969, 3934.
 H. Christol, J. Coste and F. Plénat, Bull. Soc. Chim. Fr., 1970, 2005.
- 49. P. Costantino, P. Crotti, M. Ferretti and F. Macchia, J. Org. Chem., 1982, 47, 2917.
- 50. M. P. Hartshorn and D. N. Kirk, Tetrahedron Lett., 1966, 3913; A. K. Ganguly, T. R. Govindachari and A. Manmade, Teirahedron, 1967, 23, 3847; J. W. Blunt, G. S. Boyd, M. P. Hartshorn and M. H. G. Munro, Aust. J. Chem., 1976, 29, 987; S. Yamada, S. Yamada, K. Tachibana Y. Moriyama, Y. Tanahashi, T. Tsuyuki and T. Takahashi, Bull. Chem. Soc. Jpn., 1976, 49, 1134; B. Delmond, M. Taran and J. Valade, Tetrahedron Lett., 1978, 4791.
- 51. G. Berti, F. Bottari, A. Marsili, I. Morelli and A. Mandelbaum, Tetrahedron, 1971, 27, 2143; P. Sengupta, M. Sen and S. N. Maiti, J. Indian Chem. Soc., 1980, 57, 1181.
- 52. Although not commonly isolated and characterized, dimers and oligomers are often formed in reactions of epoxides with strong acids. For an example of partly rearranged dimer from a steroidal epoxide, see K. Yoshida, Tetrahedron, 1969, 25, 1367; J. W. Blunt, M. P. Hartshorn and D. N. Kirk, Tetrahedron, 1966, 22, 3195. The nature of the oligomers is known to be sensitive to epoxide structure in simple examples. Thus propene oxide gives an unusual cyclic tetramer along with a smaller amount of dimer (dioxane), but no cyclotrimer; R. J. Kern, J. Org. Chem., 1968, 33, 388. High dilution techniques are probably generally advisable to help avoid these side products, although this feature has not been widely recognized.
- H. B. Henbest and T. I. Wrigley, J. Chem. Soc., 1957, 4765; J. M. Coxon, M. P. Hartshorn, C. N. Muir and K. E. Richards, Tetrahedron Lett., 1967, 3725; J. R. Bull, Tetrahedron Lett., 1968, 5959; J. M. Coxon, M. P. 53. Hartshorn and B. L. S. Sutherland, Tetrahedron Lett., 1969, 40 295; I. G. Guest and B. A. Marples, J. Chem. Soc., Perkin Trans. 1, 1973, 900. 54. J. K. Crandall and W. H. Machleder, J. Heterocycl. Chem., 1969, 6, 777.
- 55. J. M. Coxon, M. P. Hartshorn, A. J. Lewis, K. E. Richards and W. H. Swallow, Tetrahedron, 1969, 25, 4445.
- 56. J. M. Coxon and C.-E. Lim, Aust. J. Chem., 1977, 30, 1137.
- 57. B. N. Blackett, J. M. Coxon, M. P. Hartshorn and K. E. Richards, Aust. J. Chem., 1970, 23, 2077.
- 58. P. L. Barili, G. Berti, B. Macchia, F. Macchia and L. Monti, J. Chem. Soc. C, 1970, 1168.
- 59. M. G. Fracheboud, O. Shimomura, R. K. Hill and F. H. Johnson, Tetrahedron Lett., 1969, 3951.
- 60. B. N. Blackett, J. M. Coxon, M. P. Hartshorn, B. L. J. Jackson and C. N. Muir, Tetrahedron, 1969, 25, 1479.

- 61. J.-M. Bessiere, H. Christol and D. Panse, C. R. Hebd. Seances Acad. Sci., Ser. C, 1970, 270, 1534.
- 62. C. W. Bird, Y. C. Yeong and J. Hudec, Synthesis, 1974, 27.
- 63. E. J. Corey, R. L. Danheiser and S. Chandrasekaran, J. Org. Chem., 1976, 41, 260.
- 64. R. A. Holton, J. Am. Chem. Soc., 1984, 106, 5731.
- 66. J. M. Coxon, M. P. Hartshorn and W. J. Rae, *Tetrahedron*, 1971, 27 67. Z. Majerski and Z. Hamereck, *J. Org. Chem.* 1001, 1070, 26, 1091. 65. G. Berti, A. Marsili, I. Morelli and A. Mandelbaum, Tetrahedron, 1971, 27, 2217.
- Z. Majerski and Z. Hamersak, J. Org. Chem., 1984, 49, 1182.
- 68. I. Fleming and R. V. Williams, J. Chem. Soc., Perkin Trans. 1, 1981, 684.
- 69. H. O. House and D. J. Reif, J. Am. Chem. Soc., 1955, 77, 6525.
- 70. J. Kagan, D. A. Agdeppa, Jr., S. P. Singh, D. A. Mayers, C. G. Boyajian, C. Poorker and B. E. Firth, J. Am. Chem. Soc., 1976, 98, 4581.
- 71. H. Gerlach and W. Müller, Helv. Chim. Acta, 1972, 55, 2277.
- 72. H. S. Wilgus, III, E. N. Oftedahl, W. J. Musliner and J. W. Gates, Jr., J. Org. Chem., 1967, 32, 3208. Alkyl substitution leads to more complex chemistry, see A. Ichihara, M. Kobayashi, K. Oda and S. Sakamura, Bull. Chem. Soc. Jpn., 1978, 51, 826.
- 73. H. Hikino, N. Suzuki and T. Takemoto, Chem. Pharm. Bull., 1966, 14, 1441.
- 74. R. N. McDonald and D. G. Hill, J. Org. Chem., 1970, 35, 2942.
- 75. J. Cantacužene and D. Ricard, Bull. Soc. Chim. Fr., 1967, 1587.
- 76. R. H. Churi and C. E. Griffin, J. Am. Chem. Soc., 1966, 88, 1824.
- 77. D. F. Tavares, R. E. Estep and M. Blezard, Tetrahedron Lett., 1970, 2373; T. Durst, K.-C. Tin, F. de Reinach-Hirtzbach, J. M. Decesare and M. D. Ryan, Can. J. Chem., 1979, 57, 258.
- 78. See refs. 46 and 77.
- 79. J. M. Domagala and R. D. Bach, J. Am. Chem. Soc., 1978, 100, 1605.
- R. D. Bach and J. M. Domagala, J. Chem. Soc., Chem. Commun., 1984, 1472.
 R. D. Bach and R. C. Klix, Tetrahedron Lett., 1985, 26, 985; W. Reusch, D. F. Anderson and C. K. Johnson, J. Am. Chem. Soc., 1968, 90, 4988.
- 82. A. C. Brouwer, L. Thijs and B. Zwanenburg, Tetrahedron Lett., 1975, 807.
- 83. F. L. M. Smeets, L. Thijs and B. Zwanenburg, Tetrahedron, 1980, 36, 3269.
- 84. F. M. Hershenson and L. Christensen, Synth. Commun., 1981, 11, 615.
- 85. M. W. Creese and E. E. Smissman, J. Org. Chem., 1976, 41, 169.
- 86. J. A. Marshall and J. A. Kerschen, Synth. Commun., 1980, 10, 409; K. Maruoka, T. Ooi and H. Yamamoto, J. Am. Chem. Soc., 1989, 111, 6431.
- 87. S. P. Tanis, Y. M. Abdallah and P. G. Williard, Tetrahedron Lett., 1985, 26, 3651.
- 88. I. Kitagawa, H. Shibuya, H. Fujioka, Y. Yamamoto, A. Kajiwara, K. Kitamura, A. Miyao, T. Hakoshima and K. Tomita, Tetrahedron Lett., 1980, 21, 1963.
- 89. L. A. Paquette and M. Vazeux, Tetrahedron Lett., 1981, 22, 291. The closure can also be effected neatly by the use of an acidic resin: H. E. Simmons, III and J. E. Maggio, *Tetrahedron Lett.*, 1981, 22, 287. J. M. Coxon, M. P. Hartshorn and W. H. Swallow, *Aust. J. Chem.*, 1973, 26, 2521; K. C. Nicolaou, C. V. C.
- 90. Prasad, P. K. Somers and C.-K. Hwang, J. Am. Chem. Soc., 1989, 111, 5330, 5335.
- 91. J. M. Coxon, M. P. Hartshorn and W. H. Swallow, J. Chem. Soc., Chem. Commun., 1973, 261.
- 92. B. Bardili, H. Marschall-Weyerstahl and P. Weyerstahl, Justus Liebigs Ann. Chem., 1985, 275.
- 93. E. E. Van Tamelen, Acc. Chem. Res. 1968, 1, 111; G. A. Molander and S. W. Andrews, J. Org. Chem., 1989, 54, 3114.
- 94. E. E. Van Tamelen, S. R. Zawacky, R. K. Russell and J. G. Carlson, J. Am. Chem. Soc., 1983, 105, 142.
- 95. S. P. Tanis, Y.-H. Chuang and D. B. Head, J. Org. Chem., 1988, 53, 4929; D. J. Goldsmith and C. F. Phillips, J. Am. Chem. Soc., 1969, 91, 5862. 96. H. B. S. Conacher and F. D. Gunstone, Chem. Commun., 1967, 984.
- 97. L. Canonica, M. Ferrari, U. M. Pagnoni, F. Pelizzoni, S. Maroni and T. Salvatori, Tetrahedron, 1969, 25, 1.
- 98. E. Tsankova, Tetrahedron Lett., 1982, 23, 771.
- 99. P. Marsham, D. A. Widdowson and J. K. Sutherland, J. Chem. Soc., Perkin Trans. 1, 1974, 238.
- 100. G. D. Sargent, M. J. Harrison and G. Khoury, J. Am. Chem. Soc., 1969, 91, 4937.
- 101. H. Hart, Acc. Chem. Res., 1971, 4, 337.
- 102. R. O. Hutchins, I. M. Taffer and W. Burgoyne, J. Org. Chem., 1981, 46, 5214.
- T. Cohen and T. Tsuji, J. Org. Chem., 1961, 26, 1681; G. Hanisch and G. Henseke, Chem. Ber., 1968, 101, 2074, 4170; M. W. Miller, Tetrahedron Lett., 1969, 2545.
- 104. K. Schank and D. Wessling, Tetrahedron Lett., 1967, 1823.
- 105. R. C. Cambie, R. A. Franich and T. J. Fullerton, Aust. J. Chem., 1971, 24, 593.
- M. Tiffeneau, Bull. Soc. Chim. Fr., 1945, 12, 453.
 T. A. Geissman and R. I. Akawie, J. Am. Chem. Soc., 1951, 73, 1993.
- 108. H. O. House, J. Am. Chem. Soc., 1955, 77, 3070.
- 109. M. Pedoussaut and H. Coudane, C. R. Hebd. Seances Acad. Sci., Ser. C, 1969, 269, 567.
- 110. D. Y. Curtin and E. K. Meislich, J. Am. Chem. Soc., 1952, 74, 5905.
- 111. H. O. House, J. Am. Chem. Soc., 1955, 77, 5083.
- 112. J. K. Stille and C. N. Wu, J. Org. Chem., 1965, 30, 1222.
- 113. E. P. Kohler and M. Tisler, J. Am. Chem. Soc., 1935, 57, 217.
- 114. M. Tiffeneau, Bull. Soc. Chim. Fr., 1945, 12, 621. 115. A. J. Sisti and A. C. Vitale, J. Org. Chem., 1972, 37, 4090. This feature is also evident in the reaction of trans-2-chloro-4-t-butylcyclohexanone with PhMgBr, F. G. Bordwell and K. C. Yee, J. Am. Chem. Soc. 1970, 92, 5933.
- 116. A. J. Sisti, Tetrahedron Lett., 1967, 5327; J. Org. Chem., 1968, 33, 453.
- 117. A. J. Sisti, J. Org. Chem., 1968, 33, 3953.
- 118. A. J. Sisti, J. Org. Chem., 1970, 35, 2670.
- 119. A. J. Sisti and M. Meyers, J. Org. Chem., 1973, 38, 4431.

- 120. A. J. Sisti and G. M. Rusch, J. Org. Chem., 1974, 39, 1182.
- 121. S. M. Naqvi, J. P. Horwitz and R. Filler, J. Am. Chem. Soc., 1957, 79, 6283.
- 122. M. Rosenberger, W. Jackson and G. Saucy, Helv. Chim. Acta, 1980, 63, 1665.
- 123. T. Matsuda and M. Sugishita, Bull. Chem. Soc. Jpn., 1967, 40, 174.
- 124. P. N. Rao and J. C. Uroda, Tetrahedron Lett., 1964, 1117 and refs. therein; J. Klinot, M. Krumpolc and A. Vystrčil, Coll. Czech. Chem. Soc., 1966, 31, 3174 for a related terpene epoxide rearrangement. A steroidal β-5,6-epoxide gave the 6-ketone by hydride migration when treated with MgBr₂, J. J. Frankel, S. Julia and C. Richard-Neuville, Bull. Soc. Chim. Fr., 1968, 4870.
- 125. H. R. Nace and G. A. Crosby, J. Org. Chem., 1968, 33, 834.
- 126. L. H. Briggs, R. C. Cambie and P. S. Rutledge, J. Chem. Soc., 1963, 5374.
- 127. R. Nouri-Bimorghi, Bull. Soc. Chim. Fr., 1969, 2812.
- 128. J. Cantacužene and A. Keramat, Bull. Soc. Chim. Fr., 1968, 4540.
- 129. P. Coutrot, J.-C. Combret and J. Villiéras, Tetrahedron Lett., 1971, 1553.
- 130. P. Coutrot and C. Legris, Synthesis, 1975, 118.
- 131. P. F. Hudrlik and G. P. Withers, Tetrahedron Lett., 1976, 29.
- 132. P. F. Hudrlik, R. N. Misra, G. P. Withers, A. M. Hudrlik, R. J. Rona and J. P. Arcoleo, Tetrahedron Lett., 1976, 1453.
- 133. M. Obayashi, K. Utimoto and H. Nozaki, Tetrahedron Lett., 1977, 1807.
- 134. In contrast, the reaction of triphenylsilylethylene oxide with MgBr₂ gave triphenylsilylacetaldehyde with negligible selectivity, as shown by deuterium labeling, J. J. Eisch and J. E. Galle, J. Org. Chem., 1976, 41, 2615.
- 135. P. F. Hudrlik, A. M. Hudrlik, R. N. Misra, D. Peterson, G. P. Withers and A. K. Kulkarni, J. Org. Chem., 1980, 45, 4444.
- 136. R. M. Gerkin and B. Rickborn, J. Am. Chem. Soc., 1967, 89, 5850.
- 137. D. B. Denney, J. J. Vill and M. J. Boskin, J. Am. Chem. Soc., 1962, 84, 3944.
- 138. B. Rickborn and R. M. Gerkin, J. Am. Chem. Soc., 1968, 90, 4193.
- 139. J. Bornstein, M. A. Joseph and J. E. Shields, J. Org. Chem., 1965, 30, 801. The very rearrangement-prone 1,1-diphenylethylene oxide gave some rearranged product, 1,2,2-diphenylethanol, when treated with PhLi (LiBr) in ether. It was concluded that this did not result from LiBr-catalyzed rearrangement, since this salt alone in ether failed to cause any loss of epoxide. It was suggested that a base-induced rearrangement (A. C. Cope, P. A. Trumbull and E. R. Trumbull, J. Am. Chem. Soc., 1958, 80, 2844) might be responsible. This observation remains somewhat puzzling; although we would not expect LiBr in ether to be a potent catalyst, the base-induced rearrangement should generate diphenylacetaldehyde as its enolate, not subject to further addition of RLi.
- 140. J. K. Heeren, T. C. Flood and R. H. Cunningham, J. Organomet. Chem., 1967, 9, P18. An especially interesting product, formed from MeLi and cyclohexene oxide in the presence of LiI, is 1-methylcyclohexanol (10-20%). The authors suggested base-induced rearrangement to cyclohexanone followed by MeLi addition to account for this product, but again (cf. ref. 139) it must be noted that the enolate produced in this manner ought not to add MeLi. Significantly, 1-methylcyclohexanol was not formed in the reaction with MeLi/LiBr. Alternative LiI-catalyzed processes discussed in the text appear more likely sources of this product.
- 141. B. Rickborn and R. M. Gerkin, J. Am. Chem. Soc., 1971, 93, 1693.
- Y. Pocker and R. F. Buchholz, J. Am. Chem. Soc., 1970, 92, 2075. 142.
- 143. C. A. Bunton and M. D. Carr, J. Chem. Soc., 1963, 5854.
- 144. B. M. Trost and M. J. Bogdanowicz, J. Am. Chem. Soc., 1973, 95, 5321; it was noted that Eu(fod)₃ in chloroform also effected the rearrangement of (263; equation 112) with very high specificity, affording (264). 145. J. E. McMurry, J. H. Musser, M. S. Ahmad and L. C. Blaszczak, J. Org. Chem., 1975, 40, 1829.
- 146. E. J. Corey, M. A. Tius and J. Das, J. Am. Chem. Soc., 1980, 102, 1742.
- 147. G. L. Burge, D. J. Collins and J. D. Reitze, Aust. J. Chem., 1982, 35, 1913.
- 148. J. H. Kennedy and C. Buse, J. Org. Chem., 1971, 36, 3135.
- 149. A. Sundin, T. Frejd and G. Magnusson, J. Org. Chem., 1986, 51, 3927.
- 150. It follows from the kinetic order that the major species present in solution must also be the kinetically active form.
- 151. D. Barr, R. Snaith, D. S. Wright, R. E. Mulvey and K. Wade, J. Am. Chem. Soc., 1987, 109, 7891.
- 152. J. E. Herweh and W. J. Kauffman, Tetrahedron Lett., 1971, 809.
- 153. J. R. Salaün and J.-M. Conia, Chem. Commun., 1971, 1579; J. R. Salaün, J. Champion and J.-M. Conia, Org. Synth., 1977, 57, 36.
- 154. F. Huet, A. Lechevallier and J.-M. Conia, Chem. Lett., 1981, 1515.
- 155. M.-L. Leriverend and P. Leriverend, C. R. Hebd. Seances Acad. Sci., Ser. C, 1975, 280, 791.
- 156. S. Halazy and A. Krief, J. Chem. Soc., Chem. Commun., 1982, 1200.
- 157. B. M. Trost and L. H. Latimer, J. Org. Chem., 1978, 43, 1031.
- 158. D. R. Morton, Jr. and F. C. Brokaw, J. Org. Chem., 1979, 44, 2880.
- 159. M. C. Pirrung and S. A. Thomson, J. Org. Chem., 1988, 53, 227.
- 160. T. W. Hart and M.-T. Comte, Tetrahedron Lett., 1985, 26, 2713.
- 161. D. L. Garin, J. Org. Chem., 1971, 36, 1697.
- 162. P. R. Brook and A. J. Duke, Chem. Commun., 1970, 652; J. R. Salaün and J.-M. Conia, Chem. Commun., 1970, 1358.
- 163. Y. Lin, S. A. Lang, Jr., C. M. Seifert, R. G. Child, G. O. Morton and P. F. Fabio, J. Org. Chem., 1979, 44, 4701.
- 164. H. Banks and H. Ziffer, J. Org. Chem., 1982, 47, 3743.
- 165. G. Magnusson and S. Thorén, J. Org. Chem., 1973, 38, 1380; R. Bergman and G. Magnusson, J. Org. Chem., 1986, 51, 212.
- 166. N. Rehnberg, T. Frejd and G. Magnusson, Tetrahedron Lett., 1987, 28, 3589.
- 167. P. J. Calabretta, Int. Congr. Essent. Oils, [Pap.], 6th, 1974, 131 (Chem. Abstr., 1975, 84, 121 242y).

- 168. H. Suga and H. Miyake, Synthesis, 1988, 394.
- 169. T. Nakajima, S. Suga, T. Sugita and K. Ichikawa, Tetrahedron, 1969, 25, 1807.
- 170. E. Huq, M. Mellor, E. G. Scovell and J. K. Sutherland, J. Chem. Soc., Chem. Commun., 1978, 526.
- 171. M. Niwa, M. Iguchi and S. Yamamura, Tetrahedron Lett., 1975, 3661.
- 172. G. A. Crosby and R. A. Stephenson, J. Chem. Soc., Chem. Commun., 1975, 287; J.-L. Namy, D. Abenhaim and G. Boireau, Bull. Soc. Chim. Fr., 1971, 2943.
- 173. T. G. Waddell and P. A. Ross, J. Org. Chem., 1987, 52, 4802.
- 174. D. J. Morgans, Jr., K. B. Sharpless and S. G. Traynor, J. Am. Chem. Soc., 1981, 103, 462.
- 175. E. G. Scovell and J. K. Sutherland, J. Chem. Soc., Chem. Commun., 1978, 529.
- 176. J. Kagan, B. E. Firth, N. Y. Shih and C. G. Boyajian, J. Org. Chem., 1977, 42, 343.
- 177. E. G. E. Hawkins and R. Large, J. Chem. Soc., Perkin Trans. 1, 1973, 2169.
- 178. D. Betheil, G. W. Kenner and P. J. Powers, Chem. Commun., 1968, 227.
- 179. E. Ruiz-Hitzky and B. Casal, J. Catal., 1985, 92, 291.
- R. Grigg and G. Shelton, Chem. Commun., 1971, 1247; G. Adames, C. Bibby and R. Grigg, J. Chem. Soc., Chem. Commun., 1972, 491.
- 181. H. Alper, D. Des Roches, T. Durst and R. Legault, J. Org. Chem., 1976, 41, 3611.
- 182. M. Suzuki, A. Watanabe and R. Noyori, J. Am. Chem. Soc., 1980, 102, 2095.
- 183. Y. Bessiere-Chretien and B. Meklati, Tetrahedron Lett., 1971, 621.
- 184. T. Cohen, D. Kuhn and J. R. Falck, J. Am. Chem. Soc., 1975, 97, 4749
- 185. Y. Guindon, R. N. Young and R. Frenette, Synth. Commun., 1981, 11, 391.
- 186. N. Miyoshi, K. Kondo, S. Murai and N. Sonoda, Chem. Lett., 1979, 909.
- 187. J. B. Lewis and G. W. Hendrick, J. Org. Chem., 1965, 30, 4271.
- 188. R. L. Settine, G. L. Parks and G. L. K. Hunter, J. Org. Chem., 1964, 29, 616.
- 189. Y. Bessière-Chrétien and J. P. Bras, C. R. Hebd. Seances Acad. Sci., Ser. C, 1970, 271, 200.
- 190. B. C. Clark, Jr., T. C. Chafin, P. L. Lee and G. L. K. Hunter, J. Org. Chem., 1978, 43, 519.
- 191. H. Watanabe, J. Katsuhara and N. Yamamoto, Bull. Chem. Soc. Jpn., 1971, 44, 1328.
- 192. R. Guyon and P. Villa, Bull. Soc. Chim. Fr., 1975, 2593.

3.4 The Semipinacol and Other Rearrangements

DONAL J. COVENEY

Avondale Chemical Company, Rathdrum, Republic of Ireland

3.4.1 INT	RODUCTION	777
3.4.1.1	Semipinacol Rearrangements	777
3.4.1.2	Boron Rearrangements	779
3.4.2 SEN	IPINACOL REARRANGEMENTS	781
3.4.2.1	Rearrangements of 2-Amino Alcohols	781
3.4.2.2	Additions of Diazoalkanes to Ketones	783
3.4.2.3	Rearrangements of 2-Hydroxy Sulfides	784
3.4.2.4	Rearrangements of 2-Hydroxy Selenides	786
3.4.2.5	Rearrangements of Halohydrins	787
3.4.2.6	Rearrangements of 2-Halo Ketones and Acetals	788
3.4.2.7	Rearrangements of 2-Amino Ketones and 2-Hydroxyimines	790
3.4.2.8	Rearrangements of 2-Hydroxy Ketones	791
3.4.2.9	Tandem Cyclization-Semipinacol Rearrangements	792
3.4.3 ORGANOBORON REARRANGEMENTS		793
3.4.3.1	Rearrangements of Triorganylboranes	793
3.4.3.2	Rearrangements of Boronic Esters	796
3.4.3.3	Rearrangements of Tetraorganylboronates	798
3.4.4 REF	TERENCES	800

3.4.1 INTRODUCTION

3.4.1.1 Semipinacol Rearrangements

The semipinacol rearrangement was originally defined as a special case of the pinacol rearrangement, in which migration occurred towards the secondary center of a tertiary, secondary diol (equation 1), *i.e.* the reverse of the regiochemistry observed in the pinacol rearrangement. However, the term 'semipinacol' is now used generally to describe all such rearrangements which are related to, or reminiscent of, the pinacol rearrangement.¹ The largest class of semipinacol rearrangements are those derived from 2-heterosubstituted alcohols (1), such as halohydrins (1; X = Cl, Br), 2-amino alcohols (1; $X = NH_2$), 2-hydroxy sulfides (1; X = SR) and 2-hydroxy selenides (1; X = SeR).

The normal pathway for rearrangement of the substrates (1) is via loss of the heteroatom substituent with migration of an adjacent alkyl (or aryl) group, and concomitant ketone formation (Scheme 1, path a). An alternative migration (path b), is occasionally observed in protic media, *i.e.* protonation of the alcohol (1) and migration with loss of water generates the stabilized carbocation (2), which is then hydrolyzed.



Scheme 1

Epoxide formation (path c) is an important side reaction which can become the dominant pathway. For example, the addition of sulfur ylides to ketones (equation 2) constitutes a general synthesis of epoxides,² while 2-hydroxy sulfides undergo the semipinacol rearrangement under certain conditions (equation 3).³ Elimination (path d) is observed in some special cases such as 2-hydroxysilanes (1; $X = SiR_3$; the Peterson alkenation)⁴ and 2-hydroxyphosphonium species (1; $X = PR_3^+$; Wittig intermediates).⁵



The semipinacol rearrangement is not restricted to 2-heterosubstituted alcohols. Thus, the addition of diazoalkanes to ketones yields homologated ketones (equation 4), *via* rearrangement of the adduct (3). This process is closely related to the rearrangement of 2-amino alcohols on treatment with nitrous acid (equation 5). Similarly, 2-hydroxyimines undergo rearrangement to 2-amino ketones in a related process (equation 6).

The rearrangement of acetals of 2-haloalkyl aryl ketones is a well-documented process yielding esters of 2-arylalkanoic acids by 1,2-aryl shift (equation 7).⁶ The mechanism of this rearrangement is reminiscent of other semipinacol rearrangements. Loss of the halogen (usually assisted by Lewis acid), yields a carbocation (4), which then undergoes a 1,2-aryl shift with carbonyl group formation.

$$\begin{array}{c} O \\ R^{1} \\ R^{2} \\ R^{2} \\ R^{2} \\ R^{2} \\ R^{3} \\ R^{3} \end{array} \begin{array}{c} O \\ R^{2} \\ R^{3} \\ R^{3} \\ R^{3} \\ R^{3} \\ R^{3} \end{array} \begin{array}{c} O \\ R^{3} \\ R^{2} \\ R^{3} \\ R^{2} \\ R^{3} \end{array}$$
(4)



A number of tandem cyclization-semipinacol rearrangements have been reported,⁷ whereby cyclization is initiated by an acetal and terminated by a semipinacol rearrangement (equation 8).



In general, semipinacol rearrangements are more selective than the classical pinacol rearrangement in terms of regiochemistry and mildness of conditions. Consequently there have been many applications of the semipinacol rearrangement in the formation of carbon-carbon bonds. The carbon atom bearing the migrating group is usually tertiary, while the carbon receiving the migrating group can be primary, secondary or tertiary. Migratory aptitude is not an important factor in semipinacol rearrangements. Electronic factors, the relief of ring strain and the stereochemistry of the substrate all influence the course of migration. Inversion of configuration is observed in most cases, and this feature has been utilized in chiral synthesis. With some exceptions, the mechanism of semipinacol rearrangements has not been examined in detail. Most investigations have centered on the synthetic utility of the rearrangement.

3.4.1.2 Boron Rearrangements

Organoboranes undergo rearrangements with the formation of carbon-carbon σ -bonds. The boron atom plays a pivotal role in the rearrangement by accepting a nucleophile (5) to produce an intermediate four-coordinate organoborate species (6). Migration from boron to carbon occurs if a leaving group is positioned on the carbon center α to boron (equation 9), yielding a new neutral organoborane. For example, ylides of sulfur (5; L = SMe₂⁺) and nitrogen (5; L = NMe₃⁺) add to triorganylboranes to yield rearranged products; anions of haloalkanes (5; L = halogen) effect a similar transformation.



The carbonylation of triorganylboranes is well documented,⁸ and provides a unique route to tertiary alcohols. Thus, triorganylboranes react with carbon monoxide to yield a trimeric product (7) which is readily oxidized to a tertiary alcohol (Scheme 2). The exact mechanism of this process is not clear, but it is known to proceed by three sequential rearrangements from boron to carbon.



Scheme 2

Boronic esters (8) react with dichloromethyllithium to form intermediate borate complexes which then undergo rearrangement to form 2-haloboronic esters (9). When treated with a Grignard reagent, the 2-haloboronic esters (9) rearrange to secondary alkylboronic esters (10; Scheme 3).

i, LiCHCl₂; ii, R²MgX

Scheme 3

This iterative procedure has been utilized with great effect in asymmetric synthesis via chiral boronic esters derived from chiral alcohols.⁹

Certain tetraorganylborate salts are stable isolable species, and undergo rearrangement by migration from boron to carbon when treated with electrophiles. For example, the cyanoborates (11), when treated with acylating agents, yield the heterocyclic intermediate (12) by two successive rearrangements (Scheme 4). Oxidation of the intermediate (12) yields ketones. Similarly, alkynylborates produce alkynes when treated with iodine, *via* rearrangement and elimination (equation 10).



Scheme 4

$$R_{3}\overline{B} \longrightarrow R' \xrightarrow{I_{2}} R_{2}B \xrightarrow{I_{2}} R \xrightarrow{I_{2}} R \xrightarrow{I_{2}} R' \qquad (10)$$

There are many applications of boron rearrangements in organic synthesis and the subject has been comprehensively reviewed.¹⁰ This methodology offers considerable scope in the formation of carbon-carbon bonds.

A common mechanism involved in these rearrangements is one whereby an organic group migrates from a four-coordinate, electron-rich boron center, to an electron-deficient α -carbon center. In general, primary groups migrate before secondary, which migrate before tertiary. This migratory aptitude can be rationalized by consideration of a partial negative charge on the nucleophilic migrating group (equation 11), and hence reflects the relative stabilization of negative charge.

$$R_{3}B + C^{-}L \longrightarrow \begin{bmatrix} R_{2}^{\delta} \\ R_{2}B \\ L \end{bmatrix} \longrightarrow R_{2}B \\ R_{2}B \end{pmatrix}$$
(11)

3.4.2 SEMIPINACOL REARRANGEMENTS

3.4.2.1 Rearrangements of 2-Amino Alcohols

The semipinacol rearrangement of primary 2-amino alcohols on exposure to nitrous acid is known generally as the Tiffeneau–Demjanov rearrangement. The reaction is well documented and is often associated with ring enlargement¹¹ and, to a lesser extent, ring contraction. Thus, a cyclic ketone may be homologated in a three-step sequence *via* cyanohydrin formation, followed by reduction to an aminomethylcycloalkanol and rearrangement (Scheme 5). This sequence has been applied to the formation of five- to eight-membered rings, for example in the preparation of cyclooctanone (equation 12).¹² Trimethylsilyl cyanide forms cyanohydrin *O*-silyl ethers of ketones resistant to HCN treatment, and has been recommended for the preparation of 2-amino alcohols.¹³



i, HCN; ii, LiAlH4 or PtO2 / H2; iii, HNO2

Scheme 5



In general, the rearrangement is most effective when the hydroxy group is tertiary. However, rearrangement is possible with a secondary hydroxy group, as Woodward *et al.*¹⁴ have demonstrated in the ring contraction of the bicyclic acetal (13; equation 13), used in a synthesis of prostaglandin $F_{2\alpha}$. The degree of substitution at the carbon bearing the primary amino group is variable, but an aryl group has been shown to inhibit rearrangement.¹⁵



In ring expansion reactions, the least-substituted group shows a greater tendency to migrate (equations 14 and 15), although the selectivity may be attributable to conformational effects. The importance of conformational effects has been amply demonstrated in the rigid steroidal framework. Thus, the D-ring homosteroid (14) rearranges to the ketone (15; equation 16),¹⁷ while the alcohol epimer (16) does not rearrange under the same conditions, but instead forms the epoxide (17; equation 17).¹⁸ In marked contrast, the homosteroid (18), which is diastereoisomeric with (14) at the amino alcohol function, undergoes ring contraction (equation 18).¹⁹ In these rigid systems it is the group that is *trans* coperiplanar to the amino function which migrates. In the case of the amino alcohol (16) the hydroxy group is *trans* coperiplanar, and epoxide formation is favored (Scheme 6).





Scheme 6

Inversion of configuration is normally observed on rearrangement of cyclic 2-amino alcohol substrates. Collins *et al.*²⁰ have examined in detail the stereochemical course of the Tiffeneau–Demjanov rearrangement in acyclic substrates. In a series of 2-amino-1,2-diaryl alcohols studied, variable amounts of inversion were observed, while in some cases retention of configuration predominated (equation 19). This complexity can be attributed to the flexibility of such acyclic substrates where several conformations are possible in the transition states.



3.4.2.2 Additions of Diazoalkanes to Ketones

The homologation of ketones by the addition of diazoalkanes complements the Tiffeneau-Demjanov rearrangement. Epoxide formation is a side reaction which can be minimized if polar aprotic solvents are avoided (Scheme 7). Rearrangement (*i.e.* homologation) is maximized in ether solvents or by Lewis acid catalysis.²¹ The reaction is most effective in the ring expansion of cyclic ketones.





Scheme 7

In the addition of simple diazoalkanes to ketones, the product homologated ketone is often an effective substrate for further homologation. For example, norbornenone (19), when treated with an excess of diazomethane, yields a complex mixture of homologated ketones (equation 20).²² However, high yields of a desired homologated ketone can be obtained, as in the preparation of 5-ethoxycarbonyl-2-methylcycloheptanone from 4-ethoxycarbonylcyclohexanone (equation 21).²³ With unsymmetrical ketones a mixture, due to migration of either group, is obtained (equation 22),²⁴ and the degree of substitution of either group appears to be of little importance.²⁵ However, 2-halo ketones undergo rearrangement with exclusive migration of the nonhalogenated group, and this selectivity has been utilized in a regioselective synthesis of ketones (Scheme 8).²⁶



Ethyl diazoacetate has been used extensively in the homologation of ketones to 3-keto esters. Lewis acid is required for the reaction (equation 23).¹⁰⁵ There is a tendency for the least-substituted group to migrate, particularly if one group is fully substituted (equation 24).²⁷ In a total synthesis of (\pm) -aplysin (**20**) this selective rearrangement was applied (Scheme 9),²⁸ and a similar approach was used in a synthesis of (+)-hirsutic acid.²⁹ Ethyl diazoacetate has also been used in the homologation of acyclic



Scheme 8

ketones. The presence of a 2-halogen promotes selective migration of the nonhalogenated group. The halogen 'protection' and the ethoxycarbonyl group can be removed reductively in one operation (Scheme 10).³⁰ The anion of ethyl diazoacetate adds to ketones to produce isolable 3-hydroxy-2-diazo esters, which rearrange to 3-keto esters when exposed to metal salt catalysis (Scheme 11).³¹



3.4.2.3 Rearrangements of 2-Hydroxy Sulfides

2-Hydroxy sulfides rearrange to ketones under a variety of conditions. In aqueous acid it is the carbon bearing the sulfur group which ultimately becomes the carbonyl carbon. The rearrangement proceeds *via* protonation of the hydroxy group (Scheme 1, path b), and appears to be restricted to the expansion of small rings, particularly cyclopropanes (equation 25).³² Trost and coworkers have developed this method, whereby the anion of cyclopropyl phenyl sulfide adds to a ketone and the adduct then rearranges to a cyclobutanone.^{32,33}



Two such rearrangements were used in the preparation of the cyclobutanone (21), a key intermediate in a synthesis of grandisol (22; Scheme 12).³³ A similar rearrangement is observed if the alcohol is allylic to the sulfide (equation 26),³⁴ and the formation of cyclobutanones from 1-thiophenyl-1-vinylcyclopropanes³⁵ is a closely related process.



When exposed to soft electrophiles such as dichlorocarbene (generated *in situ*), 2-hydroxy sulfides rearrange with loss of the sulfur group and ketone formation at the hydroxy-bearing carbon,^{3,36} as in the synthesis of cuparenone (23; equation 27).³



The anion of 1,1-dithiophenoxymethane adds to cyclic ketones and the adduct then rearranges to a ring-expanded ketone on metal salt catalysis (equation 28).^{36a} The least-substituted group migrates in this case, as in the trithioalkyl derivative (**24**; equation 29), which was utilized in a synthesis of coriolin.^{36b}

In a similar vein, the adducts derived from the anion of methoxymethyl phenyl sulfone and cyclic ketones rearrange on Lewis acid catalysis (Scheme 13).³⁷ This rearrangement is quite stereospecific, in that a single diastereoisomer is formed and it is the most substituted group which migrates. Similar results were obtained with adducts derived from phenyl thiomethyl phenyl sulfone (Scheme 14).³⁷





Rearrangement is sometimes observed on the addition of sulfur ylides to ketones, yielding a homologated ketone rather than the expected epoxide (equation 30).³⁸ Hydroxy sulfides can undergo elimination to form alkenes, but the conditions are quite specific and require derivatization at oxygen³⁹ or sulfur.⁴⁰



3.4.2.4 Rearrangements of 2-Hydroxy Selenides

The semipinacolic rearrangements of 2-hydroxy selenides are closely related to those of 2-hydroxy sulfides, but they have not been studied as extensively, and the use of selenium does not seem to offer any advantages over sulfur. The rearrangement has been applied mainly to the ring expansion of cyclic ketones *via* the addition of α -selenoalkyl anions.

Paquette *et al.*⁴¹ prepared the complex tricyclic ketone (25) by a selective rearrangement of a 2-hydroxy selenide (equation 31). In this case, it was the alkyl rather than the vinyl group which migrated. However, this preference is not as marked in a less rigid system.⁴¹ A similar selectivity was noted in a cyclobutene derivative,⁴² and Krief *et al.*⁴³ found that selectivity was highly dependent on ring size and the degree of substitution at the migrating center.

Dichlorocarbene (generated *in situ*), is generally used to induce migration, but silver tetrafluoroborate has also been used, as in the preparation of cuparenone (23; equation 32).⁴⁴ Rearrangement occurs only if the selenyl moiety is attached to a fully substituted carbon. For example, the hydroxy selenide (26) does not rearrange but instead yields the epoxide (27; equation 33).⁴⁴



This limitation can be overcome using the anion of alkyl phenyl selenoxides which add to ketones and rearrange *in situ* (equation 34).⁴⁵ The amalgam reduction is required to remove the small amount of α -phenyl selenated ketones formed in the reaction.



2-Hydroxy selenides undergo elimination to alkenes when treated with an excess of methanesulfonyl chloride, but elimination is not an important side reaction in rearrangement reactions.⁴⁶

3.4.2.5 Rearrangements of Halohydrins

Halohydrins undergo semipinacolic rearrangement when exposed to base. The rearrangement of epoxides with metal halides is a closely related process and both rearrangements have been described in Chapter 3.3. However, the adducts of ketones and polyhalomethyllithium undergo rearrangements analagous to halohydrins, but by a quite different mechanism. The adducts are easily prepared by treating a mixture of polyhalomethane and ketone with an N,N-disubstituted lithium amide (equation 35).⁴⁷ When a dichloromethylcycloalkanol is treated with 2 equiv. of *n*-butyllithium, a dilithiated species (28) is formed, which then rearranges to form a homologated 2-chloro ketone (Scheme 15).⁴⁸



A dibromomethylcycloalkanol rearranges in a similar fashion, except that a different dilithio intermediate is formed via halogen-metal exchange rather than deprotonation (Scheme 16).⁴⁸ With unsymmetrical ketones, the more substituted group migrates. This selectivity has been applied to a synthesis of (\pm) -muscone (29; Scheme 17).⁴⁸ Substituted polyhalomethyllithium species can be used in the prepara-



Scheme 15

tion of 2-substituted homologated ketones (Scheme 18).⁴⁹ The O-silyl ethers of dihaloalkylcarbinols do not rearrange when treated with n-butyllithium but instead eliminate to form vinyl halides.⁵⁰



3.4.2.6 Rearrangements of 2-Halo Ketones and Acetals

The semipinacolic rearrangement of 2-halo ketones and acetals to esters or acids is a valuable synthetic procedure, and the reaction has attracted considerable interest in the synthesis of 2-arylalkanoic acids,⁶ several of which are valuable pharmaceuticals. The rearrangement is related to the Favorskii rearrangement and is sometimes referred to as the 'quasi-Favorskii' rearrangement. Under basic conditions, the semipinacolic mechanism applies with 2-halo ketones where the cyclopropanone intermediate of the Favorskii rearrangement cannot form. Such examples are ketones which do not have an α' -hydrogen (Scheme 19),⁵¹ or certain cyclobutanones (equation 36).⁵² Low yields in the base-induced rearrangement are often due to preferential displacement of the halogen by hydroxide ion. However, the rearrangement can be induced under milder conditions, such as silver nitrate in methanol, which was used in the chlorocyclobutanone ring contraction step in a recent synthesis of (±)-sirenin (**30**; Scheme 20).⁵³ Lewis acid is equally effective in the rearrangement of aryl bromoalkyl ketones (equation 37),⁵⁴ which is limited to secondary or tertiary bromides and to an electron-rich aromatic ring.



Scheme 19



The acetals of 2-haloalkyl aryl ketones rearrange to esters of 2-arylalkanoic acids under a wide range of conditions.⁶ Simply heating these substrates in aqueous alcohol containing sodium acetate provides the ester in high yield (equation 38).⁵⁵ Primary halides or electron-deficient arenes do not rearrange under these conditions. The use of silver salts in alcoholic media is more general (equation 39).⁵⁶ The rearrangement proceeds with inversion of configuration and an asymmetric synthesis of (+)-naproxen highlighting this feature has been described (Scheme 21).⁵⁷ Thus, the acetal (31), derived from (2R,3R)-tartaric acid, undergoes stereospecific bromination to yield a bromo acetal, which rearranges in high yield to the ester of (+)-naproxen. Yamauchi *et al.*⁷⁰ have demonstrated the similarity of these rearrangements by preparing esters of 2-arylalkanoic acids from ketones, acetals or enol ethers, using iodine, iodine monochloride or iodine trichloride in trimethylorthoformate solution (Scheme 22).



Scheme 21



3.4.2.7 Rearrangements of 2-Amino Ketones and 2-Hydroxyimines

Certain 2-amino ketones undergo rearrangement on thermolysis to produce new 2-amino ketones. The reaction proceeds by an initial 1,2-shift to yield a hydroxyimine, followed by a second 1,2-shift to the observed product (Scheme 23).⁵⁸ The amine must be primary or secondary and attached to a tertiary center. As would be expected, 2-hydroxyimines rearrange to 2-amino ketones (equation 40).⁵⁸ Similarly, 2-hydroxy ketones when heated with amines form rearranged 2-amino ketones *via* 2-hydroxyimines (equation 41).⁵⁸ This reaction has been applied in the synthesis of a D-ring homosteroid (equation 42).⁵⁹



The semipinacolic rearrangement of 2-hydroxyimines has also been applied to the synthesis of 2amino ketones which are not easily available by other methods (equation 43).⁶⁰ An impressive example is the biomemetic formation of the spiroindoxyl brevianamide A (32; equation 44).⁶¹



3.4.2.8 Rearrangements of 2-Hydroxy Ketones

The rearrangement of 2-hydroxy ketones is analagous to the rearrangement of 2-hydroxyimines, and is termed the acyloin rearrangement. The reaction is of limited use, however, and has not found wide application. The rearrangement is initiated by base or Lewis acid (Scheme 24).⁶² With unsymmetrical ketones it is possible for either group to migrate, depending on the conditions. With 3β , 17α -dihydroxy- 5α -pregnan-20-one, the more substituted group migrates under basic conditions (equation 45),⁶³ while Lewis acids promote migration of the least substituted group (equation 46).⁶² Ring contraction is also possible, as in the case of the steroidal substrate (33; equation 47).⁶⁴





3.4.2.9 Tandem Cyclization-Semipinacol Rearrangements

A semipinacol rearrangement in tandem with a cyclization has been shown to be a powerful method for the construction of carbocyclic rings. Here the cyclization is initiated by an acetal function and terminated by a semipinacol rearrangement (Scheme 25).⁷ In some cases, the cyclization has been found to be stereospecific at the acetal center (equation 48),⁶⁵ and up to four chiral centers can be generated in a single step (equation 49).⁶⁶ Highly substituted tetrahydrofurans are obtained from cyclic acetals (Scheme 26),⁶⁷ and stereoselective tetrahydrofuran annelations are possible using bicyclic acetals (Scheme 27).⁶⁸



Scheme 26



Scheme 27

3.4.3 ORGANOBORON REARRANGEMENTS

3.4.3.1 Rearrangements of Triorganylboranes

Trialkylboranes react with carbon monoxide in the presence of ethylene glycol to yield tertiary alkylboronic esters (Scheme 28), which on oxidation yield trialkylmethanols.^{8,69} The presence of ethylene glycol is not necessary for this rearrangement but it greatly facilitates the oxidation reaction. The alkyl groups can be primary, secondary or tertiary, but tertiary groups do not migrate easily and more vigorous conditions are required (Scheme 29).⁷¹



i, BH₃; ii, CO, (CH₂OH)₂, 100 °C; iii, NaOH, H₂O₂

Scheme 28



i, CO (70 atm), (CH₂OH)₂, 150 °C; ii, NaOH, H₂O₂, 60% (2 steps)

Scheme 29

If water is present in the carbonylation reaction, rearrangement of the third group is inhibited and ketones instead are obtained on oxidation (Scheme 30).⁷² The mechanism of this sequence is not clear but the reaction proceeds *via* the intermediate (**34**). Unsymmetrical ketones are available from thexyldialkylboranes (equation 50),⁷³ since the thexyl group does not migrate easily. Cyclic ketones are also available by this route (Scheme 31).⁷⁴ When the carbonylation is performed in the presence of reducing agents, only one alkyl group migrates. Aldehydes and alcohols are then available by this method.^{8,75}



i, CO, H₂O, 100 °C; ii, NaOH, H₂O₂, 82%

Scheme 30



Scheme 31

There are certain disadvantages to the carbonylation reaction, such as high temperature and pressure. An alternative preparation of trialkylmethanols is the use of the anion of dichloromethyl methyl ether. Addition of this anion to a trialkylborane is followed by three successive migrations to yield the trialkylmethanol on oxidation. The reaction takes place at room temperature, and tertiary groups migrate without difficulty (Scheme 32).⁷⁶



i, CHCl₂OMe, Et₃COK; ii, (CH₂OH)₂, NaOH, H₂O₂

Scheme 32

2-Halocarbonyl compounds react with triorganylboranes to yield 2-alkylated derivatives. 9-Alkyl- or 9-aryl-borabicyclononane derivatives yield the 9-alkyl/aryl group selectively (equation 51).⁷⁷ Sequential dialkylation of dichloroacetonitrile is feasible by this method.⁷⁸ Trialkylboranes also react with halo sulfones (equation 52).⁷⁹ and diazo ketones (equation 53).⁸⁰ in a similar fashion.





i, LiCH₂SMe; ii, MeI; iii, NaOH, H₂O₂

Scheme 33

Alkylbromoalkenylthexylboranes are available from thexylborane by sequential hydroboration of alkenes and bromoalkynes, and they rearrange to borinic esters when treated with methoxide. Protonation yields (E)-alkenes of high purity (Scheme 34).⁸² Rearrangement takes place with inversion of configuration at the halogen center, while protonation proceeds with retention of configuration.



Scheme 34

Monoalkylbromoboranes hydroborate terminal alkynes to yield alkylalkenylbromoboranes, which then rearrange to yield (Z)-alkenes after protonation; this useful procedure has been applied to the preparation of muscalure (35; Scheme 35).⁸³ (E)-Alkenes are available by a similar sequence if 1-bromoalkynes are employed at the hydroboration stage.⁸⁴ Similarly, dialkenylthexylboranes yield dienes on rearrangement and protonation.⁸⁵



The rearrangements of triorganylboranes have focused on the use of alkyl and alkenyl groups, due to the availability of these substrates by hydroboration. However, aryl groups are equally effective for rearrangement and this aspect warrants further study. There are many variants of the rearrangements of triorganylboranes which constitute a wide and varied methodology,^{8,10a,75,86,87} some of which have not been fully exploited in synthesis.
3.4.3.2 Rearrangements of Boronic Esters

Boronic esters are not as reactive as triorganylboranes towards nucleophiles; however, rearrangement does take place if the boronic ester is treated with an α -halocarbanion (equation 54).⁸⁸ In this case, the 1-chloroethylboronate (36) can be obtained in 95% *de* due to the diastereofacial influence of the chiral pinanediol. Similarly, chloro- or bromo-alkylboronic esters react with Grignards reagents, alkyllithiums and enolates leading to rearranged products.^{9,88}



This methodology has been developed by Matteson and coworkers, 9,87 with particular emphasis on chiral boronic esters derived from pinanediol. Sequential chiral centers can be assembled by an iterative procedure with excellent enantiomeric excess at each step. The two chiral centers in eldanolide (37), for example, were assembled in this manner (Scheme 36).⁸⁸



i, BuⁱO₂CCH₂Li; ii, LiCHCl₂, cat. ZnCl₂; iii,Me₂C=CHCH₂MgCl; iv, H₂O₂; v, TFA

Scheme 36

 $L-\alpha$ -Amino acids can also be prepared by this method. Thus, heteronucleophiles, such as azide ion, react with the chloroalkylboronic ester *via* a four-coordinate boronate, as with carbon nucleophiles (Scheme 37).⁸⁹



i, LiCHCl₂, cat. ZnCl₂; ii, NaN₃; iii, NaClO₂; iv, H₂, Pd/C, 63% (5 steps)

Scheme 37

The C_2 symmetry present in the boronic ester (38) derived from (S,S)-1,2-dicyclohexylethanediol allows the stereoselective displacement of each chlorine atom to yield the chiral pentenylboronic ester

(39), a key intermediate which has been used in a synthesis of the macrolide mycinomycin (Scheme 38).⁹⁰



(S,S)-Diisopropylethanediol and (R,R)-2,3-butanediol are also effective chiral 'directors' in boronic ester rearrangements.⁹ Both of these vicinal diols are C_2 symmetric and the butanediolboronic esters are readily hydrolyzed to boronic acids, which can then be reesterified if a change in chiral direction is required.

Brown and coworkers⁷⁵ have described an alternative synthesis of chiral alkylboronic esters. In this synthesis prochiral alkenes are hydroborated with monoisopinocamphenylborane to yield isopinocamphenylalkylboranes which are then readily transformed to chiral alkylboronic esters (Scheme 39).⁹¹ Homologation with dichloromethyllithium, followed by reduction with potassium triisopropoxyborohydride (KIPBH) and oxidation, finally yields β -chiral alcohols (Scheme 40).⁹² These alcohols are not easily prepared by other methods. Aldehydes can be prepared by homologation from chiral alkylboronic esters with LiCH(OMe)SPh and oxidation (Scheme 41).⁹¹



i, MeCHO; ii, NaOH; iii, HOCH₂CH₂CH₂OH, 85% (3 steps)

Scheme 39



Chiral boronic esters react with organolithium reagents to form diorganylalkoxyboranes (borinic esters).⁹³ Subsequent reaction with the anion of dichloromethyl methyl ether then yields chiral ketones by rearrangement of both of the groups on boron (Scheme 42).⁹³ No racemization is observed in this sequence and alkyl-, aryl-⁹³ or alkynyl-lithium⁹⁴ reagents can be used.

The rearrangements of boronic esters frequently offer advantages over those of triorganylboranes. The presence of only one organic group for migration, for example, contrasts with triorganylboranes where selectivity is often a problem. In addition, the use of boronic esters derived from chiral diols is a power-ful technique in chiral synthesis. Finally, the use of alkyl, alkenyl and aryl groups in rearrangements of boronic esters has been demonstrated. All of these features have resulted in the wide application of boronic esters in synthesis.^{9,86,87,95}



3.4.3.3 Rearrangements of Tetraorganylboronates

Triorganylboranes have been shown to react with certain carbon nucleophiles to produce stable, isolable tetraorganylboronate salts. On exposure to electrophiles, these boronates then rearrange, producing neutral triorganylboranes.^{10b,95}

Cyanoboronates are prepared by the treatment of triorganylboranes with cyanide ion. When treated with an excess of TFAA, all three groups in the triorganylborane migrate to the cyanide carbon, leading to tertiary alcohols on oxidation (Scheme 43).⁹⁶ This transformation is closely related to the carbonylation of triorganylboranes (see Section 3.4.3.1). In the presence of 1 equiv. of TFAA, the third group does not migrate and ketones are obtained instead (equation 55).⁹⁷ Unsymmetrical ketones are available from thexyldialkylcyanoboronates since the bulky thexyl group does not migrate (equation 56);⁹⁷ cyclic ketones can also be prepared by this method (Scheme 44).



Scheme 44

Triorganylboranes react with alkynide ion to yield alkynylborates. When exposed to iodine and hydroxide, rearrangement to an alkylated alkyne then takes place (Scheme 45).⁹⁸ When alkynylborates are treated with alkylating agents such as dimethyl sulfate, rearrangement and concomitant alkylation takes place. The resulting alkenylborane can then be hydrolyzed to an (E)/(Z) mixture of alkenes or oxidized to ketones (Scheme 46).⁹⁹ This alkylation-rearrangement reaction has been used, for example, in a synthesis of propylure.¹⁰⁰



Scheme 46

Alkenylalkynyldi(1,2-dimethylpropyl)boronates, when exposed to iodine and hydroxide, rearrange with selective migration of the alkenyl group to the alkynyl group. The pheromone bombykol (**40**) has been synthesized *via* this key reaction (Scheme 47).¹⁰¹





Alkenylalkyldimethoxyboronates have been shown to rearrange with inversion of geometry, leading to substituted alkenes (equation 57),¹⁰² and alkenyltrialkylboronates react with epoxides to yield 1,4-diols after oxidation (Scheme 48).¹⁰³



Scheme 48

Tetraorganylboronates offer some advantages over triorganylboranes, particularly in the preparation of ketones from cyanoborates. Where a direct comparison has been made, the cyanoborate method was found to be superior.¹⁰⁴

3.4.4 REFERENCES

- 1. J. March, 'Advanced Organic Chemistry', 3rd edn., Wiley, New York, 1985, p. 963.
- 2. E. J. Corey and M. Chaykovsky, J. Am. Chem. Soc., 1965, 87, 1353.
- 3. J.-L. Laboureur and A. Krief, Tetrahedron Lett., 1984, 25, 2713.
- 4. D. J. Ager, Synthesis, 1984, 384.
- 5. A. Maercker, Org. React. (N.Y.), 1965, 14, 270
- 6. C. Giordano, G. Castaldi and F. Uggeri, Angew. Chem., Int. Ed. Engl., 1984, 23, 413.
- 7. G. C. Hirst, P. N. Howard and L. E. Overman, J. Am. Chem. Soc., 1989, 111, 1514.
- 8. H. C. Brown, Acc. Chem. Res., 1969, 2, 65.
- 9. D. S. Matteson, Acc. Chem. Res., 1988, 21, 294.
- 10. A. Pelter and K. Smith, in 'Comprehensive Organic Chemistry', ed. D. H. R. Barton and W. D. Ollis, Pergamon Press, Oxford, 1979, vol. 4, pp. 821, 883.
- 11. P. A. S. Smith and D. R. Baer, Org. React. (N.Y.), 1960, 11, 157.
- 12. F. F. Blicke, S. J. Azuara, N. J. Doorenbos and E. B. Hotelling, J. Am. Chem. Soc., 1952, 75, 5418.
- 13. D. A. Evans, G. L. Carroll and L. K. Truesdale, J. Org. Chem., 1974, 39, 914.
- 14. R. B. Woodward, J. Gosteli, I. Ernest, R. J. Friary, G. N. Nestler, H. Raman, R. Sitrin, C. Suter and J. K. Whitesell, J. Am. Chem. Soc., 1973, 95, 6853.
- 15
- B. Tchoubar, Bull. Soc. Chim. Fr., 1949, 164. J. D. Roberts and W. F. Gorham, J. Am. Chem. Soc., 1952, 74, 2278. 16.
- 17. W. Klyne, Nature (London), 1950, 166, 559.
- 18. L. Ruzicka and H. F. Meldahl, Helv. Chim. Acta, 1941, 24, 1321.
- 19. R. J. W. Cremlyn, D. L. Garmaise and C. W. Shoppee, J. Chem. Soc., 1953, 1847.
- 20. C. J. Collins, M. M. Staum and B. M. Benjamin, J. Org. Chem., 1962, 27, 3525.
- 21. J. N. Bradley, G. W. Cowell and A. Ledwith, J. Chem. Soc., 1964, 4334.
- 22. M. A. McKinney and P. P. Patel, J. Org. Chem., 1973, 38, 4059.
- 23. J. A. Marshall and J. J. Partridge, J. Org. Chem., 1968, 33, 4090.
- J. B. Jones and P. Price, *Tetrahedron*, 1973, 29, 1941.
 M. A. McKinney and P. P. Patel, J. Org. Chem., 1973, 38, 4064.
- 26. A. E. Greene and J. P. Deprés, J. Am. Chem. Soc., 1979, 101, 4003.
- 27. H. -J. Liu and S. P. Majumdar, Synth. Commun., 1975, 5, 125.
- 28. A. K. Ghosh, S. Biswas and R. V. Venkateswaran, J. Chem. Soc., Chem. Commun., 1988, 1421.
- A. E. Greene, M. J. Luche and A. A. Serra, J. Org. Chem., 1985, 50, 3957.
 V. Dave and E. W. Warnhoff, J. Org. Chem., 1983, 48, 2590.
- 31. K. Nagao, M. Chiba and S. W. Kim, Synthesis, 1983, 197.
- 32. B. M. Trost, D. E. Keeley and M. J. Bogdanowicz, J. Am. Chem. Soc., 1973, 95, 3068.
- 33. B. M. Trost and D. E. Keeley, J. Org. Chem., 1975, 40, 2013.
- 34. B. M. Trost and L. N. Jungheim, J. Am. Chem. Soc., 1980, 102, 7910.
- 35. J. H. Byers and T. A. Spencer, Tetrahedron Lett., 1985, 26, 717.
- 36. (a) T. Cohen, D. Kuhn and J. R. Falck, J. Am. Chem. Soc., 1975, 97, 4749; (b) S. Knapp, A. F. Trope, M. S. Theodore, N. Hirata and J. J. Barchi, J. Org. Chem., 1984, 49, 608.
- 37. B. M. Trost and G. K. Mikhail, J. Am. Chem. Soc., 1987, 109, 4124.
- 38. J. K. Landquist and A. Stanier, Tetrahedron Lett., 1975, 1611.
- 39. R. L. Sowerby and R. M. Coates, J. Am. Chem. Soc., 1972, 94, 4758.
- 40. C. R. Johnson, J. R. Shanklin and R. A. Kirchhoff, J. Am. Chem. Soc., 1973, 95, 6462.
- 41. L. A. Paquette, J. R. Peterson and R. J. Ross, J. Org. Chem., 1985, 50, 5200.
- C. Schmit, S. Sahraoui-Taleb, E. Differding, C. G. Dehasse-De Lombaert and L. Ghosez, Tetrahedron Lett., 42. 1984, 25, 5043.
- 43. A. Krief and J. L. Laboureur, J. Chem. Soc., Chem Commun., 1986, 702.
- 44. S. Halazy, F. Zutterman and A. Krief, Tetrahedron Lett., 1982, 23, 4385.
- 45. R. C. Gadwood, J. Org. Chem., 1983, 48, 2098.
- 46. H. J. Reich and F. Chow, J. Chem. Soc., Chem. Commun., 1975, 790.
- 47. H. Taguchi, H. Yamamoto and H. Nozaki, Bull. Chem. Soc. Jpn., 1977, 50, 1588.
- 48. H. Taguchi, H. Yamamoto and H. Nozaki, Bull. Chem. Soc. Jpn., 1977, 50, 1592.
- 49. J. Villieras, P. Perriot and J. F. Normant, Synthesis, 1979, 968.
- 50. J. Barluenga, J. L. Fernández-Simón, J. M. Concellón and M. Yus, J. Chem. Soc., Perkin Trans. 1, 1989, 691.
- 51. E. E. Smissman and G. Hite, J. Am. Chem. Soc., 1959, 81, 1201.
- 52. X. Creary, J. Org. Chem., 1976, 41, 3734.
- 53. K. E. Harding, J. B. Strickland and J. Pommerville, J. Org. Chem., 1988, 53, 4877.
- 54. C. Giordano, G. Castaldi, F. Uggeri and F. Gurzoni, Synthesis, 1985, 436.
- 55. G. Castaldi, C. Giordano and F. Uggeri, Synthesis, 1985, 505.
- 56. C. Giordano, G. Castaldi, F. Casagrande and A. Belli, J. Chem. Soc., Perkin Trans. 1, 1982, 2575.
- 57. G. Castaldi, S. Cavicchioli, C. Giordano and F. Uggeri, J. Org. Chem., 1987, 52, 3018, 5642.
- 58. C. L. Stevens, R. D. Elliott and B. L. Winch, J. Am. Chem. Soc., 1963, 85, 1464.
- 59. D. F. Morrow, M. E. Brokke, G. W. Moersch, M. E. Butler, C. F. Klein, W. A. Neuklis and E. C. Y. Huang, J. Org. Chem., 1965, 30, 212
- 60. C. L. Stevens, US Pat. 3 254 124 (1966) (Chem. Abstr., 1966, 65, 5414b).
- 61. R. M. Williams, E. Kwast, H. Coffman and T. Glinka, J. Am. Chem. Soc., 1989, 111, 3064.
- 62. D. N. Kirk and C. R. McHugh, J. Chem. Soc., Perkin Trans. 1, 1977, 893.
- 63. D. N. Kirk and A. Mudd, J. Chem. Soc. C, 1970, 2045.
- 64. Y. Mazur and M. Nussim, *Tetrahedron Lett.*, 1961, 817.
 65. M. Sworin and W. L. Neumann, J. Org. Chem., 1988, 53, 4894.
- 66. B. M. Trost and D. C. Lee, J. Am. Chem. Soc., 1988, 110, 6556.

- 67. M. H. Hopkins and L. E. Overman, J. Am. Chem. Soc., 1987, 109, 4748.
- 68. P. M. Herrinton, M. H. Hopkins, P. Mishra, M. J. Brown and L. E. Overman, J. Org. Chem., 1987, 52, 3711.
- 69. H. C. Brown and M. W. Rathke, J. Am. Chem. Soc., 1967, 89, 2737.
- 70. T. Yamauchi, K. Hattori, K. Nakao and K. Tamaki, J. Org. Chem., 1988, 53, 4858.
- 71. E. Negishi and H. C. Brown, Synthesis, 1972, 197.
- 72. M. W. Rathke and H. C. Brown, J. Am. Chem. Soc., 1967, 89, 2738.
- 73. H. C. Brown and E. Negishi, J. Am. Chem. Soc., 1967, 89, 5285.
- 74. H. C. Brown and E. Negishi, J. Am. Chem. Soc., 1967, 89, 5478.
- 75. H. C. Brown and B. Singaram, Acc. Chem. Res., 1988, 21, 287.
- 76. H. C. Brown, J.-J. Katz and B. A. Carlson, J. Org. Chem., 1973, 38, 3968.
- 77. H. C. Brown, H. Nambu and M. M. Rogic, J. Am. Chem. Soc., 1969, 91, 6852.
- 78. H. Nambu and H. C. Brown, J. Am. Chem. Soc., 1970, 92, 5790.
- 79. W. E. Truce, L. A. Mura, P. J. Smith and F. Young, J. Org. Chem., 1974, 39, 1449.
- 80. J. Hooz and S. Linke, J. Am. Chem. Soc., 1968, 90, 5936.
- 81. E. Negishi, T. Yoshida, A. Silveira, Jr. and B. L. Chiou, J. Org. Chem., 1975, 40, 814.
- 82. E. Negishi, J.-J. Katz and H. C. Brown, Synthesis, 1972, 555
- 83. H. C. Brown and D. Basavaiah, J. Org. Chem., 1982, 47, 3806.
- 84. H. C. Brown, D. Basavaiah and S. U. Kulkarni, J. Org. Chem., 1982, 47, 3808.
- 85. E. Negishi and T. Yoshida, J. Chem. Soc., Chem. Commun., 1973, 606.
- 86. D. S. Matteson, Synthesis, 1986, 973.
- 87. D. S. Matteson, Acc. Chem. Res., 1970, 3, 186.
- 88. D. S. Matteson, K. M. Sadhu and M. L. Peterson, J. Am. Chem. Soc., 1986, 108, 810.
- 89. D. S. Matteson and E. C. Beedle, Tetrahedron Lett., 1987, 28, 4499.
- 90. K. Ditrich, T. Bube, R. Stürmer and R. W. Hoffmann, Angew Chem., Int. Ed. Engl., 1986, 25, 1028.
- 91. H. C. Brown, T. Imai, M. C. Desai and B. Singaram, J. Am. Chem. Soc., 1985, 107, 4980.
- 92. H. C. Brown, R. G. Naik, R. K. Bakshi, C. Pyun and B. Singaram, J. Org. Chem., 1985, 50, 5586.
- 93. H. C. Brown, M. Srebnik, R. K. Bakshi and T. E. Cole, J. Am. Chem. Soc., 1987, 109, 5420.

- 94. H. C. Brown, A. K. Gupta, J. V. N. V. Prasad and M. Srebnik, J. Org. Chem., 1988, 53, 1391.
 95. A. Pelter, K. Smith and H. C. Brown, 'Borane Reagents', Academic Press, London, 1988.
 96. A. Pelter, M. G. Hutchings, K. Rowe and K. Smith, J. Chem. Soc., Perkin Trans. 1, 1975, 138.
- 97. A. Pelter, K. Smith, M. G. Hutchings and K. Rowe, J. Chem. Soc., Perkin Trans. 1, 1975, 129.
- 98. M. M. Midland, J. A. Sinclair and H. C. Brown, J. Org. Chem., 1974, 39, 731.
- 99. A. Pelter, T. W. Bentley, C. R. Harrison, C. Subrahmanyam and R. J. Laub, J. Chem. Soc., Perkin Trans. 1, 1976, 2419.
- 100. K. Utimoto, M. Kitai, M. Naruse and H. Nozaki, Tetrahedron Lett., 1975, 4233.
- 101. E. Negishi, G. Lew and T. Yoshida, J. Chem. Soc., Chem. Commun., 1973, 874.
- 102. D. A. Evans, T. C. Crawford, R. C. Thomas and J. A. Walker, J. Org. Chem., 1976, 41, 3947.
- 103. K. Utimoto, K. Uchida and H. Nozaki, Tetrahedron Lett., 1973, 4527.
- 104. R. Murphy and R. H. Prager, Aust. J. Chem., 1976, 29, 617.
- 105. W. L. Mock and M. E. Hartman, J. Am. Chem. Soc., 1977, 42, 459.

3.5 Dienone-Phenol Rearrangements and Related Reactions

DONALD A. WHITING

University of Nottingham, UK

3.5.1 INTRODUCTION	803
3.5.2 INTRACYCLIC MIGRATION	804
 3.5.2.1 Formation of New C—C Bonds 3.5.2.1.1 Rearrangements forming one new C—C bond 3.5.2.1.2 Rearrangements forming more than one new C—C bond 3.5.2.2 Formation of a New C—O Bond 3.5.2.3 Formation of New C—C and C—O Bonds 	804 804 810 812 813
3.5.3 EXTRACYCLIC BOND MIGRATION	813
3.5.4 PHOTOCHEMICAL AROMATIZATION	815
3.5.5 MISCELLANEOUS AROMATIZATION	816
3.5.5.1 Dienone Aromatization with Bond Cleavage 3.5.5.2 Rearrangements in Dienone Reactions with Nucleophiles 3.5.5.3 Tropolone–Phenol Rearrangements	816 817 818
3.5.6 REFERENCES	819

3.5.1 INTRODUCTION

Dienone-phenol rearrangements involve cyclohexadienones in the great majority of cases, although other variations are possible, *e.g.* ring contractions of tropolones to phenols. Cyclohexadienones can be either cross conjugated as in the 2,5-dienones (1), or linearly conjugated as in the 2,4-dienones (2). These forms are often referred to as the '*para*' and '*ortho*' dienones respectively, and they are available synthetically from a number of reactions, including oxidative coupling of phenols, electrophilic substitution of phenols, oxidation of phenols, other aromatic rearrangements and dehydrogenation of cyclohexanones.

The cross-conjugated structures are generally more stable than their linearly conjugated isomers, but both types rearrange readily to a phenol (3) with migration of a substituent X or Y from C-4 (1) or C-6 (2). Both substituents X and Y may shift. If the migrating group moves to a site on the ring which aromatizes, as in (3), the rearrangement is termed 'intracyclic', but if a group migrates to a position outside the ring, as in (4), then the reorganization is referred to as 'extracyclic'.

Dienone-phenol rearrangements are mechanistically diverse. They may involve: 1,2-shifts of the Wagner-Meerwein type, or of the benzil-benzilic acid kind; 1,3-shifts by a Claisen-Cope mechanism; 1,5sigmatropic shifts; Favorskii-like reactions; and other types. They may also be induced photochemically. A number of reviews are available,¹ which discuss mechanistic aspects in detail. In this chapter emphasis is put on preparative aspects of these reactions and the examples are organized on a structural basis, stressing the new bond(s) formed.



3.5.2 INTRACYCLIC MIGRATION

3.5.2.1 Formation of New C-C Bonds

The major ramifications of carbon migration in the dienone-phenol rearrangement are shown in Scheme 1, for a steroid ring A dienone (5). Several hundred of such acid-catalyzed reactions have been described² and five different pathways can be discerned. Protonation at oxygen in the dienone (5) leads to cation (6), and three different carbon shifts may follow. C-9 migration to C-5 leads to a new cation (7) which can reach an aromatic system either by migration of C-6, leading to phenol (8) or by shift of C-9, giving rise to phenol (9). Going back to cation (6), two courses are possible for a methyl (C-19) shift; 1,2-migration to C-1 yields phenol (10), while shift to C-5 affords another cation (11). Intermediate (11) can undergo either a C-6 \rightarrow C-10, C-6 \rightarrow C-1 sequence to arrive at phenol (12), or a C-6 \rightarrow C-10, C-9 \rightarrow C-1 routine to yield phenol (13). Of the five phenol products, (10) and (12) are structurally identical as shown, but represent different bond reorganizations. All these pathways have been observed. The choice between, for example, the migration of methyl or methine from C-10 is finely balanced and is sensitive to reagents and conditions.³ The migration of the potentially most stable carbocation is expected to be kinetically preferred, *i.e.* the pathway which leads to (9). However, the shifts are reversible, and other factors may intervene, so that although product (9) may predominate, phenols of structures (8)and (10) are often the major products. The reader is referred to mechanistic reviews for a discussion of the controlling features. The examples collected in this chapter are organized on a structural basis, and it should be pointed out that a mechanism cannot always be reliably deduced from product structure.

3.5.2.1.1 Rearrangements forming one new C-C bond

Alkyl shifts in simple cyclohexadienones can be induced with strong acids at room temperature. Thus 4,4-dimethylcyclohexadienone (14) rearranges to 3,4-dimethylphenol (15) at 25 °C with 70% perchloric acid, in 90% yield.⁴ The 4-methyl-4-ethyl homolog (16) smoothly rearranges in trifluoroacetic acid, with preferential ethyl shift, to 3-ethyl-4-methylphenol (17).^{5,6} Preferred ethyl shift is also seen in the o-dienone (18), which rearranges to the 3-ethyl-2-methylphenol (19) with loss of a t-butyl group.⁷ In the 4hydroxy-4-methyldienone (20), methyl group shift dominates when boron trifluoride etherate is used as acid (but see Section 3.5.3) to yield methyl hydroquinone (21; 55%).⁸ Vinyl groups⁹ and ethoxycarbonyl groups⁵ migrate in preference to methyl. Electron-withdrawing groups adjacent to reacting centers slow down the reaction so that the dichloromethyl-substituted dienone (22) requires long treatment at 50 °C with concentrated sulfuric acid for conversion (84%) to 4-hydroxy-2-methylbenzaldehyde (23), with methyl migration, and concomitant hydrolysis of the gem-dichloride.¹⁰ In the decalindienone group the rearrangement of santonin (24) in acid to desmotroposantonin (25) is an example of historical interest.¹¹ The parent ring system (26) has been investigated systematically.^{3,12} With (26; R = Me) fairly good yields (50-60%) of the methyltetralol (27) were obtained using aqueous hydrochloric or sulfuric acid as catalyst; a small proportion (10%) of the isomeric tetralol (28; R = Me) was also obtained. Apparent 1.2shift of the methyl is the major pathway. With other angular alkyls, e.g. (26; $R = Pr^n$), in which both potential migrating centers are methylene, the product of 1,2-shift dominates even more *i.e.* (27; R = Pr^{n} :(28; R = Pr^{n}) = 98:2, on rearrangement with 10 M sulfuric acid.

In the steroid series the androstadiendione (29) reacts slowly with 48% hydrobromic acid at room temperature; after 5 d the main product (55%) is 1-methylestrone (30) with its isomer (31) in minor amount (11%).³ The ring bond shifts are inhibited in the 11-ketodienone (32) which essentially reacts exclusively by 1,2-methyl shift in perchloric acid-acetic anhydride to provide the 2-methyl steroid (33), with parallel peracetylation.¹³ An 11-hydroxy function has a different effect (see Section 3.5.2.1.2). The 1,2-methyl



Scheme 1

shift becomes the major pathway when C-4 is blocked, and when C-6 is an sp^2 -hybrid. Thus the 4methylandrostadienone (**34**) yields the phenol (**35**; 73%) after saponification with acetic anhydride-*p*-toluenesulfonic acid,¹⁴ and the same reagents effect the parallel conversion of the Δ^6 -dienone (**36**) to the phenol acetate (**37**).¹⁵ OH





OH



(26)



(27)

0

Bu



(28)

OH

(19)

О

Et

Bul

Et



(29)







Not all dienone-phenol rearrangements are acid catalyzed. The 4-hydroxy-4-methyl-2,5-cyclohexadienones (**38**; R = H) and (**38**; R = Me) react in refluxing aqueous alkali to afford the methylhydroquinones (**39**; R = H) and (**39**; R = Me; 95%), respectively.¹⁶ This is understandable as migration in an electronrich, vinylogous α -ketol system (**40**). A different mechanism operates in the rearrangement of the decaladienone (**41**), induced by sodium hydroxide-aqueous methanol, to the tetralindiol (**42**; 76%) which appears to proceed by a retro-aldol-aldol sequence, by way of the aldehyde (**43**).¹⁷

റ

(14)

OH

(20)

The acid-catalyzed reactions so far listed in this section all involve shift of an alkyl residue. In bicyclic spirodienones, a related ring bond shift leads to ring expansion, and rearrangement to a fused bicycle. Thus the spirooxetane (44) is converted quantitatively into the hydroxydihydrobenzofuran (45) on treatment with ethanolic hydrogen chloride (preferred carbon migration),¹⁸ and expansion of a four- to a five-membered ring is also seen in the spontaneous benzylic migration of the lactam (46), yielding (47; 62%).¹⁹ Five- to six-membered ring size increase is observed in the conversion of the dienone (48) to the octahydrophenanthrol (49; 95%) with acetic anhydride–sulfuric acid,²⁰ and in the transformation of the dienone (50) to the tetralone derivative (51), with migration of the dienone oxime (52) with acetic anhydride–sulfuric acid, which yields the phenolic amine diacetate (53; 75%). This sequence probably involves addition (of acetate) then elimination to afford a spirodienimine (54) which reacts analogously to a dienone.²² Interestingly the spirodienone (55) in aqueous citric acid reacts by a different mode of ring expansion, yielding the tropone (56; 80%), presumably through a tricyclic intermediary (57).²³



1,2-Aryl shifts are preferred, as expected, to competing alkyl shifts. Thus the 2,4-dienone (**58**) aromatizes quantitatively by phenyl migration, to trimethylphenylphenol (**59**),²⁴ and the 2,5-dienone (**60**) gives the biaryl (**61**; 51%),²⁵ in both cases with acetic anhydride–sulfuric acid. Suitable spirodienones react similarly, *e.g.* (**62**) yields the triphenylene derivative (**63**; 83%),²⁶ and the quinamide (**64**) rearranges sluggishly to the biaryl (**65**), with slow aryl migration in the protonated system.²⁷ Aryl shifts feature in the best-known applications of the dienone–phenol rearrangement for natural product synthesis, in the alkaloid area. Thus synthetic (\pm)-orientalinone (**66**) was converted into (\pm)-corydine (**67**) with dry methanolic hydrogen chloride (through preliminary hemiketalization) and into (\pm)-isocorytuberine (**68**) with an acetic acid–hydrochloric acid mixture.^{28–30} Synthetic spirodienone (**69**) gave (\pm)-multifloramine (**70**; 45%) with concentrated sulfuric acid,³¹ and its relative (**71**) rearranged with boron trifluoride to the homoaporphine (**72**) in 93% yield.³² The dienone (**73**) also reacts by 1,2-aryl shift to give homoaporphine (**74**),³³ and the demethyl homolog (**75**) is similarly transformed to (**76**; 75%),³⁴ using boron trifluoride as catalyst in both cases. The homoaporphine series can also be reached from another variation on the spirodienone structure in this series (**77**), which affords (**78**) by boron trifluoride induced ring bond and aryl shifts.³³ Finally, it is worth noting the concurrent fragmentation–aryl migration involved in the transformation of (**79**) to (**80**) in a colchicine synthesis.³⁵

















(51)























(65)



(58)





The last topic in this section concerns allyl shifts. This is a mechanistically complex area, in which migrations may proceed by one or more 1,2-shifts to cationic centers, or by 1,3- or 1,5-sigmatropic shifts, thermal or acid catalyzed. It is interesting that carbon-carbon bond formation is generally preferred to carbon-oxygen bond formation, even where the latter would be favored by aromaticity,^{1b} as in the thermal reactions of dienones (**81**) and (**82**) which lead at *ca*. 100 °C to the new dienones (**83**) and (**84**), but not detectably to the corresponding *O*-allylphenols, *e.g.* (**85**). The 2,5-dienone (**86**) reacts rapidly in acid to give a mixture of phenols (**87:88** = 2:1),³⁶ the first by an acid-catalyzed Claisen-Cope reaction, and the second *via* a Wagner-Meerwein shift. The 1,3-shift is blocked in the 2,5-dienone (**83**), and 1,2-migration is induced on an acidic clay surface yielding (**89**; 98%).³⁷ Rearrangement of the 2,5-dienone (**90**) proceeds by two sequential 1,2-shifts, rather than a 1,3, yielding (**91**).³⁸ 1,3-Shifts are observed in the aromatization of the 2,4-dienone (**92**) in trifluoroacetic acid, yielding (**93**; 63%),³⁹ with its homolog (**94**) in sulfuric acid-acetic acid, giving (**95**; 83%),³⁶ and with the androstadienone (**96**),⁴⁰ which yields both phenols (**97:98** = 3:2), essentially quantitatively and stereospecifically.

In synthetic work towards tetracyclines the tricyclic dienone (99) was rearranged at 0 °C using boron trifluoride, to the anthraquinone (100; 65%) apparently by a direct 1,5-shift.⁴¹ The quinamine (101) undergoes facile reorganization in hydrochloric acid to the biaryl (102)⁴² in an aza-Claisen-like process.

Lastly, aromatization of the dienone (103), in sodium acetate-acetic anhydride at 140 °C, proceeds by a Favorskii-style mechanism, to afford (104).⁴³



3.5.2.1.2 Rearrangements forming more than one new C-C bond

Aromatization through sequential ring bond shifts is exhibited in the acid-catalyzed conversion of the decaladienone (105; R = Me) to the methyltetralol (106; R = Me; 57%)⁴⁴ at room temperature with acetic anhydride-sulfuric acid; lactone (107) very similarly rearranges to (108; 69%).⁴⁵ Decaladienones with an angular hydroxy, *e.g.* (105; R = OH), react in parallel fashion, to give, for example, (106; R = OH),^{46,47} although their acetyl derivatives undergo oxygen migration, *vide infra*. In the steroid series the androstadienedione (109) rearranges in acetic anhydride-zinc chloride, with two shifts of C-9 (methine), to the 1-hydroxyandrosta-1,3,5-trienone (110) in 92% yield.⁴⁸ A number of similar reactions are cataloged in valuable reviews.² 10-Hydroxy steroids can also rearrange in this way, *e.g.* (111) to (112).⁴⁹ Although a 6,7 double bond usually impresses 1,2-methyl shifts, the androstatrienone (113) was converted by superacid into the ring bond rearranged product (114; 75%).⁵⁰ In the tetracycle (115) shifts of a methylene from the quaternary center, and then of a benzylic carbon, lead to the phenol acetate (116)⁵¹ in 88% conversion.















(101)









(105)







(111)

0





(113)







но



3.5.2.2 Formation of a New C-O Bond

The migration of oxygen from a quaternary center in a cyclohexadienone may be preferred to a carbon shift, when present as an ether or ester function rather than free hydroxy. Thus the *p*-quinol acetate (117) yields the orcinol monoacetate (118; 79%) on treatment at room temperature with trifluoroacetic anhydride,⁵² and the *p*-quinol ether (119) forms the resorcinol diethyl ether (120; 71%) in ethanolic sulfuric acid. In the second case, hemiketalization must intervene; also some methyl shift (12%) is observed.⁵³ With the quinol (121), treatment with acetic anhydride–sulfuric acid leads to the lactone (122); acetylation or lactonization probably precedes oxygen shift.⁵⁴ A number of related examples can be found in the steroid area.^{2,55} Thermal 1,3-shifts of *p*-quinol acetates can also be induced; acetate (117) yields catechol acetate series, 1,2-acetoxy shift is seen in (125) \rightarrow (126; 92%)⁵⁷ and in (127) \rightarrow (128; 90%),⁵⁸ both in acetic anhydride with acid catalyst. In the 2,4-dienone (129), two 1,2-acetoxy shifts are required for aromatization, leading to the hydroquinone diacetate (130).⁵⁴



A number of o-quinol diacetates have been prepared and they rearrange smoothly to triacetoxybenzenes;⁵⁹ $o \rightarrow o'$ migration is preferred unless the latter position is blocked or sterically crowded, when acetoxy shift is directed to another aryl site. The range of possibilities is indicated by the conversions of the o-quinol diacetates (131)-(134) to the 1,2,3- or 1,2,4-triacetoxybenzenes (135)-(138), in 56-90% yields.

OAc

OAc

O

OAc



The reactions in this section represent useful ways of introducing oxygen into aromatic rings, for which methodology is limited, and of reaching certain substitution patterns.

Oxygen migration may be observed in spirodienones. Thus the lactone (139) affords dihydrocoumarins (140) and (141) in 90% and 10% yields, respectively, in acetic acid-sulfuric acid; change of reagent to dilute sulfuric acid changes the yields to 5% and 95%, respectively, nicely illustrating the fine balance between pathways in these reactions.⁶⁰ The spirodienone (142) undergoes O-shift in acid, but C-shift in alkali, yielding either (143) or (144).⁶¹



Finally a rare example of C—O bond formation by migration of carbon to oxygen is found in the pyrolysis of the complex spirodienone (145) at 205 °C, which affords the depside (146; 91%).⁶²



3.5.2.3 Formation of New C-C and C-O Bonds

In a number of cyclohexadienone aromatizations both new C—C and C—O bonds are formed, in multistep reactions which may still be high yielding. Thus the tetralenone (147) gives the trimethylnaphthol acetate (148; 95%) in acetic acid.⁶³ Both of the spiroketones (149) and (150) reorganize under acetylating conditions to the tetrahydrophenanthrol acetate (151).⁶⁴ The allylquinol acetates (152; R = H) and (152; R = Me) reorganize to the monacetylallylcatechols (153) on heating at 180–190 °C, through an initial 1,5-acetyl shift to (154), followed by 1,3-allyl shift and acetyl transfer between oxygens.⁶⁵

3.5.3 EXTRACYCLIC BOND MIGRATION

The conventional dienone-phenol rearrangements of o-quinol acetates are described above. An alternative mode of restructuring to an aromatic system has been observed, exemplified by the quantitative conversion of the 2,4-dienone (155) on heating at 110 °C in acetic acid to the benzylic acetate (156).⁶⁶ The related 2,4-dienone (157) rearranges at 70-80 °C in dimethyl sulfoxide-sodium bicarbonate to the



m-hydroxybenzaldehyde (158), in 45% yield.⁶⁷ The mechanisms by which the benzylic positions are oxidized are not known. Interestingly, pyrolysis of (155) at 450 °C gave the diarylethane (159; 37%), with some of the corresponding stilbene,⁶⁸ suggesting formation of benzylic radicals. Similar halogen migrations have been recorded. Thus the *p*-bromodienone (160) quantitatively converts to the benzylic bromide (161) on standing at 0 °C, and the di-t-butyl analog (162) similarly rearranges on brief warming, to give (163; 70%).⁶⁹ Related reactions occur with polyhalomethyl groups. Thus the 2,5-cyclohexadienone (164) yields the benzoate (165) on heating with benzoyl chloride to 140 °C.⁷⁰ If phosphorus pentachloride is used then the product is the trichloroethylaryl chloride (166; 89%), but in polyphosphoric acid the acid (167) is the major product (56%).⁷¹ Again mechanisms remain obscure.



The last example in this section is the facile and quantitative rearrangement of the quinamine (168) in 10^{-5} M hydrochloric acid, to the aminoaryl ether (169), with extracyclic shift of aryl to oxygen.⁷² A number of examples are known and the mechanism appears similar to the benzidine rearrangement.



3.5.4 PHOTOCHEMICAL AROMATIZATION

The photochemistry of cyclohexadienones is a complex topic, which has been reviewed⁷³ and is discussed in detail elsewhere in this series. It is only appropriate here to point out that aromatization to a phenol is a possible outcome of such reactions, and indicate a very few synthetically useful examples.

The steroidal dienone (170) can be induced to undergo the relatively unusual reorganization to the 2hydroxy-4-methylandrosta-1,3,5-triene (171), by irradiation in refluxing acetic acid;⁷⁴ although only 48% of the acetate (171) was isolated, the related 17-ketone (36%) was also formed. The *p*-dienones (172) and (173) are restructured on irradiation in cyclohexane with net relocation of methyl and *t*-butyl groups to afford the phenols (174) and (175) in 55% and 73% yield, respectively.^{75,76} A photo-induced aryl shift in the benzylisoquinoline dienone (176) was employed to prepare the aporphine (177; 44%) *en route* to (\pm)-boldine.⁷⁷ The *p*-dienones (178; X = Cl, X = N₃) aromatize on irradiation to yield the alkylated p-hydroxybenzoates (179; *e.g.* X = N₃, 57%) as well as minor quantities of their isomers (180).⁷⁸ Mechanistic aspects of these reactions are discussed in the literature cited.



3.5.5 MISCELLANEOUS AROMATIZATION

3.5.5.1 Dienone Aromatization with Bond Cleavage

A group of predominantly steroid reactions involve a ring A or ring B dienone which aromatizes under reductive conditions with cleavage of a carbon fragment. In most cases⁷⁹ the angular A/B methyl is removed, as in the case of the androstatrienedione (181) which suffers reductive demethylation to the phenol (182) on brief refluxing with zinc in aqueous pyridine,^{79a} in 93% yield. Less commonly, fragmentation of the C(9)—C(10) ring bond is engineered, as in the reductive elimination of the 11-hydroxy derivative (183) with samarium diiodide to the B-secophenol (184; 38%).⁸⁰ Fragmentations have also been ingeniously employed in alkaloid syntheses. Thus the spirodienone (185) aromatized during alkaline deacylation; *in situ* reduction of the resulting imine gave the o,o'-bridged biaryl (186; 80%), a key intermediate in both *in vivo* and *in vitro* formation of erysodienone.⁸¹ The ketal (187), formed conventionally from the corresponding dienone, fragments (increased N-basicity) during reduction of the formamide function with lithium aluminum hydride to the N-methyl compound, to yield O-methylerybidine (188; 81%).⁸² In a synthesis of L-tyrosine, the precursor (189), from L-glutamic acid was aromatized in dilute alkali, with decarboxylation, to form N-BOC-Tyr (190; 98%).⁸³ Outside the natural product area the spirodienone (193) rearranged with C—O cleavage to yield naphthaldehyde (194; 92%).⁸⁵





(182)



(183)

(181)



HO

OMe

OH

(188)

MeO

MeO

MeO

CHO

0



(184)

OMe

OMe

(187)

MeO

MeO

MeO

HO

(185)

N-Me

COCF



(189)

0



3.5.5.2 Rearrangements in Dienone Reactions with Nucleophiles

Cyclohexadienones containing a leaving group, such as o- or p-quinol acetates, undergo various reactions with nucleophiles, most often involving an addition-elimination sequence to form a phenolic product with nucleophile as a substituent. Thus the o-quinol acetate (195) reacts with secondary amines, cvanide ion, or diethyl sodiomalonate to give the phenols (196; $X = NR_2$),⁸⁶ (196; X = CN),⁸⁷ and (196; $X = CH (CO_2Et)_2)$;⁸⁸ Wittig reagents behave similarly.⁸⁹ The site of attack may be modified by other functions, e.g. azide ion adds to C-2 in aldehyde (197) yielding azidophenol (198) as product.⁹⁰ Thiols behave in a different fashion from other nucleophiles, for example thiophenol substitutes the dienone acetate (199) at C-4 rather than C-3, to afford (200) as the major product;⁹¹ various mechanisms are possible, including S_N2' substitution-enolization. More complex sequences occur when the dienone carbaldehydes (201) and (197) are treated with cyanide. In the first case, methanolic cyanide affords the cyano ester (202), but cyanide in dimethylformamide gives the aldehyde (203), in good yield in both cases.⁹² The isomer (197) reacts with cyanide in methanol to provide the ester (204; 85%) without incorporating cyanide.⁹² For mechanistic discussions see ref. 1b. Related transformations occur with p-dienones, e.g. (205) yields cyanophenol (206), on addition-elimination with cyanide ion.⁸⁷ Different patterns of reactivity emerge with different nucleophiles; thus (205) reacts with phenyl- or methyl-magnesium bromide to afford phenols (207; R = Ph and R = Me) respectively. This is a result of 1,2-addition



Rearrangement Reactions

followed by loss of acetate and 1,2-shift of phenyl (or methyl), whereas (205) and phenyllithium yield (208) via 1,2-addition, deacylation, loss of benzylic oxygen function and 1,2-methyl shift.⁹³ Another interesting conversion occurs when the spirodienone (209) is treated with diacetyl sulfide and boron trifluoride, giving rise to the thiolactone (210), in a useful C—S bond-forming process.⁹⁴ Finally in this section we remark on the curious transformation of the dichloromethyldienone (211) into the acid (212) by hydroxide ion in ethanol;⁹⁵ an intermediate (213) is implicated. The conversion (214) \rightarrow (215) also requires a bridged intermediate.⁹⁶

3.5.5.3 Tropolone–Phenol Rearrangements

A number of cases are known of ring contraction of tropolones to phenols, with extrusion of the carbonyl group as carboxylate or carbon dioxide. Examples are the conversions of the benzotropolone (216) to the naphthol ester (217; 73%) with methanolic potash;⁹⁶ the pyridinotropolone (218) to the quinoline (219; 50%) in dilute alkali;⁹⁷ the aminotropolone (220) to the methylsalicyclic acid (221; 82%) on diazotization;⁹⁸ and the tropolonecarboxylate (222) to the dicarboxylic acid (223; 78%) on brief alkali fusion.⁹⁹



3.5.6 REFERENCES

- 1. (a) M. J. Perkins and P. Ward, in 'Mechanisms of Molecular Migrations', ed. B. S. Thyagarajan, Wiley, New York, 1971, vol. 4, p. 55; (b) B. Miller, in 'Mechanisms of Molecular Migrations', ed. B. S. Thyagarajan, Wiley, New York, 1968, vol. 1, p. 247; (c) H. J. Shine, in 'Aromatic Rearrangements', Elsevier, New York, 1967, p. 55. (d) A. J. Waring, Adv. Alicyclic Chem., 1966, 1, 207. 2. (a) R. H. Shapiro, in 'Steroid Reactions', ed. C. Djerassi, Holden-Day, San Francisco, 1963, chap. 9; (b) K.-F.
- Wedemeyer, Method. Org. Chem. (Houben-Weyl), 1976, part 2, 735.
- A. S. Dreiding, W. J. Pummer and A. J. Tomasewski, J. Am. Chem. Soc., 1953, 75, 3159.
 (a) V. P. Vitullo, J. Org. Chem., 1970, 35, 3976; (b) V. P. Vitullo and N. Grossman, J. Am. Chem. Soc., 1972, 94. 3844.
- 5. J. N. Marx, J. C. Argyle and L. R. Norman, J. Am. Chem. Soc., 1974, 96, 2121.
- 6. J. W. Pilkington and A. J. Waring, Tetrahedron Lett., 1973, 4345.
- B. Miller and H. Margulies, J. Am. Chem. Soc., 1967, 89, 1678.
 S. Goodwin and B. Witkop, J. Am. Chem. Soc., 1957, 79, 179.
- 9. J. N. Marx and Y. S. Paik Hahn, J. Org. Chem., 1988, 53, 2866.
- 10. (a) T. G. Miller, J. Org. Chem., 1962, 27, 1549; (b) H. Hemetsberger, Monatsh. Chem., 1968, 99, 1724.
- A. Andreocci and P. Bertolo, Chem. Ber., 1898, 31, 3131.
 H. J. Shine and C. E. Schoening, J. Org. Chem., 1972, 37, 2899.
- 13. E. J. Bailey, J. Elks, J. F. Oughton and L. Stephenson, J. Chem. Soc., 1961, 4535.
- 14. F. Sondheimer and Y. Mazur, J. Am. Chem. Soc., 1957, 79, 2906.
- 15. C. Djerassi, G. Rosenkranz, J. Romo, J. Pataki and S. Kaufmann, J. Am. Chem. Soc., 1950, 72, 4540.
- 16. (a) E. Bamberger and A. Rising, Chem. Ber., 1900, 33, 3636; (b) E. Bamberger, Chem. Ber., 1900, 33, 3600.
- 17. K. H. Bell, Tetrahedron Lett., 1968, 3979; Aust. J. Chem., 1972, 25, 1117.
- 18. D. Bryce-Smith, A. Gilbert and M. G. Johnson, J. Chem. Soc. C, 1967, 383.
- 19. C. W. Bird, J. Chem. Soc., 1965, 3016.
- 20. R. T. Arnold, J. S. Buckley, Jr. and R. M. Dodson, J. Am. Chem. Soc., 1950, 72, 3153.
- S. Dorling and J. Harley-Mason, Chem. Ind. (London), 1959, 1551 21.
- 22. M. Dvolaitzky and A. S. Dreiding, Helv. Chim. Acta, 1965, 48, 1988.
- 23. A. S. Kende and K. Koch, Tetrahedron Lett., 1986, 27, 6051.
- 24. E. Schinzel and F. Wessely, Monatsh. Chem., 1955, 86, 912.
- 25. V. Arkley, F. M. Dean, A. Robertson and P. Sidisunthorn, J. Chem. Soc., 1956, 2322.
- 26. H. Plieninger, G. Ege and M. I. Ullah, Chem. Ber., 1963, 96, 1610.
- 27. D. H. Hey, J. A. Leonard, T. M. Moynehan and C. W. Rees, J. Chem. Soc., 1961, 232.
- 28. A. H. Jackson and J. A. Martin, J. Chem. Soc., Chem. Commun., 1965, 142, 420.
- 29. M. Shamma and W. A. Slusarchyk, J. Chem. Soc., Chem. Commun., 1965, 528.
- 30. A. H. Jackson and J. A. Martin, J. Chem. Soc. C, 1966, 2222
- 31. A. R. Battersby, R. B. Bradbury, R. B. Herbert, M. H. G. Munro and R. Ramage, J. Chem. Soc., Chem. Commun., 1967, 450.
- S. M. Kupchan, O. P. Dhingra and C.-K. Kim, J. Org. Chem., 1978, 43, 4076.
 S. M. Kupchan, O. P. Dhingra, C.-K. Kim and V. Kameswaran, J. Org. Chem., 1978, 43, 2521.
- 34. J. P. Marino and J. M. Samanen, Tetrahedron Lett., 1973, 4553.
- 35. E. Kotani, F. Miyazaki and S. Tobinaga, J. Chem. Soc., Chem. Commun., 1974, 300.
- B. Miller, J. Am. Chem. Soc., 1965, 87, 5115.
 S. Chalais, P. Laszlo and A. Mathy, Tetrahedron Lett., 1986, 27, 2627.
- 38. B. Miller and H. Margulies, Tetrahedron Lett., 1965, 1727.
- 39. U. Widmer, J. Zsindely, H.-J. Hansen and H. Schmid, Helv. Chim. Acta, 1973, 56, 75.
- 40. A. Wunderli, J. Zsindely, H.-J. Hansen and H. Schmid, Helv. Chim. Acta, 1973, 56, 989.
- 41. K. Hayakawa, M. Aso and K. Kanematsu, J. Chem. Soc., Chem. Commun., 1987, 102.
- 42. K. Fries, R. Böker and F. Wallbaum, Justus Liebigs Ann. Chem., 1934, 509, 73.
- 43. M. U. Akpuaka, R. L. Beddoes, J. M. Bruce, S. Fitzjohn and O. S. Mills, J. Chem. Soc., Chem. Commun., 1982, 686.
- 44. R. B. Woodward and T. Singh, J. Am. Chem. Soc., 1950, 72, 494.
- 45. M. Hirakura and M. Yanagita, J. Org. Chem., 1962, 27, 2948.
- 46. S. Goodwin and B. Witkop, J. Am. Chem. Soc., 1957, 79, 179.
- 47. Y. Asahina and T. Momose, Chem. Ber., 1938, 71, 1421.
- 48. A. S. Dreiding and A. Voltman, J. Am. Chem. Soc., 1954, 76, 537.
- 49. E. Hecker, Chem. Ber., 1964, 97, 1940.
- 50. J.-C. Jacquesy, R. Jacquesy and U. H. Ly, Tetrahedron Lett., 1974, 2199.
- 51. A. L. Wilds and C. Djerassi, J. Am. Chem. Soc., 1946, 68, 1715.
- 52. E. Hecker and E. Meyer, Chem. Ber., 1964, 97, 1926.
- 53. E. Bamberger, Chem. Ber., 1907, 40, 1956.
- 54. F. Wessely and W. Metlesics, Monatsh. Chem., 1954, 85, 637.
- 55. A. M. Gold and E. Schwenk, J. Am. Chem. Soc., 1958, 80, 5683.
- 56. E. Zbiral, F. Wessely and E. Lahrmann, Monatsh. Chem., 1960, 91, 331.
- 57. F. Takacs, Monatsh. Chem., 1964, 95, 961.
- 58. H. Budzikiewicz and W. Metlesics, J. Org. Chem., 1959, 24, 1125.
- 59. H. Budzikiewicz, W. Metlesics and F. Wessely, Monatsh. Chem., 1960, 91, 117.
- 60. A. I. Scott, P. A. Dodson, F. McCapra and M. B. Meyers, J. Am. Chem. Soc., 1963, 85, 3702.
- 61. D. H. Hey, J. A. Leonard and C. W. Rees, J. Chem. Soc., 1963, 5251.
- 62. J. Gunzinger and R. Tabacchi, Helv. Chim. Acta, 1985, 68, 1940.
- 63. E. N. Marvell and E. Magoon, J. Am. Chem. Soc., 1954, 76, 5118.

- 64. (a) H. Christol, R. Jacquier and M. Mousseron, Bull. Soc. Chim. Fr., 1959, 11; (b) E. N. Marvell and A. O. Geiszler, J. Am. Chem. Soc., 1952, 74, 1259.
- 65. E. Zbiral, F. Wessely and J. Joerg, Monatsh. Chem., 1961, 92, 654.
- 66. F. Wessely and E. Schinzel, Monatsh. Chem., 1953, 84, 425.
- J. Leitich and F. Wessely, Monatsh. Chem., 1964, 95, 129.
 E. Zbiral, O. Saiko and F. Wessely, Monatsh. Chem., 1964, 95, 512.
- 69. A. A. Volodkin and V. V. Ershov, Izv. Akad. Nauk SSSR, Ser. Khim., 1962, 1292, 1213 (Chem. Abstr., 1963, **58, 6783**).
- 70. M. S. Newman and F. Bayerlein, J. Org. Chem., 1963, 28, 2804.
- 71. M. S. Newman and L. L. Wood, J. Org. Chem., 1958, 23, 1236; J. Am. Chem. Soc., 1959, 81, 6450.
- 72. B. Miller, J. Am. Chem. Soc., 1964, 86, 1135.
- 73. (a) O. L. Chapman, Adv. Photochem., 1963, 1, 330; (b) K. Schaffner, Adv. Photochem., 1966, 4, 81; (c) P. de Mayo and S. T. Reid, Q. Rev., Chem. Soc., 1961, 15, 393; (d) P. J. Kropp, Org. Photochem., 1967, 1, 1; (e) O. L. Chapman and D. S. Weiss, Org. Photochem., 1973, 3, 197; (f) D. I. Schuster, Acc. Chem. Res., 1978, 11, 65; (g) K. Schaffner and M. M. Demuth, in 'Rearrangements in Ground and Excited States', ed. P. de Mayo, Academic Press, New York, 1980, vol. 3.
- 74. (a) P. J. Kropp and W. F. Erman, J. Am. Chem. Soc., 1963, 85, 2456; (b) H. Dutler, H. Bosshard and O. Jeger, Helv. Chim. Acta, 1957, 40, 494.
- 75. B. Miller and H. Margulies, J. Am. Chem. Soc., 1967, 89, 1678.
- 76. B. Miller, J. Am. Chem. Soc., 1965, 87, 5515; 1967, 89, 1685.
- 77. S. M. Kupchan, C.-K. Kim and K. Miyano, J. Chem. Soc., Chem. Commun., 1976, 91.
- (a) A. G. Shultz, M. Macielag and M. Plummer, J. Org. Chem., 1988, 53, 391; (b) A. G. Shultz, F. P. Lavieri, M. Macielag and M. Plummer, J. Am. Chem. Soc., 1987, 109, 3991.
- 79. (a) K. Tsuda, E. Ohki and S. Nozoe, J. Org. Chem., 1963, 28, 786; (b) E. J. Bailey, A. Gale, G. H. Phillipps, P. T. Siddons and G. Smith, J. Chem. Soc., Chem. Commun., 1967, 1253; (c) H. L. Dryden, G. M. Webber and J. J. Wieczorek, J. Am. Chem. Soc., 1964, 86, 742; (d) P. Wieland and G. Anner, Helv. Chim. Acta, 1967, 50, 289; (e) K. Tsuda, E. Ohki, J. Suzuki and H. Shimizu, Chem. Pharm. Bull., 1961, 9, 131.
- 80. T. P. Ananthanarayan, P. Magnus and A. W. Norman J. Chem. Soc., Chem. Commun., 1983, 1096.
- 81. S. M. Kupchan, C.-K. Kim and J. T. Lynn, J. Chem. Soc., Chem. Commun., 1976, 86.
- 82. S. M. Kupchan, A. J. Liepa, V. Kameswaran and R. F. Bryan, J. Am. Chem. Soc., 1973, 95, 6861.
- 83. S. J. Danishefsky, J. Morris and L. A. Clizbe, *Heterocycles*, 1981, **15**, 1205. 84. R. S. Ward and K. S. Thatcher, *Tetrahedron Lett.*, 1981, **22**, 4831.
- 85. B. Eistert, H. Fink and A. Müller, Chem. Ber., 1962, 95, 2403.
- 86. F. Langer, E. Zbiral and F. Wessely, Monatsh. Chem., 1959, 90, 623.
- 87. A. Siegel, P. Stockhammer and F. Wessely, Monatsh. Chem., 1957, 88, 228.
- 88. F. Langer and F. Wessely, Monatsh. Chem., 1957, 88, 298.
- 89. E. Zbiral, Monatsh. Chem., 1960, 91, 1144.
- 90. M. Grossa and F. Wessely, Monatsh. Chem., 1966, 97, 1384.
- 91. J. Kotlan and F. Wessely, Monatsh. Chem., 1957, 88, 118.
- 92. F. Wessely and M. Grossa, Monatsh. Chem., 1966, 97, 570.
- 93. F. Wessely, L. Holzer and H. Vilcsek, Monatsh. Chem., 1952, 83, 1253.
- 94. J. S. Davies, C. H. Hassall and J. A. Schofield, J. Chem. Soc., 1964, 3126; Chem. Ind. (London), 1963, 740.
- 95. R. M. Dodson, J. R. Lewis, W. P. Webb, E. Wenkert and R. D. Youssefyeh, J. Am. Chem. Soc., 1961, 83, 938.
- 96. S. Ebine, Bull. Chem. Soc. Jpn., 1965, 38, 2029.
- 97. K. Yamane, Bull. Chem. Soc. Jpn., 1961, 34, 42.
- 98. R. D. Haworth and P. R. Jefferies, J. Chem. Soc., 1951, 2067.
- 99. G. Huber, Angew. Chem., 1951, 63, 501.

3.6 Benzil–Benzilic Acid Rearrangements

G. BRYON GILL University of Nottingham, UK

3.6.1 INTRODUCTION	821
3.6.2 SCOPE AND MECHANISM	822
3.6.2.1 Benzilic Acid Type Rearrangements 3.6.2.2 The Benzilic Ester Rearrangement	822 823
3.6.3 THE CHEMISTRY OF BENZILIC ACID REARRANGEMEN	TS 825
 3.6.3.1 Reaction Conditions 3.6.3.2 Bisaryl α-Diketones 3.6.3.3 Quinones 3.6.3.4 I-Alkyl-2-aryl-1,2-dicarbonyl Systems 3.6.3.5 Aliphatic and Alicyclic α-Diketones 3.6.3.6 Heterocyclic Systems 3.6.3.7 Related Rearrangements 	825 825 828 829 831 834 834
3.6.4 REFERENCES	836

3.6.1 INTRODUCTION

In 1838 von Liebig¹ described the hydroxide ion induced transformation of an α -diketone, benzil (1), into the salt of an α -hydroxy acid, benzilic acid (2; equation 1). This process is probably the first molecular rearrangement reaction to be recognized, and hence occupies a special place in both synthetic organic and physical organic chemistry. In its strictest sense the benzil-benzilic acid rearrangement refers to the reaction of bisaryl α -diketones with hydroxide ion leading to aryl migration. Generally, however, the reaction is chronicled in a wider sense to include the migrations of other groups (*e.g.* alkyl, acyl, nitrogen), which result from the treatment of vicinal dicarbonyl systems with hydroxide, methoxide, *t*-butoxide and other bases. This wider interpretation is used in this chapter.

$$\begin{array}{cccc}
 & O & & OH^{-} & Ph & CO_{2}^{-} \\
 & O & & Ph & Ph & OH^{-} \\
 & O & & Ph & OH^{-} \\
 & (1) & & (2) \end{array}$$
(1)

3.6.2 SCOPE AND MECHANISM

3.6.2.1 Benzilic Acid Type Rearrangements

The prototype reaction is the conversion of glyoxal into glycolic acid (equation 2),² and here the benzilic acid rearrangement mechanism coincides with that for an intramolecular Cannizzaro reaction. The reaction is observed with other purely aliphatic α -diketones such as *t*-butyl 2,3-dioxobutyrate³ and cyclohexane-1,2-dione (equations 3 and 4),⁴ but the scope is limited in the aliphatic series by competing (*e.g.* aldol) reactions. Suitably constructed heterocyclic systems also rearrange, and the conversion of alloxan (3) into alloxanic acid (4) was among the first of the benzilic acid rearrangements to be discovered (equation 5).⁵



Much of the definitive mechanistic evidence has been provided by studies of the reaction of hydroxide ion with benzil (1) and substituted benzils.⁶ The presently accepted mechanism is that which was proposed by Ingold⁷ as early as 1928, and is shown in Scheme 1; see, however, Section 3.6.2.2. The initial step involves the fast reversible addition of hydroxide to one of the carbonyl groups. In unsymmetrical α -diketones (5) the more electrophilic carbonyl (*i.e.* the one with the lower LUMO) is attacked preferentially. Thus, if R² in (5) is more electron withdrawing than R¹, the concentration of intermediate (6) greatly exceeds that of (7), and the rearrangement follows the (6) \rightarrow (9) \rightarrow (8) pathway. In physical organic terms, the observed rate constant (k_{obs}) is the product of the equilibrium constant (K) for the first step, and the rate constant (k) for the rate-determining second step. Thus $k_{obs} = Kk$ and, of the two variables, the key term is K, since the rate-limiting transition state is not far along the reaction coordinate leading from (6) to (9). It follows, therefore, that it is the more electron-withdrawing substituent R² which migrates (in the absence of steric problems).^{6,8}

With simple unsymmetrical α -diketones (5) it appears to be immaterial which substituent migrates, for the product structure is the same. Much of the information concerning migratory aptitudes has come from studies of benzils possessing one ¹⁴C=O group.⁶ Hydroxide ion attack at the labeled center results in the migration of the attached aryl substituent with the production of carboxy-labeled benzilic acid. The rearrangement is irreversible. Similar labeling studies in other α -dicarbonyl systems^{3,9-13} have indicated the preferential migration of H over Ph, Ph over Me, CONH₂ over Ph, CO₂R over CO₂⁻, and CO₂R or CO₂⁻ over cyclopropyl. These results are all consistent with initial hydroxide ion addition to the more electrophilic C=O group.



When R^1 and/or R^2 in (5) possess a chiral center, both the initial attack of hydroxide ion and the subsequent migration can be stereoselective, regioselective and stereospecific. Steroidal α -diketones are suitable vehicles for such studies, and their chemistry is considered later (see Section 3.6.3.5).

The results obtained for alloxan $(3)^9$ and related N-heterocycles¹⁴ are much more complex. The central of the three contiguous C=O groups is the most electrophilic part of the molecule (3) and, indeed, alloxan readily forms a stable hydrate through the addition of water to this center. In alkaline solution, and depending on pH, the mono-, di- and tri-anions (10)–(12) exist; (10) is much more stable than the corresponding species for benzil and does not rearrange. Thus, isotopic labeling of this C-5 carbon atom affords, on rearrangement in aqueous alkali, C-4 ring-labeled alloxanic acid. The carboxylic acid group contains none of the isotope, which would be the requirement if (10) was involved directly in the ratedetermining rearrangement to the anion of (4). Furthermore, analysis of the products of alkaline rearrangement of C-4 labeled (3), and its methyl- and phenyl-substituted derivatives, is consistent with the exclusive shift of nitrogen over carbon. In the rearrangement of N-phenyl[4-¹⁴C]alloxan the shift of NPh occurs to the exclusion of NH, whereas in the case of N-methyl[4-¹⁴C]alloxan the migration ratio of NMe to NH is about 4:1. Clearly, then, C-5 is the migration terminus, and rearrangement is triggered by hydroxide ion attack at C-4 (or the symmetry related C-6 position). Specific catalysts such as borate, which is ineffective for the benzil–benzilic acid transformation, enhance the electrophilicity of C-4 by coordinating the carbonyl oxygen atom, as shown in formula (13).



3.6.2.2 The Benzilic Ester Rearrangement

A somewhat confusing situation existed in the early literature concerning the precise timing of the migration and proton transfer steps and the apparent specificity of hydroxide ion, because of the failure of alkoxides to give benzilic esters.⁶ However, benzil is a good oxidizing agent for alcohols in the presence of their alkoxides,¹⁵ and Doering and Urban showed¹⁶ that the failure of the rearrangement with primary and secondary alkoxide ions was due to competitive hydride transfers (a Meerwein–Ponndorf–Verley– Oppenauer type of reaction). Nevertheless, boiling anhydrous methanolic sodium methoxide gave methyl benzilate (68%), benzoin (9.6%) a product of the above redox reaction, and benzilic acid (9.7%) which probably arose from a $B_{AL}2$ mechanism (*i.e.* S_N2 attack by methoxide on the methyl carbon atom of methyl benzilate). The reaction of (1) with sodium ethoxide in dry ethanol at reflux gave a mixture of rearrangement, oxidation and cleavage products. The reaction of (1) with potassium *t*-butoxide in boiling benzene is especially clean, affording *t*-butyl benzilate in 90% yield. The reaction fails with the weakly basic phenoxides, presumably because of an unfavorable equilibrium (cf. the formation of $\mathbf{6}$) in the initial attack on the α -diketone.

Rearrangement of benzil in nonhydroxylic solvents by nonoxygen bases such as sodium amide¹⁷ or acetylide,¹⁸ followed by mild hydrolytic work-up, afforded not the expected benzilamide or 1,1-diphenyl-2-oxobut-3-yn-1-ol but rather benzilic acid itself. As the mild conditions were insufficient to bring about the hydrolysis of the amide, and the only source of the required oxygen atom was another molecule of benzil, Selman and Eastham⁶ suggested the mechanism given in Scheme 2. The consistent recovery of starting material in these reactions is accounted for by the mechanism — which was also suggested as a pathway by which benzilic acid could form in some of the alkoxide-initiated processes (X = OR).



Scheme 2

In recent years, single-electron transfer (SET) mechanisms have been suggested for a number of organic processes, including a number of base-catalyzed reactions. These include the Wittig 1,2-rearrangement (see Chapter 3.11, this volume), the Cannizzaro reaction,¹⁹ Meerwein-Ponndorf-Verley reductions,²⁰ and also the benzilic acid (or ester) rearrangement.²¹ Benzil is not rearranged by lithium dialkylamides (Scheme 2; $X = NEt_2$, NPr^{i_2}), but these bases are of low nucleophilicity. Reaction with lithium t-butoxide in benzene-THF gives intense violet-colored paramagnetic solutions which exhibit an ESR spectrum consistent with the presence of the lithium semidione of benzil (14). On the basis of this evidence the SET mechanism given in Scheme 3 was proposed.²¹ Although the Bu'OLi reaction gave tbutyl benzilate in high yield (96%), similar reaction using LiOEt failed to cause rearrangement, but the semidione was detected. In a somewhat related case, the condensation products of 3-amino-2,4-dicyanocrotonic esters with benzils (see Section 3.6.3.2) undergo a benzilic acid type rearrangement in aqueous alkali, and ¹H NMR-CIDNP experiments support the assumption of a partial radical character for the processes.²² Hence, the SET mechanisms do deserve careful consideration, but the problem is that such proposals are highly controversial at the present time,²³ and it is not yet clear if the SET mechanism for benzilic acid rearrangements will achieve general acceptance. The mechanism raises a number of questions on interpretation. For example, the semidione species is a radical anion, four canonical forms of which are shown in formulae (14a-d). It is not yet clear if the electron density distribution in (14a-d) finds a satisfactory parallel in the experimentally determined migratory aptitudes of the substituents R¹ and \mathbb{R}^2 . Likewise, it is not clear why phenoxide anions, which are efficient one-electron transfer agents, fail to promote the rearrangement.





Hence, although the presence of radicals has been demonstrated, it has yet to be shown that the SET pathway is more than a minor competitive reaction, and for the remainder of this review the anionic mechanism set out in Scheme 1 will be assumed to apply.

3.6.3 THE CHEMISTRY OF BENZILIC ACID REARRANGEMENTS

3.6.3.1 Reaction Conditions

The sensitivity, or otherwise, of the α -dicarbonyl substrate determines the reaction conditions. However, variations in conditions are not especially large. Although 1 equiv. of base may be used, more usually an excess is employed — sometimes 10 equiv. or more to accelerate the reaction. Studies on the reaction of hydroxides with benzil indicate that the rate depends to some extent on the cation,^{24a} and it can be advantageous to use Ba(OH)₂ rather than the more usual NaOH or KOH.^{24b} With hydroxide ion, favored solvents are water and aqueous ethanol. Heterogeneous conditions, for example potassium hydroxide and diethyl ether²⁵ or sodamide and toluene,^{17,26} have also been employed. The reactions are conducted at room temperature (sometimes requiring up to 4 d) or under reflux (10 min–24 h). With methoxide or *t*-butoxide the corresponding anhydrous alcohol or benzene are employed as solvents.^{16,21}

3.6.3.2 Bisaryl α -Diketones

A detailed *Organic Synthesis* procedure is available for the conversion of benzil into benzilic acid.^{27a} It is advantageous to isolate the acid as its potassium salt, for this enables the removal of the more soluble potassium benzoate, which results from a competitive cleavage reaction. Benzilic acid is then obtained in 77–79% yield by acidifying an aqueous solution of potassium benzilate. Since benzil is usually obtained by the oxidation of benzoin, both this conversion and rearrangement can be performed in tandem by using alkaline sodium bromate.^{27b} Thus, benzilic acid is obtained in 84–90% yield from benzaldehyde *via* benzoin.

The rearrangement of substituted benzils is facilitated by electron-withdrawing substituents in the *meta* or *para* positions (m > p); steric hindrance retards the rate of rearrangement of *ortho*-substituted aryl groups. Some typical examples of the rearrangement of simple bisaryl α -diketones are summarized in Table 1. The very high reactivity of decafluorobenzil is noteworthy; quite exceptionally the rate of the benzilic acid rearrangement exceeds that for hydride transfer reduction by ethoxide ion (see Section 3.6.2.2) and ethyl decafluorobenzilate is obtained in 90% yield in the benzilic ester rearrangement.³² The analogous reaction with sodium methoxide was found to occur at -78 °C. The similar reaction of

Table 1 Rearrangement of Substituted Benzils



Ar ¹	Ar ²	Temperature (°C)	Time (h)	Yield (%)	Ref.
Ph Ph 4-Pr ⁱ C ₆ H₄ 2-OH-5-BrC ₆ H ₃ C ₆ F ₅ 2-Furyl	4-ClC ₆ H ₄ 2-MeC ₆ H ₄ 4-MeOC ₆ H ₄ 4-Pr ¹ C ₆ H ₄ 2-OH-5-BrC ₆ H ₃ C ₆ F ₅ 2-Furyl	25 80 80 100 25 25 25	72 0.17 0.17 17 0.08 24	70 80 60 70 68 83 88	28 29 28, 30 31 24 32 25

decafluorobenzil with pentafluorophenyllithium gave the pentafluorophenyl migration product, $(C_6F_5)_2C(OH)COC_6F_5$, after low temperature hydrolysis (see Section 3.6.3.7).

The generality of the rearrangement is further illustrated by the reaction of 2,2'-furil with hydroxide ion in dry ether (Table 1).²⁵ Likewise, 2,2'-pyridil is rearranged in hot methanol solution (40 min) to give the sodium salt of 2,2'-pyridilic acid (86%).³³ Acidification, however, affords bis(2-pyridyl)methanol by decarboxylation since 2,2'-pyridilic acid (16) is structurally similar to a β -keto acid. Benzilic rearrangement of 2,2'-pyridilic acid (17; 92%). A plausible mechanism is summarized in Scheme 4.³⁴ Rearrangement is also observed with 2,2'-quinaldil, but benzil, 2,2'-furil or 1-phenyl-2-(2'-pyridyl)ethane-1,2-dione are not susceptible to these metal template reactions.



Scheme 4

Simple benzilic acid derivatives frequently show pronounced pharmacological activity.³⁵ For example, the ester (18) is an anticholinergenic and spasmolytic agent (Minelsin, Ortyn). The reaction of benzil with urea in ethanol in the presence of potassium hydroxide affords 5,5'-diphenylhydantoin (22),³⁶ the sodium salt of which is the anticonvulsant Dilantin. The probable mechanism is shown in Scheme 5, the function of the hydroxide ion being the deprotonation of the initial adduct (19a), thereby triggering the benzilic rearrangement of the anion (20). The corresponding reaction with *N*-methylurea gives 3-methyl-5,5-diphenylhydantoin (21b; 95%), the regioselectivity resulting from the preferential formation of (19b) through the attack by the more nucleophilic nitrogen center. Substituted benzils³⁷ and ureas³⁶ or thioureas³⁸ generally react to give analogous hydantoins; however, condensation reactions without rearrangement compete in some cases.

In somewhat similar chemistry, a one-step synthesis of 4-cyano-3,3-diaryl-5-methyl-2-oxo-2,3-dihydropyrroles (23) is available through the reaction of benzils with acetonitrile in the presence of sodium hydride (Scheme 6).³⁹ Likewise, the base-catalyzed addition of malononitrile to benzil⁴⁰ yields 2-benzoyl-2-phenylethylene-1,1'-dicarbonitrile (24) which, in the presence of aqueous alkali, undergoes 1,2aryl migration (Scheme 7) and cyclization, yielding the succinimide (25).⁴¹ Condensation of benzils with 3-amino-2,4-dicyanocrotonic esters gives the ylidene cyanoacetates (26), which rearrange to the ylidene pyrrolidines (27) in the presence of aqueous alkali (Scheme 8); ¹H NMR-CIDNP experiments indicate that radical intermediates are formed under the rearrangement conditions.²² The regiochemistry of the initial condensation is determined by the more electrophilic of the two carbonyl groups of the benzil.



3.6.3.3 Quinones

The benzilic rearrangement of an *o*-quinone results in ring contraction. Among the earliest recorded examples is the synthetically useful conversion of phenanthroquinones into 9-hydroxyfluorenecarboxylic acids (equation 6).^{42–46} As with the benzilic acids, compounds such as (**28**) are susceptible to oxidative decarboxylation, especially under acidic conditions. Simply boiling a solution of (**28**) in oxygenated water gives the 9-fluorenone derivative.⁴⁴ The phenanthroquinones are readily prepared through the chromium(VI) oxidation of phenanthrenes. When phenanthrene derivatives are oxidized with alkaline permanganate, oxidation to the quinone and benzilic acid rearrangement occur in tandem; furthermore, permanganate cleaves α -hydroxy acids, and the 9-fluorenone is thus obtained directly.⁴⁵



The outcome of the reaction of acenaphthoquinone with hydroxide ion is quite different. Benzilic rearrangement, which would lead to a highly strained four-membered ring, is not observed. Instead, simple cleavage occurs to give naphthaldehyde-9-carboxylate (equation 7).^{24a} Despite the fact that the benzilic acid and ester rearrangements have been shown to be irreversible, they apparently cannot be employed routinely in the construction of strained rings (see Section 3.6.3.5). This contrasts with the semibenzilic mechanism (*e.g.* Scheme 9)⁴⁷ which operates in certain base-induced rearrangements of α -halo ketones when the conventional Favorskii mechanism is not possible (see Section 3.6.3.7).



Scheme 9

Annelated quinones such as phenanthroquinone are best regarded simply as bisaryl α -diketones. The benzilic acid rearrangement mechanism does not figure large in the chemistry of the simpler benzoquinones, presumably because there are too many other opportunities for reaction, such as oxidation and reduction, condensation, Michael addition and retro-aldol cleavage. However, benzilic rearrangements do occur, and may not be uncommon in the oxidative rearrangements of hydroxyquinones.⁴⁸ The Hooker oxidation of 2-hydroxy-3-alkyl-1,4-naphthoquinones by alkaline permanganate gives the next lower homolog, and it has been shown⁴⁹ that the alkyl and hydroxy groups exchange places. The mechanism shown in part in Scheme 10 has been proposed for the Hooker oxidation (and the related H₂O₂/OH⁻ oxidation),^{49,50} and contains a benzilic acid ring contraction step. The R group in (29) is usually saturated, but examples in which R contains unsaturation, alcohol, ether or carboxylic acid functionality have also been studied. When R is a large group (*t*-butyl or cyclohexyl), treatment with 5% aqueous alkali in the absence of oxygen leads to the formation of 2-alkylindenone-3-carboxylic acids (31) in high yield *via* the benzilic acids (30), which may be isolated under buffered conditions.^{51,52} An analogous transformation is involved in the base-promoted rearrangement of the natural pigment dunnione (32) into allodunnione

The rearrangements of hydroxy-1,4-benzoquinones in alkali are more complex.⁴⁸ Under conditions in which polyporic acid (34) is converted, by way of retro-aldol ring cleavage and benzilic rearrangement, into α -benzyl- β -phenylsuccinic acid, *cis*- and *trans*- α -benzylcinnamic acid and oxalic acid, atromentin

Benzil-Benzilic Acid Rearrangements



(35) gives the lactone (36), which affords the analogous cinnamic acid only on boiling in 50% alkali.^{54,55} Analogously, 4-hydroxycyclopentane-1,3-diones are available in good yield from 2,6-dihydroxy-1,4benzoquinones by base-induced benzilic rearrangement followed by decarboxylation; equation (8) is fairly typical.^{48,55}



3.6.3.4 1-Alkyl-2-aryl-1,2-dicarbonyl Systems

The simplest of such systems is phenylglyoxal (PhCOCHO), which undergoes H migration to give mandelic acid (PhCHOHCO₂H; 72.5%) on reaction with barium hydroxide.^{13,56} Likewise, the reaction of α -(4-isobutylphenyl) β -methyl diketone with 10% aqueous sodium hydroxide-xylene mixture at 100 °C for 5 h affords a 96% yield of α -(4-isobutylphenyl)- α -hydroxypropionic acid (4-Me₂CHCH₂C₆H₄C(Me)OHCO₂H) on work-up.⁵⁷ The rearrangement of both methyl phenyl diketone and α , α -dibromopropiophenone (PhCOCBr₂Me) occurs with 100% phenyl migration;¹⁰ the yield of

atrolactic acid ($PhC(Me)OHCO_2H$) is rather low since aldol condensation competes. It is essential, therefore, to keep the concentration of the diketone low to limit the rate of the competing bimolecular reaction.



The reaction of benzylideneacetophenone oxide (37) with aqueous sodium hydroxide in ethanol at reflux for 1.5 h affords, on work-up, a 72% yield of the acid (38). The reaction proceeds *via* benzyl phenyl diketone, and isotopic labeling indicates that exclusive benzyl migration occurs.⁵⁸ Benzyl 2-hydroxyphe-

nyl diketones are available through the base-promoted ring scission of 2-benzyl-2-hydroxy-2*H*-benzo[*b*]furan-3-ones (*e.g.* the hardwood extract maesopsin 39a)⁵⁹ and 3-hydroxyflavanones (*e.g.* 40).⁶⁰ Hence, under alkaline conditions rearrangement products are formed (*i.e.* 3-benzyl-3-hydroxy-3*H*-benzo[*b*]furan-2-ones, respectively 41 and 42) as shown in Scheme 11. A photochemical equivalent of the benzilic acid rearrangement has also been observed in this series. Thus, irradiation of 4,4',6-tri-*O*-methylmaesopsin (39b) in THF-water at 350 nm afforded the rearrangement product (43) in modest yield.⁶¹

The cyclopentapyrazoles (45; R = H, Me) are obtained in 67% and 54% yields, respectively, by benzilic acid rearrangement of the corresponding indazolediones (44).⁶² This ring contraction has many analogs in the next section, and may occur during alkaline permanganate oxidations of suitable substrates. For example, oxidation of fuscinic acid dimethyl ether (46) affords (48) and (49), implicating the α -diketone (47) as a common intermediate.⁶³



3.6.3.5 Aliphatic and Alicyclic α-Diketones

Glyoxal is the simplest member, and is readily rearranged in aqueous alkali to glycolic acid (equation 2).² This and similar rearrangements of simple aliphatic α,β -dicarbonyl and α,β,γ -tricarbonyl systems are of mechanistic rather than synthetic interest and importance. However, in handling such compounds for other synthetic purposes it is necessary to recognize their lability to base. In the rearrangement of *t*-butyl α,β -dioxobutyrate, which exists in aqueous solution as the hydrate MeCOC(OH)₂CO₂Bu^t, exclusive 1,2-shift of the CO₂Bu^t group occurs (equation 3).³ Likewise, in alkaline medium ethyl cyclopropane-2,3-dioxopropionate is transformed into PrⁿC(OH)(CO₂H)₂ by rearrangement only of CO₂Et or CO₂⁻ (not cyclopropyl)¹² and only CONH₂ migration occurs with the compounds ArCH₂COC(OH)₂CONH₂.¹²

Compounds which are converted into α -diketones by base are susceptible to benzilic acid rearrangement. Especially common is the base-catalyzed dehydration of α,β -dihydroxycarbonyl compounds, and the transformation of glyceraldehyde into lactic acid (Scheme 12) is a simple example of α -hydroxy acid formation by this route.⁶⁴ The higher carbohydrates also undergo similar alkali-induced degradative rearrangements,⁶⁵ but here an α -ketol rearrangement (see Section 3.6.3.7), for example (50) \rightarrow (51) in Scheme 13,⁶⁶ precedes the formation and benzilic rearrangement of the α -diketone. α,β -Dihydroxy ketones are also formed in the ring opening of α,β -epoxy ketones by hydroxide, and hence produce the rearranged α -hydroxy acids by way of the intermediate α -diketone; an example, (37) \rightarrow (38), was given earlier.⁵⁸



The ring contraction rearrangement of alicyclic α -diketones is much more valuable from the synthetic viewpoint (*e.g.* equation 4). There are several routes available for the synthesis of α -diketones from monoketones. These include: direct oxidation using selenium dioxide; permanganate or osmium tetroxide addition across the double bond of an enone followed by base-promoted elimination of water; bromination to give the dibromo ketone followed by hydrolysis; condensation with 4-nitroso-N,N-dimethylaniline followed by acidic hydrolysis — but there are many others. The bromination-hydrolysis route has been extensively studied by Wallach.⁴ The conversion of menthone into 1-hydroxy-



Scheme 13

2-isopropyl-5-methylcyclopentanecarboxylic acid (56) is fairly typical. In the dibromination of menthone it was proven that the bromine atoms are not attached to the same carbon atom, but rather the dibromide has the structure (52). The dihydroxy ketone formed in the initial hydrolysis step (53) presumably is transformed (cf. ref. 4) into the necessary α -diketone (55; shown in a monoenolic form, a diosphenol) by sequential α -ketol rearrangement (53) \rightarrow (54) and dehydration (54) \rightarrow (55; Scheme 14). The same acid (56) is formed in the cognate reaction of carvomenthone, the reaction also proceeding via α -diketone (55).



Scheme 14

The general strategy for ring contraction also works well with more elaborate molecules. Regiospecific sulfenylation of the diketone (57) followed by acetoxylation afforded the α -acetoxy sulfide (58; 68%).⁶⁷ Hydrolysis and *in situ* rearrangement was effected with aqueous potassium hydroxide, and oxidative cleavage of the crude α -hydroxy acid (59) gave the perhydroindanone (60; 50%), a potential intermediate in the synthesis of taxanes. Relatively unstrained cage ketones can also be ring-contracted by employing the benzilic acid rearrangement strategy (for example equation 9).^{68a} There are few examples of the production of strained rings. However, cyclobutane-1,2-diones have been shown to undergo ring contraction to α -hydroxycyclopropanecarboxylic acids and derivatives.^{68b} Relatively little attention has been paid to the use of other bases for promoting rearrangement. Cyclohexane-1,2-dione reacts in this manner with arylbiguanides (61) and with *N*-amidino-*O*-alkylisoureas (62) to give the heterocyclic derivatives (63) in fair yield (*cf.* Scheme 5).⁶⁹

Considerable interest has been shown in the rearrangement of the α -diketones of steroids and of diand tri-terpenoids. Of particular interest is the abutment of the reaction center to a chiral system of rigid stereochemistry. This allows an examination of the regiochemistry and the stereospecificity and stereoselectivity of the reaction with alkali. The isomeric diosphenols (64) and (65), respectively 3-hydroxy-5 α cholest-3-en-2-one and 2-hydroxy-5 α -cholest-1-en-3-one, are interconvertible in the presence of acid or base. On reaction of either with potassium hydroxide in propanol a single benzilic acid rearrangement


product, A-nor-5 α -cholestan-2-ol-2-carboxylic acid, was obtained.⁷⁰ The necessary common intermediate is presumably the α -diketone (**66**; Scheme 15) which can yield four possible hydroxide addition products arising from α - or β -attack at either of the two carbonyl groups. It was thought that of these (**67**) should be the most stable through the minimization of 1,3-diaxial interactions and oxygen-oxygen dipolar repulsions, and by stereospecific rearrangement should give 2β -hydroxy- 2α -carboxy-A-nor- 5α -cholestane (**68**). However, the configuration of the single product was not determined. Likewise, the diosphenol (**69**), 5α -pregn-1-en-2-ol-3,20-dione, was rearranged by treatment with potassium hydroxide in ethanol or propanol to give a single product, A-nor- 5α -pregnan-2-ol-20-one-2-carboxylic acid (90%).⁷¹ The configuration of the substituents at C-2 in the hydroxy acid (**70**) was tentatively assigned as 2β -hydroxy- 2α carboxy on the basis of similar reasoning to the above. There have been several other reports in this area of benzilic acid rearrangements in which only a single α -hydroxycarboxylic acid was isolated.⁷⁰

Reaction of a solution of 3α ,17 β -diacetoxy-11-hydroxy-5 β -androst-9(11)-en-12-one (71) in aqueous propanol with potassium hydroxide, followed by an acidic work-up, afforded the lactone (75; 76%). Three events are involved in this transformation (Scheme 16), namely retro-aldol equilibration to give the *cis*-fused c-D-ring system (72) \rightarrow (73), stereoselective benzilic acid rearrangement (73) \rightarrow (74) to



Scheme 15



provide the 11 β -carboxylic acid function, and finally lactonization (74) \rightarrow (75).⁷² Since the benzilic acid rearrangement has been shown to be irreversible,^{6,73} the formation of the lactone (75) indicates that the transformation (73) \rightarrow (74) is a highly stereoselective process.



Scheme 16

The major product of the rearrangement of 5α -cholestane-3,4-dione (**76**) was shown to be 3α -carboxy-A-nor- 5α -cholestan- 3β -ol (**78**). By using (**76**) labeled with ¹⁴C at C-4, it was shown that the carboxy group of (**78**) arose from C-3 of (**76**). The evidence, and the steric constraints imposed by the steroidal system, favors stereo- and regio-specific α -attack by hydroxide to give (**77**) and stereoselective rearrangement *via* a chair-like transition state to give the product (**78**; Scheme 17).¹¹ Rearrangement of 4,4dimethyl- 5α -cholestane-2,3-dione (**79**) gave only the hydroxy acid (**81**) with the carboxy group α . Labeling experiments indicated that 94% of the benzilic rearrangement resulted from attack at the C-3 carbonyl through a quasi-chair transition state. The remaining 6% of the rearrangement resulted from attack at the C-2 carbonyl through a quasi-boat transition state.⁷⁴ The similar reaction of 5α -cholestane-2,3-dione (**80**) gave the two hydroxy acids (**82**; 85%) and its epimer at C-2 (15%). Labeling studies reveal a more complex regio- and stereo-chemical situation than for the conversion of (**79**) into (**81**).⁷⁵

The stereochemistry of the hydroxy diacid obtained by the rearrangement of the dioxo ester (83), itself derived from (-)-abietic acid, has been determined unequivocally, and is as shown in (84).⁷⁶ Other stere-ospecific benzilic acid rearrangements have been reported, including ring contraction of a D-homo-17,17a-dione,⁷⁷ and contraction of ring A in ursolinic acid⁷⁸ and betulinic acid.⁷⁹

3.6.3.6 Heterocyclic Systems

Relatively little attention has been paid to this important area, and much of the definitive work has centered upon studies of the rearrangement of alloxan⁵ and substituted alloxan⁹ (equation 5). The essen-



tial features concerning mechanism and regiochemistry were covered in Section 3.6.2.1. These studies indicated the exclusive shift of nitrogen over carbon. A similar situation is found in the reactions of the phthalonimides (85) and (86) with aqueous alkali to give 3-hydroxyphthalimidine-3-carboxylic acids (87).¹⁴ Indeed, the ring-shrinking process can be conducted on the 1,3(2,4)-isoquinolinediones (88) or their monothio analogs (89), and their substituted derivatives, by the expedient of using excess hydrogen peroxide in aqueous alkali at room temperature (*ca*. 3 h or less). The phthalonimide intermediates (85) and (86) are converted rapidly under the reaction conditions into the hydroxy acids (87). These can be isolated provided that the excess peroxide is destroyed before acidification (46–81%), otherwise oxidative decarboxylation occurs and the corresponding phthalimides are formed. Kinetic measurements on the amide quinisatine (90) have also revealed that the rearrangement involves the shift of the carboxyamide group.^{12b} The base-promoted conversion of 2-phenylindolone (91) into 3-phenyldioxindole (92), which occurs readily, may also be classified as a benzilic acid type rearrangement.^{24,80}



3.6.3.7 Related Rearrangements

The key steps in several additional rearrangements can be represented as in equations (10) and (11), and they are effectively the 'back-end' and the 'front-end' respectively of the benzilic acid rearrange-ment itself. Selman and Eastham⁶ and others^{8,81} have provided a brief analysis of such processes, and their scope. These reactions are referred to by a variety of names; for equation (10) Selman and Eastham suggest they be called ' α -oxo alcohol rearrangements', but the 'tertiary α -ketol rearrangement' or 'acyloin rearrangement⁸¹ are more common labels. Equation (11) differs from a benzilic acid rearrangement only inasmuch as the displacement occurs on a σ -bonding electron pair rather than the π -bonding electron pair of a carbonyl group, and has been referred to earlier as a 'semibenzilic rearrangement' (see Scheme 9). Z^{-} is a stable anion such as halide or tosylate, and the reaction may operate when the Favorskii mechanism is not possible. The subsequent loss of the carboxylate proton from (93; R = H) under the basic conditions of equation (11) renders the process irreversible. However, even with alkoxides as bases (e.g. 93; R = Me) the inherent stability of the carboxylic ester function and the lack of electron pressure on the α -carbon atom, makes the reaction irreversible in the general sense. No such situation exists for equation (10), and its reversible nature frequently leads to the formation of mixtures of products. The tertiary α -ketol rearrangement occurs upon treatment of a tertiary alcohol containing an α -oxo group with base, or by reaction of an α -diketone with 1 equiv. of a Grignard reagent. The base-catalyzed reactions of α -hydroxyketosteroids have been especially well-studied. The reader is referred to the above-mentioned review articles for further insights into this area.

$$\begin{array}{c} \begin{array}{c} Y \\ X \\ \end{array} \\ \end{array} \\ \begin{array}{c} Y \\ \end{array} \\ \end{array} \\ \begin{array}{c} Y \\ \end{array} \end{array}$$
 (10)

$$\begin{array}{c}
\overbrace{} O \\ RO \\ (Z \\ R^{1} \\ (Z \\ R^{1}$$

3.6.4 REFERENCES

- 1. J. von Liebig, Justus Liebigs Ann. Chem., 1838, 25, 27.
- 2. H. Fredenhagen and K. F. Bonhoeffer, Z. Phys. Chem. Abt. A, 1938, 181, 379; see also D. O'Meara and G. N. Richards, J. Chem. Soc., 1960, 1944.
- 3. H. Gowal, Lê H. Dao and H. Dahn, Helv. Chim. Acta, 1985, 68, 173.
- 4. O. Wallach, Justus Liebigs Ann. Chem., 1918, 414, 296; J. Chem. Soc., 1918, 114/I, 442; Justus Liebigs Ann. Chem., 1924, 437, 148.
- 5. F. Wöhler and J. von Liebig, Justus Liebigs Ann. Chem., 1838, 26, 241; A. Schlieper, Justus Liebigs Ann. Chem., 1845, 55, 263; H. Biltz, M. Heyn and M. Bergius, Justus Liebigs Ann. Chem., 1917, 413, 68; G. M. Richardson and R. K. Cannan, Biochem. J., 1929, 23, 68; J. W. Patterson, Jr., A. Lazarow and S. Levey, J. Biol. Chem., 1949, 177, 187.
- 6. S. Selman and J. F. Eastham, Q. Rev., Chem. Soc., 1960, 14, 221.
- C. K. Ingold, Annu. Rep. Prog. Chem., 1928, 25, 124.
 C. J. Collins and J. F. Eastham, in 'The Chemistry of the Carbonyl Group', ed. S. Patai, Interscience, New York, 1966, p. 783; D. J. Cram, in 'Fundamentals of Carbanion Chemistry', Academic Press, New York, 1965, p. 238; T. S. Stevens and W. E. Watts, in 'Selected Molecular Rearrangements', van Nostrand Reinhold, New York, 1973, p. 126.
- 9. H. Kwart and I. M. Sarasohn, J. Am. Chem. Soc., 1961, 83, 909; H. Kwart, R. W. Spayd and C. J. Collins, J. Am. Chem. Soc., 1961, 83, 2579.
- 10. K. S. Warren, O. K. Neville and E. C. Hendley, J. Org. Chem., 1963, 28, 2152.
- 11. J. Levisalles and I. Tkatchenko, Bull. Soc. Chim. Fr., 1966, 1287, and refs. cited therein.
- (a) H. Dahn, Lê H. Dao and R. Hunma, Helv. Chim. Acta, 1982, 65, 2458; (b) H. Gowal, A. Spiess, M. 12 Ballenegger, L. Duc, H. Moll, H.-P. Schlunke and H. Dahn, Helv. Chim. Acta, 1985, 68, 2132.
- 13. W. E. Doering, T. I. Taylor and E. F. Schoenewaldt, J. Am. Chem. Soc., 1948, 70, 455; O. K. Neville, J. Am. Chem. Soc., 1948, 70, 3499.
- 14. P. A. S. Smith and R. O. Kan, J. Am. Chem. Soc., 1961, 83, 2580.
- 15. H. Adkins and R. C. Franklin, J. Am. Chem. Soc., 1941, 63, 2381.
- 16. W. E. Doering and R. S. Urban, J. Am. Chem. Soc., 1956, 78, 5938.
- 17. I. Kasiwagi, Bull. Chem. Soc. Jpn., 1926, 1, 66 (Chem. Abstr., 1926, 20, 2491).

- 18. J. Cymerman-Craig, M. Moyle, P. Rowe-Smith and P. C. Wailes, Aust. J. Chem., 1956, 9, 391.
- 19. E. C. Ashby, D. T. Coleman, III and M. P. Gamasa, Tetrahedron Lett., 1983, 24, 851; S.-K. Chung, J. Chem. Soc., Chem. Commun., 1982, 480.
- 20. C. G. Screttas and C. T. Cazianis, Tetrahedron, 1978, 34, 933; E. C. Ashby, A. B. Goel and J. N. Argyropoulos, Tetrahehron Lett., 1982, 23, 2273.
- C. G. Screttas, M. Micha-Screttas and C. T. Cazianis, Tetrahedron Lett., 1983, 24, 3287.
 B. Thierrichter and H. Junek, Monatsh. Chem., 1979, 110, 729.
- 23. M. Newcomb and D. P. Curran, Acc. Chem. Res., 1988, 21, 206; E. C. Ashby, Acc. Chem. Res., 1988, 21, 414; D. T. Sawyer and J. L. Roberts, Jr., Acc. Chem. Res., 1988, 21, 469; M. Newcomb, R. M. Sanchez and J. Kaplan, J. Am. Chem. Soc., 1987, 109, 1195; S.-U. Park, S.-K. Chung and M. Newcomb, J. Org. Chem., 1987, 52, 3275.
- 24. (a) E. Pfeil, G. Geissler, W. Jacquemin and F. Lömker, Chem. Ber., 1956, 89, 1210; (b) A. Garcia-Raso, J. V. Sinisterra and J. M. Marinas, Pol. J. Chem., 1982, 56, 1435.
- 25. T. W. Evans and W. M. Dehn, J. Am. Chem. Soc., 1930, 52, 252
- 26. S. Selman, Diss. Abstr., 1960, 20, 3083 (Chem. Abstr., 1960, 54, 10 948c).
- 27. (a) R. Adams and C. S. Marvel, Org. Synth., 1921, 1, 29; (b) D. A. Ballard and W. M. Dehn, Org. Synth., Coll. Vol., 1932, 1, 89.
- M. T. Clark, E. C. Hendley and O. K. Neville, J. Am. Chem. Soc., 1955, 77, 3280.
 J. F. Eastham, R. G. Nations and C. J. Collins, J. Org. Chem., 1958, 23, 1764.
- 30. J. D. Roberts, D. R. Smith and C. C. Lee, J. Am. Chem. Soc., 1951, 73, 618.
- 31. M. Bösler, Ber. Dtsch. Chem. Ges., 1881, 14, 323.
- 32. R. D. Chambers, M. Clark and D. J. Spring, J. Chem. Soc., Perkin Trans. 1, 1972, 2464; R. D. Chambers and M. Clark, J. Chem. Soc., Perkins Trans. 1, 1972, 2469. 33. J. Klosa, Naturwissenschaften, 1957, 44, 558; J. Prakt. Chem., 1960, 10, 335.
- 34. D. St. C. Black and R. C. Srivastava, Aust. J. Chem., 1969, 22, 1439.
- 35. J. Klosa, Arch. Pharm. Ber. Dtsch. Pharm. Ges., 1954, 287, 321; 1955, 288, 42.
- 36. H. Biltz, Ber. Dtsch. Chem. Ges., 1908, 41, 1379.
- 37. W. R. Dunnavant and F. L. James, J. Am. Chem. Soc., 1956, 78, 2740.
- 38. J. Klosa, Arch. Pharm. Ber. Disch. Pharm. Ges. 1956, 289, 223.
- 39. S. Akabori, M. Ohtomi, K. Takahashi and Y. Sakamoto and Y. Ichinohe, Synthesis, 1980, 900.
- 40. D. M. W. Anderson, F. Bell and J. L. Duncan, J. Chem. Soc., 1961, 4705.
- 41. H. Junek, B. Hornischer and H. Hamböck, Monatsh. Chem., 1969, 100, 503.
- 42. A. Baeyer, Ber. Dtsch. Chem. Ges., 1877, 10, 123; P. Friedländer, Ber. Dtsch. Chem. Ges., 1877, 10, 534.
- 43. R. Nietzki, Ber. Dtsch. Chem. Ges., 1890, 23, 3136.
- 44. J. Schmidt and K. Bauer, Ber. Disch. Chem. Ges., 1905, 38, 3737.
- R. Anschütz and F. R. Japp, Ber. Disch. Chem. Ges., 1878, 11, 211.
 D. G. Ott and G. G. Smith, J. Am. Chem. Soc., 1955, 77, 2325.
- 47. M. Mousseron and N. Phuoc Du, C. R. Hebd. Seances Acad. Sci., 1944, 218, 281; for an application in the synthesis of strained rings see R. N. McDonald and C. A. Curi, Tetrahedron Lett., 1976, 1423, and refs. cited therein.
- 48. H. W. Moore and R. J. Wikholm, in 'The Chemistry of the Quininoid Compounds', ed. S. Patai, Wiley, New York, 1974, part 1, p. 425.
- 49. L. F. Fieser, J. L. Hartwell and A. M. Seligman, J. Am. Chem. Soc., 1936, 58, 1223; L. F. Fieser and D. H. Sachs, J. Am. Chem. Soc., 1968, 90, 4129; see also L. F. Fieser and M. Fieser, J. Am. Chem. Soc., 1948, 70, 3215.
- 50. M. M. Shemyakin and L. A. Shchukina, Q. Rev., Chem. Soc., 1956, 10, 261.
- 51. L. F. Fieser, J. Am. Chem. Soc., 1948, 70, 3237; L. F. Fieser and A. R. Bader, J. Am. Chem. Soc; 1951, 73, 681
- R. G. Cooke, Nature (London), 1948, 162, 178; R. G. Cooke and T. C. Somers, Nature (London), 1950, 165, 52. 314.
- 53. J. R. Price and R. Robinson, J. Chem. Soc., 1939, 1522; R. G. Cooke and T. C. Somers, Aust. J. Sci. Res., Ser. A, 1950, 3, 487.
- 54. F. Kögl, H. Becker, A. Detzel and G. de Voss, Justus Liebigs Ann. Chem., 1928, 465, 211; F. Kögl, H. Becker, G. de Voss and E. Wirth, Justus Liebigs Ann. Chem., 1928, 465, 243.
- 55. J. F. Corbett and A. G. Fooks, J. Chem. Soc. C, 1967, 1909; J. F. Corbett, J. Chem. Soc. C, 1967, 2408.
- 56. J. Hine and G. F. Koser, J. Org. Chem., 1971, 36, 3591; D. L. Vander Jagt, L. P.-B. Han and C. H. Lehmann, J. Org. Chem., 1972, **37**, 4100.
- Y. Ichikawa, K. Sakoda, Y. Yamanaka and T. Naruchi, Jpn. Pat. 77 14 740 (1977) (Chem. Abstr., 1977, 87, 57. 134 618n).
- C. J. Collins and O. K. Neville, J. Am. Chem. Soc., 1951, 73, 2471.
 N. F. Janes, F. E. King and J. W. W. Morgan, J. Chem. Soc., 1963, 1356; G. B. Guise, E. Ritchie and W. C. Taylor, Aust. J. Chem., 1962, 15, 314.
- 60. T. Oyamada, Justus Liebigs Ann. Chem., 1939, 538, 44; D. Molho, J. Coillard and C. Mentzer, Bull. Soc. Chim. Fr., 1954, 1397; see also D. Molho and M. Chadenson, Bull. Soc. Chim. Fr., 1959, 453.
- 61. J. H. van der Westhuizen, D. Ferreira and D. G. Roux, J. Chem. Soc., Perkin Trans. 1, 1977, 1517.
- 62. A. Ya. Strakov, J. Sliede and I. A. Strakova, Latv. PSR Zinat. Akad. Vestis, Kim. Ser., 1976, 469; (Chem. Abstr., 1977, 86, 29 709z).
- 63. D. H. R. Barton and J. B. Hendrickson, J. Chem. Soc., 1956, 1028; A. J. Birch, Chem. Ind. (London), 1955, 682.
- 64. J. C. Sowden and E. K. Pohlen, J. Am. Chem. Soc., 1958, 80, 242.
- 65. J. C. Sowden, Adv. Carbohydr. Chem., 1957, 12, 35.
- 66. This may be an over-simplification see G. Machell and G. N. Richards, J. Chem. Soc., 1960, 1924.
- 67. B. M. Trost and H. Hiemstra, Tetrahedron, 1986, 42, 3323.

- 68. (a) E. Boelema, J. Strating and H. Wynberg, Tetrahedron Lett., 1972, 1175; (b) for leading refs. see J.-M. Conia and M. J. Robson, Angew. Chem., Int. Ed. Engl., 1975, 14, 473.
- 69. M. Furukawa and S. Hayashi, Synthesis, 1974, 132.
- 70. H. R. Nace and M. Inaba, J. Org. Chem., 1962, 27, 4024, see also refs. cited therein.
- H. R. Nace and D. H. Nelander, J. Org. Chem., 1964, 29, 1677.
 P. Kurath, J. Org. Chem., 1967, 32, 3626.
- 73. J. F. Eastham and S. Selman, J. Org. Chem., 1961, 26, 293.
- 74. J. Levisalles and I. Tkatchenko, Bull. Soc. Chim. Fr., 1967, 3125.
- J. Levisalles and I. Tkatchenko, Bull. Soc. Chim. Fr., 1967, 3131; see also S. Huneck, J. Levisalles and I. Tkatchenko, Bull. Soc. Chim. Fr., 1967, 3140.
- 76. T. Nakata, Y. Ohtsuka, A. Tahara and S. Takada, Chem. Pharm. Bull., 1975, 23, 2318.
- 77. V. Georgian and N. Kundu, Chem. Ind. (London), 1958, 1322.
- 78. A. M. Osman, M. El-Garby Younes and A. E. Sheta, Egypt. J. Chem., 1972, 15, 269 (Chem. Abstr., 1973, 79, 92 429d).
- 79. A. M. Osman, M. El-Garby Younes and A. Mokhtar, Indian J. Chem., 1975, 13, 193.
- 80. L. Kalb and J. Bayer, Ber. Dtsch. Chem. Ges., 1912, 45, 2150.
- 81. P. de Mayo, in 'Elucidation of Structures by Physical and Chemical Methods', 'Techniques of Organic Chemistry', ed. K. W. Bentley, Wiley, New York, 1963, vol. 11, part 2, p. 1125.

3.7 The Favorskii Rearrangement

JOHN MANN University of Reading, UK

3.7.1 INTRODUCTION	839
3.7.2 MECHANISM	840
3.7.3 SYNTHETIC UTILITY	842
 3.7.3.1 Acyclic α-Halo Ketones 3.7.3.2 α-Halocycloalkyl Ketones 3.7.3.3 Monocyclic Halo Ketones 3.7.3.4 Bicyclic Halo Ketones 3.7.3.5 Polycyclic Halo Ketones 3.7.3.6 Miscellaneous 	842 845 848 851 854 854 856
3.7.4 REFERENCES	857

3.7.1 INTRODUCTION

The rearrangement of α -halo ketones under the influence of base was first described by Favorskii in 1892,^{1,2} and the general scope of the reaction and the mechanistic implications have been the subject of a number of reviews.³⁻¹² In its most generally useful form, α -halo ketones undergo skeletal rearrangement when treated with a nucleophilic base (hydroxide, alkoxide or amine) to produce salts, esters and amides respectively (Scheme 1). With polyhalogenated ketones unsaturated acid derivatives are produced, as shown in Scheme 2.



Scheme 1



The reactions are usually carried out by adding the halo ketone to a solution or suspension of the base in either a protic solvent (water, alcohols) or an ether (diethyl ether, dioxane, dimethoxyethane). The bases employed include: hydroxides of Group I and Group II metals; alkoxides, carbonates and hydrogen-carbonates of Group I metals; ammonia, and amines. There is no one recommended set of experimental conditions, because both the mechanism of the reactions and the type of products obtained depend upon the initial choice of α -halo ketones. In addition, since the halo ketones are highly reactive molecules, side products are inevitably obtained; those most commonly produced are shown in Scheme 3.



3.7.2 MECHANISM

The various mechanisms that have been proposed for the rearrangement have been discussed in full,¹² and only a brief summary will be given here. Two main reaction mechanisms have been established: (i) a symmetrical mechanism, also known as the cyclopropanone pathway; and (ii) an unsymmetrical mechanism or semibenzylic pathway (Schemes 4 and 5).

The symmetrical mechanism necessitates an acidic hydrogen at the α' -center, and the intermediacy of a cyclopropanone or zwitterion/oxyallyl cation (Scheme 4). The likely existence of a symmetrical intermediate was first suggested by McPhee and Klingsberg¹³ when they showed that the two ketones (1) and (3) rearranged to yield the same ester (2; Scheme 6). More cogent experimental support was provided by Loftfield¹⁴ who reacted [1,2-¹⁴C₂]-2-chlorocyclohexanone with sodium pentylate and obtained two ring-contracted products. These had an identical isotope distribution to the starting ketone (Scheme 7).

Numerous more recent experiments (reviewed in ref. 12) have confirmed the general accuracy of the mechanism shown in Scheme 4, and the actual intermediate involved depends upon the relative stabilities of the zwitterions/oxyallyl cations and the corresponding cyclopropanones. These will be affected by the choice of solvent and structural features of the starting ketone, such as the degree of substitution and ring strain (in cyclic halo ketones). A recent example in which an oxyallyl intermediate and a cyclopropanone intermediate were both intercepted is shown in Scheme 8.¹⁵





Rearrangement Reactions

In the semibenzylic pathway there is a nucleophilic attack by the base at the carbonyl carbon, followed by a concerted 1,2-migration of a carbanionic moiety with concomitant expulsion of halide. Implicit in this mechanism is the requirement for an antiparallel arrangement of the C—C bond that is broken and the carbon-halogen bond. This further requires an inversion of configuration at the carbon center initially bonded to the halogen atom, and an elegant demonstration of this feature has been provided by Charpentier and coworkers (Scheme 9).¹⁶

In cyclic systems where ring strain may be large for the cyclopropanone pathway, the semibenzylic pathway is preferred, and a good example of this is shown in Scheme $10.^{17}$ Even when the ring strain is less severe, the semibenzylic pathway will be favored if there is a pseudo-equatorial halogen, thus allowing the new C—C bond to form from the pseudo-axial direction.



3.7.3 SYNTHETIC UTILITY

3.7.3.1 Acyclic α-Halo Ketones

The Favorskii rearrangement of α -monohalo ketones is of primary utility for the synthesis of branched-chain aliphatic acid derivatives, and a typical example is shown in equation (1).^{18,19} With increasing substitution at the α -carbon center, the rearrangement is favored since competing side reactions are suppressed; increasing the substitution at the α '-carbon center disfavors the rearrangement due to steric hindrance.



The structural equivalence of the starting materials with respect to their common rearrangement product has already been mentioned, and two further examples are illustrated in Scheme 11.^{20,21} In practice this often means that the ketone can be halogenated, and the mixture of halo ketones used directly in the Favorskii rearrangement.





Aryl groups generally have a labilizing effect on the substrate, and the rearrangements are particularly facile, often requiring only weak bases for their initiation. However, if aryl groups are present on both α and α' -centers, alkoxy epoxides are formed in preference to the Favorskii products (equation 2).^{22,23} Complications can also arise if an alkyl group is present at the α -center, of which the results shown in Scheme 12²⁴ are examples. It appears that the rearrangements are subject to the competitive influences of carbanion stability (secondary more stable than tertiary) and steric factors. The proportions of the two products also vary with the steric bulk of the base.



percentage of (Y) in the product mixture increases from 16 to 80% as R changes from Me to Et to Pr^{i}

Scheme 12

When the Favorskii rearrangement is carried out on a substrate which contains an internal nucleophile, this can attack the cyclopropanone intermediate to yield cyclic products. The reaction shown in Scheme 13^{25} provides a route to polysubstituted γ -butyrolactones by this kind of mechanism.



The reaction of a polyhalo ketone with base (equation 3)²⁶ was described several years before Favorskii described a similar rearrangement (equation 4).²⁷ Simple α,α - and α,α' -dihalo ketones yield the same product (Scheme 14),²⁸ so dibromination can be followed by Favorskii rearrangement without purification. These rearrangements are usually stereospecific, though subsequent *cis* to *trans* isomerization, under the influence of base, often obscures this feature. Various explanations have been provided for this stereospecificity, but the one due to Rappe (illustrated in Scheme 15) seems most plausible.²⁹

$$Br \underbrace{KOH, EtOH}_{Br} CO_2Et \underbrace{KOH, EtOH}_{CO_2H} CO_2H$$
(3)



The reactions of polyhalo ketones often proceed in good yield, and the examples shown in equations (5) and (6) are typical.^{30,31} Favorskii-type rearrangements of α, α' -dihalo ketones with enolates of diethoxyphosphinylacetic esters³² or with sodiomalonates,³³ have provided useful routes to divinyl ketones and conjugated enones respectively (Scheme 16). Most of this chemistry is of a general nature, but a reaction that provides a key intermediate for the elaboration of an analog of obtusilactone (4) is shown in Scheme 17.³⁴ These analogs are of interest because the parent lactone (R = H) has antitumor activity. Finally, the tribromo ketones (5) undergo base-induced rearrangement and elimination to yield conjugated dienes (6) with a high (Z)-stereoselectivity (Scheme 18).³⁵



Scheme 16



3.7.3.2 a-Halocycloalkyl Ketones

The reactions of monohalocycloalkyl ketones are of considerable interest because they allow access to cycloalkylcarboxylic acids; the rearrangements shown in Scheme 19 are examples. When pure stereoisomers were employed, complete racemization resulted, suggesting the intermediacy of ion pairs.³⁶ This methodology has been used for the synthesis of the potent analgesic pethidine (8) from the piperidyl ketone (7).³⁷ The lack of products from bromomethyl cyclohexyl ketone (9) was ascribed to a slow (ratedetermining) removal of hydrogen to produce a tertiary carbanion, thus allowing competing side reactions to occur.³⁸





Relatively inaccessible cycloalkylcarboxylic acids can be produced via these Favorskii rearrangements (for example equation 7),³⁹ and they have been especially useful for the introduction of alkyl groups at C-17 of the steroid nucleus. Thus the pregnenolones (10) and (11) can be converted into mixtures of rearrangement products (12) and (13), with the latter usually predominating in polar, protic solvents (Scheme

20).⁴⁰ The potent progestin (14) and the relatively inactive cortisone analog (15) have been prepared via this strategy.^{41,42}



Scheme 20

When a nonprotic, though relatively polar, solvent, such as dimethoxyethane, was employed, the $17-\alpha$ bromopregnenolone yielded primarily the $17-\beta$ -methyl product; these results have been rationalized as shown in Scheme 21.⁴³ The $17-\alpha$ -bromo steroid and the 21-bromo steroid are converted *via* separate concerted processes into discrete stereoisomeric cyclopropanones; these transformations are in competition with the formation of the common dipolar species which can yield both stereoisomers. However, one of these will usually be favored on steric grounds, and the rearrangement can thus proceed with a high degree of stereoselectivity. The steric effect of having a $12-\alpha$ -substituent or substituents at C-16 has been probed, and the result is a lowering of stereoselectivity observed in polar aprotic solvents.^{44,45}

With dihalogenated analogs of these steroids, rearrangement yields the anticipated products in high yield (Scheme 22),⁴⁶ and these unsaturated acids are appropriately substituted for elaboration of the steroid side chain.

Finally, the related trihalogenated steroids produce α -bromo acids, and the example shown in equation (8) is representative.⁴⁷



A is a polar, protic solvent; B is an aprotic, medium polarity solvent

Scheme 21



Scheme 22



3.7.3.3 Monocyclic Halo Ketones

Favorskii rearrangements have been reported with alicyclic and heterocyclic bromocycloalkanones containing 4–13 (but not 5) atoms in the rings. The important mechanistic studies with 2-bromocyclobutanone have already been mentioned (Scheme 10),¹⁷ and Scheme 23 contains a summary of other results obtained by Conia and coworkers.^{17,48} Of particular note is the fact that the rearrangement will even proceed with water as reagent, and that the stereospecificity observed is consistent with a mechanism involving (as expected) inversion of configuration at the carbon center initially attached to the bromine atom.





Reactions of this type have been much employed for the construction of cyclopropane carboxylic acids that are appropriately functionalized for elaboration into the commercially important pyrethroid insecticides (Scheme 24 and equation 9).^{49,50} The stereospecificity observed in the second report is of particular note given the superior potency of the *cis*-pyrethroids.



No Favorskii rearrangement products are produced when α -halocyclopentanones are treated with base, and products of aldol, substitution and dehydrohalogenation reactions are produced. However, when an α -halocyclopentanone is part of certain condensed systems, cubane derivatives may be obtained (see Section 3.7.3.5).

The reactions of α -halocyclohexanones have been most thoroughly studied, and the reaction shown in equation (10),⁵¹ carried out on the molar scale, is an example. In general the chloro compounds provide larger yields than the bromo and fluoro analogs, and branched-chain aliphatic alkoxides are claimed to be the most effective bases.⁷



The synthetic potential of these reactions can be illustrated by their use for the construction of polyfunctionalized cyclopentanecarboxylic acids and derivatives (Scheme 25).^{52-54a} The Michael-induced Favorskii rearrangement is of note, as is the use of carvone as a chiral starting material,^{54b} and the annelation procedure that follows the rearrangement shown in Scheme 26.⁵⁵



The corresponding rearrangements of higher homologs are also well documented (*e.g.* 2-chlorocyclododecadienone⁵⁶ and 2-bromocycloundecanone)⁵⁷ as are the base-induced rearrangements of heterocyclic haloalkanones, such as the conversion of halolactams into proline and homologs (equation 11).⁵⁸ In addition, the rearrangement of 2-chlorocaprolactam has been reported on the 3.5 mol scale.⁵⁹



When polyhalocycloalkanones are employed the anticipated products are almost invariably obtained, and the reactions of polybromomonoterpenes are of particular synthetic value. Dibromination of pulegone produced dibromo ketone (16), and reaction of this (on a 0.3 mol scale) produced the results shown in Scheme 27.⁶⁰ The *trans*-acid usually predominates because there are steric constraints to the hydrolysis of the *cis*-ester. The related tribromomonoterpene (17) provides access to carvenolide (18;

Scheme 28)⁶¹ and both rearrangement processes have potential utility for the construction of cyclopentanoid monoterpenes, *e.g.* the iridolactones which are important as constituents of insect defensive secretions.



aq. NaOH, cis:trans \approx 1:1; NaOEt, EtOH, cis:trans \approx 0:100

Scheme 27



Scheme 28

Larger ring polyhalocycloalkanones behave in a predicatable fashion. Thus 2,8-dibromocyclooctanone yields cyclohept-1-enecarboxylic acid in 96% yield when treated with aqueous NaOH, and 2,2,8-tribro-



Scheme 29

mocyclooctanone when treated with ethanolic sodium ethoxide yields 2-bromocyclohept-1-enecarboxylic acid in 83% yield.⁶² The results obtained with dibromo- undecanones and -dodecanones are shown in Scheme 29.⁶³ Finally, when various dibromomonochlorocycloalkanones are treated with sodium methoxide, there is an almost exclusive preference for displacement of bromide rather than chloride (equation 12).⁶⁴



3.7.3.4 Bicyclic Halo Ketones

Cis-fused 5,6- and *cis*-fused 6,6-bicyclic halo ketones undergo Favorskii rearrangements to produce 5,5- and 6,5-bicyclic ring systems, respectively. The *trans*-5,6-bicyclic system does not yield rearranged products, presumably due to steric constraints. Representative examples are shown in Schemes 30 and 31,^{65,66} and the high degree of retention of stereochemistry at the bridging centers is noteworthy.



An interesting synthesis of bicyclo[n.1.0] systems is shown in equation (13),⁶⁷ though the yields for n = 6 are rather poor. Of more preparative use is the chemistry illustrated in Scheme 32,⁶⁸ by which bicy-

clic analogs of proline may be prepared. These were required as components of potential inhibitors of angiotensin-converting enzyme. Once again a high degree of diastereoselectivity was observed.



With nonfused bicyclic systems, most reactions have been carried out with compounds containing a bridgehead halogen; though the chemistry of 2-bromobicyclo[3.2.1]octan-3-ones has provided interest-



NaOMe, MeOH, A, 10%, B, 5%, C, 74%; NaOMe, DME, A, 76%, endo ester, 24%

Scheme 33



ing mechanistic results (Scheme 33).^{69,70} A preponderance of the solvolysis products is obtained when methanol is used as solvent, whilst dimethoxyethane favors the Favorskii rearrangement. Protonation of the intervening enolate or zwitterion would disfavor formation of a cyclopropanone, and favor attack at the carbocation by methoxide. Interestingly, both equatorial and axial bromides of the related fused benzobicyclo[3.2.1]octanone (19) react with sodium methoxide to provide good yields of Favorskii rearrangement products (Scheme 34).⁷¹ The greater facility of this reaction has been attributed to the inherent instability of the bishomoantiaromatic carbocation (20), which would favor the alternative formation of the cyclopropanone (21), and resultant Favorskii rearrangement.

A mixture of the 2-bromobicyclo[3.3.1]nonan-3-ones (22) rearranges with great stereoselectivity and good yield to the *exo*-ester (23; equation 14).⁷²



Most other systems studied have bridgehead halogens, and special attention has been paid to the reactions of the kind shown in Scheme 35. The yields of rearrangement products are uniformly good, and deuterium incorporation results imply that the semibenzilic mechanism operates for the smaller ring



i, n = 6, Bu^tOK, Bu^tOD; R = Bu^t, 94%, 0.026 atoms D incorporated
 ii, n = 7, NaOMe, MeOD; R = Me, 96%, 0.00 atoms D incorporated
 n = 7, Bu^tOK, Bu^tOD; R = Bu^t, 67%, 0.83 atoms D incorporated
 iii, n = 8, NaOD, EtOD, D₂O; R = H, 88%, 0.90 atoms D incorporated





NaOH, THF, (26):(27) = 89:11 aq. NaOH, (26):(27) = 64:36 sizes, whilst a cyclopropanone is implicated for the larger ring size (cf. Schemes 4 and 5). This is reasonable since Bredt's rule would suggest a difficult enolization for the ring systems with n = 6 and 7. When stronger base (t-butoxide) was employed, the cyclopropanone pathway apparently also operates for the system with n = 7.73

Compound (24), with two bridgehead chlorines, yields the bicyclo[2.2.0] acid (26) and the cyclohexanecarboxylic acid (27). The relative proportion of the latter was increased when water was present, presumably because there was more quenching of the intermediate anion (25; Scheme 36).⁷⁴

The related dihalo ketone (28) rearranges smoothly at 0 °C, whilst bromo ketone (29) requires prolonged heating at 155 °C (Scheme 37).⁷⁵ This is anticipated, since formation of the intermediate (30) required for the semibenzilic mechanism is sterically disfavored.



Scheme 37

3.7.3.5 Polycyclic Halo Ketones

Ĥ

In terms of synthetic utility these rearrangements fall into two major classes: (i) reactions of halo steroids and related systems; and (ii) reactions that yield cage-like structures. Reactions of the first type are exemplified by the rearrangements illustrated in Scheme 38.^{76–78}



mainly 2 α -ester (65%) and some 3 α -ester

Scheme 38

The rearrangements of cage-like structures have been discussed in some detail in refs. 11 and 12, and can be divided into three main classes: (i) homocubane to cubane; (ii) adamantane to noradamantane; and (iii) homoadamantane to twist-brendane; and these are depicted (together with strain energy estimations from ref. 79) in Scheme 39. It can be seen that in each case rearrangement provides a system with

greater strain, and the Favorskii reaction thus allows access to strained structures that would be otherwise inaccessible. Typical examples are shown in Scheme 40.80-85



It is interesting to note that with the homocubane derivatives it is the cyclopentanone ring that contracts, whilst this process has never been observed in simple halocyclopentanones. A semibenzilic mechanism probably operates in most cases, since the requisite cyclopropanones are unlikely to be produced. Ring-opened products are not infrequent side products, and they are favored products when highly halogenated precursors are employed (Scheme 41).⁸⁶ The intermediate carbanions are stabilized by the electron-withdrawing halogens, and these lead to products that are less strained than the usual cage products.



Scheme 41

3.7.3.6 Miscellaneous

All of the examples described thus far have involved the use of bases to initiate the Favorskii rearrangements. However various Lewis acids have also been used for this purpose. An early example of this methodology is shown in Scheme 42.⁸⁷ The Lewis acid promoted ring contractions depicted in Scheme 43⁸⁸ and Scheme 44⁸⁹ are also Favorskii-type rearrangements.



KOH, Et_2O , R = OH, 34%; Na, liq. NH₃, R = NH₂, 68%; AgNO₃, aq. EtOH, R = Et, 11%; R = OH, 61%; Hg(OAc)₂, EtOH, R = Et, 71%

Scheme 42





Scheme 44



Scheme 47

3.7.4 REFERENCES

- 1. A. E. Favorskii, Zh. Russ. Fiz.-Khim. O-va., 1892, 24, 254.
- 2. A. E. Favorskii, Zh. Russ. Fiz.-Khim. O-va., 1894, 26, 556.
- 3. A. E. Favorskii, Zh. Russ. Fiz.-Khim. O-va., 1894, 26, 590.
- 4. R. Jacquier, Bull. Soc. Chim. Fr., 1950, 35.
- 5. B. Tchoubar, Bull. Soc. Chim. Fr., 1955, 1363.
- 6. A. S. Kende, Org. React. (N.Y), 1960, 11, 261.
- 7. A. A. Akhrem, T. K. Ustynyuk and Y. A. Titiv, Russ. Chem. Rev. (Engl. Transl.), 1970, 39, 732.
- 8. H. Hart and G. J. Karabatsos, Adv. Alicyclic Chem., 1971, 3, 46.
- 9. C. Rappe, in 'The Chemistry of the Carbon-Halogen Bond', ed. S. Patai, Wiley, London, 1973, p. 1084.
- 10. E. Buncel, in 'Carbanions: Mechanistic and Isotopic Aspects', Elsevier, Amsterdam, 1975, p. 143.

- 11. P. J. Chenier, J. Chem. Educ., 1978, 55, 286.
- 12. A. Baretta and B. Waegell, in 'Reactive Intermediates', ed. R. A. Abramovitch, Plenum Press, New York, 1982, vol. 2, p. 527.
- 13. W. D. McPhee and E. Klingsberg, J. Am. Chem. Soc., 1944, 66, 1132.
- R. B. Loftfield, J. Am. Chem. Soc., 1951, 73, 4707.
 M. W. Finch, J. Mann and P. D. Wilde, Tetrahedron, 1987, 43, 5431.
- 16. D. Baudry, J. P. Bégué and M. Charpentier-Morize, Bull. Soc. Chim. Fr., 1971, 1416.
- 17. J.-M. Conia and J. R. Salaün, Bull. Soc. Chim. Fr., 1964, 1957.
- 18. J. G. Aston and R. B. Greenburg, J. Am. Chem. Soc., 1940, 62, 2590.
- 19. M. Kopp-Mayer, C. R. Hebd. Seances Acad. Sci., 1957, 244, 1522.
- 20. J. G. Aston and J. D. Newkirk, J. Am. Chem. Soc., 1951, 73, 3900.
- C. L. Stevens and A. E. Sherr, J. Org. Chem., 1952, 17, 1228. 21.
- 22. W. G. Dauben, C. F. Hiskey and M. A. Muhs, J. Am. Chem. Soc., 1952, 74, 2082.
- 23. C. L. Stevens, M. L. Weiner and R. C. Freeman, J. Am. Chem. Soc., 1953, 75, 3977.
- 24. C. Rappe and L. Knutsson, Acta Chem. Scand., 1967, 21, 2205.
- 25. T. Sakai, A. Yamawaki, T. Katayama, H. Okada, M. Utaka and A. Takeda, Bull. Chem. Soc. Jpn., 1987, 60, 1067.
- 26. C. Cloez, Bull. Soc. Chim. Fr., 1890, 3, 602.
- 27. A. E. Favorskii, J. Prakt. Chem., 1895, 51, 533.
- 28. R. B. Wagner, J. Am. Chem. Soc., 1949, 71, 3214.
- 29. C. Rappe, Ark. Kemi, 1965, 24, 315.
- 30. R. B. Wagner and J. A. Moore, J. Am. Chem. Soc., 1950, 72, 974.
- C. Rappe, Org. Synth., 1973, 53, 123.
 E. Amano, K. Miyata, T. Sakai, A. Takeda and M. Utaka, Bull. Chem. Soc. Jpn., 1987, 60, 1945.
- 33. E. Amano, M. Ishikawa, T. Sakai, A. Takeda and M. Utaka, Bull. Chem. Soc. Jpn. 1987, 60, 2295.
- 34. T. Morimoto and M. Sekiya, Chem. Pharm. Bull., 1982, 30, 3513.
- 35. T. A. Engler and W. Falter, Synth. Commun., 1988, 18, 783.
- 36. E. E. Smissman and J. L. Diebold, J. Org. Chem., 1965, 30, 4005.
- 37. E. E. Smissman and G. Hite, J. Am. Chem. Soc., 1959, 81, 1201.
- 38. R. B. Loftfield and L. Schaad, J. Am. Chem. Soc., 1954, 76, 35.
- 39. B. Tchoubar, Bull. Soc. Chim. Fr., 1955, 1363.
- 40. D. F. Morrow, T. P. Culbertson and R. M. Hofer, J. Org. Chem., 1967, 32, 361. 41. C. R. Engel, K. F. Jennings and G. Just, J. Am. Chem. Soc., 1956, 78, 6153.
- 42. C. R. Engel, J. Am. Chem. Soc., 1956, 78, 4727.
- 43. C. R. Engel, Y. Mérand and J. Côté, J. Org. Chem., 1982, 47, 4485.
- 44. C. R. Engel, P. Lachance, J. Capitaine, J. Žee, D. Mukherjee and Y. Mérand, J. Org. Chem., 1983, 48, 1954.
- 45. R. T. Logan, R. G. Roy and G. F. Woods, J. Chem. Soc., Perkin Trans. 1, 1981, 2631.
- 46. J. Romo and A. Romo de Vivar, J. Am. Chem. Soc., 1957, 79, 1118.
- 47. W. J. Adams, D. K. Patel, V. Petrow and I. A. Stuart-Webb, J. Chem. Soc., 1954, 1825.
- 48. J.-M. Conia and J.-L. Ripoll, Bull. Soc. Chim. Fr., 1963, 755, 763, 773.
- 49. W. T. Brady, S. J. Norton and J. Ko, US Pat. 4 691 052 (1987) (Chem. Abstr., 1987, 107, 237 067).
- 50. D. Belluš, J. Dingwall, H. Greuter and P. Martin, Helv. Chim. Acta, 1981, 64, 2812.
- 51. D. W. Goheen and W. R. Vaughan, Org. Synth., 1959, 39, 37.
- 52. C. Bischoff and E. Schröder, J. Prakt. Chem., 1981, 323, 616.
- 53. S. Tsuboi, H. Nagae, H. Yamato and A. Takeda, Bull. Chem. Soc. Jpn., 1987, 60, 836.
- (a) S. Tsuboi, Y. Kurihara, T. Watanabe and A. Takeda, Synth. Commun., 1987, 17, 773; (b) J. M. Llera and B. Fraser-Reid, J. Org. Chem., 1989, 54, 5544. 54.
- 55. D. Wilkening and B. P. Mundy, Synth. Commun., 1984, 14, 227.
- 56. G. Haufe, Synthesis, 1983, 235.
- 57. K. Schenker and V. Prelog, Helv. Chim. Acta, 1953, 36, 896. 58. H. T. Nagasawa and J. A. Elberling, Tetrahedron Lett., 1966, 5393.
- 59. W. Kulicki and A. Zarecki, Pol. Pat. 122 826 (1983) (Chem. Abstr., 1984, 101, 90 774).
- J. Wolinsky and D. Chan, J. Org. Chem., 1965, 30, 41.
 J. Wolinsky, R. O. Hutchins and T. W. Gibson, J. Org. Chem., 1968, 33, 407.
 G. Hesse and F. Urbanek, Chem. Ber., 1958, 91, 2733.
- 63. E. W. Garbisch and J. Wohllebe, J. Chem. Soc., Chem. Commun., 1968, 306.
- 64. A. Abad, M. Arnó, J. R. Pedro and E. Seoane, J. Chem. Soc., Perkin Trans. 1., 1983, 2471.
- 65. J. Boussinesq, J.-P. Girard, P. F. Nau and R. Granger, C. R. Hebd. Seances, Acad. Sci., Ser. C, 1966, 262, 1598
- 66. G. A. Frank and H. O. House, J. Org. Chem., 1965, 30, 2948.
- 67. M. Birkhahn and E. V. Dehmlow, Liebigs Ann. Chem., 1987, 701.
- 68. R. Henning and H. Urbach, Tetrahedron Lett., 1983, 24, 5339.
- 69. P. J. Chenier and J. C. Kao, J. Org. Chem., 1976, 41, 3730.
- 70. A. Baretta and B. Waegell, Tetrahedron Lett., 1976, 753.
- 71. R. R. Rasmussen and J. W. Wilt, J. Org. Chem., 1975, 40, 1031.
- 72. T. Itooka, K. Matoba, T. Momose, O. Muraoka and T. Yamazaki, Chem. Pharm. Bull., 1986, 34, 2391. 73. W. T. Tai, E. W. Warnhoff and C. M. Wong, J. Am. Chem. Soc., 1968, 90, 514.

- C. A. Curi and R. N. McDonald, *Tetrahedron Lett.*, 1976, 1423.
 W. C. Fong, K. V. Scherer and R. Thomas, *Tetrahedron Lett.*, 1971, 3789.
- 76. H. R. Nace and B. A. Olsen, J. Org. Chem., 1967, 32, 3438. 77. A. C. De Paulet, D. E. Evans, C. W. Shoppee and F. Winterrnitz, J. Chem. Soc., 1957, 1451.
- 78. S. J. Daum, Tetrahedron Lett., 1984, 25, 4725.
- 79. N. L. Allinger, M. A. Miller and M. T. Tribble, Tetrahedron, 1972, 28, 1173.

- 80. T. W. Cole and P. E. Eaton, J. Am. Chem. Soc., 1964, 86, 3157.
- 81. V. J. DiPasquo, G. L. Dunn and J. R. E. Hoover, J. Org. Chem., 1968, 33, 1454.
- 82. J. C. Barborak, R. Pettit and L. Watts, J. Am. Chem. Soc., 1966, 88, 1328.
- 83. G. J. A. Ariaans, A. J. H. Klunder, E. A. R. M. van der Loop and B. Zwanenburg, Tetrahedron, 1986, 42, 1903.
- L. H. Sommer and O. W. Webster, J. Org. Chem., 1964, 29, 3103.
 B. R. Vogt, Tetrahedron Lett., 1968, 1579.
- 86. W. G. Dauben and L. N. Reitman, J. Org. Chem., 1975, 40, 841.
- 87. A. C. Cope and E. S. Graham, J. Am. Chem. Soc., 1951, 73, 4702.
- A. Chatterjee, S. R. R. Chaudhuri and S. B. Maiti, Synthesis, 1987, 806.
 Y. Fujimoto, H. Miura and T. Tatsuno, Synthesis, 1979, 898.
- 90. J. P. De Clerq, N. De Kimpe, M. Van Meerssche, L. Möens, N. Schamp and P. Sulmon, J. Org. Chem., 1986, **51**, 3839.
- 91. P. Bakuzis, R. J. Baumgarten, C. L. Leicht, H. P. Schenk and E. Wenkert, J. Am. Chem. Soc., 1971, 93, 3208.
- 92. P. Ceccherelli, M. Curini, R. Pellicciari and E. Wenkert, J. Org. Chem., 1982, 47, 3172.

3.8 The Ramberg–Bäcklund Rearrangement

JOHN M. CLOUGH

ICI Agrochemicals, Bracknell, UK

3.8.1	INTRODUCTION	861
3.8.2	SCOPE AND IMPORTANT USES OF THE REACTION	862
3.8.3	PREPARATION OF PRECURSORS: a-HALO SULFONES	862
3.8.4	REACTION CONDITIONS	862
3.8.5	COMPATIBILITY OF FUNCTIONAL GROUPS	865
3.8.6	MECHANISM OF THE REACTION	866
3.8.7	RELATED REACTIONS	868
3. 3.	8.7.1 Replacement of the Halide 8.7.2 Replacement of the Sulfone Group	868 870
3.8.8	SYNTHESIS OF CYCLOALKENES	871
3.8 3.8 3.8 3.8	8.8.1 Cyclobutenes 8.8.2 Cyclopentenes, Cyclopentadienes and Cyclopentenones 8.8.3 Cyclohexenes and Phenanthrenes 8.8.4 Large Ring Systems	871 874 876 877
3.8.9	SYNTHESIS OF CONJUGATED DIENES, ENYNES AND POLYENES	878
3.8.10	SYNTHESIS OF ALKYNES	883
3.8.11	REFERENCES	884

3.8.1 INTRODUCTION

In 1940, Ramberg and Bäcklund reported a new and surprising reaction, *i.e.* that on treatment with an excess of aqueous KOH, α -bromoethyl ethyl sulfone was converted in high yield into 2-butene (equation 1).¹ α -Chloroethyl ethyl sulfone reacts in the same way, although more slowly. Also surprising was the stereoselectivity of this reaction; bromination of the 2-butene was found to give racemic 2,3-dibromobutane, showing that the (Z)-isomer predominated. Since that time, this transformation, now known as the Ramberg-Bäcklund rearrangement, or simply as the Ramberg-Bäcklund reaction, has been used to synthesize a wide variety of alkenes. It is now understood to take place *via* 1,3-elimination of halide from the sulfone α -anion, and loss of SO₂ from the resulting thiirane 1,1-dioxide (Scheme 1).

$$\frac{Br}{O^{"}S_{O}} = \frac{aq. KOH, 90-100 °C}{85\%} + KBr + K_2SO_3 (1)$$



The Ramberg-Bäcklund reaction has been reviewed several times previously,² and brief reviews have appeared as recently as 1988.³ The classic and most comprehensive review was written by Paquette in 1977⁴ and, though still extremely useful, particularly with respect to mechanism and the preparation of the α -halo sulfone precursors, it is now, of course, rather out of date. Two older reviews^{5,6} are still worth citing because they discuss the mechanism of the reaction in great detail.

3.8.2 SCOPE AND IMPORTANT USES OF THE REACTION

The Ramberg-Bäcklund reaction can be used to synthesize mono-, 1,1- or 1,2-di-, tri- and tetra-substituted alkenes, including alkenes substituted with a variety of functional groups. Despite the (Z)-stereoselectivity obtained with acyclic α -halo sulfones under the original conditions of Ramberg and Bäcklund, the reaction is most useful where stereoisomeric alkenes cannot occur because high stereocontrol is not a general feature. Nevertheless, very high (E)-stereoselectivity can be achieved in favorable circumstances, especially with functionalized alkenes.

One of the great strengths of the reaction is that there is no ambiguity about the position of the newly introduced double bond; it is fixed by the position of the sulfone group in the precursor, and never migrates under the reaction conditions. Good examples are shown in equations (3) and (29), and in Scheme 20.

The reaction is extremely useful for preparing strained cycloalkenes, particularly cyclobutenes. The success of such reactions stems from the fact that they are not ring forming, but involve contraction of a less-strained saturated ring.

Thiols may be converted into alkenes with one extra carbon atom by successive chloromethylation, oxidation at sulfur, and Ramberg-Bäcklund rearrangement. If the original thiol is prepared by free radical addition of H_2S to an alkene, the sequence constitutes a one-carbon homologation of alkenes. Although these transformations were at one time important applications of the Ramberg-Bäcklund reaction,^{4,5} they have hardly been used in recent years.

Incorporation of deuterium at the vinylic positions of alkenes can be carried out very efficiently by performing the Ramberg-Bäcklund reaction in D_2O . Finally, the recently developed 'Michael-induced' and 'vinylogous' modifications of the Ramberg-Bäcklund reaction offer useful approaches to conjugated dienes and polyenes.

3.8.3 PREPARATION OF PRECURSORS: a-HALO SULFONES

 α -Halo sulfones are most commonly prepared by successive oxidation and halogenation (or vice versa) of the corresponding sulfides. These, and other methods, have been described in detail by Paquette,⁴ while more recent reviews give numerous ways to prepare sulfones⁷ and α -halo sulfides.⁸ Representative pathways to α -halo sulfones are shown in the schemes throughout this review.

3.8.4 REACTION CONDITIONS

The first rearrangements of α -halo sulfones were performed using aqueous hydroxide, and these conditions are still frequently used. However, the rearrangement takes place in a wide variety of base-solvent systems and the yield and stereochemical outcome are often strongly dependent on the conditions chosen. The work of Scholz, in which he uses the reaction to prepare a variety of alkenes with remote carboxy groups, usefully illustrates this point. Examples are shown in Schemes 2⁹ and 3.^{10,11} In each case, treatment of a cyclic β -keto sulfone with a hypohalite produces a ring-opened α -halo sulfone bearing a carboxy group; Ramberg-Bäcklund rearrangement then furnishes the unsaturated carboxylic acid. A systematic survey showed that the stereochemical outcome of these Ramberg-Bäcklund reactions is highly dependent on both the base and the solvent used, while the temperature is relatively unimportant.¹¹ Whereas aqueous hydroxide favors the (Z)-alkenes, very high (E)-stereoselectivity can be achieved in favorable circumstances using Bu^tOK in DMSO. Scholz has also reported reactions analogous to those shown in Scheme 3 involving five- or seven-membered cyclic substrates.¹² An acyclic version gives simple alkenes as products, since the carboxy group is lost during halogenation.¹³



i, Br₂, aq. NaOH, 0 °C, 2 h; ii, Bu^tOK, Bu^tOH, reflux, 4 h; iii, aq. KOH, 100 °C, 3 h

Scheme 2



i, NaOCl, NaOH, aq. EtOH, 0 °C to r.t., 15 h; ii, Bu'OK, DMSO, r.t., 12 h; iii, 0.5 M aq. NaOH, 100 °C, 18 h

Scheme 3

The Ramberg-Bäcklund reaction using hydroxide as base can also be performed under phase-transfer conditions, the rate of reaction varying markedly from one substrate to another (equations 2 and 3).¹⁴ One advantage of using these conditions is that ester groups are not hydrolyzed (equation 4; compare equation 46).

Without doubt, the most important modification of the Ramberg-Bäcklund reaction is the result of the work of Meyers. In 1969, he reported that a suspension of powdered KOH in a mixture of CCl₄ and Bu^tOH is a powerful chlorinating system for a variety of compounds with acidic methylene groups.¹⁵

 $Ph \underbrace{O_{i}^{'} S_{O}^{'} Cl}_{O_{i}^{'} S_{O}^{'} Cl} \underbrace{\frac{10\% \text{ aq. NaOH, CH}_{2}Cl_{2}}{\text{Aliquat-336, r.t., 1.5 h}}_{82\%} Ph (2)$ $\underbrace{20\% \text{ aq. NaOH, CH}_{2}Cl_{2}}_{O_{i}^{'} S_{O}^{'} Cl} \underbrace{\frac{20\% \text{ aq. NaOH, CH}_{2}Cl_{2}}{\text{Aliquat-336, reflux, }}}_{86\%} (3)$



Sulfones with α -hydrogens, for example, are converted into α -chloro sulfones on treatment with this mixture of reagents. More interesting, however, is the observation that α -chloro sulfones are formed but cannot be isolated from reactions of sulfones with both α - and α' -hydrogens; instead, further reaction, often Ramberg-Bäcklund rearrangement, takes place in the basic reaction medium. Clearly, the ability to convert sulfones into alkenes in one operation, circumventing a separate halogenation step, is extremely useful. Furthermore, the reagents required for this reaction are easy to handle, readily available and are easily separated from the reaction products. For these reasons, Meyers' variant of the Ramberg-Bäcklund reaction has been used with great frequency, and many examples appear later in this chapter. Meyers himself has written two excellent reviews on this subject.^{16,17}

Numerous experiments support the following mechanism for the chlorination of sulfones under Meyers' conditions. Reaction takes place at the surface of the KOH, where a small concentration of Bu'OK is maintained and is responsible for converting the sulfone into its α -anion. This anion transfers an electron to a molecule of CCl₄ to form an intimate ⁴caged' radical/radical anion pair in which transfer of a chlorine atom is effected by nucleophilic attack of the substrate radical on a chlorine of the CCl₄ radical anion. The by-product of this process is the trichloromethyl anion, which can eject chloride to give dichlorocarbene. If the α -chloro sulfone has an α' -hydrogen, further reaction takes place, the course of which depends on the relative rates of 1,3-elimination (which leads via a thiirane 1,1-dioxide to the Ramberg-Bäcklund alkenes) and further chlorination (which leads via an α, α -dichloro sulfone to the salt of an alkenesulfonic acid).¹⁸ With dibenzyl sulfones, 1,3-elimination clearly predominates since they are converted in high yield into stilbenes (equation 5).^{15,19} Di-s-alkyl sulfones give tetrasubstituted alkenes, although these often react further with dichlorocarbene to give dichlorocyclopropanes (equation 6).¹⁵ Fortunately, this side reaction can be suppressed by the addition of a suitable carbene trap such as phenol or an inexpensive nucleophilic alkene. Simple acyclic di-primary-alkyl sulfones normally give alkenesulfonates rather than the Ramberg–Bäcklund products (equation 7),^{17,20} although branching at the β position can divert the course of the reaction to give alkenes (Scheme 4).²¹ Primary- or s-alkyl benzhydryl sulfones give 1,1-diarylalkenes very efficiently, while primary-alkyl benzyl sulfones give mixtures of β -alkylstyrenes and alkenesulfonates.²² Whichever type of sulfone is used as substrate, carbonyl and primary or secondary hydroxy groups must be protected since they are reactive under Meyers' chlorinating conditions. 15, 16, 23

$$Ph \overbrace{O}^{\prime} S \overbrace{O}^{\prime} Ph \qquad \xrightarrow{i \text{ or } ii}_{100\%} \qquad Ph \overbrace{O}^{\prime} Ph \qquad (5)$$

i, KOH, CCl₄, Bu¹OH, H₂O, 60–80 °C, 10–60 min (ref. 15); ii, KOH, CCl₄, CH₂Cl₂, H₂O,

cat. (PhCH₂) $Et_3N^+Cl^-$, 24 h (ref. 24)





i, Na₂S•9H₂O, EtOH, heat; ii, MCPBA, NaHCO₃, CHCl₃; iii, KOH, CCl₄, BuⁱOH, 50 °C

Scheme 4

Meyers' method has been modified to allow it to be operated under phase-transfer conditions.^{14,24} This is a convenient way of converting dibenzyl sulfones directly into (E)-stilbenes (equation 5), but it is not so useful for less reactive substrates.

Vedejs has developed a method for the conversion of α -substituted- α -alkoxycarbonyl sulfones directly into α,β -unsaturated esters, whereby chlorination and Ramberg-Bäcklund rearrangement occur consecutively in the same reaction vessel (equation 8).²⁵ The α -substituent is crucial; without it, the reaction stops after chlorination, presumably because formation of the wrong anion is so strongly favored (equation 9).²⁵ Cyclic α -alkoxycarbonyl sulfones are particularly good substrates for this reaction (equation 10)²⁵ and several examples have been reported.²⁶ In some cases, the rearrangement is slow, and yields are improved if the α -alkoxycarbonyl- α -chloro sulfone is isolated and treated with Bu'OK in a separate step.^{25,27} Although similar to Meyers' variant of the Ramberg-Bäcklund reaction, Vedejs' method is not applicable to simple unactivated sulfones, which are recovered unchanged.



i, excess NaH, Cl₃CCCl₃, DME, 20 °C



i, excess NaH, Cl₃CCCl₃, DME, 20 °C



Sulfones may also be converted directly into alkenes by oxidation of their α, α' -dianions with copper(II) chloride; the intermediate thiirane 1,1-dioxides extrude SO₂ under the reaction conditions (equation 11; compare equation 47).²⁸



i, 5 equiv. BunLi, DME, 0 °C or below; ii, 5 equiv. CuCl₂, r.t.

3.8.5 COMPATIBILITY OF FUNCTIONAL GROUPS

Ramberg-Bäcklund reactions are always performed under basic conditions, and sulfones containing incompatible functional groups must obviously be protected. Carbonyl groups, for example, may be protected as acetals (e.g. Scheme 21). Certain functional groups may undergo 1,2-elimination under the reaction conditions (e.g. equation 24), and if this affects the α -halo sulfone, Ramberg-Bäcklund rearrangement may be prevented from taking place (Scheme 5²⁹ and equation 12³⁰). Further examples of undesirable competing 1,2-elimination are shown in Scheme 18 and equation (25). The α -chloro sulfone (1) undergoes 1,2-elimination when treated with an excess of KOH, but with 1 equiv. an intriguing Ramberg-Bäcklund-like rearrangement involving a 1,7- (instead of the usual 1,3-) elimination step takes place (Scheme 6).^{16,17,31}



3.8.6 MECHANISM OF THE REACTION

It is now widely accepted that α -halo sulfones are converted into alkenes under the basic conditions of the Ramberg-Bäcklund reaction *via* intermediate thiirane 1,1-dioxides (Scheme 7). Experimental data in support of this mechanism were first put forward by Bordwell in 1951,³² and extensive studies carried out since that time, particularly by Bordwell and Paquette in the late 1960s and early 1970s, have substantiated this overall scheme and have provided further insight into the mechanisms of the individual steps. Very little additional work concerning the mechanism of the reaction has been published since 1975. Consequently, only a summary of the essential features is presented here, and readers are referred to the detailed reviews, written by the leaders of the key research groups themselves, for the full data which support the following broad outline.⁴⁻⁶



Scheme 7

The transformation of an α -halo sulfone into an alkene takes place in three steps (Scheme 7). Step 1 involves reversible formation of the α' -anion. Under the basic reaction conditions, there is a rapid equilibration of the α -halo sulfone with its α - and α' -anions. Only the α' -anion leads to step 2 which is intramolecular displacement of a halide ion to give a thirane 1,1-dioxide, generally as a mixture of *cis* and *trans* isomers. This is the rate-limiting step. Finally, in step 3, the thiirane dioxides lose SO₂ to give the stereoisomeric alkenes. This transformation can take place *via* two distinct pathways; one is thermal, while the other involves base catalysis. Both pathways are stereospecific; *trans*- or *cis*-thiirane 1,1-dioxides give only (*E*)- or (*Z*)-alkenes, respectively. Consequently, the stereochemical outcome of the overall reaction is determined at the ring-forming step 2.

Strongest evidence for a rapid equilibration of the α -halo sulfone with its α - and α' -anions comes from reactions performed in D₂O. If such reactions are interrupted before they are complete, the recovered α -halo sulfones are almost fully deuterated at both the α - and α' -positions. Furthermore, the alkene products are also essentially fully deuterated, this time at the vinylic positions. When allowed to run to completion, this forms the basis of a very useful preparative method for deuterium-labeled alkenes (Scheme 8³³ and equation 24). Not only is a high percentage incorporation possible, but the source of deuterium is very cheap.



i, NaOD, D₂O, dioxane, reflux, 72 h; ii, Bu^tOK, THF, reflux, 4 h

Scheme 8

Halogens of α -halo sulfones are remarkably resistant to intermolecular nucleophilic substitution due to a combination of steric effects and the large negative field of the oxygen atoms of the sulfonyl group.³⁴ However, during the Ramberg–Bäcklund reaction, displacement of halide occurs in an intramolecular sense, leading to diastereomeric thiirane 1,1-dioxides. These intermediates are not normally deprotonated under the reaction conditions. However, if one or both of the groups R¹ or R² (Scheme 7) stabilizes an adjacent negative charge, or if a very strong base is used, reversible deprotonation and hence equilibration of the diastereoisomeric thiirane dioxides can occur. Under such circumstances, the ratio of (*E*)- to (*Z*)-alkenes does not reflect the ratio of first-formed thiirane dioxides. Instead, (*E*)-stereoselectivity can be rather high, the simple result of the greater stability of the *trans*-thiirane dioxide intermediate. For example, α , β -unsaturated ketones, sulfones and acids of (*E*)-configuration are formed when the appropriate sulfone precursors are successively brominated and treated with alkoxide (equations 13–15).³⁵

The precise mechanisms by which thiirane 1,1-dioxides are converted stereospecifically into alkenes and SO₂ (or products derived from SO₂) are not clear.^{6,36,37} In typical Ramberg–Bäcklund reactions both thermal and base-catalyzed mechanisms are likely to be in operation, with the latter generally predominating. The importance of the base-catalyzed pathway accounts for the fact that thiirane 1,1-dioxides

$$Ph \longrightarrow S \longrightarrow O \qquad \xrightarrow{i, ii} \\ 68\% \qquad Ph \longrightarrow O \qquad (13)$$

i, Br₂, aq. K₂CO₃, CHCl₃; ii, NaOMe, MeOH, reflux, 0.5 h

$$Ph \overbrace{O'' O O'' O}^{S} \underset{O}{} Ph \xrightarrow{i, ii}_{98\%} Ph \xrightarrow{O}_{S} \underset{O''}{} Ph$$
(14)

i, Br₂, aq. K₂CO₃, CHCl₃; ii, NaOMe, MeOH, reflux, 0.5 h

$$Ph \longrightarrow CO_2Et \xrightarrow{i-v} Ph \longrightarrow CO_2H$$
(15)

i, NaH, benzene, reflux, 10 min; ii, Br_2, benzene, 10 °C; iii, NaOEt, EtOH, reflux, 0.5 h;

iv, KOH, EtOH, reflux, 10 min; v, H₃O⁺

have never been isolated from Ramberg-Bäcklund reactions. Nevertheless, they can be prepared by quite independent methods,³⁸ and it is samples obtained in this way which have allowed their properties, such as the stereospecificity of their decomposition to alkenes, to be established.

Finally, no definitive mechanistic explanation has been given for one of the most intriguing aspects of the Ramberg-Bäcklund reaction, the fact that acyclic α -halo sulfones consistently give products rich in the (Z)-isomer.^{4,6}

3.8.7 RELATED REACTIONS

3.8.7.1 Replacement of the Halide

Variations of the Ramberg-Bäcklund reaction have been described in which the usual halide ion is replaced by *p*-toluenesulfonate or *p*-toluene-, alkane- or trifluoromethane-sulfinate leaving groups. For example, Meyers has described a single transformation of the Ramberg-Bäcklund type of an α -tosyloxy sulfone (Scheme 9).³⁹ The reaction is surprisingly slow, nearly 1000 times slower than that of the corresponding α -chloro sulfone under the same conditions, and this, of course, limits its value in synthesis. Another problem is the fact that the precursors are not easy to prepare.

An example of sulfinate functioning as a leaving group is shown in equation (16),⁴⁰ while another forms the basis of a useful synthesis of 3-cyclopentenones (Scheme 22). Zwanenburg has reported a transformation related to the Ramberg-Bäcklund reaction in which the halide leaving group is replaced by sulfinate and, in addition, the usual sulfone is replaced by sulfoxide.⁴¹



i, 10% aq. NaOH, 250 °C, 24 h (33%); ii, LiAlH₄, Et₂O, 24 h (61%)

More useful from a synthetic point of view than any of these variations is the discovery by Hendrickson that trifluoromethanesulfinate ('triflinate') is a very effective leaving group in Ramberg-Bäcklund reactions.^{42–44} The reaction succeeds only if the precursors, α -triflyl sulfones, are fully substituted at the α -position; if they are not, simple deprotonation occurs under the basic reaction conditions because the triflyl group is so strongly electron withdrawing. Consequently, this approach is applicable only to the synthesis of 1,1-di-, tri- and tetra-substituted alkenes.

The precursors for this reaction can be prepared by multiple alkylations of mesyltriflone (2); this can be carried out with a high degree of regiocontrol and, in fact, is facile enough to allow several successive operations in one vessel (Scheme 10).⁴⁴ The α -proton of mesyltriflone is so acidic that treatment with

1 equiv. of base (e.g. K_2CO_3) leaves an α -monoanion which is very stable and difficult to alkylate. Interestingly, a second equivalent of base (usually BuⁿLi) abstracts the second α -hydrogen in preference to an α' -hydrogen to give an α, α -dianion which can be monoalkylated without danger of over reaction. If required, further successive treatments with BuⁿLi and alkyl halides can be used to introduce either one or two alkyl groups at the α' -position, the α -anion protecting the molecule from premature Ramberg-Bäcklund rearrangement when the α' -anions are formed. Alternatively, the α, α' -dianions can be acylated at the α' -position in high yield by reaction with acid chlorides, anhydrides, esters or chloroformates. (In these reactions, an extra equivalent of base, typically LDA, is added since the β -keto sulfone products are more acidic than the starting sulfones.) Finally, the second α -position is alkylated (this can be performed only with very reactive alkylating agents and requires elevated temperatures), and treatment with another equivalent of base triggers the Ramberg-Bäcklund rearrangement. Optimum conditions depend on the nature of the substrate. For simple alkylated compounds, Bu'OK is generally the base of choice (equation 17). Formation of cyclic disulfones as by-products can be a problem, especially with trialkyl compounds (equation 18).⁴⁴ For α' -acylated substrates, the reaction proceeds smoothly on heating with K_2CO_3 (e.g. Scheme 11, which shows a synthesis of dihydrojasmone),⁴³ or on treatment with sodium hydroxide under phase-transfer conditions (e.g. Scheme 12, which shows the simplest reported synthesis of artemisia ketone).44




 $\mathbf{R} = n$ -pentyl

i, 2 equiv. $Bu^{n}Li$, THF, -78 °C; ii, H₂C=CHCHO; iii, CrO₃, aq. acetone, 0 to 5 °C; iv, 2 equiv. K₂CO₃, THF,

reflux, 3 to 5 h

Scheme 11



As usual, no significant stereocontrol is observed with these transformations. The alkylation of the mesyltriflone (2), although highly regioselective, generally gives diastereoisomeric mixtures of products, which are converted during the Ramberg-Bäcklund reaction into (E)/(Z)-mixtures of alkenes. Consequently, the sequence is most useful for the synthesis of compounds where stereoisomers are not possible, such as 1,1-disubstituted alkenes and five- or six-membered cycloalkenes.

3.8.7.2 Replacement of the Sulfone Group

In simple terms, the function of the sulfone group in Ramberg–Bäcklund reactions is first to stabilize an α -anion and then, following 1,3-elimination of halide, to be extruded efficiently from the intermediate thiirane 1,1-dioxide to yield an alkene. In fact, the sulfone group is not unique in its ability to perform these tasks: transformations related to the Ramberg–Bäcklund reaction have been described in which sulfide, sulfoxide, sulfoximine, phosphine oxide,⁴⁵ or phosphinate⁴⁶ groups replace the familiar sulfone group. On occasions, even α -halo ketones can be diverted from their usual Favorskii rearrangement to give alkenes instead of carboxylic acids. Reactions which are useful for the synthesis of alkenes are discussed in more detail in the following paragraphs.

Mitchell has reported that benzyl α -chlorobenzyl sulfides are cleanly converted into stilbenes on successive treatment with PPh₃ and Bu⁴OK (Scheme 13).⁴⁷ Desulfurization of an intermediate thiirane by PPh₃ is a likely pathway for this reaction. Benzyl α,α -dichlorobenzyl sulfides are converted into diphenylacetylenes under the same conditions.⁴⁸ Viehe has reported that pyrolysis of α -halo- α -alkylthio esters, ketones and nitriles under reduced pressure furnishes α,β -unsaturated esters, ketones and nitriles respectively; it is probable that these reactions also take place *via* intermediate thiiranes.⁴⁹ α -Chlorophenacyl phenacyl sulfide is converted into (*E*)-dibenzoylethylene on treatment with Et₃N.⁵⁰



Cyclobutenes can sometimes be prepared more efficiently from five-membered cyclic α -chloro sulfoxides rather than from the corresponding sulfones (*e.g.* Scheme 17).

Johnson has shown that α -halo-*N*-tosylsulfoximines with an α' -hydrogen atom, on treatment with KOH in refluxing methanol, undergo rearrangement to alkenes by analogy with the Ramberg-Bäcklund

reaction.⁵¹ Surprisingly, alkenes are not formed from the corresponding N-H- or N-methyl-sulfoximines under these conditions. Two types of N-tosylsulfoximine were examined. The first, alkyl- α -chloroalkylsulfoximines, give alkenes as approximately 1:1 mixtures of geometric isomers (equation 19). By contrast, alkyl- or benzyl- α -bromobenzylsulfoximines give predominantly (Z)-alkenes, together with significant quantities of sulfoximines resulting from debromination (equation 20). This (Z)-stereoselectivity probably stems from equilibration via deprotonation of the intermediate 2-phenylepisulfoximines in favor of the isomer in which substituents at the 2- and 3-positions are both trans to the bulky tosyl group.



Under basic conditions, α -halo ketones with an α' -hydrogen atom generally undergo Favorskii rearrangement via cyclopropanones to carboxylic acids. It is clear that while the ring closure step of this transformation resembles that of the Ramberg-Bäcklund reaction, the subsequent ring-opening step does not. However, under certain conditions and with particular substrates, α -halo ketones can be redirected to give alkenes via (in some cases formal) loss of carbon monoxide from the intermediate cyclopropanones. An isolated example is shown in equation (21).⁵² In this case, the intermediate cyclopropanone reacts with hydroperoxide anion to form an adduct which collapses to the alkene with loss of carbon dioxide and hydroxide ion; the Favorskii acid is a by-product. Far more useful is a preparative method for 2cyclopentenones based on the ring contraction of 2-chloro-1,3-cyclohexanediones which was discovered by Büchi (Scheme 14)⁵³ and has since been applied on numerous occasions.⁵⁴ The success of this reaction depends on the use of a non-nucleophilic base, which avoids simple dechlorination of the substrate; a systematic survey showed that Na₂CO₃ is the reagent of choice.



i, Bu^tOCl, CHCl₃, -15 °C; ii, Na₂CO₃, boiling xylene, 12 h

Scheme 14

3.8.8 SYNTHESIS OF CYCLOALKENES

3.8.8.1 Cyclobutenes

In spite of an unpromising early report,⁵⁵ the Ramberg-Bäcklund rection has been applied with great success since the early 1970s to the synthesis of cyclobutenes, particularly by Paquette and Weinges. Al-

though yields are often only moderate or low, the reaction still compares favorably with other possible approaches to these strained compounds.

The following general remarks may be made concerning the use of the Ramberg-Bäcklund reaction to prepare cyclobutenes. It is usually performed by treating α -chloro sulfones with an excess of Bu'OK in THF or, less commonly, in dioxane. These precursors are almost always prepared by successive chlorination and oxidation of the corresponding substituted tetrahydrothiophenes. Since the intermediate α -chloro sulfides and α -chloro sulfones are frequently formed as mixtures of regio- and stereo-isomers, it is convenient to use them in a crude form, and to withhold purification until after the Ramberg-Bäcklund reaction itself. The reaction is particularly suitable for the synthesis of cyclobutenes which are unsubstituted at the vinylic positions and fully substituted at the 3- and 4-positions, and the products of the reactions in the following examples are mostly compounds of this type. Meyers' method for the direct conversion of sulfones into alkenes appears not to have been applied to the synthesis of cyclobutenes.²⁴

A Diels-Alder reaction with a substituted maleic anhydride followed by a Ramberg-Bäcklund rearrangement are key steps in a reaction sequence which Paquette has used frequently for the synthesis of fused cyclobutenes.⁵⁶⁻⁵⁸ An example is shown in Scheme 15.⁵⁸ Weinges has prepared 1,4-bridged Dewar benzenes using double Ramberg-Bäcklund reactions which introduce both double bonds simultaneously (Scheme 16).⁵⁹ Although yields are low, these compounds are difficult to prepare by other methods. The interesting tetracyclic hydrocarbon pterodactyladiene was prepared for the first time using another double Ramberg-Bäcklund reaction (equation 22).⁶⁰



i, isoprene, hydroquinone, dioxane, 170 °C, 68 h; ii, LiAlH₄, THF; iii, MeSO₂Cl, Py, 0 °C; iv, Na₂S, HMPA, 130 °C, 20 h; v, NCS, CCl₄, reflux, 1 h; vi, monoperphthalic acid, Et₂O, -78 to 0°C; vii, Bu^tOK, THF, -78 to 0 °C

Scheme 15



i, NCS, CCl₄, 0 °C, 4 h; ii, monoperphthalic acid, Et₂O, r.t., 2 to 3 d; iii, Bu^tOK, THF: for n = 4, -15 °C, 24 h; for n = 5, r.t., 18 h, then reflux, 2 h

Scheme 16



Weinges has discovered that, on occasions, cyclobutenes can be prepared far more efficiently using α chloro sulfoxides rather than the conventional α -chloro sulfones as precursors (Scheme 17).⁶¹ The course of these reactions is analogous to the usual Ramberg–Bäcklund reaction, proceeding via base-catalyzed or thermal elimination of sulfur monoxide from thiirane 1-oxide intermediates.³⁷ In fact, Weinges was able to isolate such intermediates from some reactions and these were converted into cyclobutenes by treatment with Bu¹OK in refluxing THF⁶² or with LAH in THF at 55 °C.⁶³ It is not clear why sulfoxides give better yields than sulfones in these reactions.



i, Bu^tOK, THF, -70 °C to r.t. over 2 h, then 55 °C, 2 h

Scheme 17

As the preceding examples illustrate, most successful applications of the Ramberg-Bäcklund reaction to the synthesis of cyclobutenes involve as precursors 2-halotetrahydrothiophene 1,1-dioxides which are fully substituted at the 3- and 4-positions. Where this is not the case, simple 1,2-elimination of hydrogen halide can compete with the required 1,3-elimination. For example, the simple substrate (3) is converted into 1,2-dimethylenecyclobutane (4), the product of 1,2-elimination of HCl, rather than the product expected from Ramberg-Bäcklund rearrangement (Scheme 18).⁶⁴ Nevertheless, on treatment with Bu'OK, either of the epimeric α -bromo sulfones (5) and (6) are converted with surprising efficiency into the highly strained cyclobutene (7). It is postulated that competing 1,2-elimination of HBr does not occur in these instances because of unfavorable dihedral angles.⁶⁵



i, Bu^tOK, THF, 5 h at -75 °C then warm to room temperature

Cyclobutenes prepared by the Ramberg–Bäcklund reaction are almost always unsubstituted at the vinylic positions. In 1974, Paquette introduced a related and complementary ring contraction which is particularly suitable for the synthesis of 1,2-dialkylcyclobutenes.^{33,57,66,67} The reaction takes place when carbanions of five-membered cyclic sulfones are treated with LAH in refluxing dioxane (Scheme 19).⁶⁶ The mechanism of the reaction is not clear. Good yields are obtained only when the sulfone is 2,5-dialkylated; in the absence of these groups, simple reduction of sulfone to sulfide becomes a serious side reaction. Dibenzyl sulfone is converted into a mixture of stereoisomeric stilbenes (56%) and dibenzyl sulfide (23%) under the same conditions.⁶⁶



i, 2 equiv. BuⁿLi, THF, -80 °C; ii, excess MeI; iii, BuⁿLi, dioxane, 0 °C; iv, LiAlH₄, dioxane, reflux, 6 h

Scheme 19

3.8.8.2 Cyclopentenes, Cyclopentadienes and Cyclopentenones

Since the late 1970s, the Ramberg-Bäcklund reaction has been used to convert a wide variety of sixmembered cyclic α -halo sulfones into cyclopentenes, often in high yield; and useful regioselective syntheses of both 2- and 3-cyclopentenones, using sequences in which the Ramberg-Bäcklund reaction is the key step, have also been reported. Bu'OK in THF, diethyl ether or DME are the preferred conditions for these reactions and, in addition, Meyers' method has been used to convert six-membered cyclic sulfones directly into cyclopentenes. Examples of these reactions follow.

Weinges has used the reaction to prepare the stereochemically pure intermediates (8), which are useful for the synthesis of cyclopentenoid natural products (equation 23).⁶⁸ In contrast to his observations when preparing cyclobutenes (*e.g.* Scheme 17), he found that in these cases, α -chloro sulfoxides give only low yields of cyclopentenes and numerous by-products on treatment with Bu'OK. The Ramberg–Bäcklund reaction has been used to prepare (–)-(*S*)-3-methylcyclopentene (9; Scheme 20) as well as its enantiomer.⁶⁹ Although the yield is low, this method is very attractive because the product is free from double bond regioisomers. Paquette treated the trihalo sulfone (10) with an excess of Bu'OK at low temperatures in a mixture of THF and D₂O to perform a remarkable simultaneous double dehydrobromination and Ramberg–Bäcklund reaction with deuterium incorporation to give the unstable tricyclodecatriene (11; equation 24).⁷⁰ Bicyclo[2.1.1]hexenes, which, though less elaborate, are related compounds, can be made from 1,3-cyclobutanedicarboxylic acids in a sequence incorporating the Ramberg–Bäcklund reaction.⁷¹ In rigid fused-ring systems, unfavorable geometry can lead to 1,2-elimination of hydrogen halide to give a vinyl sulfone (equation 25) instead of the Ramberg–Bäcklund alkene (equation 26).⁶⁵



Scheme 20





i, Bu^tOK, THF, -15 °C, 6 h, then r.t., 18 h



i, Bu^tOK, THF, -15 °C, 6 h, then r.t., 18 h

The Ramberg-Bäcklund reaction has also been applied to the synthesis of cyclopentadienes. For example, the α -chloro sulfone (12), a single unidentified diastereoisomer, was converted regiospecifically into the cyclopentadiene (13).⁷² In principle, the acidity of the product requires the use of at least 2 equiv. of base for optimum yields, but, in practice, more than 1 equiv. of BuⁿLi causes appreciable polymerization.



Matsuyama and Taylor have reported that the Ramberg–Bäcklund reaction may be used to prepare a variety of 2-substituted-3-cyclopentenones (Scheme 21),^{73,74} 3-substituted-3-cyclopentenones (Scheme 22)^{74,75} and 2,3-disubstituted-2- or -3-cyclopentenones (Scheme 23).⁷⁶ De Waard has applied similar chemistry to the synthesis of the D-ring of a steroid.⁷⁷ In each case, the carbonyl group is protected as an acetal during the ring contraction, and careful hydrolysis is essential if migration of the carbon–carbon double bond into conjugation with the carbonyl group is to be avoided. If 2-cyclopentenones are required, hydrolysis and migration can be performed simultaneously under more strongly acidic conditions (Scheme 23). Other noteworthy features of these reactions are the use of sulfones with Meyers' one-pot basic chlorination conditions in the synthesis of 2-substituted-3-cyclopentenones (Scheme 21) and the use of 4-toluenesulfinate instead of the usual halide leaving group during the synthesis of 3-substituted-3-cyclopentenones (Scheme 22). 2-Cyclopentenones, as well as cyclopentenes, can be prepared very efficiently using Hendrickson's variant of the Ramberg–Bäcklund reaction in which trifluoromethanesulfinate functions as the leaving group (*e.g.* Scheme 11). Furthermore, 2-cyclopentenones can be prepared by ring-contraction of 2-chloro-1,3-cyclohexanediones (*e.g.* Scheme 14).



i, Bu^tOK, CCl₄, Bu^tOH, 50 °C, 20 h; ii, p-TsOH•Py, aq. acetone, reflux, 22 h

Scheme 21



R = alkyl

i, 2.5 to 3.0 equiv. NaH, 0.1 equiv. KH, DMSO, 20 to 30 °C, 24 h; ii, p-TsOH•Py, aq. acetone, reflux

Scheme 22



i, Me₃SiI, HOCH₂CH₂OH; ii, Bu^tOK, -78 °C, 0.5 h; iii, p-TsOH•Py, aq. acetone; iv, 5% aq. HCl

Scheme 23

3.8.8.3 Cyclohexenes and Phenanthrenes

The Ramberg–Bäcklund reaction has not been applied frequently to the synthesis of six-membered rings, but it has nevertheless proved useful where the target compounds are strained. Gassman chose to use the reaction to prepare *trans*-bicyclo[4.1.0]hept-3-ene (14), a strained hydrocarbon which slowly isomerizes to the more stable *cis* isomer at 120 °C (Scheme 24).⁷⁸ The 7-methyl⁷⁹ and 7,7-dimethyl derivatives⁸⁰ were prepared by similar methods (with yields of 20% and 45% respectively at the Ramberg–Bäcklund reaction step). Cyclohexenes can be prepared by Hendrickson's variant of the Ramberg–Bäcklund reaction in which trifluoromethanesulfinate functions as the leaving group.⁴⁴



i, Zn-Cu, CH₂I₂; ii, LiAlH₄; iii, TsCl, Py; iv, LiBr; v, Na₂S, high dilution; vi, NCS, CCl₄, reflux; vii, MCPBA, CH₂Cl₂, 0 to 25 °C; viii, Bu^tOK, Et₂O, 0 °C

Scheme 24

In 1964, Paquette reported that although the 9,10-double bond of simple phenanthrenes can be formed using the Ramberg-Bäcklund reaction (NaOH in aqueous dioxane), the sterically crowded 4,5-dimethylphenanthrene cannot be made by this method.⁸¹ Staab has now shown that the highly strained 'proton sponge' 4,5-bis(dimethylamino)phenanthrene (15) can be prepared using Meyers' modification of the reaction, although over-chlorination under these conditions necessitates an extra reduction step (Scheme 25).⁸²



i, KOH, CCl₄, Bu^tOH, 0 to 20 °C, 1 h; ii, Li, THF, -55 to -35 °C, 1.5 h; iii, MeOH

Scheme 25

3.8.8.4 Large Ring Systems

The Ramberg–Bäcklund reaction has been used to synthesize unsaturated cyclophanes and related bridged systems. However, it does not always compare favorably with the alternative methods for the conversion of similar precursors into the same products, namely: (i) the Wittig rearrangement of sulfides followed by repeated methylation and Hofmann elimination;⁸³ (ii) the Stevens rearrangement of sulfur ylides followed by quaternization and Hofmann elimination;⁸⁴ or (iii) the benzyne–Stevens rearrangement of sulfur solution is the preparation of the appropriate α -halo sulfones; the precursors are frequently benzylic sulfones and it has proved difficult to oxidize these to α -halo sulfones without overhalogenation. When α -halo sulfones can be obtained, the Ramberg–Bäcklund reaction itself takes place in moderate yield (Schemes 26⁸⁶ and 27⁸⁷). One way of circumventing these problems is to allow sulfones to react under Meyers' conditions (equation 27).⁸⁸ Yields are sometimes low, and over-chlorination can still occur (leading, for example, after dehydrochlorination, to alkynes; equation 28),⁸⁹ but despite these drawbacks, the ability of the reaction to introduce a double bond regiospecifically can be extremely valuable. For example, the [3.3]paracyclophanediene (16), prepared in just 2% yield by the



n = 7; 55% n = 7; 55%

i, NCS, CCl₄; ii, MCPBA, CH₂Cl₂; iii, 4 equiv. Bu¹OK, DME

Scheme 26



i, H₂O₂, cat. tungstic acid, aq. THF, 50 to 60 °C, 1 h; ii, BuⁿLi, THF, -78 °C; iii, Br₂, -78 °C; iv, NaOH, aq. dioxane, r.t., 2 h

Scheme 27



Ramberg-Bäcklund reaction (equation 29),⁹⁰ can be prepared in higher yield using the benzyne-Stevens rearrangement, but, when this approach is used, it is always contaminated with an equal amount of the inseparable regioisomeric [3.3]paracyclophanediene.

3.8.9 SYNTHESIS OF CONJUGATED DIENES, ENVNES AND POLYENES

In 1975, it was reported that treatment of each of the stereoisomeric α,β -unsaturated γ -bromo sulfones (17) and (18) with Bu'OK leads *via* a vinylogous Ramberg–Bäcklund reaction to the same mixture of stereoisomeric 1,3-dienes (19).⁹¹ More recently, Block has demonstrated that α,β -unsaturated α' -bromo sulfones are alternative and versatile precursors for vinylogous Ramberg–Bäcklund reactions, and he has developed this concept into an extremely useful method for the synthesis of dienes and polyenes. Examples are shown in equations (30)–(32).^{92–95} The (Z)-stereoselectivity of the transformation shown in equation (30) stems from an attractive interaction between the developing negative charge at the α -position and the methylene group at the δ -position of the (E)-sulfone (20), which favors the transition state leading to the (Z)-isomer. With the corresponding (Z)-sulfone (21); steric effects disfavor the transition state which leads to the (Z)-isomer and the (E)-diene is the predominant product (equation 31). In a more extreme case (equation 32), the (E)-isomer is the exclusive product.





i, 2.5 equiv. Bu^tOK, Bu^tOH:THF (7:3), -20 °C, 1 h



>99% (E)-isomer [58% yield from (E)-3-hexene]

An attractive feature of Block's vinylogous Ramberg-Bäcklund reaction is that the precursors are easily prepared by the addition of α -haloalkanesulfonyl bromides to alkenes followed by elimination of the elements of HBr (Scheme 28). The addition, which takes place for a wide range of alkenes, is almost certainly a free radical chain reaction and is regiospecific for mono-, 1,1-di- and tri-substituted alkenes, with the formation of products consistent with the addition of the sulfonyl radical first to give the more stabilized radical intermediate. Even unsymmetrical 1,2-disubstituted alkenes show a high degree of regioselectivity, and in the case of compounds with both 1,1-di- and tri-substituted double bonds, addition occurs exclusively at the less-hindered disubstituted double bond. The general procedure involves treatment of the crude addition product with Et₃N in CH₂Cl₂ at 0 °C, which gives α , β -unsaturated α' -bromo sulfones as stereoisomeric mixtures in which the (E)-isomer predominates.^{92,94} Taken sequentially, these addition and elimination steps and the vinylogous Ramberg-Bäcklund reaction allow an alkene to be converted into a 1,3-diene. Due to incomplete stereocontrol, during both the elimination and the rearrangement, the sequence is most useful where geometric isomers cannot occur, or for specific classes of compounds where high stereoselectivity has been observed. Several examples are shown in equations (33)-(38).^{93,94} In each case, the initial alkene was treated in three successive steps with BrCH₂SO₂Br, Et₃N and Bu¹OK without separating stereoisomeric intermediates. Particularly noteworthy are the transformations shown in equations (33) and (34), which illustrate how the sequence allows a methylene group to be attached to interior atoms in chains or rings. 1-Methylcyclohexene gives 1,2-dimethylenecyclohexane as the predominant product (equation 36),96 indicating that Bu'OK preferentially abstracts a proton from the methyl rather than the methylene position of the intermediate unsaturated sulfone. Block has studied the regioselectivity of this deprotonation with Bu'OK and Bu'OLi for a selection of substrates. His results can be rationalized in terms of two effects: (i) abstraction at the less-hindered position is favored with both bases; and (ii) coordination of the cations by sulfonyl oxygen favors deprotonation syn to the sulfonyl group, and this effect is far more significant with lithium than potassium. Examples are shown in equations (39) and (40).^{94,97}

Conjugated trienes can also be synthesized *via* the vinylogous Ramberg-Bäcklund reaction (Scheme 29).⁹⁴ However, it is not possible to convert conjugated trienes into conjugated tetraenes using Block's procedure: conjugated trienes fail to give adducts with BrCH₂SO₂Br, probably because the intermediate



X = Cl, Br, I; R¹ = alkyl, allyl, benzyl, etc.; R² = H, Me

Scheme 28

879

Rearrangement Reactions



pentadienyl radical is too stable to abstract a bromine atom from BrCH₂SO₂Br. Nevertheless, conjugated tetraenes can be made by the alternative procedure shown in Scheme 30, a process which involves a Ramberg–Bäcklund rearrangement through three intervening double bonds! Furthermore, terminal or internal alkynes are converted into 1,3-enynes under appropriate conditions (equations 41 and 42).⁹⁴

Block has described a further use for α -haloalkanesulfonyl bromides; radical addition to silyl enol ethers followed by simple Ramberg-Bäcklund rearrangement yields α,β -unsaturated ketones (Scheme 31).^{94,98} Ethylene oxide, an acid scavenger, is used as solvent for the radical addition to prevent hydrolysis of the silyl enol ether.

De Waard has developed a modification of the Ramberg-Bäcklund rearrangement in which the intermediate α -sulfonyl carbanion is generated by 1,4-addition of a nucleophile to a vinyl sulfone: the 'Mi-



Scheme 29





i, BrCH₂SO₂Br, ethylene oxide, hv, -15 °C; ii, DBN, CH₂Cl₂, -78 °C, 2 h, then 23 °C, 0.5 h

Scheme 31

chael-induced Ramberg-Bäcklund reaction'.⁹⁹ Scheme 32 shows this transformation in its simplest form, though in practice it appears that butadienyl sulfones have been used universally, perhaps because the equilibrium shown does not favor the sulfone α -anion with simple vinyl sulfones (compare Scheme 5). Examples from de Waard⁹⁹ and from Block⁹⁴ are shown in equations (43) and (44), respectively.

$$\sum_{\substack{O' \in O}} S \xrightarrow{Nu^{-}} \left[\begin{array}{c} Nu \xrightarrow{-} S \xrightarrow{O} S \xrightarrow{O} X \right] \xrightarrow{-SO_{2}} Nu \xrightarrow{-X} Nu \xrightarrow{-X} Scheme 32$$

$$X = Cl, Br; Nu^{-} = RSO_{2}^{-}, RO^{-}$$
Scheme 32
$$\sum_{\substack{O' \in O}} S \xrightarrow{O} Cl \xrightarrow{PhSO_{2}Na, DMSO, \\ r.t., 1 h \\ 44\% \qquad (E):(Z) = 2:1 \end{array}$$

$$(43)$$

$$\sum_{\substack{O' \in O}} S \xrightarrow{O} Br \xrightarrow{Pr^{i}ONa, Pr^{i}OH, \\ r.t., 1 h \\ 56\% \qquad (E):(Z) = 3:1 \end{aligned}$$

Rearrangement Reactions

De Waard has focused on the use of the reaction for the synthesis of conjugated isoprenoids (Scheme 33);¹⁰⁰ the regioisomeric sulfinates (22) and (23) or their stereoisomers function as the key five-carbon building blocks. Reaction of the sulfinate (22) with prenyl chloride followed by chlorination¹⁰¹ gives the α -chloro sulfone (24), precursor for the Michael-induced reaction. Further treatment with the sulfinate (22) then gives the stereoisomeric 'head-to-tail' linked pentaenes (25), ready for rechlorination and for the cycle to be repeated. Use of the alternative isoprene synthon (23) results in 'tail-to-tail' coupling of the isoprene units. When the polyisoprenoid chain reaches the required length, phenylsulfinate is allowed to trigger the rearrangement and gives an allyl phenyl sulfone, *e.g.* the conversion of (24) into the stereo-isomeric trienes (26), from which the phenylsulfonyl group may be removed by reduction.



i, prenyl chloride, DMSO, r.t., 40 h; ii, LDA, THF, -78 °C; iii, Cl₃CCCl₃, -78 °C to r.t.; iv, (22), DMSO, r.t., 68 h; v, PhSO₂Na, DMSO, r.t., 0.5 h

Scheme 33

The least attractive feature of the Michael-induced Ramberg-Bäcklund reaction is the absence of any useful stereocontrol; where several isomeric products are possible, they are all formed. In fact, both stereoisomers of the α -chloro sulfone (24) give the same mixture of trienes (26) on treatment with sodium phenylsulfinate.¹⁰⁰ The reaction is presumably most useful if the required isomer can be obtained by equilibration of the complex mixture in a subsequent step.

Bisallylic sulfones are converted into conjugated trienes under Meyers' conditions;¹⁰²⁻¹⁰⁴ an example is shown in equation (45).¹⁰² Although experiments with isomerically pure substrates have not been performed, there are indications that the geometry of the allyl groups is conserved during the rearrangement while the newly introduced double bond has mainly the (*E*)-configuration.¹⁰⁴ Meyers' conditions also allow unsaturated β -sulfonyl esters such as (27) to be converted into (*E*,*E*)-dienoic acids, hydrolysis accompanying the rearrangement (equation 46; compare equation 4). An all-(*E*)-trienoic acid was made by the same method.¹⁰⁵



β-Carotene has been prepared by treatment of an appropriate sulfone α, α' -dianion with bromine or iodine (equation 47).¹⁰² This transformation may be regarded either as proceeding through a conventional α-halo sulfone α' -anion, or *via* a two-electron oxidation to the intermediate thiirane 1,1-dioxide (compare equation 11).



i, 2 equiv. BuⁿLi, THF, 0 °C; ii, I₂ or Br₂; iii, thermal or I₂-catalyzed stereomutation

The reactive cyclodecenediyne ring system found in the calicheamicins has recently been synthesized for the first time, together with a homologous series of larger (11- to 16-membered) rings containing the same enediyne unit.¹⁰⁶ The synthesis of the first derivative (**28**) with DNA-cleaving properties by the same approach (equation 48) was reported soon afterwards.¹⁰⁷



i, 1.2 equiv. MeLi, Et₂O, -78 °C, 15 min (20% yield); ii, Buⁿ₄NF, THF, 1 h (84% yield)

3.8.10 SYNTHESIS OF ALKYNES

On treatment with appropriate tertiary amines, both α, α^{-108} and α, α' -dihalodibenzyl sulfones¹⁰⁹ are converted into diphenylthiirene 1,1-dioxides which, although far more stable than the corresponding thiirane dioxides, extrude SO₂ on heating to give diphenylacetylenes (Scheme 34).³⁷ The highly strained alkyne acenaphthyne, isolated in the form of its trimer, has been prepared in this way.¹¹⁰ α -Chloroneopentyl neopentyl sulfone reacts under Meyers' basic chlorinating conditions (KOH, CCl₄, Bu⁴OH) at above 40 °C to give di-*t*-butylacetylene in yields of up to 90%.^{16,17}



i, Et₃N, CH₂Cl₂, reflux, 3 h; ii, triethylenediamine, DMSO, r.t.; iii, boiling benzene, several hours, or melt

Scheme 34

3.8.11 REFERENCES

- 1. L. Ramberg and B. Bäcklund, Ark. Kemi, Mineral. Geol., 1940, 13A, No. 27, 1 (Chem. Abstr., 1940, 34, 4725).
- 2. P. Magnus, Tetrahedron, 1977, 33, 2019; T. Durst, in 'Comprehensive Organic Chemistry', ed. D. H. R. Barton and W. D. Ollis, Pergamon Press, Oxford, 1979, vol. 3, p. 187; E. Vedejs and G. A. Krafft, Tetrahedron, 1982, 38, 2857.
- 3. S. Oae and Y. Uchida, in 'The Chemistry of Sulphones and Sulphoxides', ed. S. Patai, Z. Rappoport and C. J. M. Stirling, Wiley, New York, 1988, chap. 12, p. 583; S. Braverman, in 'The Chemistry of Sulphones and Sulphoxides', ed. S. Patai, Z. Rappoport and C. J. M. Stirling, Wiley, New York, 1988, chap. 13, p. 665; F. S. Guziec, Jr. and L. J. Sanfilippo, Tetrahedron, 1988, 44, 6241.
- 4. L. A. Paquette, Org. React. (N.Y.), 1977, 25, 1.
- 5. F. G. Bordwell, in 'Organosulfur Chemistry', ed. M. J. Janssen, Wiley, New York, 1967, chap. 16, p. 271.
- L. A. Paquette, Acc. Chem. Res., 1968, 1, 209.
 K. Schank, in 'The Chemistry of Sulphones and Sulphoxides', ed. S. Patai, Z. Rappoport and C. J. M. Stirling, Wiley, New York, 1988, chap. 7, p. 165.
- 8. B. M. Dilworth and M. A. McKervey, Tetrahedron, 1986, 42, 3731.
- 9. D. Scholz, Chem. Ber., 1981, 114, 909.
- 10. D. Scholz, Sci. Pharm., 1984, 52, 151.
- 11. D. Scholz and P. Burtscher, Liebigs Ann. Chem., 1985, 517.
- 12. D. Scholz, Liebigs Ann. Chem., 1984, 264.
- 13. D. Scholz, Liebigs Ann. Chem., 1983, 98.
- 14. G. D. Hartman, and R. D. Hartman, Synthesis, 1982, 504.
- 15. C. Y. Meyers, A. M. Malte and W. S. Matthews, J. Am. Chem. Soc., 1969, 91, 7510; C. Y. Meyers, A. M. Malte and W. S. Matthews, Q. Rep. Sulfur Chem., 1970, 5, 229.
- 16. C. Y. Meyers, W. S. Matthews, L. L. Ho, V. M. Kolb and T. E. Parady, in 'Catalysis in Organic Syntheses', ed. G. V. Smith, Academic Press, New York, 1977, p. 197.
- 17. C. Y. Meyers, in 'Topics in Organic Sulphur Chemistry', ed. M. Tisler, University Press, Ljubljana, 1978, p. 207.
- 18. C. Y. Meyers, L. L. Ho, G. J. McCollum and J. C. Branca, Tetrahedron Lett., 1973, 1843.
- 19. O. Dann, H. Char and H. Griessmeier, Liebigs Ann. Chem., 1982, 1836.
- 20. C. Y. Meyers and L. L. Ho, Tetrahedron Lett., 1972, 4319.
- 21. L. A. Paquette, G. J. Wells and G. Wickham, J. Org. Chem., 1984, 49, 3618; for a related example see I. V. Kazimirchik, K. A. Lukin, G. F. Bebikh and N. S. Zefirov, Zh. Org. Khim., 1983, 19, 2523 (Chem. Abstr., 1984, 100, 174 296z).
- C. Y. Meyers, W. S. Matthews, G. J. McCollum and J. C. Branca, *Tetrahedron Lett.*, 1974, 1105.
 C. Y. Meyers and V. M. Kolb, *J. Org. Chem.*, 1978, 43, 1985.
- 24. S. E. Lauritzen, C. Rømming and L. Skattebøl, Acta Chem. Scand., Ser. B, 1981, 35, 263; for a further example see ref. 14.
- E. Vedejs and S. P. Singer, J. Org. Chem., 1978, 43, 4884.
 E. Vedejs, M. J. Arnost and J. P. Hagen, J. Org. Chem., 1979, 44, 3230.
- 27. V. Ceré, C. Paolucci, S. Pollicino, E. Sandri and A. Fava, J. Org. Chem., 1981, 46, 486.
- 28. J. S. Grossert, J. Buter, E. W. H. Asveld and R. M. Kellogg, Tetrahedron Lett., 1974, 2805.
- J.-H. Fuhrhop, U. Liman and V. Koesling, J. Am. Chem. Soc., 1988, 110, 6840.
 J.-M. Gaillot, Y. Gelas-Mialhe and R. Vessiere, Can. J. Chem., 1979, 57, 1958; for a related reaction see L. A. Paquette and R. W. Houser, J. Org. Chem., 1971, 36, 1015.
- 31. C. Y. Meyers and D. H. Hua, Phosphorus Sulfur, 1979, 6, 197.
- F. G. Bordwell and G. D. Cooper, J. Am. Chem. Soc., 1951, 73, 5187.
 L. A. Paquette, R. E. Wingard, Jr. and J. M. Photis, J. Am. Chem. Soc., 1974, 96, 5801.
- 34. J. Shorter, in 'The Chemistry of Sulphones and Sulphoxides', ed. S. Patai, Z. Rappoport and C. J. M. Stirling, Wiley, New York, 1988, chap. 10, p. 483; for an example of the extreme resistance of an α -halo sulfone to intermolecular nucleophilic attack see T. A. Blumenkopf, Synth. Commun., 1986, 16, 139.
- I. Shahak and E. D. Bergmann, Isr. J. Chem., 1970, 8, 589.
 K. Tatsumi, Y. Yoshioka, K. Yamaguchi and T. Fueno, Tetrahedron, 1976, 32, 1705; J. F. King, J. H. Hillhouse and K. C. Khemani, Can. J. Chem., 1985, 63, 1.
- 37. U. Zoller, in 'The Chemistry of Sulphones and Sulphoxides', ed. S. Patai, Z. Rappoport and C. J. M. Stirling, Wiley, New York, 1988, chap. 9, p. 379.
- 38. For example, thiirane 1,1-dioxides can be prepared by reaction between diazoalkanes and sulfenes. For a useful review of the preparation and properties of thiirane 1,1-dioxides, see ref. 37, p. 413.
- 39. C. Y. Meyers, D. H. Hua and N. J. Peacock, J. Org. Chem., 1980, 45, 1719.
- 40. M. W. Cronyn, J. Am. Chem. Soc., 1952, 74, 1225.
- 41. J. A. Loontjes, M. van der Leij and B. Zwanenburg, Recl. Trav. Chim. Pays-Bas, 1980, 99, 39.
- 42. J. B. Hendrickson, G. J. Boudreaux and P. S. Palumbo, Tetrahedron Lett., 1984, 25, 4617.
- 43. J. B. Hendrickson and P. S. Palumbo, J. Org. Chem., 1985, 50, 2110.
 44. J. B. Hendrickson, G. J. Boudreaux and P. S. Palumbo, J. Am. Chem. Soc., 1986, 108, 2358.
- 45. H. Quast and M. Heuschmann, Angew. Chem., Int. Ed. Engl., 1978, 17, 867; Liebigs Ann. Chem., 1981, 977; Chem. Ber., 1982, 115, 901.
- 46. A. J. Fry and L.-L. Chung, Tetrahedron Lett., 1976, 645.
- 47. R. H. Mitchell, Tetrahedron Lett., 1973, 4395.
- 48. R. H. Mitchell, J. Chem. Soc., Chem. Commun., 1973, 955.
- 49. J.-C. Pommelet, C. Nyns, F. Lahousse, R. Merényi and H. G. Viehe, Angew. Chem., Int. Ed. Engl., 1981, 20, 585.
- 50. K. Oka, Heterocycles, 1979, 12, 461.

- 51. C. R. Johnson and H. G. Corkins, J. Org. Chem., 1978, 43, 4140.
- 52. J. E. Baldwin and J. H. I. Cardellina, J. Chem. Soc., Chem. Commun., 1968, 558.
- 53. G. Büchi and B. Egger, J. Org. Chem., 1971, 36, 2021.
- 54. For a recent example see T. Kitahara, K. Miura, Y. Warita, Y. Takagi and K. Mori, Agric. Biol. Chem., 1987, 51, 1129.
- 55. T. Bacchetti and A. G. Arnaboldi, Atti Accad. Naz. Lincei Rend., Cl. Sci. Fis., Mat. Nat., 1953, 15, 75 (Chem. Abstr., 1955, 49, 2301)
- 56. L. A. Paquette, T. G. Wallis, K. Hirotsu and J. C. Clardy, J. Am. Chem. Soc., 1977, 99, 2815; L. A. Paquette, T. G. Wallis, T. Kempe, G. G. Christoph, J. P. Springer and J. Clardy, J. Am. Chem. Soc., 1977, 99, 6946; E. Chamot and L. A. Paquette, J. Org. Chem., 1978, 43, 4527.
- 57. W. D. Klobucar, L. A. Paquette and J. F. Blount, J. Org. Chem., 1981, 46, 4021.
- 58. L. A. Paquette and M. P. Trova, Tetrahedron Lett., 1986, 27, 1895; J. Am. Chem. Soc., 1988, 110, 8197.
- 59. K. Weinges and K. Klessing, Chem. Ber., 1974, 107, 1915; 1976, 109, 793; K. Weinges, H. Baake, H. Distler, K. Klessing, R. Kolb and G. Schilling, *Chem. Ber.*, 1977, 110, 2978. For other Dewar benzenes prepared by similar methods see K. Weinges and H. Baake, *Chem. Ber.*, 1977, 110, 1601; K. Weinges, G.-U. Schwarz, M. Weber and G. Schilling, Chem. Ber., 1977, 110, 2961.
- 60. H.-D. Martin, B. Mayer, M. Pütter and H. Höchstetter, Angew. Chem., Int. Ed. Engl., 1981, 20, 677.
- 61. K. Weinges, W. Kasel, J. Klein, G. Hubertus, H. Irngartinger, U. Huber-Patz and H. Rodewald, Chem. Ber., 1984, 117, 966; K. Weinges, J. Klein, W. Sipos, P. Günther, U. Huber-Patz, H. Rodewald, J. Deuter and H. Irngartinger, Chem. Ber., 1986, 119, 1540.
- 62. K. Weinges, W. Sipos, J. Klein, J. Deuter and H. Irngartinger, Chem. Ber., 1987, 120, 5.
- 63. K. Weinges, W. Kasel, U. Huber-Patz, H. Rodewald and H. Irngartinger, Chem. Ber., 1984, 117, 1868.
- 64. R. Neidlein and H. Dörr, Liebigs Ann. Chem., 1980, 1540.
- 65. K. B. Becker and M. P. Labhart, Helv. Chim. Acta, 1983, 66, 1090.
- 66. J. M. Photis and L. A. Paquette, J. Am. Chem. Soc., 1974, 96, 4715.
- 67. L. A. Paquette and J. M. Photis, J. Am. Chem. Soc., 1976, 98, 4936; Org. Synth., 1977, 57, 53.
- 68. K. Weinges and S. Haremsa, Liebigs Ann. Chem., 1987, 679.
- 69. H.-R. Sliwka and H.-J. Hansen, Helv. Chim. Acta, 1984, 67, 434. 70. J. Dressel and L. A. Paquette, J. Am. Chem. Soc., 1987, 109, 2857; J. Dressel, K. L. Chasey and L. A. Paquette, J. Am. Chem. Soc., 1988, 110, 5479.
- 71. R. G. Carlson and K. D. May, Tetrahedron Lett., 1975, 947.
- 72. J. J. Burger, T. B. R. A. Chen, E. R. de Waard and H. O. Huisman, *Heterocycles*, 1980, 14, 1739. 73. H. Matsuyama, Y. Miyazawa, Y. Takei and M. Kobayashi, *Chem. Lett.*, 1984, 833.
- 74. H. Matsuyama, Y. Miyazawa, Y. Takei and M. Kobayashi, J. Org. Chem., 1987, 52, 1703.
- 75. H. Matsuyama, Y. Miyazawa and M. Kobayashi, Chem. Lett., 1986, 433.
- 76. G. Casy and R. J. K. Taylor, J. Chem. Soc., Chem. Commun., 1988, 454.
- 77. J. Kattenberg, E. R. de Waard and H. O. Huisman, Tetrahedron Lett., 1977, 1173.
- 78. P. G. Gassman and S. M. Bonser, J. Am. Chem. Soc., 1983, 105, 667.
- 79. P. G. Gassman and K. Mlinarić-Majerski, Tetrahedron Lett., 1988, 29, 4803.
- 80. P. G. Gassman and K. Mlinarić-Majerski, J. Org. Chem., 1986, 51, 2397.
- 81. L. A. Paquette, J. Am. Chem. Soc., 1964, 86, 4085.
- 82. T. Saupe, C. Krieger and H. A. Staab, Angew. Chem., Int. Ed. Engl., 1986, 25, 451
- 83. R. H. Mitchell, T. Otsubo and V. Boekelheide, Tetrahedron Lett., 1975, 219; Y.-H. Lai and P. Chen, Tetrahedron Lett., 1988, 29, 3483.
- 84. R. H. Mitchell and V. Boekelheide, J. Am. Chem. Soc., 1970, 92, 3510.
- 85. T. Otsubo and V. Boekelheide, Tetrahedron Lett., 1975, 3381; J. Org. Chem., 1977, 42, 1085.
- 86. E. Doomes and R. M. Beard, Tetrahedron Lett., 1976, 1243.
- 87. M. P. Cooke, Jr., J. Org. Chem., 1981, 46, 1747.
- 88. S. E. Potter and I. O. Sutherland, J. Chem. Soc., Chem. Commun., 1973, 520; see also H. J. Bestmann and W. Schaper, Tetrahedron Lett., 1975, 3511. 89. M.-K. Au, T.-W. Siu, T. C. W. Mak and T.-L. Chan, Tetrahedron Lett., 1978, 4269.
- 90. M. W. Haenel and A. Flatow, Chem. Ber., 1979, 112, 249.
- 91. R. B. Mitra, M. V. Natekar and S. D. Virkar, Indian J. Chem., 1975, 13, 251.
- 92. E. Block and M. Aslam, J. Am. Chem. Soc., 1983, 105, 6164.
- 93. E. Block, M. Aslam, V. Eswarakrishnan and A. Wall, J. Am. Chem. Soc., 1983, 105, 6165.
- 94. E. Block, M. Aslam, V. Eswarakrishnan, K. Gebreyes, J. Hutchinson, R. Iyer, J.-A. Laffitte and A. Wall, J. Am. Chem. Soc., 1986, 108, 4568.
- 95. E. Block and M. Aslam, US Pat. 4 604 480 (1986) (Chem. Abstr., 1986, 105, 225 776u).
- 96. E. Block and M. Aslam, Org. Synth., 1987, 65, 90.
- 97. E. Block, V. Eswarakrishnan and K. Gebreyes, Tetrahedron Lett., 1984, 25, 5469.
- 98. E. Block, M. Aslam, R. Iyer and J. Hutchinson, J. Org. Chem., 1984, 49, 3664.
- 99. T. B. R. A. Chen, J. J. Burger and E. R. de Waard, *Tetrahedron Lett.*, 1977, 4527. 100. J. J. Burger, T. B. R. A. Chen, E. R. de Waard and H. O. Huisman, *Tetrahedron*, 1981, 37, 417.
- 101. J. J. Burger, T. B. R. A. Chen, E. R. de Waard and H. O. Huisman, Tetrahedron, 1980, 36, 1847.
- 102. G. Büchi and R. M. Freidinger, J. Am. Chem. Soc., 1974, 96, 3332.
- 103. M. Julia, D. Lavé, M. Mulhauser, M. Ramirez-Muñoz and D. Uguen, Tetrahedron Lett., 1983, 24, 1783.
- 104. F. Näf, R. Decorzant and S. D. Escher, Tetrahedron Lett., 1982, 23, 5043.
- 105. P. A. Grieco and D. Boxler, Synth. Commun., 1975, 5, 315.
- 106. K. C. Nicolaou, G. Zuccarello, Y. Ogawa, E. J. Schweiger and T. Kumazawa, J. Am. Chem. Soc., 1988, 110, 4866.
- 107. K. C. Nicolaou, Y. Ogawa, G. Zuccarello and H. Kataoka, J. Am. Chem. Soc., 1988, 110, 7247.
- 108. J. C. Philips, J. V. Swisher, D. Haidukewych and O. Morales, J. Chem. Soc., Chem. Commun., 1971, 22.

- L. A. Carpino and L. V. McAdams, III, Org. Synth., 1970, 50, 65; L. A. Carpino, L. V. McAdams, III, R. H. Rynbrandt and J. W. Spiewak, J. Am. Chem. Soc., 1971, 93, 476; L. A. Carpino and H.-W. Chen, J. Am. Chem. Soc., 1979, 101, 390.
 J. Nakayama, E. Ohshima, A. Ishii and M. Hoshino, J. Org. Chem., 1983, 48, 60.

3.9 The Wolff Rearrangement

G. BRYON GILL

University of Nottingham, UK

3.9.1 INTRODUCTION	887
3.9.1.1 Historical 3.9.1.2 Preparative Routes to α-Diazo Ketones 3.9.1.3 Methods for Initiating the Wolff Rearrangement	887 888 891
3.9.2 MECHANISM, STEREOCHEMISTRY AND COMPETING REACTIONS	891
3.9.2.1 Mechanism 3.9.2.2 Stereochemistry 3.9.2.3 Competing Reactions	891 892 893
3.9.3 THE CHEMISTRY OF WOLFF REARRANGEMENTS	897
 3.9.3.1 Homologations and Related Reactions 3.9.3.2 Ring Contractions 3.9.3.3 Cycloaddition Reactions 3.9.3.4 The Vinylogous Wolff Rearrangement 3.9.3.5 Related Reactions 	897 900 905 906 908
3.9.4 REFERENCES	909

3.9.1 INTRODUCTION

3.9.1.1 Historical

During a study of the chemistry of α -diazo ketones at about the turn of the century, Wolff found, for example, that treatment of diazoacetophenone (PhCOCHN₂) with water and silver oxide gave not the hydroxy ketone (PhCOCH₂OH), but rather the rearrangement product phenylacetic acid (PhCH₂CO₂H).^{1a} When aqueous ammonia was present in the reaction medium, the corresponding amide (PhCH₂CONH₂) was formed in good yield, even in the cold. A few years later, but independently of Wolff, Schröter^{1b} published his results of an analogous study. Occasionally, therefore, such transformations of α -diazo ketones are also called the Wolff–Schröter rearrangement.

In a sense the discoveries of Wolff and Schröter were ahead of their time, for over 20 years were to pass before convenient methods for the preparation of the diazo ketones became available. Hence, synthetic applications largely date from the early 1930s. Periodically since then a number of substantial reviews of the Wolff rearrangement have been published,²⁻¹¹ testifying to the importance that the reaction has achieved in preparative organic chemistry. Diazo ketone chemistry has achieved modern commercial importance in the photolithography industry.¹²

3.9.1.2 Preparative Routes to α -Diazo Ketones

Several methods are now available for the preparation of α -diazo ketones. Of these, the Arndt–Eistert procedure is the most versatile and popular.^{2-4,8,10,13,14} In essence, the method involves the *C*-acylation of a primary diazoalkane (RCHN₂), usually diazomethane (*i.e.* R = H), with acid chlorides or, less commonly, with acid anhydrides.¹⁵ In practice, it is essential to realize that diazo ketones are decomposed by acid. Hence, the acid chloride should be freshly purified and distilled, and precautions taken to exclude moisture. Any carboxylic acid present is converted by diazomethane into the methyl ester which contaminates the product, reducing the yield. In the classical Arndt-Eistert procedure the diazo ketones are formed in nearly quantitative yield by the slow addition of the acid chloride (1 equiv.) to an ice-cold dry ethereal (or CH₂Cl₂) solution of diazomethane (2 equiv.; for higher homologs, see below). The second equivalent of diazomethane combines with the hydrogen halide formed in the first step to give the gaseous by-products methyl chloride and nitrogen (equations 1 and 2). In the absence of the second equivalent of diazomethane,¹⁶ the HCl protonates the diazo ketone and an α -chloro ketone is then formed by the S_N2 displacement of nitrogen (equation 3).³ Triethylamine may frequently be used to take on the role of the second molecule of diazomethane in neutralizing the HCl.¹⁷ With acid anhydrides the reaction produces equimolar quantities of the diazo ketone and the methyl ester (equation 4). The addition of a mixture of a carboxylic acid (1 equiv.) and dicyclohexylcarbodiimide (1 equiv.) in ether to ice-cold ethereal diazomethane (1.25 equiv.) or diazoethane (1.25 equiv.) affords the corresponding diazoketones (max. yield 50%) by way of acid anhydride intermediates (Scheme 1).¹⁸ This route may be applicable in thoses cases where the conventional Arndt-Eistert method fails.

$$\overset{O}{\longleftarrow}_{Cl} + \overline{C}H_2\dot{N}_2 \longrightarrow \overset{O}{\longrightarrow}_{N_2^+} + HCl$$
 (1)

HCl +
$$\overline{C}H_2N_2$$
 — MeCl + N₂ (2)





$$C_6H_{11}N = C = NC_6H_{11} + RCO_2H - C_6H_{11}NHC = NC_6H_{11}$$

OCOR

$$(RCO)_{2}O + C_{6}H_{11}NHCONHC_{6}H_{11} \xrightarrow{2 R^{1}CHN_{2}} R^{1}CH_{2}OCOR + R^{1}CCOR$$

Scheme 1

Reaction conditions for the acylation of higher diazoalkanes by acid chlorides are more critical.^{19,20} Since azo coupling of the diazo ketone with the diazoalkane can occur to give azines (equation 5),²¹ it is essential to avoid an excess of the diazoalkane and to carry out the acylation at lower temperatures. Typically, the acid chloride (1 equiv.) in ether is added to a stirred solution of the diazoalkane (1 equiv.) and triethylamine (1 equiv.) in ether at -40 °C. Yields of the diazoketones are then *ca*. 74–91%, and equation (6) is fairly representative.²⁰

There are a few, mainly obvious, limitations to the Arndt-Eistert acylation procedure, in that other functional groups in the acid chloride which react with the diazoalkane require suitable protection. Be-



cause diazomethane is a fairly reactive 1,3-dipole, activated alkenes (dipolarophiles) afford cycloaddition products. Hence, α,β -unsaturated acid chlorides react with diazomethane at both the carbonyl carbon atom and at the double bond (*e.g.* equation 7).^{22,23} A more remote, unactivated double bond does not usually inhibit diazo ketone formation. On the whole, therefore, the *C*-acylation of primary diazo compounds is a widely applicable procedure.^{6,10}

The Dakin–West reaction²⁴ provides a source of *N*-acyl- α -amino ketones (*e.g.* 1), which Franzen²⁰ has exploited for the synthesis of secondary diazo ketones, thereby avoiding the need to use the higher homologs of diazomethane (Scheme 2). The ketoamide (1) is first nitrosated using N₂O₃ in glacial acetic acid. The oily *N*-nitroso derivative is separated and then decomposed with sodium methoxide in methanol to give the diazo ketone (2); yields are about 50–60%.



The synthesis of α -diazocarbonyl compounds by the transfer of a diazo group from an organic azide to a suitable substrate containing an active methylene group was studied extensively in the 1960s and 1970s, but the origins of the basic reaction are much older.^{10,25} Tosyl azide is now used almost exclusively as the diazo group transfer agent since it is stable and easily prepared, and because tosyl amide (a coproduct) is readily removed. The presence of a base is necessary, for it is the enolate anion which reacts with the tosyl azide. Commonly used solvent/base combinations are ethanol and potassium hydroxide, triethylamine, or potassium ethoxide; methanol and sodium methoxide; acetonitrile and triethylamine; and dichloromethane and piperidine. The choice of the base is, of course, dictated by the pK_a of the active methylene group in the substrate; hence the presence of two Z-groups (*i.e.* COR, CO₂R, CN, NO₂, SO₂R, *etc.*) is essential. The preparation of ethyl acetodiazoacetate (4) is illustrated in Scheme 3.²⁶ Since α -diazo- β , β '-dicarbonyl compounds such as (4), but more especially those formed from β -diketones, have a highly electrophilic diazo group, it can be difficult to completely suppress azo coupling with the enolate ion (3).²⁵ The yields of the α -diazocarbonyl compounds can be improved by employing phasetransfer catalysis. Thus, reaction of ethyl acetoacetate with tosyl azide in a mixture of pentane and saturated aqueous sodium carbonate containing tetrabutyl ammonium bromide, afforded (4) in 90% yield.²⁷



The diazo-transfer reaction strategy has to be modified for the preparation of simple diazo monoketones (e.g. 8). The presence of only one activating Z-group is circumvented by the temporary introduction of a formyl group by means of a Claisen reaction (Scheme 4; $5 \rightarrow 6$).²⁸ The formyl derivative is either preformed and then treated with tosyl azide and a tertiary amine, or the initial ketone is reacted with ethyl formate and sodium ethoxide in ethanol and then the tosyl azide is added to the Claisen adduct. Yields are generally in the range 55–90%,²⁵ and deformylative group transfer renders a wide range of α -diazocarbonyl compounds accessible; even α , β -unsaturated diazo ketones can be prepared by this route.²⁹



The need for proton-activating Z-groups may be dispensed with by employing diazo group transfer to enamines of suitable structure. The initial addition is regiospecific (cf. 7), and the triazoline so-formed decomposes spontaneously to give the diazo compound and a tosyl amidine. Although α -diazo ketones can be prepared in this manner, the other routes outlined above are generally superior. However, the method is useful for α -diazoaldehydes, and an example is given in equation (8).³⁰



Various other procedures have been developed for the synthesis of α -diazo ketones and, although less popular in modern organic chemistry, may occasionally be valuable for the preparation of specific compounds.^{7,10,25} These routes include the oxidation of α -ketoximes with chloramine,³¹ the oxidation of α -ketohydrazones with yellow mercury(II) oxide³² or manganese dioxide,³³ the hydroxide ion assisted decomposition of tosylhydrazones,³⁴ and the diazotization of α -amino ketones.^{1,35}

3.9.1.3 Methods for Initiating the Wolff Rearrangement

The Wolff rearrangement may be initiated thermally, by transition metal catalysts, or by photolysis. Thermolysis is used less frequently than the other two procedures since solution temperatures in the region of 180 °C are often required, and this can be counter-productive if the rearrangement results in the formation of a strained ring. Additionally, under thermal control nucleophiles can displace the diazo group without causing rearrangement.^{1,36} Nevertheless, the ketene intermediate produced by the rearrangement (see Section 3.9.2.1) can often be intercepted efficiently by employing a high boiling point amine or alcohol as solvent (*e.g.* aniline^{1,37} and benzyl alcohol¹⁹ which, respectively, afford the anilide and benzyl ester).

The use of transition metal species can lower appreciably the decomposition temperature of α -diazocarbonyl compounds; they can also alter the reactivity of the carbene intermediate (resulting from the initial nitrogen elimination; see Section 3.9.2.1) by complex formation. Hence, the Wolff rearrangement may occur with difficulty or, usually, not at all. Thus, some copper species (excepting, for example, CuI),³⁸ or Rh and Pd catalysts are inappropriate. Freshly prepared silver(I) oxide has been used most frequently, but silver salts (especially silver benzoate) are sometimes preferred.^{1,2,37,39} Silver-based catalysts are usually employed in combination with an alkaline reagent (*e.g.* sodium carbonate or a tertiary amine). Even under silver catalysis competing reactions may be observed, and sometimes the products of Wolff rearrangement may not be obtained (see Section 3.9.2.3).

Photolysis⁴⁰ is often successful when the thermal or catalytic procedures fail and, increasingly, has become the method of choice. The UV spectra of simple diazo ketones show two absorption bands; one in the 240–270 nm region, and a lower intensity (forbidden) band in the 270–310 nm region. Protic solvents diminish the intensity of the first band relative to the second.^{41a} Excitation of this second band accesses the lowest energy excited singlet state (¹S) of the diazo ketone by way of a $\pi \rightarrow \sigma^*$ overlap-forbidden transition.^{41b} This may be achieved by use of a medium-pressure mercury arc. Since ¹S is only moderately reactive it may be necessary to excite the higher intensity, higher energy $\pi \rightarrow \pi^*$ band by use of a low-pressure mercury arc, but this may lead to increased amounts of by-products. The photochemical procedure fails if the reaction product itself is photolabile under the reaction conditions. Since moderately low temperatures are usual (*e.g.* 0 °C), problems of thermal lability can be avoided. The presence of triplet sensitizers tends to suppress the Wolff rearrangement but other, carbene-mediated, processes may occur. Rearrangements of α -diazo ketones have also been observed under electron impact in the mass spectrometer.⁴²

3.9.2 MECHANISM, STEREOCHEMISTRY AND COMPETING REACTIONS

3.9.2.1 Mechanism

Even after more than 75 years, theoretical and mechanistic interest in the Wolff rearrangement continues unabated.⁴³ There is as yet no general agreement on all of the mechanistic detail connecting reactant to product, and it is probable that there is not a single mechanistic description for all Wolff rearrangements. Scheme 5 summarizes the main features; most of the disagreements concern the exact placement of the interconnecting arrows and the participation of the reactive intermediates (11)–(13).

Two features of the mechanism are regarded as secure. Firstly, an acyclic α -diazo ketone exists as an equilibrium mixture of the *s*-(*Z*) and *s*-(*E*) forms (9) and (10), respectively. The central C—C bond possesses double bond character, and the rotational barrier is approximately 9–18 kcal mol⁻¹ (1 kcal = 4.18 kJ).⁴⁴ The substituents R¹ and R² influence the balance of the equilibrium, and it is possible to favor entirely the *s*-(*Z*) form (*e.g.* R¹, R² = small ring) or the *s*-(*E*) form (*e.g.* R¹ = R² = *t*-butyl). Secondly, the rearrangement leads to the ketene intermediate (14),^{1b,40} which is usually trapped by a weak acid RXH (*e.g.* water, alcohols and amines) to give the product (15); stronger acids protonate the diazo ketone, and nucleophilic displacement of nitrogen occurs without rearrangement to give the product (16).

In the s-(Z) form (9) the leaving group (N₂) and the migrating group (R²) are deployed in an ideal antiperiplanar geometry for concerted extrusion/rearrangement (*i.e.* 9 \rightarrow 14). Indeed, the concerted mechanism has been proposed to account for the conformational control observed in the photolysis and thermolysis of some α -diazo ketones.⁴⁴ Product ratios obtained in direct and triplet-sensitized photolyses have led to the proposal of a concerted component from (9) and a nonconcerted (carbenic) component from (10).⁴⁵ Likewise, CIDNP studies indicate that the photochemical rearrangement of diazoacetone is concerted.⁴⁶ However, the observation of conformational control in the Wolff rearrangement is not



regarded⁴⁷ as proof *per se* for a concerted mechanism, and compelling evidence for the intermediacy of α -ketocarbenes⁴³ and for the oxirene (12)⁴⁸ has been advanced. The oxirene appears neither to be formed directly from the diazo ketone nor to rearrange directly to the ketene, but a 'symmetrical' species is needed to explain the scrambling of isotopic label which has been observed in some reactions. A wide variety of groups R¹ or R² in the carbenes (11) or (13) are able to migrate. For thermal reactions the migratory aptitudes fall in the series: H > Ph > Me > NR₂ > OR. Under photochemical control Ph and Me exchange places.¹¹

There are undoubtedly appreciable mechanistic differences in the transition metal catalyzed reactions.^{7,11} For example, Newman and Beal have suggested that a radical chain process is involved in the silver benzoate/triethylamine-promoted rearrangement of diazomethyl ketones.³⁹

3.9.2.2 Stereochemistry

The Wolff rearrangement, irrespective of the method employed for initiation, appears to occur by the intramolecular transfer of the migrating group. By and large, all stereochemical information at the migrating center is conserved. Thus, studies with optically active diazo ketones have shown predominant or complete retention of configuration of the migrating group.^{49–51} The loss, or partial loss, of stereochemical integrity has only been observed when the migrating center is a secondary carbon atom. This centre is adjacent to the C—O group of the diazo ketone, and hence partial racemization through enolization could result during diazo ketone synthesis or storage, or under the conditions of the Wolff rearrangement, as well as during the rearrangement itself. From their results on the rearrangement of s-butyldiazo-methyl compounds (RCH₂CH(Me)COCHN₂; R = H, Ph) photochemically or by a variety of silver ion catalyzed methods, Wiberg and Hutton⁵¹ concluded that either the diazo ketone was partially racemized under the reaction conditions, or racemization occurred during the rearrangement. An alternative explanation for the racemization based on a minor pathway involving C—H insertion by the carbene intermediate (Scheme 6) seems unattractive in view of the known chemistry of cyclopropanones (*i.e.* cleavage requires a strong base and takes place in the direction towards the less-substituted carbon atom).⁵²



3.9.2.3 Competing Reactions

Diazo ketones are reactive substances. The problems of compatibility towards functional groups in the same molecule are discussed in Section 3.9.1.2. Clearly, there will be a similar incompatibility in intermolecular processes if certain other compounds are present in the reaction medium during Wolff rearrangement. α -Diazo ketones are 1,3-dipoles but, according to Huisgen's index of reactivity for diazo compounds (*i.e.* CH₂N₂ > Ph₂CN₂ > EtO₂CCHN₂ > RCOC(N₂)R¹ > RCOC(N₂)CO₂Et), they are only moderately reactive.^{53a} Since dipolarophile reactivity of Z-group-activated alkenes falls in the series CO₂Ph > CO₂Et > CN > COMe > C₆H₄NO₂ > Ph,^{53b} then loss of diazo ketone through 1,3-dipolar cycloaddition is generally only important for the more electron-deficient alkenes and alkynes. Thus, for example, cycloaddition of 2-diazo-3-oxobutane to diethyl acetylenedicarboxylate affords 5-acetyl-3,4-diethoxycarbonyl-5-methyl-5*H*-pyrazole (16; equation 9).⁵⁴ Analogous intramolecular cycloaddition occurs across the nitrile group in (17) under alkaline conditions; on acidification the initial adduct undergoes prototropic rearrangement to give the triazole (18; equation 10).⁵⁵ The triazole (18) is converted by acidic hydrolysis into 2-diazoindane-1,3-dione (80%).



Diazo ketones also possess an electrophilic diazo group, and hence are susceptible to diazo-coupling reactions with suitable soft nucleophiles. Examples are given in equations $(11)^{1a}$ and (12).⁵⁶ Phosphazines such as (19) are useful synthetic intermediates in their own right. The carbon terminus of the 1,3-dipole possesses nucleophilic properties and can participate in aldol-type reactions with the particularly electrophilic carbonyl groups in 1,2-di- and 1,2,3-tri-carbonyl compounds.²⁵ Intramolecular condensations occur with greater ease (equation 13).⁵⁷ Reaction of diazo ketones of the type summarized in equations (9)–(12) have been thoroughly reviewed.^{4,6,8}

Prevention of the Wolff rearrangement is much more likely to occur through alternative reactions of the ketocarbene intermediate (11; Scheme 5). Lack of rearrangement to the ketene (14) may be due to a structural feature present in the ketocarbene (*e.g.* functional groups α to the C—O interfere with the Wolff rearrangement, or lead to secondary reactions)⁷⁻¹¹ but, not infrequently, it can be the result of an





inappropriate choice of reaction conditions. Hence, it is often worthwhile investigating the photochemical decomposition of the α -diazocarbonyl compound if the thermal or transition metal catalyzed procedures fail to effect rearrangement. Likewise, changes in temperature, solvent or coreactants should be considered.

Ketocarbenes of type (21) obtained from acyclic and medium to large ring alicyclic α -diazo ketones (20) are prone to react by way of an intramolecular [1,2] H-shift, especially under catalytic conditions, to afford α , β -unsaturated ketones (22).^{1,20,58-60} Franzen²⁰ has shown that the Wolff rearrangement of diazo ketones (20) is favored if temperatures in excess of 100 °C are used in the photochemical or Ag₂O-promoted decompositions. Related [1,2] methyl shifts can also occur, and this pathway is the principal decomposition route of 4-diazo-2,2,5,5-tetramethyl-3-hexanone.⁶¹



The transient ketocarbene is also able to effect both intramolecular and intermolecular H-abstraction. The fact that a triplet state of the carbene is often implicated follows from observations that sensitized photolyses frequently enhance the abstraction pathway.^{6,9} Abstraction and radical combination results in the formal insertion of the carbene into the C—H (or S—H, N—H, *etc.*) bond. Insertions may also occur into C—C, C—O, C—S, C—Hal, *etc.* bonds; the mechanisms are often not known, but a singlet state of the ketocarbene is probably involved in many cases. Reactions at soft basic centers (*e.g.* —S: or —Br:) generally proceed by the preliminary formation of an ylide. The multiplicity of the possible pathways is illustrated in Scheme 7,⁵⁸ equation (14) (photolysis in dioxane affords only 27, 57% and 28, 43%),⁶² and Scheme 8.⁶³

The electron deficiency in the ketocarbene intermediate can be satisfied through various types of addition reactions. Cyclopropane derivatives are the result of inter- or intra-molecular cycloaddition to alkenic double bonds. In the case of Z-group-activated C—C, the initial reaction of the diazo ketone is usually a 1,3-dipolar cycloaddition to give a pyrazoline derivative (see equation 9); on heating these decompose with nitrogen elimination, resulting in cyclopropane formation (*e.g.* $29 \rightarrow 30 \rightarrow 31$).⁶⁴ The electrophilic ketocarbenes generated by photolysis or under thermal control normally undergo the Wolff rearrangement with too great a rapidity to allow trapping by a nucleophilic double bond, even in the same molecule. Hence, the formation of the tricyclo[1.1.1.0^{4,5}]pentane skeleton by this means (equation 15) is quite exceptional.⁶⁵ Silver benzoate catalysis, however, affords only the Wolff rearrangement product (32). The production of cyclopropanes is of some importance in synthesis, and hence there have been a number of studies directed towards inhibition of the rearrangement pathway.⁹ Photolysis of diazo esters affords alkoxycarbonylcarbenes which rearrange relatively slowly and, accordingly, cyclopropanation reactions have been observed in both direct and sensitized irradiations.⁶ The *exo* addition route is favored (*e.g.* equation 16).⁶⁶ Similar results may be achieved by copper-catalyzed decomposition of the





diazo ester,⁶⁷ but loss of the metallocarbene can occur through dimerization (*e.g.* to form dimethyl fumarate as well as **33** and **34**).⁶⁶ The rearrangement of ketocarbenes is greatly inhibited through complexation by copper. Copper powder and anhydrous copper(II) sulfate are the favored catalysts, and intramolecular trapping of the metallocarbene is relatively efficient. Equations (17) (w = 1-4, x = 0-2, y = 0-1; yields *ca.* 50%)⁶⁸ and (18) (R = H, Me; X = Me, OEt; yields 30–50%)⁶⁹ hopefully indicate the scope.⁹ Allylic C—H insertion may occur in competition with cyclopropanation and Wolff rearrangement, but by the appropriate choice of the transition metal catalyst (and its ligands), a degree of control may be achieved.^{9,70} Similar reactions occur with other multiple bonds, but with nitriles, for example, formal 1,3-dipolar cycloaddition occurs to give oxazoles; yields are improved by copper catalysts (*e.g.* equation 19).⁷¹ In the absence of such trapping agents, and the use of inert solvents and copper(II) oxide as catalyst, symmetrical diacylethylenes are obtained by dimerization of the carbenone.^{6,7,72} Sometimes azines (*e.g.* RCOCH—N—N—CHCOR) are obtained by the reaction of the carbene with the original diazo compound.⁷



In the light of the large number of ways in which the reaction of α -diazocarbonyl compounds can be deflected away from the Wolff rearrangement pathway to give the ketene intermediate, it is clear that, in practice, careful thought has to be given to the choice of reaction conditions. Nevertheless, the Wolff rearrangement is a synthetic tool of very wide scope, and in the following sections only a flavor of a large body of chemistry can be given.

3.9.3 THE CHEMISTRY OF WOLFF REARRANGEMENTS

The essence of the Wolff rearrangement is the production of the ketene intermediate (14; Scheme 5). The fate of the ketene is determined by its structure and the reaction conditions. Usually the ketene is trapped *in situ* with a weak acid (RXH) to give the addition product (15). A variety of weak acids can be employed, but water, alcohols, ammonia, or primary or secondary amines are the most popular reagents (*i.e.* XR = OH, OR, NH₂, NHR, NHOH, NR₂, SR). The structure of the carboxylic acid or derivative (15) thus formed is inevitably linked to the nature of the groups R¹ and R² in the original α -diazocarbonyl compound (9). If R¹ = H, then the overall transformation is, in effect, a one-carbon homologation (*i.e.* R²CO₂H \rightarrow R²COCHN₂ \rightarrow R²CH₂COXR). This is the Arndt–Eistert synthesis,^{2-11,73} and is the principal application of the Wolff rearrangement, since it represents a rapid, efficient and economical method, of wide scope, for effecting homologation. When R¹ ≠ H, then the ultimate product (15) is an α -substituted carboxylic acid or derivative, and the overall change represents an extension to the Arndt–Eistert synthesis. However, the Wolff rearrangement, (11) \rightarrow (14), may be less efficient here because of competing decomposition pathways for the carbone intermediate (11); see Section 3.9.2.3. Hence, very much less attention has been focused on the case R¹ ≠ H. These two aspects of the Arndt–Eistert synthesis are covered in Section 3.9.3.1.

A third possibility arises when R^1 and R^2 are directly linked such that (9) is a cyclic α -diazo ketone. Rearrangement to (14) therefore represents a ring contraction, and trapping by RXH affords the ring-contracted carboxylic acid derivative (15). This area is covered in Section 3.9.3.2.

Occasionally the ketene (14) is sufficiently stable to allow its isolation in the absence of the coreactant RXH (*e.g.* $\mathbb{R}^1 = \mathbb{R}^2 = \operatorname{aryl}$).^{1b} In hydrocarbon solvents, however, the ketene is usually lost through self-dimerization or through inter- or intra-molecular cycloaddition reaction with alkenic double bonds. This aspect of the Wolff rearrangement is covered in Section 3.9.3.3. Extensions to the Wolff rearrangement strategy are covered in the concluding sections.

3.9.3.1 Homologations and Related Reactions

In the usual Arndt-Eistert homologation strategy, $(9) \rightarrow (14) \rightarrow (15)$, the nonmigrating group R¹ is a hydrogen atom and the starting material is an α -diazomethyl ketone or ester. In principle, the ketene (14) could also be generated from an α -diazo aldehyde (R¹C(N₂)CHO) in which the migrating group (R² in 9) is a hydrogen atom. However, the lack of such examples is indicative of the greater reactivity and poorer accessibility of α -diazo aldehydes.³⁰ Table 1 and the accompanying formulae provide a short summary of the many hundreds of examples of the Arndt-Eistert synthesis that have been published. Silver ion catalysis or photolysis are the favored methods for effecting the reaction. In the preparation of carboxylic acids (*i.e.* RXH = H₂O), solubility problems usually dictate the use of water as a diluant rather than as the main solvent. Hence, it can be more efficient to trap the ketene with an alcohol or amine, and subsequently hydrolyze the ester or amide obtained. The preparation of the peptides (52) and (55) indicates that the trapping agent RXH can be a fairly sophisticated molecule in its own right. In several of these examples it can be seen that migration of R² occurs with complete retention of configuration.

The migratory aptitude of R^2 in (11) varies widely with its structure (see Section 3.9.2.1), the shift of an alkoxy group being among the slowest. The formation of alkoxyketenes in the photolysis of alkyl diazoacetates is a fairly recent discovery.⁹ The major competing reactions of the carbene precursor are insertions into the C—H and O—H bonds of alcohols employed as solvents and ketene traps. The extent of Wolff rearrangement varies with structure: ethyl diazoacetate (20–25%), phenyl diazoacetate (45–60%), and *N*-methyldiazoacetamide (30%).⁹⁴ These reactions are of limited synthetic interest at present.

The value of the Arndt-Eistert method in synthesis rests with the versatility of the products obtained, and there have been a number of developments which increase the overall appeal. One of these allows the direct synthesis of homologous aldehydes by the reduction of the *N*-methylanilide with lithium aluminum hydride, or the thiol ester using Raney nickel (Scheme 9).^{4,76}

Table 2 surveys some of the results of the Arndt-Eistert method for higher order α -diazocarbonyl systems (*i.e.* 9; R¹, R² \neq H). The mildness of the photochemical procedure under neutral conditions is nicely demonstrated by the isolation of β -keto acids from the photolysis of α -diazo- β -keto thiol esters in the presence of water (entry 6).^{73,91} This work indicates that the migratory aptitude of RS is greater than RO, and occurs exclusively. Likewise, it has been shown that in the rearrangement of methyl 3-diazo-2,4-di-oxopentanoate (*i.e.* MeCOC(N₂)COCO₂Me) the migratory aptitudes fall in the series Me > MeCO₂ > MeO.⁹² Hence, although the decomposition of an unsymmetrical 2-diazo-1,3-dicarbonyl compound can give two different ketenes, one pathway usually predominates and the procedure shows sufficient

selectivity to be of value in synthesis.¹¹ The reactions in which R^1 represents a sulfonyl or phosphonate group are also of interest in this respect.

Table 1 Arndt-Eistert Homologations



Diazo ketone	RXH	Activation	Solvent/conditions	Product	Yield (%)	Ref.
(35a)	EtOH	Ag ₂ O	EtOH, reflux, 18 h	(35b)	40	74
(36a)	NH3	AgNO3	aq. NH ₃ , reflux, 2 h	(36b)	33	75
(37a)	Bu ^t OH	AgŎCOPh	Bu ^t OH, reflux, 1.25 h	(37b)	57	39
(38a)	EtSH	ν	Ét ₂ O, 8 h	(38b)	67	76
(38a)	PhNHMe	hv	PhH, 3 d	(38c)	45	76
(39a)	H ₂ O	Ag ₂ O	aq. dioxane, 65 °C, 2.5 h	(39b)	64	77
(40a)	EtOH	AgOCOPh	EtOH, reflux, 1 h	(40b)	88	78
(41a)	H ₂ O	Ag ₂ O	aq. dioxane, 25 °C, 1 h	(41b)	33	79
(42a)	H ₂ O	Ag ₂ O	H ₂ O, 50 °C	(42b)	68	49
(43a)	MeOH	AgOCOPh	MeOH, 25 °C, 15 min	(43b)	55	80
(44a)	H ₂ O	Ag ₂ O	aq. dioxane, reflux, 2 h	(44b)	33	81
(45a)	EtOH	Ağ ₂ O	EtOH, reflux, 1 h	(45b)	68	82
(46a)	H ₂ O	Ag ₂ O	aq. dioxane, 70 °C, 1.5 h	(46b)	81	83
(47a)	H_2O	$h\nu$	aq. dioxane, 2.75 h	(47b)	58	84
(48a)	MeOH	Ag ₂ O	MeOH, 100 °C, 13 h	(48b)	65	85
(49a)	MeOH	$h\bar{\nu}$	MeOH, 6–8 h	(49b)	70	86
(50)	(51)	Ag ₂ O	dioxane, 60 °C	(52)	62	87
(53)	(54)	Ag ₂ O	dioxane, 65 °C, 10 min	(55)	45	88

 $F_3C - X$

(35a) $X = COCHN_2$ (35b) $X = CH_2CO_2Et$

F	J	7	
F.C	\sim	\sim	\sim^{x}
1 30	F	F	

(36a) $X = COCHN_2$ (36b) $X = CH_2CONH_2$



(37a) $X = COCHN_2$ (37b) $X = CH_2CO_2Bu^t$

х

Me - X



(38a) X = COCHN₂ (38b) X = CH₂COSEt (38c) X = CH₂CON(Me)Ph

(39a) $X = COCHN_2$ (39b) $X = CH_2CO_2H$

х



(40a) $X = COCHN_2$

(40b) $X = CH_2CO_2Et$

(43a) $X = COCHN_2$ (43b) $X = CH_2CO_2Me$



(41a) $X = COCHN_2$ (41b) $X = CH_2CO_2H$

 $(42a) X = COCHN_2$

 $(42b) X = CH_2CO_2H$

Et

Х





(53)

F₂C

EtO₂C







Scheme 9

 Table 2
 Rearrangements Leading to α-Substituted Carboxylic Acids and Derivatives



R ²	R ¹	RXH	Activation	Solvent/conditions	Yield (%)	Ref.
4-ClC ₆ H ₄ PhCH ₂ Ph Me ₂ CH Ph H ₂ C—CHCH ₂ S MeO ₂ C Me	Me Et (CH2)3CO2Me SO2CH2Ph SO2CH2Ph CO2Me CO2Me CO2Me	PhCH ₂ OH NH ₃ PhNH ₂ PhCH ₂ OH PhCH ₂ OH H ₂ O MeOH MeOH	190 °C AgNO3 150 °C 110 °C 110 °C <i>hv</i> <i>hv</i> <i>hv</i> <i>hv</i>	PhCH ₂ OH, isoquinoline, 5 min aq. NH ₃ , 80 °C, 2 h PhNH ₂ , 5 min PhMe, 2 h PhMe, 4 h PhH, 3 h PhH, 4 h PhH, 6 h	87 60 55 86 92 70 83	19 89 59 90 90 91 92 92
	^{- ξ} PO(OMe)2 کرم PO(OMe)2	EtOH	hv hv	$CH_2Cl_2, 2-3 h$ $CH_2Cl_2, 2-3 h$	81 66	93 93

3.9.3.2 Ring Contractions

When the groups R^1 and R^2 of (9; Scheme 5) are directly linked to give a cyclic α -diazocarbonyl system, then Wolff rearrangement results in ring contraction (Scheme 10). It is in this area especially that the photochemical method comes into its own. In processes involving thermal control or transition metal catalysis, competing reaction pathways are frequently a serious problem. Application of the Wolff rearrangement to ring contraction is relatively recent, and post-dates Horner's pioneering work on the photochemical method.⁴⁰ Thus, although diazocamphor (56) yields the tricyclic ketone (57) upon heating,^{1a} photolysis in aqueous dioxane affords *exo*-1,5,5-trimethylbicyclo[2.1.1]hexane-6-carboxylic acid (59) through kinetically controlled hydration of the ketene intermediate (58) at its least-hindered *endo* face (Scheme 11).⁹⁵ This study laid the groundwork for a protocol in synthesis which has since been widely exploited, especially for the construction of strained rings since ring size imposes few restrictions on the Wolff rearrangement. This area has been reviewed previously.^{4-11,96}

There are very few examples of ring contraction from cyclobutyl to cyclopropyl. Irradiation of a benzene solution of 2-diazo-3,4-bis(diphenylmethylene)cyclobutanone (60) in the presence of water, methanol, 2-propanol or aniline afforded the carboxylic acid derivatives (61) in 13, 87, 75 and 45% yield respectively. Thermolysis of an aqueous dioxane solution of (60) surprisingly gave (61; XR = OH) in higher yield (52%; equation 20).⁹⁷

 α -Diazocyclopentanones are much easier to prepare, and consequently much more attention has been given to the contraction of cyclopentanes to cyclobutanes. Several bicyclo[2.1.1]hexane derivatives have been prepared from α -diazobicyclo[2.2.1]heptanones following similar chemistry to the transformation (56 \rightarrow 59; Scheme 11).⁹⁸ Frequently the 5-carboxylate derivatives are formed as a mixture of *endo* and



Scheme 10



exo isomers as a result of kinetically controlled addition of RXH to each of the two faces of the ketene intermediates. The *endo:exo* ratio reflects the ease of access on steric grounds. When XR represents a large grouping, however, thermodynamic control may apply and the least-hindered carboxylic acid derivative is then formed predominantly. The derivatives (62)–(66) have been prepared by the photolysis of the analogous α -diazo ketones in the presence of the appropriate trapping agent RXH. Product ratios, where given, are in the sense *endo:exo*.



Similar procedures are also successful in the construction of cyclobutane rings in other stereochemical environments, including the formation of bicyclo[3.1.1]heptane $(67)^{105}$ and bridged tricyclic systems such as (68),¹⁰⁶ $(69)^{107}$ and (70).¹⁰⁸ A variety of annelated cyclobutanes have been prepared by the photochemical procedure; as above, yields are sometimes poor or only moderate. Examples include bicyclo[2.1.0]pentanes (71),¹⁰⁹ bicyclo[2.2.0]hexanes $(72)^{110}$ and cyclobuteno aromatics (*e.g.* 73),¹¹¹ D-norsteroids (*e.g.* 74),¹¹² A-bisnorsteroids¹¹³ and triterpenes¹¹⁴ as well as the highly strained tricyclo[4.2.0.0^{1,4}]octane system (75),¹¹⁵ and the [4.4.4.5]fenestrane (76).¹¹⁶



The ring contraction of 2-diazo-1-oxonaphthenes is very difficult because of a large increase in inplane strain. The parent compound (77) failed to undergo Wolff rearrangement on thermolysis, instead giving the ketazine (81%; *cf.* equation 5).¹¹⁷ Chapman and coworkers^{43a} have observed the photochemical extrusion of nitrogen from (77) and related compounds in an argon matrix at 10–15 K. The α -ketocarbenes were characterized spectroscopically and by trapping; excitation (T_0 – T_1) led to ring contraction, the rates paralleling the increase in strain of ketene formation. In solution-phase preparative chemistry such α -ketocarbenes are, presumably, simply too reactive to survive encounters with the solvent or unreacted diazoketone.

Applications of the Wolff rearrangement in the synthesis of four-membered heterocycles have been much more limited, and have largely been directed towards the simpler mono- and bi-cyclic systems. Ring contraction of 3-diazopyrrolidine-2,4-diones affords a relatively simple route to β -lactams;¹¹⁸ carbon migrates in preference to nitrogen (*e.g.* equation 21).^{118a} The photochemical decomposition of 4-di-azopyrazolidine-3,5-diones analogously yields aza- β -lactams. Unsymmetrically substituted substrates give rise to a mixture of regioisomers (equation 22).¹¹⁹ Relative migratory aptitudes fall in the series NPh > NCHPh₂ ≈ NCH₂Ph ≈ NMe > NCH₂CO₂Et; yields were in the range 30–72% and regioselectivities varied from *ca*. 1:1 to 2.8:1. Earlier work on 2,2,5,5-tetraalkyl-4-diazotetrahydro-3-oxofurans had shown that rearrangement to oxetane-3-carboxylates occurred in good yield under photochemical, thermal or Ag₂O-promoted conditions.⁷





There are a considerable number of examples of great variety for the construction of cyclopentane rings by ring contraction.^{4-11,96} Six-membered rings in highly strained polycyclic systems are amenable to contraction, as shown by the synthesis of homocubanecarboxylic acid (78),¹²⁰ but there have been few applications in this area. Among bicyclic systems, photolysis of the appropriate diazo ketones resulted in efficient conversion into the bicyclo[2.1.1]hexane-2-carboxylic acid $(79)^{99}$ and bicyclo[2.2.1]heptane-7-carboxylic acid (80).¹²¹ In view of the earlier discussion concerning diazo ketone (77), the very low yield of ring-contracted product from photolysis of the phenanthrene derivative (81; equation 23), is not too surprising. A dihydrobenzofuran is the major product, resulting from an unprecedented reaction of the ketocarbene (or 1,3-dipole) with the aromatic solvent. On the other hand, Wolff rearrangement of 9-diazo-10-oxophenanthrene itself proceeds efficiently.¹²²



Simple 2-diazocyclohexanones are readily converted by photolysis in methanol, for example, into methyl cyclopentanecarboxylates.¹²³ However, the Favorskii method for ring contraction (see Volume 3, Chapter 3.7) is usually more convenient in these systems since the necessary starting materials are more readily prepared. Applications have been found in the synthesis of A-norsteroids^{113,124} and A-nortriterpenes. For example, photolysis of 2-diazoallobetulone (**82**) in the presence of weak acids RXH affords mainly the 2 β -carboxy-A-nor derivative (**83**).¹¹⁴ The 3,12-cycloiceane esters (**84**) are obtained in 52% yield from the photolysis of the diazo ketone (**85**) in methanol.¹²⁵ Wolff rearrangement of cyclic 2-diazo-1,3-dicarbonyl systems gives ring-contracted β -keto carboxylic acid derivatives.^{126,127} The predominant or exclusive product of rearrangement of an unsymmetrical diazo compound can be predicted from the relative migratory aptitude data discussed previously (see Sections 3.9.2.1 and 3.9.3.1). Thus, for example, exclusive carbon migration occurs in the conversion of the spirocyclic diazo compound (**86**) into (**87**; 69%).¹²⁸





The photochemical decomposition of o-diazo oxides ('o-quinone diazides') is of both preparative and commercial interest and importance. Wolff rearrangement with ring contraction gives access to the cyclopentadiene system (including heterocyclic analogs). Since the o-diazo oxides are usually prepared from o-aminophenols, the process is remarkable in that it represents the shrinking of a benzene ring. However, the products are prone to undergo diazo coupling with unreacted starting material (e.g. $88 \rightarrow 89 \rightarrow 90$),¹²⁹ but this chemistry has been exploited in the Diazotype offset photocopying process.¹³⁰ Reduction of pH inhibits the coupling process, hence the photolyses for preparative ring contraction are often conducted in acidic solvents (e.g. aq. AcOH). Reaction of benzene derivatives is decidedly more difficult than for o-diazo oxides based on naphthalene and higher aromatics where stabilization is not so great. Thermolysis can be used for the latter systems, but an intense UV source is necessary for benzenoid o-diazo oxides to enhance the rate of rearrangement. An indication of the variety of the synthetic applications is given in equations (24),¹³¹ (25),¹³¹ (26)^{132,133} and (27).¹³³ The migration of sp² nitrogen (equation 25) is particularly noteworthy. Similar routes have been used for the synthesis of a variety of indenecarboxylic acids (yields ca. 20-60%),¹³¹ related systems¹³⁴ and pentalenes such as (105; 75%).¹³⁵





Irradiation of 1,3-bisdiazocyclohexan-2-ones in hydroxylic solvents affords cyclopentene-1-carboxylic acids and esters; regiochemical control is only moderate.^{9,136} Although the course of the reactions could be accommodated by sequential Wolff rearrangement and [1,2] H-shift, studies on 1,3-bisdiazo-1,3-di-phenylpropan-2-one indicate the formation of the cyclopropenone intermediate (**106**).¹³⁷

There have been relatively few applications to the contraction of larger rings. Irradiation of solutions of 4-diazo-*trans*-bicyclo[6.1.0]nonan-5-one afforded the strained carboxylate derivatives (**107**; XR = OH, OMe) in *ca*. 22–45% yield.¹³⁸ Small to medium ring diazo ketones afford the ring-contracted carboxylic acids (**108**; n = 4-10) in 25–95% yield on photolysis in dioxane-H₂O¹³⁹ or THF-H₂O.⁵⁸ Thermolysis of the diazo ketones in aniline at 150–160 °C gave the anilides of (**108**) in 78–91% yield.²⁸ Transannular reactions of the ketocarbene intermediates can be a complication under some conditions.^{28,139} 4-Carboxy[8]paracyclophane (**109**) has been prepared (25%) by the photochemical method.¹⁴⁰

3.9.3.3 Cycloaddition Reactions

When the Wolff rearrangement is conducted in inert solvents (*i.e.* in the absence of weak acids, RXH) it is possible, in principle, to intercept the ketene intermediate; stabilized ketenes can be isolated.^{1b} Their [2 + 2] cycloadditions to activated alkenes and dienes is a valuable synthetic procedure.¹⁴¹ However, such reactions are relatively slow, and the generation of ketenes from α -diazo ketones provides a number of other species capable of acting as trapping agents. Thus, the rather labile aldoketenes react with intact α -diazo ketone to give butenolides or pyrazoles (equation 28).^{7,9,11} The *o*-diazo oxides on thermolysis afford dioxofulvalenes, resulting from the formal [1,3] dipolar cycloaddition of the ketocarbene intermediate to the ketene (equation 29).^{9,11} Nevertheless, trapping the ketene by cycloaddition to C—C, C—N, N—N and the C—O bonds of *o*-quinones has been achieved frequently.^{9,11} but such reactions have been studied mainly from a general interest viewpoint rather than for specific purposes in synthesis. Various


Rearrangement Reactions

examples of intramolecular trapping by a strategically located C—C in the molecule are also known. A particularly remarkable case is shown in equation (30).¹⁴² The major product of decomposition of diazo ketone (110) is 6,7,7-trimethyltricyclo[3.2.1.0^{3,6}]octan-4-one (111), even in boiling aqueous dioxane. On the other hand, decomposition of the more rigid diazo ketone (112) partitions between intramolecular trapping of the carbene to give (113) and the ketene to give (114) by way of a Cope rearrangement.¹⁴³ The pathway to (114) *via* the ketene is favored in the photolytic reaction, whereas copper-catalyzed decomposition affords (113) preferentially. By-and-large, therefore, this area of diazo ketone chemistry has scarcely been over-exploited.



3.9.3.4 The Vinylogous Wolff Rearrangement

This name has been applied¹⁴⁴ to the decomposition of β , γ -unsaturated diazo ketones (115) by copper salts in the presence of alcohols, leading to skeletally rearranged γ , δ -unsaturated esters, and is a comparatively recent discovery.^{144,145} The transformation represents a synthetic alternative to Claisen type rearrangements (compare equations 31 and 32). The allylic transposition of the acetate residue is thought to occur by way of a bicyclo[2.1.0]pentan-2-one (116), which is formed by intramolecular trapping of the ketocarbenoid (Scheme 12). Fragmentation of the bicyclo[2.1.0]pentan-2-one gives rise to a β , γ -unsaturated ketene (117), which is captured by the added nucleophile to afford the observed γ_{δ} -unsaturated carboxylic acid derivative (118). Similar decomposition of (115) by thermolysis, photolysis or by silver ion catalysis gives mixtures of rearranged product and the normal Wolff product, respectively (118) and (119); mixture compositions vary, but the normal product (119) generally predominates. In contrast, copper(II) ion catalysis affords mainly rearranged products (118), and Cu(OTf)₂ and benzyl alcohol give the best results. Although the initial insertion step leading to (116) seems to be stereoselective, the fragmentation reaction appears to proceed with loss of this initial stereoselectivity, hence mixtures of stereoisomers are usually produced. Illustrative examples of the reaction are given in equations (33)-(35);¹⁴⁴ the reactions were carried out in boiling cyclohexane for 4 h. In equation (35) it can be seen that the vinylogous Wolff rearrangement leading to (121), a 1:1 mixture of stereoisomers, competes effectively with cyclopropanation of the γ , δ -double bond, giving a 1:1 mixture of the stereoisomers of (120), when the β , γ -double bond is at least equally activated.





907



In a recent application, the vinylogous Wolff rearrangement has been used to provide angular functionalization in annelated polycyclic systems (e.g. equation 36).¹⁴⁶

3.9.3.5 Related Reactions

The Wolff rearrangement has much broader horizons than the decomposition pathway for α -diazocarbonyl compounds, and a generalized mechanistic route is given in Scheme 13. This allows for both a stepwise pathway (a) and concerted decomposition (b). Thus, although sometimes α -diazo ketones (122a) appear to give the ketene (124) by path b, in the majority of reactions there is strong evidence for a ketocarbene intermediate (123) preceding the ketene. Acyl azides (122c), on the other hand, suffer concerted rearrangement to isocyanates (124) on thermolysis (Curtius rearrangement); there is no evidence for nitrene intermediates in the rearrangement pathway.^{147,148} The related Schmidt, Lossen and Hofmann



reactions, which proceed via (122c; Schmidt) or (122d; LG = leaving group such as Br or OAc) also follow path b but, because of the reaction conditions, products derived from the isocyanate, rather than the isocyanate itself, are isolated.147

The electronic make-up of the assemblage WXYZ of (122) can be fulfilled in a number of other ways, as indicated by the selection of structures (122b)–(122f). Although in simple α -diazo ketones there does not appear to be a strong direct electronic interaction between the carbonyl oxygen atom (X in 122) and the terminal diazo nitrogen atom (Z in 122), when X is a softer base (e.g. NR, S, Se) the cyclic form (125) is more stable. Thus, the intermediate (123) can be accessed from (125a)-(125c), further broadening the scope. Studies on these other variants of (122) and (125) have been much more limited. Reaction pathways are determined by reaction conditions, but products arising from Wolff-type rearrangements have been observed in the thermolysis or pyrolysis of acylsulfonium ylides, 149 B-ketosulfoxonium ylides (122b),¹⁵⁰ α -diazosulfones (122e),¹⁵¹ α -diazophosphine oxides (122f),^{152,153} 1,2,3-triazoles (125a),¹⁵⁴ 1,2,3-thiadiazoles (125b),¹⁵⁵ and 1,2,3-selenadiazoles (125c).¹⁵⁶

3.9.4 REFERENCES

- 1. (a) L. Wolff, Justus Liebigs Ann. Chem., 1902, 325, 129; 1904, 333, 1; 1912, 394, 23; (b) G. Schröter, Chem. Ber., 1909, 42, 2346; 1916, 49, 2704.
- 2. W. E. Bachmann and W. S. Struve, Org. React. (N.Y.), 1942, 1, 38.
- 3. B. Eistert, in 'Newer Methods of Preparative Organic Chemistry', Interscience, New York, 1948, vol. 1, p. 513.
- 4. (a) F. Weygand and H. J. Bestmann, Angew. Chem., 1960, 72, 535; (b) F. Weygand and H. J. Bestmann, in 'Newer Methods of Preparative Organic Chemistry', Academic Press, New York, 1964, vol. 3, p. 451.
- 5. A. Mustafa, Adv. Photochem., 1964, 2, 113.
- 6. W. Ando, in 'The Chemistry of Diazonium and Diazo Groups', ed. S. Patai, Wiley, New York, 1978, part 1, p. 458.
- 7. L. L. Rodina and I. K. Korobitsyna, Russ. Chem. Rev. (Engl. Transl.), 1967, 36, 260.
- 8. B. Eistert, M. Regitz, G. Heck and H. Schwall, Method. Org. Chem. (Houben-Weyl), 1968, 10 (4), 473.
- 9. W. Kirmse, in 'Carbene Chemistry', 2nd edn., Academic Press, New York, 1971, pp. 337, 475. 10. A. L. Fridman, G. S. Ismagilova, V. S. Zalesov and S. S. Novikov, Russ. Chem. Rev. (Engl. Transl.), 1972, 41. 371.
- 11. H. Meier and K.-P. Zeller, Angew. Chem., Int. Ed. Engl., 1975, 14, 32.
- 12. H. Steppan, G. Buhr and H. Vollmann, Angew. Chem., Int. Ed. Engl., 1982, 21, 455.
- 13. F. Arndt, B. Eistert and W. Partale, Chem. Ber., 1927, 60, 1364; F. Arndt and J. Amende, Chem. Ber. 1928, 61, 1122; F. Arndt, B. Eistert and J. Amende, Chem. Ber., 1928, 61, 1949.
- 14. W. Bradley and R. Robinson, J. Chem. Soc., 1928, 1310, 1541; W. Bradley and G. Swarzenbach, J. Chem. Soc., 1928, 2904; W. Bradley and J. K. Eaton, J. Chem. Soc., 1937, 1913.
- 15. W. Bradley and R. Robinson, J. Am. Chem. Soc., 1930, 52, 1558; P. McC. Duggleby and G. Holt, J. Chem. Soc., 1962, 3579.
- 16. D. A. Clibbens and M. Nierenstein, J. Chem. Soc., 1915, 107, 1491; M. Nierenstein, D. G. Wang and J. C. Warr, J. Am. Chem. Soc., 1924, 46, 2551; H. H. Lewis, M. Nierenstein and E. M. Rich, J. Am. Chem. Soc., 1925, 47, 1728.
- 17. M. S. Newman and P. F. Beal, J. Am. Chem. Soc., 1949, 71, 1506; M. Berenbom and W. S. Fones, J. Am. Chem. Soc., 1949, **71**, 1629.
- 18. D. Hodson, G. Holt and D. K. Wall, J. Chem. Soc. (C), 1970, 971.
- 19. A. L. Wilds and A. L. Meader, Jr., J. Org. Chem., 1948, 13, 763.
- 20. V. Frantzen, Justus Liebigs Ann. Chem., 1957, 602, 199.
- 21. P. Yates, D. G. Farnum and D. W. Wiley, Chem. Ind. (London), 1958, 69.
- 22. J. A. Moore, J. Org. Chem., 1955, 20, 1607; J. Am. Chem. Soc., 1955, 77, 3417.
- 23. J. H. Wotiz and S. N. Buco, J. Org. Chem., 1955, 20, 210.
- H. D. Dakin and R. West, J. Biol. Chem., 1928, 78, 91; G. H. Cleland and C. Niemann, J. Am. Chem. Soc., 1949, 71, 841; C. S. Rondestvedt, Jr., B. Manning and S. Tabibian, J. Am. Chem. Soc., 1950, 72, 3183; S. Searles and G. J. Cvejanovich, J. Am. Chem. Soc., 1950, 72, 3200.
- 25. For reviews and leading references see M. Regitz, Synthesis, 1972, 351; M. Regitz, in 'The Chemistry of Diazonium and Diazo Groups', ed. S. Patai, Wiley, New York, 1978, part 2, p. 659. 26. M. Regitz and A. Liedhegener, *Chem. Ber.*, 1966, **99**, 3128.
- 27. H. Ledon, Synthesis, 1974, 347.
- 28. M. Regitz and J. Rüter, Chem. Ber., 1968, 101, 1263.
- 29. M. Regitz, F. Menz and A. Liedhegener, Justus Liebigs Ann. Chem., 1970, 739, 174.
- 30. J. Kučera, Z. Janoušek and Z. Arnold, Collect. Czech. Chem. Commun., 1970, 35, 3618.
- M. O. Forster, J. Chem. Soc., 1915, 107, 260; M. P. Cava, R. L. Litle and D. R. Napier, J. Am. Chem. Soc., 1958, 80, 2257; J. Meinwald, P. G. Gassman and E. G. Miller, J. Am. Chem. Soc., 1959, 81, 4751; L. Horner, W. Kirmse and K. Muth, Chem. Ber., 1958, 91, 430; L. Horner, K. Muth and H.-G. Schmelzer, Chem. Ber., 1959, 92, 2953; S. Huneck, Chem. Ber., 1965, 98, 1837; M. P. Cava, E. J. Glamkowski and P. M. Weintraub, J. Org. Chem., 1966, **31**, 2755.
- 32. T. Curtius, Chem. Ber., 1889, 22B, 2161; T. Curtius and K. Thun, J. Prakt. Chem., 1891, 44, 182; T. Curtius and B. Kastner, J. Prakt. Chem., 1911, 83, 215; R. C. Fuson, L. J. Armstrong and W. J. Shenk, Jr., J. Am.

Chem. Soc., 1944, 66, 964; B. G. Christensen, N. G. Steinberg and R. Hirschmann, Chem. Ind. (London), 1958, 1259; M. S. Newman and A. Arkell, J. Org. Chem., 1959, 24, 385.

- 33. H. Morrison, S. J. Danishefsky and P. Yates, Angew. Chem., 1961, 73, 743, J. Org. Chem., 1961, 26, 2617; S. Hauptmann, M. Kluge, K.-D. Seidig and H. Wilde, Angew. Chem., Int. Ed. Engl., 1965, 4, 688.
- 34. M. P. Cava and R. L. Litle, Chem. Ind. (London), 1957, 367; M. P. Cava R. L. Litle and D. R. Napier, J. Am. Chem. Soc., 1958, **80**, 2257; K. B. Wiberg, B. R. Lowry and T. H. Colby, J. Am. Chem. Soc., 1961, **83**, 3998; W. Ried and R. Dietrich, Chem. Ber., 1961, **94**, 387; L. Horner and W. Dürckheimer, Chem. Ber., 1962, **95**, 1206; L. Horner and K.-H. Weber, Chem. Ber., 1962, 95, 1227; Chem. Ber., 1963, 96, 1568; L. Horner, K.-H. Teichmann, K.-H. Weber and E. Geyer, Chem. Ber., 1965, 98, 1233; H. Stetter and K. Kiehs, Chem. Ber., 1965, 98, 1181.
- 35. A. Angeli, Chem. Ber., 1893, 26, 1715; T. B. Johnson, O. Baudisch and A. Hoffmann, Chem. Ber., 1931, 64, 2629; B. Eistert, H. Elias, E. Kosch and R. Wollheim, Chem. Ber., 1959, 92, 130; B. Eistert and F. Geiss, Chem. Ber., 1961, 94, 929; D. G. Farnum and P. Yates, Chem. Ind. (London), 1960, 659; J. Am. Chem. Soc., 1962, 84, 1399.
- 36. W. Jugelt and D. Schmidt, Tetrahedron, 1969, 25, 969.
- 37. F. Arndt and B. Eistert, Chem. Ber., 1935, 68, 200.
- 38. P. Yates and J. Fugger, Chem. Ind. (London), 1957, 1511.
- M. S. Newman and P. F. Beal, J. Am. Chem. Soc., 1950, 72, 5163.
 L. Horner, E. Spietschka and A. W. Gross, Justus Liebigs Ann. Chem., 1951, 573, 17; L. Horner and E. Spietschka, Chem. Ber., 1952, 85, 225.
- 41. (a) L. L. Leveson and C. W. Thomas, Tetrahedron, 1966, 22, 209; (b) I. G. Csizmadia, S. A. Houlden, O. Meresz and P. Yates, Tetrahedron, 1969, 25, 2121.
- 42. P. L. Kinson and B. M. Trost, Tetrahedron Lett., 1969, 1075; D. C. DeJongh, R. Y. Van Fossen, L. R. Dusold and M. P. Cava, Org. Mass Spectrom., 1970, 3, 31; K.-P. Zeller, H. Meier and E. Müller, Tetrahedron, 1972, 28, 5831.
- 43. For discussions and compilations of pertinent references see (a) R. J. McMahon, O. L. Chapman, R. A. Hayes, T. C. Hess and H.-P. Krimmer, J. Am. Chem. Soc., 1985, 107, 7597; (b) H. Murai, I. Safarik, M. Torres and O. P. Strausz, J. Am. Chem. Soc., 1988, 110, 1025.
- 44. F. Kaplan and G. K. Meloy, J. Am. Chem. Soc., 1966, 88, 950; F. Kaplan and M. L. Mitchell, Tetrahedron Lett., 1979, 759.
- 45. H. Tomioka, H. Okuno and Y. Izawa, J. Org. Chem., 1980, 45, 5278; H. Tomioka, M. Kondo and Y. Izawa, J. Org. Chem., 1981, 46, 1090; H. Tomioka, N. Hayashi, T. Asano and Y. Izawa, Bull. Chem. Soc. Jpn., 1983, 56, 758.
- 46. H. D. Roth and M. L. Manion, J. Am. Chem. Soc., 1976, 98, 3392; H. D. Roth, Acc. Chem. Res., 1977, 10, 85.
- 47. M. Torres, J. Ribo, A. Clement and O. P. Strausz, Can. J. Chem., 1983, 61, 996; see, however, C. Marfisi, P. Verlaque, G. Davidovics, J. Pourcin, L. Pizzala, J.-P. Aycard and H. Bodot, J. Org. Chem., 1983, 48, 533.
- 48. For recent reviews see E. G. Lewars, Chem. Rev., 1983, 83, 519; M. Torres, E. M. Lown, H. E. Gunning and O. P. Strausz, Pure Appl. Chem., 1980, 52, 1623; E. G. Lewars and S. Siddiqi, J. Org. Chem., 1985, 50, 135.
- 49. N. A. Preobrashenski, A. M. Poljakowa and W. A. Preobrashenski, Chem. Ber., 1935, 68, 850; J. Gen. Chem.
- USSR, 1939, 9, 1402 (Chem. Abstr., 1940, 34, 1658⁷). 50. J. F. Lane, J. Willenz, A. Weissberger and E. S. Wallis, J. Org. Chem., 1940, 5, 276; J. F. Lane and E. S. Wallis, J. Org. Chem., 1941, 6, 443; J. Am. Chem. Soc., 1941, 63, 1674.
- 51. C. D. Gutsche, J. Am. Chem. Soc., 1948, 70, 4150; K. J. Sax and W. Bergmann, J. Am. Chem. Soc., 1955, 77, 1910; K. B. Wiberg and T. W. Hutton, J. Am. Chem. Soc., 1956, 78, 1640; F. V. Brutcher, Jr., and D. D. Rosenfeld, J. Org. Chem., 1964, 29, 3154.
- 52. N. J. Turro and W. B. Hammond, J. Am. Chem. Soc., 1965, 87, 3258; 1966, 88, 2880; J. K. Crandall and W. H. Machleder, J. Am. Chem. Soc., 1968, 90, 7347; N. J. Turro, R. B. Gagosian, C. Rappe and L. Knutsson, J. Chem. Soc., Chem. Commun., 1969, 270; P. S. Wharton and A. R. Fritzberg, J. Org. Chem., 1972, 37, 1899.
- 53. (a) R. Huisgen, Angew. Chem., Int. Ed. Engl., 1963, 2, 565; (b) R. Huisgen, H. Stangl, H. J. Sturm and H. Wagenhofer, Angew. Chem., 1961, 73, 170.
- 54. O. Diels and H. König, Chem. Ber., 1938, 71, 1179.
- 55. G. Holt and D. K. Wall, J. Chem. Soc., 1965, 1428.
- 56. H. J. Bestmann and H. Kolm, Chem. Ber., 1963, 96, 1948.
- 57. T. L. Burkoth, Tetrahedron Lett., 1969, 5049.
- 58. M. Regitz and J. Rüter, Chem. Ber., 1969, 102, 3877.
- 59. S. Hauptmann and K. Hirschberg, J. Prakt. Chem., 1966, 34, 262.
- 60. S. A. Matlin and P. G. Sammes, J. Chem. Soc., Perkin Trans. 1, 1972, 2623.
- 61. M. S. Newman and A. Arkell, J. Org. Chem., 1959, 24, 385.
- 62. R. R. Rando, J. Am. Chem. Soc., 1970, 92, 6706.
- 63. P. A. Grieco, D. Boxler and K. Hirai, J. Org. Chem., 1973, 38, 2572.
- 64. P. C. Guha and M. S. Muthanna, Chem. Ber., 1938, 71, 2665.
- 65. S. Masamune, J. Am. Chem. Soc., 1964, 86, 735.
 66. W. E. Doering and T. Mole, Tetrahedron, 1960, 10, 65.
- 67. R. Breslow, R. Winter and M. Battiste, J. Org. Chem., 1959, 24, 415; J. Warkentin, E. Singleton and J. F. Edgar, Can. J. Chem., 1965, 43, 3456.
- W. von E. Doering, E. T. Fossel and R. L. Kaye, *Tetrahedron*, 1965, 21, 25; see also G. Stork and J. Ficini, J. Am. Chem. Soc., 1961, 83, 4678; M. M. Fawzi and C. D. Gutsche, J. Org. Chem., 1966, 31, 1390.
- 69. G. Cannic, G. Linstrumelle and S. Julia, Bull. Soc. Chim. Fr., 1968, 4913.
- 70. S. Bien and Y. Segal, J. Org. Chem., 1977, 42, 1685.
- 71. R. Huisgen, G. Binsch and L. Ghosez, Chem. Ber., 1964, 97, 2628; R. Huisgen, H. J. Sturm and G. Binsch, Chem. Ber., 1964, 97, 2864; R. Huisgen, G. Binsch and H. König, Chem. Ber., 1964, 97, 2868.

- 72. I. Ernest, Collect. Czech. Chem. Commun., 1954, 19, 1179; V. Hněvsová, V. Směléý and I. Ernest, Collect. Czech. Chem. Commun., 1956, 21, 1459; I. Ernest and J. Stanek, Collect. Czech. Chem. Commun., 1959, 24, 530; I. Ernest and Z. Linhartová, Collect Czech. Chem. Commun., 1959, 24, 1022, 2072.
- 73. V. A. Nikolaev, S. D. Kotok and I. K. Korobitsyna, Zh. Org. Khim., 1974, 10, 1334 (Chem. Abstr., 1974, 81, 179 821r); I. K. Korobitsyna and V. A. Nikolaev, Zh. Org. Khim., 1976, 12, 1251; (Chem. Abstr., 1976, 85, 142 560m).
- 74. F. Brown and W. K. R. Musgrave, J. Chem. Soc., 1953, 2087.
- 75. J. D. Park, E. R. Larsen, H. V. Holler and J. R. Loacher, J. Org. Chem., 1958, 23, 1166.
- 76. F. Weygand and H. J. Bestmann, Chem. Ber., 1959, 92, 528.
- 77. L. F. Fieser and G. W. Kilmer, J. Am. Chem. Soc., 1940, 62, 1354.
- 78. V. Lee and M. S. Newman, Org. Synth., 1970, 50, 77.
- 79. A. A. Plentl and M. T. Bogert, J. Org. Chem., 1941, 6, 669.
- 80. T. L. Westman, R. Paredes and W. S. Brey, Jr., J. Org. Chem., 1963, 28, 3512.
- 81. L. I. Zakharkin, V. N. Kalinin and V. V. Gedymin, Tetrahedron, 1971, 27, 1317.
- 82. E. M. Crook and W. Davies, J. Chem. Soc., 1937, 1697.
- 83. M. C. Kloetzel, J. Am. Chem. Soc., 1940, 62, 1708.
- 84. S. J. Field and D. W. Young, J. Chem. Soc., Perkin. Trans. 1, 1983, 2387.
- 85. R. Casanova and T. Reichstein, Helv. Chim. Acta, 1949, 32, 647.
- 86. G. Sinha and U. R. Ghatak, Indian J. Chem., Sect. B, 1981, 20, 411.
- 87. F. Weygand, P. Klinke and I. Eigen, Chem. Ber., 1957, 90, 1896; F. Weygand, H. J. Bestmann and E. Klieger, Chem. Ber., 1958, 91, 1037.
- 88. D. Fles and A. Markovac-Prpić, Croat. Chem. Acta, 1957, 29, 79 (Chem. Abstr., 1958, 52, 11 744g).
- 89. F. M. Dean and A. Robertson, J. Chem. Soc., 1948, 1674.
- Y.-C. Kuo, T. Aoyama and T. Shioiri, Chem. Pharm. Bull., 1982, 30, 526, 2787. 90.
- 91. V. Georgian, S. K. Boyer and B. Edwards, J. Org. Chem., 1980, 45, 1686.
- 92. R. R. Gallucci and M. Jones, Jr., J. Org. Chem., 1985, 50, 4404.
- 93. P. Callant, L. D'Haenens, E. Van der Eycken and M. Vandewalle, Synth. Commun., 1984, 14, 163.
- 94. H. Chaimovich, R. J. Vaughan and F. H. Westheimer, J. Am. Chem. Soc., 1968, 90, 4088.
- 95. L. Horner and E. Spietschka, Chem. Ber., 1955, 88, 934.
- 96. D. Redmore and C. D. Gutsche, Adv. Alicyclic Chem., 1971, 3, 125.
- 97. K. Ueda and F. Toda, Chem. Lett., 1975, 779; K. Ueda, M. Igaki and F. Toda, Bull. Chem. Soc. Jpn., 1976, 49, 3173.
- 98. J. Meinwald and Y. C. Meinwald, Adv. Alicyclic Chem., 1966, 1, 1.
- 99. J. Meinwald and P. G. Gassman, J. Am. Chem. Soc., 1960, 82, 2857.
- 100. J. Meinwald and J. K. Crandall, J. Am. Chem. Soc., 1966, 88, 1292.
- K. B. Wiberg, B. R. Lowry and T. H. Colby, J. Am. Chem. Soc., 1961, 83, 3998.
 J. Meinwald, C. B. Jensen, A. Lewis and C. Swithenbank, J. Org. Chem., 1964, 29, 3469.
 H. Tanida and Y. Hata, J. Am. Chem. Soc., 1966, 88, 4289; 1969, 91, 1170.
- 104. K. B. Wiberg and R. W. Ubersax, Tetrahedron Lett., 1968, 3063.
- 105. K. B. Wiberg and B. A. Hess, Jr., J. Org. Chem., 1966, 31, 2250.
- 106. L. R. Smith and J. Meinwald, J. Org. Chem., 1977, 42, 415.
- T. Miyashi, T. Nakajo and T. Mukai, J. Chem. Soc., Chem. Commun., 1978, 442. 107.
- 108. T. Tsuji, Z. Komiya and S. Nishida, Tetrahedron Lett., 1980, 21, 3583.
- 109. P. R. Brook and B. V. Brophy, Tetrahedron Lett., 1969, 4187.
- 110. K. B. Wiberg, B. L. Furtek and L. K. Olli, J. Am. Chem. Soc., 1979, 101, 7675. 111. L. Horner, W. Kirmse and K. Muth, Chem. Ber., 1958, 91, 430; L. Horner, K. Muth and H.-G. Schmelzer, Chem. Ber., 1959, 92, 2953; M. P. Cava, R. L. Litle and D. R. Napier, J. Am. Chem. Soc., 1958, 80, 2257; M. P. Cava and R. J. Spangler, J. Am. Chem. Soc., 1967, 89, 4550.
- 112. M. P. Cava and E. Moroz, J. Am. Chem. Soc., 1962, 84, 115; J. Meinwald, G. G. Curtis and P. G. Gassman, J. Am. Chem. Soc., 1962, 84, 116; G. Muller, C. Huynh and J. Mathieu, Bull. Soc. Chim. Fr., 1962, 296; J. L. Mateos, O. Chao and H. Flores R, Tetrahedron, 1963, 19, 1051; J. L. Mateos and R. Pozas, Steroids, 1963, 2, 527; A. Hassner, A. W. Coulter and W. S. Seese, Tetrahedron Lett., 1962, 759.
- 113. M. P. Cava and B. R. Vogt, Tetrahedron Lett., 1964, 2813; M. P. Cava and B. R. Vogt, J. Org. Chem., 1965, 30, 3775.
- S. Huneck, Tetrahedron Lett., 1963, 375; S. Huneck, Chem. Ber., 1965, 98, 1837, 2284. 114.
- 115. S. Wolff and W. C. Agosta, J. Chem. Soc., Chem. Commun., 1981, 118.
- 116. V. B. Rao, C. F. George, S. Wolff and W. C. Agosta, J. Am. Chem. Soc., 1985, 107, 5732.
- 117. W. Reid and H. Lohwasser, Justus Liebigs Ann. Chem., 1965, 683, 118.
- 118. (a) G. Lowe and D. D. Ridley, J. Chem. Soc., Chem. Commun., 1973, 328; G. Lowe and D. D. Ridley, J. Chem. Soc., Perkin Trans. 1, 1973, 2024; (b) G. Stork and R. P. Szajewski, J. Am. Chem. Soc., 1974, 96, 5787.
- 119. C. J. Moody, C. J. Pearson and G. Lawton, Tetrahedron Lett., 1985, 26, 3167; C. J. Moody, C. J. Pearson and G. Lawton, J. Chem. Soc., Perkin Trans. 1, 1987, 877; G. Lawton, C. J. Moody, C. J. Pearson and D. J. Williams, J. Chem. Soc., Perkin Trans 1, 1987, 885.
- 120. W. G. Dauben and D. L. Whalen, Tetrahedron Lett., 1966, 3743.
- 121. P. Yates and R. J. Crawford, J. Am. Chem. Soc., 1966, 88, 1562.
- 122. B. M. Trost and P. L. Kinson, J. Am. Chem. Soc., 1970, 92, 2591; B. M. Trost and P. L. Kinson, Tetrahedron Lett., 1973, 2675
- 123. M. Jones, Jr. and W. Ando, J. Am. Chem. Soc., 1968, 90, 2200.
- 124. M. P. Cava, P. M. Weintraub and E. J. Glamkowski, J. Org. Chem., 1966, 31, 2015.
- 125. R. Yamaguchi, K. Honda and M. Kawanisi, J. Chem. Soc., Chem. Commun., 1987, 83.

- H. Stetter and K. Kiehs, Chem. Ber., 1965, 98, 1181, 2099; H. Schwall, V. Schmidt and B. Eistert, Chem. Ber., 1969, 102, 1731; H. Veschambre and D. Vocelle, Can. J. Chem., 1969, 47, 1981; W. D. Barker, R. Gilbert, J.-P. Lapointe, H. Veschambre and D. Vocelle, Can. J. Chem., 1969, 47, 2853.
- 127. B. Eistert, H. Elias, E. Kosch and R. Wollheim, Chem. Ber., 1959, 92, 130; B. Eistert, G. Bock, E. Kosch and F. Spalink, Chem. Ber., 1960, 93, 1451.
- 128. B. Eistert and G. Heck, Justus Liebigs Ann. Chem., 1965, 681, 123; B. Eistert, D. Greiber and I. Caspari, Justus Liebigs Ann. Chem., 1962, 659, 64.
- O. Süs, Justus Liebigs Ann. Chem., 1944, 556, 65, 85; 1947, 557, 237; 1953, 579, 133; J. de Jonge and R. Dijkstra, Recl. Trav. Chim. Pays-Bas, 1948, 67, 328; J. de Jonge, R. J. H. Alink and R. Dijkstra, Recl. Trav. Chim. Pays-Bas, 1950, 69, 1448.
- 130. O. Süs, Z. Wiss. Photogr., Photophys. Photochem., 1955, 50, 476 (Chem. Abstr., 1956, 50, 8355e).
- 131. O. Süs, M. Glos, K. Möller and H.-D. Eberhardt, Justus Liebigs Ann. Chem., 1953, 583, 150; O. Süs, K. Möller, R. Dietrich, H.-D. Eberhardt, M. Glos, M. Grundkötter, H. Hoffmann and H. J. Schäfer, Justus Liebigs Ann. Chem., 1955, 593, 91.
- O. Süs, K. Möller and H. Heiss, Justus Liebigs Ann. Chem., 1956, 598, 123; O. Süs and K. Möller, Justus Liebigs Ann. Chem., 1956, 599, 233; 1958, 612, 153; O. Süs, H. Steppan and R. Dietrich, Justus Liebigs Ann. Chem., 1958, 617, 20.
- B. Stanovnik, M. Tišler, B. Kirn and I. Kovač, J. Heterocycl. Chem., 1979, 16, 195; B. Stanovnik, M. Tišler and J. T. Carlock, Synthesis, 1976, 754; W. Kloetzer, G. Doerler, B. Stanovnik and M. Tisler, Heterocycles, 1984, 22, 1763.
- 134. Y. Tamura, H. Ikeda, C. Mukai, S. Mohamad, M. Bayomi and M. Ikeda, Chem. Pharm. Bull., 1980, 28, 3430.
- 135. L. Horner, H.-G. Schmelzer, H.-U. von der Eltz and K. Habig, Justus Liebigs Ann. Chem., 1963, 661, 44.
- W. Kirmse, Angew. Chem., 1959, 71, 539; R. Tasovac, M. Štefanović and A. Stojiljković, Tetrahedron Lett., 1967, 2729; R. F. Borch and D. L. Fields, J. Org. Chem., 1969, 34, 1480.
- 137. P. J. Whitman and B. M. Trost, J. Am. Chem. Soc., 1969, 91, 7534.
- K. B. Wiberg and A. de Meijere, Tetrahedron Lett., 1969, 519; A. J. Ashe, III, Tetrahedron Lett., 1969, 523;
 P. G. Gassman, F. J. Williams and J. Seter, J. Am. Chem. Soc., 1968, 90, 6893.
- 139. A. T. Blomquist and F. W. Shlaefer, J. Am. Chem. Soc., 1961, 83, 4547; see also A. Stojiljković and R. Tasovac, Tetrahedron, 1977, 33, 65.
- 140. N. L. Allinger, L. A. Freiberg, R. B. Hermann and M. A. Miller, J. Am. Chem. Soc., 1963, 85, 1171.
- 141. W. T. Brady, Synthesis, 1971, 415; R. W. Holder, J. Chem. Educ., 1976, 53, 81.
- 142. P. Yates and A. G. Fallis, Tetrahedron Lett., 1968, 2493.
- 143. P. K. Freeman and D. G. Kuper, Chem. Ind. (London), 1965, 424; J. Meinwald and G. H. Wahl, Jr., Chem. Ind. (London), 1965, 425.
- A. B. Smith, III, J. Chem. Soc., Chem. Commun., 1974, 695; A. B. Smith, III, B. H. Toder and S. J. Branca, J. Am. Chem. Soc., 1976, 98, 7456; S. J. Branca, R. L. Lock and A. B. Smith, III, J. Org. Chem., 1977, 42, 3165; A. B. Smith, III, B. H. Toder and S. J. Branca, J. Am. Chem. Soc., 1984, 106, 3995; A. B. Smith, III, B. H. Toder, R. E. Richmond and S. J. Branca, J. Am. Chem. Soc., 1984, 106, 4001.
- 145. J. P. Lokensgard, J. O'Dea and E. A. Hill, J. Org. Chem., 1974, 39, 3355.
- 146. B. Saha, G. Bhattacharjee and U. R. Ghatak, *Tetrahedron Lett.*, 1986, 27, 3913; see also P. Ceccherelli, M. Curini, B. Porter and E. Wenkert, J. Org. Chem., 1984, 49, 2052.
- 147. J. March, in 'Advanced Organic Chemistry: Reactions, Mechanisms and Structure', 3rd edn., Wiley-Interscience, New York, 1985, p. 983.
- 148. T. L. Gilchrist and C. W. Rees, 'Carbenes, Nitrenes and Arynes', Nelson, London, 1969.
- 149. B. M. Trost, J. Am. Chem. Soc., 1966, 88, 1587; 1967, 89, 138.
- 150. E. J. Corey and M. Chaykovsky, J. Am. Chem. Soc., 1964, 86, 1640.
- 151. R. J. Mulder, A. M. van Leusen and J. Strating, Tetrahedron Lett., 1967, 3057.
- 152. M. Regitz, W. Anschütz, W. Bartz and A. Liedhegener, Tetrahedron Lett., 1968, 3171; M. Regitz, Angew. Chem., Int. Ed. Engl., 1975, 14, 222; M. Regitz, F. Bennyarto and H. Heydt, Liebigs Ann. Chem., 1981, 1044.
- 153. W. Jugelt and D. Schmidt, Tetrahedron, 1969, 25, 5569.
- 154. E. M. Burgess, R. Carithers and L. McCullagh, J. Am. Chem. Soc., 1968, 90, 1923.
- W. Kirmse and L. Horner, Justus Liebigs Ann. Chem., 1958, 614, 4; K.-P. Zeller, H. Meier and E. Müller, Tetrahedron Lett., 1971, 537; P. Krauss, K.-P. Zeller, H. Meier and E. Müller, Tetrahedron, 1971, 27, 5953; K.-P. Zeller, H. Meier and E. Müller, Justus Liebigs Ann. Chem., 1972, 766, 32; H. Meier and H. Buehl, J. Heterocycl. Chem., 1975, 12, 605.
- 156. H. Meier and I. Menzel, J. Chem. Soc., Chem. Commun., 1971, 1059; H. Meier and I. Menzel, Tetrahedron Lett., 1972, 445; H. Meier and E. Voigt, Tetrahedron, 1972, 28, 187.

3.10 The Stevens and Related Rearrangements

ISTVÁN E. MARKÓ

University of Sheffield, UK

3.10.1 INTRODUCTION	913
3.10.2 DEFINITIONS	916
3.10.3 PREPARATION OF NITROGEN AND SULFUR YLIDES	918
3.10.3.1 Indirect Methods 3.10.3.2 Direct Formation of Ylides	918 919
3.10.4 1,2-SIGMATROPIC REARRANGEMENTS	921
3.10.4.1 Synthetic Applications of the Stevens Rearrangement 3.10.4.2 Transfer of Chirality	921 927
3.10.5 3,2-SIGMATROPIC REARRANGEMENTS	932
 3.10.5.1 Introduction 3.10.5.2 Synthetic Applications in Alicyclic Systems 3.10.5.3 Configuration of the Newly Formed Double Bond 3.10.5.4 The Erythro-Threo Ratio 3.10.5.5 Transfer of Configuration 3.10.5.6 Ring Contraction Reactions 3.10.5.7 Ring Expansion Reactions 3.10.5.8 Ylidic Rearrangements Involving Alkynes and Allenes 3.10.5.9 Sommelet-Hauser Rearrangement 	932 933 943 949 953 954 957 963 965
3.10.6 CONCLUSIONS	97 1
3.10.7 REFERENCES	972

3.10.1 INTRODUCTION

Nitrogen and especially sulfur ylides are used extensively in synthetic organic chemistry. Our understanding of the sigmatropic rearrangements of these ylidic intermediates has matured over the last few decades. These rearrangements, which often proceed with high regio-, stereo- and enantio-selectivities, have become a powerful synthetic tool.

In 1928, T. S. Stevens reported the first example of a reaction which was later to be called the Stevens rearrangement.¹ He found that, upon treatment with aqueous alkali, phenacylbenzyldimethylammonium bromide (1) was converted into 1-benzoyl-2-benzyldimethylamine (2) in high yield (Scheme 1, equation a). Soon after, in 1932, Stevens reported the analogous rearrangement of the corresponding sulfonium derivative.² The sulfonium bromide (3) was smoothly transformed into a sulfide, which was initially formulated as structure (4), when (3) was treated with hot alkali (Scheme 1, equation b). Subsequent work, however, established that the correct structure of the rearranged material was (84) (Scheme 19).



A variety of groups, including allyl, propargyl and phenacyl, were also shown to be good migrating groups in ylidic rearrangements.³ Mechanistic investigations, using ¹⁴C-labeled starting material, as well as cross-over experiments, demonstrated the reaction to be intramolecular.⁴ Subsequent studies of the effect of substituents on the phenyl ring on the migrating ability of the benzyl group in reactions of (1) gave the order p-NO₂ > p-Hal > p-Me > p-MeO.⁵ This observation led Stevens to postulate that the reaction proceeded by initial formation of an ylidic intermediate (5) which, by heterolysis, yielded a benzylic anion and an iminium ion. This ion pair could then lead to the observed product (Scheme 2). Later, Kenyon showed that the migrating group retained its configuration during the course of the rearrangement.⁶ The optically-active ammonium salt (7), for example, was converted into an optically-active amine (9) with retention of chirality at the benzylic chiral carbon atom. Wittig⁷ and Hauser⁸ then proposed that the reaction was a concerted intramolecular displacement of the migrating group by the carbanionic center of the ylidic intermediate (8; Scheme 3).



If the Stevens rearrangement is a concerted reaction, then the Woodward and Hoffmann rules of conservation of orbital symmetry demonstrate that it is a symmetry-forbidden reaction.⁹ Later on, Ollis reinvestigated the reaction and demonstrated, in a very elegant manner, that the Stevens rearrangement proceeds by radical pair intermediates.¹⁰ These radicals recombine quickly while being held within a solvent cage, a fact that explains the high intramolecularity and stereoselectivity observed in these rearrangements (Scheme 4).

Another type of ylidic rearrangement, the Sommelet-Hauser rearrangement, was discovered later, in 1937, by Sommelet and studied extensively by Sommelet and by Hauser.^{11,12} This transposition involves the base-promoted rearrangement of non stabilized ammonium and sulfonium ylides. Thus, the ammonium salt (11), when heated with alkali, gave cleanly the substituted diphenylmethane derivative (12; Scheme 5). Wittig found that the dibenzyl ammonium salt (13) reacted with phenyllithium giving two



products, (14) and (15).¹³ The product (15) is produced by a competing Stevens rearrangement. Subsequent work led to the proposed mechanism for the Sommelet-Hauser rearrangement as depicted in Scheme 6.¹⁴ The ylide (17), formed upon base treatment of the ammonium salt (16), attacks the aromatic ring, leading to the unstable intermediate (19), which may be isomerized into the substituted aromatic derivative (20). It was also shown later that the two equilibrating ylides (17) and (18) were involved in these reactions; the ylide (18) being associated with the Sommelet-Hauser rearrangement.¹⁵



Scheme 5



Scheme 6

In 1969, Ollis recognised that the rearrangement of allylic ammonium and sulfonium ylides was a facile process of considerable synthetic interest,¹⁶ and Baldwin reported his generalization of the sigmatropic rearrangement processes.¹⁷ Since then, numerous examples of the synthetic utility of ylidic rearrangements have been recorded in the chemical literature and this number is growing continuously.

3.10.2 DEFINITIONS

With some substrates, several types of ylidic rearrangements can occur simultaneously, though by judicious choice of experimental conditions, this competition can be controlled. This is of paramount importance in the utilization of these rearrangements in synthesis. The two examples described below illustrate this point.

When the hypothetical ylide (21) is generated, then, in principle, a 1,2-, 3,2- or 5,2-rearrangement can occur, leading to the three products (22), (23) and (24) respectively (Scheme 7). [The 3,2-rearrangement is often reported in the literature as a 2,3-sigmatropic rearrangement. The author believes that, if the other reactions in Scheme 7 are called 1,2- and 5,2-sigmatropic rearrangements respectively, then logic would dictate that this reaction is a 3,2-sigmatropic rearrangement. We would like to encourage the use of the 'correct' 3,2 nomenclature rather than the 'incorrect', but popular, 2,3 one.] Similarly, when examining ylide (25), four different products (26), (27), (28) and (29) can, in principle, be identified: they are those derived from 1,2-, 1,4-, 3,2- and 3,4-sigmatropic rearrangement respectively (Scheme 8).





The 1,4- and 3,2-rearrangements are thermal, concerted, orbital symmetry allowed processes possessing suprafacial-suprafacial characteristics. They are facile reactions, occurring readily and under mild conditions. The 3,2-rearrangement will take place with allylic inversion, as demanded by orbital symmetry control. Thus, the sulfonium salt (30) gave the allylic sulfide (31), whereas (32) gave the isomerically pure (33; Scheme 9).¹⁸

The 1,4-rearrangement, though symmetry-allowed, is rarely observed.¹⁹ The 1,2-, 5,2- and 3,4-rearrangements are symmetry-forbidden processes which would have to take place *via* a suprafacial–anta-rafacial mode. They have been shown to proceed by radical pair intermediates.²⁰



Scheme 9

3.10.3 PREPARATION OF NITROGEN AND SULFUR YLIDES

3.10.3.1 Indirect Methods

The oldest and most commonly used method for ylide preparation is the so-called 'salt-method', where a sulfonium or ammonium cation, generated by alkylation of the corresponding sulfide or amine, is deprotonated by a strong base (Scheme 10, path a). In general, an active alkylating agent is required. Primary allylic and benzylic halides, as well as α -halocarbonyl derivatives,²¹ produce the ammonium or sulfonium salts fairly readily. A serious side reaction, which characterizes this method, is the competitive dealkylation of the salt by any nucleophilic halide ion produced (Scheme 10, path b).²² Therefore, exchange of the halide ion for a non-nucleophilic counterion employing a metal salt, *e.g.* Ag⁺, is performed routinely. Alternatively, the use of Meerwein salts,²³ *e.g.* Me₃O⁺BF₄⁻ (Scheme 11, equation a), or the more recently introduced triflate methodology²⁴ (Scheme 11, equation b) give directly the stable ammonium or sulfonium salts. The counterions in these cases, are poor nucleophiles unable to dealkylate the cations (Scheme 11).



Scheme 10



Scheme 11

A second side reaction using this approach is the Hoffmann elimination of the sulfide or amine, during attempted deprotonation of the sulfonium or ammonium salt. This reaction is especially important in the case of ammonium ylides. Finally, another problem which surfaces during the deprotonation step is the formation of a mixture of ylides, *e.g.* (45) and (46) from (44), which undergo competing rearrangements to give (47) and (48; Scheme 12).²⁵

An elegant solution to these problems was proposed independently by Vedejs²⁶ and Sato.²⁷ They generated ylides regiospecifically, under nonbasic conditions, using fluoride-catalyzed desilylation of



Scheme 12

trimethylsilyl ammonium or sulfonium salts. Not only is the extent of elimination considerably reduced, but also one reaction pathway predominates easily over the other one (Scheme 13).





3.10.3.2 Direct Formation of Ylides

The addition of a carbene or benzyne to a sulfide, and to a lesser extent to an amine, provides a direct route to ylides. Enormous progress has been realized with this approach, mainly as a result of the efforts of Ando²⁸ and Doyle.²⁹ Most noteworthy amongst examples using benzyne is the squalene synthesis reported by Ollis, in which the sulfide (55) was transformed by a 3,2-rearrangement into the squalene precursor (58) and eventually into squalene itself (Scheme 14).³⁰

The stereospecific rearrangement of *exo*-methylenecyclohexane derivatives discovered by Evans employed dichlorocarbene, generated under phase transfer conditions (Scheme 15).³¹

An extremely versatile approach to ylides involves the light-promoted or metal-catalyzed generation of carbenes from diazonium salts. In most cases, metal-catalyzed reactions give better yields and cleaner reaction mixtures than do corresponding photolytic techniques (Scheme 16, equation a).³² A classical example is to be found in penicillin chemistry (Scheme 16, equation b).³³ However, copper catalysts have their limitations in the generation of carbenoid intermediates. These limitations can be largely over-



(59) Squalene





Scheme 15

come by employing rhodium catalysts that not only afford the desired material when the copper catalysts fail to react, but also allow the reactions to be performed under much milder conditions. It is quite remarkable that rhodium catalysts allow the preparation and reaction of some oxonium and halonium ylides.³⁴ It is only recently that their synthetic use has started to be fully appreciated (Scheme 17).



3.10.4 1,2-SIGMATROPIC REARRANGEMENTS

3.10.4.1 Synthetic Applications of the Stevens Rearrangement

Although studied at great length from a mechanistic point of view, the Stevens reactions of nitrogen and sulfonium ylides have, so far, been applied in synthesis only to a limited extent. The details of the earlier work by Stevens have been summarized elsewhere,³⁵ and some excellent reviews on ylide chemistry mention a few synthetic uses of this rearrangement.³⁶ One of the classical problems associated with the 1,2-rearrangement is the co-ocurrence of competing reactions, including other rearrangements, *i.e.* 1,3 and 1,2 as well as elimination reactions (more frequent for nitrogen ylides than for sulfur ones). For

example, rearrangement of the ylide (79) gives the 1,2- and 3,2-products in 1:3 ratio, indicating the greater propensity for the concerted, symmetry-allowed 3,2-rearrangement to occur, in relation to the forbidden 1,2-rearrangement (Scheme 18).³⁷





An important breakthrough was the discovery that the ratio 1,2:3,2 was highly dependent upon the nature of the solvent and the concentration of the base. Under aprotic conditions, the Stevens rearrangement of ylide (**82**) takes place nearly exclusively,^{35a,38} whereas the 3,2-rearrangement is favored under protic basic conditions (Scheme 19).^{37,39} Not only can the 1,2:3,2 ratio be controlled, but the relative proportion of the 3,2-products (**83**) and (**84**) has been shown to depend strongly on the concentration of the base employed in these reactions.³⁷ This striking result can be understood if one looks at the fate of the primary product (**85**) obtained in the Sommelet–Hauser reaction. This intermediate can participate in two different pathways. Aromatization requires a 1,3-prototropic shift and, therefore, the presence of a proton source or protic solvent. In aprotic solvent, this transformation cannot take place and a competing homolysis will occur followed by a radical recombination giving the 1,2-product (Scheme 20).

Intermediates such as (85) have been isolated and reacted under different conditions to give the expected rearrangement products.³⁷ A few examples of simple Stevens rearrangements are presented in Table 1.

More complex and synthetically useful applications of the Stevens rearrangement have been reported. Kondo, for example, has shown that the ylide (89), formed from (88) by an intramolecular carbene re-





action, undergoes a 1,2-sigmatropic shift to give the five-membered ring (90) in 51% yield.⁴² Even smaller ring compounds, *e.g.* (94), can be prepared following this route (Scheme 21). When two ylides of similar stability can be formed and rearranged, poor selectivity is often observed. Thus, irradiation of (95) leads to the equilibrating ylides (96) and (97). It is therefore not too surprising that two Stevens products (98 and 99) are obtained from this reaction (Scheme 22).





In some cases, the Stevens rearrangement can be used to form ring-expanded products. Thus, thietane (100) reacts with dimethyl diazomalonate in the presence of a copper catalyst to provide the functionalized tetrahydrothiophene (102).⁴³ A similar and most remarkable example, where a possible competing 3,2-rearrangement is not observed, is the ring expansion of dihydrothiophene (103) to (105; Scheme 23).⁴⁴ Clearly, the increase in strain in the transition state (106), which would lead to the 3,2-rearrangement product (107) is responsible for this interesting selectivity. In general, nitrogen ylides behave in an analogous manner though sometimes they can supersede sulfur ylides (Scheme 24).

Ollis has demonstrated that the successful transformation of the dihydropyrrole (110) into the tetrahydropyridine (111) is a fairly easy process.⁴⁵ In sharp contrast, the analogous sulfonium salt (112) does not form the expected Stevens product (113) but rather (114), a product of dimerization.

Recent and interesting examples of Stevens reactions were reported by Doyle who reacted dithianes with ethyl diazoacetate in the presence of a rhodium catalyst.^{34b} Seven-membered ring bissulfides were

Table 1							
Entry	Salt	Conditions	Product	Yield (%)	Ref.		
1	$Ph + SMe BF_4$ $Ph + SMe BF_4$	NaOH, H ₂ O	Ph- Ph- Ph	21	37		
2	Ph $^{+}NMe_3 Br^{-}$	PhLi	Ph NMe_2	32	7		
3	$X \qquad Me O \\ S \qquad F \qquad Ph \qquad Br^{-}$	Δ, THF or acetone	$X \qquad SMe \\ Ph \\ O \\ X = F, Cl or Br$	70–80	35a		
4	Me Me N H Br N N H NO_2	180 °C	Me_2N-N	50	40		
5	Me Br^-	3% Na, EtOH	N-Me Ph	75	41		







obtained in variable yields, depending on the substituents present in the starting dithianes (Scheme 25). With 2-phenyl-1,3-dithiane (115), smooth Stevens 1,2-rearrangement gave (116). However, with 2-phenyl-2-methyl-1,3-dithiane (117), elimination becomes a competing process. In this case, the Stevens 1,2-rearrangement product (119) is the minor component of the reaction.

A most remarkable result is obtained when allylic halides are reacted with ethyl diazoacetate in the presence of metal catalysts.^{34a} The 1,2-rearrangement product (122) and the 3,2-rearrangement product (121) can be isolated and their relative proportions shown to be related to the nature of the catalyst em-



ployed (Scheme 26, equation a). A competing side reaction, cyclopropanation, is dependent upon the nature of the halogen atom (X) and it can be totally suppressed when X is iodide (Scheme 26, equation b). Stevens rearrangements are not limited only to heteroatom to carbon migrations; heteroatom to hetero-

atom shifts are also observed frequently. Two such examples are given in Scheme 27.^{46,47} In this case



too, the 1,2- and 3,2-rearrangements are competing but their relative importance has been shown to be substituent dependent.

Despite extensive mechanistic investigations, the Stevens 1,2-rearrangement has found only a few applications in synthesis. Standing out, however, as a landmark in this area is the beautiful work of Boekelheide, who utilized the 1,2-rearrangement of sulfur-containing macrocycles in an elegant and general approach towards cyclophanes (Scheme 28).⁴⁸ This brilliantly conceived strategy has been applied in numerous cases and two recent examples are shown in Scheme 29.^{49,50}

3.10.4.2 Transfer of Chirality

If the 1,2-rearrangement is a concerted process, then it is a disfavored one. As a consequence, there has been considerable interest in determining the stereochemistry of the migrating group. If the reaction proceeds *via* a suprafacial-antarafacial mode, inversion of configuration at the migrating center should result. In fact, a high degree of retention of configuration was observed by Kenyon⁶ and Brewster⁵¹ as early as 1952 (Scheme 30, equation a). Schollkopf⁵² and Stevens⁵³ reinvestigated the reaction and found it to proceed with at least 95% *ee*. The same conclusion was reached by Lown several years later using optically active benzylamine (Scheme 30, equation b).⁵⁴ In 1962, Jenny and Bruey showed that retention of configuration was also very high for the analogous allylic system (Scheme 30, equation c).⁵⁵ Mislow reported the fascinating case where total transmission of configuration, from a center of chirality involving a nitrogen atom into a center of chirality involving a carbon atom, took place (Scheme 31).⁵⁶

When the optically active diazepinium salt (151) was treated with phenyllithium, only the (S)-amine (152) was obtained. However, the (S)-amine (152) consists of an equilibrating mixture of (+) and (-) diastereomeric amines which interconvert by mutarotation upon heating. Similarly, Brewster demonstrated that the pure (R)-isomer of the spiro ammonium salt (153) was transformed into the alkaloid (155) having an (S) absolute configuration (Scheme 32).⁵⁷



Scheme 28





Chan and Hill have studied the transmission of configuration in ammonium salts where the only chiral center is the quaternary nitrogen atom.⁵⁸ The Stevens rearrangement proceeded in poor yield; a competing 1,4-rearrangement product being surprisingly formed as the major product, and gave the amine (157) with 58% *ee* (Scheme 33).





Ollis and coworkers reinvestigated the stereochemical outcome of the Stevens rearrangement and provided a detailed understanding of the mechanism of this reaction.^{10,59} Rearrangement of the optically pure ammonium salt (143) gave three products, (144), (159) and (160) in a ratio of 80:11:5 (Scheme 34). The amine (144) was optically active and degradation studies proved that migration occurred with more

than 95% retention of the stereochemistry at the terminus of the migrating group. The diamine (159) (and similarly the alkane 160) was a mixture of the racemate and the meso compound. Clearly, these products (159) and (160) arose from coupling of radicals which have escaped from the solvent cage. A few related examples are represented in Table 2.¹⁰

In all the cases discussed so far, retention of configuration at the terminus of the migrating group has been observed. The degree of retention was influenced strongly by the nature of the substituents present on the migrating group (Table 2, entries 1–3). It is also worth noting that the extent of retention of configuration in the case of sulfonium ylides (entry 4) is substantially lower than in the case of the ammonium analog. Comparison between entry 1 and entry 5 demonstrates how the nature of the substituent influences the enantiomeric excess. All these results can be explained by the mechanism, involving diastereomeric radical pairs, proposed by Ollis for the Stevens rearrangement and summarized in Scheme



Table 2



35.60 Homolytic fission of the chiral ammonium ylide (143) generates two diastereomeric radical pairs which evolve by two different routes: (a) cage recombination leads to the two diastereomeric ketoamines (161) and (162), where the migrating center has retained its absolute stereochemistry; (b) radical escape gives achiral free radicals that combine to yield racemic products (Scheme 35).





The radical pair recombination is a very fast process. This explains the high intramolecularity of the Stevens 1,2-rearrangement as well as the high stereoselectivity observed. Retention of configuration is ensured by the absence of rotation of the radicals in the solvent cage. The formation of the two diastereoisomeric amines (161) and (162) can be explained by the presence of an equilibrium mixture of conformational isomers of the starting ylide.⁶⁰ Indeed, the ylide (143) exists in the planar cisoid conformation C, which is in equilibrium with the two orthogonal conformational isomers A and B. These are the conformations from which migration will occur, since the σ -bond to be broken is parallel to the π -system of the enolate, ensuring maximum stabilization of the radicals as they are formed. Conformation A will lead to the radical pair (163), which eventually collapses to form the diastereoisomer (162). The same process involving conformation B leads to the diastereoisomer (161; Scheme 36).

3.10.5 3,2-SIGMATROPIC REARRANGEMENTS

3.10.5.1 Introduction

In sharp contrast with the Stevens 1,2-rearrangement, the 3,2-sigmatropic rearrangement of ylidic intermediates has been used extensively in organic synthesis. These rearrangements, being concerted symmetry-allowed processes, show high regio-, diastereo- and enantio-selectivities. It is, therefore, not too surprising that they have become rapidly accepted synthetic tools, since Baldwin published his important papers on the generalization of sigmatropic rearrangements.¹⁷



3.10.5.2 Synthetic Applications in Alicyclic Systems

3,2-Rearrangements of ylidic intermediates proceed with allylic transposition. This useful property is the key feature of one of the best preparations of 1,5-dienes (Scheme 9). Trost has used this approach to synthesize yomogi alcohol (165) in a particularly simple way by S-alkylation of sulfide (31) and solvolysis of the resulting sulfonium ion (Scheme 37).⁶¹

One of the landmarks in the early use of the 3,2-rearrangement of sulfur ylides is the squalene synthesis reported by Ollis and already discussed in Section 3.10.4.³⁰ This approach used benzyne to generate the ylidic intermediate. Frequently, however, benzyne produces a mixture of products (Scheme 38).⁶² The sulfide (168) may be formally regarded as the expected 3,2-sigmatropic rearrangement product, whereas (169) results from a competitive 1,4-rearrangement. Use of carbenes instead of benzyne gener-



Scheme 37

ally gives higher yields of 3,2-rearrangement products. Ando has shown that crotyl phenyl sulfide (170) forms mainly the sulfide (171) when the carbene is generated using $CuSO_4$.^{44,63} In contrast, the light-induced reaction affords the desired material in much lower yield (Scheme 39). Evans also used carbenes to generate sulfonium ylides of rigid systems in order to study the diastereoselectivity of the 3,2-rearrangement (Scheme 40).³¹



Scheme 39

John has investigated the BF₃-etherate and copper-catalyzed rearrangement of diazopenicillins in the presence of allylic sulfides and selenides.⁶⁴ The results are presented in Scheme 41. The reaction is believed to involve the initial formation of ylide (177) followed by a 3,2-sigmatropic rearrangement. Support for this mechanism came from utilization of substituted allylic sulfides which were shown to rearrange with net allylic transposition, a result expected on the basis of orbital symmetry considerations (Scheme 42). A similar rearrangement involving nitrogen ylides has been reported by Baldwin (Scheme 43).⁶⁵

Doyle has shown that the rhodium-catalyzed reaction of allylic sulfides and amines with ethyl diazoacetate produced smoothly the products of 3,2-rearrangement.³⁴ In contrast with the copper-catalyzed reaction, allylic amines can be used and the yields are good to high (Scheme 44); virtually no cyclopropanation is observed. These observations demonstrate the superiority of rhodium catalysts compared with either copper ones or the use of light.

Vedejs has also discussed the inefficiency of copper catalysts in ylidic 3,2-sigmatropic processes.⁶⁶ Optimization of ring expansion reactions using diazomalonates were found to be unsatisfactory and analogous reactions using diazoketones were totally unsuccessful. Takano has taken advantage of a rhodium-promoted ylide formation followed by a 3,2-rearrangement, in a useful synthesis of γ , δ -unsaturated carbonyl compounds (Scheme 45).⁶⁷

Julia, in a very elegant approach, has used sulfur ylidic 3,2-rearrangements to produce β , γ -unsaturated aldehydes (Scheme 46).⁶⁸ Following the same idea, Kociensky also used ylidic 3,2-sigmatropic rearrangements to prepare β , γ -unsaturated aldehydes; he employed sulfur ylides in conjunction with silicon substituents (Scheme 47).⁶⁹ In contrast, Mander used the nitrogen ylidic 3,2-rearrangement as an easy route towards the synthesis of β , γ -unsaturated aldehydes (Scheme 48).⁷⁰ It is noteworthy that better















Entry X

PhS BF3•Et2O 1 BF3•Et2O MeS 2 3 PhSe BF3•Et2O 4 Cu(acac)₂ PhS 5 MeS $Cu(acac)_2$ 6 PhSe $Cu(acac)_2$

Catalyst



(180) (%) 5 16

15

0

0

0

87:13 80:20 50:50

Scheme 41

47

49

48

65

60

64



overall yields are obtained using ammonium rather than sulfonium ylides. Following the same theme, Lythgoe has used 1,3-dithiane as the masked aldehydic equivalent.⁷¹ S-Alkylation of dithiane using allylic halides followed by a sulfur ylidic 3,2-rearrangement gave the homoallylic dithiane (209), which was hydrolyzed smoothly to the β , γ -unsaturated aldehyde (199; Scheme 49).

Buchi has used Mander's approach as a key step in his synthesis of α -sinensal (212; Scheme 50),⁷² and Corey has applied the ylidic 3,2-rearrangement to the efficient preparation of 3-cyclopentenones.⁷³ Thus, base treatment of the sulfonium salt (213) first gave the rearranged product (215). Thermal vinylcyclopropane ring expansion then produced the spiro compound (216). Deprotection finally gave the nonconjugated cyclopentenone (217; Scheme 51).



Cohen has used a soluble version of the Simmons–Smith reagent to promote the 3,2-rearrangement of allylic sulfides (Scheme 52).⁷⁴ He has also applied this methodology to the synthesis of sarkomycin (225).⁷⁵ The allylic sulfide (221), easily available *via* a Petrow reaction, was first alkylated and the sulfonium ylide was then generated by a fluoride-promoted desilylation. Rearrangement of the ylide (223) gave the homoallylic sulfide (224) which was eventually transformed into sarkomycin (Scheme 53).



Scheme 48





The formation and 3,2-sigmatropic rearrangement of sulfonium and ammonium ylides, generated by fluoride-promoted desilylation, has also been actively investigated by Vedejs. He showed that ylides, produced under kinetic control, could be prepared using this technique and rearranged by a 3,2-sigmatropic process. This rearrangement can take place before equilibration to the thermodynamically more stable ylide occurs (Scheme 13). In yet another version, Zbiral has used silyl-stabilized ylides to produce homoallylic sulfides.⁷⁶ An initial Peterson reaction between the ylide (**227**) and a carbonyl compound (**228**) is followed by a deprotonation to generate another ylide (**229**). Equilibration of the thermodynamically more stable ylide (**229**) gives a small amount of the isomer (**230**), which undergoes rapid 3,2-sigmatropic rearrangement, leading to the observed reaction product (Scheme 54).

Ratts and Yao reported a particularly interesting preparation of enol ethers by a 3,2-sigmatropic transposition in which the carbonyl function of stabilized ylides participates in the rearrangement (Scheme 55).³⁹ The same starting sulfonium ylide is well known to give Stevens rearrangement products. The competition between the 1,2- and 3,2-pathways is strongly dependent on the reaction conditions and good yields of either product can be obtained by carefully selecting the base concentration (Scheme 56).

During his generalization of 3,2-sigmatropic rearrangements, Baldwin observed the interesting transformation depicted in Scheme 57.⁷⁷ Base treatment of the tosylhydrazone (**236**) generates, after loss of nitrogen, the carbene (**237**), which rearranges to the dithioester *via* a 3,2-sigmatropic transposition. A most elegant application of this reaction has been reported by Evans in his bakkenolide synthesis (Scheme 58).⁷⁸ Thus, base-promoted rearrangement of the tosylhydrazone (**239**) occurred from the less hindered face of the hydrindane system and gave stereospecifically the dithioester (**241**), which was subsequently transformed into bakkenolide (**242**).

Sulfoxonium ylides also undergo 3,2-sigmatropic rearrangements.⁷⁹ A useful preparation of conjugated dienes is based on a 3,2-rearrangement followed by spontaneous elimination of methanesulfenic acid (Scheme 59, equation a). In some cases, the reaction proceeds without isolation of the intermediate ylide (Scheme 59, equation b).





5% Na, MeOH, 60 °C :	3%	78%
10% Na, MeOH, 60 °C :	48%	32%
15% Na, MeOH, 60 °C :	69%	trace


Despite the pioneering work of Ando and Doyle, few synthetic applications of oxonium ylidic rearrangements have been reported. However, three examples of synthetic relevance appeared recently (Schemes 60 to 62). When α -allyloxyacetic esters are reacted with trimethylsilyl triflate and a base, the transposed material (252), resulting from a 3,2-sigmatropic rearrangement of the transient ylide (251), could be isolated in good yields.⁸⁰ The same product ratio was also obtained upon treatment of the ketene acetal (253) with trimethylsilyl triflate. This second variant involves the direct formation of ylide (251) followed by its transformation into (252; Scheme 60).

Almost at the same time, Pirrung^{81a} and Johnson^{81b} independently described the rearrangement of oxonium ylides prepared by intramolecular rhodium-catalyzed carbene addition. The reaction appears to have a broad scope and gives an easy entry into complex oxygenated polycycles (Scheme 61). When the possibility arises, high levels of diastereocontrol are exercised (Scheme 62).



3.10.5.3 Configuration of the Newly Formed Double Bond

The 3,2-sigmatropic rearrangement of ylides proceeds in general with high stereocontrol at the newly formed carbon-carbon double bond. In the synthesis of squalene by Ollis for example, the sulfur ylide (57) rearranged into the sulfide (58) with a *trans* (E) double bond (Scheme 14). Grieco has also shown that rearrangement of the sulfide (262) gave a 9:1 ratio of (E)- to (Z)-alkenes (Scheme 63, equation a).⁸² Hill demonstrated that ammonium ylide rearrangements proceed with >95% (E)-stereoselectivity (Scheme 63, equation b),⁸³ and Julia found that the sulfur ylide (267) gave the sulfide (268) as the pure (E)-isomer (Scheme 63, equation c).⁸⁴

By analogy with the chair conformation invoked for the transition state of the 3,3-rearrangement, the transition state geometry of the 3,2-sigmatropic rearrangement could be analyzed in terms of the 'folded envelope', or puckered, conformation of cyclopentane.⁸⁵ This well-defined conformation should result in a high degree of stereocontrol of the alkene geometry and a high level of stereospecificity in the generation of chiral centers. In the absence of particular effects, examination of these puckered conformations for the transition states of the 3,2-sigmatropic rearrangement suggests that a substituent located on carbon C-1' of the ylide (**269-T**, Scheme 64) would prefer to be pseudo-equatorial. This should lead preferentially to the formation of (*E*)-alkenes (Scheme 64).

Model studies have been carried out in order to compare the stereospecificity of different types of 3,2rearrangements (Scheme 65). From these results, it appears that, in the absence of any special effect, the 3,2-ylidic rearrangement is (*E*)-selective. The Wittig rearrangement shows a greater tendency for (*Z*)stereochemistry.⁸⁹ However, the energy difference between the cisoid (**270-C**) and transoid (**270-T**) transition states in a 3,2-sigmatropic rearrangement (Scheme 64) is not very large. These rearrangements will





occur readily even if they have to proceed via the less favored cisoid transition state (270-C). This is observed when the reacting double bond is incorporated into a ring, as shown in Scheme 66. However, if the ring size allows it, the products of the favored transid transition state will be observed. The increase in strain associated with the formation of an (E)-alkene in a cyclic system might sometimes disfavor the transoid transition state (Scheme 67).⁹⁰

Rearrangement of the five-membered ring sulfonium ylide (278) gives the (Z)-sulfide (279) accompanied by a small amount of (E)-isomer. The homologous six-membered ring sulfonium ylide (280) reacts smoothly to give diastereomerically pure (E)-(281). In the formation of the sulfide (279), the transition state (278-C) leading to the (Z)-alkene suffers from an H-H steric interaction. However, the steric strain arising from the incipient *trans* double bond, in what will eventually become an eight-membered ring, is far more important and transition state (278-C) is preferred over (278-T; Scheme 68). The reverse holds true for the reaction of the ylide (280) where the transition state (280-C) leading to the (Z)-alkene is the more sterically encumbered one because of a severe H-H interaction (Scheme 69).

By placing substituents at suitable positions on the starting ylide, it may become possible to alter the stereochemical course of these reactions. Fava has actually been able to reverse completely the normal stereochemical trend in these 3,2-sigmatropic rearrangements (Scheme 70).⁹¹ Ylide (**282**) rearranges



smoothly to the (E)-sulfide (283), which is contaminated by less than 2% of the (Z)-alkene. In the transition state (282-C) leading to the (Z)-product, a severe steric interaction can be observed between an axial hydrogen and one of the alkenic methyl groups. This interaction will destabilize transition state (282-C) relative to (282-T) and a product with (E)-stereochemistry will be obtained. Similarly, the ylide (284) rearranges to the (Z)-alkenic nine-membered sulfide (285; Scheme 71).

Rearrangement Reactions



Transition state (284-T) reveals clearly a particularly destabilizing methyl-methyl interaction. It is therefore strongly disfavored compared with transition state (284-C), which shows only an axial methylmethyl interaction. This approach has been used by Fava in a highly imaginative route to betweenanenes.⁹² The sulfonium salt (286) is transformed by a stereospecific 3,2-sigmatropic rearrangement into the (*E*)-macrocycle (287). A Ramberg-Backlund reaction followed by catalytic hydrogenolysis then leads to betweenanene (289; Scheme 72, equation a). The epimeric sulfonium salt (290) rearranges smoothly to the (*Z*)-macrocycle. An identical methodology has been used by Nickon to prepare betweenanenes with enol sulfide functionalities.⁹³





(278-C)







(b)

Scheme 68







i, MCPBA; ii, NaH; iii, H₂-Pd-C

3.10.5.4 The Erythro–Threo Ratio

In 3,3-sigmatropic rearrangements, the ratio of diastereoisomers is controlled by the energy difference between boat and chair conformations. In general, this energy difference represents a significant quantity. For 3,2-sigmatropic rearrangements the diastereoisomeric ratio will be dependent upon the preference of a substituent to be located on the *exo* or *endo* position in the transition state. These are often only small energy differences and, therefore, mixtures occur frequently (Scheme 73).





When sterically feasible, an *endo* transition state will however be favored.⁹⁴ Some examples are collected in Table 3.^{94–97} From the results indicated in the table, a general pattern seems to emerge. Sulfur ylides and propargylic ammonium ylides undergo preferential 3,2-rearrangement *via* an *endo* transition state and the diastereoisomeric control is good to excellent. Acyclic, nonpropargylic ammonium ylides give a mixture of diastereoisomers and, in some cases, a strong *exo* preference is displayed (Table 4).⁹⁸ Examination of the envelope transition states reveals that the *exo* transition state is the least sterically congested and should, therefore, be the preferred one in these cases (Scheme 74).



It can also be seen that an increase in the steric bulk of the ketone substituent (Ph replaced by Bu^t) should favor the *exo* transition state even more. This is indeed observed, since the *endo:exo* ratio rises from 1:5 to 1:9. It is, however, difficult to make some predictions since subtle effects which could be present could favor either one or the other of the transition states (compare entry 5 in Table 3 with entry 1 in Table 4). Sometimes, steric hindrance in one of the transition states is so overwhelming that prediction can be made rather safely.⁹⁹ This is the case for the ylide (295) which rearranges exclusively *via* an *exo* transition state *exo*-(296), the *endo* transition state *endo*-(296) being much too hindered to be attained (Scheme 75). Similarly, the ylide (299) gives the amine (300) as a single isomer in 88% yield (Scheme 76).⁹⁹

Table 3									
Entry	Substrate	Ргод	lucts	Ratio	Transition state				
1	Ph + Me N Me Ph Ph	Ph NMe ₂	Ph NMe ₂	10:1	Endo				
2	Ph - Ph	Ph SEt	Ph SEt	_	Endo				
3	Ph Ph	Ph NMe ₂	Ph NMe ₂	86:14	Endo				
4	$Me \\ - Me \\ + - Ph \\ - Ph$	Ph NMe ₂	Ph NMe ₂	10:90	Endo				
5	Ph - Me	Ph NMe ₂	Ph , NMe2	100:0	Endo				



Entry	Substrate	Products		Ratio	Transition state
1	Ph + NMe ₂ COPh	$\frac{Ph}{Ph} \underbrace{NMe_2}_{O}$	$Ph \underbrace{\bigvee_{n_1, \dots, n_n}}_{O} NMe_2$	1:5	Ехо
2	Ph +NMe ₂ COPh	$Ph \qquad \qquad$	$Ph \underbrace{NMe_2}_{O}$	2:1	Endo
3	$\frac{Ph}{- \frac{+ NMe_2}{COBu^t}}$	Bu ^t O NMe ₂	But NMe ₂	1:9	Exo
4	+ N-Me PhOC Me	H WH Ph NMe ₂	H NMe ₂ Ph	2:1	Endo



3.10.5.5 Transfer of Configuration

In a way similar to 3,3-rearrangements, the configuration of C-1 of the allylic group can be transferred to C-3 by a 3,2-sigmatropic rearrangement. The original chiral center and the double bond configuration will control, via the envelope transition state, the chirality and configuration of the product; an (R)-(E)-starting material usually leads to an (S)-(E)-product, whereas the use of a (Z)-substrate gives the opposite result: (R)-(Z). Hill has studied the chirality transfer of simple acyclic ammonium ylides and observed that transposition of the ylide (**301**), followed by reductive cleavage, gave the ketone (**304**; 88% *ee*).^{83,100} The configuration of the carbon-carbon double bond is, as expected, retained. Similarly, the ylide (**305**) leads to the (R)-(-)-cyclic ketone (**307**; Scheme 77). In an analogous manner, rearrangement of the (S)-(E)-allylic alcohol (**308**) using the 'Buchi method' gives the (Z)-amide (**310**) of >99% *ee*, accompanied by ~13% of the (S)-(Z)-isomer resulting from the other envelope conformation.¹⁰¹ In sharp contrast, the (R)-alcohol (**308**) gives only the (R)-(E)-amide (**310**; Scheme 78).

An interesting transfer of asymmetry from a chiral heteroatom to the newly formed carbon-carbon bond has been reported by Trost.¹⁰² The sulfonium salt (311) gives the 3,2-rearrangement product (312) with 94% *ee* (Scheme 79, equation a). The high asymmetric induction obtained in this case can be rationalized by preferential rearrangement *via* transition state (313), containing a minimum of nonbonded interactions, rather than the more congested transition state (315; Scheme 79, equation b). This is, however, a rather exceptional example and, in general, the asymmetric induction remains low (Scheme 80).¹⁰³⁻¹⁰⁵

Asymmetric induction arising from remote chiral centers present in the ylide is yet another way of transferring chirality. Although only a few examples are known, an outstanding one has been described by Baldwin and Kaiser, who obtained the α -allyl penam derivative (186) as a single isomer upon rearrangement of the nitrogen ylide (185; Scheme 43).⁶⁵ Sulfonium and selenonium ylides proved to be much less efficient, as demonstrated by John (Scheme 41).⁶⁴

The last, and probably most interesting, example of the approach towards chiral products derived from ylidic 3,2-rearrangements, involves the generation of a chiral ylide by enantioselective deprotonation of a prochiral sulfonium (ammonium) salt using a chiral base. Few attempts have been reported so far, the most noteworthy coming from the laboratory of Trost and involving the reaction of the sulfonium salt (321) with the chiral base (322; Scheme 81).¹⁰³ The asymmetric induction is rather poor and represents the extent of prochiral recognition by the chiral base.



3.10.5.6 Ring Contraction Reactions

Ylidic 3,2-sigmatropic rearrangements have quite recently been used to promote ring contraction reactions, generally with high selectivity. One of the most famous examples involves the transformation of the cephalosporin derivative (70) into the penicillin derivative (72; Scheme 82).³³ A similar ring contraction was observed by Ando when the unsaturated sulfide (66) was reacted with diazomalonate under photolytic conditions.⁴⁴ The stable ylide (68) underwent 3,2-rearrangement only upon heating (Scheme 83, equation a).

In sharp contrast, the ylide (325) rearranges spontaneously above -5 °C to give a mixture of *cis*- and *trans*-sulfides (326) and (327) in a 25:1 ratio.⁹⁸ The fact that the stereochemistry of the reaction products is under kinetic control was demonstrated by the base-catalyzed epimerization of (326) and (327) to a thermodynamic 1:19 ratio. Ammonium ylides also undergo ring contraction but with complete *cis* diastereoselectivity (Scheme 84).⁹⁸ Base-catalyzed equilibration affords some *trans* isomer. The high diastereoselectivities displayed in these reactions contrasts with the poor and rather unpredictable ones observed for acyclic ylide rearrangements (*cf*. Section 3.10.5.4). In all of the cases examined, the *endo* transition state *endo*-(325) is highly favored over the *exo* transition state *exo*-(325) (Scheme 85). A



possible explanation invokes secondary orbital interactions, which can take place between the *endo* carbonyl group and the carbon-carbon double bond only in the *endo* transition state (**328**; Scheme 86).

Once again, ammonium ylides display higher diastereoselectivities than their sulfur counterparts. This observation formed the basis of a recent synthetic approach towards the alkaloid makomakine (331; Scheme 87) where the key step was the ring contraction of the azepine (329) into the *cis*-substituted piperidine ring (330).¹⁰⁶



Yoshikoshi has explored a novel strategy directed towards the enantioselective preparation of highly substituted butyrolactones, based on the 3,2-sigmatropic rearrangement of sulfur ylides (Scheme 88).¹⁰⁷ Intramolecular carbene trapping generated the eight-membered ring sulfur ylide (334), which then underwent ring contraction to give the complex lactone (335) as a single stereoisomer.

Br



3.10.5.7 Ring Expansion Reactions

One of the first ring expansion reactions of an ylidic intermediate was reported by Hauser and involves the Sommelet-Hauser reaction (see Section 3.10.5.9) of the nitrogen ylide (**336**; Scheme 89).¹⁰⁸ However, the recognition of the synthetic potential of ylidic 3,2-rearrangements for controlled ring expansions is due to Vedejs, who published in 1975 a novel 'ring-growing' reaction.¹⁰⁹ This was to be the beginning of a remarkable synthetic methodology, which eventually culminated in the preparation of numerous natural and non-natural products. The general strategy is summarized in Scheme 90.¹¹⁰ Transformation of a cyclic α -vinyl sulfide (**338**) into a stabilized ylide (**340**) is followed by a 3,2-sigmatropic rearrangement to give the eight-membered ring sulfide (**341**) as a mixture of (*Z*) (major) and (*E*) (minor) isomers. Separation and Wittig reaction regenerates an α -vinyl sulfide (**343**) ready to undergo another ring expansion reaction. The original approach used stabilized ylides but required an extra Wittig reaction in order to become an iterative process. However, by using a nonstabilized allylic ylide such as (**346**), the ring expansion product (**347**) contains an exocyclic vinylic moiety and the process can be repeated again (Scheme 91).¹¹¹



When comparing the examples described in Scheme 90 and 91, several important considerations surface. When employing nonstabilized ylides, the stereochemistry at the sulfur atom of the starting sulfonium salt can no longer be ignored. This is a real problem since the sulfonium salt, and therefore the ylide, has the vinyl group and alkyl group *trans* to each other, a result of kinetically-controlled alkylation at sulfur. The formation of (Z)-alkenes by a concerted 3,2-sigmatropic rearrangement requires a *cis* relationship between these two groups in order for it to take place. Fava has contributed extensively to the solving of this crucial problem.¹¹² He has shown that, with nonstabilized ylides, equilibration of the *cis* and *trans* sulfonium ions takes place more rapidly than deprotonation of the group present on sulfur (Scheme 92). The formation of *cis* sulfonium ylide by equilibration of the *trans* material then allows the concerted 3,2-rearrangement to take place. By blocking the epimerizable center, only the *cis* isomer rearranges smoothly, the *trans* one giving mostly the elimination product (**354**; Scheme 93).⁹⁰

The expected high preference for (E)-alkenes (as compared with acyclic precursors) is conserved in the formation of the nine (and higher) membered rings,⁶⁶ but the formation of eight-membered rings, as already discussed in Section 3.10.5.3, is (Z)-selective. However, ammonium ylides behave differently in these reactions than their sulfur counterparts.¹¹³ Both of the ylide isomers (**358**) and (**359**), derived from the ammonium salt (**357**) give smoothly the nine-membered ring amine (**360**; Scheme 94). Transition states (**358-C**) and (**358-T**) have been invoked to rationalize this result. If the piperidine ring is locked in a conformation in which the ylidic carbon center has to adopt an axial orientation, then only a *cis* macrocycle will be obtained (Scheme 95).¹¹³

Transition states (363) and (364) can be considered, with (363) possessing a correct bonding distance between the alkenic terminus and the ylidic carbon. In (364) this distance is too long for bonding and a conformational change to a boat form would be required for reaction to ensue. This is energetically too

The Stevens and Related Rearrangements





Scheme 92

KOBu^t S `S 1 Me Ă Me -78 °C (353) (354) KOBut (354) trace + **≜** Me -78 °C c (355) (356) major

Scheme 93

.

959



Scheme 94

costly and no *trans*-alkene is observed. Vedejs has used this ring expansion methodology in the preparation of macrocyclic lactones and a typical example is recorded in Scheme 96.¹¹⁴

Intramolecular S-alkylation of (366) is followed by deprotonation to form the ylide (367), which rearranges to the sulfide (368). A few more steps are then required to obtain the nine- and ten-membered ring lactones, including a particularly interesting S- to O-acyl transfer. The efficient stereocontrol usually observed for 3,2-sigmatropic rearrangements was utilized in the synthesis of the C-1 to C-9 fragment (373) of the erythromycin family (Scheme 97).¹¹⁵ This methodology was also used in a repetitive manner for methynolide synthesis (Scheme 98).¹¹⁶ Thus, the 3,2-sigmatropic rearrangement of the ylide (376) initially gave the (Z)-sulfide (377) in 40% yield. This low yield is attributed to the impossibility for the second diastereoisomeric ylide to rearrange. In this case, equilibration through the usual proton transfer is prevented by the C-4 methyl group. Six more steps are then necessary to transform (377) into (378), ready to undergo another ring expansion reaction. The 3,2-rearrangement proceeded this time uneventfully and gave the 11-membered ring (379) in 80–85% yields. Only the (E)-alkene was obtained, as expected. More surprising was the kinetic 16:1 mixture of C-10 epimers isolated, even though (378) consisted of a 1:1 mixture of diastereoisomers. The strong conformational preference for the transoid vinyl rotamers of ylides (381) and (382) accounts for this observation.

Particularly outstanding examples of the high degree of stereocontrol that can be exercised using ylidic 3,2-sigmatropic rearrangements are demonstrated in the cytochalasin $(387)^{117}$ and zygosporin $(388)^{118}$ syntheses (Scheme 99). The allylic halide (384) was obtained with high regio- and stereo-selectivity from the lactam (383). The key 3,2-sigmatropic rearrangement was performed by heating the chloride (384) with sodium iodide and K₂CO₃. A 1.3:1 mixture of epimeric sulfides (386) was obtained in 60% yield. Both isomers have the crucial (*E*)-double bond geometry. The rest of the synthesis then involved desulfurization and allylic transposition. The preparation of the highly elaborated zygosporin (388) followed essentially the same route.



Scheme 96











3.10.5.8 Ylidic Rearrangements Involving Alkynes and Allenes

Allenes can be easily prepared using the 3,2-sigmatropic rearrangement of ylidic intermediates. This methodology also provides a general route towards the synthesis of cumulenes. Some examples using sulfonium and ammonium ylides are collected in Table 5.^{119–122}

Ketene dithioacetals have been prepared by Julia by isomerization of the initially formed allene (Scheme 100),¹²³ and Viehe has utilized this approach to obtain capto-datively substituted dienes (Scheme 101).¹²⁴ The latter reaction can be stopped before isomerization is complete and the allene can be detected along with some diene. The analogous cyanodienes are also easily accessible *via* this route (Scheme 102).

The striking difference in behavior between nitrogen and sulfur ylides is illustrated once again with the examples collected in Scheme 103.¹²⁴ In most of the cases involving the rearrangement of propargylic ammonium ylides, the allene (initial reaction product) can be isolated or identified as a component of the



reaction mixture. When the substituent at C-2 is an amino group, the allene (400) is the only isolable product. Its isomerization to the diene does not occur under the reaction conditions (Scheme 103, equation a). However, in the case of the sulfur ylides, the allene cannot be detected, even after a very short period of time (Scheme 103, equation b).¹²⁵

Very few synthetic applications have appeared in this area, the most noteworthy being the extremely short synthesis of artemesia ketone (408) starting from the allyl sulfide (404; Scheme 104).¹²³ If an alkyne unit can replace a carbon-carbon double bond without apparent deleterious effect on the 3,2-rearrangement, so also can an allene unit. Dienes are then produced (Scheme 105).¹²⁶ The Stevens and Related Rearrangements



3.10.5.9 Sommelet-Hauser Rearrangement

In 1930, Ingold and Jessop reported the rearrangement of the fluorenyl ylide (412).¹²⁷ Compound (414) was obtained in what can be considered as the first example of a Sommelet-Hauser reaction (Scheme 106). In 1937, Sommelet described the base-catalyzed transformation of the benzhydryl ammonium bromide (415) into the substituted benzylic amine (416; Scheme 107, equation a).¹¹ Wittig found later that reacting the salt (417) with phenyllithium produced two amines, (418) and (419).¹³ The former is derived from a Sommelet-Hauser 3,2-rearrangement, whereas the latter comes from a Stevens



1,2-rearrangement (Scheme 107, equation b). The Stevens rearrangement often competes with the Sommelet-Hauser rearrangement. However, in protic solvents, Sommelet-Hauser rearrangements usually predominate.

The Stevens and Related Rearrangements



Scheme 107

Initial work on the Sommelet-Hauser rearrangement was mechanistically inspired. Hauser and Kantor provided a major contribution to the understanding of this transformation.¹⁴ They showed that two ylides were involved in the rearrangement. The more stable ylide (421) is initially generated but equilibrates with the less stabilized one (422). The latter ylide then undergoes a concerted, symmetry-allowed 3,2-sigmatropic shift, giving the triene (423). Aromatization finally affords the *ortho*-substituted benzylic amine (424; Scheme 108).



Similar rearrangements also occur in the case of sulfur analogs. Since the product of a Sommelet-Hauser rearrangement is a benzylic tertiary amine (or sulfide), just as the starting material was a benzylic amine (or sulfide), the process can be repeated and is a unique method to move substituents around the periphery of aromatic rings (Scheme 109).

Several limitations of this base-promoted rearrangement are worthy of mentioning: (a) when structurally feasible, both Stevens and Sommelet-Hauser reactions will compete; (b) certain substituents on the aromatic ring, *e.g.* Cl, CN, NO₂, prevent the formation of the ylide and no rearrangement takes place; (c) if a β -hydrogen is present on the ammonium salt, elimination becomes yet another undesirable side reaction. In an attempt to overcome these limitations, Sato has utilized the fluoride-induced desilylation of



triorganosilylmethyl ammonium salts (427).¹²⁸ This provided a new route for effecting the Sommelet-Hauser rearrangement under non-basic conditions (Table 6). Remarkably, the Stevens product was obtained in small amounts in most cases, except when strong electron-withdrawing groups (COR, NO₂, CN) were present on the aromatic substituent. Sato also found that, under the fluoride-induced desilylation conditions, minimum competitive Hoffmann elimination was observed (Table 7).¹²⁹ The lack of elimination products can be rationalized by the absence of ylide equilibration under fluoride-induced Sommelet-Hauser reaction conditions as compared with the base-promoted rearrangement. Desilylation affords ylide (435) which undergoes the 3,2-sigmatropic rearrangement faster than equilibration with the other ylides (434) and (436; Scheme 110).



Table 7





 R^1 R^2 Entry (431):(432) Mę Et 97:3 1234567 Pri 93:7 Me But 65:35 Me Cyclohexyl Me 89:11 Ēt Et 91:9 98:2 CH2)4 97:3 (CH₂)5-



The Sommelet–Hauser rearrangement has been mostly employed for aromatic functionalization and ring expansion. Julia for example has reported an elegant formylation using a sulfonium ylide rearrangement (Scheme 111).¹³⁰ The reaction provides an easy entry into orthoformylation of aromatics. It is interesting to note that alkylation of the thioacetal (432) followed by hydrolysis could lead to ketone preparation. Gassman has used a similar approach to formylate aniline derivatives (Scheme 112, equation a),¹³¹ and has extended his method to the useful preparation of substituted oxindoles (Scheme 112, equation b).¹³² Gammill, in his synthetic approach towards khellin, used the sulfonium ylide 3,2-sigmatropic rearrangement to transfer functionality from C-6 to C-7 in the key intermediate (441; Scheme 113).¹³³ Warpehoski employed Gassman's procedure as an important step in his synthetic route towards the indole alkaloid CC-1065 (Scheme 114).¹³⁴



Scheme 111

Aromatic functionalization coupled with ring expansion has also been applied to produce macrocyclic amines. Lednicer and Hauser for example reported the Sommelet-Hauser rearrangement of ammonium salt (336) to give the macrocycle (337; Scheme 89).¹⁰⁸ Sato reinvestigated this reaction and was able, using his silicon methodology, to isolate the intermediate triene (446).¹³⁵ Surprisingly enough, compound (446) is fairly stable and can be stored for several weeks. It reacts readily with dienophiles to provide interesting adducts (Scheme 115).

Finally, attempts have been made to induce asymmetry in the Sommelet-Hauser rearrangement, but with limited success. Campbell rearranged the optically active sulfonium salt (449) and obtained a mixture of the two Sommelet-Hauser products (450) and (451).¹³⁶ The chiral product (451) displayed only low optical activity (Scheme 116).

Rearrangement Reactions



Scheme 112





Scheme 113

i, MeOSO₂F



Scheme 114

970





3.10.6 CONCLUSIONS

Enormous progress has been realized in the last two decades both in the preparation of ylidic intermediates and in the mechanistic understanding of ylidic sigmatropic rearrangements. The Stevens 1,2sigmatropic rearrangement has been shown to proceed *via* radical pair intermediates, whereas the 3,2-sigmatropic rearrangements have been demonstrated to be concerted processes. Unfortunately, competition between these and other related rearrangements often takes place, and this has always placed a severe limitation on the use of ylidic rearrangements in synthesis. However, as a result of contemporary mechanistic insights, suitable experimental conditions have been defined that allow one of these processes to predominate largely, if not solely, over the other one. The 3,2-sigmatropic rearrangement of ylidic intermediates has been shown to be a particularly efficient and versatile synthetic tool, mainly because of the remarkable contributions by Vedejs and his group. In sharp contrast, the Stevens 1,2-sigmatropic rearrangement has, so far, led to very few synthetic applications.

ACKNOWLEDGEMENTS

The author is grateful to Professor W. D. Ollis and Professor H. G. Viehe for the use of unpublished material and thanks Professor W. D. Ollis, Dr. J. F. Stoddart and Mr. M. Bailey for reading the manuscript and providing many constructive suggestions.

3.10.7 REFERENCES

- 1. T. S. Stevens, E. M. Creighton, A. B. Gordon and M. MacNicol, J. Chem. Soc., 1928, 3193.
- 2. T. Thomson and T. S. Stevens, J. Chem. Soc., 1932, 69.
- 3. (a) T. Thomson and T. S. Stevens, J. Chem. Soc., 1932, 1932; (b) J. L. Dunn and T. S. Stevens, J. Chem. Soc., 1934, 279.
- 4. T. S. Stevens, J. Chem. Soc., 1930, 2107.
- 5. (a) T. Thomson and T. S. Stevens, J. Chem. Soc., 1932, 55; (b) J. L. Dunn and T. S. Stevens, J. Chem. Soc., 1932, 1926.
- 6. A. Campbell, A. H. J. Houston and J. Kenyon, J. Chem. Soc., 1947, 93.
- G. Wittig, R. Mangold and G. Felletschin, Justus Liebigs Ann. Chem., 1948, 560, 116. 7.
- C. R. Hauser and S. W. Kantor, J. Am. Chem. Soc., 1951, 73, 1437. 8.
- R. B. Woodward and R. Hoffmann, Angew. Chem., Int. Ed. Engl., 1969, 8, 781. 9
- W. D. Ollis, M. Rey and I. O. Sutherland, J. Chem. Soc., Perkin Trans. 1, 1983, 1009. 10.
- 11. M. Sommelet, C. R. Hebd. Seances Acad. Sci., 1937, 205, 56.
- (a) W. Q. Beard and C. R. Hauser, J. Org. Chem., 1961, 26, 371; (b) W. Q. Beard and C. R. Hauser, J. Org. 12. Chem., 1960, 25, 334; (c) G. C. Jones, W. Q. Beard and C. R. Hauser, J. Org. Chem., 1963, 28, 199.
- 13. G. Wittig, H. Tenhaeff, W. Schoch and G. Koenig, Justus Liebigs Ann. Chem., 1951, 572, 1.
- 14. S. W. Kantor and C. R. Hauser, J. Am. Chem. Soc., 1951, 73, 4122.
- 15. W. H. Puterbaugh and C. R. Hauser, J. Am. Chem. Soc., 1964, 86, 1105, 1108.
- 16. R. W. Jemison and W. D. Ollis, J. Chem. Soc., Chem. Commun., 1969, 294.
- 17. (a) J. E. Baldwin, R. E. Hackler and D. P. Kelly, J. Am. Chem. Soc., 1968, 90, 4758; (b) J. E. Baldwin, R. E. Hackler and D. P. Kelly, J. Chem. Soc., Chem. Commun., 1968, 537, 538; (c) J. E. Baldwin and D. P. Kelly, J. Chem. Soc., Chem. Commun., 1968, 899.
- 18. J. E. Baldwin and R. E. Hackler, J. Am. Chem. Soc., 1969, 91, 3646.
- 19. (a) L. Tenud, S. Farooq, J. Seibl and A. Eschenmoser, Helv. Chim. Acta, 1970, 53, 2059; (b) W. D. Ollis, I. O. Sutherland and Y. Thebtaranonth, J. Chem. Soc., Chem. Commun., 1973, 653, 654.
- (a) T. Laird, W. D. Ollis and I. O. Sutherland, J. Chem. Soc., Perkin Trans. 1, 1980, 2033; (b) K. Chantrapromma, W. D. Ollis and I. O. Sutherland, J. Chem. Soc., Perkin Trans. 1, 1983, 1049.
- (a) E. J. Corey and M. Chaykovsky, J. Am. Chem. Soc., 1965, 87, 1353; (b) M. J. Hatch, J. Org. Chem., 1969, 34, 2133; (c) R. W. LaRochelle, B. M. Trost and L. R. Krepski, J. Org. Chem., 1971, 36, 1126; (d) C. R. Hauser, S. W. Kantor and W. R. Brasen, J. Am. Chem. Soc., 1953, 75, 2660; (e) K. W. Ratts and A. N. Yao, J. Org. Chem., 1966, 31, 1185.
- 22. F. E. Ray and J. Farmer, J. Org. Chem., 1943, 8, 391.
- 23. B. M. Trost and H. C. Arndt, J. Org. Chem., 1973, 38, 3140.
- 24. E. Vedejs, D. A. Engler and M. J. Mullins, J. Org. Chem., 1977, 42, 3109.
 25. C. L. Bumgardner, H. B. Hsu, F. Afghahi, W. L. Roberts and S. T. Purrington, J. Org. Chem., 1979, 44, 2348.
- 26. (a) E. Vedejs and G. R. Martinez, J. Am. Chem. Soc., 1979, 101, 6452; (b) E. Vedejs and F. G. West, J. Org. Chem., 1983, 48, 4773.
- 27. (a) Y. Sato and H. Sakakibara, J. Organomet. Chem., 1979, 166, 303; (b) Y. Sato, Y. Yagi and M. Koto, J. Org. Chem., 1980, 45, 613.
- (a) W. Ando, Acc. Chem. Res., 1977, 179; (b) W. Ando, in 'The Chemistry of Diazonium and Diazo groups', ed. S. Patai, Wiley, New York, 1978, chap. 9.
- 29. M. P. Doyle, Acc. Chem. Res., 1986, 19, 348.
- 30. G. M. Blackburn, W. D. Ollis, C. Smith and I. O. Sutherland, J. Chem. Soc., Chem. Commun., 1969, 99.
- 31. G. C. Andrews and D. A. Evans, Tetrahedron Lett., 1972, 5121.
- 32. W. Ando, S. Kondo, K. Nakayama, K. Ichibori, H. Kohada, H. Yamoto, I. Imai, S. Nakaido and T. Nigita, J. Am. Chem. Soc., 1972, 94, 3870.
- 33. M. Yoshimoto, S. Ishihara, N. Soma and E. Nakayama, Tetrahedron Lett., 1972, 2923.
- 34. (a) M. P. Doyle, W. H. Tamblyn and V. Bagheri, J. Org. Chem., 1981, 46, 5094; (b) M. P. Doyle, J. H. Griffin, M. S. Chinn and D. van Leusen, J. Org. Chem., 1984, 49, 1917.
- (a) U. Schöllkopf, J. Schossig and G. Ostermann, Justus Liebigs Ann. Chem., 1970, 737, 158; (b) T. S. Stevens, Prog. Org. Chem., 1968, 7, 48; (c) T. S. Stevens and W. E. Watts, 'Selected Molecular Rearrangements', Van Nostrand-Reinhold, London, 1973, p. 81.
- 36. (a) B. M. Trost and L. S. Melvin, Jr., 'Sulfur Ylides', Academic Press, New York, 1975; (b) A. W. Johnson, 'Ylid Chemistry', Academic Press, New York, 1966; (c) I. Zugravescu and M. Petrovanu, 'N-Ylid Chemistry', McGraw-Hill, New York, 1976; (d) S. H. Pine, Org. React. (N.Y.) 1970, 18, chap. 4.
- 37. (a) Y. Hayashi and R. Oda, *Tetrahedron Lett.*, 1968, 5381; (b) H. E. Zimmerman, 'Molecular Rearrangements', ed. P. de Mayo, Wiley, New York, 1966, part 1, chap. 6.
- 38. U. Schöllkopf, G. Ostermann and J. Schossig, Tetrahedron Lett., 1969, 2619.
- 39. K. W. Ratts and A. N. Yao, J. Org. Chem., 1968, 33, 70.
- 40. S. Wawzonek and E. Yeakey, J. Am. Chem. Soc., 1960, 82, 5718.
- G. Wittig and H. Streib, Justus Liebigs Ann. Chem., 1953, 584, 1. 41.
- 42. K. Kondo and I. Ojima, J. Chem. Soc., Chem. Commun., 1972, 860.
- 43. W. Ando, T. Yagihara, S. Tozune, I. Imai, J. Suzuki, T. Toyama, S. Nakaido and T. Migita, J. Org. Chem., 1972, 37, 1721.
- 44. W. Ando, S. Kondo, K. Nakayama, K. Ichibori, H. Kohoda, H. Yamato, I. Imai, S. Nakaido and T. Migita, J. Am. Chem. Soc., 1972, 94, 3870.
- 45. S. Mageswaran, W. D. Ollis and I. O. Sutherland, J. Chem. Soc., Perkin Trans. 1, 1981, 1953.
- H. P. Benecke and J. H. Wikel, Tetrahedron Lett., 1971, 3479.
 K. Chantrapromma, W. D. Ollis and I. O. Sutherland, J. Chem. Soc., Perkin Trans. 1, 1983, 1029.
- 48. (a) R. H. Mitchell and V. Boekelheide, J. Am. Chem. Soc., 1974, 96, 1547; (b) V. Boekelheide, P. H. Anderson and T. A. Hylton, J. Am. Chem. Soc., 1974, 96, 1558; (c) V. Boekelheide, K. Galuszko and K. S.

Szeto, J. Am. Chem. Soc., 1974, 96, 1578; for an initial example of this strategy, see S. E. Potter and I. O. Sutherland, J. Chem. Soc., Chem. Commun., 1972, 754.

- 49. F. Diederich and H. A. Staab, Angew. Chem., Int. Ed. Engl., 1978, 17, 372.
- 50. D. J. H. Funhoff and H. A. Staab, Angew. Chem., Int. Ed. Engl., 1986, 25, 742.
- 51. J. H. Brewster and M. W. Kline, J. Am. Chem. Soc., 1952, 74, 5179.
- 52. U. Schöllkopf, U. Ludwig, G. Ostermann and M. Patsch, Tetrahedron Lett., 1969, 3415. 53. B. J. Millard and T. S. Stevens, J. Chem. Soc., 1963, 3397.
- 54. J. W. Lown and M. H. Akhtar, J. Chem. Soc., Chem. Commun., 1973, 511.
- 55. E. F. Jenny and J. Druey, Angew. Chem., Int. Ed. Engl., 1962, 1, 155.
- 56. H. Joshua, R. Gans and K. Mislow, J. Am. Chem. Soc., 1968, 90, 4884.
- J. H. Brewster and R. S. Jones, Jr., J. Org. Chem., 1969, 34, 354. 57.
- 58. R. K. Hill and T. H. Chan, J. Am. Chem. Soc., 1966, 88, 866.
- 59. K. Chantrapromma, W. D. Ollis and I. O. Sutherland, J. Chem. Soc., Perkin Trans. 1, 1983, 1049.
- 60. W. D. Ollis, unpublished material presented at the Pedler Lectures, London, 1989.
- B. M. Trost, P. Conway and J. Stanton, J. Chem. Soc., Chem. Commun., 1971, 1639.
 (a) H. Hellmann and D. Eberle, Justus Liebigs Ann. Chem., 1963, 662, 188; (b) G. M. Blackburn, W. D. Ollis, J. D. Plackett, C. Smith and I. O. Sutherland, J. Chem. Soc., Chem. Commun., 1968, 186.
- 63. W. Ando, M. Yamada, E. Matsuzaki and T. Migita, J. Org. Chem., 1972, 37, 3791.
- 64. (a) P. J. Giddings, D. I. John, E. J. Thomas and D. J. Williams, J. Chem. Soc., Perkin Trans. 1, 1982, 2757;
 (b) P. J. Giddings, D. I. John and E. J. Thomas, Tetrahedron Lett., 1980, 395, 399.
- 65. G. V. Kaiser, C. W. Ashbrook and J. E. Baldwin, J. Am. Chem. Soc., 1971, 93, 2342.
- 66. E. Vedejs, Acc. Chem. Res., 1984, 17, 358.
- 67. S. Takano, S.-I. Tomita, M. Takahashi and K. Ogasawara, Chem. Lett., 1987, 1569.
- 68. S. Julia, C. Huynh and D. Michelot, Tetrahedron Lett., 1972, 757.
- 69. P. J. Kocienski, J. Chem. Soc., Chem. Commun., 1980, 1096.
- 70. L. N. Mander and J. V. Turner, J. Org. Chem., 1973, 38, 2915.
- 71. E. Hunt and B. Lythgoe, J. Chem. Soc., Chem. Commun., 1972, 757.
- 72. G. Büchi and H. Wüest, J. Am. Chem. Soc., 1974, 96, 7573.
- 73. E. J. Corey and S. W. Walinsky, J. Am. Chem. Soc., 1972, 94, 8932.
- 74. Z. Kosarych and T. Cohen, Tetrahedron Lett., 1982, 23, 3019.
- 75. T. Cohen, Z. Kosarych, K. Suzuki and L.-C. Yu, J. Org. Chem., 1985, 50, 2965.
- 76. C. Fleischmann and E. Zbiral, Tetrahedron, 1978, 34, 317.
- J. E. Baldwin and J. A. Walker, J. Chem. Soc., Chem. Commun., 1972, 354.
 D. A. Evans, C. L. Sims and G. C. Andrews, J. Am. Chem. Soc., 1977, 99, 5453.
- 79. (a) J. Ide and Y. Kishida, Tetrahedron Lett., 1966, 1787; (b) J. Ide and Y. Kishida, Chem. Pharm. Bull. (Tokyo), 1968, 16, 793; (c) Y. Tamura, T. Miyamoto, T. Nishimura and Y. Kita, Tetrahedron Lett., 1973, 2351.
- 80. K. Mikami, O. Takahashi, T. Tabei and T. Nakai, Tetrahedron Lett., 1986, 27, 4511.
- 81. (a) M. C. Pirrung and J. A. Werner, J. Am. Chem. Soc., 1986, 108, 6060; (b) E. J. Roskamp and C. R. Johnson, J. Am. Chem. Soc., 1986, 108, 6062.
- 82. (a) P. A. Grieco, D. Boxler and K. Hiroi, J. Org. Chem., 1973, 38, 2572; (b) P. A. Grieco, J. Chem. Soc., Chem. Commun., 1972, 702.
- 83. R. K. Hill, R. Soman, M. Hojjat, Abstr. Pap. Am. Chem. Soc. 181st, Natl. Meet. 1981, Nº ORGN 105.
- 84. C. Huynh, V. Ratovelomanana and S. Julia, Bull. Soc. Chem. Fr., 1977, 710.
- 85. For a review on the 3,3-sigmatropic rearrangement, see 'Comprehensive Organic Synthesis', ed. B. M. Trost, Pergamon Press, Oxford, 1991, vol. 5, chaps. 7.1 and 7.2.
- V. Rautenstrauch, Helv. Chim. Acta, 1971, 54, 739.
 (a) D. A. Evans, G. C. Andrews, T. T. Fujimoto and D. Wells, Tetrahedron Lett., 1973, 1389; (b) P. Blatcher and S. Warren, Tetrahedron Lett., 1979, 1247; (c) S. Yamagiwa, H. Sato, N. Hoshi, H. Kosugi and H. Uda, J. Chem. Soc., Perkin Trans. 1, 1979, 570.
- 88. J. E. Baldwin, J. DeBernardis and J. E. Patrick, Tetrahedron Lett., 1970, 353.
- 89. For a review on the Wittig rearrangement, see J. A. Marshall, 'Comprehensive Organic Synthesis', ed. B. M. Trost, Pergamon Press, Oxford, 1991, vol. 3, chap. 3.11.
- 90. V. Ceré, C. Paolucci, S. Pollicino, E. Sandri and A. Fava, J. Org. Chem., 1979, 44, 4128.
- 91. V. Ceré, C. Paolucci, S. Pollicino, E. Sandri and A. Fava, J. Org. Chem., 1981, 46, 3315. 92. V. Ceré, C. Paolucci, S. Pollicino, E. Sandri and A. Fava, J. Org. Chem., 1981, 46, 486.

- A. Nickon, A. Rodriguez, V. Shirlattl and R. Ganguly, *Tetrahedron Lett.*, 1984, 25, 3555.
 R. W. Jemison, T. Laird, W. D. Ollis and I. O. Sutherland, J. Chem. Soc., Perkin Trans. 1, 1980, 1436.
- 95. R. W. C. Cose, A. M. Davies, W. D. Ollis, C. Smith and I. O. Sutherland, J. Chem. Soc., Chem. Commun., 1969, 293.
- 96. T. Laird, W. D. Ollis and I. O. Sutherland, J. Chem. Soc., Perkin Trans. 1, 1980, 2033.
- 97. K. Kondo and I. Ojima, J. Chem. Soc., Chem. Commun., 1972, 62.
- 98. S. Mageswaran, W. D. Ollis and I. O. Sutherland, J. Chem. Soc., Perkin Trans. 1, 1981, 1953.
- 99. W. D. Ollis, I. O. Sutherland and Y. Thebtaranonth, J. Chem. Soc., Perkin Trans. 1, 1981, 1963.
- 100. (a) R. K. Hill, Asymmetric Synthesis, 1984, 3, 503; (b) R. W. Hoffmann, Angew. Chem., Int. Ed. Engl., 1979, 18, 563.
- 101. (a) K. K. Chan and G. Saucy, J. Org. Chem., 1977, 42, 3828; (b) G. Büchi, M. Cushman and H. Wuest, J. Am. Chem. Soc., 1974, **96**, 5563.
- 102. B. M. Trost and R. F. Hammen, J. Am. Chem. Soc., 1973, 95, 962.
- 103. B. M. Trost and W. G. Biddlecom, J. Org. Chem., 1973, 38, 3435.
- T. Numata and S. Oae, Tetrahedron Lett., 1977, 1337.
 M. Mikoiaczyk, A. Zatorski, S. Grzejszczak, B. Costisella and W. Midura, J. Org. Chem., 1978, 43, 2518.
- 106. S. J. Neeson and P. J. Stevenson, Tetrahedron Lett., 1988, 29, 3993.

- 107. F. Kido, S. C. Sinha, T. Abiko and A. Yoshikoshi, Tetrahedron Lett., 1989, 30, 1575.
- 108. D. Lednicer and C. R. Hauser, J. Am. Chem. Soc., 1957, 79, 4449.
- 109. (a) E. Vedejs and J. P. Hagen, J. Am. Chem. Soc., 1975, 97, 6878; (b) R. Schmid and H. Schmid, Helv. Chim. Acta, 1977, 60, 1361.
- 110. (a) E. Vedejs, J. P. Hagen, B. L. Roach and K. L. Spear, J. Org. Chem., 1978, 43, 1185; (b) E. Vedejs, M. J. Arco, D. W. Powell, J. M. Renga and S. P. Singer, J. Org. Chem., 1978, 43, 4831. 111. E. Vedejs, M. J. Mullins, J. M. Renga and S. P. Singer, Tetrahedron Lett., 1978, 519.
- 112. V. Ceré, C. Paolucci, S. Pollicino, E. Sandri and A. Fava, J. Org. Chem., 1978, 43, 4826.
- 113. E. Vedejs, M. J. Arco and J. M. Renga, Tetrahedron Lett., 1978, 523.
- 114. E. Vedejs and D. W. Powell, J. Am. Chem. Soc., 1982, 104, 2046.
- 115. E. Vedejs, J. M. Dolphin and H. Mastalerz, J. Am. Chem. Soc., 1983, 105, 127.
- 116. E. Vedejs, R. A. Buchanan, P. Conrad, G. P. Meier, M. J. Mullins and Y. Watanabe, J. Am. Chem. Soc., 1987, 109, 5878.
- E. Vedejs and J. G. Reid, J. Am. Chem. Soc., 1984, 106, 4617.
 E. Vedejs, J. D. Rodgers and S. J. Wittenberger, J. Am. Chem. Soc., 1988, 110, 4822.
- 119. J. E. Baldwin, R. E. Hackler and D. P. Kelly, J. Chem. Soc., Chem. Commun., 1968, 1083.
- 120. A. Terada and Y. Kishida, Chem. Pharm. Bull., 1970, 18, 991.
- 121. S. Mageswaran, W. D. Ollis, D. A. Southam, I. O. Sutherland and Y. Thebtaranonth, J. Chem. Soc., Perkin Trans. 1, 1981, 1969.
- 122. P. Cresson, C. R. Acad. Sci. Paris, Ser. C, 1977, 284, 247.
- 123. D. Michelot, G. Linstrumelle and S. Julia, J. Chem. Soc., Chem. Commun., 1974, 10.
- 124. H. G. Viehe and N. Stevenaert, unpublished results (private communication).
- 125. (a) G. Pourcelot, L. Veniard and P. Cadiot, Bull. Soc. Chim. Fr., 1975, 1275, 1281; (b) V. Ratovelomanana and S. Julia, Synth. Commun., 1978, 8, 87.
- 126. P. A. Grieco, M. Meyers and R. S. Finkelhor, J. Org. Chem., 1974, 39, 119.
- 127. C. K. Ingold and J. A. Jessop, J. Chem. Soc., 1930, 713.
- 128. (a) M. Nakano and Y. Sato, J. Org. Chem., 1987, 52, 1844; (b) Y. Sato, Y. Yagi and M. Koto, J. Org. Chem., 1980, 45, 613; (c) Y. Sato and H. Sakakibara, J. Organomet. Chem., 1979, 166, 303.
- 129. N. Shirai and Y. Sato, J. Org. Chem., 1988, 53, 194.
- 130. S. Julia, C. Huynh and D. Michelot, Tetrahedron Lett., 1972, 3587.
- 131. (a) P. G. Gassman and G. Gruetzmacher, J. Am. Chem. Soc., 1973, 95, 588; (b) P. G. Gassman and T. J. Van Bergen, J. Am. Chem. Soc., 1973, 95, 590, 591. 132. (a) P. G. Gassman and T. J. Van Bergen, J. Am. Chem. Soc., 1973, 95, 2718; (b) P. G. Gassman and W. N.
- Schenk, J. Org. Chem., 1977, 42, 3240; (c) P. G. Gassman, G. Gruetzmacher and T. J. Van Bergen, J. Am. Chem. Soc., 1974, 96, 5512.
- 133. R. B. Gammill, J. Org. Chem., 1984, 49, 5035.
- 134. M. A. Warpehoski and V. S. Bradford, Tetrahedron Lett., 1986, 27, 2735.
- 135. N. Shirai, F. Sumiya, Y. Sato and M. Hori, J. Chem. Soc., Chem. Commun., 1988, 370.
- 136. S. J. Campbell and D. Darwish, Can. J. Chem., 1976, 54, 193.

3.11 The Wittig Rearrangement

JAMES A. MARSHALL

University of South Carolina, Columbia, SC, USA

3.11.1 INTRODUCTION	975
3.11.1.1 Historical Development	975
3.11.1.2 Rearrangement Pathways	977
3.11.1.3 Heterologs	978
3.11.2 1,2-REARRANGEMENTS	979
3.11.3 2,3-REARRANGEMENTS	981
3.11.3.1 Overview	981
3.11.3.2 2.3-Rearrangements of Allyl Lithiomethyl Ethers	982
3.11.3.3 2.3-Wittig Rearrangement of Allyl Propargyl Ethers	984
3.11.3.4 2,3-Wittig Rearrangement of Allyl Benzyl Ethers	989
3.11.3.5 2,3-Rearrangement of Dipropargyl Ethers	991
3.11.3.6 2,3-Rearrangement of Diallyl Ethers	991
3.11.3.7 Tandem and Sequential 2,3-Wittig-3,3-Sigmatropic Rearrangements	994
3.11.3.8 Rearrangement of α-Allyloxy Enolates	996
3.11.3.8.1 a-Allyloxy ketones	996
3.11.3.8.2 O-Allyl cyanohydrins	998
3.11.3.8.3 a-Allyloxycarboxylic acids	999
3.11.3.8.4 a-Allyloxycarboxylic esters	1000
3.11.3.8.5 a-Allyloxycarboxamides	1004
3.11.3.8.6 Allyloxymethyl-oxazolines and -1,3-oxazines	1005
3.11.3.9 Oxonium Ylide Rearrangements	1008
3.11.3.10 Cyclic 2,3-Rearrangement-Ring Contractions	1008
3.11.4 REFERENCES	1013

3.11.1 INTRODUCTION

3.11.1.1 Historical Development

In 1942 Wittig and Löhmann described the isomerization of α -lithiated ethers (2; R = Me or CH₂Ph) to skeletally rearranged lithio alkoxides (3; equation 1).¹ Subsequent work by Wittig and others established that a variety of ethers undergo the Wittig rearrangement, provided the α -anion is stabilized by an aryl substituent. Of the several mechanisms proposed for this reaction a sequence involving homolysis of the anion intermediate (5), followed by recombination of the radical and radical anion fragments (equation 2) best accommodates the experimental findings.¹⁻⁴

$$Ph \longrightarrow O^{-R} \xrightarrow{PhLi} Ph \longrightarrow O^{-R} \xrightarrow{PhLi} Ph \longrightarrow OLi \qquad (1)$$

$$(1) \qquad (2) \qquad (3)$$



Hauser noted that diallyl ether (8) also undergoes Wittig rearrangement upon base treatment and suggested that product formation could involve either a 1,2-shift or a 'cyclic mechanism' (equation 3).² Later studies by Schöllkopf⁵ and Makisumi⁶ with substituted allylic ethers (10, 11 and 14–16; equations 4 and 5) pointed to a cyclic (S_Ni') mechanism; a process allowed by the Woodward–Hoffmann rules.⁷ The diastereoselectivity of the reaction was not determined in these cases, but Schöllkopf subsequently found that benzyl *trans*-crotyl ether (20; equation 6) affords mainly the *anti* products upon rearrangement of ether (20) with BuLi in THF.⁸ Rautenstrauch observed a 1:1 mixture of *syn* and *anti* products upon rearrangement of ether (20) in the presence of TMEDA, whereas the *cis* isomer (23) gave only the *syn* product (22; equation 7).⁹



3.11.1.2 Rearrangement Pathways

Rearrangements of allylic ethers can follow several pathways (Scheme 1). The 2,3- and 1,4-routes are symmetry allowed concerted processes, whereas the 1,2- and 3,4-rearrangements proceed in a stepwise manner through radical dissociation and recombination (equation 2). In fact, all four pathways have been experimentally observed with bis- γ , γ -(dimethyl)allyl ether (24; equation 8).⁹ Anions derived from allyl vinylcyclopropylmethyl ethers (*e.g.* i) rearrange by competing, 1,2, 1,4, homo-2,5, homo-4,5 and radical dissociation-recombination pathways.^{9a} Nakai has shown that 1,4-rearrangement proceeds with retention of stereochemistry in the cyclohexenyl system (29; equation 9), as required by orbital symmetry considerations.¹⁰



Scheme 1 Rearrangement pathways for allylic ether anions




3.11.1.3 Heterologs

Sulfides undergo 2,3-Wittig rearrangement, but at a slower rate than their oxygen counterparts (equation 10). This rate retardation has been attributed to a decrease in the energy of the anion HOMO as a consequence of the greater stabilizing influence of sulfur.^{7c} Snider prepared the α -thiomethylcyanomethyl sulfides (**36**) through ene reaction of alkenes with methyl cyanodithioformate, and observed facile 2,3-rearrangement upon treatment with BuLi or LDA in THF-HMPA (equation 11).¹¹ In the absence of HMPA the α -cyano anions are stable and can be readily C-methylated with iodomethane. Representative examples are shown in Table 1.



 $(CH_2)_2CH(CH_2OCO_2Me)CH_2.$

The amine α -anions (**38a–c**) failed to rearrange under conditions employed for the analogous ethers.² The corresponding ammonium salts, on the other hand, readily isomerize *via* the derived ylides (Stevens rearrangement) in a process analogous to the 1,2-Wittig rearrangement.¹²

The N-benzyl β -lactams (39) and (40) undergo facile rearrangement upon base treatment.¹³ In the former case only the 2,3-product (41) is observed, whereas the latter affords a mixture of 2,3- and 1,2-products (42) and (44; equation 12). The methyl substituent of (40) evidently facilitates homolysis of the intermediate benzylic anion. The phenyl-substituted β -lactams (45) and (46) yield the 1,2-rearrangement products (47) and (48) exclusively under comparable conditions (equation 13). These rearrangements undoubtedly derive considerable driving force from relief of ring strain. Thus N,N-allylbenzylbenzamide, an acyclic counterpart of lactam (39), is recovered unchanged upon exposure to strong base.



3.11.2 1,2-REARRANGEMENTS

Early work on the 1,2-Wittig rearrangement was mechanistically inspired. As indicated in Table 2, yields of products tended to be moderate and harsh conditions were often employed. The effects of structural variation on reaction rates and product composition can be accommodated by a pathway involving homolysis of an α -anion intermediate and recombination of the radical and radical anion fragments (equation 2). In a more recent study, Schöllkopf found that the optically active benzyl ethers (51) and (52) afford the alcohols (53) and (54) of predominately retained configuration in 20% and 80% enantiomeric excess (equation 14). This finding further supports a dissociative mechanism.⁴ A concerted one-step rearrangement should proceed with inversion of the migrating center according to orbital symmetry considerations.

In the first purely synthetic application of the reaction, Schreiber employed the 1,2-Wittig rearrangement to synthesize 1,3-diol monoethers (equation 15).¹⁴ The *syn* products (56) were obtained in 14–32% yield with 90–95% diastereoselectivity (Table 3). The 1,4-product (57) was found to predominate at low temperature, whereas at 0 °C the 1,2-product was favored but the absolute yield did not increase.

The nonracemic secondary ethers (58) and (60) rearrange with over 90% retention of stereochemistry at the migrating center (equations 16 and 17). A lithium-bridged diradical species (63; Scheme 2) is postulated to account for the observed diastereoselectivity. Recombination of the radical pair must occur more rapidly than inversion of the radical center, judging from the high degree of retention observed with (58) and (60). A two-directional application of this rearrangement was used to prepare the *syn*skipped polyols (66) and (68; Scheme 3).





R	Conditions	Time (h)	Yield (%)	Ref.
PhCH ₂	a	15	61–67	2
CH2-CHCH2	а	15	65 (mixture)	2
Bu ^s	а	140	27	2
Ph	а	18	0 (recovered ether)	2
Et	а	19	0 (40% PhCH ₂ OH)	2
$CH_2 \rightarrow CH(CH_2)_2$	Ъ	24	67 (3
c-C4H7	Ъ	24	68	3
Bu ^t	С	48	50	3
1-Adamantyl	С	48	54	3

*2KNH₂, Et₂O, reflux. ^b 2MeLi, THF, r.t. ^c 5MeLi, THF, r.t.









R^1	R ²	Syn/anti	Yield (%)
H	H	92:8	32
Me	H	91:9	30
Ph	H	90:10	25
Me	Me	90:10	14



(58)

(16)

(59) 91:9 syn:anti 94% retention at C-3



Scheme 2 Chelation-controlled 1,2-rearrangement of glyceryl allyl ethers



3.11.3 2,3-REARRANGEMENTS

3.11.3.1 Overview

The 2,3-Wittig rearrangement (equation 18) proceeds under milder conditions and gives higher yields of products than its 1,2-counterpart. Furthermore, a greater degree of structural variation is possible in the starting ether. For these reasons, the 2,3-rearrangement offers considerable promise as a synthetic method. The reaction effects an interchange of sp^2 and sp^3 stereochemistry such that both (E)/(Z) and syn/anti isomerism are possible in appropriately substituted systems. Furthermore, in nonracemic ethers



the concerted nature of the rearrangement allows for control of absolute stereochemistry. In most applications, various functional groups (G) are employed to promote formation of the intermediate anion. In principle, propargylic ethers could undergo 2,3-rearrangement analogously leading to allenes (equation 19). In practice such rearrangements have infrequently been observed.



3.11.3.2 2,3-Rearrangements of Allyl Lithiomethyl Ethers

A unique variant of the 2,3-Wittig rearrangement employs an unstabilized lithiomethyl allylic ether generated *in situ* through lithiation of a (trialkylstannyl)methyl allylic ether (equation 20). This version, first introduced by Still,¹⁵ has proven exceedingly useful owing to its efficiency and potential for high stereoselectivity. Applications to acyclic systems are summarized in Table 4. The rearrangement proceeds under mild conditions to give high yields of products. Interestingly, the stereochemistry of the rearranged double bond depends upon the substitution pattern of the allylic ether. In rearrangements of secondary allylic ethers R²-substituents favor (Z)-products (76) whereas R⁴-substituents are highly (E)-directing. These trends are accommodated by the envelope transition states (A) and (B) in Scheme 4. Unfavorable steric interactions between R¹ and R² in (A) are alleviated in (B), thereby favoring formation of the (Z)-alkenic product. Transition state (B), on the other hand, imposes unfavorable 1,3-syn steric strain on substituents R¹ and R⁴. This strain is lacking in (A). When 1,2- and 1,3-interactions are both present, or when they are both absent (R² = R³ = R⁴ = H), a mixture of (Z)- and (E)-rearranged products is obtained.



Table 4 2,3-Wittig Rearrangement of Lithiomethyl Allyl Ethers



Scheme 4 Transition states for lithiomethyl allyl ether rearrangements

An additional example of such geometric control was recently observed by Midland and Kwon in the isomeric allylic ethers (78) and (79; equation 21).¹⁶ The (Z)-isomer (78) gave the (E)-rearranged alcohol (80) exclusively, in accord with transition state (A), whereas the (E)-allylic ether (79) yielded a 53:47 mixture of (E)- and (Z)-products (81) and (83). In both cases rearrangement proceeded with retention of sp^3 stereochemistry, as expected for a concerted suprafacial process. In related work, 2,3-rearrangement of the steroidal (23*R*)-ether (86) afforded (25*R*)-26-hydroxycholesterol (88), an important intermediate in bile acid biosynthesis (Scheme 5). Again, the key rearrangement step proceeded with complete (>120:1) 1,3-syn stereoselectivity.¹⁷ A similar application led to the 25,28-dihydroxylated ergosterol derivatives (91) and (92; equation 22).¹⁸



i, (R)-alpine borane, 92%, 92% ee; ii, H₂, Lindlar catalyst; iii, NaH, Me₃SnCH₂I, 62%; iv, 2.3 BuLi, THF, -78 to 0 °C, 80%; v, H₂/Pd-C; vi, AcOH, H₂O, THF

Scheme 5



Still-Wittig rearrangement of the (*E*)- and (*Z*)-17-ethylidene-16 α ethers (93) and (94) afforded the 20 α - and 20 β -methyl steroids (95) and (96) with complete stereochemical control (equation 23).¹⁹ As in other examples, *sp*³ stereochemistry depends upon double bond geometry in the starting allylic ether.

Still-Wittig rearrangements have been used to introduce an allylic hydroxymethyl substituent to a cyclohexene ring (equations 24 and 25).^{20,21} These reactions proceed with retention of stereochemistry. Al-



cohol (98) was employed in the total synthesis of (-)-punctatin A, a sesquiterpene antibiotic.²⁰ Alcohol (100) is a precursor of a synthetic intermediate for 1α , 25-dihydroxy vitamin D₃.²¹



Still-Wittig rearrangement of the nonracemic (R)-propargylic ethers (100a) and (100b) has been found to give the (R)-allenes (100c) and (100d) with complete asymmetric transfer (equation 25a).^{21a} The stere-ochemistry is consistent with a five-membered cyclic transition state.



3.11.3.3 2,3-Wittig Rearrangement of Allyl Propargyl Ethers

Allyl propargyl ethers are selectively deprotonated at the propargylic position to give anions which readily undergo 2,3-rearrangement. Typically *n*-butyllithium is employed as the base in THF or THF-hexane at -85 to 0 °C. In substituted allylic systems a high degree of diastereoselectivity is often seen, particularly with (Z)-allylic ethers (Table 5). In general, (E)-allylic ethers afford *anti* homoallylic alcohols (102), and (Z)-allylic ethers rearrange to *syn* homoallylic alcohols (103) as major 2,3-products. The TMS-propargyl (E)-allylic ethers (entries 3 and 4) are exceptional, giving rise to a predominance of *syn* products.^{22,23} However, the TMS-propargyl (Z)-allylic ether (entry 12) behaves normally to give the rearranged *syn* alcohol.²² When allylic substituents are present, as in (101; R¹ \neq H), (E)/(Z)-isomerism is possible in the rearranged products. The examples in Table 5 (entries 5–9 and 13–15) afford only the (E)-products within the limits of detection.





*1:1 THF-hexane. *Not reported. 65:35. 495:5.

An envelope transition state in which the alkynyl grouping assumes an equatorial-like orientation has been proposed to accommodate these observations (Scheme 6).^{26a} The orientation of \mathbb{R}^2 is fixed by the (E)/(Z)-geometry of the starting allylic ether. When allylic (\mathbb{R}^1) substituents are present they likewise adopt an equatorial position as in (C) and (D), thereby favoring the (E)-double bond geometry in the rearrangement product.^{26b}



Scheme 6 Envelope transition states for 2,3-Wittig rearrangements of propargyl allyl ethers

The TMS-propargyl ether (104) rearranges to an 80:20 mixture of the (E)- and (Z)-isomers (106) and (108) in an apparent violation of this transition state proposal (equation 26).²⁷ (Note that because of the stereochemical descriptor rules the sense of (E) and (Z) is inverted by the TMS substituent.) However, replacement of the vinylic TMS grouping by hydrogen led 'almost exclusively' to the (E)-product (109). Evidently, the normally favored transition state (E) is destabilized by steric interactions between the vinylic TMS substituent and the *n*-pentyl grouping (R¹). The alternative conformation (F) lacks this interaction. The situation is analogous to that noted by Still in the rearrangement of lithiomethyl allyl ethers (Scheme 4).



The 2,3-rearrangement of nonracemic propargylic ethers proceeds enantioselectively, as might be expected considering the concerted nature of the reaction and its high diastereoselectivity. Nakai examined the (S)-2-butenyl propargyl ethers (110; Table 6).²⁸ In each case a rearranged product of high stereochemical integrity was obtained in near quantitative yield. The stereochemistry of these rearrangements reflects the equatorial preference of substituents in the envelope transition state (G; R¹ = Me, R² = H) as depicted in Scheme 7.





^aMajor isomer.



Scheme 7 Transition states for 2,3-Wittig rearrangements of chiral propargyl allyl ethers

Nonracemic (Z)-allylic propargyl ethers likewise rearrange with excellent enantioselectivity (Table 7). In these examples the configuration of the starred center in (114) and (115) should be opposite for (E)and (Z)-products owing to the concerted suprafacial nature of the rearrangement. The stereochemistry at the carbinyl center is related to the equatorial/axial preference of the alkynyl grouping (Scheme 7). It should be noted that the (E)-products (114) and (115) are greatly favored because of repulsive syn-1,3 interactions between \mathbb{R}^1 and \mathbb{R}^2 in the transition states (I) and (J) leading to (Z)-products (Scheme 7). The (E)-allylic ether (116), lacking this 1,3-interaction, rearranges with complete chirality transfer but yields a mixture of *syn/anti* and (E)/(Z)-isomers (equation 27).²⁹





"The enantiomer of the depicted compound was employed. "Not reported.



Nakai has employed the propargylic 2,3-Wittig rearrangement for elaboration of side chains in steroidal systems. The (*E*)-17-ethylidene-16 α ether (119) rearranges to the expected alcohol (120) on base treatment (equation 28).³² Remarkably, the TMS derivative (121) of ether (119) rearranges with even greater facility to give the diastereomeric alcohol (122; equation 29).³³ The isomeric (*Z*)-17-ethylidene-16 β TMS-propargyl ether (123) yields the same alcohol (equation 30). These results indicate that attack by the propargylic anion derived from the 16 α ether (121) takes place with orientation of the TMS-ethynyl grouping toward the c-ring of the steroid as in (K; Scheme 8). In the 16 β -(*Z*)-isomer (123) the TMS-





Scheme 8 2,3-rearrangment of steroial 17-ethylidene-16-trimethylsilyl-propargyl ethers

ethynyl grouping adopts the opposite orientation (L), also leading to the (22R)-alcohol (122). The ethynyl anion must assume the same relative orientation as in (L) but on the α -face of the double bond *en* route to the (22S)-alcohol (120).

The effect of a stereocenter at the δ -allylic position of allyl propargyl ethers has been examined by Nakai.^{33a} The (Z)-ethers show particularly high 3,4-diastereoselectivity (Tables 7a and 7b).

The high asymmetric induction of these rearrangements is explained by transition states (La) and (Lb) as shown in Scheme 8a. The allylic oxygen substituents adopt an orientation perpendicular to the plane of the double bond to minimize allylic 1,3-repulsion. The propargylic anion attacks the double bond preferentially *anti* to these oxygens.

SiMe ₃	A HO ¹¹¹¹ 2	Me ₂ H + HO SiMe ₃	DSiBu ^t Me ₂	H H IO''''	Me ₂ + HO SiMe ₃	SiBu ¹ Me ₂
(123a)	(123b)	(123	c)	(123d)	(1236	2)
(123a)	A	Yield (%)	(123b)	Product d (123c)	istribution (123d)	(123e)
(Z) (E) (E)	BuLi, THF, -78 °C BuLi, THF, -78 °C LDA ,THF-HMPA	77 93 74	94 10 3	1 21 3	4 80 87	1 19 7

Table 7a 2,3-Wittig Rearrangement of Nonracemic Allylic Propargylic Ethers Derived from (S)-Lactaldehyde.

OC:D.IL







3.11.3.4 2,3-Wittig Rearrangement of Allyl Benzyl Ethers

As noted in Section 3.11.1, some of the earliest mechanistic work on the 2,3-Wittig rearrangement was carried out with allyl benzyl ethers. Selective benzyl deprotonation is typically effected with *n*-butyl-lithium in THF or THF-TMEDA at low temperature. Rautenstrauch's studies, summarized in Table 8, clearly showed that lower temperature favors 2,3- over 1,2-rearrangement, and established the potential



R ¹	$ \begin{array}{c} $	uLi, TMEDA R ²	R^{3} R^{4} R^{1} R^{4} R^{1} R^{4} R^{4} R^{1} R^{4} R^{4	R ¹ + R ² _{HO} ^{,,,,,,} (126)	$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	\mathbb{R}^{R^3}
R ¹	R ²	R ³	R⁴	T (°C)	(125):(126):(127)	Ref.
H H Me ^a H H Me H H	H H H H Me H H	Me H Me Me H H H	H Me H Me H OTBS ^c OTBS ^c	-80 to -25 -80 to -25 -80 to -25 -80 to -25 23 -25 -65 -65	50:50:0 100:0:0 0:100:0 (89) ^b :11 (86) ^b :14 (60) ^b :40 18:82 ^d 23:77 ^d	9 9 9 9 9 9 9 25a 25a

*Racemic. b(125) = (126). c71:29(Z):(E). d(125):(126). e93:7(Z):(E).



for (E)/(Z) and syn/anti stereoselectivity in the former.⁹ Baldwin and Patrick examined the chiral (S)-(E)-1-methyl-2-butenyl ether (**128**) and concluded that the 2,3-rearrangement is 100% suprafacial (equation 31).³⁴ A second pathway, involving nonconcerted homolysis and supported by deuterium labeling studies, was proposed to account for some 28% of racemic (E)-product (**129**). The configuration of the carbinyl center was not determined.

In connection with studies on chirality transfer in 2,3-Wittig rearrangements, Tsai and Midland examined the (E)- and (Z)-(R)-1-isopropyl-2-butenyl benzyl ethers (131; equation 32) and (135; equation 33).²⁹ Both ethers rearranged with essentially complete suprafaciality, but the (Z)-isomer (135) showed considerably higher (E)/(Z) and syn/anti selectivity in accord with transition state considerations previously discussed for the analogous propargylic ethers (Scheme 7).



A novel approach to stereocontrol in 2,3-Wittig rearrangements involves the use of chiral chromium tricarbonyl complexes (Table 8a).^{34a} The reactions, which were carried out with racemic ethers, show excellent diastereoselectivity as a consequence of a preferred transition state in which the benzylic oxygen and the R³-substituent and the vinylic grouping and Cr(CO)₃ grouping adopt *anti* orientations (Lc).





*An 80:20 mixture of (E) and (Z) ethers yielded an 80:20 mixture of syn and anti products.

3.11.3.5 2,3-Rearrangement of Dipropargyl Ethers

As noted in Section 3.11.3.3, allyl propargyl ethers undergo highly selective 2,3-rearrangement through the propargylic anion intermediate. Allenic products (138) arising from 2,3-rearrangement of the allylic anion are not observed (equation 34). This regioselectivity may result from preferred formation of the propargylic anion or from the unfavorable geometry associated with a five-membered allenic transition state. Dipropargylic ethers lack this ambiguity. In fact, such ethers afford mixtures of 2,3- and 1,2-rearrangement products on base treatment (Table 9).³⁵ A methyl substituent at the propargylic position significantly retards 2,3- in favor of 1,2-rearrangement whereas geminal substituents defeat both processes. Neither the mono- nor the di-methylated propargyl allyl ethers (144) and (145) gave an allenic rearrangement product (147). In the former case only the propargyl alcohol (146) and traces of aldehyde were isolated, whereas the latter yielded no rearranged products (equation 35).





Allenic products are formed in the anion-initiated rearrangement of cyanohydrin propargylic ethers (148; Table 10).³⁶ The intermediate cyanohydrins (149) are converted to allenyl ketones (150) upon work-up.

3.11.3.6 2,3-Rearrangement of Diallyl Ethers

Unsymmetrical diallyl ethers, like their propargyl allyl counterparts, can theoretically yield isomeric 2,3-rearrangement products depending upon the site of deprotonation. In the first systematic study of



 Table 10
 2,3-Wittig Rearrangement of Cyanohydrin Propargylic Ethers

such reactions, Nakai and coworkers found that the regioselectivity of the rearrangement is related to the stability of the presumed allylic anion intermediate (Table 11).³⁷ Thus allyl or methallyl are deprotonated more readily than crotyl or 1-methyl-2-propenyl (Table 11, entries 1 and 3–7). Allyl and methallyl, on the other hand, are deprotonated with comparable ease (entry 2). In general the less-substituted allylic anion is preferred (equation 36). (Z)-Allylic ethers afford *syn* products with high diastereoselectivity (Table 11, entries 4 and 7), whereas (E)-allylic ethers yield *anti* products, but less selectively (entries 3 and 6). Ethers with α' -substituents rearrange to (E)-products nearly exclusively (entries 1 and 5). Further investigations by Nakai and coworkers confirmed and extended these findings (Table 12).³⁸









*Artefact of sequence rule priority change. $^{\flat}70:30(Z):(E)$.

A transition state model was formulated (Scheme 9) in which the anion-stabilizing vinyl substituent adopts the lower energy equatorial orientation in the envelope conformation (M) for the (E)-allylic ether and (N) for the (Z)-allylic ether.³⁸ The alternative higher energy axial disposition of this grouping as in (O) and (P) would lead to the *syn* and *anti* products, respectively. [It should be noted that these products are enantiomers of the *syn* and *anti* products derived from (N) and (M).] Presumably, interactions between the vinyl substituent and the equatorial (Me) grouping at the migration terminus raises the energy of this transition state slightly in the (E)-allylic ethers, resulting in lower diastereoselectivity. R¹-substituents destabilize the axial vinylic grouping in transition state (O) in favor of (M) leading to an enhanced *anti:syn* ratio (Table 12, entry 3). The lower *syn:anti* ratio seen in entry 8 for the (E)-3-phenylallyl ether may stem from unfavorable interactions between the Me grouping at the migration terminus and the R³phenyl substituent in transition state (N).



Scheme 9 Envelope transition states for 2,3-Wittig rearrangements of diallylic ethers leading to diastereometric and enantiometric products

The question of conformational preferences at the nonmigrating carbinyl center was addressed by Tsai and Midland (Table 13).²⁹ Their results are consistent with the previously described transition state pic-

Rearrangement Reactions

ture for lithiomethyl and propargyl allyl ethers (Schemes 4 and 6). Accordingly, (Z)-allylic ethers rearrange to syn (E)-products with high diastereoselectivity and (E)-allylic ethers favor *anti* (E)-products. The rearrangement proceeds with complete 1,3-syn stereocontrol as expected for a concerted suprafacial process. The Nakai transition state (Scheme 6) with an equatorial-like isopropyl substituent (R¹) accommodates these results.



 Table 13
 Stereoselectivity of 2,3-Wittig Rearrangement of Chiral Diallylic Ethers

^aent-(161)/(162).

Studies on the nonracemic methylenecyclododecyl ethers (164; equation 37) led to similar conclusions.³⁹ With HMPA as a cosolvent the secondary allylic ether (164a) afforded mainly (86:14) the (E)-homoallylic alcohol (165a) of 95% *ee*. The tertiary ether (164b) yielded a 98:2 (E):(Z) mixture, but the major product (165b) was 40% racemic. A dissociative rearrangement process could account for this result. Without HMPA both ethers rearranged in poor yield (5-30%) to nearly 1:1 mixtures of (E)- and (Z)-cycloalkenes.



3.11.3.7 Tandem and Sequential 2,3-Wittig-3,3-Sigmatropic Rearrangements

The 1,5-dien-3-ols produced by 2,3-Wittig rearrangement of diallylic ethers can serve as pivotal intermediates for further elaboration through tandem or sequential 3,3-sigmatropic reactions.^{40,41} The three sequences examined to date are summarized in Scheme 10.

2,3-Wittig rearrangement of the (E)- and (Z)-diallylic ethers (166a) and (166b) afforded mixtures of the 1,5-dienes (167a) and (167b) enriched in the *anti* [from (E)] and *syn* [from (Z)] isomers, respectively. Subsequent oxy-Cope rearrangement of these dienes gave the enals (168a) and (168b) as mixtures of (E)- and (Z)-isomers whose composition depended upon the oxy-Cope reaction conditions, but not the *syn:anti* ratios of the precursor 1,5-dienes (167).⁴² Best results were obtained in refluxing decane, where-upon (E)-enals (168) of 90–95% isomeric purity were produced.

Allylation of the 2,3-Wittig products (167) leads to diallylic ethers (171) in which one of the allylic double bonds is part of a 1,5-diene system. These ethers undergo 3,3-oxy-Cope rearrangement to afford vinyl allyl ethers (170), which rearrange *in situ* by a 3,3-Claisen process to yield the (*E*)-dienals (169).⁴³ Some typical results are summarized in Table 14. The cyclopentanols (175) were shown to arise from al-dehydes (169) by an intramolecular thermal ene reaction.⁴⁴

The Wittig Rearrangement



Scheme 10 Tandem and sequential 2,3-Wittig-3,3-sigmatropic rearrangements

Table 14 Tandem Oxy-Cope-Claisen Rearrangements of 1,5-Dien-3-ol Allylic Ethers



*(CH₂)₂CH-CH₂.

A third variant of tandem rearrangements employs the dienyl allylic ether (171) in a second regioselective 2,3-Wittig rearrangement leading to the 1,5-dienol (174), which undergoes sequential oxy-Cope-Cope rearrangement to the dienal (172). Examples of the latter steps of this sequence are collected in Table 15. In the oxy-Cope rearrangement of the methyl derivative (174; $R^1 = Me$) the silyl ether proved superior to the free alcohol (entries 1 versus 2). No such improvement was realized with the geranyl-derived silyl ether (174; entries 3 versus 4). The ratio of isomeric products (172) was independent of the diastereomeric composition of the starting allylic ethers (171).





Entry	R ¹	R ²	Temperature (°C)	Yield (%)	(E):(Z) (172)
1	Me	H	202 (NMP)	41	67:33
2	Me	TMS	250 (neat)	86	a
3	CH2CH—CMe2	H	202 (NMP)	46	b
4	CH2CH—CMe2	TMS	250 (neat)	46	b

*Not determined. *Mixture of all four geometric isomers.

3.11.3.8 Rearrangement of α -Allyloxy Enolates

The anion that initiates a 2,3-Wittig rearrangement can be derived from α -allyloxy or α -propargyloxy ketones, nitriles (cyanohydrins), acids, esters and amides. Such anions have the virtue of being more easily formed under milder conditions than their methyl, benzyl, propargyl and allyl counterparts. In addition, they lead to more highly functionalized products with greater potential for synthetic applications. Furthermore, through changes in counterion or incorporation of chiral auxilliary groupings, better control of relative and absolute stereochemistry can often be realized.

3.11.3.8.1 *\alpha-Allyloxy ketones*

One of the earliest observations of ketone enolate 2,3-rearrangement was made in connection with attempts to add Grignard reagents to the tetrahydrofurfuryl ketone (176).⁴⁵ Instead of the expected adduct, the hydroxy ketone (177) was obtained (equation 38). Both *cis* and *trans* ketones (176a) and (176b) rearranged to this product. Concordant with a concerted 2,3-process, the nonracemic ketone (177) was produced from optically active ether (176). The acyclic ether (178) also underwent 2,3-rearrangement upon base treatment (equation 39). An analogous α -allyloxy ketone (180) was found to follow a 3,3-rearrangement pathway⁴³ exclusively with KOMe-KH or NaOMe-NaH in toluene or THF (Scheme 11).⁴⁶ With LiOMe-LiH as the base an 80:20 mixture of the 3,3- and 2,3-products (183) and (182) was produced.





Scheme 11 Competing 3,3- and 2,3-rearrangement pathways for a ketone enolate

The dimethylhydrazone derivatives of α -allyloxy ketones undergo exclusive 2,3-Wittig rearrangement upon deprotonation with excess KH in *t*-butyl alcohol (equation 39a).^{46a} The lack of competing 3,3-rearrangement is ascribed to the higher charge density on carbon in the anion as a result of the diminished electronegativity of the hydrazone nitrogen. In the case of (**183b**) only the (*E*)-hydrazone isomer rearranged. The apparent unreactivity of the (*Z*)-isomer is thought to stem from greater steric hindrance to deprotonation.

(183a) R = H (E:Z = 1:3) (183c) 95% yield (183b) R = Me (E:Z = 2.9:1) (183c) 73% yield (39a)

.

The cyclohexanone hydrazones (183e) rearrange similarly (Table 15a). The *cis* isomers (entries 2 and 4) required elevated temperatures for deprotonation owing to the stereoelectronically disfavored equatorial disposition of the α -proton (H-2). The diastereofacial selectivity of the rearrangement was high in all cases examined but the diastereoselectivity at the allylic center was poor (entries 3–5).

	Me ₂ NN 1 2 H H (1	$\frac{1}{R^2}$	KH, Bu'OH THF	(R3f)	
Entry	R ¹	R ²	(E)/(Z)	H-2/H-3	Yield (%)
1 2 3 4 5	H H Me Me H	H H H H Me	>50:1 4:1 >50:1 6:1 >50:1	Trans Cis Trans Cis Trans	91 92 88 (1.4:1) 45 (1.1:1) 81 (1.25:1)

 Table 15a
 2,3-Wittig Rearrangement of Cyclohexanone Hydrazone Derivatives

.

The complimentarity of hydrazone and ketone enolate rearrangements is illustrated in equation (39b).



3.11.3.8.2 O-Allyl cyanohydrins

Allylic ethers of cyanohydrins are easily prepared through phase transfer allylation.⁴⁷ Deprotonation of these ethers with LDA in THF at -78 °C effects 2,3-rearrangement to transient β , γ -unsaturated ketone cyanohydrins, which are transformed during work-up to the ketones (Table 16).48 In an extension of this work, the mixed acetal cyanohydrin ethers (187), prepared by mild acid treatment of the cyanohydrins with 2-methoxy-1,3-butadiene or 1-t-butoxyallene, rearranged to the keto enol ethers (189; equation 40).⁴⁹ Hydrolysis of the enol ethers (189) leads to 1,4-dicarbonyl compounds, which can be cyclized to cyclopentenones.





^a85:15 (*E*):(*Z*). ^bMe₂C = CHCH₂CH₂. ^c90:10 (*E*):(*Z*).

b

b

Mec

Mec



Buⁱ

Me

48

15

36

(189c) 83%

(187c) $R^1 = Me$, $R^2 = Me$, $R^3 = n - C_6 H_{13}$; 63%

Base treatment of the cyanoisopentyl β -furfurylmethyl ether (190; equation 41) affords Elsholtzia ketone (192), a naturally occurring monoterpene.⁵⁰ Even though it disrupts the aromatic furan ring, this rearrangement proceeds readily, illustrating the potency of cyanohydrin anions as 2,3-Wittig initiators.



3.11.3.8.3 *α-Allyloxycarboxylic acids*

2,3-Wittig rearrangement of α -allyloxycarboxylic acid dianions and allyl propargylic dianions (Section 3.11.3.3) might be expected to proceed analogously. In fact, the same high preference for (*E*)-products is observed, but the diastereoselectivity is reversed (Table 17, entries 5 versus 6).⁵¹ A chelated bicyclo [3.3.0] transition state readily explains the *anti* selectivity of (*Z*)-allylic ethers [Scheme 12, compare (**R**) with (**T**)]. The basis for syn selectivity observed with (*E*)-allylic ethers (**Q**) versus (**S**) is less clear.



R ¹	R^2 R^3 CO_2H	2 LDA, THF 78 to 0° C, HCl, H ₂ O	R^2 R^1 HO CO_2H (194) anti	$+ \frac{R^2}{HO}$	^{,3} 20 ₂ Н
<u></u>	(193)		(194) unit	(195) syn	
Entry	R^1	R ²	<i>R</i> ³	(E) or (Z)	Yield (%)
1 2 3 4 5 6	n-C7H15 n-C5H11 Me n-C5H11 H H	H H Me H H	H H H Me Me	(<i>E</i>) (<i>Z</i>)	80 ^a 64 ^a 87 ^a 74 ^a 60 ^b 73 ^c

*>95% (E). b92:8 syn:anti. c79:21 anti:syn.



Scheme 12 Envelope transition states for 2,3-rearrangement of α -allyloxycarboxylic acid dianions

The carboxylic acid variant of the 2,3-Wittig rearrangement has been used in steroidal systems to elaborate the 22-hydroxy substituent found in ecdysone and brassinolide natural products (equation 42).⁵² Whereas the ethyl ester (**196**; R = Et) gave a mixture of products, acid (**196**) was cleanly converted to the (22S)-alcohol (**197**) at -78 °C *via* the dianion followed by acidification and esterification.⁵³ The stereochemistry of this rearrangement is analogous to that observed for the corresponding propargylic ether (**119**) and opposite to that of the acyclic *trans* crotyl analog (Table 17, entry 5). The preference for (**197**) can be clearly seen by comparison of transition states (**U**) and (**V**) in Scheme 13.



Scheme 13 Diastereomeric transition states for carboxylic acid dianion 2,3-rearrangement of a steroidal 17-ethylidene-16α ether

2,3-Wittig rearrangement of the nonracemic (*R*)-propargylic glycolic acid ethers (197d) and (197e) to the (*R*,*S*)- α -hydroxy acids (197f) and (197g) proceeds with complete chirality transfer and 94% diastereoselectivity (equation 42a).^{51b}



3.11.3.8.4 α-Allyloxycarboxylic esters

It was previously noted that enolates of α -allyloxy ketones were capable of either 2,3- or 3,3-rearrangement, depending upon counterion, reaction conditions and substituents.⁴⁶ Ester enolates show a greater propensity for the 2,3-pathway, as illustrated by the geraniol-derived alkoxyacetate (**198**) which afforded hydroxy ester (**200**) as the sole product upon treatment with LDA in THF at -78 to 0 °C.⁵⁴ The 3,3-product (**201**) could be obtained by addition of TBDMS-Cl and HMPA to the enolate at -78 °C and thermal rearrangement of the TBS enol ether (**199b**; Scheme 14).⁴³

Zirconium enolates offer a number of advantages in the 2,3-Wittig rearrangement of α -allyloxyacetic esters. Although both lithium and zirconium enolates afford predominantly syn (Z)-products, regardless of initial allyl geometry, reactions of the latter tend to be more efficient and more selective (Table 18, entries 1, 3 and 5 versus 2, 4 and 6).⁵⁵ In addition, nonracemic secondary allyloxyacetates rearrange with high 1,3-suprafaciality (entries 6–9). These trends reflect steric interactions between the coordinated metal and the substituent R in a bicyclo [3.3.0] folded transition state (Scheme 15). Of the four possible arrangements, (U) offers the best steric environment for an R-substituent. When the allylic position is primary (R = H), then (U) and (X) are enantiomeric, as are (V) and (W).

The Wittig Rearrangement



i, LDA, THF, -78 °C; ii, -78 to 0 °C, H₂O; iii, -78 °C, THF, HMPA, Bu^tMe₂SiCl; iv, 110 °C

Scheme 14

Table 18 2,3-Wittig Rearrangement of α-Allyloxyacetic Ester Enolates



Entry	R	(E)/(Z)	Additive	Yield (%)	Syn:anti	ee (%)
1 2 3 4 5 6 7 8 9 10 11 12 13 14	H H H Me ^a Bu ⁿ n-C8H ₁₇ Pr ⁱ CH ₂ OBn CH ₂ OBn (CH ₂) ₃ OBn (CH ₂) ₃ OBn	(Z)(E)(E)(E)(E)(E)(E)(E)(E)(E)(E)(E)(E)(E)	None Cp2ZrCl2 None Cp2ZrCl2 Cp2ZrCl2 Cp2ZrCl2 Cp2ZrCl2 Cp2ZrCl2 Cp2ZrCl2 Cp2ZrCl2 Cp2ZrCl2 Cp2ZrCl2 Cp2ZrCl2 Cp2ZrCl2 Cp2ZrCl2 Cp2ZrCl2 Cp2ZrCl2 Cp2ZrCl2	20 15 14 47 72 91 81 70 26 72 37 62 57 37	67:33 86:12 82:18 98:2 80:20 99:1 98:2 99:1 98:2 98:2 98:2 98:2 98:2 98:2 98:2 98:2	>96* >96 >96 >96 >96

"Enantiomer of depicted structure. "The (E) product formed predominantly. $^{c}Ca. 2:1$ mixture of (E) and (Z).

The presence of a benzyloxymethyl substituent at the allylic position of ester (202) leads to a predominance of the (E)-syn rearranged product with Hf and Ti enolates (Table 18, entries 10 and 11). In these cases favorable chelation between the ester enolate and the benzyloxy oxygen stabilizes transition state (X; Scheme 15). When the benzyloxy substituent is more remote (Table 18, entries 12–14) such chelation is less favorable and the reaction proceeds through transition state (U).

2,3-Rearrangement of the γ -TMS-allyloxy system (204) gave the (Z)-syn- β -TMS- α -hydroxy esters (205a-c) as single isomers.⁵⁶ These were transformed through reduction and basic elimination to the (Z,E)-dienols (206a-c). Lewis acid promoted elimination converted ester (205c) to the (Z,Z)-dienoate (207; Scheme 16).

Chiral auxilliaries have been employed to direct stereochemistry in 2,3-rearrangements of ester enolates. The (-)-8-phenylmenthyl α -allyloxyacetates (208) rearrange via their lithium enolates to the syn



Scheme 15 Envelope transition states for 2,3-rearrangement of α -allyloxyacetate zirconium enolates



Scheme 16

(2S, 3R)-hydroxy esters (209) with generally high diastereo- and enantio-selectivity (Table 19).⁵⁷ Best results are obtained with allylic ethers possessing substituents at the γ -position (compare entries 1–4 with 5 and 6). The observed enantioselectivity is in accord with the envelope transition state (Y) in which the double bond approaches the enolate from the *si* face to avoid steric confrontation with the bulk of the 8-phenylmenthyl grouping (Scheme 17). Substituents (R¹, R²) on the allylic double bond accentuate this steric interaction, thereby enhancing the enantioselectivity.

Substituents at the δ -position of the allylic ether have also been found to influence diastereoselectivity in α -allyloxyacetate enolate 2,3-rearrangements. The acetonide (211) afforded the α -hydroxy ester (212) containing less than 2% of isomeric rearrangement products (equation 43).⁵⁸ In a closely related example, an analog of (211), R = Bu₃Sn, but with an (*E*) double bond, rearranged upon lithiation to an 86:14 mixture of diastereoisomers. These results can be explained by transition state (AA) in which the oxygen of the acetonide aligns perpendicular to the double bond, thereby lowering the LUMO of the allylic system through π^*/σ^* -orbital overlap (equation 44).



Table 19 2,3-Wittig Rearrangement of (-)-8-Phenylmenthyl α-Allyloxyacetates

^aMajor isomer.



Scheme 17 Diastereomeric transition states for 2,3-rearrangement of (–)-8-phenylmenthyl α -allyloxyacetates



3.11.3.8.5 *α-Allyloxycarboxamides*

The 2,3-rearrangement of α -allyloxyacetamide enolates proceeds readily. (*E*)-Crotyl ethers (213) show high syn diastereoselectivity (Table 20) in accord with acid and ester analogs (193) and (208). Amides of homochiral pyrrolidines derived from (*S*)-proline undergo moderately enantioselective rearrangement (entries 5–8).⁵⁹ 2,5-Disubstituted pyrrolidine amides, on the other hand, rearrange with appreciably higher enantioselectivity but with generally lower diastereoselectivity.⁶⁰ The (*E*)-allylic ether zirconium enolate (Table 21, entry 3) gives the best overall result. In contrast, rearrangement of the isomeric (*Z*)allylic ether zirconium enolate is unselective (entry 5). The diastereomeric cyclohexyl-substituted allylic ethers (215; $R^1 = c-C_6H_{11}$, $R^2 = H$) exhibit double stereodifferentiation. At low conversion (Table 21, entry 7) product of 87% *ee* is isolated, indicating preferential rearrangement of the mismatched diastereoisomer results in a product of lower *ee*.





Entry	R	Base	Yield (%)	Syn:anti	ee (%)*
1	н	LDA	98	96:4	<u> </u>
2	Н	LHMDS	96	96:4	
3	Н	LDA ^b	98	96:4	
4	Н	KHMDS	88	96:4	
5	CH ₂ OH	LDA	с	95:5	52
6	CH ₂ OH	LHMDS	с	95:5	20
7	CH ₂ OMEM	LDA	с	97:3	20
8	CH ₂ OMEM	d	с	95:5	12 ^e

"Major isomer. bTHF-HMPA solvent. ^c'70-80%'. ^dLDA, -100 °C; Cp₂ZrCl₂. ^eEnantiomer at C-2,C-3 of depicted structure.

Table 21 2,3-Rearrangement of Chiral α-Allyloxyacetamides

MOM		омом –	LDA, THF, -78 additive	°C ^{R¹} ∽√ H0 (21	$\int_{O} X_c +$	R ¹ HO (217) an	$\sum_{0}^{1} X_{c}$
Entry	R ¹	R ²	(E)/(Z)	Additive	Yield (%)	Syn:anti	ee (%)*
1 2 3 4 5 6 7 8	H H H n-C7H15 c-C6H11 c-C6H11	H Et Et Et H H H	(E) (E) (Z) (Z)	None None Cp2ZrCl2 None Cp2ZrCl2 None Cp2ZrCl2 Cp2ZrCl2	61 75 65 33 7 82 25 70	67:33 97:3 22:78 50:50	96 92 96 90 92 67 ^b 87 ^b 41 ^b

*Major isomer. ${}^{b}(E):(Z)$ ratio not determined.

n i

- 2

The Wittig Rearrangement

These results are consistent with the chelated transition states depicted in Scheme 18. Steric interactions between the substituent \mathbb{R}^3 and the carboxamide favor (AC) for (*E*)-allylic ethers. The \mathbb{R}^4 -substituent of a (*Z*)-allylic ether, though less affected by this interaction, still experiences a certain degree of steric strain in the *anti* transition state (AB) thus diminishing *anti* selectivity. Enantioselectivity is controlled by the substituents \mathbb{R}^1 and \mathbb{R}^2 on the pyrrolidine ring. As pictured in Scheme 18, bonding occurs preferentially on the face of the enolate *anti* to \mathbb{R}^1 . For the diastereomeric secondary allylic ethers (Table 21, entries 6–8) transition state (AB) represents the matched arrangement for $\mathbb{R}^5 = H$ and \mathbb{R}^6 = alkyl, whereas (AC) is matched for \mathbb{R}^5 = alkyl and \mathbb{R}^6 = H. The former arrangement would lead to an (*E*)-product and the latter to a (*Z*)-product.



Scheme 18 Diastereometric transition states for 2,3-rearrangement of α -allyloxyacetamide enolates

3.11.3.8.6 Allyloxymethyl-oxazolines and -1,3-oxazines

2-Oxazolines (218a) and 5,6-dihydro-1,3-oxazines (218b) with C-2 allyloxymethyl substituents, readily undergo 2,3-rearrangement via their imino-stabilized anions (Table 22).⁶¹ The (E)-crotyl ethers afford the syn products (219a) and (219b) and the oxazoline ether (Z)-(218a) yields mainly the anti product (220a). Surprisingly, the 1,3-oxazine analog (Z)-(218b) gives the syn product (219b) predominantly.

	N 0 R ¹ -R ² (218a) (218b)	base THF, -85 °C 1 h	+ + + + + + + + + + + + + + + + + + +	HO Het anti (220a) (220b)	
Entry	System	(E)/(Z)	Base	Yield (%)	Syn:anti
1 2 3 4 5 6	(218a) ^a (218a) ^a (218a) ^a (218b) ^b (218b) ^b (218b) ^b	(E) (E) (Z) (E) (E) (Z)	BuLi LDA LDA BuLi LDA LDA	98 80 80 92 86 82	66:34 84:16 22:78 98:2 96:4 65:35

Table 22 2,3-Rearrangement of Oxazoline and 1,3-Oxazine Crotyloxymethyl Anions

 ${}^{a}R^{1}-R^{2} = Me_{2}C-CH_{2}$. ${}^{b}R^{1}-R^{2} = Me_{2}C-CH_{2}CHMe$.

These findings have been extended to nonracemic oxazolines (221; Table 23). 62,63 The methoxymethyl-substituted derivatives (entries 1-3) rearrange with moderate enantioselectivity and high *syn* diastereoselectivity (entries 2 and 3). The hydroxymethyl analogs, on the other hand, rearrange with excellent enantioselectivity when KH is used as the base (entries 4, 6 and 8-10). However, the crotyl ethers show poor diastereoselectivity (entries 6, 9 and 10).

The foregoing results for the most part parallel those observed with carboxamides (Section 3.11.3.8.5) and an analogous chelated transition state has been proposed (Scheme 19). Accordingly (AD) represents the lowest energy rearrangement pathway for (E)-allylic ethers. The alternative (AF) places the R¹-sub-



Table 23 2,3-Rearrangement of Chiral Oxazoline Allyloxymethyl Anions

*Major isomer. *Minor isomer.

stituent closer to the oxazoline ring. Interactions between the R¹-substituent and the oxazoline ring must also exist to a degree in (AD) as evidenced by the lower enantioselectivity of rearrangements in systems lacking this substituent (Table 23, entries 1 *versus* 2). Evidently the methoxymethyl- and hydroxymethyl-substituted oxazoline systems (221; $R^2 = Me$ or H) rearrange through different transition states. The potassium counterion plays a major role in the latter cases (Table 23, compare entries 4 and 6 with 5 and 7). The possibility of a preferred (Z)-enolate has been suggested to explain the enantioselectivity of rearrangements carried out with KH-18-crown-6 (Table 23, compare entries 4 with 8 and 6 with 9).



Scheme 19 Transition states for 2,3-rearrangement of oxazoline allyloxy anions

The role of chelation by lithium cations in 2,3-rearrangements of oxazoline allyloxymethyl anions is exemplified by the contrasting behavior of the two tertiary ethers (224; equation 45) and (227; equation

46).⁶⁴ Rearrangement of the former affords a 72:28 mixture of (Z)- and (E)-products (225) and (226), whereas the methoxymethoxy analog (227) yields the (E)-isomer (229) as the sole product. The epimeric tertiary allylic methoxymethoxy-substituted ether (231; equation 47), on the other hand, gives rise to a 64:36 mixture of rearranged products favoring the (E)-isomer (233).



The high (E)-preference observed with ether (227) can be accounted for by the chelated transition state (AH) in which the oxazolidine enamide is held in close proximity to the allylic double bond terminus in the *s*-trans conformation (Scheme 20). The analogous transition state (AI) for the epimer (231) suffers from an eclipsing interaction between the two adjacent carbinyl methyl groupings. Accordingly, the rearrangement of (231) takes place, in part, via a nonchelated *s*-cis conformer to give a mixture of (E)- and (Z)-products. The crotyl ether (228) likewise rearranges with high (E)-selectivity, whereas the epimeric ether (232) gives rise to a mixture of isomers favoring the (E)-product (234). Selectivity for the cooperative diethers (227) and (228) decreases dramatically (to 90:10 and 55:45, respectively) in the presence of HMPA, owing to competitive coordination of the lithium cation with this cosolvent.



Scheme 20 Chelated transition states for 2,3-rearrangement of epimeric tertiary alkoxyallyl oxazolines

Greatly enhanced ratios of the (E)/(Z)-products (234):(236) (>99:1) can be realized by using MeMgBr as the base for rearrangement of the uncooperative diether (232), but the yield is only 20%. Presumably,

chelation is strong with a Mg counterion, but the anionic character of the resulting enamide is relatively weak.

An application of this chemistry to the synthesis of the nargenicins has recently been reported.^{64a}

3.11.3.9 Oxonium Ylide Rearrangements

The 2,3-rearrangement of α -allyloxyacetic esters can be effected with TMSOTf and Et₃N (Table 24).⁶⁵⁻⁷¹ A pathway involving the trimethylsilyloxonium ylide (**240**; equation 48) has been proposed. Accordingly, treatment of the TMS ketone acetals (**239**; R¹ = Me, R² = R³ = H; R¹ = R² = Me, R³ = H; R¹ = R² = H, R³ = Me) with 20 mol % of TMSOTf yields rearranged products (**241**) in ratios comparable to those obtained directly from the corresponding esters (**237**). The enhanced formation of (*Z*)-products is thought to reflect a preference for the *anti* (**AJ**) versus syn (**AK**) R¹/Me₃Si orientation in the oxonium intermediates (Scheme 21).

Table 24 Trimethylsilyl Triflate Promoted 2,3-Rearrangement of α-Allyloxyacetic Esters



Scheme 21 Transition state orientations for trimethylsilyloxonium rearrangements

3.11.3.10 Cyclic 2,3-Rearrangement–Ring Contractions

(AJ) anti

In 1986 the groups of Takahashi⁷² and Marshall⁷³ independently and simultaneously announced a new application of the 2,3-Wittig rearrangement for the construction of medium and large ring carbocycles (Scheme 22). The incorporation of the allylic ether into a cyclic structure (AL) ensures favorable

(AK) syn

proximity of the reacting centers, as in (AM), thereby lowering the activation energy of the rearrangement considerably.



Scheme 22 2,3-Wittig ring contraction

Thus, for example, the cyclic propargylic allylic ether (242; equation 49) rearranges completely within 30 min upon treatment with BuⁿLi in THF-hexane at -20 °C, whereas the acyclic analog (244; equation 50) requires 12 h under comparable conditions. Alcohol (243) serves as a useful intermediate for the synthesis of cembranoid diterpenes.⁷⁴



Takahashi and coworkers studied 2,3-Wittig ring contractions of the isomeric 13-membered diallylic ethers (246), (248) and (250).⁷² The (E,E)-diallylic ether (246; equation 51) rearranges to the (E)-trans-cyclodecenol (247). Interestingly, the (Z,Z)-ether (248; equation 52) yields the (Z)-trans-cyclodecenol (247) with the opposite relative stereochemistry from that observed with acyclic (Z)-allylic ethers (Table 12). The (Z,E)-diallylic ether (250; equation 53) is converted with high regioselectivity to a 9:91 mixture of (E)- and (Z)-products (247) and (249). The latter consists of a 50:41 mixture of trans and cis isomers. The contrasting diastereoselectivity of acyclic (E)-allylic ethers is again noteworthy (Table 12). Evidently, conformational factors in these cyclic diallylic ethers attenuate or override intrinsic transition state preferences of the acyclic analogs (Scheme 9).





The isoprenoid (E,E)-diallylic ether (252; equation 54) affords a 75:25 mixture of the regioisomeric products (253) and (254) in 98% yield. The major isomer (253) was employed in a synthesis of racemic costunolide, a medium ring sesquiterpene. In a related application the TBS ether (255; equation 55) was converted to a 48:52 mixture of regioisomers (256) and (257) in 90% yield.⁷⁵ The former consists of a 81:19 mixture of diastereoisomers (256a) and (256b). Alcohol (256a) was converted to the sesquiterpene (\pm)-haagenolide, along lines developed in the costunolide synthesis.⁷²



In contrast to the foregoing diallylic examples, the 13-membered allylic propargylic ether (258) rearranges to a single regioisomer (259).⁷⁴ The apparent preference for propargylic anion formation is in accord with analogous acyclic systems. In the first application of nonracemic chiral bases for 2,3-Wittig rearrangements, treatment of the allylic propargylic ether (258) with the lithio amide (+)-(260), prepared from (R)-(+)-1-phenylethylamine, afforded the (-)-alcohol (259) in 78% yield and 70% ee.⁷⁶ The enantiomeric base (-)-(260) gave rise to (+)-(259) in comparable yield and ee. These alcohols were converted to the germacranolide sesquiterpene aristolactone (+)-(261) and its enantiomer (Scheme 23). The bases (+)- and (-)-(260) also effect 2,3-Wittig rearrangement of the 17-membered allyl propargyl ether (242) to the cembranoid precursor (243) with an ee of 20-30%. The nine-membered diallylic ether (262; equation 56) afforded the optically active alcohol isopiperitenol (263) of 25% ee upon treatment with (-)-(260).⁷⁷ This rearrangement was considerably slower than those of the 13- and 17-membered ethers (246), (252), (258) and (242), indicative of a more highly strained transition state. The preferential formation of the cis diastereoisomer from the (E)-allylic ether moiety is contrary to results obtained with acyclic (E)-allylic ethers (Table 12). It should be noted that the diallylic ether (262) is incapable of rearranging to the regioisometric product derived from α' -deprotonation because of the strain associated with a trans-cyclohexene product.

The acyclic ethers (264a-c; equation 57) failed to give optically active rearrangement products with the nonracemic amide base (-)-(260). Accordingly, the asymmetric induction observed in the rearrangements of ethers (242), (258) and (262) appears to depend upon the chiral environment provided by the cyclic array.

Macrocyclic ethers with remote sp^3 stereocenters can undergo diastereoselective 2,3-Wittig ring contractions. Thus, the alcohol (266a; equation 58) and several of its ether derivatives (266b) and (266c) upon treatment with BuⁿLi in pentane-THF-TMEDA afforded the four diastereometric rearrangement



products (267)–(270) in varying ratios (Table 25), depending upon the nature of the substituent OR.⁷⁸ Presumably, bulkier OR groupings tend to fix the conformation of the macrocyclic ether, thereby enhancing the diastereoselectivity of deprotonation and ensuing 2,3-rearrangement. Because the controlling substituent is well removed from the reaction center, this type of directing effect is referred to as remote



diastereocontrol. A similar effect may be operational in the rearrangement of the 13-membered ether (255), although in this example the substituent is closer to one of the reacting centers and could therefore exert a direct steric effect.

 Table 25
 Remote Diastereocontrol in Macrocyclic 2,3-Wittig Ring Contraction

OR 6		OR	OH	OR	OH		● OH
(267a) (267b) (267c)	R=H R=THP R=TBS	(268)		(269)		(270)	
R	Yield (%)	(267)	(268)	(269)	(270)	Trans/cisª	Syn/anti ^b
H (a series) THP (b series) TBS (c series)	71 90 94	20 81 74	29 7 8	39 6 14	12 6 4	49:51 88:12 82:18	59:41 87:13 88:12

^a C-1/C-2 relationship. ^b C-1/C-6 relationship.

Remote diastereocontrol can be used to effect chirality transfer in these macrocyclic systems. Thus ring contraction of the nonracemic propargyl ether (+)-(**266c**), followed by Swern oxidation, afforded the ketone (**271**) as a 9:1 mixture of *syn* and *anti* isomers in over 70% yield (Scheme 24). Further transformations led to α - and β -CBT (**274**) and (**275**), two tumor inhibitory constituents of tobacco.⁷⁹ The former was prepared in nonracemic form starting from (+)-(**266c**) of the indicated (*R*)-configuration.⁸⁰ The β -isomer (**275**) was obtained as the racemate starting from racemic (**266c**).



i, BuⁿLi, pentane, -78 °C; ii, (COCl)₂, Et₃N, DMSO; iii, Na, NH₃, THF; iv, (PPh₃)RhCl, H₂, C₆H₆, EtOH

Scheme 24

3.11.4 REFERENCES

- 1. G. Wittig and L. Löhmann, Justus Liebigs Ann. Chem., 1942, 550, 260; G. Wittig and W. Happe, Justus Liebigs Ann. Chem., 1947, 557, 205.
- C. R. Hauser and S. W. Kantor, J. Am. Chem. Soc., 1951, 73, 1437.
- 3. P. T. Lansbury, V. A. Pattison, J. D. Sidler and J. B. Bieber, J. Am. Chem. Soc., 1966, 88, 78; P. T. Lansbury and V. A. Pattison, J. Org. Chem., 1962, 27, 1933; P. T. Lansbury and V. A. Pattison, J. Am. Chem. Soc., 1962, 84, 4295.
- 4. H. J. Schäfer, U. Schölikopf and D. Walter, Tetrahedron Lett., 1968, 2809; U. Schöllkopf and H. Schäfer, Justus Liebigs Ann. Chem., 1963, 663, 22; U. Schöllkopf and W. Fabian, Justus Liebigs Ann. Chem., 1961, 642, 1.
- 5. U. Schöllkopf and K. Fellenberger, Justus Liebigs Ann. Chem., 1966, 698, 80.
- 6. Y. Makisumi and S. Notzumoto, Tetrahedron Lett., 1966, 6393.
- 7. For reviews, see (a) U. Schöllkopf, Angew. Chem., Int. Ed. Engl., 1970, 9, 763; (b) R. W. Hoffmann, Angew. Chem., Int. Ed. Engl., 1979, 18, 563; (c) T. Nakai and K. Mikami, Chem. Rev., 1986, 86, 885.
- 8. U. Schöllkopf, K. Fellenberger and M. Rizk, Justus Liebigs Ann. Chem., 1970, 734, 106.
- 9. V. Rautenstrauch, J. Chem. Soc., Chem. Commun., 1970, 4; (a) L. Crombie, G. Darnbrough and G. Pattenden, J. Chem. Soc., Chem. Commun., 1976, 684.
- 10. N. Sayo, Y. Kimura and T. Nakai, Tetrahedron Lett., 1982, 23, 3931.
- 11. B. B. Snider, N. J. Hrib and L. Fuzesi, J. Am. Chem. Soc., 1976, 98, 7115.
- 12. For a review of the Stevens rearrangement see I. Markó, in 'Comprehensive Organic Synthesis', ed. B. M. Trost, Pergamon Press, Oxford, 1991, vol. 3, chap. 3.10.
- T. Durst, V. D. Elzen and M. J. Le Belle, J. Am. Chem. Soc., 1972, 94, 9261. 13.
- 14. S. L. Schreiber and M. T. Goulet, Tetrahedron Lett., 1987, 28, 1043.
- 15. W. C. Still and A. Mitra, J. Am. Chem. Soc., 1978, 100, 1927.
- M. M. Midland and Y. C. Kwon, Tetrahedron Lett., 1985, 26, 5013.
 M. M. Midland and Y. C. Kwon, Tetrahedron Lett., 1985, 26, 5021.
 M. M. Midland and Y. C. Kwon, Tetrahedron Lett., 1985, 26, 5017.

- 19. L. Castedo, J. R. Granja and A. Mouriño, Tetrahedron Lett., 1985, 26, 4959.
- 20. L. A. Paquette and T. Sugimura, J. Am. Chem. Soc., 1986, 108, 3841.
- 21. L. Castedo, J. L. Mascareñas and A. Mouriño, Tetrahedron Lett., 1987, 28, 2099; (a) J. A. Marshall, E. D. Robinson and A. Zapata, J. Org. Chem., 1989, 54, 5854.
- 22. K. Mikami, K. Azuma and T. Nakai, Tetrahedron, 1984, 40, 2303.
- 23. K. Mikami, T. Maeda and T. Nakai, Tetrahedron Lett., 1986, 27, 4189.
- 24. N. Sayo, E. Nakai and T. Nakai, Chem. Lett., 1985, 1723.
- 25. E. Nakai, E. Kitahara, N. Sayo, Y. Ueno and T. Nakai, Chem. Lett., 1985, 1725; (a) E. Nakai and T. Nakai, Tetrahedron Lett., 1988, 29, 5409.
- 26. (a) K. Mikami, Y. Kimura, N. Kishi and T. Nakai, J. Org. Chem., 1983, 48, 279; (b) Y.-D. Wu, K. N. Houk and J. A. Marshall, J. Org. Chem., 1990, 55, 1421.
- 27. A. D. Kaye, G. Pattenden, and S. M. Roberts, Tetrahedron Lett., 1986, 27, 2033.
- 28. N. Sayo, F. Shirai and T. Nakai, Chem. Lett., 1984, 255.
- 29. D. J.-S. Tsai and M. M. Midland, J. Org. Chem., 1984, 49, 1842.
- M. M. Midland and J. Gabriel, J. Org. Chem., 1985, 50, 1143.
 N. Sayo, K. Azuma, K. Mikami and T. Nakai, Tetrahedron Lett., 1984, 25, 565.
- 32. K. Mikami, K. Kawamoto and T. Nakai, Tetrahedron Lett., 1985, 26, 5799.
- 33. K. Mikami, K. Kawamoto and T. Nakai, Chem. Lett., 1985, 1719; (a) E. Nakai and T. Nakai, Tetrahedron Lett., 1988, 29, 4587.
- 34. J. E. Baldwin and J. E. Patrick, J. Am. Chem. Soc., 1971, 93, 3556; J. E. Baldwin, J. De Bernardis and J. E. Patrick, Tetrahedron Lett., 1970, 353; (a) J. Brocard, M. Mahmoudi, L. Pelinski and L. Maciejewski, Tetrahedron Lett., 1989, 30, 2549.
- 35. M. Huché and P. Cresson, Tetrahedron Lett., 1975, 367.
- 36. B. Cazes and S. Julia, Synth. Commun., 1977, 7, 273.
- 37. T. Nakai, K. Mikami, S. Taya and Y. Fujita, J. Am. Chem. Soc., 1981, 103, 6492.
- 38. K. Mikami, Y. Kimura, N. Kishi and T. Nakai, J. Org. Chem., 1983, 48, 279.
- 39. J. A. Marshall and T. M. Jenson, J. Org. Chem., 1984, 49, 1707.
- 40. For a review of tandem sigmatropic rearrangements see F. E. Ziegler, in 'Comprehensive Organic Synthesis', ed. B. M. Trost, Pergamon Press, Oxford, 1991, vol. 5, chap. 7.3.
- 41. K. Mikami, S. Taya, T. Nakai and Y. Fujita, J. Org. Chem., 1981, 46, 5447.
- 42. For a review of 3,3-oxy-Cope rearrangements see R. K. Hill, in 'Comprehensive Organic Synthesis', ed. B. M. Trost, Pergamon Press, Oxford, 1991, vol. 5, chap. 7.1.
- 43. For a review of 3,3-Claisen rearrangements see P. Wipf, in 'Comprehensive Organic Synthesis', ed. B. M. Trost, Pergamon Press, Oxford, 1991, vol. 5, chap. 7.2.
- 44. For a review of the ene reaction see B. B. Snider, in 'Comprehensive Organic Synthesis', ed. B. M. Trost, Pergamon Press, Oxford, 1991, vol. 5, chap. 1.1.
- 45. A. F. Thomas and R. Dubini, Helv. Chim. Acta, 1974, 57, 2084.
- 46. M. Koreeda and J. I. Luengo, J. Am. Chem. Soc., 1985, 107, 5572; (a) J. I. Luengo and M. Koreeda, J. Org. Chem., 1989, 54, 5419.
- 47. J. M. McIntosh, Can. J. Chem., 1977, 55, 4200.
- 48. B. Cazes and S. Julia, Tetrahedron Lett., 1974, 2077; B. Cazes and S. Julia, Bull. Soc. Chim. Fr., 1977, 925.
- 49. B. Cazes and S. Julia, Bull. Soc. Chim. Fr., 1977, 931.
- 50. B. Cazes and S. Julia, Synth. Commun., 1977, 7, 113.
- 51. T. Nakai, K. Mikami, S. Taya, Y. Kimura and T. Mimura, Tetrahedron Lett., 1981, 22, 69; (a) K. Kakinuma and H.-Y. Li, Tetrahedron Lett., 1989, 30, 4157; (b) J. A. Marshall and X. Wang, J. Org. Chem., 1990, 55, 2995.
- 52. K. Mikami, K. Kawamoto and T. Nakai, Tetrahedron Lett., 1986, 27, 4899.
- 53. M. Koreeda and D. J. Ricca, J. Org. Chem., 1986, 51, 4090.
- 54. S. Raucher and L. M. Gustavson, Tetrahedron Lett., 1986, 27, 1557.
- 55. M. Uchikawa, T. Katsuki and M. Yamaguchi, Tetrahedron Lett., 1986, 27, 4581; S. Kuroda, S. Sakaguchi, S. Ikegami, T. Hanamoto, T. Katsuki, M. Yamaguchi, Chem. Lett., Tetrahedron Lett., 1988, 29, 4768.
- 56. S. Kuroda, T. Katsuki and M. Yamaguchi, Tetrahedron Lett., 1987, 28, 803.
- 57. O. Takahashi, K. Mikami and T. Nakai, Chem. Lett., 1987, 69.
- 58. R. Brückner and H. Priepke, Angew. Chem., Int. Ed. Engl., 1988, 27, 278.
- 59. K. Mikami, O. Takahashi, T. Kasuga and T. Nakai, Chem. Lett., 1985, 1729.
- 60. M. Uchikawa, T. Hanamoto, T. Katsuki and M. Yamaguchi, Tetrahedron Lett., 1986, 27, 4577.
- 61. K. Mikami, K. Fujimoto and T. Nakai, Tetrahedron Lett., 1983, 24, 513.
- 62. K. Mikami, K. Fujimoto, T. Kasuga and T. Nakai, Tetrahedron Lett., 1984, 25, 6011.
- 63. K. Mikami, T. Kasuga, K. Fujimoto and T. Nakai, Tetrahedron Lett., 1986, 27, 4185.
- 64. M. D. Wittman and J. Kallmerten, J. Org. Chem., 1988, 53, 4631; (a) L. T. Rossano, D. J. Plata, J. Kallmerten, J. Org. Chem., 1988, 53, 5189. 65. K. Mikami, O. Takahashi, T. Tabei and T. Nakai, Tetrahedron Lett., 1986, 27, 4511.
- 66. For a review of 2,3-ylide rearrangements see R. Brückner, in 'Comprehensive Organic Synthesis', ed. B. M. Trost, Pergamon Press, Oxford, 1991, vol. 6, chap. 4.6.
- 67. Intramolecular α -diazo ketone initiated 1,2- and 2,3-rearrangements of ethers, allylic ethers and propargylic ethers: E. J. Roskamp and C. R. Johnson, J. Am. Chem. Soc. 1986, 108, 6062.
- 68. Intramolecular α-diazo ketone initiated 2,3-rearrangements of allylic ethers: M. C. Pirrung and J. A. Werner, J. Am. Chem. Soc., 1986, 108, 6060.
- 69. α-Diazo-acetate and -malonate initiated rearrangements of allylic acetals: M. P. Doyle, J. H. Griffin, M. S. Chinn and D. van Leusen, J. Org. Chem., 1984, 49, 1917.
- 70. α-Diazomalonate-initiated 2,3-ylide rearrangements of allylic ethers: W. Ando, S. Kondo, K. Nakayama, K. Ichibori, H. Kohoda, H. Yamato, I. Imai, S. Nakaido and T. Migita, J. Am. Chem. Soc., 1972, 94, 3870.
- 71. α-Diazoacetate-initiated 2,3-ylide rearrangements of allylic ethers: W. Ando, T. Yagihara, S. Kondo, K. Nakayama, H. Yamato, S. Nakaido and T. Migita, J. Org. Chem., 1971, 36, 1732. 72. T. Takahashi, H. Nemoto, Y. Kanda, J. Tsuji and Y. Fujise, J. Org. Chem., 1986, 51, 4315.
- 73. J. A. Marshall, T. M. Jenson and B. S. DeHoff, J. Org. Chem., 1986, 51, 4316; J. A. Marshall, T. M. Jenson and B. S. DeHoff, J. Org. Chem., 1987, 52, 3860.
- J. A. Marshall, J. Lebreton, B. S. DeHoff and T. M. Jenson, J. Org. Chem., 1987, 52, 3883.
 T. Takehashi, H. Nemoto, Y. Kanda and J. Tsuji, *Heterocycles*, 1987, 25, 139.
- 76. J. A. Marshall and J. Lebreton, J. Am. Chem. Soc., 1988, 110, 2925; J. A. Marshall and J. Lebreton, Tetrahedron Lett., 1987, 28, 3323.
- 77. J. A. Marshall and J. Lebreton, J. Org. Chem., 1988, 53, 4108.
- 78. J. A. Marshall, E. D. Robinson and J. Lebreton, Tetrahedron Lett., 1988, 29, 3547.
- 79. J. A. Marshall, E. D. Robinson and R. D. Adams, Tetrahedron Lett., 1988, 29, 4913.
- 80. J. A. Marshall, E. D. Robinson and J. Lebreton, J. Org. Chem., 1990, 55, 227.

4.1 Carbonylation and Decarbonylation Reactions

DAVID J. THOMPSON

ICI Specialty Chemicals, Wilmington, DE, USA

4.1.1	INTRODUCTION	1015
4.1.2 4.1 4.1 4.1 4.1 4.1 4.1	CARBONYLATION MECHANISMS 2.1 Direct Carbonylation 2.2 Substitutive Carbonylation 2.3 Additive Carbonylation 2.4 Multicomponent Carbonylations 2.5 Decarbonylation	1016 1018 1018 1019 1020 1020
4.1.3	FORMATION OF ALDEHYDES	1021
4.1.4	FORMATION OF KETONES	1023
4.1.5	FORMATION OF CARBOXYLIC ACIDS	1026
4.1.6	FORMATION OF ESTERS	1028
4.1.7	FORMATION OF LACTONES	1031
4.1.8	FORMATION OF AMIDES	1034
4.1.9	FORMATION OF LACTAMS AND RELATED N-HETEROCYCLES	1035
4.1.10	FORMATION OF ISOCYANATES	1039
4.1.11	DOUBLE CARBONYLATION	1039
4.1.12	DECARBONYLATION	1040
4.1.13	REFERENCES	1042

4.1.1 INTRODUCTION

Fifty years ago the first well-defined carbonylation reaction was discovered by Otto Roelen.¹ Whilst studying the high pressure, cobalt-catalyzed Fisher–Tropsch synthesis of hydrocarbons from carbon monoxide and hydrogen, he observed that addition of ethylene to the CO/H₂ feed gas led to formation of 1-propanal in high yield. This reaction, subsequently named hydroformylation, as it results in the addition of hydrogen to one end of the C—C double bond and a formyl group at the other, has been intensively studied over the years² and has formed the basis of several industrial processes, for example the synthesis of 1-butanal from propene, and acrylic acid from acetylene. Unfortunately, the reaction generally requires the use of high temperatures (100–300 °C) and pressures (100–1000 bar; 1 bar = 100 kPa), expensive autoclave equipment and the use of large quantities of toxic or unstable catalysts such as [Ni(CO)₄], [Fe(CO)₅] or [HCo(CO)₄].

Subsequently, up until the early 1960s carbonylation chemistry was little considered as a synthetic method for the preparation of fine organic chemicals. In recent years, however, there has been a dramatic

change in this picture, mostly brought about by the discovery of stable but extremely active catalysts based on organophosphine complexes of palladium and rhodium. Many carbonylations can now be carried out below 100 °C at atmospheric pressure, using very small quantities of involatile, air-stable catalyst precursors, such as $[Pd(PPh_3)_2Cl_2]$ or $[RhCl(CO)(PPh_3)_2]$, which are converted to the active catalytic species *in situ*. Moreover, the scope and understanding of carbonylation has grown to such an extent that it can now be regarded, like catalytic hydrogenation, as one of the more generally useful techniques of synthetic organic chemistry, with a well-developed set of guidelines for choice of catalyst and reaction conditions. In most cases the functional group tolerances of catalysts and reaction conditions have been examined and selectivities between different functional groups established, so that reactions using new substrates can be carried out with a degree of confidence impossible a few years ago.

Even after 50 years of continuous development carbonylation chemistry has still not achieved its full potential and new reactions are still being discovered, a good recent example being double carbonylation leading to α -keto carboxylic acids, α -keto esters and α -keto amides.³

Whilst a large number of transition metal complexes have been studied for carbonylation reactions, most transformations can be carried out by selection from a relatively small number of accessible metal complexes (Table 1). Wherever possible, reactions described later in this chapter will avoid the use of volatile and toxic metal carbonyls, particularly $[Ni(CO)_4]$, and will concentrate on the use of readily available metal complex catalysts or reagents which can be used in apparatus familiar to the synthetic organic chemist.

Table 1	Transition Metal	Complexes	Used in	Carbonylation Reactions
---------	------------------	-----------	---------	-------------------------

From To	Aryl halide	Alkyl, vinyl halide	Alkene	. Alkyne
Aldehyde Ketone Carboxylic acid Ester Lactone Lactam Amide	a, b a, b, j a, h a, h a, g a, g a, g a, g	d, f a, f f, g a, f, g a, b, h a a, h	d, e, g a, d, g g, h g, h c, e e	g, h c, h c

4.1.2 CARBONYLATION MECHANISMS

Despite the presence of a formally divalent carbon atom, CO is not in fact a particularly reactive molecule and much of its chemistry depends on the use of either extreme conditions, energetic reagents or some form of catalysis. Perhaps the simplest examples of such catalysis are found in the reactions of carbon monoxide with protic reagents such as alcohols or secondary amines, affording esters or amides of formic acid. These reactions are catalyzed by alkoxide or amide anions, respectively, and, as shown in Scheme 1, the key step is nucleophilic attack on CO by the catalyst to give a strongly basic alkoxyacyl or aminoacyl anion which is immediately trapped by proton transfer from the alcohol or amine, so generating the catalytic species.



Scheme 1

The aminoacyl anions can also be generated as stoichiometric reagents in the form of lithium derivatives which are stable at -75 °C and react with a variety of electrophiles, including alkyl halides, aldehydes and ketones (Scheme 2).⁴



Scheme 2

Organolithium reagents, like lithium amides, also react with carbon monoxide at low temperature $(-100 \ ^{\circ}C)$ via a formal nucleophilic attack to give acyllithium species which can be trapped cleanly by reaction with alkyl or silyl halides. Other trapping agents can also be used; aldehydes and ketones afford α -hydroxy ketones, whereas esters give good yields of α -diketones (Scheme 3). In the absence of such trapping agents, however, complex mixtures of products tend to be obtained on hydrolytic work-up, so that although reaction of phenyllithium with CO yields benzophenone as the major product, appreciable quantities of benzoin, benzil, benzpinacol and benzhydrol are also formed.⁵



Since the highest filled molecular orbital in carbon monoxide is the weakly antibonding carboncentered 5σ -orbital, it would be reasonable to expect electrophilic reagents to attack CO at carbon and, indeed, powerful electrophiles such as carbenium ions do react to give acylium ions [R₃CCO]⁺, which can be hydrolyzed to carboxylic acids *in situ* (the Koch reaction).⁶ However, carbon monoxide is an extremely weak σ -donor and does not form stable complexes with more conventional Lewis acids such as BF₃ or AlCl₃, though there is good evidence for a transient protonated species (solvated [HCO]⁺) in acidcatalyzed reactions of CO with aromatics (Gatterman–Koch synthesis, Scheme 4).⁷

In contrast, transition metals form a vast array of complexes containing coordinated CO. The effect of the metal coordination is to increase the susceptibility of CO towards nucleophilic attack, either by another ligand or by a noncoordinated nucleophile. A detailed description of the various transition metal catalyzed reactions is not appropriate here, especially as excellent reviews are available.⁸ However, it is



Scheme 4

worthwhile to outline the five major types of carbonylation which will describe nearly all the reactions to be detailed later.

4.1.2.1 Direct Carbonylation

This is the simplest of all carbonylations and is best illustrated by the conversion of iodomethane to acetyl iodide,⁹ a reaction crucial to the success of the low pressure Monsanto process for acetic acid production from methanol (equation 1).

$$RX + CO \longrightarrow R X$$
 (1)

4.1.2.2 Substitutive Carbonylation

Direct carbonylation of an organic halide is a rather rare synthetic process and systems in which halide ion is replaced by a nucleophile (equation 2) are much more frequently encountered. This is a particularly versatile type of reaction providing a wide range of 'acyl anion equivalents', which allow the synthesis of many carboxylic acid derivatives from organic halides. As an example, the elementary steps involved in the palladium-catalyzed carbonylation of a bromoarene are shown in Scheme 5, where oxidative addition is followed by CO insertion and reductive cleavage by whichever nucleophile (Nu = OH, OR, NR₂ or F) is present in the system. Substitutive carbonylation of aliphatic halides is also possible, but generally requires more vigorous conditions and the use of platinum-phosphine complexes, such as $[Pt(PPh_3)_2Cl_2].^{10}$

$$RX + CO + Nu^{-} \longrightarrow RCONu + X^{-}$$
(2)



A rather different approach to carbonylation of aliphatic halides involves the carbonyl anion $[Co(CO)_4]^-$ which, being a kinetically stable 18-electron species, attacks the halide by nucleophilic displacement rather than by undergoing oxidative addition. A probable catalytic cycle for cobalt-catalyzed carbonylation of benzyl chloride is shown in Scheme 6.¹¹



4.1.2.3 Additive Carbonylation

As noted in the introduction this class of reaction includes hydroformylation (Y = H; equation 3), the first carbonylation to be discovered. Replacing hydrogen by water or an alcohol yields carboxylic acids (hydrocarboxylation) or esters (hydroesterification), respectively, and although different catalysts may be required the general mechanistic pattern is very similar in all three cases. The reaction is initiated by alkene insertion into a metal-hydride bond and the resulting alkyl ligand migrates to coordinated CO. In cobalt-catalyzed hydroformylation, oxidative addition of hydrogen to the metal is followed by reductive elimination of aldehyde, and the original metal hydride complex is thus regenerated (Scheme 7).



Scheme 7

4.1.2.4 Multicomponent Carbonylations

One of the best illustrations of the ability of homogeneous catalysts to assemble complex organic molecules by sequential insertions is the hydroxybutenolide synthesis shown in Scheme 8, where no fewer than five separate molecules (MeI, HC=CR, H₂O and 2CO) combine under ambient conditions with high selectivity.¹²



4.1.2.5 Decarbonylation

Since most of the elementary reactions involved in carbonylation chemistry are readily reversible (equation 4), it is not surprising that metal complexes which catalyze carbonylation also catalyze decarbonylation under certain conditions. The catalytic decarbonylation of aldehydes by [RhCl(CO)(PPh_3)_2], for example, is shown in Scheme 9.



Scheme 9

4.1.3 FORMATION OF ALDEHYDES

The direct catalytic carbonylation of halides to aldehydes is not readily achieved. Aryl, heterocyclic and vinyl halides, for example, in the presence of $[Pd(PPh_3)_2Cl_2]$, a stoichiometric quantity of tertiary amine and synthesis gas (CO/H₂), are converted to aldehydes, but the conditions are somewhat drastic (80–100 bar, 80–150 °C).¹³ Alkyl halides are even less suitable for this reaction as they tend to undergo dehydrohalogenation to form alkenes, rather than carbonylation. However, using the platinum catalyst $[PtCl_2(PPh_3)_2]$, primary alkyl iodides can be successfully carbonylated to aldehydes in good yield under moderate conditions (equation 5).¹⁰

$$RI + CO + H_2 \xrightarrow{[PtCl_2(PPh_3)_2]} R H$$
(5)

0

Using poly(methylhydrosiloxane) (PMHS) as the hydrogen donor, aryl and benzyl halides can be carbonylated to aldehydes in the presence of $[Pd(PPh_3)_4]$ under mild conditions (3 bar, 80 °C).¹⁴ The reaction only works for iodides; 4-bromophenyl iodide, for example, is converted to 4-bromobenzaldehyde in 95% yield (equation 6).

$$Br \longrightarrow I + CO + PMHS \xrightarrow{[Pd(PPh_3)_4]} Br \longrightarrow CHO$$
(6)

A more versatile palladium-catalyzed formylation of organic halides takes place using tributyltin hydride and carbon monoxide (equation 7).¹⁵ The reaction works for a variety of substrates — aryl, benzyl and vinyl iodides, vinyl triflates and allyl halides. Reaction conditions are mild (1–3 bar CO, 50 °C), and a variety of functional groups can be tolerated. With unsymmetrical allyl halides formylation is regioselective, taking place at the less-substituted allylic position with retention of geometry at the allylic double bond.

$$RI + CO + Bu_3SnH \xrightarrow{[Pd(PPh_3)_4]} R H$$
(7)

Carbonylferrate salts, such as $K^{+}[HFe(CO)_{4}]^{-}$, which is readily prepared from $[Fe(CO)_{5}]$ and ethanolic KOH, react with alkyl bromides or iodides to form alkyliron complexes. Under ambient conditions these complexes undergo insertion of CO and, in the presence of excess CO, aldehyde is eliminated and $[Fe(CO)_{5}]$ is regenerated (Scheme 10).¹⁶ Unfortunately, the reaction is not catalytic and the hydridocarbonylferrate salt must be prepared in a separate step. Using the commercially available salt Na₂[Fe(CO)₄] a similar reaction takes place, but in this case the intermediate acyl complex is anionic and acid treatment is necessary to liberate the aldehyde (Scheme 11).¹⁷

$$KHFe(CO)_4 \xrightarrow{RX} R(H)Fe(CO)_4 \xrightarrow{CO} RCOFe(H)(CO)_4 \xrightarrow{CO} RCHO + [Fe(CO)_5]$$

Scheme 10

$$Na_2Fe(CO)_4 \xrightarrow{RX} Na^+[RFe(CO)_4]^- \xrightarrow{CO} [RCOFe(CO)_4] \xrightarrow{H^+} RCHO + [Fe(CO)_5]$$

Scheme 11

The cobalt-catalyzed reaction of carbon monoxide and hydrogen with an alkene, hydroformylation, is an extremely important industrial process, but it occurs under vigorous conditions (200–400 bar, 150– 200 °C) and is not a particularly selective reaction. In the presence of ligand-modified rhodium catalysts, however, hydroformylation can be carried out under extremely mild conditions (1 bar, 25 °C). The catalytic activity of such rhodium complexes is in fact 10^3-10^4 times greater than that of cobalt complexes and side reactions, such as hydrogenation, are significantly reduced. The reactivity of alkenes in hydroformylation follows a similar pattern to that observed in other carbonylation reactions, *i.e.* linear terminal alkenes react more readily than linear internal alkenes, which in turn are more reactive than branched alkenes. An interesting review describes the application of rhodium-catalyzed hydroformylation to the synthesis of a number of vitamins, terpenes and pharmacologically active compounds.¹⁸ Hydroxycitron-nellal (1), for example, can be produced by hydroformylation of 2,6-dimethyl-6-hepten-2-ol.



Rhodium complexes of the ligand α, α -TREDIP (2) give very high iso regioselectivity in the hydroformylation of styrene under mild conditions, and this has been extended to the synthesis of 2'-(2-methoxy-6-naphthyl)propanal (3), a precursor of the antiinflammatory drug naproxen.¹⁹



Using phosphite-modified rhodium catalysts, otherwise unreactive alkenes, such as 2-methyl-1hexene, limonene and cyclohexene, are hydroformylated under mild conditions.²⁰ 2-Methyl-1-hexene, for example, yields almost exclusively 3-methylheptaldehyde, which is in contrast to the result of cobalt catalysis where a compound with a quaternary carbon is formed (Scheme 12).



Optically active aldehydes can be obtained by hydroformylation of prochiral alkenes, but few successful results have been obtained. One major problem appears to be that the product aldehydes are susceptible to racemization and hence the optical yield deteriorates as the reaction proceeds. Most progress has been made with rhodium catalysts, but platinum-catalyzed reactions have also shown promise. Hydroformylation of styrene, for example, using $PtCl_2/SnCl_2$ in the presence of the chiral ligand (–)-BPPM (4) and triethyl orthoformate gives the enantiomerically pure acetal of 2-phenylpropanal, together with the acetal of 3-phenylpropanal (equation 8).²¹ A number of other chelating biphosphines, for example DIOP



and CHIRAPHOS, have been used with rhodium and platinum systems but generally optical yields do not exceed 50%.²²

$$Ph \rightarrow + CO + H_2 \xrightarrow{[Pt]} EtO \rightarrow OEt + Ph \rightarrow OEt OEt$$
(8)

4.1.4 FORMATION OF KETONES

To date, only a limited number of methods for the synthesis of unsymmetrical ketones under carbonylation conditions have been described. Aryl iodides in the presence of stoichiometric amounts of a zinccopper couple and a catalytic amount of $[Pd(PPh_3)_4]$ react with alkyl iodides in an atmosphere of CO to give unsymmetrical ketones in good yield (equation 9).²³ The reaction works best for aryl halides containing electron-donating substituents and can be applied to either primary or secondary alkyl halides.

ArI + RI + CO
$$\frac{[Pd(PPh_{3})_{4}]}{Zn/Cu} Ar R \qquad (9)$$

The palladium-catalyzed carbonylative coupling reaction of organic halides with organotin compounds is a potentially useful route to unsymmetrical ketones (equation 10).²⁴ Diallyl ketones are synthesized by the reaction of allyl chlorides and allyltin reagents in the presence of CO and a palladium catalyst under mild conditions. This reaction has been used for the high yield synthesis of egomaketone from prenyl chloride and 3-furanyltrimethyltin (equation 11).²⁴ Vinyl iodides and vinyl triflates also react with vinyl-, alkenyl-, alkynyl- and phenyl-tin compounds in the presence of CO and a palladium catalyst to yield the corresponding ketone (equation 12).²⁴ The reaction tolerates carbonyl groups in both reagents as well as alkynic groups in the tin reagents. The (*E*)-configurations of the double bonds in the vinyl iodide and the stannate are retained in the product. The (*Z*)-configuration of the double bond in the stannate, however, is lost under the usual reaction conditions and the (*E*)-isomer of the product predominates. With vinyl triflates the presence of lithium chloride is essential to ensure that the reaction proceeds smoothly. This palladium-catalyzed coupling reaction of triflates with organostannates has been extended to aryl triflates,

$$R^{1}X + CO + R^{2}SnR^{3}_{3} \xrightarrow{[Pd]} R^{1} + R^{3}_{3}SnX \qquad (10)$$





X = I, OTf

$$R^{1} \xrightarrow{\text{OTf}} + CO + R^{2} Sn R^{3}_{3} \xrightarrow{\text{[Pd]}} R^{2} \xrightarrow{\text{(13)}}$$

which react under mild conditions to give good yields of aryl ketones (equation 13).²⁵ The coupling takes place in the presence of functional groups such as alcohol, aldehyde and ester in the coupling reagents.

Methyl aryl ketones can be synthesized *via* carbonylation of aryl iodides and tetramethyltin in the presence of a nickel catalyst (equation 14).²⁶ Unfortunately the reaction cannot be extended to the synthesis of unsymmetrical diaryl ketones since tetraphenyltin does not react under these conditions.

$$ArI + CO + SnMe_4 \xrightarrow{[Ni]} Ar \xrightarrow{[Ni]} (14)$$

0

Reaction of alkyl halides or tosylates with the commercially available $Na_2[Fe(CO)_4]$ gives anionic alkyliron complexes, $[RFe(CO)_4]^-$, which can then react further with alkylating agents to give ketones in good yield (Scheme 13).²⁷ Whereas primary bromides, iodides and tosylates or secondary tosylates can be used in the first alkylation, only more reactive alkylating agents such as benzyl halides or primary iodides can be used in the second stage. Activated alkenes such as acrylonitrile or ethyl acrylate can also be used in the second step.²⁸



Scheme 13

Acyl dienes can be prepared in good yield by reaction of alkyl or acyl halides with conjugated dienes and CO in the presence of catalytic quantities of $[Co(CO)_4]^-$ and a stoichiometric quantity of base (equation 15). By using a phase transfer method increased yields can be obtained under milder conditions (1 bar, 25 °C).²⁹ The reaction is highly stereo- and regio-specific since the acyl group is normally added to the least-substituted carbon atom of the least-substituted double bond, with exclusive formation of the *trans*-acyldiene. The (acyl- π -allyl)cobalt complex (5), an intermediate in this reaction, can also be reacted with stabilized carbanions resulting in alkylation of the unsubstituted π -allyl terminus to give the product (6).³⁰ In all cases studied the alkylation only took place at the unsubstituted π -allyl terminus.



Carbonylation of the cuprate reagent, R₂(CN)CuLi₂, prepared from copper(I) cyanide and an alkyllithium, gives a product which can be used for the direct nucleophilic 1,4-acylation of α , β -unsaturated aldehydes and ketones (equation 16).³¹ The reaction works particularly well with cyclic α , β -unsaturated ketones to give high yields of the expected 1,4-diketone.

A particularly useful synthesis of cyclopentanones involves the coupling of an alkene, an alkyne and carbon monoxide in the presence of dicobalt octacarbonyl (equation 17). The reaction proceeds via an alkyne-cobalt complex (7) and with relatively unreactive alkenes such as cyclopentene it is preferable to synthesize the complex in a separate step.³² With highly strained alkenes such as norbornadiene, how-



ever, only catalytic quantities of $[Co_2(CO)_8]$ are required. Very volatile or gaseous alkenes require the use of an autoclave, but since the reaction proceeds at 1 bar pressure of CO/acetylene less volatile alkenes may be used in hydrocarbon solvents. Fused ring systems are equally readily synthesized by this route and the reaction has been extended to the synthesis of a number of natural products and oxygen heterocycles.³³ 3-Oxabicyclo[3.3.0]octa-6-en-7-ones, for example, are synthesized in moderate yields by intramolecular cyclization of the cobalt complexes of substituted allyl propargyl ethers (equation 18).^{33a}



One major problem associated with this type of reaction is the formation of regioisomeric cyclopentenones when unsymmetrically substituted alkenes are used, although unsymmetrically substituted alkynes prefer an orientation which places the larger substituent in the α -position of the cyclopentenone.^{33c} Several examples have been published which show that steric interactions can bias the regiochemical outcome of the intermolecular cycloaddition. Substituents in the allylic position, for example, exert a modest to excellent degree of regiocontrol (equation 19).³⁴ In this case the larger of the two ring fusion substituents becomes β to the new enone carbonyl.



l-Iodo-1,4-dienes undergo carbonylation in the presence of stoichiometric amounts of $[Pd(PPh_3)_4]$ to give α -methylenecyclopentenones (8).³⁵ By varying the conditions the reaction can be made catalytic in palladium, but the reaction has to be performed in methanol and the product in this case is the methyl ester (9). The reaction is also applicable to the formation of cyclohexenones and spiro compounds, with good yields being obtained in all cases.

Alkenic tosylates undergo cyclization with concomitant carbonylation when treated with Na₂[Fe(CO)₄] (equation 20).³⁶ The reaction is particularly effective for the synthesis of five- and sixmembered rings but seven-membered rings can also be prepared, although in the latter case the reaction is not regiospecific and a mixture of products is obtained. The reaction appears to be limited to monosub-



stituted alkenes since addition of a second substituent on either carbon of the double bond prevents cyclization.



4.1.5 FORMATION OF CARBOXYLIC ACIDS

Early work on the conversion of aryl halides to carboxylic acids was mostly concerned with the use of nickel tetracarbonyl, and although the reaction proceeds under mild conditions more effective and less hazardous syntheses can be achieved using palladium catalysts. In particular, aryl, benzyl, vinyl and heterocyclic halides can be converted to carboxylic acids under mild conditions in two-phase systems.³⁷ The carbonylation reaction is carried out by addition of the organic halide, triphenylphosphine, $[PdCl_2(NCPh)_2]$ and a phase transfer catalyst, such as $Bu^n_4N^+I^-$, to a two-phase system consisting of *p*-xylene and 50% aqueous sodium hydroxide, followed by stirring for several hours under CO at a pressure of 1–10 bar. A great advantage of this system is that continuous extraction of the product acid into the aqueous phase leaves the catalyst and any residual starting material in the organic layer, allowing easy product separation and effectively making heterogeneous the homogeneous catalyst. This procedure also allows selective monocarbonylation of polyhalogenated aromatic substrates such as 1,4-dibromobenzene, since transformation of the first C—X group to carboxy enables the product to be rapidly removed from the organic to the aqueous phase where it is no longer in contact with the catalyst.

A related biphasic synthesis uses $[Co(CO)_4]^-$ as the catalyst, but in this case the reaction is limited to substrates such as benzyl and naphthylmethyl halides which are susceptible to attack by the metal carbonyl anion.³⁸

Iron pentacarbonyl is also an effective catalyst precursor allowing high yield carbonylation of alkyl and aralkyl halides under relatively mild biphasic conditions (equation 21).³⁹ This represents a very interesting and useful development of the earlier procedure which used stoichiometric quantities of the preformed reagent Na₂[Fe(CO)₄]. The reaction works equally well for bromides and chlorides.

$$RX + CO \xrightarrow{[Fe(CO)_5]} RCO_2H$$
(21)

Under conditions of photostimulation $[Co_2(CO)_8]$ in aqueous alkali will catalyze carbonylation of aryl and vinyl halides, including the normally less reactive aryl chlorides, at low pressures and with high efficiency. The active catalyst under these conditions is $[Co(CO)_4]^-$ and this can be generated *in situ* from simple cobalt salts such as $CoCl_2 \cdot 6H_2O$, thus avoiding any need to handle the highly air sensitive $[Co_2(CO)_8]$ or its derived anion (equation 22).⁴⁰

$$ArX + CO + NaOH \xrightarrow{CoCl_2} ArCO_2Na \qquad (22)$$

Arenediazonium salts react stoichiometrically with nickel and iron carbonyls to give aromatic carboxylic acids in moderate yield, but a more reliable procedure involves direct, catalytic carbonylation in the presence of palladium acetate.⁴¹ The reaction proceeds at room temperature under CO pressure (9 bar) and gives a mixed anhydride as the initial product. Hydrolytic work-up produces the aromatic acid in 50–90% yield (Scheme 14).

$$ArN_2^+$$
 + CO + MeCO₂K $\xrightarrow{Pd(OAc)_2}$ $ArCO_2COMe$ \longrightarrow $ArCO_2H$ + MeCO₂H

Scheme 14

Addition of carbon monoxide and water to an alkene, *i.e.* hydrocarboxylation, is catalyzed by a variety of transition metal complexes, including [Ni(CO)₄], [Co₂(CO)₈] and [H₂PtCl₆]. Unfortunately this reaction usually leads to mixtures of products due to both metal-catalyzed alkene isomerization and the occurrence of both Markownikov and anti-Markownikov addition of the metal hydride intermediate to the alkene. The commercially available zirconium hydride [(C₅H₅)₂Zr(H)Cl] can be used as a stoichiometric reagent for conversion of alkenes to carboxylic acids under mild conditions (equation 23).⁴² In this case the reaction with linear alkenes gives exclusively terminal alkyl complexes even if the alkene double bond is internal. Insertion of CO followed by oxidative hydrolysis then leads to linear carboxylic acids in very good yield.



i, Cp₂ZrHCl; ii, CO; iii, NaOH; iv, H₂O₂

Alkynes react readily with stoichiometric quantities of [Ni(CO)₄] under essentially ambient conditions to give, in the presence of aqueous acid, good yields of cis- α , β -unsaturated carboxylic acids (equation 24).² With monosubstituted alkynes the reaction rate decreases with increasing size of the substituent group and with alkyl substituents the branched product isomer usually predominates.

The nickel-catalyzed carbonylation of allyl halides in the presence of alkynes and water produces 2,5dienoic acids in good yields under very mild conditions (equation 25).⁴³ This remarkable four-component reaction probably involves oxidative addition of the allyl chloride to the catalyst, followed by successive insertions of alkyne and CO, and finally hydrolysis. The carbon–carbon double bond derived from alkyne insertion is thus conjugated with the carbonyl group and generally has the (Z)-configuration.

$$Cl + \equiv + H_2O + CO \xrightarrow{[Ni(CO)_4]} CO_2H$$
 (25)

An interesting reaction discovered by Wakamatsu involves the cobalt-catalyzed carbonylation of aldehydes in the presence of primary amides to give *N*-acylamino acids in high yield (equation 26).⁴⁴ By combining this reaction with known catalytic routes to aldehydes, for example isomerization of allyl alcohols or hydroformylation of alkenes, it is possible to achieve the direct synthesis of *N*-acylamino acids from precursors other than aldehydes.⁴⁵

$$\begin{array}{c} O \\ R \\ H \\ H \end{array} + \begin{array}{c} O \\ H \\ R' \\ H \end{array} + \begin{array}{c} O \\ H_2 \\ H_2 \end{array} + \begin{array}{c} O \\ H_2 \\ H_2 \end{array} + \begin{array}{c} O \\ H_2 \\ R \\ CO_2 H \end{array}$$
 (26)

By using a two-phase benzene/aqueous sodium hydroxide medium and a palladium catalyst, allyl chlorides can be converted into 3-butenoic acids in high yield at atmospheric pressure and room temperature (equation 27).⁴⁶ The reaction appears to go equally well using either water soluble or water insoluble

catalysts, but in the latter case some α,β -unsaturated acid is also produced. In the absence of a two-phase system the reaction requires high pressures of CO.



4.1.6 FORMATION OF ESTERS

Nickel tetracarbonyl is an efficient catalyst for the carboalkoxylation of aryl or vinyl halides, but a much better and safer route is via palladium-catalyzed synthesis. Bromides and iodides in the presence of $1-2 \mod \%$ [Pd(PPh₃)₂Cl₂], the reactant alcohol as solvent and a tertiary amine as acid acceptor react at 60–100 °C and atmospheric pressure to produce the corresponding ester in good yield (equation 28).⁴⁷ With iodo compounds even palladium acetate, without any added phosphine, may be used as catalyst. The reaction is unaffected by the presence of a variety of functional groups (ester, ether, nitrile), and strong electron-donating or -withdrawing substituents do not affect the yield of aromatic esters to any significant extent. A more recent application of this reaction indicates that by conducting the reaction in benzyl alcohol aryl iodides are readily converted to the benzyl ester, which can then be easily cleaved to the acid by catalytic hydrogenation.⁴⁸

$$RX + CO + R'OH \xrightarrow{[Pa]} RCO_2R' + HX$$
(28)

At 60-80 °C vinyl halides can be carbonylated with almost complete retention of stereochemistry. For example, carbonylation of (E)-3-iodo-3-hexene in *n*-butanol gives 74% of the corresponding (E)-carbo-butoxylated product (10), with only 6% of the (Z)-isomer.⁴⁷

m 11



Iodoalkanes tend to undergo dehydrohalogenation rather than carbonylation, but more recently it has been shown that primary, secondary and even tertiary alkyl halides can be efficiently carbonylated to esters in the presence of platinum catalysts, under mild conditions, if the reaction takes place under UV irradiation (equation 29).⁴⁹ Alkyl bromides and chlorides are not carbonylated under these conditions and in the presence of visible light irradiation the reaction is very slow. Other transition metal carbonyls, such as $[Co_2(CO)_8]$, $[Ru_3(CO)_{12}]$, $[Os_3(CO)_{12}]$, $[Mn_2(CO)_{10}]$ and $[Re_2(CO)_{10}]$, also catalyze the reaction.

$$RI + R'OH + CO \xrightarrow{[Pi(CO)_2(PK_3)_2]} RCO_2R'$$
(29)

Benzyl halides can be catalytically carbonylated to esters in the presence of $[Rh_2(CO)_4Cl_2]^{50}$ and $[Ni(PPh_3)_2(CO)_2]$.⁴⁷ Esters can also be obtained from alkyl and benzyl halides using stoichiometric quantities of Na₂[Fe(CO)₄] (Scheme 15).⁵¹ Although aryl halides do not react with this reagent, initial

$$RX + Na_{2}[Fe(CO)_{4}] \xrightarrow{\qquad Na^{+}[RFe(CO)_{4}]^{-} \xrightarrow{I_{2}} RCOI \xrightarrow{R'OH} RCO_{2}R'$$

preparation of a Grignard reagent, followed by reaction with [Fe(CO)₅], produces an acyliron intermediate which on subsequent treatment with alcoholic halogen yields the ester (Scheme 16).⁵²

$$ArX + Mg - ArMgX - Fe(CO)_5 ArCOFe(CO)_4 - I_2 ArCO_2R$$

ROH

Scheme 16

Alkyliron complexes, $[(C_5H_5)Fe(CO)_2R]$, undergo carbonyl insertion on treatment with donor ligands, such as triphenylphosphine, to give iron acyls which are configurationally asymmetric. When resolved into their isomers such complexes allow extensive elaboration of the acyl ligand with essentially complete control of the stereochemistry at any new chiral center which may be formed (Scheme 17).⁵³ Moreover, chirality is almost invariably retained when the organic product (ester, acid or amide) is liberated by oxidative cleavage of the iron–acyl bond. Both enantiomers of the acetyl complex $[(C_5H_5)Fe(PPh_3)(CO)(COMe)]$ are now commercially available.



Organohalides, such as iodoalkanes, alkyl chloroacetates and benzyl halides, which are highly susceptible to nucleophilic attack, are readily converted to esters by reaction with carbon monoxide and an alcohol in the presence of a base using Na[Co(CO)₄] as catalyst (equation 30).⁵⁴ Reactions proceed under mild conditions (25 °C, 1 bar CO) and very good yields and selectivities can be obtained. With less reactive halides, however, higher temperatures are required leading to isomerization of the intermediate alkylcobalt complex and hence to a mixture of carbonylated products.

N. 10 (00) 1

$$RX + CO + R'OH \xrightarrow{\text{Na}[Co(CO]_4]} RCO_2R' + HX$$
(30)

By careful selection of the palladium catalyst, aryl trifluoromethanesulfonates can be converted directly to benzoate esters under mild conditions (70 °C, 1 bar CO; equation 31).⁵⁵ The best catalyst appears to be $Pd(OAc)_2-1,3$ -bis(diphenylphosphino)propane and the best solvent DMF. With aromatic compounds bearing electron-withdrawing substituents carbonylations can be carried out under even milder conditions.

$$X \longrightarrow OTf + CO + ROH \longrightarrow X \longrightarrow CO_2R$$
 (31)

The synthesis of an ester by addition of carbon monoxide and an alcohol to an alkene, *i.e.* hydroesterification, has a fairly obvious relationship to the hydrocarboxylation described in Section 4.1.5, where water replaces the alcohol and a carboxylic acid is formed. Not surprisingly, therefore, the same types of catalysts, $[Co_2(CO)_8]$, $[H_2PtCl_6]$ and $[Pd(PPh_3)_2Cl_2]$, are effective for both reactions. Unfortunately, the reaction usually requires very high pressures (200 bar) and necessitates the use of an autoclave. By varying the catalyst and reaction conditions a variety of linear, branched and cyclic alkenes can be carbonylated under these conditions to give the product in good yield (equation 32).⁵⁶ Improved selectivity to the linear ester can be obtained by addition of SnCl₂ to the catalyst system.

$$R$$
 + CO + R'OH \longrightarrow R CO_2R' + R CO_2R' (32)

Stoichiometric hydroesterification can be carried out using the zirconium complex $[(C_5H_5)_2Zr(H)Cl]$, as described in Section 4.1.5 in the context of carboxylic acid synthesis, and, apart from the use of alcoholic rather than aqueous bromine in the final cleavage reaction, the same procedure is applicable (equation 33).⁴²

$$(C_{5}H_{5})Zr(R)Cl \xrightarrow{i, CO}_{ii, Br_{2}} RCO_{2}R'$$
(33)

Palladium-catalyzed hydroesterification of styrene gives predominantly the branched ester, in which a new chiral center has been created at the 2-position (equation 34). Carrying out this reaction at low pressure (1-2 bar) in the presence of the bulky chiral phosphine neomenthyldiphenylphosphine and trifluoroacetic acid leads to significant asymmetric induction $(50\% \ ee)$;⁵⁷ this work seems to represent the only significant advance towards stereocontrol of catalytic hydroesterification reactions.



Palladium salts can bring about oxidative carbonylation of alkenes in the presence of copper(II) salts which can reoxidize Pd⁰ to Pd^{II}. Oxidative carbonylation is favored over simple hydroesterification by the presence of bases and by low temperatures (25 °C) and low pressures (3–15 bar). The products can be α,β -unsaturated esters, dicarboxylic acid esters or β -alkoxy esters. By careful optimization of the conditions (25 °C, 4 bar CO, methanol solvent, CuCl₂ reoxidant and sodium butyrate buffer) high yields of diesters can be obtained (equation 35).⁵⁸

$$R \longrightarrow + CO + R'OH \xrightarrow{Pd^{2\tau}} R \longrightarrow CO_2R'$$
(35)
$$Cu^{2+} CO_2R'$$

The carbonylation of 1-iodo-1,4-dienes in the presence of homogeneous palladium catalysts provides a good example of the ability of such catalysts to promote a complex sequence of metal-centered reactions yielding a single product with a high degree of specificity. Two carbonylation steps are involved, the first leading to formation of a cyclic ketoalkyl ligand and the second to the introduction of an ester function and release of the product from the metal (Scheme 18).³⁵



Alkynes undergo simple hydroesterification under mild conditions to give α,β -unsaturated esters. Addition of the ester function to terminal alkynes frequently occurs at the substituted carbon atom to give branched products, but linear esters may be obtained with high selectivity using palladium-tin catalysts (equation 36).⁵⁹ Under oxidative carbonylation conditions *cis*-diesters are formed.⁶⁰

$$R \longrightarrow + CO + R'OH \xrightarrow{[Pd]} R^{\gamma} \xrightarrow{CO_2R'} (36)$$

4.1.7 FORMATION OF LACTONES

 β -Lactones can be obtained by oxidative carbonylation of alkenes in the presence of water. Ethylene, for example, is converted to β -propiolactone by carbonylation in aqueous acetonitrile at -20 °C using a catalytic amount of PdCl₂ and a stoichiometric quantity of copper(II) chloride (equation 37).⁶¹ Palladium-catalyzed carbonylation of halides can also be used to prepare β -lactones under mild conditions. The reaction takes place at room temperature and pressure in the presence of [PdCl₂(PPh₃)₂] and has been applied to both bromides and chlorides (equations 38 and 39).

$$CH_2 = CH_2 + CO \qquad \frac{PdCl_2}{CuCl_2} \qquad \bigcirc O \qquad (37)$$

$$\begin{array}{c} Ph \\ Br \\ OH \end{array} + CO \qquad \underbrace{\left[PdCl_2(PPh_3)_2 \right]}_{63\%} \begin{array}{c} Ph \\ O \end{array} \\ O \end{array}$$
(38)

$$\begin{array}{c} Cl \\ OH + CO \\ \hline 52\% \\ \end{array} \begin{array}{c} PdCl_2(PPh_3)_2 \\ \hline O \end{array} \end{array}$$
(39)

In recent years, one of the most useful and best-studied areas of transition metal mediated carbonylation reactions is the formation of γ -lactones. Carbonylation of ethynyl alcohols is a very attractive route to α -methylene- γ -lactones. Earlier methods used nickel carbonyl but a more recent method uses a palladium/thiourea catalyst (equation 40).⁶³ A further improvement in the catalytic system is to use PdCl₂ with 1 equiv. of anhydrous SnCl₂ and 2 equiv. of a tertiary phosphine in dry acetonitrile.⁶⁴ Using either of these methods a variety of α -methylenelactones have been synthesized in good yield, including fused ring products. Both *trans*- and *cis*-ethynyl alcohols readily give fused-ring methylenelactones of the corresponding stereochemistry (equation 41).



 α -Methylenebutyrolactones can also be synthesized from homoallylic alcohols in the presence of a base (equation 42). Early work used [Ni(CO)₄], but more recently the reaction has been shown to proceed in the presence of [Ni(CO)₂(PPh₃)₂] and triethylamine.⁶⁵ Unfortunately neither of these reactions are catalytic but the homoallylic alcohol method has been used in the synthesis of the sesquiterpene frulanolide (11).^{65b}

$$R^{2} \xrightarrow[OH]{R^{1}} + CO \xrightarrow[Ni]{Ni} R^{3} \xrightarrow[O]{Ni} (42)$$



The conversion of hydroxy-substituted vinyl halides into α -methylenebutyrolactones can be performed catalytically in the presence of [Pd(PPh_3)_4].⁶⁶ The starting materials for this reaction are prepared from epoxides and α -silylated vinyl Grignard reagents followed by the substitution of the silyl substituent by bromine. Due to the regio- and stereo-specific nature of the ring-opening reaction of the epoxide by the vinyl Grignard reagent, and the noninvolvement of asymmetric carbons in the carbocyclization reaction, the reaction can be extended to the synthesis of optically active lactones from the corresponding optically active epoxides (Scheme 19).



Scheme 19

A further variation on this reaction is the cyclocarbonylation of hydroxy-substituted vinyl halides using a palladium acetate/triphenylphosphine catalyst system. Using this procedure five-, six- and seven-membered α -methylenelactones can be prepared (equation 43).⁶⁷



Acylcobalt tetracarbonyl complexes, formed from Na[Co(CO)₄] and alkyl or acyl halides, react with alkynes to give 2,4-pentadieno-4-lactones (Scheme 20).⁶⁸ The reaction is catalytic in cobalt and yields are around 60% for a variety of substituted alkynes and alkyl halides.



Scheme 20

Five-membered ring lactones are formed from the reaction of primary, secondary or tertiary allyl alcohols with carbon monoxide in the presence of a catalytic amount of PdCl₂/CuCl₂ (equation 44).⁶⁹ The reaction proceeds at room temperature under 1 bar of CO and yields are generally in the range 40–60%.

$$R^{1} \longrightarrow OH + CO \xrightarrow{PdCl_{2}/CuCl_{2}} R^{2} \longrightarrow O = R^{3}$$

$$(44)$$

Dilactones can be synthesized by a palladium-catalyzed stereospecific intramolecular double cyclization of 3-hydroxy-4-pentenoic acids (Scheme 21).⁷⁰ The *cis* stereochemistry of the reaction is rationalized by assuming that attack of Pd^{II} on the alkene is directed by the allylic OH group, forming the intermediate (12).



cis-3-Hydroxytetrahydrofuranacetic acid lactones have been prepared in high yield by intramolecular palladium-catalyzed oxycarbonylation of 4-pentene-1,3-diols under mild conditions (equation 45).⁷¹ A wide range of 4-pentene-1,3-diols can be prepared in essentially quantitative yield by the sequential aldol condensation (reaction of enolates, ketones, esters or lactones with unsaturated aldehydes in THF) and reduction with LiAlH₄ in ether at 100m temperature, thus allowing the ready synthesis of a number of lactones.

$$R + CO \xrightarrow{PdCl_2/CuCl_2} R \xrightarrow{H} O O \\ OH OH + CO \xrightarrow{PdCl_2/CuCl_2} R \xrightarrow{H} O O \\ H \end{pmatrix}$$
(45)

The catalytic carbonylation of halides can also be used to synthesize phthalides (equation 46). The reaction will proceed both in the presence of palladium⁶² and cobalt⁷² catalysts.



The direct carbonylation of arylthallium compounds usually requires high temperatures and pressures, but in the presence of palladium catalysts the reaction proceeds in high yield at room temperature and atmospheric pressure (equation 47).⁷³



Compared with the synthesis of five-membered rings relatively little has been done on the synthesis of δ -lactones. Homoallylic alcohols can be converted into δ -lactones by rhodium-catalyzed hydroformylation followed by oxidation (equation 48).⁷⁴ The thallation and subsequent palladium-catalyzed carbonylation described earlier can also be used for the synthesis of six-membered rings (equation 49).⁷³



Palladium-catalyzed carbonylation of halides has been applied to the synthesis of natural products such as zearalenone (13; Scheme 22)⁷⁵ and curvulin (14; Scheme 23).⁷⁶ In both cases the yields for the carbonylation steps are 70%.



Whereas most carbonylation studies have focused on the use of transition metal catalysts, one recent report describes the facile synthesis of 4-hydroxycoumarins using sulfur-assisted carbonylation (equation 50).⁷⁷ Yields are good for a variety of substrates while reaction conditions are relatively mild.



4.1.8 FORMATION OF AMIDES

Reaction of organic halides with CO and amines using stoichiometric quantities of $[Ni(CO)_4]$ or $Na[Co(CO)_4]$ indicate that metal-promoted amidation is a feasible route to carboxylic amides, but of more use is a palladium-catalyzed synthesis (equation 51).⁷⁸ Aryl, vinyl or heterocyclic halides react with primary or secondary amines in the presence of $[Pd(PPh_3)_2Cl_2]$ under mild conditions (60–100 °C; 1 bar CO) to give high yields of the appropriate amide. The amidation reaction proceeds much more rapidly than the corresponding ester synthesis (Section 4.1.6) so that bromobenzene, for example, reacts with benzylamine and CO to give N-benzyl benzamide about 17 times faster than with 1-butanol under the

same conditions. Amidation also shows greater stereoselectivity than carboalkoxylation with *cis*- and *trans*- vinyl halides, giving amides with essentially complete retention of configuration.

$$RX + CO + R'NH_2 \xrightarrow{[Pd(PPh_3)_2Cl_2]} R + HX$$
(51)

Vinyl and aryl esters of trifluoromethanesulfonic acid are readily converted to amides using a palladium-phosphine catalyst (equation 52).⁷⁹ The reaction proceeds under mild conditions (60 °C; 1 bar CO) to produce amides in excellent yield, but often appears to need the use of the exotic ligand 1,1'-bis(diphenylphosphino)ferrocene (DPPF).

$$R^{1}OTf + CO + R^{2}_{2}NH \xrightarrow{Pd(OAc)_{2}} R^{3}_{3}P R^{1} \xrightarrow{O} NHR^{2}_{2}$$
 (52)

4.1.9 FORMATION OF LACTAMS AND RELATED N-HETEROCYCLES

The synthesis of β -lactams has been achieved using a variety of transition metal mediated reactions, many of which are specific to this type of compound. The vinyl epoxide (15) reacts with [Fe₂(CO)₉] to



Scheme 24

give the lactone complex (16), which then reacts with benzylamine to give the complex (17). Oxidation with cerium(IV) ammonium nitrate produces the β -lactam, in 64% yield, which can be further elaborated into the antibiotic thienamycin (18; Scheme 24).⁸⁰

Iron-alkene complexes have also been used for the synthesis of β -lactams (Scheme 25).⁸¹ The alkene complex (19) is transformed into the pyrroline complex (20) on exposure to ammonia. Reduction with NaBH₄ gives a mixture of stereoisomeric pyrrolidine complexes which are then converted on heating to the diastereomeric chelate (21). Oxidation with air or silver oxide produces the lactam (22).



A more conventional carbonylation route to β -lactams involves the palladium-catalyzed conversion of various 2-bromo-3-aminopropene derivatives into the corresponding α -methylene- β -lactams (equation 53).⁸²

$$R^{2} \xrightarrow{\text{Br}} + CO \xrightarrow{\text{Pd}(OAc)_{2}} \xrightarrow{\text{Pd}(OAc)_{2}} (53)$$

Exposure of an azirine to carbon monoxide in the presence of a catalytic amount of $[Pd(PPh_3)_4]$ gives the bicyclic β -lactam (23) in reasonable yield.⁸³ Attempts to synthesize monocyclic β -lactams from azirines has so far failed, but aziridines can be carbonylated to β -lactams. Depending on the organometallic reagent used two different products can be obtained (Scheme 26). Using catalytic quantities of $[Rh_2(CO)_4Cl_2]$ carbon monoxide inserts selectively into the more-substituted C—N bond,⁸⁴ whereas using a stoichiometric amount of $[Ni(CO)_4]$ it is the less-substituted C—N bond that is carbonylated.⁸⁵



Scheme 26

Azetidine-2,4-diones can be prepared in good yield from aziridinones by carbonylation in the presence of a rhodium catalyst (equation 54).⁸⁶ This conversion can also be brought about using $[Co_2(CO)_8]$, but in this case the reaction is not catalytic.

A somewhat different approach to the synthesis of mono- and bi-cyclic β -lactams involves the reaction of pentacarbonyl(methoxymethyl)chromium complexes with amines under sunlight irradiation (Scheme 27).⁸⁷ The chromium complexes are readily made by reaction of [Cr(CO)₆] with alkyl- or aryl-lithium reagents, followed by alkylation with trimethyloxonium tetrafluoroborate. Sunlight photolysis is required for β -lactam formation, with the product being isolated in 40–76% yield. The reaction is applicable to a wide variety of imines and proceeds with high stereoselectivity. However, the reaction is sometimes limited by the stability of the alkoxycarbene complex and to overcome this the reaction has been extended by using the more stable aminocarbene complex (24). Photolytic reaction of these complexes with imines, oxazines, oxazolines, imidates, thiazines and thiazolines produces β -lactams in fair to good yield, with *trans* stereochemistry being observed in most cases (equation 55).



Scheme 27

 $(CO)_{5}Cr = C + R + NMe + N$

Carbonylation of allylamines can produce pyrrolidones, although relatively high pressures of CO are usually required. Homogeneous rhodium catalysts are preferred and, using similar catalysts, the pyrrolidones can be obtained from allylic halides, CO and ammonia (Scheme 28).⁸⁸



The formation of *ortho* palladium products from α -aryl nitrogen derivatives and palladium salts is well known.⁸⁹ Complexes formed from azobenzene, Schiff bases, tertiary benzylamines and oximes readily undergo insertion of CO into the metal-carbon bond to give, after work-up, a variety of heterocyclic compounds. Unfortunately, such reactions use expensive palladium salts in stoichiometric quantities. However, a number of related reactions have been shown to proceed in the presence of only catalytic quantities of palladium. Isoindolinones, for example, can be synthesized in good yield by reaction of *o*-bromoaminoalkylbenzenes with CO (100 °C, 1 bar) in the presence of catalytic amounts of Pd(OAc)₂, PPh₃ and Buⁿ₃N (equation 56).⁹⁰

Cyclization of N-substituted-o-allylanilines in the presence of carbon monoxide, methanol and a palladium catalyst gives the dihydroindoleacetic acid ester (25) in 70% yield.⁹¹



N-Substituted-1,2,3,4-tetrahydroisoquinolin-1-ones (**26**) can be prepared in good yield by carbonylation of *N*-alkyl-*o*-bromophenethylamines in the presence of catalytic amounts of $Pd(OAc)_2/PPh_3$.⁹⁰ Similarly substituted 2-(2-bromoanilino)pyridines undergo carbonyl insertion to produce pyrido[2,1-*b*]quinazoline derivatives in good yield (equation 57).⁹²



A number of berbin-8-ones have been synthesized from 2'-halogeno-1-benzylisoquinolines by reaction with metal carbonyls.⁹³ In order to optimize the yield of the required product and suppress the main side reaction, dehalogenation, it is important to choose the right metal carbonyl. In some cases $[Co_2(CO)_8]$ appears to be the right reagent (equation 58), whereas in other cases $[Fe_3(CO)_{12}]$ is more suitable (equation 59). The latter reaction can also be brought about by palladium-catalyzed carbonylation of the same starting material in 52% yield.⁹⁴



Using the same procedure as for the synthesis of the related five- and six-membered benzolactams, the benzoazepinone derivatives (27) have been synthesized in good yield.⁹⁰ This palladium-catalyzed procedure has more recently been extended to the synthesis of a number of antibiotics.⁹⁵ A key step in the synthesis of anthramycin, for example, is the palladium-catalyzed insertion of CO into the amine (28).^{95b}



4.1.10 FORMATION OF ISOCYANATES

Aromatic nitro compounds can be carbonylated to isocyanates in the presence of palladium catalysts such as PdCl₂/pyridine (equation 60).⁹⁶ Although high conversions can be obtained the reaction conditions are somewhat vigorous (200 °C, 150 bar CO). Rhodium catalysts have also been shown to be active for this transformation.⁹⁷ From an industrial point of view, however, the simplicity of the reaction offers an attractive route to isocyanates, which are important intermediates for polyurethanes.⁹⁸

$$ArNO_2 + 3CO - ArNCO + 2CO_2$$
(60)

2-Arylazirines are carbonylated under very mild conditions (5 °C, 1 bar CO) in the presence of $[Rh_2(CO)_4Cl_2]$ to give arylvinyl isocyanates in high yield (equation 61).⁹⁹ These products can be isolated or more conveniently converted to carbamates or ureas by reaction with alcohols or amines, respectively.



4.1.11 DOUBLE CARBONYLATION

The realization that doubly carbonylated products can be formed in carbonylation reactions has only emerged in the last ten years, but specific syntheses of many α -keto acids, amides, esters and related products are being developed rapidly. Although many such syntheses depend on the use of CO pressures well above ambient, variation of other parameters such as catalyst, solvent and base can lead to efficient double carbonylation even at relatively low pressures. Selective, cobalt-catalyzed formation of phenylpyruvic acid (97% selectively at 85% yield) from benzyl chloride was thus achieved at 50 bar pressure of CO in propan-2-ol, but more recent work has shown that similar selectivity and an even higher yield can be obtained at less than 2 bar when 1.2-dimethoxyethane is used as solvent (equation 62).¹⁰⁰

$$Ar Cl + 2CO \xrightarrow{[Co]} Ar \xrightarrow{CO_2H} O$$
 (62)

Palladium-catalyzed double carbonylations of aromatic halides have also been studied (equation 63).¹⁰¹ As might be expected, higher pressure is favorable for both reaction rate and selectivity. Selectivity for the formation of keto esters is also maximized by the use of sterically hindered alcohols and solvents of low polarity.

The above reactions take place in the presence of an organic tertiary amine as base, but in the presence of calcium hydroxide reductive double carbonylation takes place with the formation of α -hydroxy acids (equation 64).¹⁰² To maximize the yield of hydroxy acid it is necessary to use isopropyl alcohol as

ArX + CO + HNu
$$\xrightarrow{[PdCl_2(PPh_3)_2]}_{O}$$
 Ar
$$Ar \xrightarrow{V}_{O}$$
 Nu (63)
HNu = R_2NH, RNH_2, ROH, H_2O

solvent and a basic phosphine ligand, such as PMe₃, in combination with calcium hydroxide as the base. Formation of the α -hydroxy acid is rationalized in terms of a double carbonylation followed by Meerwein–Ponndorf-type reduction of the resulting α -keto acid by the isopropyl alcohol/calcium hydroxide mixture.

> ArX + CO + H₂O $\xrightarrow{[Pd]}$ OH Ca(OH)₂ Ar CO₂H (64)

By careful selection of the reaction conditions 3-butenols undergo efficient dialkoxycarbonylation to produce γ -butyrolactone 2-acetic acid esters under 1 bar of CO (equation 65).¹⁰³ Although the roles of the additives are not fully understood, the presence of propylene oxide and ethyl orthoacetate is essential to ensure high selectivity to the double carbonylation product.

$$R + 2CO \frac{PdCl_2/CuCl_2}{MeOH} R - CO_2Me$$
(65)

__ .

0

In the presence of catalytic amounts of $[PdCl_2(NCMe)_2]/CuI$ under CO and O₂, amines react with alcohols to form oxamates (equation 66).¹⁰⁴ The reaction proceeds at room temperature and with β -amino alcohols the corresponding cyclic oxamate is produced (equation 67).

$$R^{1}R^{2}NH + 2CO + R^{3}OH \xrightarrow{PdCl_{2}/CuI} R^{1} \xrightarrow{N} \downarrow OR^{3}$$

$$R^{1} \xrightarrow{R^{2}} + 2CO \xrightarrow{R^{1}} \bigvee_{R^{2}} R^{2} \xrightarrow{N-R^{3}}$$

$$(66)$$

$$R^{1} \xrightarrow{R^{2}} + 2CO \xrightarrow{R^{1}} \bigvee_{R^{2}} R^{2} \xrightarrow{N-R^{3}}$$

$$(67)$$

4.1.12 DECARBONYLATION

Since most of the elementary steps in carbonylation reactions are reversible, it is not surprising that transition metals and their complexes promote the decarbonylation of organic compounds in either a stoichiometric or a catalytic manner.¹⁰⁵ In stoichiometric reactions carbon monoxide removed from the organic compound is retained by the metal complex, as in equation (68), whereas for catalytic behavior this CO must be released, a reaction that often occurs only at high temperatures (>200 °C).

 $RCHO + [RhCl(PPh_3)_3] \longrightarrow RH + PPh_3 + [RhCl(CO)(PPh_3)_2]$ (68)

The rhodium complex [RhCl(PPh₃)₃] readily brings about stoichiometric decarbonylation of aldehydes, acyl halides and diketones. A typical aldehyde decarbonylation is illustrated by equation (69).¹⁰⁶ α,β -Unsaturated aldehydes are decarbonylated stereospecifically (equation 70), while with chiral aldehydes the stereochemistry is largely retained (equation 71).¹⁰⁷

Acyl halides are decarbonylated under mild conditions, but halides containing β -hydrogens also undergo dehydrohalogenation and alkenes rather than alkanes are obtained.¹⁰⁸ Early reports suggested that aromatic acyl chlorides, bromides and fluorides could be decarbonylated, but more recent work indicates that the reaction is not as simple as first thought and variable results have been obtained.¹⁰⁹



By heating above 200 °C catalytic decarbonylation is possible using [RhCl(PPh₃)₃], and is particularly suitable for aromatic aldehydes since aliphatic aldehydes tend to dehalogenate under these conditions to form alkenes (equation 72). Cationic rhodium complexes, for example [Rh(Ph₂P(CH₂)₂PPh₂)₂]⁺, are much more active catalysts and hence reactions can be carried out at below 100 °C.¹¹⁰ Because of the milder conditions aliphatic aldehydes can be decarbonylated to the alkane using this catalyst system. Rhodium catalysts can also be used to decarbonylate α - and β -diketones and keto esters (equations 73 and 74).¹¹¹

$$\begin{array}{ccc} \text{CHO} & [\text{RhCl(PPh_3)_3}] & \text{R}^1 \\ R^2 & & & \\ R^2 & & & \\ 200 \,^{\circ}\text{C} & \text{R}^2 \end{array}$$
(72)

$$\mathbf{R}^{1} \xrightarrow{\mathbf{R}^{2}} \mathbf{R}^{2} \xrightarrow{\mathbf{[Rh]}} \mathbf{R}^{1} \xrightarrow{\mathbf{R}^{2}} \mathbf{R}^{2}$$
(73)

$$R \xrightarrow{O O O} OMe \xrightarrow{[Rh]} R \xrightarrow{OMe} OMe$$
(74)

The palladium-catalyzed decarbonylation of aromatic acyl cyanides proceeds at 120 °C to give the corresponding nitriles in excellent yield.¹¹² Since acyl cyanides are readily prepared by the ruthenium-catalyzed oxidation of cyanohydrins with Bu'OOH this represents a good method for the conversion of aldehydes to nitriles under mild conditions (Scheme 29).



Scheme 29

Decarbonylation of the tricyclic bridgehead acid chloride (29) with a palladium catalyst at 130 °C in the presence of tri-*n*-butylamine gives exclusively the disubstituted alkene (30) in essentially quantitative yield.¹¹³ Under these conditions none of the bridgehead alkene is obtained, but it is assumed that this is an intermediate in the reaction.



4.1.13 REFERENCES

- 1. O. Roelen, Angew. Chem., 1948, A60, 62.
- C. W. Bird, Chem. Rev., 1962, 62, 283. 2.
- 3. T. Sakakura, H. Yamashita, T. Kobayashi, T. Hayashi and M. Tanaka, J. Org. Chem., 1987, 52, 5733.
- V. Rautenstrauch and M. Joyeux, Angew. Chem., Int. Ed. Engl., 1979, 18, 83. C. Narayana and M. Periasamy, Synthesis, 1985, 253. 4.
- 5.
- 6. H. Bahrmann, in 'New Synthesis with Carbon Monoxide', ed. J. Falbe, Springer, Berlin, 1980, p. 372.
- G. A. Olah, F. Pelizza, S. Kobayashi and J. A. Olah, J. Am. Chem. Soc., 1976, 98, 296. 7.
- 8. H. M. Colquhoun, D. J. Thompson and M. V. Twigg, 'Carbonylation', Plenum Press, New York, 1991.
- D. Forster, J. Am. Chem. Soc., 1976, 98, 846.
 R. Takeuchi, Y. Tsuji and Y. Watanabe, J. Chem. Soc., Chem. Commun., 1986, 351.
- 11. L. Cassar and M. Foa, J. Organomet. Chem., 1977, 134, C15.
- 12. H. Alper, J. K. Currie and H. des Abbayes, J. Chem. Soc., Chem. Commun., 1978, 311.
- 13. A. Schoenberg and R. F. Heck, J. Am. Chem. Soc., 1974, 96, 7761.
- 14. I. Pri-Bar and O. Buchman, J. Org. Chem., 1984, 49, 4009.
- 15. V. P. Baillargeon and J. K. Stille, J. Am. Chem. Soc., 1986, 108, 452.
- 16. Y. Takegami, Y. Watanabe, H. Masada and I. Kanaya, Bull. Chem. Soc. Jpn., 1967, 40, 1456.
- 17. M. P. Cooke, Jr., J. Am. Chem. Soc., 1970, 92, 6080.
- 18. H. Siegel and W. Himmele, Angew. Chem., Int. Ed. Engl., 1980, 19, 178.
- 19. J. M. Brown, S. J. Cook and R. Khan, Tetrahedron, 1986, 42, 5105.
- 20. P. W. N. M. Van Leeuwen and C. F. Roobeek, J. Organomet. Chem., 1983, 258, 343.
- 21. G. Parrinello and J. K. Stille, J. Am. Chem. Soc., 1987, 109, 7122.
- C. F. Hobbs and W. S. Knowles, J. Org. Chem., 1981, 46, 4422; G. Consiglio, F. Morandini, M. Scalone and 22. P. Pino, J. Organomet. Chem., 1985, 279, 193.
- 23. Y. Tamaru, H. Ochiai, Y. Yamada and Z. Yoshida, Tetrahedron Lett., 1983, 24, 3869.
- 24. J. K. Stille, Angew. Chem., Int. Ed. Engl., 1986, 25, 508.
- 25. A. M. Echavarren and J. K. Stille, J. Am. Chem. Soc., 1988, 110, 1557.
- 26. M. Tanaka, Synthesis, 1981, 47.
- 27. J. P. Collman, S. R. Winter and D. R. Clark, J. Am. Chem. Soc., 1972, 94, 1788.
- 28. M. P. Cooke, Jr. and R. M. Parlman, J. Am. Chem. Soc., 1977, 99, 5222.
- H. Alper and J. K. Currie, *Tetrahedron Lett.*, 1979, 2665.
 L. S. Hegedus and Y. Inoue, J. Am. Chem. Soc., 1982, 104, 4917.
- 31. D. Seyferth and R. C. Hui, J. Am. Chem. Soc., 1985, 107, 4551.
- 32. I. U. Khand and P. L. Pauson, J. Chem. Res. (S), 1977, 9(M), 168.
- (a) D. C. Billington and D. Willison, Tetrahedron Lett., 1984, 25, 4041; (b) P. Magnus and L. M. Principe, Tetrahedron Lett., 1985, 26, 4851; (c) M. E. Krafft, Tetrahedron Lett., 1988, 29, 999. 33.
- 34. V. Sampath, E. C. Lund, M. J. Knudsen, M. M. Olmstead and N. E. Schore, J. Org. Chem., 1987, 52, 3595.
- 35. J. M. Tour and E. Negishi, J. Am. Chem. Soc., 1985, 107, 8289.
- 36. J. E. McMurry and A. Andrus, Tetrahedron Lett., 1980, 21, 4687.
- 37. L. Cassar, M. Foa and A. Gardano, J. Organomet. Chem., 1976, 121, C55.
- 38. H. des Abbayes and A. Buloup, Tetrahedron Lett., 1980, 21, 4343.
- 39. G. C. Tustin and R. T. Hembre, J. Org. Chem., 1984, 49, 1761; G. Tanguy, B. Weinberger and H. des Abbayes, Tetrahedron Lett., 1983, 24, 4005.
- 40. K. Kudo, T. Shibata, T. Kashimura, S. Mori and N. Sugita, Chem. Lett., 1987, 577.
- 41. K. Nagira, K. Kikukawa, F. Wada and T. Matsuda, J. Org. Chem., 1980, 45, 2365.
- 42. C. A. Bertelo and J. Schwartz, J. Am. Chem. Soc., 1975, 97, 228.
- 43. L. Cassar, G. P. Chiusoli and F. Guerrieri, Synthesis, 1973, 509.
- Wakamatsu, J. Uda and N. Yamakami, J. Chem. Soc., Chem. Commun., 1971, 1540; P. Magnus and M. Slater, Tetrahedron Lett., 1987, 28, 2829.
- 45. I. Ojima, K. Hirai, M. Fujita and T. Fuchikami, J. Organomet. Chem., 1985, 279, 203.
- 46. J. Kiji, T. Okano, W. Nishiumi and H. Konishi, Chem. Lett., 1988, 957.
- 47. A. Schoenberg, I. Bartoletti and R. F. Heck, J. Org. Chem., 1974, 39, 3318.
- A. Huth, I. Beetz, I. Schumann and K. Thielert, Tetrahedron, 1987, 43, 1071. 48.
- 49. T. Kondo, Y. Tsuji and Y. Watanabe, Tetrahedron Lett., 1988, 29, 3833.
- 50. J. B. Woell and H. Alper, Tetrahedron Lett., 1984, 25, 3791; H. Alper, N. Hamel, D. J. H. Smith and J. B. Woell, Tetrahedron Lett., 1985, 26, 2273.
- 51. J. P. Collman, S. R. Winter and R. G. Komoto, J. Am. Chem. Soc., 1973, 95, 249.
- 52. M. Yamashita and R. Suemitsu, Tetrahedron Lett., 1978, 1477.
- 53. S. G. Davies, Pure Appl. Chem., 1988, 60, 13.
- 54. R. F. Heck and D. S. Breslow, J. Am. Chem. Soc., 1963, 85, 2779.
- 55. R. E. Dolle, S. J. Schmidt and L. I. Kruse, J. Chem. Soc., Chem. Commun., 1987, 904.
- 56. J. F. Knifton, J. Org. Chem., 1976, 41, 2885.
- 57. G. Cometti and G. P. Chiusoli, J. Organomet. Chem., 1982, 236, C31.

- J. K. Stille and R. Divakaruni, J. Org. Chem., 1979, 44, 3474.
 J. F. Knifton, J. Mol. Catal., 1977, 2, 293.
 H. Alper, B. Despeyroux and J. B. Woell, Tetrahedron Lett., 1983, 5691.
- 61. J. K. Stille and R. Divakaruni, J. Am. Chem. Soc., 1978, 100, 1303.
- 62. A. Cowell and J. K. Stille, J. Am. Chem. Soc., 1980, 102, 4193.
- J. R. Norton, K. E. Shenton and J. Schwartz, *Tetrahedron Lett.*, 1975, 51.
 T. F. Murray, E. G. Samsel, V. Varma and J. R. Norton, J. Am. Chem. Soc., 1981, 103, 7520.
- 65. (a) M. F. Semmelhack and S. J. Brickner, J. Org. Chem., 1981, 46, 1723; (b) M. F. Semmelhack and S. J. Brickner, J. Am. Chem. Soc., 1981, 103, 3945.

- 66. L. D. Martin and J. K. Stille, J. Org. Chem., 1982, 47, 3630.
- 67. M. Mori, Y. Washioka, T. Urayama, K. Yoshiura, K. Chiba and Y. Ban, J. Org. Chem., 1983, 48, 4058.
- 68. R. F. Heck, J. Am. Chem. Soc., 1964, 86, 2819.
- 69. H. Alper and D. Leonard, Tetrahedron Lett., 1985, 26, 5639.
- 70. Y. Tamaru, H. Higashimura, K. Naka, M. Hojo and Z. Yoshida, Angew. Chem., Int. Ed. Engl., 1985, 24, 1045. 71. Y. Tamaru, T. Kobayashi, S. Kawamura, H. Ochiai, M. Hojo and Z. Yoshida, Tetrahedron Lett., 1985, 26,
- 3207.
- 72. M. Foà, F. Francalanci, E. Bencini and A. Gardano, J. Organomet. Chem., 1985, 285, 293.
- 73. R. C. Larock and C. A. Fellows, J. Am. Chem. Soc., 1982, 104, 1900.
- 74. P. G. M. Wuts, M. L. Obrzut and P. A. Thompson, Tetrahedron Lett., 1984, 25, 4051.
- 75. T. Takahashi, T. Nagashima and J. Tsuji, Chem. Lett., 1980, 369.
- Takahashi, H. Ikeda and J. Tsuji, Tetrahedron Lett., 1980, 21, 3885. 76. Τ.
- 77. T. Mizuno, I. Nishiguchi, T. Hirashima, A. Ogawa, N. Kambe and N. Sonoda, Synthesis, 1988, 257.
- 78. A. Schoenberg and R. F. Heck, J. Org. Chem., 1974, 39, 3327.
- 79. S. Cacchi, P. G. Ciattini, E. Morera and G. Ortar, *Tetrahedron Lett.*, 1986, 27, 3931. 80. S. T. Hodgson, D. M. Hollingshead and S. V. Ley, *Tetrahedron*, 1985, 41, 5871.
- 81. S. R. Berryhill, T. Price and M. Rosenblum, J. Org. Chem., 1983, 48, 158.
- 82. M. Mori, K. Chiba, M. Okita and Y. Ban, J. Chem. Soc., Chem. Commun., 1979, 698.
- 83. H. Alper, C. P. Perera and F. R. Ahmed, J. Am. Chem. Soc., 1981, 103, 1289; H. Alper and C. P. Mahatantila, Organometallics, 1982, 1, 70.
- 84. H. Alper, F. Urso and D. J. H. Smith, J. Am. Chem. Soc., 1983, 105, 6737.
- 85. W. Chamchaang and A. R. Pinhas, J. Chem. Soc., Chem. Commun., 1988, 710.
- 86. D. Roberto and H. Alper, Organometallics, 1984, 3, 1767.
- 87. C. Borel, L. S. Hegedus, J. Krebs and Y. Satoh, J. Am. Chem. Soc., 1987, 109, 1101; L. S. Hegedus and S. D'Andrea, J. Org. Chem., 1988, 53, 3113.
- 88. J. F. Knifton, J. Organomet. Chem., 1980, 188, 223.
- 89. J. M. Thompson and R. F. Heck, J. Org. Chem., 1975, 40, 2667.
- M. Mori, K. Chiba and Y. Ban, J. Org. Chem., 1978, 43, 1684.
 L. S. Hegedus, G. F. Allen and D. J. Olsen, J. Am. Chem. Soc., 1980, 102, 3583.
- 92. J. W. Tilley, D. L. Coffen, B. H. Schaer and J. Lind, J. Org. Chem., 1987, 52, 2469.
- 93. L. S. Trifonov and A. S. Orahovats, Tetrahedron Lett., 1985, 26, 3159.
- G. D. Pandey and K. P. Tiwari, Tetrahedron, 1981, 37, 1213.
 (a) M. Mori, Y. Uozumi, M. Kimura and Y. Ban, Tetrahedron, 1986, 42, 3793; (b) M. Ishikura, M. Mori, M. Terashima and Y. Ban, J. Chem. Soc., Chem. Commun., 1982, 741.
- 96. F. J. Weigert, J. Org. Chem., 1973, 38, 1316.
- 97. S. S. Novikov, V. I. Manov-Yuvenskii, A. V. Smetanin and B. K. Nefedov, Dokl. Chem. (Engl. Transl.), 1980, 251, 135.
- 98. K. Weissermel and H. J. Arpe, 'Industrial Organic Chemistry', Verlag-Chemie, Weinheim, 1978.
- 99. T. Sakakibara and H. Alper, J. Chem. Soc., Chem. Commun., 1979, 458.
- 100. M. Foa and F. Francolanci, J. Mol. Catal., 1987, 41, 89.
- 101. T. Sakakura, H. Yamashita, T. Kobayashi, T. Hayashi and M. Tanaka, J. Org. Chem., 1987, 52, 5733.
- 102. T. Kobayashi, T. Sakakura and M. Tanaka, Tetrahedron Lett., 1987, 28, 2721.
- 103. Y. Tamaru, M. Hojo and Z. Yoshida, Tetrahedron Lett., 1987, 28, 325.
- 104. S.-I. Murahashi, Y. Mitsue and K. Ike. J. Chem. Soc., Chem. Commun., 1987, 125.
- 105. J. Tsuji and K. Ohno, Synthesis, 1969, 157; J. A. Kampmeier, S. H. Harris and I. Mergelsberg, J. Org. Chem., 1984, 49, 621.
- 106. B. M. Trost and M. J. Bogdanowicz, J. Am. Chem. Soc., 1973, 95, 2038.
- 107. H. M. Walborsky and L. E. Allen, J. Am. Chem. Soc., 1971, 93, 5465.
- 108. J. K. Stille, F. Huang and M. T. Regan, J. Am. Chem. Soc., 1974, 96, 1518.
- 109. R. E. Ehrenkaufer, R. R. MacGregor and A. P. Wolf, J. Org. Chem., 1982, 47, 2489.
- 110. D. H. Doughty and L. H. Pignolet, J. Am. Chem. Soc., 1978, 100, 7083.
- 111. K. Kaneda, H. Azuma, M. Wayaku and S. Teranishi, Chem. Lett., 1974, 215.
- 112. S.-I. Murahashi, T. Naota and N. Nakajima, J. Org. Chem., 1986, 51, 898.
- 113. K. Hori, M. Ando, N. Takaishi and Y. Inamoto, Tetrahedron Lett., 1986, 27, 4615.

4.2 Carbon–Carbon Bond Formation by C––H Insertion

DOUGLASS F. TABER

University of Delaware, Newark, DE, USA

4.2.1 INTRODUCTION	1046
4.2.2 INTERMOLECULAR CH INSERTION	1046
4.2.2.1 Abstraction-Recombination	1046
4.2.2.2 Carbenoid Insertion	1047
4.2.2.3 Metallocarbene Insertion	1047
4.2.3 INTRAMOLECULAR C-H INSERTION: CARBACYCLES	1048
4,2.3.1 Introduction	1048
4.2.3.2 Photochemical	1048
4.2.3.3 Alkylidene Carbenes	1049
4.2.3.3.1 Alkynone pyrolysis	1049
4.2.3.3.2 Diazaalkenes	1049
4.2.3.3.3 Alkylidene carbenoids	1050
4.2.3.4 Alkyl Carbenoids	1051
4.2.3.4.1 Thermal insertions	1051
4.2.3.4.2 Copper-catalyzed diazo insertions	1051
4.2.3.4.3 Rhodium-catalyzed diazo insertions	1051
4.2.4 INTRAMOLECULAR C—H INSERTION: HETEROCYCLES	1056
4.2.4.1 Rhodium-mediated Cyclization	1056
4.2.4.1.1 β-Lactam synthesis	1056
4.2.4.1.2 Furanone synthesis	1056
4.2.4.1.3 Pyrrole synthesis	1057
4.2.4.2 Photochemical Cyclization	1057
4.2.4.2.1 γ -Lactam synthesis	1057
4.2.4.2.2 Macrolactam synthesis	1037
4.2.4.2.3 Macrollae synthesis: remote functionalization	1058
4.2.5 APPLICATIONS TO NATURAL PRODUCT SYNTHESIS	1058
4.2.5.1 Photochemical: (-)-Punctatin A	1058
4.2.5.2 Alkylidene Carbene: Clovene	1058
4.2.5.3 Rhodium-mediated C-H Insertion	1059
4.2.5.3.1 Pentalenolactone E (Cane)	1059
4.2.5.3.2 (+)- α -Cuparenone	1060
4.2.5.3.5 Pentalenolactone E (Taber)	1060
4.2.3.3.4 (+)-Estrone methyl ether	1001
4.2.6 DIRECTIONS FOR THE FUTURE	1062
4.2.7 REFERENCES	1062

4.2.1 INTRODUCTION

Traditionally, carbon-carbon bond forming processes have required that both organic fragments be specifically activated. This approach is exemplified by the alkylation of an enolate anion (1) with an alkyl halide (2; equation 1).



There are isolated reports of an alternative approach, in which the carbon-carbon bond is formed by direct insertion into an unactivated C—H bond. Often, these are special cases. The conditions under which carbonylation of adamantane (4; equation 2) is observed, for instance, would not be expected to be equally efficacious for cyclohexane.¹



The purpose of this chapter is to review those methods for constructing C-C bonds from unactivated C-H bonds that appear to have some synthetic generality. Both intermolecular and intramolecular reactions will be considered.

4.2.2 INTERMOLECULAR C—H INSERTION

4.2.2.1 Abstraction-Recombination

The most common strategy for laboratory-scale hydrocarbon functionalization is hydrogen atom abstraction followed by free radical recombination. This approach is exemplified by the method of Tanner for cyanation of hydrocarbon (7) to give (8; equation 3).²



It is possible to achieve some selectivity using this method. As illustrated above, tertiary C—H sites are much more reactive than primary C—H sites. Radical-stabilizing functional groups also impart selectivity, as illustrated by the regioselective functionalization of (9; equation 4).³



Crabtree, by combining ease of free radical generation with the site selectivity imparted by functional groups that stabilize adjacent free radicals, has developed a very practical approach to hydrocarbon carbafunctionalization, as illustrated by the conversion of cyclohexane to (13; equation 5).⁴ The real elegance of this method is that it does not require highly reactive reagents. This photochemically initiated dimerization of hydrocarbons proceeds efficiently in the presence of even a very low concentration of mercury vapor.



Hill has shown that the nature of the product formed in such an insertion reaction can vary with the substrate (equations 6 and 7).⁵ It is thought that bond formation using the procedure he has developed also proceeds *via* a free radical intermediate. A secondary radical, as from (14), will react directly with acetonitrile to form a new carbon-carbon bond. A tertiary radical, as from (16), on the other hand, will be oxidized to the corresponding carbocation before reacting with acetonitrile.



4.2.2.2 Carbenoid Insertion

Intermolecular insertion of free carbenes or carbenoids into C—H bonds is not, in general, an efficient process. However, if the C—H is especially activated, as with an alkoxide such as (18; equation 8), the reaction can be preparatively useful.⁶



4.2.2.3 Metallocarbene Insertion

In 1981, Noels reported that rhodium(II) carboxylates, originally developed as cyclopropanation catalysts, smoothly catalyze the addition of ethyl diazoacetate (21; equation 9) to a variety of alkanes.^{7,8} While some differentiation between possible sites of insertion is observed, selectivity is not as high for this carbenoid process as it is for the free radical processes illustrated above.

Rhodium-mediated intermolecular C—H insertion is thought to proceed via oxidative addition of an intermediate rhodium carbene into the alkane C—H bond. Evidence that the rhodium and its ligands are directly associated with the product-determining transition state has been put forward by Callot, who ob-



served that the differential selectivity observed for C—H insertion into (24) is a function of the rhodium complex employed (equation 10).⁹



4.2.3 INTRAMOLECULAR C-H INSERTION: CARBACYCLES

4.2.3.1 Introduction

Unlike the intermolecular examples of C—H insertion mentioned above, most of which are relatively recent, intramolecular C—H insertion has long been recognized. In a late stage of the biosynthesis of cycloartenol (29), for instance, the cyclopropane ring is formed by such a process (equation 11).¹⁰



4.2.3.2 Photochemical

Photochemical C—H insertion of a ketone (30) will proceed by initial photoexcitation to give an excited state that can usefully be considered to be a 1,2-diradical. Intramolecular hydrogen atom abstraction then proceeds, to give a 1,4- or 1,5-diradical, which can collapse to form the new bond. This approach has been used to construct both four- and five-membered rings (equations 12-14).¹¹⁻¹³ Photochemically mediated cyclobutanol formation is known as the Norrish type II reaction.



1048



4.2.3.3 Alkylidene Carbenes

4.2.3.3.1 Alkynone pyrolysis

The first preparative use of intramolecular C—H insertion in organic synthesis was developed by Dreiding, who reported in 1979 that, on flash vacuum pyrolysis, a conjugated alkynyl ketone such as (36) is smoothly converted to a mixture of the cyclized enones (37) and (38) (equation 15).¹⁴ This elegant reaction apparently proceeds *via* isomerization of the alkyne to the corresponding alkylidene carbene.



It should be noted that despite a 3:2 statistical predominance of primary C—H bonds over secondary C—H bonds, a marked preference for insertion into the latter is observed.

In subsequent work, Dreiding demonstrated that the straight chain conjugated alkynone (39) cyclizes smoothly to the corresponding cyclopentenone (40; equation 16).¹⁵



4.2.3.3.2 Diazaalkenes

In 1982, Gilbert reported that an aliphatic ketone or aldehyde on exposure to dimethyl (diazomethyl)phosphonate is converted to the corresponding cyclopentene (41 \rightarrow 42 and 43; equation 17).¹⁶

1049

This reaction apparently proceeds by way of the normal phosphonate condensation product, the diazoalkylidene, which then spontaneously loses nitrogen to form the transient alkylidene carbene.



Careful work showed that, after statistical corrections were applied, the reactivity of a C—H bond toward insertion was approximately — $CH_3 = 0.003$; — CH_2 — = 1.0; Ph— CH_2 — = 7.5; R₃C—H = 18.6. These relative reactivities are very similar to those previously observed by Wolinsky for intramolecular C—H insertion by an alkylidene carbenoid generated from a vinyl bromide.¹⁷

In subsequent work, Gilbert showed that the alkylidene carbene insertion reaction proceeds with retention of absolute configuration.¹⁸ Using this approach, cyclopentene (**45**) and cyclohexene (**46**) were prepared in high enantiomeric purity (equation 18).



4.2.3.3.3 Alkylidene carbenoids

The insertion reactions described above probably proceed via the free alkylidene carbenes. The analogous alkylidene carbenoids also insert efficiently into remote C—H bonds. Ochiai has demonstrated that such alkylidene carbenoids are conveniently generated from the corresponding iodinium tosylates.¹⁹ Depending on the substitution pattern employed, either [5 + 0] cyclization to give (**49**; equation 19), or [2 + 3] cyclization to give (**52**; equation 20), can be obtained.


4.2.3.4 Alkyl Carbenoids

4.2.3.4.1 Thermal insertions

The simple thermal insertion of an unsubstituted alkyl carbene or carbenoid into a remote C—H bond is, in general, not a useful synthetic method. Activation of the C—H bond by an α -alkoxy group, as with the cyclization of (53) to (54) reported by Cohen, makes this process more efficient (equation 21).²⁰



4.2.3.4.2 Copper-catalyzed diazo insertions

There are several isolated examples of conformationally constrained α -diazo ketones that, under catalysis by Cu salts, smoothly undergo intramolecular C—H insertion.²¹ This cyclization was investigated in some detail by Wenkert, who found that it was not, in the acyclic series, a preparatively useful synthetic method (equation 22).²²



4.2.3.4.3 Rhodium-catalyzed diazo insertions

(i) Initial cyclization studies

Hubert in 1976 reported that rhodium acetate efficiently catalyzes diazo insertion into an alkene, to give the cyclopropane.⁸ In 1979, Southgate and Ponsford reported that rhodium acetate also catalyzes diazo insertion into a C—H bond.²³ Prompted by these studies, Wenkert then demonstrated that cyclization of (**58**) to (**59**) proceeded much more efficiently with the rhodium carboxylates than it had with copper salt catalysis (equation 23).²²



Concurrently, Noels had reported that rhodium carboxylates smoothly catalyze the intermolecular C—H insertion of ethyl diazoacetate into alkanes.⁷ Following up on this report, Taber demonstrated that the open chain α -diazo β -keto ester (60) cyclizes smoothly under rhodium acetate catalysis to give the corresponding cyclopentane (61; equation 24).²⁴ In contrast to the copper-mediated cyclization cited above (equation 22), the six-membered ring product is not observed. The insertion shows significant electronic selectivity. Although there is a 3:1 statistical preference for methyl C—H, only the methylene C—H insertion product (61) is observed (equation 24).



In a direct comparison, it has been found that with the same substrate (62), $Rh_2(OAc)_4$ directs the reaction toward C—H insertion, to give (63), whereas Cu bronze favors alkene insertion, to make the cyclopropane (64; Scheme 1).^{24,25}



Scheme 1

Other electron-withdrawing groups are compatible with both diazo transfer and cyclization. Both the β -keto sulfone (65) and the β -keto phosphonate (67) have been cyclized using rhodium acetate catalysis (equations 25 and 26).^{26,27} The cyclized keto phosphonate (68) can be further reacted with formaldehyde to make the α -alkylidenecyclopentanone (69; equation 26).²⁷



(ii) Selectivity

Steric and electronic selectivity in the C—H insertion process has been studied in some detail.²⁸ The cyclization of (70) was shown to give (71) with high diastereoselectivity (equation 27).

Wenkert demonstrated that cyclization of the diazomethyl ketone corresponding to $(72)^{22}$ proceeds to give predominantly the *trans*-hydrindan (equation 28).²² Taber showed that the degree of diastereoselectivity in the cyclization of (72) to (73) and (74) is affected by the ligands on rhodium (equation 28).²⁸

As with the work of Gilbert cited above, electronic selectivity can also be observed.²⁸ It is striking that cyclization of (75) proceeds to give preferentially (76) rather than (77; equation 29). The observation that



a benzylic — CH_2 — is significantly less reactive than an aliphatic — CH_2 —, in direct contrast to the 'free carbene' work of Gilbert and others, ¹⁶ led to the suggestion that the reaction is proceeding by complexation of a coordinatively unsaturated rhodium carbene complex with the electron density in the target C—H bond.²⁸ The phenyl substituent, being inductively electron-withdrawing, might then be expected to deactivate the benzylic methylene.



(iii) Mechanism

While the detailed mechanism of the rhodium-mediated cyclization is not known, a working hypothesis that accommodates all of the observations to date is that it proceeds *via* a Rh^I carbene complex such as (82; Scheme 2). Oxidative addition of the rhodium carbene into a remote C—H bond would give a metallacycle such as (83). Carbene rearrangement could then give (84), which could undergo reductive elimination to give the product (85), along with the regenerated catalyst (79).

This mechanistic hypothesis led to two predictions.²⁸ The first was that, since the product-determining step would seem to be oxidative addition of an electron deficient Rh center into the C—H bond, a C—H bond near an electron-withdrawing group should be less reactive than an isolated C—H bond. This prediction was borne out remarkably well by the work of Stork and Nakatani.²⁹ They demonstrated that cyclization of (**86**) proceeds to give exclusively (**87**; equation 30). Thus, a C—H bond even β to the electron-withdrawing carboxyl group is much less reactive toward rhodium-mediated C—H insertion than is an isolated aliphatic C—H bond.



(87) only product observed

(86)

1053





(81)









(82)

Scheme 2

The other prediction stemmed from the expectation that in the transition state leading to C—H insertion, the forming ring would adopt a lowest energy, staggered, chair-like conformation. It followed that it might be possible to design an enantiomerically pure ester that would destabilize one of the two enantiomeric transition state conformations and thus direct the absolute stereochemistry of the cyclization. Indeed, even though the nearest stereogenic center in the directing alcohol is eight atoms removed from the C—H insertion site, cyclization of (88) to (89) and (90) proceeds with reasonably good selectivity (equation 31).³⁰ The cyclized esters are diastereomers and can be separated chromatographically.



(iv) Simple α -diazo ketones and esters

It could be expected that if a simple α -diazo ketone with β -C—H bonds were exposed to the rhodium catalyst, metallocarbene formation would proceed as usual, but that β -hydride elimination would com-

pete with the desired 1,5-insertion. Such a β -hydride elimination could, in fact, be viewed as a 1,2-insertion (91 \rightarrow 93; equation 32).



In recent work, Taber and Hennessy have found that simple α -diazo ketones and esters can, in fact, be induced to undergo 1,5-insertion in preparatively useful yields.³¹ It was already known²⁸ that, in the rhodium-mediated insertion process, methyl C—H is electronically less reactive than methylene C—H or methine C—H. It therefore seemed likely that competing β -hydride elimination would be least likely with a diazoethyl ketone. In fact, on cyclization of (94), only a trace of the enone product from β -hydride elimination is observed (equation 33). The main side reaction competing with 1,5-insertion is dimer formation.



In a direct competition between 1,2- and 1,5-insertion into methylene C—H, the relative proportion of products depends on the rhodium carboxylate used. Rhodium benzoate is the most efficient catalyst so far found for the cyclization of α -diazo ester (97) to (98) (equation 34).³¹



With the long chain α -diazo ketone (100), 1,5-insertion could proceed to put the carbonyl outside the ring or to include it in the ring. In fact, only the product (101) from the first of these two cyclization modes is observed (equation 35).³¹ The alternative cyclopentane (103) is not formed. As with the α -diazo ester, the relative proportion of 1,2- and 1,5-products depends on the rhodium carboxylate used. Throughout these studies, it has been observed that the alkene (102) obtained from 1,2-elimination is cleanly cis.^{31,32}



The complete absence of the cyclopentanone product (103) in the reaction above suggests that, in the intermediate rhodium carbenoid, there is a significant preference for the conformation (104), in which the ketone carbonyl and the rhodium center are syn to one another (equation 36).



4.2.4 INTRAMOLECULAR C—H INSERTION: HETEROCYCLES

4.2.4.1 Rhodium-mediated Cyclization

4.2.4.1.1 β-Lactam synthesis

The first observation of rhodium-mediated intramolecular C—H insertion was by Southgate and Ponsford at Beecham Pharmaceuticals, who reported that (106) on exposure to a catalytic amount of rhodium acetate cyclizes cleanly to the β -lactam (107; equation 37).²³ This approach to thienamycin derivatives has been developed further by the Beecham group.^{33,34}



More recently, Doyle has shown that even an acyclic amide such as (108) can cyclize smoothly to the β -lactam (109; equation 38).³⁵ β -Lactam formation in these cases is surprising, since the electron-withdrawing heteroatom should direct insertion away from the α -C—H bond. It may be that in this situation, overlap between the filled orbital on the nitrogen atom and the C—H orbital can increase the electron density in the latter, thus making it more reactive. It may be pertinent that as (108) becomes more product-like, amide resonance decreases and the electron-donating ability of the nitrogen atom could increase.



4.2.4.1.2 Furanone synthesis

The cyclization of (110) to (111) reported by Adams is a particularly striking example of C—H activation by an adjacent heteroatom (equation 39).³⁶ Although the ring size would be the same, and conformational differences in the transition states leading to the two products should be minimal, none of the alternative bicyclic ketone (112) was observed.



4.2.4.1.3 Pyrrole synthesis

Rhodium-catalyzed intramolecular insertion into aromatic C—H bonds proceeds smoothly.^{28,37} For instance, (113) cyclizes to a mixture of the heterocyclic product (114) and the carbacyclic product (115; equation 40).³⁷ From the product ratios observed, one could conclude that the substituent on the aromatic ring can accelerate or decelerate C—H insertion. As with the results cited above (Section 4.2.3.4.3), the differences in reactivity observed can be ascribed to greater or lesser electron density in the target C—H bond.



4.2.4.2 Photochemical Cyclization

4.2.4.2.1 *y*-Lactam synthesis

Photochemical C—H insertion, by abstraction-recombination, can also be used to construct heterocyclic rings. In the example illustrated, 1,5-hydrogen atom abstraction leads to the six-membered lactam (117; equation 41).³⁸



4.2.4.2.2 Macrolactam synthesis

In some cases, abstraction-recombination can lead to still larger rings. In the example illustrated, 1,8hydrogen atom extraction leads to the formation of a seven-membered ring product (119; equation 42).³⁹ The 1,7-diradical would be thermodynamically favored because of the radical-stabilizing ability of the two sulfur atoms. The reaction may not, however, proceed by direct 1,8-abstraction. A 1,5-abstraction followed by a subsequent 1,4-abstraction would give the same product and would be sterically more reasonable.

1057



4.2.4.2.3 Macrolide synthesis: remote functionalization

In sterically constrained cases, hydrogen atom abstraction over long distances has been unequivocally established. In the steroid functionalization work of Breslow, for instance, irradiation of (120) leads, by 1,13-abstraction, to clean formation of the 12-membered macrolide (121; equation 43).⁴⁰ The chapter on 'Remote Functionalization' (Volume 7, Chapter 1.3) contains a full discussion of this family of reactions.



4.2.5 APPLICATIONS TO NATURAL PRODUCT SYNTHESIS

4.2.5.1 Photochemical: (--)-Punctatin A

The elegant route to (-)-punctatin A (125) developed by Paquette nicely illustrates the synthetic utility of photochemically mediated intramolecular C—H insertion (Scheme 3).⁴¹ The excellent stereoselectivity observed in the four-membered ring forming process (123) \rightarrow (124) was, in fact, predicted. The intermediate 1,4-diradical could cyclize to the equatorially fused *trans*-cyclobutane, as illustrated. The alternative cyclization mode, leading to the *cis*-fused cyclobutane, has a bad steric interaction with the angular alkoxymethyl group, and so is less favored.

4.2.5.2 Alkylidene Carbene: Clovene

Dreiding has used the conjugated alkynone flash vacuum pyrolysis he developed (Section 4.2.3.3) as the key step in several elegant natural product syntheses. The considerations that lead to regio- and stereo-chemical control are nicely illustrated by his approach to clovene (130; Scheme 4).⁴² Thermolysis of the (symmetrical) alkynone (126) could lead to three different enones, two of which (127 and 128) are illustrated. In fact, only (127) is isolated, in 80% yield. The structure and stereochemistry of this product enone was confirmed by straightforward conversion to the tricyclic hydrocarbon clovene (130). The outcome of the thermolysis was anticipated, on the basis of previous observations that insertion is most



Scheme 3

facile when the intermediate alkylidene carbene can achieve coplanarity with the target C—H bond. In this system, such coplanarity is easily achieved only by insertion into the boat conformation of the more flexible of the two rings.



4.2.5.3 Rhodium-mediated C-H Insertion

4.2.5.3.1 Pentalenolactone E (Cane)

After the initial observation by the Beecham group, the first detailed exploration of rhodium-mediated intramolecular C—H insertion in natural product synthesis was by Cane (Scheme 5).^{25,43} Starting from the symmetrical bicyclooctanone (131), he elaborated the corresponding diazo ester (132). Rhodium-mediated methine insertion to give (133) then proceeded smoothly. The tricyclic lactone (133) so produced had already been converted, by Paquette, to pentalenolactone E (134). In the same article (not

illustrated here), Cane also reported that rhodium-mediated cyclization provides a viable route to substituted bicyclo[3.3.0]octanones.⁴³



4.2.5.3.2 (+)- α -Cuparenone

A powerful feature of intramolecular C—H insertion is the inherent ability to transform an acyclic ternary stereogenic center into a cyclic quaternary stereogenic center.¹⁸ Taber has demonstrated that the rhodium-mediated cyclization of (135) to (136) indeed proceeds with retention of absolute configuration (Scheme 6).⁴⁴ The absolute stereochemistry of (136) was confirmed by conversion to the sesquiterpene (+)- α -cuparenone (137).



4.2.5.3.3 Pentalenolactone E (Taber)

Intramolecular C—H insertion is, essentially, a method for specific remote functionalization of hydrocarbons. An important implication of this for synthetic strategy is that the C—H insertion process can dissolve symmetry, thus leading from a simple precursor to a much more complex product. Both the clovene synthesis, by Dreiding, and the pentalenolactone E synthesis, by Cane, take advantage of this idea. It is further illustrated by the pentalenolactone E synthesis reported by Taber (Scheme 7).⁴⁵ In the key step, β -keto ester (139), which has a single stereogenic center, is transformed into the tricyclic ketone (140), which has four stereogenic centers.



Scheme 7

4.2.5.3.4 (+)-Estrone methyl ether

The key step in the enantioselective synthesis of a polycyclic target is the establishment of the first cyclic stereogenic center. Further construction can then be directed by that initial center. This principle is nicely illustrated by a synthesis of (+)-estrone methyl ether (145), reported by Taber (Scheme 8).⁴⁶ The specifically designed naphthylbornyl ester (141) is used to direct C—H insertion selectively toward one of the two diastereotopic C—H bonds. The new ternary center so created then directs the formation of the adjacent quaternary center in the course of the alkylation. Finally, the chiral skew in the product cyclopentanone (144) directs the relative and absolute course of the intramolecular cycloaddition, to give the steroid carbon skeleton.



Scheme 8

4.2.6 DIRECTIONS FOR THE FUTURE

It is apparent that intermolecular C-C bond formation by C-H insertion can be effective for the preparative functionalization of hydrocarbons. A drawback to this approach is the substantial cost of the reagents, as compared to the relatively low value of the products. These factors are perhaps best balanced in the mercury-catalyzed photochemical process developed by Crabtree.⁴

Intramolecular C—H insertion, on the other hand, is already a practical alternative for the construction of cyclobutanols,⁴¹ β -lactams^{23,33-35} and of cyclopentane-containing targets. With regard to the latter, diazo transfer can be effected on a large scale with the inexpensive methanesulfonyl azide.⁴⁷ The rhodium carboxylate catalysts are effective at very low concentration (<1 mol %) and can easily be recovered from the reaction mixture, if desired.³¹

In the rhodium carboxylate catalyzed process, the transition state leading to C-H insertion is highly ordered.^{30,46} Rhodium-mediated C-H insertion has further been shown to proceed with retention of absolute configuration.⁴⁴ It may likely be possible, therefore, to design enantiomerically pure ligands for rhodium that would direct the absolute course of insertion into a target methylene.

4.2.7 REFERENCES

- 1. O. Farooq, M. Marcelli, G. K. S. Prakash and G. A. Olah, J. Am. Chem. Soc., 1988, 110, 864.
- 2. D. D. Tanner and P. M. Rahimi, J. Org. Chem., 1979, 44, 1674.
- 3. C. Filliatre, M. Baratchart, J.-J. Villenave and R. Jaouhari, Tetrahedron Lett., 1982, 23, 3487.

- S. H. Brown and R. H. Crabtree, *Tetrahedron Lett.*, 1987, 28, 5599.
 R. F. Renneke and C. L. Hill, J. Am. Chem. Soc., 1986, 108, 3528.
 A. Oku, T. Harada, K. Hattori, Y. Nozaki and Y. Yamaura, J. Org. Chem., 1988, 53, 3089.
- A. Demonceau, A. F. Noels, A. J. Hubert and P. Teyssié, J. Chem. Soc., Chem. Commun., 1981, 688.
 A. J. Hubert, A. F. Noels, A. J. Anciaux and P. Teyssié, Synthesis, 1976, 600.

- H. J. Callot and F. Metz, Tetrahedron Lett., 1982, 23, 4321.
 J. H. Richards and J. B. Hendrickson, in 'The Biosynthesis of Steroids, Terpenes, and Acetogenins', Benjamin, New York, 1964, p. 274.
- 11. R. L. Treanor and R. G. Weiss, J. Am. Chem. Soc., 1988, 110, 2170.
- P. J. Wagner, B. P. Giri, P. Pabon and S. B. Singh, J. Am. Chem. Soc., 1987, 109, 8104.
 A. Padwa, C. S. Chou, R. J. Rosenthal and L. W. Terry, J. Org. Chem., 1988, 53, 4193.
- 14. M. Karpf and A. S. Dreiding, Helv. Chim. Acta, 1979, 62, 852
- 15. J. Ackroyd, M. Karpf and A. S. Dreiding, Helv. Chim. Acta, 1985, 68, 338.
- J. C. Gilbert, D. H. Giamalva and U. Weerasooriya, J. Org. Chem., 1983, 48, 5251.
 J. Wolinsky, G. W. Clark and P. C. Thorstenson, J. Org. Chem., 1976, 41, 745.
 J. C. Gilbert, D. H. Giamalva and M. E. Baze, J. Org. Chem., 1985, 50, 2557.

- 19. M. Ochiai, M. Kunishima, Y. Nagao, K. Fuji, M. Shiro and E. Fujita, J. Am. Chem. Soc., 1986, 108, 8281.
- 20. T. Cohen, R. H. Ritter and D. Ouellette, J. Am. Chem. Soc., 1982, 104, 7142.
- 21. S. D. Burke and P. A. Grieco, Org. React. (N.Y.), 1979, 26, 361.
- 22. E. Wenkert, L. L. Davis, B. L. Mylari, M. F. Solomon, R. R. da Silva, S. Shulman and R. J. Warnet, J. Org. Chem., 1982, 47, 3242.
- 23. R. Southgate and R. J. Ponsford, J. Chem. Soc., Chem. Commun., 1979, 846.
- 24. D. F. Taber and E. H. Petty, J. Org. Chem., 1982, 47, 4808.
- 25. D. F. Taber, S. A. Saleh and R. W. Korsmeyer, J. Org. Chem., 1980, 45, 4699.
- 26. H. J. Monteiro, Tetrahedron Lett., 1987, 28, 3459.
- 27. B. Corbel, D. Hernot, J.-P. Haelters and G. Sturtz, Tetrahedron Lett., 1987, 28, 6605.
- 28. D. F. Taber and R. E. Ruckle, Jr., J. Am. Chem. Soc., 1986, 108, 7686.
- G. Stork and K. Nakatani, *Tetrahedron Lett.*, 1988, 29, 2283.
 D. F. Taber and K. Raman, J. Am. Chem. Soc., 1983, 105, 5935.
- M. J. Hennessy, Ph.D. Thesis, University of Delaware, 1989.
 N. Ikota, N. Takamura, S. D. Young and B. Ganem, *Tetrahedron Lett.*, 1981, 22, 4163.
- 33. T. C. Smale, Tetrahedron Lett., 1984, 25, 2913.
- 34. P. Brown and R. Southgate, Tetrahedron Lett., 1986, 27, 247.
- 35. M. P. Doyle, M. S. Shanklin, S.-M. Oon, H. Q. Pho, F. R. van der Heide and W. R. Veal, J. Org. Chem., 1988, 53, 3386.
- J. Adams and R. Frenette, *Tetrahedron Lett.*, 1987, 28, 4773.
 C. W. Jefford and A. Zaslona, *Tetrahedron Lett.*, 1985, 26, 6035.
- 38. H. Aoyama, Y. Arata and Y. Omote, J. Org. Chem., 1987, 52, 4639.
- 39. S. V. Kessar, T. Singh and R. Vohra, Tetrahedron Lett., 1987, 28, 5323.
- 40. R. Breslow, S. Baldwin, T. Flechtner, P. Kalicky, S. Liu and W. Washburn, J. Am. Chem. Soc., 1973, 95, 3251.
- T. Sugimura and L. A. Paquette, J. Am. Chem. Soc., 1987, 109, 3017. 41.
- 42. J. Ackroyd, M. Karpf and A. S. Dreiding, Helv. Chim. Acta, 1984, 67, 1963.
- D. E. Cane and P. J. Thomas, J. Am. Chem. Soc., 1984, 106, 5295.
 D. F. Taber, E. H. Petty and K. Raman, J. Am. Chem. Soc., 1985, 107, 196.
 D. F. Taber and J. L. Schuchardt, J. Am. Chem. Soc., 1985, 107, 5289.
- 46. D. F. Taber, K. Raman and M. D. Gaul, J. Org. Chem., 1987, 52, 28.
- 47. D. F. Taber, R. E. Ruckle, Jr. and M. J. Hennessy, J. Org. Chem., 1986, 51, 4077.

Author Index

This Author Index comprises an alphabetical listing of the names of over 7000 authors cited in the references listed in the bibliographies which appear at the end of each chapter in this volume.

Each entry consists of the author's name, followed by a list of numbers, each of which is associated with a superscript number. For example

Abbott, D. E., 6^{12,12c}, 10⁴⁰, 573^{53,54}

The numbers indicate the text pages on which references by the author in question are cited; the superscript numbers refer to the reference number in the chapter bibliography. Citations occurring in the text, tables and chemical schemes and equations have all been included.

Although much effort has gone into eliminating inaccuracies resulting from the use of different combinations of initials by the same author, the use by some journals of only one initial, and different spellings of the same name as a result of transliteration processes, the accuracy of some entries may have been affected by these factors.

Abad, A., 85164 Abatjoglou, A. G., 124273,288, 128273,288,323, 132273,288, 134^{273,323} Abdallah, M. A., 125304, 126304 Abdallah, Y. M., 75187 Abdel-Wahab, A. M., 325160 Abdurasuleva, A. R., 30358, 321138 Abe, A., 530⁷⁸, 535⁷⁸ Abe, H., 226^{193,196} Abe, T., 61715, 61915, 62115, 62315, 62715 Abe, Y., 289⁶⁸, 446⁷⁷, 461¹⁴⁶, 541¹⁰⁷ Abecassis, J., 173⁵¹⁸ Abel, E. W., 208^{1,11}, 210^{11,11a}, 219^{11a}, 228²¹⁴, 234^{11a}, 4369,13, 52433 Abenhaim, D., 770172 Abicht, H.-P., 229224 Abiko, S., 274²¹ Abiko, T., 956¹⁰⁷ Aboujaoude, E. E., 20183 Abraham, N. A., 57261 Abramovitch, R. A., 68199, 83912, 84012, 85412 Abrams, G. D., 36289 Abramson, N. L., 38011 Abreo, M. A., 224179 Acheson, R. M., 63055 Achi, S., 47²⁵⁶ Achiwa, K., 99¹⁸⁴, 288⁶⁴ Acker, R.-D., 223159, 262162, 263162 Ackermann, M. N., 38010 Ackroyd, J., 104915, 105842 Adachi, S., 147387, 149387, 150387, 151387 Adam, G., 224¹⁷³ Adam, W., 587¹⁴⁸ Adames, G., 770180 Adamov, A. A., 640101 Adams, C., 642111 Adams, C. T., 723⁹, 731⁹ Adams, E. W., 5648, 7213 Adams, J., 28966,67, 105636 Adams, K. G., 224163 Adams, K. H., 7215 Adams, R., 296¹³, 564⁸, 721³, 825^{27a} Adams, R. D., 101279

Adams, W. J., 846⁴⁷ Addess, K. J., 224¹⁶⁵ Ade, E., 45251 Adkins, H., 82315 Adlington, R. M., 25179, 25479 Aebersold, D. R., 29933 Aebi, J. D., 40²²¹, 43²⁴⁰, 44²⁴⁰ Aeschimann, R., 41²²⁷ Afghahi, F., 91825 Agami, C., 1366, 2166,130, 22130 Agawa, T., 672⁶⁵ Agdeppa, D. A., Jr., 74770 Ager, D. J., 123248,252, 125248, 135252,350,351,352,354,355 136350,351,352, 137350,351,352,355, 140352, 141350,351,355, 778^{4} Agosta, W. C., 19103, 901115,116 Ahlbrecht, H., 656 Ahlers, H., 105²¹⁴ Ahmad, M. S., 762145 Ahmed, F. R., 103683 Ahmed, R., 692138, 693142 Ahn, K., 505¹⁶⁶ Ahn, S. H., 49²⁶¹ Ahrens, F., 582111 Aig, E. R., 28969, 30148 Aigami, K., 38345 Ainsworth, C., 62642 Akabane, Y., 695¹⁵³ Akabori, S., 826³⁹ Akawie, R. I., 754¹⁰⁷, 757¹⁰⁷ Akgun, E., 266¹⁹⁴ Akhmedov, K. N., 30358 Akhrem, A. A., 7347 Akhrem, I., 29725, 33425 Akhtar, M. H., 92754 Akiba, M., 135344, 136344, 137344, 138344 Akimoto, I., 262¹⁵⁸ Akimoto, K., 594186 Akiyama, F., 583120 Akiyama, S., 585137 Akiyama, T., 652221 Akiyoshi, K., 231243 Akpuaka, M. U., 81043

Akutagawa, K., 7135, 19736 Akuzawa, K., 470²¹⁵, 473²¹⁵, 476²¹⁵ Alami, M., 57893 Al-Aseer, M., 6717 Alazard, J.-P., 681100 Albaugh-Robertson, P., 44360 Al'bertinskii, G. L., 63544 Alberts, A. H., 124281, 125281 Albright, J. D., 48258, 19738, 19838 Alejski, K., 619²³ Alexakis, A., 209²¹, 216⁷³, 217^{21,84}, 219^{21b}, 223¹⁵⁴, 224^{166,167}, 225^{167b}, 226^{203,207}, 227²¹⁰, 245³⁰, 249⁵⁸⁻⁶¹, 258125, 259129, 263173, 265188, 42372, 466182,193 470²¹⁴, 473²¹⁴, 476²¹⁴, 485³²⁻³⁵, 486³²⁻³⁵, 494^{34,35,89}, 516⁸⁹, 579¹⁰⁰ Alexander, E. R., 72520 Alexandrou, N., 3809 Alfs, H., 30787 Al-Hassan, M. I., 232260, 25498, 25798, 49590.92, 50390 Alimardanov, R. S., 306⁸² Alink, R. J. H., 904¹²⁹ Allakhverdiev, M. A., 30469 Allan, A. R., 38348 Allen, C. F. R. H., 7214 Allen, D. S., Jr., 1152, 1752 Allen, G. F., 103891 Allen, L. E., 1040107 Allen, M. J., 56410, 56632,33, 595188, 63417 Allinger, N. L., 38236, 85479, 905140 Allport, D. C., 66434 Alper, H., 539¹⁰², 559^{55,56}, 582¹¹⁶, 770¹⁸¹, 1020¹², 1024²⁹, 1028⁵⁰, 1030⁶⁰, 1032⁶⁹, 1036^{83,84}, 1037⁸⁶, 1**039⁹⁹** Altman, L. J., 99186, 107186 Alumbaugh, R. L., 7343 Alvarez, E., 448⁹⁵ Alvernhe, G., 38128, 38228 Amamria, A., 495⁹³⁶ Amano, E., 844^{32,33} Amano, T., 244²⁵, 267²⁵, 494⁸⁴ Ameer, F., 42152 Amelotti, C. W., 57266 Amende, J., 88813 Ames, D. E., 28143, 505169, 53074,75, 53187, 53474,75, 538⁸⁷ Ammar, F., 57477 Amouroux, R., 25285 Amri, H., 24634 Amstutz, E. D., 56412 Amupitan, J. A., 34829, 35553, 35753 Ananthanarayan, T. P., 81680 Anantha Reday, P., 462149 Anciaux, A. J., 8792, 10478, 10518 Andersen, N. H., 132334, 133334, 136334, 35462, 35656 Anderskewitz, R., 30353 Anderson, B. C., 3805 Anderson, D. F., 74981 Anderson, D. M. W., 82640 Anderson, K. W., 73627 Anderson, P. C., 227²⁰⁸, 421⁶¹, 422⁶¹, 589^{161,162}, 610161,162 Anderson, P. H., 92748 Anderson, P. S., 38011 Anderson, R. A., 663²⁶, 665²⁶ Anderson, R. J., 220¹²¹, 224¹⁶³, 226¹⁹², 257¹²⁰, 265¹⁸⁷, 365⁹⁴, 419⁴³

Ando, K., 137376 Ando, M., 1041113 Ando, T., 21898 Ando, W., 887⁶, 889⁶, 893⁶, 894⁶, 896⁶, 897⁶, 900⁶, 903^{6,123}, 919^{28,32}, 923^{43,44}, 934^{44,63}, 954⁴⁴, 1008^{70,71} Andrade, J. G., 499¹²⁴, 507¹²⁴, 670⁵⁸, 673⁵⁸, 693¹⁴¹ Andreetti, G. D., 38657 Andreev, V. M., 30572, 644140.141 Andreini, B. P., 52538,40 Andrejevic, V., 38013 Andreocci, A., 80411 Andrews, G. C., 86¹⁶, 95¹⁶, 96¹⁶, 97¹⁶, 98¹⁶, 99¹⁶. 10416, 11616, 11716.235.236, 11816, 15516.233.236, 156^{235,236}, 168¹⁶, 196³⁰, 358⁶⁷, 919³¹, 934³¹, 939⁷⁸, 94687 Andrews, R. C., 100²⁰², 168²⁰², 170²⁰², 172²⁰² Andrews, S. W., 75293 Andrieux, C. P., 57477 Andrus, A., 1025³⁶ Anet, F. A. L., 164475, 38236 Angeli, A., 890³⁵ Angelini, G., 30040 Angély, L., 669⁵⁶ Angoh, A. G., 219110 Anikina, E. V., 303⁵⁷ Anklekar, T. V., 7543, 7643, 7757. 8043 An-naka, M., 461148, 541113, 543113 Anner, G., 81679 Anschütz, R., 82845 Anschütz, W., 909152 Ansell, M. F., 326^{164,165} Anson, F. C., 21351 Antonioletti, R., 512²¹³, 515²¹³ Aoki, O., 567¹⁹⁰, 595¹⁹⁰, 607¹⁹⁰ Aoki, S., 169⁵⁰⁹, 221^{131,132}, 455¹²⁵, 460¹²⁵ Aoyagi, T., 623³⁹ Aoyama, H., 105738 Aoyama, T., 900⁹⁰ Appell, H. R., 30991 Appleton, R. A., 71537, 73519 ApSimon, J. W., 35²⁰⁶, 289⁶⁷ Arad, D., 1944 Arai, M., 221¹³² Arakelova, L. V., 30468 Araki, S., 220¹¹⁶ Araki, Y., 32¹⁸⁷ Araratyan, E. A., 318123 Arase, A., 470196,197, 473196,197, 52221 Arata, K., 300^{42,43} Arata, Y., 105738 Aratani, T., 21672 Arbic, W., 63430, 64430b Arcadi, A., 53996 Arco, M. J., 261¹⁴⁸, 264¹⁴⁸, 957¹¹⁰, 958¹¹³ Arcoleo, J. P., 759¹³² Arens, J. F., 106222, 113222, 121245, 123245,246, 125246, 257115, 262160, 263160 Aresta, M., 461 145 Argyle, J. C., 804⁵ Argyropoulos, J. N., 82420 Ariaans, G. J. A., 855⁸³ Arigoni, D., 341², 352⁴⁵, 360² Arimura, T., 329¹⁸⁴ Arkell, A., 890³², 894⁶¹ Arkley, V., 80725 Armitage, J. B., 554^{18,19}

Armstrong, J. C., 53790, 53890 Armstrong, L. J., 89032 Armstrong, R. W., 231253 Arn, H., 223155 Amaboldi, A. G., 87155 Arnaud, P., 57266 Arndt, F., 563117, 582117, 88813, 89137 Arndt, H. C., 8661, 8861, 8961, 9161, 12461, 91823 Arne, K. H., 19962 Arnett, J. F., 57054 Arnó, M., 85164 Arnold, L. D., 227²¹³ Arnold, R. T., 324152, 80720 Arnold, Z., 89030, 89730 Arnost, M. J., 86526 Arnould, D., 168507, 169507, 171507 Arora, G. S., 405138 Arora, P. C., 649209 Arpe, H. J., 103998 Arrieta, J. M., 62541 Arseniyadis, S., 39²¹⁸, 48²¹⁸ Arvia, A. J., 63658 Arzoumanian, H., 19957 Asada, T., 35554, 35754 Asahina, Y., 81047 Asai, M., 56479, 57479 Asano, O., 261157 Asano, T., 89145 Asaoka, M., 20116 Asaoka, T., 2958 Ash, L., 99186, 107186 Ashbrook, C. W., 93465, 95365 Ashby, E. C., 21341,42b, 21455,58, 82419,20,23 Ashcraft, A. C., 23139 Ashe, A. J., III, 406145, 905138 Ashkenazi, P., 62845 Asker, W., 563117, 582117 Askerov, A. K., 30460 Askin, D., 45²⁴⁷ Aslam, M., 87892-95, 87992-94,96, 88094,98, 88194 Aso, M., 80941 Aston, J. G., 84218,20 Asveld, E. W. H., 86528 Atkins, M. P., 29615 Atkinson, J. G., 28967 Atlani, P. M., 96164, 98164, 99164 Attenberger, P., 53186 Au, M.-K., 87789 Aubert, C. C., 699161 Aubert, F., 41617, 41717 Auderset, P. C., 583118 Audia, V. H., 226197 Audouin, M., 73836 Augood, D. R., 505¹⁶², 507¹⁶², 512¹⁶² Augustine, R. L., 1¹, 3¹, 8¹, 11¹, 13¹, 17¹, 18¹, 21¹, 23¹, 25¹, 59¹ Aumiller, J. C., 414⁵, 422⁵ Austin, W. B., 53069, 53369, 53769 Avrutskaya, I. A., 63980 Axelrod, E. H., 20174, 597197 Aycard, J.-P., 89247 Azuara, S. J., 781¹² Azuma, H., 1041111 Azuma, K., 98422, 98522, 98731 Azuma, Y., 232267, 510185 Azumi, N., 30042

Baake, H., 87259 Baardman, F., 30249 Baba, H., 95156, 97156 Baba, S., 230239, 232239a, 238239a, 266195, 27424, 48640,41, 49540,91, 49740, 49840, 50340, 53054 Babiak, K. A., 20917 Babin, E. P., 30469 Babu, G. P., 331195 Babudri, F., 461145 Bacchetti, T., 87155 Bach, R. D., 6615, 7415, 74879, 74980,81 Bachmann, W. E., 505¹⁶¹, 507¹⁶¹, 564¹², 579⁹⁶, 724¹³, 8872, 8882, 8912, 8972 Back, T. G., 8653, 9453, 11453, 11753 Bäcklund, B., 861¹ Bäckvall, J.-E., 222139 Bacon, R. G. R., 499110.112, 501110, 505110, 509110. 512110, 66431 Bacquet, C., 63973, 65173b Badaoui, E., 8772,73, 9572 Bader, A. R., 82851 Badet, B., 159456,461, 161456,461, 162456, 165461 Badger, R. A., 20107 Baer, D. R., 78111 Baer, T. A., 41614, 41714 Baettig, K., 38010 Baeyer, A., 82842 Baggaley, K. H., 38346 Bagheri, V., 25178, 25478, 463156, 92034, 92534a, 93434 Bagotzky, V. S., 647177, 648177, 649177 Bahrmann, H., 10176 Bai, D. L., 225187 Bailey, E. J., 80413, 81679 Bailey, T. R., 6925, 7025, 7225, 7425, 7755, 232267, 514209 Bailey, W. F., 419^{29,30}, 422^{29,30} Baillargeon, V. P., 102115 Baines, D., 38668, 399118 Bainton, H. P., 24748 Baird, G. J., 47257 Baird, M. S., 38348 Baizer, M. M., 21456, 5649, 5679, 63413.20, 63820, 64920, 655²⁰ Bajgrowicz, J. A., 46254, 21566, 25175 Baker, B. W., 643117, 644150 Baker, R., 41622, 42376 Baker, S. R., 232264 Baker, W., 382³⁶ Baklan, V. F., 38347 Bakshi, R. K., 79792,93 Bakuzis, P., 85791 Balaban, A. T., 331196 Balanson, R. D., 135340, 137340, 139340, 141340 Balasubramanian, K. K., 1788 Balasubramanian, R., 3807 Baldassarre, A., 25179, 25479 Baldwin, J. E., 1898, 25281.82, 25781.82, 38234, 640108 647¹⁰⁸, 711²³, 871⁵², 916^{17,18}, 932¹⁷, 934⁶⁵, 939⁷⁷, 94688, 95365, 963119, 99034 Baldwin, S., 105840 Balf, R. J., 396103 Ballard, D. A., 82527b Ballenegger, M., 82212, 83112, 83512b Balogh, A., 334222 Balogh, D. W., 38013, 62332, 626326 Balogh, M., 125303

Balogh, V., 664²⁸, 698²⁸ Bamberger, E., 806¹⁶, 812⁵³ Bamfield, P., 501 138 Ban, Y., 103267, 103682, 103790, 103890,95,95b Bandodakar, B. S., 3804 Bandy, J. A., 47257 Banerjee, A. K., 594187 Bank, K. C., 134336 Banks, C. M., 39078,79, 39278 Banks, H., 767164 Banks, R., 38236 Banks, R. B., 43938, 4823 Bannai, K., 222144 Bansal, R. C., 322141 Banville, J., 32188 Barabadze, Sh. Sh., 318¹²⁸ Barabas, A., 416¹⁵, 417¹⁵ Baran, J., 582115 Baratchart, M., 10463 Barbeaux, P., 107²²⁸, 111^{228,230,231,232} Barborak, J. C., 85582 Barchi, J. J., 785^{36,36b} Barclay, L. R. C., 326166 Barco, A., 73837 Bard, A. J., 63422 Barden, T. C., 71123 Bardili, B., 75292 Bare, T. M., 16⁸¹ Baretta, A., 839¹², 840¹², 853⁷⁰, 854¹² Bargar, T. M., 49273, 49773 Barili, P. L., 72519, 7349, 74358 Barker, W. D., 903126 Barkley, R. M., 422 Barlow, L., 585137, 587148 Barluenga, J., 28248, 78850 Barnes, D. G., 331200a Barnetzky, E., 643124 Barnier, J.-P., 72732 Barr, D., 763¹⁵¹ Barr, P. J., 219109, 53072, 53472, 547124 Barrett, A. G. M., 25179, 25479, 6132, 6152 Barrett, G. C., 8621,22, 10421, 12121,22 Bartetzko, R., 6146, 6236, 6276 Bartlett, P. A., 3415, 3655, 36699, 3745 Bartlett, P. D., 7054, 72518 Bartlett, W. R., 36289, 369117, 371114, 372117 Bartok, M., 72623 Bartoletti, I., 102847 Bartolotti, L. J., 66¹⁶, 67¹⁶, 74^{16,42}, 75⁴², 77⁴² Barton, D. H. R., 8621.22.32,33.37.38,48,62, 10421, 12121,22, $\begin{array}{c} 147^{32,33}, 154^{32}, 158^{37,38}, 159^{37}, 163^{38}, 164^{37}, \\ 173^{37,38,62}, 386^{66}, 419^{35}, 422^{68}, 505^{158,159}, 613^2, \\ 615^2_{-1}, 660^8, 664^{30}, 681^{97}, 689^{123}, 733^2, 780^{10}, \end{array}$ 83163 Barton, J. C., 223¹⁶⁰ Bartz, W., 909152 Barua, N. C., 380¹⁰ Basavaiah, D., 795^{83,84} Bashe, R. W., 248⁵⁵, 251⁵⁵, 269⁵⁵, 419⁴⁷, 494⁸⁷, 502⁸⁷ Bassindale, A. R., 41931 Batroff, V., 62335 Bats, J. W., 56285 Batta, A. K., 644149 Battersby, A. R., 659¹, 660¹, 679¹, 681⁹⁴, 807³¹ Bättig, K., 1789 Battiste, M., 89667

Battistini, C., 7348,10 Batu, G., 219112 Baudisch, O., 89035 Baudouy, R., 256113 Baudry, D., 84216 Bauer, K., 82844 Bauer, P., 72310 Bauld, N. L., 42379, 649202, 650202c, 652202c Baume, E., 505¹⁶⁶ Baumgarten, R. J., 56737, 60637, 85791 Baumgartner, M. T., 505156 Baumstark, A. L., 588157,158 Baus, U., 39²¹³ Bayer, J., 835⁸⁰ Bayerlein, F., 81470 Bayet, P., 8766,116 Baylis, E. K., 329189 Baynham, M. K., 38013 Bayomi, M., 904134 Baze, M. E., 105018, 106018 Bazhulina, V. I., 644140,141 Beak, P., 65^{1,2}, 66¹², 67^{17,20}, 68^{2,21,22}, 69^{2,22}, 71², 74^{12,41}, 88130,131, 90130,131, 105131, 19410,11, 580108 Beal, P. F., 888¹⁷, 891³⁹, 892³⁹, 898³⁹ Beale, M. H., 222140 Beard, R. D., 159467, 166467 Beard, R. M., 87786 Beard, W. Q., 91412 Beasley, G. H., 43195,96 Beau, J.-M., 174525, 19626, 231253 Beaucourt, J.-P., 41617, 41717, 462152, 513207 Beaulieu, P. L., 21344 Beaumont, A. G., 56415 Bebernitz, G. E., 25152 Bebikh, G. F., 864²¹ Bechara, E. J. H., 588157 Beck, A. K., 87^{68,69}, 89⁶⁸, 128³²², 130³²², 134^{322,338}, 135³³⁸, 137³³⁸, 144³⁸⁴, 145³⁸⁴, 147³⁸⁴ Beck, F., 634^{25b,28}, 635^{25b,43}, 639^{25b}, 640⁹⁷, 655⁴³, 65643 Becker, E. I., 262159, 264159 Becker, H., 691135, 82954 Becker, K. B., 497102, 87365, 87465 Becker, M., 348³¹, 358³¹ Becker, P. D., 88¹³¹, 90¹³¹, 105¹³¹ Beckert, W. F., 319130 Beckett, B. A., 20105 Beckhaus, H.-D., 38236 Becking, L., 647^{197,198}, 649²⁰⁰ Beckmann, W., 62847 Beckwith, A. L. J., 107²²⁷, 600²¹² Beddoes, R. L., 810⁴³ Beeck, O., 332²⁰⁷ Beedle, E. C., 79689 Beereboom, J. J., 70711 Beers, S., 19733 Beetz, I., 102848 Begley, M. J., 407147, 586153, 604153, 610153 Begley, W. J., 623³², 677⁸⁴ Bégué, J. P., 842¹⁶ Behling, J. R., 20917, 21668, 224168 Behrens, C. H., 225185, 264181 Behrooz, M., 164478, 167478, 168478 Beissner, G., 482³ Bélanger, G., 635³⁸ Belen'kii, L. I., 6157

Beletskaya, I. P., 219¹¹¹, 436¹⁴, 503¹⁵², 524²⁹, 529^{29,48}. 530⁷³, 531⁸⁸, 534⁷³, 537⁸⁸ Belik, M. Yu., 30677 Bell, A., 20176, 7214 Bell, A. S., 510¹⁸⁴ Bell, F., 826⁴⁰ Bell, H. C., 505¹⁶⁰ Bell, K. H., 806¹⁷ Bell, T. W., 219106, 224165, 512201 Bellesia, F., 39598 Belli, A., 78956 Bellido, I. S., 396¹¹⁵ Bellucci, G., 733¹, 734⁹ Belluš, D., 84850 Belotti, D., 599^{209,210}, 602²²³, 603²²⁸ Ben, I., 591¹⁷¹ Benaim, J., 23¹⁴⁰, 30¹⁴⁰, 33¹⁴⁰, 34¹⁴⁰, 35¹⁴⁰ Bencini, E., 103372 Bencze, W. L., 56633 Benecke, H. P., 92646, 92846 Benedetti, F., 174534,535, 176534,535, 178534,535 Benetti, S., 73837 Benjamin, B. M., 328¹⁷⁹, 726²², 782²⁰ Bennett, D. A., 103²⁰⁵, 108²⁰⁵ Bennett, D. W., 3807 Bennett, S. M., 219110 Benny, D., 61927 Bennyarto, F., 909152 Bentley, K. W., 83681 Bentley, T. W., 79999 Benzinger, E., 6606 Beppu, K., 124²⁸⁰, 125²⁸⁰, 126²⁸⁰ Berenblit, V. V., 63982, 64482,137 Berenbom, M., 88817 Berezin, I. V., 499125 Bereznitskii, G. K., 30570 Berg, H., 482³ Bergbreiter, D. E., 30¹⁸¹, 31¹⁸⁵, 32¹⁸⁵, 34^{192,193}, 39¹⁹², 418²⁶ Bergius, M., 8225, 8345 Bergman, J., 511¹⁹¹ Bergman, R., 767165 Bergmann, E., 72414 Bergmann, E. D., 124²⁶⁴, 867³⁵ Bergmann, W., 89251 Bergmark, W., 580¹⁰⁷ Bergstrom, D. E., 459¹³⁵, 462^{149,151}, 470²¹⁸, 476²¹⁸ 513²⁰⁷ Berk, H. C., 62333 Berk, S. C., 20918 Berlan, J., 155430 Bernard, D., 482⁵, 499⁵, 505⁵, 509⁵ Bernard, G., 8772, 9572 Bernardini, A., 25175 Bernardon, J.-M., 262¹⁶⁷ Berner, D., 373129 Bernhard, W., 1792, 1892 Bernhardt, J. C., 470²⁰⁴, 473²⁰⁴, 484²⁴, 501²⁴ Bernstein, M. A., 227²⁰⁸ Bernstein, P. R., 1786 Berryhill, S. R., 103681 Berson, J. A., 7067 Bert, L., 2422 Bertelo, C. A., 102742, 103042 Bertholon, G., 30683 Berti, G., 72519, 7331, 74151, 74358, 74565

Bertolo, P., 80411 Bertrand, J. L., 88127, 89127, 91127,147, 92127, 93127, 109^{127,147}, 116¹²⁷ Bertrand, M., 599207 Bertucci, C., 48312 Bertz, S. H., 20912, 21024, 21127, 21343.46, 21527b, 21646, 22346 Bérubé, D., 649190 Bérubé, G., 38010 Bessiere, J.-M., 74461 Bessiere, Y., 21565 Bessiere-Chretien, Y., 771^{183,189} Bestmann, H. J., 20073, 28659, 644159, 87788, 8874, 8884, 893^{4,56}, 897^{4,76}, 898⁷⁶, 900⁴, 903⁴ Beswick, P. J., 49488 Bethell, D., 770178 Betschart, C., 580104 Bettolo, R. M., 71744 Betz, R., 25393 Beugelmans, R., 505¹⁵⁷ Bewick, A., 568^{44,45}, 636⁵⁹ Bey, P., 36563 Bezzubov, A. A., 648181 Bhanu, S., 28141 Bhatnagar, N. Y., 505159 Bhattacharjee, G., 908146 Bhattacharya, K. K., 30464 Bhushan, V., 72834 Bibby, C., 770¹⁸⁰ Bïchi, G., 936⁷² Biddlecom, W. G., 953103 Bieber, J. B., 975³, 980³ Biellmann, J.-F., 86¹⁴, 94¹⁴, 95^{14,151,155}, 96^{151,161,164,165}, 98^{151,161,164}, 99^{151,164,165,179}, 100^{14,179}, 101¹⁶⁵, 107¹⁶⁵, 149404,406, 151406, 152404,406, 153406, 19629 Bien, S., 89670 Bienz, S., 28865 Bierenbaum, R. E., 592172 Biftu, T., 67675 Bigalke, J., 277²⁹ Bigelli, C., 55427 Bigelow, W. B., 56519 Bigi, F., 311¹⁰⁰ Bihovsky, R., 216⁷⁸ Billington, D. C., 423⁷³, 424⁷³, 426⁷³, 428⁷³, 1025^{33,33a} Bilow, N., 53069, 53369, 53769 Biltz, H., 822⁵, 826³⁶, 834⁵ Binder, J., 19954 Bineeva, N. G., 30468 Binkley, E. S., 836 Binns, M. R., 8794 Binns, T. D., 63894 Binsch, G., 89671 Biran, C., 57790 Birch, A. J., 17⁹¹, 831⁶³ Birch, A. M., 38452 Birch, S. F., 331198 Bird, C. W., 744⁶², 807¹⁹, 1015², 1027² Bird, T. G. C., 159457, 173457 Birkhahn, M., 851⁶⁷ Bischoff, C., 84852 Bishop, P. M., 371115 Bishop, R., 38346 Biswas, S., 783²⁸ Bitenc, M., 652²²³, 653²²³ Bitler, S. P., 52846

Bjoerkling, F., 48961, 49561, 50461, 51161, 51561 Bjorklund, C., 499120 Black, D. St. C., 57583, 82634 Black, H. K., 55312 Black, T. H., 592172 Blackburn, G. M., 20285, 91930, 93330,62, 946³⁰ Blackett, B. N., 74257, 74357,60 Blade, R. J., 545¹²² Blagg, J., 19737 Blair, P. A., 35²⁰⁴, 66¹⁴ Blanc-Guenée, J., 147388, 151388 Blanchard, L., 62951 Blanco, L., 53891 Blankenship, R. M., 62332 Blankespoor, C. L., 325¹⁶¹ Blankley, C. J., 1783 Blanton, C. D., Jr., 62953,54 Blaschke, G., 667⁴⁶, 687^{112,113} Blasioli, C., 159⁴⁶³, 161⁴⁶³, 165⁴⁶³ Blaskó, G., 682¹⁶⁵ Blaszczak, L. C., 218⁹⁷, 762¹⁴⁵ Blatcher, P., 123²⁵¹, 124²⁶¹, 126²⁶¹, 946⁸⁷ Blatchford, T. P., 583¹¹⁹ Blatt, H., 57583 Blazejewski, J.-C., 66430 Blezard, M., 74877 Blicke, F. F., 78112 Bloch, R., 173518 Block, E., 86⁸, 121⁸, 147⁸, 154⁸, 173⁸, 878⁹²⁻⁹⁵, 879^{92-94,96,97}, 880^{94,98}, 881⁹⁴ Blomquist, A. T., 905139 Bloodworth, A. J., 3809 Bloom, S. H., 21563, 25180, 25480 Bloomfield, J. J., 597²⁰⁰, 613¹, 614^{1.5}, 615^{1,10}, 616¹, 617^{5a,14}, 619¹, 620^{1,14}, 621¹, 622¹, 623¹, 625¹, 626^{1,42}, 627¹, 628¹, 629¹, 630¹ Blough, B. E., 226199 Blount, J. F., 39085, 39285, 407148, 62332, 62632b, 67572, 87257, 87357 Blues, E. T., 4156 Blum, L., 505¹⁶³ Blum, R. B., 20120, 1946 Blumenkopf, T. A., 86734 Blunt, J. W., 741^{50,52} Boar, R. B., 613², 615² Boardman, L. D., 25178.100, 25478.100 Boaz, N. W., 21237, 21350, 21781, 23581b Bobbitt, J. M., 665⁴⁰, 666⁴⁴, 681⁹⁵, 687^{44,112} Bocelli, G., 38657 Boche, G., 27729 Bock, G., 903127 Bock, M. G., 125292, 126292 Bodot, H., 89247 Boeckman, R. K., Jr., 8⁴⁵, 15⁴⁵, 16⁴⁵, 48²⁶⁰, 252^{82,85}. 25782 Boeckmann, J., 24745, 24845 Boeckmann, R. K., Jr., 602220 Boekelheide, V., 124²⁶⁷, 125²⁶⁷, 126²⁶⁷, 127²⁶⁷, 877⁸³⁻⁸⁵, 927⁴⁸ Boelema, E., 83268a Boelens, H., 25179, 25479 Boelhouwer, C., 260145 Boes, M., 7549, 7860-62, 7960.61, 8160.61 Boese, R., 53790, 53890 Boexkes, W., 30787

Bogdanowicz, M. J., 8661, 8861, 8961, 9161, 12461, 178^{543,544}, 179⁵⁴³, 181⁵⁴³, 761¹⁴⁴, 762¹⁴⁴, 785³², 1040106 Boger, D. L., 35659 Bogert, M. T., 89879 Bogoslovskii, K. G., 648^{175,183} Bohlmann, F., 277²⁸ Böhme, H., 154421 Böhrer, G., 587143 Boireau, G., 770172 Bois-Choussy, M., 505157 Bokel, H. H., 124²⁶⁸, 125²⁶⁸, 126²⁶⁸, 127²⁶⁸, 131²⁶⁸ Böker, R., 80942 Boland, W., 220120 Boldeskul, I. E., 8643, 17943 Bollyky, L., 154417 Bolourtchian, M., 57790 Bomse, D. S., 587^{141,142} Bonakdar, A., 44045 Bond, F. T., 251⁷⁹, 254⁷⁹ Bonde, S. E., 494⁸⁵ Bonhoeffer, K. F., 822², 831² Bonnet-Delpon, D., 324155 Bonser, S. M., 87678 Boop, J. L., 39292 Boots, S. G., 369¹¹⁹, 370¹¹³, 372¹¹⁹ Borch, R. F., 905136 Borcherdt, G. T., 42889 Borden, W. T., 38010 Bordic, S., 374132 Bordner, J., 1477, 1577, 36384 Bordwell, F. G., 88¹²¹, 756¹¹⁵, 862⁵, 866^{5,32} Borel, C., 103787 Bormann, D., 66427 Bornstein, J., 760139, 774139 Borowitz, I. J., 842 Borromeo, P. S., 24643, 25843 Bory, S., 147^{390,398}, 149^{390,398,408,409,411}, 150⁴¹¹, 151^{390,408,409,411}, 152³⁹⁰, 155^{408,409} 152390, 155408,409 Bos, H. J. T., 219104 Bosch, G. K., 591 166, 168, 610 166, 168 Boskin, M. J., 760137 Bösler, M., 825³¹ Bosshard, H., 81574 Bos Vanderzalm, C. H., 57583 Bott, R. W., 56415 Bottari, F., 74151 Botteghi, C., 124286, 125286, 127286, 228222, 48745 Botteron, D. G., 564⁸, 727²⁸ Böttger, R. C., 2086 Bottino, F. A., 587148 Bottom, F. H., 23141 Boucher, R. J., 26161 Boudjouk, P., 466¹⁸⁹, 565¹⁶, 578¹⁶ Boudreaux, G. J., 174538,539, 177538,539, 86842,44, 86944. 87644 Bouma, R. J., 174⁵²³, 175⁵²³ Bounds, D. G., 642¹¹², 644^{112,148,153} Bourgain, M., 274¹⁹, 473²¹⁷, 476²¹⁷ Bourgain-Commercon, M., 246³⁵, 470²²³, 482⁴, 485⁴ Bourgeois, O. P., 730⁴⁴ Bourquelot, E., 660⁷ Boussinesq, J., 85165 Boutagy, J., 20175 Bowden, E., 30992b Bowers, K. W., 57787

Bowman, E. R., 158437, 159437, 160437, 166437 Boxler, D., 882¹⁰⁵, 894⁶³, 943⁸² Boyajian, C. G., 747⁷⁰, 770¹⁷⁶ Boyce, R., 131³³² Boyd, G. S., 74150 Boyer, S. K., 897⁹¹, 900⁹¹ Bozzini, S., 503149, 512149 Bradbury, R. B., 68194, 80731 Braden, M. L., 6615, 7415 Bradford, V. S., 969134 Bradley, J. N., 78321 Bradley, W., 88814.15 Bradlow, H. L., 5648 Bradsher, C. K., 25177, 25477, 324149 Brady, S. F., 36290 Brady, W. T., 84849, 905141 Braitsch, D. M., 48318, 50018 Branca, J. C., 864^{18,22} Branca, Q., 167485, 168485 Branca, S. J., 22133, 906144 Brand, W. W., 86³⁵, 164³⁵, 173³⁵ Brandsma, L., 87^{120,213}, 88¹²⁰, 105^{120,213,221}, 106²²², 113222, 257115, 2712, 2722, 2732d, 5212, 5222, 5516, 5526 Brandt, C. A., 136374, 141374 Brandt, S., 500132, 505132 Brannock, K. C., 21122 Bras, J. P., 771189 Brasen, W. R., 918²¹ Brattesani, D. N., 2733 Brauman, J. I., 315¹¹², 317^{112,119} Braun, M., 124255,256, 125255, 128255,256, 129255 Braverman, S., 862³ Bravetti, D., 71744 Bravo, P., 147³⁹³ Breitmaier, E., 509¹⁸¹ Breneman, C. M., 20915, 21134a, 21334a Brennan, M. P. J., 63536 Brenner, W., 39084, 39284 Breque, A., 20182 Breslow, D. S., 1029⁵⁴ Breslow, R., 896⁶⁷, 1058⁴⁰ Brettle, R., 583¹²⁶, 635³⁶, 638⁹⁴, 643¹²² Breuer, S. W., 41827 Brewer, P. D., 25177, 25477 Brewster, J. H., 927^{51,57} Brey, W. S., Jr., 898⁸⁰ Brickner, S. J., 42379, 103165,65b Brickner, W., 7051 Bridges, A. J., 139378, 154378, 155378 Briggs, L. H., 759126 Briggs, P. C., 34725 Brine, G. A., 67785 Briner, P. H., 19632 Bringmann, G., 505169, 67782 Brinker, U. H., 466¹⁸³ Brinkmeyer, R. S., 125²⁹¹, 126²⁹¹, 223^{152,153}, 363⁸⁸, 373127 Britcher, S. F., 38011 Brito, A., 327¹⁷¹ Britton, R. W., 20108 Brizzolara, A., 28170, 30170 Broadbent, H. S., 124282, 125282, 128282, 129282 Broadhurst, M. D., 43195,96 Broadhurst, M. J., 38010 Brocard, J., 99034,34a

Brock, D. J. H., 699¹⁶² Brockmann, C. J., 63418 Brockmann, H., 699160 Brocksom, T. J., 222141 Brockway, C., 224^{163,163a} Broess, A. I. A., 367100 Broger, E. A., 431101 Brohi, M. I., 53074, 53474 Broka, C. A., 125294, 126294, 167294, 168294 Brokaw, F. C., 766158 Brokke, M. E., 79059 Brook, A. G., 125314 Brook, P. R., 767162, 901109 Brookes, A., 38010 Brookes, C. J., 648187 Brophy, B. V., 901109 Broster, F. A., 41827 Brot, F. E., 35866 Brouwer, A. C., 75082 Brouwer, D. M., 330¹⁹¹ Brown, A. C., 6334 Brown, B. S., 43835 Brown, C. J., 688¹¹⁵ Brown, D. J., 56845, 63659 Brown, E., 501139, 509139, 512139 Brown, E. D., 390^{71,72}, 396^{111,114}, 398^{111,114} Brown, E. V., 5648 Brown, F., 89874 Brown, H. C., 231245, 2426, 24966, 2576, 2596, 260143, 27421, 29933, 30035,36, 42153, 48310,11, 52220, 52324 55424, 7066, 7798, 7938, 69, 71-75, 79476-78, 7958, 75, 82-84, 79775,91-95, 79895, 79998 Brown, J. D., 255¹⁰⁹, 512¹⁹⁸ Brown, J. M., 102219 Brown, L., 225187 Brown, M. J., 79268 Brown, P., 1056³⁴, 1062³⁴ Brown, R. T., 3807 Brown, S. H., 1047⁴, 1062⁴ Brown, S. L., 325¹⁶¹ Brown, S. S., 326¹⁶⁵ Brownbridge, P., 3¹², 8¹², 25¹² Browne, A. R., 62743 Brownlee, R. G., 126318 Bruce, J. M., 81043 Bruch, M., 587¹⁴¹ Brückner, R., 100258, 100866 Brun, P., 38013 Brunelle, D. J., 8⁴¹, 9⁴¹, 15⁴¹, 20⁴¹, 250⁷⁰, 288⁶³ Brunke, E.-J., 343¹⁸, 353⁵¹, 354⁵¹, 373¹³⁰ Brunner, G. L., 3807 Bruno, F., 63660 Brunold, A., 64097 Bruss, D. R., 57158 Brutcher, F. V., Jr., 89251 Brutts, D., 19733 Bruza, K. J., 252^{82,85}, 257⁸² Bryan, D. B., 27315, 27415 Bryan, R. F., 670^{57,57a}, 816⁸² Bryant, J. A., 503149, 512149 Bryce-Smith, D., 415⁶, 807¹⁸ Bryson, I., 404¹³⁶, 405¹³⁹ Bryson, T. A., 24146, 8660, 8860, 8960, 36493, 38013 Bube, T., 79790 Buchanan, A. C., 328177 Buchanan, J. G., 7331

Buchanan, R. A., 960116 Buchbauer, G., 3807 Buchholz, R. F., 761¹⁴² Büchi, G., 350³⁷, 499¹²¹, 871⁵³, 882¹⁰², 883¹⁰², 953¹⁰¹ Buchman, O., 102114 Buck, H. M., 367^{101,102} Buckley, J. S., Jr., 807²⁰ Buckner, C., 693148, 694148 Buckus, P., 328178 Buckwalter, B., 19630 Buco, S. N., 88923 Bucourt, R., 21131 Budzikiewicz, H., 81258.59 Buehl, H., 909155 Buemi, G., 38657 Bugrova, L. V., 320134 Buhr, G., 88712 Bull, D., 53075, 53187, 53475, 53887 Bull, J. R., 74153 Bullock, W. H., 25389, 26289 Bu'Lock, J. D., 66434 Buloup, A., 102638 Burnagin, N. A., 219¹¹¹, 503¹⁵², 529⁴⁸, 530⁷³, 531⁸⁸, 534⁷³, 537⁸⁸ Bumagina, I. G., 503152, 52948 Bumgardner, C. L., 88¹³⁹, 89¹³⁹, 91¹³⁹, 918²⁵ Buncel, E., 147399, 83910 Bundy, G. L., 1153 Bunes, L. A., 35970 Bunnett, J. F., 512²⁰¹ Bunton, C. A., 72311, 72415,16, 72511, 72611, 761143 Bunyan, P. J., 63988 Burckhardt, C. A., 56633 Burdon, J., 52211 Burford, C., 19851 Burgar, U., 57264 Burge, G. L., 762¹⁴⁷ Burger, J. J., 87572, 88199, 882100,101 Burger, W., 593178 Burgess, E. M., 909154 Burghard, H., 584131 Burgoyne, W., 753102 Burgstahler, A. W., 341², 360² Burk, R. M., 365% Burke, S. D., 363⁸⁶, 1051²¹ Burkert, U., 382³⁶ Burkoth, T. L., 89357 Burks, J., 176540 Burnell, D. J., 590¹⁶⁴ Burnell, R. H., 325¹⁶² Burri, K., 414³ Burrows, E. P., 4141 Bursian, N. R., 328181 Burt, E. A., 29717, 30686 Burton, A., 8786, 14286, 14486 Burton, D. J., 20286 Burtscher, P., 86211, 86311 Buse, C., 762¹⁴⁸, 769¹⁴⁸ Buser, H. R., 223155 Buser, K. R., 158440 Bushey, D. F., 19103 Busse, U., 53²⁷⁴ Buswell, R., 39292 Buter, J., 229²²⁴, 865²⁸ Butler, M. E., 79059 Butsugan, Y., 220116

Butter, S. A., 30575b Buttery, C. D., 261¹⁵³ Byers, J. H., 78535 Caballero, M. C., 396115 Cacchi, S., 53996, 103579 Cachia, P., 71122 Cadiot, P., 219103, 5513, 5523, 55311, 964125 Cadoff, B. C., 73731,32 Cadogan, J. I. G., 505161, 507161 Cady, M. A., 226198 Cahiez, G., 227²¹⁰, 245³⁰, 247⁴⁶, 249^{58,59}, 263¹⁷³, 265¹⁸⁸, 466182,193, 470214, 473214, 476214, 4825, 4995, 5055, 509⁵, 578⁹³ Cain, P. A., 67266 Caine, D., 1¹, 3^{1,10}, 8¹, 11¹, 13¹, 14⁷¹, 15^{71,79}, 17^{1,84}, 18¹, 211, 231, 251, 591, 7552 Cairncross, A., 499¹²² Calabretta, P. J., 768167 Calas, R., 57790 Calderari, G., 41227 Calianno, A., 24640, 25740 Calihan, L. E., 334214 Calingaert, G., 4157 Callahan, J. F., 39085, 39285 Callant, P., 90093 Callipolitis, A., 159454, 161454 Callot, H. J., 10489 Calo, V., 222145, 24640, 25740 Calverley, M. J., 89144, 107144, 109144, 116144 Cama, L. D., 48259 Camacho, F., 327171 Cambie, R. C., 67574, 753105, 759126 Campaigne, E., 582111,113 Campbell, A., 9146, 9276 Campbell, A. L., 20917, 21668, 224168 Campbell, B. K., 2736 Campbell, J. B., Jr., 24966, 48310,11 Campbell, K. N., 2736 Campbell, S. J., 969136 Campos, P. J., 28248 Canal, G., 28248 Cane, D. E., 24964, 105943, 106043 Cannan, R. K., 8225, 8345 Cannic, G., 89669 Canonica, L., 75297 Canonico, D. M., 328179 Cantacužene, J., 74875, 759128 Cantow, H.-J., 55748 Cantrell, T. S., 38132 Cantuniari, E. I. P., 29510 Capitaine, J., 84644 Caporusso, A. M., 21780,806, 24641, 24741, 25741, 439^{40,41}, 483¹², 491⁷⁰ Caramella, P., 1896 Carballeira, N., 587148 Carbonelle, A.-C., 681100 Carceller, E., 3807 Cardellicchio, C., 463155,166 Cardellina, J. H. I., 87152 Cardillo, G., 45250 Carel, A. B., 322141 Carey, F. A., 135³⁶⁸, 136³⁶⁸, 137³⁶⁸ Cargill, R. L., 19¹⁰³, 23¹³⁴ Caringi, J. J., 590¹⁶³ Carithers, R., 909154

Carling, R. W., 28146 Carlò, V., 28349 Carlock, J. T., 904133 Carlon, M., 19733 Carlson, B. A., 79476 Carlson, C. G., 327¹⁶⁸ Carlson, J. G., 71745, 75294 Carlson, J. L., 369120 Carlson, R. G., 103207, 87471 Carlson, R. M., 106^{223,224}, 113^{223,224}, 125²⁹⁹, 135³⁶⁷. 137377, 27834 Carmely, S., 38013, 407146 Carney, R. L., 370¹⁰⁹, 416¹⁴, 417¹⁴, 570⁵², 572⁵², 610⁵² Carpino, L. A., 883109 Carpino, P. A., 104^{208,209}, 117^{208,209} Carpita, A., 217⁹¹, 221¹²⁸, 439³⁷, 489⁵⁸, 495⁵⁸, 511⁵⁸, 515⁵⁸, 525³⁸⁻⁴⁰, 527⁴⁵, 539^{45,103,104}, 541^{45,105}, 554²⁷ Carr, M. D., 72415,16, 761143 Carretero, J. C., 162487, 168487 Carrick, W. L., 66433, 7239, 7319 Carroll, G. L., 78113 Carroll, P. J., 61821 Carruthers, W., 16, 536, 546 Carter, L. G., 19410 Carter, P. A., 27166 Carter, S. P., 168496 Cartwright, N. J., 693¹⁴⁰ Caruso, T. C., 3¹⁷, 7¹⁷ Casadei, M. A., 55284 Casagrande, F., 78956 Casal, B., 770179 Casanova, J., 3808 Casanova, J., Jr., 649²⁰², 650^{202c}, 652^{202c} Casanova, R., 89885 Casara, P., 74044 Casey, C. P., 418²⁴, 482^{3,6} Cashaw, J. L., 665³⁷ Casiraghi, G., 311¹⁰⁰ Casnati, G., 311100 Caspari, I., 903128 Casper, E. W. R., 842 Cassani, G., 48960, 49560, 50460, 51160, 51560 Cassar, L., 271¹, 530⁵⁹, 539¹⁰¹, 559⁵⁴, 1018¹¹, 1026³⁷, 1027⁴³ Cassidy, H. G., 262¹⁶⁶ Cassidy, K. C., 46255, 47255 Castaldi, G., 778⁶, 788^{6,54}, 789^{6,55-57} Castedo, L., 232²⁶², 545¹²¹, 585¹³³, 586¹⁵⁶, 591¹⁷¹, 610¹⁵⁶, 983^{19,21}, 984²¹ Castle, P. L., 24424, 464170 Castonguay, L. A., 52949 Castro, B., 19414 Castro, C. E., 21790, 21990, 5214, 5224 Casy, G., 87576 Catlin, W. E., 4159 Caubere, P., 509179 Cauquis, G., 640103 Cava, M. P., 135³⁴⁴, 136³⁴⁴, 137³⁴⁴, 138³⁴⁴, 261¹⁵⁴, 890^{31,34}, 891⁴², 901¹¹¹⁻¹¹³, 903^{113,124} Cave, R. J., 29070 Cavicchioli, S., 78957 Cazes, B., 222144, 223183, 225183, 99136, 99848,49, 99950 Cazianis, C. T., 824^{20,21}, 825²¹ Ceccherelli, P., 85792, 908146 Ceder, O., 645¹⁶⁹ Celebuski, J. E., 49380

Ceré, V., 147^{396} , 149^{413} , 151^{413} , 152^{413} , $153^{396,413}$, 155^{396} , 865^{27} , $944^{90,91}$, 946^{92} , $958^{90,112}$ Cerrini, S., 396¹¹⁰, 397¹¹⁰ Cervino, R. M., 63658 Chabert, P., 203102 Chadenson, M., 83160 Chadha, R., 46²⁵⁵, 47²⁵⁵ Chadwick, D. J., 264¹⁸⁴ Chafin, T. C., 73629, 771190 Chaimovich, H., 89794 Chalais, S., 322144, 80937 Challa, G., 5529, 10, 5579, 55957 Challenger, S., 224178 Chamberlin, A. R., 21563, 25179.80, 25479.80 Chambers, J., 264¹⁸⁴ Chambers, L., 168491, 169491, 171491 Chambers, R. D., 82532 Chamchaang, W., 103685 Chamot, E., 87256 Champagne, P. J., 644¹³⁸, 648^{138,182,184,189}, 649¹⁸⁹ Champion, J., 765¹⁵³ Chan, C., 232²⁶⁶, 488⁵⁴, 495⁵⁴ Chan, D., 84960 Chan, K. K., 953¹⁰¹ Chan, P. C.-M., 196²¹ Chan, T. H., 25157, 20071,72, 25179, 25479, 257719, 56414. 566²⁹, 607¹⁴, 929⁵⁸ Chan, T.-L., 877⁸⁹ Chandrasekaran, S., 57053, 57253.69, 57353.69, 57553. 58353, 59653, 60269, 60769, 61053,69, 72834, 74463 Chandrasekhar, J., 6612, 7412, 19411 Chandrasekhar, S., 3807 Chandwadkar, A. J., 331195 Chandy, M. J., 396104 Chang, C. K., 729³⁸ Chang, L.-J., 172513, 173513 Chang, Y.-H., 88¹³⁸, 89¹³⁸, 159¹³⁸, 161¹³⁸, 164¹³⁸, 165¹³⁸ Chano, K., 311⁹⁷ Chantegrel, B., 318127 Chantrapromma, K., 916²⁰, 926⁴⁷, 928⁴⁷, 930⁵⁹ Chao, O., 901112 Chao, S. T., 15⁷⁹ Chapdelaine, M. J., 1790, 53077, 53577 Chapman, O. L., 698^{157a}, 815⁷³, 891⁴³, 892⁴³ Char, H., 864¹⁹ Charles, J. T., 5647 Charpentier-Morize, M., 324155, 84216 Charpiot, B., 66430 Chasey, K. L., 874⁷⁰ Chassaing, G., 149407, 150407, 151407, 152407, 153407 Chatterjee, A., 85688 Chatterjee, S., 8³⁸, 12⁵⁵, 260¹⁴⁰, 385⁵⁶, 463¹⁵⁶, 469¹⁹⁹, 470¹⁹⁹, 472¹⁹⁹, 473¹⁹⁹ Chatziiosifidis, I., 25158,160 Chaudhuri, S. R. R., 85688 Chauncy, B., 507171 Chauvin, J., 25387 Chauvin, M., 21¹³⁰, 22¹³⁰ Chavdarian, C. G., 25281, 25781, 28040 Chawla, H. P. S., 73733 Chaykovsky, M., 147³⁹¹, 158³⁹¹, 778², 909¹⁵⁰, 918²¹ Chechina, O. N., 63983 Cheema, Z. K., 72622 Chekesova, M. P., 30572 Chemburkar, S. R., 7653, 7757 Chen, C.-L., 66535

Chen, F.-C., 500127, 501127 Chen, F. M. F., 618¹⁹ Chen, H.-W., 883109 Chen, J., 50²⁶⁶, 669⁵⁵ Chen, J. C., 261151 Chen, M.-H., 221129 Chen, N. Y., 30575c Chen, P., 87783 Chen, Q.-Y., 455¹²⁴, 525⁴², 530⁶⁷, 531⁶⁷, 533⁶⁷ Chen, S.-C., 73044 Chen, S.-M., 73834 Chen, T. B. R. A., 87572, 88199, 882100,101 Chen, T. L., 5662 Chen, W., 261154 Chen, Y.-Q., 369107 Chenard, B. L., 254102 Chenets, V. V., 328^{172,173,174} Cheng, C.-C., 587148 Cheng, K.-F., 36563 Cheng, L. K., 643116 Cheng, S., 369¹⁰⁸ Cheng, W., 504155, 511155, 515155 Chenier, P. J., 83911, 85369, 85411 Chernova, T. N., 3427, 34521 Cheshire, D. R., 107229 Cheung, H.-C., 168493, 169493, 171493 Cheung, Y.-F., 370¹¹² Chhabra, B. R., 402130 Chiacchio, U., 168496 Chianelli, D., 457133, 503146, 509178, 513146 Chiang, H. C., 3804 Chiba, K., 103267, 103682, 103790, 103890 Chiba, M., 784³¹ Chiba, T., 63985 Chiba, Y., 231248 Child, R. G., 767¹⁶³ Chini, M., 7348 Chinn, H. R., 224¹⁶³ Chinn, M. S., 920³⁴, 923^{34b}, 934³⁴, 1008⁶⁹ Chiong, K. N., 68195 Chiou, B. L., 79581 Chisholm, M. H., 583119 Chittattu, G. J., 107^{225,226}, 109²²⁶, 386⁶⁰ Chiu, C.-W., 124276, 126276 Chiu, S. K., 28144 Chiusoli, G. P., 42377, 102743, 103057 Chkir, M., 63973, 648180, 65173b Chliwner, I., 219¹¹⁴, 499¹⁴⁰, 501¹⁴⁰, 502¹⁴⁰ Cho, Y.-S., 41224 Chodkiewicz, W., 551³, 552³, 553¹¹ Choi, S. U., 29933 Chong, J. M., 196²¹, 225^{186,190,191} Chong, W. K. M., 683103 Chou, C. S., 104813 Chou, S.-S. P., 172514 Chou, T.-S., 172513, 173513,517 Chow, F., 8774,113, 9574, 9774, 10974, 11474, 11674, 13674.113, 14174, 145113, 157113, 78746 Chow, S. W., 42788 Chow, Y. L., 380¹¹, 689¹²³ Christ, J., 64095 Christensen, B. G., 89032 Christensen, J. R., 595¹⁹¹ Christensen, L., 75084, 76284 Christensen, L. W., 181554 Christenson, B., 21235

Christol, H., 72729, 73411, 74047,48, 74461, 81364 Christoph, G. G., 87256 Christy, M. E., 380¹¹ Chu, C., 305^{75b} Chu, C. C. C., 30¹⁷⁴ Chu, C.-Y., 24963 Chu, D. T. W., 21676 Chu, K.-H., 21560 Chuang, Y.-H., 224175, 75295 Chuaqui-Offermanns, N., 31186 Chuche, J., 57892, 61092 Chuit, C., 24638, 470221, 471221 Chukovskaya, E. Ts., 42269 Chun, M. W., 49261 Chung, K.-H., 592172 Chung, L.-L., 87046 Chung, S.-K., 82419,23 Chung, W.-K., 49²⁶¹ Churi. R. H., 74876 Ciabattoni, J., 3808 Ciaccio, J. A., 224165 Ciattini, P. G., 220118, 222118, 103579 Ciskowski, J. M., 316¹¹⁸, 317¹¹⁸ Claesson, A., 21782, 44572, 49277 Clapper, G. L., 56847 Clardy, J., 87256 Clardy, J. C., 33189, 34198, 39198, 232266, 28454, 38010, 395102, 396102, 592174,175, 594174, 698157a, 87256 Clarembeau, M., 8772,73,79,80,84,101, 88127, 89127, 91127, 92127, 93127, 9572,79,80, 104101, 10979,80,84,127, 11179,80,230, 116127 Claremon, D. A., 39216 Clark, B. C., Jr., 73629, 771190 Clark, D. E., 688115 Clark, D. R., 102427 Clark, F. R. S., 501135 Clark, G. R., 67574 Clark, G. W., 105017 Clark, J. H., 54278 Clark, J. S., 39²¹⁴ Clark, M., 82532 Clark, M. C., 87100, 104100, 110100, 117100 Clark, M. T., 82528 Clark, P. D., 87107, 105107, 106107, 114107, 120107, 157107 Clark, R. D., 21129, 158441, 159441, 28040, 34419 Clark, T., 1944, 42268 Clayton, J. F., 254101 Cleary, D. G., 226199 Cleghorn, H. P., 56844 Cleland, G. H., 88924 Clemens, A. H., 88121 Clement, A., 89247 Clement, B., 154421 Clements, G., 261152, 514211 Clerici, A., 564^{6,30,31}, 566^{30,31}, 595¹⁸⁹, 606¹⁸⁹ Cleveland, E. A., 499111 Clibbens, D. A., 88816 Cline, J. K., 582113 Clinet, J.-C., 255106, 257116, 264179, 49172 Clive, D. L. J., 8645, 8745, 9445, 107225,226,229, 109226. $114^{45}, 117^{45}, 213^{44}, 219^{110}, 386^{60}, 421^{61}, 422^{61}, 589^{161,162}, 610^{161,162}$ Clizbe, L. A., 81683 Cloez, C., 84326 Closse, A., 131³²⁷ Closson, W. D., 39393, 39495

Clough, J. M., 514²¹⁰ Coates, R. M., 844, 1544, 71434, 78639 Coburn, J. I., 73943 Cockerill, G. S., 26163 Coe, P. L., 52211, 585135, 63984, 648187 Coelho, F., 222141 Coerver, J. M., 35243b Coffen, D. L., 134336, 103892 Coffman, H., 79061 Cohen, H., 56632 Cohen, J. F., 48²⁵⁹ Cohen, J. I., 56737, 60637 Cohen, M., 56410 Cohen, S. G., 564¹¹, 567^{11,37}, 606³⁷ Cohen, T., 88^{128,129}, 89¹²⁹, 103^{204,205}, 105^{129,217} 108^{204,205}, 112¹²⁹, 124^{128,253,258}, 126²⁵³, 196²⁸, 482^{1,2}. 483^{1,2}, 491^{1,2}, 499^{1,2,114,117}, 660⁸, 753¹⁰³, 771¹⁸⁴, 785^{36,36a}, 937^{74,75}, 1051²⁰ Coillard, J., 83160 Colby, T. H., 89034 Cole, T. E., 79793 Cole, T. W., 855⁸⁰ Coleman, D., 21455 Coleman, D. T., III, 82419 Coleman, J. P., 63424, 63924, 64924 Collier, T. L., 35²⁰⁶ Collignon, N., 20183 Collins, C. J., 722⁶, 723^{6,7}, 724¹², 726²², 731^{6,7}, 782²⁰. 822^{8,9}, 823⁹, 825²⁹, 830⁵⁸, 831⁵⁸, 834⁹, 836⁸ Collins, D. J., 762¹⁴⁷ Collins, P. M., 597199 Collins, T. J., 213⁵¹ Collman, J. P., 2083, 2133b, 102427, 102851 Collum, D. B., 33189, 34198,199, 39198,199 Colomer, E., 38125, 38225 Colon, I., 500130, 509130 Colonge, J., 315¹¹⁰ Colquhoun, H. M., 10178 Colton, C. D., 380 Colvin, E. W., 20067 Comasseto, J. V., 120^{239,241}, 136³⁷⁴, 141³⁷⁴ Combret, J.-C., 2429, 759129 Cometti, G., 103057 Comins, D. L., 255109, 512198 Commercon, A., 246³⁴, 247⁴⁶, 440⁴², 441⁴², 442⁴², 473²¹⁷, 476²¹⁷, 485³¹, 486³¹, 522^{18,19} Commercon-Bourgain, M., 470226, 473226 Comte, M.-T., 767160 Conacher, H. B. S., 752⁹⁶ Conant, J. B., 565¹⁸ Concannon, P. W., 3808 Concellón, J. M., 78850 Công-Danh, N., 462152, 513207 Congson, L. N., 714³³ Conia, J.-M., 424, 20104, 72732, 765153,154, 767162, 83268b, 84217, 84817.48 Conn, R. S. E., 24426, 48528 Connor, D. S., 42270 Connors, W. J., 66535 Conrad, P., 960116 Consiglio, G., 228222, 229230,230a, 24636, 43829, 452110. 487⁴⁵, 1023²² Constantino, M., 592172 Conway, B. E., 634⁶, 636^{6,56,57}, 637^{6,61}, 639^{566,77}, 655⁶ Conway, D. C., 29933 Conway, P., 93361

Cook, J. M., 3807 Cook, L. C., 407150 Cook, S. J., 102219 Cooke, F., 198⁵¹ Cooke, M. P., Jr., 87787, 102117, 102428 Cooke, R., 41933 Cooke, R. G., 828^{52,53} Cookson, C. M., 61612 Cookson, R. C., 135³⁵⁰, 136³⁵⁰, 137³⁵⁰, 141³⁵⁰, 349³⁴, 623³⁷, 625³⁷ Coombs, R. V., 28, 118, 168, 178, 268, 653226 Cooper, G. D., 86632 Cooper, J. L., 343¹⁷, 348²⁷, 353⁴⁸, 355^{17,48}, 357^{17,48}, 35868 Cooper, K., 218100,100b, 229234, 44466 Cooper, P. J., 35243b Cooper, R., 693144 Cope, A. C., 379^{1,3}, 380^{1,5,13,16}, 381^{16a,17}, 382^{3b}, 393⁹³. 414¹, 735¹⁷, 760¹³⁹, 774¹³⁹, 856⁸⁷ Copp, L. J., 21792 Coppola, G. M., 223146 Corbel, B., 167486, 168486, 19955, 257117, 105227 Corbett, J. F., 82955 Corbin, V. L., 41943 Cordell, G. A., 682¹⁶⁵, 691¹³³ Coreil, M., 509179 Corey, E. J., 3⁹, 12⁶⁴, 17⁸⁵, 30¹⁸⁰, 34¹⁸⁰, 35^{180,200}, 86^{26,59}, 88^{59,125}, 89⁵⁹, 103^{203,206}, 108²⁰³, 112⁵⁹, 118¹²⁵, 120²⁴⁰, 12126, 12359,125, 124265,266,274,275,278 12559,266,274,275,278,292, 126292, 127265,266,274 128265,266.278, 129274,278, 131326, 132278, 134338, 135^{326,338}, 137³³⁸, 147³⁹¹, 158³⁹¹, 178^{541,545}, 181⁵⁴⁵, 1948, 20180, 21237,38, 21350, 21561,67, 21670 194, 2016, 212⁻¹⁰⁴, 213⁻¹⁰⁴, 213⁻¹⁰⁴, 218⁻¹⁰, 218⁻¹⁰, 218⁻¹⁰, 218⁻¹⁰, 218⁻¹⁰, 218⁻¹⁰, 218⁻¹⁰, 250¹²⁶, 238^{1b}, 247^{49,50}, 248⁵⁵, 249⁶⁴, 250^{49,69,70}, 251⁵⁵, 257¹²¹, 269⁵⁵, 274²⁰, 288⁶³, 356⁵⁹, 361⁷⁹, 365⁹⁶, 419⁴⁴, 420⁴⁴, 421⁵⁶, 427^{85,88}, 430⁹²⁻⁹⁴, 431⁹⁹⁻¹⁰², 546¹²³, 570⁵²⁻⁵⁴, 572^{52,53,69}, 573^{53,69,72}, 31⁹⁹⁻¹⁰², 546¹²³, 570⁵²⁻⁵⁴, 572^{52,53,69}, 573^{53,69,72}, 31⁹⁹⁻¹⁰², 546¹²³, 570⁵²⁻⁵⁴, 572^{52,53,69}, 573^{53,69,72}, 31¹⁰⁰, 31¹⁰⁰, 31¹⁰, 31¹ 57553, 58353, 59653,72, 59972, 60269,72, 60769,72, 610^{52,53,69}, 638⁹⁰, 649²⁰², 650^{202c}, 652^{202c}, 709¹⁶, 71225, 71329, 72940, 74463, 762146, 7782, 909150, 918²¹, 936⁷³ Corkins, H. G., 87151 Corley, E. G., 38011 Cornélis, A., 125303, 322144 Cornforth, J., 502141 Corriu, R. J. P., 228215, 4364, 48426, 49226, 49426, 49526. 503²⁶, 513²⁶ Cose, R. W. C., 94995 Cossec, B., 46²⁵⁴ Cossy, J., 176⁵⁴⁰, 281⁴², 599^{209,210}, 602²²³, 603²²⁸ Costantino, P., 74149 Costanzo, S. J., 260144 Coste, J., 74047,48 Costello, G., 44467 Costisella, B., 953¹⁰⁵ Côté, J., 846⁴³ Cotter, R. J., 39393 Cottrell, C. E., 163472 Coudane, H., 754109 Coulombeau, A., 5647 Coulson, D. R., 48259 Coulter, A. W., 901112 Courtois, G., 420⁵¹, 421⁵¹ Cousin, H., 693¹⁴⁵ Coutrot, P., 759129,130 Coutts, I. G. C., 689118

Covell, A. N., 28143 Cowell, A., 103362 Cowell, G. W., 783²¹ Cowley, A. H., 21128, 21528 Cox, A., 689¹²³ Cox, G. B., 63894 Cox, P. B., 232²⁶⁶, 488⁵⁴, 495⁵⁴ Cox, R. H., 62953 Coxon, J. M., 74153, 74255-57, 74357,60, 74666, 75190, 752⁹¹ Crabbé, P., 257120 Crabtree, R. H., 10474, 10624 Crafts, J. M., 29930, 317121 Crain, D. L., 319131 Cram, D. J., 114233, 124281, 125281, 154233, 164475, 503149, 512149, 72727, 8228, 8368 Crandall, J. K., 3807, 605232, 74154, 89252 Crane, R. I., 38668 Cravador, A., 8650, 8776,84,86, 10984, 13676, 14286, 144^{76,86}, 145⁷⁶ Craw, P. A., 67574 Crawford, R. J., 903121 Crawford, T. C., 799102 Creary, X., 614⁵, 788⁵² Creese, M. W., 75085 Creighton, E. M., 913¹ Cremlyn, R. J. W., 78119 Cresson, P., 963122, 99135 Crich, D., 3809 Cripe, K., 41933 Crisp, G. T., 25071, 48749, 49549, 52950, 53076, 53476 Cristea, I., 482², 483², 491², 499² Cristol, S. J., 164476, 42267 Crombie, L., 2426, 2576, 2596, 43197,98, 49486, 545122. 558⁵³, 976⁹, 977^{9,9a}, 989⁹, 990⁹ Cronyn, M. W., 86840 Crook, E. M., 89882 Crosby, A., 88¹²⁴, 118¹²⁴ Crosby, G. A., 279³⁹, 758¹²⁵, 770¹⁷² Cross, B. E., 71542 Cross, G. A., 229223 Crotti, P., 7348,10, 74149 Crouch, R. K., 842 Crouse, G. D., 44570 Crowder, D. M., 97170, 117170 Cruickshank, K. A., 53079, 53579 Crump, D. R., 42787 Crumrine, A. L., 50²⁶⁷ Crute, T. D., 226199 Csicsery, S. M., 30575b Csizmadia, I. G., 89141b Csuk, R., 57055, 58255, 58355, 63057, 63157 Culbertson, T. P., 84640 Cullin, D., 19733 Cunico, R. F., 254101 Cunningham, R. H., 760140 Cupas, C. A., 334²²⁰ Curi, C. A., 828⁴⁷, 854⁷⁴ Curini, M., 857⁹², 908¹⁴⁶ Curphey, T. J., 30174,177, 57266 Curran, D. P., 55281, 219114, 221129,130, 499140, 501140, 502¹⁴⁰, 603²²⁹, 649²⁰¹, 672⁶⁴, 824²³ Currie, J. K., 102012, 102429 Curtin, D. Y., 755¹¹⁰ Curtis, G. G., 901112 Curtis, N. J., 107226, 109226

Curtis, R. F., 55314, 690124 Curtis, V. A., 71433 Curtius, T., 89032 Cushman, M., 953101 Cutter, H. B., 56518 Cuvigny, T., 31¹⁸⁴, 45²⁴³, 123²⁴⁷, 174⁵³⁶, 176⁵³⁶, 178⁵³⁶, 448⁹⁵ Cvejanovich, G. J., 88924 Cymerman-Craig, J., 824¹⁸ Cyr, D. R., 225¹⁸⁶ Czarkie, D., 38013, 407146 Czamy, R. J., 36563 Dabbagh, G., 20912, 21024, 21127, 21343, 215276 Dadson, W. M., 71224 Daesslé, C., 35865 Dagleish, D. T., 663²⁶, 665²⁶ Dahler, P., 62338 Dahn, H., 822^{3,12}, 831^{3,12}, 835^{12b} Dailey, O. D., Jr., 135³⁶⁸, 136³⁶⁸, 137³⁶⁸ Dakin, H. D., 88924 d'Alarcao, M., 21793 Daley, R. F., 57994, 640110 Daljeet, A., 503149, 512149 Dallaire, C., 38010 Daly, J., 167485, 168485 Damiani, D., 734⁸ D'Amico, A., 45250 Damodaran, N. P., 402127 Damon, R. E., 3809, 67572 Dams, R., 583¹²², 584¹³⁰, 587¹⁴⁸ Dana, G., 572⁶⁰, 578⁹², 610⁹², 728³⁶ D'Andrea, S., 1037⁸⁷ Dang, H. P., 44043, 48530, 49130, 52744 d'Angelo, J., 1³, 197⁴², 198⁴⁴ Danheiser, R. L., 21126, 22132, 57053, 57253,69, 57353,69, 575⁵³, 583⁵³, 596⁵³, 602⁶⁹, 607⁶⁹, 610^{53,69}, 744⁶³ Daniewski, W. M., 88¹²⁸, 124^{128,258} Danishefsky, S. J., 49²⁶⁴, 503¹⁴⁹, 512¹⁴⁹, 816⁸³, 890³³ Dann, O., 86419 Dao, Lê H., 822^{3,12}, 831^{3,12} Darby, N., 71122 Darnbrough, G., 977^{9a} Darwish, D., 969136 Das, G., 362⁸² Das, J., 361⁷⁹, 762¹⁴⁶ Das, K. G., 386⁶¹, 393⁶¹ Da Settimo, F., 439^{40,41} Dasgupta, S. K., 427⁸⁷ da Silva, G. V. J., 589162, 610162 da Silva, R. R., 105122, 105222 Daterman, G. E., 124²⁷⁷ Daub, G. W., 369¹²¹, 372¹²¹ Dauben, H. J., Jr., 651²¹⁸ Dauben, W. G., 23¹³⁹, 99¹⁸⁹, 107¹⁸⁹, 110¹⁸⁹, 407¹⁵⁰, 431^{95,96}, 843²², 855⁸⁶, 903¹²⁰ Daum, H., 144³⁸⁴, 145³⁸⁴, 147³⁸⁴, 568⁴⁰, 606⁴⁰ Daum, S. J., 85478 D'Auria, M., 512^{192,213}, 515^{192,213} Dave, V., 784³⁰ Daves, G. D., Jr., 124277 Davidovics, G., 89247 Davidson, A. H., 20176 Davidson, R. S., 56737, 60637 Davidson, T. A., 67991 Davies, A. M., 94995

Davies, G. M., 511189 Davies, J. E., 63889 Davies, J. S., 81894 Davies, M., 38132 Davies, P. S., 511189 Davies, S. G., 47257, 19737, 102953 Davies, W., 89882 Davis, A. P., 3804 Davis, B. R., 325157, 35349 Davis, D. D., 565¹⁹ Davis, L. L., 105122, 105222 Davis, N. R., 125314 Davis, R. E., 38010 Davisson, M. E., 325161 Dawson, M. I., 36384 Day, A. C., 66536, 689119 Day, R. A., Jr., 57785, 57994, 595192, 640110 De, B., 274²⁰ Dean, F. M., 807²⁵, 900⁸⁹ Deardorff, D. R., 73628 Debal, A., 26317 DeBernardis, A. R., 72731 DeBernardis, J., 94688, 99034 de Brouwer, R. J., 367¹⁰² Decesare, J. M., 167486, 168486, 74877 de Clercq, P., 713²⁸ De Clerq, J. P., 85790 Decorzant, R., 882104 Degani, I., 125307 Degen, P., 24856, 25156 Degenhardt, C. R., 25496 Degner, D., 634²⁸, 640¹⁰⁰ DeHaan, F. P., 29933 de Haan, J. W., 367¹⁰¹ Dehasse-De Lombaert, C. G., 78642 Dehmlow, E. V., 509178, 85167 Dehn, W. M., 825^{25,27b}, 826²⁵ DeHoff, B. S., 100873, 100974, 101074 Deiko, S. A., 499¹²⁵, 669⁵³ Deitrich, W., 66850 De Jong, K. P., 55210 de Jonge, J., 904129 DeJongh, D. C., 89142 De Kimpe, N., 85790 de la Pradilla, R. F., 265191 Delbecq, F., 256113 Delektorsky, N., 72517 de Lera, A. R., 223¹⁴⁸, 586¹⁵⁶, 610¹⁵⁶ Della Vecchia, L., 105²¹⁹, 113²¹⁹ Delmond, B., 74150 Delorme, D., 27727 del Valle, L., 41823 Demailly, G., 34316 de Mayo, P., 379³, 386⁶⁶, 706⁷, 707^{9,10}, 815⁷³, 836⁸¹, 922³⁷, 924³⁷ de Meijere, A., 905¹³⁸ DeMember, J. R., 299^{34b}, 333^{34,208,209} De Mico, A., 512¹⁹², 515¹⁹² Demmin, T. R., 34725 Demonceau, A., 10477, 10517 Demoulin, B., 88¹²³, 91¹²³, 93¹²³, 109¹²³ Demuth, F., 124²⁸³, 125²⁸³, 127²⁸³, 132²⁸³ Demuth, M. M., 21675, 81573 Deng, C., 53274 De Nie-Sarink, M. J., 30249

Denis, J. N., 8650, 8783,89-91,110,118, 89110, 105110. 10683,110, 114110, 12083,90,110, 12183, 155427, 25391 Denmark, S. E., 21127 Denney, D. B., 760137 Denniston, A. D., 132334, 133334, 136334, 35462 de Pascual Teresa, J., 396115 De Paulet, A. C., 85477 Depezay, J.-C., 19844, 258127 Depp, M. R., 380⁷ Deprés, J.-P., 222¹⁴¹, 783²⁶ DePriest, R., 21458 DePue, R. T., 33¹⁸⁹, 34¹⁹⁸, 39¹⁹⁸ de Reinach-Hirtzbach, F., 74877 Derguini, F., 21565 Derguini-Boumechal, F., 224161, 24313, 262161, 263161, 416²⁰, 417²⁰, 464¹⁷⁶, 466^{176,188} Derieg, M. E., 4141 Dermanovic, M., 313103 Dermanovic, V., 313103 Dermer, O. C., 505¹⁶¹, 507¹⁶¹ Derocque, J.-L., 396¹⁰³ des Abbayes, H., 102012, 102638.39 Desai, M. C., 546¹²³, 797⁹¹ de Salas, E., 7065 Desauvage, S., 87⁹⁰, 120⁹⁰ Descoins, C., 246³⁴, 416²¹ Deshayes, H., 613³ Deshpande, M. N., 3807 Deslongchamps, P., 56²⁸⁶, 380¹⁰ Despeyroux, B., 103060 Des Roches, D., 770181 Dessy, R. E., 581110 Detzel, A., 82954 Deuchert, K., 19740 Deuter, J., 87261,62 Dev, S., 71327, 71537, 73733 Devaprabhakara, D., 38658.61, 39361 de Voss, G., 82954 de Vries, L., 406142 de Waal, W., 20108 de Waard, E. R., 153414, 155414, 87572.77, 88199, 882100.101 Dewar, M. J. S., 664³², 665³² D'Haenens, L., 90093 Dhanoa, D. S., 21240, 25070 Dhawan, K. L., 32188, 265193, 266194 Dhillon, R. S., 231244 Dhimane, H., 579101 Dhingra, O. P., 509¹⁷⁶, 660¹⁹, 670⁵⁷, 673⁶⁸, 679¹⁹, 681⁶⁸, 683^{57b}, 686⁶⁸, 807^{32.33} Dianin, H., 6604 Diaz, F., 327171 Dich, T. C., 124²⁸⁶, 125²⁸⁶, 127²⁸⁶ Dicke, R., 49593b Dickerson, R. E., 1477, 1577, 36384 Dickey, E. E., 693¹⁴³ Dickinson, T., 634^{25a} Dickman, D. A., 75⁵⁰, 77⁵⁵, 78⁶⁰⁻⁶², 79^{60,61}, 81^{60,61,66} Diebold, J. L., 845³⁶ Dieck, H. A., 271¹, 530⁶⁰ Dieden, R., 8787 Diederich, F., 55742, 92749 Diels, O., 89354 Diercks, R., 53790, 53890.92 Dieter, R. K., 22133, 24150 Dietl, H. K., 21¹²² Dietrich, H., 693143

Dietrich, R., 890³⁴, 904^{131,132} Differding, E., 78642 Dijkink, J., 36177, 36477, 368104 Dijkstra, R., 904¹²⁹ Dilworth, B. M., 8628 Dingwall, J., 84850 Dinh-Nguyen, N., 63429 DiPasquale, F., 732, 832 DiPasquo, V. J., 85581 Dirksen, H. W., 499113 Disanayaka, B. W., 590163 Disnar, J. R., 20295 Distler, H., 87259 Ditrich, K., 79790 Dittami, J. P., 23135 Dittmann, W., 396107, 397107 Dittmer, D. C., 72520 Dittus, G., 28454 Divakaruni, R., 103058, 103161 Dixon, B. R., 49380 Dixon, J. R., 73519 Dixon, N. J., 44468 Djahanbini, D., 223183, 225183 Djerassi, C., 804², 805¹⁵, 810^{2.51}, 812² Djuric, S., 255¹⁰⁵ Döbler, Chr., 229224 Dobrynin, V. N., 7347 Dodson, P. A., 689¹²¹, 813⁶⁰ Dodson, R. M., 807²⁰, 818⁹⁵ Doecke, C. W., 62743 Doering, W. E., 82213, 82316, 82516, 82913, 89466, 896^{66,68} Doerler, G., 904133 Doi, T., 380¹⁰ Dokuchaeva, T. G., 329¹⁸³ Dolak, L. A., 35347 Dolak, T. M., 8660, 8860, 8960 Dolby, L., 31183 Dolce, D. L., 62130 Dolgov, B. N., 318122 Dolle, R. E., 39²¹⁶, 278³¹, 289³¹, 1029⁵⁵ Dollinger, H., 656 Dolph, T., 596193 Dolphin, J. M., 960115 Domagaia, J. M., 74879, 74980 Donaldson, W. A., 27417 Dondoni, A., 232267, 511186,187, 514186 Donnelly, D. M. X., 691136 Donnelly, J. A., 73629 D'Onofrio, F., 512^{192,213}, 515^{192,213} Doomes, E., 87786 Doorenbos, N. J., 78112 Dorier, P. Ch., 242² Dorling, S., 80721 Dörr, H., 87364 Doskotch, R. W., 39077.79, 39277, 396109, 397109 Dou, H. J.-M., 505¹⁶², 507¹⁶², 512¹⁶² Dougherty, T. K., 541116, 543116 Doughty, D. H., 1041110 Douglas, E. C., 328179 Douglass, J. E., 327170 Dove, M. F. A., 29933 Dowbenko, R., 380⁹ Dowd, S. R., 30179 Doweyko, A. M., 158447, 159447 Doxsee, K. M., 503149, 512149

Author Index

Doyle, M. P., 91929, 92034, 923346, 92534a, 93434, 100869, 105635, 106235 Doyle, P., 38668 Drake, J., 3808 Drake, J. E., 46255, 47255 Dralle, G., 647¹⁹⁹ Dreiding, A. S., 804³, 807²², 810⁴⁸, 1049^{14,15}, 1058⁴² Dressel, J., 87470 Dreux, M., 257117 Drewes, S. E., 42152 Driessen-Engels, J. M. G., 367¹⁰¹ Driggs, R. J., 470²⁰⁴, 473²⁰⁴ Drover, J. C. G., 227²¹³ Druelinger, M., 36²⁰⁹ Druey, J., 92755 Dryden, H. L., 81679 Dua, S. K., 164⁴⁷⁸, 167⁴⁷⁸, 168⁴⁷⁸ Dua, S. S., 457¹²⁸ Duax, W. L., 38661, 39361 Dube, S., 96¹⁶⁴, 98¹⁶⁴, 99¹⁶⁴ Dubini, R., 99645 DuBois, G. E., 14⁷⁶, 15⁷⁶, 347²⁵, 373¹²⁶ Dubois, J. E., 257¹²¹, 636⁶⁰, 723¹⁰ Duboudin, J. G., 44045 Dubovenko, Z. V., 38668 Duc, L., 822¹², 831¹², 835^{12b} Ducep, J. B., 86¹⁴, 94¹⁴, 95^{14,151}, 96^{151,161,165}, 98^{151,161}. 99^{151,165,179}, 100^{14,179}, 101¹⁶⁵, 107¹⁶⁵, 196²⁹ Duchêne, A., 19624 Duclos, R. I., Jr., 507172 Dudfield, P., 46252 Duff, J. M., 125³¹⁴ Duffaut, N., 57790 Duffin, D., 38662,63 Dufour, J.-M., 325162 Dugar, S., 504154, 511154, 515154 Duggan, M. E., 224¹⁷⁴, 618^{20,21} Dugger, R. W., 220117 Duggleby, P. McC., 88815 Duhamel, L., 2538 Duke, A. J., 767¹⁶² du Manoir, J., 180550 Dumas, D. J., 373129 Dumont, W., 86⁵⁰, 87^{66,67,72,73,77,84,86,91-93,116-119} 89142,143, 92143, 9572, 10984, 111231, 116142, 118143, 120242, 123143, 13677, 13777, 14177, 14286, 14477.86, 14577 Duñach, E., 255106 Dunathan, H. C., 36175 Duncan, J. L., 82640 Duncan, W. G., 126318 Dundulis, E. A., 39²¹⁹, 40²¹⁹ Dunlop, M. G., 67574 Dunn, G. L., 85581 Dunn, J. L., 914^{3.5} Dunnavant, W. R., 82637 Dunne, T. J., 4²³ Dunoguès, J., 57790 Dunstan, A. E., 331 198 Dupin, C., 734⁸ Dupin, J. F., 7348 Dupre, B., 7756 Duraisamy, M., 587146, 610146 Dürckheimer, W., 89034 Duréault, A., 258127 Dürner, G., 56285

Durst, T., 86^{32,33,37,38}, 147^{32,33,392,397}, 149^{392,397,400,403} $\begin{array}{c} 150^{397,403}, \, 151^{397,403}, \, 152^{392}, \, 154^{32}, \, 158^{37,38}, \, 159^{37,460}, \\ 161^{460}, \, 163^{38,460}, \, 164^{37}, \, 167^{460,486}, \, 168^{486}, \, 173^{37,38}, \end{array}$ $180^{550}, 265^{193}, 266^{194}, 748^{77}, 770^{181}, 862^2, 979^{13}$ Dusold, L. R., 89142 Dutler, H., 81574 Dutra, G. A., 224176, 24857, 24957, 25157, 26357 Dvolaitzky, M., 80722 Dworkin, A. S., 328177 Dwyer, F. G., 30575c Dyke, S. F., 670⁶⁰ Dyrbusch, M., 30353 Dyszlewski, A. D., 253⁸⁹, 262⁸⁹ Dzhafarova, N. A., 30469 Dzieciuch, M., 63656,57, 639566,77 Eaborn, C., 41931, 56415 Eakin, M., 600²¹³ Eakin, M. A., 38013 Eapen, K. C., 450¹⁰⁴, 457¹²⁸ Earnshaw, C., 20176 Eastham, J. F., 7226, 7236, 7316, 8226.8, 8236, 8246, 825²⁹, 834^{6,73}, 836^{6.8} Eastmond, R., 55531 Easton, N. R., Jr., 38236 Eastwood, F. W., 7128 Eaton, J. K., 88814 Eaton, P. E., 57370, 85580 Ebata, T., 224177 Eberhardt, H.-D., 904131 Eberle, D., 93362 Ebermann, R., 687¹¹² Eberson, L., 634^{7,7c,19,20,22}, 636^{7,7a-c}, 637^{7,63,64}, 638^{19,20}. 63968-72, 643132, 64919,20, 65519,20 Ebert, G. W., 20919, 52210, 55317 Ebetino, F. F., 56412 Ebetino, F. H., 681100 Ebine, S., 81896 Echavarren, A. M., 232257,263, 233272, 455123, 470211, 473211, 475211, 476211, 49294, 49594, 50494, 51494, 102425 Echigo, Y., 73046 Eck, C., 386⁶⁸ Eck, C. R., 386⁶⁸, 711²², 712²⁶ Eck, S. L., 38010 Ecke, G. G., 296¹² Eckrich, T. M., 1948, 21784, 24750 Eda, S., 585¹³⁷ Edgar, J. F., 89667 Edgar, K. J., 251⁷⁷, 254⁷⁷ Edmison, M. T., 505161, 507161 Edo, K., 461 147 Edstrom, E. D., 36281 Edwards, B., 89791, 90091 Edwards, J. D., Jr., 66537 Edwards, P. D., 6925.27, 7025, 7225, 7425 Ege, G., 807²⁶ Egert, E., 30353 Eggensperger, H., 66122 Egger, B., 87153 Eggersdorfer, M., 312101 Eglington, G., 5511 Ehlinger, E., 49172 Ehrenkaufer, R. E., 1040109 Eichenauer, H., 39²¹³ Eigen, I., 89887

Eilbracht, P., 62338 Eis, M. J., 224169, 262169 Eisch, J. J., 164⁴⁷⁸, 165⁴⁷⁹, 167⁴⁷⁸, 168⁴⁷⁸, 173^{479,519}, 198⁵⁰, 579⁹⁷, 759¹³⁴ Eisenbraun, E. J., 322141 Eissenstat, M. A., 31182 Eistert, B., 81685, 8873.8, 8883.8.13, 89035, 89137, 8938, 8973,8, 9008, 9038,126,-128 Eiter, K., 28657 Ejiri, E., 593181 Ekogha, C. B. B., 25286 Elberling, J. A., 84958 El-Feraly, F. S., 39077.79, 39277 El-Garby Younes, M., 83478.79 El Hallaoui, A., 21566, 25175 Elias, H., 89035, 903127 Eliel, E. L., 7654, 124273.287.288, 128273.287.288.323, 132273,288, 134273,323 Elisseou, E. M., 220119 Elissondo, B., 19625, 453115 El-Khawaga, A. M., 325¹⁶⁰ Elks, J., 80413 Elliot, J. D., 28145 Elliott, I. W., 686¹⁰⁹, 699¹⁰⁹ Elliott, J. D., 226²⁰⁶, 360⁷² Elliott, R. D., 79058 Elliott, R. L., 221129,130, 603229 Ellis, K. L., 26161 Ellis, P. D., 19¹⁰³ Ellison, G. B., 422 Ellison, R. A., 124^{272,276}, 125²⁷², 126^{272,276}, 145³⁸⁶ Ellsworth, E. L., 21234b,c, 21346, 21646, 22346 Elofson, R. M., 507170 Elsevier, C. J., 21781, 49168,69, 53184 Elsom, L. F., 41828, 499123 Elzen, V. D., 979¹³ Elzohry, M. F., 325160 Eman, A., 8792 Emel'yanov, M. M., 35971 Emerson, D. W., 154420 Enda, J., 19733 Enders, D., 30¹⁸⁰, 34^{180,191}, 35^{180,191}, 37¹⁹¹, 39^{191,213}, 65⁷ Endo, H., 42154 Engel, C. R., 84641-44 Engel, M. R., 698¹⁵⁷ Engel, P., 623³², 626^{32b} Engelhardt, E. L., 38011 Engler, D. A., 91824 Engler, T. A., 84435 English, J., Jr., 262166 Enhsen, A., 25393 Ennis, M. D., 45²⁴⁹ Enokiya, R., 56626 Enomoto, M., 45248 Enos, H. I., Jr., 30463 Entwhistle, E., 55419 Erdelmeier, I., 230237 Erdik, E., 208¹⁰, 210¹⁰, 215^{10c}, 243¹², 250¹², 262¹², 436¹⁸ Erdtman, H., 6609, 66747 Erhardt, F., 66541 Erhardt, J. M., 226¹⁹⁵, 264¹⁸³ Erickson, B. W., 103²⁰³, 108²⁰³, 124²⁷⁵, 125^{275,313} Erickson, G. W., 36²⁰⁹, 37²¹² Erkamp, C. J. M., 21025 Erman, W. F., 7068, 81574 Ermer, O., 382³⁶, 593¹⁷⁹

Ermokhina, V. A., 321138 Ernest, I., 781¹⁴, 896⁷² Ernst, P., 6603 Ershov, V. V., 81469 Erwin, R. W., 25151 Eschenmoser, A., 1368, 3412, 34520, 34623, 35139, 3602, 916¹⁹ Escher, S. D., 126³²⁰, 882¹⁰⁴ Eskin, N. T., 63980 Eskins, K., 691129, 693129 Espy, H. H., 39393 Estep, R. E., 74877 Estermann, H., 43237 Eswarakrishnan, V., 87893,94, 87993,94,97, 88094, 88194 Etheredge, S. J., 19101 Eto, H., 691¹²⁸, 693^{128,139}, 697^{139,155,156}, 698¹²⁸ Eustache, J., 262167 Evans, B. E., 38011 Evans, C. F., 589161, 610161 Evans, D. A., 1², 2², 3², 13², 15², 23², 25², 41², 44², 45^{2,244,246,249}, 55², 86¹⁶, 95¹⁶, 96¹⁶, 97¹⁶, 98¹⁶, 99¹⁶, 10416, 11616, 11716,235,236, 11816, 135363, 136363, 139363, 142363, 15516,235,236, 156235,236,363, 16816 196³⁰, 199⁵³, 672⁶⁶, 781¹³, 799¹⁰², 919³¹, 934³¹, 939⁷⁸, 946⁸⁷ Evans, D. E., 85477 Evans, D. H., 56847 Evans, J. B., 160465, 164465, 166465 Evans, J. M., 73519 Evans, T. W., 82525, 82625 Evans, W. H., 38346 Everhardus, R. H., 87²¹³, 105²¹³ Fabian, W., 9754, 9794 Fabio, P. F., 767¹⁶³ Fabre, J.-L., 44793, 44894, 49381 Fadel, A., 41²²⁷ Fagundo, R., 327171 Fahidy, T. Z., 63650 Fairbrother, F., 29933 Falbe, J., 10176 Falck, J. R., 159464, 161464, 166464, 28863, 685106,107, 771¹⁸⁴, 785^{36,36a} Fallis, A. G., 906142 Falshaw, C. P., 66536 Falter, W., 84435 Fan, C., 222¹⁴⁴ Fan, W.-Q., 19736 Fankhauser, J. E., 104209, 117209 Fanta, P. E., 20922, 499110, 501110, 505110, 509110, 512110 Fanta, W. I., 1153 Fantin, G., 232267, 511186, 514186 Faraday, M., 6331 Farcasiu, D., 330193 Farcasiu, M., 328179 Farina, J. S., 265¹⁹¹ Farina, V., 107226, 109226, 21344, 232264 Farmer, J., 91822 Farnham, W. B., 57262 Farnia, S. M. F., 2959, 334221.221a Farnum, D. G., 888²¹, 890³⁵ Farona, M. F., 30041, 38131, 38231 Farooq, O., 2959, 332204, 334215,221,221a,b, 10461 Farooq, S., 1368, 91619 Farzaliev, V. M., 30469 Faul, D., 555³²

Faulston, D., 334220 Fauq, A. H., 224¹⁷¹, 225¹⁷¹, 264¹⁸² Fauvarque, J.-F., 44358, 450102, 454117.118 Fava, A., 147396, 149413, 151413, 152413, 153396,413, 155³⁹⁶, 865²⁷, 944^{90,91}, 946⁹², 958^{90,112} Favini, G., 38657 Favorskii, A. E., 8391-3, 84327 Favre, H., 35865 Fawzi, M. M., 89668 Fayos, J., 38010 Federici, W., 3804, 73519 Feil, M., 735²⁰ Felberg, J. D., 333²¹⁰ Feldhues, M., 642115 Felkin, H., 243¹², 246³⁸, 250¹², 262¹², 470²²¹, 471²²¹, 482⁵, 499⁵, 505⁵, 509⁵ Fellenberger, K., 976^{5,8} Felletschin, G., 9147, 9247 Fellows, C. A., 10337 Felman, S. W., 35²⁰⁸ Fenton, S. W., 379¹, 380¹ Feoktistov, L. G., 5649, 5679 Ferguson, G., 38132, 38668 Feringa, B., 586¹⁵¹, 665³⁹, 689¹²⁰ Fernandes, J. B., 246³⁸, 446⁸¹ Fernández, I., 8799, 10499 Fernández de la Pradilla, R., 226^{194,194a} Fernández-Simón, J. L., 78850 Ferrari, M., 75297 Ferreira, D., 83161 Ferreira, T. W., 44791, 44897, 456126, 503146, 513146 Ferreira, V. F., 229233 Ferretti, M., 74149 Ferro, M. P., 168491, 169491, 171491 Fesik, S. W., 216⁷⁶ Fétizon, M., 131333, 66428, 69828 Feugeas, C., 124257, 127257 Feustel, M., 28453 Fevig, T. L., 221¹³⁰, 603²²⁹ Fiandanese, V., 230235,236, 44148,49, 44687, 44948,101, 463153,154,155,166, 48529, 49279, 49329, 50329,79, 51329,79 Fichter, F., 63416, 63616 Ficini, J., 19742, 89668 Fidler, F. A., 331198 Fiedler, C., 19414 Field, L., 86², 158^{439,441}, 159⁴⁴¹ Field, S. J., 898⁸⁴ Fielding, H. C., 63984 Fields, D. L., 905136 Fieser, L. F., 82849.51, 89877 Fieser, M., 82849 Filin, V. N., 30466 Filler, R., 757¹²¹ Filley, J., 422 Filliatre, C., 10463 Finch, A. M. T., Jr., 71021 Finch, M. W., 84015 Finch, N., 124²⁷⁰, 127²⁷⁰, 128²⁷⁰, 129²⁷⁰, 629⁵¹ Findlay, J. A., 503149, 512149 Findlay, J. W. A., 66125 Finet, J.-P., 505159 Finger, A., 55²⁸⁰ Fink, H., 81685 Finke, R. G., 208³, 213^{3b} Finkelhor, R. S., 964¹²⁶ Finkelstein, M., 634²¹, 635³⁸, 649^{206,206b}, 655²¹

Finlay, J. D., 159459, 163459 Finley, K. T., 613¹, 614¹, 615¹, 616¹, 619¹, 620¹, 621¹, 622¹, 623¹, 625¹, 626¹, 627¹, 628¹, 629¹, 630¹ Finn, J., 220¹²⁴ Finocchiaro, P., 583118, 587148 Fioshin, M. Y., 635³¹, 636⁵⁵, 639⁸⁰, 648^{172,173,174} Firrell, N. F., 30176 Firth, B. E., 74770, 770176 Fischer, J. W., 62743 Fischer, W. F., Jr., 24855, 25155, 26955, 41947, 49487, 502⁸⁷ Fischli, A., 167485, 168485,498,502,503, 169498,502,503 Fisher, R. P., 19958 Fishpaugh, J. R., 24150, 21128, 21528 Fitcher, F., 66850 Fitt, J. J., 6719 Fittig, R., 563², 721² Fitzi, K., 35346 Fitzi, R., 41229 Fitzjohn, S., 81043 Fitzner, J. N., 104²⁰⁹, 117²⁰⁹ Fitzpatrick, J. M., 25070 Flanagan, P. W. K., 322141 Flasch, G. W., Jr., 63891, 64491 Flatow, A., 87890 Flechtner, T., 105840 Fleischhauer, I., 99189, 107189, 110189 Fleischmann, C., 93976 Fleming, G. H., 4157 Fleming, I., 1792, 1892-94, 23137, 25154, 27154,167, 28154. 42⁹⁴, 198⁴³, 200⁶⁹, 345²², 746⁶⁸ Fleming, M. P., 579¹²⁵, 582¹²⁵, 583^{124,125,127}, 584¹²⁵, 585^{124,125}, 587^{124,144}, 588^{125,127a}, 595¹²⁵, 596¹²⁵, 610^{125,127a} Flitsch, W., 62335 Flood, S. H., 30571 Flood, T. C., 583¹¹⁹, 760¹⁴⁰ Flores R, H., 901112 Florio, S., 461145 Floyd, C. D., 20069 Floyd, D. M., 21787, 25070,73 Fludzinski, P., 28454 Foa, M., 1018¹¹, 1026³⁷, 1033⁷², 1039¹⁰⁰ Fochi, R., 125³⁰⁷ Fogagnolo, M., 232267, 511186,187, 514186 Folkers, E. A., 1474.75, 1574.75 Folli, U., 95154, 119154 Folting, K., 583119 Fones, W. S., 88817 Fong, W. C., 85475 Fonken, G. J., 329¹⁸⁹ Font-Altaba, M., 3807 Fooks, A. G., 82955 Forbus, T. R., 328179 Forrester, A. R., 689122 Forrester, J., 399¹¹⁸, 404¹³³ Forster, D., 10189 Forster, M. O., 89031 Forsyth, C. J., 232²⁶⁶ Fort, R. C., Jr., 334²¹⁸ Fort, Y., 509179 Fossel, E. T., 89668 Foster, J. P., 242⁴, 244⁴ Fougerousse, A., 74044 Foulon, J. P., 482⁴, 485⁴ Fouquet, G., 21562, 24420, 466184

Fourneron, J.-D., 34213, 35142, 35242 Fournier, A., 3805 Fournier, F., 56848 Fournier, M., 56848 Fowler, K. W., 219114, 499140, 501140, 502140 Foxman, B. M., 219¹⁰² Fracheboud, M. G., 74359 Frajerman, C., 24638, 470221, 471221 Francalanci, F., 103372 Franck, B., 66746, 687112.113 Francke, W., 223155 Francolanci, F., 1039100 Franich, R. A., 753105 Frank, G. A., 85166 Frankel, J. J., 757¹²⁴ Franklin, R. C., 82315 Frantzen, V., 88820, 88920. 89420 Franzen, G. R., 3908t, 3928t Franzen, V., 55313 Fraser, R. R., 31¹⁸⁶, 32¹⁸⁸, 66^{9,10}, 149⁴⁰⁰ Fraser-Reid, B., 84854, 84954b Fráter, G., 41228, 43239 Fredenhagen, H., 822², 831² Freeman, P. K., 906143 Freeman, R. C., 84323 Frei, B., 133335, 136335 Freiberg, L. A., 905140 Freidinger, R. M., 882¹⁰², 883¹⁰² Freidlin, G. N., 64098,101 Freidlina, R. Kh., 42269 Freiser, H., 21130 Frejd, T., 260141,142, 261142, 466180,181, 763149, 768149,166 Frenette, R., 771185, 105636 Freudenberg, K., 693^{143,147} Frey, A. J., 346²³ Friary, R. J., 55283, 78114 Fridkin, M., 30251 Fridman, A. L., 88710, 88810, 88910, 89010, 89310, 89710. 900¹⁰, 903¹⁰ Fried, J., 27938, 57261 Fried, J. H., 946, 1046 Friedel, C., 29930, 317121 Friedländer, P., 82842 Friedman, H. M., 30463 Friedman, L., 244²¹, 247²¹, 464¹⁷³ Friedrich, K., 271², 272² Fries, K., 80942 Friesen, R. W., 488^{52,53}, 495^{52,53} Frisbee, R., 460¹⁴², 497¹⁰⁵ Fritz, G., 224¹⁷², 312¹⁰¹ Fritz, H., 38132, 62131 Fritzberg, A. R., 89252 Fröling, A., 121245, 123245,246, 125246 Fröling, M., 121245, 123245 Fronczek, F. R., 591170 Fry, A., 723^{8,9}, 731^{8,9} Fry, A. J., 87046 Frydrych-Houge, C. S. V., 594¹⁸⁷ Frye, R. L., 649²⁰⁷ Fu, J.-M., 231250 Fuchikami, T., 102745 Fuchs, C. F., 2085 Fuchs, P. L., 125300, 128300, 129300, 133300, 140380. 154³⁸⁰, 168³⁸⁰, 174³⁸⁰, 176³⁸⁰, 209¹⁶, 223¹⁶ Fuchs, W., 643¹²⁴ Fueno, T., 86736

Fuentes, L. M., 6718, 7547,48 Fugami, K., 44574 Fugger, J., 89138 Fuhrhop, J.-H., 86629 Fuji, K., 135^{341,342,343}, 136^{341,342,343}, 137^{341,342}, 1050¹⁹ Fujihara, H., 510¹⁸² Fujii, T., 652²²¹ Fujiki, M., 53063, 53263 Fujikura, S., 24429 Fujimori, M., 52847 Fujimoto, K., 100561-63 Fujimoto, N., 56522 Fujimoto, T. T., 117235.236, 155235.236, 156235.236, 94687 Fujimoto, Y., 85689 Fujimura, T., 455¹²⁵, 460¹²⁵ Fujinami, T., 57056 Fujioka, A., 437²⁵, 440²⁵, 448²⁵, 449²⁵, 450²⁵, 451²⁵. 45225, 48426, 49226, 49426, 49526, 50326, 51326 Fujioka, H., 466186, 75188 Fujioka, Y., 503¹⁴² Fujisaki, S., 321¹³⁹ Fujisawa, T., 124259, 125259, 153416, 154416, 218100, 220¹²⁵, 227^{210,211,212}, 243¹⁶, 245³⁰⁻³², 246³⁹, 257³⁹, 44678, 463158, 470212,213, 476212,213 Fujise, Y., 39079, 100872, 100972, 101072 Fujita, A., 124²⁸⁰, 125²⁸⁰, 126²⁸⁰ Fujita, E., 125³⁰⁹, 135^{341,342,343}, 136^{341,342,343}, 137^{341,342}. 168⁴⁹², 169⁴⁹², 217⁷⁹, 1050¹⁹ Fujita, H., 21795, 579101 Fujita, K., 38124, 38224, 72625 Fujita, M., 102745 Fujita, S., 593¹⁸² Fujita, T., 402¹³⁰, 404¹³² Fujita, Y., 244²⁵, 267²⁵, 428⁹⁰, 470²⁰⁹, 472²⁰⁹, 475²⁰⁹, 494⁸⁴, 992³⁷, 994⁴¹ Fujiwa, T., 445⁷¹ Fujiwara, S., 470²⁰⁸, 471²⁰⁸, 475²⁰⁸ Fujiwara, T., 137³⁷⁶ Fujiwara, Y., 29511, 30211, 56736,190, 583120, 595190, 607¹⁹⁰, 610³⁶ Fukata, G., 329187 Fukazawa, Y., 39079 Fukuda, Y., 27937 Fukui, K., 556³⁶ Fukumoto, K., 164⁴⁸⁰, 165⁴⁸⁰, 167⁴⁸², 168⁴⁸², 660¹⁷. 67786, 67917 Fukushi, H., 21672 Fukushima, M., 229²²⁶ Fukuzawa, S., 381²⁶, 382²⁶, 570⁵⁶ Fukuzumi, K., 99190 Fullerton, T. J., 753105 Funayama, M., 2958 Funfschilling, P. C., 1786 Fung, V. A., 36289 Funhoff, D. J. H., 92750 Funk, R. L., 58292 Fürstner, A., 57055.129, 58255, 58355.129, 63057, 63157 Furtek, B. L., 901110 Furukawa, M., 83269 Furukawa, N., 510182 Furukawa, S., 135^{360,361}, 136^{360,361}, 137^{360,361}, 139^{360,361}. 142360.361, 143360.361, 155429, 156360.361 Furusaki, A., 400119, 404135 Furusato, M., 96168, 104168, 108168, 117168 Furusawa, F., 168494,504, 169494,504, 170494,504 Furuta, K., 44688

Fuse, Y., 222¹⁴³ Fuson, R. C., 499111, 503149, 512149, 5631, 89032 Fuzesi, L., 97811 Gabel, R. A., 261156, 512202 Gabmeier, J., 3807 Gabriel, J., 8775, 98730 Gadallah, F. F., 507170 Gadwood, R. C., 49264, 87114, 19415, 19615, 78745 Gaginkina, E. G., 63531 Gagnier, R. P., 419^{29,30}, 422^{29,30} Gagosian, R. B., 89252 Gaillot, J.-M., 866³⁰ Gainor, J. A., 511¹⁹¹ Gains, L. H., 736²⁸ Gais, H.-J., 159466, 166466, 174530, 230237 Galamb, V., 55956 Gale, A., 81679 Galindo, A., 391⁹⁰, 393⁹⁰, 395⁹⁷ Gall, M., 4¹⁸, 7¹⁸, 8¹⁸, 11¹⁸ Gallagher, T., 67370, 67470b Galle, J. E., 165479, 173479,519, 19850, 759134 Galli, C., 55²⁸⁴ Gallina, C., 21345, 220118, 222118 Gallos, J. K., 28247 Gallucci, R. R., 89792, 90092 Galuszko, K., 92748 Galy, J. P., 124257, 127257 Galynker, I., 41226 Gamasa, M. P., 824¹⁹ Gammill, R. B., 24146, 969133 Gander-Coquoz, M., 40221, 41227 Gandour, R. D., 591170 Ganem, B., 22¹³², 224¹⁶⁹, 262¹⁶⁹, 1055³² Ganguly, A. K., 25389, 261147, 26289, 74150 Ganguly, R., 94693 Gans, R., 927⁵⁶ Ganter, C., 39083, 39283 Ganz, C. R., 380¹⁶ Gao, Y., 223156 Gapinski, R. E., 103²⁰⁴, 108²⁰⁴ Garbacik, T., 30173 Garbesi, A., 179547, 181547 Garbisch, E. W., 85163 García, B., 8799, 10499 García, G. A., 19849 Garcia-Raso, A., 82524,24b, 83524 Gardano, A., 102637, 103372 Gardette, M., 24961, 48535, 48635, 49435 Gardlik, J. M., 593¹⁷⁷ Gardner, J. O., 48260 Garigipati, R. S., 511191 Garin, D. L., 767¹⁶¹ Garland, R., 42231 Garmaise, D. L., 781¹⁹ Garst, M. E., 54²⁷⁶, 370¹¹², 537⁸⁹ Gartenmann, T. C. C., 583118 Gärtner, J., 38344 Garwood, R. F., 635³³, 640^{107,107a}, 647^{33,107} Gasc, J. C., 21131 Gassman, P. G., 334²¹⁸, 614⁵, 649²⁰⁶, 876⁷⁸⁻⁸⁰, 890³¹, 901¹¹², 903⁹⁹, 905¹³⁸, 969^{131,132} Gates, J. W., Jr., 74772 Gatilov, Y. V., 38668 Gatterman, L., 582112 Gaudenzi, M. L., 21791, 539103

Gaudino, J., 511^{188,212}, 515^{188,212} Gaul, M. D., 106146, 106246 Gautschi, F., 27313 Gavai, A. V., 220¹²⁶ Gawley, R. E., 35²⁰⁵, 66¹⁶, 67¹⁶, 74^{16,42}, 75^{42-44,51}. 76^{43,53}, 77^{42,57}, 78⁵¹, 80^{43,65}, 81⁶⁵, 685¹⁰⁸ Gaylord, N. G., 262159, 264159 Gebhart, H. J., Jr., 7215 Gebreyes, K., 87894, 87994,97, 88094, 88194 Gedymin, V. V., 89881 Geetha, K. V., 3807 Geise, H. J., 583¹²², 584¹³⁰, 587¹⁴⁸ Geiss, F., 89035 Geiss, K.-H., 86¹⁸, 94¹⁸, 95^{18,158,178}, 96^{158,178}, 97^{158,178}. 99158,178, 12118,158, 144384, 145384, 147384 Geissler, G., 825^{24,248}, 828²⁴⁸, 835²⁴ Geissman, T. A., 754¹⁰⁷, 757¹⁰⁷ Geiszler, A. O., 81364 Gelas-Mialhe, Y., 866³⁰ Gelin, R., 318¹²⁷ Gelin, S., 318¹²⁷ Gellert, E., 507171 Gellert, H. G., 1945 Gemal, A. L., 24965 Gemenden, C. W., 124²⁷⁰, 127²⁷⁰, 128²⁷⁰, 129²⁷⁰ Gendreau, Y., 470225, 473225 Genet, J.-P., 47256 George, C. F., 901116 Georgian, V., 83477, 89791, 90091 Gerdes, H. M., 136373, 137373 Gerkin, R. M., 760^{136,138,141}, 761¹⁴¹, 762¹⁴¹, 763^{138,141}. 764141 Gerlach, H., 286⁶⁰, 747⁷¹ Germain, G., 625⁴¹ German, L. S., 647¹⁹⁶ Germon, C., 258125 Gero, S. D., 126315,316,317 Gerstmans, A., 322144 Gervay, J. E., 68196 Geschwend, H. W., 124²⁷¹, 125²⁷¹, 127²⁷¹, 128²⁷¹, 129²⁷¹ Gesing, E. R. F., 583¹¹⁸ Getmanskaya, Z. I., 643¹³¹ Geyer, E., 890³⁴ Ghadiri, M. R., 27425 Gharbi-Benarous, J., 72836 Ghatak, U. R., 87¹¹⁵, 898⁸⁶, 908¹⁴⁶ Ghawgi, A. B., 325¹⁵⁸ Gheorghiu, M. D., 331196 Ghiringhelli, D., 57584 Ghosal, S., 48315, 50015 Ghose, B. N., 505¹⁶⁸ Ghosez, L., 162487, 168487, 78642, 89671 Ghosh, A. K., 546¹²³, 783²⁸ Ghosh, S., 87¹¹⁵ Ghribi, A., 226²⁰⁷, 259¹²⁹ Giacomelli, G., 28555, 48312 Giamalva, D. H., 104916, 105018, 105316, 106018 Giannis, A., 25160 Gibby, M. G., 299³³ Gibson, C. P., 20912 Gibson, T. W., 85061 Giddings, P. J., 93464, 95364 Gielen, M., 587¹⁴⁸ Giersch, W., 73629 Giese, B., 598²⁰³, 649²⁰¹ Giese, R. W., 57787

Gigg, J., 273⁷ Gigg, R., 273⁷ Gilbert, A., 80718 Gilbert, J. C., 1049¹⁶, 1050¹⁸, 1053¹⁶, 1060¹⁸ Gilbert, R., 903126 Gilchrist, T. L., 486³⁹, 491³⁹, 495³⁹, 498³⁹, 503³⁹, 908¹⁴⁸ Gilde, H. G., 647^{191,192} Gill, M., 24748 Gillespie, J. P., 683101 Gillespie, R. J., 29723 Gillet, J. P., 498107 Gilman, H., 2088.9, 24419, 4159, 511190, 56517 Ginsburg, D., 62845 Giolando, D. M., 21128, 21528 Giordano, C., 7786, 7886.54, 7896.55-57 Gipson, R. M., 39189, 39389 Girard, J.-P., 85165 Girard, Y., 277²⁷ Girgenti, S. J., 1475, 1575 Giri, B. P., 57999, 104812 Girina, G. P., 63655 Giusti, I., 21678 Gizycki, U. V., 66427 Glamkowski, E. J., 890³¹, 903¹²⁴ Glans, J. H., 49380 Glass, C., 691129, 693129 Glass, R. S., 73628 Glasstone, S., 63646 Gleason, J. G., 27315, 27415 Gleason, R. W., 3805 Gledinning, R. A., 334²²⁰ Gleiter, R., 382³⁶, 587¹⁴⁸, 592¹⁷⁵, 614⁶, 623⁶, 627⁶ Glinka, T., 79061 Glinski, M. B., 266¹⁹⁴ Glos, M., 904131 Glowinski, R., 500¹³⁴, 509¹³⁴ Godefroi, E. F., 158⁴⁴⁵, 159⁴⁴⁵, 367¹⁰² Godel, T., 46²⁵² Godfrey, A., 19627 Godfrey, C. R. A., 613², 615² Godschalx, J. P., 232²⁷¹, 469²⁰¹, 470²⁰¹, 471²⁰¹, 473²⁰¹. 475201 Godtfredsen, S., 35245 Goel, A. B., 824²⁰ Goering, H. L., 220¹²¹, 222^{121c,138,138b,142}, 393^{93,94}, 394⁹⁵ Gohdes, J. W., 328¹⁷⁹ Goheen, D. W., 848⁵¹ Goicoechea-Pappas, M., 7543,51, 7643, 7851, 8043 Goins, D. E., 29717, 30686 Gokel, G. W., 136³⁷³, 137³⁷³, 505¹⁶³ Gold, A. M., 81255 Gold, P. M., 87¹⁰³, 104¹⁰³, 106¹⁰³, 111¹⁰³, 117¹⁰³ Goldberg, S. I., 154419 Gol'dfarb, Ya. L., 6157 Gol'ding, I. R., 20920 Goldman, N. L., 28, 118, 168, 178, 268 Gol'dovskii, A. E., 30572 Goldschmidt, S., 647¹⁷⁶, 648¹⁷⁶ Goldsmith, D. J., 75295 Golfier, M., 66428, 69828 Golinski, J., 174⁵²⁸ Gololobov, Yu. G., 8643, 17943 Gomberg, M., 505¹⁶¹, 507¹⁶¹, 564¹² González, A. G., 391⁹⁰, 393⁹⁰, 395⁹⁷ Gonzalez, M. A., 71^{36,37} González, M. S., 396115

Goodburn, T. G., 28143 Goodfellow, C. L., 19737 Goodwin, S., 804⁸, 810⁴⁶ Goodwin, T. E., 97170, 117170 Gopal, M., 55956 Gopalan, A. S., 133335, 136335 Goralski, C. T., 164474, 180474, 181474 Gorbunkova, V. P., 644141 Gordon, A. B., 9131 Gordon, K. M., 8763, 11463, 11763 Gore, J., 223¹⁸³, 225¹⁸³, 256¹¹³ Gore, P. H., 499¹¹⁵ Gorham, W. F., 78216 Gorlier, J.-P., 41945, 42045 Gorzynski, J. D., 48317 Gosselin, P., 124260 Gosteli, J., 78114 Goto, F., 63544 Goto, H., 44677 Goto, T., 261157, 42682, 42982 Gotoh, Y., 446⁷⁸ Goulet, M. T., 97914 Gourley, R. N., 57478 Govindachari, T. R., 396115, 74150 Gowal, H., 822^{3,12}, 831^{3,12}, 835^{12b} Gowda, D. S. S., 386⁶¹, 393⁶¹ Gowland, B. D., 159460, 161460, 163460, 167460, 265193 Gozlan, A., 640¹⁰¹ Graefe, J., 380⁶, 386⁵⁹ Graf, B., 144³⁸⁴, 145³⁸⁴, 147³⁸⁴ Gräfing, R., 87120,213, 88120, 105120,213,221 Graham, E. S., 85687 Graham, S. H., 38346,50, 73519 Gramstad, T., 29719 Grand, P. S., 1477, 1577 Granger, R., 85165 Granja, J. R., 98319 Gränse, S., 634⁷, 636^{7,7b}, 637⁷ Gras, J.-L., 57269, 57369, 60269, 60769, 61069 Gravestock, M. B., 36493, 369118, 372118 Grayshan, R., 125^{295,296,301}, 126^{295,296,301}, 128^{296,301}. 129301, 133301 Grayson, D. H., 71019 Grayson, J. I., 123251 Grayson, M., 30035 Greaves, W. S., 643125 Greck, C., 174527, 175527c, 258127 Green, B. S., 382³⁶ Green, E. E., 649²⁰⁷ Green, G. E., 735¹⁸ Greenburg, R. B., 84218 Greene, A. E., 222¹⁴¹, 783^{26,29} Greene, J. L., 723¹⁰ Greenlee, K. W., 2735 Greenwood, J. M., 399117, 402117, 600215 Gregory, C. D., 211³¹, 213³¹ Gregson, M., 35764 Greiber, D., 903128 Greuter, H., 84850 Gribble, G. W., 261155, 262170, 31198, 644154 Grice, P., 174527, 175527c Grieco, P. A., 1787, 20113, 101506, 155431, 159462, 161462, 168⁵⁰⁶, 169⁵⁰⁶, 170⁵⁰⁶, 212³⁹, 220¹²⁴, 253⁹², 882¹⁰⁵, 89463, 94382, 964126, 105121 Griessmaver, V., 660² Griessmeier, H., 86419

Author Index

Griffin, C. E., 74876 Griffin, G. W., 57267 Griffin, J. H., 229224,224b, 92034, 92334b, 93434, 100869 Grigg, R., 484²¹, 770¹⁸⁰ Grignard, V., 24315 Griller, D., 6926 Grillot, G. F., 258124 Grimm, K., 595¹⁹¹ Grimshaw, J., 56841, 57478, 57787, 67783.84 Grinberg, V. A., 647177,196, 648177,178, 649177 Grindley, T. B., 6610 Grisar, J. M., 38016, 73517 Gröbel, B.-T., 86²⁰, 87^{82,88}, 104²⁰, 105⁸², 112⁸², 120^{20,82} 121²⁰, 125²⁰, 126²⁰, 128³²², 130³²², 131^{324,331}, 134³²², 135³²⁴ Groen, M. B., 36176, 36776,100,103 Groenewegen, P., 20¹²¹, 25¹²¹ Groh, B. L., 232²⁶⁵ Gromova, G. A., 63980 Gronowitz, S., 232259, 511188, 515188 Grosjean, F., 579¹⁰⁰ Gross, A. W., 39, 89140, 90040 Gross, S., 56²⁸⁵ Grossa, M., 81790,92 Grosse, A. V., 331197 Grossert, J. S., 86528 Grossman, N., 8044 Grotemeier, G., 45²⁵¹ Groth, U., 53²⁷⁴ Grove, J. F., 736²⁵ Groweiss, A., 38013, 395102, 396102, 407146 Gruetzmacher, G., 969131,132 Grugel, C., 57159 Grundkötter, M., 904131 Grütter, H., 34520 Grüttner, S., 30353 Grützmacher, H.-Fr., 587¹⁴⁷ Grzejszczak, S., 953105 Gschwend, H. W., 6719, 1932, 1942, 261146, 264146 Guaciaro, M. A., 26165 Guerin, P. M., 223155 Guerrieri, F., 102743 Guest, I. G., 74153 Guha, P. C., 89464 Guigina, N. I., 648175 Guiles, J., 7964 Guillerm, D., 21794 Guin, H. W., 39189, 39389 Guindon, Y., 227²⁰⁸, 289⁶⁷, 771¹⁸⁵ Guir, C., 226²⁰⁷ Guise, G. B., 83159 Guittet, E., 147395, 155395 Gunawardana, D., 7128 Gunn, P. A., 71537 Gunning, H. E., 89248 Gunstone, F. D., 752% Günther, H. J., 39082, 39282 Günther, P., 87261 Günther, W., 41228, 43239 Gunzinger, J., 81362 Gupta, A. K., 79794 Gupta, A. R., 30464 Gupta, A. S., 402127 Gupta, P., 66125 Gupta, V. K., 509179 Gupton, J. T., 24644

Gurjar, V. G., 63542 Gurzoni, F., 78854 Gustavson, L. M., 1000⁵⁴ Gut, M., 427⁸⁷ Gut, S. A., 34198, 39198 Gutiérrez, A., 39190, 39390 Gutman, A. L., 62845 Gutmann, V., 29933 Gutowski, F. D., 422⁷¹, 423⁷¹ Gutsche, C. D., 892⁵¹, 896⁶⁸, 900⁹⁶, 903⁹⁶ Gutzwiller, J., 736²⁵ Guy, J. T., Jr., 3807 Guyon, R., 771¹⁹² Guziec, F. S., Jr., 8623 Gverdtsiteli, D. D., 318128 Ha, T.-K., 41²³⁰ Haag, W. O., 330¹⁹⁴ Habermas, K. L., 21127 Habig, K., 904¹³⁵ Haces, A., 223^{147,148} Hackler, R. E., 916^{17,18}, 932¹⁷, 963¹¹⁹ Hackney, M. A., 509179 Haddon, V. R., 34725 Hadfield, J. R., 3807 Hadwick, T., 72311, 72511, 72611 Haelters, J.-P., 105227 Haemmerlé, B., 640103 Haenel, M. W., 87890 Hagen, J. P., 865²⁶, 957^{109,110} Hager, R. B., 57267 Hagihara, N., 21789, 21989, 2711, 5217, 5307,61, 53161. 532^{7,61}, 537⁶¹, 552⁸, 554²⁶, 557^{48,49} Hagiwara, I., 12⁶² Hahnvajanawong, V., 154424 Haidukewych, D., 883¹⁰⁸ Hajos, Z. G., 23142 Hakoshima, T., 751⁸⁸ Halazy, S., 86⁵⁰, 87^{70,71,78,117}, 88¹²², 89^{70,71,122}, 90^{70,71} 9171.122, 9270.71.122, 93122, 11670.71, 11970.71, 13678.122. 14478, 14578, 122, 766156, 78644 Halczenko, W., 258126, 513204 Haley, G. J., 592174,175, 594174 Haley, R. C., 24635 Halim, H., 623³⁶ Hall, T. W., 20106, 26165 Hallberg, A., 49595 Hallcher, R. C., 66644, 68744 Hallenga, K., 587148 Halpern, Y., 29934b, 322142a, 332206, 33334 Hamachi, I., 43830 Hamada, K., 168495, 169495 Hamamura, K., 153415 Hamanaka, E., 43199,100 Hamböck, H., 82641 Hamel, N., 102850 Hamersak, Z., 74667 Hamilton, G. A., 66020, 66120, 69920 Hamilton, H., 595192 Hamilton, R., 334220, 67783 Hammann, W. C., 4158 Hammen, R. F., 953¹⁰² Hammerschmidt, F.-J., 34318, 35351, 35451 Hammond, W. B., 89252 Hammoud, A., 48747, 53068, 53368 Hamon, L., 21669, 41945, 42045

Han, L. P.-B., 82956 Hanack, M., 38118, 396103,104 Hanamoto, T., 225187, 100055, 100460 Hanazaki, Y., 56626 Handa, S., 219¹¹³, 503¹⁴⁴, 505¹⁶⁷ Handa, Y., 56628, 57128, 57828 Hanessian, S., 512200, 7346 Hanisch, G., 753103 Hanna, J., 565²³, 570²³, 583²³ Hanna, J. M., Jr., 513²⁰⁸ Hannebaum, H., 63428 Hannon, F. J., 21237.38 Hänsel, R., 691134 Hansen, H.-J., 80939,40, 87469 Hansen, J. F., 507¹⁷⁴, 677⁸⁵ Hanson, G. J., 28145, 36072 Hanson, J. R., 38013, 71538,41,42, 73835 Hanson, R. M., 223156 Hansson, A.-T., 257122 Hanzawa, Y., 593¹⁷⁷ Happe, W., 9751 Hara, H., 67265 Hara, K., 135345, 174345 Hara, S., 231²⁴⁸, 251⁷⁸, 254⁷⁸, 262¹⁵⁸, 443⁵⁶, 490⁶⁶, 51166, 51566, 52325 Hara, T., 501136 Harada, M., 42153 Harada, S., 469²¹⁶, 470²¹⁶, 476²¹⁶ Harada, T., 10476 Harayama, T., 21672 Harbert, C. A., 36385, 369117, 372117 Harding, C. E., 396103 Harding, K. E., 3414, 34317, 34827, 34940, 35140. 353^{46,48,50}, 354^{4c,50}, 355^{17,48}, 357^{17,48}, 358⁶⁸, 369⁴⁰, 788⁵³ Hardy, R., 2426, 2576, 2596 Haremsa, S., 87468 Hargrove, R. J., 3804 Harirchian, B., 105215 Harley-Mason, J., 80721 Harmon, C. A., 38010 Harnisch, J., 62541 Harre, M., 42231 Harrington, J. K., 164476 Harrington, P. J., 498108 Harris, B. D., 27831, 28931 Harris, C. M., 58289 Harris, S. H., 1040105 Harris, T. D., 255107 Harris, T. M., 58288,289 Harrison, C. R., 27423, 79999 Harrison, M. J., 753100 Harrison, W., 21788 Hart, D. J., 598²⁰³ Hart, G. C., 7442, 7542,43,51, 7643, 7742, 7851, 8043 Hart, H., 73834, 753101, 8398 Hart, T. W., 767160 Hartley, F. R., 228²¹⁷, 436^{7,10} Hartman, B. C., 263177 Hartman, G. D., 258126, 513204, 86314, 86514, 88414 Hartman, M. E., 783105 Hartman, R. D., 86314, 86514, 88414 Hartmann, A. A., 124²⁷³, 128^{273,323}, 132²⁷³. 134273,323 Hartmann, J., 99182, 101182 Hartmann, R., 1785

Hartshorn, M. P., 74150,52,53, 74255,57, 74357,60, 74666, 75190, 75291 Hartwell, J. L., 82849 Hartwig, W., 53274 Harvey, S. M., 325161 Hasan, S. K., 154418, 155418 Hase, T. A., 86³⁰, 121³⁰ Hashiba, N., 404135 Hashimoto, H., 124²⁶³, 126²⁶³, 219¹¹³, 243¹⁸, 244¹⁸, 24718, 400121, 464169, 500133, 503144, 505167 Hashimoto, I., 30461, 4838 Hashimoto, K., 168⁴⁹⁷, 170⁴⁹⁷ Hashimoto, M., 100199 Hashimoto, S., 36²¹¹, 246³⁵, 349³³, 354⁶⁰, 358³³ Hashizume, A., 19741, 19941 Hashizume, K., 136³⁷⁰, 137³⁷⁰, 138³⁷⁰, 139³⁷⁰, 140³⁷⁰ Haslanger, M. F., 28863 Haslinger, E., 3807 Hassall, C. H., 66015, 690124, 81894 Hassen, W. D., 497¹⁰¹, 505¹⁰¹ Hässig, R., 87⁷⁵ Hassner, A., 901112 Haszeldine, R. N., 29719 Hata, T., 19741, 19941 Hata, Y., 902¹⁰³ Hatakeyama, S., 27830 Hatanaka, Y., 233273.274, 53895, 53997, 56734, 57034 Hatch, L. F., 30459 Hatch, M. J., 91821 Hatcher, A. S., 106224, 113224 Hatfield, G. L., 602226 Hathaway, B. J., 499118 Hatsuya, S., 44362 Hattori, K., 78970, 10476 Haufe, G., 3793, 3806,12,13, 38128, 38228,38, 38659, 84956 Haufe, J., 63425b, 63525b, 63925b, 64097 Haugen, R. D., 589¹⁶², 610¹⁶² Haung, W., 596¹⁹⁴ Hauptmann, S., 890³³, 894⁵⁹, 900⁵⁹ Hauser, C. R., 58²⁸⁸, 158⁴³⁸, 159^{438,467}, 160⁴³⁸, 166^{438,467}, 914^{8,12}, 915^{14,15}, 918²¹, 957¹⁰⁸, 967¹⁴, 969¹⁰⁸, 975², 976², 979², 980² Havinga, E. E., 552⁹, 557⁹ Havlin, R., 21790, 21990 Hawkes, G. E., 63540, 63765 Hawkins, E. G. E., 770177 Hawks, G. H., III, 54275 Hawley, D. M., 38668 Haworth, R. D., 693140, 81898 Hay, A. S., 552⁷, 557^{44,45} Hay, D. R., 67^{17,20} Hayakawa, K., 80941 Hayama, N., 463¹⁶³, 483¹⁹, 484^{20,24}, 500^{19,131}, 501²⁴, 509¹⁹ Hayama, T., 45²⁴⁸ Hayano, K., 38239, 402126, 404135, 405138 Hayasaka, K., 404137 Hayashi, J., 23143, 24143 Hayashi, K., 62339 Hayashi, M., 96169, 104169, 108169, 117169 Hayashi, N., 89145 Hayashi, S., 832⁶⁹ Hayashi, T., 17⁹⁵, 95^{156,157}, 97^{156,172,173,177}, 108¹⁷⁷, 109¹⁷⁷, 114¹⁵⁷, 116^{157,172,173}, 136³⁶⁹, 138³⁶⁹, 228219.220.221, 229225.226.227.228.229.231, 24637, 43619 43830.33, 44147, 44250, 44569.71, 45033, 45233, 455122,

457129, 470219, 471219, 472219, 49276.78, 49578, 10163, 1039101 Hayashi, Y., 36178, 36597, 469202, 470202, 473202, 92237. 92437 Hayashizaki, K., 229228.229 Hayes, R. A., 89143, 89243 Haynes, L., 38013 Haynes, R. K., 8794 Hays, H. R., 654 Hazato, A., 222144 Hazra, B. G., 67675 He, Y.-B., 455¹²⁴, 525⁴² Head, D. B., 220¹²², 224¹⁷⁵, 416^{12,13}, 417¹³, 752⁹⁵ Heap, N., 73518 Heath, R. R., 45²⁴⁵ Heathcock, C. H., 8³⁶, 20¹⁰⁷, 21¹²⁹, 39²²⁰, 178⁵⁴², 179⁵⁴², 225¹⁸⁷, 252⁸¹, 257⁸¹, 273³, 280⁴⁰, 344¹⁹ Hebert, E., 216⁷⁷ Hecht, S. S., 3808 Heck, F. R., 271¹, 530⁶⁰ Heck, G., 887^8 , 888^8 , 893^8 , 897^8 , 900^8 , $903^{8,128}$ Heck, R. F., 228^{214} , 436^{15} , 470^{220} , 471^{220} , 521^8 , 524^8 , 539^{98} , 1021^{13} , 1028^{47} , 1029^{54} , 1032^{68} , 1034^{78} , 1037^{89} Hecker, E., 810^{49} , 812^{52} Hedaya, E., 62333 Heeren, J. K., 760¹⁴⁰ Heerze, L. D., 589¹⁶², 610¹⁶² Hegedus, L. S., 208³, 213^{3b}, 423⁷⁵, 426⁸², 429⁸², 485³⁸, 486³⁸, 491³⁸, 492³⁸, 495³⁸, 498¹⁰⁸, 503³⁸, 1024³⁰, 1037⁸⁷, 1038⁹¹ Heggie, W., 386^{64,65} Heider, K., 324151 Heiligenmann, G., 45²⁵¹ Heimbach, H., 25158,160 Heimbach, P., 39084, 39284 Heinzer, J., 7546 Heiss, H., 904132 Heldrich, F. J., 464174 Hellmann, H., 93362 Hellring, S., 71³⁴, 72³⁸ Helmchen, G., 45²⁵¹ Helmlinger, D., 38344 Helquist, P. M., 135³⁶⁷, 137³⁷⁷, 243¹⁷, 249¹⁷, 251⁷⁷, 254⁷⁷, 263^{17,172}, 423⁸⁰, 430⁹³, 483¹⁷, 499¹²⁸, 500128,132, 505132, 509128 Helson, H. E., 53891,93 Hembre, R. T., 102639 Hemetsberger, H., 80410 Henbest, H. B., 734⁴, 741⁵³ Hendley, E. C., 822¹⁰, 825²⁸, 829¹⁰ Hendrick, G. W., 771187 Hendrickson, J. B., 158436, 159436, 174538,539, 177538,539, 831⁶³, 868⁴²⁻⁴⁴, 869^{43,44}, 876⁴⁴, 1048¹⁰ Henne, A. L., 2735 Hennessy, M. J., 1055³¹, 1062^{31,47} Henning, R., 85168 Hennion, G. F., 2739 Henrick, C. A., 220^{121} , 224^{163} , 257^{120} , 416^{21} , 419^{43} Henrot, S., 41^{225} , 263^{175} Henry, P. M., 38132 Henseke, G., 753103 Herbert, R. B., 67061, 68194, 80731 Hergrueter, C. A., 25177, 25477 Herissey, H., 693145 Herlihy, K. P., 725²¹, 726²¹

Hermann, R. B., 905140

Hermeling, D., 644¹⁴², 651²¹⁷ Hernandez, O., 135368, 136368, 137368 Hernot, D., 105227 Herr, M. F., 73943 Herr, R. R., 2426, 2576, 2596 Herr, R. W., 223¹⁵⁸, 226¹⁹², 263¹⁷⁷, 265¹⁸⁷ Herrin, T. R., 369117, 372117 Herrinton, P. M., 224^{172,172b}, 325¹⁶², 416¹⁹, 792⁶⁸ Herrmann, J. L., 43²³⁸, 135³⁵⁸, 136³⁵⁸, 137³⁵⁸, 139³⁵⁸, 142358, 143358 Hershberg, E. B., 242³, 244³ Hershberger, P. M., 328179 Hershberger, S. S., 497¹⁰⁰, 498¹⁰⁰ Hershenson, F. M., 750⁸⁴, 762⁸⁴ Hertenstein, U., 19740 Herunsalee, K., 154425, 155425 Hervé du Penhoat, C., 174536,537, 176536,537, 178536,537. 448⁹⁵ Herweh, J. E., 765152 Hess, B. A., Jr., 901105 Hess, T. C., 89143, 89243 Hesse, G., 85162 Hessler, E. J., 3426 Heuschmann, M., 87045 Hevesi, L., 8650,55, 8781,84,86,87,90, 89142,143, 92143, 9455 10481, 10984, 116142, 118143,238, 12090,244, 123143,238, 14286.244, 14486 Hewgill, F. R., 66951 Hey, D. H., 505¹⁶², 507¹⁶², 512¹⁶², 639⁸⁸, 807²⁷, 813⁶¹ Hey, J. P., 58291 Heydt, H., 909¹⁵² Heyn, M., 822⁵, 834⁵ Hickling, A., 6349, 63646, 6579 Hickmott, P. W., 28¹⁷¹, 30^{171,176,178}, 35¹⁷¹, 39¹⁷¹ Hidaka, T., 52847 Hiegel, G. A., 653226 Hiemstra, H., 223183, 225183, 83267 Hienuki, Y., 38010 Higaki, M., 147387, 149387, 150387, 151387 Higashimura, H., 1032⁷⁰ High, J., 197³³ Higuchi, T., 228²²⁰, 441⁴⁷ Hikino, H., 396¹¹⁵, 748⁷³ Hildebrand, R., 62131 Hill, C. L., 332204, 10475 Hill, D. G., 74874 Hill, E. A., 906145 Hill, H. A. O., 499¹¹⁰, 501¹¹⁰, 505¹¹⁰, 509¹¹⁰, 512¹¹⁰ Hill, J. H., 284⁵⁴ Hill, J. H. M., 1894, 4294 Hill, K. A., 11⁵¹ Hill, R. K., 743⁵⁹, 929⁵⁸, 943⁸³, 953^{83,100}, 994⁴² Hillhouse, J. H., 86736 Hilliger, E., 324148 Himbert, G., 28453, 55532 Himmele, W., 64097, 102218 Hindersinn, R. R., 11⁵², 17⁵² Hindriksen, B., 36176, 36776 Hine, J., 505166, 82956 Hioki, T., 229²²⁶ Hirai, K., 88^{132,133}, 90^{132,133}, 95^{132,132a,133}, 99^{132a,133} 101132a,133, 107132a,133, 118132, 89463, 102745 Hirakura, M., 81045 Hirao, I., 27733, 27935.36 Hiraoka, T., 24147 Hirashima, T., 103477

Hirata, N., 78536.36b Hiroi, K., 94382 Hirokawa, T., 66749 Hirose, Y., 347²⁶, 391⁸⁷ Hiroshima, T., 168490, 169490 Hirota, K., 219107 Hirotsu, K., 228²²⁰, 423⁷⁹, 441⁴⁷, 872⁵⁶ Hirschberg, K., 89459, 90059 Hirschmann, R., 89032 Hirst, G. C., 7797, 7927 Hirthammer, M., 53891 Hiskey, C. F., 84322 Hissom, B. R., Jr., 325161 Hitchcock, P. B., 66642 Hite, G., 788⁵¹, 845³⁷ Hite, G. A., 21127 Hiyama, T., 202^{96,97}, 216⁷⁰, 229²³⁰, 233^{273,274}, 246³³, 257³³, 421⁶³, 538⁹⁵, 539⁹⁷, 565¹⁷ Hnevsová, V., 89672 Ho, I., 57998 Ho, L.-K., 221127 Ho, L. L., 864^{16,18,20}, 866¹⁶, 883¹⁶ Ho, T.-I., 500¹²⁷, 501¹²⁷ Ho, T.-L., 86¹³, 159¹³, 421⁶², 563¹ Hoagland, S., 225187 Hoashi, K., 24644 Hobbs, A. J. W., 55853 Hobbs, C. F., 4158, 102322 Hobe, M., 95¹⁶⁰, 97¹⁶⁰, 120¹⁶⁰ Höchstetter, H., 87260 Hockerman, G. H., 325161,161a Hodge, C. N., 594¹⁷⁶, 610¹⁷⁶ Hodgins, T., 319131 Hodgson, G. L., 34932, 42786, 71226 Hodgson, S. T., 103680 Hodosan, F., 41615, 41715 Hodson, D., 88818 Hoerger, F. D., 317120 Hoey, J. G., 73629 Hofer, H., 63647 Hofer, R. M., 84640 Hoff, S., 257¹¹⁵ Hoffman, R. A., 505161, 507161 Hoffman, W. D., 57476 Hoffmann, A., 89035 Hoffmann, H., 20177, 904131 Hoffmann, R., 9149 Hoffmann, R. W., 43832, 79790, 953100, 9767 Höfle, G. A., 252⁸¹, 257⁸¹ Hofmann, A. W., 6605 Hogan, J. C., 244²⁷ Hogg, D. R., 866,10,11 Hojjat, M., 94383, 95383 Hojo, M., 1032⁷⁰, 1033⁷¹, 1040¹⁰³ Holah, D. G., 499¹¹⁸ Holder, R. W., 905141 Holl, R., 581109 Holler, H. V., 89875 Holliman, F. G., 699162 Hollingshead, D. M., 103680 Hollinshead, D. M., 613², 615² Hollister, K. R., 88¹³⁷, 95¹³⁷, 165¹³⁷, 167¹³⁷ Hollowood, F. S., 334²²⁰ Holly, F. W., 644167 Holman, R. J., 647¹⁹⁵ Holmes, A. B., 39²¹⁴, 281⁴⁶, 284⁵², 555³⁰

Holmes, S. W., 29717, 30686 Holsten, J. R., 158441, 159441 Holt, D. A., 242⁷, 257⁷ Holt, G., 888^{15,18}, 893⁵⁵ Holton, R. A., 839,40, 232256, 36595, 600216, 67988,90, 683^{90,104}, 744⁶⁴ Holy, N., 61924 Holzer, L., 81893 Honda, K., 903125 Hong, P.-K., 39292 Honig, M. L., 62029 Honwad, V. K., 21790, 21990 Hoole, R., 42152 Hoornaert, G., 332203 Hoover, J. R. E., 855⁸¹ Hooz, J., 54277, 36383, 79480 Hopkins, M. H., 79267,68 Hopkins, P. B., 104^{208,209}, 117^{208,209} Hopla, R. E., 36595 Hoppe, D., 88134, 19629 Hoppmann, A., 124268,285, 125268,285, 126268,285, 127268. 131268 Hori, I., 97173, 116173, 136369, 138369 Hori, K., 1041¹¹³ Hori, M., 969135 Horibe, I., 38657 Horie, S., 63894 Horiguchi, Y., 257¹¹⁸, 464¹⁷⁵ Horii, Z., 67781, 68681 Horikawa, H., 650^{210c,212}, 651^{210c,216} Horn, D. H. S., 553¹² Horn, R. K., 63428 Hornback, J. M., 73516 Horne, K., 19627 Horner, L., 20177, 580105,106, 596195, 89031,34, 89140, 90040,95, 901111, 904135, 909155 Hörnfeldt, A.-B., 232259 Horng, A., 19958 Hornischer, B., 82641 Hornish, R. E., 99¹⁹², 103¹⁹², 107¹⁹² Horsham, M. A., 545122, 55853 Horvath, R. F., 20072 Horwitz, J. P., 757121 Hoshi, M., 470^{196,197}, 473^{196,197}, 522²¹ Hoshi, N., 94687 Hoshino, M., 57174.75, 57474.75, 586140, 594186, 883110 Hoshino, O., 67265 Hosomi, A., 32187, 24644, 43728, 48527 Hosoya, K., 153415, 224170 Hotelling, E. B., 78112 Hou, Z., 56736,190, 595190, 607190, 61036 Houk, K. N., 4²¹, 12²¹, 18⁹⁶, 31¹⁸⁶, 66¹², 74¹², 194^{4,11}, 587142, 98526b Houlden, S. A., 89141b Houpis, I. N., 546¹²³ House, H. O., 1⁴, 2^{4,7}, 3¹⁶, 4¹⁸, 7^{18,34}, 8^{18,43}, 11^{18,54}, 13⁴, 14^{70,73}, 15⁷³, 16^{54,81}, 17^{54,83}, 18⁹⁹, 19¹⁰⁰, 20¹¹⁸, 23⁴, 26⁵⁴, 31¹¹⁸, 39⁴, 54⁴, 55⁴, 248⁵⁵, 249⁶³, 250⁷², 251⁵⁵ 264⁷², 265⁷², 269⁵⁵, 419⁴⁷, 494⁸⁷, 502⁸⁷, 563¹, 564⁷ 577⁸⁷, 606^{1b}, 746⁶⁹, 748⁶⁹, 754¹⁰⁸, 755^{108,111}, 851⁶⁶ Houser, R. W., 866³⁰ Houston, A. H. J., 9146, 9276 Howard, C. C., 29070 Howard, J., 38010 Howard, P. N., 7797, 7927 Howell, F. H., 30685

Hoye, T. R., 3429 Hrib, N. J., 978¹¹ Hsiao, C.-N., 161470, 167470 Hsieh, H. H., 390⁸³, 392⁸³ Hsu, G. J.-H., 25283 Hsu, H. B., 91825 Hu, L.-Y., 219¹⁰⁶, 512²⁰¹ Hua, D. H., 866³¹, 868³⁹ Huang, E. C. Y., 79059 Huang, F., 1040108 Huang, F.-C., 693¹⁴⁸, 694¹⁴⁸ Huang, R. L., 2426, 2576, 2596 Huang, S. J., 68195 Huang, W., 596¹⁹⁴ Hubbell, J. P., 407150 Hubbs, J. C., 38010 Huber, G., 81899 Huber, I. M. P., 7963 Huber, U., 99¹⁹¹, 103^{191a}, 107¹⁹¹ Huber-Patz, U., 87261.63 Hubert, A. J., 290⁷¹, 1047^{7,8}, 1051^{7,8} Hubert, M., 29071 Hubertus, G., 87261 Hubner, F., 25158 Hübner, H. H., 693147 Huché, M., 99135 Huckin, S. N., 58290 Hudec, J., 744⁶² Hudrlik, A. M., 34196, 34940, 35140, 36940, 759132, 760135 Hudrlik, P. F., 835, 34196, 224182, 24427, 759131.132, 760135 Huesmann, P. L., 369123 Huet, F., 765154 Huff, B. J. L., 1471, 1571 Huffman, J. C., 21795,95a, 583119 Huffman, J. W., 56457, 57264, 6075 Huffman, W. F., 369119, 372119, 39496 Hufford, C. D., 39079, 396109, 397109 Hufnal, J. M., 136373, 137373 Hughes, G. J., 3804 Hughes, G. K., 499115 Hughes, J. M., 131330 Hughes, L. R., 369^{120,122,124}. 372^{122,124} Hughes, R., 55423 Hughes, R. P., 38010 Huguenin, R., 131327 Huhtasaari, M., 640¹⁰⁹, 647^{109,197} Hui, R. C., 1024³¹ Huisgen, R., 893^{53a,b}, 896⁷¹ Huisman, H. O., 153414, 155414, 87572.77, 882100,101 Hulce, M., 5646 Hulstkamp, J., 6158 Humphreys, D. J., 689118 Huneck, S., 83475, 89031, 901114, 903114 Hung, J. C., 30^{174,177} Hünig, S., 19740, 586139 Hunma, R., 82212, 83112 Hunt, D. A., 25177, 25477 Hunt, E., 93671 Hunter, G. L. K., 73629, 771188,190 Huntington, R. D., 3807 Huq, E., 355⁵³, 357⁵³, 769¹⁷⁰, 771¹⁷⁰ Hurnaus, R., 319¹³¹ Hurst, K. M., 19953 Husband, S., 6926 Husemann, W., 587147 Hussain, N., 503149, 512149
Hussey, A. S., 2426, 2576, 2596 Hutchings, M. G., 79896.97 Hutchins, R. O., 753¹⁰², 850⁶¹ Hutchins, R. R., 103204, 108204 Hutchinson, C. R., 599211 Hutchinson, D. K., 20916, 22316 Hutchinson, J., 87894, 87994, 88094,98. 88194 Hutchinson, J. H., 50265 Huth, A., 102848 Hutton, T. W., 89251 Huxol, R. F., 88¹³⁶, 91¹³⁶, 179¹³⁶, 181¹³⁶ Huynh, C., 224¹⁶¹, 244²⁷, 262¹⁶¹, 263¹⁶¹, 264¹⁸⁰, 438³⁴. 46434, 46634,188, 901112, 93468, 94384, 969130 Hwang, C.-K., 618^{20,21}, 751⁹⁰ Hwang, K.-J., 174522 Hwu, J. R., 160⁴⁶⁹, 161⁴⁶⁹, 373¹³¹ Hyer, P. K., 14⁷⁴, 15⁷⁴ Hylton, T. A., 124²⁶⁷, 125²⁶⁷, 126²⁶⁷, 127²⁶⁷, 927⁴⁸ Hyon Yuh, Y., 647¹⁷⁹, 648¹⁷⁹ Hyuga, S., 231248, 25178, 25478, 44356, 49066, 51166, 51566 Ibuka, E., 503142 Ibuka, T., 222136,137, 226200, 62339 Ichibori, K., 91932, 92344, 93444, 95444, 100870 Ichihara, A., 693149, 694149, 74772 Ichikawa, K., 31197, 313106, 315113, 769169 Ichikawa, Y., 82957 Ichinohe, Y., 82639 Idacavage, M. J., 732, 832 Ide, J., 93979 Iemura, S., 246³⁵, 354⁶⁰ Igaki, M., 90097 Igeta, H., 461¹⁴⁶, 541¹⁰⁷ Ignatova, E., 45²⁴³ Iguchi, K., 342¹², 347¹², 381³⁰, 382³⁰, 421⁵⁹, 422⁵⁹ Iguchi, M., 39074,86, 39274, 395100, 396113,115, 397116 398113, 66538, 691128,131, 693128,139, 69738,131,139,155,156. 698¹²⁸, 769¹⁷¹ Iida, H., 136370.371, 137370, 138370.371a, 139370.371. 140^{370,371,371a,b}, 143^{371,371a,b}, 144^{371a}, 507¹⁷⁵ lida, S., 246³⁹, 257³⁹ lio, H., 35²⁰¹, 39²⁰¹, 355⁵⁴, 357⁵⁴ lítaki, Y., 380⁹, 675⁷³ Ike, K., 1040¹⁰⁴ Ikeda, H., 49²⁶³, 904¹³⁴, 1033⁷⁶ Ikeda, M., 904134 Ikeda, N., 446^{85,86,88} Ikeda, T., 599²¹¹ lkeda, Y., 446^{85.86.88} Ikegami, S., 45248, 135364, 139364, 142364, 143364, 100055 Ikehira, H., 95¹⁵² Ikemoto, Y., 579101 Ikenaga, K., 495%, 497103 Ikota, N., 1055³² Ikuta, S., 30147 Il'in, V. F., 30575a Imagawa, T., 652²²¹ Imai, I., 91932, 92343,44, 93444, 95444, 100870 Imai, K., 159⁴⁵⁵, 161⁴⁵⁵ Imai, T., 231²⁴⁵, 797⁹¹ Imamoto, T., 56734, 57034 Imuta, M., 7348 Imwinkelried, R., 580¹⁰³ Inaba, M., 83370 Inaba, S., 42155

Inaba, S.-I., 499116 Inamoto, Y., 38345, 1041113 Inanaga, J., 566²⁸, 571²⁸, 578²⁸ Ingold, C. K., 8227, 965127 Ingold, K. U., 69²⁶ Ingraham, J. N., 30992c Ingrosso, G., 7331 Inokawa, S., 500¹³¹ Inomata, K., 159451, 160451, 161451 Inoue, H., 565²² Inoue, M., 244^{29} , 311^{97} , 313^{106} , 463^{167} , 529^{52} , $555^{2\aleph}$ Inoue, S., $168^{494,501,504}$, $169^{494,501,504}$, $170^{494,501,504}$, 426⁸²⁻⁸⁴, 428^{90,91}, 429^{82,91} Inoue, Y., 500¹³³, 1024³⁰ Inubushi, Y., 62339 Ioffe, B. V., 30573, 31093, 31193 Ipatieff, V. N., 309⁹¹, 329¹⁸⁸, 331¹⁹⁷ Iqbal, J., 26¹⁶², 27¹⁶⁷, 198⁴³ Irei, H., 135³⁴⁶ Ireland, R. E., 1477, 1577, 21123, 36384.87, 36563.87 Iriuchijima, S., 147³⁹⁴ Irngartinger, H., 87261-63 Isaacs, N. S., 7331 Isacescu, D. A., 321137 Isaksson, R., 499120 Iseki, T., 3809 Ishibashi, M., 147394 Ishida, A., 585¹³⁶ Ishida, H., 224¹⁶⁹ Ishida, N., 24147, 20068, 43831, 45331, 45931 Ishiguro, H., 52325 Ishiguro, M., 244²⁵, 267²⁵, 454¹²⁰ Ishiguro, T., 406¹⁴¹, 675⁷³ Ishihara, S., 91933, 95433 Ishihara, T., 218⁹⁸ Ishii, A., 57175, 57475, 883110 Ishii, Y., 446^{76[']} Ishikawa, H., 598202 Ishikawa, M., 231252, 84433 Ishikawa, N., 244²³, 420⁵⁰, 444⁶⁴, 452¹⁰⁸, 503¹⁴⁷ Ishikawa, R., 583120 Ishikura, M., 498109, 103895,956 Ishitsuka, M., 38013 Ishiwari, H., 652221 Ishiyama, T., 231249.252, 44689, 48964, 49064, 49564, 49664, 49864, 51164, 51564 Isidor, J. L., 106223, 113223, 125299 Isihara, M., 125^{297,298}, 126²⁹⁸, 128²⁹⁷, 129²⁹⁷, 130^{297,298}, 133297, 137298 Isler, O., 698159 Ismagilova, G. S., 887¹⁰, 888¹⁰, 889¹⁰, 890¹⁰, 893¹⁰, 897¹⁰, 900¹⁰, 903¹⁰ Ismailov, R. G., 30682.84 Ismailzade, I. G., 30460 Isobe, K., 224169 Isobe, Y., 219¹⁰⁷ Isoe, S., 19852, 470208, 471208, 475208, 62849 Isoya, T., 64099 Isser, S. J., 20111 Itaya, N., 715³⁹ Itnizda, V., 4157 Ito, E., 20285 Ito, K., 565²¹ Ito, M., 168490, 169490 Itô, S., 100^{193,194,195,196,197,200,201}, 103^{193,194,195,200,201} 107194,195,201, 39069,70, 395101

Ito, Y., 1795, 45248, 225184, 227210, 229228,229, 24530, 262¹⁶⁵, 450¹⁰⁵, 463¹⁶⁷, 529⁵², 555²⁸, 623³⁹ Itoh, A., 34933, 35658, 35833 Itoh, I., 57262 Itoh, K., 262164, 44676, 466191 Itoh, M., 221134, 27421, 42153 Itoh, T., 124259, 125259, 153416, 154416, 463158, 470212, 476212 Iton, T., 28656a Itooka, T., 85372 Iuchi, Y., 168508, 169508 Iwakura, C., 63544 Iwasaki, T., 650^{210c,212}, 651^{210c,216} Iwashita, T., 135341, 136341, 137341 Iwata, C., 67781, 68681 Iwata, N., 4839 Iyer, P. S., 298²⁸, 329¹⁸⁵ Iyer, R., 87894, 87994, 88094,98, 88194 Iyoda, J., 381²¹ Iyoda, M., 42160 Izawa, Y., 89145 Izumi, T., 53081-83, 53681-83, 594185 Izzat, A. R., 66431 Jabri, N., 21784, 226203, 48532-35, 48632-35, 49434.35,89. 51689 Jachiet, D., 224166,167, 225167b Jackman, L. M., 315, 423, 5415 Jackson, A. H., 68093, 80728,30 Jackson, B. L. J., 74360 Jackson, C. B., 407147, 591169 Jackson, R. A., 56415 Jackson, R. F. W., 226204 Jackson, T. E., 23134 Jackson, W., 168491, 169491, 171491, 757122 Jackson, W. P., 34211, 35144, 35244 Jacober, W. J., 29933 Jacobs, I., 629⁵¹ Jacobs, T. L., 27311 Jacobsen, N., 642111 Jacquemin, W., 825^{24,24a}, 828^{24a}. 835²⁴ Jacquesy, J.-C., 81050 Jacquesy, R., 81050 Jacquier, R., 46²⁵⁴, 215⁶⁶, 251⁷⁵, 813⁶⁴, 839⁴ Jacquot, R., 324¹⁵⁵ Jadhav, K. P., 104²¹¹, 107²¹¹, 111²¹¹ Jadhav, P. K., 600²¹⁴ Jaen, J. C., 264¹⁸³, 265¹⁹⁰ Jaén, J. C., 226¹⁹⁴ Jagdamann, G. E., Jr., 693¹⁴¹ Jäger, V., 271², 272², 390⁸², 392⁸² Jain, A., 640¹⁰⁶ Jain, T. C., 39075.76.78.79, 39275.76.78 Jakovljevic, M., 38013 James, B. G., 587148 James, F. L., 82637 James, R., 68197 Janes, N. F., 83159 Jang, Y. M., 49261 Janiga, E. R., 88136, 91136, 179136,548, 181136 Janoušek, Z., 890³⁰, 897³⁰ Jansen, J. R., 505¹⁶⁹, 677⁸² Janssen, C. G. M., 158445, 159445 Janssen, J., 587142 Janssen, M. J., 862⁵, 866⁵ Jaouhari, R., 10463

Japp, F. R., 82845 Jarvi, E. T., 55740.41 Jarvis, B., 736²⁵ Jarvis, B. B., 422⁶⁷ Jastrzebski, J. T. B. H., 21025.26, 21132 Jaunin, R., 581¹⁰⁹ Jautelat, M., 88125, 118125, 123125, 178541.545, 181545 Jawdosiuk, M., 127321, 3807 Jean, M., 44357 Jefferies, P. R., 81898 Jeffery, B. A., 699162 Jeffery, T., 544119 Jeffery-Luong, T., 49171,72, 53185 Jefford, C. W., 105737 Jeffrey, D. A., 25151 Jeffrey-Luong, T., 21896 Jeffs, P. W., 380¹¹, 507¹⁷⁴, 677⁸⁵ Jeganathan, A., 247⁴⁷, 253⁹³ Jeger, O., 341², 360², 815⁷⁴ Jelenick, M. S., 38013 Jemison, R. W., 916¹⁶, 949⁹⁴ Jempty, T. C., 669⁵², 683⁵² Jenevein, R. M., 564¹⁰, 568⁴³ Jenkins, I. H., 514²¹¹ Jennings, K. F., 84641 Jenny, E. F., 927⁵⁵ Jenny, W., 4143 Jensen, C. B., 901¹⁰² Jensen, N. P., 36383 Jensen, U., 644^{134b,158} Jensen-Korte, U., 644¹⁶⁰ Jenson, T. M., 994³⁹, 1008⁷³, 1009⁷⁴, 1010⁷⁴ Jeremic, D., 38013 Jerris, P. J., 24149, 25149 Jerzowska-Trzebiatowska, B., 5631 Jessop, J. A., 965127 Jeyaraman, R., 73624 Ji, S., 298²⁶ Jiang, S., 298²⁷ Jobe, P. G., 57370 Joerg, J., 81365 John, D. I., 93464, 95364 John, R. A., 1255.58 Johnson, A. W., 8640, 66536, 92136 Johnson, B. F., 125294, 126294, 167294, 168294 Johnson, C. K., 74981 Johnson, C. R., 948, 1148, 8662, 88136, 91136, 17362,521. 179136,548, 181136, 21240, 223158, 224176, 226192, 248⁵⁷, 249⁵⁷, 250⁷⁰, 251⁵⁷, 263^{57,177}, 265¹⁸⁷, 786⁴⁰, 871⁵¹, 942⁸¹⁶, 1008⁶⁷ Johnson, D. A., 262170 Johnson, F., 1369, 1469 Johnson, F. H., 74359 Johnson, J. R., 27311 Johnson, M. G., 80718 Johnson, P. Y., 3809, 62540, 62951,52 Johnson, S. J., 325¹⁵⁷, 353⁴⁹ Johnson, T. B., 890³⁵ Johnson, T. R., 55531 Johnson, W. S., 1152, 1752, 126320, 133335, 136335, $\begin{array}{l} 226^{206}, 281^{45}, 341^3, 349^{40}, 351^{40}, 353^{3,46,47}, 358^{66,67}, \\ 360^{72}, 361^{75}, 362^{89,90}, 363^{83,85}, 364^{91,93}, 366^{99}, \end{array}$ 36940.107.108,117-125, 370109-113, 371114,116 372117-119,121,122,124,125, 373126,-128,129 Johnston, D. E., 334220

Johnstone, R. A. W., 229232

1088

Jolly, P. W., 228214, 4369 Jonas, D. A., 38346,50 Jonczyk, A., 158448, 159448, 168448, 174529 Jones, D. H., 38346 Jones, D. N., 114234, 35461 Jones, D. W., 690¹²⁴ Jones, E. R. H., 554^{18,19} Jones, F. N., 48422 Jones, G. C., 91412 Jones, G. E., 28452, 55530 Jones, J. B., 125^{295,296}, 126^{295,296}, 128²⁹⁶, 783²⁴ Jones, L. B., 242⁴, 244⁴ Jones, L. D., 499¹²⁸, 500^{128,132}, 505¹³², 509¹²⁸ Jones, M., Jr., 89792, 90092, 903123 Jones, M. E., 422 Jones, N. R., 124²⁷⁸, 125²⁷⁸, 128²⁷⁸, 129²⁷⁸, 132²⁷⁸ Jones, P. F., 124²⁸⁹, 125^{289,314} Jones, P. R., 135³⁵⁹, 136³⁵⁹, 137³⁵⁹, 139³⁵⁹, 142³⁵⁹, 260¹⁴⁴ Jones, R. A., 211²⁸, 215²⁸ Jones, R. G., 2089, 24419 Jones, R. S., Jr., 92757 Jones, R. V. H., 39070 Jones, R. W., 106224, 113224 Joseph, M. A., 760139, 774139 Joshi, B. S., 396115 Joshi, K. K., 38349 Joshua, H., 92756 Joukhadar, L., 613², 615² Jousseaume, B., 44045 Joyce, R. P., 511¹⁹¹ Joyeux, M., 10174 Juday, R. E., 56842 Juge, S., 47²⁵⁶ Jugelt, W., 891³⁶, 909¹⁵³ Jukes, A. E., 20810, 21010, 499110, 501110, 505110, 509110, 512110, 52214 Julia, M., 159454,456,461,463, 160471, 161454,456,461,463,471 162^{456} , 163^{471} , $165^{461,463}$, 167^{481} , $168^{481,507}$, 169^{507} , 170^{481} , 171^{507} , 172^{481} , $174^{536,537}$, $176^{536,537}$, 178^{536,537}, 342¹³, 351⁴², 352⁴², 381²⁵, 382²⁵, 447⁹³, 44894,95, 49381, 882103 Julia, S., 147395, 155395, 25286, 757124, 89669, 93468 94384, 963123, 964123, 125, 969130, 99136, 99848,49, 99950 Jun, Y. M., 587141.143 Junek, H., 824²², 826^{22,41} Jung, M. E., 1149,50, 20120, 2349, 35204, 6614, 1946, 504154, 511154, 515154, 602226 Jung, Y. H., 504154, 511154, 515154 Jungheim, L. N., 78534 Jungk, H., 300³⁶ Jurion, M., 131333 Just, G., 259132, 38010, 541115, 84641 Jutand, A., 44358, 450102, 454117,118 Kabasakalian, P., 649²⁰² Kabeta, K., 229227, 43830 Kabli, R. A., 325158 Kabuto, C., 395101 Kadam, S. R., 602²²⁵ Kaddah, A. M., 325158 Kadkhodayan, M., 464174 Kadowaki, T., 321139 Kadyrov, Ch. Sh., 30679, 315109,111, 316116, 317116 Kaeding, W. W., 30575b,c Kaenel, H. R., 298²⁹ Kagan, H. B., 56735, 57035

Kagan, J., 73938, 74770, 770176 Kagechika, K., 650²¹³ Kagotani, M., 512200 Kahn, B. E., 563¹, 570^{1h} Kahne, D., 34198, 39198 Kai, Y., 672⁶⁵ Kaiser, E. M., 158438, 159438,467, 160438, 166438,467 Kaiser, E. T., 86⁴⁴, 649²⁰², 650^{202c}, 652^{202c} Kaiser, G. V., 93465, 95365 Kaiser, W., 38118 Kaito, M., 639⁷⁵ Kaji, A., 99180, 100180, 153415, 159455, 161455, 174526, 224170 Kaji, K., 155428, 168497, 170497, 220115 Kajigaeshi, S., 321139 Kajiwara, A., 75188 Kajiwara, K., 135³⁴¹, 136³⁴¹, 137³⁴¹ Kakinuma, K., 99951 Kakisawa, H., 38013 Kakiuchi, K., 3809 Kakui, T., 469²⁰³, 470²⁰³, 473²⁰³, 483¹⁶ Kalb, L., 835⁸⁰ Kalicky, P., 105840 Kalinin, V. N., 89881 Kalinovski, I. O., 53073, 53473 Kalinowski, H.-O., 6611, 7411 Kallenberg, H., 20¹²¹, 25¹²¹ Kallmerten, J., 100764 Kalman, J. R., 505160 Kamada, M., 498¹⁰⁹ Kamat, V. N., 396115 Kambe, N., 103477 Kameswaran, V., 509¹⁷⁶, 670^{57,578}, 673⁶⁸, 681⁶⁸, 686⁶⁸. 807³³, 816⁸² Kametani, T., 164480, 165480, 167482, 168482, 66017, 677⁸⁶, 679¹⁷ Kamimura, Y., 2958 Kammerer, H., 6606 Kamneva, A. I., 648^{172,173} Kampmeier, J. A., 1040105 Kamysheva, T. P., 30460 Kamyshova, A. A., 42269 Kan, R. O., 82314, 83514 Kan, T., 100¹⁹⁹ Kanakura, A., 20297 Kanamaru, H., 259130 Kanaya, I., 102116 Kanda, Y., 39079, 100872, 100972, 101072.75 Kaneda, K., 1041111 Kaneda, T., 55635 Kanehira, K., 229226 Kanemasa, S., 20184 Kanematsu, K., 38341.44, 80941 Kanemoto, S., 48424, 50124, 73043 Kanemoto, Y., 404135 Kanevskii, L. S., 63535, 647196, 648181 Kanevsky, L. S., 647¹⁷⁷, 648¹⁷⁷, 649¹⁷⁷ Kanfer, S. J., 219¹¹⁴, 499¹⁴⁰, 501¹⁴⁰, 502¹⁴⁰ Kang, J., 217⁸⁴, 250⁶⁹ Kang, M.-C., 546¹²³ Kang, S. H., 225187 Kanjilal, P. R., 87115 Kantner, S. S., 222^{138,1385,142} Kantor, S. W., 9148, 91514, 91821, 96714, 9752, 9762, 979², 980² Kao, J. C., 85369

Kaplan, F., 89144 Kaplan, J., 824²³ Kapoor, V. M., 373127 Kappeler, H., 34520 Kappey, C.-H., 373130 Karabatsos, G. J., 8398 Karady, S., 30175, 38011 Karapetyan, K. G., 648181 Karapinka, G. L., 66433 Kariv-Miller, E., 568⁵⁷, 570⁵⁷, 599^{205,222}, 602²²² Karkour, B., 620²⁸ Karoglan, J. E., 595191 Karpf, M., 104914,15, 105842 Karras, M., 244²⁶, 485²⁸ Karrick, G. L., 325^{161,161a} Karunaratne, V., 20^{114,115} Kasahara, A., 530⁸¹⁻⁸³, 536⁸¹⁻⁸³, 594¹⁸⁵ Kasai, N., 672⁶⁵ Kascheres, A. J., 735²⁰ Kase, K., 649²⁰⁵ Kasel, W., 87261,63 Kashimura, S., 54279 Kashimura, T., 102640 Kashin, A. N., 503¹⁵² Kashman, Y., 38013, 395102, 396102, 407146 Kasiwagi, I., 824¹⁷, 825¹⁷ Kaska, D. D., 57997 Kass, S. R., 422 Kastner, B., 89032 Kasuga, K., 95¹⁵³, 107¹⁵³, 114¹⁵³, 115¹⁵³ Kasuga, T., 100459, 100562.63 Katakis, D., 565²⁰ Kataoka, F., 583¹²⁷, 587¹⁴³ Kataoka, H., 57582, 883107 Kataoka, M., 585137 Kataoka, S., 55749 Katayama, E., 73042,42b Katayama, T., 84325 Katekar, G. F., 88136, 91136, 179136, 181136 Katerinopoulos, H., 21794 Kato, M., 88124, 118124, 19733, 224162 Kato, N., 21342, 224162, 264185, 57371, 57581,82, 61071 Kato, T., 446⁷⁷, 512¹⁹⁹, 730⁴⁵ Katou, T., 530^{81,82}, 536^{81,82} Katritzky, A. R., 7135, 19736, 28247 Katsaros, N., 565²⁰ Katsuhara, J., 771 191 Katsuki, M., 30355 Katsuki, T., 45²⁴⁸, 223¹⁵⁶, 225^{185,187}, 264¹⁸¹, 1000⁵⁵, 1001⁵⁶, 1004⁶⁰ Katsumura, S., 470²⁰⁸, 471²⁰⁸, 475²⁰⁸ Katsuro, Y., 445^{69,71}, 455¹²², 492^{76,78}, 495⁷⁸ Kattenberg, J., 87577 Katz, J.-J., 79476, 79582 Katzenellenbogen, J. A., 50²⁶⁷, 288⁶⁴, 443⁶⁰ Kauffman, G. B., 20913 Kauffman, W. J., 765152 Kauffmann, T., 105²¹⁴, 203¹⁰⁰, 253⁹¹, 482^{3.5}, 499⁵, 505⁵, 5095.196, 512196, 52216 Kaufmann, K., 73520 Kaufmann, S., 805¹⁵ Kawada, M., 135^{345,346,347,348}, 136^{347,348}, 137³⁴⁷, 139347.348, 141348, 144347,348, 168508, 169508, 174345, 586¹³⁸, 650²¹¹ Kawagishi, H., 693¹⁴⁹, 694¹⁴⁹ Kawahara, I., 167484, 168484, 36174

Kawakami, S., 597²⁰¹ Kawakami, Y., 24422, 465179, 49489, 51689 Kawamoto, I., 19846 Kawamoto, K., 98732.33, 100052 Kawamoto, T., 259137 Kawamura, C., 64099 Kawamura, S., 103371 Kawanami, Y., 45248 Kawanisi, M., 652²²¹, 903¹²⁵ Kawara, T., 227²¹⁰, 245³⁰, 446⁷⁸ Kawasaki, H., 41²²⁴, 43²³⁵ Kawashima, M., 218100, 220125, 227210, 24530-32, 44678, 470212,213, 476212,213 Kawashima, S., 38013 Kawazoe, T., 470²¹⁵, 473²¹⁵, 476²¹⁵ Kaye, A. D., 985²⁷ Kaye, I. A., 72412 Kaye, P. T., 42152 Kaye, R. L., 89668 Kazakova, L. I., 63980 Kazanskii, B. A., 38132 Kazarinov, V. E., 63655 Kazi, A. B., 35352 Kazimirchik, I. V., 86421 Kazimirova, V. F., 72310 Keal, C. A., 125³⁰¹, 126³⁰¹, 128³⁰¹, 129³⁰¹, 133³⁰¹ Keay, B. A., 488⁵², 495⁵² Kebarle, P., 301⁴⁷ Keck, G. E., 57269, 57369, 60269, 60769, 61069 Keehn, P. M., 167483 Keeley, D. E., 86⁶¹, 88⁶¹, 89⁶¹, 91⁶¹, 124⁶¹, 785^{32,33} Keely, S. L., Jr., 396109, 397109 Kees, K. L., 225¹⁸⁷, 579¹²⁵, 582¹²⁵, 583¹²⁵, 584¹²⁵, 585125, 588125,159, 591165, 595125, 596125, 610125,165 Keese, R., 3804 Kegelman, M. R., 5648 Keinan, E., 1259, 229231 Kell, D. R., 38122 Kelleghan, W. J., 53069, 53369, 53769 Keller, L., 500¹³², 505¹³² Keller-Wojtkiewicz, B., 39083, 39283 Kellner, J., 53186 Kellog, M. S., 369¹⁰⁷ Kellogg, R. M., 229^{223,224,224a,b}, 512¹⁹⁹. 865²⁸ Kelly, D. P., 330¹⁹², 916¹⁷, 932¹⁷, 963¹¹⁹ Kelly, D. R., 29070 Kelly, J. T., 330190 Kelly, R. B., 20105 Kelly, R. C., 651²¹⁹ Kelman, R. D., 597²⁰⁰ Kelsey, D. R., 500¹³⁰, 509¹³⁰ Kempe, T., 872⁵⁶ Kempf, D. J., 6717 Kendall, P. E., 41826 Kende, A. S., 50²⁶⁶, 51²⁶⁹, 217⁹⁵, 219¹¹⁴, 284⁵⁴, 483¹⁸, 499¹⁴⁰, 500¹⁸, 501¹⁴⁰, 502¹⁴⁰, 672⁶⁴, 674⁷¹, 681^{1(K)}, 80723, 8396 Kendrick, D. A., 55530 Kennedy, J. H., 762¹⁴⁸, 769¹⁴⁸ Kennedy, J. P., 331199 Kenner, G. W., 770178 Kenny, C., 57473, 57573, 59973,208, 61073 Kenyon, J., 9146, 9276 Keough, A. H., 380⁵ Keramat, A., 759128

Kerkman, D. J., 629^{51,52}

Kern, D. H., 56838,39 Kern, R. J., 74152 Kerr, C. A., 316115 Kerschen, J. A., 75086 Kerton, N. A., 49486 Kessar, S. V., 105739 Kessler, H., 56285 Kettenring, J., 62845 Kevelam, H. J., 55210 Khalaf, A. A., 294⁵, 299⁵, 300⁵, 303⁵, 304⁵, 306⁸⁰. 323145, 324147, 325156,158,160, 327170 Khalikov, S. S., 30679 Khan, R., 102219 Khan, W. A., 655 Khand, I. U., 102432 Kharasch, M. S., 2084,5, 2104 Kharasch, N., 505¹⁶⁴, 507^{164,173}, 512¹⁶⁴, 515¹⁶⁴ Khemani, K. C., 867³⁶ Khoury, G., 753¹⁰⁰ Khromov, S. I., 38132 Khrustova, Z. S., 644140 Kibayashi, C., 507¹⁷⁵ Kidd, J. M., 29719 Kido, F., 956¹⁰⁷ Kieczykowski, G. R., 33190, 34197 Kiefer, G. E., 509179 Kiehs, K., 890³⁴, 903¹²⁶ Kiel, W. A., 107226, 109226 Kiely, J. S., 466¹⁸⁹ Kienzle, F., 505165 Kierstead, R. W., 643117 Kiessling, L. L., 21795, 545120 Kihara, M., 677⁸⁷ Kihira, K., 644¹⁴⁹ Kiji, J., 55529, 102746 Kikukawa, K., 495%, 497103, 53078, 53578, 102641 Kilger, R., 53274 Kilian, R. J., 391⁸⁸ Kilmer, G. W., 89877 Kilpatrick, M., 297²² Kim, C.-K., 67368, 68168,98, 68668,110, 80732,33, 81577. 81681 Kim, C.-M., 509176 Kim, D., 49²⁶¹ Kim, H., 38013 Kim, H. K., 727³³ Kim, H. S., 49²⁶¹ Kim, I. O., 49261 Kim, S., 28863 Kim, S. W., 784³¹ Kim, Y., 57158, 596193, 72837 Kimura, K., 638⁹⁴ Kimura, M., 103895 Kimura, Y., 977¹⁰, 985^{26a}, 992³⁸, 993³⁸, 999⁵¹ King, A. O., 266¹⁹³, 450¹⁰³, 453¹⁰³, 486⁴⁴, 495⁴⁴, 503^{150,151}, 524³⁵⁻³⁷, 529⁵³ King, F. D., 500126, 512126 King, F. E., 83159 King, J. F., 70710, 86736 King, J. R., 330¹⁹⁰ King, R. B., 208³, 380¹⁰ King, T. J., 665³⁶ Kingsbury, C. A., 114233, 154233 Kinney, W. A., 168⁵⁰⁵ Kinoshita, H., 159451, 160451, 161451 Kinsman, D. V., 73514

Kinson, P. L., 89142, 903122 Kinzig, C. M., 273¹⁵, 274¹⁵ Kirby, G. W., 681⁹⁷, 689¹²³ Kirchen, R. P., 374133 Kirchhoff, R. A., 78640 Kirilov, M., 20289 Kirk, D. N., 74150,52, 79162,63 Kirkpatrick, D., 27423 Kirmse, W., 706⁶, 887⁹, 890³¹, 893⁹, 894⁹, 896⁹, 897⁹. 900⁹, 901¹¹¹, 903⁹, 905^{9,136}, 909¹⁵⁵ Kirn, B., 904¹³³ Kirrmann, A., 73942 Kirrstetter, R. G. H., 61925 Kirschleger, B., 20291 Kirst, H. A., 431102 Kirtane, J. G., 223157, 262159, 264159 Kishi, M., 38012 Kishi, N., 985²⁶, 992³⁸, 993³⁸ Kishi, Y., 125²⁹⁴, 126²⁹⁴, 167²⁹⁴, 168²⁹⁴, 225¹⁸⁷, 231²⁵³, 466¹⁸⁶ Kishida, Y., 88^{132,133}, 90^{132,133}, 95^{132,132a,133}, 97¹⁷¹ 99132a,133, 101132a,133, 107132a,133, 118132,171, 93979, 963120 Kiso, Y., 228^{218,222}, 437^{21,25}, 440²⁵, 448²⁵, 449²⁵, 450²⁵, 45125,106, 45225,106,107,109,111, 460107, 48426, 48745, 492²⁶, 494²⁶, 495²⁶, 503²⁶, 513²⁶ Kita, Y., 93979 Kitade, Y., 219107 Kitagawa, I., 75188 Kitagawa, O., 42154 Kitagawa, T., 45248 Kitagawa, Y., 24635, 34933, 35460, 35833 Kitahara, E., 985²⁵ Kitahara, T., 49²⁶², 287⁶², 871⁵⁴ Kitai, M., 799¹⁰⁰ Kitamura, K., 51²⁷¹, 198⁴⁸, 751⁸⁸ Kitamura, T., 219¹⁰⁵, 522²² Kitano, Y., 223¹⁸³, 225¹⁸³ Kitatani, K., 202%, 21670 Kitazume, T., 420⁵⁰, 444⁶⁴ Kiyoi, T., 229²²⁸ Klaubert, D. H., 499¹²¹ Klaver, W. J., 223¹⁸³, 225¹⁸³ Kleijn, H., 21782, 49168, 53184 Kleiman, R., 691¹²⁹, 693¹²⁹ Klein, C. F., 79059 Klein, G., 38343 Klein, H. A., 20299 Klein, J., 1682, 87261.62 Kleinberg, J., 56412, 57476 Klessing, K., 87259 Klieger, E., 89887 Klima, W. L., 486⁴², 498⁴² Kline, D. N., 168496 Kline, M. W., 92751 Klinger, T. C., 8635, 16435, 17335 Klingsberg, E., 84013 Klingstedt, T., 260141,142, 261142, 466180,181 Klinke, P., 89887 Klinkert, G., 29070 Klinot, J., 757124 Klix, R. C., 74981 Klobucar, W. D., 87257, 87357 Kloek, L. A., 369122, 372122 Kloetzel, M. C., 89883 Kloetzer, W., 904133

Klopman, G., 4²⁰ Klosa, J., 82633,35,38 Kluge, M., 89033 Klunder, A. J. H., 85583 Klunder, J. M., 223¹⁵⁶, 224¹⁸⁰ Klünenberg, H., 644157 Klyne, W., 781¹⁷ Knapp, S., 785^{36,36b} Kneen, G., 43197,98 Knesel, G. A., 390⁸¹, 392⁸¹ Knifton, J. F., 102956, 103059, 103788 Knight, D. W., 2426, 2576, 2596, 261153, 27832 Knight, H. M., 330190 Knochel, P., 20918, 257122 Knol, K. E., 5529, 5579 Knolle, J., 640^{105,107}, 641^{107b}, 644^{162,163,165}, 647¹⁰⁷. 648¹⁰⁷⁶ Knorr, R., 587¹⁴³ Knotter, D. M., 21025 Knowles, W. S., 102322 Knox, S. A. R., 380¹⁰ Knudsen, M. J., 1025³⁴ Knutsson, L., 843²⁴, 892⁵² Ko, J., 848⁴⁹ Ko, S. Y., 223¹⁵⁶ Kobal, V. M., 135340, 137340, 139340, 141340 Kobayashi, K., 19733 Kobayashi, M., 231²⁴¹, 266¹⁹⁶, 443^{51,53,54}, 453⁵³, 628⁴⁹, 74772, 87573-75 Kobayashi, S., 295⁷, 302⁵², 638⁹², 677⁸⁷, 1017⁷ Kobayashi, T., 380⁴, 529⁵¹, 1016³, 1033⁷¹, 1039^{101,102} Kobayashi, Y., 100²⁰¹, 103²⁰¹, 107²⁰¹, 224¹⁶², 421⁵⁴, 48962, 49562, 50462, 51162, 51562, 63985 Kobori, Y., 228²²⁰, 441⁴⁷ Köbrich, G., 55315 Koch, K., 681¹⁰⁰, 807²³ Koch, P., 158⁴⁴³, 159⁴⁴³, 160⁴⁴³, 161⁴⁴³, 167⁴⁴³, 168⁴⁴³ Koch, R. W., 581110 Köcher, M., 593179 Kochi, J., 415¹⁰, 418¹⁰ Kochi, J. K., 210²³, 228⁵³, 243¹³, 244¹⁹, 418²⁵, 436^{1,2,6}, 437^{20,24}, 438^{2,20}, 439²⁰, 441²⁰, 442²⁰, 464²⁴, 482⁷, 49484, 4997 Kocienski, P. J., 26¹⁶³, 135³⁴⁹, 136³⁴⁹, 137³⁴⁹, 140³⁴⁹, $141^{349}, 159^{452}, 162^{452}, 163^{452}, 218^{100,1005}, 224^{163,163a}, 229^{234}, 264^{178}, 444^{66-68}, 934^{69}$ Kocsi, W. P., 333^{211b} Kodama, A., 168490, 169490 Kodama, H., 463¹⁶⁵ Kodama, M., 99¹⁹⁰, 100^{193,194,195,196,197,200,201} 103193,194,195,200,201, 107194,195,201, 39069.70, 395101 Kodama, S., 437²⁵, 440²⁵, 448²⁵, 449²⁵, 450²⁵, 451²⁵, 452^{25,107}, 459¹³⁷, 460^{107,137}, 461¹³⁷, 484²⁶, 492²⁶, 49426, 49526, 50326, 510206, 51326,206 Kodovsky, P., 591167, 610167, 70915 Koehi, W. J., Jr., 649²⁰³ Koenig, G., 915¹³, 965¹³ Koemer, M., 20917, 21348, 25176, 26176, 26476 Koesling, V., 866²⁹ Kofron, W. G., 35²⁰³, 66¹³ Koga, K., 36²¹¹, 41²²⁴, 43²³⁵, 217⁹⁵, 675⁷³ Koga, T., 380^{13'} Kögl, F., 829⁵⁴ Kohada, H., 91932 Kohama, T., 396115 Kohler, E. P., 756¹¹³

Kohlheim, K., 322¹⁴³ Kohoda, H., 92344, 93444, 95444, 100870 Koizumi, N., 125²⁹³, 126²⁹³, 127²⁹³, 128²⁹³ Kojima, T., 100²⁰⁰, 103²⁰⁰ Kojima, Y., 21342, 264185 Kolb, M., 131324,331, 135324 Kolb, R., 87259 Kolb, V. M., 86416,23, 86616, 88316 Kolbe, H., 633² Kolhe, J. N., 602²²⁵ Kolka, A. J., 29612 Koller, E., 35037 Kolm, H., 89356 Kolodny, N. H., 57787 Kolodziejski, W., 322144 Komarewsky, V. I., 331197 Komeno, T., 38012 Komiya, Z., 901108 Komoto, H., 63981 Komoto, R. G., 102851 Kondo, A., 38341,44 Kondo, I., 4839 Kondo, K., 99¹⁸⁷, 107¹⁸⁷, 110¹⁸⁷, 159⁴⁵⁸, 161⁴⁵⁸, 162⁴⁵⁸, 167⁴⁵⁸, 202^{85,88}, 437²², 438^{22,36}, 440³⁶, 771¹⁸⁶, 923⁴², 949⁹⁷ Kondo, M., 89145 Kondo, S., 91932, 92344, 93444, 95444, 100870,71 Kondo, T., 500133, 102849 Kondo, Y., 406¹⁴¹, 460¹⁴³, 461^{143,148}, 530⁶⁴, 533⁶⁴, 541 108-111,113,114, 543 108-111,113,117,118 Kondrikov, N. B., 648183 König, H., 89354, 89671 König, J., 509¹⁹⁶, 512¹⁹⁶ König, L., 466¹⁸³ Konishi, H., 55529, 102746 Konishi, M., 228^{219,220}, 229^{226,227,231}, 246³⁷, 438³³ 44147, 44250, 45033, 45233, 470219, 471219, 472219 Konno, C., 396¹¹⁵ Konno, S., 461¹⁴⁷ Kononov, N. F., 30572 Konopelski, J. P., 70917,18 Konstantatos, J., 56520 Konz, W. E., 36595 Koolpe, G. A., 87¹⁰⁸, 89¹⁴⁵, 90¹⁴⁵, 105¹⁴⁵, 106¹⁰⁸, 114¹⁰⁸. 120108, 136108, 144108 Koosha, K., 155430 Köppelmann, E., 4823 Kopp-Mayer, M., 84219 Koreeda, M., 24¹⁴⁸, 125²⁹³, 126²⁹³, 127²⁹³, 128²⁹³, 996⁴⁶, 997^{46a}, 1000^{46,53} Kornienko, A. G., 63544, 648172,173,174 Korniski, T. J., 154420 Korobeinicheva, I. K., 38668 Korobitsyna, I. K., 887⁷, 890⁷, 892⁷, 893⁷, 896⁷, 897^{7,73}. 900⁷, 902⁷, 903⁷, 905⁷ Korostova, S. E., 259128 Korshak, V. V., 319132 Korsmeyer, R. W., 1052²⁵, 1059²⁵ Korte, W. D., 419^{32,33} Korzeniowski, S. H., 505¹⁶³ Kos, A. J., 1944 Kosarych, Z., 93774,75 Kosch, E., 89035, 903127 Koser, G. F., 82956 Koshchii, V. A., 30677 Koskimies, J. K., 8630, 12130

Köster, F.-H., 34828 Kosugi, H., 94687 Kosugi, K., 446⁸⁰, 456⁸⁰, 470⁸⁰, 471⁸⁰, 493⁸² Kosugi, M., 12⁶², 453^{113,114}, 454¹²⁰, 463^{160,164}, 469²¹⁶, 470215,216, 473215, 476215,216 Kotake, H., 159451, 160451, 161451 Kotani, E., 67989, 683102,105, 69389, 695153, 696154, 80735 Kotani, S., 541106, 55850 Kotlan, J., 81791 Koto, M., 91827, 968128 Kotok, S. D., 89773 Koukoua, G., 3808 Koumaglo, K., 20071 Kovad, I., 904133 Kovsman, E. P., 64098,101, 647177, 648177, 649177 Kow, R., 19960,66 Kowalczyk-Przewloka, T., 56285 Kozar, L. G., 34419 Kozerski, L., 38236 Kozikowski, A. P., 131326, 135326, 224169 Kozlikovskii, Ya. B., 30677 Kozlova, L. M., 31094, 31194 Kozlowski, J. A., 209¹⁵, 211^{34a}, 213^{34a,46,46c,54}, 216⁴⁶, 223⁴⁶, 224¹⁶⁷, 250⁷², 251⁷⁴, 264^{72,186}, 265⁷², 491⁷⁰ Krafft, G. A., 8612, 8622 Krafft, M. E., 839,40, 232256, 102533,33c Kramar, V., 27 Kramer, R., 263¹⁷² Krammer, R., 27728 Krantz, A., 21792 Kranzlein, G., 2943 Krapcho, A. P., 39²¹⁹, 40²¹⁹, 727²⁹ Krasavtsev, I. I., 643¹¹⁸ Kraus, C. A., 29933 Krauss, P., 909155 Krebs, A., 55633 Krebs, E.-P., 19843 Krebs, J., 103787 Kreft, A., 220¹²¹ Kremer, K. A. M., 39²¹³ Krepski, L. R., 179⁵⁴⁶, 579¹²⁵, 582¹²⁵, 583¹²⁵, 584¹²⁵, 585¹²⁵, 586¹⁵², 588^{125,152}, 595¹²⁵, 596¹²⁵, 610^{125,152}, 918²¹ Kretchmer, R. A., 39188, 500134, 509134 Kreuzfeld, H. J., 229²²⁴ Krief, A., 86^{23,25,49,50,52,55} 87^{23,25,66,67,70-73,76-80,83,84,86,89-93,101,110,116-119}, 88^{25,127}, 8923,25,52,70,71,110,127,140,142,143, 9025,70,71,140 9123.25,52.71,127,140,148,149,150, 9223,70,71,127,140,143,148, 9325,127, 9423,25,52,55, 9523,52,72,79,80, 104¹⁰¹, 105¹¹⁰, 106^{83,110}, 107^{140,228}, 109^{79,80,84,127,140}, 11179.80.228.230.231, 11423.52,110, 11652.70.71,127,140,142, 11752, 118143.237.238, 11952.70.71, 12083,90,110,149,150,242,243, 12183, 123143,238, 12425, 13676-78, 13777, 14177, 14286,243,379, 14476-78,86. 145^{25,76-78}, 155⁴²⁷, 193¹, 253⁹¹, 766¹⁵⁶, 778³, 785³, 78643.44 Krieger, C., 87782 Krieger, J. K., 24855, 25155, 26955, 4826 Krimer, M. Z., 3427 Krimmer, H.-P., 89143, 89243 Krishna, R. R., 73733 Krishnamurthy, N., 604230 Krishnamurthy, V. V., 334215 Kröger, C.-F., 322142b Krolski, M. E., 232269

Kronberger, K., 57787 Kronja, O., 374132 Kropf, H., 5631 Kropp, P. J., 81573,74 Krumpolc, M., 757124 Kruse, C. G., 24210 Kruse, L. I., 1898, 102955 Kryuchkova, V. E., 30574 Ku, A., 255¹⁰⁴ Ku, H., 255¹⁰⁴ Ku, Y. Y., 512²⁰³ Kubota, Y., 7548 Kucerovy, A., 196²³, 257¹¹⁵ Kucherov, V. F., 342^{7,14}, 345²¹, 346²⁴, 351⁴¹, 361^{73,80} Kudera, J., 890³⁰, 897³⁰ Kudo, K., 102640 Kuehne, M. E., 30173, 55282, 3809 Kuétegan, M., 219103 Kuhlmann, D., 482³ Kuhn, D., 771¹⁸⁴, 785^{36,36a} Kuhn, L. P., 574⁸⁰ Kuhn, S. J., 30571, 320135, 3312008 Kuhnle, J. A., 66645 Kulicki, W., 84959 Kulkarni, A. K., 760135 Kulkarni, S. B., 328175, 331195 Kulkarni, S. U., 79584 Kumada, M., 20068, 228216,217,218,219,220,221,222 229^{225,226,227,231}, 246³⁷, 436^{3,8,10,19}, 437^{21,25,26} 43830,31,33, 44025, 44147, 44250, 44569,71, 44825, 44925,99,100, 45025,26,33, 45125,106, 45225,33,106,107,111 453³¹, 455¹²², 457^{129,130}, 459^{31,137,139}, 460^{107,130,137}, 461137, 464171, 469203, 470203,219, 471219, 472219, 473²⁰³, 483¹⁶, 487⁴⁵, 492^{74,76,78}, 495⁷⁸, 497¹⁰⁴, 503¹⁴⁵, 510183,206, 513205,206, 52427,32 Kumai, S., 72625 Kumazawa, T., 883¹⁰⁶ Kume, A., 197⁴¹, 199⁴¹ Kundu, N., 83477 Kunesch, G., 263173 Kunishima, M., 286^{56a}, 1050¹⁹ Kunze, M., 38236 Kuo, C. H., 689119 Kuo, D. L., 26¹⁶⁴ Kuo, P. C., 694¹⁵⁰ Kuo, S. J., 219¹¹⁴, 499¹⁴⁰, 501¹⁴⁰, 502¹⁴⁰ Kuo, Y.-C., 90090 Kuo, Y. H., 693146, 694150 Kupchan, S. M., 507173, 509176, 67057,578,59, 67368 68168,98, 68357b, 68668,110, 80732,33, 81577, 81681,82 Kuper, D. G., 906¹⁴³ Kuppers, H., 19413 Kurath, P., 83472 Kurbanov, M., 34214, 36180 Kurcherov, V. F., 34214 Kurihara, K., 52847 Kurihara, T., 53063, 53263 Kurihara, Y., 84854 Kurita, Y., 218¹⁰⁰ Kuriyama, K., 386⁵⁷ Kurobe, H., 164⁴⁸⁰, 165⁴⁸⁰, 167⁴⁸², 168⁴⁸² Kuroda, K., 698^{157b} Kuroda, S., 593^{181,182}, 1000⁵⁵, 1001⁵⁶ Kurono, M., 3804 Kuroyama, Y., 53061, 53161, 53261, 53761 Kurozumi, S., 222¹⁴⁴

Author Index

Kurth, M. J., 630, 224179, 3429 Kurtz, P., 27418 Kusumi, T., 38013 Kusumoto, T., 56734, 57034 Kutney, J. P., 3807 Kuwahara, S., 396115 Kuwajima, I., 628, 828, 1263, 1428, 19733, 221131, 257118. 454119, 455125, 460125, 61715, 61915, 62115, 62315,34, 62715, 72733 Kuyama, M., 167484, 168484, 36174 Kuyper, L. F., 124²⁸⁸, 128²⁸⁸, 132²⁸⁸ Kwa, T. L., 260¹⁴⁵ Kwart, H., 8229, 8239, 8349 Kwast, E., 79061 Kwiatkowski, G. T., 66433 Kwon, Y. C., 983¹⁶⁻¹⁸ Kwun, O. C., 299³³ Kyler, K. S., 39²¹⁸, 48²¹⁸, 215⁶⁷, 483¹⁵, 500¹⁵ Laali, K., 334²¹⁵ Labana, L. L., 73623 Labana, S. S., 73623 Labar, D., 8650, 87117, 89142, 116142 Labhart, M. P., 87365, 87465 Laborde, E., 226^{194,194}, 265¹⁹¹ Laboureur, J.-L., 87117,119, 7783, 7853, 78643 Labovitz, J. N., 36492, 41943 Lachance, P., 84644 Lachkova, V., 20289 Ladduwahetty, T., 220¹¹⁹, 224¹⁷⁴ Laffitte, J.-A., 87894, 87994, 88094, 88194 Lahav, M., 38236 Lahousse, F., 87049 Lahrmann, E., 81256 Lai, Y.-H., 563¹, 606¹⁸, 607¹⁸, 877⁸³ Lai, Y.-S., 220123 Laipanov, D. Z., 316116, 317116 Laird, B. B., 38010 Laird, T., 91620, 94994,96 Lakhvich, F. A., 24145 Lal, D., 69²⁶ Lalima, N. J., Jr., 250⁶⁸ Lailemand, J. Y., 224^{174,174c}, 263¹⁷³ La Londe, R. T., 649²⁰², 650^{202c}, 652^{202c} Lamarre, C., 63538 Lamba, D., 396110, 397110 Lambert, C., 279³⁷ Lambert, R. L., Jr., 20069 Lammer, O., 224164, 225164 Lammertsma, K., 333²¹⁰ Lamothe, S., 56²⁸⁶, 380¹⁰ Lampilas, M., 224^{174,174c} Lampman, G. M., 414⁵, 422⁵ Landais, Y., 509177 Landesman, H., 28170, 30170 Landmann, B., 43832 Landor, P. D., 522¹² Landor, S. R., 223146, 52212 Landquist, J. K., 78638 Lane, J. F., 892⁵⁰ Lang, S. A., Jr., 767¹⁶³ Lange, B. C., 3¹⁵, 54¹⁵ Lange, G. L., 38010 Langer, F., 817^{86.88} Langer, V., 70915 Langler, R. F., 587141

Lansbury, P. T., 1476, 1576, 25151,152, 3414, 34725, 35869, 975³, 980³ Lantos, I., 500127, 501127 Lanza, T. J., 541112, 543112 Lapierre, J. C., 327¹⁶⁸ LaPierre, R. B., 328179 Lapointe, J.-P., 903126 Lappert, M. F., 124²⁸⁹, 125²⁸⁹ Larchevêque, M., 31¹⁸⁴, 41²²⁵, 45²⁴³, 215⁶⁶, 225^{189,190}, 263174,175 Lardicci, L., 217^{80,80b}, 246⁴¹, 247⁴¹, 257⁴¹, 285⁵⁵, 439^{40,41}, 483¹², 491⁷⁰ Large, R., 770177 Larin, N. A., 318122 La Rochelle, R. W., 179546, 91821 Larock, R. C., 470^{204} , 473^{204} , 484^{23-25} , 489^{56} , $497^{23,56,100}$, 498^{100} , 500^{23} , $501^{23,24}$, 509^{23} , 1033^{73} Larossi, D., 95154, 119154 Larsen, E. R., 89875 Larsen, R. D., 57158 Larson, G. L., 20070, 41823 Laszlo, P., 125303, 142379, 30044, 31044, 31144, 322144, 582115, 80937 Latham, D. W., 643122 Latimer, L. H., 766157 Lau, K. S. Y., 53069, 53369, 53769, 541116, 543116 Laub, R. J., 79999 Lauer, R. F., 8763, 11463, 11763 Laurent, A., 38128, 38228, 63541 Laurent, E., 63541 Lauritzen, S. E., 86424, 86524, 87224 LaVaute, T., 5646 Lavé, D., 882¹⁰³ Lavie, D., 693144 Lavielle, S., 149408,409, 151408,409, 155408,409 Lavieri, F. P., 81578 Lavoie, A. C., 44684 LaVoie, E. J., 503149, 512149 Lawitz, K., 511188, 515188 Lawton, G., 902119 Layer, R. W., 324153 Layloff, T. P., 57266 Layman, W. J., 24644 Lazarow, A., 8225, 8345 Lazurin, E. A., 30681 Lê, T. Q., 87¹¹⁷ Leach, D. R., 48956, 49756 Leake, P. H., 507170 Leavitt, R. K., 46255, 47255 Le Belle, M. J., 97913 Le Berre, V., 66956 Lebioda, L., 3808, 62744 Leboff, A., 681100 Lebreton, J., 380¹⁰, 1009⁷⁴, 1010^{74,76,77}, 1011⁷⁸, 1012⁸⁰ Lebrun, A., 509177 Lechevallier, A., 765154 Lecomte, P., 111231 Leder, J., 466186 Lednicer, D., 957¹⁰⁸, 969¹⁰⁸ Ledon, H., 88927 Ledwith, A., 783²¹ Lee, A. O., 105²¹⁹, 113²¹⁹ Lee, A. W. M., 225¹⁸⁵, 264¹⁸¹ Lee, B., 68²¹ Lee, C. C., 825³⁰ Lee, D. C., 79266

Lee, G. C. M., 63055 Lee, P. L., 73629, 771190 Lee, R. A., 21¹²⁵ Lee, S., 167⁴⁸³, 738³⁴ Lee, S.-J., 172⁵¹⁴, 173⁵¹⁷ Lee, T. G., 643¹²¹ Lee, T. V., 26¹⁶¹ Lee, V., 89878 Lee, Y.-S., 41823 Lefferts, J. L., 20069 Lefker, B. A., 42²³⁴ Leftin, M. H., 44792, 513208 Legault, R., 770181 Legler, J., 4823 Le Goaller, R., 57266 Legris, C., 759130 Lehmann, C. H., 82956 Lehmann, R., 1947 Lehn, J.-M., 55739 Lehr, P., 53274 Lehrer, M., 692138 Leicht, C. L., 85791 Leighton, J. P., 52212 Leimner, J., 584132 Leitich, J., 81467 Lelandais, D., 63973, 648180, 65173b Lellouche, J. P., 41617, 41717 Lemaire, M., 229224 Lemal, D. M., 158435, 173435 Lemmen, P., 587¹⁴⁹ Lenain, V., 509177 Lenfant, M., 30175 Lenoir, D., 583¹²⁸, 584¹³¹, 587^{142,149} Lentz, C. M., 6²⁷, 8⁴¹, 9⁴¹, 15⁴¹, 17^{27,90}, 20⁴¹ Leonard, D., 103269 Leonard, J. A., 80727, 81361 Leonard, N. J., 38011 Leopold, E. J., 34940, 35140, 36383, 36940 Lepley, A. R., 655 Leriverend, M.-L., 765155 Leriverend, P., 765155 Leroux, Y., 104²¹⁰, 111²¹⁰, 242⁹, 257¹¹⁵ Lesimple, P., 19626 Lessi, A., 541¹⁰⁵ Lester, D. J., 505158 Leszczweski, D., 221129 Letendre, L. J., 226195, 264183, 38010 Lett, R., 147398, 149398,407,411,412, 150407,411, 151407,411, 152⁴⁰⁷, 153⁴⁰⁷, 288⁶³ Leu, L.-J., 328¹⁸⁰ Levason, W., 21351 Levenberg, P. A., 24¹⁴⁹, 25¹⁴⁹ Lever, J. R., 88¹³⁹, 89¹³⁹, 91¹³⁹, 211³³ Lever, O. W., Jr., 86²⁹, 121²⁹, 124²⁹, 125²⁹, 127²⁹, 25281.82, 25781,82 Leveson, L. L., 89141a Levey, S., 822⁵, 834⁵ Levif, G., 47²⁵⁶ Levin, A. I., 63983 Levin, D., 20176 Levin, D. Z., 30572.75a Levine, A. M., 735²⁰ Levine, R., 261^{147,150} Levisalles, J., 1366, 2166, 130, 22130, 21669, 41945, 42045, 738³⁶, 822¹¹, 834^{11,74,75} Levy, A. B., 25068, 262158

Levy, E. C., 693144 Levy, P. F., 643¹¹⁶ Lew, G., 48957, 49557, 52323, 799101 Lewars, E. G., 89248 Lewellyn, M., 62744 Lewin, A. H., 499¹¹⁴ Lewis, A., Lewis, A. J., 74255 Lewis, H. B., 507¹⁷³ Lewis, H. H., 888¹⁶ Lewis, J. B., 771¹⁸⁷ Lewis, J. J., 1893 Lewis, J. R., 660¹¹, 661²⁵, 688¹¹⁴, 690^{114,124}, 818⁹⁵ Lewis, M. C., 635³⁴, 638³⁴ Lewis, R. T., 21795,95a Lex, J., 593¹⁷⁹ Ley, S. V., 174^{527,527a,b}, 175^{527a-c}, 342¹¹, 351⁴⁴, 352⁴⁴, 380¹⁰, 1036⁸⁰ Lezzi, A., 52745, 53945, 54145 Li, H.-Y., 999⁵¹ Li, T., 369¹¹⁷, 372¹¹⁷ Li, Y., 596¹⁹⁴ Liang, G., 587145 Liang, W. C., 20118, 31118 Liang Chen, Y. P., 24148 Libertini, E., 583118, 587148 Libit, L., 24964 Libman, J., 38344 Lichtenwalter, G. D., 56517 Liebeskind, L. S., 47257, 48318, 50018, 67264, 67471 Liedhegener, A., 889²⁶, 890²⁹, 909¹⁵² Lien, A. P., 297²⁰, 298²⁰ Liepa, A. J., 575⁸³, 670^{57,57a,59}, 671⁶², 816⁸² Liétjé, S., 20183 Ligon, R. C., 35350, 35450 Lilley, G. L., 396106, 397106 Lilley, K. J., 396¹⁰⁶, 397¹⁰⁶ Lim, C.-E., 74256 Lim, H., 588¹⁶⁰, 610¹⁶⁰ Liman, U., 866²⁹ Lin, H. S., 56286 Lin, J. J., 21342,426 Lin, K., 178545, 181545 Lin, L.-J., 691 133 Lin, M.-T., 19628 Lin, S. T., 693146, 694150 Lin, W. S., 639^{76[']} Lin, Y., 767¹⁶³ Lind, J., 103892 Linday, L. B., 88137, 95137, 165137, 167137 Lindell, S. D., 226²⁰⁶, 369¹²⁵, 372¹²⁵, 373¹²⁸ Linderman, R. J., 19627, 21787, 263176 Lindert, A., 7³¹, 8³¹ Lindley, J., 52215 Lindsey, R. V., Jr., 648¹⁷¹ Lindstedt, E.-L., 21347 Lindsten, G., 499120 Lindy, L. B., 158444, 161444, 164444, 167444 Lines, R., 63424, 63924, 64924 Linhartová, Z., 89672 Linke, S., 794⁸⁰ Linstead, R. P., 351³⁸, 642¹¹², 643^{117,125,130}, 644^{112,147,148,148a,150,151,153,166} Linstrumelle, G., 215⁶⁵, 217^{91,94}, 218⁹⁶, 224¹⁶¹, 243¹³, 244²⁷, 247^{45,46}, 248^{45,55}, 251⁵⁵, 256¹¹³, 257^{114,116}, 262¹⁶¹, 263¹⁶¹, 264^{179,180}, 269⁵⁵, 416²⁰, 417²⁰, 437²⁷.

438³⁴, 440⁴³, 449⁹⁸, 464^{34,176}, 466^{34,176,188,192}, 485³⁰ 4874547, 49130,71,72, 52744, 53068, 53185, 53368, 53999, 545⁹⁹, 896⁶⁹, 963¹²³, 964¹²³ Lion, C., 124²⁶², 257¹²¹ Liotta, C. L., 3¹⁷, 7¹⁷ Liotta, D., 8654, 8754, 9454, 9554, 11754 Lipovich, V. G., 328172,173,174 Lipshutz, B. H., 20915,17, 21134a, 21234b,c 213^{34a,46,46c,48,54}, 214^{56,57}, 216^{46,68}, 223^{46,160} 224^{167,168}, 250^{70,72,73}, 251^{74,76}, 261⁷⁶, 264^{72,76,186}, 26572, 49170 Lis, L. G., 24145 Lis, R., 220¹²⁴ Lishchiner, I. I., 30572,75a Lissel, M., 28451 Litle, R. L., 890^{31,34}, 901¹¹¹ Little, R. D., 160468, 21456 Liu, C.-L., 172⁵¹⁴ Liu, H.-J., 221127, 78327 Liu, S., 105840 Livinghouse, T., 263177, 28761, 36281 Llera, J. M., 848⁵⁴, 849^{54b} Llewellyn, D. R., 723¹¹, 725¹¹, 726¹¹ Loacher, J. R., 89875 Lobkina, V. V., 30468 Lo Cicero, B., 1366, 2166 Lock, R. L., 906144 Loev, B., 297¹⁶, 500¹²⁷, 501¹²⁷ Loewe, M. F., 7239, 7549, 8167,68 Loftfield, R. B., 840¹⁴, 845³⁸ Logan, R. T., 84645 Logusch, E. W., 24144 Lohmann, J.-J., 7130,31,33 Löhmann, L., 975¹ Lohwasser, H., 902117 Lokensgard, J. P., 906145 Lömker, F., 825^{24,24a}, 828^{24a}, 835²⁴ Lomölder, R., 642115 Loontjes, J. A., 86841 Loots, M. J., 48314 Lopez, L., 222¹⁴⁵, 246⁴⁰, 257⁴⁰, 283⁴⁹ Lorne, R., 247⁴⁶, 416²⁰, 417²⁰ Lothrop, W. C., 505166 Lotvin, B. M., 648178 Lourak, M., 509179 Lovich, S. F., 52543 Lowe, G., 902118,118a Lowe, J. A., 35²⁰⁴, 66¹⁴ Lowe, J. U., Jr., 319¹³⁰ Löwe, U., 25¹⁶⁰ Lown, E. M., 89248 Lown, J. W., 92754 Lowndes, P. R., 3804 Lowry, B. R., 89034 Lubell, W. D., 44242 Lubinskaya, O. V., 3427 Luborsky, F. E., 29722 Lubosch, W., 6611, 7411, 1949 Lucchetti, J., 86⁵⁰, 87^{78.84}, 91¹⁴⁸, 92¹⁴⁸, 109⁸⁴, 120²⁴³, 136⁷⁸, 142²⁴³, 144⁷⁸, 145⁷⁸ Luche, J. L., 24965 Luche, M. J., 149408,409, 151408,409, 155408,409, 78329 Ludwig, J. W., 34193 Ludwig, U., 927⁵² Luedtke, A. E., 587¹⁴¹

Luengo, J. I., 996⁴⁶, 997^{46a}, 1000⁴⁶ Luftmann, H., 640109, 647109 Lukacs, G., 126316,317 Lukas, K. L., 159466, 166466, 174530 Luke, G. P., 48315, 50015 Lukenbach, E. R., 124²⁷⁶, 126²⁷⁶ Lukin, K. A., 864²¹ Lund, E. C., 102534 Lund, H., 5649, 5679, 63422 Lunn, W. H., 35346 Lunt, J. C., 643130, 644151,166 Lunt, L. C., 644147 Luo, F.-T., 1256,57, 44044, 460142, 463156, 48537, 48637, 491³⁷, 495³⁷, 497¹⁰⁵, 503³⁷ Luo, T.-T., 52543 Lupi, A., 71744 Lutomski, K. A., 53273, 503149, 512149 Lüttke, W., 587¹⁴² Lüttringhaus, A., 499113 Luu, B., 158443, 159443, 160443, 161443, 167443, 168443, 406140 Luyten, M., 3804 Ly, N. D., 261149 Ly, U. H., 810⁵⁰ Lygo, B., 174^{527,527a,b}, 175^{527a,b} Lyle, S. B., 325^{161,161a} Lyle, T. A., 369¹²¹, 372¹²¹ Lynd, R. A., 259134 Lyness, W. I., 147389 Lynn, J. T., 68198, 81681 Lyon, D. R., 38236 Lysenko, V. P., 8643, 17943 Lythgoe, B., 936⁷¹ McAdams, L. V., III, 883109 Macaione, D. P., 55747 McAndrews, C., 21125 McBee, E. T., 319131 McBride, B. J., 54276, 53789 McCague, R., 232260 McCapra, F., 681%, 689121, 81360 McCarry, B. E., 369118, 370113, 372118 McCarthy, J. R., 49273, 49773 McCarthy, K. E., 264186 McCartney, R. L., 57266 McCaulay, D. A., 297²⁰, 298²⁰ Macchia, B., 725¹⁹, 734⁹, 743⁵⁸ Macchia, F., 725¹⁹, 734⁸⁻¹⁰, 741⁴⁹, 743⁵⁸ McClard, R. W., 20181 McClory, M. R., 147³⁹⁷, 149^{397,400}, 150³⁹⁷, 151³⁹⁷ McCloskey, C. J., 1784, 588¹⁵⁸ McCloskey, J. E., 390^{75,76,78,79}, 392^{75,76,78} Macco, A. A., 367^{101,102} McCollum, G. J., 86418,22 McCombie, S. W., 25389, 261147, 26289 McConnell, W. V., 73515 MacCorquodale, F., 3809 McCowan, J. R., 49273, 49773 McCoy, P. A., 396¹⁰⁶, 397¹⁰⁶ McCrae, D. A., 31183, 132334, 133334, 136334 McCrae, W., 551¹ McCrindle, R., 71537 McCullagh, L., 909154 McCullough, K. J., 593177 McCurry, P., 99191, 1031914, 107191 McDaniel, W. C., 5647

McDonald, C. E., 220117 McDonald, P. D., 660²⁰, 661²⁰, 699²⁰ McDonald, R. N., 73939-41, 74874, 82847 MacDonald, S. J. F., 497¹⁰¹, 505¹⁰¹ Macdonald, T. L., 196^{23,30}, 223^{152,153} McDougal, P. G., 25284 McElvain, S. M., 61922 McEwen, W. E., 56412, 57476 McFarland, J. W., 158439 McFarland, P. E., 226201 McGarry, D. G., 61820 McGarvey, G., 254⁹⁸, 257⁹⁸ McGarvey, G. J., 18⁹⁷, 42⁹⁷, 43⁹⁷, 196²³, 254⁹⁸, 257⁹⁸ McGhie, J. F., 613², 615² McGlinchey, M. J., 38010 MacGregor, I. R., 324153 MacGregor, R. R., 1040109 McGregor, S. D., 158435, 173435 McGuirk, P. R., 24317, 24917, 26317,172 Machell, G., 83166 Machida, H., 594186 Machleder, W. H., 74154, 89252 McHugh, C. R., 79162 Maciejewski, L., 99034,34a Macielag, M., 51270, 81578 McIntosh, J. M., 46255, 47255, 38240, 73518, 99847 McKean, D. R., 232²⁶⁸, 495^{93a} McKenna, J. F., 316¹¹⁷, 317¹¹⁷ McKennis, H., Jr., 158437, 159437, 160437, 166437 McKenzie, A., 7215 MacKenzie, R. K., 38236 McKenzie, T. C., 363⁸⁷, 365⁸⁷, 497¹⁰¹, 505¹⁰¹ McKervey, M. A., 334²²⁰, 379³, 380¹³, 382^{3b}, 402¹²⁸, 422⁶⁸, 862⁸ McKillop, A., 54275, 41828, 499123,124, 505165, 507124, 670⁵⁸, 673^{58,69}, 693¹⁴¹ McKillop, T. F. W., 38667,68 McKinney, M. A., 783^{22,25} McKinnie, B. G., 88130, 90130 McLean, W. N., 229232 McMahon, R. J., 89143, 89243 MacMillan, J., 71543 McMorris, T. C., 226193 McMullen, G., 621³¹ McMurry, J. E., 1580, 20110-112, 21899, 21999, 232256. 239⁹⁹, 436¹⁷, 487⁴⁸, 489⁴⁸, 492⁴⁸, 495⁴⁸, 579¹²⁵, 582¹²⁵, 583^{118,124-127}, 584¹²⁵, 585^{124,125}, 586¹⁵², 587124,144, 588125,1278,152,159, 591165,166,167,168 592174.175, 594174.176, 595125, 596125, 610125,127a,152,165,166,167,168,176, 63056, 63156,58, 762145, 102536 McNeil, D., 623³³ MacNicol, D. D., 38236 MacNicol, M., 9131 McPhail, A. T., 407149, 57264 McPhee, W. D., 840¹³ McQuillin, F. J., 38122 McQuillin, J. F., 23¹⁴¹ MacSweeney, D. F., 34932, 42786, 71226 McWhorter, W. W., Jr., 225187 Madhava Reddy, S., 587¹⁴⁶, 610¹⁴⁶ Maeda, K., 496⁹⁹, 498⁹⁹, 511⁹⁹, 515⁹⁹ Maeda, M., 592¹⁷³ Maeda, T., 984²³, 985²³ Maekawa, K., 96168, 104168, 108168, 117168 Maercker, A., 778⁵

Maeta, M., 30356 Maetzke, Th., 41227 Mafunda, B. G., 216⁷³, 423⁷² Mageswaran, S., 92345, 94998, 95498, 963121 Maggio, J. E., 751⁸⁹ Magid, R. M., 257121, 467194 Magnenat, J.-P., 581109 Magnus, P., 27166, 8636, 105215, 15836, 16336, 16936, 173³⁶, 174³⁶, 198⁵¹, 217^{95,95a}, 255¹⁰⁵, 491⁷², 673⁷⁰, 674^{70b}, 816⁸⁰, 862², 1025³³, 1027⁴⁴ Magnus, P. D., 8648 Magnusson, G., 763149, 767165, 768149,166 Magoon, E., 81363 Mahachi, T. J., 56857, 57057, 599205,222, 602222 Mahalanabis, K. K., 44²⁴¹, 728³⁵ Mahatantila, C. P., 103683 Mahmoudi, M., 99034,34a Mahoney, L. R., 66124 Mahoungou, J. R., 420⁵¹, 421⁵¹ Maier, W. F., 25¹⁶⁰ Mais, R. H. B., 38349 Maiti, S. B., 85688 Maiti, S. N., 74151 Maitte, P., 324154 Majerski, Z., 74667 Majerus, G., 5647 Majetich, G., 20¹¹³, 212³⁹, 253⁹² Majewski, M., 50²⁶⁸, 589¹⁶², 610¹⁶² Majumdar, D., 19963 Majumdar, S. P., 78327 Mak, T. C. W., 55738, 87789 Maki, Y., 219107 Makisumi, Y., 9766 Makosza, M., 127321, 158446, 159446, 174528 Malacria, M., 44046 Malamidou-Xenikaki, E., 3809 Maldonado, L., 19739 Malhotra, S., 602²¹⁸, 607²¹⁸ Malhotra, S. K., 23138 Malinowski, M., 583¹²², 584¹³⁰ Mallamo, J. P., 155433, 157433, 230238 Mallart, S., 47²⁵⁶ Malmberg, H., 219114, 512195 Malova, T. N., 30465 Malte, A. M., 21790, 21990, 86315, 86415 Malwitz, D., 587¹⁴² Maly, N. A., 381³¹, 382³¹ Mamdapur, V. R., 416¹⁶, 417¹⁶ Mamedaliev, G. M., 30684 Marnedov, I. M., 30460 Mami, I. S., 68092 Manabe, H., 566²⁶ Manabe, K., 222144 Manchand, P. S., 168⁴⁹¹, 169⁴⁹¹, 171⁴⁹¹, 407¹⁴⁸ Mandai, T., 135^{345,346,347,348}, 136^{347,348}, 137³⁴⁷, 139^{347,348}, 141³⁴⁸, 144^{347,348}, 168⁵⁰⁸, 169⁵⁰⁸, 174³⁴⁵, 586¹³⁸, 639⁷⁵, 652²²⁰ Mandel, N. G., 135³⁶³, 136³⁶³, 139³⁶³, 142³⁶³, 156³⁶³ Mandelbaum, A., 741⁵¹, 745⁶⁵ Mandell, L., 57785, 57994, 595192, 640110 Mander, L. N., 21123, 71540, 93470 Mandolini, L., 55284 Mangeney, P., 223¹⁵⁴, 226²⁰⁷, 579¹⁰⁰ Mangold, R., 9147, 9247 Manion, M. L., 89146 Manmade, A., 74150

Mann, A., 35657 Mann, F. G., 382³⁶ Mann, G., 379³ Mann, I. S., 514²¹⁰ Mann, J., 840¹⁵ Manning, B., 889²⁴ Manning, D. T., 647¹⁹³ Manning, M. J., 254¹⁰² Manov-Yuvenskii, V. I., 103997 Mansilla, H., 391⁹⁰, 393⁹⁰, 395⁹⁷ Mantione, R., 104²¹⁰, 111²¹⁰ Mantlo, N. B., 255109, 512198 Mar, E. K., 225¹⁸⁶ Marburg, S., 262¹⁶⁶ Marcelli, M., 1046¹ March, J., 777¹, 908¹⁴⁷, 909¹⁴⁷ Marchese, G., 208², 217², 230^{235,236}, 283⁴⁹, 436¹¹, 441⁴⁸, 446⁸⁷, 449^{48,101}, 463^{153,154,155,166}, 485²⁹, 492⁷⁹, 493²⁹, 503^{29,79}, 513^{29,79} Mardis, W. S., 103207 Maréchal, E., 331 199 Marek, I., 223154, 226207 Marfat, A., 243¹⁷, 249¹⁷, 263^{17,172}, 500¹³², 505¹³² Marfisi, C., 89247 Margaretha, P., 57995 Margerum, D. W., 21351 Margulies, H., 804⁷, 809³⁸, 815⁷⁵ Marinas, J. M., 82524,246, 83524 Marinelli, E. R., 262¹⁵⁸ Marinelli, F., 539% Marinier, A., 38010 Marino, J. P., 21787, 226193,194,1948,196, 263176, 264183. 265190,191, 66748, 68748, 80734 Mark, H. B., Jr., 459138 Markezich, R. L., 370113 Märki, H.-P., 225¹⁸⁷ Märkl, G., 53186, 593178 Markó, I., 97912 Markovac-Prpic, A., 89888 Maroni, S., 75297 Marples, B. A., 74153 Marquet, A., 147390,398, 149390,398,407,408,409,410,411,412 150407,410,411, 151390,407,408,409,410,411, 152390,407, 153407. 155408,409,410 Marr, G., 160465, 164465, 166465 Marra, J. M., 6924 Marron, B. E., 61820 Marrs, P. S., 44357 Marsais, F., 261146, 264146 Marschall-Weyerstahl, H., 75292 Marsh, C. R., 52211 Marshall, J. A., 1153, 100202, 135365.366, 136365.366, 139365.366, 142365.366, 168202, 170202, 172202, 223149, 226^{197,199}, 356⁵⁵, 380^{8,10}, 394⁹⁶, 592¹⁷², 627⁴⁴, 750⁸⁶, 78323, 94389, 98321, 98421,21a, 98526b, 99439, 99951 1000^{51b}, 1008⁷³, 1009⁷⁴, 1010^{74,76,77}, 1011⁷⁸, 1012^{79,80} Marshall, J. L., 19102 Marsham, P., 75399 Marsi, M., 218101 Marsili, A., 74151, 74565 Marson, S., 99186, 107186 Martens, H., 332203 Martin, H.-D., 38236, 87260 Martin, J., 38013, 38667.68, 595191, 600213 Martin, J. A., 68093, 80728.30 Martin, L. D., 466190, 103266

Martin, M. M., 3793, 38236 Martin, P., 84850 Martin, R. A., 6145, 6175a Martin, S. F., 21128, 21528 Martin, S. J., 28350 Martin, V. S., 225185, 264181 Martina, V., 463154 Martinelli, L. C., 62953,54 Martinengo, T., 460141 Martinez, G. R., 91826 Martinez, R. A., 72733 Maruoka, K., 48313, 75086 Maruyama, K., 43²³⁶, 87^{97,98}, 99^{98,183}, 100^{98,183}. 105¹⁸³. 15798, 196³¹, 483¹⁰ Marvel, C. S., 825278 Marvell, E. N., 3804, 73519, 81363,64 Marx, J. N., 8045,9 Masada, H., 102116 Masaki, Y., 101506, 155428, 159462, 161462, 168497.506, 169506, 170497,506, 220115, 28656a Masamune, H., 223156 Masamune, S., 89465 Masamune, T., 125³⁰⁶, 126³⁰⁶, 735²² Mascareñas, J. L., 98321, 98421 Mascolo, G., 230²³⁵, 446⁸⁷ Massardo, P., 48960, 49560, 50460, 51160, 51560 Masse, J. P., 228²¹⁵, 436⁴, 484²⁶, 492²⁶, 494²⁶, 495²⁶, 50326, 51326 Massengale, J. T., 29716 Masson, S., 124260 Massuda, D., 25179, 25479 Mastalerz, H., 960115 Mastrorilli, E., 733¹ Masuda, S., 300⁴⁶, 302⁴⁶, 314¹⁰⁸, 318¹²⁹ Masuda, Y., 470196,197, 473196,197, 52221 Matar, S., 30459 Mateescu, G. D., 330192 Mateos, J. L., 901112 Mathey, F., 20182 Mathieu, J., 521³, 901¹¹² Mathre, D. J., 45²⁴⁹ Mathy, A., 30044, 31044, 31144, 322144, 80937 Matlin, S. A., 89460 Matoba, K., 85372 Matsubara, S., 48424, 50124 Matsuda, H., 88¹³³, 90¹³³, 95¹³³, 99¹³³, 101¹³³, 107¹³³ Matsuda, I., 262¹⁶³ Matsuda, S., 483⁹ Matsuda, T., 484²⁶, 492²⁶, 494²⁶, 495^{26,96}, 497¹⁰³, 503²⁶. 513²⁶, 530⁷⁸, 535⁷⁸, 757¹²³, 1026⁴¹ Matsue, H., 125306, 126306 Matsuhashi, Y., 168494,504, 169494,504, 170494,504 Matsui, K., 20285, 53997 Matsui, M., 99¹⁹¹, 107¹⁹¹, 287⁶², 644¹⁶¹, 715³⁹ Matsui, S., 174526 Matsui, T., 71432 Matsuki, Y., 100^{193,194}, 103^{193,194}, 107¹⁹⁴, 390^{69,70} Matsumiya, K., 34195, 35202 Matsumoto, H., 229225, 48316, 499116 Matsumoto, K., 650^{210c,212}, 651^{210c,216}, 721⁵ Matsumoto, M., 99¹⁸⁷, 107¹⁸⁷, 110¹⁸⁷, 698¹⁵⁷⁶ Matsumoto, S., 30354 Matsumoto, T., 100199, 223183, 225183, 30355,56, 38239, 38657, 400119-124, 402125,126,130,131, 404132,135,137, 405138, 55850, 71435

Matsumoto, Y., 1795

Matsumura, C., 555²⁹ Matsunaga, I., 172⁵¹⁶, 173⁵¹⁶ Matsunaga, S., 19852 Matsuoka, R., 257118 Matsuoka, Y., 64099 Matsushima, H., 402¹²⁶ Matsushita, H., 1255, 259138, 260139,140, 44352,53, 45353, 460¹⁴², 469^{198,199,200}, 470^{198,199,200,210}, 472¹⁹⁹. 473 198, 199, 200, 475 200, 210, 497 105 Matsuyama, H., 87573-75 Matsuzaki, E., 93463 Matteson, D. S., 19622, 19959,61-65, 7809, 79586.87, 796^{9,87-89}, 797^{9,86,87} Matthes, H. W. D., 406140 Matthews, R. S., 14^{74,75}, 15^{74,75}, 382³⁷, 384⁵¹, 393³⁷ Matthews, W. S., 863¹⁵, 864^{15,16,22}, 866¹⁶, 883¹⁶ Mattox, J. R., 322141 Matuyama, Y., 125²⁹⁷, 128²⁹⁷, 129²⁹⁷, 130²⁹⁷, 133²⁹⁷ Matz, J. R., 88^{129} , 89^{129} , 105^{129} , 112^{129} , 196^{28} , 591^{165} , $592^{174,175}$, 594^{174} , 610^{165} Maue, M., 31095, 31195 Mauli, R. M., 736²⁵ Mauzé, B., 20298 May, G. L., 505160 May, K. D., 87471 Mayer, B., 87260 Mayer, F., 324¹⁵⁰ Mayer, H., 168498,502,503, 169498,502,503, 698159 Mayer, R., 66643 Mayer, R. P., 726²⁴ Mayers, D. A., 74770 Maynard, S. C., 25152 Mayr, H., 3312005.202 Mazur, Y., 66952, 68352, 79164, 80514 Mazzocchi, P. H., 224172 Meader, A. L., Jr., 88819, 89119, 90019 Meakins, G., 264¹⁸⁴ Medici, A., 232²⁶⁷, 511^{186,187}, 514¹⁸⁶ Meerwein, H., 705², 725¹⁷ Meguriya, N., 44688 Mehrotra, K. N., 579⁹⁹ Mehta, G., 21¹²⁴, 384⁵⁴, 385⁵⁵, 604²³⁰ Meidar, D., 30045 Meier, G. P., 960116 Meier, H., 666⁴², 887¹¹, 891⁴², 892¹¹, 893¹¹, 897¹¹, 898¹¹, 900¹¹, 903¹¹, 905¹¹, 909^{155,156} Meijer, J., 217⁸², 254⁹⁶, 491⁶⁸, 531⁸⁴ Meinders, H. C., 552¹⁰, 559⁵⁷ Meinwald, J., 395¹⁰², 396¹⁰², 736²³, 737^{31,32}, 890³¹, 900⁹⁸, 901^{106,112}, 903⁹⁹, 906¹⁴³ Meinwald, Y. C., 90098 Meisels, A., 638⁸⁹ Meislich, E. K., 755110 Meister, M., 30990 Mekelburger, H. B., 313 Mekki, M. S. T., 325¹⁵⁸ Meklati, B., 771¹⁸³ Meldahl, H. F., 781¹⁸ Melega, W. P., 25496 Mellor, M., 35553, 35753, 769170, 771170 Meloy, G. K., 89144 Melvin, L. S., Jr., 86⁴¹, 87⁴¹, 178⁴¹, 179⁴¹, 288⁶³, 921³⁶ Melvin, T., 3809 Menapace, H., 381³¹, 382³¹ Menchen, S. M., 107²²⁶, 109²²⁶ Mendelson, L., 500132, 505132

Mendoza, A., 19964 Mentzer, C., 831⁶⁰ Menz, F., 89029 Menzel, I., 909156 Mérand, Y., 84643,44 Merényi, R., 87049 Meresz, O., 89141b Mergelsberg, I., 1040105 Merkel, C., 123250, 124250, 125250 Merlini, L., 691^{132,134} Merritt, V. Y., 62130 Mersereau, J. M., 640101 Mertes, K., 220120 Meshram, N. R., 328175 Mesnard, D., 420⁵¹, 421⁵¹ Messerotti, W., 38657 Messier, A., 25387 Mestdagh, H., 53891 Metcalf, B. W., 126320 Metlesics, W., 81254,58,59 Métra, P., 322144 Metz, F., 10489 Metzger, J., 505¹⁶², 507¹⁶², 512¹⁶², 698¹⁵⁹ Meyer, B., 587¹⁴² Meyer, E., 81252 Meyer, N., 8769, 19518, 255108 Meyer, A. I., 31^{185} , 32^{185} , 36^{209} , 37^{212} , $42^{231,232,233,234}$, $53^{272,273}$, 67^{18} , 68^{23} , $69^{24,25,27}$, 70^{25} , $71^{34,36,37}$, $72^{25,38-40}$, $74^{25,41}$, 75^{47-50} , $77^{55,56,58}$, 78^{60-62} , $79^{60,61,64}$, 81^{40,60,61,66-68}, 125²⁹¹, 126²⁹¹, 181⁵⁵², 255¹⁰³, 261103,156, 28865, 503149, 512149,197,202 Meyers, C. Y., 863¹⁵, 864^{15-18,20,22,23}, 866^{16,17,31}, 868³⁹, 883^{16,17} Meyers, M., 757¹¹⁹, 964¹²⁶ Meyers, M. B., 689¹²¹, 813⁶⁰ Meyers, P. L., 50²⁶⁵ Meyers, R. F., 370110 Miah, M. A. J., 242⁵ Miccoli, G., 44149, 48529, 49329, 50329, 51329 Michaelis, K., 644159 Michaelis, R., 644¹⁵⁹, 653²²⁷, 654²²⁷ Micha-Screttas, M., 88126, 95126, 96126, 107126, 109126, 123¹²⁶, 125¹²⁶, 824²¹, 825²¹ Michel, S. T., 19415, 19615 Micheli, R. A., 23142 Michelot, D., 256¹¹³, 257¹¹⁴, 485³⁰, 491^{30,72}, 934⁶⁸, 963¹²³, 964¹²³, 969¹³⁰ Michelotti, E. L., 229²³³, 246³⁸, 444⁶⁵, 446⁸¹, 447^{91,92}, 456¹²⁶, 492⁷⁵, 503⁷⁵, 513²⁰⁸ Midland, M. M., 274²¹, 421⁵³, 79998, 983¹⁶⁻¹⁸, 987^{29,30}, 990²⁹, 993²⁹ Midorikawa, H., 97177, 108177, 109177, 136369, 138369 Midura, W., 953105 Miethchen, R., 3221426,143 Migdal, S., 62845 Miginiac, L., 420⁵¹, 421⁵¹ Migita, T., 1262, 453113,114, 454120, 463160,164, 469216 470^{215,216}, 473²¹⁵, 476^{215,216}, 923^{43,44}, 934^{44,63}, 954⁴⁴, 100870,71 Mihailovic, M. L., 380¹³ Mikami, K., 215⁶⁴, 942⁸⁰, 976⁷, 984^{22,23}, 985^{22,23,26a}, 987³¹⁻³³, 992^{37,38}, 993³⁸, 994⁴¹, 999⁵¹, 1000⁵², 100257, 100459, 100561-63, 100865 Mikhail, G. K., 78537 Mikoiaczyk, M., 953105 Miles, D. E., 136373, 137373

Miles, D. H., 36493 Miles, W. H., 498¹⁰⁸ Millar, J. G., 223155, 224174 Millard, A. A., 44359 Millard, B. J., 92753 Miller, B., 803¹, 804⁷, 809^{1b,36,38}, 815^{72,75,76}, 817^{1b} Miller, D., 57785 Miller, D. B., 72⁴⁰, 81⁴⁰ Miller, D. D., 41227, 583126, 63056, 63156,58 Miller, E. G., 890³¹ Miller, J. A., 1367, 24635, 259133, 463156, 48640, 49540. 497⁴⁰, 498⁴⁰, 503⁴⁰, 553¹⁶ Miller, J. M., 54278 Miller, L. L., 66952, 68352,101, 685106,107 Miller, M. A., 85479, 905140 Miller, M. W., 753¹⁰³ Miller, R. B., 232²⁶⁰, 254⁹⁸, 257⁹⁸, 489⁶¹, 495^{61,90}, 50390, 50461,154, 51161,154, 51561,154, 602219 Miller, R. D., 25174, 62130 Miller, R. L., 483¹³ Miller, R. W., 407149 Miller, T. G., 80410 Millidge, A. F., 351³⁸ Millon, J., 24746, 466192 Mills, H. H., 38236 Mills, J. E., 672⁶⁴ Mills, O. S., 81043 Mills, R. W., 42786, 71226 Mills, S., 67¹⁷ Milne, G. M., 20174 Milstein, D., 453112, 463161,162, 504153 Milstein, N., 313¹⁰⁴, 316¹¹⁴ Mimura, T., 999⁵¹ Minachev, Kh. M., 30572 Minagawa, M., 466¹⁸⁷ Minakata, H., 222¹³⁷ Minakata, M., 62339 Minami, H., 99¹⁹⁰ Minami, N., 61715, 61915, 62115, 62315, 62715 Minami, T., 67265 Minato, A., 43725, 44025, 44825,96, 44925, 45025, 45125. 45225, 457129,130,131, 459137,139, 460130,137, 461137 48426, 49226, 49426, 49526, 497104, 50326, 510183,206, 51326,205,206 Minato, I., 63894 Minato, M., 38657 Mine, N., 29511, 30211 Minowa, N., 168501, 169501, 170501 Miocque, M., 147388, 151388 Mioskowski, C., 203102 Mirkind, L. A., 63544, 648172,173,174,175,183 Mirrington, R. N., 20109 Mirzaeva, A. K., 30989 Mishra, P., 46255, 47255, 79268 Mislankar, S. G., 24148 Mislow, K., 643¹²⁷, 927⁵⁶ Misra, R. A., 640¹⁰⁶ Misra, R. N., 759132, 760135 Misra, S. C., 613², 615² Misumi, S., 400119-122, 402126, 404135, 55635 Mita, N., 43722, 43822, 48536, 49136, 49436, 49736 Mitani, M., 599²⁰⁴, 600²¹⁷ Mitchell, J., 592174, 594174 Mitchell, M. L., 89144 Mitchell, R. H., 87^{104,105}, 95^{104,105}, 97¹⁰⁴, 109¹⁰⁵, 114¹⁰⁴, 116104.105, 505168, 87047.48, 87783.84, 92748

Mitchell, T. N., 49593b Mitoh, H., 232²⁶⁷, 510¹⁸⁵ Mitra, A., 982¹⁵ Mitra, R. B., 72940, 87891 Mitsue, Y., 1040¹⁰⁴ Miura, H., 85689 Miura, I., 135³⁴¹, 136³⁴¹, 137³⁴¹ Miura, K., 87154 Miura, M., 44679,80,82, 45679,80, 459136, 460136, 461136, 470^{79,80}, 471⁸⁰, 473⁷⁹ Miura, N., 543¹¹⁸ Miyake, F., 104²⁰⁸, 117²⁰⁸ Miyake, H., 769¹⁶⁸ Miyake, M., 38010 Miyake, N., 228222, 48426, 49226, 49426, 49526, 50326, 513²⁶ Miyamoto, K., 172516, 173516 Miyamoto, O., 168494,504, 169494,504, 170494,504 Miyamoto, T., 93979 Miyamoto, Y., 99190 Miyano, K., 686¹¹⁰, 815⁷⁷ Miyano, S., 219113, 503144, 505167 Miyao, A., 75188 Miyashi, T., 901107 Miyata, K., 84432 Miyaura, N., 221134, 231246,247,249,252, 24966, 27421. 446⁸⁹, 465¹⁷⁸, 469²⁰⁶, 470^{178,205,206,207} 473178,205,206,207, 48958,59,61,63,64, 49064,65, 49558,59,61,63-65,97, 49664,65,98,99, 49859,64,65,98,99, 50461,63,97,154, 51158,59,61,63-65,97-99,154 51558,59,61,63-65,97-99,154, 53057 Miyazaki, F., 683102, 80735 Miyazaki, S., 644¹³⁹ Miyazawa, Y., 875⁷³⁻⁷⁵ Miyoshi, M., 650^{210c,212}, 651^{210c,216} Miyoshi, N., 771¹⁸⁶ Mizugami, M., 461147 Mizuno, T., 10347 Mizuta, N., 55420 Mlinaric-Majerski, K., 87679.80 Mlotkiewicz, J. A., 404^{134,136} Mo, Y. K., 330¹⁹², 332²⁰⁶, 333²⁰⁹ Mock, W. L., 783105 Möens, L., 85790 Moerck, R. E., 62333 Moersch, G. W., 79059 Moest, M., 63647 Moffatt, M. E., 30571 Mohamad, S., 904134 Mohamadi, F., 34198, 39198 Mohan, R., 26162 Mohar, A. F., 63537, 63987 Moir, M., 404¹³³ Moir, R. Y., 147399 Moiseenkov, A. M., 181553, 7347 Mojé, S., 21790, 21990 Mokhtar, A., 83479 Molander, G. A., 49485, 52220, 52324, 57473, 57573, 59973,208, 61073, 75293 Mole, T., 89466, 89666 Molho, D., 83160 Molin, M., 147³⁹², 149³⁹², 152³⁹² Moll, H., 822¹², 831¹², 835^{12b} Möller, K., 904^{131,132} Mollere, P. D., 587¹⁴² Molnar, A., 72623

Moloney, M. G., 28656b Momongan, M., 34192, 39192 Momose, D., 42159, 42259 Momose, T., 81047, 85372 Mondon, M., 35²⁰¹, 39²⁰¹, 257¹²¹ Money, T., 26¹⁶⁴, 349³², 427⁸⁶, 681⁹⁶, 710²⁰, 711²², 712^{24,26} Moniot, J. L., 21676 Monkiewicz, J., 20178 Monn, J. A., 7129 Monot, M. R., 57892, 61092 Monro, M. H. G., 68194 Montanucci, M., 509178 Montecalvo, D., 38236 Monteils, Y., 321¹³⁶ Monteiro, H. J., 105226 Monti, L., 72519, 74358 Monti, S. A., 73044 Moody, C. J., 902119 Moody, R. J., 19961 Mooiweer, H. H., 21781 Moolenaar, M. J., 153414, 155414, 223183, 225183 Moon, S., 38013,16, 4141 Moon, S.-S., 226¹⁹³ Moore, D. R., 1787 Moore, H. W., 82848, 82948 Moore, J. A., 844³⁰, 889²² Moore, M. W., 44355 Moore, R. E., 43835 Moore, W. H., 73515 Moorthy, S. N., 38661, 39361 Moosmayer, A., 66123 Morales, O., 883¹⁰⁸ Morandini, F., 229^{230,230a}, 246³⁶, 438²⁹, 452¹¹⁰. 1023²² Moreau, B., 147³⁹⁸, 149^{398,408,409,411}, 150⁴¹¹. 151408,409,411, 155408,409 Morelli, I., 74151, 74565 Morera, E., 103579 Moretti, R., 20917 Morgan, J. W. W., 83159 Morgan, K. D., 168493, 169493, 171493 Morgan, L. R., 699¹⁶¹ Morgan, T., 224¹⁷² Morgans, D. J., Jr., 770¹⁷⁴ Mori, A., 197³³, 573⁷¹, 610⁷¹ Mori, I., 565¹⁷ Mori, K., 49²⁶², 99¹⁹¹, 107¹⁹¹, 124²⁶³, 126²⁶³, 224^{162,177}, 28762, 396115, 55737, 63986, 644161, 71539, 87154 Mori, M., 650²¹³, 1032⁶⁷, 1036⁸², 1037⁹⁰, 1038^{90,95,95b} Mori, S., 102640 Moriarty, R. M., 512203 Moriconi, E. J., 574⁸⁰ Morii, S., 426⁸², 428⁹¹, 429^{82,91} Morikawa, T., 421⁵⁴ Morimoto, H., 55749 Morimoto, T., 30461, 84434 Morisaki, K., 501137, 509137 Morita, A., 3809 Morita, Y., 426, 526, 1026, 225187 Moritani, I., 436⁵, 437⁵ Moriyama, T., 135³⁴⁸, 136³⁴⁸, 139³⁴⁸, 141³⁴⁸, 144³⁴⁸ Moriyama, Y., 741⁵⁰ Morizawa, Y., 730⁴³ Moroz, E., 901112 Morris, J., 81683 Morrison, G. A., 735²¹

Morrison, H., 89033 Morrison, J. D., 1², 2², 3², 13², 15², 23², 25², 30¹⁸¹ 34¹⁹¹, 35¹⁹¹, 37¹⁹¹, 39¹⁹¹, 41^{2,224}, 44², 45², 53²⁷³, 55², 76⁵⁴, 228²²¹, 341⁵, 365⁵, 374⁵, 436¹⁹ Morrow, D. F., 79059, 84640 Morrow, G. W., 695152 Mortikov, E. S., 30572,75a Morton, D. R., 370110 Morton, D. R., Jr., 766158 Morton, G. O., 767¹⁶³ Morton, H. E., 248^{52,53} Morton, J. A., 35144, 35244 Morton, T. H., 587141,142 Mortreux, A., 583121 Mosbach, E. H., 644149 Moser, J.-F., 36563 Moskowitz, H., 147388, 151388 Moss, N., 42161, 42261 Motherwell, W. B., 505^{158,159}, 594¹⁸⁷, 664³⁰ Motoki, S., 63974, 643123, 644146 Motter, R. F., 105²¹² Moulineau, C., 257121 Mouriño, A., 232²⁶², 545¹²¹, 983^{19,21}, 984²¹ Mousseron, M., 81364, 82847 Moyano, A., 3807 Moyle, M., 824¹⁸ Moynehan, T. M., 80727 Mpango, G. B., 50²⁶⁸ Mualla, M., 605²³² Muchmore, S., 21671 Muck, D. L., 63539 Mudd, A., 79163 Mühlstädt, M., 3793, 3806,12, 38659 Muhs, M. A., 84322 Muir, C. N., 74153, 74360 Mukai, C., 904134 Mukai, T., 901¹⁰⁷ Mukaiyama, T., 25¹⁵³, 86¹⁵, 96^{15,168,169}, 104^{168,169}, 108^{15,168,169}, 117^{15,168,169}, 125³⁰⁸, 226²⁰⁵, 227²⁰⁵, 28658, 42682, 42982, 5631, 56523, 57023, 58323, 585136, 598²⁰², 730⁴⁶ Mukherjee, D., 84644 Mukherjee, P. N., 329182 Mukherji, S. M., 325159 Mukhin, O. N., 329¹⁸³ Mulamba, T., 466¹⁸² Mulder, R. J., 909¹⁵¹ Mulhauser, M., 882¹⁰³ Müllen, K., 594¹⁸⁴ Müller, A., 322143, 81685 Muller, B., 43195,96 Müller, E., 324151, 4144, 55425, 5631, 66122,23, 66643, 891⁴², 909¹⁵⁵ Muller, F., 211³² Muller, G., 901112 Müller, N., 373130, 648188 Müller, St., 41²²⁷ Müller, U., 41228, 43239, 652222, 653227, 654227 Müller, W., 747⁷¹ Mullins, M. J., 91824, 957111, 960116 Mulvey, R. E., 763151 Mulzer, J., 224164, 225164 Mumtaz, M., 44241 Munasinghe, V. R. N., 597199 Mundy, B. P., 571⁵⁸, 596¹⁹³, 727^{30,31}, 728³⁷, 849⁵⁵ Mundy, D., 114²³⁴

Muñoz, H., 19849 Munoz-Madrid, F., 5648 Munro, M. H. G., 74150, 80731 Muntyan, G. E., 36180 Mura, A. J., Jr., 103²⁰⁵, 108²⁰⁵ Mura, L. A., 79479 Murabayashi, S., 586140 Murae, T., 39599, 402129 Murahashi, S., 437²³, 438³⁶, 440³⁶ Murahashi, S.-I., 222143, 223150, 24967, 259130, 4365. 4375.22, 43822, 48310, 48536, 49136, 49436, 49736, 1040104, 1041112 Murai, A., 125306, 126306, 73522 Murai, F., 57175, 57475 Murai, H., 89143, 89243 Murai, S., 771¹⁸⁶ Murakami, M., 226²⁰⁵, 227²⁰⁵, 463¹⁶⁷, 529⁵², 555²⁸ Muralidharan, S., 21130 Muraoka, O., 85372 Murase, H., 125306, 126306 Murata, E., 552⁸, 557⁴⁹ Murata, S., 25¹⁵⁹, 402¹³⁰, 404¹³⁵ Murata, Y., 159451, 160451, 161451, 402125, 405138, 71435 Murphy, D. K., 734^{3,5} Murphy, F. G., 691136 Murphy, G. J., 20298 Murphy, R., 799¹⁰⁴ Murphy, W. S., 131^{328,329,332}, 132³²⁹, 135^{328,329}, 683¹⁰⁴ Murray, M. J., Jr., 643121 Murray, P. J., 503149, 512149 Murray, R. W., 73624 Murray, T. F., 103164 Murray-Rust, J., 404134 Murray-Rust, P., 404134 Mursakulov, I. G., 34935 Murthy, K. S. K., 589162, 610162 Murtiashaw, C. W., 36386 Musgrave, W. K. R., 89874 Musliner, W. J., 74772 Musser, J. H., 762145 Musso, H., 66012, 66427 Mustafa, A., 887⁵, 897⁵, 900⁵, 903⁵ Mustafaev, E. Kh., 30682 Mustafaeva, M. T., 34214 Muth, K., 890³¹, 901¹¹¹ Muthanna, M. S., 89464 Muthard, J. L., 614⁶, 623^{6,32}, 627⁶ Muthukrishnan, R., 99182, 101182 Mwinkelried, R. I., 40²²³, 41²²³, 42²²³ Mychajlowskij, W., 257119 Myers, R. F., 36493, 39394 Mylari, B. L., 105122, 105222 Myong, S. O., 160468 Nabeyama, K., 38012 Nabney, J., 689119 Nacco, R., 124286, 125286, 127286 Nace, H. R., 758¹²⁵, 833^{70,71}, 854⁷⁶ Nachtwey, P., 563117, 582117 Nadebaum, P. R., 63650 Naef, R., 40^{221,222}, 41²²⁷, 212³⁸ Näf, F., 24856, 25156, 882104 Nagae, H., 84853 Nagai, Y., 729³⁹ Nagakura, 1., 212³⁹ Nagao, K., 784³¹

Nagao, Y., 125³⁰⁹, 217⁷⁹, 286^{56a}, 1050¹⁹ Nagarajan, S. C., 19415, 19615 Nagasawa, H. T., 849⁵⁸ Nagashima, T., 49²⁶³, 51²⁷¹, 198⁴⁷, 396¹¹⁵, 1033⁷⁵ Nagendrappa, G., 34¹⁹⁶, 380⁴, 386⁵⁸ Nagira, K., 102641 Nagy, T., 35²⁰⁵ Naik, R. G., 797⁹² Naito, S., 137376 Najafi, A., 700163 Najai, C., 647¹⁹⁴ Najera, C., 25390 Naka, H., 469²¹⁶, 470²¹⁶, 476²¹⁶ Naka, K., 1032⁷⁰ Nakadaira, Y., 23¹⁴³, 24¹⁴³, 380⁴ Nakagawa, M., 245³¹, 556^{34,36}, 585¹³⁷ Nakagawa, N., 20184 Nakahara, Y., 124²⁸⁰, 125²⁸⁰, 126²⁸⁰ Nakai, E., 985^{24,25a}, 987^{24,33}, 988^{33a}, 989^{25a}. 993^{25a} Nakai, T., 97^{174,175,176}, 103^{175,176}, 108¹⁷⁴, 109^{175,176}, 117¹⁷⁴, 136³⁷², 942⁸⁰, 976⁷, 977¹⁰, 984^{22,23}, 985^{22,25a,26a}, 986²⁸, 987^{24,31-33}, 988^{33a}, 989^{25a}, 992^{37,38}, 993^{25a,38}, 994⁴¹, 999⁵¹, 1000⁵², 1002⁵⁷, 100459, 100561-63, 100865 Nakaido, S., 919³², 923^{43,44}, 934⁴⁴, 954⁴⁴, 1008^{70,71} Nakajima, I., 43725, 44025, 44825, 44925, 45025, 45125, 452²⁵, 459¹³⁷, 460¹³⁷, 461¹³⁷, 484²⁶, 492²⁶, 494²⁶, 49526, 50326, 510206, 51326.206 Nakajima, M., 19741, 19941 Nakajima, N., 1041112 Nakajima, R., 501136 Nakajima, T., 135³⁴⁵, 174³⁴⁵, 300⁴⁶, 302⁴⁶, 313¹⁰⁵. 314¹⁰⁸, 315¹¹³, 318¹²⁹, 769¹⁶⁹ Nakajo, E., 225184, 262165 Nakajo, T., 901107 Nakamoto, Y., 313105 Nakamura, E., 6²⁸, 8²⁸, 14²⁸, 221^{131,132}, 455¹²⁵, 460¹²⁵, 623³⁴, 727³³ Nakamura, K., 147387, 149387, 150387, 151387 Nakamura, M., 136³⁷¹, 138^{371a}, 139³⁷¹, 140^{371,371a}. 143371,371a, 144371a Nakamura, S., 57371, 61071 Nakamura, T., 231242, 42048, 44361, 44561, 463159 Nakamura, W., 136374, 141374 Nakane, R., 30038 Nakanishi, K., 23143, 24143, 3804, 57581 Nakanishi, S., 565²¹ Nakano, K., 35202 Nakano, M., 968128 Nakao, K., 78970 Nakao, T., 222136 Nakashita, Y., 67781, 68681 Nakata, T., 83476 Nakatani, K., 105329 Nakatani, Y., 158443, 159443, 160443, 161443, 167443. 168443 Nakatsuji, S.-i., 585137 Nakatsuka, M., 450105 Nakatsuka, S., 261157 Nakatsuka, T., 452107, 460107 Nakaya, T., 664³², 665³² Nakayama, E., 91933, 95433 Nakayama, J., 57174,75, 57474,75, 586140, 594186, 883110 Nakayama, K., 91932, 92344, 93444, 95444, 100870.71 Nakayama, Y., 135348, 136348, 139348, 141348, 144348

Nakazaki, M., 592173, 62849 Nakazawa, T., 4839 Nallaiah, C., 61718 Nambu, H., 242⁶, 257⁶, 259⁶, 260¹⁴³, 794^{77,78} Nametkin, S., 72517 Namikawa, M., 402129 Namy, J.-L., 56735, 57035, 770172 Nantz, M. H., 140380, 154380, 168380, 174380, 176380, 20916, 22316 Naota, T., 437²³, 1041¹¹² Nap, I., 217⁸² Napier, D. R., 890^{31,34}, 901¹¹¹ Napolitano, J. P., 29612 Naqvi, S. M., 757¹²¹ Narasaka, K., 96^{168,169}, 104^{168,169}, 108^{168,169}, 117^{168,169}, 125308, 28658 Narayana, C., 1017⁵ Narayanan, K., 497¹⁰⁰, 498¹⁰⁰ Narimatsu, S., 262¹⁶⁴, 466¹⁹¹ Narr, B., 66643 Naruchi, T., 82957 Naruse, K., 227²¹¹, 243¹⁶, 245^{30,32}, 470²¹³, 476²¹³ Naruse, M., 274²², 799¹⁰⁰ Naser-ud-Din, 635³³, 640^{107,107a}, 643¹²⁰, 644^{120,134c}. 64733,107 Nasipuri, D., 36282 Naso, F., 208², 217², 230^{235,236}, 436¹¹, 441^{48,49}, 446⁸⁷. 449^{48,101}, 485²⁹, 492⁷⁹, 493²⁹, 503^{29,79}, 513^{29,79} Natarajan, R. K., 3807 Natekar, M. V., 87891 Nations, R. G., 82529 Nativi, C., 35657 Natsume, M., 512198 Nau, P. F., 85165 Naves, Y.-R., 349³⁶ Nawata, Y., 172516, 173516 Nayak, U. R., 406144, 600214 Ncube, S., 124²⁶¹, 126²⁶¹ Ndibwami, A., 38010 Nealy, D. L., 38117 Nechiporenko, V. P., 648175 Nechvatal, A., 34623 Necula, A., 331¹⁹⁶ Nedelec, L., 21131 Nederlof, P. J. R., 153414, 155414, 373129 Neeman, M., 73413 Neenan, T. X., 55746 Neeson, S. J., 955¹⁰⁶ Nefedov, B. K., 103997 Negashi, E., 104²¹¹, 107²¹¹, 111²¹¹ Negishi, A., 20288 Negishi, E., 7³², 8^{32,36}, 12⁵⁵⁻⁵⁸, 21¹²⁸, 208¹, 230²³⁹, 231^{241,243}, 232^{239a}, 233^{1b}, 238^{239a}, 251^{78,99,100}, 254^{78,99,100}, 259¹³⁸, 260^{139,140}, 266^{195,196}, 274²⁴, 436¹², 44044, 44351-55, 450103, 45353,103, 460142, 463156 469198,199,200, 470198,199,200,210, 472199, 473198,199,200. 475^{200,210}, 485³⁷, 486^{37,40-43}, 489⁵⁷, 491³⁷ 49537.40.43.57.91, 49740.105, 49840.42, 50337.40.150.151 5219, 52323, 5249,35,36, 5309,56, 79371,73,74, 79581,82,85, 799¹⁰¹, 1025³⁵, 1030³⁵ Negishi, E.-i., 25178, 25478, 260140 Negishi, Y., 454¹²⁰ Negrini, E., 511¹⁸⁷ Neidlein, R., 87364 Nelan, D. R., 698¹⁵⁸ Nelander, D. H., 83371

Nelke, J. M., 613¹, 614^{1,5}, 615¹, 616¹, 617^{5a,14}, 619¹, 620^{1,14}, 621¹, 622¹, 623¹, 625¹, 626¹, 627¹, 628¹, 629¹. 630¹ Nelson, A. L., 30463 Nelson, D. J., 35243b Nelson, J. A., 55²⁸² Nelson, R. E., 242² Nemoto, H., 51271, 164480, 165480, 167482, 168482, 19848, 226²⁰⁰, 390⁷⁹, 1008⁷², 1009⁷², 1010^{72,75} Nenitzescu, C. D., 295¹⁰, 321¹³⁷ Nes, W. R., 406143 Nesmeyanov, A. N., 8643, 17943 Nesterova, T. N., 30465 Nestler, G. N., 78114 Neubauer, H.-J., 53274 Neuenschwander, K., 257115 Neuklis, W. A., 79059 Neukom, C., 168⁴⁹³, 169⁴⁹³, 171⁴⁹³, 289⁶⁹ Neumann, H., 247⁴⁷ Neumann, P., 414² Neumann, S. M., 49484 Neumann, W. L., 79265 Neumann, W. P., 57159 Nevell, T. P., 706⁵ Neville, O. K., 822^{10,13}, 825²⁸, 829^{10,13}, 830⁵⁸, 831⁵⁸ Newcomb, M., 30³⁸¹, 31¹⁸⁵, 32¹⁸⁵, 34¹⁹³, 824²³ Newkirk, J. D., 84220 Newkome, G. R., 509179, 587143 Newman, M. S., 814^{70,71}, 888¹⁷, 890³², 891³⁹, 892³⁹. 894⁶¹, 898^{39,78} Newton, C., 369¹²⁵, 372¹²⁵ Newton, R. F., 29070 Ng, J. S., 20917, 21668, 224168 Ng, L. K., 66⁹ Nguyen, D., 216⁶⁸, 224¹⁶⁸ Nguyen, N. H., 503¹⁴⁹, 512¹⁴⁹ Nguyen, S. L., 224¹⁶⁷, 264¹⁸⁶ Nichikova, P. R., 643¹³¹ Nicholas, K. M., 21671 Nickel, S., 582112 Nickon, A., 164⁴⁷⁷, 386⁶⁶, 709¹⁴, 946⁹³ Nicolaou, K. C., 39²¹⁶, 86⁵⁶, 94⁵⁶, 117⁵⁶, 217⁹⁴, 220¹¹⁹, 224¹⁷⁴, 278³¹, 288⁶³, 289^{31,68}, 558^{51,52}, 618^{20,21}, 75190, 883106,107 Nieh, M. T., 583119 Niem, T., 369¹²², 372¹²² Niemann, C., 88924 Nierenstein, M., 88816 Nietzki, R., 82843 Nieuwland, J. A., 2739 Nigh, W. G., 551⁴, 552⁴ Nigita, T., 919³² Nikolaev, V. A., 89773 Nilsson, A., 67163, 67680, 68563 Nilsson, H. G., 645169 Nilsson, M., 213⁴⁷, 219¹¹⁴, 257¹²², 499^{119,120}, 512^{193,195} Nilsson, S., 639⁷¹ Nishida, A., 321139 Nishida, I., 6²⁹ Nishida, S., 380¹⁰, 583¹²⁷, 587¹⁴³, 676⁷⁷, 901¹⁰⁸ Nishide, H., 36178 Nishiguchi, I., 599²⁰⁴, 602²²¹, 726²⁵, 1034⁷⁷ Nishihata, K., 149^{401,402,405}, 150⁴⁰², 151^{402,405} Nishii, S., 222136 Nishikawa, K., 586¹³⁸ Nishimura, J., 31199, 464175

Nishimura, K., 39187 Nishimura, S., 460143, 461143 Nishimura, T., 93979 Nishio, M., 149401,402,405, 150402, 151402,405 Nishiumi, W., 102746 Nishiwaki, T., 42154 Nishiyama, A., 390⁷⁴, 392⁷⁴, 665³⁸, 691^{128,131}, 693^{128,139}, 697^{38,131,139,155,156}, 698¹²⁸ Nishiyama, H., 262¹⁶⁴, 466¹⁹¹ Nishiyama, K., 3809 Nishiyama, S., 67679, 690125, 69579 Nishizawa, M., 36178, 36597 Niwa, M., 369¹²¹, 372¹²¹, 390^{74,86}, 392⁷⁴, 395¹⁰⁰. 396^{113,115}, 397¹¹⁶, 398¹¹³, 769¹⁷¹ Nobayashi, Y., 27936 Noel, M., 2429 Noels, A. F., 1047^{7,8}, 1051^{7,8} Nogradi, M., 35²⁰⁷ Noguchi, I., 665⁴⁰, 681⁹⁵ Noguchi, M., 321139 Nohe, H., 63428 Noire, J., 19742 Nokami, J., 650²¹¹ Noltes, J. G., 20811, 21011,11a, 21911a, 23411a Nominé, G., 1265 Nomoto, T., 169510, 172510, 173510, 38011, 55636 Nonaka, T., 56479, 57479 Nonet, S., 95159 Nonhebel, D. C., 66326, 66526 Norell, J. R., 38120 Norin, T., 39073, 48961, 49561, 50461, 51161, 51561 Norinder, U., 594184 Norman, A. W., 81680 Norman, L. R., 8045 Norman, R. O. C., 501135 Norman, T. C., 223148 Normant, 1. C., 223¹⁰⁵ Normant, H., 123²⁴⁷, 194¹⁴, 242⁹ Normant, J. F., 202^{87,92,93,95}, 208¹⁰, 209²¹, 210¹⁰, 215^{21a}, 216⁷³, 217^{21,84}, 219^{21a,b}, 223¹⁵⁴, 224¹⁶⁶, 226^{203,207}, 227²¹⁰, 242¹, 243^{1,14}, 245³⁰, 246³⁵, 247⁴⁶, 249^{14,58,61}, 250⁷², 258¹²⁵, 259¹²⁹, 263^{14,173}, 264⁷², 265^{72,188}, 274¹⁹, 419^{39,41}, 423⁷², 440⁴², 441⁴², 442⁴², 464¹⁷², 466¹¹⁸, 193, 470³¹⁴, 323^{25,276}, 470^{314,173}, 253⁷², 258 466182,193, 470214,223,225,226, 473214,217,225,226 476^{214,217}, 482^{4,5}, 485^{4,31-35}, 486³¹⁻³⁵, 487⁴⁵ 49434,35,89, 498107, 4995, 5055, 5095, 51689, 52218,19, 525⁴¹, 579¹⁰⁰, 788⁴⁹ Norris, J. F., 30992c Norrish, H. K., 20176 Northington, D. J., Jr., 39081, 39281 Norton, J. R., 2083, 2133b, 103163,64 Norton, S. J., 84849 Nossin, P. M. M., 367¹⁰² Nott, A. P., 261¹⁵³ Notzumoto, S., 9766 Nouri-Bimorghi, R., 739⁴², 759¹²⁷ Novack, V. J., 203¹⁰¹ Novak, J., 464¹⁶⁸ Novakovskii, E. M., 643131 Novikov, N. A., 648175 Novikov, S. S., 88710, 88810, 88910, 89010, 89310, 89710, 90010, 90310, 103997 Noyori, R., 4²⁶, 5²⁶, 6²⁹, 9⁴⁷, 10^{26,47}, 25¹⁵⁹, 103^{203,206}. 108203, 402130, 404135, 771182 Nozaki, H., 96162,166,167, 99162,163,166,167, 103166,167 104162,163, 105166,216, 112216, 120162,163,166, 121162,163 131162.163, 155166, 20296.97, 21670, 24311, 24429, 24635,

25496,97, 259135, 27422, 27937, 34933, 35460, 35658, 358³³, 421⁶³, 445⁷³, 449⁷³, 484²⁴, 501²⁴, 565¹⁷, 730⁴³, 759¹³³, 787^{47,48}, 799^{100,103} Nozaki, Y., 10476 Nozawa, S., 42153 Nozawa, Y., 593¹⁸² Nozoe, S., 71329, 81679,79a Nozulak, J., 53274 Nsunda, K. M., 8781, 10481 Nugent, S. T., 21456 Numata, T., 953¹⁰⁴ Nunn, C. M., 21128, 21528 Nunokawa, Y., 470¹⁹⁷, 473¹⁹⁷ Nunomoto, S., 244²², 465¹⁷⁹, 494⁸⁹, 516⁸⁹ Nussim, M., 79164 Nutaitis, C. F., 31198 Nutt, R. F., 644¹⁶⁷ Nwaukwa, S. O., 167⁴⁸³ Nwokogu, G. C., 530^{70,71}, 534^{70,71} Nyberg, K., 634²², 637⁶⁴ Nye, S. A., 53894 Nyi, K., 573⁷⁰ Nyman, F., 38349 Nyns, C., 87049 Oae, S., 86^{7,19,31,35,42,44}, 104³¹, 121^{7,19}, 147³¹, 154³¹, 164³⁵, 173³⁵, 862³, 953¹⁰⁴ Obayashi, M., 20285, 25496, 56517, 759133 Oberhansli, P., 407150 Obrecht, J. P., 35245 O'Brien, M. J., 630 Obrzut, M. L., 103374 Ochiai, H., 221131, 231242, 42048,49, 44361, 44561, 463^{157,159}, 1023²³, 1033⁷¹ Ochiai, M., 168492, 169492, 21779, 28656a, 105019 O'Connor, B., 21128, 259132 O'Connor, D. E., 147389 O'Connor, S., 512¹⁹⁸ O'Connor, W. F., 57480 Oda, K., 747⁷² Oda, M., 421⁶⁰ Oda, R., 381²⁴, 382²⁴, 922³⁷, 924³⁷ Oda, T., 391⁸⁸ Odaira, Y., 3809, 63894 Odaka, T., 63974, 643123 O'Dea, J., 906145 Odham, G., 643129 Odic, Y., 733, 20119 Oehlschlager, A. C., 223155, 259136 Oei, H.-A., 56840, 56949, 60640, 60749 Oftedahl, E. N., 74772 Ogasawara, K., 34194, 224181, 93467 Ogata, T., 453¹¹³ Ogawa, A., 103477 Ogawa, K., 216⁷² Ogawa, M., 672⁶⁵ Ogawa, T., 124²⁸⁰, 125²⁸⁰, 126²⁸⁰ Ogawa, Y., 883^{106,107} Ognyanov, I., 39073 O'Grodnick, J. S., 734¹³ Ogumi, Z., 647¹⁹⁴ Ogura, F., 135353, 136353, 137353, 141353, 142353 Ogura, K., 135^{356,357,360,361,362}, 136^{356,357,360,361,362,370,371} $\begin{array}{c} \text{ura}, \text{ k.}, 133 \\ 137357,360,361,370, 138357,370,371a, 139356,360,361,362,370,371, \\ 140^{370,371,371a,b}, 142^{357,360,361}, 143^{357,360,361,371,371a,b}, \\ 144^{371a}, 155^{429}, 156^{360,361,362} \end{array}$

Oguri, T., 46²⁵³ Ohashi, K., 470²¹⁵, 473²¹⁵, 476²¹⁵ Ohashi, S., 466¹⁸⁷ Ohbayashi, A., 464175 Ohbuchi, S., 57582 Ohfune, Y., 20113, 38239, 400119-121, 402126 Oh-hashi, N., 99184 Ohira, N., 135353, 136353, 137353, 141353, 142353 Ohki, E., 816^{79,79} Ohki, M., 99¹⁹¹, 107¹⁹¹ Ohloff, G., 73629 Ohmizu, H., 597²⁰¹, 599²⁰⁴ Ohno, A., 147³⁸⁷, 149³⁸⁷, 150³⁸⁷, 151³⁸⁷, 463¹⁶³ Ohno, K., 1040105 Ohno, M., 72940 Ohsaka, T., 66749 Ohsawa, A., 461¹⁴⁶, 541¹⁰⁷ Ohsawa, H., 172^{511,512}, 173⁵¹² Ohshima, E., 883110 Ohshiro, Y., 67265 Ohta, H., 222¹⁴³ Ohta, T., 30678, 498109, 681100 Ohtomi, M., 82639 Ohtsuka, T., 400¹²⁰⁻¹²⁴, 402¹²⁵, 404^{132,135,137}, 405¹³⁸. 714³⁵ Ohtsuka, Y., 83476 Ohtsuki, K., 136371, 138371a, 139371, 140371,371a. 143^{371,371a}, 144^{371a} Ohuchi, Y., 24425, 26725, 49484 Ohwada, T., 306⁷⁸ Ohyama, T., 30039 Oida, T., 95¹⁵² Oishi, T., 95¹⁵⁷, 97^{172,173}, 114¹⁵⁷, 116^{157,172,173} Ojima, I., 92342, 94997, 102745 Ojima, J., 593^{181,182} Oka, K., 870⁵⁰ Oka, S., 147³⁸⁷, 149³⁸⁷, 150³⁸⁷, 151³⁸⁷, 463¹⁶³, 484²⁴, 501²⁴ Okabe, H., 135³⁶⁴, 139³⁶⁴, 142³⁶⁴, 143³⁶⁴ Okabe, K., 30678 Okada, H., 843²⁵ Okada, J., 55737,38 Okada, K., 35²⁰¹, 39²⁰¹, 644^{134a} Okada, M., 503142 Okada, N., 227²¹² Okada, S.-i., 503⁽⁴⁴ Okajima, T., 39079 Okamoto, T., 463¹⁶³, 484²⁴, 501²⁴ Okamoto, Y., 229²²⁷, 445⁷¹, 455¹²², 492⁷⁸, 495⁷⁸ Okamura, H., 446^{79,80,82}, 456^{79,80}, 459¹³⁶, 460¹³⁶, 461¹³⁶, 470^{79,80,224}, 471⁸⁰, 472²²⁴, 473⁷⁹ Okamura, N., 222¹⁴⁴ Okamura, W. H., 223147,148 Okano, M., 87111, 95152, 106111, 114111, 38126,27,29, 38226,27,29 Okano, T., 55529, 102746 Okarma, P. J., 590¹⁶³ Okawara, M., 97174,175,176, 103175,176, 108174, 109175,176, 117¹⁷⁴, 136³⁷², 169⁵⁰⁹ Okawara, R., 650211 Okida, Y., 644133 Okita, M., 1036⁸² Oku, A., 464175, 10476 Okubo, T., 67677 Okude, Y., 42163 O'Kuhn, S., 21897

Okukado, N., 230²³⁹, 232^{239a}, 238^{239a}, 260¹³⁹, 443⁵², 450103, 453103, 48642,44, 49544, 49842, 503150, 52435, 52543, 52953, 53054 Okumura, K., 100²⁰¹, 103²⁰¹, 107²⁰¹ Okuno, H., 89145 Olah, G. A., 294^{1,2}, 295^{2,6,7,9}, 297^{18,24}, 298^{2,28}, 299^{34a,b}, 300^{37,39,45}, 302⁵², 303^{58,58b}, 305^{71,76}, 311⁹⁹, 320¹³⁵, 321², 322^{142a}, 326¹⁶⁶, 327¹⁶⁸, 329¹⁸⁵, 330¹⁹², 331^{200a} 332^{204,205,206}, 333^{34,208,209,210,212}, 334^{213,215,216,221,221a,b}, 335^{1,6}, 339⁶, 421^{62,64}, 587^{142,145}, 706⁶, 1017⁷, 1046¹ Olah, J. A., 2959, 30037,39, 30358,58b, 322142a, 333209, 10177 Oliver, J. P., 2083 Oliver, M. A., 591170 Oliver, R., 219104, 52217 Oliveto, E. P., 23142 Olli, L. K., 901110 Ollis, W. D., 86^{21,22,32,33,37,38,48,62}, 104²¹, 121^{21,22} $147^{32,33}$, 154^{32} , $158^{37,38}$, 159^{37} , 163^{38} , 164^{37} , $173^{37,38,62}$, 419^{35} , 422^{68} , 660^{15} , 688^{115} , 780^{10} , 862^{2} 914¹⁰, 916^{16,19,20}, 919³⁰, 923⁴⁵, 926⁴⁷, 928⁴⁷, 930^{10,59}, 931¹⁰, 932⁶⁰, 933^{30,62}, 946³⁰, 949^{94-96,98,99}, 954⁹⁸, 963¹²¹ Olmstead, H. D., 418, 718, 818, 1118, 1470 Olmstead, M. M., 21343, 102534 Olmstead, T. A., 58292 Olofsson, B., 6347, 6367,76, 6377 Olsen, B. A., 85476 Olsen, D. J., 103891 Olson, D. H., 29614, 330194 Olson, G. L., 168⁴⁹³, 169⁴⁹³, 171⁴⁹³, 362⁹⁰ Olson, R. E., 87103, 104103, 106103, 111103, 117103 Olsson, L. I., 21782 Olszanowski, A., 619²³ O'Meara, D., 822², 831² Omizu, H., 602²²¹ Omote, Y., 1057³⁸ Onaka, M., 42682, 42982 Onami, T., 224¹⁸⁰ O'Neill, M. E., 208ⁱ Onishi, A., 168⁵⁰¹, 169⁵⁰¹, 170⁵⁰¹ Ono, M., 541¹⁰⁶, 735²² Onoe, A., 381²⁹, 382²⁹ Onuma, K., 24318, 24418, 24718, 464169 Ooi, T., 750⁸⁶ Oon, S.-M., 105635, 106235 Oostveen, J. M., 21785 Ootake, K., 159⁴⁵⁵, 161⁴⁵⁵ Opalko, A., 505¹⁶⁹ Oparaeche, N. N., 597199 Oplinger, J. A., 36386 Oppolzer, W., 1789, 46252, 159457, 173457, 178541, 19632, 728³⁵ Oprean, I., 41615, 41715 Orahovats, A. S., 103893 Orbaugh, B., 61613 Orena, M., 45²⁵⁰ O'Riordan, E. A., 131332 Orlinkov, A., 29725, 33425 Ornstein, P. L., 44683 Orsini, F., 42165 Ort, B., 586139 Ortar, G., 103579 Osawa, E., 38345, 38657, 402131 Osawa, K., 644^{134a} Oshima, H., 30354

Oshima, K., 96162,166,167, 99162,163,166,167, 103166,167 104162,163, 105166,216, 112216, 120162,163,166, 121162,163 131162,163, 155166, 35658, 44573,74, 44973, 48424, 50124, 583123, 73043 Oshino, H., 19733, 221131 Osman, A. M., 83478,79 Ostermann, G., 92135, 92235a,38, 92435a, 92752 O'Sullivan, R. D., 24423 Ota, S., 42890 Ota, T., 494⁸⁴ Otaka, K., 24644, 48527 Otera, J., 135^{345,346,347,348}, 136^{347,348}, 137³⁴⁷, 139^{347,348}, 141348, 144347, 348, 168508, 169508, 174345, 586138 Oth, J. F. M., 7546 Otsubo, T., 135353, 136353, 137353, 141353, 142353. 87783,85 Otsuji, Y., 565²¹ Otsuka, H., 42160 Ott, D. G., 82846 Ott, E., 284⁵⁴ Otvos, J. W., 332207 Otzenberger, R. D., 72731 Ouellette, D., 105120 Oughton, J. F., 80413 Ourisson, G., 158443, 159443, 160443, 161443, 167443. 168443, 38344, 406140, 5647 Ousset, J. B., 203102 Overberger, C. G., 649202 Overchuck, N. A., 30571 Overman, L. E., 7797, 7927,67,68 Owada, H., 87¹¹¹, 106¹¹¹, 114¹¹¹ Owen, D. M., 648¹⁸⁷ Owens, G. D., 213⁵¹ Owsley, D. C., 21790, 21990, 597200, 6131, 6141, 6151, 616¹, 619¹, 620¹, 621¹, 622¹, 623¹, 625¹, 626^{1,42}, 627¹, 628¹, 629¹, 630¹ Owston, P. G., 38349 Oyama, N., 66749 Oyamada, T., 83160 Oyler, A. R., 27834 Ozasa, S., 457¹³¹, 503¹⁴² Ozawa, F., 528⁴⁷ Pabon, H. J. J., 24962 Pabon, P., 104812 Paddon-Row, M. N., 421, 1221, 1896 Padwa, A., 168496, 25389, 255104, 26289, 580107, 104813 Page, P. C. B., 225188 Paget, W. E., 511189 Pagnoni, U. M., 38657, 39598, 75297 Paik Hahn, Y. S., 8049 Paisley, S. D., 220121, 222121c Pakhomova, I. E., 30572 Paknikar, S. K., 223157, 262159, 264159 Pally, M., 38013 Palmer, R., 71433 Palmquist, U., 67163, 67680, 68563 Palumbo, P. S., 174538,539, 177538,539, 86842-44, 86943,44, 87644 Pan, X., 596194, 63893 Pan, Y., 20916, 22316 Pandell, A. J., 317119 Pandey, G. D., 103894 Pandey, U. C., 36598 Panek, E. J., 41824, 4823 Panitkova, E. S., 63982, 64482,137

Pankratova, A. F., 30989 Panse, D., 744⁶¹ Pant, C., 224^{163,163a} Panyachotipun, C., 154424 Paolucci, C., 147396, 153396, 155396, 86527, 94490,91, 946⁹², 958^{90,112} Pap, A. A., 24145 Papahatjis, D. P., 39216 Papoula, M. T. B., 505158 Pappalardo, P., 49172 Pappo, R., 1152, 1752 Paquette, L. A., 158434, 159453, 160453, 163434,453,472,473, $\begin{array}{c} 168^{505}, 200^{67}, 254^{96}, 334^{222}, 380^{10,13}, 381^{33}, 383^{43}, \\ 445^{70}, 572^{62}, 586^{154,155}, 593^{177}, 610^{155}, 614^{6}, \\ 623^{6,32,33}, 626^{32b}, 627^{6,43}, 751^{89}, 786^{41}, 862^{4,6}, 864^{21} \end{array}$ 8664,6,30, 8676,33, 8684.6, 87256-58, 87333,57,66,67, 87470, 87681, 98320, 98420, 105841, 106241 Parady, T. E., 86416, 86616, 88316 Paredes, R., 89880 Parham, W. E., 105²¹² Park, J. D., 89875 Park, M.-K., 56516, 57816 Park, P., 125294, 126294, 167294, 168294 Park, S.-U., 82423 Park, S. W., 49261 Parker, D., 1894, 4294, 21354, 25174 Parker, D. A., 213⁴⁸, 224¹⁶⁷, 251⁷⁶, 261⁷⁶, 264^{76,186} Parker, K. A., 48²⁶⁰, 370¹¹¹ Parker, R. E., 7331 Parker, V. D., 67267, 67680 Parker, W., 38013, 38667,68, 399118, 404133,134, 600213 Parkins, A. W., 24423 Parks, G. L., 771188 Parlman, R. M., 102428 Parmigiani, G., 7349 Parnell, D. R., 55747 Parola, A., 564¹¹, 567¹¹ Parr, J. E., 88¹³⁷, 95¹³⁷, 165¹³⁷, 167¹³⁷ Parra, T., 396¹¹⁵ Parratt, M. J., 20285 Parrinello, G., 232268, 49593a, 102221 Parrish, D. R., 23142 Parry, R. J., 36493 Parsons, G. H., Jr., 56411, 56711 Partale, W., 88813 Partch, R. E., 38013 Partridge, J. J., 78323 Parvez, M., 58292 Pasau, P., 111231 Pascual, J., 5648 Pashayan, D., 580107 Passannanti, S., 6610 Pasto, D. J., 49167 Patai, P., 521¹, 551⁵, 552⁵, 556^{34,35} Patai, S., 86^{52,53}, 89⁵², 91⁵², 94^{52,53}, 95⁵², 114^{52,53}, 116⁵², 117^{52,53}, 119⁵², 208², 217², 223¹⁴⁶, 271², 272², 4367,10,11, 582111, 63419, 63819, 64919, 65519, 7226, 7236, 72623, 7316, 8228, 82848, 82948, 8368, 8399, 862^{3,7}, 867^{34,37}, 872³⁷, 883³⁷, 884³⁷, 887⁶, 889^{6,25}, 890²⁵, 893^{6.25}, 894⁶, 896⁶, 897⁶, 900⁶, 903⁶, 919²⁸ Pataki, J., 80515 Patel, B. A., 53998 Patel, D. K., 84647 Patel, K. M., 21125 Patel, P. P., 78322,25 Patel, S. V., 219106, 512201

Paterson, I., 25^{156,157}, 26¹⁵⁶, 28¹⁶⁸ Patil, D. G., 73733 Patra, A., 396112, 398112 Patra, S. K., 87115 Patricia, J. J., 41930, 42230 Patrick, D. W., 8763, 11463, 11763 Patrick, J. E., 94688, 99034 Patsch, M., 92752 Patten, A. D., 503149, 512149 Pattenden, G., 50²⁶⁵, 278³², 384^{52,53}, 400⁵³, 407¹⁴⁷, 43197,98, 49486, 585137, 586153, 587148, 591169, 603227, 604¹⁵³, 605²³¹, 610¹⁵³, 976⁹, 977^{9,9a}, 985²⁷, 989⁹, 990⁹ Patterson, J. W., Jr., 946, 1046, 20107, 8225, 8345 Patterson, W., 266196 Pattison, F. L. M., 63979 Pattison, V. A., 9753, 9803 Paugam, J.-P., 19955, 257117 Paulmier, C., 8651 Pauson, P. L., 66326, 66526, 102432 Payet, C. R., 580108 Peacock, N. J., 86839 Pearce, A., 34522 Pearce, H. L., 35200 Pearson, A. J., 220123 Pearson, C. J., 902119 Pearson, J. R., 371115 Pearson, R. G., 419, 21131, 21331 Pearson, W. H., 49380 Pecora, A. J., 1256 Pedersen, S. F., 579¹⁰², 596¹⁹⁶, 597¹⁹⁶, 610¹⁰² Pedler, A. E., 648187 Pedoussaut, M., 754109 Pedrini, P., 232²⁶⁷, 511¹⁸⁶, 514¹⁸⁶ Pedro, J. R., 87⁹⁹, 104⁹⁹, 851⁶⁴ Peel, M. R., 35461 Peel, T. E., 29723 Peet, W. G., 458134 Peevey Pratt, D. V., 104209, 117209 Pelcman, B., 511191 Pélerin, G., 599207 Peleties, N., 8764.65, 13664.65, 14464.65, 14564.65, 14764 Pelinski, L., 99034.34a Pelizza, F., 10177 Pelizzoni, F., 42165, 75297 Pellicciari, R., 85792 Pelter, A., 124²⁶¹, 126²⁶¹, 199^{56,66}, 261¹⁵², 274²³, 511¹⁸⁹. 514²¹¹, 554^{22,23}, 691¹³⁴, 780¹⁰, 797⁹⁵, 798⁹⁵⁻⁹⁷, 799⁹⁹ Peña, M. R., 487⁵¹, 495⁵¹ Pennanen, S. I., 39215 Pennetreau, P., 142379 Penning, T. D., 948, 1148 Pennings, M. L. M., 367101 Pennington, W. T., 57264 Penrose, A. B., 38668 Pentegova, V. A., 38668 Pepermans, H., 587148 Peppard, D. J., 43195,96 Perera, C. P., 103683 Pereyre, M., 3¹⁴, 7³³, 8¹⁴, 20¹¹⁹, 195¹⁶, 196²⁵, 453¹¹⁵, 524³⁰, 529³⁰ Periasamy, M., 1017⁵ Perkins, M. J., 505162, 507162, 512162, 8031 Pernet, A. G., 21676 Perrin, P., 41617, 41717 Perrin, R., 30683 Perriot, P., 20292,93,95, 78849

Perry, C. W., 35037 Perry, R., 28966 Pesce, G., 222145, 24640, 25740, 28349 Pesnelle, P., 43195.96 Petasis, N. A., 8656, 9456, 11756, 55851.52 Pete, J. P., 28142, 602223, 603228, 6133 Pete, J. R., 599²⁰⁹ Peter, R., 42158 Peters, J. A. M., 371116 Peterson, C. J., 57997 Peterson, D., 224¹⁸², 760¹³⁵ Peterson, D. J., 65⁴, 86^{57,58}, 88⁵⁸ Peterson, J. C., 380⁸, 627⁴⁴ Peterson, J. R., 78641 Peterson, M. L., 648171, 79688 Peterson, P. E., 28144, 380^{5.16}, 381^{16a}, 735¹⁷ Petit, F., 583121 Petit, M., 583¹²¹ Petit, Y., 21566, 225189, 190, 263174 Petragnani, N., 39²¹⁷, 87⁸⁵, 120²⁴¹, 136³⁷⁴, 141³⁷⁴ Petraitis, J. J., 242⁸, 257⁸ Pétrier, C., 24965 Petrovanu, M., 92136 Petrovich, J. P., 5649, 5679 Petrow, V., 84647 Petrzilka, M., 1789 Pettersson, B., 643129 Pettersson, T., 67163, 68563 Pettit, G. R., 126310 Pettit, R., 85582 Petty, E. H., 105124, 105224, 106044, 106244 Pfau, A. S., 27312 Pfeiffer, J. G., 38119 Pfeil, E., 825^{24,24a}, 828^{24a}, 835²⁴ Pham, C. V., 459¹³⁸ Pham, P. Q., 466185 Pham, P. T. K., 682¹⁶⁴ Philips, J. C., 883108 Phillipps, G. H., 81679 Phillips, B. T., 513204 Phillips, C. F., 75295 Phillips, H., 324150 Phillips, S., 381³² Phillips, W. V., 316, 1899 Phillipson, J. D., 7759 Pho, H. Q., 105635, 106235 Photis, J. M., 159453, 160453, 163453, 86733, 87333,66,67 Phuoc Du, N., 82847 Piancatelli, G., 512192.213, 515192.213 Piccardi, P., 48960, 49560, 50460, 51160, 51560 Piccolo, O., 229230,230a, 24636, 312102, 452110, 460141 Pichat, L., 462152, 513207 Picon, M., 2734 Pictet, J., 27312 Pierini, A. B., 505156 Pierre, J.-L., 57266 Piers, E., 20106,108,114,115, 21239, 21560, 24852-54, 25394,95, 3807, 44357, 48852,53, 48955, 49552,53,55 Pierson, W. G., 564¹⁰, 595¹⁸⁸ Pieter, R., 95178, 96178, 97178, 99178 Pietrasanta, F., 72729 Pietrusiewicz, K. M., 20178,79 Pigiere, Ch., 46254, 21566, 25175 Pignolet, L. H., 1041110 Pigou, P. E., 107227 Pilkington, J. W., 8046

Pillai, P. M., 74045 Pillay, K. S., 38011 Pillay, M. K., 73624 Pil'shchikov, V. A., 30465 Pim, F. B., 331198 Pine, S. H., 92136 Pines, H., 30991, 329188, 331197 Pinhas, A. R., 103685 Pinhey, J. T., 28656b, 505160 Pinnick, H. W., 88¹³⁸, 89¹³⁸, 159¹³⁸, 161¹³⁸, 164¹³⁸, 165138 Pino, P., 43829, 102322 Pinsonnault, J., 25157 Piquard, J. L., 120244, 142244 Pirrung, M. C., 713³⁰, 766¹⁵⁹, 942^{81a}, 1008⁶⁸ Pisanenko, D. A., 30570 Pistorius, R., 63762, 647170, 648170 Pitchford, A. T., 42152 Pitt, C. G., 125305, 126305, 127305 Pitteloud, R., 38010 Pitteroff, W., 19413 Pizzala, L., 89247 Plackett, J. D., 93362 Plakhotnik, V. A., 30575a Plashkin, V. S., 644137 Plata, D. J., 100764, 100864a Plattner, P., 27312 Plavac, F., 39220 Plénat, F., 73411, 74047,48 Plentl, A. A., 89879 Plieninger, H., 807²⁶ Ploner, K.-J., 39084, 39284 Plummer, M., 81578 Pöckel, I., 7054 Pocker, Y., 721⁵, 723¹¹, 725¹¹, 726¹¹, 736^{26,27}, 761¹⁴² Podlaha, J., 70915 Podlahová, J., 70915 Poeckel, I., 72518 Poeth, T., 482¹, 483¹, 491¹, 499¹ Pogonowski, C. S., 34197 Pohlen, E. K., 83164 Polishchuk, V. P., 647196 Polishchyuk, V. R., 647¹⁷⁷, 648¹⁷⁷, 649¹⁷⁷ Polizzi, C., 217^{80,80b}, 246⁴¹, 247⁴¹, 257⁴¹, 491⁷⁰ Poljakowa, A. M., 89249 Polla, E., 374132 Pollak, I. E., 258124 Pollicino, S., 147396, 149413, 151413, 152413, 153396.413, 155³⁹⁶, 865²⁷, 944^{90,91}, 946⁹², 958^{90,112} Pollini, G. P., 73837 Polonski, J., 691136 Polston, N. L., 48312, 48957, 49557 Polunin, E. V., 181553 Polyakova, S. G., 644140,141 Pommelet, J.-C., 87049 Pommerville, J., 78853 Ponomaryov, A. B., 219111, 53188, 53788 Pons, J.-M., 563¹, 572⁶³, 577^{85,86,91}, 607¹¹ Ponsford, R. J., 1051²³, 1056²³, 1062²³ Poochaivatananon, P., 154426 Poon, Y.-F., 493⁸⁰ Poorker, C., 747⁷⁰ Pop, L., 416¹⁵, 417¹⁵ Porta, O., 564^{6.30,31}, 566^{30,31}, 595¹⁸⁹, 606¹⁸⁹ Portella, C., 599209 Porter, B., 908146

Porter, R. D., 330192 Portnova, S. L., 36173 Posler, J., 55283 Poster, J., 33 Posner, G. H., 6²⁷, 8⁴¹, 9⁴¹, 15⁴¹, 17^{27,90}, 20⁴¹, 91¹⁴⁶, 155⁴³³, 157⁴³³, 208¹⁰, 210¹⁰, 212³⁹, 214⁵⁵, 216^{70,74}, 217⁸⁸, 226^{39a.d.201}, 230²³⁸, 248⁵⁵, 250^{70,72}, 251⁵⁵, 257¹²¹, 264⁷², 265⁷², 269⁵⁵, 419^{37,38.42,44.46}, 420^{44.46}, 421⁵⁶, 521^{5.6}, 522^{5.6}, 733¹ Pospišil, J., 664²⁹ Possel, O., 158449, 174449,523, 175449,523 Posthumus, T. A. P., 371116 Postlethwaite, J. D., 499118 Potter, S. E., 87788, 92748 Potts, K. T., 513²⁰⁸, 538⁹⁴ Poulter, C. D., 24643, 25843 Poulter, D. D., 131330 Pourcelot, G., 964125 Pourcin, J., 89247 Powell, D. W., 957¹¹⁰, 960¹¹⁴ Powell, L. A., 57788 Power, J. M., 21128, 21528 Power, P. P., 21343 Powers, P. J., 770178 Pozas, R., 901112 Prabhu, A. V., 316 Pradhan, S. K., 57264, 602224,225 Prager, R. H., 799104 Prakash, G. K. S., 2956, 29718, 29828, 329185, 332204. 334215,221,2216, 3356, 3396, 42164, 587142,145, 7066, 1046¹ Prangé, T., 691136 Prasad, C. V. C., 75190 Prasad, J. S., 589¹⁶², 610¹⁶² Prasad, J. V. N. V., 79794 Prasad, R., 30464 Prelog, V., 379^{2,3}, 564⁴, 566⁴, 849⁵⁷ Preobrashenski, N. A., 89249 Preobrashenski, W. A., 89249 Prestwich, G. D., 41²²⁵, 364⁹², 395¹⁰², 396¹⁰² Prewo, R., 313¹⁰³ Pri-Bar, I., 102114 Price, C. C., 299³¹, 300³¹, 309⁹⁰, 316¹¹⁸, 317¹¹⁸ Price, E. M., 178542, 179542 Price, J. R., 82853 Price, P., 783²⁴ Price, T., 103681 Priddy, D. B., 30462 Pridgen, L. N., 457132, 460140, 503143 Priepke, H., 100258 Priest, M. A., 3809, 62540 Priest, N. A., 3809 Prince, T. L., 582116 Principe, L. M., 102533 Prinzbach, H., 62131 Pristach, H. A., 325161 Procter, G., 224178 Prokopiou, P. A., 613², 615² Prout, K., 47257 Puckett, P. M., 348²⁷, 353⁴⁸, 355⁴⁸, 357⁴⁸, 358⁶⁸ Puckett, W. E., 509179 Puglisi, V. J., 56847 Pummer, W. J., 8043 Purrington, S. T., 88¹³⁹, 89¹³⁹, 91¹³⁹, 918²⁵ Pusch, J., 563¹¹⁷, 582¹¹⁷ Puterbaugh, W. H., 91515 Puttaraja, 38661, 39361

Pütter, M., 87260 Pyne, S. G., 57372, 59672, 59972, 60272, 60772 Pyun, C., 79792 Quader, A., 44572, 49277 Quan, P. M., 501138 Quast, H., 64095, 87045 Quayle, P., 174533, 177533, 179533 Queguiner, G., 261146, 264146 Quinkert, G., 56²⁸⁵ Quintard, J.-P., 314, 814, 19516, 19624.25, 453115, 52430, 529³⁰ Quirici, M. G., 21791, 48958, 49558, 51158, 51558, 539103 Qureshi, I. H., 600²¹⁵ Rabideau, P. W., 6134, 6194 Rabinovitz, M., 61926,27 Rabjohn, N., 63891, 64491 Rabo, J. A., 30575b Radhakrishnan, T. V., 602224,225 Radisson, X., 140380, 154380, 168380, 174380, 176380 Radwan-Pytlewski, T., 158448, 159448, 168448, 174529 Rae, I. D., 316115 Rae, W. J., 746⁶⁶ Raggon, J. W., 325¹⁶² Rahimi, P. M., 1046² Rahm, A., 195¹⁶, 524³⁰, 529³⁰ Rahn, B. J., 49485 Rainey, W. T., 72412 Rajagopalan, K., 3807 Rajamannar, T., 1788 Rajeswari, K., 30467 Raju, N., 21567, 3807 Rakiewicz, D. M., 221129 Rall, G. J. H., 693141 Ralston, A. W., 61716 Ramachandran, S., 3807 Ramachandran, V., 683101 Ramage, R., 681⁹⁴, 807³¹ Ramaiah, M., 598²⁰³ Ramamoorthy, B., 38342 Raman, H., 78114 Raman, K., 105430, 106044, 106146, 106230,44,46 Ramanathan, V., 261147.150 Ramaswamy, M., 27417 Rambaud, M., 20291.94 Rambault, D., 509177 Ramberg, L., 861 Ramirez-Muñoz, M., 882103 Ramphal, J. Y., 28968 Ramsey, J. S., 568⁴¹ Ran, R., 298²⁷ Rand, C. L., 104211, 107211, 111211, 25178,100, 25478,100, 44044, 44353.55, 45353 Rand, L., 63537, 63987, 652224 Randall, E. W., 38236 Rando, R. R., 89462 Ranganayakulu, K., 30467, 374133 Ransley, D. L., 318^{124,126} Ranu, B. C., 71431 Rao, A. S., 223157, 262159, 264159 Rao, B., 396¹⁰³ Rao, C. S., 652²²⁴ Rao, K. S., 21¹²⁴, 384⁵⁴, 385⁵⁵ Rao, P. N., 727³³, 757¹²⁴ Rao, V. B., 901116

Raphael, R. A., 226²⁰⁴, 273¹⁰, 382³⁶, 386⁶⁷, 551¹ Rapoport, H., 44²⁴², 507¹⁷² Rappe, C., 839⁹, 843^{24,29}, 844³¹, 892⁵² Rappoport, Z., $86^{52.53}$, 89^{52} , 91^{52} , $94^{52.53}$, 95^{52} , $114^{52.53}$, 11652, 11752.53, 11952, 2082, 2172, 2712, 2722, 43611, 521¹, 551⁵, 552⁵, 862^{3,7}, 867^{34,37}, 872³⁷, 883³⁷, 884³⁷ Rasmussen, R. R., 85371 Rassat, A., 5647 Raston, C. L., 4367 Ratajczak, A., 164475 Ratcliff, D. G., 97170, 117170 Ratcliffe, B. E., 35866, 36385 Rathke, M. W., 7³¹, 8³¹, 199^{60,66}, 257¹²¹, 443⁵⁹, 793^{69,72} Ratnasamy, P., 328175, 331195 Ratovelomanana, V., 21791, 43727, 44998. 48745-47. 53068, 53368, 53999, 54599, 94384, 964125 Ratts, K. W., 918²¹, 922³⁹, 939³⁹ Raubenheimer, H. G., 56627 Raucher, S., 87¹⁰⁸, 89¹⁴⁵, 90¹⁴⁵, 105¹⁴⁵, 106¹⁰⁸, 114¹⁰⁸, 120¹⁰⁸, 136¹⁰⁸, 144¹⁰⁸, 1000⁵⁴ Rauk, A., 147^{,399} Rausch, M. D., 56412 Rautenstrauch, V., 100198, 103198, 57051, 57264, 60651, 946⁸⁶, 976⁹, 977⁹, 989⁹, 990⁹, 1017⁴ Ravid, U., 24857, 24957, 25157, 26357 Ravikumar, P. R., 693¹⁴⁸, 694¹⁴⁸ Rawal, V. H., 135344, 136344, 137344, 138344 Rawlings, F. F., 643121 Ray, F. E., 91822 Ray, P. S., 219108 Rayner, C. M., 225188 Raynolds, P. W., 254102 Reagan, D. R., 223153 Reagan, J., 227²⁰⁹ Reale, A., 587¹⁴⁸ Reamer, R. A., 45²⁴⁷ Recca, A., 583118, 587148 Reday, P. A., 513207 Reddy, G. N., 21559 Reddy, G. S., 457¹²⁷ Reddy, K. Bal., 61820 Reddy, P. A., 459135 Redmore, D., 900%, 903% Reece, C. A., 126³¹⁸ Rees, C. W., 80727, 81361, 908148 Reese, P. B., 73835 Reetz, M. T., 25^{155,158,160}, 27¹⁵⁵, 421^{57,58} Regan, M. T., 1040108 Regis, R. R., 158447, 159447 Regitz, M., 887⁸, 888⁸, 889^{25,26}, 890^{25,28,29}, 893^{8,25}. 894⁵⁸, 897⁸, 900⁸, 903⁸, 905^{28,58}, 909¹⁵² Regondi, V., 125³⁰⁷ Rehnberg, N., 768166 Reich, H. J., 8646.47.54 8747.54,74.95,96,100.102.103,106,107,109,112.113 9454 9546,54.74,106, 9774, 10495,96,100,102,103, 105107, 106103,107 $109^{74,106}, 110^{95,100}, 111^{102,103}, 114^{47,74,107}, 116^{74}$ 11746.47,54,95,100.102,103, 120107, 13674,106,113, 14174,106, 145113, 157107,112,113, 24851, 38010, 78746 Reich, I. L., 87%, 104%, 24851, 38010 Reich, R., 2087 Reichstein, T., 89885 Reid, D. H., 86^{1,3,5} Reid, J. G., 960117 Reid, S. T., 81573 Reid, W., 902117

Reif, D. J., 746⁶⁹, 748⁶⁹ Rein, K., 7543, 7643,53, 8043 Reingold, I. D., 380⁸ Reitman, G. A., 30468 Reitman, L. N., 85586 Reitz, D. B., 65^{1,2}, 68², 69², 71², 88¹³⁰, 90¹³⁰, 194¹⁰ Reitze, J. D., 762147 Rej, R. N., 503149, 512149 Reliquet, A., 20287 Rémion, J., 8650 Remy, D. C., 38011 Renaldo, A. F., 232^{268,269}, 495^{93a} Renard, G., 73411 Renard, M., 8781, 10481 Renaud, P., 645¹⁶⁸, 650^{210b} Renaud, R. N., 643¹¹⁹, 644¹³⁸, 648^{138,182,184,189}, 649^{189,190} Renga, J. M., 957^{110,111}, 958¹¹³ Renneke, R. F., 10475 Resnati, G., 147393 Restivo, R., 38132 Retrakul, V., 154425, 155425 Reuman, M., 255103, 261103, 503149, 512149 Reusch, W., 21125, 23136, 24136, 595191, 74981 Reutov, O. A., 503152 Reutrakul, V., 154422,423,424,426, 155422,423 Rey, M., 914¹⁰, 930¹⁰, 931¹⁰ Reychler, A., 2429 Reynolds, D. P., 29070 Rhee, I., 554^{20,21} Rhyne, L. D., 499116 Ribo, J., 89247 Ricard, D., 74875 Ricca, D. J., 100053 Ricca, G., 42165 Rice, J. E., 503149, 512149 Rice, K. C., 7129 Rich, D. H., 369117, 372117 Rich, E. M., 888¹⁶ Richard, H., 38011 Richard-Neuville, C., 757124 Richards, G. N., 822², 831^{2,66} Richards, J. H., 34215, 104810 Richards, K. E., 74153, 74255,57, 74357 Richardson, G. M., 8225, 8345 Richardson, K. A., 26¹⁶¹ Richardson, L. J., 325¹⁶¹ Richardson, S. K., 24747, 25393 Richardson, W. H., 585134 Richey, H. G., Jr., 43835, 73514 Richman, J. E., 34¹⁹⁷, 135³⁵⁸, 136³⁵⁸, 137³⁵⁸, 139³⁵⁸. 142³⁵⁸, 143³⁵⁸ Richmond, R. E., 906144 Richmond, T. G., 213⁵¹ Rickards, R. W., 222135, 24748 Rickborn, B., 263¹⁷⁷, 734^{3,5}, 760^{136,138,141}, 761¹⁴¹, 762141, 763138,141, 764141 Rico, J. G., 25284 Ridby, J. H., 86⁶¹, 88⁶¹, 89⁶¹, 91⁶¹, 124⁶¹ Riddell, F. G., 404¹³⁴ Ridella, J., 19733 Ridley, D. D., 902118.118a Ried, W., 890.34 Ried, W. B., 582114 Riediker, M., 469²⁰², 470²⁰², 473²⁰² Riefling, B., 48425 Riehl, J. J., 74044

Rieke, R. D., 20919, 226202, 263171, 42155, 499116, 52210, 55317, 5631, 5701h Rieker, A., 661^{22,23}, 666^{42,43} Rieker, W. F., 67¹⁸, 69²⁵, 70²⁵, 72²⁵, 74²⁵ Ringold, H. J., 23138 Rink, H.-P., 505169 Ripley, S., 582116 Ripoll, J.-L., 84848 Risaliti, A., 503149, 512149 Rising, A., 806¹⁶ Riskibaev, E. R., 30358 Ritchie, E., 83159 Ritter, R. H., 1051²⁰ Ritterskamp, P., 21675 Rivierè, H., 7348 Rizk, M., 9768 Roach, B. L., 957110 Robbins, C. M., 380⁴ Roberto, D., 103786 Roberts, D. A., 159457, 173457, 510184 Roberts, D. H., 600²¹² Roberts, D. L., 407¹⁵⁰ Roberts, D. T., Jr., 334²¹⁴ Roberts, J. D., 382³⁶, 782¹⁶, 825³⁰ Roberts, J. L., Jr., 24643, 25843. 82423 Roberts, J. S., 386⁶⁷, 404^{133,134,136}, 405¹³⁹ Roberts, M. F., 77⁵⁹ Roberts, R. E., 123249 Roberts, R. M., 294⁵, 299⁵, 300⁵, 303⁵, 304⁵, 306⁸⁰, 323145, 324147, 325156, 327170, 328179, 329189 Roberts, S. M., 232²⁶⁶, 290⁷⁰, 488⁵⁴, 495⁵⁴, 985²⁷ Roberts, W. L., 91825 Robertson, A., 80725, 90089 Robertson, G. M., 586¹⁵³, 604¹⁵³, 605²³¹, 610¹⁵³ Robertson, J. M., 38668 Robertson, R. E., 62642 Robeson, C. D., 698158 Robien, W., 3807 Robin, J.-P., 501139, 509139,177, 512139 Robins, M. J., 219¹⁰⁹, 530⁷², 534⁷², 547¹²⁴ Robinson, E. D., 983²¹, 984^{21,21a}, 1011⁷⁸, 1012^{79,80} Robinson, G. M., 24642, 25842 Robinson, J., 4159 Robinson, R., 24642, 25842, 82853, 88814.15 Robson, M. J., 83268b Rochas, P., 315110 Roche, E. G., 28656b Rochefort, M. P., 691134 Rockell, C. J. M., 26161 Rodé, L., 35350, 35450 Rode, T., 509¹⁸¹ Röder, A., 62541 Rodewald, H., 87261,63 Rodewald, L. B., 635³⁴, 638³⁴ Rodewald, P. G., 29614 Rodgers, J. D., 960118 Rodgman, A., 407¹⁵⁰ Rodin, J. O., 126318 Rodina, L. L., 887⁷, 890⁷, 892⁷, 893⁷, 896⁷, 897⁷, 900⁷. 902⁷, 903⁷, 905⁷ Rodriguez, A., 94693 Rodriguez, H. R., 193², 194², 261¹⁴⁶, 264¹⁴⁶ Roe, D. K., 57787 Roe, R., 29613 Roelen, O., 10151 Roger, R., 7215

Rogers, A. O., 242² Rogers, D. Z., 7331 Rogic, M. M., 2426, 2576, 2596, 260143, 79477 Rohde, C., 1944 Rohm, A., 314, 814 Rohrer, C., 570⁵⁵, 582⁵⁵, 583⁵⁵, 630⁵⁷, 631⁵⁷ Röhrig, P., 593¹⁷⁹ Röhse, C., 322142b Rohwedder, W., 691¹²⁹, 693¹²⁹ Rokach, J., 273¹⁶, 274¹⁶, 277²⁷, 289^{66,67} Roller, G. G., 39393 Romadane, I., 31094, 31194 Roman, C., 257115 Romañuk, M., 643128 Römer, R., 62541 Romero, M., 45250 Rømming, C., 864²⁴, 865²⁴, 872²⁴ Romo, J., 80515, 84646 Romo de Vivar, A., 84646 Rona, P., 257¹²⁰ Rona, R. J., 224¹⁸², 759¹³² Ronald, B. P., 721⁵, 736^{26,27} Roncetti, L., 229230,230a Rondan, N. G., 1896, 31186, 6612, 7412, 1944,11 Rondarev, D. S., 63982, 64482.137 Rondestvedt, C. S., Jr., 497¹⁰¹, 505¹⁰¹, 889²⁴ Rondestvedt, E., 324¹⁵² Rone, A.-M., 27419 Ronlan, A., 66747, 67163, 67267, 67680, 68563 Ronman, P., 255¹¹¹ Rönneberg, H., 222135 Ronzini, L., 230235,236, 44148,49, 44687, 44948,101, 461145 463^{153,154,155,166}, 485²⁹, 492⁷⁹, 493²⁹, 503^{29,79}, 513^{29,79} Roobeek, C. F., 1022²⁰ Rooney, J. J., 334²²⁰ Roper, J. M., 587¹⁴³ Rosario, O., 20070 Röscheisen, G., 414⁴ Rose, B. F., 67990, 68390 Rose, C. B., 265¹⁸⁷, 390⁸⁰, 392⁸⁰, 396¹⁰⁵, 397¹⁰⁵ Rosen, P., 2⁸, 11⁸, 16⁸, 17⁸, 26⁸ Rosen, T., 21676 Rosenberger, M., 168491, 169491, 171491, 28969, 757122 Rosenblum, L. D., 41943 Rosenblum, M., 218¹⁰¹, 219¹⁰², 1036⁸¹ Rosenfeld, D. D., 89251 Rosenkranz, G., 80515 Rosenthal, R. J., 104813 Rosevear, J., 329¹⁸⁶ Rosewater, W. H., 499114 Roskamp, E. J., 579102, 596196, 597196, 610102, 94281b, 100867 Rösner, M., 55315 Ross, P. A., 770¹⁷³ Ross, R. J., 78641 Ross, S. D., 63421, 63538, 649206,206b, 65521 Ross, W. C. J., 341¹ Rossano, L. T., 100764, 100864a Rosser, A., 634³⁰, 644^{30b} Rossi, R., 21791, 221128, 43937, 48958, 49558, 505156 51158, 51558, 52538-40, 52745, 53945,103,104, 54145,105, 554²⁷ Roth, B., 30148 Roth, G. P., 255107 Roth, H. D., 89146 Roth, H. J., 737³⁰

Roth, K., 644159 Roth, Z., 229231 Rothenberger, S. D., 223149 Rothrock, T. S., 723⁸, 731⁸ Rotunno, D., 230236 Rouessac, F., 20104, 34210 Rougier, M., 57266 Roush, D. M., 178⁵⁴², 179⁵⁴² Rousse, G., 24638, 470221, 471221 Rouwette, P. H. F. M., 174524, 175524 Roux, D. G., 83161 Rowe, K., 798^{96,97} Rowe-Smith, P., 824¹⁸ Rowland, R. L., 407150 Rowlands, M., 261152, 514211 Roy, G., 198⁵¹ Roy, R. G., 84645 Royals, E. E., 42266 Rozen, S., 124²⁶⁴ Rubin, M. B., 597¹⁹⁸, 628⁵⁰ Rubino, M. R., 19415, 19615 Ruckle, R. E., Jr., 1052²⁸, 1053²⁸, 1055²⁸, 1057²⁸, 1062⁴⁷ Rudd, E. F., 634²¹, 635³⁸, 655²¹ Ruddock, K. S., 510¹⁸⁴ Rudisill, D. E., 232269, 52949 Rüegg, R., 698¹⁵⁹ Ruel, O., 25286 Rühlmann, K., 616¹¹, 618¹¹, 619¹¹, 620¹¹, 623¹¹, 625¹¹ Ruicheng, R., 29826 Ruitenberg, K., 49168, 53184 Ruiz-Hitzky, E., 770179 Ruiz Montes, J., 47256 Rukachaisirikul, V., 154422, 155422 Rusch, G. M., 757¹²⁰ Rusik, C. A., 47257 Rusling, J. F., 56632 Russell, C. E., 485³⁸, 486³⁸, 491³⁸, 492³⁸, 495³⁸, 503³⁸ Russell, C. G., 107²²⁶, 109²²⁶ Russell, G. A., 127321 Russell, R. K., 717⁴⁵, 752⁹⁴ Rüter, J., 890²⁸, 894⁵⁸, 905^{28,58} Ruth, J. A., 39496 Ruth, J. L., 470²¹⁸, 476²¹⁸ Rutledge, P. S., 672⁶⁴, 675⁷⁴, 759¹²⁶ Rutledge, T. F., 273⁸, 551², 552² Ruzicka, L., 341², 360², 705³, 781¹⁸ Ryabov, A. D., 499¹²⁵, 669⁵³ Ryan, J. D., 35868 Ryan, K. M., 45²⁴⁷ Ryan, M. D., 74877 Ryang, M., 483⁸, 554^{20,21} Ryde-Petterson, G., 63763 Rydjeski, D. R., 1477, 1577 Rynbrandt, R. H., 883109 Ryono, L. S., 126³¹⁹, 500¹³², 505¹³² Saá, J. M., 585¹³³, 591¹⁷¹ Saavedra, J. E., 658 Sable, H. Z., 7331 Sachdev, H. S., 136³⁷⁵, 141³⁷⁵ Sachdev, K., 136³⁷⁵, 141³⁷⁵ Sachs, D. H., 82849 Sacks, C. E., 135363, 136363, 139363, 142363, 156363 Sackville, M. A., 125³⁰¹, 126³⁰¹, 128³⁰¹, 129³⁰¹, 133³⁰¹ Sadhu, K. M., 79688 Sadykh-Zade, S. I., 30460

Author Index

Saegusa, T., 21786, 259137, 450105 Safarik, I., 89143, 89243 Saha, B., 908¹⁴⁶ Saha, M., 21671 Sahlberg, C., 44572, 49277 Sahli, M. S., 154419 Sahm, W., 482³ Sahraoui-Taleb, S., 78642 Saijo, K., 27830 Saiko, O., 81468 Sainsbury, M., 499110, 501110, 505110, 509110, 512110, 660¹⁸, 676⁷⁸, 686⁷⁸, 700¹⁶³ Saito, K., 42682.84, 42982 Saito, M., 43728 Saito, R., 594186 Saito, Y., 8797,98, 9998, 10098, 15798, 19631 Sakaguchi, S., 100055 Sakai, K., 400122 Sakai, S., 512¹⁹⁸, 570⁵⁶ Sakai, T., 84325, 84432,33 Sakaitani, M., 42160 Sakakibara, H., 91827, 968128 Sakakibara, T., 103999 Sakakibara, Y., 463¹⁶³, 484²⁴, 501²⁴ Sakakura, T., 529⁵¹, 1016³, 1039^{101,102} Sakamoto, T., 460143, 461143,147,148, 53064, 53364, 541 108-111,113,114, 543 108-111,113,117,118 Sakamoto, Y., 82639 Sakamura, S., 693149, 694149, 74772 Sakanishi, K., 587148 Sakano, M., 2958 Sakashita, T., 125308 Sakata, J., 629 Sakito, Y., 327167 Sakoda, K., 82957 Sakuma, K., 155428, 220115 Sakurai, A., 95157, 114157, 116157 Sakurai, H., 32187, 24644, 43728, 48527 Sakurai, K., 27830 Salaün, J. R., 62028, 765153, 767162, 84217, 84817 Saleh, S. A., 105225, 105925 Salem, G., 4367 Salemink, C. A., 464¹⁶⁸ Salen, G., 644149 Salerno, G., 42377 Salfeld, J. C., 505166 Salmin, L. A., 648^{172,173,174} Salvatori, T., 75297 Sam, T. W., 3809, 39072, 396111,114,115, 398111,114 Samain, D., 246³⁴ Samanen, J. M., 80734 Samkoff, D. E., 38010 Sammes, P. G., 89460 Sampath, V., 102534 Samplavskaya, K. K., 319132 Samsel, E. G., 103164 Sanchez, R. M., 82423 Sanda, F., 463157 Sandberg, B., 63969 Sandefur, L. O., 844, 1544 Sanderson, J. E., 643116 Sandri, E., 147396, 149413, 151413, 152413, 153396,413, 155³⁹⁶, 865²⁷, 944^{90,91}, 946⁹², 958^{90,112} Sandri, S., 45250 Sands, R. D., 72728 Sanechika, K., 530⁸⁰, 535⁸⁰

San Filippo, J., Jr., 24855, 25155, 26955, 41824, 41947, 482³, 494⁸⁷, 502⁸⁷ Sanfilippo, L. J., 8623 Sanjoh, H., 34212, 34712, 38130, 38230 Sankarappa, S. K., 3807 Sanner, C., 225190 Sano, H., 38121, 453113, 454120, 469216, 470215.216. 473215, 476215,216 Sano, T., 24425, 26725, 30043, 49484 Sanstead, J. K., 644154 Santarsiero, B. D., 21351 Santelli, M., 563¹, 572⁶³, 577^{85,86,91}, 607¹¹ Santiago, M. L., 88135, 90135 Santilli, D. S., 328176 Saotome, K., 63981 Sapino, C., Jr., 232264 Sarandeses, L. A., 232²⁶², 545¹²¹ Sarasohn, I. M., 8229, 8239, 8349 Sargent, G. D., 753100 Sarges, R., 124269 Sarkanen, K. V., 691130, 693130 Sarkar, D. C., 71431 Sarkar, M., 87115 Sarkar, T., 72835 Sarkar, T. K., 255105, 35656 Sarma, D. N., 38010 Sarmah, P., 36598 Sartori, G., 311100 Sasaki, H., 231²⁵² Sasaki, K., 48319, 50019, 50919 Sasaki, T., 383^{41,44} Sasazawa, K., 453114 Sass, V. P., 63982, 64482,137 Sasse, W. H. F., 509180 Sato, F., 41224, 223183, 224162, 225183, 24427.28, 463165, 464177, 48962, 49484, 49562, 50462, 51162, 51562 Sato, H., 23143, 24143, 3804, 94687 Sato, K., 168494,501,504, 169494,501,504, 170494,501,504 42682-84, 42890.91, 42982.91 Sato, M., 24427,28, 44573, 44677, 44973, 463165, 464177. 583123 Sato, N., 649^{204,205} Sato, O., 592173 Sato, S., 735²² Sato, S., 753-Sato, T., 124²⁵⁹, 125²⁵⁹, 153⁴¹⁶, 154⁴¹⁶, 218¹⁰⁰, 227^{210,211,212}, 243¹⁶, 245³⁰⁻³², 246³⁹, 257³⁹, 259¹³¹, 380^{9,11}, 446⁷⁸, 463¹⁵⁸, 470^{212,213}, 476^{212,213}, 565²³, 570²³, 583²³, 617¹⁵, 619¹⁵, 621¹⁵, 623¹⁵, 627¹⁵ Sato, Y., 918²⁷, 968^{128,129}, 969¹³⁵ 100⁴¹⁵, 105⁴¹⁶, 413⁴¹⁵, 105⁴¹⁶⁵, 105⁴¹⁶⁵, 105⁴¹⁶⁵ Satoh, M., 231246.247.249.252, 48964, 49064.65, 49564.65. 49664,65,99, 49864,65,99, 51164,65,99, 51564,65,99 Satoh, Y., 52325, 67786, 103787 Satou, M., 530⁸¹, 536⁸¹ Sattar, A., 404133,134, 405139 Saucy, G., 168491,493, 169491,493, 171491,493, 757122, 953101 Sauer, J., 57159 Sauers, R. R., 63890 Saugier, R. K., 99189, 107189, 110189 Saulnier, M. G., 261155 Saunders, J. O., 36386 Saunders, M., 7066 Saupe, T., 87782 Sausen, G. N., 1152, 1752 Sauvêtre, R., 498¹⁰⁷, 525⁴¹ Savard, M., 648¹⁸⁹, 649¹⁸⁹

Saveant, J. M., 57477

Savignac, P., 20182,83, 257117 Savoia, D., 168488, 169488 Savu, P. M., 50267 Sawada, H., 25178,100, 25478,100 Sawada, S., 381²⁶, 382²⁶ Sawamura, H., 30042 Sawamura, M., 54279 Sawicki, R. A., 38011 Sawyer, D. T., 82423 Sawyer, J. S., 19623 Sax, K. J., 892⁵¹ Sayed, Y., 125³⁰⁵, 126³⁰⁵, 127³⁰⁵ Sayer, T. S. B., 1899, 19100 Sayo, N., 97710, 98524,25, 98628, 98724,31 Scalone, M., 102322 Scarborough, R. M., Jr., 24149, 25149 Scettri, A., 512²¹³, 515²¹³ Schaad, L., 84538 Schaaf, T. K., 36491 Schaap, A., 262160, 263160 Schaefer, J. P., 61510 Schaer, B. H., 103892 Schäfer, H., 9754, 9794 Schäfer, H. J., 5649, 5679, 5989c, 63412,14,15, 640105,109, 642115, 64312, 644135,142,157,158,159,160,162,165, 646135, 647109,170,197,198, 648170,186, 649200, 651217, 65315,227, 654²²⁷, 904¹³¹, 975⁴, 979⁴ Schäfer, W., 592175 Schaffner, K., 21675, 81573 Schall, C., 63648 Schallner, O., 62332 Schamp, N., 85790 Schank, K., 174526, 753104, 8627 Schaper, W., 87788 Scharver, J. D., 38011 Schaub, F., 53062, 53262 Schaumann, E., 105²¹⁸ Schaus, J. M., 31182 Scheffler, K., 661^{22,23}, 666⁴³ Scheffold, R., 40²²³, 41²²³, 42²²³, 209²¹, 213⁴⁹, 215^{21a}, 217²¹, 219^{21a} Scheiner, P., 38117 Scheinmann, F., 28141 Schenk, H. P., 85791 Schenk, W. N., 969132 Schenker, K., 379², 849⁵⁷ Scherer, K. V., 85475 Scherrer, R. A., 42788 Schick, H., 55280 Schierloh, C., 56285 Schigeko, E. J., 53066, 53366 Schiketanz, A., 331196 Schill, G., 8785, 123250, 124250, 125250, 135359, 136359. 137359, 139359, 142359, 24745, 24845, 62847 Schilling, G., 87259 Schilling, P., 322142a Schinz, H., 345²⁰, 351³⁹, 358⁶⁵ Schinzel, E., 80724, 81366 Schinzer, D., 348^{30,31}, 358³¹ Schlecht, M. F., 226¹⁹⁴, 265¹⁸⁹, 380¹³ Schlecker, R., 1949 Schlessinger, R. H., 33190, 34197, 43238, 135358, 136358, 137³⁵⁸, 139³⁵⁸, 142³⁵⁸, 143³⁵⁸, 675⁷² Schleyer, P. v. R., 66¹², 74¹², 194^{4,11}, 334^{219,220} Schlieper, A., 822⁵, 834⁵ Schlingloff, G., 66746

Schlosberg, R. H., 29934b, 330192,193, 33334,208,211b, 334213 Schlosser, M., 99182, 101182, 1947, 21562, 24420, 261149. 466184 Schlubach, H. H., 55313 Schlunke, H.-P., 82212, 83112, 83512b Schmalz, H.-G., 56285 Schmalzl, K. J., 20109 Schmelzer, H.-G., 890³¹, 901¹¹¹, 904¹³⁵ Schmerling, L., 320¹³³, 331²⁰¹ Schmickler, H., 593179 Schmid, A., 223155 Schmid, B., 105²²⁰, 113²²⁰, 155⁴³² Schmid, H., 809^{39,40}, 957¹⁰⁹ Schmid, R., 369123,124, 372124, 957109 Schmidt, B., 580104 Schmidt, D., 89136, 909153 Schmidt, F., 55739 Schmidt, G. M. J., 38236 Schmidt, J., 82844 Schmidt, M., 134339, 135339 Schmidt, R. R., 105²²⁰, 113²²⁰, 155⁴³², 253⁹³ Schmidt, S. J., 102955 Schmidt, V., 903126 Schmierer, R., 45251 Schmit, C., 78642 Schmitt, H.-G., 174526 Schmitt, J. L., 95155 Schmuff, N. R., 168489, 169489 Schneider, D. F., 62332 Schneider, H.-P., 66642 Schneider, M., 587141 Schneider, R., 53062, 53262 Schneider, U., 49593b Schneiders, G. E., 219112, 67676 Schobert, R., 28659 Schoch, W., 91513, 96513 Schoenberg, A., 102113, 102847, 103478 Schoenewaldt, E. F., 822¹³, 829¹³ Schoening, C. E., 80412 Schöffer, A., 62541 Schofield, J. A., 81894 Schofield, K., 689118 Scholl, T., 73730 Schöllkopf, U., 53274, 19413, 19517, 19734, 30353, 41936. 921³⁵, 922^{35a,38}, 924^{35a}, 927⁵², 975⁴, 976^{5,7,8}, 979⁴ Scholz, D., 8629-11, 86311-13 Schönberg, A., 563¹¹⁷, 564¹¹, 567¹¹, 582^{111,112,117} Schönfelder, M., 4823 Schore, N. E., 102534 Schossig, J., 92135, 92235a,38, 92435a Schott, A., 482³ Schow, S. R., 26¹⁶⁵ Schrall, R., 691135 Schramm, S. B., 325^{161,161#} Schrecken, H., 4823 Schreiber, F. G., 693142 Schreiber, S. L., 21795, 227209, 545120, 97914 Schreibmann, A. A. P., 56413 Schröder, E., 84852 Schroeder, F., 174526 Schroeder, M. C., 87%, 104% Schröter, G., 887^{1b}, 890¹, 891^{1,1b}, 894¹, 897^{1b}, 905^{1b} Schubert, H., 39213 Schubert, M., 687112 Schubert, U., 369108 Schuchardt, J. L., 106045

Schuchardt, W., 72414 Schuda, A. D. C., 224172 Schudel, P., 698159 Schulte-Frohlinde, D., 66541 Schultz, A. G., 23135, 51270 Schultz, J., 67370, 67470b Schulz, H.-J., 509178 Schulze, K., 38238 Schumacher, J. N., 407150 Schumaker, R. R., 135340, 137340, 139340, 141340 Schumann, I., 102848 Schuster, D. I., 81573 Schuster, H. F., 223¹⁴⁶ Schütz, O., 582112 Schwab, P. A., 73939 Schwager, H., 587150 Schwall, H., 887⁸, 888⁸, 893⁸, 897⁸, 900⁸, 903^{8,126} Schwartz, A., 66748, 68748 Schwartz, A. M., Schwartz, J., 469²⁰², 470²⁰², 473²⁰², 483¹⁴, 1027⁴². 1030⁴², 1031⁶³ Schwartz, L. H., 73520 Schwartz, M. A., 55283, 67988.90, 68092, 682164, 68390 Schwartz, R. H., 24427 Schwarz, G.-U., 87259 Schwarz, H., 55²⁸⁰, 623³⁶ Schwarz, R. A., 159467, 166467 Schwarz, S., 55²⁸⁰ Schweiger, E. J., 883106 Schweitzer, H., 640104 Schwenk, E., 81255 Scott, A. E., 66015 Scott, A. I., 66014, 67991, 68196, 689119,121, 81360 Scott, C. J., 635³³, 640^{107,107a}, 647^{33,107} Scott, E. J. Y., 328179 Scott, F., 21673, 24314, 24914, 26314, 42372, 464172, 4 66¹⁸² Scott, S. W., 67988,90, 68390 Scott, W. J., 21899, 21999, 232256,258,270, 23999, 25071 436¹⁷, 446⁷⁵, 454¹¹⁶, 487⁴⁸⁻⁵⁰, 489⁴⁸, 492⁴⁸, 495⁴⁸⁻⁵⁰ Scovell, E. G., 34829, 35553, 35753, 38235, 769170, 770175, 771170 Screttas, C. G., 88¹²⁶, 95¹²⁶, 96¹²⁶, 107¹²⁶, 109¹²⁶, 123¹²⁶, 125¹²⁶, 194¹², 824^{20,21}, 825²¹ Scriven, C. E., 585135 Seager, J. L., 643121 Searles, S., 314107, 88924 Secrist, J. A., 28863 Sedelmeier, G., 62131 Sedrani, R., 226207 Seebach, D., 40^{221,222,223}, 41^{223,227,229,230}, 42²²³, 43^{237,240}. 44240, 653,7, 6611, 7130-33, 7411, 7546, 7963, 8618.20.27.28.59, 8764.65.68.69.75.82.88, 8859, 8959.68, 9418, 9518.158.178, 96158.178, 97158.178, 99158.178, 10420 10527,82, 11259,82, 12020,82, 12118,20,27,28,158, 12359, 12427,28,255,256,265,266,274,278,279,283,284 12520,27,28,59,255,266,274,278,279,283,284,290,313, 12620,279. $\begin{array}{c} 12727,28.265.266.274.283, \\ 129255.274.278, \\ 130^{322}, \\ 131^{324.331}, \\ 132^{278.283}, \\ 134^{28.322.338}, \\ 135^{324.338}, \\ 136^{64.65}, \\ 137^{338}, \\ 136^{54.65}, \\ 137^{338}, \\ 136^{54.65}, \\ 137^{338}, \\ 136^{54.65}, \\ 137^{338}, \\ 136^{54.65}, \\ 137^{338}, \\ 136^{54.65}, \\ 137^{338}, \\ 136^{54.65}, \\ 137^{338}, \\ 136^{54.65}, \\ 137^{338}, \\ 136^{54.65}, \\ 137^{338}, \\ 136^{54.65}, \\ 137^{338}, \\ 136^{54.65}, \\ 137^{338}, \\ 136^{54.65}, \\ 137^{338}, \\ 136^{54.65}, \\ 137^{338}, \\ 136^{54.65}, \\ 137^{338}, \\ 136^{54.65}, \\ 137^{338}, \\ 136^{54.65}, \\ 137^{338}, \\ 136^{54.65}, \\ 137^{338}, \\ 136^{54.65}, \\ 137^{338}, \\ 136^{54.65}, \\ 137^{338}, \\ 136^{54.65}, \\ 137^{34}, \\ 137^{34}, \\ 136^{54.65}, \\ 137^{34}, \\ 137^{34}, \\ 137^{34}, \\ 136^{54.65}, \\ 137^{34}, \\ 137^{34}, \\ 137^{34}, \\ 136^{54.65}, \\ 137^{34}, \\ 137^{34}, \\ 136^{54.65}, \\ 137^{34}, \\ 137^{34}, \\ 136^{54.65}, \\ 137^{34}, \\ 137^{34}, \\ 136^{54.65}, \\ 137^{34}, \\ 137^{34}, \\ 136^{54.65}, \\ 137^{34},$ 14428.64.65.381.382.384, 14564.65.381.382.384.385 14764.381.382.384, 1949, 224173, 24747, 255108, 257122, 5644, 5664.27, 56840, 56949, 580103,104, 60640, 60749, 645¹⁶⁸, 650^{210b} Seeman, J. I., 7545 Seese, W. S., 901112

Segal, R., 396¹⁰⁸, 397¹⁰⁸ Segal, Y., 896⁷⁰ Segi, M., 314¹⁰⁸, 318¹²⁹ Segnitz, A., 554²⁵ Seibel, W. L., 21561 Seibl, J., 1368, 91619 Seidel, M. C., 73731 Seidel, W., 644^{135,155,160,162}, 646¹³⁵ Seidig, K.-D., 89033 Seifert, C. M., 767163 Seifert, P., 57159 Seijas, J. A., 586¹⁵⁶, 591¹⁷¹, 610¹⁵⁶ Seiler, M. P., 99¹⁸⁵ Seitzinger, N. K., 97¹⁷⁰, 117¹⁷⁰ Sekine, M., 19741, 19941 Sekine, T., 649^{204,205} Sekiya, A., 244²³, 452¹⁰⁸, 503¹⁴⁷ Sekiya, K., 221131,132 Sekiya, M., 844³⁴ Selby, W. M., 61716 Self, C. R., 12⁶⁰ Selick, C., 21678 Seligman, A. M., 82849 Selim, A., 45251 Selleck, M. E., 326164 Sellén, M., 222139 Sellers, S. F., 63984 Selman, S., 8226, 8236, 8246, 82526, 8346.73, 8366 Seltzer, S., 52213 Seltzman, H. H., 125³⁰⁵, 126³⁰⁵, 127³⁰⁵ Semenovsky, A. V., 181553, 3427.8.14, 34521, 34624, 349³⁵, 351⁴¹, 361^{73,80} Semigran, M. J., 588¹⁵⁷ Semmelhack, M. F., 124254, 126254, 319, 137254, 35347, 423^{74,79,80}, 426⁷⁴, 427⁸⁵, 430⁹⁴, 483¹⁷, 499¹²⁸, 500^{128,132}, 505¹³², 509¹²⁸, 712²⁵, 1031^{65,65b} Sen, M., 741⁵¹ Sengupta, P., 74151 Seno, K., 125309 Sen Sharma, D. K., 30147 Seoane, E., 85164 Sepelak, D. J., 20299 Sépulchre, A. M., 126315,316,317 Serebryakov, B. R., 30466 Serelis, A. K., 35869 Sergeeva, O. R., 328^{173,174} Serra, A. A., 783²⁹ Serratosa, F., 3807 Servin, R., 6376 Set, L., 107²²⁹ Seter, J., 905138 Seto, K., 454¹²¹ Settine, R. L., 771¹⁸⁸ Seubert, J., 380⁴, 735¹⁹ Seuring, B., 95158,178, 96158,178, 97158,178, 99158,178, 121158. 124279, 125279, 126279 Sevrin, M., 86⁵⁰, 87¹¹⁰, 89^{110,140,141,143}, 90^{140,141} 91^{140,141,149,150}, 92^{140,143}, 105¹¹⁰, 106¹¹⁰, 107¹⁴⁰, 109¹⁴⁰, 114¹¹⁰, 116¹⁴⁰, 118^{141,143,237,238}, 120110,149,150,242, 123141,143,238, 25391 Seyferth, D., 86¹⁸, 94¹⁸, 95¹⁸, 121¹⁸, 200⁶⁹, 202^{98,99}, 208³, 1024³¹ Sgarabotto, P., 38657 Shadmanov, K. M., 318125 Shadrova, V. N., 643131 Shah, J. N., 125304, 126304

Shah, S. K., 8774.102.103.106.112.113, 9574.106, 9774, 104102.103 $106^{103}, 109^{74,106}, 111^{102,103}, 114^{74}, 116^{74}, 117^{102,103}.$ 13674,106,113, 14174,106, 145113, 157112,113 Shahak, I., 124264, 86735 Shamma, M., 68093, 80729 Shani, A., 244²¹, 247²¹, 391⁹¹, 464¹⁷³ Shank, R. C., 499121 Shankar, B. B., 25389, 261147, 26289 Shankaran, K., 265192 Shankland, R. V., 72413 Shanklin, J. R., 78640 Shanklin, M. S., 105635, 106235 Shapiro, H., 4157 Shapiro, R. H., 804², 810², 812² Sharma, G. M., 681% Sharma, K. S., 325159 Sharma, M. L., 41618, 41718 Sharma, R. K., 505¹⁶⁴, 507¹⁶⁴, 512¹⁶⁴, 515¹⁶⁴ Sharma, R. P., 36598, 38010 Sharma, S., 21346, 21646, 22346, 259136 Sharp, J. T., 255110 Sharp, M. J., 231250, 504155, 511155, 515155 Sharpe, L. A., 38010 Sharpless, K. B., 8763, 11463, 11763, 223156, 224180, 225^{185,191}, 264¹⁸¹, 583¹¹⁹, 770¹⁷⁴ Shavandin, Yu. A., 328181 Shaver, A., 56629 Shaw, A., 53077, 53577 Shcherbik, P. K., 643131 Shchukina, L. A., 82850 Shea, K. J., 466185 Shea, R. G., 104²⁰⁹, 117²⁰⁹ Shearing, E. A., 689117 Shechter, H., 161470, 167470 Sheffy, F. K., 232²⁷¹, 469²⁰¹, 470²⁰¹, 471²⁰¹, 473²⁰¹, 475201 Sheldrake, P. W., 28863 Shelton, G., 770180 Shemyakin, M. M., 82850 Shen, G.-Y., 223148 Shen, J., 29827, 332206, 333209 Shenk, W. J., Jr., 89032 Shenton, K. E., 103163 Shenvi, A., 369122, 372122 Shenvi, A. B., 28660 Shephard, B. R., 643⁽²⁵, 644^(47,15) Sheppard, W. A., 499122 Sher, F., 125299 Sherr, A. E., 842²¹ Sheta, A. E., 83478 Shibasaki, M., 650213 Shibata, A., 38130, 38230 Shibata, T., 102640 Shibutani, M., 593181 Shibutani, T., 510182 Shibuya, H., 75188 Shieh, H.-M., 41225 Shields, J. E., 760139, 774139 Shih, C., 256112 Shih, E. M., 73834 Shih, H. C., 503149, 512149 Shih, N. Y., 770¹⁷⁶ Shiina, K., 541106, 55850 Shimada, A., 31095, 31195 Shimada, K., 100196,197, 395101 Shimasaki, C., 2958

Shimazaki, T., 224162, 48962, 49562, 50462, 51162, 51562 Shimazaki, Y., 34194 Shimizu, H., 227²¹⁰, 245³⁰, 816⁷⁹ Shimizu, I., 53081,83, 53681,83, 594185 Shimizu, K., 38010 Shimizu, M., 6²⁸, 8²⁸, 14²⁸, 174⁵³¹, 176⁵³¹, 177⁵³¹, 179⁵³¹ Shimizu, Y., 453¹¹⁴, 463^{160,164} Shimomura, O., 74359 Shimuk, N. P., 328181 Shine, H. J., 803¹, 804¹² Shiner, C. S., 24144 Shinkai, I., 45247 Shinker, W. M., 66535 Shinoda, N., 39187 Shintani, Y., 501136 Shiobara, Y., 99¹⁹⁰ Shioiri, T., 46253, 90090 Shiono, H., 97174,175,176, 103175,176, 108174, 109175,176. 117174 Shiono, M., 24425, 26725, 470209, 472209, 475209, 49484, 730⁴⁶ Shiozaki, M., 71539 Shiragami, H., 27937 Shirahama, H., 100199, 38239, 38657, 400119-124 402125,126,130,131, 404132,135,137, 405138, 71435 Shirai, F., 98628 Shirai, N., 968129, 969135 Shiraishi, H., 567190, 595190, 607190 Shiraiwa, M., 541114 Shirlattl, V., 94693 Shirley, D. A., 511¹⁹⁰ Shiro, M., 105019 Shive, W., 39189, 39389 Shizuri, Y., 694151, 696151 Shlaefer, F. W., 905139 Shmelev, L. V., 34214 Shoji, F., 461147 Shokoka, E. A., 38132 Shone, R. L., 592¹⁷² Shono, T., 54279, 597201, 599204, 600217, 602221, 72625 Shoolery, J. N., 3807 Shoppee, C. W., 781¹⁹, 854⁷⁷ Shorter, J., 86734 Shroot, B., 262167 Shu, C.-S., 500127, 501127 Shubert, D. C., 49485 Shudo, K., 30678 Shukla, S. N., 63657 Shulman, J. I., 120240 Shulman, S., 105122, 105222 Shultz, A. G., 81578 Shuojian, J., 29826 Siahaan, T. J., 21234b.c Sibarov, D. A., 329183 Sibi, M. P., 242⁵, 683¹⁰⁴ Sicken, M., 593179 Siddall, J. B., 220121, 257120 Siddigi, S., 89248 Siddiqui, M. A., 231251 Siddons, P. T., 81679 Sidisunthorn, P., 80725 Sidler, J. D., 975³, 980³ Sidorova, N. G., 30989 Sieburth, S. M., 242^{7.8}, 257^{7.8} Siegel, A., 817⁸⁷ Siegel, H., 312101, 102218

Sieloff, R. F., 5647 Sierakowski, A. F., 502141 Sigg, H. P., 73625 Sih, C. J., 693148, 694148 Sih, J. C., 27938 Silveira, A., Jr., 7³², 8³², 12⁵⁶, 266¹⁹⁶, 524³⁶, 795⁸¹ Silverstein, R. M., 126318, 24857, 24957, 25157, 26357 Simmons, H. E., 380⁵ Simmons, H. E., III, 75189 Simon, R. M., 20299 Simon, W., 168503, 169503 Simondi, L. I., 5211, 5515, 5525 Simonet, J., 66956 Simonet-Guéguen, N., 66956 Simonsen, S. H., 39189, 39389 Simonson, J., 3411 Simpkins, N. S., 173⁵²⁰, 253⁸⁸ Simpson, J. H., 232^{259,259b,261} Sims, C. L., 93978 Sinaÿ, P., 174525, 19626 Sinclair, J. A., 274²¹, 554²⁴, 799⁹⁸ Singaram, B., 199⁶⁶, 793⁷⁵, 795⁷⁵, 797^{75,91,92} Singer, M. S., 164476 Singer, S. P., 8763, 11463, 11763, 86525, 957110,111 Singh, A., 421⁶¹, 422⁶¹ Singh, G., 125³¹³ Singh, J., 837, 602220 Singh, P. K., 30467 Singh, R., 38010, 541115 Singh, S. B., 104812 Singh, S. K., 2428, 2578 Singh, S. P., 73938, 74770 Singh, T., 81044, 105739 Singleton, E., 89667 Singleton, T., 464174 Sinha, G., 89886 Sinha, N. D., 219114, 499140, 501140, 502140 Sinha, S. C., 956¹⁰⁷ Sinisterra, J. V., 82524,24b, 83524 Sinoway, L., 25070 Sioda, R. E., 57789 Sipos, W., 87261,62 Siragusa, J. A., 56410, 595188 Siret, P., 57269, 57369, 60269, 60769, 61069 Siskin, M., 333^{211a,b} Sisti, A. J., 756115, 757116-120 Sitrin, R., 78114 Siu, T.-W., 87789 Siverns, M., 73835 Skaletz, D. H., 580^{105,106}, 596¹⁹⁵ Skameikina, T. I., 30684 Skattebøl, L., 86424, 86524, 87224 Skeean, R. W., 35143a Skell, P. S., 390⁸², 392⁸² Skerlj, R. T., 21560, 25394.95, 48955, 49555 Skinner, C. E. D., 255110 Skinner, C. G., 39189, 39389 Skipper, P. L., 158436, 159436 Skundín, A. M., 648181 Sladkov, A. M., 20920 Slater, M., 102744 Slates, H. L., 689119 Sliede, J., 83162 Sliwka, H.-R., 87469 Slobbe, J., 651²¹⁵ Sloneker, J., 691129, 693129

Slusarchyk, W. A., 68093, 80729 Smale, T. C., 1056³³, 1062³³ Smeets, F. L. M., 75083 Smeley, V., 89672 Smetanin, A. V., 103997 Smiles, S., 689117 Smirnova, V. V., 644140.141 Smissman, E. E., 75085, 78851, 84536,37 Smit, C. J., 56846,47, 56950 Smit, V. A., 3427.8.14, 34521, 34624, 34935, 35141, 36173.80 Smith, A. B., III, 22¹³³, 24¹⁴⁹, 25¹⁴⁹, 26¹⁶⁵, 709^{17.18}, 906144 Smith, A. L., 75⁵¹, 77⁵⁷, 78⁵¹ Smith, B. H., 62848 Smith, C., 91930, 93330,62, 94630, 94995 Smith, C. A., 21795, 681100 Smith, C. W., Jr., 265¹⁸⁷ Smith, D. J. H., 159459, 163459, 102850, 103684 Smith, D. R., 825³⁰ Smith, G., 81679 Smith, G. A., 7543, 7643, 8043.65, 8165, 685108 Smith, G. G., 82846 Smith, G. P., 328177 Smith, G. V., 86416, 86616, 88316 Smith, H. A., 1579 Smith, J., 54277 Smith, J. G., 223157, 262159, 264159, 500132, 505132, 57998 Smith, J. K., 31185, 32185 Smith, K., 124261, 126261, 55422.23, 78010, 79795, 79895-97 Smith, L. R., 901106 Smith, M., 7344 Smith, P. A. S., 78111, 82314, 83514 Smith, P. J., 79479 Smith, R. G., 124277 Smith, R. S., 437²⁰, 438²⁰, 439²⁰, 441²⁰, 442²⁰ Smith, S. A., 349³⁴, 623³⁷, 625³⁷ Smith, S. G., 67^{17,20} Smith, W. B., 647^{179,191,192,193}, 648¹⁷⁹, 724¹² Smith, W. E., 723¹⁰ Snaith, R., 763151 Sneen, R. A., 1264 Snell, J. M., 61922 Snider, B. B., 244²⁶, 485²⁸, 978¹¹, 994⁴⁴ Snieckus, V., 44241, 50268, 231250,251, 2425, 265192, 504155, 511155, 515155 Snowden, R. L., 196³², 264¹⁸⁴, 345²² So, J.-H., 56516, 57816 So, Y. H., 683101 Sobota, P., 5631 Sobti, R. R., 71537 Soderquist, J. A., 25283 Soejima, T., 54279 Sohani, S. V., 602225 Sohda, T., 8168 Sokoloff, S., 396¹⁰⁸, 397¹⁰⁸ Sokolov, S. V., 63982,83, 64482,137 Solaja, B., 313103 Solans, X., 380⁷ Solheim, B. A., 395¹⁰², 396¹⁰² Solladié, G., 86³⁴, 147³⁴, 149³⁴, 150³⁴, 151³⁴, 343¹⁶, 439³⁹ Solladié-Cavallo, A., 315112, 317112 Solomon, M. D., 39071, 399117, 402117 Solomon, M. F., 1051²², 1052²² Solomon, P. A., 382³⁷, 393³⁷ Solter, L. E., 159467, 166467

Sólyom, S., 34830,31, 35831 Soma, N., 91933, 95433 Soman, R., 94383, 95383 Somers, P. K., 75190 Somers, T. C., 828^{52,53} Sommelet, M., 91411, 96511 Sommer, J., 2956, 29718.24, 3356, 3396 Sommer, L. H., 41932, 85584 Sondheimer, F., 27310, 80514 Sonehara, S., 21672 Song, Z., 67²⁰ Sonnet, P. E., 45245 Sonobe, A., 466¹⁸⁷ Sonoda, A., 222¹⁴³, 249⁶⁷, 259¹³⁰, 483¹⁰ Sonoda, N., 771186, 103477 53161, 532^{7,61}, 537⁶¹, 541¹⁰⁶, 552⁸, 554²⁶, 557^{48,49}, 558⁵⁰ Sonogashira, K., 21789, 21989, 2711, 5217, 5307.61 Soong, J., 639⁷⁶ Sooriyakumaran, R., 466189 Sorensen, T. S., 374¹³³, 708¹³ Sorm, F., 643¹²⁸ Soroos, H., 4156.7 Sotheeswaran, S., 154424 Sotirious, C., 72938 Sotoyama, T., 262158 Soulie, J., 224^{174,174c} Souma, Y., 381²¹ Souppe, J., 56735, 57035 Southam, D. A., 963¹²¹ Southgate, R., 105123, 105623,34, 106223,34 Sowa, F. J., 316¹¹⁷, 317¹¹⁷ Sowden, J. C., 831^{64,65} Sowerby, D. B., 29933 Sowerby, R. L., 78639 Sowinski, A. F., 599²⁰⁶ Spagnuolo, C. J., 25152 Spalink, F., 903¹²⁷ Spaltenstein, A., 104208,209, 117208,209 Spanagel, H.-D., 661²² Spangler, R. J., 901111 Sparapani, C., 30040 Spayd, R. W., 8229, 8239, 8349 Speake, R. N., 71542 Spear, K. L., 957110 Spears, G. W., 1472, 1572, 1672, 1772 Speckamp, W. N., 223¹⁸³, 225¹⁸³, 341⁴, 361⁷⁷, 364⁷⁷, 368104 Speer, H., 155432 Spencer, C. F., 379¹, 380¹ Spencer, R. W., 21792 Spencer, T. A., 20117, 55283, 78535 Speranza, M., 30040 Spessard, G. O., 21559 Spicer, M. D., 213⁵¹ Spiegel, B. I., 48644, 49544, 52953 Spies, J. W., 582¹¹³ Spiess, A., 822¹², 831¹², 835^{12b} Spietschka, E., 89140, 90040.95 Spiewak, J. W., 883109 Spitznagel, G. W., 1944 Sprague, J. T., 382³⁶ Sprague, L. G., 20286 Spreafico, F., 312¹⁰² Spring, D. J., 825³² Springer, J. P., 698^{157a}, 872⁵⁶

Spyroudis, S., 53891 Srebnik, M., 79793.94 Sreekumar, C., 19520 Sridar, V., 3809 Srinivasa, R., 596¹⁹³, 727^{30,31} Srinivasan, K. G., 3807 Srinivasan, V., 3807 Srivastava, R. C., 826³⁴ Staab, H. A., 87782, 92749,50 Stadler, P. A., 126311,312, 34623, 35139 Ställberg-Stenhagen, S., 643126 Stam, C. H., 21025, 21132 Stanek, J., 89672 Stang, P. J., 219105, 3804, 52222 Stangl, H., 89353b Stanier, A., 786³⁸ Stanovnik, B., 904133 Stanton, J., 93361 Staral, J. S., 322142a Stark, C. J., Jr., 173521 Staskun, B., 369117, 372117 Stassinopoulou, C. F., 56520 Staub, A. P. A., 7346 Stauffacher, D., 345²⁰ Stauffer, R. D., 500132, 505132 Staum, M. M., 782²⁰ Steege, S., 322142b Steele, B. R., 19412 Steele, J., 373128 Steele, J. A., 406143 Stefanovic, M., 313103, 905136 Steffen, J., 34830 Stegmann, H. B., 66122 Steinbach, R., 42157.58 Steinberg, H., 72727 Steinberg, I. V., 643127 Steinberg, N. G., 89032 Steindl, H., 3807 Steinmüller, D., 124^{283,284}, 125^{283,284}, 127²⁸³, 132²⁸³ Steliou, K., 693141 Stelling, G. D., 35866 Stemke, J. E., 25179, 25479 Stener, A., 503149, 512149 Stenhagen, E., 643129 Stepanov, F. N., 38347 Stephens, C. J., 649¹⁹⁰ Stephens, R. D., 5214, 5224 Stephenson, L., 80413 Stephenson, M., 334221,221a Stephenson, R. A., 279³⁹, 770¹⁷² Steppan, H., 887¹², 904¹³² Steppel, R. N., 73941 Sterling, J. J., 8⁴¹, 9⁴¹, 15⁴¹, 20⁴¹, 212³⁹, 226³⁹², 250⁷⁰ Stermitz, F. R., 461144, 683101, 685106,107 Stern, M., 30251 Stern, R., 262¹⁶⁶ Sternfeld, F., 174527.527a, 175527a Sternhell, S., 505160 Stetter, H., 38344, 89034, 903126 Stevenaert, N., 963124 Stevens, C. L., 740⁴⁵, 790^{58,60}, 842²¹, 843²³ Stevens, R. V., 28761 Stevens, T. S., 822⁸, 836⁸, 913^{1,2}, 914³⁻⁵, 921³⁵, 927⁵³ Stevenson, D. P., 332207 Stevenson, J. R., 38667 Stevenson, P., 48421

Stevenson, P. J., 955106 Stevenson, R., 219¹¹², 676^{75,76}, 692¹³⁸, 693¹⁴² Stevenson, T., 46252 Stewart, K. R., 21133 Stewart, O. J., 499112 Stewart, R. F., 683101 Stiles, M., 726²⁴ Still, W. C., 41226, 19519,20, 19630, 203101, 602220, 98215 Stille, J. K., 213⁴⁹, 231²⁵⁴, 232^{257,258,259,259,261,263,265,268,269,271}, 233²⁷², 250⁷¹, 43616, 44675, 453112, 455123, 463161,162, 466190, 469²⁰¹, 470^{201,211}, 471²⁰¹, 473^{201,211}, 475^{201,211} 476²¹¹, 485³⁰, 487⁴⁹⁻⁵¹, 491³⁰, 492⁹⁴, 495^{49-51,938,94}, 50494,153, 51494, 52428,31, 52928,31, 755112, 102115, 1022²¹, 1023²⁴, 1024²⁵, 1030⁵⁸, 1031⁶¹, 1032⁶⁶, 103362, 1040108 Stipanovic, R. D., 36385 Stirling, C. J. M., 86⁴², 159⁴⁵⁰, 164⁴⁵⁰, 174^{534,535}, 176^{534,535}, 178^{534,535}, 862^{3,7}, 867^{34,37}, 872³⁷, 883³⁷, 88437 St-Jacques, M., 38236 Stocker, J. H., 56410, 56838,39,43 Stockhammer, P., 817⁸⁷ Stöckl, E., 647¹⁷⁶, 648¹⁷⁶ Stockwell, D. L., 53079, 53579 Stojiljkovic, A., 38013, 905136,139 Stoll, A. T., 25199, 25499, 463156 Stoll, M., 6158, 644145 Stoller, H.-J., 168503, 169503 Stolyarov, B. V., 30573 Stone, F. G. A., 208^{1,11}, 210^{11,11a}, 219^{11a}, 228²¹⁴, 234^{11a}, 380¹⁰, 436^{9,13}, 524³³ Stone, T. J., 66121 Stork, G., 28, 835.37, 118.50, 1478, 1578.80, 168, 178, 21126, $\begin{array}{c} 22^{132}, 23^{140}, 24^{144}, 26^8, 28^{170}, 30^{140,170,179}, 33^{140}, \\ 34^{140}, 35^{140}, 48^{259,260}, 125^{302}, 134^{302}, 197^{39}, 198^{44-46}, \end{array}$ 341², 351³⁹, 360², 602^{218,220}, 607²¹⁸, 638⁸⁹, 896⁶⁸, 902¹¹⁸, 1053²⁹ Stork, L., 648186 Stothers, J. B., 63979 Stotter, P. L., 11⁵¹, 99¹⁹², 103¹⁹², 107¹⁹² Stout, D. M., 181552, 512197 Stowell, J. C., 1⁵, 53⁵, 54⁵, 193³, 208³ Stowell, M. H. B., 20181 Strachan, R. G., 644167 Strakov, A. Ya., 83162 Strakova, I. A., 83162 Straley, J. M., 2088 Strasser, B. L., 38132 Strating, J., 649²⁰⁸, 832⁶⁸⁸, 909¹⁵¹ Strausz, O. P., 89143, 89243,47,48 Streib, H., 92441 Streinz, L., 643128 Strickland, J. B., 78853 Striepe, W., 331200b Stroezel, M., 63428 Strozier, R. W., 31186 Struve, G., 52213 Struve, W. S., 887², 888², 891², 897² Struwe, H., 34318, 35351, 35451 Stuart, A. P., 327169 Stuart-Webb, I. A., 84647 Stuhmiller, L. M., 226193 Sturm, H. J., 89353b, 89671 Sturmer, D., 3804, 73519 Stürmer, R., 79790

Sturtz, G., 19955, 105227 Stürzenhofecker, F., 396107, 397107 Stütz, P., 126311.312 Suau, R., 585133, 591171 Subba Rao, G. S. R., 1791 Subrahamanian, K. P., 595191 Subrahmanyam, C., 79999 Subramanian, R., 602224 Sudol, J. J., 62333 Sudweeks, W. B., 124²⁸², 125²⁸², 128²⁸², 129²⁸² Suemitsu, R., 34195, 35202, 102952 Suga, H., 201⁸⁴, 769¹⁶⁸ Suga, S., 300⁴⁶, 302⁴⁶, 313¹⁰⁵, 314¹⁰⁸, 315¹¹³, 318¹²⁹, 769¹⁶⁹ Sugawara, R., 44677 Sugihara, T., 224¹⁸¹ Sugimura, H., 446⁸², 459¹³⁶, 460¹³⁶, 461¹³⁶, 462¹⁵⁰. 472222, 513208 Sugimura, T., 983²⁰, 984²⁰, 1058⁴¹, 1062⁴¹ Suginome, H., 469²⁰⁶, 470^{205,206,207}, 473^{205,206,207}, 48959,61,63, 49559,61,63, 49699, 49859,99, 50461,63 51159,61,63,99, 51559,61,63,99, 53058 Sugishita, M., 757¹²³ Sugita, K., 300³⁸, 400¹²⁰ Sugita, N., 1026⁴⁰ Sugita, T., 31197, 313106, 315113, 769169 Sugiura, S., 222¹⁴⁴ Sugiyama, T., 42159, 42259 Suhara, Y., 644139 Sukh Dev, 402127 Sukhoverkhov, V. D., 38347 Sulbaran de Carrasco, M. C., 594187 Suld, G., 327169 Sullivan, D. E., 643119 Sullivan, J. W., 644¹⁵⁴ Sulmon, P., 85790 Sulsky, R. B., 33190 Sultana, M., 512²⁰³ Sultanbawa, M. U. S., 35347 Sum, F. W., 99188, 107188, 110188 Sumi, K., 135^{341,342,343}, 136^{341,342,343}, 137^{341,342}, 217⁷⁹ Sumiki, Y., 715³⁹ Sumitani, K., 228^{216,218}, 436³, 437^{25,26}, 440²⁵, 448²⁵, 44925, 45025,26, 45125,106, 45225,106, 503145, 52427 Sumitomo, H., 99190 Sumiya, F., 969135 Sumiya, T., 12⁶², 453¹¹³ Summersell, R. J., 486³⁹, 491³⁹, 495³⁹, 498³⁹, 503³⁹ Summons, R. E., 67162 Sundararaman, P., 51²⁷⁰ Sundin, A., 763¹⁴⁹, 768¹⁴⁹ Surleraux, D., 111231 Suryawanshi, S. N., 406144 Süs, O., 904^{129,130,131,132} Susz, B., 27312 Suter, C., 43195,96, 78114 Sutherland, B. L. S., 74153 Sutherland, B. L. S., 741^{-10} Sutherland, I. O., 225^{188} , 877^{88} , 914^{10} , $916^{19,20}$, 919^{30} , 923^{45} , 926^{47} , 927^{48} , 928^{47} , $930^{10,59}$, 931^{10} , $933^{30,62}$, 946^{30} , $949^{94-96,98,99}$, 954^{98} , 963^{121} Sutherland, J. K., 341⁴, 342^{4d}, 348²⁹, 353⁵², 355⁵³, 357⁵³, 371¹¹⁵, 379³, 380⁹, 382³⁵, 386^{3c,62-65}, 390^{71,72}, 396^{111,114,115}, 398^{111,114}, 399¹¹⁷, 402¹¹⁷, 600²¹⁵, 753⁹⁹, 769170, 770175, 771170 Sutherland, M. D., 39070

Sutton, K. H., 19737 Sutton, L. E., 38236 Suzukamo, G., 327167 Suzuki, A., 221134, 231244,246,247,248,249,252, 24966, 25178, 25478, 262158, 27421, 326163, 42153, 44356, 44689, 465178, 469206, 470178,205,206,207, 473178,205,206,207, 48958,59,61,63,64, 49064-66, 49558,59,61,63-65,97 49664.65.98.99, 49859.64.65.98.99, 50461.63.97.154 51158,59,61,63-66,97-99,154, 51558,59,61,63-66,97-99,154 523^{25,26}, 530⁵⁷ Suzuki, H., 169510, 172510, 173510, 56626 Suzuki, J., 816⁷⁹, 923⁴³ Suzuki, K., 168508, 169508, 30355,56, 44896, 457129,130, $\begin{array}{l} 459^{137,139}, \ 460^{130,137}, \ 461^{137}, \ 497^{104}, \ 510^{183,206}, \\ 513^{205,206}, \ 730^{42,42b}, \ 740^{46}, \ 937^{75} \end{array}$ Suzuki, M., 4²⁶, 5²⁶, 9⁴⁷, 10^{26,47}, 25¹⁵⁹, 135^{356,360,361}, 136356,360,361,371, 137360,361, 139356,360,361,371 140371,3716, 142360,361, 143360,361,371,3716, 156360,361, 56522, 771182 Suzuki, N., 74873 Suzuki, S., 470²⁰⁹, 472²⁰⁹, 475²⁰⁹ Suzuki, T., 172511,512,515,516, 173512,515,516, 19846, 44363, 503144, 66749 Suzuki, Y., 690¹²⁵ Svadkovskaya, G. E., 63411, 63511 Swallow, W. H., 742⁵⁵, 751⁹⁰, 752⁹¹ Swaminathan, S., 380⁷ Swann, S., Jr., 63532 Swartz, J. E., 599²²², 602²²² Swarzenbach, G., 88814 Sweeney, K. M., 328179 Swenson, W., 28454 Swenton, J. S., 254¹⁰², 256¹¹², 695¹⁵² Swierczewski, G., 24312, 24638, 25012, 26212, 470221, 471²²¹, 482⁵, 499⁵, 505⁵, 509⁵ Swindell, C. S., 229²³³, 246³⁸, 444⁶⁵, 446⁸¹, 492⁷⁵, 503⁷⁵ Swingle, R. B., 149⁴⁰⁰ Swisher, J. V., 883¹⁰⁸ Swithenbank, C., Swoboda, J. J., 36175 Sworin, M., 79265 Syfrig, M. A., 71^{31,33} Szajewski, R. P., 902118 Szary, A. C., 124289, 125289 Szeimies, G., 62541 Szeto, K. S., 92748 Szmuszkovicz, J., 28¹⁷⁰, 30¹⁷⁰ Szöcs, G., 223155 Tabacchi, R., 81362 Tabata, M., 55422.23 Tabei, H., 53063, 53263 Tabei, T., 942⁸⁰, 1008⁶⁵ Taber, D. F., 602²²⁰, 1051²⁴, 1052^{24,25,28}, 1053²⁸, 1054³⁰, 1055²⁸, 1057²⁸, 1059²⁵, 1060^{44,45}, 1061⁴⁶, 1062^{30,44,46,47} Tabibian, S., 88924 Tabor, T. E., 73940 Tabushi, I., 38124, 38224 Tachibana, K., 74150 Tada, H., 38657, 39079 Taddei, F., 95154, 119154 Taddei, M., 356⁵⁷ Taddia, R., 738³⁷ Tadzhimukhamedov, Kh. S., 30358 Taeger, E., 19414

Taffer, I. M., 753102 Tagaki, W., 86¹⁹, 121¹⁹ Tagat, J., 251⁷⁷, 254⁷⁷ Taguchi, H., 78747.48 Taguchi, T., 42154 Tahara, A., 83476 Tai, W. T., 85473 Taikov, B. F., 643131 Taimr, L., 66429 Tait, T., 331198 Taits, S. Z., 6157 Takacs, F., 81257 Takacs, J. M., 45²⁴⁴, 199⁵³ Takada, S., 834⁷⁶ Takagi, K., 463¹⁶³, 483¹⁹, 484^{20,24}, 500^{19,131}, 501²⁴, 509¹⁹ Takagi, Y., 42682, 42982, 87154 Takahashi, E., 44677 Takahashi, H., 96¹⁶⁶, 99¹⁶⁶, 103¹⁶⁶, 105^{166,216}, 112²¹⁶. 120¹⁶⁶, 155¹⁶⁶, 530⁵⁶ Takahashi, J., 644161 Takahashi, K., 136^{370,371}, 137³⁷⁰, 138^{370,371a}, 139^{370,371}, 140^{370,371,371a}, 143^{371,371a}, 144^{371a}, 826³⁹ Takahashi, M., 34194, 644134a, 150, 152, 93467 Takahashi, O., 94280, 100257, 100459, 100865 Takahashi, S., 454121, 53061, 53161, 53261, 53761, 5528, 55748.49. 68199 Takahashi, T., 49²⁶³, 51²⁷¹, 95¹⁵³, 100²⁰⁰, 103²⁰⁰, 107¹⁵³, 114¹⁵³, 115¹⁵³, 198⁴⁵⁻⁴⁸, 230²³⁹, 232^{239a}, 238^{239a}, 38010, 39079, 39599,101, 402129, 48641, 503151, 53054.55, 74150, 100872, 100972, 101072, 103375,76 Takahashi, Y., 326163 Takai, K., 44573, 44973 Takaishi, N., 1041113 Takamine, K., 56736,190, 595190, 607190, 61036 Takamura, N., 105532 Takano, S., 34¹⁹⁴, 224¹⁸¹, 278³⁰, 934⁶⁷ Takaoka, Y., 286^{56a} Takata, Y., 63985 Takatani, M., 225185, 264181 Takaya, T., 181⁵⁵² Takayama, H., 169510, 172510,511,512,515,516 173510,512,515,516, 38011, 44363 Takebe, M., 314¹⁰⁸ Takeda, A., 843²⁵, 844^{32,33}, 848^{53,54} Takeda, K., 386⁵⁷, 390⁷⁹ Takeda, T., 137³⁷⁶ Takegami, Y., 102116 Takehara, Z., 647194 Takehashi, T., 101075 Takehira, Y., 49167 Takei, H., 20116, 44679,80,82, 45679,80, 459136, 460136 461136, 462150, 47079,80,224, 47180, 472222,224, 47379, 513²⁰⁸ Takei, Y., 875^{73,74} Takematsu, A., 300³⁸ Takemoto, T., 396¹¹⁵, 406¹⁴¹, 748⁷³ Takenaka, Y., 124²⁶³, 126²⁶³ Takenouchi, K., 20116 Takeshita, H., 57371, 57581,82, 61071 Takeshita, M., 135³⁴⁷, 136³⁴⁷, 137³⁴⁷, 139³⁴⁷, 144³⁴⁷ Takeshita, T., 30043 Takeuchi, M., 227²¹², 470²¹², 476²¹² Takeuchi, N., 683105 Takeuchi, R., 101810, 102110 Takeya, T., 676⁷⁷, 695¹⁵³, 696¹⁵⁴ Taki, K., 2958

Takigawa, T., 1787, 124263, 126263 Takinami, S., 251⁷⁸, 254⁷⁸, 490⁶⁶, 511⁶⁶, 515⁶⁶ Takken, H. J., 251⁷⁹, 254⁷⁹ Takundwa, C., 531⁸⁷, 538⁸⁷ Talapatra, B., 396¹¹², 398¹¹² Talapatra, S. K., 396¹¹², 398¹¹² Tam, C. C., 131325 Tam, T. F., 21792 Tam, W., 457¹²⁷, 458¹³⁴ Tamai, K., 245³², 470²¹³, 476²¹³ Tamaki, K., 789⁷⁰ Tamao, K., 20068, 225184, 228216,217,218,222, 229225. 262¹⁶⁵, 436^{3,10}, 437^{21,25,26}, 438³¹, 440²⁵, 448^{25,96}, 44925,99,100, 45025,26, 45125,106, 45225,106,107,111, 45331. 457129,130, 45931,137,139, 460107,130,137, 461137, 464171, 469²⁰³, 470²⁰³, 473²⁰³, 483¹⁶, 487⁴⁵, 492⁷⁴, 497¹⁰⁴, 503145, 510183,206, 513205,206, 52427 Tamariz, J., 19849 Tamarkin, D., 619^{26,27} Tamaru, Y., 221¹³¹, 231²⁴², 420^{48,49}, 443⁶¹, 445⁶¹. 463^{157,159}, 1023²³, 1032⁷⁰, 1033⁷¹, 1040¹⁰³ Tamblyn, W. H., 92034, 92534a, 93434 Tamborski, C., 450¹⁰⁴, 457¹²⁸ Tamm, C., 73625 Tamura, H., 63544 Tamura, M., 21023, 24313, 24419, 41510, 41810.25 4361,2.6, 43724, 4382, 46424, 4827, 49484, 4997 Tamura, Y., 386⁵⁷, 904¹³⁴, 939⁷⁹ Tanabe, K., 30043 Tanabe, M., 34195 Tanabe, Y., 470²⁰⁷, 473²⁰⁷ Tanahashi, Y., 74150 Tanaka, H., 99¹⁸¹, 101¹⁸¹, 501¹³⁷, 509¹³⁷, 579¹⁰¹, 650²¹¹, 652220 Tanaka, K., 99180, 100180, 159455, 161455, 174526 Tanaka, M., 226²⁰⁰, 507¹⁷⁵, 529⁵¹, 1016³, 1024²⁶, 1039101,102 Tanaka, N., 159455, 161455 Tanaka, S., 57582, 74046 Tanaka, T., 222¹⁴⁴ Tanaka, Y., 244^{27,28}, 464¹⁷⁷ Taneja, H. R., 125299 Taneja, K. L., 325159 Tang, H., 63893 Tang, P. W., 155433, 157433, 230238 Tang, Q., 505157 Tanguy, G., 102639 Tani, S., 28656a Tanida, H., Tanigawa, Y., 222143, 223150, 259130, 43723 Taniguchi, H., 295¹¹, 302¹¹, 567^{36,190}, 595¹⁹⁰, 607¹⁹⁰, 61036 Taniguchi, Y., 45248 Tanikaga, R., 153415, 224170 Tanimoto, S., 95¹⁵² Tanis, S. P., 220122, 224172,1726,175, 325162, 41611-13,19, 417¹³, 751⁸⁷, 752⁹⁵ Tanner, D., 593180.183, 594184 Tanner, D. D., 10462 Tao, Y.-T., 172514 Taran, M., 74150 Tardella, P. A., 425, 725 Tarhouni, R., 20291 Tarkhanov, G. A., 640¹⁰¹ Tashiro, M., 295⁷, 302⁵², 307⁸⁸, 329^{88,184,185,187} Tasovac, R., 905^{136,139}

Tatlow, J. C., 52211, 63984, 648187 Tatone, D., 124286, 125286, 127286 Tatsumi, K., 86736 Tatsuno, T., 85689 Tat'yanina, E. M., 328181 Taub, D., 689119 Tavares, D. F., 74877 Taveras, A. G., 51270 Tawney, P. O., 2084, 2104 Taya, S., 992³⁷, 994⁴¹, 999⁵¹ Taylor, C. E., 53272 Taylor, D. A., 26161 Taylor, E. C., 54²⁷⁵, 219¹⁰⁸, 418²⁸, 499^{123,124}, 505¹⁶⁵. 507¹²⁴, 551¹, 670⁵⁸, 673^{58,69}, 693¹⁴¹ Taylor, J. A., 55314 Taylor, R. J. K., 311, 811, 87576 Taylor, R. T., 25496 Taylor, S. K., 325^{161,161a}, 390⁸⁰, 392⁸⁰, 396^{105,106}. 397105,106 Taylor, T. I., 822¹³, 829¹³ Taylor, W. C., 83159 Taylor, W. I., 659¹, 660¹, 679¹ Tchoubar, B., 781¹⁵, 839⁵, 845³⁹ Teague, S. J., 384⁵³, 400⁵³, 603²²⁷ Tefertiller, B. A., 1470 Teichmann, K.-H., 89034 Teisseire, P., 599207 Teissier, B., 66122 Tejero, T., 579¹⁰⁰ Telfer, S. J., 369¹⁰⁸ Tellier, F., 52541 Telschow, J. E., 23¹³⁶, 24¹³⁶ Tember, G. A., 643¹³¹ Temple, J. S., 469²⁰², 470²⁰², 473²⁰² Templeton, D. H., 178⁵⁴², 179⁵⁴² Templeton, J. L., 47257 Templeton, L. K., 178⁵⁴², 179⁵⁴² Tenge, B. J., 104²⁰⁹, 117²⁰⁹ Tenhaeff, H., 91513, 96513 Ten Hoeve, W., 6823, 7134 Tenud, L., 1368, 91619 Terada, A., 963¹²⁰ Terada, K., 503142 Terada, Y., 264¹⁸⁵, 665³⁸, 691^{128,131}, 693^{128,139}, 697^{38,131,139,155}, 698¹²⁸ Teranishi, S., 583120, 1041111 Terao, K., 87¹¹¹, 106¹¹¹, 114¹¹¹ Terashima, M., 498¹⁰⁹, 1038^{95,95b} Terauchi, M., 99180, 100180 Terem, B., 57789 Terrell, R., 28170, 30170 Terry, L. W., 104813 Tessier, D., 57477 Testaferri, L., 457¹³³, 493⁸³, 503¹⁴⁶, 509¹⁷⁸, 513^{83,146} Teter, L. A., 20913 Teulade, M. P., 20183 Teyssié, P., 1047^{7,8}, 1051^{7,8} Tezuka, H., 38126, 38226 Thaker, V. B., 602225 Thakker, K. R., 57264 Thal, C., 681 100 Thamnusan, P., 154423, 155423 Thatcher, K. S., 81684 Thebtaranonth, Y., 916¹⁹, 949⁹⁹, 963¹²¹ Theis, A. B., 209¹⁴ Theodore, M. S., 785^{36,36b}

Theodoropoulos, S., 62333 Thielert, K., 102848 Thiem, J., 5631 Thierrichter, B., 82422, 82622 Thies, H., 623³⁶ Thies, R. W., 39292 Thiessen, G. W., 643121 Thiessen, W. E., 407150 Thijs, L., 75082.83 Thirring, K., 683103 Thomalla, M., 63541 Thomas, A., 734⁴ Thomas, A. F., 99645 Thomas, C. A., 2944 Thomas, C. B., 501135 Thomas, C. W., 89141a Thomas, E. J., 3804, 93464, 95364 Thomas, E. M., 217⁹² Thomas, J. A., 714³² Thomas, M. T., 50²⁶⁸ Thomas, P. J., 105943, 106043 Thomas, R., 20175, 85475 Thomas, R. C., 799102 Thomas, S. L. S., 643125 Thömel, F., 390⁸⁴, 392⁸⁴ Thompson, D. J., 10178 Thompson, H., 602²¹⁸, 607²¹⁸ Thompson, J. M., 103789 Thompson, M. E., 180551 Thompson, P. A., 103374 Thompson, W. J., 511^{188,212}, 515^{188,212} Thomson, R. H., 689122 Thomson, S. A., 766159 Thomson, T., 913², 914^{3.5} Thorén, S., 767165 Thornmen, W., 57264 Thorsett, E., 461144 Thorstenson, P. C., 105017 Thottathil, J. K., 21676 Thuan, S. L. T., 72836 Thuillier, A., 124260 Thun, K., 89032 Thyagarajan, B. S., 158442, 164442, 507173, 66010, 8031, 809^{1b}, 817^{1b} Tidwell, T. T., 587141 Tiecco, M., 44790, 45690, 457133, 49383, 503146, 509178, 51383.146 Tiffeneau, M., 754106, 756114 Tilhard, H. J., 105²¹⁴ Tilley, J. W., 53065, 53365, 103892 Timberlake, J. W., 587141,143 Tin, K.-C., 74877 Ting, J.-S., 21455, 21674 Tingoli, M., 229²³³, 444⁶⁵, 447⁹⁰, 456⁹⁰, 457¹³³, 492⁷⁵, 493⁸³, 503^{75,146}, 509¹⁷⁸, 513^{83,146} Tirado-Rives, J., 591170 Tirpak, J. G., 499117 Tischler, A., 541¹¹², 543¹¹² Tisler, M., 756¹¹³ Tišler, M., 904133 Tissler, M., 864¹⁷, 866¹⁷, 883¹⁷ Tissot, P., 57995, 63978, 644156 Titiv, Y. A., 839⁷, 848⁷ Tius, M. A., 224171, 225171, 264182, 36179, 497106, 762146 Tiwari, K. K., 329¹⁸² Tiwari, K. P., 103894

Tkachev, A. V., 38668 Tkatchenko, I., 82211, 83411.74.75 Tobe, Y., 380⁹ Tobinaga, S., 67677, 67989, 683102,105, 69389, 695153. 696¹⁵⁴, 807³⁵ Tobita, M., 219113, 505167 Toda, F., 49167, 55737.38, 90097 Toder, B. H., 22133, 26165, 51269, 906144 Tohda, Y., 21789, 21989, 2711, 5217, 5307, 5327, 55426 Tohjima, K., 650213 Tojo, G., 585¹³³, 591¹⁷¹ Tokarev, Yu. I., 30684 Tökes, L., 257¹²⁰ Tokoroyama, T., 35²⁰¹, 39²⁰¹, 355⁵⁴, 357⁵⁴ Tolman, R. L., 262¹⁶⁶ Toma, K., 39599 Tomasewski, A. J., 8043 Tomesch, J. C., 124²⁵⁴, 126²⁵⁴, 137²⁵⁴, 423⁷⁹ Tomilov, A. P., 639⁸⁰ Tomimori, K., 222144 Tominaga, Y., 24644 Tomioka, H., 89145 Tomioka, K., 41224, 43235, 21795, 67573 Tomita, B., 34726 Tomita, K., 75188 Tomita, S.-I., 93467 Tomooka, K., 73042 Tomotaki, Y., 99181, 101181 Tomuro, Y., 463¹⁶⁵ Tonker, T. L., 47257 Toong, Y.-C., 595191 Topchii, V. A., 30250 Tori, K., 38657, 407149 Torii, S., 99¹⁸¹, 101¹⁸¹, 125^{297,298}, 126²⁹⁸, 128²⁹⁷, 129²⁹⁷, 130297,298, 133297, 137298, 167484, 168484,495,499,500, 169495,499,500, 36174, 501137, 509137, 579101, 63414.23, 649²³, 650²¹¹, 652²²⁰ Torisawa, Y., 135³⁶⁴, 139³⁶⁴, 142³⁶⁴, 143³⁶⁴ Toro, J. L., 498¹⁰⁸ Toromanoff, E., 21131 Torr, R. S., 20176 Torre, A., 390^{71,72}, 399¹¹⁷, 402¹¹⁷ Torres, M., 89143, 89243,47,48 Toscano, V. G., 20177 Toshimitsu, A., 87¹¹¹, 106¹¹¹, 114¹¹¹, 381^{26,27}, 382^{26,27} Toth, J. E., 21351 Toth, M., 223155 Toubiana, M.-J., 407149 Toubiana, R., 407149 Touboul, E., 57260 Tour, J. M., 25178,100, 25478,100, 102535, 103035 Toure, V., 380⁸ Townsend, C. A., 20914 Toy, M. S., 640102 Toyama, T., 92343 Toyoshima, T., 52847 Tozune, S., 92343 Traas, P. C., 25179, 25479 Traencker, H. J., 19413 Trahanovsky, W. S., 5514, 5524, 66019.20, 66120, 67919, 699²⁰ Trammell, G. L., 35143a Trammell, M. H., 261¹⁴⁸, 264¹⁴⁸ Tranchepain, 1., 258127 Trave, R., 38657, 39598 Traynham, J. G., 379³, 390^{81,83}, 392^{81,83}, 649²⁰⁷

Author Index

Traynor, S. G., 770174 Treanor, R. L., 104811 Trecker, D., 334²²⁰ Treco, B. G. R. T., 21351 Trehan, S., 497¹⁰⁶ Tremble, J., 257120 Triaca, W. E., 63658 Tribble, M. T., 85479 Trifan, D., 653225 Trifonov, L. S., 103893 Trifunac, A. D., 258124 Trimitsis, G., 19733 Trinks, R., 594184 Trivedi, S. V., 416¹⁶, 417¹⁶ Trocha-Grimshaw, J., 67783 Trofimov, B. A., 259¹²⁸ Trombini, C., 168488, 169488 Trometer, J. D., 226¹⁹⁹ Trope, A. F., 785^{36,36b} Trost, B. M., 3¹³, 7³⁴, 11⁵⁴, 12^{59,60}, 16⁵⁴, 17^{54,86}, 26⁵⁴, 34¹⁹², 39¹⁹², 55²⁸¹, 56²⁸⁷, 58²⁸⁷, 75⁵², 86^{9,17,39,41,61}, 87⁴¹, 88^{17,61,135}, 89⁶¹, 90¹³⁵, 91⁶¹, 94¹⁷, 95¹⁷, 124^{17,61}, 139³⁷⁸, 154³⁷⁸, 155³⁷⁸, 168⁴⁸⁹, 169⁴⁸⁹, 174^{531,532,533}, 176^{531,532,540}, 177^{39,531,532,333}, 178^{41,532,543,544}, 17941,531,533,543,546, 181543, 228214, 27425, 28350, 38010, 423⁷⁸, 436¹³, 446^{83,84}, 524³³, 564⁵, 583¹²⁶, 607⁵, 761¹⁴⁴, 762¹⁴⁴, 766¹⁵⁷, 785^{32-34,37}, 792⁶⁶, 832⁶⁷, 89142, 903122, 905137, 909149, 91821,23, 92136, 93361, 953^{102,103}, 979¹², 994^{40,43,44}, 996⁴³, 1000⁴³, 1008⁶⁶, 1040106 Trova, M. P., 87258 Trowbridge, B. D., 20176 Truce, W. E., 8635, 88137, 95137, 123249, 158440,444, 161444, 16435,444,474, 165137, 167137,444, 17335, 180^{474,549}, 181^{474,554}, 317¹²⁰, 794⁷⁹ Truesdale, L. K., 781¹³ Truesdell, D., 38010 Trumbull, E. R., 760¹³⁹, 774¹³⁹ Trumbull, P. A., 760¹³⁹, 774¹³⁹ Trust, R. I., 36387, 36563.87 Trybulski, E. J., 28863 Tsai, D. J.-S., 987²⁹, 990²⁹, 993²⁹ Tsan-Hsi Yang, 31093, 31193 Tsankova, E., 390⁷³, 396¹¹⁰, 397¹¹⁰, 752⁹⁸ Tse, H. L. A., 248⁵⁴ Tseng, C. C., 220¹²¹, 222^{121c,142} Tso, H.-H., 172513, 173513,517 Tsubaki, K., 221131, 42048,49 Tsuboi, S., 848^{53,54} Tsuchida, E., 67781, 68681 Tsuchihashi, G., 135^{356,357,360,361,362}, 136^{356,357,360,361,362,371}, 137^{357,360,361}, 138³⁵⁷ 139356,360,361,362,371, 140371,3716, 142357,360,361 143^{357,360,361,371,371b}, 147³⁹⁴, 155⁴²⁹, 156^{360,361,362}, 303⁵⁶, 730^{42,42b} Tsuda, K., 816^{79,79a} Tsuda, T., 21786, 259137 Tsuge, O., 20184 Tsuji, J., 2⁸, 11⁸, 16⁸, 17⁸, 26⁸, 28¹⁶⁹, 51²⁷¹, 95¹⁵³, 107¹⁵³, 114¹⁵³, 115¹⁵³, 198^{47,48}, 380¹⁰, 390⁷⁹, 524³⁴, 639⁷⁵, 1008⁷², 1009⁷², 1010^{72,75}, 1033⁷⁶, 1040¹⁰⁵ Tsuji, T., 49²⁶³, 51²⁷¹, 380¹⁰, 753¹⁰³, 901¹⁰⁸ Tsuji, Y., 101810, 102110, 102849 Tsukamoto, M., 35554, 35754 Tsukervanik, I. P., 30992a, 315109,111, 320134, 321138,140 Tsukida, K., 168490, 169490

Tsunoda, T., 100²⁰¹, 103²⁰¹, 107²⁰¹ Tsutsui, T., 592¹⁷³ Tsutsumi, S., 4838, 55420.21 Tsuyama, K., 136370, 137370, 138370, 139370, 140370 Tsuyuki, T., 741⁵⁰ Tsuzuki, K., 49²⁶⁴ Tu, C.-Y., 709¹⁷ Tubiana, W., 72941 Tueting, D. R., 233²⁷², 470²¹¹, 473²¹¹, 475²¹¹, 476²¹¹ Tuller, F. N., 14⁷¹, 15⁷¹ Tuncay, A., 512²⁰³ Tunemoto, D., 159458, 161458, 162458, 167458, 20288 Tung, J. S., 507¹⁷² Turedek, F., 70915 Turnbull, L. B., 158437, 159437, 160437, 166437 Turnbull, M. M., 219102 Turner, A. B., 66013 Turner, D. W., 68197 Turner, J. V., 93470 Turner, R. W., 41622 Turrell, A. G., 499¹²⁴, 507¹²⁴, 673⁶⁹ Turro, N. J., 89252 Tustin, G. C., 1026³⁹ Twigg, M. V., 1017⁸ Twine, C. E., Jr., 125³⁰⁵, 126³⁰⁵, 127³⁰⁵ Tyrala, A., 158446, 159446 Tyrlik, S., 56524, 57024, 58324 Ubersax, R. W., Uchibayashi, M., 602²¹⁸, 607²¹⁸ Uchida, A., 483⁹ Uchida, K., 243¹¹, 254⁹⁷, 259¹³⁵, 799¹⁰³ Uchida, N., 168⁵⁰¹, 169⁵⁰¹, 170⁵⁰¹ Uchida, S., 677⁸¹, 686⁸¹ Uchida, T., 469²⁰³, 470²⁰³, 473²⁰³ Uchida, Y., 862³ Uchikawa, M., 100055, 100460 Uchiyama, H., 45²⁴⁸ Uđa, H., 946⁸ Uda, J., 102744 Ueda, E., 464¹⁷⁵ Ueda, J., 644^{134a} Ueda, K., 648¹⁸⁵, 900⁹⁷ Ueda, M., 135^{341,342,343}, 136^{341,342,343}, 137^{341,342} Uehara, H., 421⁵⁴ Uemura, S., 87¹¹¹, 106¹¹¹, 114¹¹¹, 381^{26,27,29}, 382^{26,27,29} Uenishi, J., 231253, 55851 Ueno, Y., 169509, 98525 Uggeri, F., 7786, 7886,54, 7896,55,57 Uguen, D., 159454, 161454, 167481, 168481, 170481, 172481. 882103 Ukai, J., 446^{85,86} Ukita, T., 168492, 169492 Ulery, H. E., 34215 Ullah, M. I., 807²⁶ Ullenius, C., 21235, 512193.194 Ulrich, P., 25070 Umani-Ronchi, A., 168488, 169488 Umbreit, M. A., 583¹¹⁹ Umen, M. J., 14⁷³, 15⁷³, 249⁶³ Umezawa, B., 672⁶⁵ Umezu, K., 220125 Underhill, E. W., 224174 Underiner, T. L., 222138 Underwood, G. R., 38342 Unelius, C. R., 489⁶¹, 495⁶¹, 504⁶¹, 511⁶¹, 515⁶¹
Uneyama, K., 125^{297,298}, 126²⁹⁸, 128²⁹⁷, 129²⁹⁷ 130297,298, 133297, 137298, 167484, 168484,495,499,500, 169495,499,500, 36174, 648185 Uno, H., 729³⁹ Uno, M., 454121 Uozumi, Y., 103895 Urabe, H., 1263, 454119 Urayama, T., 103267 Urbach, H., 85168 Urban, R. S., 82316, 82516 Urbanek, F., 85162 Uroda, J. C., 757124 Urso, F., 103684 Ursprung, J. J., 70916 Usifer, D., 513208 Ustynyuk, T. K., 8397, 8487 Utaka, M., 84325, 84432.33 Utimoto, K., 24311, 24429, 25496,97, 259135, 27422, 27937, 44574, 48424, 50124, 759133, 799100,103 Utley, J. H. P., 57789, 6348.20.24, 63540, 6368, 63765.66, 6388.20, 6398.24, 642111, 6438, 647195, 6498.20.24, 65066, 6558,20 Utsunomiya, I., 512198 Uyehara, T., 73045 Vagt, U., 619²⁵ Valade, J., 74150 Valente, L. F., 231241, 266196, 44351 Valette, G., 31184 Valls, J., 1265 Valoti, E., 312102 Valyocsik, E. W., 29614 Van Bergen, T. J., 512¹⁹⁹, 969^{131,132} van den Goorbergh, J. A. M., 58293 Vanderesse, R., 509179 Van der Eycken, E., 90093 van der Gen, A., 20121, 25121, 58293, 24210, 36175 van der Heide, F. R., 105635, 106235 van der Helm, D., 21671 Vander Jagt, D. L., 82956 van der Leij, M., 86841 van der Loop, E. A. R. M., 85583 Van Derveer, D., 1784 Vanderveer, D., 5647 Van der Werf, C. A., 5648 van der Westhuizen, J. H., 83161 Vandewalle, M., 71328, 90093 Van Duyne, G., 33¹⁸⁹, 34¹⁹⁸, 39¹⁹⁸, 592¹⁷⁵ Van Emster, K., 705² Van Ende, D., 8650, 8767.76.77.84.117, 89140, 90140, 91140. 92¹⁴⁰, 107¹⁴⁰, 109^{84,140}, 116¹⁴⁰, 136^{76,77}, 137⁷⁷, 141⁷⁷, 14476,77, 14576,77 Van Fossen, R. Y., 89142 Vangermain, E., 30787 van Helden, R., 30249 Van Horn, D. E., 230239, 232239a, 238239a, 44355, 486⁴²⁻⁴⁴, 495^{43,44}, 498⁴², 529⁵³, 530⁵⁴ Van Horssen, L. W., 5529, 5579 van Koten, G., 20811, 21011,11a,25,26, 21132, 21911a, 23411a van Kruchten, E. M. G. A., 223147 Van Leeuwen, P. W. N. M., 102220 van Leusen, A. M., 158449, 174449,523,524, 175449,523,524 909¹⁵¹ van Leusen, D., 174524, 175524, 92034, 92334b, 93434, 100869 Van Meerssche, M., 85790

van Mourik, G. L., 24962 van Tamelen, E. E., 99^{185,191}, 103^{191a}, 107¹⁹¹. 126³¹⁰. 201⁷⁴, 341⁴, 342⁴^{a,6}, 365^{94,95}, 373¹³¹, 717⁴⁵, 752^{93,94} van Tilborg, W. J. M., 568^{46,47}, 569⁵⁰ van Vliet, N. P., 367100, 371116 Van Zorge, J. A., 649²⁰⁸ van Zütphen, L., 324150 Varaprath, S., 42682, 42982 Vardi, S., 640101 Varma, V., 103164 Vartanyan, A. G., 318123 Vartanyan, S. A., 318123 Vasil'ev, Y. B., 63531,35 Vass, G., 126315,316,317 Vassil'ev, Y. B., 64098, 647177, 648177,178,181, 649177 Vatakencherry, P. A., 1785, 72940 Vaughan, K., 49²⁶⁴ Vaughan, R. J., 89794 Vaughan, W. R., 710²¹, 848⁵¹ Vaughn, T. H., 273⁹ Vaultier, M., 88135, 90135 Vazeux, M., 75189 Veach, C. D., 57998 Veal, P. L., 688115 Veal, W. R., 105635, 106235 Veale, C. A., 21794 Veber, D. F., 644¹⁶⁷ Vedejs, E., 8612, 125300, 128300, 129300, 133300, 8622, 86525,26, 91824,26, 93466, 957109-111, 95866.113, 960114-118 Vederas, J. C., 227²¹³ Velluz, L., 1265 Veniard, L., 964125 Venkataramani, P. S., 595¹⁹¹ Venkateswaran, R. V., 78328 Venus-Danilova, E. D., 72310 Verboom, W., 219¹⁰⁴ Verevkin, S. P., 30465 Verhoeven, T. R., 228²¹⁴, 436¹³, 524³³ Verkruijsse, H. D., 87120, 88120, 105120 Verlaque, P., 89247 Vermeer, P., 21781,82,85, 219104, 25496, 49168.69, 53184 Vernin, G., 505162, 507162, 512162 Verpeaux, J.-N., 44793, 44894, 49381 Veschambre, H., 903¹²⁶ Vesely, J. A., 30463 Vessiere, R., 86630 Vetter, W., 62847 Viallefont, Ph., 46²⁵⁴, 215⁶⁶, 251⁷⁵ Viani, F., 147³⁹³ Viau, R., 147397, 149397,400,403, 150397,403, 151397,403 Vicens, J. J., 96161,164, 98161,164, 99164, 149404,406, 151406, 152404,406, 153406 Vickers, J. A., 688¹¹⁴, 690¹¹⁴ Viehe, H. G., 271², 272², 551³, 552³, 556³³, 870⁴⁹, 963¹²⁴ Vig, O. P., 41618, 41718 Vigneron, J.-P., 263173, 55739 Vijh, A. K., 6346, 63538, 6366, 637661, 6556 Vilcsek, H., 81893 Vill, J. J., 760137 Villa, P., 771¹⁹² Villacorta, G. M., 211²⁷ Villani, F. J., Jr., 524³⁶ Villarica, R. M., 55²⁸³ Villaverde, M. C., 586156, 610156

Villemin, D., 219103, 53066, 53366 Villenave, J.-J., 1046³ Villieras, J., 20287,91-95, 24314, 24634,35, 24746, 24914, 263¹⁴, 440⁴², 441⁴², 442⁴², 464¹⁷², 470^{223,225,226}, 473217,225,226, 476217, 48531, 48631, 52218,19 Villiéras, J., 759129, 78849 Vinokur, E., 56414, 60714 Virkar, S. D., 87891 Visentin, G., 312102 Vishnuvajjala, B., 67990, 68390 Viskocil, J. F., Jr., 382³⁶ Vitale, A. C., 756115 Vite, G. D., 20117 Viti, S. M., 225185, 264181 Vitulio, V. P., 8044 Vlattas, I., 105²¹⁹, 113²¹⁹ Vocelle, D., 903126 Vogel, A. I., 499¹¹¹ Vogel, E., 593179,180, 72626 Vogiazoglou, D., 257121 Vogt, B. R., 62333, 85585, 901113, 903113 Vogt, H. R., 345²⁰ Vogt, R. R., 273⁹ Vögtle, F., 414² Vohra, N., 416¹⁸, 417¹⁸ Vohra, R., 105739 Voigt, E., 909156 Voitkevich, S. A., 63411, 63511 Volante, R. P., 45²⁴⁷, 201⁸⁰ Volkmann, R. A., 35867 Vollhardt, J., 159466, 166466, 174530 Vollhardt, K. P. C., 255¹⁰⁶, 440⁴⁶, 537⁹⁰, 538⁹⁰⁻⁹³, 539¹⁰⁰ Vollmann, H., 887¹² Volodkin, A. A., 81469 Vol'pin, M., 29725, 33425 Voltman, A., 81048 von Auwers, K., 324148 von Bézard, D. A., 43829 von Braun, J., 324151 von der Eltz, H.-U., 904135 von Liebig, J., 821¹, 822⁵, 834⁵ Vora, M., 619²⁴ Voronenkov, V. V., 30681 Vostrowsky, O., 644159 Vrachnou-Astra, E., 565²⁰ Vrencur, D. J., 180549 Vriesema, B. K., 229224,224a Vysotskii, A. V., 328¹⁷² Vystrdil, A., 757124 Wachtmeister, C. A., 6609 Wada, F., 495%, 497103, 53078, 53578, 102641 Waddell, T. G., 735²², 770¹⁷³ Wade, K., 2081, 763151 Wadman, S., 218^{100,100b}, 229²³⁴, 444^{66,68} Wadsworth, A. H., 29070 Waefler, J. P., 63978 Waegell, B., 380^{8,13}, 729⁴¹, 839¹², 840¹², 853⁷⁰, 854¹² Wagenhofer, H., 89353b Wagner, A., 582111 Wagner, C. D., 332²⁰⁷ Wagner, E. R., 49273, 49773 Wagner, G., 705¹ Wagner, P. J., 104812 Wagner, R. B., 843²⁸, 844³⁰ Wagnon, J., 41945, 42045

Wahl, G. H., Jr., 73623, 906143 Wailes, P. C., 82418 Waiss, A. C., Jr., 66645 Wakamatsu, H., 102744 Wakasa, N., 229²³⁰, 246³³, 257³³ Wakefield, B. J., 86⁴, 419^{34,35} Wakselman, C., 257121 Walborsky, H. M., 255¹¹¹, 439³⁸, 482³, 587¹⁴⁶, 610¹⁴⁶. 1040107 Walinsky, S. W., 93673 Walker, E. R. H., 71543 Walker, F. J., 225185, 264181 Walker, J., 6334 Walker, J. A., 528⁴⁶, 799¹⁰², 939⁷⁷ Walker, J. C., 47²⁵⁷ Walker, O. J., 63645,57 Walker, W. E., Jr., 407¹⁵⁰ Wall, A., 878^{93,94}, 879^{93,94}, 880⁹⁴, 881⁹⁴ Wall, D. K., 888¹⁸, 893⁵⁵ Wallace, R. H., 42233, 57264 Wallace, T. W., 502141 Wallach, O., 822⁴, 831⁴, 832⁴ Wallbaum, F., 80942 Wallenberger, F. T., 57480 Walling, C., 154417, 649202 Wallis, A. F. A., 690¹²⁶, 691¹³⁰, 693¹³⁰ Wallis, E. S., 892⁵⁰ Wallis, T. G., 87256 Walpole, A. L., 35138 Walsh, R., 382³⁶ Walter, D., 423⁸¹, 975⁴, 979⁴ Walter, W., 105²¹⁸ Walters, M. E., 61613 Walton, D. R. M., 219¹⁰⁴, 419³¹, 500¹²⁶, 512¹²⁶, 522¹⁷, 555³¹ Walton, J. C., 380⁹ Walzer, E., 56²⁸⁵ Wanat, R. A., 33189, 34198, 199, 39198, 199 Wang, C.-L. J., 212³⁹, 253⁹² Wang, D. G., 888¹⁶ Wang, K. K., 215⁶⁰ Wang, T., 163⁴⁷² Wang, T.-F., 368105 Wang, X., 999⁵¹, 1000^{51b} Wanner, K. Th., 42232 Ward, P., 160471, 161471, 163471, 8031 Ward, R. S., 692137, 81684 Wardleworth, J. M., 511¹⁸⁹ Ware, R. S., 68195 Waring, A. J., 803¹, 804⁶ Warita, Y., 871⁵⁴ Warkentin, J., 89667 Warner, R. W., 174532, 176532, 177532, 178532 Warnet, R. J., 57158, 596193, 72837, 105122, 105222 Warnhoff, E. W., 3807, 78430, 85473 Warpehoski, M. A., 969134 Warr, J. C., 888¹⁶ Warren, C. D., 2737 Warren, K. S., 82210, 82910 Warren, P., 67060 Warren, S., 123²⁵¹, 124²⁶¹, 126²⁶¹, 201⁷⁶, 946⁸⁷ Warrington, B. H., 688¹¹⁴, 690¹¹⁴ Warshawsky, A., 30251 Warwick, P. J., 53077, 53577 Washburn, W., 105840 Washioka, Y., 103267

Wasmuth, D., 43²⁴⁰, 44²⁴⁰ Wasserman, E., 62846 Wassmundt, F. W., 503149, 512149 Wasylishen, R., 38236 Wat, E. K. W., 43092 Watanabe, A., 771¹⁸² Watanabe, H., 244^{27,28}, 464¹⁷⁷, 771¹⁹¹ Watanabe, J., 136³⁷¹, 139³⁷¹, 140^{371,371b}, 143^{371,371b} Watanabe, K., 42683 Watanabe, R., 543117 Watanabe, S., 259¹³¹ Watanabe, T., 72625, 84854 Watanabe, Y., 507¹⁷⁵, 960¹¹⁶, 1018¹⁰, 1021^{10,16}, 1028⁴⁹ Waters, W. A., 660¹⁶, 661²¹ Waters, W. L., 56842 Waterson, D., 1792, 1892,94, 4294 Watkins, J. J., 21341,42b Watt, D. S., 39²¹⁸, 48²¹⁸, 247⁴⁷, 253⁹³ Wattanasin, S., 131328,329, 132329, 135328,329 Watts, L., 85582 Watts, W. E., 8228, 8368, 92135 Waugh, M. A., 34196 Wautier, H., 120244, 142244 Wawzonek, S., 92440 Wayaku, M., 1041111 Weaver, T. D., 55283 Webb, I. D., 42889 Webb, W. P., 81895 Webber, G. M., 81679 Webber, S. E., 21794 Weber, A., 174⁵²⁶ Weber, K.-H., 89034 Weber, M., 87259 Weber, T., 40²²³, 41^{223,227}, 42²²³ Weber, Th., 41227 Webster, O. W., 85584 Wedemeyer, K.-F., 804², 810², 812² Wee, A. G. H., 589¹⁶², 610¹⁶² Weedon, A. C., 590163 Weedon, B. C. L., 553¹², 577⁸⁹, 633⁵, 634^{5,24}, 635^{5,33}, 639^{5,24}, 640^{107,107a}, 642¹¹², 643^{5,117,125,130}, 644112,147,148,1482,150,151,153,166, 64733,107, 64924 Weeks, P. D., 20118, 31118 Weerasooriya, U., 104916, 105316 Weglein, R. C., 70914 Wehman, E., 210²⁵ Wehmeyer, R. M., 209¹⁹, 226²⁰², 263¹⁷¹ Wehner, G., 19740 Wehrli, P. A., 168491, 169491, 171491 Wehrli, S., 3807 Weidhaup, K., 36290 Weidlich, H. A., 6159 Weidmann, H., 57055,129, 58255, 58355,129, 63057, 63157 Weigert, F. J., 103996 Weiler, L., 58290, 99188, 107188, 110188, 396103 Weil-Raynal, J., 5213 Weinberg, H. R., 634¹⁰, 638¹⁰ Weinberg, N. L., 634^{8,10}, 636⁸, 638^{8,10}, 639⁸, 643⁸, 649⁸, 655⁸ Weinberg, R. B., 38010 Weinberger, B., 102639 Weiner, M. L., 84323 Weiner, S. A., 661²⁴ Weinges, K., 872^{59,61-63}, 874⁶⁸ Weinreb, S. M., 257¹¹⁵, 499¹²¹, 511¹⁹¹ Weinshenker, N. M., 380¹³

Weinstein, B., 30575b Weinstock, L. M., 38011 Weintraub, P. M., 890³¹, 903¹²⁴ Weiper, A., 642¹¹³, 643¹¹³, 644¹¹³ Weir, J. R., 53998 Weis, C. D., 38132 Weisenfeld, R. B., 88128, 105217, 124128 Weisgraber, K. H., 66540 Weiss, D. S., 81573 Weiss, J., 63645 Weiss, R. G., 104811 Weiss, U., 380⁷ Weissberger, A., 53²⁷², 892⁵⁰ Weissermel, K., 103998 Weissflog, E., 134339, 135339 Welch, J., 373129 Welch, M., 51270 Welch, R. W., 320133 Welker, M. E., 47257 Wells, D., 117^{235,236}, 155^{235,236}, 156^{235,236}, 946⁸⁷ Wells, G. J., 586154, 86421 Welner, S., 62850 Welzel, P., 583118 Wender, P. A., 31182, 226195, 2427.8, 2577.8, 264183, 38010 Wenderoth, B., 42158 Wendler, N. L., 689119 Wenisch, F., 63428 Wenkert, E., 229²³³, 246³⁸, 380⁷, 444⁶⁵, 446⁸¹, 447⁹⁰⁻⁹². 44897, 45690,126, 457133, 49275, 49383, 50375,146 509¹⁷⁸, 513^{83,146,208}, 572⁶⁸, 715³⁶, 818⁹⁵, 857^{91,92}, 908146, 105122, 105222 Wennerström, O., 499119,120, 593180,183, 594184 Werner, J. A., 94281a, 100868 Werner, L. H., 62951 Wernert, G. T., 7128 Werstiuk, N. H., 86²⁴, 95²⁴, 159²⁴, 164⁴⁷⁷ Werth, R. G., 72622 Wesseler, E. P., 319131 Wessely, F., 80724, 81254,56.59, 81365,66, 81467,68, 81786-88,90-92, 81893 Wessling, D., 753104 West, F. G., 91826 West, J. P., 320¹³³ West, R., 889²⁴ Westberg, H. H., 651²¹⁸ Westdrop, I., 584130 Westerlund, C., 49595 Westermann, J., 42157.58 Westheimer, F. H., 89794 Westman, T. L., 898⁸⁰ Westmijze, H., 21782,85, 219104, 25496 Westphalen, K.-O., 53274 Wettlaufer, D. G., 21788 Wetzel, D. M., 613⁴, 619⁴ Wexler, B. A., 70917 Weyerstahl, P., 124^{268,285}, 125^{268,285}, 126^{268,285}, 127²⁶⁸. 131²⁶⁸, 584¹³², 752⁹² Weygand, F., 887⁴, 888⁴, 893⁴, 897^{4,76}, 898⁷⁶, 900⁴, 903⁴ Whalen, D. L., 903120 Whangbo, M. -H., 21133 Wharton, P. S., 39188, 653226, 89252 Whitby, R., 229²³⁴, 444^{66,67} White, A. M., 333²¹² White, D. A., 642¹¹⁴ White, D. H., 592175 White, F. H., 72⁴⁰, 81⁴⁰

White, J. D., 215⁵⁹, 261¹⁴⁸, 264¹⁴⁸, 351^{43a}, 683¹⁰³, 714³² White, J. F., 30041 White, S., 36²⁰⁹, 37²¹² Whitehouse, R. D., 114234 Whitesell, J. K., 28172, 30172, 31172, 35208, 36210, 38237, 38451, 39337, 78114 Whitesell, M. A., 28172, 30172, 31172, 36210 Whitesides, G. M., 24855, 25155, 26955, 41824,26, 41947, 422⁷¹, 423⁷¹, 482^{3,6}, 494⁸⁷, 502⁸⁷, 557⁴⁶, 599²⁰⁶ Whitham, G. H., 3804, 61612, 73518 Whiting, D. A., 688¹¹⁶ Whiting, M. C., 55418,19 Whitlock, B. J., 55740,43 Whitlock, H. W., Jr., 55740,41,43, 693148, 694148 Whitman, P. J., 905137 Whitmore, F. C., 4157 Whitney, C. C., 483¹² Whitney, R. A., 135³⁶³, 136³⁶³, 139³⁶³, 142³⁶³, 156³⁶³ Whitney, S., 20915 Whittaker, G., 63984 Whitten, C. E., 8⁴¹, 9⁴¹, 15⁴¹, 20⁴¹, 212³⁹, 226^{39a,d,201}, 25070 Whittle, A. J., 34211 Whybrow, D., 43198 Wiberg, K. B., 381¹⁹, 406¹⁴⁵, 890³⁴, 892⁵¹, 901^{105,110}. 905138 Wickenkamp, R., 49593b Wickham, G., 586154, 86421 Widdowson, D. A., 244²⁴, 464¹⁷⁰, 494⁸⁸, 503¹⁴⁸, 514²¹⁰. 681⁹⁷, 753⁹⁹ Widiger, G. N., 57054 Widmer, U., 80939 Wieczorek, J. J., 81679 Wiedhaup, K., 36175 Wieland, D. M., 223¹⁵⁸, 263¹⁷⁷ Wieland, P., 81679 Wiemann, J., 57892, 61092 Wiemer, D. F., 395¹⁰², 396¹⁰² Wierenga, W., 99185 Wiersema, A. K., 66645 Wierzchowski, R., 45251 Wigfield, Y. Y., 149400 Wightman, R. M., 57788 Wijers, H. E., 106222, 113222 Wijkens, P., 21781 Wijsman, A., 24210 Wikel, J. H., 926⁴⁶, 928⁴⁶ Wikholm, R. J., 82848, 82948 Wilcox, C. S., 313 Wilde, H., 89033 Wilde, P. D., 84015 Wilds, A. L., 810⁵¹, 888¹⁹, 891¹⁹, 900¹⁹ Wiley, D. W., 888²¹ Wilgus, H. S., III, 74772 Wilhelm, R. S., 213^{46,46c,54}, 214^{56,57}, 216⁴⁶, 223⁴⁶, 250^{72,73}, 264^{72,186}, 265⁷², 491⁷⁰ Wilka, E.-M., 125²⁹⁰ Wilke, G., 423⁸¹, 587¹⁵⁰ Wilkening, D., 849⁵⁵ Wilkins, C. W., Jr., 438³⁵ Wilkins, J. M., 843, 24963 Wilkinson, G., 208^{1,11}, 210^{11,11a}, 219^{11a}, 228²¹⁴, 234^{11a}, 4369.13, 52433 Wilkinson, J. B., 73521 Willard, J. E., 299³³ Willem, R., 587¹⁴⁸

Willenz, J., 89250 Willert, I., 128322, 130322, 134322 Willhalm, B., 57264 Williams, A. D., 600²¹⁶ Williams, C. C., 145386 Williams, D. J., 902¹¹⁹, 934⁶⁴, 953⁶⁴ Williams, D. L., 125³⁰⁵, 126³⁰⁵, 127³⁰⁵ Williams, D. R., 31185, 32185, 36209, 37212 Williams, F. J., 905138 Williams, G. H., 505162, 507162, 512162 Williams, J. H., 57266 Williams, J. M., 1897, 4297, 4397 Williams, J. R., 39085, 39285, 693142 Williams, L., 19966 Williams, M. E., 505168 Williams, N. R., 73412 Williams, R. M., 27726, 79061 Williams, R. V., 74668 Williams, T. W., 690124 Williams, V. Z., Jr., 334220 Williamson, H., 5111189 Williamson, K. L., 73943 Williard, P. G., 58291, 75187 Willis, H. B., 4156 Willis, W. W., Jr., 87^{100,107,109}, 104¹⁰⁰, 105¹⁰⁷, 106¹⁰⁷. 110100, 114107, 117100, 120107, 157107 Willison, D., 1025^{33,33a} Wills, G. O., 721⁵ Wilshire, J. F. K., 329¹⁸⁶ Wilson, C. L., 7065 Wilson, E. R., 42270, 63539 Wilson, F. B., 38236 Wilson, J. W., 19965,66 Wilson, S. R., 38011, 61613 Wilt, J. W., 714³³, 853⁷¹ Winch, B. L., 79058 Windle, J. J., 66645 Wingard, R. E., Jr., 867³³, 873³³ Winkler, J. D., 24¹⁴⁴, 58²⁹¹, 218⁹⁷, 380⁹ Winkler, T., 381³² Winn, L. S., 539100 Winstein, S., 406142, 653225, 7066 Winter, M., Winter, R., 89667 Winter, R. E. K., 62029 Winter, S. R., 102427, 102851 Wintermitz, F., 85477 Winton, P. M., 41622 Wipf, P., 99443, 99643, 100043 Wirth, E., 82954 Wirth, W.-D., 4141 Withers, G. P., 759131,132, 760135 Witkop, B., 8048, 81046 Witt, K. E., 588¹⁵⁸ Witte, J. F., 201⁸¹ Wittenberg, D., 38123, 38223, 56517 Wittenberger, S. J., 960¹¹⁸ Wittenbrink, C., 644¹⁶⁴ Wittig, G., 914⁷, 915¹³, 924^{7,41}, 965¹³, 975¹ Wittman, M. D., 100764 Wladislaw, B., 644^{148,148a} Woell, J. B., 102850, 103060 Woessner, W. D., 124²⁷², 125²⁷², 126²⁷², 144³⁸³, 145383,386 Wogan, G. N., 499121 Wohl, A., 640¹⁰⁴

Wöhler, F., 8225, 8345 Wohllebe, J., 85163 Wolber, G. J., 66¹⁵, 74¹⁵ Wold, S., 721⁵ Wolf, A., 382³⁸ Wolf, A. P., 1040¹⁰⁹ Wolf, H., 34828, 373130 Wolf, W., 505164, 507164,173, 512164, 515164 Wolfe, S., 147³⁹⁹, 154⁴¹⁸, 155⁴¹⁸ Wölfel, G., Wolff, L., 887^{1a}, 890¹, 891¹, 893^{1a}, 894¹, 900^{1a} Wolff, R. E., 30175 Wolff, S., 19103, 901115,116 Wolinsky, J., 84960, 85061, 105017 Wollenberg, R. H., 24749, 25049 Wollheim, R., 89035, 903127 Wolochowicz, I., 565²⁴, 570²⁴, 583²⁴ Woltermann, A., 105²¹⁴, 509¹⁹⁶, 512¹⁹⁶ Wong, C. K., 107^{225,226}, 109²²⁶, 386⁶⁰ Wong, C. M., 85473 Wong, H., 168491, 169491, 171491 Wong, R. Y., 67266 Wonnacott, A., 174^{527,527a,b}, 175^{527a-c} Woo, S. L., 28040 Woo, S.-O., 689¹²² Wood, A. F., 688¹¹⁶ Wood, A. M., 38349 Wood, G., 564⁸, 727²⁸ Wood, G. W., 380⁵, 381¹⁷ Wood, H. C. S., 24635 Wood, L. L., 81471 Woodard, S. S., 225¹⁸⁵, 264¹⁸¹ Woodgate, P. D., 325¹⁵⁷, 675⁷⁴ Woods, G. F., 84645 Woods, L. A., 2089, 24419 Woods, M. C., 3804 Woodward, P., 38010 Woodward, R. B., 781¹⁴, 810⁴⁴, 914⁹ Woolford, R. G., 63430, 639308.76.79, 64430b, 649209 Worakun, T., 484²¹ Wormser, H. C., 507173 Wovkulich, P. M., 24149, 25149, 26165 Wragg, R. T., 134337 Wright, D. S., 763¹⁵¹ Wright, J. R., 402128 Wrigley, T. I., 74153 Wrobel, J. E., 262169 Wu, A., 50²⁶⁸, 334^{221,221b} Wu, C. N., 755¹¹² Wu, G.-S., 629⁵³ Wu, J. C., 328¹⁸⁰ Wu, T.-C., 209¹⁹, 263¹⁷¹, 353⁵⁰, 354⁵⁰, 565¹⁷ Wu, Y.-D., 985^{26b} Wu, Y.-J., 590¹⁶⁴ Wudl, F., 52846 Wüest, H., 93672, 953101 Wunderli, A., 80940 Wurtz, A., 633³ Wuts, P. G. M., 135^{365,366}, 136^{365,366}, 139^{365,366}, 142365,366, 35655, 103374 Wyatt, J., 67678, 68678 Wykypiel, W., 66¹¹, 71³⁰, 74¹¹ Wynberg, H., 551¹, 586¹⁵¹, 649²⁰⁸, 665³⁹, 689¹²⁰, 832⁶⁸ Wynne-Jones, W. F. K., 63425* Wyvratt, M. J., 623^{32,33}

Xia, Y., 509179 Xie, J., 669^{54,55} Xu, B., 63893 Yadav, A. K., 640106, 644156 Yadav, J. S., 73733 Yagi, H., 665⁴⁰ Yagi, Y., 918²⁷, 968¹²⁸ Yagihara, T., 92343, 100871 Yahata, N., $136^{370,371}$, 137^{370} , $138^{370,371a}$, $139^{370,371}$, $140^{370,371,371a}$, $143^{371,371a}$, 144^{371a} Yamada, H., 100195, 103195, 107195, 36597, 39070 Yamada, J., 44362, 73045 Yamada, K., 469²⁰⁶, 470²⁰⁶, 473²⁰⁶, 489^{58,63}, 495^{58,63}, 504⁶³, 511^{58,63}, 515^{58,63}, 530⁵⁷, 740⁴⁶ Yamada, M., 934⁶³ Yamada, N., 464¹⁷⁵ Yamada, S., 46²⁵³, 99¹⁸⁴, 169⁵¹⁰, 172^{510,511,512,515,516}, 173^{510,512,515,516}, 741⁵⁰ Yamada, T., 639⁷⁵ Yamada, Y., 31199, 34212, 34712, 38130, 38230, 42159, 422⁵⁹, 644¹⁴⁶, 1023²³ Yamagatsu, T., 30043 Yamagiwa, S., 946⁸⁷ Yamaguchi, H., 45²⁴⁸, 135³⁵³, 136³⁵³, 137³⁵³, 141³⁵³. 142353, 541106, 55850, 586138 Yamaguchi, K., 512¹⁹⁸, 867³⁶ Yamaguchi, M., 45²⁴⁸, 225¹⁸⁷, 277³³, 279^{35,36}, 286⁵⁸. 100055, 100156, 100460 Yamaguchi, R., 903125 Yamaji, T., 244²⁷, 464¹⁷⁷ Yamakami, N., 102744 Yamamoto, A., 528⁴⁷, 530⁸⁰, 535⁸⁰ Yamamoto, H., 96162,166,167, 99162,163,166,167, 103166,167 $\begin{array}{c} 104^{162,163}, 105^{166,216}, 112^{216}, 120^{162,163,166}, 121^{162,163}, \\ 131^{162,163}, 155^{166}, 202^{97}, 229^{225}, 246^{35}, 349^{33}, 354^{60}, \end{array}$ 358³³, 446^{85,86,88}, 469²⁰³, 470²⁰³, 473²⁰³, 483¹³, 750⁸⁶, 787^{47,48} Yamamoto, K., 228222, 592173, 593181,182, 62849 Yamamoto, M., 74046 Yamamoto, N., 771¹⁹¹ Yamamoto, T., 53080, 53580 Yamamoto, Y., 43²³⁶, 87^{97,98}, 99^{98,183}, 100^{98,183}, 105¹⁸³, 157⁹⁸, 196³¹, 212³⁶, 221³⁶, 222¹³⁶, 226^{36,200}, 232²⁶⁷, 24967, 262168, 35462, 44362, 48310, 510185, 62339, 75188 Yamamura, M., 436⁵, 437^{5,22}, 438²², 485³⁶, 491³⁶. 49436, 49736 Yamamura, S., 390^{74,86}, 392⁷⁴, 395¹⁰⁰, 396^{113,115}, 397¹¹⁶, 398¹¹³, 665³⁸, 676⁷⁹, 690¹²⁵, 691^{128,131}, 693^{128,139}, 694¹⁵¹, 695⁷⁹, 696¹⁵¹, 697^{38,131,139,155,156}, 698¹²⁸, 131 769171 Yamana, M., 21898 Yamanaka, H., 460143, 461143, 147, 148, 512199, 53064, 53364, 541108-111,113,114, 543108-111,113,117,118 Yamanaka, Y., 82957 Yamane, K., 818⁹⁷ Yamaoka, S., 571⁷⁴, 574⁷⁴ Yamashina, N., 231248, 44356 Yamashita, A., 42379 Yamashita, H., 1016³, 1039¹⁰¹ Yamashita, J., 466¹⁸⁷, 500¹³³ Yamashita, M., 34¹⁹⁵, 35²⁰², 135^{356,360,361,362}, 136^{356,360,361,362,371}, 137^{360,361}, 139^{356,360,361,362,371}, 140^{371,371b}, 142^{360,361}, 143^{360,361,371,371b}, 156^{360,361,362,371}, 156^{360,361,361,371,371b}, 156^{360,361,362,371}, 156^{360,361,371,371b}, 156^{360,361,362,371}, 156^{360,361,362,371}, 133^{360,361,371,371b}, 156^{360,361,362,371}, 156^{360,361,361,362}, 156^{360,361,361,361,362}, 156^{360,361,361,362}, 156^{360,361,361,362}, 156^{360,361,361,360,361,360,361,361,360,360,361,36} 102952

Yamashita, Y., 24422, 465179, 49489, 51689 Yamataka, K., 64099, 67265 Yamato, H., 84853, 92344, 93444, 95444, 100870,71 Yamato, T., 329184,185 Yamauchi, T., 789⁷⁰ Yamaura, Y., 10476 Yamawaki, A., 843²⁵ Yamaya, M., 243¹¹ Yamazaki, T., 639⁸¹, 853⁷² Yamoto, H., 91932 Yan, T.-H., 586155, 610155 Yanagi, T., 504¹⁵⁴, 511¹⁵⁴, 515¹⁵⁴ Yanagisawa, A., 9⁴⁷, 10⁴⁷ Yanagisawa, K., 43722, 43822, 48536, 49136, 49436, 49736 Yanagita, M., 81045 Yanagiya, M., 100199 Yanase, M., 224¹⁸¹ Yang, J., 669⁵⁵ Yang, N. C., 38344 Yang, Y., 159464, 161464, 166464, 232259 Yang, Y.-L., 7304 Yang, Z.-u., 323146 Yang, Z.-Y., 53067, 53167, 53367, 53893 Yannakopoulou, K., 28247 Yano, T., 24966, 465178, 470178, 473178 Yao, A. N., 918²¹, 922³⁹, 939³⁹ Yarneil, T. M., 370¹¹⁰, 373¹²⁷ Yasuda, K., 41224 Yasuoka, N., 67265 Yatagai, H., 8798, 9998,183, 10098,183, 105183, 15798, 196³¹, 249⁶⁷, 470¹⁹⁵, 473¹⁹⁵, 483¹⁰ Yates, G. B., 635⁴⁰, 637^{65,66}, 650⁶⁶ Yates, P., 597¹⁹⁸, 888²¹, 890^{33,35}, 891^{38,41b}, 903¹²¹, 906¹⁴² Yatsimirsky, A. K., 499125, 66953 Yau, C.-C., 1899, 19100 Yavari, I., 382³⁶ Yax, E., 5647 Yeakey, E., 92440 Yeates, C., 229²³⁴, 264¹⁷⁸, 444^{66.67} Yee, K. C., 8⁴², 756¹¹⁵ Yeh, M. C. P., 20918 Yeh, M. K., 35²⁰³, 66¹³, 197³⁵, 500¹²⁷, 501¹²⁷ Yelm, K. E., 248⁵¹ Yeo, H. M., 693142 Yeong, Y. C., 744⁶² Yi, C. S., 629⁵⁴ Yimenu, T., 34196 Yoder, J. E., 57268 Yogo, T., 446⁷⁶ Yokoo, S., 100195, 103195, 107195, 39069,70 Yokota, K., 228²¹⁹, 246³⁷, 442⁵⁰, 639⁸⁵ Yokota, K.-i., 470²¹⁹, 471²¹⁹, 472²¹⁹ Yokoyama, H., 262¹⁶⁴, 466¹⁹¹ Yokoyama, M., 567³⁴, 570³⁴ Yonashiro, M., 39²¹⁷ Yoneda, F., 21672, 62339 Yoneda, N., 326163 Yonekawa, Y., 124259, 125259 Yoo, S.-e., 28863 Yordy, J. D., 595¹⁹¹ Yoshida, J., 19852, 469203, 470203, 473203, 48316, 500129 Yoshida, K., 741⁵² Yoshida, T., 217⁸⁶, 259¹³⁷, 489⁵⁷, 495⁵⁷, 523²³, 795^{81,85}, 799101 Yoshida, Z., 221131, 231242, 42048,49, 44361, 44561, 463157,159, 102323, 103270, 103371, 1040103

Yoshifuji, M., 71³¹⁻³³, 483¹⁴ Yoshikawa, K., 244²⁵, 267²⁵, 494⁸⁴ Yoshikoshi, A., 956¹⁰⁷ Yoshimoto, M., 24147, 91933, 95433 Yoshioka, Y., 86736 Yoshiura, K., 103267 Yoshizawa, S., 647194 Yost, W. L., 56633 Young, D. M., 6134, 6194 Young, D. W., 499¹²⁴, 507¹²⁴, 898⁸⁴ Young, F., 79479 Young, G. B., 323146 Young, L. B., 30575b Young, M. W., 8763, 11463, 11763 Young, R. N., 28967, 771 185 Young, S. D., 105532 Youssefyeh, R. D., 81895 Yu, C.-F., 173⁵¹⁷ Yu, L.-C., 124253, 126253, 93775 Yuan, S.-S., 73044 Yue, S., 599²¹¹ Yuh, Y., 382³⁶ Yukizaki, H., 246³⁹, 257³⁹ Yuldashev, Kh. Yu., 321¹⁴⁰ Yulina, V. I., 306⁸¹ Yurchenko, A. G., 30250, 38347 Yus, M., 253⁹⁰, 788⁵⁰ Yusupov, N. U., 30358 Zabkiewicz, J. A., 382³⁶ Zablocka, M., 20178,79 Zagatti, P., 263¹⁷³ Zago, P., 595189, 606189 Zahler, R. E., 512201 Záhorszky, U. I., 66427 Zahra, J.-P., 57791 Zajdel, W. J., 65², 66¹², 68^{2,22}, 69^{2,22}, 71², 74¹², 194¹¹ Zakharkin, L. I., 30357, 89881 Zalar, F. V., 649²⁰⁶ Zalesov, V. S., 887¹⁰, 888¹⁰, 889¹⁰, 890¹⁰, 893¹⁰, 897¹⁰. 90010, 90310 Zamarlik, H., 34210 Zambon, R., 27316, 27416 Zamecnik, J., 20105 Zanarotti, A., 691127,132,134 Zanasi, R., 38657 Zapata, A., 983²¹, 984^{21,21a} Zapevalov, A. Y., 644137 Zarecki, A., 84959 Zaslona, A., 105737 Zatorski, A., 953105 Zavarin, E., 686111 Zavgorodnii, S. V., 30570,74 Zawacky, S. R., 71745, 75294 Zawoiski, S., 53065, 53365 Zbiral, E., 19954, 81256, 81365, 81468, 81786,89, 93976 Zee, J., 84644 Zeelen, F. J., 361⁷⁶, 367^{76,100,103}, 371¹¹⁶ Zefirov, N. S., 86421 Zeigler, F. E., 588¹⁶⁰, 610¹⁶⁰ Zelle, R. E., 220¹²⁴ Zeller, K.-P., 887¹¹, 891⁴², 892¹¹, 893¹¹, 897¹¹, 898¹¹, 90011, 90311, 90511, 909155 Zembayashi, M., 437^{21,25}, 440²⁵, 448²⁵, 449^{25,99,100} 45025, 45125, 45225, 464171, 48426, 48745, 49226,74, 494²⁶, 495²⁶, 500¹²⁹, 503²⁶, 513²⁶

Zenk, M. H., 77⁵⁹ Zerby, G. A., 136³⁷³, 137³⁷³ Zhai, D., 277²⁶ Zhai, W., 277²⁶ Zhang, C., 669^{54,35} Zhang, L., 638⁹³ Zhang, Y., 21¹²⁸, 251⁷⁸, 254⁷⁸ Zhang, Y., 21¹²⁸, 251⁷⁸, 254⁷⁸ Zhao, K., 125³⁰², 134³⁰² Zhao, M., 638⁹³ Zheikovskaya, V. P., 643¹³¹ Zhou, J., 669^{54,55} Zhukov, A. G., 302⁵⁰ Zibuck, R., 224¹⁷³ Ziegenbein, W., 271², 272² Ziegler, F. E., 131³²⁵, 215⁶⁴, 219¹¹⁴, 226¹⁹⁸, 368¹⁰⁵, 499¹⁴⁰, 501¹⁴⁰, 502¹⁴⁰, 994⁴⁰ Ziegler, K., 194⁵ Ziffer, H., 734⁸, 767¹⁶⁴ Zikra, N., 249⁶⁵

Zimero, C., 57266 Zimmer, H., 459138 Zimmerman, H. E., 158442, 164442, 92237, 92437 Zimmerman, J., 41230 Zimmermann, R. G., 43939 Zipkin, R. E., 278³¹, 289³¹, 558^{51,52} Zoller, U., 86⁷³⁷, 872³⁷, 883³⁷, 884³⁷ Zook, H. D., 13⁶⁷, 723¹⁰ Zoutendam, P., 595191 Zovko, M. J., 499114 Zsindely, J., 80939,40 Zuccarello, F., 38657 Zuccarello, G., 883106,107 Zugravescu, I., 92136 Zuman, P., 56632 Zutterman, F., 78644 Zwanenburg, B., 750^{82,83}, 868⁴¹ Zweifel, G., 199^{57,58}, 259^{133,134}, 483^{12,13}, 486⁴⁰, 489⁵⁷, 495^{40,57}, 497⁴⁰, 498⁴⁰, 503⁴⁰, 553¹⁶ Zwenger, C., 6603

Subject Index

Abietic acid dioxo ester rearrangement, 834 Abstraction hydrogen atom recombination, 1046 Acenaphthenes synthesis Friedel-Crafts cycloalkylation, 325 Acenaphthoquinone reaction with hydroxides, 828 Acenaphthyne synthesis Ramberg-Bäcklund rearrangement, 883 Acerogenin related ethers synthesis, 688 Acetals cyclization Lewis acid induced, 362 reactions with organocopper compounds, 226 Acetals, allylic reaction with organocopper compounds, 227 Acetals, 2-halo rearrangements, 788 Acetals, propargylic reaction with organocopper compounds, 227 Acetamide, α -allyloxy-Wittig rearrangement, 1004 Acetic acid, α -allyloxyesters, Wittig rearrangement, 1008 zirconium enolates, 1000 8-phenylmenthyl ester Wittig rearrangement, 1001 Acetic acid, 2,2-dialkylsynthesis, 53 Acetic acid, diazoethyl ester C-H insertion reactions, 1051 Acetic acid, phenylethyl ester acyloin coupling reaction, 619 methyl ester acyloin coupling reaction, 619 Acetic acid, trialkylesters synthesis, 644 Acetic acid, trifluorocatalysis epoxide ring opening, 738 Acetic acid, trimethylsilylethyl ester acyloin coupling reaction, 619 Acetoacetic acid methyl ester γ-alkylation, 58 Acetodiazoacetic acid

ethyl ester synthesis, 889 Acetone, geranylcyclization, 349 Acetonitrile, bromocoupling reactions with arylzinc reagents, 260, 466 Acetonitrile, dichloroalkylation, 794 Acetonitrile, (diethoxyphosphoryl)oxide reaction with alkenes, 201 Acetophenone, benzylideneoxide benzylic acid rearrangement, 830 Acetophenone, diazorearrangements, 887 Acetophenone, p-ethylethylation Friedel-Crafts reaction, 301 Acetylene (see also Alkynes) monometallation, 271 Acetylene, di-t-butylsynthesis Ramberg-Bäcklund rearrangement, 883 Acetylene, dilithiosynthesis, 271 Acetylene, diphenylsynthesis Ramberg-Bäcklund rearrangement, 883 Acetylene, ethoxyhydrozirconation, 498 Acetylene, lithiosynthesis, 271 Acetylene, sodiosynthesis, 271 Acetylene, trimethylsilylin terminal alkyne synthesis, 531 Acetylides organometallic coupling reactions, 1-haloalkynes, 553 oxidative coupling reactions, 554 Acid anhydrides acyloin coupling reaction, 617 Acid chlorides α,β -unsaturated reaction with diazomethane, 889 Acidic chalcogenides catalysts Friedel-Crafts reaction, 296 Acidic oxides catalysts Friedel-Crafts reaction, 296 Acidic sulfides catalysts Friedel-Crafts reaction, 296 Acrolein, a-lithio-

1132

Subject Index

synthesis, 253 Acrylates synthesis rearrangement of epoxides, 760 Acrylates, B-lithiosynthesis, 253 Acrylates, a-vinylsynthesis copper catalysts, 217 Actinidine synthesis, 599 Acyl anion equivalents selenium containing alkylation, 134, 136 sulfur containing alkylation, 134 Acyl chlorides acyloin coupling reaction, 617 coupling reactions with sp³ organometallics, 463 Acyl cyanides decarbonylation palladium-catalyzed, 1041 Acvl halides decarbonylation, 1040 reactions with organocopper reagents, 226 Acylimonium ions initiators polyene cyclization, 342 Acyloin rearrangement 2-hydroxy ketones, 791 Acyloins coupling reactions, 613-631 heterocyclic systems, 629 cvclic synthesis, 620 synthesis epoxide ring opening, 753 Acyloxallyl cations initiators polyene cyclization, 343 Adamantane alkylation Friedel-Crafts reaction, 334 arylation Friedel-Crafts reaction, 322 rearrangements, 854 synthesis Friedel-Crafts reaction, 334 Adamantane, 1-bromoreaction with naphthalene Friedel-Crafts reaction, 302 2,4,6,8-Adamantane, 1,3-dilithio-5,7-dimethylmethylation, 134 Adamantane-1-carboxylic acid ethyl ester acyloin coupling reaction, 619 Adenosine, 8-bromocoupling reactions with Grignard reagents, 462 Adrenosterone synthesis, 24 Aerothionin biosynthesis, 689 Africane biosynthesis, 404

Africanol biosynthesis, 404 Alane, alkenylalkylation, 259 reactions nickel catalysis, 230 Alane, methylalkenylsynthesis, 529 Alane, α -silylallylation, 259 Alane, B-stannylallylation, 259 Alanes semipinacol rearrangement, 730 Alanine bislactim ethers from, 53 Alaninol, S-phenylmethyl ether lithiated imine, 37 Alcohols arene alkylation Friedel-Crafts reaction, 309 β-chiral synthesis, 797 synthesis organoboranes, 793 Alcohols, 2-aminodiastereoselective synthesis, 596 rearrangements, 778, 781 semipinacol rearrangements, 777 Alcohols, 2-amino-1,2-diaryl rearrangement, 782 Alcohols, 2,3-epoxyreaction with organocopper compounds, 225 Alcohols, tertiary synthesis cyanoboronates, 798 triorganylboranes, 780 Aldehydes α -alkylated enantioselective synthesis, 35 synthesis, 26 alkylation, 20 alkyl enol ether derivatives alkylation, 25 homologation, 897 intermolecular pinacol coupling reactions, 570 metal enolates alkylation, 3 optically active synthesis, hydroformylation of prochiral alkenes, 1022 γ-οχο synthesis, 103 saturated metal enolates alkylation, 20 synthesis alkylboronic esters, 797 carbonylation, 1021 a-heterosubstituted sulfides and selenides, 141 organoboranes, 793 tri-n-butyltin enolates alkylation, 20 α , β -unsaturated dienolates, alkylation, 25 β,γ -unsaturated

synthesis, 934 γ,δ -unsaturated synthesis, 103 Aldehydes, *β*-keto γ-alkylation, 58 Aldimines lithiated α,β -unsaturated alkylation, 33 pinacol coupling reactions, 580 Aldoximes dianions alkylation, 35 Alkanes alkylation Friedel-Crafts reaction, 332 arene alkylation Friedel-Crafts reaction, 322 Alkanes, 1,1-bis(ethylthio)alkylation, 123 Alkanes, chlorophenylcycloalkylations Friedel-Crafts reaction, 324 Alkanes, diazoaddition to ketones, 783 Alkanes, a, w-dibromomonoarylation with aryl Grignard reagents, 464 Alkanes, a.w-dichlorobenzene alkylation by Friedel-Crafts reaction, 318 Alkanes, α,ω-diethynyloxidative coupling, 557 Alkanes, 1,1-diselenocarbonyl compound synthesis from, 142 Alkanes, 1,1-disulfinylreaction with allylic epoxides synthesis of macrolides, 177 Alkanes, *β*-hydroxyarylsynthesis Friedel-Crafts reaction, 313 Alkanes, 1-metallo-1,1-bis(alkylthio)in synthesis, 123 Alkanes, 1-metallo-1, 1-bis(dithio)alkylation, 121 Alkanes, 1-metallo(phenylthio)in synthesis, 123 Alkanesulfonic acid, perfluorocatalyst Friedel-Crafts reaction, 297 coupling reactions with sp^3 organometallics, 455 Alkanesulfonyl bromide, a-haloreaction with alkenes, 879 Alkanoates enolates, 45 Alkanoic acid, arylesters synthesis, 778 optically active esters synthesis, Friedel-Crafts reaction, 312 synthesis, 788 Friedel-Crafts reaction, 316 Alkanoic acid, w-chlorobenzene alkylation Friedel-Crafts reaction, 303 Alkanols, aryl-

cycloalkylation Friedel-Crafts reaction, 325 Alkatrienes synthesis, 644 Alkenes alkylation Friedel-Crafts reaction, 331 arene alkylation Friedel-Crafts reaction, 304 bicyclic oxides opening, 734 carbonylation palladium salt catalyst, 1030 coupling reactions, 482 with aryl compounds, 492 with heteroaryl compounds, 497 crossed coupling reactions, 484 deuterium-labeled synthesis, 867 difunctional coupling reactions with sp³ organometallics, 448 dimerization, 482 one-carbon homologation via Ramberg-Bäcklund rearrangement, 862 pinacol coupling reactions with carbonyl compounds, 598 reaction with Kolbe radicals, 646 remote carboxyl groups synthesis, 862 stereochemistry in coupling reactions, 436 synthesis alkenylalkyldimethoxyboronates, 799 alkenylboranes, 795 α -alkylation of γ -substituted allyl phosphonates, 202 alkylboranes, 795 π -allylnickel halides, 426 carboxylic acids, 652 1,1-dibromoalkanes, deprotonation, 202 sulfides or selenides, 114 tetrasubstituted synthesis, 864 Alkenes, w-bromosynthesis, 247 Alkenes, 1,1-diarylsynthesis, 864 Alkenes, 1,1-diselenoreduction, 106 Alkenes, fluorosynthesis, 420 2-Alkenoic acid deconjugated alkylation, 51 6-Alkenoic acid Kolbe electrolysis, 640 Alkenyl bromides coupling reactions with Grignard reagents and alkyllithium reagents, 437 Alkenyl chlorides coupling reactions with Grignard reagents, 437 Alkenyl halides coupling reactions with sp³ organometallics, 436 reaction with 1-alkynes, 539 reaction with organocopper compounds, 217

Alkenyl iodides coupling reactions with Grignard reagents, 439 Alkenyl pentafluorosilicates coupling butadiene synthesis, 483 Alkylarsino compounds halogenolysis, 203 Alkylation acyclic ketone enolates extraannular chirality transfer, 17 acyl anion equivalents sulfur or selenium derivatives, 134 aldehydes metal enolates, 3 alkanes Friedel-Crafts reaction, 332 alkenes Friedel-Crafts reaction, 331 alkyl sulfonates, sultones and sulfonamides, 179 alkynes Friedel-Crafts reaction, 332 alkynides, 272 alkyl halides, 272 epoxides, 277 alkynyl carbanions, 271-292 allene carbanions, 256 angular 1-decalone lithium 1(9)-enolate, 16 arenes Friedel-Crafts reaction, 298 polyfunctional alkylating agents, 317 with alcohols, 309 with alkanes, 322 with alkenes, 304 with alkyl halides, 299 with epoxides, 309 with esters, 309 with ethers, 309 with lactones, 309 aryl carbanions, 259 axial 4-t-butylcyclohexanone, 13 carbanions heteroatom-stabilized, 193-204 nitrogen-stabilized, 65-82 nonstabilized, 207-233 sulfur- and selenium-containing, 85-181 carbonyl compound nitrogen derivatives regiochemistry, 28 stereochemistry, 28 diastereoselective acyclic carboxylic acids, 44 acyclic enolates of carboxylic acid derivatives, 42 carboxylic acid enolates, 39 β-dicarbonyl compounds, 54, 58 dienolates α , β -unsaturated carboxylic acids, 50 1,1-(dithio)allyl metals, 131 1,1-(dithio)propargyl metals, 131 enantioselective synthesis, 35 enolates, 1-58 stereochemistry, 12 enols, 1-58 equatorial 4-t-butylcyclohexanone, 13

Friedel-Crafts, 293-335 heteroaromatic carbanions, 260 ketones metal enolates, 3 masked carboxylic acid anions asymmetric syntheses, 53 metal dienolates α,β -unsaturated ketones, 21 metal enolates carboxylic acid derivatives, 39 α-metalloalkyl selenoxides, 157 α-metalloalkyl sulfones, 158 α-metalloalkyl sulfoxides and selenoxides, 147 1-metallo-1,1-bis(dithio)alkanes synthetic applications, 121 a-metalloorthoselenoformates, 144 α -metalloorthothioformates, 144 α -metallovinyl selenides, 104 a-metallovinyl selenoxides, 157 α-metallovinyl sulfides, 104 α -metallovinyl sulfone, 173 S_N2' process, 257 α -selenoalkyllithium, 88 α -selenoallyllithium, 95 a-selenobenzylmetal, 94 a-selenopropargylic lithium derivatives, 104 silyl enol ethers, 25 sp^2 centers epoxides, 262 stabilized metal enolates, 54 sulfur ylides, 178 synthesis saturated aldehyde metal enolates, 20 α -thioalkyllithium, 88 α -thioallyllithium, 95 a-thiobenzylmetal, 94 a-thiopropargylic lithium derivatives, 104 vinyl- and aryl-lithium compounds, 247 vinyl carbanions, 241-266 alkyl halides, 242 heteroatom substituted, 252 vinyl Grignard reagents, 242 Alkyl carbenoids insertion reactions, 1051 Alkyl fluorides Friedel-Crafts reactions, 294 mixture with antimony fluoride Friedel-Crafts reaction, intermediate, 299 Alkyl halides alkylation arenes, 299 sulfur- and selenium-stabilized carbanions, 86 vinyl carbanions, 242 carbonylation formation of esters, 1028 coupling reactions sodium metal, 414 with sp^3 carbon centers, 426 with sp^2 organometallics, 464 Friedel-Crafts reactions alkylating agents, 294 haloalkylation, 118 reactions with π -allylnickel halides, 424 reactions with organocopper reagents, 215 secondary coupling reactions with sp^2 organometallics, 466

 α -Alkylidenation sulfur oxidative removal, 26 Alkylidene carbenes insertion reactions, 1049 Alkylidene carbenoids insertion reactions, 1050 Alkyl sulfinates reactions with organocopper reagents, 215 Alkyl tosylates coupling reactions with sp^2 organometallics, 466 Alkyneboron difluorides reaction with oxiranes, 279 Alkynes alkylated synthesis, 799 alkylation Friedel-Crafts reaction, 332 carbonylation nickel tetracarbonyl catalyst, 1027 cyclic synthesis, 553, 556 heterocyclic coupling reactions with alkenyl bromides, 539 hydroboration organopalladium catalysis, 231 hydroesterification formation of α , β -unsaturated esters, 1030 metallation, 271 pinacol coupling reactions with carbonyl compounds, 602 synthesis organoboranes, 780 organocopper compounds, 217 Ramberg-Bäcklund rearrangement, 883 terminal coupling reactions, 551 oxidative homocoupling, 552 reaction with sp^2 carbon halides, 530 stereospecific synthesis, 539 synthesis, 531 ylidic rearrangements, 963 1-Alkynes hydroalumination asymmetrical diene synthesis, 486 hydrozirconation asymmetrical diene synthesis, 486 reaction with alkenyl halides, 539 synthesis from dichloromethyllithium, 202 vinylation, 521 4,5-Alkynes cyclization, 344 6,7-Alkynes cyclization selectivity, 344 Alkynes, alkoxysynthesis organocopper compounds, 217 Alkynes, a-aminosynthesis, 282 Alkynes, arylconjugated one-pot synthesis, 539 one-pot synthesis, 541 Alkynes, bromo-

Chodkiewicz-Cadiot reaction, 553 reaction with trialkylaluminum, 285 Alkynes, chlororeaction with tertiary enolates, 284 Alkynes, dialkylcyclization selectivity, 344 Alkynes, halocoupling reactions organometallic acetylides, 553 electrophilic substitution, 284 Alkynes, silylstannylreaction with alkenyl iodide, 539 Alkynes, trimethylsilylmethylsynthesis, 281 Alkynides alkylation, 272 alkyl halides, 272 sulfates, 272 synthesis, 272 Alkynones pyrolysis, 1049 Alkynylation vinyl organometallic reagents, 521 Alkynyl halides cross-coupling reactions organometallic reagents, 522 reactions with 1-alkenyl metals, 529 reaction with organocopper compounds, 219 Allene carbanions alkylation, 256 Allenes epoxides rearrangement, 741 pinacol coupling reactions with carbonyl compounds, 605 synthesis $sp^2 - sp^2$ coupling, 491 ylidic rearrangements, 963 Allenes, bromocoupling reactions alkyl Grignard reagents, 439 dimerization, 491 reaction with alkynes, 531 reaction with cyanocuprates, 491 reaction with lithium dialkylcuprates, 217 Allenes, iodoreaction with arylchlorozinc, 491 Allenes, methoxydeprotonation, 256 α-Allenyl phosphates reaction with organocopper reagents, 223 Allobetulone, 2-diazophotolysis, 903 Allodunnione synthesis, 828 Allogeraniol cyclization, 345 Alloxan rearrangement, 822, 834 labeling studies, 823 Alloxanic acid synthesis, 822 Allylamines carbonylation formation of pyrrolidones, 1037

1136

Subject Index

synthesis, 258 Allylamines, 2-arylsynthesis, 492 Allylation enolates palladium-catalyzed regioselective, 12 a-selenoalkylmetals, 91 sulfur- and selenium-stabilized carbanions, 88 Allylic alcohols arene alkylation Friedel-Crafts reaction, 322 N,N-diisopropyl carbamates oxa-allylic anions, 196 Allylic alcohols, 1-(trimethylsilyl)rearrangement formation of lithium homoenolates, 197 Allylic cations initiators polyene cyclization, 342 Allylic electrophiles reaction with organocopper compounds, 220 Allylic halides arene haloalkylation Friedel-Crafts reaction, 321 carbonylation formation of aldehydes, 1021 coupling reactions with sp^3 carbon centers, 428 with sp^2 organometallics, 467 haloalkylation, 118 reaction with ethyl diazoacetate, 925 reaction with vinyltin compounds organopalladium catalysis, 232 Allyl shifts cyclohexadienones, 809 Alnusone synthesis, 126, 505 Aluminates, tetraalkylcoupling reactions with acyl chlorides and acid anhydrides, 463 Aluminum, alkenylalkylation, 259 coupling reactions with allylic chlorides, 475 with aryl halides, 495 with vinyl halides, 486 Aluminum, alkylthioallylreaction with allylic halides, 99 Aluminum, trimethylcoupling reactions with difunctional alkenes, 449 Aluminum alkynide, ethylreaction with 3,4-epoxycyclopentene, 279 Aluminum alkynides alkylation, 274 Aluminum chloride Friedel-Crafts alkylations catalyst, 294 propylene oxide, 769 Aluminum isopropoxide epoxide ring opening, 770 Aluminum phenoxide catalyst Friedel-Crafts reaction, 296 Dowex resin bond catalyst, Friedel-Crafts reaction, 297

Aluminum reagents organopalladium catalysis, 230 Aluminum tribromide catalyst Friedel-Crafts reaction, 295 Aluminum trichloride catalyst Friedel-Crafts reaction, 295 graphite-intercalated, catalyst Friedel-Crafts reaction, 298 Amberlite IR-112 catalyst Friedel-Crafts reaction, 296 Amides cyclic deprotonation, 66 deprotonation, 65 synthesis carbonylation, 1034 Amides, N.N-dialkyldeprotonation with lithium dialkylamides, 45 Amidines, methylthioalkylation, 88 Amines macrocyclic synthesis, 969 Amino acids asymmetric synthesis, 53 α -Amino acids esters imines, alkylation, 46 L-α-Amino acids synthesis, 796 Amino acids, N-acylsynthesis cobalt-catalyzed carbonylation, 1027 α -Amino acids, β , γ -unsaturated synthesis, 117 Aminoacyl anions reactions with electrophiles carbonylation reaction, 1017 a-Amino ketones diazotization synthesis of α -diazo ketones, 890 Aminomethylation Grignard reagents, 258 Ammonium bromide, phenacylbenzyldimethyl-Stevens rearrangement, 913 Ammonium salts, allyltrialkylreaction with Grignard reagents, 246 Ammonium salts, tetraalkylintermolecular pinacol coupling reactions, 568 Amoxycillin synthesis 2-arylglycines, 303 Amphimedine synthesis organopalladium catalysts, 232 Ampicillin synthesis 2-arylglycines, 303 Anacyclin synthesis, 558 Ancistrocladine synthesis, 506

Androstadienedione rearrangement, 804, 810 Androstadienone, 4-methylrearrangement, 805 Androstenone epoxide rearrangement, 738 5B-Androst-9(11)-en-12-one, 3a, 17B-diacetoxy-11-hydroxyrearrangement, 833 Aniline derivatives formylation, 969 Anilines isopropylation Friedel-Crafts reaction, 302 Anilines, o-alkynylsynthesis, 543 Anilines, o-allyl-N-substituted carbonylation, 1038 Anisole Friedel-Crafts alkylation, 300 oxidative coupling, 669 Anisole, 2,6-dimethylbenzylation, 300 Anisole, 2-methoxythiometallated alkylation, 135 Anisole, trimethylsilylmethylthioalkylation, 137 metallated alkylation, 135 Anisomycin synthesis, 77 Annulenes synthesis, 594 Annulenes, dehydrosynthesis, 556 Anthramycin synthesis, 487 palladium-catalyzed carbonylation, 1038 Antibiotic A-23187 — see Calcimycin Antimonic acid, fluorocatalyst Friedel-Crafts reaction, 297 Antimony trifluoride mixture with alkyl fluoride Friedel-Crafts reaction, intermediate, 299 Aphidicolin synthesis, 717 epoxide ring opening, 752 rearrangement of allylic epoxides, 762 Aplysiatoxin synthesis, 126, 168 Aplysiatoxin, debromosynthesis, 126 Aplysin synthesis, 783 Apogossypol hexamethyl ether synthesis, 665 Aporphine synthesis, 507 Aquillochin synthesis, 691 Arachidonic acid

eicosanoid metabolites synthesis, 217 Arachidonic acid, 3-dehydrosynthesis, 250 Arachidonic acid, 5,6-dehydrosynthesis, 247 Arenediazonium salts carbonylation, 1026 Arenes alkylation Friedel-Crafts reaction, 298 regiospecific alkylation Friedel-Crafts reaction, 303 Arenes, bromocarbonylation palladium catalysts, 1018 Arenes, methylintramolecular isomerization Friedel-Crafts reaction, 328 Argentilactone synthesis, 168 Aristolactone synthesis Wittig rearrangement, 1010 Arndt-Eistert synthesis a-diazo ketones
synthesis, 888 Wolff rearrangement, 897 Aromadendrene synthesis, 390 Aromatization photochemical cyclohexadienones, 813 Aroyl chlorides acyloin coupling reaction, 617 Arsonium ylides reaction with carbonyl compounds formation of alkenes and epoxides, 203 Artemisia ketone synthesis, 869 use of ylidic rearrangements, 964 a-Arylation metal enolates regioselectivity, 12 Aryl carbanions alkylation, 259 Aryl compounds coupling reactions, 499 with alkenes, 492 crossed coupling reactions, 501 dimerization, 499 intramolecular coupling reactions, 505 Aryl compounds, allylchiral synthesis, 246 Aryl halides carbonylation palladium catalysts, 1021 cross-coupling reactions organometallic reagents, 522 reaction with organocopper compounds, 219 Aryloxy radicals oxidative coupling, 660 Aryl phosphates coupling reactions with sp³ organometallics, 455

Aryl triflates carbonylation formation of esters, 1029 coupling reactions with sp^3 organometallics, 455 with vinylstannane, 495 cross-coupling reactions with terminal alkynes, 531 reaction with cyanocuprates, 219 Asatone synthesis, 697 Asatone, bisdemethoxysynthesis, 697 Aspartamine synthesis, 543 Atherosperminine synthesis, 586 Atisiran-15-one synthesis, 715 Atrolactic acid synthesis, 829 Atromentin synthesis, 828 Azacycloheptanes synthesis acyloin coupling reaction, 629 3-Aza[10]paracyclophane synthesis acyloin coupling reaction, 629 Azelaic acid Kolbe electrolysis, 640 Azepine, perhydroformamidines alkylation, 72 Azepinone synthesis pinacol rearrangement, 729 Azetidine-2,4-diones synthesis rhodium-catalyzed carbonylation, 1037 Azetidinone, 4-acetoxysynthesis, 651 Azides, acyl rearrangement, 908 Aziridine, 2-methylarene alkylation by Friedel-Crafts reaction, 316 Aziridines arene alkylation by Friedel-Crafts reaction, 316 carbonylation formation of B-lactams, 1036 reaction with organocopper reagents, 224 Azirine carbonylation formation of bicyclic B-lactams, 1036 Azirine, 2-arylcarbonylation formation of isocyanates, 1039 Azulene, hydrosynthesis, 394 Bakkenolide synthesis, 939

Subject Index

Baldwin's rules polyene cyclization, 344 Benzaldehyde, allylsynthesis, 255 Benzaldehyde, 2-bromodimerization, 501 Benzaldehyde, 4-bromosynthesis carbonylation, 1021 Benzaldehyde, hydroxyelectropinacolization, 568 Benzamide, N,N-allylbenzyl-Wittig rearrangement, 979 Benzamide, 2-methoxyreduction, 51 Benzene, alkyltransalkylation Friedel-Crafts reaction, 327 Benzene, allylsynthesis vinyl carbanion alkylation, 242 Benzene, dibromomonoalkylation with primary alkyl Grignard reagents or benzylzinc halides, 457 Benzene, 1,4-dibromocarbonylation selective, 1026 Benzene, o-di-t-butylisomerization Friedel-Crafts reaction, 327 Benzene, dichlorodialkylation coupling reactions with primary alkyl Grignard reagents, 450 monoalkylation with primary alkyl Grignard reagents, 457 Benzene, m-diethynylpolymerization, 557 Benzene, 1,3-dimethylalkylation with 2,5-dichloro-2,5-dimethylhexane Friedel-Crafts reaction, 318 Benzene, ethylsynthesis Friedel-Crafts reaction, 304 Benzene, hexaethylsynthesis Friedel-Crafts reaction, 301 Benzene, 2-iodonitro-Ullmann reaction, 499 Benzene, isopropylsynthesis Friedel-Crafts reaction, 304 Benzene, o-mercaptonitroin peptide synthesis, 302 Benzene, 1,2-methylenedioxyoxidative trimerization, 669 Benzene, n-pentylsynthesis, 415 Benzene, polyalkyltransalkylation Friedel-Crafts reaction, 327 Benzene, n-propylsynthesis, 415 Benzene, 1,3,5-tribromo monoalkylation

with primary alkyl Grignard reagents or benzylzinc halides, 457 Benzene, trichlorodialkylation coupling reactions with primary alkyl Grignard reagents, 450 monoalkylation with primary alkyl Grignard reagents, 457 Benzene, 1,2,4-trimethylreaction with isoprene Friedel-Crafts reaction, 322 Benzenesulfonic acid, trimethyl-2,4,6-trinitroalkylation by, 16 Benzenium ion, t-butylstability, 301 Benzil rearrangements, 821-836 nonhydroxylic solvents, 824 reaction conditions, 825 Benzil, decafluororearrangement, 825 Benzilic acid esters rearrangements, 823 pharmacological activity, 826 rearrangements, 821-836 chemistry, 825 labeling studies, 822 mechanism, 822, 824 migratory aptitudes, 822 reaction conditions, 825 synthesis reaction conditions, 825 Benzimidazole, 2-methylthioreaction with Grignard reagents, 461 Benzimidazolin-2-one reaction with crotonic acid, 306 Benzoates reduction to 1,4-dihydrobenzoates, 613 Benzoazepinone synthesis palladium-catalyzed carbonylation, 1038 Benzobicyclo[3.2.1]octanone, bromo-Favorskii rearrangement, 853 Benzocyclobutanes synthesis, 161 Parham-type cyclization, 251 Benzocyclobutanol synthesis, 265 Benzo[j]fluoranthene, 8-methylsynthesis Friedel-Crafts cycloalkylation, 325 Benzofuran coupling reactions with alkyl Grignard reagents, 444 synthesis, 494 Benzofuran, 2-bromoreaction with arylzinc bromide, 514 Benzofuran, dihydrosynthesis, 265 Benzofuran, 2-trimethylstannylsynthesis, 514 Benzofuran-3-ols, 3-methyl-2,3-dihydrosynthesis Friedel-Crafts reaction, 312

2H-Benzo[b]furan-3-one, 2-benzylring scission, 830 3H-Benzo[b]furan-2-one, 3-benzyl-3-hydroxysynthesis, 831 Benzoic acid, aryl-Birch reduction enolate generation, 51 Benzoic acid, 2,4,6-triisopropylalkyl esters metallation, 194 Benzophenone synthesis carbonylation of phenyllithium, 1017 Benzophenone, 2,3,4'-trihydroxyoxidative coupling mechanism, 661 Benzoquinone, 2,3-dichloro-2,6-dicyanophenolic coupling, 661 1,4-Benzoquinone, 2,6-dihydroxybenzilic rearrangement, 829 1,4-Benzoquinone, hydroxyrearrangements, 828 Benzothiazole, 2-allyloxyreaction with Grignard reagents, 246 Benzothiazole, 2-chlorocoupling reactions with Grignard reagents, 461 Benzothiazole, 2-hydroxyreaction with copper alkynides, 283 Benzothiazole, 2-methylthiocoupling reactions with Grignard reagents, 461 Benzothiophene coupling reactions with primary alkyl Grignard reagents, 447 methylation, 456 Benzotriazoline N-substituted reaction with lithium alkynides, 282 Benzotropolone rearrangements, 818 Benzoxazepinones reduction enolate generation, 51 Benzpinacolone label redistribution pinacol rearrangement, mechanism of, 723 Benzylamine, *a*-alkylstereoselective synthesis, 76 Benzylation sulfur- and selenium-stabilized carbanions, 88 Benzyl halides carbonylation formation of esters, 1028 palladium catalysts, 1021 Benzylic electrophiles reaction with organocopper compounds, 220 Berbin-8-one synthesis carbonylation, 1038 Bergamotene synthesis, 108, 249 Betulaprenols synthesis, 170 Betulinic acid ring A contraction, 834

Betweenanenes synthesis, 591, 946 intramolecular acyloin coupling reaction, 627 Beverene rearrangement, 715 Biaryls formation in phenol ether couplings, 668 synthesis, 663 aryl radical insertion, 677 **Bicyclic systems** Wagner-Meerwein rearrangment, 706 Bicyclization polyalkenes mechanism, 374 Bicyclo[1.1.0]butane synthesis Wurtz reaction, 414, 422 Bicyclo[4.4.0]decadiene synthesis, 390 Bicyclo[5.3.0]deca-2,10-diene synthesis, 399 Bicyclo[4.4.0]decane synthesis, 389, 391 Bicyclo[4.3.1]decan-10-one synthesis, 58 Bicyclo[4.4.0]decene synthesis, 393 polyene cyclization, 345 Bicyclo[4.4.0]decenol synthesis, 392 Bicyclogermacrene transannular cyclization, 390 Bicyclo[3.1.1]heptane synthesis, 901 Bicyclo[2.2.1]heptane-7-carboxylic acid synthesis, 903 Bicyclo[2.2.1]heptanol carbocations rearrangement, 707 Bicyclo[2.2.1]heptanone lithium enolate exoalkylation, 17 synthesis, 19 Bicyclo[2.2.1]heptanone, a-diazo-Wolff rearrangement, 900 trans-Bicyclo[4.1.0]hept-3-ene synthesis Ramberg-Bäcklund rearrangement, 876 Bicyclo[8.8.8]hexacosane synthesis intramolecular acyloin coupling reaction, 628 Bicyclo[2.1.1]hexane synthesis, 900 Bicyclo[2.2.0]hexane synthesis, 901 Bicyclo[2.1.1]hexane-2-carboxylic acid synthesis, 903 Bicyclo[2.1.1]hexane-6-carboxylic acid, exo-1,5,5-trimethylsynthesis, 900 Bicyclo[2.1.1]hexenes synthesis Ramberg-Bäcklund rearrangement, 874 Bicyclohumulenediol synthesis, 404 Bicyclo[3.3.1]nonan-3-one, 2-bromo-

Favorskii rearrangement, 853 Bicyclo[6.1.0]nonan-5-one, 4-diazo-transirradiation, 905 Bicyclo[3.3.1]non-2-ene epoxide transannular hydride shifts, 735 Bicyclo[3.3.1]non-2-en-4-one, 5-(2-ethylallyl)-1-methylsynthesis, 23 Bicyclo[3.3.0]octadiene synthesis, 489 Bicyclo[3.3.0]octa-2,6-diol synthesis, 382 Bicyclo[3.2.1]octane ring formation, 380 Bicyclo[3.3.0]octane ring formation, 380 Bicyclo[4.2.0]octane rearrangement, 714 synthesis, 382 Bicyclo[3.3.0]octanecarbaldehyde synthesis, 383 Bicyclo[3.3.0]octanol synthesis, 384 Bicyclo[2.2.2]octan-2-one synthesis, 19 Bicyclo[3.3.0]octanone synthesis, 139 C-H insertion reactions, 1060 Bicyclo[3.2.1]octan-3-ones, 2-bromo-Favorskii rearrangement, 852 Bicyclo[3.3.0]octene synthesis, 380 Bicyclo[2.2.2]octenol methanesulfonates rearrangement, 717 Bicyclo[2.1.0]pentane synthesis, 901 Bicyclo[2.1.0]pentan-2-one vinylogous Wolff rearrangement, 906 Bicyclo[4.3.0] rings polyene cyclization, 359 Bicyclo[4.4.0] rings polyene bicyclization, 360 Bicyclo[6.3.0]undecane synthesis, 406 Bicyclo[4.4.1]undecan-7-one synthesis, 58 Binaphthyls synthesis, 499, 503 nickel catalysts, 229 use of vanadium oxytrichloride, 664 Binaphthyls, tetrahydroxysynthesis use of potassium ferricyanide, 664 **Biomimetic synthesis** Wagner-Meerwein rearrangement, 714 Biotin synthesis, 151 Biphenol synthesis, 664 Biphenyl alkylation Friedel-Crafts reaction, 304 fluorination synthesis, 499

polyoxygenated synthesis, 503 synthesis Negishi method, 503 Biphenyl, 2,2'-dihydroxyoxidative coupling, 666 Biphenyl, 4,4'-dihydroxysynthesis use of vanadium tetrachloride, 664 Biphenyl, 4,4'-dimethoxysynthesis, 669 Biphenyl, 2,6-dinitrosynthesis, 501 Biphenyl, 3,3',4,4'-tetramethoxysynthesis, 668 2.2'-Bipyridy] reaction with phenyllithium, 512 synthesis, 509 y-Bisabolene synthesis, 215 Bisbenzocycloheptatriene synthesis in steganone synthesis, 673 **Bislactim** ethers alkylation, 53 Bismuth carbonate, triphenylbiaryl synthesis, 505 **A-Bisnorsteroids** synthesis, 901 Bisquinonemethides synthesis, 698 **Bisthiazoles** synthesis, 511 Bisthiophenes coupling reactions, 512 synthesis, 515 Block copolymers styrene-ethylene-butene-styrene Friedel-Crafts alkylation, 303 Boldine synthesis, 686, 815 Bombykol synthesis, 489, 799 9-Borabicyclo[3.3.1]nonane, B-methyldeprotonation alkylation of anion, 199 Borane, alkenyl-1,3-butadiene synthesis, 483 coupling reactions, 489 sp^2 organometallics, 473 with benzyl bromide, 465 cross-coupling reactions with 1-alkynyl halides, 530 Borane, alkynylcoupling reactions, 523 Borane, catecholcoupling reactions with aryl iodides, 496 Borane, dimesityldeprotonation, 199 Borane, monoisopinocamphenylalkene hydroboration, 797 Borane, phenylalkylation, 260 Borane, pyridylcoupling reactions with vinyl bromides, 498

Borane, trialkylreaction with aryl Grignard reagents, 243 reaction with carbon monoxide, 793 Borane, triorganylcarbonylation route to tertiary alcohols, 779 rearrangements, 793 Borane, vinylboron-lithium exchange, 254 Borates, alkynylcoupling reactions, 554 synthesis, 799 Borates, tetraorganylrearrangement, 780 Borepane, 3,6-dimethylhydroborating agent, 199 Boron alkynides alkylation, 274 Boronates, alkenylcoupling reactions, 489 with alkenyl halides, 496 reactions organopalladium catalysts, 231 Boronates, 1-bromo-1-alkenylfrom 1-alkynes, 490 Boronates, cyanorearrangements, 798 Boronates, *β*-ethoxycoupling reactions with aryl iodides, 496 Boronates, tetraorganylrearrangements, 798 Boronic acid, alkenylbiaryl synthesis, 504 synthesis, 489 Boronic acid, 1-chloroethylester synthesis, 796 **Boronic esters** chiral asymmetric synthesis, 780 synthesis, 796 rearrangements, 780, 796 Boron reagents organopalladium catalysis, 231 Boron trifluoride catalyst Friedel-Crafts reaction, 295 epoxide ring opening, 741 etherate organocopper compound reactions, 212 Botryodiplodin synthesis rearrangement of epoxides, 768 Bouveault-Blanc reduction esters conversion to primary alcohols, 613 Brassinolide synthesis Wittig rearrangement, 1000 Brefeldin synthesis, 287 Brevianamide A synthesis, 790 Brevicomin synthesis, 644

1142

Subject Index

Brönsted acids catalysts Friedel--Crafts reaction, 297 Brönsted-Lewis superacid catalysts Friedel-Crafts reaction, 297 Bromohydrin coupling reactions with aryl Grignard reagents, 464 reaction with magnesium halides, 757 Brown-Walker electrolysis of halfester dimerization, 640 Brunke steroid synthesis diene cyclization, 373 1,3-Butadiene, 1-bromosynthesis vinylic coupling, 490 Butadienes (Z,Z)synthesis, 485 1,3-Butadienes symmetrical synthesis, 482 Butadienes, chlorosynthesis vinylic coupling, 487 Butadienes, silylsynthesis, 487 vinylic coupling, 485 1,3-Butadiyne alkylation, 284 1,3-Butadiyne, 1-alkyl-4-(N,N-dialkylamino)synthesis, 284 1,3-Butadiyne, bis(trimethylsilyl)alkylation, 284 1,3-Butadiyne, 1-(N,N-dialkylamino)lithium derivative synthesis, 284 **Butadiynes** synthesis, 551 Butadiynes, 1,4-dialkynylsynthesis, 554 Butadiynes, 1,4-diarylsynthesis, 554 1-Butanal synthesis hydroformylation of propene, 1015 Butanamide, diethylalkylation, 68 n-Butane isomerization Friedel-Crafts reaction, 334 Butane, 1-chloro-3-methyl-3-phenylsynthesis Friedel-Crafts reaction, 320 Butane, 1,3-dichloro-3-methylbenzene alkylation by Friedel-Crafts reaction, 320 Butane, 1,2-epoxybenzene alkylation with Friedel-Crafts reaction, 313 Butane, 2,3-epoxyreaction with magnesium halides epoxide ring opening, 755 rearrangement boron trifluoride catalyzed, 742

2.3-Butanediol boronic esters, 797 pinacol rearrangement, 725 Butanoic acid, 3-benzoylaminodilithium dianions alkylation, 43 Butanoic acid, 3-hydroxyethyl esters alkylation, 44 Butanoic acid, 3-methylethyl ester reduction, 617 2-Butene synthesis Ramberg-Bäcklund rearrangement, 861 2-Butene, 1-bromoalkylation, 253 2-Butene, 1-iodo-3-trimethylsilylalkylation by, 11 3-Butenoic acids synthesis carbonylation of allylic chlorides, 1027 Butenolides synthesis, 905 Butenolides, hydroxysynthesis multicomponent carbonylation, 1020 3-Butenyl bromide coupling reactions with phenyl Grignard reagents, 464 1-Butyne, 3-methoxy-3-methylorganocopper compounds, 212 3-Butyn-2-ol, 2-methylin terminal alkyne synthesis, 531 Butyric acid, 2,3-dioxot-butyl ester rearrangement, 822, 831 Butyric acid, a-haloaryl esters cycloalkylation, 324 Butyrolactone, 5-ethenylsynthesis, 245 γ-Butyrolactone, β-hydroxyalkylation, 41 Butyrolactone, a-methylenesynthesis carbonylation of homoallylic alcohols, 1031 γ -Butyrolactone 2-acetic acid esters synthesis carbonylation, 1040 y-Butyrolactones polysubstituted synthesis, 843 synthesis enantioselectivity, 956 Cadiot-Chodkiewicz coupling, 553 alkynes organocopper compounds, 219 Cadmium, arylalkylation, 260 Caged compounds transannular reactions, 382 Cage-like structures synthesis, 854 Calcimycin

synthesis, 126, 139 Calichemicins synthesis copper catalysts, 217 Ramberg-Bäcklund rearrangement, 883 Camphene rearrangement, 705 Camphenilol rearrangement, 706 Campherenone synthesis, 427 Camphor rearrangement, 710 Camphor, 3-endo-bromorearrangement, 711 Camphor, diazo-Wolff rearrangement, 900 Camphor, 3,3-dibromo-Wagner-Meerwein rearrangement, 712 Camphor-9-sulfonic acid synthesis, 710 Camphor-10-sulfonic acid synthesis from camphor, 710 Camphor-9-sulfonic acid, 3-endo-bromosynthesis, 711 Cannabinoids synthesis, 127 Cannithrene II synthesis, 591 Cannizzaro reaction electron transfer mechanism, 824 Capnellane synthesis, 389 Capnellene synthesis, 384, 404 via carbonyl-alkyne cyclization, 602 $\Delta^{9(12)}$ -Capnellene synthesis, 20, 288 $\Delta^{9(12)}$ -Capnellene-8 β , 10 α -diol synthesis, 603 Caprolactam, 2-chlororearrangements, 849 Carbamates α' -lithioalkyl alkylation, 88 Carbamates, allyl-N-phenylreaction with lithium cuprates, 222 Carbamazepin epoxide ring opening, 737 Carbamic acid, allythioalkylation, 103 sigmatropic rearrangement, 103 Carbamic acid, N.N-dimethyldithiomethylthiomethyl ester alkylation, 136 Carbamic acid, dithioa-alkylated allylic rearrangement, 117 allyl ester reduction, 108 Carbamic acid, y-methylthioallylalkylation, 103 Carbanions α -alkoxy

from protected cyanohydrins, 197 silicon-stabilized, alkylation, 198 alkynyl alkylation, 271-292 allylic heteroatom-stabilized alkylation, 196 antimony-stabilized alkylation, 203 arsenic-stabilized alkylation, 203 benzylic α -alkoxy alkylation, 196 bismuth-stabilized alkylation, 203 boron-stabilized alkylation, 199 α-epoxyphosphorus-stabilized, alkylation, 199 germanium-stabilized alkylation, 203 halogen-stabilized alkylation, 202 heteroaromatic alkylation, 260 heteroatom-stabilized alkylation, 192 lead-stabilized alkylation, 203 nitrogen-stabilized alkylations, 65-82 nonstabilized alkylations, 207-233 oxygen-stabilized alkylation, 193 phosphorus-stabilized alkylation, 200 selenium-containing alkylation, 85-181 silicon-stabilized alkylation, 200 sulfur-containing alkylation, 85-181 tin-stabilized alkylation, 203 Carbenium ions non-Kolbe electrolysis, 649 Carbenoids alkylation, 202 insertion, 1047 Carbinol, bis(2-pyridyl)synthesis, 826 Carboalkoxylation halides aryl and vinyl, 1028 Carboalumination vinylalanes, 266 Carbohydrates a-ketol rearrangement, 831 β-Carbolines synthesis, 72 Carbon-carbon bonds formation C-H insertion, 1045-1062 Carbon monoxide reactions with alcohols and amines mechanism, 1016

Carbonochloridothioates ketone synthesis from Grignard reagents, 463 Carbonylation, 1015-1041 additive mechanism, 1019 catalysts, 1016 direct mechanism, 1018 double, 1039 mechanisms, 1016 multicomponent mechanism, 1020 substitutive mechanism, 1018 Carbonyl compounds aliphatic intermolecular pinacol coupling reactions, 570 α -alkylated asymmetric synthesis, enantioselectivity, 30 diastereoselective synthesis, 34 aromatic intermolecular pinacol coupling reactions, 564 deprotonation enolization, 1 a-diazosynthesis, 889 nitrogen derivatives alkylation, 28 alkylation, regiochemistry, 28 alkylation, stereochemistry, 28 pinacol coupling reactions with alkenes, 598 with alkynes, 602 with allenes, 605 reductive coupling reactions, 563 with alkenes, 583 synthesis from alkyl vinyl sulfides and selenides, 120 α,β -unsaturated pinacol coupling reactions, 577 synthesis, 161 Carboxamides, α -allyloxy-Wittig rearrangement, 1004 Carboxylic acid esters, a-allyloxy-Wittig rearrangement, 1000 Carboxylic acids acyclic diastereoselective alkylation, 44 acyclic enolates diastereoselective alkylation, 42 coupling reactions with alkyl Grignard reagents, 463 α, α -dialkyl asymmetric synthesis, 53 endocyclic enolates diastereoselective alkylation, 39 enolates cycloalkylation, 48 intramolecular cyclization, 49 exocyclic enolates diastereoselective alkylation, 39 Kolbe electrolysis cross-coupling, 642 symmetrical coupling, 637 masked anions

alkylation, 53 metal enolates alkylation, 39 synthesis carbonylation, 1026 unsaturated synthesis, 862 α,β -unsaturated reaction with allylic halides, 50 cis-a, B-unsaturated carbonylation of alkynes, 1027 α,β -unsaturated dienolates alkylation, 50 Carboxylic acids, a-allyloxy-Wittig rearrangement, 999 Carene epoxides ring opening, 736 β-Carotene synthesis, 169, 585 Ramberg-Bäcklund rearrangement, 883 Carpanone synthesis, 698 2.3-Carvene oxiranes rearrangement, 771 Carvenolide synthesis, 849 Carvomenthene oxiranes rearrangement, 771 Carvomenthone rearrangement, 832 Caryolan-1-ol synthesis, 386 Caryophyllane synthesis, 389 Caryophyllene synthesis, 386, 399 α -Caryophyllene alcohol synthesis, 400, 713 Casbene synthesis, 431 Castro reaction copper(I) alkynides reaction with aryl halides, 522 Catechol oxidative trimerization, 669 Catenane synthesis intramolecular acyloin coupling reaction, 628 Cation-exchange resins acidic catalyst, Friedel-Crafts reaction, 296 Cation-forming agents metathetic, catalysts Friedel-Crafts reaction, 298 CBT synthesis, 1012 CC-1065 synthesis Sommelet-Hauser rearrangement, 969 Cectopia juvenile hormone synthesis, 99, 107 Cembramolides synthesis, 99

Cembrene synthesis, 431 Cephalosporin rearrangements, 954 synthesis organopalladium catalysts, 232 Cerium use in cycloalkanone coupling reactions, 570 use in pinacol coupling reactions, 567 Chalcogenides synthetic catalysts, Friedel-Crafts reaction, 296 Chirality self-reproduction alkylation of enolates, 40 Chlorohydrins synthesis, 224 Chloromethylation arenes Friedel-Crafts reaction, 321 Cholestane, 2,3-epoxyreaction with Grignard reagents, 757 Cholestane, 24-hydroxysynthesis, 168 5a-Cholestane-2,3-dione rearrangement, 834 5α -Cholestane-3,4-dione rearrangement, 834 5α-Cholestane-2,3-dione, 4,4-dimethylrearrangement, 834 5a-Cholest-1-en-3-one, 2-hydroxyrearrangements, 832 5a-Cholest-3-en-2-one, 3-hydroxyrearrangements, 832 Cholesterol, (20S)-hydroxysynthesis, 127 Cholesterol, 24-hydroxysynthesis, 161 Cholesterol, (25R)-26-hydroxysynthesis, 983 Chromium, chloroarene tricarbonylcoupling reactions with tetrabutyltin, 454 Chromium, pentacarbonyl(methoxymethyl)reaction with amines β-lactam synthesis, 1037 Chromium trichloride catalyst Wurtz reaction, 421 Chromium(II) salts use in intermolecular pinacol coupling reactions, 565 Cinnamic acid, hydroxyoxidative dimerization, 692 Citronnellal, hydroxysynthesis rhodium-catalyzed hydroformylation, 1022 Clay stabilized pillared catalysts Friedel-Crafts reaction, 296 Cleomiscosin A synthesis silver oxide oxidation, 691 Cleomiscosin B synthesis silver oxide oxidation, 691 Clovene

synthesis, 23, 386 C-H insertion reactions, 1058 Coal hydrocracking, 328 Cobalt, octacarbonyldicatalyst carbonylation of aryl and vinyl halides, 1026 Cobalt(I), chlorotris(triphenylphosphine)catalyst Wurtz reaction, 421 Codamine, N-trifluoroacetyloxidative coupling, 670 Colchicine synthesis, 807 electrooxidation, 683 Compactin synthesis, 589 Compactin, dihydrosynthesis, 161 Copacamphene synthesis, 20, 712 Copaene synthesis, 20 Copper activated powder, catalyst alkene dimerization, 482 polymer complexes oxidative coupling catalyst, 559 Ullmann reaction, 499 Copper, alkylthioallylallylation, 99 Copper, iodofluoroacetates synthesis, 421 Copper, methylstructure, 210 synthesis, 208 Copper, phenylstability, 210 structure, 210 synthesis, 208 Copper, vinylreaction with alkynyl halides, 219 reaction with vinyl halides, 217 Copper borates reaction with allylic halides, 221 Copper compounds zero-valent organocopper compounds from, 209 Copper(I), alkenylcoupling reactions with alkenyliodinium tosylates, 522 substitution at C=C, 522 synthesis, 522 Copper(I), (diethoxyphosphoryl)methylalkylation, 201 Copper(I), tetrakis[iodo(tri-n-butylphosphine)coupling, 418 Copper(I) acetylides Cadiot-Chodkiewicz coupling, 219 hydrolysis, 210 reaction with propargylic electrophiles, 223 synthesis, 208, 209 Copper(I) alkynides reaction with aryl halides, 522 Copper(I) bromide purification, 209

1146

Subject Index

Copper(I) compounds catalysts Grignard couplings, 419 Copper(I) cyanide purification, 209 Copper(I) iodide in alkylation α -thioalkyllithium, 88 purification, 209 vinyl Grignard reagent alkylation catalyst, 243 Copper(I) salts lithium enolate polyalkylation, 6 Copper(I) triflate catalyst alkene dimerization, 482 Ullmann reaction, 499 Corannulene synthesis intramolecular acyloin coupling reaction, 625 Coriolin synthesis, 603, 785 organocopper compounds, 221 Corticosteroids synthesis, 126 Corydine synthesis, 807 Costunolide synthesis Wittig rearrangement, 1010 Coumarin, 4-hydroxysynthesis sulfur-assisted carbonylation, 1034 Coupling reactions acyloins, 613--631 alkenes, 482 sp carbon centers, 551-559 sp^2 and sp carbon centers, 521–549 sp^2 carbon centers, 481–515 sp^3 and sp^2 carbon centers, 435–476 sp³ carbon centers, 413–432 sp³ organometallics and alkenyl halides, 436 p-Cresol oxidative coupling, 665 Cresols cycloalkylation Friedel-Crafts reaction, 304 disproportionation Friedel-Crafts reaction, 328 Crotonates synthesis, 263 Crotonic acid, 3-amino-2,4-dicyanoreaction with benzil, 824 Crown ethers in sulfide metallation, 86 synthesis, 591 Cryptopleurine synthesis, 507 use of thallium trifluoroacetate, 670 Cubane synthesis, 848, 854 Cumene synthesis Friedel-Crafts reaction, 294 Cumulenes synthesis

3,2-sigmatropic rearrangement, 963 Cuparene synthesis, 588 epoxide ring opening, 744 Cuparenone synthesis, 785, 786 C-H insertion reactions, 1060 Cuprates higher order, 213 synthesis, 209 immobilization solid supports, 211 reaction with tosylates, 248 Cuprates, benzylsynthesis, 209 Cuprates, (a-carbalkoxyvinyl)reaction with activated halides, 217 Cuprates, cyano(2-thienyl)alkylation, 251 stability, 213 Cuprates, dialkylcyanocarbonylation with α , β -unsaturated ketones 1,4-diketone synthesis, 1024 Cuprates, di-t-butylphosphidostability, 211 Cuprates, dicyclohexylamidostability, 211 Cuprates, dicyclohexylphosphidostability, 211 Cuprates, diphenylphosphidostability, 211 Cuprates, divinylalkylation, 259 Cuprates, phenylthioreaction with acyl halides, 226 Cuprates, a-selenoalkylallylation, 91 Cuprates, vinylsynthesis, 209 Curcumene synthesis nickel catalysts, 229 β-Curcumin synthesis, 127 Curtius rearrangement acyl azides, 908 Curvulin synthesis palladium-catalyzed carbonylation, 1033 Cyanation hydrocarbons, 1046 Cyanodienes synthesis use of ylidic rearrangements, 963 β-Cyano esters metal enolates alkylation, 54 Cyanohydrins intramolecular alkylation, 48 protected a-alkoxy carbanions from, 197 O-trimethylsilyl alkylation, 197 unsaturated anions intramolecular reactions, 51 Cyanohydrins, O-allyl-

Wittig rearrangement, 998 Cyanohydrins, ketone β,γ-unsaturated synthesis, 998 Cyclic compounds synthesis allylic halides, 429 Wurtz reaction, 422 Cyclizations polyenes, 341-375 initiation, 342 mechanism, 374 propagation, 343 termination, 345 tandem semipinacol rearrangements, 792 transannular electrophilic, 379-407 Cycloaddition reactions phenol ethers, 696 Wolff rearrangement, 905 Cycloalkadienes monoepoxides reaction with lithium homocuprates, 226 Cycloalkanecarboxylic acid, 1,2-dialkylsynthesis intramolecular alkylation, 49 Cycloalkanes synthesis Wurtz reaction, 422 Cycloalkanones pinacol coupling reactions cerium-induced, 570 spiroannelation, 88 Cycloalkanones, 2-alkyl-2-phenylsynthesis, 36 Cycloalkanones, dibromomonochlororearrangement, 851 Cycloalkanones, polyhalolarger ring rearrangements, 850 rearrangements, 849 Cycloalkenes synthesis intramolecular McMurry reaction, 588 toluene alkylation with Friedel-Crafts reaction, 304 transannular reactions, 379 Cycloalkenes, 1-alkylsynthesis, 247 Cycloalkylation arenes Friedel-Crafts reaction, 309 carboxylic acid derivatives enolates, 48 Friedel-Crafts reaction, 323 saturated ketones, 18 Cycloalkylcarboxylic acids synthesis, 845 Cycloalkylidene epoxides, a-methylenemacrocyclic reaction with organocopper compounds, 226 Cycloartenol biosynthesis, 1048 Cyclobutane, 1,2-dimethylenesynthesis, 873 Cyclobutanediols pinacol rearrangement, 727

Cyclobutane-1,2-diones ring contraction, 832 Cyclobutanes synthesis intramolecular acyloin coupling reaction, 620 Wurtz reaction, 422 Cyclobutanol synthesis photochemically mediated, 1048 Cyclobutanone, 2-bromo-Favorskii rearrangements mechanism, 848 Cyclobutanone, 2-diazo-3,4-bis(diphenylmethylene)irradiation Wolff rearrangement, 900 Cyclobutanone, 2-hydroxysynthesis intramolecular acyloin coupling reaction, 620 Cyclobutanone, 2-methylsynthesis, 175 Cyclobutanones polyalkylation side reaction to monoalkylation, 4 synthesis, 785 rearrangement of epoxides, 765 rearrangement of oxaspiropentanes, 761 Cyclobutenes synthesis, 163 intramolecular McMurry reaction, 588 Ramberg–Bäcklund rearrangement, 862, 871 Cyclobutenes, alkylidenesynthesis, 116 Cyclobutenes, 1,2-dialkylsynthesis via Ramberg-Bäcklund reaction, 873 Cyclocuprate Wurtz coupling, 423 Cyclodecadienes monoepoxides transannular cyclization, 396 transannular reactions, 389 2.6-Cyclodecadienone synthesis from protected cyanohydrins, 198 Cyclodecenes transannular cyclization, 388 Cyclodecenols synthesis Wittig rearrangement, 1009 transannular cyclization, 393 Cyclodec-5-ynol transannular cyclization, 396 Cyclododecadienone, 2-chlororearrangements, 849 Cyclododecanone lithiated imines alkylation, 37 Cyclododecanone, 2,3-epoxyrearrangement epoxide ring opening, 753 Cyclododecanone, 2-methylsynthesis, 37 Cyclofenchene synthesis, 709 Cycloheptanes synthesis

Subject Index

intramolecular acyloin coupling reaction, 626 Cycloheptanone, 5-ethoxycarbonyl-2-methylsynthesis, 783 Cycloheptene oxide rearrangement, lithium perchlorate catalyzed, 761 2-Cycloheptenone alkyl-substituted synthesis, 202 Cyclohexadecanone synthesis from protected cyanohydrins, 198 1,3-Cyclohexadiene, 1-trimethylsiloxyt-butylation, 27 1,4-Cyclohexadienes synthesis, 653 2,4-Cyclohexadienone 1,2-diaryl shifts, 806 Cyclohexadienone, 4,4-dimethylrearrangements, 804 Cyclohexadienone, 4-ethyl-4-methylrearrangements, 804 2,5-Cyclohexadienone, 4-hydroxy-4-methylmethyl group shift, 804 rearrangement, 806 Cyclohexadienones alkyl shifts, 804 aromatization bond cleavage, 816 conjugation rearrangements, 803 oxygen migration, 812 photochemical aromatization, 815 reactions with nucleophiles rearrangements, 817 rearrangements, 803 synthesis phenol ether coupling, 683 Cyclohexane benzene alkylation with Friedel-Crafts reaction, 322 Cyclohexane, cyanointramolecular cyclization, 48 Cyclohexane, 1,2-dimethylenesynthesis Ramberg-Bäcklund rearrangement, 879 Cyclohexane, exo-methylenestereospecific rearrangement, 919 Cyclohexanecarboxylic acid methyl ester acyloin coupling reaction, 619 1,2-Cyclohexanedicarboxylic acid dimethyl ester acyloin coupling reaction, 623 intramolecular acyloin coupling reaction, 621 synthesis intramolecular acyloin coupling reaction, 622 1,2-Cyclohexanediol, 1,2-dimethylpinacol rearrangement, 724, 761 1,2-Cyclohexanediols synthesis intramolecular pinacol coupling reactions, 575 1,2-Cyclohexanedione reaction with arylbiguanides, 832 rearrangement, 822 1,3-Cyclohexanedione, 2-chloro-

ring contraction, 871, 875 1,3-Cyclohexanedione, 2-methylalkylation, 55 Cyclohexanes synthesis intramolecular acyloin coupling reaction, 625 Cyclohexanol, 2-methylsynthesis epoxide ring opening, 753 Cyclohexanone enamines axial alkylation, 30 pyrrolidine enamine dialkylation, 29 methylation, 30 tri-n-butyltin enolates alkylation, 7 Cyclohexanone, 3-alkyl-1-enolates alkylation, 15 Cyclohexanone, 2-allyl-2-methylsynthesis regioselective alkylation, 28 Cyclohexanone, 1,3-bisdiazoirradiation, 905 Cyclohexanone, t-butyllithium enolate methylation, 15 Cyclohexanone, 4-t-butylchiral lithium enolate alkylation, 13 stabilized metal enolates metallation, 55 Cyclohexanone, 5-t-butyla-methyl substituents axial alkylation, 14 Cyclohexanone, 2-n-butyl-2-methylsynthesis alkylation of unsymmetrical enolate, 8 Cyclohexanone, 1,2-[14C2]-2-chlororeaction with sodium pentylate, 840 Cyclohexanone, 2,3-dialkyl-1-enolates alkylation, 15 synthesis, 8 Cyclohexanone, 3,5-dialkyllithium 1-enolate alkylation, 8 Cyclohexanone, 2-diazophotolysis, 903 Cyclohexanone, 2,4-di-t-butylsynthesis, 26 Cyclohexanone, 2,2-dimethylsynthesis alkylation of enolate, 2 Cyclohexanone, 2,6-dimethylsynthesis, 34 Cyclohexanone, 4-ethoxycarbonylrearrangement, 783 Cyclohexanone, 2-ethylsynthesis, 35 Cyclohexanone, 2-halo-Favorskii rearrangements, 848 Cyclohexanone, 2-methylenolates, 2 enol ethers

alkylation, 8 lithiated dimethylhydrazones crystal structure, 34 lithium enolate alkylation, 8 regioselective alkylation, 8 TMS enol derivative alkylation, 28 TMS enol ethers t-butylation, 25 Cyclohexanone, 2-methyl-6-allylsynthesis regioselective alkylation, 28 Cyclohexanone, 2-methyl-4-t-butylsynthesis, 32 Cyclohexanone, 2-methyl-6-butylsynthesis alkylation of unsymmetrical enolate, 8 Cyclohexanone, 3-methyl-5-t-butyllithium 1-enolate stereoselectivity of alkylation, 15 Cyclohexanone, 2-methyl-3-(4-tosyloxybutyl)exocycloalkylation, 20 Cyclohexanone, 2-phenylsynthesis, 257 Cyclohexanone, 3-(2-tosyloxyethyl)endocycloalkylation, 19 Cyclohexene hydroformation phosphite-modified rhodium catalysts, 1022 oxide anti opening, stereoelectronic aspects, 733 initiators, polyene cyclization, 356 reaction with Grignard reagents, ring opening, 754 rearrangement, 760 rearrangement, lithium halide catalyzed, 764 synthesis Ramberg-Bäcklund rearrangement, 876 Cyclohexene, 1-benzyloxide syn-opening, 741 Cyclohexene, 1,2-dimethyloxide rearrangement, lithium halide catalyzed, 764 rearrangement, lithium perchlorate catalyzed, 761 Cyclohexene, 1-methyloxide cyclization, 342 syn-opening, 741 reaction with magnesium bromide, 757 rearrangement, 753 rearrangement, lithium halide catalyzed, 763 rearrangement, lithium perchlorate catalyzed, 761 Cyclohexene, 1-phenyloxide opening, 734 Cyclohexene, 1-trimethylsiloxy-4-t-butylt-butylation diastereoselectivity, 26 2-Cyclohexenone, 3-alkoxylithium dienolates α' -alkylation, 21 2-Cyclohexenone, 4-alkylsynthesis, 21 2-Cyclohexenone, 3-aminoextended dienolates

y-alkylation, 24 2-Cyclohexenone, 4-(3-chloropropyl)-4-methyl-6-(2-ethylallyl)cycloalkylation, 23 Cyclohexenones α' -alkylation, 21 pinacol coupling reactions, 577 spiroannulation, 22 synthesis carbonylation, 1025 Cyclohexylidene epoxides, a-alkenylreaction with lithium homocuprates, 226 1,5-Cyclononadiene transannular cyclization, 386 Cyclooctadecane, 1,9-bis(3-methoxycarbonylpropyl)acyloin coupling reaction intramolecular, 628 1,5-Cyclooctadiene transannular cyclization, 381 1,5-Cyclooctadiene, 1,5-dimethyltransannular cyclization, 382 Cyclooctadienes monoepoxides transannular hydride shifts, 735 transannular reactions, 383 Cyclooctanecarbaldehyde, 5-methylenetransannular cyclization, 383 Cyclooctanes synthesis via cyclization of 1,8-dialdehydes, 575 Cyclooctanone synthesis, 781 Cyclooctanone, 2,8-dibromorearrangement, 850 Cyclooctanone, 2,2,8-tribromorearrangement, 850 Cyclooctatetraene monoepoxide rearrangement, 757 Cyclooctene oxide solvolysis, 735 transannular electrophilic cyclization, 380 Cyclooctene, 6-iodoreactions with lithium cuprates mechanism, 213 4-Cyclooctenone electrochemical transannulation, 600 2-Cyclopentadecenone synthesis, 51 Cyclopentadienes synthesis Ramberg-Bäcklund rearrangement, 874, 875 Cyclopentane, acetylsynthesis polyene cyclization, 347 Cyclopentane, ylidenesynthesis, 251 Cyclopentanecarbaldehyde synthesis, 769 Cyclopentanecarboxylic acid, 1-hydroxy-2-isopropyl-5-methylsynthesis, 831 Cyclopentanecarboxylic acids methyl ester synthesis, 903

polyfunctionalized synthesis, 848 1,2-Cyclopentanediols synthesis intramolecular pinacol coupling reactions, 574 1,3-Cyclopentanedione, 4-hydroxysynthesis, 829 1,3-Cyclopentanedione, 2-methylenolates alkylation, 55 1,3-Cyclopentanediones 2,2-disubstituted synthesis, pinacol rearrangement, 728 Cyclopentanes stereospecific synthesis, 653 synthesis, 647 intramolecular acyloin coupling reaction, 623 Cyclopentanoid monoterpenes synthesis, 850 Cyclopentanone, 3-alkenyl-1-enolate alkylation, 17 Cyclopentanone, 3-alkyl-1-enolate alkylation, 17 Cyclopentanone, 2-allylsynthesis alkylation of enolate, 6 Cyclopentanone, 2-benzylsynthesis alkylation of enolate, 6 Cyclopentanone, 3-(2-bromoethyl)endocycloalkylation, 19 Cyclopentanone, 2,3-dialkylsynthesis conjugate addition-enolate alkylation, 9 Cyclopentanone, 2-diazosynthesis, 900 Cyclopentanone, 2-halorearrangements, 848 Cyclopentanone, 2-methylsynthesis alkylation of enolate, 4 Cyclopentanone, 3-methyl-3-phenyl-1-enolate alkylation, 17 Cyclopentanone, 3-phenyldeprotonation, 17 1-enolate alkylation, 17 Cyclopentanone, 3-(2-tosyloxyethyl)endocycloalkylation, 19 Cyclopentanones polyalkylation side reaction to monoalkylation, 4 synthesis carbonylation, 1024 Cyclopentapyrazoles synthesis, 831 Cyclopentene, 3,4-epoxyreaction with ethylaluminum alkynide, 279 reaction with Grignard reagents, 265 Cyclopentene, 4-hydroxy-4-(1-hexynyl)synthesis, 279 Cyclopentene, 3-methylsynthesis

Ramberg-Bäcklund rearrangement, 874 3-Cyclopentene, 2-phenylsulfonylmethyl ester, acetate synthesis, 654 1-Cyclopentenecarboxylic acid synthesis, 905 3,5-Cyclopentenediols synthesis prostaglandin precursor, 155 Cyclopentenes carbonylation cobalt carbonyl catalyst, 1024 oxide rearrangement, lithium halide catalyzed, 764 synthesis Ramberg-Bäcklund rearrangement, 874 Cyclopentenone, 3-alkyllithium dienolates methylation, 22 Cyclopentenone, 3-aminoextended dienolates γ -alkylation, 24 Cyclopentenone, 4,4-dialkylsynthesis, 42 Cyclopentenone, 4,5-dihydroxyenolates alkylation, 11 Cyclopentenone, 4-hydroxysynthesis, 10 Cyclopentenones α' -alkylation, 21 synthesis, 936 from 2-chloro-1,3-cyclohexanediones, 871 Ramberg-Bäcklund rearrangement, 868, 874, 875 Cyclopentenones, a-methylenesynthesis carbonylation of 1-iodo-1,4-dienes, 1025 Cyclophanes synthesis, 557, 591, 594 1,2-rearrangement, 927 unsaturated synthesis, 877 *m*-Cyclophanes synthesis coupling reactions, 452 Cyclopropane, alkylidenesynthesis, 116 Cyclopropane, arylsynthesis, 120 Cyclopropane, 1,1-bis(methylthio)ketones from, 124 Cyclopropane, 1,1-bis(phenylseleno)synthesis, 136 Cyclopropane, 1-bromo-1-tributylstannylsynthesis by transmetallation, 196 Cyclopropane, dimethylsynthesis, 216 Cyclopropane, 1,1-dithiosynthesis, 124 Cyclopropane, 1-ethoxy-1-lithiosynthesis metallation, 194 Cyclopropane, phenylthioallylation, 88 Cyclopropane, siloxy-

coupling reactions with Grignard reagents, 460 with sp^3 organometallics, 455 Cyclopropanecarboxylic acids synthesis, 848 1,1-Cyclopropanedicarboxylic acid, 2-alkenylintermolecular alkylation, 56 Cyclopropanediols synthesis intramolecular pinacol coupling, 572 Cyclopropane-2,3-dioxopropionic acid ethyl ester rearrangement, 831 Cyclopropanes neighboring group epoxide ring opening, 736, 752, 753 ring expansion, 785 synthesis, 163 reduction of malonate, 620 Wurtz reaction, 422 Cyclopropanesulfomorpholine synthesis, 181 Cyclopropanesulfonic acid t-butyl esters synthesis, 180 neopentyl ester synthesis, 181 Cyclopropanone Favorskii rearrangement, 840 Cyclopropenes synthesis intramolecular McMurry reaction, 588 Cycloundecanone, 2-bromorearrangements, 849 Cycloundecene transannular cyclization, 398 Cytochalasin 3,2-sigmatropic rearrangement synthesis, stereocontrol, 960 Dactylol synthesis, 404 Dakin-West reaction N-acyl-a-amino ketones, 889 Damsin synthesis, 20 Daphneticin synthesis, 691 Darzens aldehyde synthesis epoxide ring opening, 738 Daucenone synthesis, 586 Dauricine synthesis, 687 Dealkylation Friedel-Crafts reaction, 327 Debenzhydrylation Friedel-Crafts reaction, 328 De-t-butylation Friedel-Crafts reaction, 329 Decaladienones aromatization, 810 Decalins cis synthesis, 360 trans

synthesis, 360 synthesis intramolecular cyclization of cyanocyclohexanes, 48 polyene bicyclization, 360 polyene cyclization, 350 1-Decalone alkylation, 11 lithium 1(9)-enolate angular alkylation, 16 synthesis exocycloalkylation, 20 TMS enol ether phenylthiomethylation, 26 2-Decalone lithium 2-enolate alkylation, 16, 21 lithium 1(2)-enolates methylation, 15 3-substituted enolates alkylation, diastereoselectivity, 55 1-Decalone, cis-9-methylsynthesis exocycloalkylation, 20 1-Decalone, 10-methyllithium 1(9)-enolate angular alkylation, 16 2-Decalone, 10-methyllithium enolate alkylation, 15 stabilized metal enolates metallation, 55 2-Decalone, trans-10-methyllithium 1-enolate alkylation, 2 Decalones synthesis regiospecific alkylation, 11 1,10-Decanedioic acid dimethyl ester intramolecular acyloin coupling reaction, 626 Decanesulfonic acid, perfluorocatalyst, solid superacid Friedel-Crafts reaction, 298, 305 Decarbonylation, 1015-1041 mechanism, 1020, 1040 1-Decene benzene alkylation with Friedel-Crafts reaction, 304 synthesis, 248 Dendrolasin synthesis, 99 from furan-3-carbaldehyde, 195 reduction of sulfides, 107 Denudatin A synthesis, 694 Deplancheine asymmetric synthesis, 81 Desmosterol synthesis, 427 Desmotroposantonin synthesis, 804 Dewar benzene 1,4-bridged synthesis, 872 Dextrorphan

1152

Subject Index

synthesis, 77 Dialkylation enolates equilibration, 4 Dialkyl phosphonates alkylation, 201 Diamantane synthesis Friedel-Crafts reaction, 334 1,2-Diamines coupling reactions with imines, 564 Diamondoid hydrocarbons synthesis Friedel-Crafts reaction, 334 Diastereofacial differentiation asymmetric synthesis, 72 Diazaalkenes insertion reactions, 1049 Diazabicyclooctane in sulfide metallation, 86 α -Diazo aldehydes synthesis, 890 Diazo esters photolysis, 894 Diazo insertion reactions rhodium-catalyzed, 1051 Diazo ketones secondary synthesis, 889 α,β -unsaturated synthesis, 890 β,γ -unsaturated vinylogous Wolff rearrangement, 906 UV spectra, 891 α-Diazo ketones rearrangements, 887 synthesis, 888 Diazonium salts, arylcoupling reactions with alkenes, 497 o-Diazo oxides photochemical decomposition, 904 Diazotype offset photocopying o-diazo oxides, 904 Dibenzocyclobutane side product in coupling reaction, 505 Dibenzothiophene methylation, 456 Dibromides geminal double alkyl substitution, 216 vicinal reaction with dialkyl cuprates, 216 1,2-Dicarbonyl compounds diazo-coupling reactions, 893 1,3-Dicarbonyl compounds alkylation, 54 cyclic 2-diazo-Wolff rearrangement, 903 dianions γ-alkylation, 58 dienolates γ -alkylation, l 1,2-Dicarbonyl compounds, 1-alkyl-2-arylbenzylic rearrangement, 829

1,3-Dicarbonyl compounds, 2-ethynylsynthesis, 286 Diconiferyl alcohol, dehydrosynthesis, 693 Dicyanomethylation arvl iodides with sodium salt of malononitrile, 454 Dienes arene alkylation Friedel-Crafts reaction, 322 conjugated synthesis, 878 monoepoxides rearrangement, 770 'skipped' synthesis, 244, 265 synthesis 3,2-sigmatropic rearrangement, 964 1,3-Dienes reaction with Kolbe radicals, 647 synthesis copper catalysts, 217 from alkenes, 879 organopalladium catalysis, 232 1,4-Dienes synthesis, 249 coupling reactions of allylic halides, 473 1,5-Dienes cvclic synthesis, 429 medium rings conformation, 386 synthesis, 428 organopalladium catalysis, 231 phosphonium ylide alkylation, 201 Dienes, acylsynthesis carbonylation, 1024 1,3-Dienes, 1-iodocarbonylation palladium catalysts, 1030 1.4-Dienes, 1-iodocarbonylation palladium catalyst, 1025 1,3-Dienes, 1-silylsynthesis organopalladium catalysis, 232, 233 Dienoic acids synthesis, 882 2.5-Dienoic acids synthesis nickel-catalyzed carbonylation, 1027 Dienolates copper(I)alkylation, 50 diastereoselective alkylation solvent effects, 24 extended alkylation, stereochemistry, 23 monoalkylation, 23 heteroannular extended equatorial alkylation, 24 α,β-unsaturated carboxylic acids alkylation, 50 1,5-Dien-3-ols synthesis

Wittig rearrangement, 994 Dienone-phenol rearrangements, 803-820 intracyclic migrations, 804 2,4-Dienones rearrangements, 803 Dienynes conjugated one-pot synthesis, 539 Diesters synthesis oxidative carbonylation of alkynes, 1030 vicinal reduction, 614 **B**-Diesters enolates reaction with allylic acetate, 56 metal enolates alkylation, 54 α,ω-Diethynyl monomers oxidative polymerization, 557 Diisoeugenol, dehydrosynthesis, 693 Diketene coupling reactions with sp³ organometallics, 446 Diketone, α -(4-isobutylphenyl)- β -methyl benzylic acid rearrangement, 829 Diketones decarbonylation, 1041 synthesis bis-dithiane dialkylation, 128 1,2-Diketones alicyclic benzilic acid rearrangement, 831 ring contraction rearrangement, 831 aliphatic benzilic acid rearrangement, 831 reactions with π -allylnickel halides, 424 synthesis carbonylation of lithium amides, 1017 1,3-Diketones y-alkylation, 58 cyclic enolates alkylation, 55 metal enolates alkylation, 54 1,2-Diketones, bisaryl rearrangements, 825 Dilactones synthesis palladium-catalyzed carbonylation, 1032 Dilantin anticonvulsant synthesis, 826 Dilithium alkylcyano(2-thienyl)cuprate alkylation, 261 Dilithium tetrachlorocuprate alkene dimerization, 482 alkylation Grignard reagents, 244 vinyl Grignard reagent alkylation catalyst, 243 Wurtz coupling, 415 1.2-Diols synthesis coupling reactions, 597

from β,γ -epoxy alcohols, 264 reductive coupling of carbonyl compounds, 563 1.3-Diols monoethers synthesis, 979 1,4-Diols synthesis alkenyltrialkylboronates, 799 1.3-Diones synthesis from 2,3-epoxyketones, 771 1,6-Diones synthesis coupling of α , β -unsaturated carbonyl compounds, 577 Diosphenols rearrangements, 832 Dioxalones enolates diastereoselective alkylation, 40 1.4-Dioxenes synthesis, 651 Dioxindole, 3-phenylsynthesis, 835 Dioxirane, dimethyloxidant reaction with quadricyclane, 736 1,3-Dioxolane,2-(2-methoxycarbonylethyl)-2-methylacyloin coupling reaction, 619 Dioxolenes iron complexes reaction with organocopper compounds, 218 Diphenoquinone synthesis, 664 Diplodialides synthesis, 286 Diploicin synthesis manganese dioxide oxidation, 688 Disiloxane, 1,3-bis(dimethylethynyl)oxidative coupling, 557 Disparlure synthesis, 224, 286, 644 Disproportionation Friedel-Crafts reaction, 327 1,1-Disulfones alkylation, 176 Diterpenoids tetracyclic synthesis, 715 1,3-Dithiane, 2-chlororeaction with aryl Grignard reagents, 242 1,3-Dithiane, 2-ethynylalkylation, 132 1,3-Dithiane, 1-lithioreaction with epoxides, 127 1,3-Dithiane, 2-lithioin synthesis, 126 reaction with oxiranes, 128 1,3-Dithiane, 2-metalloin synthesis, 124 1,3-Dithiane, 2-phenyl-Stevens rearrangement, 925 1.3-Dithianes acyl anion equivalent, 144 1-oxide

1154

Subject Index

alkylation, 137 Dithiatropazine synthesis acyloin coupling reaction, 618 Diynes conjugated synthesis, 525, 554 synthesis phase-transfer catalysts, 559 1,4-Diynes synthesis organocopper compounds, 223 1,3-Diynes, 1-trimethylsilylsynthesis, 553 Dodeca-3,5-diyn-1-ol synthesis, 273 Dodecahedrane, 1,16-dimethylsynthesis Friedel-Crafts reaction, 334 Dodecanone, dibromorearrangement, 851 Dodecylamine, dimethyla-deprotonation, 65 Dolastane synthesis, 488 Domesticine synthesis photochemical oxidation, 677 Douglas fir tussock moth sex pheromone synthesis, 161 Dowex 50 catalyst Friedel-Crafts reaction, 296 Drimnanes synthesis farnesol bicyclization, 342 Dubamine synthesis, 514 Dunnione rearrangement, 828 Ecdysone synthesis Wittig rearrangement, 1000 Egomaketone synthesis palladium-catalyzed carbonylation, 1023 Elaidic acid Kolbe electrolysis, 642 Eldanolide synthesis, 796 Electrochemical pinacolization aromatic compounds, 567 Elemol synthesis, 431 Elsholtzia ketone synthesis, 999 Enamines alkylation, 28 reversibility, 29 cyclohexanone axial alkylation, 30 hindered aldehyde C-alkylation, 30 Enamines, chloro-

Favorskii rearrangement, 857 Enantioselective synthesis alkylation, 35 Endiandric acid synthesis, 558 Enedivnes synthesis cobalt-catalyzed cyclizations, 255 Enolates acyclic, carboxylic acid derivatives diastereoselective alkylation, 42 aggregation geometry, 4 alkylation, 1-58 stereochemistry, 12 allylation palladium-catalyzed regioselective, 12 carboxylic acids cycloalkylation, 48 endocyclic stereochemical alkylation, 41 equilibration alkylation, 4 lithium α -alkylation, 3 stabilized intramolecular alkylation, 55 sterically congested stereoselectivity of alkylation, 15 2-substituted distortion, 14 transmetallation tri-n-butyltin chloride, 10 Enolates, α -allyloxy Wittig rearrangement, 996 Enol carboxylates coupling reactions with sp³ organometallics, 444 Enol ethers coupling reactions with sp³ organometallics, 444 rearrangements, 789 Enol phosphates coupling reactions with sp³ organometallics, 444 reaction with dialkylcuprates, 218 Enols alkylation, 1-58 Enol silyl ethers coupling reactions with aromatic bromides, 454 with primary alkyl Grignard reagents, 445 Enol sulfonates coupling reactions with sp³ organometallics, 444 Enol triflates coupling reactions with lithium diarylcuprates, 492 with lithium divinylcuprates, 487 with sp³ organometallics, 445 Enol trimethylsilyl ethers synthesis from α -trimethylsilyl epoxides reaction with Grignard reagents, 759 Enones conjugated synthesis, 844

 α,β -dialkylation conjugate addition-enolate alkylation, 8 vicinal dialkylation, 8 Enynes conjugated synthesis, 878 fluorinated synthesis, 525 in vitamin D synthesis, 545 skipped synthesis, 274 stereospecific synthesis, 539 synthesis palladium catalysis, 217 1,3-Enynes synthesis from alkynes, 880 stereoselective, 522 1,5-Enynes synthesis, 104, 107 organopalladium catalysis, 231 Envnes, 1-chlorostereospecific synthesis, 539 Enynes, dithienylsynthesis, 527 Ephedrine amides enolates diastereoselective alkylation, 45 6a-Epipretazettine synthesis, 683 Epi-B-santalene synthesis, 427 Epoxides (see also Oxiranes) a-acetoxy steroidal rearrangement, 739 alkylation, 262 alkynides, 277 alkylation with sulfur- and selenium-stabilized carbanions, 86 allylic rearrangement, 762 α -amino rearrangement, 740 arene alkylation Friedel-Crafts reaction, 309 α -carbonvl rearrangement, 738 a-chloro acid-catalyzed rearrangement, 739 thermal rearrangement, 739 α -electron withdrawing group rearrangement, 746 formation semipinacol rearrangement, 778 Friedel-Crafts reactions, 769 opening anti, 734 hydroxy neighboring group, 735 stereochemistry, 733 syn, 734 reactions with organocopper compounds, 223 reaction with α -selenoalkyllithium, 91 rearrangements acid-catalyzed, 733-771 protic acid catalyzed, 734 ring opening

boron trifluoride catalyzed, 741 magnesium halide catalysis, 754 mechanism in aqueous acid, 736 with Grignard reagents, 466 a-substituted rearrangement, 738 α -trimethylsilyl reaction with Grignard reagents, 759 β-trimethylsilyl synthesis, 759 Epoxides, acyclic vinyl reaction with organocopper compounds, 226 Epoxides, α -lithio from transmetallation, 198 Epoxides, vinyl reaction with organocuprates, 225 2,3-Epoxysqualene synthesis, 178 Eremophilane biosynthesis, 388 Eremophilone synthesis rearrangement of allylic epoxides, 762 Ergosterol 25,28-dihydroxylated synthesis, 983 Erucic acid Kolbe electrolysis, 642 Erybidine, O-methylsynthesis, 816 Erysodienone synthesis, 816 use of alkaline ferricyanide, 681 Erythrina alkaloids synthesis, 505 Erythromycin partial synthesis stereocontrol, 960 Erythronolide B synthesis, 288 Esperamicin synthesis, 545 copper catalysts, 217 Esperamicin A synthesis, 27 Ester enolates acyclic alkylation, 42 Esters acyloin coupling reaction heterogeneous conditions, 614 homogeneous conditions, 615 necessary reaction conditions, 614 preferred reaction conditions, 617 with ketones, 630 arene alkylation Friedel-Crafts reaction, 309 hindered aryl deprotonation, 194 β-hydroxy alkylation, 43 reduction alkali metals, 613 synthesis carbonylation, 1028 α,β -unsaturated

1156

Subject Index

synthesis, 865 Esters, a-diazo C-H insertion reactions, 1054 Esters, B-keto γ -alkylation, 58 synthesis, 783, 784 Esters, γ -stannyl- α , β -unsaturated coupling reactions with alkenyl halides, 443 Esters, α,β -unsaturated synthesis Ramberg-Bäcklund rearrangement, 870 Esters, β , γ -unsaturated synthesis coupling reactions, 443 Estrone cyclization, 371 methyl ether synthesis, 1061 synthesis polyene bicyclization, 360 polyene cyclization, 366 Ethane ethylation Friedel-Crafts reaction, 333 Ethane, 1,2-diaryldimerization, 673 Ethane, 1,2-dihaloarene alkylation Friedel-Crafts reaction, 318 Ethane, 1,2-dihalo-2-phenylarene alkylation Friedel-Crafts reaction, 318 Ethane, 1,2-diphenylsynthesis, 638 Ethane, hexafluorosynthesis, 640 Ethane, hexamethylsynthesis, 415 Ethane, 1,1,1-trifluoro-2,2-diarylsynthesis Friedel-Crafts reaction, 311 Ethanediol, 1,2-dicyclohexylboronic esters, 796 Ethanediol, diisopropylboronic esters, 797 Ethanol, 2-bromoacetate reaction with aryl Grignard reagents, 243 reaction with aryl Grignard reagents, 243 Ethanol, α-(2-hydroxyphenyl)lactate Friedel-Crafts reaction, 311 Ether, benzyl methyl deprotonation by *n*-butyllithium, 197 Ether, bis- γ , γ -(dimethyl)allyl Wittig rearrangement mechanism, 977 Ether, t-butyl methyl potassium salts synthesis, 194 Ether, chloromethyl methyl α -halometallation, 194 Ether, 2-cyclohexen-1-yl methyl synthesis, 651

Ether, dimethyl deprotonation with n-butyllithium, 194 potassium salts synthesis, 194 Ether, methyl vinyl reaction with t-butyllithium, 252 Ethers arene alkylation Friedel-Crafts reaction, 309 cyclic allylic Wittig rearrangement, 1008 cyclic propargylic Wittig rearrangement, 1008 Ethers, alkyl haloalkyl arene haloalkylation by Friedel-Crafts reaction, 321 Ethers, allyl benzyl Wittig rearrangement, 989 Ethers, allylic Wittig rearrangement mechanism, 977 Ethers, allyl lithiomethyl Wittig rearrangement, 982 Ethers, allyl propargyl carbonylation use of cobalt complexes catalysts, 1025 Wittig rearrangement, 984 Ethers, allyl silyl reaction with aryl Grignard reagents, 246 Ethers, aryl oxidation radical cation reactions, 662 synthesis, 686 Ethers, aryl alkyl synthesis C-O coupling, 690 Ethers, *a*-arylamino reaction with lithium alkynides, 282 Ethers, aryl tetrazolyl substitution reactions nickel catalysts, 229 Ethers, benzothiazolyl reaction with organocopper compounds, 222 Ethers, benzyl α -alkoxy carbanions from Wittig rearrangement, 197 Ethers, benzyl trans-crotyl Wittig rearrangement, 976 Ethers, a-chlorodialkyl Grignard reagents preparation, 194 Ethers, crotyl Wittig rearrangement, 1004 Ethers, diallyl Wittig rearrangement, 976, 991 Ethers, dichloromethyl methyl anion trialkylcarbinol synthesis, 794 Ethers, dipropargyl Wittig rearrangement, 991 Ethers, epoxy reaction with organocopper compounds, 225 Ethers, farnesyl synthesis, 429 Ethers, geranyl

synthesis, 429 Ethers, a-halo reaction with aryl Grignard reagents, 242 Ethers, *β*-halovinyl coupling reactions with aryl Grignard reagents, 492 Ethers, 1-isopropyl-2-butenyl benzyl Wittig rearrangement, 990 Ethers, ketoxime methyl deprotonation, 35 Ethers, propargylic Wittig rearrangement, 986 Ethers, (trialkylstannyl)methyl allylic lithiation, 982 Ethylamine, β-arylsynthesis Friedel-Crafts reaction, 316 Ethyl bromoacetate coupling reactions with arylzinc reagents, 466 Ethyl diazoacetate ketone homologation, 783 Ethylene, trans-1,2-bis(tri-n-butylstannyl)alkylation, 247 Ethylene, 1-bromo-2-phenylthiocoupling reaction with alkyl Grignard reagents, 449 with secondary alkyl Grignard reagents, 441 reaction with Grignard reagents palladium catalysts, 230 tandem couplings, 492 Ethylene, 1,1-dichlorocoupling reactions with alkyl Grignard reagents, 448 Ethylene, 1,2-dichlorocoupling reactions with alkyl Grignard reagents, 449 with vinylic Grignard reagents, 487 Ethylene, diphenylarsenoreaction with organolithium compounds formation of α -arseno anions, 203 Ethylene, polychlorocoupling reactions, 487 Ethylene, tetraphenyl-Wurtz reaction catalyst, 414 Ethylenediamine, N.N.N',N'-tetramethylin sulfide metallation, 86 Ethylene-1,1'-dicarbonitrile, 2-benzoyl-2-phenylsynthesis, 826 Ethyl halides arene alkylation Friedel-Crafts reaction, 300 Ethyl iodide ethylation with stereochemistry, 14 Eudesmane rearrangement, 388 synthesis, 396 β-Eudesmol synthesis, 20 Europium salts use in intermolecular pinacol coupling reactions, 565 Fusiderin synthesis use of silver oxide, 691

Farnesol bicyclization, 342 cyclization, 360 synthesis, 170 Farnesol, 10,11-epoxysynthesis, 99 Fatty acids synthesis, 643 Favorskii rearrangement, 839-857 in synthesis, 842 Lewis acids, 856 mechanism, 840 reaction conditions, 840 side-products, 840 stereospecificity, 848 Fenchenes synthesis from fenchyl alcohol, 709 Fenchyl alcohol rearrangement, 709 [4.4.4.5]Fenestrane synthesis, 901 Ferrocenophanes synthesis, 594 Ferruginol synthesis, 169 Ferulic acid oxidation, 693 Five-membered rings formation polyene cyclization, 347 Flavanone, 3-hydroxyring scission, 831 Flavinantine, methylsynthesis, 81 anodic oxidation, 685 electrooxidation, 685 Flexibilene synthesis, 591 Fluorene synthesis, 543 Fluorenecarboxylic acid, 9-hydroxysynthesis, 828 9-Fluorenone synthesis, 828 Fluorohydrin synthesis epoxide ring opening, 749 Fluorohydrocarbons synthesis, 640 Fluorosulfuric acid catalyst Friedel-Crafts reaction, 297 Formaldine, methylthioalkylation, 137 metallated alkylation, 135 Formaldines α -heterosubstituted carbonyl compound synthesis from, 141 Formamides synthesis, 420 Formamidines alkylation, 68 Formates, chlorothioketone synthesis from

Grignard reagents, 463 Formates, a-metalloorthoselenoester precursors, 144 Formates, a-metalloorthothioester precursors, 144 Fragranol synthesis, 103 Fredericamycin A synthesis organocopper compounds, 219 Friedel-Crafts reaction aliphatic, 294 alkylating agents, 294 alkylation, 293-335 arene alkylation kinetics, 300 asymmetric alkylation, 302 catalysts, 294, 295 cocatalysts, 295 epoxides, 769 Friedelin backbone rearrangement, 709 Frullanolide synthesis, 1031 Fumarates, iododimerization, 482 Furan, aminomethylsynthesis, 258 Furan, 3-bromomethylalkylation by cuprates, 250 Furan, dihvdrocoupling reactions with alkyl Grignard reagents, 444 reaction with Grignard reagents nickel catalysts, 229 reaction with organocopper compounds, 218 Furan, 2-lithioalkylation, 261 reaction with propylene oxide, 264 Furan, 3-lithioreaction with epoxides, 264 Furan, 2-methyltetrahydroalkylation Friedel-Crafts reaction, 317 benzene alkylation Friedel-Crafts reaction, 315 Furan, tetrahydroarene alkylation by Friedel-Crafts reaction, 315 deprotonation, 194 potassium salts synthesis, 194 3-substituted synthesis, 647 synthesis, 792 Furanacetic acid lactones, cis-3-hydroxytetrahydrosynthesis palladium-catalyzed oxycarbonylation, 1033 Furanone synthesis C-H insertion reactions, 1056 3(2H)-Furanone, 2,2-disubstituted-5-alkylextended dienolates y-alkylation, 24 Furans

coupling reactions with sp³ organometallics, 459 intermolecular dimerization, 509 2,2'-Furil rearrangement, 826 Furopyridines synthesis, 543 Furst-Plattner rule epoxides opening, 734 Fuscinic acid dimethyl ether oxidation, 831 Fusicoccin synthesis, 575 Futoene synthesis, 696 Galbulin synthesis, 696 Gallium, trimethyllithium alkynides reaction with oxiranes, 279 Gallium trichloride polystyrene-divinylbenzene copolymer beads, catalyst Friedel-Crafts reaction, 298 Gatterman reaction thiocarbonyl compounds, 582 Geodoxin synthesis use of lead dioxide, 690 Geraniol cyclization, 345, 347, 352 Geraniol, geranylcyclization, 362 Geranyl synthesis, 428 Geranylacetone cyclization, 346 Germacranolides transannular cyclization, 396 Germacrenes transannular reactions, 389 Germacrones intramolecular cyclization epoxide ring opening, 769 synthesis from protected cyanohydrins, 198 Gibberellic acid synthesis, 572, 602 Gibberellins rearrangement, 715 synthesis rearrangement of epoxides, 766 Wagner-Meerwein rearrangement, 715 Gilman reagents reaction with epoxides, 223 Ginkgolide B synthesis, 546 organocopper compounds, 220 Glaucine synthesis use of vanadium oxytrifluoride, 670 Gloeosporone synthesis, 281 Glutaric acid

diethyl ester acyloin coupling reaction, 623 Glutaric acid, perfluoro-Kolbe electrolysis, 640 Glyceraldehyde rearrangement, 831 Glyceric acid reaction with pivaldehyde, 40 Glycidic esters reaction with organocuprates, 225 Glycidic esters, 3-phenylrearrangement migratory preferences, 747 Glycidyl tosylate reaction with lithium cyanodiphenylcuprates, 224 Glycine bislactim ethers from, 53 t-butyl ester, camphor imine alkylation, 46 Glycine, 2-arylesters synthesis, Friedel-Crafts reaction, 303 Glycollic acid synthesis, 822 C-Glycosides synthesis copper catalysts, 216 Glycosides, C-aryl synthesis Friedel-Crafts reaction, 303 Glycosyl fluorides Friedel-Crafts reaction, 303 α-Glycosyl halides reaction with dialkyl homocuprates, 216 Glyoxal benzilic acid rearrangement, 831 rearrangement, 822 Glyoxal, phenylbenzilic acid rearrangement, 829 Gomberg-Bachmann-Hey process, 505 Gomberg-Bachmann process, 505 Gomberg process, 505 Grahamimycin A synthesis, 575 Grandisol synthesis, 48, 103, 785 Grignard reagents alkenyl configuration in coupling reactions, 464 alkvnic coupling reaction with 1-haloalkynes, 553 alkynide alkylation, 272 allyl coupling reactions with heteroaromatic halides, 461 coupling reaction with bromobenzene, 451 aryl alkylation, 242 dimerization, 499 2-butadienyl coupling reactions with alkyl halides, 465 coupling, 415 cross-coupling reactions with organic halides, 436 cyclopropyl coupling reactions with bromobenzene, 452

primary alkyl coupling reactions with alkenyl halides, 436 coupling reactions with aromatic halides, 450 reactions with alkenyl halides organonickel catlysis, 228 reaction with epoxides ring opening, 754 secondary alkyl coupling reactions with alkenyl halides, 440 coupling reactions with aromatic halides, 452 α -sulfonyl alkylation, 159 tertiary alkyl coupling reactions with alkenyl halides, 441 coupling reactions with aromatic halides, 452 vinyl alkylation, 242 coupling reactions, 484 Griseofulvin, dehydrosynthesis oxidation of griseophenone A, 689 Griseofulvoxin, dehydrosynthesis use of manganese dioxide, 690 Guaiane rearrangement, 388 synthesis, 396 Haagenolide synthesis Wittig rearrangement, 1010 Halides aromatic coupling reactions with primary alkyl Grignard reagents, 450 coupling reactions with secondary and tertiary alkyl Grignard reagents, 452 coupling reaction with sp³ organometallics, 450 double carbonylation, palladium-catalyzed, 1039 carbonylation, 1021 heteroaromatic coupling reactions with sp³ organometallics, 459 Haloalkylation alkyl and allyl halides, 118 arenes Friedel-Crafts reaction, 320 Halohydrins rearrangements, 787 semipinacol rearrangements, 777 synthesis epoxide ring opening, 754 Halonium ions, dialkylpreparation of Friedel-Crafts reaction, intermediate, 299 Halopropenylation alkyl and allyl halides, 118 Harman, tetrahydrosynthesis, 81 Hedycaryol transannular cyclization, 390 11-Heneicosene synthesis, 644 Heptanal reductive allylation, 109 Heptanal, 3-methylsynthesis
Subject Index

hydroformylation of 2-methyl-1-hexene, 1022 Heptanol, 5,6-epoxyring opening stereospecificity, 751 2-Heptanone lithium 2-enolates benzylation, 7 **5-HETE** synthesis, 289 12-HETE synthesis, 289 copper-catalyzed, 216 Heteroaryl compounds coupling reactions, 509 with aryl compounds, 512 Heterocuprates, 211 synthesis, 209 Heterocyclic systems benzilic acid rearrangement, 834 Heterotropantrione synthesis, 697 Hexadecanedioic acid dimethyl ester synthesis, 642 11-Hexadecynoic acid synthesis, 646 Hexadiynene synthesis, 528 Hexa-1,5-diyn-3-ene, 3-alkyl-4-(1-alkynyl)synthesis, 554 Hexamethylphosphoramide in sulfide metallation, 86 Hexanal, 2-ethylpotassium enolates alkylation, 20 Hexane, 2,5-dichloro-2,5-dimethylalkylation of 1,3-dimethylbenzene Friedel-Crafts reaction, 318 Hexane, 2,4-dihalobenzene alkylation by Friedel-Crafts reaction, 318 Hexanedioic acid dimethyl ester acyloin coupling reaction, 625 Hexanedioic acid, 3,4-diphenyldiethyl ester acyloin coupling reaction, 615 Hexanedioic acid, 2,2,5,5-tetramethyldimethyl ester acyloin coupling reaction, 625 Hexanoic acid, 2-ethylethyl ester acyloin coupling reaction, 619 Hexanol, 4,5-epoxyring opening stereospecificity, 751 2-Hexanone, 6-bromo-3,3-dimethylterminal lithium enolates cycloalkylation, 18 3-Hexanone, 4-diazo-2,2,5,5-tetramethylsynthesis, 894 3-Hexanone, 4-hydroxy-2,2,5,5-tetramethylsynthesis acyloin coupling reaction, 619 4-Hexanone, 3-methylsynthesis, 37

1-Hexene, 2-methylhydroformation phosphite-modified rhodium catalysts, 1022 3-Hexyne benzylation Friedel-Crafts reaction, 332 1-Hexyne, 1-bromoboronate reaction with 2-thienyllithium, 498 Hirsutane biosynthesis, 404 synthesis, 389 Hirsutene synthesis, 402, 590 via carbonyl-alkyne cyclization, 602 Hirsutic acid synthesis, 783 13-HODE synthesis, 488 Hoffmann elimination ylide preparation, 918 Hoffmann reaction, 908 Homoadamantane rearrangements, 854 Homoallylic alcohols anti synthesis, 984 synthesis, 263 Homoaporphine synthesis, 807 Homocubane rearrangements, 854 Homocubanecarboxylic acid synthesis, 903 Homocuprates, 211 mixed, 212 synthesis, 209 Homo-Favorskii rearrangement, 857 Homoglaucine synthesis anodic oxidation, 673 S-Homolaudanosine synthesis, 79 Homologation Wolff rearrangement, 897 **D-Homosteroids** synthesis polyene cyclizations, 369 Hooker oxidation 2-hydroxy-3-alkyl-1,4-naphthoquinones, 828 Hopane epoxide rearrangement, 745 Human leukocyte elastase ynenol lactone inhibitors synthesis, 217 Humulene rearrangement, 389 synthesis, 431, 591 coupling reaction of alkenylboranes, 473 transannular cyclization, 399 Wagner-Meerwein rearrangement, 714 Humulene 1,2-epoxide transannular cyclization, 402 Humulene 4,5-epoxide transannular cyclization, 404

Humulene 8,9-epoxide transannular reactions, 405 Humulene epoxides transannular cyclization, 402 Humulol synthesis, 399 Hybridalactone synthesis, 290 Hydantoin, 5,5'-diphenylsynthesis, 826 Hydantoin, 3-methyl-5,5-diphenylsynthesis, 826 Hydrazones deprotonation, 34 metallated metal enolate equivalents, 30 unsymmetrical deprotonation, 34 Hydrazones, N,N-dialkylmetallated alkylation, 34 Hydrazones, N.N-dimethylcyclic α,β -unsaturated alkylation, 34 deprotonation, 34 lithiated axial alkylation, 34 Hydrazulene synthesis, 406 cis-Hydrindane synthesis, 386 Hydrindanediols pinacol rearrangement, 727 Hydrindanes synthesis, 359, 602, 1052 intramolecular cyclization of cyanocyclohexanes, 48 1-Hydrindanone alkylation, 11 Hydrindanones angular alkylation stereochemistry, 17 synthesis regiospecific alkylation, 11 Hydroalumination 1-alkynes asymmetrical diene synthesis, 486 symmetrical diene synthesis, 483 vinylalanes, 266 Hydroboration alkynes organopalladium catalysis, 231 Hydrocarbons dimerization mercury-catalyzed, 1047 Hydrocarboxylation catalysts, 1027 mechanism, 1019 Hydroesterification catalysts, 1029 mechanism, 1019 styrene palladium catalyst, 1030 Hydroformylation cobalt catalysts, 1021 formation of 1-propanal, 1015

mechanism, 1019 rhodium catalysts, 1021 a-Hydroxy acids synthesis double carbonylation, 1039 **B-Hydroxy** acids 6-membered ring O,O-acetals, endocyclic enolates alkylation, 41 γ-Hydroxyalkyl bromides synthesis, 120 γ-Hydroxyalkyl iodides synthesis, 120 Hydrozirconation 1-alkynes asymmetrical diene synthesis, 486 symmetrical diene synthesis, 483 Hyperacyloin condensation synthesis of phenanthrquinone, 619 Hypericin synthesis, 699 Hypnophilin synthesis, 603 organocopper compounds, 221 Icosanoic acid, 10,16-dimethylsynthesis, 644 Illudane biosynthesis, 404 Imidazolidiones enolates diastereoselective alkylation, 45 Imides chiral 2-oxazolidones diastereoselective alkylation, 45 Imides, *a*-allenylreaction with organocopper reagents, 223 Imidoyl chloride coupling reactions with primary alkyl Grignard reagents, 463 Imines coupling reactions with 1,2-diamines, 564 halomagnesium derivatives alkylation, 31 lithiated alkylation, 31 axial alkylation, 32 metallated metal enolate equivalents, 30 metallated chiral asymmetric alkylation, 35 pinacol coupling reactions, 579 intermolecular, 579 intramolecular, 581 with ketones, 596 Imines, 2-hydroxyrearrangement, 790 semipinacol rearrangement, 778 Immonium cations initiators polyene cyclization, 343 Indane-1,3-dione, 2-diazosynthesis, 893 Indanone, perhydrosynthesis, 832 Indazolediones

Subject Index

benzilic acid rearrangement, 831 Indenecarboxylic acid synthesis, 904 Indenone, 2,3-diethylsynthesis Friedel-Crafts reaction, 332 Indenone-3-carboxylic acid, 2-alkylsynthesis, 828 Indole coupling reactions, 511 Indole, 4-bromo-3-iodosynthesis, 498 Indole, 2-phenylsynthesis, 513 Indole, 3-phenylsynthesis, 512 Indoleacetic acid, dihydroester, synthesis carbonylation, 1038 Indole alkaloids synthesis, 81 Indolizine, aminosynthesis, 541 Indolone, 2-phenylrearrangement, 835 Insertion reactions C-H carbon-carbon bond formation, 1045-1062 intermolecular, 1046 intramolecular, carbacycles, 1048 intramolecular, heterocycles, 1056 photochemical, 1048, 1057 Iodinium tosylates, alkenylcoupling reactions with alkenyl copper(I) compounds, 522 Iodonium tosylates, alkynylphenylreaction with vinylcopper compounds, 219 **w**-lonone cyclization, 349 Ionones pinacol coupling reactions, 577 Iridolactones synthesis, 850 Iridomyrmecin synthesis, 384 Iron, pentacarbonylcatalyst carbonylation of alkyl and aralkyl halides, 1026 pinacol coupling reactions aromatic aldehydes, 565 Iron complexes carbonylation formation of asymmetric iron acyls, 1029 Iron(III) chloride epoxide ring opening, 770 Iron(III) complexes catalysts aryl Grignard reagent reaction with alkenyl halides, 494 w-Irone cyclization Lewis acid induced, 349 Ishwarane synthesis, 20 Ishwarone synthesis, 20

Isoamijiol synthesis, 586, 603 Isoasatone, demethoxysynthesis, 697 Isoboldine synthesis, 679 Isobutene arene alkylation Friedel-Crafts reaction, 306 Isobutylamine, n-butylmethylation, 30 Isobutyraldehyde potassium enolates alkylation, 20 Isocarbacyclin synthesis, 139 Isocaryophyllene transannular cyclization, 387 Isocomene synthesis, 385, 713 Isocorytuberine synthesis, 807 Isocoumarins synthesis, 543 Isocyanates synthesis carbonylation, 1039 Isocyanides ortho-lithiated aryl synthesis, 255 Isoeugenol oxidation, 690 Isogermacrone epoxide rearrangement, 752 Isogibberellin from gibberellin Wagner-Meerwein rearrangement, 715 Isoharringtonine synthesis, 596 Isoheterotropantrione synthesis, 697 Isoindoline alkylation stereoselective and regioselective, 77 Isoindolinones synthesis carbonylation of o-bromoaminoalkyl benzenes, 1037 Isokaurene epoxide rearrangement, 715 Isokhusimone synthesis, 590 Isolongifolene rearrangement, 737 synthesis, 713 Isomerization Friedel-Crafts reaction, 327 Isopilocereine synthesis use of ferricyanide, 686 Isopiperitenol synthesis Wittig rearrangement, 1010 Isopodophyllotoxone synthesis, 695

Isoprene reaction with 1,2,4-trimethylbenzene Friedel-Crafts reaction, 322 Isoprene monoepoxide coupling reactions with alkenyl Grignard reagents, 476 Isoprenoids conjugated synthesis, 882 Isoquinoline, benzylasymmetric synthesis, 81 Isoquinoline, 2-chlorocoupling reactions with Grignard reagents, 461 Isoquinoline, 4-chlorocoupling reactions with Grignard reagents, 461 Isoquinoline, p-hydroxybenzyltetrahydroanodic oxidation, 666 Isoquinoline, tetrahydroalkylation, 71 oxidative coupling, 665 Isoquinoline alkaloids synthesis, 77 1,3(2,4)-Isoquinolinediones ring contraction, 835 Isoquinoline formamidine, tetrahydroalkylation selectivity, 75 Isoquinolin-1-ones, 1,2,3,4-tetrahydro-*N*-substituted, synthesis carbonylation, 1038 Isorenieratene synthesis, 585 Isosalutaridine synthesis, 679 Isosilvbin synthesis, 691 Isostegane synthesis use of vanadium oxytrifluoride, 675 Isosteviol from steviol Wagner-Meerwein rearrangement, 715 Jasmone, dihydrosynthesis, 869 Jasmonic acid methyl ester synthesis, 653 Jatrophone synthesis, 26 Juvabione synthesis from protected cyanohydrins, 198 Juvenile hormone synthesis, 99, 107 Karahanaenone synthesis pinacol rearrangement, 728 Kaurene rearrangement, 715 Ketene dithioacetals coupling reactions with alkyl Grignard reagents, 448

Ketenes acetals silyl enol derivatives, 50 Ketimines metallated alkylation, 31 unsymmetrical deprotonation, 32 α-Keto acids synthesis double carbonylation, 1039 β-Keto aldehydes metal enolates alkylation, 54 Ketocarbenes rearrangement inhibition by copper, 896 Wolff rearrangement, 893 Keto esters synthesis double carbonylation, 1039 **B**-Keto esters metal enolates alkylation, 54 a-Ketohydrazones oxidation synthesis of α -diazo ketones, 890 α-Ketol rearrangement pinacol rearrangement comparison with, 722 Ketones acyclic α -alkylated, synthesis, 26 lithiated imines, 37 regiospecific alkylation, 3 acyclic enolates alkylation, 17 acyloin coupling reactions with esters, 630 addition to diazoalkanes, 783 α -alkylated enantioselective synthesis, 35 α' -alkylated synthesis, 28 alkyl enol ether derivatives alkylation, 25 α -alkyl- β , γ -unsaturated synthesis, 23 bicyclic alkylation, 11 cyclic α -alkylated, synthesis, 26 homologation, 781 regiospecific alkylation, 3 dienolates intramolecular y-alkylation, 25 dimethylthioacetal S,S-dioxides ketone synthesis from, 143 fluoro synthesis, epoxide ring opening, 748 homologation, 783 intermolecular pinacol coupling reactions, 570 macrocyclic lithiated imines, 37 metal enolates alkylation, 3

α-metallated formation, 3 O-metallated tautomers formation, 3 α -methyl α , β -unsaturated synthesis, 33 reaction with diazoalkanes homologation, 778 saturated cycloalkylation, 18 steroidal synthesis, regiospecific alkylation, 11 synthesis α-alkoxy carbanions, 197 alkylboronic esters, 797 alkynylborates, 799 carbonylation, 1023 coupling reactions with organometallics, 463 cyanoboronates, 798 α -heterosubstituted sulfides and selenides, 141 organoboranes, 780, 793 tricyclic synthesis, regiospecific alkylation, 11 α,β -unsaturated metal dienolates, alkylation, 21 methylation, 23 synthesis, 880, 894 unsymmetrical regioselective alkylation, 2 synthesis, 199 unsymmetrical enamines alkylation, 30 unsymmetrical enolates regioselective alkylation, 7 Ketones, allenyl synthesis, 991 Ketones, 2-allyloxy Wittig rearrangement, 996 Ketones, 2-amino rearrangement, 790 Ketones, 2-arylated synthesis Friedel-Crafts reaction, 306 Ketones, aryl methyl synthesis carbonylation, 1024 Ketones, γ -aryl- α -trifluoromethyl cycloalkylation Friedel-Crafts reaction, 324 Ketones, bicyclic halo Favorskii rearrangement, 851 Ketones, 4-t-butylcyclohexylmethyl exocyclic enolate methylation, 16 Ketones, 2-chloro homologation, 787 Ketones, diallyl synthesis carbonylation, 1023 Ketones, diazo alkylation, 794 C---H insertion reactions, 1054 copper-catalyzed, 1051 Ketones, divinyl synthesis, 844 Ketones, 2,3-epoxy

ring opening to 1,3-diones, 771 Ketones, 2-ethenyl synthesis vinylmagnesium halide alkylation, 242 Ketones, 2-halo acyclic Favorskii rearrangement, 842 alkene synthesis from, 871 homologation, 783 reaction with vinyl and aryl Grignard reagents, 242 rearrangements, 788, 828, 839 Ketones, 2-haloalkyl aryl ketals rearrangement, 789 Ketones, 2-halocycloalkyl rearrangements, 845 Ketones, 2-hydroxy rearrangements, 791 synthesis carbonylation of lithium amides, 1017 Ketones, 3-hydroxy synthesis from α,β -epoxy ketones, 264 Ketones, ω-iodoalkyl zinc compounds from coupling reactions with alkenyl halides, 443 Ketones, monocyclic halo Favorskii rearrangements, 848 Ketones, phenylthienyl synthesis, 515 Ketones, polycyclic halo Favorskii rearrangement, 854 Ketones, polyhalo Favorskii rearrangement, 843 Ketones, 2-substituted N-(9-phenylfluorene-9-yl)amino deprotonation, alkylation, 44 Ketones, tetrahydrofurfuryl reactions with Grignard reagents, 996 Ketones, α , β -unsaturated synthesis Ramberg-Bäcklund rearrangement, 870 Ketones, β , γ -unsaturated synthesis π -allylnickel halides, 424 Ketoximes unsymmetrical deprotonation, 35 a-Ketoximes oxidation synthesis of α -diazo ketones, 890 Khellin synthesis Sommelet-Hauser rearrangement, 969 Koch reaction carboxylic acid synthesis, 1017 Kolbe electrolysis anode material, 635 cathode material, 636 critical potential, 634 current densities, 634 ionic additives, 634 mechanism, 636 pH, 634 reaction conditions, 634

solvents, 635 Kolbe radicals addition to double bonds, 646 Kolbe reactions, 633-658 Kreysiginone synthesis use of vanadium oxytrifluoride, 681 Labda-7,14-dien-13-ol synthesis, 168 Lactams bicyclic alkylation, 42 synthesis carbonylation, 1035 **B-Lactams** synthesis, 902 C-H insertion reactions, 1056 y-Lactams synthesis C-H insertion reactions, 1057 B-Lactams, azasynthesis, 902 β-Lactams, N-benzyl-Wittig rearrangement, 979 Lactams, halorearrangements, 849 Lactones arene alkylation Friedel-Crafts reaction, 309 five-membered ring synthesis, carbonylation of allylic alcohols, 1032 macrocyclic synthesis, 431 synthesis carbonylation, 1031 epoxide ring opening, 752 β-Lactones Grignard reagent alkylation, 245 reaction with organocopper compounds, 227 β-substituted enolates alkylation, 41 γ -Lactones β - and γ -substituted alkylation, 41 synthesis carbonylation, 1031 δ-Lactones enolates alkylation, 41 synthesis carbonylation of homoallylic alcohols, 1033 γ -Lactones, α -hydroxysynthesis from protected cyanohydrins, 198 Lactones, vinyl ring-opening and coupling reactions with Grignard reagents, 476 Lanthanide trihalides catalysts Friedel-Crafts reaction, 295 Lanthanoids use in pinacol coupling reactions, 567 Lanthanoid trichlorides Friedel-Crafts catalysts, 302 benzylation of benzene, 302

Laudanosine oxidation, 685 oxidative coupling, 670 synthesis, 80 Laudanosoline methiodide oxidative coupling, 666 Laurencin synthesis, 126 Laurenene synthesis, 385 Lavandulol cyclization, 345 Lead triacetate, arylaromatic arylation reactions, 505 Leucine t-butyl ester, enamines alkylation, 36 Leukotriene A4 synthesis, 289 Leukotriene B4 synthesis, 489 Leukotriene B₄, 14,15-dehydrosynthesis, 289 Leukotrienes synthesis, 289 organocopper compounds, 217 Lewis acids Friedel-Crafts reaction catalysts, 295 Lewis superacids catalysts Friedel-Crafts reaction, 297 Libocedrol synthesis use of alkaline ferricyanide, 686 Limonene hydroformation phosphite-modified rhodium catalysts, 1022 oxiranes rearrangement, 771 synthesis from α- and Mf128b-pinene, 708 Linalool cyclization, 352 synthesis, 170 Linaloyl oxide synthesis, 126 Lithium, alkenylcoupling reactions with alkenyl halides, 485 Lithium, alkylprimary coupling reactions with alkenyl halides, 436 Lithium, allenylreaction with epoxides, 264 Lithium, allyl-Wurtz coupling, 419 Lithium, arylalkylation, 247 coupling reactions with alkenyl halides, 494 Lithium, benzyldithiocarbamatoalkylation, 95 Lithium, benzylphenylthioalkylation, 95

Subject Index

Lithium, benzylthiothiazolinoalkylation, 95 Lithium, bis(phenylthio)benzylsynthesis, 123 Lithium, bis(phenylthio)methylsynthesis, 123 Lithium, t-butoxymethylsynthesis, 194 Lithium, t-butylcoupling dihalides, 419 Lithium, cyclohexadienylalkylation, 255 Lithium, cyclopentenylsynthesis, 247 Lithium, (dialkoxyphosphoryl)trimethylsilylalkylalkylation, 201 Lithium, 1,1-dichloroallylsynthesis alkylation, 202 Lithium, diethoxymethylsynthesis by transmetallation, 196 Lithium, (diethoxyphosphoryl)dichloromethylalkylation, 202 Lithium, 1,1-difluoroallylsynthesis alkylation, 202 Lithium, (dimethylphosphoryl)methylalkylation, 201 Lithium, diphenylarsinomethylalkylation with epoxides, 203 synthesis by transmetallation, 203 Lithium, (diphenylphosphinoyl)alkylalkylation with epoxides, 201 Lithium, 1,1-(diseleno)alkylsynthesis, 87 Lithium, 1,1-(diseleno)benzylsynthesis, 87 Lithium, 1,1-(dithio)allylalkylation, 131 synthesis, 131 Lithium, 1,1-(dithio)propargylalkylation, 131 Lithium, 2-furylcoupling reactions with alkenyl bromides, 497 Lithium, glycosylsynthesis by transmetallation, 196 Lithium, methylselenomethylalkylation, 90 Lithium, 1-phenylseleno-1-hexylalkylation, 90 Lithium, phenylselenomethylalkylation, 90 Lithium, a-selenoalkylacyl anion equivalents synthesis, 121 alkylation, 88 allylation, 91 reactions, 88 synthesis, 87

Lithium, α -selenoallenylsynthesis, 87 Lithium, α -selenoallylalkylation, 95 synthesis, 87 Lithium, α -selenobenzylalkylation, 94, 95 synthesis, 87 Lithium, 1-selenocyclobutylalkylation, 90 Lithium, 1-selenocyclopropylalkylation, 90 Lithium, a-selenopropargylalkylation, 104 synthesis, 87 Lithium, a-selenovinylsynthesis, 87 Lithium, 2-thienylcoupling reactions with alkenyl bromides, 497 Lithium, a-thioalkylacyl anion equivalents synthesis, 121 alkylation, 88 reactions, 88 synthesis, 87 Lithium, a-thioallylalkylation, 95 reaction with allyl halides, 99 reaction with epoxides, 100 Lithium, α -thiobenzylalkylation, 94 Lithium, a-thiopropargylalkylation, 104 Lithium, trialkylsilyldichloromethylalkylation, 200 Lithium, 3-triethylsilyloxypentadienylalkylation, 196 Lithium, a-trimethylsilylallylalkylation regioselectivity, 200 Lithium, trimethylsilyl(phenylthio)methylalkylation, 137 Lithium, tris(trimethylsilyl)methylalkylation, 200 Lithium, vinylalkylation, 247 reaction with alkyl halides, 247 Lithium alkylcyano(2-thienyl)cuprate preparation, 213 Lithium alkynides alkylation alkyl halides, 272 reaction with methyl triflate, 281 Lithium alkynylcuprates reaction with haloallenes, 274 Lithium amides reaction with carbon monoxide mechanism, 1017 Lithium compounds use in intermolecular pinacol coupling reactions aliphatic carbonyl compounds, 570 Lithium dialkylcuprates conjugate additions enolate synthesis, 8 Lithium dialkylcyanocuprate

structure, 213 Lithium diallenylcuprates alkylation, 256 Lithium dimethylcuprate mixture with water structure, 212 stability, 211 synthesis, 208 tetrahydrofuran solution structure, 211 Lithium diorganocopper compounds coupling, 419 Lithium diphenylcuprate reaction with alkyl bromide, 248 reaction with allylic acetate, 257 Lithium divinylcuprates coupling reactions with enol triflates, 487 vinyl halide coupling, 482 Lithium enolates alkylation, 2 a-methyl substituents axial alkylation, 14 Lithium halides catalysts epoxide rearrangement, 760, 763 Lithium halocarbenoids alkylation, 202 Lithium halocarbenoids, cyclopropylidenesynthesis by halogen-lithium exchange alkylation, 202 Lithium hexamethyldisilylamide ketone enolates synthesis, 3 Lithium iodophenylcuprate coupling with allylic alcohols, 259 Lithium naphthalide reduction, 263 Lithium t-octyl-t-butylamide ketone enolate synthesis, 3 Lithium pentamethyltricuprate structure, 213 Lithium perchlorate catalyst epoxide ring opening, 760, 761 Lithium salts catalysts epoxide ring opening, 760 Lithium tetramethyldiphenyldisilylamide ketone enolates synthesis, 3 Lithium thiocyanate catalysts epoxide ring opening, 767 Lithium trimethyldicuprate structure, 213 Lithium triphenylcuprate structure, 213 Loganin synthesis, 599 Longifolene synthesis, 20, 357 pinacol rearrangement, 729 Wagner-Meerwein rearrangement, 713 Looplure synthesis, 644

Lophotoxin synthesis, 497 Lossen reaction, 908 Luciferin aldehyde synthesis epoxide ring opening, 743 Lysergic acid, 2-methylsynthesis, 126 McMurry reaction, 583 intermolecular, 585 intramolecular, 588 Macrocyclic ethers Wittig ring contractions, 1010 Macrolactams synthesis C-H insertion reactions, 1057 Macrolides synthesis, 286 C-H insertion reactions, 1058 Macrophyllate, ethylsynthesis use of alkaline potassium ferricyanide, 665 Maesopsin ring scission, 831 Maesopsin, 4,4',6-tri-O-methylirradiation, 831 Magic acid catalyst Friedel-Crafts reaction, 297 Magnesium graphite-suspended use in intermolecular pinacol coupling reactions, 570 Magnesium, alkynylhalocross-coupling reactions with vinyl iodides, 527 Magnesium, bromoethynylsynthesis, 271 Magnesium, bromo(a-silylvinyl)alkylation, 244 Magnesium, bromo(2-thienyl)reaction with vinyloxirane, 265 Magnesium, bromovinylalkylation, 243 Magnesium, chloro(diisopropoxymethyl)silylmethylhydroxymethylation with, 200 Magnesium, chloroprenylalkylation copper catalysis, 215 Magnesium, ethynylidenebis(bromosynthesis, 271 Magnesium halides epoxide ring opening, 754 Magnocurarine methiodide model reaction dimerization, 687 Makomakine synthesis use of ammonium ylides, 955 Maleates, iododimerization, 482 Malic acid diethyl ester alkylation, 44 Malonic acid, (w-bromoalkyl)-

Subject Index

diethyl ester intramolecular alkylation, 55 Marasmane biosynthesis, 404 Maritimol synthesis, 717 Maytansin synthesis, 126 Mecambrine synthesis alkaline photolysis, 686 Meerwein arylation, 505 Meerwein-Ponndorf-Verley reductions electron transfer mechanism, 824 Menthadiene cyclization, 349 Menthol, 8-phenylchiral malonic esters intermolecular alkylation, 56 Menthone rearrangement, 831 Mercury, arylchlororeaction with vinyl cuprates, 497 Mercury, chlorovinylcoupling reactions with vinyl cuprates, 489 Mercury, diarylextrusion of mercury, 501 Mercury, vinyldimerization diene synthesis, 484 Mercury bis(trifluoro)acetate polyene cyclization, 342 Mercury(II) salts, aryldimerization, 501 Metacyclophanes synthesis, 126 Metal alkoxides catalysts Friedel-Crafts reaction, 296 Metal alkyls catalysts Friedel-Crafts reaction, 296 Metal dienolates α,β -unsaturated ketones alkylation, 21 Metal enolates C-alkylation, 4 O-alkylation competition with C-alkylation, 4 carboxylic acid derivative alkylation, 39 chirality transfer, 13 molecular aggregates dependence on solvent, 3 saturated aldehydes alkylation, 20 stabilized alkylations, 54 Metallation alkynes, 271 Metallocarbene insertion, 1047 Metazocine synthesis, 77 Methane

ethylation Friedel--Crafts reaction, 333 Methane, alkoxybis(sulfonyl)alkylation, 177 Methane, bis(N,N-dimethyldithiocarbamato)methylthiomethyl ester alkylation, 136 Methane, bis(1,3,2-dioxaborin-2-yl)deprotonation alkylation of anion, 199 Methane, diazo-C-acylation, 888 Methane, dichlorodiphenylsynthesis Friedel-Crafts reaction, 320 Methane, iodocarbonylation formation of acetyl iodide, 1018 Methane, methoxybis(trimethylsilyl)deprotonation with s-butyllithium, 198 Methane, methoxy(trimethylsilyl)deprotonation with s-butyllithium, 198 Methane, polyhaloreaction with ketones, 787 Methane, tris(dimethoxyboryl)cleavage alkylation of anion from, 199 Methanesulfinic acid, trifluoro-Ramberg-Bäcklund rearrangement, 868 Methanesulfonic acid, trifluorocatalyst Friedel-Crafts reaction, 297 esters conversion to amides by carbonylation, 1035 Methanesulfonyl chloride 2-hydroxyselenide elimination reactions, 787 Methanols, trialkylsynthesis, 793, 794 Methyl ceriferate synthesis, 99 Methyl iodide alkylation with, 14 Methyl vernolate rearrangement, 752 Methynolide synthesis stereocontrol, 960 Mevinolin synthesis, 589 Minelsin anticholinergenic and spasmolytic agent, 826 Mitosene, 7-methoxysynthesis, 261 MK-801 synthesis, 71 Mokupalide synthesis, 99 Molybdenum, arenetricarbonylcatalyst Friedel-Crafts reaction, 300 Monsanto process acetic acid production, 1018 Montmorillonite catalyst

Friedel-Crafts reaction, 300 Mukapolide synthesis reduction of sulfides, 107 Multifloramine synthesis, 807 use of ferricyanide, 681 Muscalure synthesis, 644 Muscone synthesis, 168, 787 Muscopyridine synthesis coupling reactions, 460 Mycinomycin synthesis, 797 Myrcene synthesis, 429 Nafion resin catalyst, solid superacid Friedel-Crafts reaction, 298 Nametkin rearrangement, 706 Naphthalene alkylation 1-bromoadamantane, 302 Friedel-Crafts reaction, 304 competitive alkylation Friedel-Crafts reaction, 300 isopropylation Friedel-Crafts reaction, 304 regioselective isopropylation Friedel-Crafts reaction, 305 Naphthalene, 2-(1-adamantyl)synthesis Friedel-Crafts reaction, 302 Naphthalene, t-butylsynthesis Friedel-Crafts reaction, 311 Naphthalene, dihydrothienylsynthesis, 497 Naphthalene, methylisomerization Friedel-Crafts reaction, 327 Naphthalene, tetrahydrochiral derivatives synthesis, 327 synthesis Friedel-Crafts reaction, 311 Naphthene, 2-diazo-1-oxoring contraction, 902 2-Naphthol oxidative dimerization, 665 1,4-Naphthoguinone, 2,3-dichloromonoalkylation with tetraalkyltins or alkylzirconium complexes, 458 Naproxen asymmetric synthesis, 789 synthesis, 1022 Napthaldehyde-9-carboxylic acid synthesis, 828 Narwedine synthesis, 683 Neoclovene synthesis, 386

Neolignan synthesis use of silver oxide, 691 Neopentane synthesis, 415 Nerol synthesis stereoselectivity, 180 Nerol, nerylsynthesis, 170 Nerolidol synthesis, 170 Nickel catalysts cross-coupling reactions, 523 Nickel, π -allylhalochemoselectivity, 424 preparation, 423 reactions, 423 Nickel, bis(1,5-cyclooctadiene)alkenyl halide dimerization diene synthesis, 483 catalyst Ullmann reaction, 500 Nickel, dichloro(1,2-bis(diphenylphosphino)ethane)catalyst Grignard reagents, 228 Nickel, dichloro(1,3-bis(diphenylphosphino)propane)catalyst crossed alkene coupling, 484 Grignard reagents, 228 Nickel, dichlorobis(trialkylphosphine) catalyst Ullmann reaction, 500 Nickel, dichlorobis(triphenylphosphine)catalyst crossed alkene coupling, 484 Grignard reagents, 228 Nickel, tetrakis(triphenylphosphine)catalyst crossed alkene coupling, 484 Ullmann reaction, 500 Nickel catalysis carbanion alkylations, 227 Nickel complexes catalysts Grignard reagent alkylation, 244 Wurtz reaction, 421 Nickel complexes, π -allylregioselectivity, 426 stereoselectivity, 426 Nicotelline synthesis, 510 Nicotinates, 5-arvlsynthesis, 515 Nitriles carbanions intramolecular alkylation, 49 intramolecular alkylation, 48 Nitriles, epoxyaromatic α -cleavage, 748 Nitriles, α , β -unsaturated synthesis Ramberg-Bäcklund rearrangement, 870 Nitrogen ylides

Subject Index

preparation, 918 Nitrosamines anions alkylation, 66 deprotonation, 65 1,8-Nonadiyne oxidative polymerization, 552 1,9-Nonanedioic acid, 5-methylenedimethyl ester intramolecular acyloin coupling reaction, 625 Nonanoic acid, 2-methyldimethyl ester intramolecular acyloin coupling reaction, 626 Non-Kolbe electrolysis, 634 carbenium ions, 649 experimental procedure, 654 Noradamantane synthesis, 854 epoxide ring opening, 746 Norbornadiene carbonylation cobalt carbonyl catalyst, 1024 oxide rearrangement, 736 Norbornane carbocations rearrangement, 707 Norbornane-2-carboxylic acid enolates diastereoselective alkylation, 39 Norbornene oxide rearrangement, lithium halide catalyzed, 764 rearrangement, lithium perchlorate catalyzed, 761 2,3-exo-oxides rearrangement, 740 Norbornene, 2-chloroexo-oxide rearrangement, 739 Norbornenone homologation of ketones, 783 A-Nor-5 α -cholestan-3 β -ol, 3 α -carboxysynthesis, 834 A-Nor-5α-cholestan-2β-ol-2-carboxylic acid rearrangements, 832 synthesis, 833 Norcoralydine synthesis, 81 18-Nor-D-homo steroids angular alkylation, 17 A-Nor-5α-pregnan-2-ol-20-one-2-carboxylic acid synthesis, 833 Norpyrenophorin synthesis, 126 Norrish type II reaction cyclobutanol, 1048 19-Norsteroids synthesis polyene cyclization, 371 A-Norsteroids synthesis, 903 **D**-Norsteroids synthesis, 901 Nortricyclene, 3-methoxysynthesis, 653 A-Nortriterpenes

synthesis, 903 Nuciferal synthesis, 161 Nucleosides, 6-alkylpurine synthesis coupling reactions, 462 Obtusilactone synthesis, 844 Ocoteine intracoupling reaction with benzyltetrahydroisoquinoline, 670 synthesis, 81 2,6-Octadiene cyclization, 342 1(9)-Octalin-2-one α' -alkylation, 21 cross conjugated lithium dienolate metallation, 21 1(9)-Octalin-2-one, 10-methylcyclohexylamine methylation, 33 Octalins conformation, 354 Octane, 1,2-epoxyhydride migration epoxide ring opening, 742 1,8-Octanedioic acid, 2,7-dimethyldimethyl ester synthesis, 623 4-Octanone, 5-hydroxysynthesis acyloin coupling reaction, 619 Octatetraynediamines synthesis, 555 1,4,7-Octatriyne-3,6-diol, 3,6-di-t-butylsynthesis, 557 2-Octyne, 1-bromoreaction with Grignard reagents, 273 2-Octyne, 1-iodoreaction with 3-butyn-1-ol, 273 13(18)-Oleanene synthesis, 709 Oleic acid Kolbe electrolysis, 642 Oligomerization aliphatic alkenes Friedel-Crafts reaction, 331 α -Onocerin synthesis, 638 Ophiobolins synthesis, 575 Orellanine synthesis, 509 Organoboranes rearrangements, 779 Wurtz coupling, 418 Organoboron compounds rearrangements, 793 Organocopper compounds alkylations nonstabilized carbanions, 208 catalysts, 210 preparation, 208 coupling, 415 cross-coupling reactions

unsaturated halides, 522 reaction with electrophiles mechanism, 213 synthesis, 208, 419 Organolithium compounds chiral dipole-stabilized stereoselective alkylation, 75 Wurtz coupling, 419 Organomagnesium compounds primary coupling reactions with alkenyl halides, 436 Wurtz coupling, 415 Organometallic compounds 1-alkenyl reaction with 1-alkynyl halides, 529 cross-coupling reactions with unsaturated halides, 522 vinyl alkynylation, 521 Organometallic polymers synthesis, 557 Organonickel compounds catalysts carbanion alkylations, 228 Organopalladium compounds catalysts Grignard reactions, 230 Organosodium compounds coupling, 414 Organostannanes α -alkoxy organolithiums from, 195 Organotin compounds coupling reactions with alkenyl halides, 442 with aromatic halides, 452 primary alkyl coupling reactions with aromatic halides, 453 Organozinc compounds copper-catalyzed reactions, 221 coupling reactions with alkenyl halides, 442 with aromatic halides, 452 nickel catalysts, 228 perfluoroalkyl with alkenyl halides, 444 primary alkyl coupling reactions with alkenyl halides, 442 coupling reactions with aromatic halides, 453 secondary alkyl coupling reactions with alkenyl halides, 442 Wurtz coupling, 420 Orientalone synthesis use of potassium ferricyanide, 680 Orthoformylation aromatic compounds, 969 Ortyn anticholinergenic and spasmolytic agent, 826 Ovatodiolide transannular cyclization, 407 Oxabetweenallenes synthesis organocopper compounds, 223 3-Oxabicyclo[3.3.0]oct-6-en-7-ones synthesis use of cobalt complexes, 1025

Oxamates synthesis carbonylation of amines and alcohols, 1040 Oxaporphine synthesis oxidative coupling, 670 1-Oxaspiro[2.6]nonane solvolysis transannular hydride shifts, 735 Oxaspiropentanes rearrangement lithium perchlorate catalyzed, 761 1.3-Oxathiane metallated alkylation, 135 1.3-Oxathiane, 2-lithioalkylation, 137 Oxathiolane, 4,4-dimethyl-3,3-dioxide alkylation, 136 1,3-Oxazine, 2-alkyldihydroalkylation, 53 1,3-Oxazine, allyloxymethyl-Wittig rearrangement, 1005 Oxazole, dihydroenolates diastereoselective alkylation, 40 Oxazole, 2-stannylcoupling reactions, 511 with bromobenzenes, 514 Oxazolidine enolates diastereoselective alkylation, 40 2-Oxazolidones synthesis from organic isocyanates and terminal alkene epoxides, 765 Oxazoline metallation, 261 Oxazoline, 2-alkylalkylation, 53 Oxazoline, allyloxymethyl-Wittig rearrangement, 1005 Oxazoline, allyloxymethyl anions Wittig rearrangement lithium cation chelation, 1006 Oxepinobenzofuran synthesis ferricyanide oxidation, 666 Oxetane, 2-methylbenzene alkylation by Friedel-Crafts reaction, 314 3-Oxetanecarboxylic acid ester Wolff rearrangement, 902 Oxetanes alkylation with sulfur- and selenium-stabilized carbanions, 86 arene alkylation by Friedel-Crafts reaction, 314 coupling reactions with phenyllithium, 466 reaction with α -selenoalkyllithium, 91 Oxidative coupling copper-polymer complex catalysts, 559 organometallic acetylides, 554

phenols, 659-700 terminal alkynes, 552 Oxidative homocoupling reactions terminal alkynes, 552 2,3-Oxidosqualene synthesis, 99 Oximes metallated metal enolate equivalents, 30 pinacol coupling reactions with ketones, 596 Oxindoles synthesis Sommelet-Hauser rearrangement, 969 Oxirane, vinylreaction with aryl Grignard reagents, 265 Oxiranes (see also Epoxides) neighboring group epoxide opening, 735 Oxocrinine synthesis, 683 Oxonium ions initiators polyene cyclization, 343, 354 Oxonium ylides rearrangements, 942 Wittig rearrangement, 1008 Oxyallyl cations polyene cyclization, 354 Oxy-Cope rearrangement following Wittig rearrangement, 994 Oxymercuration humulene, 400 Palladium, allylchlorocatalyst TASF reaction with organic halides, 233 Palladium, bis(acetonitrile)dichlorocatalyst vinyl iodide reaction with organotin compounds, 232 Palladium, dichlorobis(triphenylphospine)catalysis halide carbonylation, 1021 vinyl iodide reaction with organotin compounds, 232 Palladium, phenylbis(triphenylphosphine)catalysis arylmagnesium halide reaction with alkyl halides, 244 Palladium, tetrakis(triphenylphosphine)catalyst aryl halide reaction with organotin compounds, 232 coupling reactions between organolithium and vinyl halides, 485 halide carbonylation, 1021 vinylic Grignard coupling, 485 Palladium catalysis carbanion alkylations, 227 cross-coupling reactions, 523 synthesis of enynes, 217 Palladium complexes ferrocene catalyst, Grignard reagent alkylation, 244 Palladium complexes, π -allylreactions, 423

Palladium dichloride catalyst alkene dimerization, 482 alkenyl halide dimerization, 484 [8]Paracyclophane, 4-carboxysynthesis, 905 [3.3]Paracyclophanediene synthesis, 877 Paracyclophanes synthesis intramolecular acyloin coupling reaction, 628 Parham-type cyclization benzocyclobutanes, 251 Parinaric acid synthesis, 116 Payne rearrangement epoxides hydroxy neighboring group, 735 Penicillins, diazorearrangement, 934 Penicillins, semisynthetic synthesis 2-arylglycines, 303 2,4-Pentadieno-4-lactone synthesis carbonylation of alkynes, 1032 Pentalenane synthesis, 389 Pentalene synthesis pinacol rearrangement, 726 Pentalenene synthesis, 20, 384, 400 Pentalenene, hydroxysynthesis, 400 Pentalenic acid synthesis, 400 Pentalenolactone E synthesis, 400 C-H insertion reactions, 1059, 1060 Pentalenolactone F synthesis, 400 Pentalenolactone G synthesis, 766 Pentalenolactones synthesis, 400 I-Pentalol, 4-phenylsynthesis Friedel-Crafts reaction, 315 Pentane, 2,2,4-trimethylbenzene alkylation with Friedel-Crafts reaction, 322 1.5-Pentanedioic acid dimethyl ester acyloin coupling reaction, 615 Pentanoic acid, 3-diazo-2,4-dioxomethyl ester Wolff rearrangement, 897 2-Pentanol, 3-bromoreaction with magnesium halides, 755 3-Pentanol, 2-bromoreaction with magnesium halides, 755 1-Pentanol, 3,4-epoxyring opening stereospecificity, 751 2-Pentanone, 5-bromo-3,3-dimethyl-

terminal lithium enolates cycloalkylation, 18 4-Pentene-1,3-diol synthesis formation of lactones on oxycarbonylation, 1033 4-Pentenoic acid, 3-hydroxypalladium-catalyzed carbonylation formation of dilactones, 1032 Peptides synthesis Friedel-Crafts benzylation, 302 Peracetic acid, trifluoroboron trifluoride mixture oxidant, 753 Perinaphthenones synthesis Friedel-Crafts cycloalkylation, 325 Permutit Q catalyst Friedel-Crafts reaction, 297 Peroxides photolysis, 642 Pestalotin synthesis, 278 Peterson alkene synthesis Hudrlik version, 224 Pethidine synthesis, 845 [2.2]Phanes synthesis, 414 Phase-transfer catalysts diyne synthesis, 559 Phenanthraquinone synthesis acyloin coupling reaction, 619 Phenanthrene, 4,5-bis(dimethylamino)synthesis Ramberg-Bäcklund rearrangement, 876 Phenanthrene, 9-diazo-10-oxo-Wolff rearrangement, 903 Phenanthrene, hydrosynthesis epoxide ring opening, 753 Phenanthrene, octahydrosynthesis Friedel-Crafts cycloalkylation, 325 Phenanthrene, perhydrosynthesis, 578, 640 Phenanthrenes automerization Friedel-Crafts reaction, 331 synthesis, 507 Ramberg-Bäcklund rearrangement, 876 regiospecific alkylation, 11 Phenanthridines synthesis organopalladium catalysts, 231 Phenanthrols synthesis organopalladium catalysts, 231 Phenanthroquinones benzilic acid rearrangement, 828 synthesis, 828 Phenazine-1,6-dicarboxylic acid synthesis, 699 Phenol, o-amino-

o-diazo oxides from, 904 Phenol, p-t-octylsynthesis Friedel-Crafts reaction, 307 Phenol ethers oxidative coupling, 659-700 biaryls, 668 mechanism, 660 trimerization, 669 Phenols alkylation branched alkenes, 304 o-alkylation by 1-hexene Friedel-Crafts reaction, 306 alkylation with isobutene Friedel-Crafts reaction, 306 cycloalkylation Friedel-Crafts reaction, 304 oxidative coupling, 659-700 electron transfer, 661 mechanism, 660 radical substitution, 661 phenol coupling, 663 postoxidative coupling, 662 a-Phenylsulfonyl esters enolates reaction with allylic acetate, 56 Phenylthioalkylation silyl enol ethers, 25 Phenylthiomethylstannylations silyl enol ethers, 26 Pheromones synthesis, 643, 644 Phosphazines diazo-coupling reactions, 893 Phosphine oxide, alkylalkylation, 201 Phosphine oxide, α -diazo-Wolff rearrangement, 909 Phosphine oxide, ethyl[(menthoxycarbonyl)methyl]phenyldeprotonation alkylation, decarboxylation, 201 Phosphines, β-(dimethylamino)alkylnickel compounds Grignard reagent catalysts, 228 Phosphines, ferrocenylnickel compounds Grignard reagent catalysts, 228 Phosphites, trimethylsilyldiethyl ester reaction with aldehydes, 199 Phosphonates, α, α -difluoroalkyl alkylation, 202 Phosphonates, a-fluoro alkylation, 202 Phosphonic acid, bis(dimethylamido)allyl ester deprotonation, 199 Phosphonic acid, ethoxymethyl(2-trimethylsilyl)diethyl ester alkylation, 199 Phosphorus ylides alkylation formation of phosphonium salts, 200 solubilizer

Subject Index

lithium halides, 760 Photochemical pinacolization aromatic compounds, 567 Photolithography diazo ketones, 887 Phthalides synthesis carbonylation of halides, 1033 Phthalimidine-3-carboxylic acid, 3-hydroxysynthesis, 835 Phthalonimides rearrangements, 835 Piceatannol, dihydrooxidative coupling, 666 Picrolichenic acid synthesis use of manganese dioxide oxidation, 679 Picropodophyllone synthesis, 696 Pilocerine synthesis, 687 Pinacol coupling reactions, 563-605 intermolecular, 564 intramolecular, 572 mixed, 595 Pinacolone label redistribution pinacol rearrangement, mechanism of, 723 synthesis pinacol rearrangement, 721 Pinacol rearrangement, 721-730 applications, 726 definition, 721 mechanism, 722 migratory aptitudes, 726 Pinacols, thiosynthesis, 582 Pinanediol boronic esters, 796 3-Pinanone, 2-hydroxyglycinate esters, enolates alkylation, 46 α -Pinene oxide rearrangement, 771 rearrangement, 705 Pinenes fragmentation, 708 Pipercide, dehydrosynthesis palladium-catalyzed coupling reactions, 545 Piperidine, dehydroasymmetric alkylation, 77 Piperidine, N-methyldeprotonation, 65 **Piperidine** amides alkylation, 69 Piperidine formamidine, α -allylic metalation, 70 Piperidine formamidines alkylation, 69 Piperitone oxiranes rearrangement, 771 Pivalaldehyde synthesis

epoxide ring opening, 742 Pivalic acid synthesis Friedel-Crafts dealkylation, 330 Platinum(II) complexes, halomethyl(arylphosphine)metallocyclization Friedel-Crafts reaction, 323 Pleraplysillin 1 synthesis, 487 Plysiatoxin, debromosynthesis, 168 Podototarin synthesis use of alkaline potassium ferricyanide, 665 Polyalkenes cyclization mechanism, 374 Polyalkylation enolates equilibration, 4 Polyalkynes linear synthesis, 555 synthesis, 551 Polyenepolyynes cyclic synthesis, 556 Polyenes bicyclizations, 359 conjugated synthesis, 878 cyclization, 341-375 monocyclization, 347 tetracyclization, 362 tricyclization, 362 Polyenes, ω,ω'-biazulenylsynthesis, 586 Polyenes, bis-2-thienylsynthesis, 586 Polymers synthesis from α, ω -diethynyl monomers, 557 Polyporic acid rearrangement, 828 Polyynes platinum polymer, 558 Porphycenes synthesis, 594 Porphyrins pinacol rearrangement, 729 Potassium, phenylthioallylmethylation selectivity, 99 Potassium, phenylthioprenylmethylation selectivity, 99 Potassium carbonylferrate halide carbonylation formation of aldehydes, 1021 Potassium enolates α, α -disubstituted aldehydes alkylation, 20 Potassium triisopropoxyborohydride alkene reduction, 797 5α-Pregn-1-en-2-ol-3,20-dione rearrangement, 833

Pregnenolone, 17a-bromorearrangements, 846 Progesterone, 11a-hydroxyenantiospecific synthesis, 371 synthesis, 126 Progestin synthesis, 846 Prolinol amides alkylation, 45 Propacin synthesis, 691 Propanal, 2'-(2-methoxy-6-naphthyl)synthesis rhodium-catalyzed hydroformylation, 1022 Propanal, 2-methyl-3-phenyllithiation with tributylstannyllithium, 195 Propanal, 2-phenylacetal synthesis, hydroformylation of styrene, 1022 Propanal, 3-phenylacetal synthesis, hydroformylation of styrene, 1022 Propane propylation Friedel-Crafts reaction, 333 Propane, 3-chloro-2-methyl-1-phenylthiometallation, 89 Propane, 1-chloro-2-phenylbenzene alkylation Friedel-Crafts reaction, 300, 302 Propane, 2-chloro-1-phenylbenzene alkylation Friedel-Crafts reaction, 300, 302 Propane, 1,1-diphenylsynthesis Friedel-Crafts reaction, 311 Propane, 1,2-diphenylsynthesis Friedel-Crafts reaction, 300 1,2-Propanediol pinacol rearrangement, 725 1-Propanol, 2,3-epoxyopening Payne rearrangement, 735 1-Propanol, 2-phenylbenzene alkylation with Friedel-Crafts reaction, 311 synthesis Friedel-Crafts reaction, 313 2-Propanol, 1-phenylbenzene alkylation with Friedel-Crafts reaction, 311 synthesis Friedel-Crafts reaction, 313 2-Propanone, 1,3-bisdiazo-1,3-diphenyl-Wolff rearrangement intermediates, 905 Propargyl alcohol dilithiated reaction with α, ω -dibromoalkanes, 281 Propargyl alcohol, arylsynthesis, 537 Propargylic acid dilithium salt

reaction with oxiranes, 278 Propargylic electrophiles reaction with organocopper compounds, 220 Propellane synthesis, 573 4,3,2-Propellanols rearrangement, 709 Propene, phenylsynthesis vinyl carbanion alkylation, 242 **β-Propiolactone** reaction with phenyl Grignard reagents, 466 synthesis carbonylation of ethylene, 1031 β-Propiolactone, β-ethynylreaction with organocopper compounds, 227 Propionic acid, 2-arylsynthesis, 244 Propionic acid, 2-bromot-butyl ester catalyst, Grignard reagent alkylation, 244 Propionic acid, 2-haloaryl esters cycloalkylation, 324 Propionic acid, 2-(mesyloxy)benzene alkylation with Friedel-Crafts reaction, 312 Propionic acid, methyl 2-(chlorosulfonyloxy)benzene alkylation with Friedel-Crafts reaction, 312 Propionic acid, methyl-2-phenylsynthesis Friedel-Crafts reaction, 312 Propionic acid, 3-phenylethyl ester acyloin coupling reaction, 619 Propionic acid, 3-thienylethyl ester acyloin coupling reaction, 619 Propionic acid, 3-trimethylsilylethyl ester acyloin coupling reaction, 619 Propylene oxide Friedel-Crafts reaction, 769 Propylene, 3-chloroarene alkylation by Friedel-Crafts reaction, 321 Propylure synthesis, 799 Propyne, dilithioalkylation, 281 Prostacyclin synthesis rearrangement of epoxides, 767 Prostaglandin, 11-deoxysynthesis enolate alkylation, 9 Prostaglandin $F_{2\alpha}$ synthesis, 290, 781 Prostaglandins precursors synthesis, 139 synthesis, 103, 126, 279, 289, 649 conjugate addition-enolate alkylation, 9 from protected cyanohydrins, 198

Subject Index

Prostaglandins, 5,6-didehydrosynthesis enolate alkylation, 10 Protoilludane biosynthesis, 404 synthesis, 389 Protostephanone enantioselective synthesis, 685 synthesis, 685 Pschorr ring closure, 507 Pseudoguianane biosynthesis, 388 Pteridines, substituted synthesis organocopper compounds, 219 Pterins, substituted synthesis organocopper compounds, 219 Pterodactyladiene synthesis, 872 Pulegone oxiranes rearrangement, 771 Pummerer's ketone synthesis use of silver carbonate, 664 Punctatin A synthesis, 984 photochemical C-H insertion reactions, 1058 Purine, 6-chlorocoupling reactions with primary alkyl Grignard reagents, 462 Purine, 6-methylthiocoupling reactions with primary alkyl Grignard reagents, 462 Pyran, tetrahydroarene alkylation by Friedel-Crafts reaction, 315 2H-Pyran-2-one, 4-isopropyl-6-methylalkylation, 24 Pyrans, dihydrocoupling reactions with alkyl Grignard reagents, 444 metallation, 252 reaction with Grignard reagents nickel catalysts, 229 reaction with organocopper compounds, 218 5H-Pyrazole, 5-acetyl-3, 4-diethoxycarbonyl-5-methylsynthesis cycloaddition method, 893 Pyrazoles synthesis, 905 Pyrazolidine-3,5-dione, 4-diazodecomposition, 902 Pyrenophorin synthesis, 126 Pyrethroids insecticides synthesis, 848 Pyridazine, chloroalkylation with primary alkyl Grignard reagents, 461 2,2'-Pyridil rearrangement, 826 2,2'-Pyridilic acid synthesis, 826

Pyridine, 2-bromocoupling reactions with primary alkyl Grignard reagents, 460 Pyridine, 2-chlorocoupling reactions with primary alkyl Grignard reagents, 460 Pyridine, 2,6-dichlorocoupling reactions with primary alkyl Grignard reagents, 460 Pyridine, iodocoupling reactions with alkylzinc reagents, 460 Pyridine, 2-mercaptopolymer-bonded synthesis, Friedel-Crafts reaction, 302 Pyridine, 4-phenylsynthesis, 513 Pyridines coupling reactions with sp^3 organometallics, 460 Pyridines, 2-arylsynthesis, 512 Pyridines, 4-arylsynthesis, 512 Pyridines, halocoupling reactions, 509 reaction with magnesium dialkylcuprates, 219 Pyridines, substituted synthesis organocopper compounds, 219 Pyridines, vinylsynthesis, 498 Pyridinotropolone rearrangements, 818 Pyridizine N-oxide reaction with aryl Grignard reagents, 494 Pyridopyrimidines synthesis, 543 Pyrido[2,1-b]quinazoline synthesis carbonylation, 1038 Pyridyl triflate coupling reactions with Grignard reagents, 460 Pyrimidine, 2,4-diamino-5-(3,4,5-triethylbenzyl)synthesis, 301 Pyrimidine, 2-methylthioalkylation with primary alkyl Grignard reagents, 461 Pyrimidine, trichloroalkylation with primary alkyl Grignard reagents, 461 Pyrimidines heterocyclic fused synthesis, 543 Pyrone, α -cuprioreactivity, 217 Pyrrole synthesis C-H insertion reactions, 1057 Pyrrole, 4-cyano-3,3-diaryl-5-methyl-2-oxo-2,3-dihydrosynthesis, 826 Pyrrole, 2-lithioalkylation, 261 Pyrrolidine, 1-amino-2-methoxymethyl-

lithiated hydrazone asymmetric alkylation, 37 Pyrrolidine, 2,5-dimethylcyclohexanone enamine from alkylation, 35 Pyrrolidine, 2-methoxysynthesis, 651 Pyrrolidine, N-methyldeprotonation, 65 Pyrrolidine-2,4-dione, 3-diazoring contraction route to β-lactams, 902 Pyrrolidines alkylation, 69 synthesis, 647 Pyrrolidinium tetrafluoroborate, dimethylelectropinacolization aliphatic carbonyl compounds, 570 **Pyrrolidones** synthesis carbonylation of allylamines, 1037 Δ^2 -Pyrroline asymmetric alkylation, 77 Pyrrolizidine synthesis from epoxides, 736 Pyrrolizidine sulfide homologation, 493 Pyruvates synthesis β-cleavage of epoxides, 759 Pyruvic acid, phenylsynthesis cobalt-catalyzed double carbonylation, 1039 Quadricyclane reaction with dimethyldioxirane, 736 Quadrone synthesis, 573, 709 Quasi-Favorskii rearrangement 2-arylalkanoic acids, 788 2,2'-Quinaldil rearrangement, 826 Quinisatine rearrangement, 835 Quinodimethanes precursors synthesis, 255 synthesis, 161, 173 o-Quinol acetates extracyclic migrations, 813 diacetates rearrangements, 812 Quinoline, 3-bromocoupling reactions with Grignard reagents, 461 Quinoline, 2-chlorocoupling reactions with Grignard reagents, 461 Quinolines coupling reactions with sp³ organometallics, 460 Quinolines, chlorosynthesis, 513 Quinolines, halo-

coupling reactions, 509 Quinones benzilic rearrangement, 828 reactions with π -allylnickel halides, 424 Quinones, hydroxybenzilic rearrangement, 828 Ramberg-Bäcklund rearrangement, 861-883 functional group compatibility, 865 mechanism, 866 Michael-induced, 880 phase-transfer conditions, 863 reaction conditions, 862 scope, 862 stereoselectivity, 862 synthesis of alkenes, 163 uses, 862 variations, 868 Recifeiolide synthesis, 286 Recombination hydrogen atom abstraction, 1046 Red bollworm moth sex pheromone synthesis, 169 Reduction selenides use in synthesis, 106 sulfides use in synthesis, 106 Reformatsky reagents coupling reactions with alkenyl halides, 443 with aromatic halides, 454 Reframoline synthesis, 81 Rethrolones synthesis, 126 Reticuline intracoupling to morphines, 679 synthesis, 79 Retinal pinacol coupling reactions, 577 Retronecic acid synthesis, 420 Rhodium, chlorotris(triphenylphosphine)catalyst decarbonylation, 1040 Rhodium, dicarbonylchlorobislithium chloride salt catalyst, alkenyl halide dimerization, 484 Rhodium acetate catalyst C-H insertion reactions, 1051 Rimuene synthesis, 21 Ring contractions Wolff rearrangement, 900 ylides 3,2-sigmatropic rearrangements, 954 Ring expansion Wagner-Meerwein reactions, 713 ylides, 957 **Ring-growing reactions** 3,2-rearrangement, 957 Rotaxanes

synthesis intramolecular acyloin coupling reaction, 628 Rule of diaxial opening epoxides, 734 Salsolidine synthesis, 78 Samarium diiodide intermolecular pinacol coupling reactions aliphatic carbonyl compounds, 570 intramolecular pinacol coupling reactions, 574 pinacol coupling reactions, 567 α -Santalene synthesis, 161, 427, 712 β-Santalene synthesis, 712 Santene synthesis, 706 Santonin rearrangement, 804 Sarcophine transannular cyclization, 407 Sarkomycin synthesis, 937 Sativene synthesis, 20, 712 Schizandrin, deoxysynthesis use of vanadium oxytrifluoride, 676 Schizandrin C synthesis use of thallium trifluoroacetate, 669 Schmidt reaction, 908 Schweizer's reagent reaction with divinylcuprates, 259 Sebacic acid synthesis, 640 Secologanin synthesis, 599 1,2,3-Selenadiazoles Wolff rearrangement, 909 Selenides alkenes from, 114 alkyl and allyl halides from, 118 alkylated use in synthesis, 106 allylic oxidative rearrangement, 117 α -heterosubstituted carbonyl compound synthesis from, 141 reduction use in synthesis, 106 Selenides, alkenyl coupling reactions with sp³ organometallics, 446 Selenides, alkyl vinyl carbonyl compounds from, 120 Selenides, allenyl phenyl synthesis, 106 Selenides, allyl oxidation, 117 Selenides, aryl coupling reactions with Grignard reagents, 456 Selenides, aryl vinyl alkylation, 106

Selenides, homoallyl synthesis, 91 Selenides, 2-hydroxyelimination reactions, 787 rearrangement, 786 semipinacol rearrangements, 777 Selenides, γ -hydroxyalkyl oxidation, 120 Selenides, α -metallovinyl alkylation, 104 Selenides, α -silylalkyl carbonyl compound synthesis from, 141 Selenides, vinyl reaction with Grignard reagents, 493 synthesis, 253 Selenium carbanions stabilized by alkylation, 85-181 Selenones, a-metalloalkyl synthesis, 87 Selenonium salts, a-metalloalkyl synthesis, 87 Selenophene coupling reactions with primary alkyl Grignard reagents, 447 Selenoxides, alkyl alkylation, 147, 157 Selenoxides, allyl rearrangement, 117 Selenoxides, α -lithioalkyl synthesis alkylation, 157 Selenoxides, α -metalloalkyl alkylation, 147, 157 synthesis, 87 Selenoxides, *a*-metallovinyl alkylation, 157 synthesis, 87 Semibenzilic pathway Favorskii rearrangement, 840 Semibenzilic rearrangement mechanism, 828, 836 Semibullvalene synthesis, 640 Semipinacol rearrangements, 777-799 definition, 777 pinacol rearrangement comparison with, 722 tandem cyclization reactions, 792 Senecioic acid dicopper(I) dianion alkylation, 50 Senecioic acid amide, N-isopropyldianions alkylation, 50 Senoxyn-4-en-3-one, 8-oxysynthesis, 404 Serine β-lactone reaction with organocopper compounds, 227 reaction with pivaldehyde, 40 Sesamol, benzyloxidative coupling, 669 Sesquicarene synthesis, 288 Sesquifenchene

synthesis, 161 Sesquiterpenes polycyclic biosynthesis, 388 synthesis, 288 Seven-membered rings formation polyene cyclization, 357 Seychellene synthesis, 20 Shapiro reaction vinyllithium generation, 251 Sigmatropic rearrangements Baldwin's rules, 915 1,2-Sigmatropic rearrangements, 921 chirality transfer, 927 3,2-Sigmatropic rearrangements, 932 ylides stereocontrol, 943 3,3-Sigmatropic rearrangements erythro-threo ratio, 949 Silanes, alkynylcoupling reactions with aryl halides, 538 Silanes, allylsynthesis coupling reactions, 437, 445 nickel catalysts, 229 Silanes, bisallylsynthesis, 482 Silanes, dimethylfluororeaction with alkenyl iodides organopalladium catalysis, 233 Silanes, epoxyreaction with organocopper compounds, 224 Silanes, α,β -epoxyalkyldeprotonation, 198 Silanes, B-keto(aldehydo)synthesis from α -trimethylsilyl epoxides reaction with Grignard reagents, 759 Silanes, vinylcoupling reactions with aryl bromides, 495 Silicon reagents organopalladium catalysis, 233 Silver catalysts Grignard reagent coupling, 418 Silver benzoate Wolff rearrangement initiator, 891 Silver nitrate in halohydrin rearrangements formation of aldehydes, 758 Silver salts Wurtz coupling, 422 Silver(I) oxide Wolff rearrangement initiator, 891 Silvbin synthesis, 691 Silyl chloride, trimethylacyloin coupling reaction trapping agent, 615 Silyl dienol ethers cross-conjugated

alkylation, 28 extended y-alkylation, 27 homoannular alkylation, 27 Silyl enol ethers aldehyde allylation, 28 alkylation, 25 coupling reactions with aryl Grignard reagents, 492 cycloalkylation, 27 intramolecular alkylation, 26 synthesis from lithium homoenolates, 197 Sinapic acid oxidation, 692 α-Sinensal synthesis, 936 **B**-Sinensal synthesis, 429 Sirenin synthesis, 288, 788 Six-membered rings formation polyene cyclization, 349 Sodium cyanotrihydroborate boron trifluoride mixture epoxide reduction, 753 Sodium-potassium alloy ester reduction heterogeneous conditions, 615 Sodium tetracarbonylcobaltate catalysts carbonylation of alkyl halides, 1029 Sodium tetracarbonylferrate catalyst carbonylation of alkyl and aralkyl halides, 1026 halide carbonylation formation of aldehydes, 1021 ketone synthesis carbonylation, 1024 Solanesol synthesis, 170 Sommelet-Hauser rearrangement, 965 asymmetry, 969 ylidic, 914 Soybean lipoxygenase irreversible inhibitors synthesis, 217 Spiroacetals synthesis, 252 Spiroannulation Wurtz coupling, 423 Spirocyclization polyenes, 354 Spirodienones oxygen migration, 813 synthesis, 679 aryl radical insertion, 686 C---C phenol-phenol coupling, 679 Spiroethers synthesis, 688 Squalene synthesis, 99, 170 phosphonium ylides, 201

reduction of sulfides, 107 3,2-sigmatropic rearrangement, 943 sulfur ylides, 3,2-rearrangement, 933 ylides, 919 Squalene, 2,3-epoxysynthesis, 126 Squalene, perhydrosynthesis, 586 2-Stanna-1,3-dioxolane synthesis, 571 Stannane, acylasymmetric reduction to a-alkoxy organostannanes, 196 Stannane, aryltri-n-butyldimerization, 500 Stannane, ethoxy-a-chloromethyltributylreaction with Grignard reagents preparation of O-ethyl organostannanes, 196 Stannane, pyridylcoupling reactions with bromopyridine, 510 Stannane, *β*-silylvinylcoupling reactions with alkenyl halides, 495 Stannane, *a*-sulfonylalkylcoupling reactions with alkenyl halides, 443 Stannane, vinylcoupling reactions butadiene synthesis, 483 with alkenyl halides, 495 with vinyl iodides, 488 Steganacin synthesis use of vanadium oxytrifluoride, 674 Stegane synthesis use of vanadium oxytrifluoride, 675 Steganone synthesis, 150, 501 ring expansion, 674 use of thallium trifluoroacetate, 673 Stemodin synthesis, 717 Stephens-Castro coupling alkynic ketones, 226 copper acetylide intermediates, 217 Stereospecificity anti epoxide ring opening, 733 Steroids aromatic synthesis, 366 exo-methylene epoxides, opening, 743 nonaromatic, synthesis polyene cyclization, 369 synthesis polyene cyclization, 362 Steroids, 17a-bromo rearrangements, 846 Steroids, 21-bromo rearrangements, 846 Steroids, halo ring A contractions, 854 Sterpurene

synthesis, 402, 714 Stevens rearrangement, 913-971 Steviol rearrangement, 715 Stilbene oxide reaction with Grignard reagents epoxide ring opening, 755 Stilbenes synthesis, 497 Ramberg-Bäcklund rearrangement, 864, 865 Stilbenes, chloroperoxy acid reaction epoxides as reactive intermediates, 739 Still-Wittig rearrangement, 983 Stork enamine reaction, 28 Stork-Eschenmoser hypothesis polyalkene cyclization, 341 Styrene hydroesterification palladium catalyst, 1030 hydroformylation platinum catalysts, 1022 oxide reaction with organocopper compounds, 224 rearrangement, lithium halide catalyzed, 764 rearrangement, lithium perchlorate catalyzed, 761 synthesis, 495 Friedel-Crafts reaction, 294 vinylic coupling, 485 Succinic acid 2.3-disubstituted synthesis, 638 Succinic acid, a-benzyl-B-phenylsynthesis, 828 Succinic acid, 2-methyldimethyl ester intramolecular acyloin coupling reaction, 621 Succinoin, di-t-butylsynthesis intramolecular acyloin coupling reaction, 621 Sulfide, dimethyl solvent alkylcopper compound reactions, 210 Sulfides alkenes from, 114 alkyl and allyl halides from, 118 alkylated use in synthesis, 106 anions reaction with boranes, 795 heteroaromatic coupling reactions with sp³ organometallics, 459 α -heterosubstituted carbonyl compound synthesis from, 141 metallation use of additives, 86 reduction use in synthesis, 106 α -thiometallation, 196 Wittig rearrangement, 978 Sulfides, alkenyl coupling reactions with sp³ organometallics, 446 Sulfides, alkyl vinyl carbonyl compounds from, 120 Sulfides, 1-alkynyl

metallation, 106 Sulfides, allyl oxidation, 116 reduction selectivity, 107 Sulfides, allyl benzyl metallation selectivity, 99 Sulfides, allyl 2-pyridyl reduction, 108 Sulfides, aryl coupling reactions with Grignard reagents, 456 Sulfides, benzyl α -chlorobenzyl Ramberg-Bäcklund rearrangement, 870 Sulfides, benzyl a, a-dichlorobenzyl Ramberg-Bäcklund rearrangement, 870 Sulfides, bisreaction with vinyl magnesium halides regioselectivity, 493 Sulfides, α -chlorophenacyl phenacyl Ramberg-Bäcklund rearrangement, 870 Sulfides, cycloalkyl phenyl synthesis, 88 Sulfides, 1,3-dienyl alkylation, 105 Sulfides, haloalkyl phenyl rearrangement, 88 Sulfides, hydroxyelimination reactions, 786 Sulfides, 2-hydroxyrearrangement, 784 semipinacol rearrangements, 777, 778 Sulfides, α -metallovinyl alkylation, 104 Sulfides, α -methoxyalkenyl phenyl carbonyl compound synthesis from, 141 Sulfides, α -methoxy allyl α -methylenated acyl anion equivalent, 144 Sulfides, 2-pyridyl coupling reactions with Grignard reagents, 460 Sulfides, α -silylalkyl phenyl carbonyl compound synthesis from, 141 Sulfides, α -thiomethylcyanomethyl Wittig rearrangement, 978 Sulfides, vinyl reaction with alkenylaluminum, 492 reaction with Grignard reagents nickel catalysts, 229 3-Sulfolene reaction with alkyl iodides selectivity, 172 Sulfolenes 1,3-dienes from, 173 Sulfonamides, alkyl alkylation, 179 Sulfonates alkylation vinyl carbanions, 242 Sulfonates, alkyl alkylation, 179 Sulfone, α -bromoethyl ethyl Ramberg-Bäcklund rearrangement, 861 Sulfone, α -chloroethyl ethyl Ramberg-Bäcklund rearrangement, 861

Sulfone, dibenzyl Ramberg-Bäcklund rearrangement, 864 Sulfone, methoxymethyl phenyl anions reaction with cyclic ketones, 785 Sulfone, methyl methylthiomethyl alkylation, 139 Sulfone, methyl phenyl alkylation, 159 Sulfone, methylthiomethyl p-tolyl methylthiomethyl ester alkylation, 136 Sulfone, methylthiomethyl p-tosyl alkylation, 139 Sulfone, phenylthiomethyl phenyl anions reaction with cyclic ketones, 785 Sulfone, α -triflyldimethyl alkylation, 177 Sulfones alkylation, 158 chlorination mechanism, 864 Sulfones, alkenyl coupling reactions with sp³ organometallics, 446 Sulfones, alkyl in synthesis, 160 a-metallo, a-heterosubstituted alkylation, 174 Sulfones, allyl in synthesis, 169 Sulfones, aryl coupling reactions with Grignard reagents, 456 Sulfones, a-chloro synthesis, 864 Sulfones, di-s-alkyl Ramberg-Bäcklund rearrangement, 864 Sulfones, a-diazo-Wolff rearrangement, 909 Sulfones, a-halo reaction with trialkylboranes, 794 synthesis, 862 Sulfones, α -isocyanoalkyl alkylation, 175 Sulfones, *β*-keto metal enolates alkylation, 54 Sulfones, α -metalloalkyl reactions, 158 Sulfones, α -metalloallyl reactions, 168 Sulfones, α -metallovinyl reactions, 173 Sulfones, a-sulfinyl reactions, 176 Sulfones, a-tosyloxy-Ramberg-Bäcklund rearrangement, 868 Sulfones, vinyl deprotonation, 253 reaction with Grignard reagents, 493 Sulfonium salts, α -metalloalkyl synthesis, 87 Sulfonium ylides, acyl-Wolff rearrangement, 909

Subject Index

Sulfoxide, t-butyl thiomethyl alkylation, 139 Sulfoxide, ethyl ethylthio alkylation, 139 Sulfoxide, methyl methylthio alkylation, 139 Sulfoxide, methyl methylthiomethyl alkylation, 139 metallated alkylation, 135 Sulfoxide, methyl thiomethyl alkylation, 137 Sulfoxide, phenyl thiomethyl alkylation, 139 Sulfoxides alkenes from sulfenic acid elimination, 154 alkylated in synthesis, 154 Sulfoxides, 1-alkenyl aryl alkylation, 155 Sulfoxides, alkyl alkylation, 147 reduction, 155 Sulfoxides, allylic alkylation, 155 metallation, 155 Sulfoxides, aryl coupling reactions with Grignard reagents, 456 Sulfoxides, a-chloro cyclobutene synthesis from, 872 Sulfoxides, B-keto metal enolates alkylation, 54 Sulfoxides, a-metalloalkyl alkylation, 147 Sulfoxides, thioacetal carbonyl compound synthesis from, 142 Sulfoximides alkylation, 173 Sulfoximines, alkenylreaction with organozinc reagents nickel catalysis, 230 Sulfoximines, alkyl a-chloroalkyl Ramberg-Bäcklund rearrangement, 871 Sulfoximines, α -halo N-tosyl Ramberg-Bäcklund rearrangement, 870 Sulfoxonium ylides 3,2-sigmatropic rearrangement, 939 Sulfoxonium ylides, β-keto-Wolff rearrangement, 909 Sulfur carbanions stabilized by alkylation, 85-181 Sulfur ylides alkylation, 178 preparation, 918 Sultones arene alkylation Friedel-Crafts reaction, 317 Sultones, alkyl alkylation, 179 Superacids catalysts Friedel-Crafts reaction, 297

solid, catalysts Friedel-Crafts reaction, 297 Suzuki couplings alkenylboron species, 489 Syringaresinol synthesis, 693 Tamoxifen synthesis, 585 Tartaric acid acetal stereospecific bromination, 789 Taxanes synthesis, 832 epoxide ring opening, 744 Tellurides, alkenyl coupling reactions with sp^3 organometallics, 446 Tellurium compounds catalyst Wurtz reaction, 421 Tellurophene coupling reactions with primary alkyl Grignard reagents, 447 Terephthaloyl chloride acyloin coupling reaction, 617 Ternaphthyls synthesis nickel catalysts, 229 Terpene oxides epoxide ring opening zinc bromide catalysis, 771 Terpenes cyclic biogenetic origins, 380 synthesis, 428 Terpenes, polybromomono-Favorskii rearrangements, 849 Terphenyl synthesis, 503 Terpineol synthesis linalool cyclization, 352 α -Terpineol synthesis from α - and β -pinene, 708 Terpyridine synthesis, 512 Testosterone synthesis polyene cyclization, 371 Tetracycles polyene cyclization, 369 Tetracyclines synthesis, 809 Tetraenes conjugated synthesis, 880 Tetralin, 1,4-dimethylsynthesis Friedel-Crafts reaction, 318 β -Tetralone pyrrolidine enamine monomethylation, 29 Tetralones synthesis

Friedel-Crafts cycloalkylation, 326 Tetraphenyl synthesis, 501 1,3,5,7-Tetrathiacyclooctane tetraanion methylation, 134 Thalictuberine synthesis, 586 Thaliphorphine acetate synthesis, 672 Thallium, arylcarbonylation palladium catalysts, 1033 Thallium salts catalysts alkyl halide coupling, 418 Thallium trifluoroacetate, arylbiaryl synthesis, 505 Thallium tris(trifluoroacetate) dimerization, 499 Thebaine synthesis oxidation by thallium tris(trifluoroacetate), 680 Thiacyclohexane, 4-t-butylmetallation, 151 1,2,3-Thiadiazoles Wolff rearrangement, 909 Thiazole, 2-bromocoupling reactions, 511 Thiazoline, 2-alkylalkylation, 53 Thiazoline, a-lithioalkylthioalkylation, 88 Thienamycin synthesis, 1036 Thienopyridines synthesis, 543 Thiirane 1.1-dioxide Ramberg-Bäcklund rearrangement, 866, 867 Thioacetophenone pinacol coupling reactions, 582 Thiobenzaldehyde dianions alkylation and allylation selectivity, 95 Thiocarbonyl compounds pinacol coupling reactions, 582 Thioesters α' -lithioalkyl alkylation, 88 reductive coupling, 618 Thiols, aryl coupling reactions with Grignard reagents, 456 Thiophene, 2-bromohalogen-metal exchange with Grignard reagents, 459 Thiophene, 3-bromocoupling reactions with Grignard reagents, 459, 513 Thiophene, diarylsynthesis, 513 Thiophene, 2-lithioalkylation, 261 Thiophene, vinylsynthesis, 497

Thiophenes acyloin coupling reaction esters, 615 coupling reactions with sp³ organometallics, 459 with primary alkyl Grignard reagents, 447 intermolecular dimerization, 509 Thiophenophane 2,5-bridged synthesis, acyloin coupling reaction, 630 Thiopyran, 2,3-dihydroring-opening coupling reaction with primary alkyl Grignard reagents, 447 Thymol, methoxyoxidation use of alkaline ferricyanide, 686 Thymoquinone, libocedroxysynthesis use of alkaline ferricyanide, 686 Tiffeneau-Demjanov rearrangement 2-amino alcohols, 781 pinacol rearrangement comparison with, 722 Tiglic acid allylation, 50 Tiglic acid, y-iodot-butyl ester alkylation by, 11 Tin, alkynylcoupling reactions, 529 Tin, allylcarbonylation palladium catalysts, 1023 coupling reactions with acyl chlorides, 463 with aromatic halides, 453 Tin, benzylcoupling reactions with acyl chlorides, 463 Tin, cyanomethyltributylcyanomethylation aryl bromides, 454 Tin, hydroxymethylcoupling reactions with aromatic halides, 453 Tin, methoxymethylcoupling reactions with aromatic halides, 453 Tin, methylcoupling reactions with aromatic halides, 453 Tin, tetraphenylreaction with aryl halides, 504 Tin chloride epoxide ring opening, 770 Tin compounds lithium exchange formation of α -alkoxylithiums, 195 organopalladium catalysts, 231 Titanium, chloromethylalkyl halide methylation, 421 Titanium compounds use in intermolecular pinacol coupling reactions aliphatic carbonyl compounds, 570 use in intramolecular pinacol coupling reactions, 572 use in pinacol coupling reactions, 565

Subject Index

use in reductive coupling reactions carbonyls with alkenes, 583 Titanium isopropoxide epoxide ring opening, 770 Titanium trichloride catalyst Wurtz reaction, 421 a-Tocopherol oxidative coupling cycloaddition, 698 synthesis, 644 Toluene alkylation Friedel-Crafts reaction, 300, 304 with *n*-butene, 304 isopropylation Friedel-Crafts reaction, 311 Toluene, n-alkylisomerization Friedel--Crafts reaction, 327 Tosylhydrazones hydroxide ion assisted decomposition synthesis of a-diazo ketones, 890 Totarol oxidative coupling alkaline potassium ferricyanide, 665 Transalkylation Friedel--Crafts reaction, 327 Transannular cyclizations electrophilic, 379-407 imines, 581 Transbenzhydrylation Friedel-Crafts reaction, 328 Transbenzylation Friedel-Crafts reaction, 329 Trans-t-butylation Friedel-Crafts reaction, 329 Triamantane synthesis Friedel-Crafts reaction, 334 Triannulenes synthesis intramolecular acyloin coupling reaction, 627 1.2.3-Triazoles Wolff rearrangement, 909 1,2,3-Tricarbonyl compounds diazo-coupling reactions, 893 Trichostatin A synthesis from protected cyanohydrins, 198 Trichothecenes epoxide opening, 736 synthesis, 714 Tricyclic diterpenoids synthesis, 325 Tricyclohumuladiol synthesis, 399, 402 Tricyclo[4.2.0.0^{1,4}]octane synthesis, 901 Tricyclo[3.2.1.0^{3,6}]octan-4-one, 6,7,7-trimethylsynthesis, 906 Tricyclo[2.1.1.0^{4.5}]pentane synthesis, 894 Trienes conjugated synthesis, 879, 882

Trienoic acids synthesis, 882 Triethylamine α -deprotonation, 65 Triflates catalysts Friedel-Crafts reaction, 295 Triflone, mesylalkylation, 868 Triphyophylline, O-methyltetradehydrosynthesis photochemical oxidation, 677 Triquinane synthesis, 384 Triquinane sesquiterpenes synthesis organocopper compounds, 221 Tris(diethylamino)sulfonium difluorotrimethylsilicate reaction with organic halides palladium catalysis, 233 Trisporone, deoxysynthesis, 169 Triterpenes synthesis polyalkene cyclization, 364 2,6,7-Trithiabicyclo[2.2.2]octane, 1-lithio-4-methylalkylation, 145 1,3,5-Trithiane metallation, 134 Trogodermal synthesis, 243 Tropolone, aminorearrangements, 818 Tropolonecarboxylic acid rearrangements, 818 Tropolone-phenol rearrangements, 817 Tuberculostearic acid synthesis, 644 α -Turmerone synthesis, 126 ar-Turmerone synthesis, 28 Twistbrendane synthesis, 854 exo-Twistbrendan-2-ol brosylate acetolysis, 709 Tylophorine synthesis use of thallium trifluoroacetate, 670 use of vanadium oxytrifluoride, 670 L-Tyrosine synthesis, 816 Ullmann reaction biaryl synthesis, 482, 499 organocopper compounds, 209, 219 Undeca-2,5-diyl-1-ol synthesis, 273 Undecanone, dibromorearrangement, 851 Untriacontane, 3-methylsynthesis, 414 Urea, tetramethyllithium halide sensitizer epoxide ring opening, 763

Uridine, allylsynthesis allylation of mercury intermediates, 476 Ursolinic acid ring A contraction, 834 Valerane synthesis, 20 γ-Valerolactone benzene alkylation by Friedel-Crafts reaction, 317 Valerolactone, 5-ethenylsynthesis, 245 Valine bislactim ethers from, 53 t-butyl ester, enamines alkylation, 36 Vanadium chloride catalyst Wurtz reaction, 421 Vanadium compounds use in intermolecular pinacol coupling reactions, 565 Veraguensin synthesis, 693 Veratrole electrolytic oxidation, 668 Veratrole, 4-methyloxidative coupling, 669 Verhenol synthesis, 126 Vermiculin antibiotics synthesis, 126 Vernolepin synthesis, 280 Verticillene synthesis, 591 α -Vetispirene synthesis, 586 **B**-Vetivone synthesis, 20, 22 Vinyl aluminate alkylation copper-catalyzed, 215 Vinylation 1-alkynes, 521 alkynic iodides palladium-catalyzed, 544 metal enolates regioselectivity, 12 Vinyl carbanions alkylation, 241-266 heteroatom substituted alkylation, 252 Vinvl epoxides reaction with arylstannanes organopalladium catalysis, 232 Vinyl halides carbonylation formation of esters, 1028 cross-coupling reactions organometallic reagents, 522 cyclocarbonylation formation of α -methylene lactones, 1032 reaction with lithium dialkylcuprates, 217 a-substituted arene alkylation, 322

synthesis, 788 Vinyl iodides carbonylation formation of ketones, 1023 reaction with organotin compounds organopalladium catalysis, 232 Vinyl magnesiocuprates synthesis alkylmagnesium halide reactions with alkynes, 243 Vinyl triflates carbonylation formation of aldehydes, 1021 formation of ketones, 1023 cross-coupling reactions, 529 intermediate in dolastane synthesis, 488 reaction with tin compounds organopalladium catalysts, 232 synthesis organocopper compounds, 218 Virescenol A homo-Favorskii rearrangement, 857 Virolin synthesis use of silver oxide, 691 Vitamin A epoxide ring opening, 757 synthesis, 169, 170 organocopper compounds, 223 Vitamin D synthesis, 109, 545 organocopper compounds, 223 organopalladium catalysts, 232 Vitamin D₃ synthesis, 168, 173 Vitamin D₃, 1α,25-dihydroxy synthesis, 984 Wagner-Meerwein rearrangement, 705-717 bicyclic systems, 706 definition, 706 stereoelectronic features, 709 use in synthesis, 710 Wheland intermediate Friedel-Crafts reaction arene alkylation, 298 Wilsonirine, N-trifluoroacetylsynthesis use of diphenyl selenoxide, 666 use of vanadium oxytrifluoride, 670 Wittig rearrangement, 975-1012 1,2-rearrangement electron transfer mechanism, 824 mechanism, 979 2,3-rearrangement, 981 stereochemistry, 943 tandem and sequential rearrangements, 994 Wolff rearrangement, 887–909 chemistry, 897 competing reactions, 893 initiation, 891 photolysis, 891 thermolysis, 891 transition metal catalysts, 891 mechanism, 891 stereochemistry, 891 vinylogous, 906

Subject Index

Wurtz reaction classical, 414 coupling reactions, 413 intramolecular, 422 variants, 414 Wuweizisu-C synthesis use of vanadium oxytrifluoride, 676 Xanthone, 2,6-dihydroxysynthesis use of ferricyanide, 688 **Xylenes** isomerization Friedel-Crafts reaction, 327 **Xylenols** transalkylation Friedel-Crafts reaction, 329 Ylangene synthesis, 20 Ylides direct formation, 919 Ylidic rearrangements definition, 916 Yohimbone asymmetric synthesis, 81 synthesis, 72 Yomogi alcohol synthesis 3,2-rearrangement, 933 Ytterbium use in pinacol coupling reactions, 567 Zearalenone synthesis, 49 palladium-catalyzed carbonylation, 1033 Zeolites modified, catalysts Friedel-Crafts reaction, 296 polyfunctional catalysts acidity, Friedel-Crafts reaction, 305 shape selective catalysts Friedel-Crafts reaction, 296 Zerumbone 8.9-epoxide transannular cyclization, 406 Ziegler-Natta catalysts metal alkyls, 296 Zinc, alkenylcoupling reactions with aryl iodides, 495 Zinc, alkynylchlororeaction with alkenyl halides palladium-catalyzed, 524 Zinc, arylalkylation, 260 Zinc, arylchloro-

coupling reactions with alkenyl bromides, 495 Zinc, benzylcoupling reactions with aromatic halides, 453 Zinc, bromo(diethoxyphosphoryl)difluoromethylalkylation with allylic halides, 202 Zinc, chloro-2-furylcoupling reactions with alkenyl iodides, 497 Zinc, chloroheteroarylalkylation, 261 Zinc, chloropyridylcoupling reactions, 510 Zinc, homoallylcoupling reactions with aromatic halides, 453 Zinc, homopropargylcoupling reactions with aromatic halides, 453 Zinc, phenylethylcoupling reactions with aromatic halides, 453 Zinc. silvlmethylcoupling reactions with aromatic halides, 453 Zinc chloride catalyst vinylic Grignard coupling, 485 Zinc halides epoxide ring opening, 771 Zinc reagents organopalladium catalysis, 230 Zirconium, alkenylcoupling reactions with aryl iodides, 495 reactions nickel catalysis, 230 Zirconium, chlorocyclopentadienylhydridocatalyst hydroesterification, 1030 Zirconium complexes, alkylmonoalkylation 2,3-dichloro-1,4-naphthoquinone, 458 Zirconium hydride catalysts carbonylation, 1027 Zirconium reagents organopalladium catalysis, 230 ZSM-5 zeolite catalyst Friedel-Crafts reaction, 305 Zygosporin 3,2-sigmatropic rearrangement synthesis, stereocontrol, 960